BELLSOUTH® / CLEC Agreement

Customer Name: Knology

Knology (Reneg 2002)	2
Table of Contents	3
General Terms and condtions	5
Att 1 - Resale	30
Att 1 - Resale Discounts and Rates	50
Attachment 2 - UNE	56
Attachment 2 - Rates	131
Attachment 3 - Interconnection	411
Attachment 3 Rates	439
Att 4 - Collocation - Central Office	451
Att 4 - Collocation - Remote Site	489
Att 4 - Collocation Rates	524
Att 5 - Access to Numbers and Number Portability	549
Att 6 - Ordering	553
Attachment 7	560
Att 7 - ODUF/ADUF/EODUF/CMDS Rates	576
Att 8 - Rights of Way	582
Att 9 - Performance Measurements	584
Att 10 - Disaster Recovery Plan	737
Att 11 - BFR and NBR Process	746

INTERCONNECTION AGREEMENT BETWEEN BELLSOUTH TELECOMMUNICATIONS INC. AND Knology

TABLE OF CONTENTS

General Terms and Conditions

- Definitions
- 1. Purpose
- 2. Term of the Agreement
- 3. Ordering Procedures
- 4. Parity and Performance
- 5. White Pages Listings
- 6. Blank
- 7. Court Ordered Requests for Call Detail Records and Other Subscriber Information
- 8. Liability and Indemnification
- 9. Intellectual Property Rights and Indemnification
- 10. Treatment of Proprietary and Confidential Information
- 11. Assignments
- 12. Resolution of Disputes
- 13. Taxes
- 14. Force Majeure
- 15. Blank
- 16. Modification of Agreement
- 17. Waivers
- 18. Governing Law
- 19. Notices
- 20. Implementation of Agreement
- 21. Filing of Agreement
- 22. Entire Agreement
- 23. Survival
- 24. Blank
- 25. Branding and Customer Contracts
- 26. Customer Information
- 27. Miscellaneous
- 28. Compliance with Applicable Law
- 29. Necessary Approval
- 30. Good Faith Performance
- 31. Nonexclusive Dealings
- 32. CLEC Certification

TABLE OF CONTENTS (cont'd)

- Attachment 1 Resale
- **Attachment 2 Network Elements and Other Services**
- **Attachment 3 Network Interconnection**
- **Attachment 4 Physical Collocation**
- **Attachment 5 Access to Numbers and Number Portability**
- Attachment 6 Pre-Ordering, Ordering, Provisioning, Maintenance and Repair
- Attachment 7 Billing
- Attachment 8 Rights-of-Way, Conduits and Pole Attachments
- **Attachment 9 Performance Measurements**
- **Attachment 10- BellSouth Disaster Recovery Plan**
- Attachment 11–Bona Fide Request/New Business Request Process

AGREEMENT

THIS AGREEMENT is made by and among BellSouth Telecommunications, Inc., ("BellSouth"), a Georgia corporation, and Knology of Alabama, Inc., Knology of Florida, Inc., Knology of Georgia, Inc., Knology of Kentucky, Inc., Knology of South Carolina, Inc., and Knology of Tennessee, Inc., Delaware corporations (hereinafter collectively referred to as "Knology"), and shall be effective on the Effective Date, as defined herein. This Agreement may refer to either BellSouth or Knology or both as a "Party" or "Parties."

WITNESSETH

WHEREAS, BellSouth is a local exchange telecommunications company authorized to provide telecommunications services in the States of Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina and Tennessee; and

WHEREAS, Knology of Alabama, Inc. is or seeks to become a competitive local exchange telecommunications company ("CLEC") authorized to provide telecommunications services in the State of Alabama; Knology of Florida, Inc. is or seeks to become a CLEC authorized to provide telecommunications services in the State of Florida; Knology of Georgia, Inc. is or seeks to become a CLEC authorized to provide telecommunications services in the State of Georgia; Knology of Kentucky, Inc. is or seeks to become a CLEC authorized to provide telecommunications services in the State of South Carolina, Inc. is or seeks to become a CLEC authorized to provide telecommunications services in the State of South Carolina; and Knology of Tennessee, Inc. is or seeks to become a CLEC authorized to provide telecommunications services in the State of Tennessee; and

WHEREAS, the Parties wish to resell BellSouth's telecommunications services and/or interconnect their facilities, purchase network elements and other services, and exchange traffic specifically for the purposes of fulfilling their respective obligations pursuant to sections 251 and 252 of the Telecommunications Act of 1996 ("the Act").

NOW THEREFORE, in consideration of the mutual agreements contained herein, BellSouth and Knology agree as follows:

Definitions

Affiliate is defined as a person that (directly or indirectly) owns or controls, is owned or controlled by, or is under common ownership or control with, another person. For purposes of this paragraph, the term "own" means to own an equity interest (or equivalent thereof) of more than 10 percent.

Commission is defined as the appropriate state regulatory agency in each of BellSouth's nine-state region, Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina, and Tennessee.

Competitive Local Exchange Carrier (CLEC) means a telephone company certificated by the Commission to provide local exchange service within BellSouth's franchised area.

Effective Date is defined as the date that the Agreement becomes effective in each state, which date shall be sixty (60) days after the date of the last signature executing the Agreement. Future amendments for rate changes will be effective thirty (30) days after the Effective Date of the amendment, which shall be the date of the last signature as evidenced on the last page of such amendment.

End User means the ultimate user of a Telecommunications Service.

FCC means the Federal Communication Commission.

General Terms and Conditions means this document including all of the terms, provisions and conditions set forth herein.

Telecommunications means the transmission, between or among points specified by the user, of information of the user's choosing, without change in the form or content of the information as sent and received.

Telecommunications Service means the offering of telecommunications for a fee directly to the public, or to such classes of users as to be effectively available directly to the public, regardless of the facilities used.

Telecommunications Act of 1996 ("Act") means Public Law 104-104 of the United States Congress effective February 8, 1996 and the FCC's rules promulgated thereunder. The Act amended the Communications Act of 1934 (47 U.S.C. Section 1 et. seq.).

1. Purpose

The Parties agree that the rates, terms and conditions contained within this Agreement, including all Attachments, comply and conform with each Parties' obligations under sections 251 and 252 of the Act. The resale, access and interconnection obligations contained herein enable Knology to provide competing local exchange service to residential and business subscribers within the territory of BellSouth. The Parties agree that Knology will not be considered to have offered telecommunications services to the public in any state within BellSouth's region until such time as it has ordered services for resale or interconnection facilities for the purposes of providing business and/or residential local exchange service to customers.

2. Term of the Agreement

- 2.1 The term of this Agreement shall be two (2) years from the Effective Date of this Agreement and shall apply to the BellSouth territory in the state(s) of Alabama, Florida, Georgia, Kentucky, South Carolina and Tennessee.
- 2.2 The Parties agree that by no earlier than two hundred seventy (270) days and no later than one hundred and eighty (180) days prior to the expiration of this Agreement, they shall commence negotiations for a new agreement to be effective beginning on the expiration date of this Agreement ("Subsequent Agreement").
- 2.3 If, within one hundred and thirty-five (135) days of commencing the negotiation referred to in Section 2.2 above, the Parties are unable to negotiate new terms, conditions and prices for a Subsequent Agreement, either Party may petition the Commission to establish appropriate terms, conditions and prices for the Subsequent Agreement pursuant to 47 U.S.C. 252.
- 2.4 In the event the Commission does not issue its order prior to the expiration date of this Agreement, or if the Parties continue beyond the expiration date of this Agreement to negotiate the Subsequent Agreement, this Agreement shall be deemed extended on a month-to-month basis. Upon conversion to a month-to-month term, either Party, in its discretion may terminate this Agreement upon thirty (30) days notice to the other Party; provided, however, that in the event the Parties are in arbitration with the Commission regarding the Subsequent Agreement, this Agreement cannot be terminated prior to 180 days after the original expiration date. In the event that BellSouth terminates this Agreement as provided above, BellSouth shall continue to provide services to Knology pursuant to (1) the terms, conditions and rates set forth in BellSouth's standard interconnection agreement then in effect and made available to CLECs requesting negotiations pursuant to Section 251 of the Act, or (2) an agreement adopted by Knology pursuant to Section 13 of this Agreement. Neither Party shall refuse to provide services to the other Party during the negotiation of the Subsequent Agreement or the transition from this Agreement to the Subsequent Agreement. In the event that the Parties begin operating under BellSouth's standard interconnection agreement or an agreement adopted by Knology, the Parties may continue to negotiate a Subsequent Agreement or may continue to pursue arbitration of a Subsequent Agreement before the Commission. The terms of such Subsequent Agreement shall be effective as of the effective date stated in such Subsequent Agreement and shall not be applied retroactively to the expiration date of this Agreement unless the Parties agree otherwise.

3. Ordering Procedures

3.1 Knology shall provide BellSouth its Carrier Identification Code (CIC), Operating Company Number (OCN), Group Access Code (GAC) and Access Customer Name and Address (ACNA) code as applicable prior to placing its first order.

3.2 The Parties agree to adhere to the BellSouth Business rules for Local Ordering, which may be found at <u>http://www.interconnection.bellsouth.com/guides/guides/html</u> to the extent appropriate for the services ordered.

3.3 Knology shall pay charges for Operational Support Systems (OSS) (i) as set forth in this Agreement in Attachment 1 and/or in Attachment 2, 3, 5 and 7 as applicable or (ii) if otherwise provided by the Commission or by applicable law at any time during the term of this Agreement in which case the Parties shall amend this Agreement pursuant to Section 16.

4. **Parity and Performance**

4.1 When Knology purchases, orders or obtains any telecommunications services for resale pursuant to Attachment 1 from BellSouth, BellSouth shall provide said services to Knology on a nondiscriminatory basis so that the services are at least equal in quality, subject to the same conditions, and provided within the same provisioning time intervals that BellSouth provides to itself, to its Affiliates, Subsidiaries, to any other telecommunications carrier or to any BellSouth End Users. To the extent technically feasible, BellSouth shall ensure that the quality of any Network Element prescribed to be unbundled by the FCC, as well as the quality of the access to such Network Element provides by BellSouth to Knology shall be at least equal in quality to that which BellSouth provides to itself, a subsidiary, an Affiliate, or any other Party. The interconnection facilities shall be designed to meet the same technical criteria and service standards that are used within BellSouth's network and shall extend to a consideration of service quality as perceived by end users and service quality as perceived by BellSouth.

4.2 The services shall be provided and each of the services under this Agreement will be provided and will operate, consistent with the criteria set forth in (i) this Agreement, (ii) any applicable BellSouth tariffs if dealing with resold services and (iii) as established by the Commission.

5. White Pages Listings

BellSouth shall provide Knology and their customers access to white pages directory listings under the following terms:

5.1 <u>Listings</u>. Knology shall provide all new, changed and deleted listings on a timely basis and BellSouth or its agent will include Knology residential, business and government customer listings in the appropriate alphabetical directories. Directory listings shown will make no distinction between Knology and BellSouth subscribers.

5.2 <u>Rates</u>. Subscriber primary listing information in the White Pages shall be provided to Knology at no charge other than applicable service order charges.

5.3 <u>Procedures for Submitting Knology Subscriber Information</u>. These procedures are found in BellSouth's Ordering Guide for manually processed listings and in the Local Exchange Ordering Guide for mechanically submitted listings.

5.4 <u>Unlisted/Non-Published Subscribers</u>. Knology will be required to provide to BellSouth the names, addresses and telephone numbers coded appropriately of all Knology customers that wish to be omitted from directories.

5.5 <u>Inclusion of Knology Customers in Directory Assistance Database</u>. BellSouth will include and maintain Knology subscriber listings in BellSouth's Directory Assistance databases at no charge and Knology shall provide such Directory Assistance listings at no charge. BellSouth and Knology will formulate appropriate procedures regarding lead time, timeliness, format and content of listing information.

5.6 <u>Listing Information Confidentiality</u>. BellSouth will treat Knology's directory listing information with the same level of confidentiality that BellSouth treats its own directory listing information, and BellSouth shall limit access to Knology's customer proprietary confidential directory information to those BellSouth employees who are involved in the preparation of listings.

5.7 <u>Optional Listings</u>. Additional listings and optional listings will be offered by BellSouth at tariffed rates as set forth in the General Subscriber Services Tariff.

5.8 <u>Delivery.</u> BellSouth or its agent shall deliver directories to Knology subscribers in accordance with prevailing standards.

6. Section 6 is left blank intentionally

7. Court Ordered Requests for Call Detail Records and Other Subscriber Information

- 7.1 Where BellSouth provides resold services or local switching, BellSouth maintains and shall maintain call detail records for Knology end users for limited time periods and may respond to subpoenas and court ordered requests for this information. BellSouth shall maintain such information for Knology end users for the same length of time it maintains such information for its own end users.
- 7.2 BellSouth, to the extent permitted by law, may respond to subpoenas and court ordered requests delivered directly to BellSouth for the purpose of providing call detail records when the targeted telephone numbers belong to Knology end users. Billing for such requests will be generated by BellSouth and directed to the law enforcement agency initiating the request.
- 7.3 Knology agrees that in cases where Knology receives subpoenas or court ordered requests for call detail records for targeted telephone numbers belonging to Knology end users, Knology will advise the law enforcement agency initiating the request to redirect the

subpoena or court ordered request to BellSouth. Billing for call detail information will be generated by BellSouth and directed to the law enforcement agency initiating the request.

- 7.4 In cases where the timing of the response to the law enforcement agency prohibits Knology from having the subpoena or court ordered request redirected to BellSouth by the law enforcement agency, Knology will furnish the official request to BellSouth for providing the call detail information. BellSouth will provide the call detail records to Knology and bill Knology for the information. Knology agrees to reimburse BellSouth for the call detail information provided.
- 7.5 Knology will provide Knology end user and/or other customer information that is available to Knology in response to subpoenas and court orders for Knology customer records. BellSouth will redirect subpoenas and court ordered requests for Knology end user and/or other customer information to Knology for the purpose of providing this information to the law enforcement agency.

1. Liability and Indemnification

- 8.1 <u>Knology Liability</u>. In the event that Knology consists of two (2) or more separate entities as set forth in this Agreement and/or any Amendments hereto, all such entities shall be jointly and severally liable for the obligations of Knology under this Agreement.
- 8.2 <u>Liability for Acts or Omissions of Third Parties</u>. Neither Party shall be liable to the other Party for any act or omission of another telecommunications company providing services to said other Party.
- 8.3 <u>Limitation of Liability</u>
- 8.3.1 Except for any indemnification obligations of the Parties hereunder, each Party's liability to the other for any action, loss, cost, claim, judgement, damages, injury or liability or expense, including without limitation reasonable attorney's fees (collectively, "Loss") relating to or arising out of any negligent act or omission in its performance of this Agreement whether in contract or in tort, shall be limited to a credit for the actual cost of the services or functions not performed or improperly performed.

Notwithstanding the foregoing, this limitation of liability shall not apply in the event of either Party's (1) gross negligence, (2) willful misconduct, or (3) material breach of any material term of this Agreement which has not been cured pursuant to the following terms. If either Party believes the other Party has materially breached this Agreement, the aggrieved party must notify the breaching Party and the breaching Party shall have sixty (60) days to cure such breach. Willful misconduct as used in this Section shall not include actions in reliance upon either Party's reasonable interpretation of this Agreement even if such term is ultimately

found to be erroneous by the Commission, FCC, or court of competent jurisdiction. Further, failure to pay any undisputed sums due under this Agreement shall not be subject to the notice and cure provisions of this Section.

- 8.3.2 Limitations in Tariffs. A Party may, in its sole discretion, provide in its tariffs and contracts with its End Users and third parties that relate to any service, product or function provided or contemplated under this Agreement, that to the maximum extent permitted by applicable law, such Party shall not be liable to the End User or third party for (i) any Loss relating to or arising out of this Agreement, whether in contract, tort or otherwise, that exceeds the amount such Party would have charged that applicable person for the service, product or function that gave rise to such Loss or (ii) for Consequential Damages (defined hereunder). To the extent that a Party elects not to place in its tariffs or contracts such limitations of liability, and the other Party incluse a loss as a result thereof, such Party shall indemnify and reimburse the other Party for that portion of the Loss that would have been limited had the first Party included in its tariffs and contracts the limitations of liability that such other Party included in its own tariffs at the time of such Loss.
- 8.3.3 Neither BellSouth nor Knology shall be liable for damages to the other Party's terminal location, equipment or End User premises resulting from the furnishing of a service, including, but not limited to, the installation and removal of equipment or associated wiring, except to the extent caused by a Party's negligence or willful misconduct or by a Party's failure to ground properly a local loop after disconnection.
- 8.3.4 Under no circumstance shall a Party be responsible or liable for indirect, incidental, or consequential damages, including, but not limited to, economic loss or lost business or profits, damages arising from the use or performance of equipment or software, or the loss of use of software or equipment, or accessories attached thereto, delay, error, or loss of data (collectively, "Consequential Damages"). In connection with this limitation of liability, each Party recognizes that the other Party may, from time to time, provide advice, make recommendations, or supply other analyses related to the services, or facilities described in this Agreement, and, while each Party shall use diligent efforts in this regard, the Parties acknowledge and agree that this limitation of liability shall apply to provision of such advice, recommendations, and analyses.
- 8.3.5 To the extent any specific provision of this Agreement purports to impose liability, or limitation of liability, on either Party different from or in conflict with the liability or limitation of liability set forth in this Section, then with respect to any facts or circumstances covered by such specific provisions, the liability or limitation of liability contained in such specific provision shall apply.
 - 8.4 <u>Indemnification for Certain Claims</u>. The Party providing services to the other hereunder, its Affiliates and its parent company, shall be indemnified, defended and held harmless by the Party receiving services hereunder from and against any Loss

arising from the receiving Party's use of the services provided by the providing Party under this Agreement to the extent such Loss pertains to (1) claims for libel, slander or invasion of privacy arising from the content of the receiving Party's own communications, or (2) any Loss claimed by the End User of the Party receiving services arising from such receiving Party's use or reliance on the providing Party's services, actions, duties, or obligations arising out of this Agreement.

- 8.4.1 <u>Obligation to Defend</u>. If the indemnifying Party, within a reasonable time following (but in any event no later than thirty (30) calendar days from the date of) receipt of notice of any Claim, either refuses or fails to undertake the defense of any Claim, then the indemnified Party, upon further written notice to the indemnifying Party, shall have the right (but not the obligation) to undertake its own defense, compromise or settlement of such Claim. If the indemnifying Party shall remain fully liable for (and shall promptly pay as and when incurred) all Losses related thereto, including without limitation any and all reasonable costs and expenses incurred by the indemnified Party in said defense, compromise or settlement.
- 8.4.2 <u>Compromise and Settlement</u>. The indemnifying Party shall consult with the indemnified Party prior to undertaking any compromise or settlement of any Claim, and the indemnified Party, will have the right, at its sole option and discretion, by written notice to the indemnifying Party, to refuse any such compromise or settlement that (in the indemnified Party's reasonable opinion) might prejudice the rights of the indemnified Party, and, to take over the defense, compromise or settlement of such Claim; provided, however, upon receipt of such written notice of refusal from the indemnified Party, the indemnifying Party will be deemed automatically released from any further obligation to indemnify, defend or hold the indemnified Party harmless from or against any Loss arising from such Claim (but such release will not affect the Parties rights or obligations arising under any other Claim).
- 8.5 Disclaimer. EXCEPT AS SPECIFICALLY PROVIDED TO THE CONTRARY IN THIS AGREEMENT. NEITHER PARTY MAKES ANY REPRESENTATIONS OR WARRANTIES TO THE OTHER PARTY CONCERNING THE SPECIFIC QUALITY OF ANY SERVICES, OR FACILITIES PROVIDED UNDER THIS AGREEMENT. THE PARTIES DISCLAIM, WITHOUT LIMITATION, ANY WARRANTY OR GUARANTEE OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE, ARISING FROM COURSE OF PERFORMANCE, COURSE OF DEALING, OR FROM USAGES OF TRADE.

9. Intellectual Property Rights and Indemnification

9.1 <u>No License</u>. No patent, copyright, trademark or other proprietary right is licensed, granted or otherwise transferred by this Agreement. Each Party is strictly prohibited from any use, including but not limited to in sales, in marketing or advertising of telecommunications services, of any name, service mark or trademark (collectively, the "Marks") of the other Party. The Marks include those Marks owned directly by each Party and those Marks that each Party has a legal and valid license to use.

9.2 <u>Ownership of Intellectual Property</u>. Any intellectual property that belongs to or is developed by a Party shall remain the exclusive property of that Party. Except for a limited right to use patents or copyrights to the extent necessary for the Parties to use any facilities or equipment (including software) or to receive any service solely as provided under this Agreement, no license in or right to use any patent, copyright, trademark or trade secret, or other proprietary or intellectual property right now or hereafter owned, controlled or licensable by a Party, is granted to the other Party or shall be implied or arise by estoppel. It is the responsibility of each Party to ensure at no additional cost to the other Party that it has obtained any necessary licenses in relation to intellectual property of third Parties used in its network that may be required to enable the other Party to use any facilities or equipment (including software), to receive any service, or to perform its respective obligations under this Agreement.

9.3 <u>Indemnification</u>. The Party providing a service pursuant to this Agreement will defend the Party receiving such service or data provided as a result of such service against claims of infringement arising solely from the use by the receiving Party of such service and will indemnify the receiving Party for any damages awarded based solely on such claims in accordance with Section 8 of this Agreement.

9.4 <u>Claim of Infringement</u>. In the event that use of any facilities or equipment (including software), becomes, or in reasonable judgment of the Party who owns the affected network is likely to become, the subject of a claim, action, suit, or proceeding based on intellectual property infringement, then said Party shall promptly and at its sole expense, but subject to the limitations of liability set forth below:

9.4.1 modify or replace the applicable facilities or equipment (including software) while maintaining form and function, or

9.4.2 obtain a license sufficient to allow such use to continue.

9.4.3 In the event 9.4.1 or 9.4.2 are commercially unreasonable, then said Party may, discontinue, upon reasonable notice, the provision of the services provided through use of, the affected facilities or equipment (including software), but solely to the extent required to avoid the infringement claim. In said event, the providing Party shall exercise its best reasonable commercial efforts to mitigate the effect of such discontinuation on the receiving Party, including, without limitation, providing functionally equivalent services to the receiving Party.

9.5 <u>Exception to Obligations</u>. Neither Party's obligations under this Section to the other Party ("Indemnitee") shall apply to the extent the infringement is caused by: (i)

modification of the facilities or equipment (including software) by the indemnitee; (ii) use by the indemnitee of the facilities or equipment (including software) in combination with equipment or facilities (including software) not provided or authorized by the indemnitor provided the facilities or equipment (including software) would not be infringing if used alone; or (iii) conformance to specifications of the indemnitee which would necessarily result in infringement or (iv) continued use by the indemnitee of the affected facilities or equipment (including software) after being placed on notice to discontinue use as set forth herein.

9.6 <u>Exclusive Remedy</u>. The foregoing shall constitute the Parties' sole and exclusive remedies and obligations with respect to a third Party claim of intellectual property infringement arising under this Agreement.

9.7 <u>Dispute Resolution</u>. Any claim arising under this Section 9 shall be excluded from the dispute resolution procedures set forth in Section 12 and shall be brought in a court of competent jurisdiction.

10. Treatment of Proprietary and Confidential Information

10.1 <u>Proprietary and Confidential Information</u>: Defined. It may be necessary for BellSouth and Knology, each as the "Discloser," to provide to the other Party, as "Recipient," certain proprietary and confidential information (including trade secret information) including but not limited to technical, financial, marketing, staffing and business plans and information, strategic information, proposals, request for proposals, specifications, drawings, prices, costs, procedures, processes, business systems, software programs, techniques, customer account data, call detail records and like information (collectively the Discloser's "Information"). To qualify for confidential treatment, all Information provided by Discloser to Recipient must be either (i) in written or other tangible or electronic form, clearly marked with a confidential and proprietary at the time of such disclosure and reduced to writing marked with a confidential and proprietary notice and provided to Recipient within thirty (30) calendar days after such oral or visual disclosure.

10.2 <u>Use and Protection of Information</u>. Recipient shall use the Information solely or the purpose(s) of performing this Agreement and Recipient shall protect Information from any use, distribution or disclosure except as permitted hereunder. Recipient will use the same standard of care to protect Information as Recipient uses to protect its own similar confidential and proprietary information, but not less than a reasonable standard of care. Recipient may disclose Information solely to the Authorized Representatives of the Recipient who (a) have a substantive need to know such Information in connection with performance of the Agreement; and either (i) have been advised of the confidential and proprietary nature of the Information or (ii) are under a professional obligation to preserve client confidences. "Authorized Representatives" are the officers, directors and employees of Recipient and its Affiliates, as well as Recipient's and its Affiliates' consultants, contractors, counsel and agents. 10.3 **Ownership, Copying & Return of Information**. Information remains at all times the property of Discloser. Recipient may make tangible or electronic copies, notes, summaries or extracts of Information only as necessary for use as authorized herein. All such tangible or electronic copies, notes, summaries or extracts must be marked with the same confidential and proprietary notice as appears on the original. Upon Discloser's request, all or any requested portion of the Information (including, but not limited to, tangible and electronic copies, notes, summaries or extracts of any information) will be promptly returned to Discloser or destroyed and Recipient will provide Discloser with written certification stating that such Information has been returned or destroyed.

10.4 <u>Exceptions</u>. Discloser's Information does not include: (a) any information publicly disclosed by Discloser; (b) any information Discloser in writing authorizes Recipient to disclose without restriction; (c) any information already lawfully known to Recipient at the time it is disclosed by the Discloser, without an obligation to keep confidential; (d) any information Recipient lawfully obtains from any source other than Discloser, provided that such source lawfully disclosed and/or independently developed such information or (e) any information independently developed by the Recipient without violation of this Section 10. If Recipient is required to provide Information to any court or government agency pursuant to written court order, subpoena, regulation or process of law, Recipient must first provide Discloser with prompt written notice of such requirement and cooperate with Discloser to appropriately protect against or limit the scope of such disclosure. To the fullest extent permitted by law, Recipient will continue to protect as confidential and proprietary all Information disclosed in response to a written court order, subpoena, regulation or process of law.

10.5 <u>Equitable Relief</u>. Recipient acknowledges and agrees that any breach or threatened breach of this Agreement is likely to cause Discloser irreparable harm for which money damages may not be an appropriate or sufficient remedy. Recipient therefore agrees that Discloser or its Affiliates, as the case may be, are entitled to receive injunctive or other equitable relief to remedy or prevent any breach or threatened breach of this Agreement, and to recover all costs in doing so (including, without limitation, reasonable attorneys' fees.) Such remedy is not the exclusive remedy for any breach or threatened breach of this Agreement, but is in addition to all other rights and remedies available at law or in equity.

10.6 <u>Survival of Confidentiality Obligations</u>. The Parties' rights and obligations under this Section 10 shall survive and continue in effect until two (2) years after the expiration or termination date of this Agreement with regard to all Information exchanged during the term of this Agreement. Thereafter, the Parties' rights and obligations hereunder survive and continue in effect with respect to any Information that is a trade secret under applicable law.

11. Assignments

Any assignment by either Party to any entity of any right, obligation or duty, or of any other interest hereunder, in whole or in part, without the prior written consent of the other Party shall be void, except as otherwise hereinafter provided. A Party may assign this Agreement in its

entirety to an Affiliate of the Party or any entity succeeding a Party by sale, merger, or acquisition without the consent of the other Party; provided, however, that the assignee is authorized as a CLEC in all states covered by this Agreement. The assigning Party shall notify the other Party in writing of such assignment thirty (30) days prior to the Effective Date thereof and, provided further, if the assignee is an assignee of Knology, the assignee must provide evidence of Commission CLEC certification. The Parties shall amend this Agreement to reflect such assignments and shall work cooperatively to implement any changes required due to such assignment. All obligations and duties of any Party under this Agreement shall be binding on all successors in interest and assigns of such Party. No assignment or delegation hereof shall relieve the assignor of its obligations under this Agreement in the event that the assignee fails to perform such obligations. In addition to the foregoing, either Party may assign its rights to payments hereunder to a commercial lending institution (a "Commercial Lender") upon sixty (60) days notice to the other Party; provided that any such assignment shall not affect either Party's obligations under this Agreement, regardless of whether the assigning Party is in default as to the assignee. In the event that a Party to this Agreement (the "Assigning Party") assigns its rights to receive payment hereunder to a Commercial Lender pursuant to the previous sentence, the other Party hereto ("the Obligated Party") shall retain the right to set-off against such assigned payment obligations any amounts owed to the Obligated Party by the Assigning Party, notwithstanding any notification given by the Commercial Lender to the Obligated Party, and notwithstanding any other provision set forth in Section 9-404(a) or (b) of the Uniform Commercial Code or equivalent thereof as in effect pursuant to the Governing Law section of this Agreement. Any such assignment to a Commercial Lender that could result in the assignee or a designee of the assignee operating as a carrier under this Agreement shall be void.

12. Resolution of Disputes

Except as otherwise stated in this Agreement, if any dispute arises as to the interpretation of any provision of this Agreement or as to the proper implementation of this Agreement, either Party, shall petition the Commission for a resolution of the dispute. However, each Party reserves any rights it may have to seek judicial review of any ruling made by the Commission concerning this Agreement. Nothing in this Section shall be deemed a waiver of any other remedies of either Party under this Agreement.

13. Taxes

13.1 <u>Definition</u>. For purposes of this Section, the terms "taxes" and "fees" shall include but not be limited to federal, state or local sales, use, excise, gross receipts or other taxes or taxlike fees of whatever nature and however designated (including tariff surcharges and any fees, charges or other payments, contractual or otherwise, for the use of public streets or rights-of-way, whether designated as franchise fees or otherwise) which are imposed, or sought to be imposed, on or with respect to the services furnished hereunder or measured by the charges or payments therefore.

13.2 Taxes and Fees Imposed Directly On Either Providing Party or Purchasing Party.

13.2.1 Taxes and fees imposed on the providing Party, which are not permitted or required to be passed on by the providing Party to its customer, shall be borne and paid by the providing Party.

13.2.2 Taxes and fees imposed on the purchasing Party, which are not required to be collected and/or remitted by the providing Party, shall be borne and paid by the purchasing Party.

13.3 <u>Taxes and Fees Imposed on Purchasing Party But Collected And Remitted</u> <u>By Providing Party</u>.

13.3.1 Taxes and fees imposed on the purchasing Party shall be borne by the purchasing Party, even if the obligation to collect and/or remit such taxes or fees is placed on the providing Party.

13.3.2 To the extent permitted by applicable law, any such taxes and/or fees shall be shown as separate items on applicable billing documents between the Parties. Notwithstanding the foregoing, the purchasing Party shall remain liable for any such taxes and fees regardless of whether they are actually billed by the providing Party at the time that the respective service is billed.

13.3.3 If the purchasing Party determines that in its opinion any such taxes or fees are not payable, the providing Party shall not bill such taxes or fees to the purchasing Party if the purchasing Party provides written certification, reasonably satisfactory to the providing Party, stating that it is exempt or otherwise not subject to the tax or fee, setting forth the basis therefore, and satisfying any other requirements under applicable law. If any authority seeks to collect any such tax or fee that the purchasing Party has determined and certified not to be payable, or any such tax or fee that was not billed by the providing Party, the purchasing Party shall have the right, at its own expense, to contest the same in good faith, in its own name or on the providing Party's behalf. In any such contest, the purchasing Party shall promptly furnish the providing Party with copies of all filings in any proceeding, protest, or legal challenge, all rulings issued in connection therewith, and all correspondence between the purchasing Party and the taxing authority.

13.3.4 In the event that all or any portion of an amount sought to be collected must be paid in order to contest the imposition of any such tax or fee, or to avoid the existence of a lien on the assets of the providing Party during the pendency of such contest, the purchasing Party shall be responsible for such payment and shall be entitled to the benefit of any refund or recovery.

13.3.5 If it is ultimately determined that any additional amount of such a tax or fee is due to the imposing authority, the purchasing Party shall pay such additional amount, including any interest and penalties thereon.

13.3.6 Notwithstanding any provision to the contrary, the purchasing Party shall protect, indemnify and hold harmless (and defend at the purchasing Party's expense) the providing Party from and against any such tax or fee, interest or penalties thereon, or other charges or payable expenses (including reasonable attorney fees) with respect thereto, which are incurred by the providing Party in connection with any claim for or contest of any such tax or fee.

13.3.7 Each Party shall notify the other Party in writing of any assessment, proposed assessment or other claim for any additional amount of such a tax or fee by a governmental authority; such notice to be provided, if possible, at least ten (10) days prior to the date by which a response, protest or other appeal must be filed, but in no event later than thirty (30) days after receipt of such assessment, proposed assessment or claim.

13.4 <u>Taxes and Fees Imposed on Providing Party But Passed On To Purchasing Party</u>.

13.4.1 Taxes and fees imposed on the providing Party, which are permitted or required to be passed on by the providing Party to its customer, shall be borne by the purchasing Party.

13.4.2 To the extent permitted by applicable law, any such taxes and/or fees shall be shown as separate items on applicable billing documents between the Parties. Notwithstanding the foregoing, the purchasing Party shall remain liable for any such taxes and fees regardless of whether they are actually billed by the providing Party at the time that the respective service is billed.

13.4.3 If the purchasing Party disagrees with the providing Party's determination as to the application or basis for any such tax or fee, the Parties shall consult with respect to the imposition and billing of such tax or fee. Notwithstanding the foregoing, the providing Party shall retain responsibility for determining whether and to what extent any such taxes or fees are applicable. The providing Party shall further retain responsibility for determining whether and how to contest the imposition of such taxes and fees; provided, however, that any such contest undertaken at the request of the purchasing Party shall be at the purchasing Party's expense. In the event that such contest must be pursued in the name of the providing Party, the providing Party shall permit the purchasing Party to pursue the contest in the name of the providing Party and the providing Party shall have the opportunity to participate fully in the preparation of such contest.

13.4.4. If after consultation in accordance with the preceding Section, the purchasing Party does not agree with the providing Party's final determination as to the application or basis of a particular tax or fee, and if the providing Party, after receipt of a written request by the purchasing Party to contest the imposition of such tax or fee with the imposing authority, fails or refuses to pursue such contest or to allow such contest by the purchasing Party, the purchasing Party may raise the dispute with the appropriate taxing authority or a court of competent jurisdiction. The filing of such dispute shall not relieve the purchasing Party from liability for any tax or fee billed by the providing Party pursuant to this subsection during the pendency of such proceeding. In the event that the purchasing Party prevails in such proceeding, it shall be entitled to a refund in accordance with the final decision therein. Notwithstanding the

foregoing, if at any time prior to a final decision in such proceeding the providing Party initiates a contest with the imposing authority with respect to any of the issues involved in such proceeding, the proceeding shall be dismissed as to such common issues and the final decision rendered in the contest with the imposing authority shall control as to such issues.

13.4.5 In the event that all or any portion of an amount sought to be collected must be paid in order to contest the imposition of any such tax or fee, or to avoid the existence of a lien on the assets of the providing Party during the pendency of such contest, the purchasing Party shall be responsible for such payment and shall be entitled to the benefit of any refund or recovery.

13.4.6 If it is ultimately determined that any additional amount of such a tax or fee is due to the imposing authority, the purchasing Party shall pay such additional amount, including any interest and penalties thereon.

13.4.7 Notwithstanding any provision to the contrary, the purchasing Party shall protect, indemnify and hold harmless (and defend at the purchasing Party's expense) the providing Party from and against any such tax or fee, interest or penalties thereon, or other reasonable charges or payable expenses (including reasonable attorney fees) with respect thereto, which are incurred by the providing Party in connection with any claim for or contest of any such tax or fee.

13.4.8 Each Party shall notify the other Party in writing of any assessment, proposed assessment or other claim for any additional amount of such a tax or fee by a taxing authority; such notice to be provided, if possible, at least ten (10) days prior to the date by which a response, protest or other appeal must be filed, but in no event later than thirty (30) days after receipt of such assessment, proposed assessment or claim.

13.5 <u>Mutual Cooperation</u>. In any contest of a tax or fee by one Party, the other Party shall cooperate fully by providing records, testimony and such additional information or assistance as may reasonably be necessary to pursue the contest. Further, the other Party shall be reimbursed for any reasonable and necessary out-of-pocket copying and travel expenses incurred in assisting in such contest.

14. Force Majeure

In the event performance of this Agreement, or any obligation hereunder, is either directly or indirectly prevented, restricted, or interfered with by reason of fire, flood, earthquake or like acts of God, wars, revolution, civil commotion, explosion, acts of public enemy, embargo, acts of the government in its sovereign capacity, labor difficulties, including without limitation, strikes, slowdowns, picketing, or boycotts, unavailability of equipment from vendor, or any other circumstances beyond the reasonable control and without the fault or negligence of the Party affected, the Party affected, upon giving prompt notice to the other Party, shall be excused from such performance on a day-to-day basis to the extent of such prevention, restriction, or interference (and the other Party shall likewise be excused from performance of its obligations on a day-to-day basis until the delay, restriction or interference has ceased); provided however, that the Party so affected shall promptly use diligent efforts to avoid or remove such causes of nonperformance and both Parties shall proceed whenever such causes are removed or cease, provided further, that neither Party shall be required to compensate the other for services or service not performed as a result of such causes.

15. Section Blank Intentionally

16. Modification of Agreement

16.1 BellSouth shall make available, pursuant to 47 USC § 252(i) and the FCC rules and regulations regarding such availability, to Knology, selected at its sole option, election, and discretion (i) any interconnection, service(s), Network Element(s), or any combination thereof (said interconnection, service(s), network element(s), or combinations collectively referred to herein as "Other Terms") provided under any other agreement filed and approved pursuant to 47 USC § 252 ("Other Agreement"), or (ii) the rates, terms, and conditions of any such Other Agreement in their entirety. Upon Knology's election of the foregoing alternative (i) all rates, terms and conditions that are interrelated or negotiated in exchange for such Other Terms shall be included in said Other Terms and this Agreement. The Parties shall execute an amendment to adopt such Other Terms or Other Agreement. Provided, however, the adopted Other Terms or Other Agreement, as the case may be, shall only apply to the same states in which said Other Agreement had been filed and approved pursuant to 47 USC Section 252; provided, further, except as otherwise required by the Act, the term of the amendment or replacement, as the case may be, is to be deemed coterminous with the term of such Other Agreement.

16.2 If Knology changes its name or makes changes to its company structure or identity due to a merger, acquisition, transfer or any other reason, it is the responsibility of Knology to notify BellSouth of said change and the Parties shall promptly execute an amendment to this Agreement, if necessary, to reflect said change.

16.3 No modification, amendment, supplement to, or waiver of the Agreement or any of its provisions shall be effective and binding upon the Parties unless it is made in writing and duly signed by the Parties.

16.4 Execution of this Agreement by either Party does not confirm or imply that the executing Party agrees with any decision(s) issued pursuant to the Telecommunications Act of 1996 or with the effect of those decisions on specific language in this Agreement. Neither Party waives its rights to appeal or otherwise challenge any such decision(s) and each Party reserves all of its rights to pursue any and all legal and/or equitable remedies, including appeals of any such decision(s).

16.5 In the event that any effective legislative, regulatory, judicial or other legal action materially affects any material terms of this Agreement, or the ability of Knology or BellSouth to perform any material terms of this Agreement, Knology or BellSouth may, on thirty (30) days' written notice require that such terms be renegotiated, and the Parties shall renegotiate in good faith such mutually acceptable new terms as may be required. In the event that such new terms are not renegotiated within ninety (90) days after such notice, the Dispute shall be referred to the Dispute

Resolution procedure set forth in Section 12 of the General Terms and Conditions – Part A of this Agreement.

The Parties intend that this Agreement be indivisible and nonseverable, and each of the 16.6 Parties acknowledges that it has assented to all of the covenants and promises in this Agreement as a single whole and that all of such covenants and promises, taken as a whole, constitute a single contract. Without limiting the generality of the foregoing, each of the Parties acknowledges that any provision by BellSouth of collocation space under this Agreement is solely for the purpose of facilitating the provision of other services under this Agreement and that neither Party would have contracted with respect to the provisioning of collocation space if the covenants and promises of the other Party with respect to the other services provided for under this Agreement had not been made. The Parties further acknowledge that this Agreement is intended to constitute a single transaction, that the obligations of the Parties under this Agreement are interdependent, and that any payment obligations or credits due by one Party to the other Party under this Agreement are intended to be recoupable against other payment obligations or credits due by one Party to the other Party under this Agreement. Notwithstanding the foregoing, in the event that Congress, the FCC, the Commission or any court of competent jurisdiction renders any decision that invalidates or makes illegal any provision contained in this Agreement, or otherwise finds any provision herein to be invalid or illegal, the Parties will immediately commence negotiations for a replacement agreement that does not contain the invalid or illegal provision. Any services ordered by the Parties during the negotiation of such replacement agreement will be provisioned at the rates set forth in this Agreement until such time as the replacement agreement is executed.

17. Waivers

A failure or delay of either Party to enforce any of the provisions hereof, to exercise any option which is herein provided, or to require performance of any of the provisions hereof shall in no way be construed to be a waiver of such provisions or options, and each Party, notwithstanding such failure, shall have the right thereafter to insist upon the specific performance of any and all of the provisions of this Agreement.

18. Governing Law

This Agreement shall be governed by, and construed and enforced in accordance with, the laws of the State of Georgia, without regard to its conflict of laws principles.

19. Notices

19.1 Every notice, consent, approval, or other communications required or contemplated by this Agreement shall be in writing and shall be delivered in person or given by postage prepaid mail, return receipt requested addressed to:

BellSouth Telecommunications, Inc.

CLEC Account Team 9th Floor 600 North 19th Street Birmingham, Alabama 35203

and

General Attorney - COU Suite 4300 675 W. Peachtree St. Atlanta, GA 30375

KNOLOGY

1241 O.G. Skinner Drive West Point, Georgia 31833 Attn: Chad Wachter, Esq. Vice President & General Counsel (Telephone) (706) 634-2663 (Facsimile) (706) 645-0148 (E-mail) chad.wachter@knology.com

with a copy to:

Walt Sapronov, Esq. Gerry & Sapronov, LLP Three Ravinia Drive, Suite 1455 Atlanta, Georgia 30346 (Telephone) (770) 399-9100 (Facsimile) (770) 395-0505 (E-mail) wsapronov@gstelecomlaw.com

or at such other address as the intended recipient previously shall have designated by written notice to the other Party.

19.2 Where specifically required, notices shall be by certified or registered mail. Unless otherwise provided in this Agreement, notice by mail shall be effective on the date it is officially recorded as delivered by return receipt or equivalent.

19.3 Notwithstanding the foregoing, BellSouth may provide Knology notice via Internet posting of price changes and changes to the terms and conditions of services available for resale per Commission Orders. BellSouth will also post changes to business processes and policies, notices of new service offerings, and changes to service offerings not requiring an amendment to this Agreement, notices required to be posted to BellSouth's website, and any other information of general applicability to CLECs. Any website posting that directly conflicts with a provision of

the Agreement shall not be effective unless this Agreement is amended to incorporate the provision contained in such web posting.

20. Implementation of Agreement

If Knology is a facilities based provider or a facilities based and resale provider entering into an Interconnection Agreement with BellSouth for the first time, this section shall apply. Within 60 days of the execution of this Agreement, the Parties will adopt a schedule for the implementation of the Agreement. The schedule shall state with specificity time frames for submission of including but not limited to, network design, interconnection points, collocation arrangement requests, pre-sales testing and full operational time frames for the business and residential markets. An implementation template to be used for the implementation schedule is contained in Attachment 10 of this Agreement.

21. Filing of Agreement

Upon execution of this Agreement BellSouth shall file this Agreement with the appropriate state regulatory agency pursuant to the requirements of Section 252 of the Act, and the Parties shall share equally any filing fees therefore. If the regulatory agency imposes any filing or public interest notice fees regarding the filing or approval of the Agreement, Knology shall be responsible for publishing the required notice and the publication and/or notice costs shall be borne by Knology. Notwithstanding the foregoing, this Agreement shall not be submitted for approval by the appropriate state regulatory agency unless and until such time as Knology is duly certified as a local exchange carrier in such state, except as otherwise required by a Commission.

22. Entire Agreement

This Agreement, together with its preamble, recitals and all Attachments and attached Schedules, incorporated herein by this reference, sets forth the Parties entire understanding and supersedes prior agreements between the Parties relating to the subject matter contained herein and merges all prior discussions between them, and neither Party shall be bound by any definitions, conditions, provision, representation, warrenty, covenant or promise other than as expressly stated in this Agreement or as is contemporaneously or subsequently set forth in writing and executed by a duly authorized officer or representative of the Party to be bound thereby. In the event of any conflict between the term(s) of this Agreement and those of an applicable tariff, the terms of this Agreement shall control.

This Agreement includes the following Attachments:

Attachment 1	Resale
Attachment 2	Network Elements and Other Services
Attachment 3	Network Interconnection
Attachment 4	Physical and Remote Site Collocation
A 1	

Attachment 5 Access to Numbers and Number Portability

Attachment 6 Pre-Ordering, Ordering and Provisioning, Maintenance and Repair

Attachment 7 Billing

Attachment 8 Rights of Way, Conduits and Pole Attachments

Attachment 9 Performance Measurements

Attachment 10 BellSouth Disaster Recovery Plan

Attachment 11 Bona Fide Request and New Business Requests Process

The following services are included as options for purchase by Knology pursuant to the terms and conditions set forth in this Agreement. Knology may elect to purchase said services by written request to its Account Manager if applicable:

Optional Daily Usage File (ODUF) Enhanced Optional Daily Usage File (EODUF) Access Daily Usage File (ADUF) Line Information Database (LIDB) Storage Centralized Message Distribution Service (CMDS) Calling Name (CNAM) LNP Data Base Query Service

23. Survival.

The following provisions are intended to survive the expiration or termination, for any reason whatsoever, of this Agreement for a period two (2) years following the date of expiration or termination: Section 8 (Liability and Indemnification), Section 9 (Intellectual Property Rights and Indemnification), Section 10 (Treatment of Proprietary and Confidential Information), Section 13 (Taxes), and any other provision of this Agreement which, by its terms or any reasonable interpretation thereof, is intended to survive.

24. Section left blank intentionally

25. Branding and Customer Contacts

25.1 In all cases of operator and directory assistance services Knology provides using services provided by BellSouth for Knology under this Agreement, BellSouth shall, where technically feasible, at Knology's sole discretion and expense, brand any and all such services in accordance with Attachment 2 to this Agreement.

25.2 Knology shall provide the exclusive interface to Knology subscribers, except as otherwise provided specifically in this Agreement.

25.3. All forms, business cards or other business materials furnished by BellSouth to Knology subscribers shall bear no corporate name, logo, trademark or trade name.

25.4 Neither Party will disparage or make any disparaging, derogatory or defamatory statements whatsoever concerning the other Party (or the services provided by the other Party) to any of the other Party's end-users or other customers or prospective customers. BellSouth shall

not solicit Knology end users as a result of Knology submitting a Local Service Request to BellSouth for services ordered via the terms and conditions of this Agreement.

26. Customer Information

Each Party shall fully comply with all Customer Proprietary Network Information ("CPNI") and carrier information set forth in Section 222 of the Act and the FCC's rules and regulations implementing, or promulgated under, Section 222 of the Act.

27. Miscellaneous

27.1 This Agreement may be executed in any number of counterparts, each of which shall be deemed an original, and which together shall constitute a single Agreement. A facsimile copy of a Party's execution of this Agreement shall be valid and binding upon the Party and must be followed as soon as practicable thereafter by the original version of such execution.

27.2 The Parties are independent contractors and nothing herein shall be construed to imply that they are partners, joint venturers or agents of one another.

27.3 Except as otherwise expressly provided in this Agreement, each of the remedies provided under this Agreement is cumulative and is in addition to any remedies that may be available at law, in equity or elsewhere in this Agreement.

27.4 Except as may be specifically set forth in this Agreement, this Agreement does not provide and shall not be construed to provide any person not a Party or proper assignee or successor hereunder with any beneficial interest, remedy, claim, liability, reimbursement, cause of action, or other privilege arising under or relating to this Agreement.

27.5 This Agreement was executed after arm's length negotiations between the undersigned Parties and reflects the conclusion of the undersigned that this Agreement is in the best interests of all Parties.

27.6 The headings of Articles and Sections of this Agreement are for convenience of reference only, and shall in no way define, modify or restrict the meaning or interpretation of the terms or provisions of this Agreement.

27.7 Pronouns used herein shall be construed as masculine, feminine, or neuter, and both singular and plural, as the context may require, and the term "person" shall include an individual, corporation, association, partnership, trust, and other organization.

27.8 All terms used but not otherwise defined in this Agreement have their customary meanings in accordance with the ordinary usage in the telecommunications industry.

28. Compliance with Applicable Law

Each Party shall comply at its own expense with Applicable Law.

29. Necessary Approvals

Each Party shall be responsible for obtaining and keeping in effect all approvals from, and rights granted by, governmental authorities, building and property owners, other carriers, and any other persons that may be required in connection with the performance of its obligations under this Agreement. Each Party shall reasonably cooperate with the other Party in obtaining and maintaining any required approvals and rights for which such Party is responsible.

30. Good Faith Performance

Each Party shall act in good faith in its performance under this Agreement and, in each case in which a Party's consent or agreement is required or requested hereunder, such Party shall not unreasonably withhold or delay such consent or agreement.

31. Nonexclusive Dealings

This Agreement does not prevent either Party from providing or purchasing services to or from any other person nor, except as provided in Section 252(i) of the Act, does it obligate either Party to provide or purchase any services (except insofar as the Parties are obligated to provide access to Interconnection, services and Network Elements to Knology as a requesting carrier under the Act).

32. CLEC Certification

32.1 Knology agrees to provide BellSouth in writing the certificate number, company number or docket number, for the docket pending certification, for all states covered by this Agreement except Kentucky prior to BellSouth filing this Agreement with the appropriate commission for approval.

32.2 Additionally, Knology will notify BellSouth in writing when it becomes certified or has a docket pending certification to operate in any other state in the BellSouth region. Upon notification, BellSouth will file this Agreement with the appropriate commission for approval.

IN WITNESS WHEREOF, the Parties have caused their duly appointed representatives to execute this Agreement the day and year above first written.

BellSouth Telecommunications, Inc.

Knology of Alabama, Inc.

<u>Signature on File</u>

Elizabeth R. A. Shiroishi Name Assistant Director Title

<u>12/11/2002</u> Date <u>Signature on File</u>

Felix L. Boccucci

Name

_Vice President Title

<u>11/19/2002</u> Date

Knology of Florida, Inc.

Signature on File Signature

Felix L. Boccucci
Name

Vice President_____ Title

<u>11/19/2002</u> Date

Knology of Georgia, Inc.

General Terms and Conditions – Part A Page 24

Signature on File Signature

Felix L. Boccucci

_11/19/2002_____

Date

Knology of Kentucky, Inc.

Signature of File

Felix L. Boccucci

Name

<u>Vice President</u> Title

<u>11/19/2002</u> Date

General Terms and Conditions – Part A Page 25

Knology of South Carolina, Inc.

<u>Signature On File</u>

_Felix L. Boccucci_____ Name

___Vice President_____ Title

____11/19/20002_____

Date

Knology of Tennessee, Inc.

Signature On File Signature

_Felix L. Boccucci Name

Vice President Title

___11/19/2002_

Date

This page is

intentionally blank

Attachment 1 Page 1

Attachment 1

Resale

Table of Contents

1.	Discount Rates	3
2.	Definition of Terms	3
3.	General Provisions	4
4.	BellSouth's Provision of Services to Knology	8
5.	Maintenance of Services	9
6.	Establishment of Service	10
7.	Discontinuance of Service	10
8.	Operator Services (Operator Call Processing and Directory Assistance)	11
9.	Line Information Database (LIDB)	15
10	. RAO Hosting	15
Re	esale Restrictions Exhibit	t A
Li	ne Information Database (LIDB) Storage Agreemt Exhibit	t B
Re	esale Discounts and Rates Exhibit	t C

RESALE

1. Discount Rates

- 1.1 The discount rates applied to Knology purchases of BellSouth Telecommunications Services for the purpose of resale shall be as set forth in Exhibit C. Such discounts have been determined by the applicable Commission to reflect the costs avoided by BellSouth when selling a service for wholesale purposes.
- 1.2 The telecommunications services available for purchase by Knology for the purposes of resale to Knology's End Users shall be available at BellSouth's tariffed rates less the discount set forth in Exhibit C to this Agreement and subject to the exclusions and limitations set forth in Exhibit A to this Agreement.

2. Definition of Terms

- 2.1 COMPETITIVE LOCAL EXCHANGE COMPANY (CLEC) means a telephone company certificated by the Commission to provide local exchange service within BellSouth's franchised area.
- 2.2 CUSTOMER OF RECORD means the entity responsible for placing application for service; requesting additions, rearrangements, maintenance or discontinuance of service; payment in full of charges incurred such as non-recurring, monthly recurring, toll, directory assistance, etc.
- 2.3 DEPOSIT means assurance provided by a customer in the form of cash, surety bond or bank letter of credit to be held by BellSouth.
- 2.4 END USER means the ultimate user of the Telecommunications Service.
- 2.5 END USER CUSTOMER LOCATION means the physical location of the premises where an End User makes use of the telecommunications services.
- 2.6 NEW SERVICES means functions, features or capabilities that are not currently offered by BellSouth. This includes packaging of existing services or combining a new function, feature or capability with an existing service.
- 2.7 RESALE means an activity wherein a certificated CLEC, such as Knology, subscribes to the telecommunications services of BellSouth and then offers those telecommunications services to the public.

3. General Provisions

- 3.1 All of the negotiated rates, terms and conditions set forth in this Attachment pertain to the resale of BellSouth's retail telecommunications services and other services specified in this Attachment. Subject to effective and applicable FCC and Commission rules and orders, BellSouth shall make available to Knology for resale those telecommunications services BellSouth makes available, pursuant to its General Subscriber Services Tariff and Private Line Services Tariff, to customers who are not telecommunications carriers.
- 3.1.1 When Knology provides Resale service in a cross boundary area (areas that are part of the local serving area of another state's exchange) the rates, regulations and discounts for the tariffing state will apply. Billing will be from the serving state.
- 3.1.2 In Tennessee, if Knology does not resell Lifeline services to any end users, and if Knology agrees to order an appropriate Operator Services/Directory Services block as set forth in BellSouth's General Subscriber Services Tariff, the discount shall be 21.56%.
- 3.1.2.1 In the event Knology resells Lifeline service to any end user in Tennessee, BellSouth will begin applying the 16% discount rate to all services. Upon Knology and BellSouth's implementation of a billing arrangement whereby a separate Master Account (Q-account) associated with a separate Operating Customer Number (OCN) is established for billing of Lifeline service end users, the discount shall be applied as set forth in 3.1.2 preceding for the non-Lifeline affected Master Account (Q-account).
- 3.1.2.2 Knology must provide written notification to BellSouth within 30 days prior to providing its own operator services/directory services or orders the appropriate operator services/directory assistance blocking, to qualify for the higher discount rate of 21.56%.
- 3.2 Knology may purchase resale services from BellSouth for their own use in operating their business. The resale discount will apply to those services under the following conditions:
- 3.2.1 Knology must resell services to other End Users.
- 3.2.2 Knology cannot be a competitive local exchange telecommunications company for the single purpose of selling to themselves.
- 3.3 Knology will be the customer of record for all services purchased from BellSouth. Except as specified herein, BellSouth will take orders from, bill and receive payment from Knology for said services.
- 3.4 Knology will be BellSouth's single point of contact for all services purchased pursuant to this Agreement. BellSouth shall have no contact with the End User

except to the extent provided for herein. Each Party shall provide to the other a nation wide (50 states) toll-free contact number for purposes of repair and maintenance.

- 3.5 BellSouth will continue to bill the End User for any services that the End User specifies it wishes to receive directly from BellSouth. BellSouth maintains the right to serve directly any End User within the service area of Knology. BellSouth will continue to market directly its own telecommunications products and services and in doing so may establish independent relationships with End Users of Knology. Neither Party shall interfere with the right of any person or entity to obtain service directly from the other Party.
- 3.5.1 When a subscriber of Knology or BellSouth elects to change his/her carrier to the other Party, both Parties agree to release the subscriber's service to the other Party concurrent with the due date of the service order, which shall be established based on the standard interval for the subscriber's requested service as set forth in the BellSouth Product and Services Interval Guide.
- 3.5.2 BellSouth and Knology will refrain from contacting subscribers who have placed or whose selected carrier has placed on their behalf an order to change his/her service provider from BellSouth or Knology to the other Party until such time that the order for service has been completed.
- 3.6 Current telephone numbers may normally be retained by the End User and are assigned to the service furnished. However, neither Party nor the End User has a property right to the telephone number or any other call number designation associated with services furnished by BellSouth, and no right to the continuance of service through any particular central office. BellSouth reserves the right to change such numbers, or the central office designation associated with such numbers, or both, whenever BellSouth deems it necessary to do so in the conduct of its business and in accordance with BellSouth practices and procedures on a nondiscriminatory basis.
- 3.7 Where BellSouth provides local switching or resold services to Knology, BellSouth will provide Knology with on line access to intermediate telephone numbers as defined by applicable FCC rules and regulations on a first come first served basis. Knology acknowledges that such access to numbers shall be in accordance with the appropriate FCC rules and regulations. Knology acknowledges that there may be instances where there is a shortage of telephone numbers in a particular Common Language Location Identifier Code (CLLIC); and in such instances, Knology shall return unused intermediate telephone numbers to BellSouth upon BellSouth's request. BellSouth shall make all such requests on a nondiscriminatory basis.
- 3.8 BellSouth will allow Knology to designate up to 100 intermediate telephone numbers per CLLIC, for Knology's sole use. Assignment, reservation and use of

telephone numbers shall be governed by applicable FCC rules and regulations. Knology acknowledges that there may be instances where there is a shortage of telephone numbers in a particular CLLIC and BellSouth has the right to limit access to blocks of intermediate telephone numbers. These instances include: 1) where jeopardy status has been declared by the North American Numbering Plan (NANP) for a particular Numbering Plan Area (NPA); or 2) where a rate center has less than six months supply of numbering resources.

- 3.9 Service is furnished subject to the condition that it will not be used for any unlawful purpose.
- 3.10 Service will be discontinued if any law enforcement agency advises that the service being used is in violation of the law.
- 3.11 BellSouth can refuse service when it has grounds to believe that service will be used in violation of the law.
- 3.12 BellSouth will cooperate with law enforcement agencies with subpoenas and court orders relating to Knology's End Users, pursuant to Section 6 of the General Terms and Conditions.
- 3.13 If Knology or its End Users utilize a BellSouth resold telecommunications service in a manner other than that for which the service was originally intended as described in BellSouth's retail tariffs, Knology has the responsibility to notify BellSouth. BellSouth will only provision and maintain said service consistent with the terms and conditions of the tariff describing said service.
- 3.14 Facilities and/or equipment utilized by BellSouth to provide service to Knology remain the property of BellSouth.
- 3.15 White page directory listings for Knology End Users will be provided in accordance with Section 5 of the General Terms and Conditions.
- 3.16 Service Ordering and Operational Support Systems (OSS)
- 3.16.1 Knology must order services through resale interfaces, i.e., the Local Carrier Service Center (LCSC) and/or appropriate Complex Resale Support Group (CRSG) pursuant to this Agreement. BellSouth has developed and made available interactive interfaces by which Knology may submit LSRs electronically as set forth in Attachment 6 of this Agreement. Service orders will be in a standard format designated by BellSouth.
- 3.16.2 LSRs submitted by means of one of these interactive interfaces will incur an OSS electronic charge as set forth in Exhibit C to this Agreement. An individual LSR will be identified for billing purposes by its Purchase Order Number (PON). LSRs submitted by means other than one of these interactive interfaces (Mail, fax, courier, etc.) will incur a manual order charge as set forth in Exhibit C to this

Agreement. Supplements or clarifications to a previously billed LSR will not incur another OSS charge.

- 3.16.3 <u>Denial/Restoral OSS Charge</u>. In the event Knology provides a list of customers to be denied and restored, rather than an LSR, each location on the list will require a separate PON and therefore will be billed as one LSR per location.
- 3.16.4 Cancellation OSS Charge. Knology will incur an OSS charge for an accepted LSR that is later canceled.
- 3.17 Where available to BellSouth's End Users, BellSouth shall provide the following telecommunications services at a discount to allow for voice mail services:
 - Message Waiting Indicator ("MWI"), stutter dialtone and message waiting light feature capabilities
 - Call Forward Busy Line ("CF/B")
 - Call Forward Don't Answer ("CF/DA")

Further, BellSouth messaging services set forth in BellSouth's Messaging Service Information Package shall be made available for resale without the wholesale discount.

- 3.18 BellSouth shall provide branding for, or shall unbrand, voice mail services for Knology per the Bona Fide Request/New Business Request process as set forth in Attachment 11 of the General Terms and Conditions.
- 3.19 BellSouth's Inside Wire Maintenance Service Plan is available for resale at rates, terms and conditions as set forth by BellSouth and without the wholesale discount.
- 3.20 In the event Knology acquires an end user whose service is provided pursuant to a BellSouth Special Assembly, BellSouth shall make available to Knology that Special Assembly at the wholesale discount at Knology's option. Knology shall be responsible for all terms and conditions of such Special Assembly including but not limited to termination liability if applicable.
- 3.21 BellSouth shall provide 911/E911 for Knology customers in the same manner that it is provided to BellSouth customers. BellSouth shall provide and validate Knology customer information to the PSAP. BellSouth shall use its service order process to update and maintain, on the same schedule that it uses for its customers, the Knology customer service information in the ALI/DMS (Automatic Location Identification/Location Information) databases used to support 911/E911 services.
- 3.22 BellSouth shall bill, and Knology shall pay, the End User line charge associated with implementing Number Portability as set forth in BellSouth's FCC No. 1 tariff. This charge is not subject to the wholesale discount.
3.23 Pursuant to 47 CFR Section 51.617, BellSouth will bill to Knology, and Knology shall pay, End User common line charges identical to the End User common line charges BellSouth bills its End Users.

4. BellSouth's Provision of Services to Knology

- 4.1 Resale of BellSouth services shall be as follows:
- 4.1.1 The resale of telecommunications services shall be limited to users and uses conforming to the class of service restrictions.
- 4.1.2 Hotel and Hospital PBX services are the only telecommunications services available for resale to Hotel/Motel and Hospital End Users, respectively. Similarly, Access Line Service for Customer Provided Coin Telephones is the only local service available for resale to Payphone Service Provider (PSP) customers. Shared Tenant Service customers can only be sold those local exchange access services available in BellSouth's A23 Shared Tenant Service Tariff in the states of Florida, Georgia, North Carolina and South Carolina, and in A27 in the states of Alabama, Kentucky, Louisiana, Mississippi and Tennessee.
- 4.1.3 BellSouth reserves the right to periodically audit services purchased by Knology to establish authenticity of use. Such audit shall not occur more than once in a calendar year. Knology shall make any and all records and data available to BellSouth or BellSouth's auditors on a reasonable basis. BellSouth shall bear the cost of said audit. Any information provided by Knology for purposes of such audit shall be deemed Confidential Information pursuant to the General Terms and Conditions of this Agreement.
- 4.2 Subject to Exhibit A hereto, resold services can only be used in the same manner as specified in BellSouth's Tariffs. Resold services are subject to the same terms and conditions as are specified for such services when furnished to an individual End User of BellSouth in the appropriate section of BellSouth's Tariffs. Specific tariff features (e.g. a usage allowance per month) shall not be aggregated across multiple resold services.
- 4.3 Knology may resell services only within the specific service area as defined in its certificate of operation approved by the Commission.
- 4.4 If Knology cancels an order for resold services, any costs incurred by BellSouth in conjunction with provisioning of such order will be recovered in accordance with BellSouth's General Subscriber Services Tariffs and Private Line Services Tariffs.
- 4.5 Service Jointly Provisioned with an Independent Company or Competitive Local Exchange Company Areas

- 4.5.1 BellSouth will in some instances provision resold services in accordance with the General Subscriber Services Tariff and Private Line Tariffs jointly with an Independent Company or other Competitive Local Exchange Carrier.
- 4.5.2 When Knology assumes responsibility for such service, all terms and conditions defined in the Tariff will apply for services provided within the BellSouth service area only.
- 4.5.3 Service terminating in an Independent Company or other Competitive Local Exchange Carrier area will be provisioned and billed by the Independent Company or other Competitive Local Exchange Carrier directly to Knology.
- 4.5.4 Knology must establish a billing arrangement with the Independent Company or other Competitive Local Exchange Carrier prior to assuming an end user account where such circumstances apply.
- 4.5.5 Specific guideline regarding such service are available on BellSouth's website @ www.interconnection.bellsouth.com.

5. Maintenance of Services

- 5.1 Services resold pursuant to this Attachment and BellSouth's General Subscriber Service Tariff and Private Line Service Tariff and facilities and equipment provided by BellSouth shall be maintained by BellSouth.
- 5.2 Knology or its End Users may not rearrange, move, disconnect, remove or attempt to repair any facilities owned by BellSouth except with the written consent of BellSouth.
- 5.3 Knology accepts responsibility to notify BellSouth of situations that arise that may result in a service problem.
- 5.4 Knology will contact the appropriate repair centers in accordance with procedures established by BellSouth.
- 5.5 For all repair requests, Knology shall adhere to BellSouth's prescreening guidelines prior to referring the trouble to BellSouth.
- 5.6 BellSouth will bill Knology for handling troubles that are found not to be in BellSouth's network pursuant to its standard time and material charges. The standard time and material charges will be no more than what BellSouth charges to its retail customers for the same services.
- 5.7 BellSouth reserves the right to contact Knology's End Users, if deemed necessary, for maintenance purposes.

6. Establishment of Service

- 6.1 After receiving certification as a local exchange company from the appropriate regulatory agency, Knology will provide the appropriate BellSouth service center the necessary documentation to enable BellSouth to establish a master account for Knology's resold services. Such documentation shall include the Application for Master Account, proof of authority to provide telecommunications services, an Operating Company Number ("OCN") assigned by the National Exchange Carriers Association ("NECA") and a tax exemption certificate, if applicable.
- 6.2 Knology shall provide to BellSouth a blanket letter of authorization ("LOA") certifying that Knology will have End User authorization prior to viewing the End User's customer service record or switching the End User's service. BellSouth will not require End User confirmation prior to establishing service for Knology's End User customer. Knology must, however, be able to demonstrate End User authorization upon request.
- 6.3 BellSouth will accept a request directly from the End User for conversion of the End User's service from Knology to BellSouth or will accept a request from another CLEC for conversion of the End User's service from Knology to such other CLEC. Upon completion of the conversion BellSouth will notify Knology that such conversion has been completed.

7. Discontinuance of Service

- 7.1 The procedures for discontinuing service to an End User are as follows:
- 7.1.1 BellSouth will deny service to Knology's End User on behalf of, and at the request of, Knology. Upon restoration of the End User's service, restoral charges will apply and will be the responsibility of Knology.
- 7.1.2 At the request of Knology, BellSouth will disconnect a Knology End User customer.
- 7.1.3 All requests by Knology for denial or disconnection of an End User for nonpayment must be in writing.
- 7.1.4 Knology will be made solely responsible for notifying the End User of the proposed disconnection of the service.
- 7.1.5 BellSouth will continue to process calls made to the Annoyance Call Center and will advise Knology when it is determined that annoyance calls are originated from one of its End User's locations. BellSouth shall be indemnified, defended and held harmless by Knology and/or the End User against any claim, loss or damage arising from providing this information to Knology. It is the responsibility of Knology to take the corrective action necessary with its End Users who make

annoying calls. (Failure to do so will result in BellSouth's disconnecting the End User's service.)

8.0 Operator Services (Operator Call Processing and Directory Assistance)

- 8.1 Operator Services provides: (1) operator handling for call completion (for example, collect, third number billing, and manual calling-card calls). (2) operator or automated assistance for billing after the end user has dialed the called number (for example, calling card calls); and (3) special services including but not limited to Busy Line Verification and Emergency Line Interrupt (ELI), Emergency Agency Call and Operator-assisted Directory Assistance.
- 8.2 Upon request for BellSouth Operator Call Processing, BellSouth shall:
- 8.2.1 Process 0+ and 0- dialed local calls
- 8.2.2 Process 0+ and 0- intraLATA toll calls.
- 8.2.3 Process calls that are billed to Knology end user's calling card that can be validated by BellSouth.
- 8.2.4 Process person-to-person calls.
- 8.2.5 Process collect calls.
- 8.2.6 Provide the capability for callers to bill a third party and shall also process such calls.
- 8.2.7 Process station-to-station calls.
- 8.2.8 Process Busy Line Verify and Emergency Line Interrupt requests.
- 8.2.9 Process emergency call trace originated by Public Safety Answering Points.
- 8.2.10 Process operator-assisted directory assistance calls.
- 8.2.11 Adhere to equal access requirements, providing Knology local end users the same IXC access that BellSouth provides its own operator service.
- 8.2.12 Exercise at least the same level of fraud control in providing Operator Service to Knology that BellSouth provides for its own operator service.
- 8.2.13 Perform Billed Number Screening when handling Collect, Person-to-Person, and Billed-To-Third-Party calls.
- 8.2.14 Direct customer account and other similar inquiries to the customer service center designated by Knology.

- 8.2.15 Provide call records to Knology in accordance with ODUF standards.
- 8.2.16 The interface requirements shall conform to the interface specifications for the platform used to provide Operator Services as long as the interface conforms to industry standards.
- 8.3 Directory Assistance Service
- 8.3.1 Directory Assistance Service provides local end user telephone number listings with the option to complete the call at the caller's direction separate and distinct from local switching.
- 8.3.2 Directory Assistance Service shall provide up to two listing requests per call, if available and if requested by Knology's end user. BellSouth shall provide calleroptional directory assistance call completion service at rates contained in Exhibit C to one of the provided listings.
- 8.3.3 Directory Assistance Service Updates
- 8.3.3.1 BellSouth shall update end user listings changes daily. These changes include:
- 8.3.3.1.1 New end user connections
- 8.3.3.1.2 End user disconnections
- 8.3.3.1.3 End user address changes
- 8.3.3.2 These updates shall also be provided for non-listed and non-published numbers for use in emergencies.
- 8.4 Branding for Operator Call Processing and Directory Assistance
- 8.4.1 BellSouth's branding feature provides a definable announcement to Knology end users using Directory Assistance (DA)/ Operator Call Processing (OCP) prior to placing such end users in queue or connecting them to an available operator or automated operator system. This feature allows Knology's name on whose behalf BellSouth is providing Directory Assistance and/or Operator Call Processing. Rates for the branding features are set forth in Exhibit C.
- 8.4.2 BellSouth offers three branding offering option to Knology when ordering BellSouth's Directory Assistance and Operator Call Processing: BellSouth Branding, Unbranding and Custom Branding.
- 8.4.3 Upon receipt of the branding order from Knology, the order is considered firm after ten (10) business days. Should Knology decide to cancel the order, written notification to Knology's BellSouth Account Executive is required. If Knology

decides to cancel after ten (10) business days from receipt of the branding order, Knology shall pay all charges per the order.

- 8.4.4 Selective Call Routing using Line Class Codes (SCR-LCC)
- 8.4.4.1 Where Knology resells BellSouth's services and utilizes an operator services provider other than BellSouth, BellSouth will route Knology's end user calls to that provider through Selective Call Routing.
- 8.4.4.2 Selective Call Routing using Line Class Codes (SCR-LCC) provides the capability for Knology to have its OCP/DA calls routed to BellSouth's OCP/DA platform for BellSouth provided Custom Branded or Unbranded OCP/DA or to its own or an alternate OCP/DA platform for Self-Branded OCP/DA. SCR-LCC is only available if line class code capacity is available in the requested BellSouth end office switches.
- 8.4.4.3 Custom Branding for Directory Assistance is not available for certain classes of service, including but not limited to Hotel/Motel services, WATS service and certain PBX services.
- 8.4.4.4 Where available, Knology specific and unique line class codes are programmed in each BellSouth end office switch were Knology intends to service end users with customized OCP/DA branding. The line class codes specifically identify Knology's end users so OCP/DA calls can be routed over the appropriate trunk group to the request OCP/DA platform. Additional line class codes are required in each end office if the end office serves multiple NPAs (i.e., a unique LCC is required per NPA), and/or if the end office switch serves multiple rate areas and Knology intends to provide Knology-branded OCP/DA to its end users in these multiple rate areas.
- 8.4.4.5 SCR-LCC supporting Custom Branding and Self Branding require Knology to order dedicated transport and trunking from each BellSouth end office identified by Knology, either to the BellSouth Traffic Operator Position System (TOPS) for Custom Branding or to the Knology Operator Service Provider for Self Branding. Separate trunk groups are required for Operator Services and for Directory Assistance. Rates for transport and trunks are as set forth in applicable BellSouth Tariffs.
- 8.4.4.6 The rates for SCR-LCC are as set forth in Exhibit C of this Attachment. There is a nonrecurring charge for the establishment of each Line Class Code in each BellSouth central office.
- 8.4.4.7 Unbranded Directory Assistance and/or Operator Call Processing calls ride common trunk groups provisioned by BellSouth from those end offices identified by Knology to the BellSouth Tops. The calls are routed to "No Announcement."
- 8.4.5 Branding via Originating Line Number Screening (OLNS)

- 8.4.5.1 BellSouth Branding, Unbranding and Custom Branding are also available for Directory Assistance, Operator Call Processing or both via OLNS software. When utilizing this method of Unbranding or Custom Branding, Knology shall not be required to purchase direct trunking.
- 8.4.5.2 For Bellsouth to provide Unbranding or Custom Branding via OLNS software for Operator Call Processing or for Directory Assistance, Knology must have its Operating Company Number ("OCN(s)") and telephone numbers reside in BellSouth's LIDB; however, a BellSouth LIDB Storage Agreement is not required. To implement Unbranding and Custom Branding via OLNS software, Knology must submit a manual order form which requires, among other things, Knology's OCN and a forecast for the traffic volume anticipated for each BellSouth TOPS during the peak busy hour. Knology shall provide updates to such forecast on a quarterly basis and at any time such forecasted traffic volumes are expected to change significantly. Upon Knology's purchase of Unbranding or Custom Branding using OLNS software for any particular TOPS, all Knology end users served by that TOPS will receive the Unbranded "no announcement" or the Custom Branded announcement.
- 8.4.5.3 Rates for Unbranding and Custom Branding via OLNS software for Directory Assistance and for Operator Call Processing are as set forth in Exhibit C of this Attachment. Notwithstanding anything to the contrary in this Agreement, to the extent BellSouth is unable to bill Knology applicable charges currently, BellSouth shall track such charges and will bill the same retroactively at such time as a billing process is implemented. In addition to the charges for Unbranding and Custom Branding via OLNS software, Knology shall continue to pay BellSouth applicable labor and other charges for the use of BellSouth's Directory Assistance and Operator Call Processing platforms as set forth in Exhibit C of this Attachment.
- 8.4.5.4 Customized Branding includes charges for the recording of the branding announcement and the loading of the audio units in each TOPS Switch and Network Applications Vehicles (NAV) equipment for which Knology requires service.
- 8.4.5.5 Directory Assistance customized branding uses:
- 8.4.5.5.1 the recording of Knology
- 8.4.5.5.2 the loading of-the recording in switch.
- 8.4.5.6 Operator Call Processing customized branding uses:
- 8.4.5.6.1 the recording of Knology
- 8.4.5.6.2 the loading of the_recording each switch

8.4.5.6.3 the loading on the Network Applications Vehicle (NAV). All NAV shelves within the region where the customer is offering service must be loaded.

9. Line Information Database (LIDB)

- 9.1 BellSouth will store in its Line Information Database (LIDB) records relating to service only in the BellSouth region. The LIDB Storage Agreement is included in this Attachment as Exhibit B.
- 9.2 BellSouth will provide LIDB Storage upon written request to Knology's Account Manager stating a requested activation date.

10. RAO Hosting

10.1 RAO Hosting is not required for resale in the BellSouth region.

Attachment 1 Page 16 Exhibit A

EXCLUSIONS AND LIMITATIONS ON SERVICES AVAILABLE FOR RESALE (Note 3)

			AL		FL	(GA	K	Y]	LA	Ι	MS]	NC	5	SC	r	ГN
	Type of Service	Resale	Discount	Resale	Discount	Resale	Discount	Resale	Discou nt	Resale	Discount	Resale	Discount	Resale	Discount	Resale	Discount	Resale	Discount
1	Grandfathered Services (Note 1)	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
2	Promotions - > 90 Days(Note 2)	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
3	Promotions - ≤ 90 Days (Note 2)	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No
4	Lifeline/Link Up Services	Yes	Yes	Yes	Yes	Yes	Yes	No	No	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
5	911/E911 Services	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	No	No	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
6	N11 Services	Yes	Yes	Yes	Yes	Yes	Yes	No	No	No	No	Yes	Yes	Yes	Yes	No	No	Yes	Yes
7	MemoryCall [®] Service	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No
8	Mobile Services	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No
9	Federal Subscriber Line Charges	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No
10	Non-RecurCharges	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	No
11	End User Line Chg- Number Portability	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No
12	Public Telephone Access Svc(PTAS)	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	No	Yes	Yes
13	Inside Wire Maint Service Plan	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No
	Applicable No																		
	1. Grandfathere																		
	2. Where availabl	e for res	sale, prom	otions v	will be ma	de avail	able only t	to End Us	sers who	would h	nave quali	fied for	the promo	tion had	l it been p	rovided	by BellSo	uth dire	etly.
	3. Some of BellSo	outh's lo	cal exchan	nge and	toll telecon	mmunic	cations serv	vices are	not avail	able in	certain cei	ntral off	ices and an	reas.					

LINE INFORMATION DATA BASE (LIDB)

RESALE STORAGE AGREEMENT

I. Definitions (from Addendum)

- A. Billing number a number used by BellSouth for the purpose of identifying an account liable for charges. This number may be a line or a special billing number.
- B. Line number a ten-digit number assigned by BellSouth that identifies a telephone line associated with a resold local exchange service, or with a SPNP arrangement.
- C. Special billing number a ten-digit number that identifies a billing account established by BellSouth in connection with a resold local exchange service or with a SPNP arrangement.
- D. Calling Card number a billing number plus PIN number assigned by BellSouth.
- E. PIN number a four-digit security code assigned by BellSouth that is added to a billing number to compose a fourteen-digit calling card number.
- F. Toll billing exception indicator associated with a billing number to indicate that it is considered invalid for billing of collect calls or third number calls or both, by Knology.
- G. Billed Number Screening refers to the activity of determining whether a toll billing exception indicator is present for a particular billing number.
- H. Calling Card Validation refers to the activity of determining whether a particular calling card number exists as stated or otherwise provided by a caller.
- I. Billing number information information about billing number or Calling Card number as assigned by BellSouth and toll billing exception indicator provided to BellSouth by Knology.

II. General

A. This Agreement sets forth the terms and conditions pursuant to which BellSouth agrees to store in its LIDB certain information at the request of Knology and pursuant to which BellSouth, its LIDB customers and Knology shall have access to such information. In addition, this Agreement sets forth the terms and conditions for Knology's provision of billing number information to BellSouth for inclusion in BellSouth's LIDB. Knology understands that BellSouth provides access to

information in its LIDB to various telecommunications service providers pursuant to applicable tariffs and agrees that information stored at the request of Knology, pursuant to this Agreement, shall be available to those telecommunications service providers. The terms and conditions contained herein shall hereby be made a part of this Interconnection/Resale Agreement upon notice to Knology's account team and/or Local Contract Manager to activate this LIDB Storage Agreement. The General Terms and Conditions of the Interconnection/Resale Agreement shall govern this LIDB Storage Agreement. The terms and conditions contained in the attached Addendum are hereby made a part of this LIDB Storage Agreement as if fully incorporated herein.

- B. BellSouth will provide responses to on-line, call-by-call queries to billing number information for the following purposes:
 - 1. Billed Number Screening

BellSouth is authorized to use the billing number information to determine whether Knology has identified the billing number as one that should not be billed for collect or third number calls.

2. Calling Card Validation

BellSouth is authorized to validate a 14-digit Calling Card number where the first 10 digits are a line number or special billing number assigned by BellSouth, and where the last four digits (PIN) are a security code assigned by BellSouth.

3. Fraud Control

BellSouth will provide seven days per week, 24-hours per day, fraud monitoring on Calling Cards, bill-to-third and collect calls made to numbers in BellSouth's LIDB, provided that such information is included in the LIDB query. BellSouth will establish fraud alert thresholds and will notify Knology of fraud alerts so that Knology may take action it deems appropriate.

III. Responsibilities of the Parties

- BellSouth will administer all data stored in the LIDB, including the data provided by Knology pursuant to this Agreement, in the same manner as BellSouth's data for BellSouth's End User customers. BellSouth shall not be responsible to Knology for any lost revenue which may result from BellSouth's administration of the LIDB pursuant to its established practices and procedures as they exist and as they may be changed by BellSouth in its sole discretion from time to time.
- B. Billing and Collection Customers

BellSouth currently has in effect numerous billing and collection agreements with various interexchange carriers and billing clearing houses and as such these billing and collection customers ("B&C Customers") query BellSouth's LIDB to determine whether to accept various billing options from End Users. Until such time as BellSouth implements in its LIDB and its supporting systems the means to differentiate Knology's data from BellSouth's data, the following shall apply:

- (1) BellSouth will identify Knology end user originated long distance charges and will return those charges to the interexchange carrier as not covered by the existing B&C agreement. Knology is responsible for entering into the appropriate agreement with interexchange carriers for handling of long distance charges by their end users.
- (2) BellSouth shall have no obligation to become involved in any disputes between Knology and B&C Customers. BellSouth will not issue adjustments for charges billed on behalf of any B&C Customer to Knology. It shall be the responsibility of Knology and the B&C Customers to negotiate and arrange for any appropriate adjustments.

C. SPNP ARRANGEMENTS

- 1. BellSouth will include billing number information associated with resold exchange lines or SPNP arrangements in its LIDB. Knology will request any toll billing exceptions via the Local Service Request (LSR) form used to order resold exchange lines, or the SPNP service request form used to order SPNP arrangements.
- 2. Under normal operating conditions, BellSouth shall include the billing number information in its LIDB upon completion of the service order establishing either the resold local exchange service or the SPNP arrangement, provided that BellSouth shall not be held responsible for any delay or failure in performance to the extent such delay or failure is caused by circumstances or conditions beyond BellSouth's reasonable control. BellSouth will store in its LIDB an unlimited volume of the working telephone numbers associated with either the resold local exchange lines or the SPNP arrangements. For resold local exchange lines or for SPNP arrangements, BellSouth will issue line-based calling cards only in the name of Knology. BellSouth will not issue line-based calling cards in the name of Knology's individual End Users. In the event that Knology wants to include calling card numbers assigned by Knology in the BellSouth LIDB, a separate agreement is required.

IV. Fees for Service and Taxes

A. Knology will not be charged a fee for storage services provided by BellSouth to Knology, as described in this LIDB Resale Storage Agreement.

Attachment 1 Page 20 of 20 Exhibit B

B. Sales, use and all other taxes (excluding taxes on BellSouth's income) determined by BellSouth or any taxing authority to be due to any federal, state or local taxing jurisdiction with respect to the provision of the service set forth herein will be paid by Knology in accordance with the tax provisions set forth in the General Terms and Conditions of this Agreement.

RESA	LE DIS	COUNTS AND RATES - Alabama												Attachr	nent: 1	Exhi	bit: C
												Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEG	ORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
												-	-	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							D	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)	•	
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
APPLIC	CABLE D	DISCOUNTS															
		Residence %					16.3										
		Business %					16.3										
		CSAs %					16.3										
OPERA	TIONAL	SUPPORT SYSTEMS (OSS) RATES															
		Electronic LSR				SOMEC		3.50	3.50	3.50	3.50						
		Manual LSR				SOMAN		19.99	19.99	19.99	19.99						

RESA	LE DIS	COUNTS AND RATES - Florida												Attachr	nent: 1	Exhi	bit: C
												Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEG	ORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
												-	-	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							Dee	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
APPLIC	CABLE D	DISCOUNTS															
		Residence %					21.83										
		Business %					16.81										
		CSAs %					16.81										
OPERA	TIONAL	SUPPORT SYSTEMS (OSS) RATES															
		Electronic LSR				SOMEC		3.50	3.50	3.50	3.50						
		Manual LSR				SOMAN		19.99	19.99	19.99	19.99						

RESA	LE DIS	COUNTS AND RATES - Georgia												Attachr	nent: 1	Exhi	bit: C
												Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEG	ORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
												-	-	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							D	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)	•	
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
APPLIC	CABLE D	DISCOUNTS															
		Residence %					20.3										
		Business %					17.3										
		CSAs %					17.3										
OPERA	TIONAL	SUPPORT SYSTEMS (OSS) RATES															
		Electronic LSR				SOMEC		3.50	3.50	3.50	3.50						
		Manual LSR				SOMAN		19.99	19.99	19.99	19.99						

RESA	LE DIS	COUNTS AND RATES - Kentucky												Attachr	nent: 1	Exhi	bit: C
												Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEG	ORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
												-	-	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							Dee	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
APPLIC	CABLE D	DISCOUNTS															
		Residence %					16.79										
		Business %					15.54										
		CSAs %					15.54										
OPERA	TIONAL	. SUPPORT SYSTEMS (OSS) RATES															
		Electronic LSR				SOMEC		3.50	3.50	3.50	3.50						
		Manual LSR				SOMAN		19.99	19.99	19.99	19.99						

RESA	LE DIS	COUNTS AND RATES - South Carolina												Attachi	nent: 1	Exhi	bit: C
												Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEG	ORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
												•		Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							_	Nonrec	urrina	Nonrecurring	Disconnect			OSS	Rates(\$)		1
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
APPLIC	ABLE	DISCOUNTS															
		Residence %					14.8										
		Business %					14.8										
		CSAs %					8.98										
OPERA	TIONAL	SUPPORT SYSTEMS (OSS) RATES															
		Electronic LSR				SOMEC		3.50	3.50	3.50	3.50						
		Manual LSR				SOMAN		19.99	19.99	19.99	19.99						

RESA	LE DIS	COUNTS AND RATES - Tennessee												Attachr	nent: 1	Exhi	bit: C
												Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEG	ORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
												-	-	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							Rec	Nonrecurring		Nonrecurring	g Disconnect			OSS	Rates(\$)		•
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
APPLIC	CABLE D	DISCOUNTS															
		Residence %					16										
		Business %					16										
		CSAs %					16										
OPERA	TIONAL	. SUPPORT SYSTEMS (OSS) RATES															
		Electronic LSR				SOMEC		3.50	3.50	3.50	3.50						
		Manual LSR				SOMAN		19.99	19.99	19.99	19.99						

Attachment 2

Network Elements and Other Services

TABLE OF CONTENTS

1	INTRODUCTION
2	UNBUNDLED LOOPS
3	HIGH FREQUENCY SPECTRUM NETWORK ELEMENT
4	LOCAL SWITCHING
5	UNBUNDLED NETWORK ELEMENT COMBINATIONS 42
6	TRANSPORT, CHANNELIZATION AND DARK FIBER 48
7 SCR	BELLSOUTH SWITCHED ACCESS ("SWA") 8XX TOLL FREE DIALING TEN DIGIT EENING SERVICE
8	LINE INFORMATION DATABASE (LIDB)
9	SIGNALING
10	OPERATOR SERVICES (OPERATOR CALL PROCESSING AND DIRECTORY ASSISTANCE). 62
11	AUTOMATIC LOCATION IDENTIFICATION/DATA MANAGEMENT SYSTEM (ALI/DMS) 68
12	CALLING NAME (CNAM) DATABASE SERVICE
13 ADV	SERVICE CREATION ENVIRONMENT AND SERVICE MANAGEMENT SYSTEM (SCE/SMS) ANCED INTELLIGENT NETWORK (AIN) ACCESS
14	BASIC 911 AND E911 70
15	OPERATIONAL SUPPORT SYSTEMS (OSS)
LID	B Storage Agreement Exhibit A
Rat	es Exhibit B

ACCESS TO NETWORK ELEMENTS AND OTHER SERVICES

1 Introduction

- 1.1 This Attachment sets forth rates, terms and conditions for Network Elements and combinations of Network Elements that BellSouth agrees to offer to Knology in accordance with its obligations under Section 251(c)(3) of the Act. Additionally, this Attachment sets forth the rates, terms and conditions for other services BellSouth makes available to Knology. The rates for each Network Element and combination of Network Elements and other services are set forth in Exhibit B of this Agreement. Additionally, the provision of a particular Network Element or service may require Knology to purchase other Network Elements or services.
- 1.2 For purposes of this Agreement, "Network Element" is defined to mean a facility or equipment Knology used in the provision of a telecommunications service. For purposes of this Agreement, combinations of Network Elements shall be referred to as "Combinations."
- 1.3 BellSouth shall, upon request of Knology, and to the extent technically feasible, provide to Knology access to its Network Elements for the provision of Knology's telecommunications services. If no rate is identified in this Agreement, the rate for the specific service or function will be as set forth in the applicable BellSouth tariff or as negotiated by the Parties upon request by either Party.
- 1.4 Knology may purchase Network Elements and other services from BellSouth for the purpose of combining such network elements in any manner Knology chooses to provide telecommunication services to its intended users, including recreating existing BellSouth services. With the exception of the sub-loop Network Elements which are located outside of the central office, BellSouth shall deliver the Network Elements purchased by Knology to the demarcation point associated with Knology's collocation arrangement.
- 1.5 BellSouth shall comply with the requirements as set forth in the technical references within this Attachment 2.
- 1.6 Knology may not purchase unbundled network elements (UNEs) or convert special access circuits to UNEs if such network elements will be used to provide wireless telecommunications services.
- 1.7 Rates
- 1.7.1 The prices that Knology shall pay to BellSouth for Network Elements and Other Services are set forth in Exhibit B to this Attachment. If Knology purchases a service(s) from a tariff, all terms and conditions and rates as set forth in such tariff shall apply.

- 1.7.2 Rates, terms and conditions for order cancellation charges and Service Date Advancement Charges will apply in accordance with Attachment 6 and are incorporated herein by this reference.
- 1.7.3 If Knology modifies an order (Order Modification Charge (OMC)) after being sent a Firm Order Confirmation (FOC) from BellSouth, any costs incurred by BellSouth to accommodate the modification will be paid by Knology in accordance with FCC No. 1 Tariff, Section 5.
- 1.7.4 A one-month minimum billing period shall apply to all UNE conversions or new installations.

2 Unbundled Loops

- 2.1 General
- 2.1.1 The local loop Network Element ("Loop") is defined as a transmission facility between a distribution frame (or its equivalent) in BellSouth's central office and the loop demarcation point at an end-user customer premises, including inside wire owned by BellSouth. The local loop Network Element includes all features, functions, and capabilities of the transmission facilities, including dark fiber and attached electronics (except those used for the provision of advanced services, such as Digital Subscriber Line Access Multiplexers) and line conditioning.
- 2.1.2 The provisioning of a Loop to Knology's collocation space will require crossoffice cabling and cross-connections within the central office to connect the Loop to a local switch or to other transmission equipment. These cross-connects are separate components that are not considered a part of the Loop, and thus, have a separate charge.
- 2.1.3 To the extent available within BellSouth's network at a particular location, BellSouth will offer Loops capable of supporting telecommunications services. If a requested loop type is not available and cannot be made available through BellSouth's Unbundled Loop Modification process, then Knology can use the Special Construction process to request that BellSouth place facilities in order to meet Knology's loop requirements. Standard Loop intervals shall not apply to the Special Construction process.
- 2.1.4 Where facilities are available, BellSouth will install Loops in compliance with BellSouth's Products and Services Interval Guide available at the website at http://www.interconnection.bellsouth.com. For orders of 15 or more Loops, the installation and any applicable Order Coordination as described below will be handled on a project basis, and the intervals will be set by the BellSouth project manager for that order. When Loops require a Service Inquiry (SI) prior to issuing the order to determine if facilities are available, the interval for the SI process is separate from the installation interval.

- 2.1.5 The Loop shall be provided to Knology in accordance with BellSouth's TR73600 Unbundled Local Loop Technical Specification and applicable industry standard technical references.
- 2.1.6 Knology may utilize the unbundled Loops to provide telecommunications services as long as such services are consistent with industry standards and BellSouth's TR73600.
- 2.1.7 BellSouth will only provision, maintain and repair the Loops to the standards that are consistent with the type of Loop ordered. In those cases where Knology has requested that BellSouth modify a Loop so that it no longer meets the technical parameters of the original Loop type (e.g., voice grade, ISDN, ADSL, etc.), the resulting Loop will be maintained as an unbundled copper Loop (UCL), and Knology shall pay the recurring and non-recurring charges for a UCL. For nonservice specific loops (e.g. UCL, Loops modified by Knology using the Unbundled Loop Modification (ULM) process), BellSouth will only support that the Loop has copper continuity and balanced tip-and-ring.

2.1.8 Loop Testing/Trouble Reporting

- 2.1.8.1 Knology will be responsible for testing and isolating troubles on the Loops. Knology must test and isolate trouble to the BellSouth portion of a designed/nondesigned unbundled loop (e.g., UVL-SL2, UCL-D, UVL-SL1, UCL-ND, etc.) before reporting repair to the UNE Customer Wholesale Interconnection Network Services (CWINS) Center. At the time of the trouble report, Knology will be required to provide the results of the Knology test which indicate a problem on the BellSouth provided loop.
- 2.1.8.2 Once Knology has isolated a trouble to the BellSouth provided Loop, and had issued a trouble report to BellSouth on the Loop, BellSouth will take the actions necessary to repair the Loop if a trouble actually exists. BellSouth will repair these Loops in the same time frames that BellSouth repairs similarly situated Loops to its end users.
- 2.1.8.3 If Knology reports a trouble on a non-designed or designed loop and no trouble actually exists, BellSouth will charge Knology for any dispatching and testing (both inside and outside the CO) required by BellSouth in order to confirm the loop's working status.

2.1.9 Order Coordination and Order Coordination-Time Specific

2.1.9.1 "Order Coordination" (OC) allows BellSouth and Knology to coordinate the installation of the SL2 Loops, Unbundled Digital Loops (UDL) and other Loops where OC may be purchased as an option, to Knology's facilities to limit end user service outage. OC is available when the Loop is provisioned over an existing circuit that is currently providing service to the end user. OC for physical

conversions will be scheduled at BellSouth's discretion during normal working hours on the committed due date. OC shall be provided in accordance with the chart set forth below.

2.1.9.2 "Order Coordination - Time Specific" (OC-TS) allows Knology to order a specific time for OC to take place. BellSouth will make every effort to accommodate Knology's specific conversion time request. However, BellSouth reserves the right to negotiate with Knology a conversion time based on load and appointment control when necessary. This OC-TS is a chargeable option for all Loops except Unbundled Copper Loops (UCL) and Universal Digital Channel (UDC), and is billed in addition to the OC charge. Knology may specify a time between 9:00 a.m. and 4:00 p.m. (location time) Monday through Friday (excluding holidays). If Knology specifies a time outside this window, or selects a time or quantity of Loops that requires BellSouth technicians to work outside normal work hours, overtime charges will apply in addition to the OC and OC-TS charges. Overtime charges will be applied based on the amount of overtime worked and in accordance with the rates established in the Access Services Tariff, Section E13.2, for each state. The OC-TS charges for an order due on the same day at the same location will be applied on a per Local Service Request (LSR) basis.

2.1.10 <u>CLEC to CLEC Conversions for Unbundled Loops</u>

- 2.1.10.1 The CLEC to CLEC conversion process for unbundled Loops may be used by Knology when converting an existing unbundled Loop from another CLEC for the same end user. The Loop type being converted must be included in Knology's Interconnection Agreement before requesting a conversion.
- 2.1.10.2 To utilize the CLEC to CLEC conversion process, the Loop being converted must be the same Loop type with no requested changes to the Loop, must serve the same end user location from the same serving wire center, and must not require an outside dispatch to provision.
- 2.1.10.3 The Loops converted to Knology pursuant to the CLEC to CLEC conversion process shall be provisioned in the same manner and with the same functionality and options as described in this Attachment for the specific Loop type.

	Order Coordination (OC)	Order Coordination – Time Specific (OC-TS)	Test Points	DLR	Charge for Dispatch and Testing if No Trouble Found
SL-1	Chargeable Option	Chargeable Option	Not available	Chargeable Option –	Charged for Dispatch inside and outside
(Non- Designed)				ordered as Engineering Information	Central Office

				Document	
UCL-ND (Non- Designed)	Chargeable Option	Not Available	Not Available	Chargeable Option – ordered as Engineering Information Document	Charged for Dispatch inside and outside Central Office
Unbundled Voice Loops - SL-2 (including 2- and 4-wire UVL) (Designed)	Included	Chargeable Option	Included	Included	Charged for Dispatch outside Central Office
Unbundled Digital Loop (Designed)	Included	Chargeable Option (except on Universal Digital Channel)	Included (where appropriate)	Included	Charged for Dispatch outside Central Office
Unbundled Copper Loop (Designed)	Chargeable in accordance with Section 2	Not available	Included	Included	Charged for Dispatch outside Central Office

For UVL-SL1 and UCLs, Knology must order and will be billed for both OC and OC-TS if requesting OC-TS.

2.2 Unbundled Voice Loops (UVLs)

- 2.2.1 BellSouth shall make available the following UVLs:
- 2.2.1.1 2-wire Analog Voice Grade Loop SL1 (Non-Designed)
- 2.2.1.2 2-wire Analog Voice Grade Loop SL2 (Designed)
- 2.2.1.3 4-wire Analog Voice Grade Loop (Designed)
- 2.2.2 Unbundled Voice Loops (UVL) may be provisioned using any type of facility that will support voice grade services. This may include loaded copper, non-loaded copper, digital loop carrier systems, fiber or a combination of any of these facilities. BellSouth, in the normal course of maintaining, repairing, and configuring its network, may also change the facilities that are used to provide any given voice grade circuit. This change may occur at any time. In these situations, BellSouth will only ensure that the newly provided facility will support voice grade services. BellSouth will not guarantee that Knology will be able to continue to provide any advanced services over the new facility. BellSouth will offer UVL in

two different service levels - Service Level One (SL1) and Service Level Two (SL2).

- 2.2.3 Unbundled Voice Loop SL1 (UVL-SL1) loops are 2-wire loop start circuits, will be non-designed, and will not have remote access test points. OC will be offered as a chargeable option on SLI loops when reuse of existing facilities has been requested by Knology. Knology may also order OC-TS when a specified conversion time is requested. OC-TS is a chargeable option for any coordinated order and is billed in addition to the OC charge. An Engineering Information (EI) document can be ordered as a chargeable option. The EI document provides loop make up information which is similar to the information normally provided in a Design Layout Record. Upon issuance of a non-coordinated order in the service order system, SL1 loops will be activated on the due date in the same manner and time frames that BellSouth normally activates POTS-type loops for its end users.
- 2.2.4 For an additional charge BellSouth will make available Loop Testing so that Knology may request further testing on new UVL-SL1 loops. Rates for Loop Testing are as set forth in Exhibit B of this Attachment.
- 2.2.5 Unbundled Voice Loop SL2 (UVL-SL2) loops may be 2-wire or 4-wire circuits, shall have remote access test points, and will be designed with a Design Layout Record provided to Knology. SL2 circuits can be provisioned with loop start, ground start or reverse battery signaling. OC is provided as a standard feature on SL2 loops. The OC feature will allow Knology to coordinate the installation of the loop with the disconnect of an existing customer's service and/or number portability service. In these cases, BellSouth will perform the order conversion with standard order coordination at its discretion during normal work hours.

2.3 Unbundled Digital Loops

- 2.3.1 BellSouth will offer Unbundled Digital Loops (UDL). UDLs are service specific, will be designed, will be provisioned with test points (where appropriate), and will come standard with OC and a Design Layout Record (DLR). The various UDLs are intended to support a specific digital transmission scheme or service.
- 2.3.2 BellSouth shall make available the following UDLs:
- 2.3.2.1 2-wire Unbundled ISDN Digital Loop
- 2.3.2.2 2-wire Universal Digital Channel (IDSL Compatible)
- 2.3.2.3 2-wire Unbundled ADSL Compatible Loop
- 2.3.2.4 2-wire Unbundled HDSL Compatible Loop
- 2.3.2.5 4-wire Unbundled HDSL Compatible Loop

- 2.3.2.6 4-wire Unbundled DS1 Digital Loop
- 2.3.2.7 4-wire Unbundled Digital Loop/DS0 64 kbps, 56 kbps and below
- 2.3.2.8 DS3 Loop
- 2.3.2.9 STS-1 Loop
- 2.3.2.10 OC-3 Loop
- 2.3.2.11 OC-12 Loop
- 2.3.2.12 OC-48 Loop
- 2.3.3 2-Wire Unbundled ISDN Digital Loops will be provisioned according to industry standards for 2-Wire Basic Rate ISDN services and will come standard with a test point, Order Coordination, and a DLR. Knology will be responsible for providing BellSouth with a Service Profile Identifier (SPID) associated with a particular ISDN-capable loop and end user. With the SPID, BellSouth will be able to adequately test the circuit and ensure that it properly supports ISDN service. BellSouth will not reconfigure its ISDN-capable loop to support IDSL service.
- 2.3.3.1 The Universal Digital Channel (UDC) (also known as IDSL-compatible Loop) is intended to be compatible with IDSL service and has the same physical characteristics and transmission specifications as BellSouth's ISDN-capable loop. These specifications are listed in BellSouth's TR73600.
- 2.3.3.2 The UDC may be provisioned on copper or through a Digital Loop Carrier (DLC) system. When UDC Loops are provisioned using a DLC system, the Loops will be provisioned on time slots that are compatible with data-only services such as IDSL.
- 2.3.4 2-Wire ADSL-Compatible Loop. This is a designed loop that is provisioned according to Revised Resistance Design (RRD) criteria and may be up to 18kft long and may have up to 6kft of bridged tap (inclusive of loop length). The loop is a 2-wire circuit and will come standard with a test point, Order Coordination, and a DLR.
- 2.3.5 2-Wire or 4-Wire HDSL-Compatible Loop. This is a designed loop that is provisioned according to Carrier Serving Area (CSA) criteria and may be up to 12,000 feet long and may have up to 2,500 feet of bridged tap (inclusive of loop length). It may be a 2-wire or 4-wire circuit and will come standard with a test point, Order Coordination, and a DLR.
- 2.3.6 4-Wire Unbundled DS1 Digital Loop. This is a designed 4-wire loop that is provisioned according to industry standards for DS1 or Primary Rate ISDN services and will come standard with a test point, Order Coordination, and a DLR.

A DS1 Loop may be provisioned over a variety of loop transmission technologies including copper, HDSL-based technology or fiber optic transport systems. It will include a 4-Wire DS1 Network Interface at the end-user's location.

- 2.3.7 4-Wire Unbundled Digital/DS0 Loop. These are designed 4-wire loops that may be configured as 64kbps, 56kbps, 19kbps, and other sub-rate speeds associated with digital data services and will come standard with a test point, Order Coordination, and a DLR.
- 2.3.8 DS3 Loop. DS3 Loop is a two-point digital transmission path which provides for simultaneous two-way transmission of serial, bipolar, return-to-zero isochronous digital electrical signals at a transmission rate of 44.736 megabits per second (Mbps) that is dedicated to the use of the ordering CLEC in its provisioning of local exchange and associated exchange access services. It may provide transport for twenty-eight (28) DS1 channels, each of which provides the digital equivalent of twenty-four analog voice grade channels. The interface to unbundled dedicated DS3 transport is a metallic-based electrical interface.
- 2.3.9 STS-1 Loop. STS-1 Loop is a high-capacity digital transmission path with SONET VT1.5 mapping that is dedicated for the use of the ordering customer for the purpose of provisioning local exchange and associated exchange access services. It is a two-point digital transmission path which provides for simultaneous two-way transmission of serial bipolar return-to-zero synchronous digital electrical signals at a transmission rate of 51.84 megabits per second (Mbps). It may provide transport for twenty-eight (28) DS1 channels, each of which provides the digital equivalent of twenty-four analog voice grade channels. The interface to unbundled dedicated STS-1 transport is a metallic-based electrical interface.
- 2.3.10 OC-3 Loop/OC-12 Loop/OC-48 Loop. OC-3/OC-12/OC-48 Loops are optical two-point transmission paths that are dedicated to the use of the ordering CLEC in its provisioning of local exchange and associated exchange access services. The physical interface for all optical transport is optical fiber. This interface standard allows for transport of many different digital signals using a basic building block or base transmission rate of 51.84 megabits per second (Mbps). Higher rates are direct multiples of the base rate. The following rates are applicable: OC-3 155.52 Mbps; OC-12 622.08 Mbps; and OC-48 2488 Mbps.
- 2.3.11 DS3 and above services come with a test point and a DLR. Mileage is airline miles, rounded up and a minimum of one mile applies. BellSouth TR 73501
 LightGate[®]Service Interface and Performance Specifications, Issue D, June 1995 applies to DS3 and above services.

2.4 Unbundled Copper Loops (UCL)

2.4.1 BellSouth shall make available Unbundled Copper Loops (UCLs). The UCL is a copper twisted pair Loop that is unencumbered by any intervening equipment (e.g., filters, load coils, range extenders, digital loop carrier, or repeaters) and is not intended to support any particular telecommunications service. The UCL will be offered in two types – Designed and Non-Designed.

2.4.2 Unbundled Copper Loop – Designed (UCL-D)

- 2.4.2.1 The UCL-D will be provisioned as a dry copper twisted pair loop that is unencumbered by any intervening equipment (e.g., filters, load coils, range extenders, digital loop carrier, or repeaters). The UCL-D will be offered in two versions - Short and Long.
- 2.4.2.2 A short UCL-D (18,000 feet or less) is provisioned according to Resistance Design parameters, may have up to 6,000 feet of bridged tap and will have up to 1300 Ohms of resistance.
- 2.4.2.3 The long UCL-D (beyond 18,000 feet) is provisioned as a dry copper twisted pair longer than 18,000 feet and may have up to 12,000 feet of bridged tap and up to 2800 Ohms of resistance.
- 2.4.2.4 The UCL-D is a designed circuit, is provisioned with a test point, and comes standard with a DLR. OC is a chargeable option for a UCL-D; however, OC is always required on UCLs where a reuse of existing facilities has been requested by Knology.
- 2.4.2.5 These loops are not intended to support any particular services and may be utilized by Knology to provide a wide-range of telecommunications services as long as those services do not adversely affect BellSouth's network. This facility will include a Network Interface Device (NID) at the customer's location for the purpose of connecting the loop to the customer's inside wire.
- 2.4.2.6 BellSouth will make available the following UCL-Ds:
- 2.4.2.6.1 2-Wire UCL-D/short
- 2.4.2.6.2 2-Wire UCL-D/long
- 2.4.2.6.3 4-Wire UCL-D/short
- 2.4.2.6.4 4-Wire UCL-D/long

2.4.3 <u>Unbundled Copper Loop – Non-Designed (UCL-ND)</u>

2.4.3.1 The UCL–ND is provisioned as a dedicated 2-wire metallic transmission facility from BellSouth's Main Distribution Frame to a customer's premises (including the NID). The UCL-ND will be a "dry copper" facility in that it will not have any intervening equipment such as load coils, repeaters, or digital access main lines ("DAMLs"), and may have up to 6,000 feet of bridged tap between the end user's premises and the serving wire center. The UCL-ND typically will be 1300 Ohms resistance and in most cases will not exceed 18,000 feet in length, although the UCL-ND will not have a specific length limitation. For loops less than 18,000 feet and with less than 1300 Ohms resistance, the loop will provide a voice grade transmission channel suitable for loop start signaling and the transport of analog voice grade signals. The UCL-ND will not be designed and will not be provisioned with either a DLR or a test point.

- 2.4.3.2 The UCL-ND facilities may be mechanically assigned using BellSouth's assignment systems. Therefore, the Loop Make Up process is not required to order and provision the UCL-ND. However, Knology can request Loop Make Up for which additional charges would apply.
- 2.4.3.3 For an additional charge, BellSouth also will make available Loop Testing so that Knology may request further testing on the UCL-ND. Rates for Loop Testing are as set forth in Exhibit B of this Attachment.
- 2.4.3.4 UCL-ND loops are not intended to support any particular service and may be utilized by Knology to provide a wide-range of telecommunications services as long as those services do not adversely affect BellSouth's network. The UCL-ND will include a Network Interface Device (NID) at the customer's location for the purpose of connecting the loop to the customer's inside wire.
- 2.4.3.5 Order Coordination (OC) will be provided as a chargeable option and may be utilized when the UCL-ND provisioning is associated with the reuse of BellSouth facilities. Order Coordination -Time Specific (OC-TS) does not apply to this product.
- 2.4.3.6 Knology may use BellSouth's Unbundled Loop Modification (ULM) offering to remove bridge tap and/or load coils from any loop within the BellSouth network. Therefore, some loops that would not qualify as UCL-ND could be transformed into loops that do qualify, using the ULM process.

2.5 Unbundled Loop Modifications (Line Conditioning)

- 2.5.1 Line Conditioning is defined as the removal from the Loop of any devices that may diminish the capability of the Loop to deliver high-speed switched wireline telecommunications capability, including xDSL service. Such devices include, but are not limited to, load coils, bridged taps, low pass filters, and range extenders.
- 2.5.2 BellSouth shall condition Loops, as requested by Knology, whether or not BellSouth offers advanced services to the End User on that Loop.
- 2.5.3 In some instances, Knology will require access to a copper twisted pair loop unfettered by any intervening equipment (e.g., filters, load coils, range extenders,

etc.), so that Knology can use the loop for a variety of services by attaching appropriate terminal equipment at the ends. Knology will determine the type of service that will be provided over the loop. BellSouth's Unbundled Loop Modifications (ULM) process will be used to determine the costs and feasibility of conditioning the loops as requested. Rates for ULM are as set forth in Exhibit B of this Attachment.

- 2.5.4 In those cases where Knology has requested that BellSouth modify a Loop so that it no longer meets the technical parameters of the original Loop type (e.g., voice grade, ISDN, ADSL, etc.), the resulting modified Loop will be ordered and maintained as a UCL.
- 2.5.5 The Unbundled Loop Modifications (ULM) offering provides the following elements: 1) removal of devices on 2-wire or 4-wire Loops equal to or less than 18,000 feet; 2) removal of devices on 2-wire or 4-wire Loops longer than 18,000 feet; and 3) removal of bridged-taps on loops of any length.
- 2.5.6 Knology shall request Loop make up information pursuant to this Attachment prior to submitting a service inquiry and/or a LSR for the Loop type that Knology desires BellSouth to condition.
- 2.5.7 When requesting ULM for a loop that BellSouth has previously provisioned for Knology, Knology will submit a service inquiry to BellSouth. If a spare loop facility that meets the loop modification specifications requested by Knology is available at the location for which the ULM was requested, Knology will have the option to change the loop facility to the qualifying spare facility rather than to provide ULM. In the event that BellSouth changes the loop facility in lieu of providing ULM, Knology will not be charged for ULM but will only be charged the service order charges for submitting an order.

2.6 Loop Provisioning Involving Integrated Digital Loop Carriers

2.6.1 Where Knology has requested an Unbundled Loop and BellSouth uses Integrated Digital Loop Carrier (IDLC) systems to provide the local service to the end user and BellSouth has a suitable alternate facility available, BellSouth will make such alternative facilities available to Knology. If a suitable alternative facility is not available, then to the extent it is technically feasible, BellSouth will make alternative arrangements available to Knology (e.g. hairpinning).

2.6.2 BellSouth will select one of the following arrangements:

- 1. Roll the circuit(s) from the IDLC to any spare copper that exists to the customer premises.
- 2. Roll the circuit(s) from the IDLC to an existing DLC that is not integrated.
- 3. If capacity exists, provide "side-door" porting through the switch.
- 4. If capacity exists, provide "DACS-door" porting (if the IDLC routes through a DACS prior to integration into the switch).

- 2.6.3 Arrangements 3 and 4 above require the use of a designed circuit. Therefore, nondesigned loops such as the SL1 voice grade and UCL-ND may not be ordered in these cases.
- 2.6.4 If no alternate facility is available, BellSouth will utilize its Special Construction (SC) process to determine the additional costs required to provision the loop facilities. Knology will then have the option of paying the one-time SC rates to place the loop.

2.7 Network Interface Device (NID)

- 2.7.1 The NID is defined as any means of interconnection of end-user customer premises wiring to BellSouth's distribution plant, such as a cross-connect device used for that purpose. The NID is a single-line termination device or that portion of a multiple-line termination device required to terminate a single line or circuit at the premises. The NID features two independent chambers or divisions that separate the service provider's network from the end user's customer-premises wiring. Each chamber or division contains the appropriate connection points or posts to which the service provider and the end user each make their connections. The NID provides a protective ground connection and is capable of terminating cables such as twisted pair cable.
- 2.7.2 BellSouth shall permit Knology to connect Knology's Loop facilities to the enduser's customer-premises wiring through the BellSouth NID or at any other technically feasible point.

2.7.3 Access to NID

- 2.7.3.1 Knology may access the end user's customer-premises wiring by any of the following means and Knology shall not disturb the existing form of electrical protection and shall maintain the physical integrity of the NID:
- 2.7.3.1.1 1) BellSouth shall allow Knology to connect its loops directly to BellSouth's multi-line residential NID enclosures that have additional space and are not used by BellSouth or any other telecommunications carriers to provide service to the premises.
- 2.7.3.1.22) Where an adequate length of the end user's customer premises wiring is present and environmental conditions permit, either Party may remove the customer premises wiring from the other Party's NID and connect such wiring to that Party's own NID;
- 2.7.3.1.3 3) Enter the subscriber access chamber or dual chamber NID enclosures for the purpose of extending a connect divisioned or spliced jumper wire from the customer premises wiring through a suitable "punch-out" hole of such NID enclosures; or

- 2.7.3.1.4 4) Request BellSouth to make other rearrangements to the end user customer premises wiring terminations or terminal enclosure on a time and materials cost basis.
- 2.7.3.2 In no case shall either Party remove or disconnect the other Party's loop facilities from either Party's NIDs, enclosures, or protectors unless the applicable Commission has expressly permitted the same and the disconnecting Party provides prior notice to the other Party. In such cases, it shall be the responsibility of the Party disconnecting loop facilities to leave undisturbed the existing form of electrical protection and to maintain the physical integrity of the NID. It will be Knology's responsibility to ensure there is no safety hazard and will hold BellSouth harmless for any liability associated with the removal of the BellSouth loop from the BellSouth NID. Furthermore, it shall be the responsibility of the disconnecting Party, once the other Party's loop has been disconnected from the NID, to reconnect the disconnected loop to a nationally recognized testing laboratory listed station protector, which has been grounded as per Article 800 of the National Electrical Code. If no spare station protector exists in the NID, the disconnected loop must be appropriately cleared, capped and stored.
- 2.7.3.3 In no case shall either Party remove or disconnect ground wires from BellSouth's NIDs, enclosures, or protectors.
- 2.7.3.4 In no case shall either Party remove or disconnect NID modules, protectors, or terminals from BellSouth's NID enclosures.
- 2.7.3.5 Due to the wide variety of NID enclosures and outside plant environments, BellSouth will work with Knology to develop specific procedures to establish the most effective means of implementing this section if the procedures set forth herein do not apply to the NID in question.
- 2.7.4 Technical Requirements
- 2.7.4.1 The NID shall provide an accessible point of interconnection and shall maintain a connection to ground.
- 2.7.4.2 If an existing NID is accessed, it shall be capable of transferring electrical analog or digital signals between the end user's customer premises and the Distribution Media and/or cross connect to Knology's NID.
- 2.7.4.3 Existing BellSouth NIDs will be provided in "as is" condition. Knology may request BellSouth to do additional work to the NID on a time and material basis. When Knology deploys its own local loops with respect to multiple-line termination devices, Knology shall specify the quantity of NIDs connections that it requires within such device.

2.8 <u>Sub-loop Elements</u>

2.8.1 Where facilities permit, BellSouth shall offer access to its Unbundled Sub-Loop (USL) and Unbundled Sub-loop Concentration (USLC) System.

2.8.2 Unbundled Sub-Loop Distribution

2.8.2.1 The unbundled sub-loop distribution facility is a dedicated transmission facility that BellSouth provides from an end user's point of demarcation to a BellSouth crossconnect device. The BellSouth cross-connect device may be located within a remote terminal (RT) or a stand-alone cross-box in the field or in the equipment room of a building. The unbundled sub-loop distribution media is a copper twisted pair that can be provisioned as a 2-Wire or 4-Wire facility. BellSouth will make the following available sub-loop distribution offerings where facilities permit:

> Unbundled Sub-Loop Distribution – Voice Grade Unbundled Copper Sub-Loop Unbundled Sub-Loop Distribution – Intrabuilding Network Cable (aka riser cable)

- 2.8.2.2 Unbundled Sub-Loop Distribution Voice Grade (USLD-VG) is a sub-loop facility from the cross-box in the field up to and including the point of demarcation at the end user's premises and may have load coils.
- 2.8.2.3 Unbundled Copper Sub-Loop (UCSL) is a copper facility of any length provided from the cross-box in the field up to and including the end-user's point of demarcation. If available, this facility will not have any intervening equipment such as load coils between the end-user and the cross-box.
- 2.8.2.4 If Knology requests a UCSL and it is not available, Knology may request the Sub-Loop facility be modified pursuant to the ULM process request to remove load coils and/or bridged taps. If load coils and/or bridged taps are removed, the facility will be classified as a UCSL.
- 2.8.2.5 Unbundled Sub-Loop Distribution Intrabuilding Network Cable (USLD-INC) is the distribution facility inside a building or between buildings on the same continuous property that is not separated by a public street or road. USLD-INC includes the facility from the cross-connect device in the building equipment room up to and including the point of demarcation at the end user's premises.
- 2.8.2.6 BellSouth will install a cross connect panel in the building equipment room for the purpose of accessing USLD-INC pairs from a building equipment room. The cross-connect panel will function as a single point of interconnection (SPOI) for USLD-INC and will be accessible by multiple carriers as space permits. BellSouth will place cross-connect blocks in 25-pair increments for Knology's use on this cross-connect panel. Knology will be responsible for connecting its facilities to the 25-pair cross-connect block(s).

- 2.8.2.7 Unbundled Sub-Loop distribution facilities shall support functions associated with provisioning, maintenance and testing of the Unbundled Sub-Loop. For access to Voice Grade USLD and UCSL, Knology shall install a cable to the BellSouth cross-box pursuant to the terms and conditions for physical collocation for remote sites set forth in this Agreement. This cable would be connected by a BellSouth technician within the BellSouth cross-box during the set-up process. Knology's cable pairs can then be connected to BellSouth's USL within the BellSouth cross-box by the BellSouth technician.
- 2.8.2.8 Through the Service Inquiry (SI) process, BellSouth will determine whether access to Unbundled Sub-Loops at the location requested by Knology is technically feasible and whether sufficient capacity exists in the cross-box. If existing capacity is sufficient to meet Knology's request, then BellSouth will perform the site set-up as described in the CLEC Information Package, located at the Website address: http://www.interconnection.bellsouth.com/products/html/unes.html. If any work must be done to modify existing BellSouth facilities or add new facilities (other than adding the cross-connect panel in a building equipment room to accommodate Knology's request for Unbundled Sub-Loops, Knology may request BellSouth's Special Construction (SC) process to determine additional costs required to provision the Unbundled Sub-Loops. Knology will have the option to proceed under the SC process to modify the BellSouth facilities.
- 2.8.2.9 The site set-up must be completed before Knology can order sub-loop pairs. For the site set-up in a BellSouth cross-connect box in the field, BellSouth will perform the necessary work to splice Knology's cable into the cross-connect box. For the site set-up inside a building equipment room, BellSouth will perform the necessary work to install the cross-connect panel and the connecting block(s) that will be used to provide access to the requested USLs.
- 2.8.2.10 Once the site set-up is complete, Knology will request sub-loop pairs through submission of a Local Service Request (LSR) form to the Local Carrier Service Center (LCSC). Order Coordination is required with USL pair provisioning when Knology requests reuse of an existing facility and is in addition to the USL pair rate. For expedite requests by Knology for sub-loop pairs, expedite charges will apply for intervals less than 5 days.
- 2.8.2.11 Unbundled Sub-Loops will be provided in accordance with technical reference TR73600.

2.8.3 Unbundled Network Terminating Wire (UNTW)

2.8.3.1 Unbundled Network Terminating Wire (UNTW) is unshielded twisted copper wiring that is used to extend circuits from an intra-building network cable terminal or from a building entrance terminal to an individual customer's point of demarcation. It is the final portion of the Loop that in multi-subscriber
configurations represents the point at which the network branches out to serve individual subscribers.

2.8.3.2 This element will be provided in Multi-Dwelling Units (MDUs) and/or Multi-Tenants Units (MTUs) where either Party owns wiring all the way to the end-users premises. Neither Party will provide this element in locations where the property owner provides its own wiring to the end-user's premises, where a third party owns the wiring to the end-user's premises or where the property owner will not allow the other Party to place its facilities to the end user.

2.8.3.3 Requirements

- 2.8.3.3.1 On a multi-unit premises, upon request of the other Party ("Requesting Party"), the Party owning the network terminating wire ("Provisioning Party") will provide access to UNTW pairs on an Access Terminal that is suitable for use by multiple carriers at each Garden Terminal or Wiring Closet.
- 2.8.3.3.2 The Provisioning Party shall not be required to install new or additional NTW beyond existing NTW to provision the services of the Requesting Party.
- 2.8.3.3.3 In existing Multi-Dwelling Units (MDUs) and/or Multi-Tenant Units (MTUs) in which BellSouth does not own or control wiring (INC/NTW) to the end users premises, Knology will install UNTW Access Terminals for BellSouth at no additional charge.
- 2.8.3.3.4 In situations in which BellSouth activates a UNTW pair, BellSouth will compensate Knology for each pair activated commensurate to the price specified in Knology's Agreement.
- 2.8.3.3.5 Upon receipt of the UNTW Service Inquiry (SI) requesting access to the Provisioning Party's UNTW pairs at a multi-unit premises, representatives of both Parties will participate in a meeting at the site of the requested access. The purpose of the site visit will include discussion of the procedures for installation and location of the Access Terminals. By request of the Requesting Party, an Access Terminal will be installed either adjacent to each Provisioning Party's Garden Terminal or inside each Wiring Closet. Requesting Party will deliver and connect its central office facilities to the UNTW pairs within the Access Terminal. Requesting Party may access any available pair on an Access Terminal. A pair is available when a pair is not being utilized to provide service or where the end user has requested a change in its local service provider to the Requesting Party. Prior to connecting Requesting Party's service on a pair previously used by Provisioning Party, Requesting Party is responsible for ensuring the end-user is no longer using Provisioning Party's service or another CLEC's service before accessing UNTW pairs.

- 2.8.3.3.6 Access Terminal installation intervals will be established on an individual case basis.
- 2.8.3.3.7 Requesting Party is responsible for obtaining the property owner's permission for Provisioning Party to install an Access Terminal(s) on behalf of the Requesting Party. The submission of the SI by the Requesting Party will serve as certification by the Requesting Party that such permission has been obtained. If the property owner objects to Access Terminal installations that are in progress or subsequent to completion and demands removal of Access Terminals, Requesting Party will be responsible for costs associated with removing Access Terminals and restoring property to its original state prior to Access Terminals being installed.
- 2.8.3.3.8 The Requesting Party shall indemnify and hold harmless the Provisioning Party against any claims of any kind that may arise out of the Requesting Party's failure to obtain the property owner's permission. Requesting Party will be billed for non-recurring and recurring charges for accessing UNTW pairs at the time the Requesting Party activates the pair(s). The Requesting Party will notify the Provisioning Party each time it activates UNTW pairs using the LSR form.
- 2.8.3.3.9 Requesting Party will isolate and report troubles in the manner specified by the Provisioning Party. Requesting Party must tag the UNTW pair that requires repair. If Provisioning Party dispatches a technician on a reported trouble call and no UNTW trouble is found, Provisioning Party will charge Requesting Party for time spent on the dispatch and testing the UNTW pair(s).
- 2.8.3.3.10 If Requesting Party initiates the Access Terminal installation and the Requesting Party has not activated at least one pair on the Access Terminal installed pursuant to Requesting Party's request for an Access Terminal within 6 months of installation of the Access Terminal, Provisioning Party will bill Requesting Party a non-recurring charge equal to the actual cost of provisioning the Access Terminal.
- 2.8.3.3.11 If Provisioning Party determines that Requesting Party is using the UNTW pairs without reporting the activation of the pairs, the following charges shall apply:
- 2.8.3.3.11.1 If Requesting Party issued a LSR to disconnect an end-user from Provisioning Party in order to use a UNTW pair, Requesting Party will be billed for the use of the pair back to the disconnect order date.
- 2.8.3.3.11.2 If Requesting Party activated a UNTW pair on which Provisioning Party was not previously providing service, Requesting Party will be billed for the use of that pair back to the date the end-user began receiving service using that pair. Upon request, Requesting Party will provide copies of its billing record to substantiate such date. If Requesting Party fails to provide such records, then Provisioning Party will bill the Requesting Party back to the date of the Access Terminal installation.

2.8.4 Unbundled Sub-Loop Feeder

- 2.8.4.1 Unbundled Sub-Loop Feeder (USLF) provides connectivity between BellSouth's central office and cross-box (or other access point) that serves an end user location.
- 2.8.4.2 USLF utilized for voice traffic can be configured as 2-wire voice (USLF-2W/V) or 4-wire voice (USLF-4W/V).
- 2.8.4.3 USLF utilized for digital traffic can be configured as 2-wire ISDN (USLF-2W/I); 2-wire Copper (USLF-2W/C); 4-wire Copper (USLF-4W/C); 4-wire DS0 level loop (USLF-4W/D0); or 4-wire DS1 and ISDN (USLF-4W/DI).
- 2.8.4.4 USLF will provide access to both the equipment and the features in the BellSouth central office and BellSouth cross box necessary to provide a 2-wire or 4-wire communications pathway from the BellSouth central office to the BellSouth crossbox. This element will allow for the connection of Knology's loop distribution elements onto BellSouth's feeder system.
- 2.8.4.5 Requirements
- 2.8.4.5.1 Knology will extend a compatible cable to BellSouth's cross-box. BellSouth will connect the cable to a cross-connect panel inside the BellSouth cross-box to the requested level of feeder element. In those cases in which there is no room in the BellSouth cross-box to accommodate the additional cross-connect panels mentioned above, Knology may request, through the BellSouth Special Construction process, a determination of costs to provide the sub-loop feeder element to Knology. Knology will then have the option of paying the special construction charges or canceling the order.
- 2.8.4.5.2 USLF will be a designed circuit and BellSouth will provide a Design Layout Record (DLR) for this element.
- 2.8.4.5.3 BellSouth will provide USLF elements in accordance with applicable industry standards for these types of facilities. Where industry standards do not exist, BellSouth's TR73600 will be used to determine performance parameters.
- 2.8.4.6 Unbundled Sub-Loop Feeder (USLF DS3 and above)
- 2.8.4.6.1 USLF DS3 and above provides connectivity between a BellSouth Serving Wire Center (SWC) and the Remote Terminal (RT) associated with the SWC that serves an end user location.
- 2.8.4.6.2 The sub-loop feeder is intended to be utilized for voice traffic and digital traffic. It can be configured at DS3, STS-1, OC-3, OC-12, or OC-48 transmission capacities.

- 2.8.4.6.3 The OC-48 Sub-Loop Feeder will consist of four (4) OC12 interfaces.
- 2.8.4.6.4 Both 2-fiber and 4-fiber-protect applications will be supported for OC-3 level and higher.
- 2.8.4.7 Requirements
- 2.8.4.7.1 Access in the SWC and RT will be via a Collocation cross-connect.
- 2.8.4.7.2 USLF DS3 and above will be a designed circuit. BellSouth will provide a Design Layout Record (DLR) for this network element.
- 2.8.4.7.3 Rates. Rates for these services are as set forth in Exhibit B of this Attachment. Mileage is based on airline miles.
- 2.8.4.7.4 BellSouth will provide USLF DS3 and above elements in accordance with applicable industry standards.

2.8.5 Unbundled Loop Concentration (ULC)

- 2.8.5.1 BellSouth will provide to Knology Unbundled Loop Concentration (ULC). Loop concentration systems in the central office concentrate the signals transmitted over local loops onto a digital loop carrier system. The concentration device is placed inside a BellSouth central office. BellSouth will offer ULC with a TR008 interface or a TR303 interface.
- 2.8.5.2 ULC will be offered in two system options. System A will allow up to 96 BellSouth loops to be concentrated onto two or more DS1s. The high-speed connection from the concentrator will be at the electrical DS1 level and will connect to Knology at Knology's collocation site. System B will allow up to 192 BellSouth loops to be concentrated onto 4 or more DS1s. System A may be upgraded to a System B. A minimum of two DS1s is required for each system (i.e., System A requires two DS1s and System B would require an additional two DS1s or four in total). All DS1 interfaces will terminate to Knology's collocation space. ULC service is offered with concentration (2 DS1s for 96 channels) or without concentration (4 DS1s for 96 channels) and with or without protection. A Loop Interface element will be required for each loop that is terminated onto the ULC system.

2.8.6 Unbundled Sub-Loop Concentration (USLC)

- 2.8.6.1 Where facilities permit, Knology may concentrate its sub-loops onto multiple DS1s back to the BellSouth Central Office.
- 2.8.6.2 USLC, using the Lucent Series 5 equipment, will be offered in two system options. System A will allow up to 96 of Knology's sub-loops to be concentrated onto two or more DS1s. System B will allow an additional 96 of Knology's sub-loops to be

concentrated onto two or more additional DS1s. One System A may be supplemented with one System B and they both must be physically located in a single Series 5 dual channel bank. A minimum of two DS1s is required for each system (i.e., System A requires two DS1s and System B would require an additional two DS1s or four in total). The DS1 level facility that connects the Remote Terminal site with the serving wire center is known as a Feeder Interface. All DS1 Feeder Interfaces will terminate to Knology's demarcation point associated with Knology's collocation space within the SWC that serves the remote terminal (RT). USLC service is offered with or without concentration and with or without a protection DS1.

2.8.6.3 Knology is required to deliver its sub-loops to its own cross-box, RT, or other similar device and deliver a single cable to the BellSouth RT. This cable shall be connected by a BellSouth technician to a cross-connect panel within the BellSouth RT/cross-box and shall allow Knology's sub-loops to be placed on the USLC and transported to Knology's collocation space at a DS1 level.

2.8.7 Dark Fiber Loop

- 2.8.7.1 Dark Fiber Loop is an unused optical transmission facility, without attached signal regeneration, multiplexing, aggregation or other electronics, from an end user's premises connected via a cross connect to the demarcation point associated with Knology's collocation space in the end user's serving wire center. Dark Fiber Loops may be strands of optical fiber existing in aerial or underground structure. BellSouth will not provide line terminating elements, regeneration or other electronics necessary for Knology to utilize Dark Fiber Loops.
- 2.8.7.2 Requirements
- 2.8.7.2.1 BellSouth shall make available Dark Fiber Loop where it exists in BellSouth's network and where, as a result of future building or deployment, it becomes available. Dark Fiber Loop will not be deemed available if: (1) it is used by BellSouth for maintenance and repair purposes; (2) it is designated for use pursuant to a firm order placed by another customer; (3) it is restricted for use by all carriers, including BellSouth, because of transmission problems or because it is scheduled for removal due to documented changes to roads and infrastructure; or (4) BellSouth has plans to use the fiber within a two-year planning period. BellSouth is not required to place the fiber for Dark Fiber Loop if none is available.
- 2.8.7.2.2 Knology is solely responsible for testing the quality of the Dark Fiber to determine its usability and performance specifications.
- 2.8.7.2.3 BellSouth shall use its commercially reasonable efforts to provide to Knology information regarding the location, availability and performance of Dark Fiber

Loop within ten (10) business days after receiving a Service Inquiry ("SI") from Knology.

2.8.7.2.4 If the requested Dark Fiber Loop is available, BellSouth shall use commercially reasonable efforts to provision the Dark Fiber Loop to Knology within twenty (20) business days after Knology submits a valid, error free LSR. Provisioning includes identification of appropriate connection points (e.g., Light Guide Interconnection (LGX)) to enable Knology to connect Knology provided transmission media (e.g., optical fiber) or equipment to the Dark Fiber Loop.

2.9 Loop Makeup (LMU)

2.9.1 Description of Service

- 2.9.1.1 BellSouth shall make available to Knology LMU information so that Knology can make an independent judgment about whether the Loop is capable of supporting the advanced services equipment Knology intends to install and the services Knology wishes to provide. This section addresses LMU as a preordering transaction, distinct from Knology ordering any other service(s). Loop Makeup Service Inquiries (LMUSI) for preordering loop makeup are likewise unique from other preordering functions with associated service inquiries (SI) as described in this Agreement.
- 2.9.1.2 BellSouth will provide Knology LMU information consisting of the composition of the loop material (copper/fiber); the existence, location and type of equipment on the Loop, including but not limited to digital loop carrier or other remote concentration devices, feeder/distribution interfaces, bridged taps, load coils, pair-gain devices; the loop length; the wire gauge and electrical parameters.
- 2.9.1.3 BellSouth's LMU information is provided to Knology as it exists either in BellSouth's databases or in its hard copy facility records. BellSouth does not guarantee accuracy or reliability of the LMU information provided.
- 2.9.1.4 BellSouth's provisioning of LMU information to the requesting CLEC on facilities is contingent upon either BellSouth or the requesting CLEC owning the loop(s) that serve the service location for which LMU information has been requested by the CLEC. The requesting CLEC is not authorized to receive LMU information on a facility owned by another CLEC unless BellSouth receives a Letter of Authorization (LOA) from the voice CLEC (owner) or its authorized agent on the LMUSI (Loop Makeup Service Inquiry) submitted by the requesting CLEC.
- 2.9.1.5 Knology may choose to use equipment that it deems will enable it to provide a certain type and level of service over a particular BellSouth Loop as long as that equipment does not disrupt other services on the BellSouth network. The determination shall be made solely by Knology and BellSouth shall not be liable in any way for the performance of the advanced data services provisioned over said

Loop. The specific Loop type (ADSL, HDSL, or otherwise) ordered on the LSR must match the LMU of the loop reserved taking into consideration any requisite line conditioning. The LMU data is provided for informational purposes only and does not guarantee Knology's ability to provide advanced data services over the ordered loop type. Further, if Knology orders loops that do not require a specific facility medium (i.e. copper only) or loops that are not intended to support advanced services (such as UV-SL1, UV-SL2, or ISDN compatible loops) and that are not inventoried as advanced services loops, the LMU information for such loops is subject to change at any time due to modifications and/or upgrades to BellSouth's network. Knology is fully responsible for any of its service configurations that may differ from BellSouth's technical standard for the loop type ordered.

2.9.2 Submitting Loop Makeup Service Inquiries

- 2.9.2.1 Knology may obtain LMU information by submitting a LMU Service Inquiry (LMUSI) mechanically or manually. Mechanized LMUSIs should be submitted through BellSouth's Operational Support Systems interfaces. After obtaining the Loop information from the mechanized LMUSI process, if Knology needs further loop information in order to determine loop service capability, Knology may initiate a separate Manual Service Inquiry for a separate nonrecurring charge as set forth in Exhibit B of this Attachment.
- 2.9.2.2 Manual LMUSIs shall be submitted by electronic mail to BellSouth's Complex Resale Support Group (CRSG) utilizing the Preordering Loop Makeup Service Inquiry form. The service interval for the return of a Loop Makeup Manual Service Inquiry is three business days. Manual LMUSIs are not subject to expedite requests. This service interval is distinct from the interval applied to the subsequent service order.

2.9.3 Loop Reservations

- 2.9.3.1 For a Mechanized LMUSI, Knology may reserve up to ten Loop facilities. For a Manual LMUSI, Knology may reserve up to three Loop facilities.
- 2.9.3.2 Knology may reserve facilities for up to four (4) business days for each facility requested on a LMUSI from the time the LMU information is returned to Knology. During and prior to Knology placing an LSR, the reserved facilities are rendered unavailable to other customers, including BellSouth. If Knology does not submit an LSR for a UNE service on a reserved facility within the four-day reservation timeframe, the reservation of that spare facility will become invalid and the facility will be released.
- 2.9.3.3 Charges for preordering LMUSI are separate from any charges associated with ordering other services from BellSouth.

2.9.4 Ordering of Other UNE Services

- 2.9.4.1 All LSRs issued for reserved facilities shall reference the facility reservation number as provided by BellSouth. Knology will not be billed any additional LMU charges for the loop ordered on such LSR. If, however, Knology does not reserve facilities upon an initial LMUSI, Knology's placement of an order for an advanced data service type facility will incur the appropriate billing charges to include service inquiry and reservation per Exhibit B of this Attachment.
- 2.9.4.2 Where Knology has reserved multiple Loop facilities on a single reservation, Knology may not specify which facility shall be provisioned when submitting the LSR. For those occasions, BellSouth will assign to Knology, subject to availability, a facility that meets the BellSouth technical standards of the BellSouth type Loop as ordered by Knology. If the ordered Loop type is not available, Knology may utilize the Unbundled Loop Modification process or the Special Construction process, as applicable, to obtain the Loop type ordered.

3 High Frequency Spectrum Network Element

- 3.1 General
- 3.1.1 BellSouth shall provide Knology access to the high frequency spectrum of the local loop as an unbundled network element only where BellSouth is the voice service provider to the end user at the rates set forth in this Attachment.
- 3.1.2 The High Frequency Spectrum is defined as the frequency range above the voiceband on a copper loop facility carrying analog circuit-switched voiceband transmissions. Access to the High Frequency Spectrum is intended to allow Knology the ability to provide Digital Subscriber Line ("xDSL") data services to the end user for which BellSouth provides voice services. The High Frequency Spectrum shall be available for any version of xDSL complying with Spectrum Management Class 5 of ANSI T1.417, American National Standard for Telecommunications, Spectrum Management for Loop Transmission Systems. BellSouth will continue to have access to the low frequency portion of the loop spectrum (from 300 Hertz to at least 3000 Hertz, and potentially up to 3400 Hertz, depending on equipment and facilities) for the purposes of providing voice service. Knology shall only use xDSL technology that is within the PSD mask for Spectrum Management Class 5 as found in the above-mentioned document.
- 3.1.3 Access to the High Frequency Spectrum requires an unloaded, 2-wire copper Loop. An unloaded Loop is a copper Loop with no load coils, low-pass filters, range extenders, DAMLs, or similar devices and minimal bridged taps consistent with ANSI T1.413 and T1.601.
- 3.1.4 BellSouth will provide Loop Modification to Knology on an existing Loop in accordance with procedures developed in the Line Sharing Collaborative. High

Frequency Spectrum (Central Office Based) Unbundled Loop Modification is a separate distinct service from Unbundled Loop Modification set forth in Section 2.5 of this Attachment. Procedures for High Frequency Spectrum (Central Office Based) Unbundled Loop Modification were developed in the Line Sharing Collaborative and may be found posted to the web at http://www.interconnection.bellsouth.com/html/unes.html. Nonrecurring rates for this UNE offering may be found in Exhibit B of this Attachment. BellSouth is not required to modify a Loop for access to the High Frequency spectrum if modification of that Loop significantly degrades BellSouth's voice service. If Knology requests that BellSouth modify a Loop longer than 18,000 ft. and such modification significantly degrades the voice services on the Loop, Knology shall pay for the Loop to be restored to its original state.

- 3.1.5 The High Frequency Spectrum shall only be available on Loops on which BellSouth is also providing, and continues to provide, analog voice service directly to the end user. In the event the end-user terminates its BellSouth provided voice service for any reason, or in the event BellSouth disconnects the end user's voice service pursuant to its tariffs or applicable law, and Knology desires to continue providing xDSL service on such Loop, Knology shall be required to purchase a full stand-alone Loop unbundled network element. To the extent commercially practicable, BellSouth shall give Knology notice in a reasonable time prior to disconnect, which notice shall give Knology an adequate opportunity to notify BellSouth of its intent to purchase such Loop. In those cases in which BellSouth no longer provides voice service to the end user and Knology purchases the full stand-alone loop, Knology may elect the type of loop it will purchase. Knology will pay the appropriate recurring and non-recurring rates for such Loop as set forth in Exhibit B to this Attachment. In the event Knology purchases a voice grade Loop, Knology acknowledges that such Loop may not remain xDSL compatible.
- 3.1.6 Only one competitive local exchange carrier (CLEC) shall be permitted access to the High Frequency Spectrum of any particular loop.

3.2 **Provisioning of High Frequency Spectrum and Splitter Space**

- 3.2.1 BellSouth will provide Knology with access to the High Frequency Spectrum as follows:
- 3.2.1.1 To order High Frequency Spectrum on a particular Loop, Knology must have a Digital Subscriber Line Access Multiplexer (DSLAM) collocated in the central office that serves the end-user of such Loop.
- 3.2.1.2 Knology may provide its own splitters or may order splitters in a central office once it has installed its DSLAM in that central office. BellSouth will install splitters within thirty-six (36) calendar days of Knology's submission of an error

free Line Splitter Ordering Document ("LSOD") to the BellSouth Complex Resale Support Group.

- 3.2.1.3 Once a splitter is installed on behalf of Knology in a central office in which Knology is located, Knology shall be entitled to order the High Frequency Spectrum on lines served out of that central office. BellSouth will bill and Knology shall pay the electronic or manual ordering charges as applicable when Knology orders High Frequency Spectrum for end-user service.
- 3.2.1.4 BellSouth shall test the data portion of the loop to ensure the continuity of the wiring for Knology's data.

3.3 BellSouth Provided Splitter

- 3.3.1 BellSouth will select, purchase, install, and maintain a central office POTS splitter and provide Knology access to data ports on the splitter. The splitter will route the High Frequency Spectrum on the circuit to Knology's xDSL equipment in Knology's collocation space. At least 30 days before making a change in splitter suppliers, BellSouth will provide Knology with a carrier notification letter, informing Knology of change. Knology shall purchase ports on the splitter in increments of 8, 24, or 96 ports in Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina and South Carolina. Knology shall purchase ports on the splitter in increments of 24 or 96 ports in Tennessee.
- 3.3.2 BellSouth will install the splitter in (i) a common area close to Knology's collocation area, if possible; or (ii) in a BellSouth relay rack as close to Knology's DS0 termination point as possible. Knology shall have access to the splitter for test purposes, regardless of where the splitter is placed in the BellSouth premises. For purposes of this section, a common area is defined as an area in the central office in which both Parties have access to a common test access point. A Termination Point is defined as the point of termination for Knology on the main distributing frame in the central office and is not the demarcation point set forth in Attachment 4 of this Agreement. BellSouth will cross-connect the splitter data ports to a specified Knology DS0 at such time that a Knology end user's service is established.

3.4 CLEC Provided Splitter

3.4.1 Knology may at its option purchase, install and maintain central office POTS splitters in its collocation arrangements. Knology may use such splitters for access to its customers and to provide digital line subscriber services to its customers using the High Frequency Spectrum. Existing Collocation rules and procedures and the terms and conditions relating to Collocation set forth in Attachment 4 shall apply.

3.4.2 Any splitters installed by Knology in its collocation arrangement shall comply with ANSI T1.413, Annex E, or any future ANSI splitter Standards. Knology may install any splitters that BellSouth deploys or permits to be deployed for itself or any BellSouth affiliate.

3.5 Ordering

- 3.5.1 Knology shall use BellSouth's Line Splitter Ordering Document ("LSOD") to order splitters from BellSouth and to activate and deactivate DS0 Collocation Connecting Facility Assignments (CFA) for use with High Frequency Spectrum.
- 3.5.2 BellSouth will provide Knology the Local Service Request ("LSR") format to be used when ordering the High Frequency Spectrum.
- 3.5.3 BellSouth will provision High Frequency Spectrum in compliance with BellSouth's Products and Services Interval Guide available at the website at http://www.interconnection.bellsouth.com.
- 3.5.4 BellSouth will provide Knology access to Preordering Loop Makeup (LMU) in accordance with the terms of this Agreement. BellSouth shall bill and Knology shall pay the rates for such services, as described in Exhibit B.

3.6 Maintenance and Repair

- 3.6.1 Knology shall have access for repair and maintenance purposes to any loop for which it has access to the High Frequency Spectrum. If Knology is using a BellSouth owned splitter, Knology may access the loop at the point where the combined voice and data signal exits the central office splitter via a bantam test jack. If Knology provides its own splitter, it may test from the collocation space or the Termination Point.
- 3.6.2 BellSouth will be responsible for repairing voice services and the physical line between the network interface device at the customer's premises and the Termination Point. Knology will be responsible for repairing data services. Each Party will be responsible for maintaining its own equipment.
- 3.6.3 Knology shall inform its end users to direct data problems to Knology, unless both voice and data services are impaired, in which event the end users should call BellSouth.
- 3.6.4 Once a Party has isolated a trouble to the other Party's portion of the loop, the Party isolating the trouble shall notify the end user that the trouble is on the other Party's portion of the Loop.
- 3.6.5 Notwithstanding anything else to the contrary in this Agreement, when BellSouth receives a voice trouble and isolates the trouble to the physical collocation arrangement belonging to Knology, BellSouth will notify Knology. Knology will

provide at least one but no more than two (2) verbal connecting facility assignments (CFA) pair changes to BellSouth in an attempt to resolve the voice trouble. In the event a CFA pair change resolves the voice trouble, Knology will provide BellSouth an LSR with the new CFA pair information within 24 hours. If the owner of the collocation space fails to resolve the trouble by providing BellSouth with the verbal CFA pair changes, BellSouth may discontinue Knology's access to the High Frequency Spectrum on such loop. BellSouth will not be responsible for any loss of data as a result of this action.

3.7 Line Splitting

- 3.7.1 General
- 3.7.2 Line splitting allows a provider of data services (a "Data LEC") and a provider of voice services (a "Voice CLEC") to deliver voice and data service to end-users over the same loop. The Voice CLEC and Data LEC may be the same or different carriers. Knology shall provide BellSouth with a signed Letter of Authorization ("LOA") between it and the Data LEC or Voice CLEC with which it desires to provision Line Splitting services, if Knology will not provide voice and data services.
- 3.7.3 End Users currently receiving voice service from a Voice CLEC through a UNE platform (UNE-P) may be converted to Line Splitting arrangements by Knology or its authorized agent ordering Line Splitting Service. If the CLEC wishes to provide the splitter, the UNE-P arrangement will be converted to a stand-alone UNE loop, a UNE port, two collocation cross connects and the high frequency spectrum line activation. If BellSouth owns the splitter, the UNE-P arrangement will be converted to a stand-alone UNE loop, port, and one collocation cross connection.
- 3.7.4 When end users on Loops using High Frequency Spectrum CO Based line sharing service are converted to Line Splitting, BellSouth will discontinue billing Knology for the High Frequency Spectrum. BellSouth will continue to bill the Data LEC for all associated splitter charges if the Data LEC continues to use a BellSouth splitter. It is the responsibility of Knology or its authorized agent to determine if the loop is compatible for Line Splitting Service. Knology or its authorized agent may use the existing loop unless it is not compatible with the Data LEC's data service and Knology or its authorized agent submits an LSR to BellSouth to change the loop.

3.8 **Provisioning Line Splitting and Splitter Space**

3.8.1 The Data LEC, Voice CLEC or BellSouth may provide the splitter. When Knology or its authorized agent owns the splitter, Line Splitting requires the following: a non-designed analog loop from the serving wire center to the network interface device (NID) at the end user's location; a collocation cross connection connecting the loop to the collocation space; a second collocation cross connection from the collocation space connected to a voice port; the high frequency spectrum line activation, and a splitter. The loop and port cannot be a loop and port combination (i.e. UNE-P), but must be individual stand-alone network elements. When BellSouth owns the splitter, Line Splitting requires the following: a non designed analog loop from the serving wire center to the network interface device (NID) at the end user's location with CFA and splitter port assignments, and a collocation cross connection from the collocation space connected to a voice port.

- 3.8.2 An unloaded 2-wire copper loop must serve the end user. The meet point for the Voice CLEC and the Data LEC is the point of termination on the MDF for the Data LEC's cable and pairs.
- 3.8.3 The foregoing procedures are applicable to migration to Line Splitting Service from a UNE-P arrangement, BellSouth Retail Voice Service, BellSouth High Frequency Spectrum (CO Based) Line Sharing.
- 3.8.4 For other migration scenarios to line splitting, BellSouth will work cooperatively with CLECs to develop methods and procedures to develop a process whereby a Voice CLEC and a Data LEC may provide services over the same loop.

3.9 Ordering

- 3.9.1 Knology shall use BellSouth's Line Splitter Ordering Document ("LSOD") to order splitters from BellSouth and to activate and deactivate DS0 Collocation Connecting Facility Assignments (CFA) for use with Line Splitting.
- 3.9.2 BellSouth shall provide Knology the Local Service Request ("LSR") format to be used when ordering Line Splitting service.
- 3.9.3 BellSouth will provision Line Splitting service in compliance with BellSouth's Products and Services Interval Guide available at the website at http://www.interconnection.bellsouth.com.
- 3.9.4 BellSouth will provide Knology access to Preordering Loop Makeup (LMU) in accordance with the terms of this Agreement. BellSouth shall bill and Knology shall pay the rates for such services as described in Exhibit B.
- 3.9.5 BellSouth will provide loop modification to Knology on an existing loop in accordance with procedures developed in the Line Sharing Collaborative. High Frequency Spectrum (CO Based) Unbundled Loop Modification is a separate distinct service from Unbundled Loop Modification set forth in Section 2.5 of this Attachment. Procedures for High Frequency Spectrum (CO Based) Unbundled Loop Modification may be found on the web at:
 <u>HTTP://www.interconnection.bellsouth.com/html/unes.html</u>. Nonrecurring rates for this UNE offering may be found in Exhibit B of this Attachment.

3.10 Maintenance

- 3.10.1 BellSouth will be responsible for repairing voice services and the physical line between the network interface device at the customer's premises and the Termination Point. Knology will be responsible for repairing data services. Each Party will be responsible for maintaining its own equipment.
- 3.10.2 Knology shall inform its end users to direct data problems to Knology, unless both voice and data services are impaired, in which event the end users should call BellSouth.
- 3.10.3 Once a Party has isolated a trouble to the other Party's portion of the loop, the Party isolating the trouble shall notify the end user that the trouble is on the other Party's portion of the Loop.
- 3.10.4 When BellSouth receives a voice trouble and isolates the trouble to the physical collocation arrangement belonging to owner of the collocation space, BellSouth will notify the owner of the collocation space. The owner of the collocation space will provide at least one but no more than two (2) verbal CFA pair changes to BellSouth in an attempt to resolve the voice trouble. In the event the CFA pair is changed, the owner of the collocation space will provide BellSouth an LSR with the new CFA pair information within 24 hours. If the owner of the collocation space fails to resolve the trouble by providing BellSouth with the verbal CFA pair changes, BellSouth may discontinue the owner of the collocation space access to the High Frequency Spectrum on such loop.
- 3.10.5 If Knology is not the data provider, Knology shall indemnify, defend and hold harmless BellSouth from and against any claims, losses, actions, causes of action, suits, demands, damages, injury, and costs including reasonable attorney fees, which arise out of actions related to the data provider.

3.11 Remote Site High Frequency Spectrum

- 3.11.1 General
- 3.11.2 BellSouth shall provide Knology access to the high frequency spectrum of the local sub-loop as an unbundled network element (UNE) only where BellSouth is the voice service provider to the end user at the rates set forth in this Attachment.
- 3.11.3 The High Frequency Spectrum is defined as the frequency range above the voiceband on a copper sub-loop facility carrying analog circuit-switched voiceband transmissions. Access to the High Frequency Spectrum is intended to allow Knology the ability to provide Digital Subscriber Line ("xDSL") data services to the end user for whom BellSouth provides voice services. The High Frequency Spectrum shall be available for any version of xDSL complying with Spectrum Management Class 5 of ANSI T1.417, American National Standard for Telecommunications, Spectrum Management for Loop Transmission Systems.

BellSouth will continue to have access to the low frequency portion of the subloop spectrum (from 300 Hertz to at least 3000 Hertz, and potentially up to 3400 Hertz, depending on equipment and facilities) for the purposes of providing voice service. Knology shall only use xDSL technology that is within the PSD mask for Spectrum Management Class 5 as found in the above-mentioned document.

- 3.11.4 Access to the High Frequency Spectrum requires an unloaded, 2-wire (Non-Designed) copper sub-loop. An unloaded copper sub-loop has no load coils, lowpass filters, range extenders, DAMLs, or similar devices and minimal bridged taps consistent with ANSI T1.413 and T1.601.
- 3.11.5 BellSouth will provide Loop Modification to Knology on an existing sub-loop in accordance with procedures developed in the Line Sharing Collaborative. Procedures for High Frequency Spectrum (Remote Site) Unbundled Loop Modification were developed in the Line Sharing Collaborative and may be found posted to the web at http://www.interconnection.bellsouth.com/html/unes.html. Nonrecurring rates for this UNE offering may be found in Exhibit B of this Attachment. BellSouth is not required to modify a loop for access to the High Frequency spectrum if modification of that loop significantly degrades BellSouth's voice service. If Knology requests modifications on a sub-loop longer than 18,000 ft. and requested modifications significantly degrades the voice services on the loop, Knology shall pay for the loop to be restored to its original state.
- 3.11.6 The High Frequency Spectrum shall only be available on sub-loops provided by BellSouth that continues to provide analog voice service directly to the end user. In the event the end-user terminates its BellSouth provided voice service for any reason, or in the event BellSouth disconnects the end user's voice service pursuant to its tariffs or applicable law, and Knology desires to continue providing xDSL service on such sub-loop, Knology shall be required to purchase a full stand-alone sub-loop. To the extent commercially practicable, BellSouth shall give Knology notice in a reasonable time prior to disconnect, which notice shall give Knology an adequate opportunity to notify BellSouth of its intent to purchase such sub-loop. In those cases where BellSouth no longer provides voice service to the end user and Knology purchases the full stand-alone sub-loop, Knology may elect the type of sub-loop it will purchase. Knology will pay the appropriate recurring and nonrecurring rates for such sub-loop as set forth in Exhibit B to this Attachment. In the event Knology purchases a voice grade Loop, Knology acknowledges that such sub-loop may not remain xDSL compatible.
- 3.11.7 Only one competitive local exchange carrier shall be permitted access to the High Frequency Spectrum of any particular sub-loop.

3.12 **Provisioning of High Frequency Spectrum and Splitter Space**

3.12.1 BellSouth will provide Knology with access to the High Frequency Spectrum as follows:

- 3.12.1.1 To order High Frequency Spectrum on a particular sub-loop, Knology must have a Digital Subscriber Line Access Multiplexer (DSLAM) collocated at the remote site that serves the end-user of such sub-loop.
- 3.12.1.2 Knology may provide its own splitters or may order splitters in a remote site once the Knology has installed its DSLAM at that remote site. BellSouth will install splitters within thirty-six (36) calendar days of Knology's submission of an error free Line Splitter Ordering Document ("LSOD") to the BellSouth Complex Resale Support Group.
- 3.12.1.3 Once a splitter is installed on behalf of Knology in a remote site in which Knology is located, Knology shall be entitled to order the High Frequency Spectrum on lines served out of that remote site. BellSouth will bill and Knology shall pay applicable for High Frequency Spectrum end-user activation.

3.13 BellSouth Owned Splitter

- 3.13.1 BellSouth will select, purchase, install and maintain a splitter at the remote site. The Knology's meet point is at the BellSouth "cross connect" point located at the Feeder Distribution Interface (FDI). Knology will provide a cable facility to the BellSouth FDI. BellSouth will splice the Knology's cable to BellSouth's spare binding post in the FDI and use "cross connects" to connect the Knology's cable facility to the BellSouth splitter. The splitter will route the high frequency portion of the circuit to the Knology's xDSL equipment in their collocation space. Access to the high frequency spectrum is not compatible with foreign exchange (FX) lines, ISDN, and other services listed in the technical section of this document.
- 3.13.2 The BellSouth splitter bifurcates the digital and voice band signals. The low frequency voice band portion of the circuit is routed back to the BellSouth switch. The high frequency digital traffic portion of the circuit is routed to the xDSL equipment in the Knology's Remote Terminal (RT) collocation space and routed back to the Knology's network. At least 30 business days before making a change in splitter suppliers, BellSouth will provide Knology with a carrier notification letter informing Knology of change. Knology shall purchase ports on the splitter in increments of 24 ports.
- 3.13.3 BellSouth will install the splitter in (i) a common area close to Knology's collocation area, if possible; or (ii) in a BellSouth relay rack as close to Knology's DS0 termination point as possible. Knology shall have access to the splitter for test purposes regardless of where the splitter is placed in the BellSouth premises. For purposes of this section, a common area is defined as an area in the remote site in which both Parties have access to a common test access point. BellSouth will cross-connect the splitter data ports to a specified Knology DS0 at such time that a Knology end user's service is established.

3.14 **CLEC Owned Splitter**

- 3.14.1 Knology may at its option purchase, install and maintain splitters in its collocation arrangements. Knology may use such splitters for access to its customers and to provide digital line subscriber services to its customers using the High Frequency Spectrum. Existing Collocation rules and procedures shall apply. Knology will be required to activate cable pairs in no less than 8 (eight) pair increments.
- 3.14.2 Any splitters installed by Knology in its collocation arrangement shall comply with ANSI T1.413, Annex E, or any future ANSI splitter Standards. Knology may install any splitters that BellSouth deploys or permits to be deployed for itself or any BellSouth affiliate.

3.15 Ordering

- 3.15.1 Knology shall use BellSouth's Remote Splitter Ordering Document ("RSOD") to order and activate splitters from BellSouth or to activate CLEC owned splitters at an RT for use with High Frequency Spectrum.
- 3.15.2 BellSouth will provide Knology the Local Service Request ("LSR") format to be used when ordering the High Frequency Spectrum.
- 3.15.3 BellSouth will provision High Frequency Spectrum in compliance with BellSouth's Products and Services Interval Guide available at the website at http://www.interconnection.bellsouth.com.
- 3.15.4 BellSouth will provide Knology access to Preordering Loop Makeup (LMU) in accordance with the terms of this Agreement. BellSouth shall bill and Knology shall pay the rates for such services as described in Exhibit B.
- 3.15.5 BellSouth shall test the data portion of the sub-loop to ensure the continuity of the wiring for Knology's data.

3.16 Maintenance and Repair

- 3.16.1 Knology (Reneg 2002) shall have access for repair and maintenance purposes to any sub-loop for which it has access to the High Frequency Spectrum. If Knology is using a BellSouth owned splitter, Knology may access the sub-loop at the point where the data signal exits. If Knology provides its own splitter, it may test from the collocation space or the Termination Point.
- 3.16.2 BellSouth will be responsible for repairing voice services and the physical line between the network interface device at the customer's premises and the Termination Point. Knology will be responsible for repairing data services. Each Party will be responsible for maintaining its own equipment.
- 3.16.3 Knology shall inform its end users to direct data problems to Knology, unless both voice and data services are impaired, in which event the end users should call BellSouth.

- 3.16.4 Once a Party has isolated a trouble to the other Party's portion of the sub-loop, the Party isolating the trouble shall notify the end user that the trouble is on the other Party's portion of the sub-loop.
- 3.16.5 Notwithstanding anything else to the contrary in this Agreement, when BellSouth receives a voice trouble and isolates the trouble to the physical collocation arrangement belonging to Knology, BellSouth will notify Knology. Knology will provide at least one but no more than two (2) verbal connecting facility assignments (CFA) pair changes to BellSouth in an attempt to resolve the voice trouble. In the event a CFA pair change resolves the voice trouble, Knology will provide BellSouth an LSR with the new CFA pair information within 24 hours. If the owner of the collocation space fails to resolve the trouble by providing BellSouth with the verbal CFA pair changes, BellSouth may discontinue Knology's access to the High Frequency Spectrum on such sub-loop. BellSouth will not be responsible for any loss of data as a result of this action.

4 Local Switching

4.1 BellSouth shall provide non-discriminatory access to local circuit switching capability and local tandem switching capability on an unbundled basis, except as set forth in the Sections below to Knology for the provision of a telecommunications service. BellSouth shall provide non-discriminatory access to packet switching capability on an unbundled basis to Knology for the provision of a telecommunications service only in the limited circumstance described below in Section 4.5.

4.2 Local Circuit Switching Capability, including Tandem Switching Capability

- 4.2.1Local circuit switching capability is defined as: (A) line-side facilities, which include but are not limited to the connection between a loop termination at a main distribution frame and a switch line card; (B) trunk-side facilities, which include but are not limited to the connection between trunk termination at a trunk-side cross-connect panel and a switch trunk card; (C) switching provided by remote switching modules; and (D) all features, functions, and capabilities of the switch, which include but are not limited to: (1) the basic switching function of connecting lines to lines, line to trunks, trunks to lines, and trunks to trunks, as well as the same basic capabilities made available to BellSouth's customers, such as a telephone number, white page listings, and dial tone; and (2) all other features that the switch is capable of providing, including but not limited to customer calling, customer local area signaling service features, and Centrex, as well as any technically feasible customized routing functions provided by the switch. Any features that are not currently available but are technically feasible through the switch can be requested through the BFR/NBR process.
- 4.2.2 Notwithstanding BellSouth's general duty to unbundle local circuit switching, BellSouth shall not be required to unbundle local circuit switching for Knology

when Knology serves an end-user with four (4) or more voice-grade (DS-0) equivalents or lines served by BellSouth in one of the following MSAs: Atlanta, GA; Miami, FL; Orlando, FL; Ft. Lauderdale, FL; Charlotte-Gastonia-Rock Hill, NC; Greensboro-Winston Salem-High Point, NC; Nashville, TN; and New Orleans, LA, and BellSouth has provided non-discriminatory cost based access to the Enhanced Extended Link (EEL) throughout Density Zone 1 as determined by NECA Tariff No. 4 as in effect on January 1, 1999.

- 4.2.3 In the event that Knology orders local circuit switching for an end user with four (4) or more DS0 equivalent lines within Density Zone 1 in an MSA listed above, BellSouth shall charge Knology the market based rates in Exhibit B for use of the local circuit switching functionality for the affected facilities. If a market rate is not set forth in Exhibit B, such rate shall be negotiated by the Parties.
- 4.2.4 Unbundled Local Switching consists of three separate unbundled elements: Unbundled Ports, End Office Switching Functionality, and End Office Interoffice Trunk Ports.
- 4.2.5 Unbundled Local Switching combined with Common Transport and, if necessary, Tandem Switching provides to Knology's end user local calling and the ability to presubscribe to a primary carrier for intraLATA and/or to presubscribe to a primary carrier for interLATA toll service.
- 4.2.6 Provided that Knology purchases unbundled local switching from BellSouth and uses the BellSouth CIC for its end users' LPIC or if a BellSouth local end user selects BellSouth as its LPIC, then the Parties will consider as local any calls originated by a Knology local end user, or originated by a BellSouth local end user and terminated to a Knology local end user, where such calls originate and terminate in the same LATA, except for those calls originated and terminated through switched access arrangements (i.e., calls that are transported by a Party other than BellSouth). For such calls, BellSouth will charge Knology the UNE elements for the BellSouth facilities utilized. Neither Party shall bill the other originating or terminating switched access charges for such calls. Intercarrier compensation for local calls between BellSouth and Knology shall be as described in BellSouth's UNE Local Call Flows set forth on BellSouth's web site.
- 4.2.7 Where Knology purchases unbundled local switching from BellSouth but does not use the BellSouth CIC for its end users' LPIC, BellSouth will consider as local those direct dialed telephone calls that originate from a Knology end user and terminate within the basic local calling area or within the extended local calling areas and that are dialed using 7 or 10 digits as defined and specified in Section A3 of BellSouth's General Subscriber Services Tariffs. For such local calls, BellSouth will charge Knology the UNE elements for the BellSouth facilities utilized. Intercarrier compensation for local calls between BellSouth and Knology shall be as described in BellSouth's UNE Local Call Flows set forth on BellSouth's web site.

4.2.8 For any calls that originate and terminate through switched access arrangements (i.e., calls that are transported by a party other than BellSouth), BellSouth shall bill Knology the UNE elements for the BellSouth facilities utilized. Each Party may bill the toll provider originating or terminating switched access charges as appropriate.

4.2.9 Unbundled Port Features

- 4.2.9.1 Charges for Unbundled Port are as set forth in Exhibit B, and as specified in such exhibit, may or may not include individual features.
- 4.2.9.2 Where applicable and available, non-switch-based services may be ordered with the Unbundled Port at BellSouth's retail rates.
- 4.2.9.3 Any features that are not currently available but are technically feasible through the switch can be requested through the- BFR/NBR process.
- 4.2.9.4 BellSouth will provide to Knology selective routing of calls to a requested Operator System platform pursuant to Section 10 of Attachment 2. Any other routing requests by Knology will be made pursuant to the BFR/NBR Process as set forth in Attachment 11.

4.2.10 **Remote Call Forwarding**

- 4.2.10.1 As an option, BellSouth shall make available to Knology an unbundled port with Remote Call Forwarding capability ("URCF service"). URCF service combines the functionality of unbundled local switching, tandem switching and common transport to forward calls from the URCF service telephone number (the number dialed by the calling party) to another telephone number selected by the URCF service subscriber. When ordering URCF service, Knology will ensure that the following conditions are satisfied:
- 4.2.10.1.1 That the end user of the forward-to number (service) agrees to receive calls forwarded using the URCF service (if such end user is different from the URCF service end user);
- 4.2.10.1.2 That the forward-to number (service) is equipped with sufficient capacity to receive the volume of calls that will be generated from the URCF service;
- 4.2.10.1.3 That the URCF service will not be utilized to forward calls to another URCF or similar service; and
- 4.2.10.1.4 That the forward-to number (service) is not a public safety number (e.g. 911, fire or police number).
- 4.2.10.2 In addition to the charge for the URCF service port, BellSouth shall charge Knology the rates set forth in Exhibit B for unbundled local switching, tandem

switching, and common transport, including all associated usage incurred for calls from the URCF service telephone number (the number dialed by the calling party) to the forward- to number (service).

4.2.11 Provision for Local Switching

- 4.2.11.1 BellSouth shall perform routine testing (e.g., Mechanized Loop Tests (MLT) and test calls such as 105, 107 and 108 type calls) and fault isolation on a mutually agreed upon schedule.
- 4.2.11.2 BellSouth shall control congestion points such as those caused by radio station call-ins and network routing abnormalities. All traffic shall be restricted in a non-discriminatory manner.
- 4.2.11.3 BellSouth shall perform manual call trace and permit customer originated call trace. BellSouth shall provide Switching Service Point (SSP) capabilities and signaling software to interconnect the signaling links destined to the Signaling Transfer Point Switch (STPS). These capabilities shall adhere to the technical specifications set forth in the applicable industry standard technical references.
- 4.2.11.4 BellSouth shall provide interfaces to adjuncts through Telcordia standard interfaces. These adjuncts can include, but are not limited to, the Service Circuit Node and Automatic Call Distributors. BellSouth shall offer to Knology all AIN triggers in connection with its SMS/SCE offering.
- 4.2.11.5 BellSouth shall provide access to SS7 Signaling Network or Multi-Frequency trunking if requested by Knology.

4.2.12 Local Switching Interfaces.

- 4.2.12.1 Knology shall order ports and associated interfaces compatible with the services it wishes to provide as listed in Exhibit B. BellSouth shall provide the following local switching interfaces:
- 4.2.12.1.1 Standard Tip/Ring interface including loopstart or groundstart, on-hook signaling (e.g., for calling number, calling name and message waiting lamp);
- 4.2.12.1.2 Coin phone signaling;
- 4.2.12.1.3 Basic Rate Interface ISDN adhering to appropriate Telcordia Technical Requirements;
- 4.2.12.1.4 Two-wire analog interface to PBX;
- 4.2.12.1.5 Four-wire analog interface to PBX;

- 4.2.12.1.6 Four-wire DS1 interface to PBX or customer provided equipment (e.g. computers and voice response systems);
- 4.2.12.1.7 Primary Rate ISDN to PBX adhering to ANSI standards Q.931, Q.932 and appropriate Telcordia Technical Requirements;
- 4.2.12.1.8 Switched Fractional DS1 with capabilities to configure Nx64 channels (where N = 1 to 24); and
- 4.2.12.1.9 Loops adhering to Telcordia TR-NWT-08 and TR-NWT-303 specifications to interconnect Digital Loop Carriers.

4.3 Tandem Switching

- 4.3.1 The Tandem Switching capability Network Element is defined as: (i) trunkconnect facilities, which include, but are not limited to, the connection between trunk termination at a cross connect panel and switch trunk card; (ii) the basic switch trunk function of connecting trunks to trunks; and (iii) the functions that are centralized in the Tandem Switches (as distinguished from separate end office switches), including but not limited to call recording, the routing of calls to operator services and signaling conversion features.
- 4.3.2 <u>Technical Requirements</u>
- 4.3.2.1 Tandem Switching shall have the same capabilities or equivalent capabilities as those described in Telcordia TR-TSY-000540 Issue 2R2, Tandem Supplement, 6/1/90. The requirements for Tandem Switching include but are not limited to the following:
- 4.3.2.1.1 Tandem Switching shall provide signaling to establish a tandem connection;
- 4.3.2.1.2 Tandem Switching will provide screening as jointly agreed to by Knology and BellSouth;
- 4.3.2.1.3 Tandem Switching shall provide Advanced Intelligent Network triggers supporting AIN features where such routing is not available from the originating end office switch, to the extent such Tandem switch has such capability;
- 4.3.2.1.4 Tandem Switching shall provide access to Toll Free number database;
- 4.3.2.1.5 Tandem Switching shall provide connectivity to PSAPs where 911 solutions are deployed and the tandem is used for 911; and
- 4.3.2.1.6 Where appropriate, Tandem Switching shall provide connectivity for the purpose of routing transit traffic to and from other carriers.

- 4.3.2.2 BellSouth may perform testing and fault isolation on the underlying switch that is providing Tandem Switching. Such testing shall be testing routinely performed by BellSouth. The results and reports of the testing shall be made available to Knology.
- 4.3.2.3 BellSouth shall control congestion points and network abnormalities. All traffic will be restricted in a non-discriminatory manner.
- 4.3.2.4 Tandem Switching shall process originating toll-free traffic received from Knology's local switch.
- 4.3.2.5 In support of AIN triggers and features, Tandem Switching shall provide SSP capabilities when these capabilities are not available from the Local Switching Network Element to the extent such Tandem Switch has such capability.
- 4.3.3 Upon Knology's purchase of overflow trunk groups, Tandem Switching shall provide an alternate routing pattern for Knology's traffic overflowing from direct end office high usage trunk groups.

4.4 <u>AIN Selective Carrier Routing for Operator Services, Directory Assistance</u> and Repair Centers

- 4.4.1 BellSouth will provide AIN Selective Carrier Routing at the request of Knology.
 AIN Selective Carrier Routing will provide Knology with the capability of routing operator calls, 0+ and 0- and 0+ NPA (LNPA) 555-1212 directory assistance, 1+411 directory assistance and 611 repair center calls to pre-selected destinations.
- 4.4.2 Knology shall order AIN Selective Carrier Routing through its Account Team and/or Local Contract Manager. AIN Selective Carrier Routing must first be established regionally and then on a per central office per state basis.
- 4.4.3 AIN Selective Carrier Routing is not available in DMS 10 switches.
- 4.4.4 Where AIN Selective Carrier Routing is utilized by Knology, the routing of Knology's end user calls shall be pursuant to information provided by Knology and stored in BellSouth's AIN Selective Carrier Routing Service Control Point database. AIN Selective Carrier Routing shall utilize a set of Line Class Codes (LCCs) unique to a basic class of service assigned on an "as needed" basis. The same LCCs will be assigned in each central office where AIN Selective Carrier Routing is established.
- 4.4.5 Upon ordering AIN Selective Carrier Routing Regional Service, Knology shall remit to BellSouth the Regional Service Order non-recurring charges set forth in Exhibit B of this Attachment. There shall be a non-recurring End Office Establishment Charge per office due at the addition of each central office where AIN Selective Carrier Routing will be utilized. Said non-recurring charge shall be as set forth in Exhibit B of this Attachment. For each Knology end user activated,

there shall be a non-recurring End User Establishment charge as set forth in Exhibit B of this Attachment. Knology shall pay the AIN Selective Carrier Routing Per Query Charge set forth in Exhibit B of this Attachment.

- 4.4.6 This Regional Service Order non-recurring charge will be non-refundable and will be paid with 1/2 due up-front with the submission of all fully completed required forms including: Regional Selective Carrier Routing (SCR) Order Request-Form A, Central Office AIN Selective Carrier Routing (SCR) Order Request Form B, AIN_SCR Central Office Identification Form Form C, AIN_SCR Routing Options Selection Form Form D, and Routing Combinations Table Form E. BellSouth has 30 days to respond to Knology's fully completed firm order as a Regional Service Order. With the delivery of this firm order response to Knology, BellSouth considers that the delivery schedule of this service commences. The remaining 1/2 of the Regional Service Order payment must be paid when at least 90% of the Central Offices listed on the original order have been turned up for the service.
- 4.4.7 The non-recurring End Office Establishment Charge will be billed to Knology following BellSouth's normal monthly billing cycle for this type of order.
- 4.4.8 End-User Establishment Orders will not be turned-up until the second payment is received for the Regional Service Order. The non-recurring End-User Establishment Charges will be billed to Knology following BellSouth's normal monthly billing cycle for this type of order.
- 4.4.9 Additionally, the AIN Selective Carrier Routing Per Query Charge will be billed to Knology following the normal billing cycle for per query charges.
- 4.4.10 All other network components needed, for example, unbundled switching, unbundled local transport, etc., will be billed per contracted rates.

4.5 Packet Switching Capability

- 4.5.1 The packet switching capability network element is defined as the function of routing or forwarding packets, frames, cells or other data units based on address or other routing information contained in the packets, frames, cells or other data units.
- 4.5.2 BellSouth shall be required to provide non-discriminatory access to unbundled packet switching capability only where each of the following conditions are satisfied:
- 4.5.2.1 BellSouth has deployed digital loop carrier systems, including but not limited to, integrated digital loop carrier or universal digital loop carrier systems; or has deployed any other system in which fiber optic facilities replace copper facilities in the feeder section (e.g., end office to remote terminal, pedestal or environmentally controlled vault);

- 4.5.2.2 There are no spare copper loops capable of supporting the xDSL services Knology seeks to offer;
- 4.5.2.3 BellSouth has not permitted Knology to deploy a DSLAM at the remote terminal, pedestal or environmentally controlled vault or other interconnection point, nor has Knology obtained a virtual collocation arrangement at these sub-loop interconnection points as defined by 47 CFR § 51.319 (b); and
- 4.5.2.4 BellSouth has deployed packet switching capability for its own use.
- 4.5.3 If there is a dispute as to whether BellSouth must provide Packet Switching, such dispute will be resolved according to the dispute resolution process set forth in Section 10 of the General Terms and Conditions of this Agreement incorporated herein by this reference.

5 Unbundled Network Element Combinations

5.1 For purposes of this Section, references to "Currently Combined" network elements shall mean that the particular network elements requested by Knology are in fact already combined by BellSouth in the BellSouth network. References to "Ordinarily Combined" network elements shall mean that the particular network elements requested by Knology are not already combined by BellSouth in the location requested by Knology but are elements that are typically combined in BellSouth's network. References to "Not Typically Combined" network elements shall mean that the particular network elements requested by Knology are not elements that BellSouth combines for its use in its network.

5.2 Enhanced Extended Links (EELs)

- 5.2.1 EELs are combinations of unbundled loops and unbundled dedicated transport as defined in Section 6. BellSouth shall provide Knology with EELs where they are available.
- 5.2.2 BellSouth will provide access to EELs in the combinations set forth in Section 5.4.1 below.
- 5.2.3 EELs are intended to provide service connectivity from an end user's location through that end user's SWC to Knology's collocation space in a BellSouth central office. The circuit must be connected to the Knology's switch for the purpose of provisioning circuit telephone exchange service to the Knology's end-user customers. Knology may connect EELs within the Knology's collocation space to other transport terminating into Knology's switch. Knology may also connect the local loops listed in Section 5.3.1.3 to an appropriate Unbundled Local Channel to form additional EELs which terminate in Knology's switch. Provided that the entire EEL circuit meets the criteria set forth in Section 5.3.1.3 below, the circuit may, upon Knology's request, terminate to a CLEC's Point of Presence ("POP"). Knology will provide a significant amount of local exchange service over the

requested combination, as described in Section 5.3.1 et seq. below. Upon BellSouth's request, Knology shall indicate under what local usage option Knology seeks to qualify. Knology shall be deemed to providing a significant amount of local exchange service over the requested combination if one of the options listed in Section 5.3.1 et seq. is met. BellSouth shall have the right to audit Knology's EELs as specified in Section 5.3.3 below.

5.3 Conversions from Special Access Service to EELs

- 5.3.1 Knology may not convert existing special access services to combinations of loop and transport network elements, whether or not Knology self-provides its entrance facilities (or obtains entrance facilities from a third party), unless Knology uses the combination to provide a significant amount of local exchange service, in addition to exchange access service, to a particular customer. To the extent Knology requests to convert any special access services to combinations of loop and transport network elements at UNE prices, Knology shall provide to BellSouth a certification that Knology is providing a significant amount of local exchange service (as described in this Section) over such combinations. The certification shall also indicate under what local usage option Knology seeks to qualify for conversion of special access circuits. Knology shall be deemed to be providing a significant amount of local exchange service over such combinations if one of the following options is met:
- 5.3.1.1 **Option 1:** Knology certifies that it is the exclusive provider of an end user's local exchange service. The loop-transport combinations must terminate at Knology's collocation arrangement in at least one BellSouth central office. This option does not allow loop-transport combinations to be connected to BellSouth's tariffed services. Under this option, Knology is the end user's only local service provider, and thus is providing more than a significant amount of local exchange service. Knology can then use the loop-transport combinations that serve the end user to carry any type of traffic, including using them to carry 100 percent interstate access traffic; or
- 5.3.1.2 **Option 2:** Knology certifies that it provides local exchange and exchange access service to the end user customer's premises and handles at least one third of the end user customer's local traffic measured as a percent of total end user customer local dial tone lines; and for DS1 circuits and above, at least 50 percent of the activated channels on the loop portion of the loop-transport combination have at least 5 percent local voice traffic individually, and the entire loop facility has at least 10 percent local voice traffic. When a loop-transport combination includes multiplexing, each of the individual DS1 circuits must meet this criterion. The loop-transport combination must terminate at Knology's collocation arrangement in at least one BellSouth central office. This option does not allow loop-transport

combinations to be connected to BellSouth tariffed services; or

- 5.3.1.3 **Option 3:** Knology certifies that at least 50 percent of the activated channels on a circuit are used to provide originating and terminating local dial tone service and at least 50 percent of the traffic on each of these local dial tone channels is local voice traffic, and that the entire loop facility has at least 33 percent local voice traffic. When a loop-transport combination includes multiplexing, each of the individual DS1 circuits must meet this criterion. This option does not allow loop-transport combinations to be connected to BellSouth's tariffed services. Under this option, collocation is not required. Knology does not need to provide a defined portion of the end user's local service, but the active channels on any loop-transport combination, and the entire facility, must carry the amount of local exchange traffic specified in this option.
- 5.3.2 In addition, there may be extraordinary circumstances where Knology is providing a significant amount of local exchange service but does not qualify under any of the three options set forth in Section 5.3.1 et seq. In such case, Knology may petition the FCC for a waiver of the local usage options set forth above. If a waiver is granted, then upon Knology's request the Parties shall amend this Agreement to the extent necessary to incorporate the terms of such waiver for such extraordinary circumstance.
- 5.3.3 BellSouth may, at its sole discretion, audit Knology's records in order to verify compliance with the local usage option provided by Knology pursuant to Section 5.3.1. The audit shall be conducted by a third party independent auditor, and Knology shall be given thirty days written notice of scheduled audit. Such audit shall occur no more than one time in a calendar year unless results of an audit find noncompliance with the significant amount of local exchange service requirement. In the event of noncompliance, Knology shall reimburse BellSouth for the cost of the audit. If, based on the audit, Knology is not providing a significant amount of local exchange traffic over the combinations of loop and transport network elements, BellSouth will convert such combinations of loop and transport network elements to special access services in accordance with BellSouth's tariffs and will bill Knology for appropriate retroactive reimbursement. If the Parties disagree as to whether the audits indicate that Knology is not providing a significant amount of local exchange traffic, the dispute will be resolved according to the dispute resolution process set forth in Section 10 of the General Terms and Conditions of this Agreement incorporated herein by this reference.
- 5.3.4 In the event Knology converts special access circuits to combinations of loop and transport UNEs pursuant to the terms of this Section, Knology shall be subject to the termination liability provisions in the applicable special access tariffs, if any.

1 1

DOIT

cc*

- - - -

5.4.1	Currently Combined EELs listed below in Sections 5.4.1.1-5.4.1.14 shall be billed
	at the nonrecurring switch-as-is charge and recurring charges for that combination
	as set forth in Exhibit B of this Attachment. Currently Combined EELs not listed
	below shall be billed at the sum of the nonrecurring and recurring charges for the
	individual network elements that comprise the combination as set forth in Exhibit
	B of this Attachment.

5.4.1.1	DST Interoffice Channel + DST Channelization + 2-wire VG Local Loop
5.4.1.2	DS1 Interoffice Channel + DS1 Channelization + 4-wire VG Local Loop
5.4.1.3	DS1 Interoffice Channel + DS1 Channelization + 2-wire ISDN Local Loop

- 5.4.1.4 DS1 Interoffice Channel + DS1 Channelization + 4-wire 56 kbps Local Loop
- 5.4.1.5 DS1 Interoffice Channel + DS1 Channelization + 4-wire 64 kbps Local Loop
- 5.4.1.6 DS1 Interoffice Channel + DS1 Local Loop
- 5.4.1.7 DS3 Interoffice Channel + DS3 Local Loop
- 5.4.1.8 STS-1 Interoffice Channel + STS-1 Local Loop
- 5.4.1.9 DS3 Interoffice Channel + DS3 Channelization + DS1 Local Loop
- 5.4.1.10 STS-1 Interoffice Channel + DS3 Channelization + DS1 Local Loop
- 5.4.1.11 2-wire VG Interoffice Channel + 2-wire VG Local Loop
- 5.4.1.12 4wire VG Interoffice Channel + 4-wire VG Local Loop

5.4.1.13 4-wire 56 kbps Interoffice Channel + 4-wire 56 kbps Local Loop

- 5.4.1.14 4-wire 64 kbps Interoffice Channel + 4-wire 64 kbps Local Loop
- 5.4.2 Ordinarily Combined EELs listed above shall be billed the sum of the nonrecurring and recurring charges for that combination as set forth in Exhibit B of this Attachment. Ordinarily combined EELs not listed in Sections 5.4.1.1-5.4.1.14 shall be billed the sum of the nonrecurring charges and recurring charges for the individual network elements that comprise the combination as set forth in Exhibit B of this Attachment.
- 5.4.3 To the extent that Knology requests an EEL combination Not Typically Combined in the BellSouth network, the rates, terms and conditions shall be determined pursuant to the Bona Fide Request Process.

5.5 UNE Port/Loop Combinations

- 5.5.1 Combinations of port and loop unbundled network elements along with switching and transport unbundled network elements provide local exchange service for the origination or termination of calls. Port/ loop combinations support the same local calling and feature requirements as described in the Unbundled Local Switching or Port section of this Attachment 2 and the ability to presubscribe to a primary carrier for intraLATA toll service and/or to presubscribe to a primary carrier for interLATA toll service.
- 5.5.2 BellSouth shall make available UNE port/loop combinations, regardless of whether such combinations are Currently Combined, as long as such combinations are Ordinarily Combined in BellSouth's network.
- 5.5.3 Except as set forth in Section 5.5.4 below, BellSouth shall provide UNE port/loop combinations described in Section 5.5.6 below that are Currently Combined or Ordinarily Combined in BellSouth's network at the cost-based rates in Exhibit B. Except as set forth in Section 5.5.4 below, BellSouth shall provide UNE port/loop combinations not described in Section 5.5.6 below or Not Typically Combined Combinations in accordance with the Bona Fide Request process.
- 5.5.4 BellSouth is not required to provide combinations of port and loop network elements on an unbundled basis in locations where, pursuant to FCC rules, BellSouth is not required to provide circuit switching as an unbundled network element.
- 5.5.4.1 BellSouth shall not be required to provide local circuit switching as an unbundled network element in density Zone 1, as defined in 47 CFR 69.123 as of January 1,

1999 of the Atlanta, GA; Miami, FL; Orlando, FL; Ft. Lauderdale, FL; Charlotte-Gastonia-Rock Hill, NC; Greensboro-Winston Salem-High Point, NC; Nashville, TN; and New Orleans, LA, MSAs to Knology if Knology's customer has 4 or more DS0 equivalent lines.

- 5.5.4.2 Notwithstanding the foregoing, BellSouth shall provide combinations of port and loop network elements on an unbundled basis where, pursuant to FCC rules, BellSouth is not required to provide local circuit switching as an unbundled network element and shall do so at the market rates in Exhibit B. If a market rate is not set forth in Exhibit B for a UNE port/loop combination, such rate shall be negotiated by the Parties.
- 5.5.5 BellSouth shall make 911 updates in the BellSouth 911 database for Knology's UNE port/loop combinations. BellSouth will not bill Knology for 911 surcharges. Knology is responsible for paying all 911 surcharges to the applicable governmental agency.
- 5.5.6 Combination Offerings
- 5.5.6.1 2-wire voice grade port, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 5.5.6.2 2-wire voice grade Coin port, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 5.5.6.3 2-wire voice grade DID port, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 5.5.6.4 2-wire CENTREX port, voice grade loop, CENTREX intercom functionality, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 5.5.6.5 2-wire ISDN Basic Rate Interface, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 5.5.6.6 4-wire ISDN Primary Rate Interface, DS1 loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 5.5.6.7 4-wire DS1 Trunk port, DS1 Loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.

5.5.6.8 4-wire DS1 Loop with normal serving wire center channelization interface, 2-wire voice grade ports (PBX), 2-wire DID ports, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.

5.6 **Other UNE Combinations**

- 5.6.1 BellSouth shall provide other Currently Combined and Ordinarily Combined and Not Typically Combined UNE Combinations to Knology in addition to those specifically referenced in this Section 5 above, where available. Such combinations shall not be connected to BellSouth tariffed services. To the extent Knology requests a combination for which BellSouth does not have methods and procedures in place to provide such combination, rates and/or methods and procedures for such combination will be developed pursuant to the BFR/NBR process.
- 5.6.2 Rates
- 5.6.3 The rates for Ordinarily Combined UNE Combinations shall be the sum of the recurring rates and nonrecurring rates for the stand-alone network elements as set forth in Exhibit B of this Attachment. The rates for Currently Combined UNE Combinations shall be the sum of the recurring rates for the stand-alone network elements as set forth in Exhibit B, in addition to a nonrecurring charge set forth in Exhibit B. To the extent Knology requests a Not Typically Combined Combination, or to the extent Knology requests any combination for which BellSouth has not developed methods and procedures to provide such combination, rates and/or methods and procedures for such combination shall be established pursuant to the BFR/NBR process.

6 Transport, Channelization and Dark Fiber

6.1 **Transport**

- 6.1.1 BellSouth shall provide nondiscriminatory access, in accordance with FCC Rule
 51.311 and Section 251(c)(3) of the Act, to interoffice transmission facilities on an unbundled basis to Knology for the provision of a telecommunications service. Interoffice transmission facility network elements include:
- 6.1.1.1 Dedicated transport, defined as BellSouth's transmission facilities, is dedicated to a particular customer or carrier that provides telecommunications between wire centers or switches owned by BellSouth, or between wire centers and switches owned by BellSouth and Knology.
- 6.1.1.2 Dark Fiber transport, defined as BellSouth's optical transmission facilities without attached signal regeneration, multiplexing, aggregation or other electronics;

- 6.1.1.3 Common (Shared) transport, defined as transmission facilities shared by more than one carrier, including BellSouth, between end office switches, between end office switches and tandem switches, and between tandem switches, in BellSouth's network. Where BellSouth Network Elements are connected by intraoffice wiring, such wiring is provided as part of the Network Element and is not Common (Shared) Transport.
- 6.1.2 BellSouth shall:
- 6.1.2.1 Provide Knology exclusive use of interoffice transmission facilities dedicated to a particular customer or carrier, or shared use of the features, functions, and capabilities of interoffice transmission facilities shared by more than one customer or carrier;
- 6.1.2.2 Provide all technically feasible transmission facilities, features, functions, and capabilities of the transport facility for the provision of telecommunications services;
- 6.1.2.3 Permit, to the extent technically feasible, Knology to connect such interoffice facilities to equipment designated by Knology, including but not limited to, Knology's collocated facilities; and
- 6.1.2.4 Permit, to the extent technically feasible, Knology to obtain the functionality provided by BellSouth's digital cross-connect systems.
- 6.1.3 Technical Requirements of Common (Shared) Transport
- 6.1.3.1 Common (Shared) Transport provided on DS1 or VT1.5 circuits shall at a minimum meet the performance, availability, jitter, and delay requirements specified for Central Office to Central Office ("CO to CO") connections in the applicable industry standards.
- 6.1.3.2 Common (Shared) Transport provided on DS3 circuits, STS-1 circuits, and higher transmission bit rate circuits shall at a minimum meet the performance, availability, jitter, and delay requirements specified for CO to CO connections in the applicable industry standards.
- 6.1.3.3 BellSouth shall be responsible for the engineering, provisioning, and maintenance of the underlying equipment and facilities that are used to provide Common (Shared) Transport.
- 6.1.3.4 At a minimum, Common (Shared) Transport shall meet all of the requirements set forth in the applicable industry standards.

6.2 **Dedicated Transport**

6.2.1 Dedicated Transport is composed of the following Unbundled Network Elements:

- 6.2.1.1 Unbundled Local Channel, defined as the dedicated transmission path between Knology's Point of Presence ("POP") and Knology's collocation space in the BellSouth Serving Wire Center for Knology's POP, and
- 6.2.1.2 Unbundled Interoffice Channel, defined as the dedicated transmission path that provides telecommunication between BellSouth's Serving Wire Centers' collocations.
- 6.2.1.3 BellSouth shall offer Dedicated Transport in each of the following ways:
- 6.2.1.3.1 As capacity on a shared UNE facility.
- 6.2.1.3.2 As a circuit (e.g., DS0, DS1, DS3) dedicated to Knology.
- 6.2.1.4 Dedicated Transport may be provided over facilities such as optical fiber, copper twisted pair, and coaxial cable, and shall include transmission equipment such as line terminating equipment, amplifiers, and regenerators.
- 6.2.2 Technical Requirements
- 6.2.2.1 The entire designated transmission service (e.g., DS0, DS1, DS3) shall be dedicated to Knology designated traffic.
- 6.2.2.2 For DS1 or VT1.5 circuits, Dedicated Transport shall at a minimum meet the performance, availability, jitter, and delay requirements specified for Customer Interface to Central Office ("CI to CO") connections in the applicable industry standards.
- 6.2.2.3 For DS3 circuits, Dedicated Transport shall at a minimum meet the performance, availability, jitter, and delay requirements specified for CI to CO connections in the applicable industry standards.
- 6.2.2.4 BellSouth shall offer the following interface transmission rates for Dedicated Transport:
- 6.2.2.4.1 DS0 Equivalent;
- 6.2.2.4.2 DS1;
- 6.2.2.4.3 DS3; and
- 6.2.2.4.4 SDH (Synchronous Digital Hierarchy) Standard interface rates in accordance with International Telecommunications Union (ITU) Recommendation G.707 and Plesiochronous Digital Hierarchy (PDH) rates per ITU Recommendation G.704.
- 6.2.2.5 BellSouth shall design Dedicated Transport according to its network infrastructure. Knology shall specify the termination points for Dedicated Transport.

- 6.2.2.6 At a minimum, Dedicated Transport shall meet each of the requirements set forth in the applicable industry technical references.
- 6.2.2.7 BellSouth Technical References:
- 6.2.2.7.1 TR-TSY-000191 Alarm Indication Signals Requirements and Objectives, Issue 1, May 1986.
- 6.2.2.7.2 TR 73501 LightGate[®] Service Interface and Performance Specifications, Issue D, June 1995.
- 6.2.2.7.3 TR 73525 MegaLink[®] Service, MegaLink Channel Service and MegaLink Plus Service Interface and Performance Specifications, Issue C, May 1996.

6.3 Unbundled Channelization (Multiplexing)

- 6.3.1 Unbundled Channelization (UC) provides the multiplexing capability that will allow a DS1 (1.544 Mbps) or DS3 (44.736 Mbps) or STS-1 (51.84 Mbps) Unbundled Network Element (UNE) or collocation cross-connect to be multiplexed or channelized at a BellSouth central office. Channelization will be offered with both the high and low speed sides to be connected to collocation. Channelization can be accomplished through the use of a stand-alone multiplexer or a digital cross-connect system at the discretion of BellSouth. Once UC has been installed, Knology may request channel activation on an as-needed basis and BellSouth shall connect the requested facilities via Central Office Channel Interfaces (COCIs). The COCI must be compatible with the lower capacity facility and ordered with the lower capacity facility.
- 6.3.2 BellSouth shall make available the following channelization systems and COCIs:
- 6.3.2.1 DS3/STS-1 Channelization System: channelizes a DS3 signal into 28 DS1s.
- 6.3.2.2 DS1 COCI, which can be activated on a DS3 Channelization System.
- 6.3.2.3 DS1 Channelization System: channelizes a DS1 signal into 24 DS0s.
- 6.3.2.4 Voice Grade, Digital Data and ISDN can be activated on a DS1 Channelization System through the use of a COCI.
- 6.3.2.5 Data COCI, which can be activated on a DS1 Channelization System.
- 6.3.2.6 AMI and B8ZS line coding with either Super Frame (SF) and Extended Super Frame (ESF) framing formats will be supported as an optional feature on DS1 facilities.
- 6.3.3 Technical Requirements

- 6.3.3.1 In order to assure proper operation with BellSouth provided central office multiplexing functionality, Knology's channelization equipment must adhere strictly to form and protocol standards. Knology must also adhere to such applicable industry standards for the multiplex channel bank, for voice frequency encoding, for various signaling schemes, and for sub rate digital access.
- 6.3.3.2 DS0 to DS1 Channelization
- 6.3.3.2.1 The DS1 signal must be framed utilizing the framing structure defined in ANSI T1.107, Digital Hierarchy Formats Specifications and ANSI T1.403.02, DS1 Robbed-bit Signaling State Definitions.
- 6.3.3.3 DS1 to DS3 Channelization
- 6.3.3.3.1 The DS3 signal must be framed utilizing the framing structure define in ANSI T1.107, Digital Hierarchy Formats Specifications. The asynchronous M13 multiplex format (combination of M12 and M23 formats) is specified for terminal equipment that multiplexes 28 DS1s into a DS3.
- 6.3.3.4 DS1 to STS Channelization
- 6.3.3.4.1 The STS-1 signal must be framed utilizing the framing structure define in ANSI T1.105, Synchronous Optical Network (SONET) Basic Description Including Multiplex Structure, Rates and Formats and T1.105.02, Synchronous Optical Network (SONET) Payload Mappings.

6.4 Dark Fiber Transport

- 6.4.1 Dark Fiber Transport is an unused optical transmission facility without attached signal regeneration, multiplexing, aggregation or other electronics. Dark Fiber Transport is offered in two configurations: Interoffice Channel, between Knology's collocation arrangement within the POP serving wire center and the end user service wire center and Local Channel, from Knology's POP to Knology's collocation arrangement in the POP serving wire center. It may be strands of optical fiber existing in aerial or underground structure. BellSouth will not provide line terminating elements, regeneration or other electronics necessary for Knology to utilize Dark Fiber Transport.
- 6.4.2 Requirements
- 6.4.2.1 BellSouth shall make available Dark Fiber Transport where it exists in BellSouth's network and where, as a result of future building or deployment, it becomes available. Dark Fiber Transport will not be deemed available if (1) it is used by BellSouth for maintenance and repair purposes, (2) it is designated for use pursuant to a firm order placed by another customer, (3) it is restricted for use by all carriers, including BellSouth, because of transmission problems or because it is scheduled for removal due to documented changes to roads and infrastructure, or

(4) BellSouth has plans to use the fiber within a two-year planning period. BellSouth is not required to place fibers for Dark Fiber Transport if there are none available.

- 6.4.2.2 Knology is solely responsible for testing the quality of the Dark Fiber Transport to determine its usability and performance specifications.
- 6.4.2.3 BellSouth shall use its best efforts to provide to Knology information regarding the location, availability and performance of Dark Fiber Transport within ten (10) business days after receiving a request from Knology. Within such time period, BellSouth shall send written confirmation of availability of the Dark Fiber Transport.
- 6.4.2.4 If the requested Dark Fiber Transport is available, BellSouth shall use its commercially reasonable efforts to provision the Dark Fiber Transport to Knology within twenty (20) business days after Knology submits a valid, error free LSR. Provisioning includes identification of appropriate connection points (e.g., Light Guide Interconnection (LGX)) to enable Knology to connect Knology provided transmission media (e.g., optical fiber) or equipment to the Dark Fiber Transport.

7 BellSouth Switched Access ("SWA") 8XX Toll Free Dialing Ten Digit Screening Service

- 7.1 The BellSouth SWA 8XX Toll Free Dialing Ten Digit Screening Service database ("8XX SCP Database") is a Signaling control Point ("SCP") that contains customer record information and the functionality to provide call-handling instructions for 8XX calls. The 8XX SCP IN software stores data downloaded from the national SMS/8XX database and provides the routing instructions in response to queries from the Switching Service Point ("SSP") or tandem. The BellSouth SWA 8XX Toll Free Dialing Ten Digit Screening Service ("8XX TFD Service") utilizes the 8XX SCP Database to provide identification and routing of the 8XX calls, based on the ten digits dialed. At Knology's option, 8XX TFD Service is provided with or without POTS number delivery, dialing number delivery, and other optional complex features as selected by Knology.
- 7.2 The 8XX SCP Database is designated to receive and respond to queries using the ANSI Specification of Signaling System Seven (SS7) protocol.

8 Line Information Database (LIDB)

8.1 The Line Information Database (LIDB) is a transaction-oriented database accessible through Common Channel Signaling (CCS) networks. For access to LIDB, Knology must purchase appropriate signaling links pursuant to Section 9 of this Attachment. LIDB contains records associated with end user Line Numbers and Special Billing Numbers. LIDB accepts queries from other Network Elements and provides appropriate responses. The query originator need not be the owner
of LIDB data. LIDB queries include functions such as screening billed numbers that provides the ability to accept Collect or Third Number Billing calls and validation of Telephone Line Number based non-proprietary calling cards. The interface for the LIDB functionality is the interface between BellSouth's CCS network and other CCS networks. LIDB also interfaces to administrative systems.

- 8.2 Technical Requirements
- 8.2.1 BellSouth will offer to Knology any additional capabilities that are developed for LIDB during the life of this Agreement.
- 8.2.2 BellSouth shall process Knology's customer records in LIDB at least at parity with BellSouth customer records, with respect to other LIDB functions. BellSouth shall indicate to Knology what additional functions (if any) are performed by LIDB in the BellSouth network.
- 8.2.3 Within two (2) weeks after a request by Knology, BellSouth shall provide Knology with a list of the customer data items, which Knology would have to provide in order to support each required LIDB function. The list shall indicate which data items are essential to LIDB function and which are required only to support certain services. For each data item, the list shall show the data formats, the acceptable values of the data item and the meaning of those values.
- 8.2.4 BellSouth shall provide LIDB systems for which operating deficiencies that would result in calls being blocked shall not exceed 30 minutes per year.
- 8.2.5 BellSouth shall provide LIDB systems for which operating deficiencies that would not result in calls being blocked shall not exceed 12 hours per year.
- 8.2.6 BellSouth shall provide LIDB systems for which the LIDB function shall be in overload no more than 12 hours per year.
- 8.2.7 All additions, updates and deletions of Knology data to the LIDB shall be solely at the direction of Knology. Such direction from Knology will not be required where the addition, update or deletion is necessary to perform standard fraud control measures (e.g., calling card auto-deactivation).
- 8.2.8 BellSouth shall provide priority updates to LIDB for Knology data upon Knology's request (e.g., to support fraud detection), via password-protected telephone card, facsimile, or electronic mail within one hour of notice from the established BellSouth contact.
- 8.2.9 BellSouth shall provide LIDB systems such that no more than 0.01% of Knology customer records will be missing from LIDB, as measured by Knology audits. BellSouth will audit Knology records in LIDB against DBAS to identify record mismatches and provide this data to a designated Knology contact person to resolve the status of the records and BellSouth will update system appropriately.

BellSouth will refer record of mis-matches to Knology within one business day of audit. Once reconciled records are received back from Knology, BellSouth will update LIDB the same business day if less than 500 records are received before 1:00PM Central Time. If more than 500 records are received, BellSouth will contact Knology to negotiate a time frame for the updates, not to exceed three business days.

- 8.2.10 BellSouth shall perform backup and recovery of all of Knology's data in LIDB including sending to LIDB all changes made since the date of the most recent backup copy, in at least the same time frame BellSouth performs backup and recovery of BellSouth data in LIDB for itself. Currently, BellSouth performs backups of the LIDB for itself on a weekly basis; and when a new software release is scheduled, a backup is performed prior to loading the new release.
- 8.2.11 BellSouth shall provide Knology with LIDB reports of data which are missing or contain errors, as well as any misrouted errors, within a reasonable time period as negotiated between Knology and BellSouth.
- 8.2.12 BellSouth shall prevent any access to or use of Knology data in LIDB by BellSouth personnel that are outside of established administrative and fraud control personnel, or by any other Party that is not authorized by Knology in writing.
- 8.2.13 BellSouth shall provide Knology performance of the LIDB Data Screening function, which allows a LIDB to completely or partially deny specific query originators access to LIDB data owned by specific data owners, for Customer Data that is part of an NPA-NXX or RAO-0/1XX wholly or partially owned by Knology at least at parity with BellSouth Customer Data. BellSouth shall obtain from Knology the screening information associated with LIDB Data Screening of Knology data in accordance with this requirement. BellSouth currently does not have LIDB Data Screening capabilities. When such capability is available, BellSouth shall offer it to Knology under the BFR/NBR process as set forth in Attachment 11.
- 8.2.14 BellSouth shall accept queries to LIDB associated with Knology customer records and shall return responses in accordance with industry standards.
- 8.2.15 BellSouth shall provide mean processing time at the LIDB within 0.50 seconds under normal conditions as defined in industry standards.
- 8.2.16 BellSouth shall provide processing time at the LIDB within 1 second for 99% of all messages under normal conditions as defined in industry standards.
- 8.3 Interface Requirements
- 8.3.1 BellSouth shall offer LIDB in accordance with the requirements of this subsection.

- 8.3.2 The interface to LIDB shall be in accordance with the technical references contained within.
- 8.3.3 The CCS interface to LIDB shall be the standard interface described herein.
- 8.3.4 The LIDB Data Base interpretation of the ANSI-TCAP messages shall comply with the technical reference herein. Global Title Translation shall be maintained in the signaling network in order to support signaling network routing to the LIDB.
- 8.3.5 The application of the LIDB rates contained in Exhibit B to this Attachment will be based on a Percent CLEC LIDB Usage ("PCLU") factor. Knology shall provide BellSouth a PCLU. The PCLU will be applied to determine the percentage of total LIDB usage to be billed to the other Party at local rates. Knology shall update its PCLU on the first of January, April, July and October and shall send it to BellSouth to be received no later than thirty (30) calendar days after the first of each such month based on local usage for the past three months ending the last day of December, March, June and September, respectively. Requirements associated with PCLU calculation and reporting shall be as set forth in BellSouth's Jurisdictional Factors Reporting Guide, as it is amended from time to time.

9 Signaling

9.1 BellSouth shall offer access to signaling and access to BellSouth's signaling databases subject to compatibility testing and at the rates set forth in this Attachment. BellSouth may provide mediated access to BellSouth signaling systems and databases. Available signaling elements include signaling links, signal transfer points and service control points. Signaling functionality will be available with both A-link and B-link connectivity.

9.2 Signaling Link Transport

- 9.2.1 Signaling Link Transport is a set of two or four dedicated 56 kbps transmission paths between Knology-designated Signaling Points of Interconnection that provide appropriate physical diversity.
- 9.2.2 Technical Requirements
- 9.2.3 Signaling Link Transport shall consist of full duplex mode 56 kbps transmission paths and shall perform in the following two ways:
- 9.2.3.1 As an "A-link" Signaling Link Transport is a connection between a switch or SCP and a home Signaling Transfer Point switch pair; and
- 9.2.3.2 As a "B-link" Signaling Link Transport is a connection between two Signaling Transfer Point switch pairs in different company networks (e.g., between two Signaling Transfer Point switch pairs for two CLECs).

- 9.2.4 Signaling Link Transport shall consist of two or more signaling link layers as follows:
- 9.2.4.1 An A-link layer shall consist of two links.
- 9.2.4.2 A B-link layer shall consist of four links.
- 9.2.4.3 A signaling link layer shall satisfy interoffice and intraoffice diversity of facilities and equipment, such that:
- 9.2.4.4 No single failure of facilities or equipment causes the failure of both links in an Alink layer (i.e., the links should be provided on a minimum of two separate physical paths end-to-end); and
- 9.2.4.5 No two concurrent failures of facilities or equipment shall cause the failure of all four links in a B-link layer (i.e., the links should be provided on a minimum of three separate physical paths end-to-end).
- 9.2.5 Interface Requirements
- 9.2.5.1 There shall be a DS1 (1.544 Mbps) interface at Knology's designated SPOIs. Each 56 kbps transmission path shall appear as a DS0 channel within the DS1 interface.

9.3 Signaling Transfer Points (STPs)

- 9.3.1 A Signaling Transfer Point is a signaling network function that includes all of the capabilities provided by the signaling transfer point switches (STPs) and their associated signaling links that enables the exchange of SS7 messages among and between switching elements, database elements and signaling transfer point switches.
- 9.3.2 Technical Requirements
- 9.3.2.1 Signaling Transfer Point s shall provide access to BellSouth Local Switching or Tandem Switching and to BellSouth Service Control Points/Databases connected to BellSouth SS7 network. Signaling Transfer Point also provide access to thirdparty local or tandem switching and Third-party-provided Signaling Transfer Points.
- 9.3.2.2 The connectivity provided by Signaling Transfer Points shall fully support the functions of all other Network Elements connected to the BellSouth SS7 network. This includes the use of the BellSouth SS7 network to convey messages that neither originate nor terminate at a signaling end point directly connected to the BellSouth SS7 network (i.e., transit messages). When the BellSouth SS7 network is used to convey transit messages, there shall be no alteration of the Integrated

Services Digital Network User Part or Transaction Capabilities Application Part (TCAP) user data that constitutes the content of the message.

- 9.3.2.3 If a BellSouth tandem switch routes traffic, based on dialed or translated digits, on SS7 trunks between a Knology local switch and third party local switch, the BellSouth SS7 network shall convey the TCAP messages that are necessary to provide Call Management features (Automatic Callback, Automatic Recall, and Screening List Editing) between Knology local STPs and the STPs that provide connectivity with the third party local switch, even if the third party local switch is not directly connected to BellSouth STPs.
- 9.3.2.4 STPs shall provide all functions of the SCCP necessary for Class 0 (basic connectionless) service as defined in Telcordia ANSI Interconnection Requirements. This includes Global Title Translation (GTT) and SCCP Management procedures, as specified in ANSI T1.112.4. Where the destination signaling point is a Knology or third party local or tandem switching system directly connected to BellSouth SS7 network, BellSouth shall perform final GTT of messages to the destination and SCCP Subsystem Management of the destination. In all other cases, BellSouth shall perform intermediate GTT of messages to a gateway pair of STPs in an SS7 network connected with BellSouth SS7 network and shall not perform SCCP Subsystem Management of the destination. If BellSouth performs final GTT to a Knology database, then Knology agrees to provide BellSouth with the Destination Point Code for Knology database.
- 9.3.2.5 STPs shall provide all functions of the OMAP as specified in applicable industry standard technical references, which may include, where available in BellSouth's network, MTP Routing Verification Test (MRVT) and SCCP Routing Verification Test (SRVT).
- 9.3.2.6 Where the destination signaling point is a BellSouth local or tandem switching system or database, or is a Knology or third party local or tandem switching system directly connected to the BellSouth SS7 network, STPs shall perform MRVT and SRVT to the destination signaling point. In all other cases, STPs shall perform MRVT and SRVT to a gateway pair of STPs in an SS7 network connected with the BellSouth SS7 network. This requirement may be superseded by the specifications for Internetwork MRVT and SRVT when these become approved ANSI standards and available capabilities of BellSouth STPs.

9.4 SS7 Advanced Intelligent Network (AIN) Access

9.4.1 When technically feasible and upon request by Knology, SS7 AIN Access shall be made available in association with switching. SS7 AIN Access is the provisioning of AIN 0.1 triggers in an equipped BellSouth local switch and interconnection of the BellSouth SS7 network with Knology's SS7 network to exchange TCAP queries and responses with a Knology SCP.

9.4.2	SS7 AIN Access shall provide Knology SCP access to an equipped BellSouth
	local switch via interconnection of BellSouth's SS7 and Knology SS7 Networks.
	BellSouth shall offer SS7 AIN Access through its STPs. If BellSouth requires a
	mediation device on any part of its network specific to this form of access,
	BellSouth must route its messages in the same manner. The interconnection
	arrangement shall result in the BellSouth local switch recognizing the Knology
	SCP as at least at parity with BellSouth's SCPs in terms of interfaces, performance
	and capabilities.

- 9.4.3 Interface Requirements
- 9.4.3.1 BellSouth shall provide the following STP options to connect Knology or Knology-designated local switching systems to the BellSouth SS7 network:
- 9.4.3.1.1 An A-link interface from Knology local switching systems; and,
- 9.4.3.1.2 A B-link interface from Knology local STPs.
- 9.4.3.2 Each type of interface shall be provided by one or more layers of signaling links.
- 9.4.3.3 The Signaling Point of Interconnection for each link shall be located at a crossconnect element in the Central Office (CO) where the BellSouth STP is located. There shall be a DS1 or higher rate transport interface at each of the SPOIs. Each signaling link shall appear as a DS0 channel within the DS1 or higher rate interface.
- 9.4.3.4 BellSouth shall provide intraoffice diversity between the Signaling Point of Interconnection and BellSouth STPs so that no single failure of intraoffice facilities or equipment shall cause the failure of both B-links in a layer connecting to a BellSouth STP.
- 9.4.3.5 STPs shall provide all functions of the MTP as defined in the applicable industry standard technical references.
- 9.4.4 Message Screening
- 9.4.4.1 BellSouth shall set message screening parameters so as to accept valid messages from Knology local or tandem switching systems destined to any signaling point within BellSouth's SS7 network where the Knology switching system has a valid signaling relationship.
- 9.4.4.2 BellSouth shall set message screening parameters so as to pass valid messages from Knology local or tandem switching systems destined to any signaling point or network accessed through BellSouth's SS7 network where the Knology switching system has a valid signaling relationship.

9.4.4.3 BellSouth shall set message screening parameters so as to accept and pass/send valid messages destined to and from Knology from any signaling point or network interconnected through BellSouth's SS7 network where the Knology SCP has a valid signaling relationship.

9.5 Service Control Points/Databases

- 9.5.1 Call Related Databases provide the storage of, access to, and manipulation of information required to offer a particular service and/or capability. BellSouth shall provide access to the following Databases: Local Number Portability, LIDB, Toll Free Number Database, Automatic Location Identification/Data Management System, and Calling Name Database. BellSouth also provides access to Service Creation Environment and Service Management System (SCE/SMS) application databases and Directory Assistance.
- 9.5.2 A Service Control Point (SCP) is deployed in a SS7 network that executes service application logic in response to SS7 queries sent to it by a switching system also connected to the SS7 network. Service Management Systems provide operational interfaces to allow for provisioning, administration and maintenance of subscriber data and service application data stored in SCPs.
- 9.5.3 Technical Requirements for SCPs/Databases
- 9.5.3.1 BellSouth shall provide physical access to SCPs through the SS7 network and protocols with TCAP as the application layer protocol.
- 9.5.3.2 BellSouth shall provide physical interconnection to databases via industry standard interfaces and protocols (e.g. SS7, ISDN and X.25).
- 9.5.3.3 The reliability of interconnection options shall be consistent with requirements for diversity and survivability.

9.6 Local Number Portability Database

9.6.1 The Permanent Number Portability (PNP) database supplies routing numbers for calls involving numbers that have been ported from one local service provider to another. BellSouth agrees to provide access to the PNP database at rates, terms and conditions as set forth by BellSouth and in accordance with an effective FCC or Commission directive.

9.7 SS7 Network Interconnection

9.7.1 SS7 Network Interconnection is the interconnection of Knology local signaling transfer point switches or Knology local or tandem switching systems with BellSouth signaling transfer point switches. This interconnection provides connectivity that enables the exchange of SS7 messages among BellSouth switching systems and databases, Knology local or tandem switching systems, and other third-party switching systems directly connected to the BellSouth SS7 network.

- 9.7.2 The connectivity provided by SS7 Network Interconnection shall fully support the functions of BellSouth switching systems and databases and Knology or other third-party switching systems with A-link access to the BellSouth SS7 network.
- 9.7.3 If traffic is routed based on dialed or translated digits between a Knology local switching system and a BellSouth or other third-party local switching system, either directly or via a BellSouth tandem switching system, then it is a requirement that the BellSouth SS7 network convey via SS7 Network Interconnection the TCAP messages that are necessary to provide Call Management services (Automatic Callback, Automatic Recall, and Screening List Editing) between the Knology local signaling transfer point switches and BellSouth or other third-party local switch.
- 9.7.4 SS7 Network Interconnection shall provide:
- 9.7.4.1 Signaling Data Link functions, as specified in ANSI T1.111.2;
- 9.7.4.2 Signaling Link functions, as specified in ANSI T1.111.3; and
- 9.7.4.3 Signaling Network Management functions, as specified in ANSI T1.111.4.
- 9.7.5 SS7 Network Interconnection shall provide all functions of the SCCP necessary for Class 0 (basic connectionless) service as specified in ANSI T1.112. This includes Global Title Translation (GTT) and SCCP Management procedures as specified in ANSI T1.112.4. Where the destination signaling point is a BellSouth switching system or DB, or is another third-party local or tandem switching system directly connected to the BellSouth SS7 network, SS7 Network Interconnection shall include final GTT of messages to the destination and SCCP Subsystem Management of the destination. Where the destination signaling point is a Knology local or tandem switching system, SS7 Network Interconnection shall include intermediate GTT of messages to a gateway pair of Knology local STPs and shall not include SCCP Subsystem Management of the destination.
- 9.7.6 SS7 Network Interconnection shall provide all functions of the Integrated Services Digital Network User Part as specified in ANSI T1.113.
- 9.7.7 SS7 Network Interconnection shall provide all functions of the TCAP as specified in ANSI T1.114.
- 9.7.8 If Internetwork MRVT and SRVT become approved ANSI standards and available capabilities of BellSouth STPs, SS7 Network Interconnection may provide these functions of the OMAP.
- 9.7.9 Interface Requirements

- 9.7.9.1 The following SS7 Network Interconnection interface options are available to connect Knology or Knology-designated local or tandem switching systems or signaling transfer point switches to the BellSouth SS7 network:
- 9.7.9.1.1 A-link interface from Knology local or tandem switching systems; and
- 9.7.9.1.2 B-link interface from Knology STPs.
- 9.7.9.2 The Signaling Point of Interconnection for each link shall be located at a crossconnect element in the central office where the BellSouth STP is located. There shall be a DS1 or higher rate transport interface at each of the Signaling Points of interconnection. Each signaling link shall appear as a DS0 channel within the DS1 or higher rate interface.
- 9.7.9.3 BellSouth shall provide intraoffice diversity between the Signaling Points of Interconnection and the BellSouth STP, so that no single failure of intraoffice facilities or equipment shall cause the failure of both B-links in a layer connecting to a BellSouth STP.
- 9.7.9.4 The protocol interface requirements for SS7 Network Interconnection include the MTP, ISDNUP, SCCP, and TCAP. These protocol interfaces shall conform to the applicable industry standard technical references.
- 9.7.9.5 BellSouth shall set message screening parameters to accept messages from Knology local or tandem switching systems destined to any signaling point in the BellSouth SS7 network with which the Knology switching system has a valid signaling relationship.

10 Operator Services (Operator Call Processing and Directory Assistance)

- 10.1 Operator Call Processing provides: (1) operator handling for call completion (for example, collect, third number billing, and manual calling-card calls); (2) operator or automated assistance for billing after the end user has dialed the called number (for example, calling card calls); and (3) special services including but not limited to Busy Line Verification and Emergency Line Interrupt (ELI), Emergency Agency Call, and Operator-assisted Directory Assistance.
- 10.2 Upon request for BellSouth Operator Call Processing, BellSouth shall:
- 10.2.1 Process 0+ and 0- dialed local calls.
- 10.2.2 Process 0+ and 0- intraLATA toll calls.
- 10.2.3 Process calls that are billed to Knology end user's calling card that can be validated by BellSouth.
- 10.2.4 Process person-to-person calls.

10.2.5 Process collect calls.

- 10.2.6 Provide the capability for callers to bill to a third party and shall also process such calls.
- 10.2.7 Process station-to-station calls.
- 10.2.8 Process Busy Line Verify and Emergency Line Interrupt requests.
- 10.2.9 Process emergency call trace originated by Public Safety Answering Points.
- 10.2.10 Process operator-assisted directory assistance calls.
- 10.2.11 Adhere to equal access requirements, providing Knology local end users the same IXC access as provided to BellSouth end users.
- 10.2.12 Exercise at least the same level of fraud control in providing Operator Service to Knology that BellSouth provides for its own operator service.
- 10.2.13 Perform Billed Number Screening when handling Collect, Person-to-Person, and Billed-to-Third-Party calls.
- 10.2.14 Direct customer account and other similar inquiries to the customer service center designated by Knology.
- 10.2.15 Provide call records to Knology in accordance with ODUF standards specified in Attachment 7.
- 10.2.16 The interface requirements shall conform to the interface specifications for the platform used to provide Operator Services as long as the interface conforms to industry standards.

10.3 Directory Assistance Service

- 10.3.1 Directory Assistance Service provides local and non-local end user telephone number listings with the option to complete the call at the caller's direction separate and distinct from local switching.
- 10.3.2 Directory Assistance Service shall provide up to two listing requests per call. If available and if requested by Knology's end user, BellSouth shall provide calleroptional directory assistance call completion service at rates contained in this Attachment to one of the provided listings.

10.3.3 Directory Assistance Service Updates

- 10.3.3.1 BellSouth shall update end user listings changes daily. These changes include:
- 10.3.3.1.1 New end user connections;

- 10.3.3.1.2 End user disconnections;
- 10.3.3.1.3 End user address changes.
- 10.3.3.2 These updates shall also be provided for non-listed and non-published numbers for use in emergencies.

10.4 Branding for Operator Call Processing and Directory Assistance

- 10.4.1 BellSouth's branding feature provides a definable announcement to Knology end users using Directory Assistance (DA)/Operator Call Processing (OCP) prior to placing such end users in queue or connecting them to an available operator or automated operator system. This feature allows Knology to have its calls custom branded with Knology's name on whose behalf BellSouth is providing Directory Assistance and/or Operator Call Processing. Rates for the branding features are set forth in this Attachment.
- 10.4.2 BellSouth offers three branding offering options to Knology when ordering BellSouth's Directory Assistance and Operator Call Processing: BellSouth Branding, Unbranding and Custom Branding.
- 10.4.3 Upon receipt of the custom branding order from Knology, the order is considered firm after ten business days. Should Knology decide to cancel the order, written notification to Knology's Local Contract Manager is required. If Knology decides to cancel after ten business days from receipt of the custom branding order, Knology shall pay all charges per the order.

10.4.4 Selective Call Routing Using Line Class Codes (SCR-LCC)

- 10.4.4.1 Where Knology purchases unbundled local switching from BellSouth and utilizes an Operator Services Provider other than BellSouth, BellSouth will route Knology's end user calls to that provider through Selective Call Routing.
- 10.4.4.2 Selective Call Routing using Line Class Codes (SCR-LCC) provides the capability for Knology to have its OCP/DA calls routed to BellSouth's OCP/DA platform for BellSouth provided Custom Branded or Unbranded OCP/DA or to its own or an alternate OCP/DA platform for Self-Branded OCP/DA. SCR-LCC is only available if line class code capacity is available in the requested BellSouth end office switches.
- 10.4.4.3 Custom Branding for Directory Assistance is not available for certain classes of service, including but not limited to Hotel/Motel services, WATS service, and certain PBX services.
- 10.4.4.4 Where available, Knology specific and unique line class codes are programmed in each BellSouth end office switch where Knology intends to serve end users with customized OCP/DA branding. The line class codes specifically identify Knology's

end users so OCP/DA calls can be routed over the appropriate trunk group to the requested OCP/DA platform. Additional line class codes are required in each end office if the end office serves multiple NPAs (i.e., a unique LCC is required per NPA), and/or if the end office switch serves multiple rate areas and Knology intends to provide Knology -branded OCP/DA to its end users in these multiple rate areas.

- 10.4.4.5 BellSouth Branding is the default branding offering.
- 10.4.4.6 SCR-LCC supporting Custom Branding and Self Branding require Knology to order dedicated trunking from each BellSouth end office identified by Knology, either to the BellSouth Traffic Operator Position System (TOPS) for Custom Branding or to the Knology Operator Service Provider for Self Branding. Separate trunk groups are required for Operator Services and for Directory Assistance. Rates for trunks are set forth in applicable BellSouth tariffs.
- 10.4.4.7 Unbranding Unbranded Directory Assistance and/or Operator Call Processing calls ride common trunk groups provisioned by BellSouth from those end offices identified by Knology to the BellSouth TOPS. These calls are routed to "No Announcement."
- 10.4.4.8 The Rates for SCR-LCC are as set forth in this Attachment. There is a nonrecurring charge for the establishment of each Line Class Code in each BellSouth central office. Furthermore, for Unbranded and Custom Branded OCP/DA provided by BellSouth Operator Services with unbundled ports and unbundled port/loop switch combinations, monthly recurring usage charges shall apply for the UNEs necessary to provide the service, such as end office and tandem switching and common transport. A flat rated end office switching charge shall apply to Self-Branded OCP/DA when used in conjunction with unbundled ports and unbundled port/loop switch combinations.
- 10.4.4.9 UNE Provider Branding via Originating Line Number Screening (OLNS)
- 10.4.4.10 BellSouth Branding, Unbranding and Custom Branding are also available for Directory Assistance, Operator Call Processing or both via Originating Line Number Screening (OLNS) software. When utilizing this method of Unbranding or Custom Branding, Knology shall not be required to purchase dedicated trunking.
- 10.4.4.11 For BellSouth to provide Unbranding or Custom Branding via OLNS software for Operator Call Processing or for Directory Assistance, Knology must have its Operating Company Number ("OCN(s)") and telephone numbers reside in BellSouth's LIDB; however, a BellSouth LIDB Storage Agreement is not required. To implement Unbranding and Custom Branding via OLNS software, Knology must submit a manual order form which requires, among other things, Knology's OCN and a forecast for the traffic volume anticipated for each

BellSouth TOPS during the peak busy hour. Knology shall provide updates to such forecast on a quarterly basis and at any time such forecasted traffic volumes are expected to change significantly. Upon Knology's purchase of Unbranding or Custom Branding using OLNS software for any particular TOPS, all Knology end users served by that TOPS will receive the Unbranded "no announcement" or the Custom Branded announcement.

- 10.4.4.12 BellSouth Branding is the default branding offering.
- 10.4.4.13 Rates for Unbranding and Custom Branding via OLNS software for Directory Assistance and for Operator Call Processing are as set forth in this Attachment. Notwithstanding anything to the contrary in this Agreement, to the extent BellSouth is unable to bill Knology applicable charges currently, BellSouth shall track such charges and will bill the same retroactively at such time as a billing process is implemented. In addition to the charges for Unbranding and Custom Branding via OLNS software, Knology shall continue to pay BellSouth applicable labor and other charges for the use of BellSouth's Directory Assistance and Operator Call Processing platforms as set forth in this Attachment. Further, where Knology is purchasing unbundled local switching from BellSouth, UNE usage charges for end office switching, tandem switching and transport, as applicable, shall continue to apply.

10.4.5 Facilities Based Carrier Branding

- 10.4.5.1 All Service Levels require Knology to order dedicated trunking from their end office(s) point of interface to the BellSouth TOPS Switches. Rates for trunks are set forth in applicable BellSouth tariffs.
- 10.4.5.2 Unbranding is the default branding offering.
- 10.4.5.3 Rates for Custom Branded OCP/DA are set forth in this Attachment.
- 10.4.5.4 Customized Branding includes charges for the recording of the branding announcement and the loading of the audio units in each TOPS Switch and Network Applications Vehicle (NAV) equipment for which Knology requires service.
- 10.4.5.5 Directory Assistance customized branding uses:
- 10.4.5.5.1 the recording of Knology;
- 10.4.5.5.2 the loading of the recording in each switch.
- 10.4.5.6 Operator Call Processing customized branding uses:
- 10.4.5.6.1 the recording of Knology;

- 10.4.5.6.2 the loading of the recording in each switch (North Carolina);
- 10.4.5.6.3 the loading on the Network Applications Vehicle (NAV). All NAV shelves within the region where the customer is offering service must be loaded.

10.5 Directory Assistance Database Service (DADS)

- 10.5.1 BellSouth shall make its Directory Assistance Database Service (DADS) available at the rates set forth in this Attachment solely for the expressed purpose of providing Directory Assistance type services to Knology end users. The term "end user" denotes any entity that obtains Directory Assistance type services for its own use from a DADS customer. Directory Assistance type service is defined as Voice Directory Assistance (DA Operator assisted) and Electronic Directory Assistance (Data System assisted). Knology agrees that DADS will not be used for any purpose that violates federal or state laws, statutes, regulatory orders or tariffs. For the purposes of provisioning a Directory Assistance type service, all terms and conditions of GSST A38 apply and are incorporated by reference herein. Except for the permitted uses, Knology agrees not to disclose DADS to others and shall provide due care in providing for the security and confidentiality of DADS.
- 10.5.2 BellSouth shall initially provide Knology with a Base File of subscriber listings via magnetic tape. DADS is available and may be ordered on a Business, Residence or combined Business and Residence listings basis for each central office requested. BellSouth will require approximately 30-45 days after receiving an order from Knology to prepare the Base File.
- 10.5.3 BellSouth will provide updates on either a daily or weekly basis reflecting all listing change activity occurring since Knology's previous update. Delivery of updates will commence immediately after Knology receives the Base File. Updates will be provided via magnetic tape unless BellSouth and Knology mutually develop CONNECT: Direct TM electronic connectivity. Knology will pay all costs associated with CONNECT: Direct TM connectivity, which will vary depending upon volume and mileage.
- 10.5.4 Knology authorizes the inclusion of Knology Directory Assistance listings in the BellSouth Directory Assistance products including but not limited to DADS. Any other use is not authorized.

10.6 Direct Access to Directory Assistance Service

10.6.1 Direct Access to Directory Assistance Service (DADAS) will provide Knology's directory assistance operators with the ability to search, using a standard directory assistance search format, the same listing information that is available to BellSouth operators including all available BellSouth subscriber listings, all available listings associated with lines resold by competitive local exchange carriers, and all available listings associated with lines provisioned by local exchange carriers that

provide their listings to BellSouth. DADAS will also provide Knology with the ability to search all listings BellSouth obtains from sources other than the provider of the local exchange lines associated with the listings. The search format will be provided to Knology by BellSouth upon subscription to the service. Subscription to DADAS requires that Knology utilize its own switch, operator workstations, directory assistance operators, transport facilities, and optional audio subsystems.

10.6.2 Rates, terms and conditions for provisioning DADAS are as set forth in the FCC tariff No. 1.

11 Automatic Location Identification/Data Management System (ALI/DMS)

- 11.1 The ALI/DMS Database contains end user information (including name, address, telephone information, and sometimes special information from the local service provider or end user) used to determine to which Public Safety Answering Point ("PSAP") to route the call. The ALI/DMS database is used to provide enhanced routing flexibility for E911.
- 11.2 Technical Requirements
- 11.2.1 BellSouth shall provide Knology access to the ALI/DMS database. BellSouth shall provide error reports from the ALI/DMS database to Knology after Knology provides end user information for input into the ALI/DMS database.
- 11.2.2 When BellSouth is responsible for administering the ALI/DMS database in its entirety, ported number NXXs entries for the ported numbers should be maintained unless Knology requests otherwise and shall be updated if Knology requests, provided Knology supplies BellSouth with the updates.
- 11.2.3 When Remote Call Forwarding (RCF) is used to provide number portability to the local end user and a remark or other appropriate field information is available in the database, the shadow or "forwarded-to" number and an indication that the number is ported shall be added to the customer record.
- 11.2.4 If BellSouth is responsible for configuring PSAP features (for cases when the PSAP or BellSouth supports an ISDN interface), it shall ensure that CLASS Automatic Recall (Call Return) is not used to call back to the ported number. Although BellSouth currently does not have ISDN interface, BellSouth agrees to comply with this requirement once ISDN interfaces are in place.
- 11.3 Interface Requirements
- 11.3.1 The interface between the E911 Switch or Tandem and the ALI/DMS database for Knology end users shall meet industry standards.

12 Calling Name (CNAM) Database Service

- 12.1 CNAM is the ability to associate a name with the calling party number, allowing the end user (to which a call is being terminated) to view the calling party's name before the call is answered. This service also provides Knology the opportunity to load and store its subscriber names in the BellSouth CNAM SCPs.
- 12.2 Knology shall submit to BellSouth a notice of its intent to access and utilize BellSouth CNAM Database Services. Said notice shall be in writing no less than 60 days prior to Knology's access to BellSouth's CNAM Database Services and shall be addressed to Knology's Local Contract Manager.
- 12.3 BellSouth's provision of CNAM Database Services to Knology requires interconnection from Knology to BellSouth CNAM Service Control Points (SCPs). Such interconnections shall be established pursuant to Attachment 3 of this Agreement, incorporated herein by this reference.
- 12.4 In order to formulate a CNAM query to be sent to the BellSouth CNAM SCP, Knology shall provide its own CNAM SSP. Knology's CNAM SSPs must be compliant with TR-NWT-001188, "CLASS Calling Name Delivery Generic Requirements".
- 12.5 If Knology elects to access the BellSouth CNAM SCP via a third party CCS7 transport provider, the third party CCS7 provider shall interconnect with the BellSouth CCS7 network according to BellSouth's Common Channel Signaling Interconnection Guidelines and Telcordia's CCS Network Interface Specification document, TR-TSV-000905. In addition, the third party provider shall establish CCS7 interconnection at the BellSouth Local Signal Transfer Points (LSTPs) serving the BellSouth CNAM SCPs that Knology desires to query.
- 12.6 If Knology queries the BellSouth CNAM SCP via a third party national SS7 transport provider, the third party SS7 provider shall interconnect with the BellSouth CCS7 network according to BellSouth's Common Channel Signaling Interconnection Guidelines and Telcordia's CCS Network Interface Specification document, TR-TSV-000905. In addition, the third party provider shall establish SS7 interconnection at one or more of the BellSouth Gateway Signal Transfer Points (STPs). The payment of all costs associated with the transport of SS7 signals via a third party will be established by mutual agreement of the Parties and this Agreement shall be amended in accordance with modification of the General Terms and Conditions incorporated herein by this reference.
- 12.7 The mechanism to be used by Knology for initial CNAM record load and/or updates shall be determined by mutual agreement. The initial load and all updates shall be provided by Knology in the BellSouth specified format and shall contain records for every working telephone number that can originate phone calls. It is the responsibility of Knology to provide accurate information to BellSouth on a current basis.

- 12.8 Updates to the SMS shall occur no less than once a week, reflect service order activity affecting either name or telephone number, and involve only record additions, deletions or changes.
- 12.9 Knology CNAM records provided for storage in the BellSouth CNAM SCP shall be available, on a SCP query basis only, to all Parties querying the BellSouth CNAM SCP. Further, CNAM service shall be provided by each Party consistent with state and/or federal regulation.

13 Service Creation Environment and Service Management System (SCE/SMS) Advanced Intelligent Network (AIN) Access

- 13.1 BellSouth's Service Creation Environment and Service Management System (SCE/SMS) Advanced Intelligent Network (AIN) Access shall provide Knology the capability to create service applications in a BellSouth SCE and deploy those applications in a BellSouth SMS to a BellSouth SCP.
- 13.2 BellSouth's SCE/SMS AIN Access shall provide access to SCE hardware, software, testing and technical support (e.g., help desk, system administrator) resources available to Knology. Training, documentation, and technical support will address use of SCE and SMS access and administrative functions but will not include support for the creation of a specific service application.
- 13.3 BellSouth SCP shall partition and protect Knology service logic and data from unauthorized access.
- 13.4 When Knology selects SCE/SMS AIN Access, BellSouth shall provide training, documentation, and technical support to enable Knology to use BellSouth's SCE/SMS AIN Access to create and administer applications.
- 13.5 Knology access will be provided via remote data connection (e.g., dial-in, ISDN).
- 13.6 BellSouth shall allow Knology to download data forms and/or tables to BellSouth SCP via BellSouth SMS without intervention from BellSouth.

14 Basic 911 and E911

- 14.1 Basic 911 and E911 provides a caller access to the applicable emergency service bureau by dialing 911.
- 14.2 <u>Basic 911 Service Provisioning.</u> BellSouth will provide to Knology a list consisting of each municipality that subscribes to Basic 911 service. The list will also provide, if known, the E911 conversion date for each municipality and, for network routing purposes, a ten-digit directory number representing the appropriate emergency answering position for each municipality subscribing to 911. Knology will be required to arrange to accept 911 calls from its end users in municipalities that subscribe to Basic 911 service and translate the 911 call to the

appropriate 10-digit directory number as stated on the list provided by BellSouth. Knology will be required to route that call to BellSouth at the appropriate tandem or end office. When a municipality converts to E911 service, Knology will be required to begin using E911 procedures.

- 14.3 E911 Service Provisioning. Knology shall install a minimum of two dedicated trunks originating from the Knology serving wire center and terminating to the appropriate E911 tandem. The dedicated trunks shall be, at a minimum, DS0 level trunks configured either as a 2-wire analog interface or as part of a digital (1.544 Mb/s) interface. Either configuration shall use CAMA-type signaling with multifrequency ("MF") pulsing that will deliver automatic number identification ("ANI") with the voice portion of the call. If the user interface is digital, MF pulses as well as other AC signals shall be encoded per the u-255 Law convention. Knology will be required to provide BellSouth daily updates to the E911 database. Knology will be required to forward 911 calls to the appropriate E911 tandem along with ANI based upon the current E911 end office to tandem homing arrangement as provided by BellSouth. If the E911 tandem trunks are not available, Knology will be required to route the call to a designated 7-digit local number residing in the appropriate Public Service Answering Point ("PSAP"). This call will be transported over BellSouth's interoffice network and will not carry the ANI of the calling party. Knology shall be responsible for providing BellSouth with complete and accurate data for submission to the 911/E911 database for the purpose of providing 911/E911 to its end users.
- 14.4 <u>Rates.</u> Charges for 911/E911 service are borne by the municipality purchasing the service. BellSouth will impose no charge on Knology beyond applicable charges for BellSouth trunking arrangements.
- 14.5 Basic 911 and E911 functions provided to Knology shall be at least at parity with the support and services that BellSouth provides to its end users for such similar functionality.
- 14.6 The detailed practices and procedures for 911/E911 services are contained in the E911 Local Exchange Carrier Guide For Facility-Based Providers as amended from time to time during the term of this Agreement.

15 Operational Support Systems (OSS)

15.1 BellSouth has developed and made available the following electronic interfaces by which Knology may submit LSRs electronically.

LENS	Local Exchange Navigation System
EDI	Electronic Data Interchange
TAG	Telecommunications Access Gateway

15.2 LSRs submitted by means of one of these electronic interfaces will incur an OSS electronic ordering charge. An individual LSR will be identified for billing

purposes by its Purchase Order Number (PON). LSRs submitted by means other than one of these interactive interfaces (mail, fax, courier, etc.) will incur a manual order charge. All OSS charges are specified in Rate Exhibit B of this Attachment 2.

- 15.3 Denial/Restoral OSS Charge
- 15.3.1 In the event Knology provides a list of customers to be denied and restored, rather than an LSR, each location on the list will require a separate PON and therefore will be billed as one LSR per location.
- 15.4 Cancellation OSS Charge
- 15.4.1 Knology will incur an OSS charge for an accepted LSR that is later canceled.
- 15.4.2 Supplements or clarifications to a previously billed LSR will not incur another OSS charge.
- 15.4.3 Network Elements and Other Services Manual Additive
- 15.4.4 The Commissions in some states have ordered per-element manual additive nonrecurring charges (NRC) for Network Elements and Other Services ordered by means other than one of the interactive interfaces. These ordered Network Elements and Other Services manual additive NRCs will apply in these states, rather than the charge per LSR. The per-element charges are listed on the Rate Tables in Exhibit B.

EXHIBIT A

LINE INFORMATION DATA BASE (LIDB)

FACILITIES BASED STORAGE AGREEMENT

I. Definitions

- A. Billing number a number that Knology creates for the purpose of identifying an account liable for charges. This number may be a line or a special billing number.
- B. Line number a ten-digit number that identifies a telephone line administered by Knology.
- C. Special billing number a ten-digit number that identifies a billing account established by Knology.
- D. Calling Card number a billing number plus PIN number.
- E. PIN number a four-digit security code assigned by Knology that is added to a billing number to compose a fourteen-digit calling card number.
- F. Toll billing exception indicator associated with a billing number to indicate that it is considered invalid for billing of collect calls or third number calls or both, by Knology.
- G. Billed Number Screening refers to the activity of determining whether a toll billing exception indicator is present for a particular billing number.
- H. Calling Card Validation refers to the activity of determining whether a particular calling card number exists as stated or otherwise provided by a caller.
- I. Billing number information information about billing number, Calling Card number and toll billing exception indicator provided to BellSouth by Knology.

II. General

A. This Agreement sets forth the terms and conditions pursuant to which BellSouth agrees to store in its LIDB certain information at the request of Knology and pursuant to which BellSouth, its LIDB customers and Knology shall have access to such information. In addition, this Agreement sets forth the terms and conditions for Knology's provision of billing number information to BellSouth for inclusion in BellSouth's LIDB. Knology understands that BellSouth provides access to information in its LIDB to various telecommunications service providers pursuant to applicable tariffs and agrees that information stored at the request of Knology, pursuant to this Agreement, shall be available to those telecommunications service providers. The terms and conditions contained herein shall hereby be made a part of this Interconnection Agreement upon notice to Knology's account team and/or Local Contract Manager to activate this LIDB Storage Agreement. The General Terms and

Conditions of the Interconnection/Resale Agreement shall govern this LIDB Storage Agreement.

- B. BellSouth will provide responses to on-line, call-by-call queries to billing number information for the following purposes:
 - 1. Billed Number Screening

BellSouth is authorized to use the billing number information to determine whether Knology has identified the billing number as one that should not be billed for collect or third number calls.

2. Calling Card Validation

BellSouth is authorized to validate a 14-digit Calling Card number where the first 10 digits are a line number or special billing number assigned by BellSouth and where the last four digits (PIN) are a security code assigned by BellSouth.

3. Fraud Control

BellSouth will provide seven days per week, 24-hours per day, fraud monitoring on Calling Cards, bill-to-third and collect calls made to numbers in BellSouth's LIDB, provided that such information is included in the LIDB query. BellSouth will establish fraud alert thresholds and will notify Knology of fraud alerts so that Knology may take action it deems appropriate.

III. Responsibilities of the Parties

- BellSouth will administer all data stored in the LIDB, including the data provided by Knology pursuant to this Agreement, in the same manner as BellSouth's data for BellSouth's end user customers. BellSouth shall not be responsible to Knology for any lost revenue which may result from BellSouth's administration of the LIDB pursuant to its established practices and procedures as they exist and as they may be changed by BellSouth in its sole discretion from time to time.
- B. Billing and Collection Customers

BellSouth currently has in effect numerous billing and collection agreements with various interexchange carriers and billing clearinghouses and as such these billing and collection customers ("B&C Customers") query BellSouth's LIDB to determine whether to accept various billing options from end users. Until such time as BellSouth implements in its LIDB and its supporting systems the means to differentiate Knology's data from BellSouth's data, the following terms and conditions shall apply:

- BellSouth will identify Knology's end user originated long distance charges and will return those charges to the interexchange carrier as not covered by the existing B&C agreement with interexchange carriers for handling of long distance charges by their end users.
- 2. BellSouth shall have no obligation to become involved in any disputes between Knology and B&C Customers. BellSouth will not issue adjustments for charges billed on behalf of any B&C Customer to Knology. It shall be the responsibility of Knology and the B&C Customers to negotiate and arrange for any appropriate adjustments.

C. SPNP Arrangements

- 1. BellSouth will include billing number information associated with exchange lines or SPNP arrangements in its LIDB. Knology will request any toll billing exceptions via the Local Service Request (LSR) form used to order exchange lines, or the SPNP service request form used to order SPNP arrangements.
- 2. Under normal operating conditions, BellSouth shall include the billing number information in its LIDB upon completion of the service order establishing either the local exchange service or the SPNP arrangement, provided that BellSouth shall not be held responsible for any delay or failure in performance to the extent such delay or failure is caused by circumstances or conditions beyond BellSouth's reasonable control. BellSouth will store in its LIDB an unlimited volume of the working telephone numbers associated with either the local exchange lines or the SPNP arrangements. For local exchange lines or for SPNP arrangements, BellSouth will issue line-based calling cards only in the name of Knology. BellSouth will not issue line-based calling cards in the name of Knology's individual End Users. In the event that Knology wants to include calling card numbers assigned by Knology in the BellSouth LIDB, a separate agreement is required.

IV. Fees for Service and Taxes

- A. Knology will not be charged a fee for storage services provided by BellSouth to Knology as described in this LIDB Facilities Based Storage Agreement.
- B. Sales, use and all other taxes (excluding taxes on BellSouth's income) determined by BellSouth or any taxing authority to be due to any federal, state or local taxing jurisdiction with respect to the provision of the service set forth herein will be paid by Knology in accordance with the tax provisions set forth in the General Terms and Conditions of this Agreement.

UNBU	NDLEI	D NETWORK ELEMENTS - Alabama			1		1						r -	Attachment:			bit: B
CATEG	ORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
									-							DISC 1St	Disc Add I
							Rec	Nonree			Disconnect	0.01150	001111		Rates(\$)	SOMAN	001141
	The "Zo	l one" shown in the sections for stand-alone loops or loops as	nart of	a com	bination refers to Ge	ographically	v Deaveraged U	First NF Zones, To	Add'l view Georgran	First bically Deave	Add'l aged UNF Zor		SOMAN ons by C O		SOMAN	SOMAN	SOMAN
		vww.interconnection.bellsouth.com/become_a_clec/html/inter				ograpinoun;	y Deaveragea o		new deorgrap	Sincarly Deave	agea one zon	ie Desiguiti		, refer to filter	net Webbite.		
OPERA	TIONAL	SUPPORT SYSTEMS															
		(1) Electronic Service Order: CLEC should contact its contract															is rate
		is the BellSouth regional electronic service ordering charge.															
		(2) Any element that can be ordered electronically will be bill															
		elements that cannot be ordered electronically at present per t og charge, SOMAN, will be applied to a CLECs bill when it sub				e in this cate	gory reflects th	e charge that v	would be billed	to a CLEC on	ce electronic o	ordering cap	abilities co	me on-line to	r that elemen	. Otherwise,	the manual
	oraerin	Electronic OSS Charge, per LSR, submitted via BST's OSS	omits an	LSK	o BellSouth.	1	1					r					
		interactive interfaces (Regional)				SOMEC		3.50									
		Manual Service Order Charge, per LSR, Disconnect Only (AL)				SOMAN	1	2.50		1.97		1				1	
		DATE ADVANCEMENT CHARGE															
	NOTE:	The Expedite charge will be maintained commensurate with	BellSou	th's F	CC No.1 Tariff, Section	on 5 as appl	icable.										
		UNE Expedite Charge per Circuit or Line Assignable USOC, per				00.000		000.00									
					ALL UNE	SDASP		200.00							_		
		ANALOG VOICE GRADE LOOP				<u> </u>						<u> </u>					
	2-1111	2-Wire Analog Voice Grade Loop - Service Level 1- Zone 1		1	UEANL	UEAL2	12.58	37.81	17.56	23.49	5.30		15.66				
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 1		2	UEANL	UEAL2	21.05	37.81	17.56	23.49	5.30	t	15.66		-		
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 3		3	UEANL	UEAL2	34.34	37.81	17.56	23.49	5.30		15.66				
		Loop Testing - Basic 1st Half Hour			UEANL	URET1		34.16					15.66				
		Loop Testing - Basic Additional Half Hour			UEANL	URETA		19.85					15.66				
		CLEC to CLEC Conversion Charge Without Outside Dispatch															
		(UVL-SL1)			UEANL	UREWO		15.78	8.94				15.66				
		Unbundled Voice Loop, Unbundled Non-Design Voice Loop, billing for BST providing make-up			UEANL	UEANM		13.44									
		Manual Order Coordination for UVL-SL1s (per loop)			UEANL			8.15									
		Order Coordination for Specified Conversion Time for UVL-SL1			ULANL	OLANO		0.15									
		(per LSR)			UEANL	OCOSL		18.09									
	2-WIRE	Unbundled COPPER LOOP															
		2-Wire Unbundled Copper Loop - Non-Designed Zone 1		1	UEQ	UEQ2X	11.20	34.14	15.10	21.25	4.15		15.66				
		2 Wire Unbundled Copper Loop - Non-Designed - Zone 2	I	2	UEQ	UEQ2X	13.27	34.14	15.10	21.25	4.15		15.66				
		2 Wire Unbundled Copper Loop - Non-Designed - Zone 3	I	3	UEQ	UEQ2X	15.07	34.14	15.10	21.25	4.15		15.66				
		Order Coordination 2 Wire Unbundled Copper Loop - Non-			1150	1000100		0.45									
┝──┤		Designed (per loop) Unbundled Copper Loop, Non-Designed Billing for BST			UEQ	USBMC		8.15				<u> </u>					
		providing make-up			UEQ	UEQMU		13.44					15.66				
		Loop Testing - Basic 1st Half Hour			UEQ	URET1	1	34.16		1		t	15.66		-		
		Loop Testing - Basic Additional Half Hour			UEQ	URETA		19.85					15.66				
		CLEC to CLEC Conversion Charge Without Outside Dispatch															
		(UCL-ND)			UEQ	UREWO	1	14.27	7.43				15.66				
		XCHANGE ACCESS LOOP															
┝──┤	2-WIRE	ANALOG VOICE GRADE LOOP					-					<u> </u>			L		
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zone 1		4	UEPSR UEPSB	UEALS	12.58	37.81	17.56	23.49	5.30		15.66				
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-			JEI ON DEF 3D	JEALO	12.30	57.01	17.30	23.49	5.30	-	10.00				
		Zone 1		1	UEPSR UEPSB	UEABS	12.58	37.81	17.56	23.49	5.30		15.66				
		2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting-				-											
		Zone 2		2	UEPSR UEPSB	UEALS	21.05	37.81	17.56	23.49	5.30		15.66				
		2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting-															
		Zone 2		2	UEPSR UEPSB	UEABS	21.05	37.81	17.56	23.49	5.30		15.66				
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-		_				07.04	47 - 0				45.00				
┝──┤		Zone 3		3	UEPSR UEPSB	UEALS	34.34	37.81	17.56	23.49	5.30	<u>├</u> ──	15.66		L		1
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zone 3		3	UEPSR UEPSB	UEABS	34.34	37.81	17.56	23.49	5.30		15.66				
┝──┤	UNFLO	pop Rates for Line Splitting		3	ULF ON ULFOD	ULADO	34.34	37.01	17.30	23.49	5.30	<u> </u>	10.00	{		ł	1
		2-Wire Voice Grade Loop (SL1) for Line Splitting - Zone 1		1	UEPRX	UEPLX	12.70			1		t			-		
		2-Wire Voice Grade Loop (SL1) for Line Splitting - Zone 2	1	2	UEPRX	UEPLX	21.19							İ			
		2-Wire Voice Grade Loop (SL1)for Line Splitting - Zone 3	-		UEPRX	UEPLX	34.80					1					1

JNBUNDLE	D NETWORK ELEMENTS - Alabama			I							Г		Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	EXCHANGE ACCESS LOOP															
2-WIRI	E ANALOG VOICE GRADE LOOP															
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling - Zone 1		1	UEA	UEAL2	14.38	88.00	55.00	47.24	7.44		15.66				
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling - Zone 2		2	UEA	UEAL2	22.85	88.00	55.00	47.24	7.44		15.66				
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling - Zone 3		3	UEA	UEAL2	36.14	88.00	55.00	47.24	7.44		15.66				
	Order Coordination for Specified Conversion Time (per LSR)		-	UEA	OCOSL		18.09									
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse															
	Battery Signaling - Zone 1		1	UEA	UEAR2	14.38	88.00	55.00	47.24	7.44		15.66				
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zone 2		2	UEA	UEAR2	22.85	88.00	55.00	47.24	7.44		15.66				
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zone 3		3	UEA	UEAR2	36.14	88.00	55.00	47.24	7.44		15.66				
	Order Coordination for Specified Conversion Time (per LSR)		Ű	UEA	OCOSL	00.11	18.09	00.00				10.00				
	CLEC to CLEC Conversion Charge without outside dispatch			UEA	UREWO		87.72	36.36				15.66				
4-WIRI	E ANALOG VOICE GRADE LOOP															
	4-Wire Analog Voice Grade Loop - Zone 1		1	UEA	UEAL4	25.34	131.97	94.51	59.14	14.50		15.66				
	4-Wire Analog Voice Grade Loop - Zone 2		2	UEA	UEAL4	38.58	131.97	94.51	59.14	14.50		15.66				
	4-Wire Analog Voice Grade Loop - Zone 3		3	UEA	UEAL4	60.02	131.97	94.51	59.14	14.50		15.66				
	Order Coordination for Specified Conversion Time (per LSR)			UEA	OCOSL		18.09									
	CLEC to CLEC Conversion Charge without outside dispatch			UEA	UREWO		87.72	36.36				15.66				
2-WIRI	E ISDN DIGITAL GRADE LOOP															
	2-Wire ISDN Digital Grade Loop - Zone 1		1	UDN	U1L2X	21.88	117.24	79.77	52.88	10.54		15.66				
	2-Wire ISDN Digital Grade Loop - Zone 2		2	UDN	U1L2X	32.85	117.24	79.77	52.88	10.54		15.66				
	2-Wire ISDN Digital Grade Loop - Zone 3		3	UDN	U1L2X	48.55	117.24	79.77	52.88	10.54		15.66				
	Order Coordination For Specified Conversion Time (per LSR)			UDN	OCOSL		18.09	11.10				45.00				
2 W/D	CLEC to CLEC Conversion Charge without outside dispatch			UDN	UREWO		91.63	44.16				15.66				
2-WIRI	E Universal Digital Channel (UDC) COMPATIBLE LOOP [2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone		1	UDC	UDC2X	21.88	117.24	79.77	52.88	10.54		15.66				
	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone		<u> </u>	000	0202/	21.00			02.00	10.01		10.00				
	2 2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone	I	2	UDC	UDC2X	32.85	117.24	79.77	52.88	10.54		15.66				
	3	Т	3	UDC	UDC2X	48.55	117.24	79.77	52.88	10.54		15.66				
	CLEC to CLEC Conversion Charge without outside dispatch			UDC	UREWO		91.63	44.16				15.66				
2-WIRI	E ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMP	ATIBLE	LOOP													
	2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation - Zone 1		1	UAL	UAL2X	11.01	110.00	68.00	47.24	7.44		15.66				
	2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation - Zone 2		2	UAL	UAL2X	12.73	110.00	68.00	47.24	7.44		15.66				
	2 Wire Unbundled ADSL Loop including manual service inquiry					44.00	440.00	00.00	17.04	7.44		45.00				
	& facility reservation - Zone 3		3	UAL UAL	UAL2X	14.30	110.00	68.00	47.24	7.44		15.66				
	Order Coordination for Specified Conversion Time (per LSR) 2 Wire Unbundled ADSL Loop without manual service inquiry &			-	OCOSL		18.09									
	facility reservaton - Zone 1 2 Wire Unbundled ADSL Loop without manual service inquiry &		1	UAL	UAL2W	11.01	90.00	57.00	47.24	7.44		15.66				
	facility reservaton - Zone 2		2	UAL	UAL2W	12.73	90.00	57.00	47.24	7.44		15.66				
	2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 3		3	UAL	UAL2W	14.30	90.00	57.00	47.24	7.44		15.66				
	Order Coordination for Specified Conversion Time (per LSR)			UAL	OCOSL		18.09									
	CLEC to CLEC Conversion Charge without outside dispatch			UAL	UREWO		86.20	40.40				15.66				
2-WIRI	E HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPA	TIBLE	LOOP													
	2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation - Zone 1		1	UHL	UHL2X	8.74	110.00	68.00	47.24	7.44		15.66				
	2 Wire Unbundled HDSL Loop including manual service inquiry						-					15.66				

UNBUNDLI	ED NETWORK ELEMENTS - Alabama			1							1		Attachment:			ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2 Wire Unbundled HDSL Loop including manual service inquiry								17.04							
	& facility reservation - Zone 3		3	UHL	UHL2X	11.44	110.00	68.00	47.24	7.44		15.66				
	Order Coordination for Specified Conversion Time (per LSR) 2 Wire Unbundled HDSL Loop without manual service inquiry			UHL	OCOSL		18.09									<u> </u>
	and facility reservation - Zone 1		1	UHL	UHL2W	8.74	90.00	57.00	47.24	7.44		15.66				
	2 Wire Unbundled HDSL Loop without manual service inquiry			UTIL	UTILZW	0.74	90.00	57.00	47.24	7.44		15.00				
	and facility reservation - Zone 2		2	UHL	UHL2W	10.17	90.00	57.00	47.24	7.44		15.66				
	2 Wire Unbundled HDSL Loop without manual service inquiry			-	-											
	and facility reservation - Zone 3		3	UHL	UHL2W	11.44	90.00	57.00	47.24	7.44		15.66				
	Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		18.09									
	CLEC to CLEC Conversion Charge without outside dispatch			UHL	UREWO		86.14	40.40				15.66				
4-WIF	RE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPA	TIBLE	LOOP													
	4 Wire Unbundled HDSL Loop including manual service inquiry		1			10.67	4 40 00	00.00	F4	0		45.00				
	and facility reservation - Zone 1		1	UHL	UHL4X	13.95	148.36	68.00	51.70	9.73		15.66				
	4-Wire Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 2		2	UHL	UHL4X	15.56	148.36	68.00	51.70	9.73		15.66				
	4-Wire Unbundled HDSL Loop including manual service inquiry		- 4			10.00	140.30	00.00	51.70	9.73		10.00				
	and facility reservation - Zone 3		3	UHL	UHL4X	15.25	148.36	68.00	51.70	9.73		15.66				
	Order Coordination for Specified Conversion Time (per LSR)		Ŭ	UHL	OCOSL	10.20	18.09	00.00	01.10	0.10		10.00				
	4-Wire Unbundled HDSL Loop without manual service inquiry															
	and facility reservation - Zone 1		1	UHL	UHL4W	13.95	94.00	57.00	51.70	9.73		15.66				
	4-Wire Unbundled HDSL Loop without manual service inquiry															
	and facility reservation - Zone 2		2	UHL	UHL4W	15.56	94.00	57.00	51.70	9.73		15.66				
	4-Wire Unbundled HDSL Loop without manual service inquiry															
	and facility reservation - Zone 3		3	UHL	UHL4W	15.25	94.00	57.00	51.70	9.73		15.66				
	Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		18.09									
	CLEC to CLEC Conversion Charge without outside dispatch			UHL	UREWO		86.14	40.40				15.66				
4-WIF	RE DS1 DIGITAL LOOP		1			82.55	252.47	157.54	44.70	11.71		45.00				
	4-Wire DS1 Digital Loop - Zone 1 4-Wire DS1 Digital Loop - Zone 2			USL USL	USLXX USLXX	82.55	252.47	157.54	44.70	11.71		15.66 15.66				
	4-Wire DS1 Digital Loop - Zone 3		3	USL	USLXX	314.52	252.47	157.54	44.70	11.71		15.66				
	Order Coordination for Specified Conversion Time (per LSR)		5	USL	OCOSL	314.32	18.09	107.04	44.70	11.71		13.00				
	CLEC to CLEC Conversion Charge without outside dispatch			USL	UREWO		101.09	43.05				15.66				
4-WIF	RE 19.2, 56 OR 64 KBPS DIGITAL GRADE LOOP															
	4 Wire Unbundled Digital 19.2 Kbps		1	UDL	UDL19	26.09	126.27	88.80	59.14	14.50		15.66				
	4 Wire Unbundled Digital 19.2 Kbps		2	UDL	UDL19	35.95	126.27	88.80	59.14	14.50		15.66				
	4 Wire Unbundled Digital 19.2 Kbps			UDL	UDL19	37.88	126.27	88.80	59.14	14.50		15.66				
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 1		1	UDL	UDL56	26.09	126.27	88.80	59.14	14.50		15.66				
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 2		2	UDL	UDL56	35.95	126.27	88.80	59.14	14.50		15.66				
├──	4 Wire Unbundled Digital Loop 56 Kbps - Zone 3		3	UDL	UDL56	37.88	126.27 18.09	88.80	59.14	14.50		15.66				<u> </u>
	Order Coordination for Specified Conversion Time (per LSR)		1	UDL UDL	OCOSL UDL64	26.09	18.09	88.80	59.14	14.50		15.66				
├──	4 Wire Unbundled Digital Loop 64 Kbps - Zone 1 4 Wire Unbundled Digital Loop 64 Kbps - Zone 2		2	UDL	UDL64	26.09	126.27	88.80	59.14	14.50		15.66				
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 2			UDL	UDL64	35.95	126.27	88.80	59.14	14.50		15.66				ł
	Order Coordination for Specified Conversion Time (per LSR)		Ŭ	UDL	OCOSL	07.00	18.09	00.00	00.14	14.00		10.00				
	CLEC to CLEC Conversion Charge without outside dispatch	1	1	UDL	UREWO		102.13	49.75	1			15.66			1	<u> </u>
2-WIF	RE Unbundled COPPER LOOP	1													l	1
	2-Wire Unbundled Copper Loop/Short including manual service		1													
	inquiry & facility reservation - Zone 1		1	UCL	UCLPB	11.01	112.46	65.30	47.24	7.44		15.66				
	2-Wire Unbundled Copper Loop/Short including manual service		1													
	inquiry & facility reservation - Zone 2		2	UCL	UCLPB	12.73	112.46	65.30	47.24	7.44		15.66				
	2 Wire Unbundled Copper Loop/Short including manual service		_													
	inquiry & facility reservation - Zone 3		3	UCL	UCLPB	14.30	112.46	65.30	47.24	7.44		15.66				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.15	8.15								<u> </u>
	2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 1		1	UCL	UCLPW	11.01	91.46	54.30	47.24	7.44	1	15.66				
	2-Wire Unbundled Copper Loop/Short without manual service			UUL	0001.00	11.01	31.40	54.30	41.24	7.44		10.00				<u>├</u>
	inquiry and facility reservation - Zone 2		2	UCL	UCLPW	12.73	91.46	54.30	47.24	7.44		15.66				

UNBUNDLE	D NETWORK ELEMENTS - Alabama												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 3		3	UCL	UCLPW	14.30	91.46	54.30	47.24	7.44		45.00				1
	Order Coordination for Unbundled Copper Loops (per loop)		3	UCL	UCLMC	14.30	8.15	8.15	47.24	7.44		15.66				
	2-Wire Unbundled Copper Loop/Long - includes manual srvc.			UOL	OOLINO		0.10	0.10								
	inquiry and facility reservation - Zone 1		1	UCL	UCL2L	31.42	112.46	65.30	47.24	7.44		15.66				1
	2-Wire Unbundled Copper Loop/Long - includes manual svc.															
	inquiry and facility reservation - Zone 2		2	UCL	UCL2L	55.01	112.46	65.30	47.24	7.44		15.66				ا ا
	2-Wire Unbundled Copper Loop/Long - includes manual svc.		2	UCL		00.00	110.10	05.00	47.04	7.44		45.00				1
	inquiry and facility reservation - Zone 3 Order Coordination for Unbundled Copper Loops (per loop)		3	UCL	UCL2L UCLMC	80.00	112.46 8.15	65.30 8.15	47.24	7.44		15.66				
	2-Wire Unbundled Copper Loop/Long - without manual service			002	OOLIVIO		0.15	0.15								
	inquiry and facility reservation - Zone 1	I.	1	UCL	UCL2W	31.42	91.46	54.30	47.24	7.44		15.66				1
İ	2-Wire Unbundled Copper Loop/Long - without manual service															
	inquiry and facility reservation - Zone 2		2	UCL	UCL2W	55.01	91.46	54.30	47.24	7.44		15.66				,
	2-Wire Unbundled Copper Loop/Long - without manual service		~			00.00	04.40	54.00	17.04			45.00				1
	inquiry and facility reservation - Zone 3 Order Coordination for Unbundled Copper Loops (per loop)		3	UCL UCL	UCL2W UCLMC	80.00	91.46 8.15	54.30 8.15	47.24	7.44		15.66				
	CLEC to CLEC Conversion Charge without outside dispatch			UCL	OCLIVIC		0.15	0.15								
	(UCL-Des)			UCL	UREWO		97.23	42.48				15.66				1
4-WIRE	COPPER LOOP															
	4-Wire Copper Loop/Short - including manual service inquiry and facility reservation - Zone 1		1	UCL	UCL4S	17.36	135.21	88.05	51.70	9.73		15.66				
	4-Wire Copper Loop/Short - including manual service inquiry		~	UCL		00.70	105.01	00.05	54 70	0.70		45.00				1
	and facility reservation - Zone 2 4-Wire Copper Loop/Short - including manual service inquiry		2	UCL	UCL4S	20.76	135.21	88.05	51.70	9.73		15.66				
	and facility reservation - Zone 3		3	UCL	UCL4S	28.21	135.21	88.05	51.70	9.73		15.66				1
	Order Coordination for Unbundled Copper Loops (per loop)		-	UCL	UCLMC		8.15	8.15								
	4-Wire Copper Loop/Short - without manual service inquiry and															
	facility reservation - Zone 1	1	1	UCL	UCL4W	17.36	114.21	67.05	51.70	9.73		15.66				ا ا
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation - Zone 2	I	2	UCL	UCL4W	20.76	114.21	67.05	51.70	9.73		15.66				
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation - Zone 3		3	UCL	UCL4W	28.21	114.21	67.05	51.70	9.73		15.66				, ,
-	Order Coordination for Unbundled Copper Loops (per loop)		Ŭ	UCL	UCLMC	20.21	8.15	8.15	01.70	0.10		10.00				
	4-Wire Unbundled Copper Loop/Long - includes manual svc.															
	inquiry and facility reservation - Zone 1		1	UCL	UCL4L	49.35	135.21	88.05	51.70	9.73		15.66				,
	4-Wire Unbundled Copper Loop/Long - includes manual svc.		2	UCL	UCL4L	92.45	135.21	88.05	51.70	9.73		15.66				1
	inquiry and facility reservation - Zone 2 4-Wire Unbundled Copper Loop/Long - includes manual svc.		2	UCL	UCL4L	92.45	135.21	88.05	51.70	9.73		10.00				
	inquiry and facility reservation - Zone 3		3	UCL	UCL4L	127.39	135.21	88.05	51.70	9.73		15.66				1
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.15	8.15								
	4-Wire Unbundled Copper Loop/Long - without manual svc.															
	inquiry and facility reservation - Zone 1		1	UCL	UCL40	49.35	114.21	67.05	51.70	9.73		15.66				
	4-Wire Unbundled Copper Loop/Long - without manual svc.		2	UCL	UCL4O	92.45	114.21	67.05	51.70	9.73		15.66				1
	inquiry and facility reservation - Zone 2 4-Wire Unbundled Copper Loop/Long - without manual svc.		2	UUL	UUL4U	92.45	(14.21	07.05	51.70	9.73		00.01				
	inquiry and facility reservation - Zone 3	1	3	UCL	UCL4O	127.39	114.21	67.05	51.70	9.73		15.66				1
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.15	8.15			1					
	CLEC to CLEC conversion Charge without outside dispatch		[UCL	UREWO		97.23	42.48				15.66				
	Unbundled Loop Modification, Removal of Load Coils - 2 Wire			UAL, UHL, UCL, UEQ, ULS, UEA, UEANL, UDL, UDC, UDN, UDL, USL,												
	pair less than or equal to 18k ft			UEPSR, UEPSB	ULM2L		0.00	0.00				15.66				
	Unbundled Loop Modification, Removal of Load Coils - 2 wire greater than 18k ft Unbundled Loop Modification Removal of Load Coils - 4 Wire	I		UCL, ULS, UEQ, UEPSR, UEPSB	ULM2G		170.51	170.51				15.66				
	less than or equal to 18K ft	Т		UHL, UCL	ULM4L		0.00	0.00				15.66				1

UNBUNDLE	ED NETWORK ELEMENTS - Alabama												Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
							Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Unbundled Loop Modification Removal of Load Coils - 4 Wire															
	pair greater than 18k ft	1		UCL	ULM4G		170.51	170.51				15.66				
	Unbundled Loop Modification Removal of Bridged Tap Removal, per unbundled loop	1		UAL, UHL, UCL, UEQ, UEF, ULS, UEA, UEANL, UDL, UDC, UDN, UDL, USL, UEPSR, UEPSB	ULMBT		32.41	32.41				15.66				
SUB-LOOPS																
Sub-L	oop Distribution															
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-				110000		<u> </u>									
ı — I —	Up			UEANL	USBSA		244.42					15.66				
1	Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-Up	1	1	UEANL	USBSB		22.64					15.66				
<u> </u>	Sub-Loop - Per Building Equipment Room - CLEC Feeder			0E/ WE	00000		22.04					10.00				
	Facility Set-Up	1		UEANL	USBSC		177.45					15.66				
	Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel															
	Set-Up	1		UEANL	USBSD		55.15					15.66				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop -		1			11.01	05.00		45.05	0.70		45.00				
	Zone 1 Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop -		1	UEANL	USBN2	11.21	65.80	30.96	45.25	6.70		15.66				
	Zone 2		2	UEANL	USBN2	11.94	65.80	30.96	45.25	6.70		15.66				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop -		-	0E/ WE	000112	11.04	00.00	00.00	40.20	0.70		10.00				
	Zone 3		3	UEANL	USBN2	16.86	65.80	30.96	45.25	6.70		15.66				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		8.15	8.15								
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop -															
	Zone 1		1	UEANL	USBN4	8.46	79.03	44.19	49.71	9.07		15.66				
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 2		2	UEANL	USBN4	16.67	79.03	44.19	49.71	9.07		15.66				
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop -		2	UEANL	USDIN4	10.07	79.03	44.19	49.71	9.07		15.00				
	Zone 3		3	UEANL	USBN4	32.57	79.03	44.19	49.71	9.07		15.66				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		8.15	8.15								
	Sub-Loop 2-Wire Intrabuilding Network Cable (INC)	- 1		UEANL	USBR2	2.27	53.01	18.17	45.25	6.70		15.66				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 4-Wire Intrabuilding Network Cable (INC)			UEANL UEANL	USBMC USBR4	5.16	8.15 59.25	8.15 24.41	49.71	9.07		15.66				
	Sub-Loop 4-wire intrabuliding Network Cable (INC)	1		UEANL	USBR4	5.16	59.25	24.41	49.71	9.07		10.00				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		8.15	8.15								
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 1		1	UEF	UCS2X	6.22	65.80	30.96	45.25	6.70		15.66				
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2		2	UEF	UCS2X	8.76	65.80	30.96	45.25	6.70		15.66				
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 3		3	UEF	UCS2X	11.27	65.80	30.96	45.25	6.70		15.66				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair	<u> </u>	1	UEF UEF	USBMC UCS4X	6.11	8.15 79.03	8.15 44.19	49.71	9.07		15.66				
┝───┤───	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 1 4 Wire Copper Unbundled Sub-Loop Distribution - Zone 2			UEF	UCS4X UCS4X	6.11 12.61	79.03	44.19	49.71 49.71	9.07		15.66				
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 4 Wire Copper Unbundled Sub-Loop Distribution - Zone 3			UEF	UCS4X	15.36	79.03	44.19	49.71	9.07		15.66				
			Ť	1						0.01		.0.00				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC		8.15	8.15								
Unbu	ndled Sub-Loop Modification															
	Unbundled Sub-Loop Modification - 2-W Copper Dist Load															
	Coil/Equip Removal per 2-W PR	I		UEF	ULM2X		175.78	5.10			ļ	15.66				
	Unbundled Sub-loop Modification - 4-W Copper Dist Load Coil/Equip Removal per 4-W PR			UEF	ULM4X		175.78	5.10				15.66				
┝───┤───	Coll/Equip Removal per 4-W PR Unbundled Sub-loop Modification - 2-w/4-w Copper Dist Bridged			UEF	ULIVI4X		1/5./8	5.10				15.66				
	Tap Removal, per PR unloaded			UEF	ULM4T		278.20	6.11				15.66				
Unbu	ndled Network Terminating Wire (UNTW)	1			200071		210.20	0.71			1	10.00				
	Unbundled Network Terminating Wire (UNTW) per Pair		1	UENTW	UENPP	0.40	30.01					15.66				İ

UNBUNDL	ED NETWORK ELEMENTS - Alabama												Attachment:	2	Exhi	ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I		Incremental Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)	_	
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Netwo	ork Interface Device (NID)						40.00	00.00				45.00				
	Network Interface Device (NID) - 1-2 lines			UENTW UENTW	UND12 UND16		43.23	28.38				15.66				
	Network Interface Device (NID) - 1-6 lines Network Interface Device Cross Connect - 2 W			UENTW	UND16 UNDC2		63.97 5.87	49.11 5.87				15.66 15.66				
	Network Interface Device Cross Connect - 2 W	-		UENTW	UNDC2 UNDC4		5.87	5.87				15.66				
SUB-LOOPS	Network Interface Device Closs Connect - 4W			OLIVIW	UNDO4		5.07	5.07				10.00				
	.oop Feeder															
	USL-Feeder, DS0 Set-up per Cross Box location - CLEC			UEA,												
	Distribution Facility set-up			UDN,UCL,UDL,UDC	USBFW		244.42					15.66				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair			UEA,												
	set-up			UDN,UCL,UDL,UDC	USBFX		22.64	22.64				15.66				
	USL Feeder DS1 Set-up at DSX location, per DS1 termination			USL	USBFZ		519.95	11.32				15.66				
	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice									40		1				
	Grade - Zone 1		1	UEA	USBFA	8.03	93.00	56.48	54.51	13.67		15.66				───
	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 2		2	UEA	USBFA	12.00	93.00	56.48	54.51	13.67		15.66				
	Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start,									10.07		15.00				
	Voice Grade - Zone 3		3	UEA	USBFA	20.39	93.00	56.48	54.51	13.67		15.66				
	Order Coordination for Specified Conversion Time, per LSR Unbundlde Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice			UEA	OCOSL		18.09									
	Grade - Zone 1		1	UEA	USBFB	8.03	93.00	56.48	54.51	13.67		15.66				
	Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2		2	UEA	USBFB	12.00	93.00	56.48	54.51	13.67		15.66				
	Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice	-	2	UEA	USDFD	12.00	93.00	30.40	54.51	13.07		15.00				
	Grade - Zone 3		3	UEA	USBFB	20.39	93.00	56.48	54.51	13.67		15.66				
	Order Coordination for Specified Time Conversion, per LSR		Ŭ	UEA	OCOSL	20.00	18.09	00.40	04.01	10.07		10.00				
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1		1	UEA	USBFC	8.03	93.00	56.48	54.51	13.67		15.66				
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery,															
	Voice Grade - Zone 2		2	UEA	USBFC	12.00	93.00	56.48	54.51	13.67		15.66				
	Unbundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse															
	Battery, Voice Grade - Zone 3		3	UEA	USBFC	20.39	93.00	56.48	54.51	13.67		15.66				
	Order Coordination For Specified Conversion Time, per LSR			UEA	OCOSL		18.09									
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice					10.01				17.10		15.00				
	Grade - Zone 1		1	UEA	USBFD	19.21	107.56	70.09	62.05	17.40		15.66				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2		2	UEA	USBFD	23.47	107.56	70.09	62.05	17.40		15.66				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice	-			00010	23.47	107.30	70.09	02.05	17.40		13.00				<u> </u>
	Grade - Zone 3		3	UEA	USBFD	39.63	107.56	70.09	62.05	17.40		15.66				
	Order Coordination For Specified Conversion Time, Per LSR		<u> </u>	UEA	OCOSL		18.09									1
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice	1												İ		
	Grade - Zone 1		1	UEA	USBFE	19.21	107.56	70.09	62.05	17.40		15.66				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2		2	UEA	USBFE	23.47	107.56	70.09	62.05	17.40		15.66				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice															
	Grade - Zone 3		3	UEA	USBFE	39.63	107.56	70.09	62.05	17.40		15.66				
	Order Coordination For Specified Conversion Time, Per LSR			UEA	OCOSL		18.09									Ļ
	Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 1			UDN	USBFF	14.87	106.16	68.69	55.64	13.29		15.66		ļ		L
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 2			UDN	USBFF	21.69	106.16	68.69	55.64	13.29		15.66				───
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 3 Order Coordination For Specified Conversion Time, Per LSR	-	3	UDN UDN	USBFF OCOSL	32.51	106.16 18.09	68.69	55.64	13.29	-	15.66				<u> </u>
├ ── ├ ──	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)		1	UDN	USBFS	14.87	18.09	68.69	55.64	13.29		15.66				╂─────
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)	-	2	UDC	USBFS	21.69	106.16	68.69	55.64	13.29		15.66				<u> </u>
	Unbundled Sub-Loop Feeder, 2 Wire ODC (IDSL compatible)			UDC	USBFS	32.51	106.16	68.69	55.64	13.29		15.66				1
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 1	l		USL	USBFG	55.09	101.85	64.38	62.05	17.40		15.66	1		1	1
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 2	l		USL	USBFG	124.69	101.85	64.38	62.05	17.40	1	15.66	l		1	1
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 3		3	USL	USBFG	294.62	101.85	64.38	62.05	17.40		15.66				
	Order Coordination For Specified Conversion Time, Per LSR			USL	OCOSL		18.09									
	Unbundled Sub-Loop Feeder, 2-Wire Copper Loop - Zone 1		1	UCL	USBFH	5.75	83.78	46.32	53.02	10.67		15.66				

UNBUND	DLED NETWORK ELEMENTS - Alabama												Attachment:	2	Exhi	bit: B
CATEGOR		Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st		Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge -
						Baa	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		•
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone															
	2		2	UCL	USBFH	4.93	83.78	46.32	53.02	10.67		15.66				
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone		3	UCL	USBFH	3.96	83.78	46.32	53.02	10.67		15.66				
├ ── ├ ──	Order Coordination For Specified Conversion Time, per LSR		3	UCL	OCOSL	3.90	18.09	40.32	55.02	10.67		15.00				
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 1		1	UCL	USBFJ	12.71	100.99	63.53	57.90	13.26		15.66				
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 2		2	UCL	USBFJ	9.69	100.99	63.53	57.90	13.26		15.66				
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 3			UCL	USBFJ	14.37	100.99	63.53	57.90	13.26		15.66				
	Order Coordination For Specified Conversion Time, per LSR			UCL	OCOSL		18.09									
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		1	UDL	USBFN	19.20	101.85	64.38	62.05	17.40		15.66				
\vdash	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		2	UDL	USBFN	21.64	101.85	64.38	62.05	17.40		15.66				
\vdash	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		3	UDL	USBFN	23.75	101.85	64.38	62.05	17.40		15.66				
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone 1		1	UDL	USBFO	19.20	101.85	64.38	62.05	17.40		15.66				
\vdash	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -			ODL	030-0	19.20	101.00	04.38	C2.U0	17.40		10.00	ł		ł	1
	Zone 2		2	UDL	USBFO	21.64	101.85	64.38	62.05	17.40		15.66				
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -		<u> </u>		202.0	204		000	02.00			.0.00				İ
	Zone 3		3	UDL	USBFO	23.75	101.85	64.38	62.05	17.40		15.66				
	Order Coordination For Specified Time Conversion, per LSR			UDL	OCOSL		18.09									
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -															
	Zone 1		1	UDL	USBFP	19.20	101.85	64.38	62.05	17.40		15.66				
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -						101.05					15.00				
	Zone 2 Sub Lean Fooder, Der 4 Wire 64 Khao Digital Crade Lean		2	UDL	USBFP	21.64	101.85	64.38	62.05	17.40		15.66				
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 3		3	UDL	USBFP	23.75	101.85	64.38	62.05	17.40		15.66				
	Order Coordination For Specified Conversion Time, per LSR		3	UDL	OCOSL	23.75	18.09	04.30	02.03	17.40		15.00				
SUB-LOOP				ODE	CCCCL		10.00									
	ub-Loop Feeder															
	Sub Loop Feeder - DS3 - Per Mile Per Month	I		UE3	1L5SL	13.55										
	Sub Loop Feeder - DS3 - Facility Termination Per Month			UE3	USBF1	332.40	3,400.58	407.00	160.47	90.97		15.66				
	Sub Loop Feeder – STS-1 – Per Mile Per Month			UDLSX	1L5SL	13.55										
	Sub Loop Feeder - STS-1 - Facility Termination Per Month			UDLSX	USBF7	357.36	3,400.58	407.00	160.47	90.97		15.66				
	Sub Loop Feeder – OC-3 – Per Mile Per Month			UDLO3	1L5SL	10.28			-		-					
	Sub Loop Feeder - OC-3 - Facility Termination Protection Per Month			UDLO3	USBF5	54.89										
<u> </u>	Sub Loop Feeder - OC-3 - Facility Termination Per Month			UDLO3	USBF2	538.69	3,400.58	407.00	160.47	90.97		15.66				
	Sub Loop Feeder - OC-12 - Per Mile Per Month	i		UDL12	1L5SL	12.66	0,400.00	407.00	100.47	50.57		10.00				
	Sub Loop Feeder - OC-12 - Facility Termination Protection Per															
	Month	I.		UDL12	USBF6	620.18										
	Sub Loop Feeder - OC-12 - Facility Termination Per Month			UDL12	USBF3	1,729.00	3,400.58	407.00	160.47	90.97		15.66				
	Sub Loop Feeder - OC-48 - Per Mile Per Month			UDL48	1L5SL	41.51										
	Sub Loop Feeder - OC-48 - Facility Termination Protection Per	,				c / c - c -										
\vdash	Month	1		UDL48	USBF9	310.30	2 500 50	407.00	400 47	00.07		45.00				
\vdash	Sub Loop Feeder - OC-48 - Facility Termination Per Month Sub Loop Feeder - OC-12 Interface On OC-48	1		UDL48 UDL48	USBF4 USBF8	1,495.00 350.09	3,586.58	407.00 407.00	160.47 160.47	90.97 90.97		15.66				
	LED LOOP CONCENTRATION	I			USBFÖ	300.09	804.67	407.00	100.47	90.97		15.66				
	Unbundled Loop Concentration - System A (TR008)			ULC	UCT8A	364.17	325.41	325.41				15.66				
	Unbundled Loop Concentration - System B (TR008)			ULC	UCT8B	43.70	135.59	135.59				15.66	1		1	1
	Unbundled Loop Concentration - System A (TR303)			ULC	UCT3A	395.12	325.41	325.41								İ
	Unbundled Loop Concentration - System B (TR303)			ULC	UCT3B	73.64	135.59	135.59				15.66				
	Unbundled Loop Concentration - DS1 Loop Interface Card			ULC	UCTCO	4.16	63.29	46.07	16.79	4.70		15.66				
	Unbundled Loop Concentration - ISDN Loop Interface (Brite			l					I T							
\vdash	Card)			UDN	ULCC1	6.60	10.54	10.48	5.39	5.36		15.66		ļ		
	Unbundled Loop Concentration - UDC Loop Interface (Brite					0.00	10.54	40.40	F 00	F 00		45.00				
\vdash	Card) Unbundled Loop Concentration2 Wire Voice-Loop Start or			UDC	ULCCU	6.60	10.54	10.48	5.39	5.36		15.66				
	Ground Start Loop Interface (POTS Card)			UEA	ULCC2	1.65	10.54	10.48	5.39	5.36		15.66				
			1			00			5.00	5.00	1					
	Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery															

CATEGONT RATE ELEMENTS Image April Logs Logs <thlogs< th=""> <thlogs< th=""> Logs<th>UNBUNDLE</th><th>D NETWORK ELEMENTS - Alabama</th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th>Attachment:</th><th>2</th><th>Exhi</th><th>bit: B</th></thlogs<></thlogs<>	UNBUNDLE	D NETWORK ELEMENTS - Alabama												Attachment:	2	Exhi	bit: B
Image: Constraint - Wire Vale Loop Inertials - Wire Vale Loop Inertial - Wire Vale Loop I	CATEGORY	RATE ELEMENTS		Zone	BCS	USOC						Submitted Elec	Submitted Manually	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Charge - Manual Svc Order vs. Electronic-	Charge -
Unbuilded Loge Convention - 4 WT volage loss from the set of the							Rec										
Species Card UPA U.CCV 5.55 10.54 0.38 5.36 15.66 Hellow District Card Contrainton Tegr UN To Ray Data Loop U.C. U.C. 0.55 10.54 0.38 5.36 11.66 Hellow Internance Logic Locations To Digit 6 Roys Data Loop U.C. 0.67 10.54 10.44 5.38 5.38 11.66 Hellow Internance Logic Locations To Digit 6 Roys Data Loop U.C. 6.67 10.54 10.44 5.38 5.38 11.60 10.50 10.54 10.44 5.38 5.38 11.60 10.50 10.54 10.44 5.38 5.38 10.64 10.44 5.38 10.58 10.58 10.54 10.44 5.38 5.38 10.58 10.58 10.58 10.58 10.58 10.58 10.58 10.58 10.58 10.58 10.58 10.58 10.58 10.58 10.58 <t< td=""><td><u> </u></td><td>Linbundled Loop Concentration 4 Wire Voice Loop Interface</td><td></td><td></td><td></td><td></td><td></td><td>First</td><td>Add'l</td><td>First</td><td>Add'l</td><td>SOMEC</td><td>SOMAN</td><td>SOMAN</td><td>SOMAN</td><td>SOMAN</td><td>SOMAN</td></t<>	<u> </u>	Linbundled Loop Concentration 4 Wire Voice Loop Interface						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Ubburdleid Lags Consentation - TBST GROUT Case ULC UCTC 28.60 10.52 50.80 50.80 50.80 50.80 Working Consentation - Opilit 38 Mys Data Loop interiae UDL ULC 6.97 10.25 10.48 5.30 5.36 15.66 15.66 10.64 10.64 10.64 10.64 10.64 5.30 5.36 10.56 10.60 <td></td> <td></td> <td></td> <td></td> <td>UEA</td> <td>ULCC4</td> <td>5.85</td> <td>10.54</td> <td>10.48</td> <td>5.39</td> <td>5.36</td> <td></td> <td>15.66</td> <td></td> <td></td> <td></td> <td> </td>					UEA	ULCC4	5.85	10.54	10.48	5.39	5.36		15.66				
Interface UPC ULC27 8.07 10.54 10.49 5.39 5.30 15.66 Whorkson Concentration - Digital 64 Apps Dia Loop Interface UDL UCC23 8.07 10.54 10.48 5.30 5.36 15.66 Image UND CONSTRUMENT - Digital 64 Apps Dia Loop Interface UDL UCC3 8.07 10.54 10.48 5.30 5.36 15.66 Image																	
Unbounded Log Concentration: Digite 54 Rbys Data Loop based on any Concentration: UDL LLCCS 4.00 10.44 6.30 6.30 11.66 Weburded Log Concentration: UDL ULCCS 4.00 10.54 10.48 5.39 5.38 15.66 10.00 Weburded Log Concentration: USL ULCCS 4.807 10.04 5.39 5.38 15.66 10.00 Weburded Contract Network Over In Network One To Network Over In Netwo																	
Interlead UDC ULC2 6.07 10.54 10.48 5.39 15.66 UNE OFFER, PEOVISIONEO ON Y, NO RATE UDC 8.07 10.44 0.43 5.39 15.60 Image: Control of the ND instance Im					UDL	ULCC7	8.67	10.54	10.48	5.39	5.36		15.66				
Image: Solution of the VLD state o		Interface			UDL	ULCC5	8.67	10.54	10.48	5.39	5.36		15.66				
UNE OF REF. PEOVISIONED ONLY - NO RATE Image: Control of the MD installation Image: Control of the MD installing instal							0.07	10.54	40.40	5.00	5.00		45.00				
NIC - Degate and Service Order for No mataliation UENTW (KPCRC 0.00 0.00 0.00 Ubunded Cruited Maines Previousing Only - No Rate UENTW (KPCRC 0.00					UDL	ULCC6	8.67	10.54	10.48	5.39	5.36		15.66				
UNITY Clock the statistication of provide and provide and provide and provide constraints of provide and provide an	ONE OTHER, I				UENTW	UNDBX	0.00	0.00							1		
Under Contract Name, Provisioning Only - No Rate ENTW UNECA 0.00 0.00 0.00 We OTHER, Provisioning Only - No Rate INALUGLUDC, UDL, UDL, UDL, UDL, UDL, UDL, UDL, UDL						UENCE	0.00	0.00									
UNE OFTICE, PROVISIONNO ONLY - NO RATE Image: Control barry how how how how how how how how how how																	
Lisbundled Cartact Name, Provisioning Only - no rate UALUCLUCU, UNIC 0.00 0.00 0.00 Using Sub-Loop Feeder-Wire Cross Box Jumper - no rate UEAUROLUCLUDC USER 0.00 0.00 0.00 0.00 Using Sub-Loop Feeder-Wire Cross Box Jumper - no rate UEAUROLUCLUDC USER 0.00 0.00 0.00 0.00 Ubundled DS Loop - Superfame Fermat Option - no rate UEAUROLUCLUDC USE CCOSP 0.00 0.00 Unbundled DS Loop - Superfame Fermat Option - no rate USE CCOSP 0.00 0.00 0.00 Hight Capacity Unbundlet Local Loop - DS3 - Per Mile per Imonth ULS USBPX 0.80 0.00 0.00 0.00 Hight Capacity Unbundlet Local Loop - DS3 - Per Mile per Imonth ULSX LESPX 0.80 845162 283.94 119.49 83.58 15.66 LOP MARE-UP ULSX ULSX ULSX ULSX 1LSND 8.38 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 <					ENTW	UNECN	0.00	0.00									ļ
Unbundled Contract Name, Providening Onty no rate UDNUELAULUC UNECN 0.00 0.00 0.00 0.00 Unbundled Sub-Loop, Feeder 4 Wire Cross Box Jumper - no UEA.UDN.UEL.UDC USFPR 0.00	UNE OTHER, P																
Ubundled Contact Name, Providening Orly - no rate UDR/UEL/UEL/UE UNECAL O 0.00 0.00 0.00 0.00 Ubundled Sub-Loop Feeder-4 Wire Cross Box Jurger - no UEAUDNUELUDC USEFQ 0.00					UAL.UCL.UDC.UDL.												
rate UEALUNUCLUCU USBR0 0.00 0.00 0.00 Hubunded 301-Loop - Segnetifame Format Option - no rate USL CCCOSF 0.00 0.00 0.00 0.00 0.00 Hubunded 305 Loop - Segnetifame Format Option - no rate USL CCCOSF 0.00 <		Unbundled Contact Name, Provisioning Only - no rate				UNECN	0.00	0.00									
Unbundled Sub-Loop Feeder- Wire Cross Box Jumper - no rate UEA, USL, UCL, ULL USBFR 0.00 0.00 0.00 0.00 Unbundled St Loop - Superframe Format Option - no rate USL CCOSF 0.00 0.00 0.00 0.00 Unbundled St Loop - Spanded Superframe Format Option - no rate USL CCOSF 0.00																	
Inde UEAUSL_UCLUD USER 0.00 0.00 Unbunded DS1 Loop - Superframe Format Option - nor net USL CCOSF 0.00 0.00 0.00 0.00 HIGH CAPACIT. Optimum CSL Loop - Superframe Format Option - nor net USL CCOSF 0.00 <td< td=""><td></td><td></td><td></td><td></td><td>UEA,UDN,UCL,UDC</td><td>USBFQ</td><td>0.00</td><td>0.00</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></td<>					UEA,UDN,UCL,UDC	USBFQ	0.00	0.00									
Unbundled DS1 Logo - Superframe Formal Option - no rate USL CCOSF 0.00 0.00 0.00 In orate USL CCOEF 0.00 0.00 0.00 0.00 Info Capacity Unbundled Local Loop - DS3 - Peri Mie per month ULSL CCOEF 0.00 0.00 0.00 Info Capacity Unbundled Local Loop - DS3 - Peri Mie per month ULSL ULSN 8.38 0 <t< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td>0.00</td><td>0.00</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></t<>							0.00	0.00									
Unbundled DS1 Loop - Expanded Superframe Format option - no rate USL CCOEF 0.00 0.00 0.00 0.00 HGH CAPACITY UNBUNDLED LOCAL LOOP USL CCOEF 0.00 0																	
High Capacity UnsubleD LOCAL LOOP Image: Coop - DS3 - Parting or month month LE3 LLSND 8.38 Image: Coop - DS3 - Parting or Loop - DS3 - Facility Image: Coop - DS3 - Facip - DS3 - Facility Image: Coop -																	
High Capacity Unbundled Local Loop - DS3 - Per Mie per month UE3 LISND 8.38 Image: Comparison of the comparison of the					USL	CCOEF	0.00	0.00									
month UE3 LLND 8.38 month u month month u month u month month u month u month u month month u month u month	HIGH CAPACIT																ļ
Image: Image:					UE3	1L5ND	8.38										
High Capacity Unbundled Load Loop - STS-1 - Per Mile per month UDLSX 1LSND 8.38 High Capacity Unbundled Load Loop - STS-1 - Facility Termination per month UDLSX UDLSX UDLSX 119.49 83.58 115.66 LOOP MAKE-UP Fordering Without Reservation, per working or spane facility queried (Manual). UMK UMK 20.00 20.00 20.00																	
month UDLSX 1LSND 8.38 Image: Constraint of the constraint of the					UE3	UE3PX	308.98	451.52	263.94	119.49	83.58		15.66				
Termination per month UDLS1 319.83 451.52 28.34 119.49 83.58 15.66 LOOP MAKEUP Image: Comparison of the servation, per working or pare facility queried (Manual). Image: UMK VMK VMK 20.00 Image: Comparison of the servation, per working or pare facility queried (Manual). Image: Comparison of the servation, per working or pare facility queried (Manual). Image: Comparison of the servation, per working or pare facility queried (Machanizad). Image: Comparison of the servation, per working or pare facility queried (Machanizad). Image: Comparison of the servation, per working or pare facility queried (Machanizad). Image: Comparison of the servation, per working or pare facility queried (Machanizad). Image: Comparison of the servation, per working or pare facility queried (Machanizad). Image: Comparison of the servation, per working or pare facility queried (Machanizad). Image: Comparison of the servation of th		month			UDLSX	1L5ND	8.38										
LOOP MAKE-UP Image: Coop Makeup Preordering Without Reservation, per working or spare facility queried (Manual). UMK							210.92	451 52	262.04	110.40	92 59		15.66				
Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manua). UMK UMKLW 20.00 20.00 Composition	LOOP MAKE-U				ODLGA	ODLOI	319.03	431.32	203.94	115.45	05.50		15.00				
Loop Makey Producting With Reservation, per spare facility queried (Manua). UMK UMKLP 21.00 100		Loop Makeup - Preordering Without Reservation, per working or			LIMK			20.00	20.00								
queried (Manual). UMK UMK UMK UMK Queried (Manual). Composition of spare facility queried (Mechanized) Composition of spare facility que						ONINEW		20.00	20.00						1		
Image: Space facility queried (Mechanized) UMK PSUMK 0.59 0.59 0.59 0.59 HIGH FREQUENCY SPECTRUM Image: Constraint of the constraint of the		queried (Manual).			UMK	UMKLP		21.00	21.00								
HIGH FREQUENCY SPECTRUM Image: Constraint of the second seco									0								
LINE SHARING Image: Constraint of the state					UMK	PSUMK		0.59	0.59								
SPLITTERS-CENTRAL OFFICE BASED U Image: Constraint of the const																	
Line Sharing Splitter, per System 24 Line Capacity ULS ULSDB 38.99 188.79 0.00 177.98 0.00 15.66 Line Sharing Splitter, per System, 8 Line Capacity I ULS ULSDB 12.73 377.58 0.00 15.66 Image: Comparison of the comparison o				1													
Line Sharing Splitter, Per System, 8 Line CapacityIULSULSD812.73377.580.00355.960.0015.66Line Sharing-DLEC Owned Splitter in CO-CFA activaton- deactivation (per LSOD)ULSULSDG86.470.0049.840.0015.6615.66END USER ORDERING-CENTRAL OFFICE BASED-HIGH FREQUENCY SPECTRUM AKA LINE SHARINGULSULSDC0.6118.5110.6010.014.9215.6616.60Line Sharing - per Line Activation (BST Owned splitter)ULSULSDC0.6118.5110.6010.014.9215.6616.60Line Sharing - per Subsequent Activity per Line Rearrangement(DLEC Owned Splitter)ULSULSDS16.398.1915.6616.6015.66Line Sharing - per Line Activation (DLEC owned Splitter)ULSULSCS16.398.1915.6616.6015.6616.60Line Sharing - per Line Activation (DLEC owned Splitter)ULSULSCS16.398.1915.6616.6016.6016.6016.60Line Sharing - per Line Activation (DLEC owned Splitter)ULSULSCS16.398.1915.6616.6016.6016.6016.6016.60Line Sharing - per Line Activation (DLEC owned Splitter)ULSULSC0.6147.4419.3120.029.8315.6616.6016.6016.60Line Sharing - per Line Activation (DLEC owned Splitter)ULSULSC0.6147.4419.3120.029.8315.6616.6016.60 <td< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></td<>																	
Line Sharing-DLEC Owned Splitter in CO-CFA activaton- deactivation (per LSOD) ULS ULSOG 86.47 0.00 49.84 0.00 15.66 6 END VSER ORDERING-CENTRAL OFFICE BASED-HIGH FREQUENCY SPECTRUM AKA LINE SHARING 0 15.66 0			L														
Image: deactivation (per LSOD) ULS ULSDG 86.47 0.00 49.84 0.00 15.66 Image: constraint of the constraint	├ ── ├ ──	Line Sharing Splitter, Per System, 8 Line Capacity			ULS	ULSU8	12.73	377.58	0.00	355.96	0.00		15.66				┟────┤
END USER ORDERING-CENTRAL OFFICE BASED-HIGH FREQUENCY SPECTRUM AKA LINE SHARING Image: Content of the starting o					ULS	ULSDG		86,47	0.00	49,84	0.00		15.66				1
Line Sharing - per Subsequent Activity per Line Rearrangement(BST Owned Splitter ULS ULSS 16.39 8.19 15.66 15.66 Line Sharing - per Subsequent Activity per Line Rearrangement(DLEC Owned Splitter ULS ULSS 16.39 8.19 15.66 15.66 Line Sharing - per Line Activity per Line Rearrangement(DLEC Owned Splitter) I ULS ULSC 16.39 8.19 15.66 Line Sharing - per Line Activity on Line Rearrangement(DLEC Owned Splitter) I ULS ULSC 0.61 47.44 19.31 20.02 9.83 15.66 LINE SPLITTING I <td>END U</td> <td></td> <td>Y SPEC</td> <td>TRUM</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>2.50</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	END U		Y SPEC	TRUM							2.50						
Rearrangement(BST Owned Splitter ULS ULSDS 16.39 8.19 15.66 Line Sharing - per Subsequent Activity per Line Rearrangement(DLEC Owned Splitter) ULS ULSCS 16.39 8.19 15.66 Line Sharing - per Line Activation (DLEC owned Splitter) I ULS ULSCS 16.39 8.19 15.66 Line Sharing - per Line Activation (DLEC owned Splitter) I ULS ULSCC 0.61 47.44 19.31 20.02 9.83 15.66 LINE SPLITTING Image: Comparison of the transmit of					ULS	ULSDC	0.61	18.51	10.60	10.01	4.92		15.66				
Line Sharing - per Subsequent Activity per Line Rearrangement(DLEC Owned Splitter ULS ULSCS 16.39 8.19 15.66 Line Sharing - per Line Activation (DLEC owned Splitter) I ULS ULSCC 0.61 47.44 19.31 20.02 9.83 15.66 LINE SPLITTING I ULS ULSCC 0.61 47.44 19.31 20.02 9.83 15.66 END USER ORDERING-CENTRAL OFFICE BASED I					ULS	ULSDS		16.39	8.19				15.66				
Line Sharing - per Line Activation (DLEC owned Splitter) I ULSC 0.61 47.44 19.31 20.02 9.83 15.66 LINE SPLITTING I ULSC 0.61 47.44 19.31 20.02 9.83 15.66 Image: Control of the second sec		Line Sharing - per Subsequent Activity per Line															[
LINE SPLITTING			<u> </u>	<u> </u>								ļ			ļ		ļ
END USER ORDERING-CENTRAL OFFICE BASED					ULS	ULSCC	0.61	47.44	19.31	20.02	9.83		15.66				├ ───┤
												<u> </u>			<u> </u>		<u> </u>
Line Splitting - per line activation DLEC owned splitter I UEPSR UEPSB UREOS 0.61		Line Splitting - per line activation DLEC owned splitter	1	1	UEPSR UEPSB	UREOS	0.61					t			<u> </u>		

UNBUI	NDLE	D NETWORK ELEMENTS - Alabama			r									Attachment:			ibit: B
CATEG	ORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Charge -	Charge -
							Rec	Nonrec		Nonrecurring					Rates(\$)		
								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Line Splitting - per line activation BST owned - physical Line Splitting - per line activation BST owned - virtual			UEPSR UEPSB UEPSR UEPSB	UREBP UREBV	0.61	37.01 37.01	21.19 21.19	20.02 20.02	9.83 9.83		15.66 15.66				
	REMOT	E SITE HIGH FREQUENCY SPECTRUM	1		UEFSK UEFSD	UREDV	0.61	37.01	21.19	20.02	9.03		15.00				
		ERS-REMOTE SITE															
	-	Remote Site Line Share BellSouth Owned Splitter, 24 Port	I		ULS	ULSRB	38.18	221.09	0.00	254.79	0.00		15.66				
		Remote Site Line Share Cable Pair Activation CLEC Owned at															
		RS and Deactivation	I		ULS	ULSTG		74.38	0.00	46.77	0.00		15.66				
	END US	SER ORDERING-REMOTE SITE HIGH FREQUENCY SPECTRU	M AKA	REMOT	E SITE LINE SHAR	ING											
		Remote Site Line Share Line Activationfor End User Served at RS, BST Splitter			ULS	ULSRC	0.61	37.01	21.19	20.02	9.83		15.66				
		RS Line Share Line Activation for End User served at RS, CLEC	-		ULS	ULSKU	0.01	37.01	21.19	20.02	9.65		15.00				
		Splitter	1		ULS	ULSTC	0.61	37.01	21.19	20.02	9.83		15.66				
UNBUN	DLED D	DEDICATED TRANSPORT															
		INTEROFFICE CHANNEL DEDICATED TRANSPORT - minimu	m billin	g perio	d - below DS3=one	month, DS3/	STS-1=four mor	nths									
_	INTERC	OFFICE CHANNEL - DEDICATED TRANSPORT															
		Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade -															
		Per Mile per month			U1TVX	1L5XX	0.008838										
		Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade - Facility Termination			U1TVX	U1TV2	21.13	40.54	27.41	16.74	6.90		15.66				
		Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade				01172	21.13	40.34	27.41	10.74	0.90		15.00				
		Rev Bat Per Mile per month			U1TVX	1L5XX	0.008838										
		Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat	-		0.11.11	120/01	0.000000										
		Facility Termination			U1TVX	U1TR2	21.13	40.54	27.41	16.74	6.90		15.66				
		Interoffice Channel - Dedicated Transport - 4-Wire Voice Grade -	-														
		Per Mile per month			U1TVX	1L5XX	0.008838										
		Interoffice Channel - Dedicated Transport - 4- Wire Voice Grade					10 70	10 5 1									
		- Facility Termination Interoffice Channel - Dedicated Transport - 56 kbps - per mile			U1TVX	U1TV4	18.73	40.54	27.41	16.74	6.90		15.66				
		per month			U1TDX	1L5XX	0.008838										
		Interoffice Channel - Dedicated Transport - 56 kbps - Facility			OTIBA	TE3///	0.000000										
		Termination			U1TDX	U1TD5	15.12	40.54	27.41	16.74	6.90		15.66				
		Interoffice Channel - Dedicated Transport - 64 kbps - per mile			-	1											
		per month			U1TDX	1L5XX	0.008838										
		Interoffice Channel - Dedicated Transport - 64 kbps - Facility															
		Termination			U1TDX	U1TD6	15.12	40.54	27.41	16.74	6.90		15.66				
		Interoffice Channel - Dedicated Channel - DS1 - Per Mile per			U1TD1	1L5XX	0.18										
		month Interoffice Channel - Dedicated Tranport - DS1 - Facility			UTIDI	ILSAA	0.16			-							
		Termination			U1TD1	U1TF1	60.16	89.27	81.81	16.35	14.44		15.66				
		Interoffice Channel - Dedicated Transport - DS3 - Per Mile per			-	-											
		month			U1TD3	1L5XX	4.09										
		Interoffice Channel - Dedicated Transport - DS3 - Facility															
		Termination per month			U1TD3	U1TF3	703.52	278.75	162.76	60.20	58.46		15.66				
		Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per			114704	41 5307	4.00										
		month Interoffice Channel - Dedicated Transport - STS-1 - Facility			U1TS1	1L5XX	4.09										
		Termination			U1TS1	U1TFS	701.37	278.75	162.76	60.20	58.46		15.66				
		CHANNEL - DEDICATED TRANSPORT			01101	01110	101.01	210.10	102.10	00.20	00.40		10.00				
		LOCAL CHANNEL DEDICATED TRANSPORT - minimum billin	g perio	d - belo	w DS3=one month	DS3/STS-1=	four months							1	1		
		Local Channel - Dedicated - 2-Wire Voice Grade			ULDVX	ULDV2	13.97	193.10	33.17	36.64	3.20		15.66				
		Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat			ULDVX	ULDR2	13.97	193.10	33.17	36.64	3.20		15.66				
		Local Channel - Dedicated - 4-Wire Voice Grade		<u>.</u>	UNDVX	ULDV4	14.93	193.53	33.60	27.11	3.67	ļ	15.66				l
		Local Channel - Dedicated - DS1 - Zone 1 Local Channel - Dedicated - DS1 - Zone 2		1	ULDD1	ULDF1	35.76 49.98	177.47	153.72	22.19	15.26		15.66				<u> </u>
\vdash		Local Channel - Dedicated - DS1 - Zone 2 Local Channel - Dedicated - DS1 - Zone 3		2	ULDD1 ULDD1	ULDF1 ULDF1	49.98	177.47 177.47	153.72 153.72	22.19 22.19	15.26 15.26		15.66 15.66				
		Local Channel - Dedicated - DS1 - 2016 3 Local Channel - Dedicated - DS3 - Per Mile per month		3	ULDD3	1L5NC	6.92	(77.47	155.72	22.19	15.20	-	10.00				
		Local Channel - Dedicated - DS3 - Fei Mile per month	-		ULDD3	ULDF3	416.54	451.52	463.94	119.49	83.58	1	15.66				
		Local Channel - Dedicated - STS-1- Per Mile per month			ULDS1	1L5NC	6.92				22100					İ	
		Local Channel - Dedicated - STS-1 - Facility Termination	1	1	ULDS1	ULDFS	408.49	451.52	463.94	119.49	83.58	1	15.66	ĺ	İ	l	

CATEGON BATE BLEMENTS Intel D Zone BCS UNC FRATESION Boothood Sciences (Based) Boothood Sciences (Based) Boothood Sciences (Based) Diversion Sciences (Based) <thdiversion sciences<br="">(Based) Diversi</thdiversion>	UNBUNDLE	D NETWORK ELEMENTS - Alabama												Attachment:	2	Exhi	bit: B
CAREGOM AND E LEIRENTS Image And E	ONDONDEE											Svc Order	Svc Order				1
LATE PLANENT Non Res Non Procession "><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>Charge -</td></t<>																	Charge -
CLT EXP RATE ELEMENTs Non 200 BC USC VICE			Indani														
Image: Problem in the second problem in the	CATEGORY	RATE ELEMENTS		Zone	BCS	USOC			RATES(\$)								Order vs.
Image: constraint of the second se			m									percon	per Loix				Electronic-
Image: Construction Image: Construction																	Disc Add'l
Image: Process of the second part of the second																Disc Tst	DISC AUU I
DARK FIRST Image							Rec	Nonrec	curring	Nonrecurring	g Disconnect						
Day Flow: Day Flow: Performance <							Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Interest per method User >DARK FIBER</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	DARK FIBER																
INP C bick Filer-Load Channel USP USP 172.4 197.85																	i
Date Flaw, Four Flaw Strate, For Road Web P Flaxon International Strate Provided Strat							60.32										
Interface Theory per conv. stretchic Quantal UDF 11.00F 22.24 0 1 0 10 MCC Mark Tear Interface UDF UDF 4 68.00 02.01 317.00 11.60 11.60 11.60 WCC Mark Tear Interface UDF UDF 0.02 0.02 12.00 11.60					UDF	UDFC4		639.09	137.87	317.06	197.66		15.66				
NRC Dia Floar - Insurfix Oran 4 UDF																	i
Dark Flaw, Faur Rele Stands, Per Role Mar Grastion UCP LLOL 00.2 Image of a month - local loop Image of a month - local loop BXX Access Ten Dig Steering, Per Coll DAP UPP-1 00000 127.87 317.66 127.87 <td< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td>22.34</td><td>000.00</td><td>407.07</td><td>047.00</td><td>407.00</td><td></td><td>45.00</td><td></td><td></td><td></td><td>ł</td></td<>							22.34	000.00	407.07	047.00	407.00		45.00				ł
There of per month Local Loop UOP 11.50L 40.22 Image					UDF	UDF14		639.09	137.87	317.06	197.66		15.66				ł
NRC Dak Rev Log DP UDF UDF UDF 06300 197.47 377.06 197.46 15.66 NACCESS TR MDF SCREEMA NUMBER SCREEMAN, INFORMA, INFORMAN, INFORMA, INFORMA, INFORMAN, INFORMAN, INFORMA, INFORMAN, INFORMAN, INFO							co 00										i
BitX ACCESS TON DIGT SCREENNAL Image: Control of the screen							00.32	630.00	107.07	217.06	107.66		15.66				i
BXX Access True Dig Streaming, Per Gall OHD 0.0006 Image: Control of	ANY ACCESS				UDF	UDFL4		639.09	137.87	317.06	197.00		15.66				i
BXX Access Tem Digit Streening, Pare XX No. Established WU0 OH Net X 2.06 0.44 10.66 BXX Access Tem Digit Streening, Pare XX No. Established WU0 OHD Not X 5.84 0.81 4.57 0.64 15.66 Image: Construction of the stablished WU0	UNA ACCESS				ОНО	1	0 00056										
Number Reserved OHO NBR1X 2.58 0.44 10.66 Image: Constraint of the constraint of the						1	0.00000										
BXX Access Tran Dig Screening, PEXX No. Established W/W OHD NBTX 5.94 0.01 4.57 0.64 15.66 BXX Access Tran Dig Screening, Detroxing Area of Service OHD NBTX 5.94 0.81 4.57 0.54 15.66 Image: Constraint of Service of			1	1	ОНД	N8R1X		2 58	0.44				15.66				1
Image: POTS Transitions OHD Model 5.84 0.81 4.87 0.64 15.66 Image: POTS Transitions OHD NPTX 5.94 0.81 4.87 0.54 15.66 Image: PotS Transitions Image: PotS Transitions OHD NPTX 5.94 0.81 4.57 0.54 15.66 Image: PotS Transitions Image: PotS Transitions OHD NPTX 5.94 0.81 4.57 0.54 15.66 Image: PotS Transitions Image: PotS Transitions OHD NPTX 5.94 0.81 4.57 0.54 15.66 Image: PotS Transitions Image: PotS Transitions OHD NPTX 3.02 1.7 Image: PotS Transitions Image	<u> </u>		1	-	0.10			2.00	0.44				10.00				
BXA Access Tam Digit Screening, Per 8X No. Established With POT Smallakors Ord NBFTX 5.64 0.81 4.57 0.54 15.66 BXA Access Tam Digit Screening, Mulpite Net ATA CXR Routing Per CXR Requested Per 8X No. Ord NBFTX 5.64 0.81 4.57 0.54 15.66 Image: Constraint Mulpite Net ATA CXR Routing Per CXR Requested Per 8X No. 0.40 NBFTX 3.02 0.44 15.66 Image: Constraint Mulpite Net ATA CXR Routing Per CXR Requested Per 8X No. 0.40 NBFTX 3.02 0.44 15.66 Image: Constraint Mulpite Net ATA CXR Routing Per CXR Requested Per 8X No. 0.40 NBFTX 3.02 0.44 15.66 Image: Constraint Mer 41 A CXR Routing Per CXR Requested Per 8X No. 0.40 NBFTX 3.02 0.44 15.66 Image: Constraint Mer 41 A CXR Routing Per CXR Requested Per 8X No. 0.44 15.66 Image: Constraint Mer 41 A CXR Routing Per CXR Requested Per 8X No. 0.44 15.66 Image: Constraint Mer 41 A CXR Routing Per CXR Requested Per 8X No. 0.000565 Image: Constraint Mer 41 A CXR Routing Per CXR Requested Per 8X No. 0.000565 Image: Constraint Mer 41 A CXR Routing Per 8X No. 0.000565 Image: Constraint Mer 41 A CXR Routing Per 8X No. 0.000565 Image: Constraint M			1	1	OHD			5 94	0.81	4 57	0.54		15.66				1
POTS Translations OHD NBTX 5.94 0.61 4.57 0.54 15.66 BXA Access Ten Digt Screening, Unkliph Intel AT XOR OHD NBFCX 2.66 1.29 15.66 15.66 BX Access Ten Digt Screening, Carbon Tento Market OHD NBFCX 3.02 1.73 15.66 16.68 BX Access Ten Digt Screening, Carbon Tentoming and Destination OHD NBFCX 2.08 15.66 16.66 BX Access Ten Digt Screening, Carbon Tentoming and Destination OHD NBFCX 2.08 16.66 16.66 BX Access Ten Digt Screening, Carbon Tentoming and Destination OHD 0.000565 16.66 16.66 16.66 LINE NFORMATION DATA ASEE ACCESS (LINP) OHD 0.000565 16.66 16.66 16.66 16.66 SIGNALINK (CS) OGU NPRSX 3.02 4.20 16.66 16.66 16.66 16.66 16.66 16.66 16.66 16.66 16.66 16.66 16.66 16.66 16.66 16.66 16.66 16.66 16.66 16.66						1		0.04	0.01		5.04		.0.00		1	1	1
BXX Access Tim Dgit Screening, Cutatorized Area of Service OHD NSFCX 2.58 1.29 15.66 BXX Access Tim Dgit Screening, Multiple InterLATACXR OHD NSFXX 3.02 1.73 15.66 15.66 BXX Access Tim Dgit Screening, Call Handling and Destination OHD NSFXX 3.02 1.73 15.66 15.66 BXX Access Tim Dgit Screening, Call Handling and Destination OHD NSFXX 3.02 0.44 15.66 15.66 BXX Access Tim Dgit Screening, W ST, No. Delvey OHD 0.000585 15.66 15.66 UNE INFORMATION DATA BASE ACCESS (LDB) OT 0.00002 15.66 16.66 16.66 16.67 17.64 16.66 16.66 16.67 16.66 16.67 16.67 16.67 16.67 16.67 16.67 16.67 16.67 16.67 16.68 16.67 <td></td> <td></td> <td></td> <td></td> <td>OHD</td> <td>N8FTX</td> <td></td> <td>5.94</td> <td>0.81</td> <td>4.57</td> <td>0.54</td> <td></td> <td>15.66</td> <td></td> <td></td> <td></td> <td>1</td>					OHD	N8FTX		5.94	0.81	4.57	0.54		15.66				1
Pre BXX Number OHD NFCX 2.58 1.29 15.66 15.66 BXX Access Ten Digl Screening, Charge Durge Pur Request OHD NBFAX 3.02 1.73 15.66 15.66 BXX Access Ten Digl Screening, Charge Durge Pur Request OHD NBFAX 3.02 0.44 15.66 15.66 BXX Access Ten Digl Screening, Call Handling and Destination OHD NBFAX 3.02 0.44 15.66 15					-	-				-							[
Bit X Access Ten Digli Screening, Multiple Intel LATA CXR OHD NEFAX 3.02 1,73 15.66 15.66 Bit X Access Ten Digli Screening, Charge Charge PR equest OHD NBFAX 3.02 0.44 15.66 15.66 Bit X Access Ten Digli Screening, Charge Charge PR equest OHD NBFAX 2.58 15.66 15.66 15.66 Bit X Access Ten Digli Screening, VPC Ten No. Delavery OHD 0.000565 16.66 15.66 16.66 Bit X Access Ten Digli Screening, VPC Ten No. Delavery OHD 0.000565 16.66 16.66 16.66 UBE VerCMATCM DATA BASE ACCESS (UB9) OHD 0.0002 16.66 16.66 16.66 UBD Validator Per Courty OQU NO 15.66 16.66 16.66 CCSY Signaling Tennation. Per StRX ps Facility 16.47 15.46 35.53 35.53 16.44 16.64 16.66 16.66 CCSY Signaling Tennation. Per StRX ps Facility 10.08 17.64 35.53 35.53 16.44 16.44 15.66 16.44 16.44 16.66 16.44					OHD	N8FCX		2.58	1.29				15.66				i
Image: CRR Requested Per SXN. OHD NRFLX 3.02 1.73 Image: CRR Requested Per Request OHD NRFLX 3.02 1.73 Image: CRR Requested Per Request OHD NRFLX 3.02 1.73 Image: CRR Requested Per Request OHD NRFLX 3.02 0.44 Image: CRR Requested Per Request OHD NRFLX 3.02 0.44 Image: CRR Requested Per Request OHD NRFLX 3.02 0.44 Image: CRR Requested Per Request OHD NRFLX 3.02 0.44 Image: CRR Requested Per Request OHD NRFLX 3.02 0.44 Image: CRR Requested Per Request OHD NRFLX 3.02 0.45 Image: CRR Requested Per Request OHD NRFLX 3.02 0.00000000000000000000000000000000000		8XX Access Ten Digit Screening, Multiple InterLATA CXR															
BXX Access Ten Dig Screening, Call Handling and Destination Features OHD NEFX 2.58 15.66 15.66 BXX Access Ten Dig Screening, WFL No. Delwey OHD 0.000565					OHD	N8FMX		3.02	1.73				15.66				i
Image: Construction Open of the particular set of the partin set of the particular set of the partin set of the pa		8XX Access Ten Digit Screening, Change Charge Per Request			OHD	N8FAX		3.02	0.44				15.66				[
BXX Access Ten Digit Screening, wi FL: ND. Elevery OHD 0.000565 Image: Construction of the C		8XX Access Ten Digit Screening, Call Handling and Destination															í
BXX Access Ten Digit Screening, w POTS No. Delivery OHD 0.000565 <th< td=""><td></td><td></td><td></td><td></td><td></td><td>N8FDX</td><td></td><td>2.58</td><td></td><td></td><td></td><td></td><td>15.66</td><td></td><td></td><td></td><td><u> </u></td></th<>						N8FDX		2.58					15.66				<u> </u>
LINE RPGRMATION DATA BASE ACCESS QLIDB) Image of the property of the p																	ļ
LUB Common Transport Per Query OOT 0.00002 Image: Common Transport Per Query OOU 0.000002 LUBB Validation Per Query OOU 0.012002 4 1 1 1 StankLink (CCS7) OCU NRPBX 34.32 42.08 15.66 1 1 CCS7 Signaling Connection, Per 56Kbps Facility Image: CCS7 1 15.46 35.53 35.53 16.44 16.44 15.66 1 1 CCS7 Signaling Connection, Per SK Message Image: CCS7 Mark Message Image: CCS7 Signaling Connection, Per link (A link) UDB 1					OHD		0.000565										I
LUB Validation Per Query OQU 0.012002 - <	LINE INFORM																
LUBB Originating Point Code Establishment or Change OQT, OQU NRPBX 34.32 44.08 15.66 SIGNALURS (CS7) I																	
SIGENALING (CCS7) CCS7 Signaling Connection, Per 56kbps Faciliy Image: Concent on Per 56kbps Faciliy Image: CCS7 Signaling Connection, Per 51P Port Image: CCS7 Signaling Connection, Per 51P Port Image: CCS7 Signaling Connection, Per 10k A link) I						NEEDY	0.012002			10.00			15.00				l
CCS7 Signaling Connection, Per Str Port UDB PT85X 130.83 35.53 16.44 16.66 Image: Constraint of the str Port CCS7 Signaling Usage, Per Call Setup Message UDB 0.0000142 Image: Constraint of the str Port Image: Constr Port Image: Constraint of the str Por					OQT, OQU	NRPBX		34.32		42.08			15.66				ł
CCS7 Signaling Termination. Per STP Port UDB PT85X 130.83 Image: CCS7 Signaling Usage, Per CAP Message Image: CCS7 Signaling Connection, Per link (also known as D Image: CCS7 Signaling Connection, Per link (also known as D Image: CCS7 Signaling Connection, Per link (also known as D Image: CCS7 Signaling Usage, Per (SUP Message Image: CCS7 Signaling Usage Sure (SUP Message Image: CCS7 Signaling Usage Sure	SIGNALING (C					_	45.40	25.52	25.52	40.44	40.44		45.00				ł
Image: CCS7 Signaling Usage, Per Call Setup Message UDB 0.0000142 Image: CCS7 Signaling Usage, Per Call Setup Message Image: CCS7 Signaling Connection, Per link (A link) Image: CCS7 Signaling Connection, Per link (B link) (also known as D link) Image: CCS7 Signaling Connection, Per link (B link) (also known as D link) Image: CCS7 Signaling Usage, Per ISUP Message Image: CCS7 Signaling Point Code Image: CCS7 Signaling Point Co						DTOEV		30.03	30.03	16.44	16.44		15.66				i
CCS7 Signaling Usage, Per TCAP Message UDB 0.0000569					UDB	F 103A											<u> </u>
CCS7 Signaling Connection, Per link (A link) UDB TPP++ 15.46 35.53 35.53 16.44 16.44 15.66 Image: Connection of the connecone of the connection of the connection of the connectic					אַסו	-											i
CCS7 Signaling Connection, Per link (B link) (also known as D UDB TPP+++ 15.46 35.53 36.53 16.44 16.44 15.66 Image: Constraint of the constrai	<u>├</u> ──┤───					TPP++		25 52	35 53	16 //	16 //		15.66				
link/ UDB TPP++ 15.46 35.53 35.53 16.44 16.44 15.66 Image: Case Signaling Usage Surrogate, per link per LATA UDB STU56 650.33 Image: Case Signaling Usage Surrogate, per link per LATA UDB STU56 650.33 Image: Case Signaling Usage Surrogate, per link per LATA UDB STU56 650.33 Image: Case Signaling Usage Surrogate, per link per LATA UDB STU56 650.33 Image: Case Signaling Usage Surrogate, per link per LATA UDB STU56 650.33 Image: Case Signaling Usage Surrogate, per link per LATA Image: Case Signaling Usage Surrogate, per link per LATA Image: Case Signaling Usage Surrogate, per link per LATA Image: Case Signaling Usage Surrogate, per link per LATA Image: Case Signaling Usage Surrogate, per link per LATA Image: Case Signaling Usage Surrogate, per link per LATA Image: Case Signaling Usage Surrogate, per link per LATA Image: Case Signaling Usage Surrogate, per link per LATA Image: Case Signaling Usage Surrogate, per link per LATA Image: Case Signaling Usage Surrogate, per link per LATA Image: Case Signaling Usage Surrogate, per link per LATA Image: Case Signaling Usage Surrogate, per link per LATA Image: Case Signaling Usage Surrogate, per link per LATA Image: Case Signaling Usage Surrogate, per link per LATA Image: Case Signaling Usage Surrogate, per link per LATA <t< td=""><td><u>├</u>──┤───</td><td></td><td> </td><td></td><td>000</td><td>11.1.97</td><td>10.40</td><td>55.55</td><td>55.55</td><td>10.44</td><td>10.44</td><td> </td><td>10.00</td><td></td><td></td><td></td><td></td></t<>	<u>├</u> ──┤───				000	11.1.97	10.40	55.55	55.55	10.44	10.44		10.00				
CC37 Signaling Usage, Per ISUP Message UDB 0.0000142 Image: Constraint of the cons			1	1	UDB	TPP++	15 46	35.53	35.53	16 44	16 44		15.66				1
CCS7 Signaling Usage Surrogate, per Ink per LATA UDB STU56 660.33 Image: CCS7 Signaling Point Code, per Originating Point Code Image: CCS7 Signaling Point Code, per Originating Point Code Image: CCS7 Signaling Point Code, per Originating Point Code Image: CCS7 Signaling Point Code, per Originating Point Code Image: CCS7 Signaling Point Code, per Originating Point Code Image: CCS7 Signaling Point Code, per Originating Point Code Image: CCS7 Signaling Point Code Image: CCS7 Signaling Point Code, per Originating Point Code Image: CCS7 Signaling Point Code Image: CCS7 Signaling Point Code, per Originating Point Code Image: CCS7 Signaling Point Code Image: CCS7 Sig		/	1	1		1		00.00	00.00	10.44	10.44	1	10.00				
CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affected UDB CCAPO 29.01 29.01 35.57 15.66 E911 SERVICE <			1			STU56				1	1			1	1	1	
Establishment or Change, per STP affected UDB CCAPO 29.01 29.01 35.57 35.57 15.66 Image: Constraint of the constraint of th			1			2.000	300.00			1	1			1	1	1	
E911 SERVICE Image: Channel - Dedicated - 2-wr Voice Grade Image: Channel - Dedicated - 2-wr Voice Grade Per Mile Image: Channel - Dedicated - 2-wr Voice Grade Per Mile Image: Channel - Dedicated - 2-wr Voice Grade Per Mile Image: Channel - Dedicated - 2-wr Voice Grade Per Facility Image: Channel - Dedicated - 2-wr Voice Grade Per Facility Image: Channel - Dedicated - 2-wr Voice Grade Per Facility Image: Channel - Dedicated - 2-wr Voice Grade Per Facility Image: Channel - Dedicated - 2-wr Voice Grade Per Facility Image: Channel - Dedicated - 2-wr Voice Grade Per Facility Image: Channel - Dedicated - DS1 - Zone 1 Image: Channel - Dedicated - DS1 - Zone 1 Image: Channel - Dedicated - DS1 - Zone 2 Image: Channel - Dedicated - DS1 - Zone 2 Image: Channel - Dedicated - DS1 - Zone 2 Image: Channel - Dedicated - DS1 - Zone 2 Image: Channel - Dedicated - DS1 - Zone 3 Image: Channel - Dedicated - DS1 - Zone 3 Image: Channel - Dedicated - DS1 - Zone 3 Image: Channel - Dedicated - DS1 - Zone 3 Image: Channel - Dedicated - DS1 - Zone 3 Image: Channel - Dedicated - DS1 - Zone 3 Image: Channel - Dedicated - DS1 - Zone 3 Image: Channel - Dedicated - DS1 - Zone 3 Image: Channel - Dedicated - DS1 - Zone 3 Image: Channel - Dedicated - DS1 - Zone 3 Image: Channel - Dedicated - DS1 - Zone 3 Image: Channel - DEdicated - DS1 - Zone 3 Image: Channel - DEdicated - DS1 - Zone 3 Image: Channel - DS1 - Zone 3 Image: Channel - DS1 - Zone 3 Image: Channel - DS1 - Zone 3		Establishment or Change, per STP affected	1	1	UDB	CCAPO		29.01	29.01	35.57	35.57		15.66				1
Local Channel - Dedicated - 2-wr Voice Grade Per Mile 13.97 193.10 33.17 36.64 3.20 15.66 Image: Constraint of the constra	E911 SERVICE											1			İ	İ	
Interoffice Transport - Dedicated - 2-wr Voice Grade Per Mile 0.008838 c <thc< th=""> c <thc< th=""> c c c</thc<></thc<>							13.97	193.10	33.17	36.64	3.20		15.66				[
Interoffice Transport - Dedicated - 2-wr Voice Grade Per Facility Termination 21.13 40.54 27.41 16.74 6.90 15.66 Image: Constraint of the second se							0.008838										
Local Channel - Dedicated - DS1 - Zone 1 35.76 177.47 153.72 22.19 15.26 15.66 15.66 Local Channel - Dedicated - DS1 - Zone 3 49.98 177.47 153.72 22.19 15.26 15.66 <t< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>Í</td></t<>																	Í
Local Channel - Dedicated - DS1 - Zone 2 49.98 177.47 153.72 22.19 15.26 15.66 1 Local Channel - Dedicated - DS1 - Zone 3 107.63 177.47 153.72 22.19 15.26 15.66 1 Interoffice Transport - Dedicated - DS1 Per Mile 0.18 1 <td< td=""><td></td><td>Termination</td><td></td><td></td><td></td><td></td><td>21.13</td><td></td><td></td><td></td><td></td><td></td><td>15.66</td><td></td><td></td><td></td><td>L</td></td<>		Termination					21.13						15.66				L
Local Channel - Dedicated - DS1 - Zone 3 107.63 177.47 153.72 22.19 15.66 Image: Constraint of the cons																	
Interoffice Transport - Dedicated - DS1 Per Mile 0 0.18 0.18 0																	
Interoffice Transport - Dedicated - DS1 Per Facility Termination 60.16 89.27 81.81 16.35 14.44 15.66 60.16 60.16 CALLING NAME (CNAM) SERVICE 0 <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>177.47</td> <td>153.72</td> <td>22.19</td> <td>15.26</td> <td></td> <td>15.66</td> <td></td> <td></td> <td></td> <td>ļ</td>								177.47	153.72	22.19	15.26		15.66				ļ
CALLING NAME (CNAM) SERVICE O O O O O CNAM For DB Owners - Service Establishment OQV 22.95 21.11 O O CNAM For DB Owners - Service Establishment OQV 22.95 21.11 O O CNAM For DB Owners - Service Porvisioning With Point Code OQV 22.95 21.11 O O		Interoffice Transport - Dedicated - DS1 Per Mile					0.18										ļ
CALLING NAME (CNAM) SERVICE O O O O O CNAM For DB Owners - Service Establishment OQV 22.95 21.11 O O CNAM For DB Owners - Service Establishment OQV 22.95 21.11 O O CNAM For DB Owners - Service Porvisioning With Point Code OQV 22.95 21.11 O O			_														1
CNAM For DB Owners - Service Establishment OQV 22.95 21.11 Image: Constraint of the constraint of the							60.16	89.27	81.81	16.35	14.44		15.66				I
CNAM For Non DB Owners - Service Establishment OQV 22.95 21.11 CNAM For DB Owners - Service Provisioning With Point Code <t< td=""><td>CALLING NAM</td><td>IE (CNAM) SERVICE</td><td>I</td><td>I</td><td>0.01/</td><td></td><td></td><td></td><td></td><td></td><td></td><td> </td><td></td><td></td><td></td><td></td><td>I</td></t<>	CALLING NAM	IE (CNAM) SERVICE	I	I	0.01/												I
CNAM For DB Owners - Service Provisioning With Point Code		CNAM For DB Owners - Service Establishment	<u> </u>									I					ł
	┝──┤───		<u> </u>		UQV			22.95		21.11		I					
		CNAM For DB Owners - Service Provisioning With Point Code Establishment	1	1	OQV			990.88	732.84	268.93	197.74						1

	1	1														bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs.	Charge - Manual Svc Order vs. Electronic- Add'l	Charge -	Charge - Manual Svo Order vs.			
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	CNAM For Non DB Owners - Service Provisioning With Point			001/			0.40.00	045.44	075.05	107.71						
	Code Establishment CNAM for DB Owners, Per Query			OQV OQV	-	0.000902	342.33	245.14	275.25	197.74						
	CNAM for Non DB Owners, Per Query			OQV		0.000902										
LNP Query Ser						0.000302										
-iti daeiy ooi	LNP Charge Per query					0.000757										
	LNP Service Establishment Manual					0.000101	12.52		11.51			15.66				
	LNP Service Provisioning with Point Code Establishment						593.49	303.20	268.93	197.74		15.66				
	ALL PROCESSING															
	Oper. Call Processing - Oper. Provided, Per Min Using BST															
	LIDB					1.20										
	Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIDB					1.24										
	Oper. Call Processing - Fully Automated, per Call - Using BST LIDB					0.20										
	Oper. Call Processing - Fully Automated, per Call - Using Foreign LIDB					0.20										
	ATOR SERVICES					0.20										
	Inward Operator Services - Verification, Per Minute					1.15										
	Inward Operator Services - Verification and Emergency Interrupt - Per Minute					1.15										
BRANDING - C	PERATOR CALL PROCESSING					1.10										
	based CLEC															
	Recording of Custom Branded OA Announcement				CBAOS		7,000.00	7,000.00				15.66				
	Loading of Custom Branded OA Announcement per shelf/NAV per OCN				CBAOL		500.00	500.00				15.66				
UNEP (CLEC															
	Recording of Custom Branded OA Announcement						7,000.00	7,000.00				15.66				
	Loading of Custom Branded OA Announcement per shelf/NAV per OCN						500.00	500.00				15.66				
	ding via OLNS for UNEP CLEC															
	Loading of OA per OCN (Regional)						1,200.00	1,200.00				15.66				
DIRECTORY A	SSISTANCE SERVICES															
	TORY ASSISTANCE ACCESS SERVICE															
	Directory Assistance Access Service Calls, Charge Per Call					0.275										
DIRECT	FORY ASSISTANCE CALL COMPLETION ACCESS SERVICE (E	DACC)														
	Directory Assistance Call Completion Access Service (DACC), Per Call Attempt					0.10										
	ER SERVICES INTERCEPT ACCESS SERVICE			<u> </u>	+	0.10							-		ł	
	SSISTANCE SERVICES				1											
	TORY ASSISTANCE DATA BASE SERVICE (DADS)	1		1	1										1	
	Directory Assistance Data Base Service Charge Per Listing					0.04										
	Directory Assistance Data Base Service, per month	1			DBSOF	150.00										1
	IRECTORY ASSISTANCE															
Facility	Based CLEC															
	Recording and Provisioning of DA Custom Branded Announcement			AMT	CBADA		6,000.00	6,000.00				15.66				
	Loading of Custom Branded Announcement per Switch			AMT	CBADC		1,170.00	1,170.00				15.66				
UNEP C																
	Recording of DA Custom Branded Announcement						3,000.00	3,000.00				15.66				
	Loading of DA Custom Branded Announcement per Switch per OCN						1,170.00	1,170.00				15.66				
	ding via OLNS for UNEP CLEC															
	Loading of DA per OCN (1 OCN per Order)						420.00	420.00				15.66				
	Loading of DA per Switch per OCN	l					16.00	16.00				15.66				
SELECTIVE RO																
	Selective Routing Per Unique Line Class Code Per Request Per Switch				USRCR		84.70	84.70	14.11	14.11		15.66				

UNBUNDLE	D NETWORK ELEMENTS - Alabama											1	Attachment:			ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs.	Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -			
					1	Rec	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Virtual Collocation - Application Cost			AMTFS	EAF		1,205.26	1,205.26	0.51	0.51		15.66				
	Virtual Collocation - Cable Installation Cost, per cable			AMTFS	ESPCX		859.71	859.71	22.49	22.49		15.66				
	Virtual Collocation - Floor Space, per sq. ft.			AMTFS AMTFS	ESPVX ESPAX	3.22 7.83					-					
	Virtual Collocation - Power, per fused amp Virtual Collocation - Cable Support Structure, per entrance			AIVITES	ESPAA	1.03										
	cable			AMTES	ESPSX	14.97										
	Virtual Collocation - 2-wire Cross Connects (loop)			UEANL, UEA, UDN, U DC, UAL, UHL, UCL, U EQ, AMTFS, UDL, UNCVX, UNCDX, UNCNX	UEAC2	0.03	12.30	11.80	6.03	5.44		15.66				
	Virtual Conocation - 2-wire Cross Connects (1000)			UEA,UHL,UCL,UDL, AMTFS, UAL, UDN,	UEAC2	0.03	12.30	11.60	6.03	5.44		15.66				
	Virtual Collocation - 4-wire Cross Connects (loop)			UNCVX, UNCDX	UEAC4	0.05	12.39	11.87	6.39	5.73		15.66				
				AMTFS,UDL12, UDLO3, U1T48, U1T12, U1T03, ULDO3, ULD12,												
	Virtual Collocation - 2-Fiber Cross Connects			ULD48, UDF AMTFS,UDL12,	CNC2F	2.84	20.89	15.20	7.38	5.92		15.66				
	Virtual Collocation - 4-Fiber Cross Connects			UDLO3, U1T48, U1T12, U1T03, ULD03, ULD12, ULD48, UDF USL,ULC,AMTFS, ULR, UXTD1, UNC1X, ULDD1,	CNC4F	5.69	25.55	19.86	9.71	8.25		15.66				
	Virtual collocation - Special Access & UNE, cross-connect per DS1			U1TD1, USLEL, UNLD1	CNC1X	1.11	22.03	15.93	6.40	5.79		15.66				
	Virtual collocation - Special Access & UNE, cross-connect per DS3			UNLD1 USL, ULC, AMTFS, U E3, U1TD3, UXTS1, UXTD3, UNC3X, UNCSX, ULDD3, U1TS1, ULDS1, UDLSX, UNLD3	CND3X	14.16	22.03	15.20	7.38	5.92		15.66				
	Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable			UDL3X, UNLD3	CIND3A	14.10	20.09	15.20	7.30	5.92		15.00				
<u> </u>	Support Structure, per linear foot Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax			AMTFS	VE1CB	0.0026										
<u> </u>	Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable	<u> </u>		AMTFS	VE1CD	0.0038										
⊢	Support Structure,per cable Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax			AMTFS	VE1CC		535.37					15.66				
1	Cable Support Structure, per cable			AMTES	VE1CE		535.37					15.66				
	Virtual Collocation Cable Records - per request	1		AMTES	VE10L		1,518.57	1,518.57	265.99	265.99	<u> </u>	15.66				<u> </u>
	Virtual Collocation Cable Records - VG/DS0 Cable, per cable record	1	1	AMTES	VE1BB		653.83	653.83	378.24	378.24	İ	15.66				
	Virtual Collocaiton Cable Records - VG/DS0 Cable, per each	1	l		1						1					<u> </u>
┟───┤────	100 pair		<u> </u>	AMTES	VE1BC		9.62	9.62	11.79	11.79		15.66				l
┟───┤────	Virtual Collocation Cable Records - DS1, per T1TIE	+		AMTFS AMTFS	VE1BD VE1BE		4.50 15.75	4.50 15.75	5.52 19.32	5.52 19.32	<u>├</u> ───	15.66 15.66		L		───
	Virtual Collocation Cable Records - DS3, per T3TIE Virtual Collocation Cable Records - Fiber Cable, per 99 fiber			AMTES	VE1BE		15.75	15.75						<u> </u>		<u> </u>
┢───┤────	records Virtual collocation - Security Escort - Basic, per half hour			AMTES	VE1BF SPTBX		168.97	168.97	154.25	154.25		15.66 15.66				
<u>├───</u>	Virtual collocation - Security Escort - Dasic, per half hour	1		AMTES	SPTOX		22.05	13.86			-	15.66				
	Virtual collocation - Security Escort - Overtime, per half hour	+		AMTES	SPTPX		22.05	16.98				15.66				<u> </u>
1	Virtual collocation - Maintenance in CO - Basic, per half hour			AMTES	CTRLX	1	27.93	10.33	1 1		1	15.66	1		1	
	Virtual collocation - Maintenance in CO - Overtime, per half hour			AMTFS	SPTOM		36.47	13.86				15.66				

UNBUN	DLE	D NETWORK ELEMENTS - Alabama		-									Svc Order	Attachment:			bit: B
CATEGO	RY	RATE ELEMENTS	Interi m	Zone	BCS	USOC		RATES(\$)						Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
							Rec	Nonrec		Nonrecurring					SS Rates(\$)		
						-		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Virtual collocation - Maintenance in CO - Premium per half hour			AMTFS	SPTPM		45.02	16.98				15.66				1
VIRTUAL	COLL					0		10.02	10.00				10.00				
		Virtual Collocation - 2-wire Cross Connect, Exchange Port 2-															
		Wire Analog - Res			UEPSR	VE1R2	0.03	12.30	11.80	6.03	5.44		15.66				L
		Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-					0.03	12.30	44.00	C 02	E 44		45.00				
		Wire Line Side PBX Trunk - Bus Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire			UEPSP	VE1R2	0.03	12.30	11.80	6.03	5.44		15.66				
		Voice Grade PBX Trunk - Res			UEPSE	VE1R2	0.03	12.30	11.80	6.03	5.44		15.66				
		Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire															
		Analog Bus			UEPSB	VE1R2	0.03	12.30	11.80	6.03	5.44		15.66				L
		Virtual Collocation 2-Wire Cross Connect, Exchnage Port 2-Wire						10.00									
		ISDN Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire			UEPSX	VE1R2	0.03	12.30	11.80	6.03	5.44		15.66				┣────
		ISDN			UEPTX	VE1R2	0.03	12.30	11.80	6.03	5.44		15.66				
		Virtual Collocation 4-Wire Cross Connect, Exchange Port 4-Wire				VEITE	0.00	12.00	11.00	0.00	0.44		10.00				
		ISDN DS1			UEPEX	VE1R4	0.05	12.39	11.87	6.39	5.44		15.66				
VIRTUAL	. COLL	OCATION															
		Virtual Collocation-2 Wire Cross Connects (Loop) for Line				1/541.0	0.00	40.00	44.00	0.00	5 44		45.00				
PHYSICA		Splitting LOCATION			UEPSR, UEPSB	VE1LS	0.03	12.30	11.80	6.03	5.44		15.66				
FITSICA		Physical Collocation-2 Wire Cross Connects (Loop) for Line															<u> </u>
		Splitting			UEPSR, UEPSB	PE1LS	0.03	12.30	11.80	6.03	5.44		15.66				
AIN SELE	ECTIV	E CARRIER ROUTING															
		Regional Service Establishment			SRC	SRCEC		101,098.91		8,590.70			15.66				
		End Office Establishment			SRC	SRCEO		169.88	169.88	1.70	1.70		15.66				L
		Query NRC, per query			SRC		0.002749										
AIN - BEI	LLSOU	JTH AIN SMS ACCESS SERVICE AIN SMS Access Service - Service Establishment, Per State,				-											<u> </u>
		Initial Setup			A1N	CAMSE		39.44	39.44	40.69	40.69		15.66				
		AIN SMS Access Service - Port Connection - Dial/Shared Access			A1N	CAMDP		7.83	7.83	9.09	9.09		15.66				
		AIN SMS Access Service - Port Connection - Dian Shared Access AIN SMS Access Service - Port Connection - ISDN Access			AIN	CAM1P		7.83	7.83	9.09	9.09		15.66				<u> </u>
		AIN SMS Access Service - User Identification Codes - Per User				0,		1.00	1.00	0.00	0.00		10.00				
		ID Code			A1N	CAMAU		35.00	35.00	27.06	27.06		15.66				
		AIN SMS Access Service - Security Card, Per User ID Code,															
		Initial or Replacement			A1N	CAMRC	0.000400	41.88	41.88	11.71	11.71		15.66				ļ
		AIN SMS Access Service - Storage, Per Unit (100 Kilobytes) AIN SMS Access Service - Session, Per Minute				-	0.002188										<u> </u>
		AIN SMS Access Service - Session, Per Minute AIN SMS Access Service - Company Performed Session, Per					0.59										ł
		Minute					0.73										
AIN - BEI	LLSOU	JTH AIN TOOLKIT SERVICE															
		AIN Toolkit Service - Service Establishment Charge, Per State,															
		Initial Setup			CAM	BAPSC		39.44	39.44	40.69	40.69		15.66				L
		AIN Toolkit Service - Training Session, Per Customer				BAPVX		4,202.17	4,202.17				15.66				ł
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt				BAPTT		7.83	7.83	9.09	9.09		15.66				
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per				DAITI		7.05	7.05	3.03	3.03		13.00				<u> </u>
		DN, Off-Hook Delay				BAPTD		7.83	7.83	9.09	9.09		15.66				1
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per															
		DN, Off-Hook Immediate				BAPTM		7.83	7.83	9.09	9.09		15.66				Ļ
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per				PADTO		04 47	04.47	44.00	44.00		45.00				1
├		DN, 10-Digit PODP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per				BAPTO		34.47	34.47	14.36	14.36		15.66				
		DN. CDP				BAPTC		34.47	34.47	14.36	14.36		15.66				1
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per				2/11/0		0-1.47	0-1.47	14.00	14.00		10.00				<u> </u>
		DN, Feature Code				BAPTF		34.47	34.47	14.36	14.36		15.66				L
		AIN Toolkit Service - Query Charge, Per Query					0.05										

UNBUNDLE	D NETWORK ELEMENTS - Alabama												Attachment:	2	Exhi	bit: B
											Svc Order	Svc Order	Incremental		Incremental	
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
		Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svo
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR		Order vs.	Order vs.	Order vs.	Order vs.
											-	-	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
							Name		Nenneermine	- Disseminant			220		iJ	L
						Rec	Nonreo First		Nonrecurring First		COMEC	SOMAN	SOMAN	Rates(\$) SOMAN	SOMAN	SOMAN
	AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit		-				FIrSt	Add'l	FIISt	Add'l	SOMEC	SOWAN	SOMAN	SOMAN	SOWAN	SOWAN
	Subscription, Per Node, Per Query					0.00582								1	1	
	AIN Toolkit Service - SCP Storage Charge, Per SMS Access					0.00302										<u> </u>
	Account, Per 100 Kilobytes					0.05								1	1 1	
	AIN Toolkit Service - Monthly report - Per AIN Toolkit Service															
	Subscription			CAM	BAPMS	10.17	7.83	7.83	5.50	5.50		15.66		1	1	
	AIN Toolkit Service - Special Study - Per AIN Toolkit Service															
	Subscription			CAM	BAPLS	2.87	8.66	8.66				15.66			1	
	AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service														í I	
	Subscription			CAM	BAPDS	7.39	7.83	7.83	5.50	5.50		15.66		<u> </u>		
	AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit														1 7	1 -
	Service Subscription		L	CAM	BAPES	0.10	8.66	8.66				15.66			µ!	
	XTENDED LINK (EELs)		<u> </u>				0.1								µ′	L
	New Density Zone 1 EELs are available in the following MSAs					Atlanta, Ga; Nev	v Orleans, LA,								ļ	
	Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-														<u> </u>	<u> </u>
	In all states, EEL network elements shown below also apply to												UNES.(NON-re	curring rates	do not apply.	<u>) </u>
	In All States the EEL network elements apply to ordinarily cor E VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INT				Itch As Is Cha	arge.) when or	dering ordinar	ily combined i	network elemen	nts, Non-recur	ring rates d	o appiy.				<u> </u>
2-99160	First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport	ERUFF		ANSPORT (EEL)												<u> </u>
	Combination - Zone 1		1	UNCVX	UEAL2	14.38	88.00	55.00	47.24	7.44		15.66		1	1 1	
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed		<u> </u>	UNCVA	ULALZ	14.30	00.00	55.00	47.24	7.44		15.00				
	Transport Combination - Zone 2		2	UNCVX	UEAL2	22.85	88.00	55.00	47.24	7.44		15.66		1	1 1	
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed		-		O E / LE	22.00	00.00	00.00	77.27	7.44		10.00				<u> </u>
	Transport Combination - Zone 3		3	UNCVX	UEAL2	36.14	88.00	55.00	47.24	7.44		15.66		1	1 1	
	Interoffice Transport - Dedicated - DS1 combination - Per Mile		-													
	per month			UNC1X	1L5XX	0.18								1	1 1	
	Interoffice Transport - Dedicated - DS1 combination - Facility															
	Termination per month			UNC1X	U1TF1	60.16	89.27	81.81	16.35	14.44		15.66		1	1 1	
	DS1 Channelization System Per Month			UNC1X	MQ1	107.19	91.04	62.57	10.54	9.79		15.66				
	Voice Grade COCI - DS1 To Ds0 Interface - Per Month			UNCVX	1D1VG	0.56	6.58	4.72				15.66			í l	
	Each Additional 2-Wire VG Loop(SL 2) in the same DS1													1	1 1	
	Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL2	14.38	88.00	55.00	47.24	7.44		15.66			ļ	L
	Each Additional 2-Wire VG Loop(SL2) in the same DS1		_											1	1 1	
	Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL2	22.85	88.00	55.00	47.24	7.44		15.66			ļ]	<u> </u>
	Each Additional 2-Wire VG Loop(SL2) in the same DS1								17.04			15.00		1	1 1	
	Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL2	36.14	88.00	55.00	47.24	7.44		15.66			↓	ł
	Voice Grade COCI - DS1 to DS0 Channel System combination -				1D1VG	0.56	6.58	4.72				45.00		1	1	
	per month Nonrecurring Currently Combined Network Elements Switch -As-			UNCVX	IDIVG	0.00	86.0	4.72				15.66			↓	
	Is Charge			UNC1X	UNCCC		5.59	5.59	6.98	6.98		15.66		1	1 1	
4-WIR	E VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INT	FROFE			UNCCC		5.55	5.59	0.90	0.90		15.00				
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice				1											
	Transport Combination - Zone 1	1	1	UNCVX	UEAL4	25.34	131.97	94.51	59.14	14.50		15.66		1	1 1	1
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice		1								1				l	
	Transport Combination - Zone 2	1	2	UNCVX	UEAL4	38.58	131.97	94.51	59.14	14.50		15.66		1	1 1	1
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice														1	
	Transport Combination - Zone 3		3	UNCVX	UEAL4	60.02	131.97	94.51	59.14	14.50		15.66		I		
	Interoffice Transport - Dedicated - DS1 combination - Per Mile														1	
	Per Month			UNC1X	1L5XX	0.18						15.66		·	L	
	Interoffice Transport - Dedicated - DS1 - Facility Termination Per	1	1		l					l				1	1	1
	Month		<u> </u>	UNC1X	U1TF1	60.16	89.27	81.81	16.35	14.44		15.66			µ]	
	Channelization - Channel System DS1 to DS0 combination Per	1	1			107 10			40 - 1	0.70		45.00		1	1	1
1 1	Month		<u> </u>	UNC1X	MQ1	107.19	91.04	62.57	10.54	9.79		15.66			ا	───
\vdash	Voice Grade COCI - DS1 to DS0 Channel System combination -	1	1	UNCVX	1D1VG	0.56	6.58	4.72				15.00		1	1	1
	por month						b.58	4.72	1	1		15.66			1	L
	per month Additional 4-Wire Analog Voice Grade Loop in same DS1				IDIVO	0.50									1	
	Additional 4-Wire Analog Voice Grade Loop in same DS1		1						50 14	14 50		15 66				
			1	UNCVX	UEAL4	25.34	131.97	94.51	59.14	14.50		15.66				
UNBUNDLE	D NETWORK ELEMENTS - Alabama												Attachment:			bit: B
----------	--	-------------	--------	-----------------	----------	--------	--------	-----------	--------------	-------	-------	---	--	---	---	---
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Additional 4-Wire Analog Voice Grade Loop in same DS1		3	UNCVX	UEAL4	60.02	404.07	04.54	50.44	44.50		45.00				ł
	Interoffice Transport Combination - Zone 3 Voice Grade COCI - DS1 to DS0 Channel System combination -		3	UNCVX	UEAL4	60.02	131.97	94.51	59.14	14.50		15.66				J
	per month			UNCVX	1D1VG	0.56	6.58	4.72				15.66				ļ
	Nonrecurring Currently Combined Network Elements Switch -As-				110000		5 50	5 50	0.00	0.00		45.00				1
4 WID	Is Charge E 56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1		FEICE	UNC1X	UNCCC		5.59	5.59	6.98	6.98		15.66				i
4-9916	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice	INTERC	FFICE	TRANSPORT (EEL)												i
	Transport Combination - Zone 1		1	UNCDX	UDL56	26.09	126.27	88.80	59.14	14.50		15.66				ł
	First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice			0.102/1	00200	20.00	120.21	00.00	00.11	11.00		10.00				(
	Transport Combination - Zone 2		2	UNCDX	UDL56	35.95	126.27	88.80	59.14	14.50		15.66				1
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice		1		1											1
	Transport Combination - Zone 3		3	UNCDX	UDL56	37.88	126.27	88.80	59.14	14.50		15.66				I
	Interoffice Transport - Dedicated - DS1 combination - Per Mile															
	Per Month			UNC1X	1L5XX	0.18										1
	Interoffice Transport - Dedicated - DS1 - combination Facility															1
	Termination Per Month			UNC1X	U1TF1	60.16	89.27	81.81	16.35	14.44		15.66				l
	Channelization - Channel System DS1 to DS0 combination Per					107.10		00.57	10.51	0.70		45.00				1
	Month			UNC1X	MQ1	107.19	91.04	62.57	10.54	9.79		15.66				
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kbs)			UNCDX	1D1DD	1.19	6.58	4.72				15.66				ł
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1			UNCDA	סטוטו	1.19	0.00	4.72				15.00				1
	Interoffice Transport Combination - Zone 1		1	UNCDX	UDL56	26.09	126.27	88.80	59.14	14.50		15.66				1
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1			ONODA	ODLOO	20.00	120.21	00.00	00.14	14.00		10.00				[
	Interoffice Transport Combination - Zone 2		2	UNCDX	UDL56	35.95	126.27	88.80	59.14	14.50		15.66				1
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1		_													1
	Interoffice Transport Combination - Zone 3		3	UNCDX	UDL56	37.88	126.27	88.80	59.14	14.50		15.66				1
	OCU-DP COCI (data) - DS1 to DS0 Channel System -															i
	combination per month (2.4-64kbs)			UNCDX	1D1DD	1.19	6.58	4.72				15.66				l
	Nonrecurring Currently Combined Network Elements Switch -As-															l
	Is Charge			UNC1X	UNCCC		5.59	5.59	6.98	6.98		15.66				1
4-WIR	E 64 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1	INTERC	FFICE	TRANSPORT (EEL)	1											l
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice					00.00	100.07		50.44	11.50		45.00				ł
	Transport Combination - Zone 1		1	UNCDX	UDL64	26.09	126.27	88.80	59.14	14.50		15.66				ł
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2		2	UNCDX	UDL64	35.95	126.27	88.80	59.14	14.50		15.66				1
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice		2	UNCDA	UDL64	30.95	120.27	00.00	59.14	14.50		15.00				1
	Transport Combination - Zone 3		3	UNCDX	UDL64	37.88	126.27	88.80	59.14	14.50		15.66				1
	Interoffice Transport - Dedicated - DS1 combination - Per Mile		Ŭ	ONODA	ODLOT	07.00	120.21	00.00	00.14	14.00		10.00				[
	Per Month			UNC1X	1L5XX	0.18										l I
	Interoffice Transport - Dedicated - DS1 combination - Facility															i
	Termination Per Month			UNC1X	U1TF1	60.16	89.27	81.81	16.35	14.44		15.66				ł
	Channelization - Channel System DS1 to DS0 combination Per															Ĩ
	Month			UNC1X	MQ1	107.19	91.04	62.57	10.54	9.79		15.66				
	OCU-DP COCI (data) - DS1 to DS0 Channel System				1		_									ł
	combination - per month (2.4-64kbs)			UNCDX	1D1DD	1.19	6.58	4.72				15.66				I
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1	1				00.00	400.07	00.00	50.11	44.50		45.00				ł
┝──┼──	Interoffice Transport Combination - Zone 1 Additional 4-Wire 64Kbps Digital Grade Loopin same DS1		1	UNCDX	UDL64	26.09	126.27	88.80	59.14	14.50		15.66				
	Interoffice Transport Combination - Zone 2		2	UNCDX	UDL64	35.95	126.27	88.80	59.14	14.50		15.66				ł
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1		-	ONODA	5 DL04	55.85	120.27	00.00	55.14	14.30		15.00				[
	Interoffice Transport Combination - Zone 3		3	UNCDX	UDL64	37.88	126.27	88.80	59.14	14.50		15.66				ł
	OCU-DP COCI (data) - DS1 to DS0 Channel System				1											i
	combination - per month (2.4-64kbs)			UNCDX	1D1DD	1.19	6.58	4.72				15.66				ł
	Nonrecurring Currently Combined Network Elements Switch -As-															1
	Is Charge			UNC1X	UNCCC		5.59	5.59	6.98	6.98		15.66				
4-WIR	E DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTE	ROFFI	CE TRA	NSPORT (EEL)	<u> </u>											
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice				1101.567		c=c +-					/= o-				ł
	Transport - Zone 1		1	UNC1X	USLXX	82.55	252.47	157.54	44.70	11.71		15.66				

UNBUNDLE	ED NETWORK ELEMENTS - Alabama		1		1								Attachment:			ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						1.00	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2		2	UNC1X	USLXX	454.40	050.47	157.54	44.70	44 74		45.00				
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice		2	UNCIA	USLAA	154.18	252.47	157.54	44.70	11.71		15.66				
	Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile		3	UNC1X	USLXX	314.52	252.47	157.54	44.70	11.71		15.66				
	Per Month			UNC1X	1L5XX	0.18										
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Month			UNC1X	U1TF1	60.16	89.27	81.81	16.35	14.44		15.66				
	Nonrecurring Currently Combined Network Elements Switch -As-					00.16										
4 14/15		DOFF		UNC1X	UNCCC		5.59	5.59	6.98	6.98		15.66				
4-WIR	RE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTE First DS1Loop in DS3 Interoffice Transport Combination - Zone	ROFFI		ANSPORT (EEL)												├───
	1 First DS1Loop in DS3 Interoffice Transport Combination - Zone		1	UNC1X	USLXX	82.55	252.47	157.54	44.70	11.71		15.66				<u> </u>
	2		2	UNC1X	USLXX	154.18	252.47	157.54	44.70	11.71		15.66				
	First DS1Loop in DS3 Interoffice Transport Combination - Zone 3		3	UNC1X	USLXX	314.52	252.47	157.54	44.70	11.71		15.66				
	Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Month			UNC3X	1L5XX	4.09										
	Interoffice Transport - Dedicated - DS3 - Facility Termination per					700 50	070 75	100 70	00.00	50.40		45.00				
	month DS3 to DS1 Channel System combination per month			UNC3X UNC3X	U1TF3 MQ3	703.52 176.20	278.75 178.14	162.76 93.97	60.20 33.26	58.46 31.83		15.66 15.66				
	DS3 to DS1 Charnel System combination per month			UNC1X	UC1D1	13.47	6.58	4.72	33.20	31.03		15.66				
	Additional DS1Loop in DS3 Interoffice Transport Combination -		1						44.70							
	Zone 1 Additional DS1Loop in DS3 Interoffice Transport Combination -			UNC1X	USLXX	82.55	252.47	157.54	44.70	11.71		15.66				
	Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination -		2	UNC1X	USLXX	154.18	252.47	157.54	44.70	11.71		15.66				
	Zone 3		3	UNC1X	USLXX	314.52	252.47	157.54	44.70	11.71		15.66				
	DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	13.47	6.58	4.72								-
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNC3X	UNCCC		5.59	5.59	6.98	6.98		15.66				
2-WIR	RE VOICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INT	EROFF	ICE TR		UNCCC		5.59	5.55	0.90	0.90		13.00				
	2-WireVG Loop used with 2-wire VG Interoffice Transport															
	Combination - Zone 1 2-WireVG Loop used with 2-wire VG Interoffice Transport		1	UNCVX	UEAL2	14.38	88.00	55.00	47.24	7.44		15.66				<u> </u>
	Combination - Zone 2		2	UNCVX	UEAL2	22.85	88.00	55.00	47.24	7.44		15.66				
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL2	36.14	88.00	55.00	47.24	7.44		15.66				
	Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month			UNCVX	1L5XX	0.008838										
	Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility Termination per month			UNCVX	U1TV2	21.13	40.54	27.41	16.74	6.90		15.66				
	Nonrecurring Currently Combined Network Elements Switch -As-					20										
4.WIR	IS Charge RE VOICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INT	FROFF	ICE TR	UNCVX	UNCCC		5.59	5.59	6.98	6.98		15.66				+
4-1411	4-WireVG Loop used with 4-wire VG Interoffice Transport	LIGFF			1											<u> </u>
	Combination - Zone 1 4-WireVG Loop used with 4-wire VG Interoffice Transport		1	UNCVX	UEAL4	25.34	131.97	94.51	59.14	14.50		15.66				<u> </u>
	Combination - Zone 2		2	UNCVX	UEAL4	38.58	131.97	94.51	59.14	14.50		15.66				<u> </u>
	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL4	60.02	131.97	94.51	59.14	14.50		15.66				
	Interoffice Transport - Dedicated - 4-wire VG combination - Per Mile Per Month			UNCVX	1L5XX	0.008838										
	Interoffice Transport - Dedicated - 4- Wire Voice Grade combination - Facility Termination per month			UNCVX	U1TV4	18.73	40.54	27.41	16.74	6.90		15.66				
	Nonrecurring Currently Combined Network Elements Switch -As-		1								1				1	
	Is Charge			UNCVX	UNCCC		5.59	5.59	6.98	6.98		15.66				
DS3 D	DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFIC	E TRAI	NSPOR	T (EEL)												

UNBUNDLE	D NETWORK ELEMENTS - Alabama		1								·		Attachment:		Exhi	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	High Capacity Unbundled Local Loop - DS3 combination - Per Mile per month			UNC3X	1L5ND	8.89										
	High Capacity Unbundled Local Loop - DS3 combination -			UNC3X	TLOND	8.89							-		-	
	Facility Termination per month			UNC3X	UE3PX	327.71	451.52	263.94	119.49	83.58		15.66				
	Interoffice Transport - Dedicated - DS3 - Per Mile per month			UNC3X	1L5XX	4.09										
	Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per month			UNC3X	U1TF3	703.52	278.75	162.76	60.20	58.46		15.66				
	Nonrecurring Currently Combined Network Elements Switch -As-															
	Is Charge			UNC3X	UNCCC		5.59	5.59	6.98	6.98		15.66				
STS1	DIGITAL EXTENDED LOOP WITH DEDICATED STS1 INTEROFI	FICE TR	RANSPO	ORT (EEL)												
	High Capacity Unbundled Local Loop - STS1 combination - Per Mile per month			UNCSX	1L5ND	8.89										
	High Capacity Unbundled Local Loop - STS1 combination -					0.00							-		-	
	Facility Termination per month			UNCSX	UDLS1	339.21	451.52	263.94	119.49	83.58		15.66				
	Interoffice Transport - Dedicated - STS1 combination - Per Mile															
	per month			UNCSX	1L5XX	4.09										
	Interoffice Transport - Dedicated - STS1 combination - Facility Termination per month			UNCSX	U1TFS	701.37	278.75	162.76	60.20	58.46		15.66				
	Nonrecurring Currently Combined Network Elements Switch -As-			UNCOX	01110	701.57	210.15	102.70	00.20	30.40		10.00				
	Is Charge			UNCSX	UNCCC		5.59	5.59	6.98	6.98		15.66				
2-WIR	E ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPOR	RT (EEL)													
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination															
	Transport - Zone 1 First 2-Wire ISDN Loop in a DS1 Interoffice Combination		1	UNCNX	U1L2X	21.88	117.24	79.77	52.88	10.54		15.66				
	Transport - Zone 2		2	UNCNX	U1L2X	32.85	117.24	79.77	52.88	10.54		15.66				
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination		2	ONONA	UTLEX	52.05	117.24	13.11	52.00	10.54		10.00				
	Transport - Zone 3		3	UNCNX	U1L2X	48.55	117.24	79.77	52.88	10.54		15.66				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile			UNC1X	1L5XX	0.18										
	Interoffice Transport - Dedicated - DS1 combinition - Facility								10.05							
	Termination per month Channelization - Channel System DS1 to DS0 combination -			UNC1X	U1TF1	60.16	89.27	81.81	16.35	14.44		15.66	-		-	
	per month			UNC1X	MQ1	107.19	91.04	62.57	10.54	9.79		15.66				
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System															
	combination - per month			UNCNX	UC1CA	2.56	6.58	4.72				15.66				
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport															
	Combination - Zone 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport		1	UNCNX	U1L2X	21.88	117.24	79.77	52.88	10.54		15.66				
	Combination - Zone 2		2	UNCNX	U1L2X	32.85	117.24	79.77	52.88	10.54		15.66				
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport										1		-		-	
	Combination - Zone 3		3	UNCNX	U1L2X	48.55	117.24	79.77	52.88	10.54		15.66				
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System				110404	0.50	0.50	1 70								
	combintaion- per month Nonrecurring Currently Combined Network Elements Switch -As-			UNCNX	UC1CA	2.56	6.58	4.72								
	Is Charge			UNC1X	UNCCC		5.59	5.59	6.98	6.98		15.66				
4-WIR	E DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 IN	TEROF	FICE TI				0.00	0.00	0.00	0.00		.0.00				
	First DS1 Loop in STS1 Interoffice Transport Combination -															
	Zone 1		1	UNC1X	USLXX	82.55	252.47	157.54	44.70	11.71		15.66				
	First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2		2	UNC1X	USLXX	154.18	252.47	157.54	44.70	11.71		15.66				
	First DS1 Loop in STS1 Interoffice Transport Combination -		2		3000	134.10	232.41	107.04	44.70	11.71		13.00				
	Zone 3		3	UNC1X	USLXX	314.52	252.47	157.54	44.70	11.71		15.66				
1	Interoffice Transport - Dedicated - STS1 combination - Per Mile															
	Per Month			UNCSX	1L5XX	4.09										
	Interoffice Transport - Dedicated - STS1 combination - Facility Termination			UNCSX	U1TFS	701.37	278.75	162.76	60.20	58.46		15.66				
	STS1 to DS1 Channel System conbination per month			UNCSX	MQ3	176.20	278.75	93.97	60.20 33.26	31.83		15.66				
	DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	13.47	6.58	4.72	55.20	51.05		13.00				
	Additional DS1Loop in STS1 Interoffice Transport Combination -			-								1				
	Zone 1		1	UNC1X	USLXX	82.55	252.47	157.54	44.70	11.71		15.66				

UNBUNDL	ED NETWORK ELEMENTS - Alabama												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec			g Disconnect				Rates(\$)		
	Additional DC4Loop in CTC4 Interaffice Transport Combination				-		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 2		2	UNC1X	USLXX	154.18	252.47	157.54	44.70	11.71		15.66				1
	Additional DS1Loop in STS1 Interoffice Transport Combination -		~	UNUTX	UGEXX	104.10	232.41	107.04	44.70	11.71		10.00				
	Zone 3		3	UNC1X	USLXX	314.52	252.47	157.54	44.70	11.71		15.66				i
	DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	13.47	6.58	4.72								
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNCSX	UNCCC		5.59	5.59	C 00	6.98		45.00				i
4-WI	RE 56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTERO	FFICE	RANS		UNCCC		5.59	5.59	6.98	6.98		15.66				i
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport															i
	Combination - Zone 1		1	UNCDX	UDL56	26.09	126.27	88.80	59.14	14.50		15.66				1
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport		_													i
	Combination - Zone 2 4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport		2	UNCDX	UDL56	35.95	126.27	88.80	59.14	14.50		15.66				1
	Combination - Zone 3		3	UNCDX	UDL56	37.88	126.27	88.80	59.14	14.50		15.66				i
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination -															1
	Per Mile			UNCDX	1L5XX	0.008838										I
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination -				LIATOR	15.10	10.54	07.44	10.74	0.00		45.00				i
	Facility Termination Nonrecurring Currently Combined Network Elements Switch -As-			UNCDX	U1TD5	15.12	40.54	27.41	16.74	6.90		15.66				ļ
	Is Charge			UNCDX	UNCCC		5.59	5.59	6.98	6.98		15.66				1
4-WI	RE 64 KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTERO	FFICE 1	RANS		0.1000		0.00	0.00	0.00	0.00		10.00				1
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport															Í
	Combination - Zone 1		1	UNCDX	UDL64	26.09	126.27	88.80	59.14	14.50		15.66				ļ
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 2		2	UNCDX	UDL64	35.95	126.27	88.80	59.14	14.50		15.66				i
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport		2	UNCDA	UDL04	35.95	120.27	00.00	59.14	14.50		15.00				l
	Combination - Zone 3		3	UNCDX	UDL64	37.88	126.27	88.80	59.14	14.50		15.66				i
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination -															ſ
	Per Mile			UNCDX	1L5XX	0.008838										ļ
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility Termination			UNCDX	U1TD6	15.12	40.54	27.41	16.74	6.90		15.66				1
	Nonrecurring Currently Combined Network Elements Switch -As-			ONODA	01100	10.12	40.04	27.41	10.74	0.50		15.00				ł
	Is Charge			UNCDX	UNCCC		5.59	5.59	6.98	6.98		15.66				1
	NETWORK ELEMENTS															1
	n used as a part of a currently combined facility, the non-recurr															1
	n used as ordinarily combined network elements in All States, th ecurring Currently Combined Network Elements "Switch As Is"					AS IS Charge C	ides not.									l
	Nonrecurring Currently Combined Network Elements Switch -As-								1		<u> </u>					
	Is Charge - 2 wire/4-Wire VG			UNCVX	UNCCC		5.59	5.59	6.98	6.98		15.66				1
	Nonrecurring Currently Combined Network Elements Switch -As-								0.00	0.00		45.00				i i
├── ┼──	Is Charge - 56/64 kbps Nonrecurring Currently Combined Network Elements Switch -As-			UNCDX	UNCCC		5.59	5.59	6.98	6.98	<u> </u>	15.66				i
	Is Charge - DS1		1	UNC1X	UNCCC		5.59	5.59	6.98	6.98		15.66				1
	Nonrecurring Currently Combined Network Elements Switch -As-		1								İ					
	Is Charge - DS3		1	UNC3X	UNCCC		5.59	5.59	6.98	6.98	ļ	15.66				
	Nonrecurring Currently Combined Network Elements Switch -As-		1	UNCSX	UNCCC		5.59	5.59	6.98	6.98		15.66				1
NOT	Is Charge - STS1 E: Local Channel - Dedicated Transport - minimum billing period	d - Belo	w DS3-			r months	5.59	5.59	0.98	0.98	<u> </u>	00.01				i
	Local Channel - Dedicated - 2-Wire Voice Grade			UNCXV	ULDV2	13.97	193.10	33.17	36.64	3.20	<u> </u>	15.66				
	Local Channel - Dedicated - 4-Wire Voice Grade			UNCXV	ULDV4	14.93	193.53	33.60	37.11	3.67		15.66				
	Local Channel - Dedicated - DS1 per month Zone 1		1	UNC1X	ULDF1	35.76	177.47	153.72	22.19	15.26		15.66				
	Local Channel - Dedicated -DS1 Per Month Zone 2 Local Channel - Dedicated - DS1- Per Month Zone 3		2	UNC1X UNC1X	ULDF1 ULDF1	49.98 107.63	177.47 177.47	153.72 153.72	22.19 22.19	15.26 15.26	<u> </u>	15.66 15.66				
	Local Channel - Dedicated - DS1 - Per Mile per month		5	UNC3X	1L5NC	6.92	177.47	100.12	22.19	15.20		13.00				
	Local Channel - Dedicated - DS3 - Facility Termination		<u> </u>	UNC3X	ULDF3	416.54	451.52	263.94	119.49	83.58		15.66				
	Local Channel - Dedicated - STS-1- Per Mile per month			UNCSX	1L5NC	6.92										
	Local Channel - Dedicated - STS-1 - Facility Termination			UNCSX	ULDFS	408.49	451.52	263.94	119.49	83.58	<u> </u>	15.66				
	onal Features & Functions: TIPLEXERS			<u> </u>												'
INIOL		L	1	[I			I	I	I	I	I	I	I	L

UNBU	INDLE	D NETWORK ELEMENTS - Alabama			n								1	Attachment:			ibit: B
CATEG	ORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Charge -
							Rec	Nonrec		Nonrecurring					Rates(\$)		
								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Channelization - DS1 to DS0 Channel System			UXTD1	MQ1	101.06	91.04	62.57	10.54	9.79		15.66				
		OCU-DP COCI (data) - DS1 to DS0 Channel System - per			UDL	1D1DD	1.12	6.58	4.72				15.66				
		month (2.4-64kbs) 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per			UDL	10100	1.12	86.0	4.72				15.00				
		month			UDN	UC1CA	2.41	6.58	4.72				15.66				
		Voice Grade COCI - DS1 to DS0 Channel System - per month			UEA	1D1VG	0.53	6.58	4.72				15.66				
		DS3 to DS1 Channel System per month			UXTD3	MQ3	166.13	178.14	93.97	33.26	31.83		15.66				
		STS1 to DS1 Channel System per month		1	UXTS1	MQ3	166.13	178.14	93.97	33.26	31.83		15.66				
		DS3 Interface Unit (DS1 COCI) used with Loop per month			USL	UC1D1	12.70	6.58	4.72				15.66				
		DS3 Interface Unit (DS1 COCI) used with Local Channel per															
		month			ULDD1	UC1D1	12.70	6.58	4.72				15.66				
		DS3 Interface Unit (DS1 COCI) used with Interoffice Channel															
		per month			U1TD1	UC1D1	12.70	6.58	4.72				15.66			ļ	
		op Feeder															
		Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Statewide		SW	UNC1X	USBFG		101.05			17.10						
		Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 1		1	UNC1X	USBFG	55.09	101.85	64.38	62.05	17.40						
		Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 2 Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 3		2	UNC1X UNC1X	USBFG USBFG	124.69 294.62	101.85 101.85	64.38 64.38	62.05 62.05	17.40 17.40						
		Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 3		4	UNC1X	USBFG	294.02	101.65	04.30	62.05	17.40						
		OCAL EXCHANGE SWITCHING(PORTS)		4	UNCIX	036-0								-			
UNBOIN		ige Ports															
-		Although the Port Rate includes all available features in GA, I	KY. LA	& TN. t	he desired feature	s will need to h	e ordered usin	g retail USOCs									
-		VOICE GRADE LINE PORT RATES (RES)		<u> </u>	lo uconou routuro			g rotali eeeet									
		Exchange Ports - 2-Wire Analog Line Port- Res.		1	UEPSR	UEPRL	1.38	2.38	2.27	1.42	1.33		15.66				
		Exchange Ports - 2-Wire Analog Line Port with Caller ID - Res.			UEPSR	UEPRC	1.38	2.38	2.27	1.42	1.33		15.66				
		Exchange Ports - 2-Wire Analog Line Port outgoing only - Res.			UEPSR	UEPRO	1.38	2.38	2.27	1.42	1.33		15.66				
		Exchange Ports - 2-Wire VG unbundled AL extended local															
		dialing parity Port with Caller ID - Res.			UEPSR	UEPAR	1.38	2.38	2.27	1.42	1.33		15.66				
		Exchange Ports - 2-Wire VG unbundled res, low usage line port															
-		with Caller ID (LUM)			UEPSR	UEPAP	1.38	2.38	2.27	1.42	1.33		15.66				
		Exchange Ports - 2-Wire VG Alabama Residence Dialing Plan			115000		1.00	0.00	0.07	4.40	1.00		45.00				
		without Caller Id		-	UEPSR	UEPWA	1.38	2.38	2.27	1.42	1.33		15.66				
		2-Wire voice unbundled Low Usage Line Port without Caller ID					4.00	0.00	0.07	4.40	4.00		45.00				
		Capability Subsequent Activity			UEPSR UEPSR	UEPRT USASC	1.38 0.00	2.38 0.00	2.27	1.42	1.33		15.66 15.66				
-	FEATU			-	UEPSK	USASC	0.00	0.00	0.00				15.00				
		All Available Vertical Features			UEPSR	UEPVF	1.98	0.00	0.00				15.66				
		VOICE GRADE LINE PORT RATES (BUS)			02.01	02. 11		0.00	0.00				10.00				
		Exchange Ports - 2-Wire Analog Line Port without Caller ID -															
		Bus			UEPSB	UEPBL	1.38	2.38	2.27	1.42	1.33		15.66				
		Exchange Ports - 2-Wire VG unbundled Line Port with				-											
		unbundled port with Caller+E484 ID - Bus.			UEPSB	UEPBC	1.38	2.38	2.27	1.42	1.33		15.66				
		Exchange Ports - 2-Wire Analog Line Port outgoing only - Bus.			UEPSB	UEPBO	1.38	2.38	2.27	1.42	1.33		15.66				
	_	Exchange Ports - 2-Wire VG unbundled AL extended local		1												I	
	ļ	dialing parity Port with Caller ID - Bus.		I	UEPSB	UEPAW	1.38	2.38	2.27	1.42	1.33		15.66				
		Exhange Ports - 2-Wire VG unbundled incoming only port with		1												1	
L		Caller ID - Bus	<u> </u>		UEPSB	UEPB1	1.38	2.38	2.27	1.42	1.33		15.66				<u> </u>
		Exchange Ports - 2-Wire Voice Alabama Business Dialing Plan	1	1			4.00	0.00	0.07	4 40	4.00		45.00			1	
		without Caller ID 2-Wire voice unbundled Incoming Only Port without Caller ID	l		UEPSB	UEPWB	1.38	2.38	2.27	1.42	1.33		15.66			<u> </u>	
		2-wire voice unbundled incoming Only Port without Caller ID Capability	1	1	UEPSB	UEPBE	1.38	2.38	2.27	1.42	1.33		15.66			1	
		Subsequent Activity			UEPSB	USASC	0.00	0.00	0.00	1.42	1.33		15.66			ł	
	FEATU		-	1	521 00	00000	0.00	0.00	0.00				10.00			1	
		All Available Vertical Features	1	1	UEPSB	UEPVF	1.98	0.00	0.00				15.66				1
		NGE PORT RATES (DID & PBX)	1	1				0.00	0.00			<u> </u>					1
	EXCHA																

UNBUNDLE	D NETWORK ELEMENTS - Alabama												Attachment:			bit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus			UEPSP	UEPPC	1.38	31.27	14.85	13.94	0.90		15.66				
	2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus			UEPSP	UEPPO	1.38	31.27	14.85	13.94	0.90		15.66				
	2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus			UEPSP	UEPP1	1.38	31.27	14.85	13.94	0.90		15.66				
	2-Wire Analog Long Distance Terminal PBX Trunk - Bus 2-Wire Voice Unbundled 2-Way PBX Alabama Calling Port			UEPSP UEPSP	UEPLD UEPA2	1.38 1.38	31.27 31.27	14.85 14.85	13.94 13.94	0.90		15.66 15.66				
	2-Wire Voice Unbundled 2-Way PBX Alabama Calling Port 2-Wire Voice Unbundled PBX LD Terminal Ports			UEPSP	UEPAZ	1.38	31.27	14.85	13.94	0.90		15.66				
	2-Wire Vice Unbundled 2-Way PBX Usage Port			UEPSP	UEPXA	1.38	31.27	14.85	13.94	0.90		15.66				
	2-Wire Vice Unbundled PBX Toll Terminal Hotel Ports			UEPSP	UEPXB	1.38	31.27	14.85	13.94	0.90		15.66				
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPSP	UEPXC	1.38	31.27	14.85	13.94	0.90		15.66				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPSP	UEPXD	1.38	31.27	14.85	13.94	0.90		15.66				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD						-									
	Capable Port			UEPSP	UEPXE	1.38	31.27	14.85	13.94	0.90		15.66				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy															
	Administrative Calling Port			UEPSP	UEPXL	1.38	31.27	14.85	13.94	0.90		15.66				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy															
	Room Calling Port			UEPSP	UEPXM	1.38	31.27	14.85	13.94	0.90		15.66				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital															
	Discount Room Calling Port			UEPSP	UEPXO	1.38	31.27	14.85	13.94	0.90		15.66				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPSP	UEPXS	1.38	31.27	14.85	13.94	0.90		15.66				
	Subsequent Activity			UEPSP	USASC	0.00	0.00	0.00				15.66				
FEATU						1.00						15.00				
EXOL	All Available Vertical Features			UEPSP UEPSE	UEPVF	1.98	0.00	0.00				15.66				
EXCH	ANGE PORT RATES (COIN)				-	4.00	0.00	2.27	4.40	4.00		15.66				
NOTE	Exchange Ports - Coin Port	ultak ad			ainavit avvitala	1.38	2.38		1.42	1.33	at a d with 0					
NOTE:	 Transmission/usage charges associated with POTS circuit sw Access to B Channel or D Channel Packet capabilities will be 	vitched	usage	will also apply to o	Rusinose Po	auost Procoss	Pates for the	nackot canabi	lission by B-Ch	tormined via t	aleu with 2	WITE ISDN P	Now Rusinger	Poqueet Bro	0000	
	LOCAL EXCHANGE SWITCHING(PORTS)	avalla			V Busiliess Re	quest Flocess.	Rates for the		intes will be de	termineu via t	THE DUTIA FIC	le Request	New Busiliess	s Request FIC		
	ANGE PORT RATES				1											
	Exchange Ports - 2-Wire DID Port			UEPEX	UEPP2	8.05	119.31	18.74	59.90	3.76		15.66				
	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID			-	-											
	capability			UEPDD	UEPDD	60.09	202.02	95.69	72.59	2.46		15.66				
	Exchange Ports - 2-Wire ISDN Port (See Notes below.)			UEPTX UEPSX	U1PMA	9.79	72.77	52.99	47.79	10.74		15.66				
	All Features Offered			UEPTX UEPSX	UEPVF	1.98	0.00	0.00								
	: Transmission/usage charges associated with POTS circuit sv															
NOTE:	: Access to B Channel or D Channel Packet capabilities will be	availa	ble only						lities will be de	termined via t	he Bona Fic	le Request/	New Business	s Request Pro	cess.	
	Exchange Ports - 2-Wire ISDN Port Channel Profiles			UEPTX UEPSX	U1UMA	0.00	0.00	0.00								
	Exchange Ports - 4-Wire ISDN DS1 Port			UEPEX	UEPEX	84.32	203.81	101.56	79.18	20.06		15.66				
	NDLED PORT with REMOTE CALL FORWARDING CAPABILITY	,			_											
UNBU	NDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE											48.00				
	Unbundled Remote Call Forwarding Service, Area Calling, Res		1	UEPVR	UERAC	1.38	2.38	2.27	1.42	1.33		15.66				
						1.00										
							0.00	0.07	4.40	4.00		45.00				
	Unbundled Remote Call Forwarding Service, Local Calling - Res			UEPVR	UERLC	1.38	2.38	2.27	1.42	1.33		15.66				
	Unbundled Remote Call Forwarding Service, InterLATA - Res			UEPVR	UERTE	1.38 1.38	2.38	2.27	1.42	1.33		15.66				
Non B	Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res					1.38										
Non-R	Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res lecurring			UEPVR	UERTE	1.38 1.38	2.38	2.27	1.42	1.33		15.66				
Non-R	Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res tecurring Unbundled Remote Call Forwarding Service - Conversion -			UEPVR UEPVR	UERTE UERTR	1.38 1.38	2.38 2.38	2.27 2.27	1.42	1.33		15.66 15.66				
Non-R	Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res tecurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is			UEPVR	UERTE	1.38 1.38	2.38	2.27	1.42	1.33		15.66				
Non-R	Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res ecurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with			UEPVR UEPVR UEPVR	UERTE UERTR USAC2	1.38 1.38	2.38 2.38 0.10	2.27 2.27 0.10	1.42	1.33		15.66 15.66 15.66				
	Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res eccurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC)			UEPVR UEPVR	UERTE UERTR	1.38 1.38	2.38 2.38	2.27 2.27	1.42	1.33		15.66 15.66				
	Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res ecurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with			UEPVR UEPVR UEPVR	UERTE UERTR USAC2	1.38 1.38	2.38 2.38 0.10	2.27 2.27 0.10	1.42	1.33		15.66 15.66 15.66				
	Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res eccurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC)			UEPVR UEPVR UEPVR	UERTE UERTR USAC2	1.38 1.38	2.38 2.38 0.10	2.27 2.27 0.10	1.42	1.33		15.66 15.66 15.66				
	Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res tecurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) NDLED REMOTE CALL FORWARDING - Bus			UEPVR UEPVR UEPVR UEPVR	UERTE UERTR USAC2 USACC	1.38 1.38 1.38	2.38 2.38 0.10 0.10	2.27 2.27 0.10 0.10	1.42 1.42	1.33 1.33		15.66 15.66 15.66 15.66				
	Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res tecurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) NDLED REMOTE CALL FORWARDING - Bus			UEPVR UEPVR UEPVR UEPVR	UERTE UERTR USAC2 USACC	1.38 1.38 1.38	2.38 2.38 0.10 0.10	2.27 2.27 0.10 0.10	1.42 1.42	1.33 1.33		15.66 15.66 15.66 15.66				
	Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res tecurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) NDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus			UEPVR UEPVR UEPVR UEPVR UEPVB	UERTE UERTR USAC2 USACC USACC	1.38 1.38 1.38 1.38	2.38 2.38 0.10 0.10 2.38	2.27 2.27 0.10 0.10 2.27	1.42 1.42 1.42	1.33 1.33 1.33		15.66 15.66 15.66 15.66 15.66				
	Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res eccurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) NDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus			UEPVR UEPVR UEPVR UEPVR UEPVB UEPVB	UERTE UERTR USAC2 USACC UERAC UERAC UERLC	1.38 1.38 1.38 1.38 1.38 1.38	2.38 2.38 0.10 0.10 2.38 2.38	2.27 2.27 0.10 0.10 2.27 2.27 2.27	1.42 1.42 1.42	1.33 1.33 1.33 1.33 1.33		15.66 15.66 15.66 15.66 15.66 15.66				
	Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res tecurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) NDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus			UEPVR UEPVR UEPVR UEPVR UEPVB UEPVB UEPVB UEPVB UEPVB	UERTE UERTR USAC2 USACC UERAC UERAC UERLC UERTE UERTR	1.38 1.38 1.38 1.38 1.38 1.38 1.38 1.38	2.38 2.38 0.10 2.38 2.38 2.38 2.38 2.38 2.38	2.27 2.27 0.10 0.10 2.27 2.27 2.27 2.27 2.27	1.42 1.42 1.42 1.42 1.42 1.42 1.42 1.42	1.33 1.33 1.33 1.33 1.33 1.33 1.33		15.66 15.66 15.66 15.66 15.66 15.66 15.66				
UNBU	Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res eccurring Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) NDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus			UEPVR UEPVR UEPVR UEPVR UEPVB UEPVB UEPVB	UERTE UERTR USAC2 USACC UERAC UERAC UERLC UERTE	1.38 1.38 1.38 1.38 1.38 1.38 1.38	2.38 2.38 0.10 2.38 2.38 2.38 2.38	2.27 2.27 0.10 0.10 2.27 2.27 2.27	1.42 1.42 1.42 1.42 1.42 1.42 1.42	1.33 1.33 1.33 1.33 1.33 1.33		15.66 15.66 15.66 15.66 15.66 15.66 15.66				

	NDLE	D NETWORK ELEMENTS - Alabama												Attachment:			bit: B
CATEG	ORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'I
							Rec	Nonrec		Nonrecurring					Rates(\$)		
	-							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Unbundled Remote Call Forwarding Service - Conversion -			UEPVB	USAC2		0.10	0.10				15.66				
		Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with			UEFVD	03AC2	ł	0.10	0.10				15.00			-	
		allowed change (PIC and LPIC)			UEPVB	USACC		0.10	0.10				15.66				
UNBUN	DLED I	LOCAL SWITCHING, PORT USAGE															
	End Of	ffice Switching (Port Usage)															
		End Office Switching Function, Per MOU					0.0007025										
	T I	End Office Trunk Port - Shared, Per MOU					0.0001638										
	Tande	m Switching (Port Usage) (Local or Access Tandem) Tandem Switching Function Per MOU					0.000095										
		Tandem Trunk Port - Shared, Per MOU					0.0002015										
	Comm	on Transport	1	1	1	1	0.0002010					1					
		Common Transport - Per Mile, Per MOU	1	1	1	1	0.000023										
		Common Transport - Facilities Termination Per MOU		1	1	1	0.0003224										
		PORT/LOOP COMBINATIONS - COST BASED RATES															
\square	Cost B	ased Rates are applied where BellSouth is required by FCC and	nd/or S	tate Co	ommission rule to p	rovide Unbur	dled Local Swi	tching or Swite	h Ports.								
	Featur	es shall apply to the Unbundled Port/Loop Combination - Cos	st Based	Rate	section in the same	manner as th	ey are applied	to the Stand-A	one Unbundle	d Port section	of this Rate E	xhibit.	n Dewt [#] -	Combined			
		ffice and Tandem Switching Usage and Common Transport Us															
		st and additional Port nonrecurring charges apply to Not Curr E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)		mamo	ed Combos. For Cu	riently Comb		le nonrecurring	g charges sha	i be triose idei	itilied in the r	Ionrecurring	- Currentity	Compined Se	cuons.		
		ort/Loop Combination Rates															
	0.12	2-Wire VG Loop/Port Combo - Zone 1		1			12.70										
		2-Wire VG Loop/Port Combo - Zone 2		2			21.19										
		2-Wire VG Loop/Port Combo - Zone 3		3			34.80										
	UNE L	oop Rates															
	-	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPRX	UEPLX	11.55										
		2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPRX	UEPLX	20.04										
-	2-Wiro	2-Wire Voice Grade Loop (SL1) - Zone 3 Voice Grade Line Port Rates (Res)	-	3	UEPRX	UEPLX	33.65										
		Voice Glade Line Fort Rates (Res)															
		2-Wire voice unbundled port - residence	1		LIEDRY	LIEPRI	1 15	/0.19	10.83	2/ 01	6.63		15.66				
		2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res			UEPRX UEPRX	UEPRL	1.15	40.19 40.19	19.83 19.83	24.91 24.91	6.63 6.63		15.66 15.66				
		2-Wire voice unbundled port with Caller ID - res			UEPRX UEPRX UEPRX	UEPRC	1.15	40.19	19.83	24.91	6.63		15.66				
					UEPRX			40.19 40.19									
		2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade unbundled Alabama extended local dialing			UEPRX UEPRX	UEPRC UEPRO	1.15 1.15	40.19	19.83 19.83	24.91 24.91	6.63 6.63		15.66 15.66				
		2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade unbundled Alabama extended local dialing parity port with Caller ID - res 2-Wire voice unbundles res, low usage line port with Caller ID (LUM) 2-Wire Voice Unbundled Alabama Residence Dialing Plan			UEPRX UEPRX UEPRX UEPRX	UEPRC UEPRO UEPAR UEPAP	1.15 1.15 1.15 1.15	40.19 40.19 40.19 40.19 40.19	19.83 19.83 19.83 19.83	24.91 24.91 24.91 24.91 24.91	6.63 6.63 6.63 6.63		15.66 15.66 15.66 15.66				
		2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade unbundled Alabama extended local dialing parity port with Caller ID - res 2-Wire voice unbundles res, low usage line port with Caller ID (LUM) 2-Wire Voice Unbundled Alabama Residence Dialing Plan without Caller ID			UEPRX UEPRX UEPRX	UEPRC UEPRO UEPAR	1.15 1.15 1.15	40.19 40.19 40.19	19.83 19.83 19.83	24.91 24.91 24.91	6.63 6.63 6.63		15.66 15.66 15.66				
		2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade unbundled Alabama extended local dialing parity port with Caller ID - res 2-Wire voice unbundles res, low usage line port with Caller ID (LUM) 2-Wire Voice Unbundled Alabama Residence Dialing Plan without Caller ID 2-Wire voice unbundled Low Usage Line Port without Caller ID			UEPRX UEPRX UEPRX UEPRX UEPRX	UEPRC UEPRO UEPAR UEPAP UEPWA	1.15 1.15 1.15 1.15 1.15 1.15	40.19 40.19 40.19 40.19 40.19 40.19	19.83 19.83 19.83 19.83 19.83 19.83	24.91 24.91 24.91 24.91 24.91 24.91	6.63 6.63 6.63 6.63 6.63		15.66 15.66 15.66 15.66 15.66				
		2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade unbundled Alabama extended local dialing parity port with Caller ID - res 2-Wire voice unbundles res, low usage line port with Caller ID (LUM) 2-Wire Voice Unbundled Alabama Residence Dialing Plan without Caller ID 2-Wire voice unbundled Low Usage Line Port without Caller ID Capability			UEPRX UEPRX UEPRX UEPRX	UEPRC UEPRO UEPAR UEPAP	1.15 1.15 1.15 1.15	40.19 40.19 40.19 40.19 40.19	19.83 19.83 19.83 19.83	24.91 24.91 24.91 24.91 24.91	6.63 6.63 6.63 6.63		15.66 15.66 15.66 15.66				
	FEATU	2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade unbundled Alabama extended local dialing parity port with Caller ID - res 2-Wire voice unbundles res, low usage line port with Caller ID (LUM) 2-Wire Voice Unbundled Alabama Residence Dialing Plan without Caller ID 2-Wire voice unbundled Low Usage Line Port without Caller ID Capability RES			UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPRC UEPRO UEPAR UEPAP UEPWA UEPRT	1.15 1.15 1.15 1.15 1.15 1.15 1.15	40.19 40.19 40.19 40.19 40.19 40.19 40.19	19.83 19.83 19.83 19.83 19.83 19.83 19.83	24.91 24.91 24.91 24.91 24.91 24.91	6.63 6.63 6.63 6.63 6.63		15.66 15.66 15.66 15.66 15.66 15.66				
	FEATU	2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade unbundled Alabama extended local dialing parity port with Caller ID - res 2-Wire voice unbundles res, low usage line port with Caller ID (LUM) 2-Wire Voice Unbundled Alabama Residence Dialing Plan without Caller ID 2-Wire voice unbundled Low Usage Line Port without Caller ID Capability RES All Features Offered			UEPRX UEPRX UEPRX UEPRX UEPRX	UEPRC UEPRO UEPAR UEPAP UEPWA	1.15 1.15 1.15 1.15 1.15 1.15	40.19 40.19 40.19 40.19 40.19 40.19	19.83 19.83 19.83 19.83 19.83 19.83	24.91 24.91 24.91 24.91 24.91 24.91	6.63 6.63 6.63 6.63 6.63		15.66 15.66 15.66 15.66 15.66				
	FEATU	2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade unbundled Alabama extended local dialing parity port with Caller ID - res 2-Wire voice unbundles res, low usage line port with Caller ID (LUM) 2-Wire Voice Unbundled Alabama Residence Dialing Plan without Caller ID 2-Wire voice unbundled Low Usage Line Port without Caller ID Capability IRES AII Features Offered NUMBER PORTABILITY			UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPRC UEPAR UEPAR UEPAP UEPWA UEPRT UEPVF	1.15 1.15 1.15 1.15 1.15 1.15 1.15 1.15	40.19 40.19 40.19 40.19 40.19 40.19 40.19	19.83 19.83 19.83 19.83 19.83 19.83 19.83	24.91 24.91 24.91 24.91 24.91 24.91	6.63 6.63 6.63 6.63 6.63		15.66 15.66 15.66 15.66 15.66 15.66				
	FEATU	2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade unbundled Alabama extended local dialing parity port with Caller ID - res 2-Wire voice unbundles res, low usage line port with Caller ID (LUM) 2-Wire Voice Unbundled Alabama Residence Dialing Plan without Caller ID 2-Wire voice unbundled Low Usage Line Port without Caller ID Capability RES All Features Offered			UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPRC UEPRO UEPAR UEPAP UEPWA UEPRT	1.15 1.15 1.15 1.15 1.15 1.15 1.15	40.19 40.19 40.19 40.19 40.19 40.19 40.19	19.83 19.83 19.83 19.83 19.83 19.83 19.83	24.91 24.91 24.91 24.91 24.91 24.91	6.63 6.63 6.63 6.63 6.63		15.66 15.66 15.66 15.66 15.66 15.66				
	FEATU	2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade unbundled Alabama extended local dialing parity port with Caller ID - res 2-Wire voice unbundles res, low usage line port with Caller ID (LUM) 2-Wire Voice Unbundled Alabama Residence Dialing Plan without Caller ID 2-Wire voice unbundled Low Usage Line Port without Caller ID Capability RES All Features Offered NUMBER PORTABILITY Local Number Portability (1 per port) ECURRING CHARGES (NRCs) - CURRENTLY COMBINED 2-Wire Voice Grade Loop / Line Port Combination - Conversion -			UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPRC UEPAR UEPAR UEPAP UEPWA UEPWA UEPVF LNPCX	1.15 1.15 1.15 1.15 1.15 1.15 1.15 1.15	40.19 40.19 40.19 40.19 40.19 40.19 0.00	19.83 19.83 19.83 19.83 19.83 19.83 19.83 0.00	24.91 24.91 24.91 24.91 24.91 24.91	6.63 6.63 6.63 6.63 6.63		15.66 15.66 15.66 15.66 15.66 15.66 15.66				
	FEATU	2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade unbundled Alabama extended local dialing parity port with Caller ID - res 2-Wire voice unbundles res, low usage line port with Caller ID (LUM) 2-Wire Voice Unbundled Alabama Residence Dialing Plan without Caller ID 2-Wire voice unbundled Low Usage Line Port without Caller ID 2-Wire voice unbundled Low Usage Line Port without Caller ID 2-Wire Voice Unbundled Low Usage Line Port without Caller ID 2-Wire Voice Unbundled Low Usage Line Port without Caller ID Capability RES All Features Offered - NUMBER PORTABILITY Local Number Portability (1 per port) 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is			UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPRC UEPAR UEPAR UEPAP UEPWA UEPRT UEPVF	1.15 1.15 1.15 1.15 1.15 1.15 1.15 1.15	40.19 40.19 40.19 40.19 40.19 40.19 40.19	19.83 19.83 19.83 19.83 19.83 19.83 19.83	24.91 24.91 24.91 24.91 24.91 24.91	6.63 6.63 6.63 6.63 6.63		15.66 15.66 15.66 15.66 15.66 15.66				
	FEATU	2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade unbundled Alabama extended local dialing parity port with Caller ID - res 2-Wire voice unbundles res, low usage line port with Caller ID (LUM) 2-Wire voice Unbundled Alabama Residence Dialing Plan without Caller ID 2-Wire voice Unbundled Low Usage Line Port without Caller ID 2-Wire voice unbundled Low Usage Line Port without Caller ID 2-Wire voice Unbundled Low Usage Line Port without Caller ID 2-Wire voice Unbundled Low Usage Line Port without Caller ID 2-Wire voice Unbundled Low Usage Line Port without Caller ID Capability RES All Features Offered NUMBER PORTABILITY Local Number Portability (1 per port) CURRING CHARGES (NRCs) - CURRENTLY COMBINED 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is IoNAL NRCs			UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPRC UEPAR UEPAR UEPAP UEPWA UEPWA UEPVF LNPCX	1.15 1.15 1.15 1.15 1.15 1.15 1.15 1.15	40.19 40.19 40.19 40.19 40.19 40.19 0.00	19.83 19.83 19.83 19.83 19.83 19.83 19.83 0.00	24.91 24.91 24.91 24.91 24.91 24.91	6.63 6.63 6.63 6.63 6.63		15.66 15.66 15.66 15.66 15.66 15.66 15.66				
	FEATU	2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade unbundled Alabama extended local dialing parity port with Caller ID - res 2-Wire voice unbundled Rabama Residence Dialing Plan without Caller ID 2-Wire voice Unbundled Alabama Residence Dialing Plan without Caller ID 2-Wire voice unbundled Low Usage Line Port without Caller ID Capability RES All Features Offered AUMBER PORTABILITY Local Number Portability (1 per port) CURRING CHARGES (NRCs) - CURRENTLY COMBINED 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is 2-Wire Voice Grade Loop/Line Port Combination - Subsequent			UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPRC UEPAR UEPAR UEPAP UEPWA UEPRT UEPVF LNPCX USAC2	1.15 1.15 1.15 1.15 1.15 1.15 1.15 1.98 0.35	40.19 40.19 40.19 40.19 40.19 40.19 40.19 0.00 0.00	19.83 19.83 19.83 19.83 19.83 19.83 19.83 0.00 0.10	24.91 24.91 24.91 24.91 24.91 24.91	6.63 6.63 6.63 6.63 6.63		15.66 15.66 15.66 15.66 15.66 15.66 15.66				
	FEATU	2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade unbundled Alabama extended local dialing parity port with Caller ID - res 2-Wire voice unbundles res, low usage line port with Caller ID (LUM) 2-Wire voice Unbundled Alabama Residence Dialing Plan without Caller ID 2-Wire voice unbundled Low Usage Line Port without Caller ID Capability RES All Features Offered			UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPRC UEPAR UEPAR UEPAP UEPWA UEPWA UEPVF LNPCX	1.15 1.15 1.15 1.15 1.15 1.15 1.15 1.15	40.19 40.19 40.19 40.19 40.19 40.19 0.00	19.83 19.83 19.83 19.83 19.83 19.83 19.83 0.00	24.91 24.91 24.91 24.91 24.91 24.91	6.63 6.63 6.63 6.63 6.63		15.66 15.66 15.66 15.66 15.66 15.66 15.66				
	FEATU LOCAL NONRI ADDIT 2-WIRE	2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade unbundled Alabama extended local dialing parity port with Caller ID - res 2-Wire voice unbundles res, low usage line port with Caller ID (LUM) 2-Wire voice Unbundled Alabama Residence Dialing Plan without Caller ID 2-Wire voice Unbundled Low Usage Line Port without Caller ID 2-Wire voice Unbundled Low Usage Line Port without Caller ID 2-Wire voice Unbundled Low Usage Line Port without Caller ID 2-Wire Voice Grade Low (1) RES All Features Offered			UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPRC UEPAR UEPAR UEPAP UEPWA UEPRT UEPVF LNPCX USAC2	1.15 1.15 1.15 1.15 1.15 1.15 1.15 1.98 0.35	40.19 40.19 40.19 40.19 40.19 40.19 40.19 0.00 0.00	19.83 19.83 19.83 19.83 19.83 19.83 19.83 0.00 0.10	24.91 24.91 24.91 24.91 24.91 24.91	6.63 6.63 6.63 6.63 6.63		15.66 15.66 15.66 15.66 15.66 15.66 15.66				
	FEATU LOCAL NONRI ADDIT 2-WIRE	2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade unbundled Alabama extended local dialing parity port with Caller ID - res 2-Wire voice unbundles res, low usage line port with Caller ID (LUM) 2-Wire voice Unbundled Alabama Residence Dialing Plan without Caller ID 2-Wire voice unbundled Low Usage Line Port without Caller ID Capability RES All Features Offered NUMBER PORTABILITY Local Number Portability (1 per port) ECURRING CHARGES (NRCs) - CURRENTLY COMBINED 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is 2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS) ort/Loop Combination Rates			UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPRC UEPAR UEPAR UEPAP UEPWA UEPRT UEPVF LNPCX USAC2	1.15 1.15 1.15 1.15 1.15 1.15 1.15 0.35 0.35	40.19 40.19 40.19 40.19 40.19 40.19 40.19 0.00 0.00	19.83 19.83 19.83 19.83 19.83 19.83 19.83 0.00 0.10	24.91 24.91 24.91 24.91 24.91 24.91	6.63 6.63 6.63 6.63 6.63		15.66 15.66 15.66 15.66 15.66 15.66 15.66				
	FEATU LOCAL NONRI ADDIT 2-WIRE	2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade unbundled Alabama extended local dialing parity port with Caller ID - res 2-Wire voice unbundles res, low usage line port with Caller ID (LUM) 2-Wire Voice Unbundled Alabama Residence Dialing Plan without Caller ID 2-Wire voice unbundled Low Usage Line Port without Caller ID 2-Wire voice unbundled Low Usage Line Port without Caller ID 2-Wire voice unbundled Low Usage Line Port without Caller ID 2-Wire Voice Unbundled Low Usage Line Port without Caller ID Capability RES All Features Offered			UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPRC UEPAR UEPAR UEPAP UEPWA UEPRT UEPVF LNPCX USAC2	1.15 1.15 1.15 1.15 1.15 1.15 1.15 1.98 0.35 0.35	40.19 40.19 40.19 40.19 40.19 40.19 40.19 0.00 0.00	19.83 19.83 19.83 19.83 19.83 19.83 19.83 0.00 0.10	24.91 24.91 24.91 24.91 24.91 24.91	6.63 6.63 6.63 6.63 6.63		15.66 15.66 15.66 15.66 15.66 15.66 15.66				
	FEATU LOCAL NONRI ADDIT 2-WIRE	2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade unbundled Alabama extended local dialing parity port with Caller ID - res 2-Wire voice Unbundles res, low usage line port with Caller ID (LUM) 2-Wire voice Unbundled Alabama Residence Dialing Plan without Caller ID 2-Wire voice Unbundled Low Usage Line Port without Caller ID 2-Wire voice unbundled Low Usage Line Port without Caller ID 2-Wire voice Unbundled Low Usage Line Port without Caller ID 2-Wire voice Unbundled Low Usage Line Port without Caller ID 2-Wire voice Unbundled Low Usage Line Port without Caller ID Capability IRES All Features Offered - NUMBER PORTABILITY Local Number Portability (1 per port) ECURRING CHARGES (NRCs) - CURRENTLY COMBINED 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is EVOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS) ort/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2			UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPRC UEPAR UEPAR UEPAP UEPWA UEPRT UEPVF LNPCX USAC2	1.15 1.15 1.15 1.15 1.15 1.15 1.15 0.35 0.35	40.19 40.19 40.19 40.19 40.19 40.19 40.19 0.00 0.00	19.83 19.83 19.83 19.83 19.83 19.83 19.83 0.00 0.10	24.91 24.91 24.91 24.91 24.91 24.91	6.63 6.63 6.63 6.63 6.63		15.66 15.66 15.66 15.66 15.66 15.66 15.66				
	FEATU LOCAL NONRI ADDIT 2-WIRE P	2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade unbundled Alabama extended local dialing parity port with Caller ID - res 2-Wire voice unbundles res, low usage line port with Caller ID (LUM) 2-Wire Voice Unbundled Alabama Residence Dialing Plan without Caller ID 2-Wire voice unbundled Low Usage Line Port without Caller ID 2-Wire voice unbundled Low Usage Line Port without Caller ID 2-Wire voice unbundled Low Usage Line Port without Caller ID 2-Wire Voice Unbundled Low Usage Line Port without Caller ID Capability RES All Features Offered			UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPRC UEPRO UEPAR UEPAP UEPWA UEPRT UEPVF UEPVF USAC2 USAS2	1.15 1.15 1.15 1.15 1.15 1.15 1.15 0.35 0.35 0.35 0.35 0.00 0.00 12.70 21.19 34.80	40.19 40.19 40.19 40.19 40.19 40.19 40.19 0.00 0.00	19.83 19.83 19.83 19.83 19.83 19.83 19.83 0.00 0.10	24.91 24.91 24.91 24.91 24.91 24.91	6.63 6.63 6.63 6.63 6.63		15.66 15.66 15.66 15.66 15.66 15.66 15.66				
	FEATU LOCAL NONRI ADDIT 2-WIRE P	2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade unbundled Alabama extended local dialing parity port with Caller ID - res 2-Wire voice unbundles res, low usage line port with Caller ID (LUM) 2-Wire Voice Unbundled Alabama Residence Dialing Plan without Caller ID 2-Wire voice unbundled Low Usage Line Port without Caller ID 2-Wire voice unbundled Low Usage Line Port without Caller ID 2-Wire voice unbundled Low Usage Line Port without Caller ID 2-Wire voice Unbundled Low Usage Line Port without Caller ID Capability RES All Features Offered - NUMBER PORTABILITY Local Number Portability (1 per port) CURRING CHARGES (NRCs) - CURRENTLY COMBINED 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is IONAL NRCs 2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS) ort/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 oop Rates 2-Wire Voice Grade Loop (SL1) - Zone 1		3	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPRC UEPRO UEPAR UEPAP UEPWA UEPRT UEPVF UEPVF USAC2 USAS2 USAS2	1.15 1.15 1.15 1.15 1.15 1.15 1.98 0.35 0.35 0.35 0.00 12.70 21.19 34.80 11.55	40.19 40.19 40.19 40.19 40.19 40.19 40.19 0.00 0.00	19.83 19.83 19.83 19.83 19.83 19.83 19.83 0.00 0.10	24.91 24.91 24.91 24.91 24.91 24.91	6.63 6.63 6.63 6.63 6.63		15.66 15.66 15.66 15.66 15.66 15.66 15.66				
	FEATU LOCAL NONRI ADDIT 2-WIRE P	2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade unbundled Alabama extended local dialing parity port with Caller ID - res 2-Wire voice Unbundles res, low usage line port with Caller ID (LUM) 2-Wire Voice Unbundled Alabama Residence Dialing Plan without Caller ID 2-Wire voice unbundled Low Usage Line Port without Caller ID Capability RES All Features Offered			UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPRC UEPRO UEPAR UEPAP UEPWA UEPRT UEPVF UEPVF USAC2 USAS2	1.15 1.15 1.15 1.15 1.15 1.15 1.15 0.35 0.35 0.35 0.35 0.00 0.00 12.70 21.19 34.80	40.19 40.19 40.19 40.19 40.19 40.19 40.19 0.00 0.00	19.83 19.83 19.83 19.83 19.83 19.83 19.83 0.00 0.10	24.91 24.91 24.91 24.91 24.91 24.91	6.63 6.63 6.63 6.63 6.63		15.66 15.66 15.66 15.66 15.66 15.66 15.66				

UNBUNDL	ED NETWORK ELEMENTS - Alabama												Attachment:			ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge - Manual Sv Order vs.
						Rec	Nonrec	urring	Nonrecurring	g Disconnect				Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
2-Wi	re Voice Grade Line Port (Bus)															<u> </u>
	2-Wire voice unbundled port without Caller ID - bus			UEPBX	UEPBL	1.15	40.19	19.83	24.91	6.63		15.66				<u> </u>
	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPBX	UEPBC	1.15	40.19	19.83	24.91	6.63		15.66			-	
	2-Wire voice unbundled port outgoing only - bus 2-Wire voice Grade unbundled Alabama extended local dialing			UEPBX	UEPBO	1.15	40.19	19.83	24.91	6.63		15.66				
	parity port with Caller ID - bus			UEPBX	UEPAW	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire voice unbundled incoming only port with Caller ID - Bus			UEPBX	UPEB1	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Unbundled Alabama Business Dialing Plan without			OEI DX	OF ED I	1.10	40.10	10.00	24.01	0.00		10.00				ł
	Caller ID			UEPBX	UEPWB	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire voice unbundled Incoming Only Port without Caller ID															1
	Capability			UEPBX	UEPBE	1.15	40.19	19.83	24.91	6.63		15.66				
LOC	AL NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPBX	LNPCX	0.35										
FEA	TURES															
	All Features Offered			UEPBX	UEPVF	1.98	0.00	0.00				15.66		ļ	ļ	───
NON	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -															
400	Switch-as-is			UEPBX	USAC2		0.10	0.10				15.66				
ADD	TIONAL NRCs				-											+
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent						0.00	0.00				45.00				
2 WI	Activity RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)			UEPBX	USAS2		0.00	0.00				15.66				
	Port/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			12.70										+
	2-Wire VG Loop/Port Combo - Zone 2		2			21.19										+
	2-Wire VG Loop/Port Combo - Zone 3		3			34.80										1
UNE	Loop Rates		Ű			0 1.00										1
-	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEPRG	UEPLX	11.55										1
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEPRG	UEPLX	20.04										1
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEPRG	UEPLX	33.65										1
2-Wi	re Voice Grade Line Port Rates (RES - PBX)															
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port -															T
	Res			UEPRG	UEPRD	1.15	69.08	32.41	37.43	6.20		15.66				
LOC	AL NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPRG	LNPCP	3.15	0.00	0.00				15.66				
FEA						1.00										
				UEPRG	UEPVF	1.98	0.00	0.00				15.66				
NON	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -				-											+
	Conversion - Switch-As-Is			UEPRG	USAC2		7.91	1.90				15.66				
	TIONAL NRCs			UEFRG	USACZ		7.91	1.90				15.00				+
ADD	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -															+
	Subsequent Activity			UEPRG	USAS2	0.00	0.00	0.00				15.66				
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt	1				0.00	0.00	0.00	1	1		.0.00	1		ł	1
	Group	1	1				7.32	7.32				15.66			1	
2-WI	RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)	1	1												1	1
	Port/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			12.70										
	2-Wire VG Loop/Port Combo - Zone 2		2			21.19										
	2-Wire VG Loop/Port Combo - Zone 3		3			34.80									ļ	
UNE	Loop Rates															───
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEPPX	UEPLX	11.55										───
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEPPX	UEPLX	20.04									ļ	
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEPPX	UEPLX	33.65					<u> </u>				ļ	<u> </u>
2-Wi	re Voice Grade Line Port Rates (BUS - PBX)														<u> </u>	
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPPX	UEPPC	4 45	60.00	22.44	27 /2	6.00		15 60			1	
	TLINE SIDE UNDUNDIED COMPINATION 2-WAY PBA (TUNK PORT - BUS	1				1.15	69.08	32.41	37.43	6.20		15.66			L	
	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPPX	UEPPO	1.15	69.08	32.41	37.43	6.20		15.66				

UNBUNDL	ED NETWORK ELEMENTS - Alabama												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Unbundled 2-Way Combination PBX Alabama															
	Calling Port			UEPPX	UEPA2	1.15	69.08	32.41	37.43	6.20		15.66				
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPPX	UEPLD	1.15	69.08	32.41	37.43	6.20		15.66				L
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPPX	UEPXA	1.15	69.08	32.41	37.43	6.20		15.66				<u> </u>
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports 2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPPX UEPPX	UEPXB UEPXC	1.15 1.15	69.08 69.08	32.41 32.41	37.43 37.43	6.20 6.20		15.66 15.66				┢────
	2-Wire Voice Unbundled PBX LD DDD Terminals Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPPX	UEPXC	1.15	69.08	32.41	37.43	6.20		15.66				<u> </u>
	2-Wire Voice Unbundled PBX LD Terminal Switchboard PDT 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD			UEFFA	UEFAD	1.15	69.06	32.41	37.43	0.20		15.00				<u> </u>
	Capable Port			UEPPX	UEPXE	1.15	69.08	32.41	37.43	6.20		15.66				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy			OLITX	OLIXE	1.15	03.00	52.41	57.45	0.20		15.00				
	Administrative Calling Port			UEPPX	UEPXL	1.15	69.08	32.41	37.43	6.20		15.66				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy	1			JEI AL	1.13	03.00	52.71	51.45	0.20	1	10.00				
	Room Calling Port			UEPPX	UEPXM	1.15	69.08	32.41	37.43	6.20		15.66				1
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital						00.00	02.71	00	0.20		.0.00	1	1	1	<u> </u>
	Discount Room Calling Port			UEPPX	UEPXO	1.15	69.08	32.41	37.43	6.20		15.66				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPPX	UEPXS	1.15	69.08	32.41	37.43	6.20		15.66				
LOCA	AL NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPPX	LNPCP	3.15	0.00	0.00				15.66				
FEAT	URES															
	All Features Offered			UEPPX	UEPVF	1.98	0.00	0.00				15.66				
NONE	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -															
	Conversion - Switch-As-Is			UEPPX	USAC2		7.91	1.90				15.66				
ADDI	TIONAL NRCs															
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -															
	Subsequent Activity			UEPPX	USAS2	0.00	0.00	0.00				15.66				
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt						7.00	=				1 = 0.0				
0 14/15	Group RE VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN POP	. <u>.</u>					7.32	7.32				15.66				ł
					-											
UNE	Port/Loop Combination Rates 2-Wire VG Coin Port/Loop Combo – Zone 1		1			12.70										
	2-Wire VG Coin Port/Loop Combo – Zone 1		2			21.19										ł
	2-Wire VG Coin Port/Loop Combo – Zone 3		3			34.80										l
UNF	Loop Rates		Ŭ			04.00										
0.12	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPCO	UEPLX	11.55										
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPCO	UEPLX	20.04										
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPCO	UEPLX	33.65										
2-Wir	e Voice Grade Line Ports (COIN)															
1	2-Wire Coin 2-Way without Operator Screening and without															
	Blocking (AL, KY, LA, MS)			UEPCO	UEPRF	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Coin 2-Way with Operator Screening (AL, KY)			UEPCO	UEPRE	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Coin 2-Way with Operator Screening and Blocking: 011,															1
	900/976, 1+DDD (AL, KY, LA, MS)			UEPCO	UEPRA	1.15	40.19	19.83	24.91	6.63		15.66				ļ
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking															1
	(AL, LA, MS)			UEPCO	UEPRB	1.15	40.19	19.83	24.91	6.63	ļ	15.66				I
	2-Wire Coin 2-Way with Operator Screening & Blocking:			115000			10.10	10.00	04.54	0.00		45.00				1
	900/976, 1+DDD, 011+, & Local (AL, KY, LA, MS)			UEPCO	UEPCD	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Coin Outward with Operator Screening and 011 Blocking (AL, FL)			UEPCO	UEPRK	1.15	40.19	19.83	24.91	6.63		15.66				1
	2-Wire Coin Outward with Operator Screening and Blocking:				ULFIXN	1.10	40.19	19.03	24.91	0.03	1	10.00	ł	ł	ł	<u> </u>
	2-wire Con Outward with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, KY, LA, MS)			UEPCO	UEPRH	1.15	40.19	19.83	24.91	6.63		15.66				1
	2-Wire Coin Outward Operator Screening & Blocking: 900/976,	1			021101	1.13	40.13	13.03	27.31	0.03	1	10.00				
	1+DDD, 011+, and Local (AL, KY, LA, MS)			UEPCO	UEPCN	1.15	40.19	19.83	24.91	6.63		15.66				1
	2-Wire 2-Way Smartline with 900/976 (all states except LA)			UEPCO	UEPCK	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Coin Outward Smartline with 900/976 (all states except	1								2.50	1					
	LA)			UEPCO	UEPCR	1.15	40.19	19.83	24.91	6.63		15.66				1
ADDI	TIONAL UNE COIN PORT/LOOP (RC)	1									1					
	UNE Coin Port/Loop Combo Usage (Flat Rate)			UEPCO	URECU	1.56	40.19	19.83	24.91	6.63	1	15.66				
	AL NUMBER PORTABILITY										1					

UNBUNDL	ED NETWORK ELEMENTS - Alabama												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Bee	Nonred	curring	Nonrecurring	g Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Local Number Portability (1 per port)			UEPCO	LNPCX	0.35										
NON	RECURRING CHARGES - CURRENTLY COMBINED 2-Wire Voice Grade Loop / Line Port Combination - Conversion															
	Switch-as-is	-		UEPCO	USAC2		0.10	0.10				15.66				
ADD	ITIONAL NRCs			02.00	00/102		0.110	0.10				10.00				
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent															
0.14/	Activity			UEPCO	USAS2		0.00	0.00				15.66				
	RE VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIR Port/Loop Combination Rates			RES)												
0.11	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			15.76										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			24.23										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			37.52										
UNE	Loop Rates 2-Wire Voice Grade Loop (SL2) - Zone 1	<u> </u>	4	UEPFR	UECF2	14.38					ļ	ļ				
	2-Wire Voice Grade Loop (SL2) - Zone 1 2-Wire Voice Grade Loop (SL2) - Zone 2		1	UEPFR	UECF2 UECF2	14.38 22.85						<u> </u>				
	2-Wire Voice Grade Loop (SL2) - Zone 2 2-Wire Voice Grade Loop (SL2) - Zone 3	1	3	UEPFR	UECF2	36.14					1	1				
2-W	re Voice Grade Line Port Rates (Res)															
	2-Wire voice unbundled port - residence			UEPFR	UEPRL	1.38	90.38	57.27	48.66	8.77		15.66				
	2-Wire voice unbundled port with Caller ID - res			UEPFR UEPFR	UEPRC UEPRO	1.38	90.38	57.27	48.66	8.77		15.66				
	2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade unbundled Alabama extended local dialing	-		UEPFR	UEPRO	1.38	90.38	57.27	48.66	8.77		15.66				
	parity port with Caller ID - res			UEPFR	UEPAR	1.38	90.38	57.27	48.66	8.77		15.66				
	2-Wire voice unbundles res, low usage line port with Caller ID															
	(LUM)			UEPFR	UEPAP	1.38	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Unbundled Alabama Residence Dialing Plan without Caller ID			UEPFR	UEPWA	1.38	90.38	57.27	48.66	8.77		15.66				
INTE	ROFFICE TRANSPORT			OLFTR	OLFWA	1.30	90.30	51.21	40.00	0.77		13.00				
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility															
	Termination			UEPFR	U1TV2	21.13	40.54	27.41	16.74	6.90						
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile				1L5XX	0.000000										
FEA	or Fraction Mile TURES	-		UEPFR	1L5XX	0.008838										
1 50	All Features Offered			UEPFR	UEPVF	1.98	0.00	0.00				15.66				
LOC	AL NUMBER PORTABILITY			-	-											
	Local Number Portability (1 per port)			UEPFR	LNPCX	0.35										
NON	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port Combination - Conversion - Switch-as-is			UEPFR	USAC2		8.48	1.87				15.66				
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port			0EITIK	00/102		0.40	1.07				10.00				
	Combination - Conversion - Switch-With-Change			UEPFR	USACC		8.48	1.87				15.66				
	RE VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIR	E LINE I	PORT (BUS)												
UNE	Port/Loop Combination Rates		4			45.70										
├	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1 2-Wire VG Loop/IO Tranport/Port Combo - Zone 2	+	1	{		15.76 24.23					<u> </u>	<u> </u>				
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3	1	3			37.52					1	1				
UNE	Loop Rates		Ĺ													
	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFB	UECF2	14.38										
	2-Wire Voice Grade Loop (SL2) - Zone 2 2-Wire Voice Grade Loop (SL2) - Zone 3	<u> </u>	2	UEPFB UEPFB	UECF2 UECF2	22.85 36.14					<u> </u>					
2-W	re Voice Grade Line Port (Bus)		3	ULPED	UEUF2	30.14					+	+				
	2-Wire voice unbundled port without Caller ID - bus	1	1	UEPFB	UEPBL	1.38	90.38	57.27	48.66	8.77		15.66				
	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPFB	UEPBC	1.38	90.38	57.27	48.66	8.77		15.66				
	2-Wire voice unbundled port outgoing only - bus			UEPFB	UEPBO	1.38	90.38	57.27	48.66	8.77		15.66				
	2-Wire voice Grade unbundled Alabama extended local dialing parity port with Caller ID - bus	1		UEPFB	UEPAW	1.38	90.38	57.27	48.66	8.77		15.66				
	2-Wire voice unbundled incoming only port with Caller ID - Bus			UEPFB	UEPAW UEPB1	1.38	90.38 90.38	57.27	48.66	8.77	+	15.66				
	2-Wire Voice Unbundled Alabama Business Dialing Plan without	1	1		02.01	1.00	00.00	01.21	40.00	0.11		10.00				
	Caller ID			UEPFB	UEPWB	1.38	90.38	57.27	48.66	8.77		15.66				
LOC	AL NUMBER PORTABILITY															

UNBUNDL	ED NETWORK ELEMENTS - Alabama												Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						Rec	Nonrec		Nonrecurring					Rates(\$)	•	
	Level March Bratel We (American)						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Local Number Portability (1 per port) ROFFICE TRANSPORT			UEPFB	LNPCX	0.35										┢────
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility															
	Termination			UEPFB	U1TV2	21.13	40.54	27.41	16.74	6.90						
1	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile															
	or Fraction Mile			UEPFB	1L5XX	0.008838										
FEA	TURES															
	All Features Offered			UEPFB	UEPVF	1.98	0.00	0.00				15.66				
NON	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port Combination - Conversion - Switch-as-is			UEPFB	USAC2		8.48	1.87				15.66				
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port			OLFIB	03A02		0.40	1.07				13.00				
	Combination - Conversion - Switch with change			UEPFB	USACC		8.48	1.87				15.66				
2-WI	RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)			-												
UNE	Port/Loop Combination Rates															
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			15.76										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			24.23										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			37.52										
UNE	Loop Rates				115050											l
	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFP	UECF2	14.38 22.85										I
	2-Wire Voice Grade Loop (SL2) - Zone 2 2-Wire Voice Grade Loop (SL2) - Zone 3			UEPFP UEPFP	UECF2 UECF2	36.14										
2-Wi	re Voice Grade Line Port Rates (BUS - PBX)		3	UEFFF	UECF2	30.14										<u> </u>
2-111	The voice Grade Line Fort Rates (DOG - T DA)															
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPFP	UEPPC	1.38	119.27	69.85	61.18	8.34		15.66				
	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPFP	UEPPO	1.38	119.27	69.85	61.18	8.34		15.66				
	Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPFP	UEPP1	1.38	119.27	69.85	61.18	8.34		15.66				
	2-Wire Voice Unbundled 2-Way Combination PBX Alabama															
	Calling Port			UEPFP	UEPA2	1.38	119.27	69.85	61.18	8.34		15.66				
<u> </u>	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPFP	UEPLD	1.38	119.27	69.85	61.18	8.34		15.66				
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port 2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPFP UEPFP	UEPXA UEPXB	1.38 1.38	119.27 119.27	69.85 69.85	61.18 61.18	8.34 8.34		15.66 15.66				
	2-Wire Voice Unbundled PBX Toll Terminal Hoter Ports			UEPFP	UEPXC	1.38	119.27	69.85	61.18	8.34		15.66				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPFP	UEPXD	1.38	119.27	69.85	61.18	8.34		15.66				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD															
	Capable Port			UEPFP	UEPXE	1.38	119.27	69.85	61.18	8.34		15.66				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy															
	Administrative Calling Port			UEPFP	UEPXL	1.38	119.27	69.85	61.18	8.34		15.66				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy	1	1													1
┝──┼──	Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital	 		UEPFP	UEPXM	1.38	119.27	69.85	61.18	8.34		15.66				
	2-wire voice Unbundled 1-way Outgoing PBX Hotel/Hospital Discount Room Calling Port			UEPFP	UEPXO	1.38	119.27	69.85	61.18	8.34		15.66				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPFP	UEPXS	1.38	119.27	69.85	61.18	8.34		15.66				
LOC	AL NUMBER PORTABILITY	1						66.50	00	0.04	1	.0.00	1	1	1	
	Local Number Portability (1 per port)			UEPFP	LNPCP	3.15	0.00	0.00				15.66				
INTE	ROFFICE TRANSPORT															
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility	1														1
	Termination	<u> </u>	L	UEPFP	U1TV2	21.13	40.54	27.41	16.74	6.90	ļ					Ļ
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile															
	or Fraction Mile TURES	 		UEPFP	1L5XX	0.008838										
FEA	All Features Offered	<u> </u>		UEPFP	UEPVF	1.98	0.00	0.00				15.66				<u> </u>
NON	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED	<u> </u>				1.98	0.00	0.00		ł	<u> </u>	10.00	ł	ł	{	t
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port	1														<u> </u>
	Combination - Conversion - Switch-as-is	1	1	UEPFP	USAC2		8.48	1.87				15.66				1
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port	1									1					
	Combination - Conversion - Switch with change			UEPFP	USACC		8.48	1.87				15.66				
	D PORT/LOOP COMBINATIONS - COST BASED RATES															
2-WI	RE VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK	PORT														

UNBUNDL	ED NETWORK ELEMENTS - Alabama												-	Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	E	SCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
							Rec	Nonrec	curring	Nonrecurring	g Disconnect			OSS	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
UNE	Port/Loop Combination Rates																
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1		1				22.40										
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2 2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3	_	2				30.88 44.17										
LINE	2-Wire VG Loop/2-Wire DID Trunk Port Combo - ONE Zone 3		3				44.17										
UNE	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1		1	UEPPX		UECD1	14.38					1					
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2		2	UEPPX		UECD1	22.85										
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3		3	UEPPX		UECD1	36.14										
UNE	Port Rate			02.17		02001	00.11										
-	Exchange Ports - 2-Wire DID Port			UEPPX		UEPD1	8.02	207.31	73.74	107.14	11.20		15.66				
NON	RECURRING CHARGES - CURRENTLY COMBINED																
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination -	·															
	Switch-as-is			UEPPX		USAC1		7.31	1.87								
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion						Т										
	with BellSouth Allowable Changes			UEPPX		USA1C		7.31	1.87								
ADDI	TIONAL NRCs																
Talar	2-Wire DID Subsequent Activity - Add Trunks, Per Trunk	_		UEPPX		USAS1		26.78	26.78								
l elep	bhone Number/Trunk Group Establisment Charges DID Trunk Termination (One Per Port)			UEPPX		NDT	0.00	0.00	0.00								
	Additional DID Numbers for each Group of 20 DID Numbers	-		UEPPX		ND4	0.00	0.00	0.00								
	DID Numbers, Non- consecutive DID Numbers , Per Number			UEPPX		ND4 ND5	0.00	0.00	0.00								
	Reserve Non-Consecutive DID numbers			UEPPX		ND6	0.00	0.00	0.00								
	Reserve DID Numbers			UEPPX		NDV	0.00	0.00	0.00								
LOCA	AL NUMBER PORTABILITY																
	Local Number Portability (1 per port)			UEPPX		LNPCP	3.15	0.00	0.00								
2-WI	RE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LI	NE SIDE	PORT														
UNE	Port/Loop Combination Rates																
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 1		1	UEPPB	UEPPR		27.28										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 2		2	UEPPB	UEPPR		37.86										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 3		3	UEPPB	UEPPR		53.84										
LINE	Loop Rates		3	UEPPB	UEPPR		53.84										
ONL	2-Wire ISDN Digital Grade Loop - UNE Zone 1		1	UEPPB	UEPPR		19.03										
	2-Wire IODIN Digital Grade Loop - GINE Zone 1			OLITO	OLITIK	UULZA	13.05										
	2-Wire ISDN Digital Grade Loop - UNE Zone 2		2	UEPPB	UEPPR	USL2X	29.62										
	2-Wire ISDN Digital Grade Loop - UNE Zone 3		3	UEPPB	UEPPR		45.60										
UNE	Port Rate																
	Exchange Port - 2-Wire ISDN Line Side Port			UEPPB	UEPPR	UEPPB	8.24	190.01	132.76	100.67	21.28		15.66				
NON	RECURRING CHARGES - CURRENTLY COMBINED																
	2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port																
	Combination - Conversion	 		UEPPB	UEPPR	USACB	0.00	38.51	27.02				15.66		ļ		L
LOCA	AL NUMBER PORTABILITY	 		UEPPB	UEPPR	LNPCX	0.35	0.00	0.00								
B.CU	Local Number Portability (1 per port)		<u> </u>	UEPPB	UEPPR	LINPUX	0.35	0.00	0.00								
в-Сп	CVS/CSD (DMS/5ESS)	+		UEPPB	UEPPR	U1UCA	0.00	0.00	0.00								
	CVS/CSD (DNS/SESS) CVS (EWSD)	+	<u> </u>	UEPPB	UEPPR	U1UCB	0.00	0.00	0.00			1					
	CSD	1		UEPPB	UEPPR	U1UCC	0.00	0.00	0.00			1					
B-CH	IANNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS S	C.MS. 8	TN)		22	2.000	0.00	0.00	0.00					İ			
	CVS/CSD (DMS/5ESS)	, ,,,	.,	UEPPB	UEPPR	U1UCD	0.00	0.00	0.00			1		l		1	
	CVS (EWSD)	1		UEPPB	UEPPR	U1UCE	0.00	0.00	0.00			1		l		1	
İ.	CSD			UEPPB	UEPPR	U1UCF	0.00	0.00	0.00								
USEF	R TERMINAL PROFILE																
	User Terminal Profile (EWSD only)			UEPPB	UEPPR	U1UMA	0.00	0.00	0.00								
VERT	TICAL FEATURES																
	All Vertical Features - One per Channel B User Profile	I		UEPPB	UEPPR	UEPVF	1.98	0.00	0.00			ļ					
INTE	ROFFICE CHANNEL MILEAGE	1															

BUNDLED	D NETWORK ELEMENTS - Alabama												Attachment:			ibit: B
EGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Charge -	Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Interoffice Channel mileage each, including first mile and															
	facilities termination			UEPPB UEPPR	M1GNC	21.14	40.54	27.41	16.74	6.90						
	Interoffice Channel mileage each, additional mile			UEPPB UEPPR	M1GNM	0.008838	0.00	0.00				0.00				
	DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK	PORT														
	ort/Loop Combination Rates															
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE															
	Zone 1		1	UEPPP		166.87										
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE		_													
	Zone 2		2	UEPPP		238.50										
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE															
	Zone 3		3	UEPPP		398.85										l
	oop Rates										l					
	4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPPP	USL4P	82.55										L
	4-Wire DS1 Digital Loop - UNE Zone 2		2	UEPPP	USL4P	154.18										
	4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPPP	USL4P	314.52										
	ort Rate															
	Exchange Ports - 4-Wire ISDN DS1 Port			UEPPP	UEPPP	84.32	456.28	259.10	123.88	31.77		15.66				
	CURRING CHARGES - CURRENTLY COMBINED															
	4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port															
	Combination - Conversion -Switch-as-is			UEPPP	USACP	0.00	119.07	78.56				15.66				
ADDITI	ONAL NRCs															
	4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy-															
	Inward/two way Tel Nos. (except NC)			UEPPP	PR7TF		0.49									
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port -															
	Outward Tel Numbers (All States except NC)			UEPPP	PR7TO		11.51									
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port -						-									
	Subsequent Inward Tel Numbers			UEPPP	PR7ZT		23.02									
LOCAL	NUMBER PORTABILITY		1													
	Local Number Portability (1 per port)			UEPPP	LNPCN	1.75										
	ACE (Provsioning Only)															
	Voice/Data			UEPPP	PR71V	0.00	0.00	0.00								
	Digital Data			UEPPP	PR71D	0.00	0.00	0.00								
	Inward Data			UEPPP	PR71E	0.00	0.00	0.00								
	Additional "B" Channel			02		0.00	0.00	0.00								
	New or Additional - Voice/Data B Channel			UEPPP	PR7BV	0.00	14.53									
	New or Additional - Digital Data B Channel			UEPPP	PR7BF	0.00	14.53									
	New of Additional Inward Data B Channel			UEPPP	PR7BD	0.00	14.53									
CALL T				02111	11000	0.00	14.00									
	Inward			UEPPP	PR7C1	0.00	0.00	0.00								-
	Outward			UEPPP	PR7C0	0.00	0.00	0.00								-
	Two-way	<u> </u>	+	UEPPP	PR7CC	0.00	0.00	0.00			t					1
	ice Channel Mileage			ULFFF	FRICC	0.00	0.00	0.00								
	Fixed Each Including First Mile			UEPPP	1LN1A	60.34	89.27	81.81	16.35	14.44		15.66				
	Each Airline-Fractional Additional Mile			UEPPP	1LN1A 1LN1B	0.18	09.27	01.01	10.35	14.44		10.00				├───
	Each Ainine-Fractional Additional Mile			ULPPP	ILINID	0.18										
	DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRONK PORT					├		1								
			1	UEPDC		140.64		1								
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1	-	1			142.64 214.26					+					
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2	-	2	UEPDC							+					
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3	-	3	UEPDC		374.61					+					
	A Wire DS1 Digital Loop LINE Zopo 1		1			00.55										
	4-Wire DS1 Digital Loop - UNE Zone 1	-		UEPDC	USLDC	82.55					+					
	4-Wire DS1 Digital Loop - UNE Zone 2		2	UEPDC	USLDC	154.18					l					
	4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPDC	USLDC	314.52					l					
UNE Po			 			00.00	151.10	050.00	117.00	44.25	l	45.00				
	4-Wire DDITS Digital Trunk Port		 	UEPDC	UDD1T	60.09	454.49	253.23	117.29	14.17	l	15.66				───
INONRE	CURRING CHARGES - CURRENTLY COMBINED		 								l					───
		1	1	1	1				1	1	1	1			1	1
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Switch-as-is			UEPDC	USAC4		129.49	67.02				15.66				

IBUNDLE	D NETWORK ELEMENTS - Alabama		-		-								Attachment:		Exhi	bit: B
TEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted	Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						1.00	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
	- Conversion with DS1 Changes			UEPDC	USAWA		129.49	67.02				15.66				
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with Change - Trunk			UEPDC	USAWB		129.49	67.02				45.00				
	IONAL NRCs			UEPDC	USAWB		129.49	67.02				15.66				
ADDIT	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC -															
	Subsequent Channel Activation/Chan - 2-Way Trunk			UEPDC	UDTTA		14.48	14.48				15.66				
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent															
	Channel Activation/Chan - 1-Way Outward Trunk			UEPDC	UDTTB		14.48	14.48				15.66				
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel															
	Activation/Chan Inward Trunk w/out DID			UEPDC	UDTTC		14.48	14.48				15.66				
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan		1													
_	Activation Per Chan - Inward Trunk with DID		<u> </u>	UEPDC	UDTTD		14.48	14.48				15.66				
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan		1									45.00				
BIROL	Activation / Chan - 2-Way DID w User Trans AR 8 ZERO SUBSTITUTION			UEPDC	UDTTE		14.48	14.48				15.66				
DIFUL	B8ZS -Superframe Format			UEPDC	CCOSF		0.00	600.00								
	B8ZS - Extended Superframe Format			UEPDC	CCOEF		0.00	600.00								
Alterna	ate Mark Inversion		1	021 00	OODEI		0.00	000.00								
/	AMI -Superframe Format			UEPDC	MCOSF		0.00	0.00			1					
	AMI - Extended SuperFrame Format			UEPDC	MCOPO		0.00	0.00								
Teleph	one Number/Trunk Group Establisment Charges															
	Telephone Number for 2-Way Trunk Group			UEPDC	UDTGX	0.00										
	Telephone Number for 1-Way Outward Trunk Group			UEPDC	UDTGY	0.00										
	Telephone Number for 1-Way Inward Trunk Group Without DID			UEPDC	UDTGZ	0.00										
	DID Numbers for each Group of 20 DID Numbers			UEPDC	ND4	0.00	0.00									
	DID Numbers, Non- consecutive DID Numbers , Per Number			UEPDC	ND5	0.00										
	Reserve Non-Consecutive DID Nos. Reserve DID Numbers			UEPDC UEPDC	ND6	0.00	0.00	0.00								
Dadiaa	ted DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1	1 Digito	Lloon		NDV	0.00	0.00	0.00								
Deulca	Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities	Digita	ГСООР	with 4-wire DDI13												
	Termination)			UEPDC	1LNO1	60.16	89.27	81.81	16.35	14.44		15.66				
			1	021 00	ILINO I	00.10	00.27	01.01	10.00	14.44		10.00				
	Interoffice Channel Mileage - Additional rate per mile - 0-8 miles			UEPDC	1LNOA	0.18	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities															
	Termination)			UEPDC	1LNO2	0.00	0.00	0.00								
	Interoffice Channel Mileage - Additional rate per mile - 9-25															
	miles			UEPDC	1LNOB	0.18	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities	1	1													
	Termination)		<u> </u>	UEPDC	1LNO3	0.00	0.00	0.00	0.00							
	Interoffice Channel Mileage - Additional rate per mile - 25+ miles	1	1	UEPDC	1LNOC	0.18	0.00	0.00								
-	Local Number Portability, per DS0 Activated		-	UEPDC	LNPCP	3.15	0.00	0.00	0.00		<u> </u>		ł	-	-	
-	Central Office Termininating Point		1	UEPDC	CTG	0.00	0.00	0.00	0.00							<u> </u>
4-WIRE	E DS1 LOOP WITH CHANNELIZATION WITH PORT		<u> </u>			0.00					t		1			-
	n is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Act	ivations	5	İ							1		l			l
	system can have up to 24 combinations of rates depending on			ber of ports used												
UNE D	S1 Loop															
	4-Wire DS1 Loop - UNE Zone 1		1	UEPMG	USLDC	82.55	0.00	0.00								
	4-Wire DS1 Loop - UNE Zone 2	I	2	UEPMG	USLDC	154.18	0.00	0.00			ļ					
	4-Wire DS1 Loop - UNE Zone 3	<u> </u>	3	UEPMG	USLDC	314.52	0.00	0.00								<u> </u>
UNE D	SO Channelization Capacities (D4 Channel Bank Configuration	ns)			VLIM24	101.40	0.00	0.00				ļ				
_	24 DSO Channel Capacity - 1 per DS1 48 DSO Channel Capacity - 1 per 2 DS1s	<u> </u>		UEPMG UEPMG	VUM24 VUM48	101.40 202.80	0.00	0.00			-					
	96 DSO Channel Capacity - 1 per 2 DS1s 96 DSO Channel Capacity -1per 4 DS1s			UEPMG	VUIVI48 VUM96	405.60	0.00	0.00								
1	144 DS0 Channel Capacity - 1 per 6 DS1s			UEPMG	VUM14	405.60 608.40	0.00	0.00			-					<u> </u>
			1		1.00014	500.40	0.00	0.00			1				L	<u> </u>
				UEPMG	VUM19	811.20	0.00	0.00								
	192 DS0 Channel Capacity -1 per 8 DS1s 240 DS0 Channel Capacity - 1 per 10 DS1s			UEPMG UEPMG	VUM19 VUM20	811.20 1,014.00	0.00	0.00								

UNBUNDL	ED NETWORK ELEMENTS - Alabama		-			-							Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						_	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)	1	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN		SOMAN	SOMAN	SOMAN
	384 DS0 Channel Capacity - 1 per 16 DS1s			UEPMG	VUM38	1,622.40	0.00	0.00								
	480 DS0 Channel Capacity - 1 per 20 DS1s			UEPMG	VUM40	2,028.00	0.00	0.00								
	576 DS0 Channel Capacity -1 per 24 DS1s			UEPMG	VUM57	2,433.60	0.00	0.00								
	672 DS0 Channel Capacity - 1 per 28 DS1s			UEPMG	VUM67	2,839.20	0.00	0.00								
	Recurring Charges (NRC) Associated with 4-Wire DS1 Loop with						stem									-
	nimum System configuration is One (1) DS1, One (1) D4 Channe iples of this configuration functioning as one are considered Ac															
Multi	NRC - Conversion (Currently Combined) with or without		r the fi	inninum system co	Iniguration is	counted.										
	BellSouth Allowed Changes			UEPMG	USAC4	0.00	150.48	8.36				15.66				
Syste	em Additions at End User Locations Where 4-Wire DS1 Loop wit	th Chan	nelizat					0.50				13.00				
	(Not Currently Combined) in all states, except in Density Zone 1															1
	1 DS1/D4 Channel Bank - Additionally Add NRC for each Port															
	and Assoc Fea Activation			UEPMG	VUMD4	0.00	716.11	468.04	148.75	17.65		15.66				
Bipol	lar 8 Zero Substitution															
	Clear Channel Capability Format, superframe - Subsequent															
	Activity Only			UEPMG	CCOSF	0.00	0.00	600.00								
	Clear Channel Capability Format - Extended Superframe -											1				
	Subsequent Activity Only			UEPMG	CCOEF	0.00	0.00	600.00								
Alterr	nate Mark Inversion (AMI)															
	Superframe Format			UEPMG	MCOSF	0.00	0.00	0.00								
Evolu	Extended Superframe Format ange Ports Associated with 4-Wire DS1 Loop with Channelizati	on with	Dort	UEPMG	MCOPO	0.00	0.00	0.00								
	ange Ports Associated with 4-wire DS1 Loop with Channelization	on with	Port		_											
EACING					-											
	Line Side Combination Channelized PBX Trunk Port - Business			UEPPX	UEPCX	1.15	0.00	0.00	0.00	0.00		15.66				
	Line Side Outward Channelized PBX Trunk Port - Business			UEPPX	UEPOX	1.15	0.00	0.00	0.00	0.00		15.66				
	Line Side Inward Only Channelized PBX Trunk Port without DID			UEPPX	UEP1X	1.15	0.00	0.00	0.00	0.00		15.66				
	2-Wire Trunk Side Unbundled Channelized DID Trunk Port			UEPPX	UEPDM	8.05	0.00	0.00	0.00	0.00		15.66				
	2-Wire Channelized PBX Area Calling Service Combination Port															
	(AL Only)			UEPPX	UEPA4	1.15	0.00	0.00				15.66				
	2 Wire Channelized PBX Area Calling Service Outgoing Only															
_	Port (AL Only)			UEPPX	UEPA3	1.15	0.00	0.00				15.66				
Featu	ure Activations - Unbundled Loop Concentration				-											
	Feature (Service) Activation for each Line Port Terminated in D4 Bank			UEPPX	1PQWM	0.56	54.55					45.00				
	Feature (Service) Activation for each Trunk Port Terminated in	-		UEPPX	TPQVVIVI	0.56	54.55					15.66				
	D4 Bank			UEPPX	1PQWU	0.56	77.03					15.66				
Telen	phone Number/ Group Establishment Charges for DID Service			52117		0.00	11.03				1	10.00				
	DID Trunk Termination (1 per Port)	l		UEPPX	NDT	0.00	0.00	0.00			1		1		1	1
	DID Numbers - groups of 20 - Valid all States			UEPPX	ND4	0.00	0.00	0.00								
	Non-Consecutive DID Numbers - per number	1		UEPPX	ND5	0.00	0.00	0.00			1			İ		
	Reserve Non-Consecutive DID Numbers			UEPPX	ND6	0.00	0.00	0.00								
	Reserve DID Numbers			UEPPX	NDV	0.00	0.00	0.00								
Local	I Number Portability															
	Local Number Portability - 1 per port			UEPPX	LNPCP	3.15	0.00	0.00								
	FURES - Vertical and Optional		ļ													
Local	I Switching Features Offered with Line Side Ports Only					4.00	0.00	0.00								
	All Features Available 2-Wire Voice Unbundled Alabama Business Dialing Plan without			UEPPX	UEPVF	1.98	0.00	0.00								
	2-Wire Voice Unbundled Alabama Business Dialing Plan without Caller ID			UEPBX	UEPWB	14.00	90.00	90.00				15.66				
2-1/10	RE VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE		POPT /		JEFWB	14.00	90.00	90.00				10.00				
	Port/Loop Combination Rates															
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			28.38			l							
1	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2	1	2	1	1	36.85							1	1	1	1
1	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3		1	50.14					1					
UNE	Loop Rates	1	-	İ							1		ĺ	ĺ	ĺ	1
	2-Wire Voice Grade Loop (SL2) - Zone 1	1	1	UEPFR	UECF2	14.38					1		ĺ	ĺ	ĺ	1
	2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFR	UECF2	22.85					1					1

UNBUNDLE	ED NETWORK ELEMENTS - Alabama												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFR	UECF2	36.14										
2-Wire	e Voice Grade Line Port Rates (Res)					44.00	105.00	00.00	70.00	45.00		45.00				
	2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res			UEPFR UEPFR	UEPRL UEPRC	14.00 14.00	125.00 125.00	80.00 80.00	70.00 70.00	15.00 15.00		15.66 15.66				
	2-Wire voice unbundled port outgoing only - res			UEPFR	UEPRO	14.00	125.00	80.00	70.00	15.00		15.66				
	2-Wire voice Grade unbundled Alabama extended local dialing parity port with Caller ID - res			UEPFR	UEPAR	14.00	125.00	80.00	70.00	15.00		15.66				
	2-Wire voice unbundles res, low usage line port with Caller ID (LUM)			UEPFR	UEPAP	14.00	125.00	80.00	70.00	15.00		15.66				
	2-Wire Voice Unbundled Alabama Residence Dialing Plan without Caller ID			UEPFR	UEPWA	14.00	125.00	80.00	70.00	15.00		15.66				
INTER	ROFFICE TRANSPORT															
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility Termination			UEPFR	U1TV2	21.13	40.54	27.41	16.74	6.90						
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile or Fraction Mile			UEPFR	1L5XX	0.008838										
FEAT	URES															
	All Features Offered			UEPFR	UEPVF	0.00	0.00	0.00				15.66				
LOCA					LNDOV	0.05										
NONE	Local Number Portability (1 per port) ECURRING CHARGES (NRCs) - CURRENTLY COMBINED			UEPFR	LNPCX	0.35										
NONE	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
	Combination - Conversion - Switch-as-is			UEPFR	USAC2		8.48	1.87				15.66				
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port			OLITIK	00,102		0.40	1.07				10.00				
	Combination - Conversion - Switch-With-Change			UEPFR	USACC		8.48	1.87				15.66				
2-WIR	E VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE	LINE F	ORT (BUS)												
UNE F	Port/Loop Combination Rates															
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			28.38										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			36.85										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			50.14										
UNE L	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFB	UECF2	14.38										
	2-Wire Voice Grade Loop (SL2) - Zone 1 2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFB	UECF2 UECF2	22.85										
	2-Wire Voice Grade Loop (SL2) - Zone 2 2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFB	UECF2	36.14										
2-Wire	e Voice Grade Line Port (Bus)		0	OLIT D	02012	00.14										
	2-Wire voice unbundled port without Caller ID - bus			UEPFB	UEPBL	14.00	125.00	80.00	70.00	15.00		15.66				
	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPFB	UEPBC	14.00	125.00	80.00	70.00	15.00		15.66				
	2-Wire voice unbundled port outgoing only - bus			UEPFB	UEPBO	14.00	125.00	80.00	70.00	15.00		15.66				
	2-Wire voice Grade unbundled Alabama extended local dialing															
	parity port with Caller ID - bus			UEPFB	UEPAW	14.00	125.00	80.00	70.00	15.00		15.66				
	2-Wire voice unbundled incoming only port with Caller ID - Bus			UEPFB	UEPB1	14.00	125.00	80.00	70.00	15.00		15.66				
	2-Wire Voice Unbundled Alabama Business Dialing Plan without Caller ID			UEPFB	UEPWB	14.00	125.00	80.00	70.00	15.00		15.66				
LOCA	L NUMBER PORTABILITY			UEPFB	LNPCX	0.35										
	Local Number Portability (1 per port) ROFFICE TRANSPORT			ULFFD	LINFUX	0.35			├				ł		}	
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility Termination			UEPFB	U1TV2	21.13	40.54	27.41	16.74	6.90						
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile or Fraction Mile			UEPFB	1L5XX	0.008838	40.34	21.41	10.74	0.90						
FFAT	URES				ILONA	0.000000	i									
	All Features Offered			UEPFB	UEPVF	0.00	0.00	0.00				15.66	1		1	
NONR	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED														İ	
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
	Combination - Conversion - Switch-as-is			UEPFB	USAC2		8.48	1.87				15.66				
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
	Combination - Conversion - Switch with change E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)			UEPFB	USACC		8.48	1.87				15.66				
0.1													1			1

UNBUNDLE	D NETWORK ELEMENTS - Alabama	-	r										Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			28.38										L
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			36.85										L
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			50.14										L
UNE LO	pop Rates		1	UEPFP	UECF2	14.38										<u> </u>
	2-Wire Voice Grade Loop (SL2) - Zone 1 2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFP	UECF2	22.85										ł
	2-Wire Voice Grade Loop (SL2) - Zone 2 2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFP	UECF2	36.14										-
2-Wire	Voice Grade Line Port Rates (BUS - PBX)		Ŭ	OLITI	02012	00.14										
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPFP	UEPPC	14.00	119.27	69.85	61.18	8.34		15.66				
	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPFP	UEPPO	14.00	119.27	69.85	61.18	8.34		15.66				
	Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPFP	UEPP1	14.00	119.27	69.85	61.18	8.34		15.66				
	2-Wire Voice Unbundled 2-Way Combination PBX Alabama															1
	Calling Port			UEPFP	UEPA2	14.00	119.27	69.85	61.18	8.34		15.66				ļ
	2-Wire Voice Unbundled PBX LD Terminal Ports	L		UEPFP	UEPLD	14.00	119.27	69.85	61.18	8.34		15.66				<u> </u>
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPFP	UEPXA	14.00	119.27	69.85		8.34		15.66				
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports 2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPFP UEPFP	UEPXB UEPXC	14.00 14.00	119.27 119.27	69.85 69.85	61.18 61.18	8.34 8.34		15.66 15.66				ł
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPFP	UEPXD	14.00	119.27	69.85	61.18	8.34		15.66				<u> </u>
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD			OLITI		14.00	113.27	03.05	01.10	0.54		13.00				
	Capable Port			UEPFP	UEPXE	14.00	119.27	69.85	61.18	8.34		15.66				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy															
	Administrative Calling Port			UEPFP	UEPXL	14.00	119.27	69.85	61.18	8.34		15.66				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy															
	Room Calling Port			UEPFP	UEPXM	14.00	119.27	69.85	61.18	8.34		15.66				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital															
	Discount Room Calling Port			UEPFP UEPFP	UEPXO UEPXS	14.00 14.00	119.27 119.27	69.85 69.85	61.18 61.18	8.34 8.34		15.66 15.66				ł
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port NUMBER PORTABILITY			UEPFP	UEPAS	14.00	119.27	69.85	61.18	8.34		10.00				<u> </u>
LOCAL	Local Number Portability (1 per port)			UEPFP	LNPCP	3.15	0.00	0.00				15.66				ł
INTER	DFFICE TRANSPORT			OLITI		0.10	0.00	0.00				10.00				
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility															
	Termination			UEPFP	U1TV2	21.13	40.54	27.41	16.74	6.90						
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile															
	or Fraction Mile			UEPFP	1L5XX	0.008838										
FEATU																
	All Features Offered			UEPFP	UEPVF	0.00	0.00	0.00				15.66				l
NONRE	CURRING CHARGES (NRCs) - CURRENTLY COMBINED															<u> </u>
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port			UEPFP	USAC2		0.40	4.07				45.00				1
<u>├──</u>	Combination - Conversion - Switch-as-is 2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port			UEPFP	USAC2		8.48	1.87				15.66				<u> </u>
	Combination - Conversion - Switch with change			UEPFP	USACC		8.48	1.87				15.66				1
UNE La	contrained - conversion - owner with change				00,000		0.40	1.07				10.00				├ ───
	CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES	5							1		1		İ	İ	İ	L
	Based Rates are applied where BellSouth is required by FCC		State 0	Commission rule t	o provide Unb	undled Local S	witching or Sw	vitch Ports.	1		1	İ	l	l	l	<u> </u>
2. Feat	ures shall apply to the Unbundled Port/Loop Combination -	ost Bas	ed Rat	e section in the sa	ame manner as	they are applie	ed to the Stand	-Alone Unbun								
3. End	Office and Tandem Switching Usage and Common Transport	Usage	rates in	the Port section	of this rate exh	ibit shall apply	to all combination	ations of loop/	/port network el	lements excep	t for UNE C	oin Port/Lo	op Combinat	ions.		
	first and additional Port nonrecurring charges apply to Not Cu	urrently	Comb	ined Combos. Fo	or Currently Co	mbined Combo	os, the nonrecu	irring charges	shall be those	identified in t	he Nonrecu	rring - Curr	ently Combine	ed sections.	Additional NR	Cs may
	also and are categorized accordingly.															
	ket Rates for Unbundled Centrex Port/Loop Combination will I		otiated	on an Individual	Case Basis, un	til further notic	e.									
	CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only))									ļ	ļ				
	VG Loop/2-Wire Voice Grade Port (Centrex) Combo															ļ
UNE Po	ort/Loop Combination Rates (Non-Design)	ļ														I
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -					10.70										1
├ ── ├ ──	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		1	UEP91		12.70										<u> </u>
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		2	UEP91		21.19										1
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		-	02131		21.19					1					<u> </u>
	Non-Design		3	UEP91	1	34.80										1
L			, v			000							l	l	l	·

INBUNDLE	ED NETWORK ELEMENTS - Alabama			r								1	Attachment:			bit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Increment Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrec		Nonrecurring					Rates(\$)	•	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
UNE F	Port/Loop Combination Rates (Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -		1	115004		15.50										
	Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		1	UEP91		15.53										
	Design		2	UEP91		24.00										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		2	OLF91		24.00										
	Design		3	UEP91		37.29										
UNE L	Loop Rate		-													
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP91	UECS1	11.55										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP91	UECS1	20.04										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP91	UECS1	33.65										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP91	UECS2	14.38										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP91	UECS2	22.85										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP91	UECS2	36.14										
UNE F											-					
All Sta	ates (Except North Carolina and Sout Carolina)			UEP91	UEPYA	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex) Basic Local Area 2-Wire Voice Grade Port (Centrex 800 termination)Basic Local			UEF91	UEPTA	1.15	40.19	19.03	24.91	0.03		15.00				
	Area			UEP91	UEPYB	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local			OLI SI	OLITE	1.15	40.13	13.05	24.31	0.05		13.00				
	Area			UEP91	UEPYH	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire			OEI OI	OEI III	1.10	40.10	10.00	24.01	0.00	1	10.00				
	Center)2 Basic Local Area			UEP91	UEPYM	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service				-					-						
	Term - Basic Local Area			UEP91	UEPYZ	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent															
	- Basic Local Area			UEP91	UEPY9	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port Terminated on 800 Service Term -															
	Basic Local Area			UEP91	UEPY2	1.15	40.19	19.83	24.91	6.63		15.66				
AL, K	Y, LA, MS, & TN Only			115504			10.10	10.00				15.00				
	2-Wire Voice Grade Port (Centrex) 2-Wire Voice Grade Port (Centrex 800 termination)			UEP91 UEP91	UEPQA UEPQB	1.15 1.15	40.19 40.19	19.83 19.83	24.91 24.91	6.63 6.63		15.66 15.66				
	2-Wire Voice Grade Port (Centrex 800 termination) 2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP91 UEP91	UEPQB	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire			OLF91	ULFQIT	1.15	40.19	19.03	24.91	0.05		15.00		-		
	Center)2			UEP91	UEPQM	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			OEI OI	OEI QIVI	1.10	00.00	01.27	40.00	0.11	1	10.00				
	Term			UEP91	UEPQZ	1.15	90.38	57.27	48.66	8.77		15.66				
											1				İ	İ
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP91	UEPQ9	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP91	UEPQ2	1.15	40.19	19.83	24.91	6.63		15.66				
Local	Switching															
	Centrex Intercom Funtionality, per port			UEP91	URECS	0.5488										L
Local	Number Portability				LNDCC	0.05					1					
F /	Local Number Portability (1 per port)			UEP91	LNPCC	0.35							ļ			
Featu	All Standard Features Offered, per port			UEP91	UEPVF	1.98					-		1			
	All Standard Features Offered, per port All Select Features Offered, per port			UEP91 UEP91	UEPVF	0.00	405.52			-	<u> </u>				{	
	All Centrex Control Features Offered, per port			UEP91	UEPVC	1.98	+00.JZ									<u> </u>
NARS					02.00	1.00					1					
	Unbundled Network Access Register - Combination			UEP91	UARCX	0.00	0.00	0.00							İ	
<u>l</u>	Unbundled Network Access Register - Indial			UEP91	UAR1X	0.00	0.00	0.00								
	Unbundled Network Access Register - Outdial			UEP91	UAROX	0.00	0.00	0.00								
	Ilaneous Terminations															
2-Wire	e Trunk Side															
	Trunk Side Terminations, each		L	UEP91	CENA6	8.05	119.31	18.74	59.90	3.76	ļ	15.66				
Intero	ffice Channel Mileage - 2-Wire				14050	01.10	10 5 1	07 <i>1</i> ·	10.71	0.00		45.00	ļ			<u> </u>
	Interoffice Channel Facilities Termination - Voice Grade Interoffice Channel mileage, per mile or fraction of mile			UEP91 UEP91	M1GBC M1GBM	21.13 0.008838	40.54	27.41	16.74	6.90	ļ	15.66				<u> </u>
1		e		OFAI	IVIIGBIVI	0.008838					L					L

UNBUNDL	ED NETWORK ELEMENTS - Alabama												Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec			g Disconnect		r		Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
D4 Ci	hannel Bank Feature Activations				1PQWS	0.50										───
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP91	TPQW5	0.56										<u> </u>
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP91	1PQW6	0.56										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop															
	Slot			UEP91	1PQW7	0.56										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -															
	Different Wire Center			UEP91	1PQWP	0.56										Ļ
					100100/	0.50										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop		<u> </u>	UEP91	1PQWV	0.56										<u> </u>
	Slot			UEP91	1PQWQ	0.56										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP91	1PQWA	0.56			1	1	-					
Non-	Recurring Charges (NRC) Associated with UNE-P Centrex	t	1			0.00					1					<u> </u>
	Conversion - Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP91	USAC2		0.10	0.10				15.66				
	Conversion of Existing Centrex Common Block			UEP91	USACN		37.75	16.58				15.66				
	New Centrex Standard Common Block			UEP91	M1ACS	0.00	667.21					15.66				
	New Centrex Customized Common Block			UEP91	M1ACC	0.00	667.21					15.66				
	Secondary Block, per Block			UEP91	M2CC1	0.00	78.02					15.66				
	NAR Establishment Charge, Per Occasion			UEP91	URECA	0.00	72.73					15.66				
	P CENTREX - 5ESS (Valid in All States)															
	e VG Loop/2-Wire Voice Grade Port (Centrex) Combo															
UNE	Port/Loop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	Non-Design		1	UEP95		12.70										ļ
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Non-Design		2	UEP95		21.19										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		_													
	Non-Design		3	UEP95		34.80										<u> </u>
UNE	Port/Loop Combination Rates (Design)															I
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -		1	UEP95		15.53										
	Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		-	UEF95		15.55										
	Design		2	UEP95		24.00										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		2	01 33		24.00										
	Design		3	UEP95		37.29										
UNE	Loop Rate															
	2-Wire Voice Grade Loop (SL 1) - Zone 1	1	1	UEP95	UECS1	11.55			1	1	1			ĺ	İ	<u> </u>
	2-Wire Voice Grade Loop (SL 1) - Zone 2	I	2	UEP95	UECS1	20.04			1	1						
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP95	UECS1	33.65										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP95	UECS2	14.38										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP95	UECS2	22.85										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP95	UECS2	36.14										
	Port Rate								ļ	ļ						
All St			l													I
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP95	UEPYA	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex 800 termination) 2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local			UEP95	UEPYB	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area		1	UEP95	UEPYH	1.15	40.19	19.83	24.91	6.63		15.66				1
	2-Wire Voice Grade Port (Centrex from diff Serving Wire			0LF 90		1.10	40.19	19.63	24.91	0.03		10.00				<u> </u>
	Center)2 Basic Local Area			UEP95	UEPYM	1.15	90.38	57.27	48.66	8.77		15.66				
<u> </u>	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			52.00	561 11	1.15	00.00	01.21		5.11	1	10.00				<u> </u>
	Term - Basic Local Area		1	UEP95	UEPYZ	1.15	90.38	57.27	48.66	8.77		15.66				1
1	2-Wire Voice Grade Port terminated in on Megalink or equivalent									1	1				İ	<u> </u>
	- Basic Local Area		1	UEP95	UEPY9	1.15	40.19	19.83	24.91	6.63		15.66				1
1	2-Wire Voice Grade Port Terminated on 800 Service Term -															
	Basic Local Area		1	UEP95	UEPY2	1.15	40.19	19.83	24.91	6.63		15.66				1
	(Y, LA, MS, SC, & TN Only	l	1						1	1	1					

RONDLE	ED NETWORK ELEMENTS - Alabama		r			1							Attachment:			bit: B
FEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Increment Charge Manual S Order vs Electroni Disc Add
						Dee	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)	•	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port (Centrex)			UEP95	UEPQA	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPQB	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP95	UEPQH	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire															
	Center)2			UEP95	UEPQM	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			UEP95	UEPQZ	1 15	90.38	E7 07	48.66	8.77		15.66				
	Term			UEP95	UEPQZ	1.15	90.38	57.27	48.00	8.77		15.66				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP95	UEPQ9	1.15	40.19	19.83	24.91	6.63		15.66				
_	2-Wire Voice Grade Fort terminated in of Weganink of equivalent 2-Wire Voice Grade Port Terminated on 800 Service Term			UEP95	UEPQ2	1.15	40.19	19.83	24.91	6.63		15.66				
Local	Switching			01 33		1.15	40.13	13.00	24.31	0.05		13.00				
	Centrex Intercom Funtionality, per port	1		UEP95	URECS	0.5488									İ	
Local	Number Portability								i l						İ	
	Local Number Portability (1 per port)			UEP95	LNPCC	0.35										
Featu					1											
	All Standard Features Offered, per port			UEP95	UEPVF	1.98										
	All Select Features Offered, per port			UEP95	UEPVS	0.00	405.52									
	All Centrex Control Features Offered, per port			UEP95	UEPVC	1.98										
NARS																
	Unbundled Network Access Register - Combination			UEP95	UARCX	0.00	0.00	0.00								
	Unbundled Network Access Register - Indial			UEP95	UAR1X	0.00	0.00	0.00								
	Unbundled Network Access Register - Outdial			UEP95	UAROX	0.00	0.00	0.00								
	Ilaneous Terminations															
2-Wire	e Trunk Side				05115.0	0.05			50.00			15.00				
4 14/5	Trunk Side Terminations, each e Digital (1.544 Megabits)			UEP95	CEND6	8.05	119.31	18.74	59.90	3.76		15.66				
4-99116	DS1 Circuit Terminations, each			UEP95	M1HD1	60.09	202.02	95.69	72.59	2.46		15.66				
	DS0 Channels Activated, each			UEP95	M1HD0	0.00	14.46	95.69	72.59	2.40	-	15.66				
Intero	office Channel Mileage - 2-Wire			021 35	WITTEO	0.00	14.40					13.00				
intero	Interoffice Channel Facilities Termination			UEP95	MIGBC	21.13	40.54	27.41	16.74	6.90		15.66				
	Interoffice Channel mileage, per mile or fraction of mile			UEP95	MIGBM	0.008838	10.01	2	10.71	0.00		10.00				
Featu	re Activations (DS0) Centrex Loops on Channelized DS1 Service	e:														
	annel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP95	1PQWS	0.56										
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP95	1PQW6	0.56										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop															
	Slot			UEP95	1PQW7	0.56										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -	1														
	Different Wire Center			UEP95	1PQWP	0.56										
				LIEDOS	100100											
	Feature Activation on D-4 Channel Bank Private Line Loop Slot	I		UEP95	1PQWV	0.56						L				
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop				10014/0	0.50										
_	Slot Feature Activation on D-4 Channel Bank WATS Loop Slot	I		UEP95 UEP95	1PQWQ 1PQWA	0.56										
Non 5	Recurring Charges (NRC) Associated with UNE-P Centrex			05432	IPQWA	0.56										
NON-P	NRC Conversion Currently Combined Switch-As-Is with allowed															
	changes, per port	1		UEP95	USAC2		0.10	0.10				15.66				
	Conversion of Existing Centrex Common Block, each	1		UEP95	USACN		37.75	16.58			1	15.66				
-	New Centrex Standard Common Block	1		UEP95	MIACS	0.00	667.21				1	15.66		1	1	
	New Centrex Customized Common Block			UEP95	MIACC	0.00	667.21					15.66			İ	
1	NAR Establishment Charge, Per Occasion	1	1	UEP95	URECA	0.00	72.73		i i		1	15.66		ĺ	İ	
	P CENTREX - DMS100 (Valid in All States)	1	1		-		-		i i		1			ĺ	İ	
	e VG Loop/2-Wire Voice Grade Port (Centrex) Combo				1											
UNE F	Port/Loop Combination Rates (Non-Design)	Γ														
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo															
	Non-Design	I	1	UEP9D		12.70										
1	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	1	1	1	1						1	1			1	1

UNBUNDLE	D NETWORK ELEMENTS - Alabama	1	1	1		1							Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
							Nonred	curring	Nonrecurring	a Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Non-Design		3	UEP9D	_	34.80										
UNE	Port/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	Design		1	UEP9D		15.53										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		2	UEP9D		24.00										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		~	021 00		24.00										
	Design		3	UEP9D		37.29										
UNE I	oop Rate															
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9D	UECS1	11.55										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9D	UECS1	20.04										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP9D	UECS1	33.65										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9D	UECS2	14.38										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9D	UECS2	22.85										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9D	UECS2	36.14										
	Port Rate															
ALLS	TATES					4.45	10.10	40.00	04.04	0.00	-	45.00				
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9D	UEPYA	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area			UEP9D	UEPYB	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3Basic Local Area			UEP9D	UEPYC	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3Basic Local				UEPYD	1.15	40.19	40.00	04.04	0.00		45.00				
	Area 2-Wire Voice Grade Port (Centrex / EBS-M5209))3 Basic Local			UEP9D	UEPYD	1.15	40.19	19.83	24.91	6.63		15.66				
	Area			UEP9D	UEPYE	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex / EBS-M5112))3 Basic Local Area			UEP9D	UEPYF	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex / EBS-M5312))3Basic Local			02.00	02.11		10110	10.00	21101	0.00		10.00				
	Area			UEP9D	UEPYG	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex / EBS-M5008))3 Basic Local															
	Area 2-Wire Voice Grade Port (Centrex / EBS-M5208))3 Basic Local			UEP9D	UEPYT	1.15	40.19	19.83	24.91	6.63		15.66				
	Area			UEP9D	UEPYU	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex / EBS-M5216))3 Basic Local						10.10									
	Area			UEP9D	UEPYV	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex / EBS-M5316))3 Basic Local Area			UEP9D	UEPY3	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex with Caller ID) Basic Local Area			UEP9D	UEPYH	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp															
	Indication))3 Basic Local Area			UEP9D	UEPYW	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication))3 Basic Local Area			UEP9D	UEPYJ	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center) 2 Basic Local Area			UEP9D	UEPYM	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3								10.6-			15.0-				
┝──┼──	Basic Local Area		<u> </u>	UEP9D	UEPYO	1.15	90.38	57.27	48.66	8.77	ļ	15.66				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3 Basic Local Area			UEP9D	UEPYP	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3 Basic Local Area			UEP9D	UEPYQ	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3		1							_						
	Basic Local Area		Ļ	UEP9D	UEPYR	1.15	90.38	57.27	48.66	8.77	ļ	15.66				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3 Basic Local Area			UEP9D	UEPYS	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3 Basic Local Area			UEP9D	UEPY4	1.15	90.38	57.27	48.66	8.77		15.66				

JNBUNDLE	D NETWORK ELEMENTS - Alabama		-									-	Attachment:			bit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge - Manual Sy Order vs. Electronic Disc Add
						Rec	Nonrec	curring	Nonrecurring	J Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3															i
	Basic Local Area			UEP9D	UEPY5	1.15	90.38	57.27	48.66	8.77		15.66				I
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3 Basic Local Area			UEP9D	UEPY6	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3															i
	Basic Local Area			UEP9D	UEPY7	1.15	90.38	57.27	48.66	8.77	-	15.66				<u> </u>
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP9D	UEPYZ	1.15	90.38	57.27	48.66	8.77		15.66				1
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			DEP9D	UEPTZ	1.15	90.36	57.27	40.00	0.77		15.00				
	Basic Local Area			UEP9D	UEPY9	1.15	40.19	19.83	24.91	6.63		15.66				1
	2-Wire Voice Grade Port Terminated on 800 Service Term Basic															
	Local Area			UEP9D	UEPY2	1.15	40.19	19.83	24.91	6.63		15.66				
AL, K	r, LA, MS, SC, & TN Only															
	2-Wire Voice Grade Port (Centrex)			UEP9D	UEPQA	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP9D	UEPQB	1.15	40.19	19.83	24.91	6.63		15.66				ļ
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3			UEP9D	UEPQC	1.15	40.19	19.83	24.91	6.63		15.66				ļ
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3			UEP9D	UEPQD	1.15	40.19	19.83	24.91	6.63		15.66				ļ
	2-Wire Voice Grade Port (Centrex / EBS-M5209)3			UEP9D UEP9D	UEPQE UEPQF	1.15	40.19 40.19	19.83 19.83	24.91	6.63		15.66 15.66				<u> </u>
	2-Wire Voice Grade Port (Centrex / EBS-M5112)3 2-Wire Voice Grade Port (Centrex / EBS-M5312)3			UEP9D UEP9D	UEPQF	1.15 1.15	40.19	19.83	24.91 24.91	6.63 6.63		15.66				├ ───
	2-Wire Voice Grade Port (Centrex / EBS-M5312)3 2-Wire Voice Grade Port (Centrex / EBS-M5008)3			UEP9D UEP9D	UEPQG	1.15	40.19	19.83	24.91	6.63		15.66				
_	2-Wire Voice Grade Port (Centrex / EBS-M5006)3 2-Wire Voice Grade Port (Centrex / EBS-M5208)3			UEP9D	UEPQU	1.15	40.19	19.83	24.91	6.63		15.66		-		<u> </u>
	2-Wire Voice Grade Port (Centrex / EBS-M5206)3			UEP9D	UEPQV	1.15	40.19	19.83	24.91	6.63		15.66		-		<u> </u>
	2-Wire Voice Grade Port (Centrex / EBS-M5216)3			UEP9D	UEPQ3	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex with Caller ID)			UEP9D	UEPQH	1.15	40.19	19.83	24.91	6.63		15.66				<u> </u>
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp			021 00	OEI QII	1.10	40.10	10.00	24.01	0.00	1	10.00				-
	Indication)3			UEP9D	UEPQW	1.15	40.19	19.83	24.91	6.63		15.66				1
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication)3			UEP9D	UEPQJ	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)															
	2			UEP9D	UEPQM	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			UEP9D	UEPQO	1.15	90.38	57.27	48.66	8.77		15.66				
																1
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3			UEP9D	UEPQP	1.15	90.38	57.27	48.66	8.77		15.66				L
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3			UEP9D	UEPQQ	1.15	90.38	57.27	48.66	8.77		15.66				
	ONES VERSE OF A DET (OF THE ATTEND ON OF DO NETTONO O					4.45	00.00	57.07	10.00	0.77		45.00				1
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			UEP9D	UEPQR	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3			UEP9D	UEPQS	1.15	90.38	57.27	48.66	8.77		15.66				i
				021 00	OLI QU	1.10	50.00	01.21	40.00	0.11		10.00				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3			UEP9D	UEPQ4	1.15	90.38	57.27	48.66	8.77		15.66				1
																(
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3			UEP9D	UEPQ5	1.15	90.38	57.27	48.66	8.77		15.66				l
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3			UEP9D	UEPQ6	1.15	90.38	57.27	48.66	8.77		15.66				ļ
					I T	Т										1
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			UEP9D	UEPQ7	1.15	90.38	57.27	48.66	8.77		15.66				L
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service				115007		00.00	F7 07	10.00	0		45.00				i
_	Term			UEP9D	UEPQZ	1.15	90.38	57.27	48.66	8.77		15.66		L		
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9D	UEPQ9	1.15	40.19	19.83	24.91	6.63		15.66				i
	2-Wire Voice Grade Port terminated in on Megalink or equivalent 2-Wire Voice Grade Port Terminated on 800 Service Term			UEP9D UEP9D	UEPQ9	1.15	40.19	19.83	24.91	6.63	-	15.66				
Local	Switching			02.00	561 62	1.15	40.13	10.00	2-1.31	0.00	t	10.00				<u> </u>
	Centrex Intercom Funtionality, per port			UEP9D	URECS	0.5488					t					<u> </u>
Local	Number Portability														İ	[
	Local Number Portability (1 per port)			UEP9D	LNPCC	0.35										(
Featur				-							İ 👘					(
	All Standard Features Offered, per port			UEP9D	UEPVF	1.98										
	All Select Features Offered, per port			UEP9D	UEPVS	0.00	405.52									
	All Centrex Control Features Offered, per port	_		UEP9D	UEPVC	1.98										

NBUNDLE	ED NETWORK ELEMENTS - Alabama		-	-									Attachment:			bit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
NARS																
	Unbundled Network Access Register - Combination			UEP9D	UARCX	0.00	0.00	0.00								
	Unbundled Network Access Register - Inward Unbundled Network Access Register - Outdial			UEP9D UEP9D	UAR1X UAROX	0.00	0.00	0.00								
Misso	Ilaneous Terminations			UEF9D	UARUA	0.00	0.00	0.00								
	e Trunk Side															
2-1110	Trunk Side Terminations, each			UEP9D	CEND6	8.05	119.31	18.74	59.90	3.76		15.66				
4-Wire	e Digital (1.544 Megabits)			02.00	02.120	0.00	110101	10.71	00.00	0.10		10.00				
	DS1 Circuit Terminations, each			UEP9D	M1HD1	60.09	202.02	95.69	72.59	2.46		15.66				
	DS0 Channels Activiated per Channel			UEP9D	M1HDO	0.00	14.46					15.66				
Intero	ffice Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination			UEP9D	MIGBC	21.13	40.54	27.41	16.74	6.90		15.66				
	Interoffice Channel mileage, per mile or fraction of mile			UEP9D	MIGBM	0.008838										
	re Activations (DS0) Centrex Loops on Channelized DS1 Servic	e														
D4 Ch	annel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9D	1PQWS	0.56										
	Facture Activities on D.4 Channel Dark EV line Cide Lean Clat			UEP9D	1PQW6	0.50										
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot Feature Activation on D-4 Channel Bank FX Trunk Side Loop			UEP9D	1PQW6	0.56										
	Slot			UEP9D	1PQW7	0.56										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -			UEP9D	TPQW7	0.56										
	Different Wire Center			UEP9D	1PQWP	0.56										
	Different wife Genter				II QWI	0.50										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9D	1PQWV	0.56										
	Feature Activation on D-4 Channel Bank Tile Line/Trunk Loop			02.00		0.00										
	Slot			UEP9D	1PQWQ	0.56										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9D	1PQWA	0.56										
Non-R	Recurring Charges (NRC) Associated with UNE-P Centrex															
	NRC Conversion Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP9D	USAC2		0.10	0.10				15.66				
	Conversion of existing Centrex Common Block, each			UEP9D	USACN		37.75	16.58				15.66				
	New Centrex Standard Common Block			UEP9D	M1ACS	0.00	667.21					15.66				
	New Centrex Customized Common Block			UEP9D	M1ACC	0.00	667.21					15.66				
	NAR Establishment Charge, Per Occasion			UEP9D	URECA	0.00	72.73					15.66				
	CENTREX - EWSD (Valid in AL, FL, KY, LA, MS & TN)															
	e VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design)															
UNE F	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	Non-Design		1	UEP9E		12.70										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			51. 0L		12.70										
	Non-Design		2	UEP9E		21.19										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	1			1	25			1							
	Non-Design		3	UEP9E		34.80										
UNE F	Port/Loop Combination Rates (Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	Design		1	UEP9E		15.53										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			<u> </u>												
	Design		2	UEP9E		24.00										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design		3	UEP9E	+	37.29				-						
UNEL	oop Rate			UEP9E	UECS1	11.55										
	2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2		1	UEP9E UEP9E	UECS1 UECS1	11.55 20.04										
	2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3		2	UEP9E UEP9E	UECS1 UECS1	20.04 33.65								1	1	
	2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1		3	UEP9E UEP9E	UECS1 UECS2	33.65 14.38										
	2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9E	UECS2 UECS2	22.85										
	2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9E	UECS2	36.14										
UNE F	Port Rate	1			52002	00.14										<u> </u>
	L, KY, LA, MS, & TN only				1											

UNDLE	D NETWORK ELEMENTS - Alabama	1	-	1		1					-		Attachment:			ibit: B
GORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Charg
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMA
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9E	UEPYA	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local			UEP9E	UEPYB	1.15	40.19	40.02	24.04	C C2		45.00				
	Area 2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local			UEP9E	UEPTB	1.15	40.19	19.83	24.91	6.63		15.66				
	Area			UEP9E	UEPYH	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire			OEI OE	OEI III	1.10	40.10	10.00	24.01	0.00		10.00			1	
	Center)2 Basic Local Area			UEP9E	UEPYM	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term - Basic Local Area			UEP9E	UEPYZ	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent															
	- Basic Local Area			UEP9E	UEPY9	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port Terminated on 800 Service Term -						10.10					15.00				
	Basic Local Area	_		UEP9E	UEPY2	1.15	40.19	19.83	24.91	6.63		15.66			-	
AL, KT	, LA, MS, & TN Only 2-Wire Voice Grade Port (Centrex)			UEP9E	UEPQA	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP9E	UEPQA	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex odd termination)			UEP9E	UEPQH	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire			02.02	02. 0		10110	10.00	2	0.00		10.00			1	
	Center)2			UEP9E	UEPQM	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term			UEP9E	UEPQZ	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent	:		UEP9E	UEPQ9	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP9E	UEPQ2	1.15	40.19	19.83	24.91	6.63		15.66				
Local S	Switching				URECS	0.5488										
	Centrex Intercom Funtionality, per port Iumber Portability			UEP9E	URECS	0.5488										
LUCAIN	Local Number Portability (1 per port)			UEP9E	LNPCC	0.35										
Feature				OLI OL	LIVI 00	0.00										
	All Standard Features Offered, per port			UEP9E	UEPVF	1.98										
	All Select Features Offered, per port			UEP9E	UEPVS	0.00	405.52									
	All Centrex Control Features Offered, per port			UEP9E	UEPVC	1.98										
NARS																
	Unbundled Network Access Register - Combination			UEP9E	UARCX	0.00	0.00	0.00								
	Unbundled Network Access Register - Indial			UEP9E	UAR1X	0.00	0.00	0.00								
	Unbundled Network Access Register - Outdial	_		UEP9E	UAROX	0.00	0.00	0.00				-			-	
	aneous Terminations Trunk Side															
2-44116	Trunk Side Terminations, each	+		UEP9E	CEND6	8.05	119.31	18.74	59.90	3.76	+	15.66	-		ł	l
4-Wire	Digital (1.544 Megabits)	1			02.100	0.00	110.01	10.74	00.00	0.70	1	10.00			<u> </u>	1
	DS1 Circuit Terminations, each			UEP9E	M1HD1	60.09	202.02	95.69	72.59	2.46		15.66			1	1
	DS0 Channel Activated Per Channel	1	1	UEP9E	M1HDO	0.00	14.46				1	15.66		ĺ	1	1
Interoff	ice Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination			UEP9E	MIGBC	21.13	40.54	27.41	16.74	6.90		15.66				
	Interoffice Channel mileage, per mile or fraction of mile			UEP9E	MIGBM	0.008838					ļ	ļ		ļ	ļ	I
	Activations (DS0) Centrex Loops on Channelized DS1 Service	ce		1											ł	<u> </u>
D4 Cha	nnel Bank Feature Activations Feature Activation on D-4 Channel Bank Centrex Loop Slot	+		UEP9E	1PQWS	0.56									<u> </u>	
-	r earare Activation on D-4 Ghannel Dank Centrex Loop SIO	+			IFQVIO	0.00				-	+	<u> </u>	-		ł	l
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9E	1PQW6	0.56									1	
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop	1				0.00			İ		1				1	1
	Slot			UEP9E	1PQW7	0.56									1	
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -	1	1								1			ĺ	1	1
	Different Wire Center			UEP9E	1PQWP	0.56										
				ULE DOE	1001407	0.50					1	1	1		1	1
	Feature Activation on D-4 Channel Bank Private Line Loop Slot		-	UEP9E	1PQWV	0.56										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot			UEP9E	1PQWV 1PQWQ	0.56										

NBUNDLE	D NETWORK ELEMENTS - Alabama		1										Attachment:			ibit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted	Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Non-R	ecurring Charges (NRC) Associated with UNE-P Centrex															
	NRC Conversion Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP9E	USAC2		0.10	0.10				15.66				
	Conversion of Existing Centrex Common Block, each			UEP9E	USACN		37.75	16.58				15.66				
	New Centrex Standard Common Block			UEP9E	M1ACS	0.00	667.21					15.66				
	New Centrex Customized Common Block			UEP9E	M1ACC	0.00	667.21					15.66				
	NAR Establishment Charge, Per Occasion			UEP9E	URECA	0.00	72.73					15.66				
	CENTREX - DCO - Valid in AL, KY, LA, MS, & TN)															
	VG Loop/2-Wire Voice Grade Port (Centrex) Combo				_											
UNE F	Port/Loop Combination Rates (Non-Design)				_											
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -					10 70										
	Non-Design	l	1	UEP93	-	12.70					L					l
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Non-Design	l	2	UEP93	-	21.19										l
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Non-Design		3	UEP93		34.80										
UNE F	Port/Loop Combination Rates (Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	Design		1	UEP93		15.53										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design		2	UEP93		24.00										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design		3	UEP93		37.29										
UNE L	.oop Rate															
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP93	UECS1	11.55										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP93	UECS1	20.04										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP93	UECS1	33.65										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP93	UECS2	14.38										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP93	UECS2	22.85										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP93	UECS2	36.14										
	Port Rate															
AL, K	Y, LA, MS, & TN only															
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP93	UEPYA	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local															
	Area			UEP93	UEPYB	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local															
	Area			UEP93	UEPYH	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire															
	Center)2 Basic Local Area			UEP93	UEPYM	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term - Basic Local Area			UEP93	UEPYZ	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent															
	- Basic Local Area			UEP93	UEPY9	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port Terminated on 800 Service Term -															
	Basic Local Area			UEP93	UEPY2	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex)			UEP93	UEPQA	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP93	UEPQB	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP93	UEPQH	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire															
	Center)2			UEP93	UEPQM	1.15	90.38	57.27	48.66	8.77		15.66				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service						Т									
	Term			UEP93	UEPQZ	1.15	90.38	57.27	48.66	8.77		15.66				
																1
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP93	UEPQ9	1.15	40.19	19.83	24.91	6.63		15.66				
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP93	UEPQ2	1.15	40.19	19.83	24.91	6.63		15.66				
Local	Switching															
	Centrex Intercom Funtionality, per port			UEP93	URECS	0.5488										
Local	Number Portability										1					
	Local Number Portability (1 per port)		1	UEP93	LNPCC	0.35					1		1			1

INBUNDLED NETWORK ELEMENTS - Alabama												Attachment:	2	Exhi	bit: B
										Svc Order	Svc Order	Incremental	Incremental	Incremental	Incrementa
											Submitted	Charge -	Charge -	Charge -	Charge -
										Elec				Manual Svc	
ATEGORY RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)								
	m	20116	000	0000						per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
												Electronic-	Electronic-	Electronic-	Electronic-
												1st	Add'l	Disc 1st	Disc Add'l
		-		-		Neuros		Neuroeumine	Discoursed			000			
					Rec	Nonrec First	urring Add'l	Nonrecurring First	Add'l	SOMEC	SOMAN	SOMAN	Rates(\$) SOMAN	SOMAN	SOMAN
Features						THOU	Add I	That	Auui	SOMEC	JONIAN	JONIAN	JONIAN	JONIAN	JONIAN
All Standard Features Offered, per port		-	UEP93	UEPVF	1.98										
All Centrex Control Features Offered, per port			UEP93	UEPVF											
			UEP93	UEPVC	1.98										
NARS															
Unbundled Network Access Register - Combination			UEP93	UARCX	0.00	0.00	0.00								
Unbundled Network Access Register - Indial		_	UEP93	UAR1X	0.00	0.00	0.00								
Unbundled Network Access Register - Outdial			UEP93	UAROX	0.00	0.00	0.00								
Miscellaneous Terminations															
2-Wire Trunk Side															
Trunk Side Terminations, each			UEP93	CEND6	8.05	119.31	18.74	59.90	3.76		15.66				
4-Wire Digital (1.544 Megabits)															
DS1 Circuit Terminations, each			UEP93	M1HD1	60.09	202.02	95.69	72.59	2.46		15.66				
DS0 Channels Activated, Per Channel			UEP93	M1HDO	0.00	14.46					15.66				
Interoffice Channel Mileage - 2-Wire															
Interoffice Channel Facilities Termination		-	UEP93	MIGBC	21.13	40.54	27.41	16.74	6.90		15.66				
Interoffice Channel mileage, per mile or fraction of mile			UEP93	MIGBM	0.008838	40.54	27.41	10.74	0.30		13.00				
Feature Activations (DS0) Centrex Loops on Channelized DS1 Se	ndee	-	ULF 93	IVIIGBIVI	0.000030										
D4 Channel Bank Feature Activations	IVICE	-		-											
			LIEDOO	45014/0	0.50										
Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP93	1PQWS	0.56										
Feature Activation on D-4 Channel Bank FX Line Side Loop S			UEP93	1PQW6	0.56										
Feature Activation on D-4 Channel Bank FX Trunk Side Loop															
Slot			UEP93	1PQW7	0.56										
Feature Activation on D-4 Channel Bank Centrex Loop Slot -															
Different Wire Center			UEP93	1PQWP	0.56										
Feature Activation on D-4 Channel Bank Private Line Loop SI	ot		UEP93	1PQWV	0.56										
Feature Activation on D-4 Channel Bank Tie Line/Trunk Loop															
Slot			UEP93	1PQWQ	0.56										
Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP93	1PQWA	0.56										
Non-Recurring Charges (NRC) Associated with UNE-P Centrex		-	OEI 00	II Service	0.00										
NRC Conversion Currently Combined Switch-As-Is with allow	od .														
changes, per port	5u	1	UEP93	USAC2		0.10	0.10			1	15.66				
Conversion of Existing Centrex Common Block, each		+	UEP93	USAC2 USACN		37.75	16.58				15.66				
		+			0.00		10.58								
New Centrex Standard Common Block		+	UEP93	M1ACS	0.00	667.21				l	15.66			ł	
New Centrex Customized Common Block		-	UEP93	M1ACC	0.00	667.21					15.66				
NAR Establishment Charge, Per Occasion			UEP93	URECA	0.00	72.73				L	15.66				
Note 1 - Required Port for Centrex Control in 1AESS, 5ESS & EW	SD														
Note 2 - Requres Interoffice Channel Mileage															
Note 3 - Requires Specific Customer Premises Equipment															
Note: Rates displaying an "R" in Interim column are interim and	subject to	rate tru	ue-up as set forth in	General Terr	ns and Conditio	ons.									

UNBUND	DLED	NETWORK ELEMENTS - Florida			1		n					T -	-	Attachment:			ibit: B
CATEGOR	۲Y	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
							Rec	Nonrec	urring	Nonrecurring	g Disconnect			OSS	Rates(\$)	11	
								First	Add'l	First	Add'l		SOMAN			SOMAN	SOMAN
		ne" shown in the sections for stand-alone loops or loops as	•			eographically	y Deaveraged U	NE Zones. To	view Georgrap	hically Deaver	aged UNE Zor	ne Desiganti	ons by C O	, refer to Inter	net Website:		
		ww.interconnection.bellsouth.com/become_a_clec/html/inter SUPPORT SYSTEMS	connec	tion.ht	m	T.	1					1		1			
		1) Electronic Service Order: CLEC should contact its contract	t nego	iator i	it prefers the state	specific elec	tronic service o	rdering charge	s as ordered h	v the State Co	mmissions T	he electron	ic service o	rdering charg	e currentiv co	untained in th	is rate
		is the BellSouth regional electronic service ordering charge.															lo futo
tho	ose el	2) Any element that can be ordered electronically will be bille lements that cannot be ordered electronically at present per ti g charge, SOMAN, will be applied to a CLECs bill when it sub	he BBR	-LO, tł	ne listed SOMEC rate												
	I	Manual Service Order Charge, per LSR, Disconnect Only (FL)				SOMAN				1.83							
		Electronic OSS Charge, per LSR, submitted via BST's OSS															
		interactive interfaces (Regional) DATE ADVANCEMENT CHARGE				SOMEC		3.50								└──── ′	───
		DATE ADVANCEMENT CHARGE The Expedite charge will be maintained commensurate with E	Relisou	th'e Fí	C No 1 Tariff Scoti	on 5 as annli	icable					1				┨────┘	+
		UNE Expedite Charge per Circuit or Line Assignable USOC, per	Senoul	arsru	Service Franni, Secu											├──── /	<u> </u>
		Day			ALL UNE	SDASP		200.00									
UNBUNDL		XCHANGE ACCESS LOOP										1				ł – – ł	
2-V		ANALOG VOICE GRADE LOOP															
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 1		1	UEANL	UEAL2	10.69	49.57	22.83	25.62	6.57		11.90				
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 2		2	UEANL	UEAL2	15.20	49.57	22.83	25.62	6.57		11.90			!	
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 3		3	UEANL	UEAL2	26.97	49.57	22.83	25.62	6.57		11.90			Į/	
		Loop Testing - Basic 1st Half Hour Loop Testing - Basic Additional Half Hour			UEANL UEANL	URET1 URETA		48.65 23.95					11.90 11.90			ļ/	
		CLEC to CLEC Conversion Charge Without Outside Dispatch			UEANL	URETA		23.95					11.90			├─── ┦	
		(UVL-SL1)			UEANL	UREWO		15.78	8.94				11.90				
		Unbundled Voice Loop, Unbundled Non-Design Voice Loop,															
		billing for BST providing make-up			UEANL	UEANM		13.49								ļ!	
		Manual Order Coordination for UVL-SL1s (per loop) Order Coordination for Specified Conversion Time for UVL-SL1			UEANL	UEAMC		9.00								ļ/	
		(per LSR)			UEANL	OCOSL		23.02									
2-V		Unbundled COPPER LOOP			OLANE	OCCOL		20.02									
		2-Wire Unbundled Copper Loop - Non-Designed Zone 1	1	1	UEQ	UEQ2X	7.69	44.98	20.90	19.65	5.09		11.90			l	
		2 Wire Unbundled Copper Loop - Non-Designed - Zone 2	I	2	UEQ	UEQ2X	10.92	44.98	20.90	19.65	5.09		11.90			1	
		2 Wire Unbundled Copper Loop - Non-Designed - Zone 3	Ι	3	UEQ	UEQ2X	19.38	44.98	20.90	19.65	5.09		11.90				
		Order Coordination 2 Wire Unbundled Copper Loop - Non-															
		Designed (per loop)	-		UEQ	USBMC		9.00									
		Unbundled Copper Loop, Non-Designed Billing for BST						12.40					44.00				
		providing make-up Loop Testing - Basic 1st Half Hour			UEQ UEQ	UEQMU URET1	ł	13.49 48.65					11.90 11.90			┨────┘	+
		Loop Testing - Basic Additional Half Hour	1		UEQ	URETA		23.95					11.90			┟────┦	+
		CLEC to CLEC Conversion Charge Without Outside Dispatch				5	1	20.00				1	11.00				
	((UCL-ND)			UEQ	UREWO		14.27	7.43				11.90				
		XCHANGE ACCESS LOOP															
2-V		ANALOG VOICE GRADE LOOP														└─────────────────────────────────────	\square
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zone 1		1	UEPSR UEPSB	UEALS	10.69	49.57	22.83	25.62	6.57		11.90				
	2	2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zone 1		1	UEPSR UEPSB	UEABS	10.69	49.57	22.83	25.62	6.57		11.90				
	2	2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting-														├ ──┤	
		Zone 2 2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting-		2	UEPSR UEPSB	UEALS	15.20	49.57	22.83	25.62	6.57		11.90				
	2	Zone 2 2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-		2	UEPSR UEPSB	UEABS	15.20	49.57	22.83	25.62	6.57		11.90				
	2	Zone 3 2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-		3	UEPSR UEPSB	UEALS	26.97	49.57	22.83	25.62	6.57		11.90			ļ	<u> </u>
	2	Zone 3		3	UEPSR UEPSB	UEABS	26.97	49.57	22.83	25.62	6.57		11.90			ļ'	
LIN		op Rates for Line Splitting		<u> </u>				0.10-	o 10-							↓ ′	
UN				1	UEPRX	UEPLX	12.94	0.102	0.102			1	1	1		· · · ·	1
		2-Wire Voice Grade Loop (SL1) for Line Splitting - Zone 1 2-Wire Voice Grade Loop (SL1) for Line Splitting - Zone 2			UEPRX	UEPLX	17.06	0.102	0.102							I	

	NETWORK ELEMENTS - Florida	1	1	1									Attachment:			bit: B
TEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	XCHANGE ACCESS LOOP															
	ANALOG VOICE GRADE LOOP															
(2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling - Zone 1		1	UEA	UEAL2	12.24	135.75	82.47	63.53	12.01		11.90				
(2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling - Zone 2		2	UEA	UEAL2	17.40	135.75	82.47	63.53	12.01		11.90				
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling - Zone 3		3	UEA	UEAL2	30.87	135.75	82.47	63.53	12.01		11.90				
	Order Coordination for Specified Conversion Time (per LSR)			UEA	OCOSL		23.02									
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse															
	Battery Signaling - Zone 1 2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse		1	UEA	UEAR2	12.24	135.75	82.47	63.53	12.01		11.90				
6	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse		2	UEA	UEAR2	17.40	135.75	82.47	63.53	12.01		11.90				
E	Battery Signaling - Zone 3		3	UEA	UEAR2	30.87	135.75	82.47	63.53	12.01		11.90				
	Order Coordination for Specified Conversion Time (per LSR)			UEA	OCOSL		23.02									
	CLEC to CLEC Conversion Charge without outside dispatch			UEA	UREWO		87.71	36.35				11.90				
	ANALOG VOICE GRADE LOOP		4			10.00	407.00	445 45	07.00	45.50		44.00				-
	4-Wire Analog Voice Grade Loop - Zone 1		1	UEA	UEAL4	18.89	167.86	115.15	67.08	15.56		11.90				
	4-Wire Analog Voice Grade Loop - Zone 2			UEA UEA	UEAL4 UEAL4	26.84 47.62	167.86 167.86	115.15 115.15	67.08 67.08	15.56 15.56		11.90 11.90				
	4-Wire Analog Voice Grade Loop - Zone 3 Order Coordination for Specified Conversion Time (per LSR)		3	UEA	OCOSL	47.62	23.02	115.15	67.08	15.50		11.90				
	CLEC to CLEC Conversion Charge without outside dispatch			UEA	UREWO		87.71	36.35				11.90				1
	ISDN DIGITAL GRADE LOOP			ULA	UKLWO		07.71	30.33				11.90				-
	2-Wire ISDN Digital Grade Loop - Zone 1		1	UDN	U1L2X	19.28	147.69	94.41	62.23	10.71		11.90				
	2-Wire ISDN Digital Grade Loop - Zone 1		2	UDN	U1L2X	27.40	147.69	94.41	62.23	10.71		11.90				
	2-Wire ISDN Digital Grade Loop - Zone 2			UDN	U1L2X	48.62	147.69	94.41	62.23	10.71		11.90				
	Order Coordination For Specified Conversion Time (per LSR)		Ŭ	UDN	OCOSL	40.02	23.02	04.41	02.20	10.71		11.00				
	CLEC to CLEC Conversion Charge without outside dispatch			UDN	UREWO		91.61	44.15				11.90				
	Universal Digital Channel (UDC) COMPATIBLE LOOP			05.1	0.12.110		0.1.01					11100				1
	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone		1	UDC	UDC2X	19.28	147.69	94.41	62.23	10.71		11.90				
	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone															
	2 2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone		2	UDC	UDC2X	27.40	147.69	94.41	62.23	10.71		11.90				
3	3 CLEC to CLEC Conversion Charge without outside dispatch		3	UDC UDC	UDC2X UREWO	48.62	147.69 91.61	94.41 44.15	62.23	10.71		11.90 11.90				
	ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMP				UREWO		91.01	44.15				11.90				
	2 Wire Unbundled ADSL Loop including manual service inquiry															
6	& facility reservation - Zone 1		1	UAL	UAL2X	8.30	149.53	103.85	75.05	15.63		11.90				
ł	2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation - Zone 2		2	UAL	UAL2X	11.80	149.53	103.85	75.05	15.63		11.90				
	2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation - Zone 3		3	UAL	UAL2X	20.94	149.53	103.85	75.05	15.63		11.90				
(Order Coordination for Specified Conversion Time (per LSR)			UAL	OCOSL		23.02									
	2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 1		1	UAL	UAL2W	8.30	124.83	71.12	60.64	9.12		11.90				
2	2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 2		2	UAL	UAL2W	11.80	124.83	71.12	60.64	9.12		11.90				
2	2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 3		3	UAL	UAL2W	20.94	124.83	71.12	60.64	9.12		11.90				
	Order Coordination for Specified Conversion Time (per LSR)		5	UAL	OCOSL	20.94	23.02	11.12	00.04	5.12		11.90				
	CLEC to CLEC Conversion Charge without outside dispatch			UAL	UREWO		86.19	40.39	-			11.90				<u> </u>
	HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPA		LOOP	0, 12	SILLING		00.19	+0.35				11.90				
	2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation - Zone 1		1	UHL	UHL2X	7.22	159.09	112 14	75.05	15.00		11.00				
L L						172	159.09	113.41	75.05	15.63		11.90		1	1	1

UNBUNDLE	ED NETWORK ELEMENTS - Florida												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring			_		Rates(\$)	_	
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2 Wire Unbundled HDSL Loop including manual service inquiry		2			40.04	150.00	440.44	75.05	45.00		44.00				
	& facility reservation - Zone 3 Order Coordination for Specified Conversion Time (per LSR)		3	UHL	UHL2X OCOSL	18.21	159.09 23.02	113.41	75.05	15.63		11.90				
	2 Wire Unbundled HDSL Loop without manual service inquiry				OCOSL		23.02									
	and facility reservation - Zone 1		1	UHL	UHL2W	7.22	134.40	80.69	60.64	9.12		11.90				
	2 Wire Unbundled HDSL Loop without manual service inquiry		· ·	OTIL	OTILE	1.22	104.40	00.00	00.04	0.12	1	11.00				
	and facility reservation - Zone 2		2	UHL	UHL2W	10.26	134.40	80.69	60.64	9.12		11.90				
	2 Wire Unbundled HDSL Loop without manual service inquiry															
	and facility reservation - Zone 3		3	UHL	UHL2W	18.21	134.40	80.69	60.64	9.12		11.90				
	Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		23.02									
	CLEC to CLEC Conversion Charge without outside dispatch			UHL	UREWO		86.12	40.39				11.90				
4-WIR	E HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPA	TIBLE	LOOP													
	4 Wire Unbundled HDSL Loop including manual service inquiry		1	UHL		10.00	100.01	100.00	77.45	10.01		11.00				
	and facility reservation - Zone 1		1	UHL	UHL4X	10.86	193.31	138.98	77.15	12.61		11.90				
	4-Wire Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 2		2	UHL	UHL4X	15.44	193.31	138.98	77.15	12.61		11.90				
	4-Wire Unbundled HDSL Loop including manual service inquiry		2	UHL		15.44	193.31	130.90	77.15	12.01		11.90				
	and facility reservation - Zone 3		3	UHL	UHL4X	27.39	193.31	138.98	77.15	12.61		11.90				
	Order Coordination for Specified Conversion Time (per LSR)		Ŭ	UHL	OCOSL	21.00	23.02	100.00	11.10	12.01		11.00				
	4-Wire Unbundled HDSL Loop without manual service inquiry															
	and facility reservation - Zone 1		1	UHL	UHL4W	10.86	168.62	115.47	62.74	11.22		11.90				
	4-Wire Unbundled HDSL Loop without manual service inquiry															
	and facility reservation - Zone 2		2	UHL	UHL4W	15.44	168.62	115.47	62.74	11.22		11.90				
	4-Wire Unbundled HDSL Loop without manual service inquiry															
	and facility reservation - Zone 3		3	UHL	UHL4W	27.39	168.62	115.47	62.74	11.22		11.90				
	Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		23.02									
	CLEC to CLEC Conversion Charge without outside dispatch			UHL	UREWO		86.12	40.39				11.90				
4-WIR	E DS1 DIGITAL LOOP		1			70.74	040.75	181.48	61.22	13.53	-	11.90				-
	4-Wire DS1 Digital Loop - Zone 1 4-Wire DS1 Digital Loop - Zone 2			USL USL	USLXX USLXX	100.54	313.75 313.75	181.48	61.22	13.53		11.90				
	4-Wire DS1 Digital Loop - Zone 2 4-Wire DS1 Digital Loop - Zone 3		3	USL	USLXX	178.39	313.75	181.48	61.22	13.53		11.90				
	Order Coordination for Specified Conversion Time (per LSR)		5	USL	OCOSL	170.00	23.02	101.40	01.22	15.55		11.50				
	CLEC to CLEC Conversion Charge without outside dispatch			USL	UREWO		101.07	43.04			1	11.90				
4-WIR	E 19.2, 56 OR 64 KBPS DIGITAL GRADE LOOP															
	4 Wire Unbundled Digital 19.2 Kbps		1	UDL	UDL19	22.20	161.56	108.85	67.08	15.56		11.90				
	4 Wire Unbundled Digital 19.2 Kbps		2	UDL	UDL19	31.56	161.56	108.85	67.08	15.56		11.90				
	4 Wire Unbundled Digital 19.2 Kbps		3	UDL	UDL19	55.99	161.56	108.85	67.08	15.56		11.90				
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 1		1	UDL	UDL56	22.20	161.56	108.85	67.08	15.56		11.90				ļ
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 2		2	UDL	UDL56	31.56	161.56	108.85	67.08	15.56		11.90				ļ
┝──-	4 Wire Unbundled Digital Loop 56 Kbps - Zone 3		3	UDL	UDL56	55.99	161.56	108.85	67.08	15.56	ļ	11.90				ł
├ ── ├ ──	Order Coordination for Specified Conversion Time (per LSR)		4	UDL UDL	OCOSL UDL64	22.20	23.02 161.56	108.85	67.08	15.56		11.90				
├──	4 Wire Unbundled Digital Loop 64 Kbps - Zone 1 4 Wire Unbundled Digital Loop 64 Kbps - Zone 2		2	UDL	UDL64 UDL64	22.20 31.56	161.56	108.85	67.08	15.56		11.90				ł
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 2			UDL	UDL64	55.99	161.56	108.85	67.08	15.56	<u> </u>	11.90				ł
	Order Coordination for Specified Conversion Time (per LSR)		5	UDL	OCOSL	55.55	23.02	100.00	07.00	10.00		11.30				
	CLEC to CLEC Conversion Charge without outside dispatch		1	UDL	UREWO		102.11	49.74			1	11.90	1		1	1
2-WIR	E Unbundled COPPER LOOP										1		1		l	İ
	2-Wire Unbundled Copper Loop/Short including manual service		1									1		İ		Ì
	inquiry & facility reservation - Zone 1		1	UCL	UCLPB	8.30	148.50	102.82	75.05	15.63		11.90				
	2-Wire Unbundled Copper Loop/Short including manual service		1													
	inquiry & facility reservation - Zone 2		2	UCL	UCLPB	11.80	148.50	102.82	75.05	15.63	ļ	11.90		ļ		
	2 Wire Unbundled Copper Loop/Short including manual service		_													1
	inquiry & facility reservation - Zone 3		3	UCL	UCLPB	20.94	148.50	102.82	75.05	15.63	L	11.90				<u> </u>
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		9.00	9.00								
	2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 1	1	1	UCL	UCLPW	8.30	123.81	70.09	60.64	9.12		11.90				1
			+	UUL	0001.00	0.00	123.01	70.09	00.04	3.1Z	t	11.90				1
	2-Wire Unbundled Copper Loop/Short without manual service															

UNBUNDLI	ED NETWORK ELEMENTS - Florida	1		1	,								Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring			0		Rates(\$)		
						100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Unbundled Copper Loop/Short without manual service						100.01	== ==								
	inquiry and facility reservation - Zone 3		3	UCL UCL	UCLPW UCLMC	20.94	123.81 9.00	70.09	60.64	9.12	-	11.90				───
	Order Coordination for Unbundled Copper Loops (per loop) 2-Wire Unbundled Copper Loop/Long - includes manual srvc.			UCL	UCLIVIC		9.00	9.00								<u> </u>
	inquiry and facility reservation - Zone 1		1	UCL	UCL2L	17.42	148.50	102.82	75.05	15.63		11.90				
	2-Wire Unbundled Copper Loop/Long - includes manual svc.			UUL	UULZL	17.42	140.00	102.02	13.03	10.00		11.30				
	inquiry and facility reservation - Zone 2		2	UCL	UCL2L	24.76	148.50	102.82	75.05	15.63		11.90				
	2-Wire Unbundled Copper Loop/Long - includes manual svc.					-										
	inquiry and facility reservation - Zone 3		3	UCL	UCL2L	43.94	148.50	102.82	75.05	15.63		11.90				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		9.00	9.00								
	2-Wire Unbundled Copper Loop/Long - without manual service															
	inquiry and facility reservation - Zone 1		1	UCL	UCL2W	17.42	123.81	70.09	60.64	9.12		11.90				L
	2-Wire Unbundled Copper Loop/Long - without manual service						100 -			a :-						1
	inquiry and facility reservation - Zone 2		2	UCL	UCL2W	24.76	123.81	70.09	60.64	9.12		11.90				L
	2-Wire Unbundled Copper Loop/Long - without manual service		~			10.04	100.01	70.00	00.04	0.40		44.00				
	inquiry and facility reservation - Zone 3		3	UCL UCL	UCL2W UCLMC	43.94	123.81 9.00	70.09	60.64	9.12		11.90				
	Order Coordination for Unbundled Copper Loops (per loop) CLEC to CLEC Conversion Charge without outside dispatch			UCL	UCLIVIC		9.00	9.00								
	(UCL -Des)			UCL	UREWO		97.21	42.47				11.90				
4-W/IE	E COPPER LOOP			UCL	UKLWO		97.21	42.47			-	11.50		-		ł
	4-Wire Copper Loop/Short - including manual service inquiry															
	and facility reservation - Zone 1		1	UCL	UCL4S	11.83	177.87	132.76	77.15	17.73		11.90				
	4-Wire Copper Loop/Short - including manual service inquiry			UOL	COLIC	11.00	111.01	102.10	11.10	11.10		11.00				
	and facility reservation - Zone 2		2	UCL	UCL4S	16.81	177.87	132.76	77.15	17.73		11.90				
	4-Wire Copper Loop/Short - including manual service inquiry								-	-						
	and facility reservation - Zone 3		3	UCL	UCL4S	29.82	177.87	132.76	77.15	17.73		11.90				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		9.00	9.00								
	4-Wire Copper Loop/Short - without manual service inquiry and															
	facility reservation - Zone 1		1	UCL	UCL4W	11.83	153.18	100.03	62.74	11.22		11.90				
	4-Wire Copper Loop/Short - without manual service inquiry and															
	facility reservation - Zone 2		2	UCL	UCL4W	16.81	153.18	100.03	62.74	11.22		11.90				
	4-Wire Copper Loop/Short - without manual service inquiry and						150.10									
	facility reservation - Zone 3		3	UCL	UCL4W	29.82	153.18	100.03	62.74	11.22		11.90				<u> </u>
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		9.00	9.00								ł
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 1		1	UCL	UCL4L	31.10	177.87	132.76	77.15	17.73		11.90				1
	4-Wire Unbundled Copper Loop/Long - includes manual svc.			UCL	UUL4L	31.10	177.87	132.76	11.15	17.73		11.90				<u> </u>
	inquiry and facility reservation - Zone 2		2	UCL	UCL4L	44.20	177.87	132.76	77.15	17.73		11.90				1
	4-Wire Unbundled Copper Loop/Long - includes manual svc.	<u> </u>	-		J JLTL	20	111.01	102.70	11.15	11.15		11.30				<u> </u>
	inquiry and facility reservation - Zone 3		3	UCL	UCL4L	78.42	177.87	132.76	77.15	17.73		11.90				1
	Order Coordination for Unbundled Copper Loops (per loop)		Ŭ	UCL	UCLMC		9.00	9.00								
	4-Wire Unbundled Copper Loop/Long - without manual svc.	1														
	inquiry and facility reservation - Zone 1		1	UCL	UCL4O	31.10	153.18	100.03	62.74	11.22		11.90				
	4-Wire Unbundled Copper Loop/Long - without manual svc.	Γ														
	inquiry and facility reservation - Zone 2		2	UCL	UCL4O	44.20	153.18	100.03	62.74	11.22		11.90				ļ
	4-Wire Unbundled Copper Loop/Long - without manual svc.				Ι											1
	inquiry and facility reservation - Zone 3		3	UCL	UCL40	78.42	153.18	100.03	62.74	11.22		11.90				ļ
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		9.00	9.00			<u> </u>	44.00				<u> </u>
	CLEC to CLEC Conversion Charge without outside dispatch			UCL	UREWO		97.21	42.47				11.90				
OOP MODIF				UAL, UHL, UCL,	├ ───┤											<u> </u>
				UAL, UHL, UCL, UEQ, ULS, UEA,												1
	Unbundled Loop Modification, Removal of Load Coils - 2 Wire			UEANL, UDL, UDC,												1
	pair less than or equal to 18k ft			UDN, UDL, USL	ULM2L		0.00	0.00				11.90				
	Unbundled Loop Modification, Removal of Load Coils - 2 wire	1					0.00	0.00				11.00				<u> </u>
	greater than 18k ft			UCL, ULS, UEQ	ULM2G		343.12	343.12				11.90				1
	Unbundled Loop Modification Removal of Load Coils - 4 Wire	1		,,	<u> </u>		,								ĺ	<u> </u>
	less than or equal to 18K ft			UHL, UCL	ULM4L		0.00	0.00				11.90				1

	DLED NETWORK ELEMENTS - Florida		1		1	1					Come Card	Con Cost	Attachment:			ibit: B
CATEGOR	RY RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Unbundled Loop Modification Removal of Load Coils - 4 pair greater than 18k ft	Wire		UCL	ULM4G		242.42	343.12				11.90				
	Unbundled Loop Modification Removal of Bridged Tap R per unbundled loop	emoval,		UAL, UHL, UCL, UEQ, UEF, ULS, UEA, UEANL, UDL, UDC, UDN, UDL, USL	ULMBT		343.12	10.52				11.90				
SUB-LOOP																
Su	ub-Loop Distribution															
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facil Up	ity Set-		UEANL	USBSA		487.23					11.90				
	Sub-Loop - Per Cross Box Location - Per 25 Pair Panel S			UEANL	USBSB		6.25					11.90				
	Sub-Loop - Per Building Equipment Room - CLEC Feed Facility Set-Up	er		UEANL	USBSC		160.25					11.90				
\vdash	Sub-Loop - Per Building Equipment Room - Per 25 Pair	Panel	+	OLAINL	USDOL		169.25					11.90				
	Set-Up	1		UEANL	USBSD		38.65					11.90				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Lo Zone 1		1	UEANL	USBN2	6.46	60.19	21.78	47.50	5.26		11.90				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Lo Zone 2		2	UEANL	USBN2	9.18	60.19	21.78	47.50	5.26		11.90				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Lo Zone 3	oop -	3	UEANL	USBN2	16.29	60.19	21.78	47.50	5.26		11.90				
	Order Coordination for Unbundled Sub-Loops, per sub-l	oop pair		UEANL	USBMC		9.00									
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Lo Zone 1	oop -	1	UEANL	USBN4	7.37	68.83	30.42	49.71	6.60		11.90				
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Lo Zone 2	oop -	2	UEANL	USBN4	10.47	68.83	30.42	49.71	6.60		11.90				
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Lo Zone 3	oop -	3	UEANL	USBN4	18.58	68.83	30.42	49.71	6.60		11.90				
	Order Coordination for Unbundled Sub-Loops, per sub-l	oop pair		UEANL	USBMC		9.00									
	Sub-Loop 2-Wire Intrabuilding Network Cable (INC)	1		UEANL	USBR2	3.96	51.84	13.44	47.50	5.26		11.90				
	Order Coordination for Unbundled Sub-Loops, per sub-I	oop pair		UEANL	USBMC		9.00									
	Sub-Loop 4-Wire Intrabuilding Network Cable (INC)			UEANL	USBR4	9.37	55.91	17.51	49.71	6.60		11.90				
					1000											
\vdash	Order Coordination for Unbundled Sub-Loops, per sub-			UEANL	USBMC UCS2X	F 45	9.00	21.78	47.50	F 00	<u> </u>	44.00				──
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone			UEF UEF	UCS2X UCS2X	5.15 7.31	60.19 60.19	21.78	47.50	5.26 5.26		11.90 11.90				
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone		3	UEF	UCS2X UCS2X	12.98	60.19	21.78	47.50	5.26	<u> </u>	11.90				<u> </u>
	Order Coordination for Unbundled Sub-Loops, per sub-		5	UEF	USBMC	12.30	9.00	21.70	47.50	5.20		11.30				
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone		1	UEF	UCS4X	5.36	68.83	30.42	49.71	6.60		11.90				
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone		2	UEF	UCS4X	7.61	68.83	30.42	49.71	6.60		11.90				
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone			UEF	UCS4X	13.51	68.83	30.42	49.71	6.60		11.90				
	Order Coordination for Unbundled Sub-Loops, per sub-l	oop pair		UEF	USBMC		9.00									ļ
Un	Unbundled Sub-Loop Modification - 2-W Copper Dist Lo.	ad														
	Coil/Equip Removal per 2-W PR			UEF	ULM2X		10.11					11.90				
	Unbundled Sub-loop Modification - 4-W Copper Dist Loa Coil/Equip Removal per 4-W PR			UEF	ULM4X		10.11					11.90				
	Unbundled Sub-loop Modification - 2-w/4-w Copper Dist Tap Removal, per PR unloaded	Bridged		UEF	ULM4T		15.58					11.90				
Un	nbundled Network Terminating Wire (UNTW)															
	Unbundled Network Terminating Wire (UNTW) per Pair			UENTW	UENPP	0.4572	18.02					11.90				
Ne	etwork Interface Device (NID)				1											

UNBUNDLE	D NETWORK ELEMENTS - Florida	1		1								-	Attachment:			ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Network Interface Device (NID) - 1-2 lines			UENTW	UND12		71.49	48.87				11.90				
	Network Interface Device (NID) - 1-6 lines			UENTW	UND16		113.89	89.07				11.90				
	Network Interface Device Cross Connect - 2 W			UENTW	UNDC2		7.63	7.63				11.90				
	Network Interface Device Cross Connect - 4W			UENTW	UNDC4		7.63	7.63				11.90				
SUB-LOOPS	Lan Fooder															<u> </u>
Sub-L	Dop Feeder USL-Feeder, DS0 Set-up per Cross Box location - CLEC			UEA,												<u> </u>
	Distribution Facility set-up			UEA, UDN,UCL,UDL,UDC			487.23					11.90				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair			UEA,	0381 11		407.23					11.90				
	set-up			UDN,UCL,UDL,UDC	USBFX		6.25	6.25				11.90				
	USL Feeder DS1 Set-up at DSX location, per DS1 termination			USL	USBFZ		522.41	11.32				11.90				
	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice															
	Grade - Zone 1		1	UEA	USBFA	6.41	92.75	51.24	58.45	13.07		11.90				
	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice															
	Grade - Zone 2		2	UEA	USBFA	9.10	92.75	51.24	58.45	13.07		11.90				
	Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start,															
	Voice Grade - Zone 3		3	UEA	USBFA	16.15	92.75	51.24	58.45	13.07		11.90				
	Order Coordination for Specified Conversion Time, per LSR			UEA	OCOSL		23.02									
	Unbundlde Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice															
	Grade - Zone 1		1	UEA	USBFB	6.41	92.75	51.24	58.45	13.07		11.90				L
	Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice		~			0.40	00.75	54.04	50.45	40.07		44.00				
	Grade - Zone 2		2	UEA	USBFB	9.10	92.75	51.24	58.45	13.07		11.90				
	Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice Grade - Zone 3		3	UEA	USBFB	16.15	92.75	51.24	58.45	13.07		11.90				
	Order Coordination for Specified Time Conversion, per LSR		3	UEA	OCOSL	10.15	23.02	51.24	30.43	13.07		11.90				
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery,			OLA	OCO3L		23.02									
	Voice Grade - Zone 1		1	UEA	USBFC	6.41	92.75	51.24	58.45	13.07		11.90				
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery,		-													
	Voice Grade - Zone 2		2	UEA	USBFC	9.10	92.75	51.24	58.45	13.07		11.90				
	Unbundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse															
	Battery, Voice Grade - Zone 3		3	UEA	USBFC	16.15	92.75	51.24	58.45	13.07		11.90				
	Order Coordination For Specified Conversion Time, per LSR			UEA	OCOSL		23.02									
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice															
	Grade - Zone 1		1	UEA	USBFD	12.47	106.92	64.46	63.54	14.83		11.90				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice					17.70	100.00									
	Grade - Zone 2		2	UEA	USBFD	17.73	106.92	64.46	63.54	14.83		11.90				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zone 3		3	UEA	USBFD	31.45	106.92	64.46	63.54	14.83		11.90				
	Order Coordination For Specified Conversion Time, Per LSR	-	3	UEA	OCOSL	31.43	23.02	04.40	03.34	14.03		11.90				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice				OCCOL		20.02									
	Grade - Zone 1		1	UEA	USBFE	12.47	106.92	64.46	63.54	14.83		11.90				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice			02.1	000.2		100.02	01110	00.01	1.00		11.00				
	Grade - Zone 2		2	UEA	USBFE	17.73	106.92	64.46	63.54	14.83		11.90				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice															
	Grade - Zone 3		3	UEA	USBFE	31.45	106.92	64.46	63.54	14.83		11.90				
	Order Coordination For Specified Conversion Time, Per LSR			UEA	OCOSL		23.02									
	Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 1		1	UDN	USBFF	14.83	109.71	66.68	60.21	12.49		11.90				
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 2			UDN	USBFF	21.07	109.71	66.68	60.21	12.49		11.90				
├──	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 3	ļ	3	UDN	USBFF	37.39	109.71	66.68	60.21	12.49		11.90				───
	Order Coordination For Specified Conversion Time, Per LSR		1	UDN	OCOSL USBFS	44.00	23.02	00.00	60.21	40.40		44.00				l
├── 	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible) Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)			UDC UDC	USBES USBES	14.83 21.07	109.71 109.71	66.68 66.68	60.21 60.21	12.49 12.49		11.90 11.90				───
├──	Unbundled Sub-Loop Feeder, 2 Wire ODC (IDSL compatible) Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)	<u> </u>	2	UDC	USBFS	37.39	109.71	66.68	60.21	12.49		11.90				╂─────
	Unbundled Sub-Loop Feeder Loop, 4-Wire DSL compatible)	-	3	USL	USBFG	42.59	133.77	78.02	85.16	21.21		11.90				├ ────
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 1		2	USL	USBFG	60.53	133.77	78.02	85.16	21.21		11.90				1
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 2		3	USL	USBFG	107.39	133.77	78.02	85.16	21.21		11.90	1	1		t
	Order Coordination For Specified Conversion Time, Per LSR			USL	OCOSL		23.02	. 0.0E	00.10	2					İ	1
	Unbundled Sub-Loop Feeder, 2-Wire Copper Loop - Zone 1	1	1	UCL	USBFH	3.76	85.27	42.24	58.54	10.82		11.90				1

UNBUND	ED NETWORK ELEMENTS - Florida											1	Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone		2	UCL		5.05	05.07	10.01	50.54	40.00		44.00				
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone		2	UCL	USBFH	5.35	85.27	42.24	58.54	10.82		11.90				
	3		3	UCL	USBFH	9.49	85.27	42.24	58.54	10.82		11.90				
	Order Coordination For Specified Conversion Time, per LSR		-	UCL	OCOSL		23.02									
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 1		1	UCL	USBFJ	7.32	99.66	57.20	60.98	12.28		11.90				
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 2		2	UCL	USBFJ	10.40	99.66	57.20	60.98	12.28		11.90				
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 3		3	UCL	USBFJ	18.46	99.66	57.20	60.98	12.28		11.90				
	Order Coordination For Specified Conversion Time, per LSR			UCL	OCOSL		23.02									L
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		1	UDL	USBFN	14.48	100.62	58.16	63.54	14.83		11.90				ł
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop			UDL UDL	USBFN USBFN	20.59 36.53	100.62 100.62	58.16 58.16	63.54 63.54	14.83 14.83		11.90 11.90				
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop -		3	UDL	USBEIN	30.55	100.62	56.10	03.34	14.03		11.90				ł
	Zone 1		1	UDL	USBFO	14.48	100.62	58.16	63.54	14.83		11.90				1
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -	1	1													1
	Zone 2		2	UDL	USBFO	20.59	100.62	58.16	63.54	14.83		11.90				
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -		ľ													
	Zone 3		3	UDL	USBFO	36.53	100.62	58.16	63.54	14.83		11.90				
	Order Coordination For Specified Time Conversion, per LSR			UDL	OCOSL		23.02									L
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -						100.00	== +=								
	Zone 1		1	UDL	USBFP	14.48	100.62	58.16	63.54	14.83		11.90				┢────
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 2		2	UDL	USBFP	20.59	100.62	58.16	63.54	14.83		11.90				
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -		2	ODL	CODIT	20.55	100.02	50.10	00.04	14.00		11.30				
	Zone 3		3	UDL	USBFP	36.53	100.62	58.16	63.54	14.83		11.90				
	Order Coordination For Specified Conversion Time, per LSR		-	UDL	OCOSL		23.02									
SUB-LOOPS	3 · · · · · · · · · · · · · · · · · ·															
Sub	-Loop Feeder															
	Sub Loop Feeder - DS3 - Per Mile Per Month	1		UE3	1L5SL	15.69										L
	Sub Loop Feeder - DS3 - Facility Termination Per Month	1		UE3	USBF1	347.59	3,402.59	407.15	166.83	94.58		11.90				
	Sub Loop Feeder – STS-1 – Per Mile Per Month			UDLSX	1L5SL	15.69	2 402 50	407.45	400.00	04.50		44.00				
	Sub Loop Feeder - STS-1 - Facility Termination Per Month Sub Loop Feeder – OC-3 – Per Mile Per Month			UDLSX UDLO3	USBF7 1L5SL	402.09 11.90	3,402.59	407.15	166.83	94.58		11.90				<u> </u>
	Sub Loop Feeder - OC-3 - Facility Termination Protection Per			UDL03	TLUGE	11.90										ł
	Month	1		UDLO3	USBF5	62.98										
	Sub Loop Feeder - OC-3 - Facility Termination Per Month	Í		UDLO3	USBF2	547.22	3,402.59	407.15	166.83	94.58		11.90				
	Sub Loop Feeder - OC-12 - Per Mile Per Month	1		UDL12	1L5SL	14.65										
	Sub Loop Feeder - OC-12 - Facility Termination Protection Per															
	Month	I		UDL12	USBF6	502.47										L
	Sub Loop Feeder - OC-12 - Facility Termination Per Month	1	Ļ	UDL12	USBF3	1,577.00	3,402.59	407.15	166.83	94.58		11.90		ļ		
├ ── ├ ──	Sub Loop Feeder - OC-48 - Per Mile Per Month			UDL48	1L5SL	48.06										ł
	Sub Loop Feeder - OC-48 - Facility Termination Protection Per Month	1	1	UDL48	USBF9	251.80										1
	Sub Loop Feeder - OC-48 - Facility Termination Per Month			UDL48 UDL48	USBF9 USBF4	1,589.00	3,588.59	407.15	168.35	95.43	<u> </u>	11.90	ł		ł	ł
	Sub Loop Feeder - OC-12 Interface On OC-48	i	1	UDL48	USBF8	331.15	804.98	407.15	168.35	95.43		11.90				
UNBUNDLE	D LOOP CONCENTRATION	· ·		00210	000.0	001110	00 1100	101110	100.00	00.10		11.00				
	Unbundled Loop Concentration - System A (TR008)	1		ULC	UCT8A	449.49	359.42	359.42			1	11.90				
	Unbundled Loop Concentration - System B (TR008)			ULC	UCT8B	53.44	149.76	149.76				11.90				<u> </u>
	Unbundled Loop Concentration - System A (TR303)			ULC	UCT3A	487.33	359.42	359.42				11.90				L
	Unbundled Loop Concentration - System B (TR303)			ULC	UCT3B	90.05	149.76	149.76			ļ	11.90				Ļ
	Unbundled Loop Concentration - DS1 Loop Interface Card	-		ULC	UCTCO	5.04	71.70	51.52	18.49	4.82	l	11.90				
	Unbundled Loop Concentration - ISDN Loop Interface (Brite		1		111.004	0.00	40.50	40 50	0.77	0.70		44.00				1
	Card) Unbundled Loop Concentration - UDC Loop Interface (Brite			UDN	ULCC1	8.00	16.59	16.50	6.77	6.73		11.90				┟────
	Card)		1	UDC	ULCCU	8.00	16.59	16.50	6.77	6.73		11.90				1
	Unbundled Loop Concentration2 Wire Voice-Loop Start or	-	1	000	JLCCU	0.00	10.39	10.50	0.77	0.75		11.90				ł
	Ground Start Loop Interface (POTS Card)		1	UEA	ULCC2	2.00	16.59	16.50	6.77	6.73		11.90				1
	Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery		1	İ	-						1		1		1	<u> </u>
	Loop Interface (SPOTS Card)	1		UEA	ULCCR	11.90	16.59	16.50	6.77	6.73		11.90				1

Lettern Name Norm	UNBUNDLE	D NETWORK ELEMENTS - Florida												Attachment:	2	Exhi	bit: B
Image: Second Log Convention - Wire Vide Log Yaming - U Price Addr Frid Addr Frid Addr Stock				Zone	BCS	USOC						Submitted Elec	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
Interval Unit Print <							Rec										
Special Card USA UCC4 7:0 10.50 0.77 0.72 11.00 Michael Card Construct - Topic Registrature Herbag USA USC UTC 558 11.50 0.77 6.73 11.00 Michael								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
International Log Concentration - 1551 CRUIT Corr ULC UCPTIC 34.06 16.50 4.77 6.73 11.80 Ubbriefed Log Concentration - Digit IS Reps Data Loop treation U.S. U.S.COS 10.51 16.53 16.73 11.90 Integration Weinbace Loop Concentration - Digit IS Reps Data Loop treation U.S. U.S.COS 10.51 16.53 16.77 6.73 11.90 Integration Weinbace Loop Concentration - Digit IS Reps Data Loop treations U.S. U.S.COS 10.51 16.53 16.77 6.73 11.90 Integration Weinbace U.S.COR Mark Tool Concentration - Digits RE Reps Data Loop Loop Concentration - Digits RE Reps Data Loop Loop Concentration - Digits RE Reps Data Loop Loop Concentration - Digits RE Reps Data Loop Loop Concentration - Digits RE Reps Data Loop - Distributing Digits RE Reps Data Loop - Distributing Digits RE Reps Data Loop - Distributing Digits RE Reps Data Loop - Distributing Digits RE Reps Data Loop - Distributing Digits RE Reps Data Loop - Distributing Digits RE Reps Data Loop - Distributing Digits RE Reps Data Loop - Distributing Digits RE Reps Data Loop - Distributing Digits RE Reps Data Loop - Distributing Distributing Digits RE Reps Data Loop - Distributing Distributing Distributing Distributing Distributing Distributing Distributing Distributing Distributing Distributing Distributing Distributing Distributing Distributing Distributing Distributing Distributing Distributing Distributing D					IIEA		7 10	16 59	16 50	6.77	6.73		11 90				
Unbuilded Log Concentration - Opini 28 Stops Data Log ULC Hash																	
Unbuinded Log Concentration - Digital 54 Maps Data Log UDL ULCGS 10.0 16.00 6.77 6.73 11.0 International concentration - Digital 54 Maps Data Log UDL ULCGS 10.01 16.60 16.60 6.77 6.73 11.00 International Concentration - Digital 54 Maps Data Log International Concentration - Digital 54 Maps Data Log International Concentration - Digital 54 Maps Data Log International Concentration - Digital 54 Maps Data Log International Concentration - Digital 54 Maps Data Log International Concentration - Digital 54 Maps Data Log International Concentration - Digital 54 Maps Data Log International Concentration - Digital 54 Maps Data Log International Concentrational Concentencentrenal Concentrational Concentrational Concentrational Conc																	
Interior UDC ULCC 101 160 6.77 6.73 1100 UPC OFFER, PEOPLOSIONED OUX - NO RATE UDL UDC.00 164 1650 6.77 6.73 1100 UPC OFFER, PEOPLOSIONED OUX - NO RATE UDL UDC.00 0.00 0.00 <					UDL	ULCC7	10.51	16.59	16.50	6.77	6.73		11.90				
Interface UDU ULCs 100 16:50 6:77 6:73 11:00 Interface UPO (THE, FROM Stands out Y, No Rate UERNU, UPU (Service) 6:50 6:77 6:73 11:00 0 Interface UPU (Service) 10:80% 0:80% 0:00 <t< td=""><td></td><td>Interface</td><td></td><td></td><td>UDL</td><td>ULCC5</td><td>10.51</td><td>16.59</td><td>16.50</td><td>6.77</td><td>6.73</td><td></td><td>11.90</td><td></td><td></td><td></td><td></td></t<>		Interface			UDL	ULCC5	10.51	16.59	16.50	6.77	6.73		11.90				
NDC - Degrade and Service Order for NDL registration UP/NV UMBXX 0.00 0.00 INFO - Degrade and Service Order for NDL registration US/NV (FUE) UP/NV		Interface			UDL	ULCC6	10.51	16.59	16.50	6.77	6.73		11.90				
UNTW Clock of Equilations Experiment Provide sing Only - No Rate UPNU_UEVEC 0.00 0.00 0.00 0.00 0.00 UNE OTHER, PROVISIONS ONLY - NO RATE UPNU_UEVEVEV UPNU_OTHER, PROVISIONS ONLY - NO RATE UPNU_UEVEVEV 0.00	UNE OTHER, P						0.00	0.00									
International Contract Name, Provisioning Dhi) - No Rate UNECYNER, PROVISIONEG ONL''' NO RATE UNECYNER, PROVISIONEG ONL''' NO RATE UNE OTHER, PROVISIONEG ONL''' NO RATE UNEL UCLUDCUCK, UNECN 0.00 0.00 0.00 Ubbundled Contract Name, Provisioning Only - no rate UNEL/LICLUDCUCK, UNECN 0.00 0.00 0.00 Ubbundled Sul-Loop Feeder 4 Wire Cross Box Jumper - no UBLUDLULCUC USPT 0.00 0.00 0.00 0.00 Ubbundled Sul-Loop Feeder 4 Wire Cross Box Jumper - no UBLUDLULCUC USPT 0.00 0.00 0.00 0.00 Ubbundled Sul-Loop Feeder 4 Wire Cross Box Jumper - no UBLUDLULCUC USPT 0.00	├ ── ├ ──											<u> </u>					
Internet Unkonded Contract Name, Provisioning Only - No Rate ENTW UNECN 0.00 0.00 0.00 0.00 Inter OTHER, Provisioning Only - no rate UNLUCLUDCUDC, UNLUCLUDCUDCU, UNLUCLUCLUDCUDCU, UNLUCLUCLUDCUDCU, UNLUCLUCLUDCUDCU, UNLUCLUDCUDCU, UNLUCLUCLUDCUDCU, UNLUCLUCLUDCUDCU, UNLUCLUCLUDCUDCU, UNLUCLUCLUCLUCLUDCUDCU		erter ender a Lotabionment, i Tovisioning Only - No Kale				SLINGL	0.00	0.00					-				
Unbundled Contact Name. Providening Only - no rate UAL UCLUDC UDL UDNURAVELUC 0.00 0.00 0.00 0.00 0.00 Homored Sol-Loop Fooder-2 Wer Cross Box Jumper - no UEAUNCLUDC UDC USSPC 0.00 0.0						UNECN	0.00	0.00									
Ubundled Contact Name, Providening Only - no rate UDNUERAUHULUC UNERN 0.00 0.00 0.00 Underded Sub-Loop Feeder 4 Wire Cross Box Jumper - no ate UEA UDNUCLUD USBFR 0.00				1								1					
Inte ULAUDRUCLUC USEFO 0.00 0.00 0.00 Hubundle Ski-Loop Feder4 Wire Cross Box Jumper - no UEAUSLUCLUDU USER 0.00		Unbundled Contact Name, Provisioning Only - no rate				UNECN	0.00	0.00									
Intel UEAUSLUCULU UserF 0.00 0.00 0.00 Hubunded DS1 Loop - Expanded Superframe Format option - no rate USL CCOSF 0.00		rate			UEA,UDN,UCL,UDC	USBFQ	0.00	0.00									
Unbundled DS1 Loop - Expanded Superframe Format option - no rate USL CCOEF 0.00 0.00 0.00 HGH CAPACITY UNBUNDLED LOCAL LOOP USL CCOEF 0.00 0.00 0.00 0.00 0.00 HGH CAPACITY UNBUNDLED LOCAL LOOP - DS3 - Per Mile per month UE3 1LSND 10.92 0.00		rate															
In orate USL CCOEF 0.00 0.00 0.00 High CAPACT/UNBUNDED LOCAL LOOP Image: Coefficient of the coeffic					USL	CCOSF	0.00	0.00									
HIGH CAPACITY UNEUROLED LCALL LOOP Image: Control of the control of the						00055	0.00	0.00									
High Capacity Unbundled Local Loop - DS3 - Fer Mile per month UE3 LSND 10.92 Image: Constraint of the constraint of the	HIGH CAPACI				USL	CLUEF	0.00	0.00									
High Capacity Unbundled Local Loop - DS3 - Facility UE3 UE3PX 386.88 556.37 343.01 139.13 96.84 11.90 Image: Control C		High Capacity Unbundled Local Loop - DS3 - Per Mile per			UE3	1L5ND	10.92										
High Capacity Unbundled Load Loop - STS-1 - Fer Mile per month UDLSX 1LSND 10.92 Image: Capacity Unbundled Load Loop - STS-1 - Feality Image: Capacity Unbundled Load Loop - STS-1 - Feality Image: Capacity Unbundled Load Loop - STS-1 - Feality Image: Capacity Unbundled Load Loop - STS-1 - Feality Image: Capacity Unbundled Load Loop - STS-1 - Feality Image: Capacity Unbundled Load Loop - STS-1 - Feality Image: Capacity Unbundled Load Loop - STS-1 - Feality Image: Capacity Unbundled Load Loop - STS-1 - Feality Image: Capacity Unbundled Load Loop - STS-1 - Feality Image: Capacity Unbundled Load Loop - STS-1 - Feality Image: Capacity Unbundled Load Loop - STS-1 - Feality Image: Capacity Unbundled Load Loop - STS-1 - Feality Image: Capacity Unbundled Load Loop - STS-1 - Feality Image: Capacity Unbundled Load Loop - STS-1 - Feality Image: Capacity Unbundled Load Loop - STS-1 - Feality Image: Capacity Unbundled Load Loop - STS-1 - Feality Image: Capacity Capacity - StS-1 Image: Capacity - StS-1 Image: Capacity - StS-1 Image: Capacity - StS-1 Image: Capacity - StS-1 Image: Capacity - StS-1 Image: Capacity - StS-1 Image: Capacity - StS-1 Image: Capacity - StS-1 Image: Capacity - StS-1 Image: Capacity - StS-1 Image: Capacity - StS-1 Image: Capacity - StS-1 Image: Capacity - StS-1 Image: Capacity - StS-1 Image: Capacity - StS-1 Image: Capacity - StS-1 Image: Capacity - StS-1		High Capacity Unbundled Local Loop - DS3 - Facility						556 37	343.01	139 13	96.84		11 90				
High Capacity Unbundled Local Loop - STS-1 - Facility UDLSX UDLS1 426.60 556.37 343.01 139.13 96.84 11.90 1.83 LOOP MAKE-UP Spare facility queried (Manual). UMK UMK UMK 52.17		High Capacity Unbundled Local Loop - STS-1 - Per Mile per						000.01	0 10.01	100.10	00.01		11100				
LOOP MAKE-UP Image: Coop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual). Image: Coop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual). Image: Coop Makeup - Preordering With Reservation, per spare facility queried (Manual). Image: Coop Makeup - Preordering With Reservation, per working or spare facility queried (Manual). Image: Coop Makeup - Preordering With Reservation, per working or gueried (Manual). Image: Coop Makeup - Preordering With Reservation, per working or gueried (Manual). Image: Coop Makeup - Preordering With Reservation, per working or gueried (Machanized) Image: Coop Makeup - Preordering With Reservation, per working or gueried (Machanized) Image: Coop Makeup - Preordering With Reservation, per working or gueried (Machanized) Image: Coop Makeup - Preordering With Reservation, per working or gueried (Machanized) Image: Coop Makeup - Preordering With Reservation, per working or gueried (Machanized) Image: Coop Makeup - Preordering With Reservation, per working or gueried (Machanized) Image: Coop Makeup - Preordering With Reservation, per working or gueried (Machanized) Image: Coop Makeup - Preordering With Reservation, per working or gueried (Machanized) Image: Coop Makeup - Preordering With Reservation, per working or gueried (Machanized) Image: Coop Makeup - Preordering With Reservation, per working or gueried (Machanized) Image: Coop Makeup - Preordering With Reservation, per working or gueried (Machanized) Image: Coop Makeup - Preordering With Reservation, per working or gueried (Machanized) Image: Coop Makeup - Preordering With Reservation - gueried (Machanized)		High Capacity Unbundled Local Loop - STS-1 - Facility						556.37	343 01	139 13	96.84		11 90			1.83	
Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual). UMK UMKLW 52.17 52.17 Image: Constraint of the servation of	LOOP MAKE-U				0020/	00201	120.00	000.01	010101	100110	00.01		11.00				
queried (Manual). UMK UMK UMK WMKLP 55.07		Loop Makeup - Preordering Without Reservation, per working or			UMK	UMKLW		52.17	52.17								
Image: Indity queried (Mechanized) UMK PSUMK 0.6784 0.60 11.90 0.6784					UMK	UMKLP		55.07	55.07								
LINE SHARING Image: Constraint of the constraint of the					UMK	PSUMK		0.6784	0.6784								
SPLITTERS-CENTRAL OFFICE BASED Image: Constraint of the constraint of the																	
Line Sharing Splitter, per System 96 Line Capacity - True up pending approval by PSC R ULS ULSDA 119.72 379.13 0.00 347.90 0.00 11.90 Interview Line Sharing Splitter, per System 24 Line Capacity - True up pending approval by PSC R ULS ULSDB 29.93 379.13 0.00 347.90 0.00 11.90 Interview Inte												<u> </u>					
Image: pending approval by PSC R ULS ULSDA 119.72 379.13 0.00 347.90 0.00 11.90 Image: pending approval by PSC Line Sharing Splitter, per System 24 Line Capacity - True up pending approval by PSC R ULS ULSDB 29.93 379.13 0.00 347.90 0.00 11.90 Image: pending approval by PSC P Image: pending approval by PSC(BST Owned Splitter) P Image: pending approval by PSC(BST Owned Splitter) P Image: pend	SPLIT			<u> </u>													
Image: pending approval by PSC R ULS ULSDB 29.93 379.13 0.00 347.90 0.00 11.90 Image: pending approval by PSC Line Sharing Splitter, Per System, 8 Line Capacity I ULS ULSDB 8.33 379.13 0.00 347.90 0.00 11.90 Image: pending approval by PSC(BST Owned Splitter) R ULS ULSDS Image: pending approval by PSC(BST Owned Splitter) Image: pending approval by PSC(BST Owned Splitter) R ULS Image: pending approval by PSC(BST Owned Splitter) Image: pending approval by PSC(BST Owned Splitter) R Image: pending approval by PSC(BST Owned Splitter) Image: pending approval by PSC(BST Owned Splitter) R Image: pen		pending approval by PSC	R		ULS	ULSDA	119.72	379.13	0.00	347.90	0.00		11.90				
Line Sharing-DLEC Owned Splitter in CO-CFA activation- (deactivation (per LSOD) ULS ULSDG 173.66 0.00 97.42 0.00 11.90 Integration		pending approval by PSC	R														
deactivation (per LSOD) ULS ULSDG 173.66 0.00 97.42 0.00 11.90 0 0 0 END USER ORDERING-CENTRAL OFFICE BASED-HIGH FREQUENCY SPECTRUM AKA LINE SHARING 0 0 97.42 0.00 11.90 0 0 0 Line Sharing - per Line Activation - (BST Owned Splitter) ULS ULSDC 0.61 29.68 21.28 19.57 9.61 11.90 0		Line Sharing Splitter, Per System, & Line Capacity			ULO	ULSUS	8.33	3/9.13	0.00	347.90	0.00		11.90				
END USER ORDERING-CENTRAL OFFICE BASED-HIGH FREQUENCY SPECTRUM AKA LINE SHARING Image: Constraint of the const					ULS	ULSDG		173.66	0.00	97.42	0.00		11.90				
Line Sharing - per Subsequent Activity per Line Rearrangement - True up pending approval by PSC(BST Owned Splitter) R ULS ULSDS 21.68 16.44 11.90 11.90 11.90 Line Sharing - per Subsequent Activity per Line Rearrangement ULS ULSDS 21.68 16.44 11.90 11.90 11.90 11.90	END U	SER ORDERING-CENTRAL OFFICE BASED-HIGH FREQUENCY	SPEC														
- True up pending approval by PSC(BST Owned Splitter) R ULS ULSDS 21.68 16.44 11.90 11.90		Line Sharing - per Line Activation -(BST Owned Splitter)			ULS	ULSDC	0.61	29.68	21.28	19.57	9.61		11.90				
Line Sharing - per Subsequent Activity per Line Rearrangement		Line Sharing - per Subsequent Activity per Line Rearrangement - True up pending approval by PSC(BST Owned Solitter)	R		ULS	ULSDS		21.68	16.44				11.90				
- True up pending approval by PSC(DLEC Owned Splitter) R ULS ULSCS 21.68 16.44 11.90		Line Sharing - per Subsequent Activity per Line Rearrangement															

INBUND	DLED	O NETWORK ELEMENTS - Florida	1	1		1	1					.		Attachment:			bit: B
ATEGOR	Y	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'I
							Rec	Nonrec		Nonrecurring					Rates(\$)	_	
		Line Charing and Line Activistics (DLEC surged Californ)			ULS	ULSCC		First 47.44	Add'l 19.31	First 20.67	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
LIN		Line Sharing - per Line Activation (DLEC owned Splitter) PLITTING			ULS	ULSCC	0.61	47.44	19.31	20.67	12.74		11.90				<u> </u>
		ER ORDERING-CENTRAL OFFICE BASED															<u> </u>
	000	Line Splitting - per line activation DLEC owned splitter	1		UEPSR UEPSB	UREOS	0.61										
		Line Splitting - per line activation BST owned - physical	I		UEPSR UEPSB	UREBP	0.61	29.68	21.28	19.57	9.61		11.90				
		Line Splitting - per line activation BST owned - virtual	1		UEPSR UEPSB	UREBV	1.134	29.68	21.28	19.57	9.61		11.90				
		E SITE HIGH FREQUENCY SPECTRUM															
SP	LITT	ERS-REMOTE SITE															
		Remote Site Line Share BellSouth Owned Splitter, 24 Port	I		ULS	ULSRB	25.00	150.00	0.00	150.00	0.00		11.90				L
		Remote Site Line Share Cable Pair Activation CLEC Owned at RS and deactivation			ULS	ULSTG		74.38	0.00	46.77	0.00		11.90				
EN		RS and deactivation ER ORDERING-REMOTE SITE HIGH FREQUENCY SPECTRUM		PEMOT				74.38	0.00	40.77	0.00		11.90				<u> </u>
	5 03	Remote Site Line Share Line Activation for End User Served at				1	 					1					<u> </u>
		RS, BST Splitter RS Line Share Line Activation for End User served at RS, CLEC	I		ULS	ULSRC	0.61	40.00	22.00	19.57	9.61		11.90				
		Splitter			ULS	ULSTC	0.61	40.00	22.00	19.57	9.61		11.90				
NBUNDLE					010	02010	0.01	40.00	22.00	10.07	0.01		11.00				
		INTEROFFICE CHANNEL DEDICATED TRANSPORT - minimu	m billin	g perio	d - below DS3=one	month, DS3/	STS-1=four mo	nths									
		FFICE CHANNEL - DEDICATED TRANSPORT		ľ													
		Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade - Per Mile per month			U1TVX	1L5XX	0.0091										
		Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade - Facility Termination			U1TVX	U1TV2	25.32	47.35	31.78	18.31	7.03		11.90				
		Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade Rev Bat Per Mile per month			U1TVX	1L5XX	0.0091										
		Interoffice Channel - Declated Transport- 2- Wire VG Rev Bat Facility Termination			U1TVX	U1TR2	25.32	47.35	31.78	18.31	7.03		11.90				
		Interoffice Channel - Dedicated Transport - 4-Wire Voice Grade - Per Mile per month			U1TVX	1L5XX	0.0091	47.55	51.70	10.01	7.00		11.50				
		Interoffice Channel - Dedicated Transport - 4- Wire Voice Grade			01177	TLJAA	0.0091										
		- Facility Termination Interoffice Channel - Dedicated Transport - 56 kbps - per mile			U1TVX	U1TV4	22.58	47.35	31.78	18.31	7.03		11.90				<u> </u>
		per month Interoffice Channel - Dedicated Transport - 56 kbps - Facility			U1TDX	1L5XX	0.0091										l
		Termination			U1TDX	U1TD5	18.44	47.35	31.78	18.31	7.03		11.90				
		Interoffice Channel - Dedicated Transport - 64 kbps - per mile per month			U1TDX	1L5XX	0.0091										
		Interoffice Channel - Dedicated Transport - 64 kbps - Facility Termination			U1TDX	U1TD6	18.44	47.35	31.78	18.31	7.03		11.90				
		Interoffice Channel - Dedicated Channel - DS1 - Per Mile per						47.33	51.76	10.51	7.05		11.90				
		month Interoffice Channel - Dedicated Tranport - DS1 - Facility			U1TD1	1L5XX	0.1856										
	_	Termination Interoffice Channel - Dedicated Transport - DS3 - Per Mile per			U1TD1	U1TF1	88.44	105.54	98.47	21.47	19.05		11.90				<u> </u>
		month Interoffice Channel - Dedicated Transport - DS3 - Facility			U1TD3	1L5XX	3.87										
		Termination per month Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per			U1TD3	U1TF3	1,071.00	335.46	219.28	72.03	70.56		11.90				ł
	_	month Interoffice Channel - Dedicated Transport - STS-1 - Facility			U1TS1	1L5XX	3.87										
LO		Termination CHANNEL - DEDICATED TRANSPORT			U1TS1	U1TFS	1,056.00	335.46	219.28	72.03	70.56		11.90				<u> </u>
		LOCAL CHANNEL DEDICATED TRANSPORT - minimum billin	g perio	d - belo	ow DS3=one month,	, DS3/STS-1=1	four months					İ					
		Local Channel - Dedicated - 2-Wire Voice Grade - Zone 1		1	ULDVX	ULDV2	19.66	265.84	46.97	37.63	4.00		11.90				
		Local Channel - Dedicated - 2-Wire Voice Grade - Zone 2		2	ULDVX	ULDV2	27.94	265.84	46.97	37.63	4.00		11.90				
		Local Channel - Dedicated - 2-Wire Voice Grade - Zone 3		3	UNDVX	ULDV2	49.58	265.84	46.97	37.63	4.00		11.90				<u> </u>
		Local Channel - Dedicated - 2-Wire Voice Grade Rev. Bat Zone 1		1	ULDVX	ULDR2	19.66	265.84	46.97	37.63	4.00		11.90				

CATE OF RATE ELEMENTS Image Row Row	JNBUNDLED	ONETWORK ELEMENTS - Florida		r								1		Attachment:			bit: B
Image: Control - Dockard - 2-Wire Vood Grade Net Mer. Image: Control - Dockard - 2-Wire V	ATEGORY	RATE ELEMENTS		Zone	BCS	USOC			RATES(\$)			Submitted Elec	Submitted Manually	Charge - Manual Svc Order vs. Electronic-	Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
Load Orward - Deduced - 2Mine Visco Gade Res. 2 LDXX ULDX ULDX ULDX Parts Add? Parts Parts Parts Parts Parts<							Boc	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		
Image Image <th< th=""><th></th><th></th><th></th><th></th><th></th><th></th><th>Nec</th><th>First</th><th>Add'l</th><th>First</th><th>Add'l</th><th>SOMEC</th><th>SOMAN</th><th>SOMAN</th><th>SOMAN</th><th>SOMAN</th><th>SOMAN</th></th<>							Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Load Channel: Deck deal Ave: Log deal Log deal <thlog deal<="" th=""></thlog>									10.07								
Zon 3 J. ULOX ULDX 44.958 26.54 46.07 77.63 4.00 11.00 Image: Constrained State St				2	ULDVX	ULDR2	27.94	265.84	46.97	37.63	4.00		11.90	-			l
Load Obsend: Descard: 4 We Vision Grade - Zane 1 1 UNX ULV 20.46 4767 44.22 5.31 11.90 Image: Control - Contro				3			19 58	265.84	46.97	37.63	4.00		11 90				
Land Dimend - Johnson 4 - Varie Vong Guarde - Zone 2 2 UNDYX UDVA 206 30 47.87 44.22 6.33 11.80 Load Dimend - Declination - Varie Vong Guarde - Zone 3 3 UDVA UDVA 15.88 265.24 47.87 44.22 6.33 11.80 Impact - Declination - Dec														1			
Local Channel - Soldiand - 4-We yoog Grade - Zoes 3 5 UNOW ULDW 518 268-51 47.7 44.22 5.33 11.00 Image Local Channel - Soldiand - Sist Zwei 1 2 ULDDY ULDY 64.60 165.54 24.50 165.65 11.00														1			
Local Channel Desiculate DS1 - Zona 2 2 LLDDT ULDF1 S18 216.85 183.54 24.30 16.85 11.80 Local Dennel Desiculate DS1 - Zona 3 3 LLDD1 LLDD2 1200 216.65 153.5 / 243.01 163.65 11.80 — Local Dennel Desiculate DS1 - Zona 3 0 LLDD1 LLDD2 66.637 343.01 131.15 — …						ULDV4							11.90				
Local Channel - Dedication DS1 - Zman 3 3 ULDP1 ULDP1 ULDP1 ULDP1 ULDP3 21.66 13.80 11.80 Model Local Channel - Dedication DS3 - Fully Termination LLDD3 ULDP3 S1.81 565.77 340.01 138.13 66.84 11.90 Model Local Channel - Dedication IDS3 - Fully Termination LLDD3 ULPP3 S1.81 565.17 340.01 139.13 66.84 11.90 Model				1			36.49	216.65			16.95		11.90				
Load Datawet-Decisioned -DSS - Factor Firmation ULDD3 U.SAC R.S.D				2				216.65					11.90				
Local Channel - Dedicate - DS3 - Facility Termination ULDDS VLDDF S51.1 556.57 343.01 193.13 96.84 11.00 Machinel Local Channel - Dedicate - STS1 - Facility Termination ULDDS1 ULDPS 540.69 556.57 343.01 159.13 06.84 11.00 Machinel Mac				3				216.65	183.54	24.30	16.95		11.90				
Load Channel - Dedicated -STS1 - Facility Flemination ULDS1 ULSN USN ULSN ULSN ULSN ULSN ULSN ULSN ULSN ULSN ULSN ULSN ULSN ULSN ULSN ULSN ULSN ULSN ULSN ULSN USN <th< td=""><td>L</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>ļ</td></th<>	L																ļ
Load Channel - Dedicated - STS-1 - Facility Termination ULDS1 ULDFS 540.69 550.73 343.01 139.13 98.64 11.90 Dark Fiber, Tour Fiber Strands, Per Route Mile of Fraction The Dark Fiber, Coal Channel UDF ULDC 55.04	L							556.37	343.01	139.13	96.84		11.90				ļ
DARK Fleer, Dark Fleer, Fleer								FF0 07	0.10.01	400.40	00.01		11.00	ļ			l
But Fiber, Four Fiber Strands, Per Route Mile or Fraction UDF 1LSC 55.4 Image: Constraint of the cons		Local Channel - Dedicated - STS-1 - Facility Termination			ULDS1	ULDES	540.69	556.37	343.01	139.13	96.84		11.90				 '
Image of per north - Local Channel UDP 1LSDC 55.54 - <td></td> <td>Dark Fiber, Four Fiber Strands, Par Pouto Milo or Fraction</td> <td> </td> <td></td> <td></td> <td></td> <td> </td> <td></td> <td></td> <td>├</td> <td></td> <td></td> <td></td> <td><u> </u></td> <td></td> <td></td> <td> '</td>		Dark Fiber, Four Fiber Strands, Par Pouto Milo or Fraction								├				<u> </u>			 '
INRC Dark Flor - Local Channel UDF UDFAL 751.34 193.88 11.00 10.00 Back Flore, Flore Flore Channel UDF 1.50F 26.85 193.88 11.90 10.00 NRC Dark Flore, Flore Flore Channel UDF UDF14 751.34 193.88 11.90 10.00 Dark Flore, Flore, Flore Flore Strands, Par Route Mile or Fraction UDF UDF14 751.34 193.88 11.90 10.00 BXX Access Ten Digit Screening, Reservation Charge Per 8XX UDF UDF14 751.34 193.88 11.90 11.90 10.00 BXX Access Ten Digit Screening, Reservation Charge Per 8XX UDF UDF14 751.34 193.88 11.80 11.90 11						11.500	55.04										1
Dark Flow, Four, Flow Strands, Per Route Mile of Frazion UDF 1LSDF 28.85 C							55.04	751.34	193.88				11.90				<u> </u>
Image of performed - Interofice Channel UDF 1LOPF14 751.34 113.88 Image of the constraints of t					05.	00101		101101	100.00	1			11.00				
Dark Fber, Four Fiber Strands, Per Route Mie or Fraction UpF LDR 55.04 Four Fiber Local Loop Four Fiber Local Loop Four Fiber Local Loop Four Fiber Local Loop Four Fiber Local Loop Four Fiber Four Fib					UDF	1L5DF	26.85										1
Image of pronch - Local Loop UDF 11,50L 55.04 C <thc< th=""></thc<>						UDF14		751.34	193.88				11.90				
INRC Dark Flbar - Local Loop UDF UDFL4 751.34 193.88 11.90 10 8X ACCESS Ten Digit Screening, Per Call 0+D 0.0006252 -	[Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction															
BXX Access Tro Noid Screening. Per Call OH <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>55.04</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>							55.04										
Image: Note Access Ten Digit Screening, Per Call OHD 0.0006252 Image: Note Access Ten Digit Screening, Reservation Charge Per RXX Number Reservation Charge Per RXX Number Reservation Charge Per RXX Number Reservation Charge Per RXX Number Reservation Charge Per RXX Number Reservation Charge Per RXX Number Reservation Charge Per RXX Number Reservation Charge Per RXX Number Reservation Charge Per RXX Number Reservation Charge Per RXX Number Reservation Charge Per RXX Number Reservation Charge Per RXX Number Reservation Charge Per RXX Number Reservation Charge Per RXX Number Reservation Charge Per RXX Number Reservation Charge Per RXX Number Reservation Charge Per RXX Number RXX N					UDF	UDFL4		751.34	193.88				11.90				
BXX Access Ten Digl Streening, Reservation Charge Per 8XX OHD NBR1X 4.15 0.70 11.90 11.90 11.90 8XX Access Ten Digl Streening, Per 8XX No. Established Win 0HD NBR1X 8.78 1.18 5.77 0.70 11.90<					0115												'
Number Reserved OHD NBR1X 4.15 0.70 Image: Constraints of the constraints of					OHD		0.0006252										
BXX Access Ten Digit Screening, Per 8XX No. Established With POTS Translations OHD NBFTX 8.78 1.18 5.77 0.70 11.90 Image: Constraints BXX Access Ten Digit Screening, Customized Area of Service Per 8XX Number OHD NBFTX 8.78 1.18 5.77 0.70 11.90 Image: Constraints BXX Access Ten Digit Screening, Multiple InterLATA CXR Routing Per CXR Requested Per 8XX No. OHD NBFCX 4.15 2.07 11.90 Image: Constraints 11.90 Image: Constraints 11.90 Image: Constraints Image: Constraints 11.90 Image: Constraints Image: Con								4.15	0.70				11.00				
POTS Translations OHD 8.78 1.18 5.77 0.70 11.90 Image: constraints POTS Translations OHD NBFTX 8.78 1.18 5.77 0.70 11.90 Image: constraints POTS Translations OHD NBFTX 8.78 1.18 5.77 0.70 11.90 Image: constraints Per XX Access Ten Diglt Screening, Wiltiple Intel/LTA CXR Routing Per CXR Requested Per 8XX No. OHD NBFXX 4.85 2.78 Image: constraints			-		UHD	INOR IA		4.15	0.70				11.90				
BXX Access Ten Digit Screening, Der BXX No. Established With Port's Translations OHD NBFTX 8.78 1.18 5.77 0.70 11.90 Image: Constraint of the constraint of the					ОНО			8 78	1 18	5 77	0.70		11 90				
POTS Translations OHD NBTX 8.78 1.18 5.77 0.70 11.90 Image: constraints BXX Access Ten Digit Screening, Customized Area of Service Per 8XX Number OHD NBFCX 4.15 2.07 11.90					OND			0.70	1.10	5.11	0.70		11.30				
BXX Access Ten Digit Screening, Customized Area of Service Per 8XX Number OHD NBFCX 4.15 2.07 11.90 11.90 BXX Access Ten Digit Screening, Multiple InterLATA CXR Routing Per CXR Requested Per 8XX No. OHD NBFRX 4.85 2.78 11.90					OHD	N8FTX		8.78	1.18	5.77	0.70		11.90				
Per 8XX Number OHD N8FCX 4.15 2.07 11.9 1																	
Routing Per CXR Requested Per 8XX No. OHD NBFMX 4.85 2.78 (11.90) (11.9					OHD	N8FCX		4.15	2.07				11.90				
8XX Access Ten Digit Screening, Call Handling and Destination Reatures 0HD N8FAX 4.85 0.70 11.90 11.90 11.90 8XX Access Ten Digit Screening, Call Handling and Destination Features 0HD N8FAX 4.15 4.15 11.90	8	8XX Access Ten Digit Screening, Multiple InterLATA CXR															
BXX Access Ten Digit Screening, VaFL No. Delivery, per query OHD N8FDX 4.15 4.15 11.9 11.90	F	Routing Per CXR Requested Per 8XX No.															
Image: services of the services of the service of	8	8XX Access Ten Digit Screening, Change Charge Per Request			OHD	N8FAX		4.85	0.70				11.90				L
Bit Access Ten Digit Screening, w/ 8FL No. Delivery, per query OHD 0.0006252 Image: Control of the stability of the sta					0115	NOFEN		4.45					44.00				
BXX Access Ten Digit Screening, w/ POTS No. Delivery, per query OHD 0.0006252 Image: Constraint of the constraint of	1	Features			OHD	N8FDX		4.15	4.15				11.90				
BXX Access Ten Digit Screening, w/ POTS No. Delivery, per query OHD 0.0006252 Image: Constraint of the constraint of		8XX Access Ten Digit Screening w/ 8EL No. Delivery, por guory			ОНР		0.0006252							1			1
query OHD 0.0006252 Image: Constraint of the c					OND		0.0000232										
LINE INFORMATION DATA BASE ACCESS (LIDB) OQT O.0000203 Image: Constraint of the co					OHD		0.0006252										1
LIDB Validation Per Query OQU 0.0136959 Image: Constraint of the constraint of t	INE INFORMAT	TION DATA BASE ACCESS (LIDB)			-												
LDB Originating Point Code Establishment or Change OQT, OQU NRPBX 55.13 55.13 55.13 55.13 11.90 Income	L	LIDB Common Transport Per Query			OQT		0.0000203										
SIGNALING (CCS7) 0							0.0136959										
CC\$7 Signaling Termination, Per STP Port UDB PT8SX 135.05 Image: Comparison of the comparison of the					OQT, OQU	NRPBX		55.13	55.13	55.13	55.13		11.90				
CCS7 Signaling Usage, Per TCAP Message UDB 0.0000607 0																	L
CCS7 Signaling Connection, Per link (A link) UDB TPP++ 17.93 43.57 43.57 18.31 11.90 Image: Connection (Con						PT8SX								ļ			
CCS7 Signaling Connection, Per link (B link) (also known as D link) UDB TPP++ 17.93 43.57 43.57 18.31 11.90 11			ļ			TDD		40.57	40.57	40.01	40.04		44.00	ł			ł
link) UDB TPP++ 17.93 43.57 43.57 18.31 18.30 11.90 Implementation CCS7 Signaling Usage, Per ISUP Message UDB 0.0000152 Implementation <td< td=""><td></td><td></td><td> </td><td></td><td>UDB</td><td>122++</td><td>17.93</td><td>43.57</td><td>43.57</td><td>18.31</td><td>18.31</td><td></td><td>11.90</td><td><u> </u></td><td></td><td></td><td> </td></td<>					UDB	122++	17.93	43.57	43.57	18.31	18.31		11.90	<u> </u>			
CCS7 Signaling Usage, Per ISUP Message UDB 0.0000152 Image: Cost of the second s					UDB	TPP++	17 03	43 57	43 57	18 31	18 31		11 00	1			1
CCS7 Signaling Usage Surrogate, per link per LATA UDB STU56 694.32 Image: CCS7 Signaling Point Code, per Originating Point Code CCS7 Signaling Point Code, per Originating Point Code Image: CCS7 Signaling Point Code Image: CCS7 Signaling Point Code Image: CCS7 Signaling Point Code						11 1 77		-5.57	+3.37	10.31	10.31		11.90	1			
CCS7 Signaling Point Code, per Originating Point Code						STU56				† †				1	1		l
		CCS7 Signaling Point Code, per Originating Point Code	I		-							1	1	1		l	
	E	Establishment or Change, per STP affected			UDB	CCAPO		46.03	46.03	46.03	46.03		11.90				1
E911 SERVICE Local Channel - Dedicated - 2-wr Voice Grade - Zone 1 21.94 265.84 46.97 37.63 4.00 11.90 0																	
UNBUNDLE	D NETWORK ELEMENTS - Florida		-	-									Attachment:			bit: B	
---------------	--	-------------	----------	------------	-------	----------	----------	-----------	--------------	--------	---	----------	--	--	----------	----------	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Charge -	
						Rec	Nonrecu		Nonrecurring					Rates(\$)			
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	
	Local Channel - Dedicated - 2-wr Voice Grade - Zone 2					29.62	265.84	46.97	37.63	4.00		11.90					
	Local Channel - Dedicated - 2-wr Voice Grade - Zone 3					57.22	265.84	46.97	37.63	4.00		11.90					
	Interoffice Transport - Dedicated - 2-wr Voice Grade Per Mile				-	0.0091					-						
	Interoffice Transport - Dedicated - 2-wr Voice Grade Per Facility Termination					25.32	47.35	31.78	18.31	7.03		11.90					
	Local Channel - Dedicated - DS1 - Zone 1				-	25.32	216.65	183.54	21.47	19.05		11.90					
	Local Channel - Dedicated - DS1 - Zone 2					47.63	216.65	183.54	21.47	19.05		11.90					
	Local Channel - Dedicated - DS1 - Zone 3					92.01	216.65	183.54	21.47	19.05		11.90					
	Interoffice Transport - Dedicated - DS1 Per Mile					0.1856	210.00	100.01	2	10.00	1	11.00					
	Interoffice Transport - Dedicated - DS1 Per Facility Termination					88.44	105.54	98.47	21.47	19.05		11.90					
CALLING NAM	IE (CNAM) SERVICE																
	CNAM For DB Owners - Service Establishment			OQV			25.35	25.35	19.01	19.01		11.90					
	CNAM For Non DB Owners - Service Establishment			OQV			25.35	25.35	19.01	19.01		11.90					
	CNAM For DB Owners - Service Provisioning With Point Code				1												
	Establishment			OQV			1,592.00	1,177.00	352.36	259.09		11.90					
	CNAM For Non DB Owners - Service Provisioning With Point			001/			540.54	000.00	050.00	050.00		11.00					
	Code Establishment			OQV OQV	-	0.001001	546.51	393.82	358.06	259.09	-	11.90					
	CNAM for DB Owners, Per Query CNAM for Non DB Owners, Per Query				-	0.001024											
LNP Query Ser				UQV	-	0.001024											
LINF QUERY SE	LNP Charge Per query			OQV	-	0.000852											
	LNP Service Establishment Manual			0QV		0.000002	13.83	13.83	12.71	12.71		11.90					
	LNP Service Provisioning with Point Code Establishment						655.50	334.88	297.03	218.40	1	11.90					
OPERATOR C	ALL PROCESSING																
	Oper. Call Processing - Oper. Provided, Per Min Using BST LIDB					1.20											
	Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIDB					1.24											
	Oper. Call Processing - Fully Automated, per Call - Using BST LIDB					0.20											
	Oper. Call Processing - Fully Automated, per Call - Using Foreign LIDB					0.20											
INWARD OPER	RATOR SERVICES																
	Inward Operator Services - Verification, Per Call					1.00											
	Inward Operator Services - Verification and Emergency Interrupt - Per Call					1.95											
	PERATOR CALL PROCESSING																
Facility	y based CLEC																
	Recording of Custom Branded OA Announcement				CBAOS		7,000.00	7,000.00				11.90					
	Loading of Custom Branded OA Announcement per shelf/NAV per OCN				CBAOL		500.00	500.00				11.90					
UNEP		I					7 000 00	7 000 00			L	44.00					
	Recording of Custom Branded OA Announcement		<u> </u>	+	+		7,000.00	7,000.00			ļ	11.90					
	Loading of Custom Branded OA Announcement per shelf/NAV per OCN						500.00	500.00				11.90					
	nding via OLNS for UNEP CLEC	I	<u> </u>				1 000 00	4 000 00				44.00					
	Loading of OA per OCN (Regional)			+	+		1,200.00	1,200.00			ļ	11.90					
	SSISTANCE SERVICES TORY ASSISTANCE ACCESS SERVICE	<u> </u>	+						I								
DIREC	Directory Assistance Access Service Calls, Charge Per Call		+	+	+	0.275					<u> </u>		ł		}	1	
DIREC	TORY ASSISTANCE CALL COMPLETION ACCESS SERVICE (E	DACC)				0.275											
	Directory Assistance Call Completion Access Service (DACC), Per Call Attempt					0.10											
	SSISTANCE SERVICES		+			0.10					<u> </u>						
	TORY ASSISTANCE DATA BASE SERVICE (DADS)	1	1	1	1						1						
	Directory Assistance Data Base Service Charge Per Listing	1	1		1	0.04									İ	İ	
	Directory Assistance Data Base Service, per month	1	1	1	DBSOF	150.00					1	1	1	1	1	1	
	Directory Assistance Data Base Service, per month DIRECTORY ASSISTANCE																

UNBUND	LED	NETWORK ELEMENTS - Florida												Attachment:			bit: B
CATEGOR	Y	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
							Rec	Nonred		Nonrecurring					Rates(\$)		
							Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Fac		Based CLEC															
		Recording and Provisioning of DA Custom Branded															
		Announcement			AMT	CBADA CBADC		6,000.00	6,000.00				11.90				L
		Loading of Custom Branded Announcement per Switch			AMT	CBADC		1,170.00	1,170.00				11.90				<u> </u>
UNI		Recording of DA Custom Branded Announcement						3,000.00	3,000.00				11.90		-	-	
		Loading of DA Custom Branded Announcement per Switch per						3,000.00	3,000.00				11.30				
		OCN						1,170.00	1,170.00				11.90				
Unk		ding via OLNS for UNEP CLEC						.,	.,								
		Loading of DA per OCN (1 OCN per Order)						420.00	420.00				11.90				
		Loading of DA per Switch per OCN						16.00	16.00				11.90				
SELECTIVE																	
		Selective Routing Per Unique Line Class Code Per Request Per															
		Switch				USRCR		93.55	93.55	12.71	12.71		11.90				
VIRTUAL C																	L
		Virtual Collocation - Application Cost			AMTES	EAF		4,122.00	1,249.00				11.90				
		Virtual Collocation - Cable Installation Cost, per cable			AMTES	ESPCX	12.45	965.00					11.90				L
		Virtual Collocation - Floor Space, per sq. ft.			AMTES	ESPVX ESPAX	4.25 6.95										
		Virtual Collocation - Power, per fused amp			AMTFS	ESPAX	6.95										┢────
		Virtual Collocation - Cable Support Structure, per entrance cable			AMTES	ESPSX	13.35										
					UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ, AMTFS, UDL,		10.00										
		Virtual Collocation - 2-wire Cross Connects (loop)			UNCVX, UNCDX, UNCNX	UEAC2	0.0502	11.57	11.57				11.90				
					UEA,UHL,UCL,UDL, AMTFS, UAL, UDN,												
		Virtual Collocation - 4-wire Cross Connects (loop)			UNCVX, UNCDX	UEAC4	0.0502	11.57	11.57				11.90				
		Virtual Collocation - 2-Fiber Cross Connects			AMTFS,UDL12, UDLO3, U1T48, U1T12, U1T03, ULDO3, ULD12, ULD48, UDF	CNC2F	6.71	2,431.00					11.90				
		Virtual Collocation - 4-Fiber Cross Connects			AMTFS,UDL12, UDLO3, U1T48, U1T12, U1T03, ULDO3, ULD12, ULD48, UDF	CNC4F	6.71	2,431.00					11.90				
		Virtual collocation - Special Access & UNE, cross-connect per			USL,ULC,AMTFS, ULR, UXTD1, UNC1X, ULDD1, U1TD1, USLEL,												
		DS1			UNLD1 USL,ULC,AMTFS,U E3, U1TD3, UXTS1, UXTD3, UNC3X, UNCSX, ULDD3,	CNC1X	7.50	155.00	14.00				11.90				
		Virtual collocation - Special Access & UNE, cross-connect per DS3 Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable			U1TS1, ULDS1, UDLSX, UNLD3	CND3X	56.25	151.90	11.83				11.90				ļ
		Support Structure, per linear foot Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable			AMTFS,CLO	VE1CB	0.0028										
		Cable Support Structure, per linear ft			AMTFS, CLO	VE1CD	0.0041										1
		Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure,per cable			AMTES	VE1CC	5.00.11	535.54					11.90				
		Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per cable			AMTFS	VE1CE		535.54					11.90				<u> </u>

UNBUNDLE	D NETWORK ELEMENTS - Florida				-	•						1	Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrec	urring	Nonrecurring	g Disconnect				Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Virtual Collocation Cable Records - per request			AMTFS	VE1BA		1,525.00	1,525.00	267.08	267.08						
	Virtual Collocation Cable Records - VG/DS0 Cable, per cable															
	record			AMTFS	VE1BB		656.50	656.50	379.78	379.78						
	Virtual Collocation Cable Records - VG/DS0 Cable, per each 100 pair			AMTES	VE1BC		9.66	9.66	11.84	11.84						
	Virtual Collocation Cable Records - DS1, per T1TIE			AMTES	VE1BD		4.52	4.52	5.54	5.54						
	Virtual Collocation Cable Records - DS1, per T1TLE			AMTES	VE1BE		15.82	15.82	19.40	19.40						
	Virtual Collocation Cable Records - Fiber Cable, per 99 fiber				12.02		10.02	10102	10.10	10.10						
	records			AMTFS	VE1BF		169.67	169.67	154.89	154.89						
	Virtual collocation - Security Escort - Basic, per quarter hour			AMTFS	SPTBQ		10.89					11.90				
	Virtual collocation - Security Escort - Overtime, per quarter hour		L	AMTFS	SPTOQ		13.64				L	11.90				
					ODTRO		10.10					44.00				
	Virtual collocation - Security Escort - Premium, per quarter hour			AMTFS	SPTPQ		16.40					11.90				
	Virtual Collocation - DS-1/DCS Cross Connects, PER 28 CKTS			AMTES	VE11S	226.39	1,950.00					11.90				
				AMITO	VEITO	220.33	1,350.00					11.30				
	Virtual Collocation - DS-1.DSX Cross Connects, PER 28 CKTS			AMTFS	VE11X	11.51	1.950.00					11.90				
	Virtual Collocation - DS-3/DCS Cross Connects, PER CKT			AMTFS	VE13S	56.97	528.00					11.90				
	Virtual Collocation - DS-3/DSC Cross Connects, PER CKT			AMTFS	VE13X	10.06	528.00					11.90				
	Virtual collocation - Maintenance in CO - Basic, per quarter hour			AMTFS	SPTRE		10.89					11.90				
	Virtual collocation - Maintenance in CO - Overtime, per quarter															
	hour			AMTFS	SPTOE		13.64					11.90				
	Virtual collocation - Maintenance in CO - Premium per quarter hour			AMTES	SPTPE		16.40					11.90				
VIRTUAL COL				AIVITES	SPIPE	-	16.40					11.90				
VINTUAL COL	Virtual Collocation - 2-wire Cross Connect, Exchange Port 2-															
	Wire Analog - Res			UEPSR	VE1R2	0.0502	11.57	11.57				11.90				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-															
	Wire Line Side PBX Trunk - Bus			UEPSP	VE1R2	0.0502	11.57	11.57				11.90				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire															
	Voice Grade PBX Trunk - Res			UEPSE	VE1R2	0.0502	11.57	11.57				11.90				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire															
	Analog Bus			UEPSB	VE1R2	0.0502	11.57	11.57				11.90				
	Virtual Collocation 2-Wire Cross Connect, Exchnage Port 2-Wire ISDN			UEPSX	VE1R2	0.0502	11.57	11.57				11.90				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire			UEPSA	VEIRZ	0.0502	11.57	11.57				11.90				
	ISDN			UEPTX	VE1R2	0.0502	11.57	11.57				11.90				
	Virtual Collocation 4-Wire Cross Connect, Exchange Port 4-Wire	1		1	1	0.0002			1		1		1		1	
	ISDN DS1			UEPEX	VE1R4	0.0502	11.57	11.57				11.90				
VIRTUAL COL																
	Virtual Collocation-2 Wire Cross Connects (Loop) for Line															
	Splitting			UEPSR, UEPSB	VE1LS	0.0502	11.57					11.90				
PHYSICAL CO																
	Physical Collocation-2 Wire Cross Connects (Loop) for Line Splitting			UEPSR, UEPSB	PE1LS	0.0276	8.22	7.22	5.74	4.58		11.90				
	/E CARRIER ROUTING			UEPSR, UEPSB	PEILS	0.0276	8.22	1.22	5.74	4.58		11.90				
	Regional Service Establishment		<u> </u>	SRC	SRCEC		193,444.00		7,737.00		-	11.90				
	End Office Establishment	1		SRC	SRCEO		187.36	187.36	0.69	0.69	1	11.90				
	Query NRC, per query	1		SRC		0.0031868			2.00	2100						
AIN - BELLSO	UTH AIN SMS ACCESS SERVICE															
	AIN SMS Access Service - Service Establishment, Per State,															
	Initial Setup			A1N	CAMSE		43.56	43.56	44.93	44.93		11.90				
			1		0.0055											
	AIN SMS Access Service - Port Connection - Dial/Shared Access AIN SMS Access Service - Port Connection - ISDN Access	l		A1N A1N	CAMDP CAM1P		8.64 8.64	8.64 8.64	10.03 10.03	10.03 10.03		11.90 11.90				
	AIN SMS Access Service - Port Connection - ISDN Access AIN SMS Access Service - User Identification Codes - Per User				CAWIP		8.04	ö.ö4	10.03	10.03		11.90				
				1	1				1		1		1	1	1	1

UNBUNDLE	D NETWORK ELEMENTS - Florida												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st			Incremental Charge -
						_	Nonrec	urring	Nonrecurring	a Disconnect			OSS	Rates(\$)		<u> </u>
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	AIN SMS Access Service - Security Card, Per User ID Code,															
	Initial or Replacement			A1N	CAMRC		75.10	75.10	12.93	12.93		11.90				
	AIN SMS Access Service - Storage, Per Unit (100 Kilobytes)					0.0028										L
	AIN SMS Access Service - Session, Per Minute					0.7809										
	AIN SMS Access Service - Company Performed Session, Per Minute					0.4609										
AIN - BELLSO	JTH AIN TOOLKIT SERVICE					0.4009										<u> </u>
	AIN Toolkit Service - Service Establishment Charge, Per State,				-											ł
	Initial Setup			CAM	BAPSC		43.56	43.56	44.93	44.93		11.90				
	AIN Toolkit Service - Training Session, Per Customer				BAPVX		8,439.00	8,439.00				11.90				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per															
	DN, Term. Attempt		<u> </u>		BAPTT		8.64	8.64	10.03	10.03		11.90				ł
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay				BAPTD		8.64	8.64	10.03	10.03		11.90				1
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per						0.04	0.04	10.03	10.03		11.90				ł
	DN, Off-Hook Immediate				BAPTM		8.64	8.64	10.03	10.03		11.90				1
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per										1		İ	İ	İ	
	DN, 10-Digit PODP				BAPTO		38.06	38.06	15.86	15.86		11.90				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per															
	DN, CDP				BAPTC		38.06	38.06	15.86	15.86		11.90				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN. Feature Code				DADTE		28.06	29.06	15.96	15.96		11.90				
	AIN Toolkit Service - Query Charge, Per Query				BAPTF	0.0535927	38.06	38.06	15.86	15.86		11.90				
	AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit					0.0000027										<u> </u>
	Subscription, Per Node, Per Query					0.0063698										
	AIN Toolkit Service - SCP Storage Charge, Per SMS Access															
	Account, Per 100 Kilobytes					0.06										
	AIN Toolkit Service - Monthly report - Per AIN Toolkit Service			.												
				CAM	BAPMS	8.34	8.64	8.64	6.08	6.08		11.90				ł
	AIN Toolkit Service - Special Study - Per AIN Toolkit Service Subscription			CAM	BAPLS	3.73	9.56	9.56				11.90				
	AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service			CAM	BAFL3	5.75	9.50	9.30				11.90				<u> </u>
	Subscription			CAM	BAPDS	4.73	8.64	8.64	6.08	6.08		11.90				
	AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit			-												
	Service Subscription			CAM	BAPES	0.12	9.56	9.56				11.90				
	(TENDED LINK (EELs)															L
	New Density Zone 1 EELs are available in the following MSA					Atlanta, Ga; Nev	v Orleans, LA,									ļ
	Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem In all states, EEL network elements shown below also apply t					arted to LINE ra	tes A Switch	s le Charge a	nnlies to curre	ntly combined	l facilities co	nverted to	LINEs (Non-re	curring rates	do not apply	
	In All States the EEL network elements apply to ordinarily co													ourning rates	do not apply	<u>/</u>
	VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INT				1		g	,]					
	First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport															
	Combination - Zone 1		1	UNCVX	UEAL2	12.24	127.59	60.54	42.79	2.81	L	11.90				
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed															1
┝──┤───	Transport Combination - Zone 2	<u> </u>	2	UNCVX	UEAL2	17.40	127.59	60.54	42.79	2.81		11.90				<u> </u>
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 3		3	UNCVX	UEAL2	30.87	127.59	60.54	42.79	2.81		11.90				1
	Interoffice Transport - Dedicated - DS1 combination - Per Mile			001/	3-1,1-2	50.07	121.59	00.04	72.13	2.01	1	11.30				
	per month	1		UNC1X	1L5XX	0.1856										1
	Interoffice Transport - Dedicated - DS1 combination - Facility		I													[
	Termination per month			UNC1X	U1TF1	88.44	174.46	122.46	45.61	17.95	L	11.90				
	DS1 Channelization System Per Month			UNC1X	MQ1	146.77	51.83	10.75	0.71	1.01		11.90				<u> </u>
┝──┼──	Voice Grade COCI - DS1 To Ds0 Interface - Per Month			UNCVX	1D1VG	1.38	12.16	8.77	6.71	4.84		11.90				
	Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL2	12.24	127.59	60.54	42.79	2.81		11.90				1
	Each Additional 2-Wire VG Loop(SL2) in the same DS1			0.101/1	J L / 12	12.24	121.09	00.04	72.13	2.01	1	11.30				
	Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL2	17.40	127.59	60.54	42.79	2.81		11.90				1
	Each Additional 2-Wire VG Loop(SL2) in the same DS1		I													[
	Interoffice Transport Combination - Zone 3	1	3	UNCVX	UEAL2	30.87	127.59	60.54	42.79	2.81		11.90				L

UNBUNDLE	ED NETWORK ELEMENTS - Florida												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
	Vaine Crade COCL DS1 to DS0 Channel System combination						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Voice Grade COCI - DS1 to DS0 Channel System combination - per month			UNCVX	1D1VG	1.38	12.16	8.77	6.71	4.84		11.90				1 1
	Nonrecurring Currently Combined Network Elements Switch -As-				IDIVO	1.00	12.10	0.77	0.71	4.04		11.30				
	Is Charge			UNC1X	UNCCC		8.98	8.98	8.98	8.98		11.90				
4-WIR	E VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INT	EROFF	ICE TR	ANSPORT (EEL)												ļ!
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL4	18.89	127.59	60.54	42.79	2.81		11.90				1 1
├ ──┤───	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice		1	UNCVX	UEAL4	18.89	127.59	60.54	42.79	2.81		11.90				i
	Transport Combination - Zone 2		2	UNCVX	UEAL4	26.84	127.59	60.54	42.79	2.81		11.90				i
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice															1
	Transport Combination - Zone 3		3	UNCVX	UEAL4	47.62	127.59	60.54	42.79	2.81		11.90				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile					0.4050										i
	Per Month Interoffice Transport - Dedicated - DS1 - Facility Termination Per			UNC1X	1L5XX	0.1856										
	Month			UNC1X	U1TF1	88.44	174.46	122.46	45.61	17.95		11.90				i
	Channelization - Channel System DS1 to DS0 combination Per															1
	Month			UNC1X	MQ1	146.77	51.83	10.75				11.90				ļ
	Voice Grade COCI - DS1 to DS0 Channel System combination -				104/0	4.00	10.10	0.77	0.74			44.00				i
	per month Additional 4-Wire Analog Voice Grade Loop in same DS1			UNCVX	1D1VG	1.38	12.16	8.77	6.71	4.84		11.90				
	Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL4	18.89	127.59	60.54	42.79	2.81		11.90				1
	Additional 4-Wire Analog Voice Grade Loop in same DS1				02/21	10.00	121100	00.01	12.110	2.01						l .
	Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL4	26.84	127.59	60.54	42.79	2.81		11.90				1
	Additional 4-Wire Analog Voice Grade Loop in same DS1															l I
	Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL4	47.62	127.59	60.54	42.79	2.81		11.90				1
	Voice Grade COCI - DS1 to DS0 Channel System combination - per month			UNCVX	1D1VG	1.38	12.16	8.77	6.71	4.84		11.90				i
	Nonrecurring Currently Combined Network Elements Switch -As-				10110		12.10	0.11	0.71							1
	Is Charge			UNC1X	UNCCC		8.98	8.98	8.98	8.98		11.90				1
4-WIR	E 56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1	INTERC	FFICE	TRANSPORT (EEL))											4
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1		1	UNCDX	UDL56	22.20	127.59	60.54	42.79	2.81		11.90				i
	First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice		1	UNCDA	UDL56	22.20	127.59	60.54	42.79	2.01		11.90				i
	Transport Combination - Zone 2		2	UNCDX	UDL56	31.56	127.59	60.54	42.79	2.81		11.90				i
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice															[
	Transport Combination - Zone 3		3	UNCDX	UDL56	55.99	127.59	60.54	42.79	2.81		11.90				I
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month			UNC1X	1L5XX	0.1856										i
	Interoffice Transport - Dedicated - DS1 - combination Facility			UNCIX	ILSAA	0.1656										i
	Termination Per Month			UNC1X	U1TF1	88.44	174.46	122.46	45.61	17.95		11.90				1
	Channelization - Channel System DS1 to DS0 combination Per															ſ
	Month			UNC1X	MQ1	146.77	51.83	10.75				11.90				l
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kbs)			UNCDX	1D1DD	2.10	12.16	8.77	6.71	4.84		11.90				i
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1			UNCDX	סטוטו	2.10	12.10	8.77	0.71	4.84		11.90				i
	Interoffice Transport Combination - Zone 1		1	UNCDX	UDL56	22.20	127.59	60.54	42.79	2.81		11.90				i
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1															ſ
	Interoffice Transport Combination - Zone 2		2	UNCDX	UDL56	31.56	127.59	60.54	42.79	2.81		11.90				ļ
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1		3			FF 00	407 50	00 F 4	40.70	0.04		44.00				
├──┼ ──	Interoffice Transport Combination - Zone 3 OCU-DP COCI (data) - DS1 to DS0 Channel System -		3	UNCDX	UDL56	55.99	127.59	60.54	42.79	2.81	<u> </u>	11.90				l
	combination per month (2.4-64kbs)			UNCDX	1D1DD	2.10	12.16	8.77	6.71	4.84		11.90				1
	Nonrecurring Currently Combined Network Elements Switch -As-										1					
	Is Charge			UNC1X	UNCCC		8.98	8.98	8.98	8.98	L	11.90				ļ
4-WIR	E 64 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1	INTERC	OFFICE	TRANSPORT (EEL))						ļ					
1 1	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1		1	UNCDX	UDL64	22.20	127.59	60.54	42.79	2.81		11.90				1
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice		· ·		5DL04	22.20	121.39	00.04	72.13	2.01	<u> </u>	11.90				
	Transport Combination - Zone 2		2	UNCDX	UDL64	31.56	127.59	60.54	42.79	2.81		11.90				I '

UNBUNDLE	D NETWORK ELEMENTS - Florida			1							-	-	Attachment:			ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec	urring	Nonrecurring	g Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice		~			55.00	107.50	00.54	40.70	0.04		44.00				
	Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile		3	UNCDX	UDL64	55.99	127.59	60.54	42.79	2.81		11.90				
	Per Month			UNC1X	1L5XX	0.1856										
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Month			UNC1X	U1TF1	88.44	174.46	122.46	45.61	17.95		11.90				
	Channelization - Channel System DS1 to DS0 combination Per Month			UNC1X	MQ1	146.77	51.83	10.75				11.90				
	OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs)			UNCDX	1D1DD	2.10	12.16	8.77	6.71	4.84		11.90				
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 1		1	UNCDX	UDL64	22.20	127.59	60.54	42.79	2.81		11.90				
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 2		2	UNCDX	UDL64	31.56	127.59	60.54	42.79	2.81		11.90				
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1		3	UNCDX	UDL64	55.99	127.59	60.54	42.79	2.81		11.90				<u> </u>
	Interoffice Transport Combination - Zone 3 OCU-DP COCI (data) - DS1 to DS0 Channel System		3													
	combination - per month (2.4-64kbs) Nonrecurring Currently Combined Network Elements Switch -As-			UNCDX	1D1DD	2.10	12.16	8.77	6.71	4.84		11.90				
4.14/15		DOFE		UNC1X	UNCCC		8.98	8.98	8.98	8.98		11.90				
4-WIR	E DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTE 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice	ROFFIC		ANSPORT (EEL)												
	Transport - Zone 1		1	UNC1X	USLXX	70.74	217.75	121.62	51.44	14.45		11.90				
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2		2	UNC1X	USLXX	100.54	217.75	121.62	51.44	14.45		11.90				
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3		3	UNC1X	USLXX	178.39	217.75	121.62	51.44	14.45		11.90				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month			UNC1X	1L5XX	0.1856										
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Month			UNC1X	U1TF1	88.44	174.46	122.46	45.61	17.95		11.90				
	Nonrecurring Currently Combined Network Elements Switch -As-				UNCCC	00.44	8.98	8.98	8.98	8.98						
4-WIR	IS Charge E DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTE	ROFFIC	CF TR/	UNC1X	UNCCC		8.98	8.98	8.98	8.98		11.90				
	First DS1Loop in DS3 Interoffice Transport Combination - Zone		1		USLXX	70.74	217.75	121.62	51.44	14.45		11.90				
	First DS1Loop in DS3 Interoffice Transport Combination - Zone		_													<u> </u>
	2 First DS1Loop in DS3 Interoffice Transport Combination - Zone		2	UNC1X	USLXX	100.54	217.75	121.62	51.44	14.45		11.90				<u> </u>
	3 Interoffice Transport - Dedicated - DS3 combination - Per Mile		3	UNC1X	USLXX	178.39	217.75	121.62	51.44	14.45		11.90				
	Per Month			UNC3X	1L5XX	3.87										
	Interoffice Transport - Dedicated - DS3 - Facility Termination per month			UNC3X	U1TF3	1,071.00	314.45	130.88	38.60	18.23		11.90				
	DS3 to DS1 Channel System combination per month			UNC3X UNC3X	MQ3	211.19	115.60	59.93	5.45	0.00		11.90				
	DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	13.76	12.16	8.77	6.71	4.84		11.90				
	Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 1		1	UNC1X	USLXX	70.74	217.75	121.62	51.44	14.45		11.90				
	Additional DS1Loop in DS3 Interoffice Transport Combination -		2	UNC1X	USLXX	100.54	217.75	121.62	51.44	14.45		11.90				<u> </u>
	Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination -															
	Zone 3 DS3 Interface Unit (DS1 COCI) combination per month		3	UNC1X UNC1X	USLXX UC1D1	178.39 13.76	217.75 12.16	121.62 8.77	51.44 6.71	14.45 4.84		11.90 11.90				
	Nonrecurring Currently Combined Network Elements Switch -As-					13.70										<u> </u>
2 14/15/	Is Charge E VOICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INT	EDOEF			UNCCC		8.98	8.98	8.98	8.98		11.90				───
2-vVIR	2-WireVG Loop used with 2-wire VG Interoffice Transport	LKOPP		ANGFORI (EEL)	+ +											
	Combination - Zone 1		1	UNCVX	UEAL2	12.24	127.59	60.54	42.79	2.81		11.90				

UNBUNDLE	D NETWORK ELEMENTS - Florida				r	1							Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
	2-WireVG Loop used with 2-wire VG Interoffice Transport						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Combination - Zone 2		2	UNCVX	UEAL2	17.40	127.59	60.54	42.79	2.81		11.90				
	2-WireVG Loop used with 2-wire VG Interoffice Transport		-	onom	0L/1LL		121100	00.01	12.110	2.01						
	Combination - Zone 3		3	UNCVX	UEAL2	30.87	127.59	60.54	42.79	2.81		11.90				
	Interoffice Transport - Dedicated - 2-wire VG combination - Per															
	Mile Per Month Interoffice Transport - Dedicated - 2- Wire Voice Grade			UNCVX	1L5XX	0.0091										
	combination - Facility Termination per month			UNCVX	U1TV2	25.32	94.70	52.59	50.49	21.53		11.90				
	Nonrecurring Currently Combined Network Elements Switch -As-			onovix	01112	20.02	54.10	02.00	00.40	21.00		11.00				
	Is Charge			UNCVX	UNCCC		8.98	8.98	8.98	8.98		11.90				
4-WIRI	E VOICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INT	EROFF	ICE TR	ANSPORT (EEL)												
	4-WireVG Loop used with 4-wire VG Interoffice Transport					10.00	107 50	00 F f	10.70			44.00				
	Combination - Zone 1 4-WireVG Loop used with 4-wire VG Interoffice Transport		1	UNCVX	UEAL4	18.89	127.59	60.54	42.79	2.81		11.90				
	Combination - Zone 2		2	UNCVX	UEAL4	26.84	127.59	60.54	42.79	2.81		11.90				
	4-WireVG Loop used with 4-wire VG Interoffice Transport		-	5	32,127	20.04	121.55	00.04	72.13	2.01		11.30				
	Combination - Zone 3		3	UNCVX	UEAL4	47.62	127.59	60.54	42.79	2.81		11.90				
	Interoffice Transport - Dedicated - 4-wire VG combination - Per															
	Mile Per Month			UNCVX	1L5XX	0.0091										
	Interoffice Transport - Dedicated - 4- Wire Voice Grade combination - Facility Termination per month			UNCVX	U1TV4	22.58	94.70	52.59	50.49	21.53		11.90				
	Nonrecurring Currently Combined Network Elements Switch -As-			UNCVA	01174	22.30	94.70	52.59	50.49	21.55		11.90				
	Is Charge			UNCVX	UNCCC		8.98	8.98	8.98	8.98		11.90				
DS3 D	GITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFIC	E TRA	SPOR	T (EEL)	1											
	High Capacity Unbundled Local Loop - DS3 combination - Per															
	Mile per month			UNC3X	1L5ND	10.92										
	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month			UNC3X	UE3PX	386.88	249.97	162.05	67.10	26.82		11.90				
	Interoffice Transport - Dedicated - DS3 - Per Mile per month			UNC3X	1L5XX	3.87	249.97	102.05	07.10	20.02		11.90				
	Interoffice Transport - Dedicated - DS3 combination - Facility															
	Termination per per month			UNC3X	U1TF3	1,071.00	314.45	130.88	38.60	18.23		11.90				
	Nonrecurring Currently Combined Network Elements Switch -As-															
ete4	Is Charge DIGITAL EXTENDED LOOP WITH DEDICATED STS1 INTEROFI			UNC3X	UNCCC		8.98	8.98	8.98	8.98		11.90				
31311	High Capacity Unbundled Local Loop - STS1 combination - Per		ANGE													
	Mile per month			UNCSX	1L5ND	10.92										
	High Capacity Unbundled Local Loop - STS1 combination -															
	Facility Termination per month			UNCSX	UDLS1	426.60	249.97	162.05	67.10	26.82		11.90				
	Interoffice Transport - Dedicated - STS1 combination - Per Mile			INCOV	41 5307	0.07										
	per month Interoffice Transport - Dedicated - STS1 combination - Facility	-		UNCSX	1L5XX	3.87										-
	Termination per month			UNCSX	U1TFS	1,056.00	314.45	130.88	38.60	18.23		11.90				
	Nonrecurring Currently Combined Network Elements Switch -As-					.,										
	Is Charge			UNCSX	UNCCC		8.98	8.98	8.98	8.98		11.90				
2-WIRI	ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPOR	T (EEL)													
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1		1	UNCNX	U1L2X	19.28	127.59	60.60	42.79	2.81		11.90				
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination				UILZA	19.28	127.39	00.00	42.19	2.01		11.90				
	Transport - Zone 2		2	UNCNX	U1L2X	27.40	127.59	60.60	42.79	2.81		11.90				
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination															
	Transport - Zone 3	<u> </u>	3	UNCNX	U1L2X	48.62	127.59	60.60	42.79	2.81		11.90				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile			UNC1X	1L5XX	0.1856										
	Interoffice Transport - Dedicated - DS1 combintion - Facility Termination per month			UNC1X	U1TF1	88.44	174.46	122.46	45.61	17.95		11.90				
	Channelization - Channel System DS1 to DS0 combination -					00.+4	174.40	122.40	-0.01	17.55		11.00				-
	per month			UNC1X	MQ1	146.77	51.83	10.75				11.90				
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System															
	combination - per month			UNCNX	UC1CA	3.66	12.16	8.77	6.71	4.84		11.90				

UNBUNDLE	ED NETWORK ELEMENTS - Florida	1	1		, , ,							• · · ·	Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonred		Nonrecurring					Rates(\$)		
					-		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1		1	UNCNX	U1L2X	19.28	127.59	60.60	42.79	2.81		11.90			1	
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2		2	UNCNX	U1L2X	27.40	127.59	60.60	42.79	2.81		11.90				
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3		3	UNCNX	U1L2X	48.62	127.59	60.60	42.79	2.81		11.90				
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System															
	combintaion- per month Nonrecurring Currently Combined Network Elements Switch -As-			UNCNX	UC1CA	3.66	12.16	8.77	6.71	4.84		11.90				<u> </u>
	Is Charge			UNC1X	UNCCC		8.98	8.98	8.98	8.98		11.90			ļ'	Ļ
4-WIR	E DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 IN First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1	TEROF	FICE T			70.74	047.75	404.00	54.44	44.45		44.00			1	
	First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2		2	UNC1X UNC1X	USLXX	70.74 100.54	217.75	121.62	51.44 51.44	14.45		11.90 11.90				
	First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3		3	UNC1X	USLXX	178.39	217.75	121.62	51.44	14.45		11.90				
	Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month			UNCSX	1L5XX	3.87										
	Interoffice Transport - Dedicated - STS1 combination - Facility Termination			UNCSX	U1TFS	1,056.00	314.45	130.88	38.60	18.23		11.90				
	STS1 to DS1 Channel System conbination per month			UNCSX	MQ3	211.19	10.10	3.39	0.74						ļ'	Ļ
	DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination -			UNC1X	UC1D1	13.76	12.16	8.77	6.71	4.84		11.90			Į′	───
	Zone 1 Additional DS1Loop in STS1 Interoffice Transport Combination -		1	UNC1X	USLXX	70.74	217.75	121.62	51.44	14.45		11.90				
	Zone 2 Additional DS1Loop in STS1 Interoffice Transport Combination - Additional DS1Loop in STS1 Interoffice Transport Combination -		2	UNC1X	USLXX	100.54	217.75	121.62	51.44	14.45		11.90				
	Zone 3		3	UNC1X	USLXX	178.39	217.75	121.62	51.44	14.45		11.90			1	
	DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	13.76	12.16	8.77	6.71	4.84		11.90				
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNCSX	UNCCC		8.98	8.98	8.98	8.98		11.90				
4-WIR	E 56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTERO	FFICE 1	TRANS	PORT (EEL)											ļ!	ļ
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 1		1	UNCDX	UDL56	22.20	127.59	60.54	42.79	2.81		11.90				
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 2 4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport		2	UNCDX	UDL56	31.56	127.59	60.54	42.79	2.81		11.90				
	Combination - Zone 3		3	UNCDX	UDL56	55.99	127.59	60.54	42.79	2.81		11.90			1 1	
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Per Mile			UNCDX	1L5XX	0.0091										
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Termination			UNCDX	U1TD5	18.44	94.70	52.59	50.49	21.53		11.90				
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge				UNCCC		8.98	8.98	8.98	8.98		11.90				
4-WIR	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport	FFICE	RANS	PORT (EEL)											┝───┘	<u> </u>
	Combination - Zone 1 4-wire 64 kbps Loop/4-wire 64 kbps interoffice Transport 4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport		1	UNCDX	UDL64	22.20	127.59	60.54	42.79	2.81		11.90			ļ!	
	Combination - Zone 2 4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport		2	UNCDX	UDL64	31.56	127.59	60.54	42.79	2.81		11.90				
	Combination - Zone 3 Interoffice Transport - Dedicated - 4-wire 64 kbps combination -		3	UNCDX	UDL64	55.99	127.59	60.54	42.79	2.81		11.90				
	Per Mile		<u> </u>	UNCDX	1L5XX	0.0091										
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination -															
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility Termination Nonrecurring Currently Combined Network Elements Switch -As- Is Charge				U1TD6 UNCCC	18.44	94.70	52.59 8.98	50.49 8.98	21.53 8.98		11.90 11.90				

UNBUNDL	ED NETWORK ELEMENTS - Florida												Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		-
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	n used as a part of a currently combined facility, the non-recurr															L
	n used as ordinarily combined network elements in All States, th					h As Is Charge o	loes not.									L
Nonr	ecurring Currently Combined Network Elements "Switch As Is" Nonrecurring Currently Combined Network Elements Switch -As-	Charge	(One a	applies to each com	bination)											 '
	Is Charge - 2 wire/4-Wire VG			UNCVX	UNCCC		8.98	8.98	8.98	8.98		11.90				
	Nonrecurring Currently Combined Network Elements Switch -As-			UNOVA	UNCCC		0.30	0.30	0.30	0.50		11.50				
	Is Charge - 56/64 kbps			UNCDX	UNCCC		8.98	8.98	8.98	8.98		11.90				
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - DS1			UNC1X	UNCCC		8.98	8.98	8.98	8.98		11.90				
	Nonrecurring Currently Combined Network Elements Switch -As-				UNCCC		0.30	0.30	0.30	0.50		11.50				
	Is Charge - DS3			UNC3X	UNCCC		8.98	8.98	8.98	8.98		11.90				
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - STS1			UNCSX	UNCCC		8.98	8.98	8.98	8.98		11.90				1
NOTE	E: Local Channel - Dedicated Transport - minimum billing period	l - Relo	W DS3			r months	0.90	0.90	0.98	0.98	<u> </u>	11.90		-	ł	ł
	Local Channel - Dedicated - 2-Wire Voice Grade Zone 1			UNCVX	ULDV2	19.66	265.84	46.97	37.63	4.00		11.90				-
	Local Channel - Dedicated - 2-Wire Voice Grade Zone 1		2	UNCVX	ULDV2	27.94	265.84	46.97	37.63	4.00		11.90				
	Local Channel - Dedicated - 2-Wire Voice Grade Zone 2		3	UNCXV	ULDV2	49.58	265.84	46.97	37.63	4.00	1	11.90				
	Local Channel - Dedicated - 4-Wire Voice Grade Zone 1		1	UNCVX	ULDV4	20.45	266.54	47.67	44.22	5.33		11.90				
	Local Channel - Dedicated - 4-Wire Voice Grade Zone 2		2	UNCVX	ULDV4	29.06	266.54	47.67	44.22	5.33		11.90				
	Local Channel - Dedicated - 4-Wire Voice Grade Zone3		3	UNCXV	ULDV4	51.56	266.54	47.67	44.22	5.33		11.90				
	Local Channel - Dedicated - DS1 per month Zone 1		1	UNC1X	ULDF1	36.49	216.65	183.54	24.30	16.95		11.90				
	Local Channel - Dedicated -DS1 Per Month Zone 2		2	UNC1X	ULDF1	51.85	216.65	183.54	24.30	16.95		11.90				
	Local Channel - Dedicated - DS1- Per Month Zone 3		3	UNC1X	ULDF1	92.00	216.65	183.54	24.30	16.95		11.90				
	Local Channel - Dedicated - DS3 - Per Mile per month			UNC3X	1L5NC	8.50										
	Local Channel - Dedicated - DS3 - Facility Termination			UNC3X	ULDF3	531.91	556.37	343.01	139.13	96.84		11.90				<u> </u>
	Local Channel - Dedicated - STS-1- Per Mile per month			UNCSX	1L5NC ULDFS	8.50 540.69	556.37	343.01	139.13	96.84		11.90				
Ontio	Local Channel - Dedicated - STS-1 - Facility Termination onal Features & Functions:			UNCSX	ULDF5	540.69	556.37	343.01	139.13	96.84		11.90				<u> </u>
	TIPLEXERS															
	Channelization - DS1 to DS0 Channel System			UXTD1	MQ1	146.77	101.42	71.62	11.09	10.49		11.90				
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per			O/TD1	NIGE 1	140.17	101.42	11.02	11.00	10.40		11.00				
	month (2.4-64kbs)			UDL	1D1DD	2.10	10.07	7.08				11.90				
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per															
	month			UDN	UC1CA	3.66	10.07	7.08				11.90				
	Voice Grade COCI - DS1 to DS0 Channel System - per month			UEA	1D1VG	1.38	10.07	7.08				11.90				L
	DS3 to DS1 Channel System per month			UXTD3	MQ3	211.19	199.28	118.64	40.34	39.07		11.90				L
	STS1 to DS1 Channel System per month DS3 Interface Unit (DS1 COCI) used with Loop per month			UXTS1 USL	MQ3 UC1D1	211.19 13.76	199.28 10.07	118.64 7.08	40.34	39.07		11.90 11.90				
	DS3 Interface Unit (DS1 COCI) used with Loop per month DS3 Interface Unit (DS1 COCI) used with Local Channel per			USL	00101	13.70	10.07	7.00				11.90				<u> </u>
	month			ULDD1	UC1D1	13.76	10.07	7.08				11.90				
	DS3 Interface Unit (DS1 COCI) used with Interoffice Channel															1
	per month		1	U1TD1	UC1D1	13.76	10.07	7.08				11.90				1
Sub-	Loop Feeder				1											
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Statewide		SW	UNC1X	USBFG											
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 1		1	UNC1X	USBFG	42.59	133.77	78.02	85.16	21.21						
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 2		2	UNC1X	USBFG	60.53	133.77	78.02	85.16	21.21	ļ					
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 3		3	UNC1X	USBFG	107.39	133.77	78.02	85.16	21.21		L				
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 4 D LOCAL EXCHANGE SWITCHING(PORTS)		4	UNC1X	USBFG											ł
	ange Ports				+						<u> </u>			-	ł	ł
	E: Although the Port Rate includes all available features in GA, F	(Y, I A	& TN. +	he desired features	will need to b	be ordered usin	a retail USOCs									
	RE VOICE GRADE LINE PORT RATES (RES)	.,_,											i			1
	Exchange Ports - 2-Wire Analog Line Port- Res.			UEPSR	UEPRL	1.40	3.74	3.63	1.88	1.80		11.90	i			<u> </u>
	- U										İ					
	Exchange Ports - 2-Wire Analog Line Port with Caller ID - Res.		L	UEPSR	UEPRC	1.40	3.74	3.63	1.88	1.80	ļ	11.90				ļ
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Res.			UEPSR	UEPRO	1.40	3.74	3.63	1.88	1.80		11.90				1
	Exchange Ports - 2-Wire VG unbundled Florida area calling with															
	Caller ID - Res.		1	UEPSR	UEPAF	1.40	3.74	3.63	1.88	1.80		11.90				1

Characterization Control	JNBUNDLED	NETWORK ELEMENTS - Florida												Attachment:			bit: B
Extra Point Control First Add?	ATEGORY	RATE ELEMENTS		Zone	BCS	USOC			RATES(\$)			Submitted Elec	Submitted Manually	Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
is primage Points - 2Win VC subunded Floxia Brandmon Amp UPPS VPR VPR VPR <							Rec									•	
Caling Pin. Hundle Galer Docadality ULPPR ULPPR ULPPR ULPPR ULPPR ULPR <th></th> <th></th> <th></th> <th></th> <th></th> <th></th> <th>Nec</th> <th>First</th> <th>Add'l</th> <th>First</th> <th>Add'l</th> <th>SOMEC</th> <th>SOMAN</th> <th>SOMAN</th> <th>SOMAN</th> <th>SOMAN</th> <th>SOMAN</th>							Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Exhappe Point CEPSR UEPA 1.4 3.7 3.85 1.86 1.80 1.80 1.80 Exhappe Point Setting port for use with CREX7 and Caller D capability UEPSR UEPA 1.60 3.74 3.85 1.88 1.80 1.10 <td< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>1</td></td<>																	1
deling port to warch CREX and Caller D UEPSR UEPA1 1.0 3.24 3.63 1.88 1.80 11.00 Beckings Ports - XVW v Unducide serviced UEPSR UEPSR UEPAR 1.40 3.74 3.63 1.88 1.80 11.00 Beckings Ports - XVW v Unducide serviced UEPSR UEPAR 1.40 3.74 3.63 1.88 1.80 11.00 Beckings Ports - XVW v Unducide serviced UEPSR UEPSR 1.40 3.74 3.63 1.88 1.80 11.00 SWM vs cons unbundled for Urage Line Port whord Caller D UEPSR UEPSR 2.28 0.00 0.00 1.10 PARAMANES Vortunit Feature UEPSR UEPSR 1.40 3.74 3.63 1.88 1.80 1.10 Balance Ports - XWM vs Constructure Port whord Caller D - Mala Vs Constructure Port whord Caller D - Mala Vs Constructure Port whord Caller D - Mala Vs Constructure Port Ws Constructure Port Ws Constructure Port Ws Constructure Port Ws Constructure Port Ws Constructure Port Ws Constructure Port Ws Constructure Port Ws Constructure Port Ws Constructure Port Ws Const					UEPSR	UEPA9	1.40	3.74	3.63	1.88	1.80		11.90	-			ł
daring port to use with CREX7, whence Caler ID UPERR UPERR UPAR 1.0 3.74 3.63 1.88 1.60 11.00 Exhange Potra - XWE votubed IES, for Votand Law Longe Line Port without Caler ID UPERR UPAR 1.40 3.74 3.63 1.88 1.80 11.00 11.00 Capability UPERR UPERR UPERR 1.40 3.74 3.63 1.88 1.80 11.00	d	dialing port for use with CREX7 and Caller ID			UEPSR	UEPA1	1.40	3.74	3.63	1.88	1.80		11.90				
Exchange Pots - 2-Wire VG unbundled roe, low usage ine port upper value upp							1 40	2.74	2.62	1 00	1 90		11.00				i
which caller (DLUM) UEPSR UEPAR 1.40 3.74 3.68 1.88 1.80 11.00 Capability UEPSR UEPSR UEPSR 0.83 7.44 3.68 1.88 1.80 11.00 11.00 FAILURES UEPSR UEPSR UEPSR 0.000 0.00 0.00 1.88 1.80 11.00 1.00 FAILURES UEPSR UEPSR UEPSR UEPSR 0.00 0.00 0.00 1.00 1.100 1.00<					ULFSK	ULFAO	1.40	3.74	3.03	1.00	1.00		11.90				
2-Wite vice influence UPPS					UEPSR	UEPAP	1.40	3.74	3.63	1.88	1.80		11.90				i
Capabily UEPS UEPS OPEN 3.74 3.63 1.88 1.80 1.100 FRATM UEPS UEPS UEPS 0.00 0.00 0.00 <																	l .
PEATURES DEPX					UEPSR			3.74	3.63	1.88	1.80		11.90				1
MI Available Vertical Features UEPSR UEPVF 2.28 0.00 0.00 11.90 2 WIRE VOIC GRADE LLER PORT RATES (BLM) UEPSR UEPSR UEPSR UEPSR 1.40 3.74 3.63 1.88 1.40 11.90 But Anage Ports - 2-Wire Avails (Line Port without Caller Port with Callers Edds ID - Sus. UEPSR UEPSR UEPSR UEPSR 1.40 3.74 3.63 1.88 1.80 11.90 Exchange Ports - 2-Wire Avails Avails (Mark Origing only - Bus. UEPSR UEPSR UEPSR UEPSR UEPSR 1.40 3.74 3.63 1.88 1.80 11.90 Exchange Ports - 2-Wire Avails Avails (Mark Origing only - Bus. UEPSR UEPSR UEPSR 0.00 0.00 1.88 1.80 11.90 11.90 Caller D - Bus UEPSR UEPSR UEPSR UEPSR 0.00 0.00 0.00 1.80 11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90<					UEPSR	USASC	0.00	0.00	0.00				11.90				
2-WRE VOICE GRADE LINE PORT RAINING Line PORT WRING Caller D Desc.																	1
Exchange Ports - 2-Wire Analog Lue Port without Caller ID - Bus Analog Ports - 2-Wire Valuational Lue Port with Caller ID- Exchange Ports - 2-Wire Valuational Lue Port with Caller ID- Bus Analog Ports - 2-Wire Valuational Lue Port with Caller ID- Bus Analog Ports - 2-Wire Valuational Lue Port with Caller ID- Bus Analog Ports - 2-Wire Valuational Lue Port without Caller ID- Bus Analog Ports - 2-Wire Valuational Lue Port without Caller ID- Bus Analog Ports - 2-Wire Valuational Lue Port without Caller ID- Bus Analog Ports - 2-Wire Valuational Lue Port without Caller ID- Bus Analog Ports - 2-Wire Valuational Lue Port without Caller ID- Bus Analog Ports - 2-Wire Valuational Lue Port without Caller ID- Bus Analog Ports - 2-Wire Valuational Lue Port Wire Valuational Lue Ports Caller ID - Bus Analog Valuational Lue Port Waluational Lue Ports Caller ID- Bus Analog Valuational Lue Port Wire Valuational Lue Ports Caller ID- Bus Analog Valuational Lue Port Waluational Lue Ports Caller ID- Bus Analog Valuational Lue Port Waluational Lue Ports Caller ID- Bus Analog Valuational Lue Port Waluational Lue Ports Port Valuational Lue Port Port Port Port Port Port Port Port					UEPSR	UEPVF	2.26	0.00	0.00				11.90				
Bis UEPS UEPS UEPS UEPS State 1.80 1.80 1.80 1.80 Exchange Ports - 2Wire VG unbundled Incoming only or with unbundled port with Caller L-840 D-Bus. UEPSB UEPSB UEPSB UEPSB UEPSB 1.80 1.100 1.80 1.100 1.80 1.100 1.80 1.100 1.80 1.100 1.80 1.100																	
Impundied port with Calier-EdeN LD-BOS. UEPSB	E	Bus			UEPSB	UEPBL	1.40	3.74	3.63	1.88	1.80		11.90				
Exchange Ports - 2-Wire Vier Wortwick UEPSB UEPSB UEPSB 1.40 3.74 3.63 1.88 1.80 1.10 Exchange Ports - 2-Wire Vier Wortwick UEPSB UEPSB UEPSB 1.40 3.74 3.63 1.88 1.80 1.100 1.100 2/Wire vice unbundled incoming Only Port without Caller ID UEPSB UEPSB UEPSB UEPSB 3.63 1.88 1.80 1.100 1.100 1.0							4.40	0.74	0.00	4.00	4.00		44.00				i
Extransp Ports - 2-Wire VG unbundled incoming only port with Caller ID - Bus UEPSB UEPSC <	U	unbundled port with Caller+E484 ID - Bus.			UEPSB	UEPBC	1.40	3.74	3.63	1.88	1.80		11.90	-			ł
Caller ID- Bus UEPS UEPS UEPS UEPS 1.40 3.74 3.63 1.88 1.80 11.90 Capability UEPS UEPS UEPS 0.00 <td></td> <td></td> <td></td> <td></td> <td>UEPSB</td> <td>UEPBO</td> <td>1.40</td> <td>3.74</td> <td>3.63</td> <td>1.88</td> <td>1.80</td> <td></td> <td>11.90</td> <td></td> <td></td> <td></td> <td> </td>					UEPSB	UEPBO	1.40	3.74	3.63	1.88	1.80		11.90				
2-Wire volce unbundled hooming Only Port without Caller ID UEPSE							4 40	0.74	2.02	4.00	4.00		44.00				i
Capability UEPSB UEPSB UAB 3.74 3.63 1.88 1.80 11.90 FEATURES UEPSB USASC 0.00					UEPSB	UEPB1	1.40	3.74	3.63	1.88	1.80		11.90				l
Bubsequent Activity UEPSB USASC 0.00					LIEDSB		1.40	3 74	3.63	1.88	1.80		11 00				i
FEATURES Image: Control of Contrel of Control of Control of										1.00	1.00						i
Nit Natisable Vertical Features UEPSB UEPVF 2.28 0.00 0.00 11.90 11.90 EXCHANCE PORT RATES (DID & PRX) Image: Constraint of the constraint of					ULF 3D	USAGC	0.00	0.00	0.00				11.90				
EXCHANCE PORT RATES (DID & PBX) Image: https://wite VG <thttps: thr="" vg<="" wite=""> Image: https://wite VG</thttps:>					UEPSB	UEPVF	2.26	0.00	0.00				11.90				i
2-Wire VG Line Side Unbundied 2-Way PBX Trunk - Bus UEPSP UEPSP UEPSP UEPSP 11.00 30.06 18.18 12.35 0.7187 11.90 2-Wire VG Line Side Unbundied Duward PBX Trunk - Bus UEPSP UEPSP UEPP1 1.40 39.06 18.18 12.35 0.7187 11.90 2-Wire VG Line Side Unbundied PLX Trunk - Bus UEPSP UEPSP UEPLD 1.40 39.06 18.18 12.35 0.7187 11.90 2-Wire Voice Unbundied PLX DTerminal PDX Trunk - Bus UEPSP UEPLD 1.40 39.06 18.18 12.35 0.7187 11.90 2-Wire Voice Unbundied PLX TomTerminal PDX Trunk - Bus UEPSP UEPXA 1.40 39.06 18.18 12.35 0.7187 11.90 2-Wire Voice Unbundied PLX LD DDTerminal Nethel Ports UEPSP UEPXA 1.40 39.06 18.18 12.35 0.7187 11.90 2-Wire Voice Unbundied PLX LD DDTerminal Switchboard PDT UEPSP UEPXC 1.40 39.06 18.18 12.35 0.7187 11.90 2-						-	-										[
2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus UEPSP UEPSP UEP1 1.40 39.06 18.18 12.35 0.7187 11.90 11.90 2-Wire VG Line Side Unbundled DxWard PBX Trunk - Bus UEPSP UEPN 1.40 39.06 18.18 12.35 0.7187 11.90	2	2-Wire VG Unbundled 2-Way PBX Trunk - Res			UEPSE	UEPRD	1.40	39.06	18.18	12.35	0.7187		11.90				
2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus UEPSP UEPP1 1.40 39.06 18.18 12.35 0.7187 11.90 2-Wire VG Line Side Unbundled PBX LD Terminal Ports UEPSP UEPLD 1.40 39.06 18.18 12.35 0.7187 11.90 2-Wire Voice Unbundled PBX LD Terminal Ports UEPSP UEPAL 1.40 39.06 18.18 12.35 0.7187 11.90 2-Wire Voice Unbundled PBX LD Terminal Ports UEPSP UEPXA 1.40 39.06 18.18 12.35 0.7187 11.90 2-Wire Voice Unbundled PBX LD Torminal Sequence UEPSP UEPXB 1.40 39.06 18.18 12.35 0.7187 11.90 2-Wire Voice Unbundled PBX LD DDD Terminals Port UEPSP UEPXC 1.40 39.06 18.18 12.35 0.7187 11.90 2-Wire Voice Unbundled PBX LD Terminal Switchboard DD UEPSP UEPXC 1.40 39.06 18.18 12.35 0.7187 11.90 2-Wire Voice Unbundled PBX Hote/Hospital Economy UEPSP UEPXL 1.40 3								39.06	18.18	12.35							ĺ
2-Wire Voice Unbundled PX LD Terminal P6X Trunk - Bus UEPSP UEPLD 1.40 39.06 18.18 12.35 0.7187 11.90 2-Wire Voice Unbundled 2-Way PBX Usage Port UEPSP UEPLD 1.40 39.06 18.18 12.35 0.7187 11.90 11.90 2-Wire Voice Unbundled 2-Way PBX Usage Port UEPSP UEPXA 1.40 39.06 18.18 12.35 0.7187 11.90 11.90 2-Wire Voice Unbundled PX ID DDD Terminals Port UEPSP UEPXC 1.40 39.06 18.18 12.35 0.7187 11.90 11.90 2-Wire Voice Unbundled PBX LD DDD Terminal Switchboard Port UEPSP UEPXD 1.40 39.06 18.18 12.35 0.7187 11.90 11																	
2-Wire Voice Unbundled PEX LD Terminal Ports UEPSP UEPSP UEPXA 1.40 39.06 18.18 12.35 0.7187 11.90 11.90 2-Wire Voice Unbundled 2-Way PBX Usage Port UEPSP UEPXB 1.40 39.06 18.18 12.35 0.7187 11.90 11.90 2-Wire Voice Unbundled PEX LD IDTerminal Hotel Ports UEPSP UEPXB 1.40 39.06 18.18 12.35 0.7187 11.90 11.90 2-Wire Voice Unbundled PEX LD Terminal Switchboard Port UEPSP UEPXD 1.40 39.06 18.18 12.35 0.7187 11.90 11.90 2-Wire Voice Unbundled PEX LD Terminal Switchboard IDD UEPSP UEPXD 1.40 39.06 18.18 12.35 0.7187 11.90 1.90 2-Wire Voice Unbundled 2-Way PBX Hote/Hospital Economy UEPSP UEPXL 1.40 39.06 18.18 12.35 0.7187 11.90 1.90																	I
2-Wire Vice Unbundled 2-Way PBX Usage Port UEPSP UEPSP UEPSP UEPSP 1.40 39.06 18.18 12.35 0.7187 11.90 11.9																	l
2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports UEPSP UEPSP UEPXB 1.40 39.06 18.18 12.35 0.7187 11.90 11.90 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port UEPSP UEPXD 1.40 39.06 18.18 12.35 0.7187 11.90 11.90 2-Wire Voice Unbundled PBX LD Terminal Switchboard PDT UEPSP UEPXD 1.40 39.06 18.18 12.35 0.7187 11.90 11.90 2-Wire Voice Unbundled PBX LD Terminal Switchboard PDT UEPSP UEPSP UEPXE 1.40 39.06 18.18 12.35 0.7187 11.90 <td< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>ļ</td></td<>																	ļ
2:Wire Voice Unbundled PBX LD DDD Terminals Port UEPSP UEPXC 1.40 33.06 18.18 12.35 0.7187 11.90 2:Wire Voice Unbundled PBX LD Terminal Switchboard IDD UEPSP UEPXD 1.40 39.06 18.18 12.35 0.7187 11.90 Capable Port UEPSP UEPSP UEPXE 1.40 39.06 18.18 12.35 0.7187 11.90 11.90 2:Wire Voice Unbundled PBX LD Terminal Switchboard IDD UEPSP UEPXE 1.40 39.06 18.18 12.35 0.7187 11.90 11.9																	l
2-Wire Voice Unbundled PBX LD Terminal Switchboard Port UEPSP UEPXD 1.40 39.06 18.18 12.35 0.7187 11.90 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD UEPSP UEPXE 1.40 39.06 18.18 12.35 0.7187 11.90 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy UEPSP UEPXL 1.40 39.06 18.18 12.35 0.7187 11.90 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy UEPSP UEPXL 1.40 39.06 18.18 12.35 0.7187 11.90 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy UEPSP UEPXL 1.40 39.06 18.18 12.35 0.7187 11.90 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital UEPSP UEPXD 1.40 39.06 18.18 12.35 0.7187 11.90 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port UEPSP UEPXD 1.40 39.06 18.18 12.35 0.7187 11.90 11.90 14.90 14.90 14.90 14.90			-														i
2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD UEPSP UEPXE 1.40 39.06 18.18 12.35 0.7187 11.90 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy UEPSP UEPXE 1.40 39.06 18.18 12.35 0.7187 11.90 11.90 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy UEPSP UEPSP UEPXL 1.40 39.06 18.18 12.35 0.7187 11.90 1			-														i
Capable Port UEPSP UEPXE 1.40 39.06 18.18 12.35 0.7187 11.90 2-Wire Voice Unbundled 2-Way PBX Hote//Hospital Economy UEPSP UEPSP UEPXL 1.40 39.06 18.18 12.35 0.7187 11.90 0 2-Wire Voice Unbundled 2-Way PBX Hote//Hospital Economy UEPSP UEPXL 1.40 39.06 18.18 12.35 0.7187 11.90 0 2-Wire Voice Unbundled 2-Way PBX Hote//Hospital UEPSP UEPXM 1.40 39.06 18.18 12.35 0.7187 11.90 0 0 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital UEPSP UEPXO 1.40 39.06 18.18 12.35 0.7187 11.90 0						0EI XB	1.40	00.00	10.10	12.00	0.7107		11.00				
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port UEPSP UEPSP UEPXL 1.40 39.06 18.18 12.35 0.7187 11.90 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port UEPSP UEPSP UEPXM 1.40 39.06 18.18 12.35 0.7187 11.90 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port UEPSP UEPSP UEPXS 1.40 39.06 18.18 12.35 0.7187 11.90 0 2-Wire Voice Unbundled 1-Way Outgoing PBX Motel/Hospital Discount Room Calling Port UEPSP UEPSP UEPXS 1.40 39.06 18.18 12.35 0.7187 11.90 0					UEPSP	UEPXE	1.40	39.06	18.18	12.35	0.7187		11.90				i
Administrative Calling Port UEPSP UEPSP UEPXL 1.40 39.06 18.18 12.35 0.7187 11.90 11.90 2.Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port UEPSP UEPXM 1.40 39.06 18.18 12.35 0.7187 11.90 11.90 2.Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port UEPSP UEPXN 1.40 39.06 18.18 12.35 0.7187 11.90 11.			İ 👘		-									1			
Room Calling Port UEPSP UEPXM 1.40 39.06 18.18 12.35 0.7187 11.90 11.90 2-Wire Voice Unbundled 1-Way Outgoing PBX Hote/Hospital UEPSP UEPXO 1.40 39.06 18.18 12.35 0.7187 11.90 11.90 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port UEPSP UEPXO 1.40 39.06 18.18 12.35 0.7187 11.90	A	Administrative Calling Port			UEPSP	UEPXL	1.40	39.06	18.18	12.35	0.7187		11.90				I
Discount Room Calling Port UEPSP UEPSP UEPXS 1.40 39.06 18.18 12.35 0.7187 11.90 2.Wire Voice Unbundled 1-Way Outgoing PBX Measured Port UEPSP UEPXS 1.40 39.06 18.18 12.35 0.7187 11.90 Subsequent Activity UEPSP UEPSP UEPXS 1.40 39.06 18.18 12.35 0.7187 11.90 11.90 FEATURES UEPSP USASC 0.00 0.00 0.00 0.00 11.90 0 All Available Vertical Features UEPSP UEPSE UEPVF 2.26 0.00 0.00 0.00 11.90 0 EXCHANGE PORT RATES (COIN) UEPSP UEPSE UEPVF 2.26 0.00 0.00 0.00 11.90 0 NOTE: Transmission/usage charges associated with POTS circuit switched usage will also apply to circuit switched voice and/or circuit switched data transmission by B-Channels associated with 2-wire ISDN ports. NOTE: Access to B Channel or D Channel Packet capabilities will be available only through BFR/New Business Request Process. Rate for the packet capabilities will be determined via the Bona Fide Request/New Business Request Process. UNBUNDLED LOCAL EXCHANGE PORT RATES UBUNDLED LOCAL EXCHANGE					UEPSP	UEPXM	1.40	39.06	18.18	12.35	0.7187		11.90				
Discount Room Calling Port UEPSP UEPSP 1.40 39.06 18.18 12.35 0.7187 11.90 11.90 2.Wire Voice Unbundled 1-Way Outgoing PBX Measured Port UEPSP UEPSP UEPXS 1.40 39.06 18.18 12.35 0.7187 11.90	2	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital															(
Subsequent Activity UEPSP USASC 0.00 0.00 0.00 11.90 1 FEATURES All Available Vertical Features Image: Constraint of the sector of the sec																	I
FEATURES Image: Construct State							-			12.35	0.7187			ļ			
All Available Vertical Features UEPSP UEPSE UEPVF 2.26 0.00 0.00 11.90 ExCHANGE PORT RATES (COIN) Image: Construct on Port Image: Construct on Port 1.40 3.74 3.63 1.88 1.80 11.90 NOTE: Transmission/usage charges associated with POTS circuit switched usage will also apply to circuit switched voice and/or circuit switched data transmission by B-Channels associated with 2-wire ISDN ports. NOTE: Access to B Channel or D Channel Packet capabilities will be available only through BFR/New Business Request Process. Rates for the packet capabilities will be determined via the Bona Fide Request/New Business Request Process. Rates for the packet capabilities will be determined via the Bona Fide Request/New Business Request Process. UNBUNDLED LOCAL EXCHANGE SWITCHING(PORTS) Image: Construct on the Port RATES Image: Construct on the Port RATES				L	UEPSP	USASC	0.00	0.00	0.00	ļ		ļ	11.90	ļ			l
EXCHANGE PORT RATES (COIN) Image Ports - Coin Port Image Port - Coin Port Image Port - Coin Port Image Port - Coin Port Image Port - Coin Port Image Port - Coin Port Image Port - Coin Port Image Port - Coin Port Image Port - Coin Port Image Port - Coin Port Image Port - Coin Port Image Port - Coin Port Image Port - Coin Port Image Port - Coin Port Image Port - Coin Port Image Port - Coin Port <td< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td> </td></td<>																	
Exchange Ports - Coin Port 1.40 3.74 3.63 1.88 1.80 11.90 NOTE: Transmission/usage charges associated with POTS circuit switched usage will also apply to circuit switched voice and/or circuit switched data transmission by B-Channels associated with 2-wire ISDN ports. NOTE: Access to B Channel or D Channel Packet capabilities will be available only through BFR/New Business Request Process. Rates for the packet capabilities will be determined via the Bona Fide Request/New Business Returned Via the Bona Fide Request/New Business Returned Via the Bona Fide Request/New Business Returned Via the Bona Fide Request/New Business Returned Via the Bona Fide Return					UEPSP UEPSE	UEPVF	2.26	0.00	0.00				11.90	<u> </u>			
NOTE: Transmission/usage charges associated with POTS circuit switched usage will also apply to circuit switched voice and/or circuit switched data transmission by B-Channels associated with 2-wire ISDN ports. NOTE: Access to B Channel or D Channel Packet capabilities will be available only through BFR/New Business Request Process. Rates for the packet capabilities will be determined via the Bona Fide Request/New Business Request Process. UNBUNDLED LOCAL EXCHANGE SWITCHING(PORTS)			<u> </u>			+	1 40	2.74	363	1 00	1.90		11 00	<u> </u>			l
NOTE: Access to B Channel or D Channel Packet capabilities will be available only through BFR/New Business Request Process. Rates for the packet capabilities will be determined via the Bona Fide Request/New Business Request Process. Rates for the packet capabilities will be determined via the Bona Fide Request/New Business Request Process. Rates for the packet capabilities will be determined via the Bona Fide Request/New Business Request Process. Rates for the packet capabilities will be determined via the Bona Fide Request/New Business Request Process. Rates for the packet capabilities will be determined via the Bona Fide Request/New Business Request Process. Rates for the packet capabilities will be determined via the Bona Fide Request/New Business Request Process. Rates for the packet capabilities will be determined via the Bona Fide Request/New Business Request Process. Rates for the packet capabilities will be determined via the Bona Fide Request/New Business Request Process. Rates for the packet capabilities will be determined via the Bona Fide Request/New Business Request Process. Rates for the packet capabilities will be determined via the Bona Fide Request/New Business Request Process. Rates for the packet capabilities will be determined via the Bona Fide Request/New Business Request Process. Rates for the packet capabilities will be determined via the Bona Fide Request/New Business Request Process. Rates for the packet capabilities will be determined via the Bona Fide Request/New Business Request Process. Rates for the packet capabilities will be determined via the Bona Fide Request/New Business Request Process. Rates for the packet capabilities will be determined via the Bona Fide Request/New Business Request Process. Rates for the packet capabilities will be determined via the Bona Fide Request/New Business Request Process. Rates for the packet Process. Rates for the packet Process. Rates for the packet Process. Rates for the packet Process. Rates for the packet Process. Rates for the packet Proces			witched	lisade	will also apply to c	ircuit switche								ports.			
UNBUNDLED LOCAL EXCHANGE SWITCHING(PORTS)															Request Pro	cess.	
EXCHANGE PORT RATES					,				, capabi								1
			1			1						1		1		ĺ	(
		Exchange Ports - 2-Wire DID Port			UEPEX	UEPP2	8.73	78.41	15.82	41.94	4.26		11.90			1.83	
Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID			ſ			i											
capability UEPDD UEPDD 54.95 151.11 77.75 48.81 3.10 11.90	c	capability			UEPDD	UEPDD	54.95	151.11	77.75	48.81	3.10		11.90			1.83	I

UNBU	NDLE	D NETWORK ELEMENTS - Florida												Attachment:	2	Exhi	bit: B
CATEG	iory	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
							Rec	Nonrec	urring	Nonrecurring	g Disconnect			OSS	Rates(\$)		
								First	Add'l	First	Add'l	SOMEC		SOMAN	SOMAN	SOMAN	SOMAN
		Exchange Ports - 2-Wire ISDN Port (See Notes below.)			UEPTX UEPSX	U1PMA	8.83	46.83	50.68	27.64	11.93		11.90			1.83	
	NOTE	All Features Offered			UEPTX UEPSX	UEPVF	2.26	0.00	0.00	iacian hu D Ch		at a d with 0	11.90			1.83	-
		Transmission/usage charges associated with POTS circuit sv Access to B Channel or D Channel Packet capabilities will be													Boguost Br	0000	
	NOTE.	Exchange Ports - 2-Wire ISDN Port Channel Profiles	avalla		UEPTX UEPSX	U1UMA	0.00	0.00	0.00	intes will be de			le Request	New Busilies:	s Request FR		-
		Exchange Ports - 4-Wire ISDN DS1 Port			UEPEX	UEPEX	82.74	174.61	95.17	49.80	18.23		11.90			1.83	
	UNBUN	IDLED PORT with REMOTE CALL FORWARDING CAPABILITY				OEI EX	02.14	114.01	50.17	40.00	10.20		11.00			1.00	
		IDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE															
		Unbundled Remote Call Forwarding Service, Area Calling, Res			UEPVR	UERAC	1.40	3.74	3.63	1.88	1.80		11.90				
		Unbundled Remote Call Forwarding Service, Local Calling - Res			UEPVR	UERLC	1.40	3.74	3.63	1.88	1.80		11.90				
		Unbundled Remote Call Forwarding Service, InterLATA - Res			UEPVR	UERTE	1.40	3.74	3.63	1.88	1.80		11.90				
		Unbundled Remote Call Forwarding Service, IntraLATA - Res			UEPVR	UERTR	1.40	3.74	3.63	1.88	1.80		11.90				
	Non-Re	ecurring		I								ļ					
		Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is			UEPVR	USAC2		0.102	0.102				11.90				
		Unbundled Remote Call Forwarding Service - Conversion with		1												1	
		allowed change (PIC and LPIC)			UEPVR	USACC		0.102	0.102								
	UNBUN	IDLED REMOTE CALL FORWARDING - Bus															
		Unbundled Remote Call Forwarding Service, Area Calling - Bus			UEPVB	UERAC	1.40	3.74	3.63	1.88	1.80		11.90				
		Unbundled Remote Call Forwarding Service, Local Calling - Bus			UEPVB	UERLC	1.40	3.74	3.63	1.88	1.80		11.90				
		Unbundled Remote Call Forwarding Service, Local Calling - Bus			UEPVB	UERTE	1.40	3.74	3.63	1.88	1.80		11.90				
		Unbundled Remote Call Forwarding Service, Interactive Bus		1	UEPVB	UERTR	1.40	3.74	3.63	1.88	1.80		11.90				
		Unbundled Remote Call Forwarding Service Expanded and				OLIVIN	1.40	0.14	0.00	1.00	1.00		11.00			1	
		Exception Local Calling			UEPVB	UERVJ	1.40	3.74	3.63	1.88	1.80		11.90				
	Non-Re	ecurring															
		Unbundled Remote Call Forwarding Service - Conversion -															
		Switch-as-is			UEPVB	USAC2		0.102	0.102				11.90				
		Unbundled Remote Call Forwarding Service - Conversion with															
		allowed change (PIC and LPIC)			UEPVB	USACC		0.102	0.102								
		OCAL SWITCHING, PORT USAGE															
	End Of	fice Switching (Port Usage)				-	0.0007000					-				-	-
		End Office Switching Function, Per MOU End Office Trunk Port - Shared, Per MOU				_	0.0007662 0.000164										
	Tander	n Switching (Port Usage) (Local or Access Tandem)					0.000164					<u> </u>				ł	ł
	ranuel	Tandem Switching Function Per MOU			1	1	0.0001319					-				 	1
	1	Tandem Trunk Port - Shared, Per MOU		1		1	0.000235					1					
	Commo	on Transport		1		1										1	
	1	Common Transport - Per Mile, Per MOU		1			0.0000035									1	
		Common Transport - Facilities Termination Per MOU					0.0004372										
UNBUN		ORT/LOOP COMBINATIONS - COST BASED RATES															
		ased Rates are applied where BellSouth is required by FCC an															
		es shall apply to the Unbundled Port/Loop Combination - Cos														ļ	L
		fice and Tandem Switching Usage and Common Transport Us														ļ	ļ
		st and additional Port nonrecurring charges apply to Not Curr	ently C	ombine	ed Combos. For Cu	rrently Comb	ined Combos th	ne nonrecurring	g charges sha	II be those ider	ntified in the N	onrecurring	g - Currently	Combined s	ections.	ļ	ļ
-		VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)		<u> </u>								ļ				ł	ł
	UNE PO	prt/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1		1		_	10.94										
		2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2		1		+	10.94				-	<u> </u>		ł		ł	ł
		2-Wire VG Loop/Port Combo - Zone 3		3			25.80									1	1
		pop Rates					20.00									1	1
	2	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPRX	UEPLX	9.77					1					
		2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPRX	UEPLX	13.88			1					1	1	1
	1	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPRX	UEPLX	24.63					1	l	1		1	İ
	2-Wire	Voice Grade Line Port Rates (Res)											1			1	1
1		2-Wire voice unbundled port - residence			UEPRX	UEPRL	1.17	53.31	26.46	27.50	8.37		11.90				
		2-Wire voice unbundled port with Caller ID - res			UEPRX	UEPRC	1.17	53.31	26.46	27.50	8.37		11.90				

NRONDLE	D NETWORK ELEMENTS - Florida										-	-	Attachment:			bit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge Manual S Order vs Electronic Disc Add
						Rec	Nonrec		Nonrecurring					Rates(\$)		
			-				First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
_	2-Wire voice unbundled port outgoing only - res			UEPRX	UEPRO	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire voice unbundled Florida Area Calling with Caller ID - res			UEPRX	UEPAF	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire voice unbundles res, low usage line port with Caller ID			021101	02.74		00.01	20.10	21.00	0.01		11.00				
	(LUM)			UEPRX	UEPAP	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire voice unbundled Florida extended dialing port for use															
	with CREX7 and Caller ID			UEPRX	UEPA1	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire voice unbundled Florida extended dialing port for use															
	with CREX7, without Caller ID capability		-	UEPRX	UEPA8	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire voice unbundled Florida Area Calling Port without Caller ID Capability			UEPRX	UEPA9	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire voice unbundled Low Usage Line Port without Caller ID			ULFRA	ULFAS	1.17	55.51	20.40	27.30	0.57		11.90				
	Capability			UEPRX	UEPRT	1.17	53.31	26.46	27.50	8.37		11.90				
FEAT																
	All Features Offered			UEPRX	UEPVF	2.26	0.00	0.00				11.90				
LOCA	L NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPRX	LNPCX	0.35										
NONR	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -															
	Switch-as-is			UEPRX	USAC2		0.102	0.102				11.90				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with change	-		UEPRX	USACC		0.102	0.102				11.90				
	TONAL NRCs			ULFRA	USACC		0.102	0.102				11.90				
ADDII	2-Wire Voice Grade Loop/Line Port Combination - Subsequent		1													
	Activity			UEPRX	USAS2	0.00	0.00	0.00				11.90				
2-WIR	E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)															
UNE F	Port/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			10.94										
	2-Wire VG Loop/Port Combo - Zone 2		2			15.05										
	2-Wire VG Loop/Port Combo - Zone 3		3			25.80										
UNEL	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPBX	UEPLX	9.77										
	2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPBX	UEPLX	9.77										
	2-Wire Voice Grade Loop (SL1) - Zone 2		3	UEPBX	UEPLX	24.63										
2-Wire	Voice Grade Line Port (Bus)			02. 5/	02.2.	21.00										
	2-Wire voice unbundled port without Caller ID - bus			UEPBX	UEPBL	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPBX	UEPBC	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire voice unbundled port outgoing only - bus			UEPBX	UEPBO	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire voice unbundled incoming only port with Caller ID - Bus			UEPBX	UPEB1	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire voice unbundled Incoming Only Port without Caller ID	1	1				FO O I	00.10	07.50	0.07		44.00				
1.004				UEPBX	UEPBE	1.17	53.31	26.46	27.50	8.37		11.90				
LUCA	L NUMBER PORTABILITY Local Number Portability (1 per port)			UEPBX	LNPCX	0.35										
FEAT			1			0.35										
	All Features Offered	1	1	UEPBX	UEPVF	2.26	0.00	0.00				11.90		1		
NONR	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED	1	1								1			l		
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -															
	Switch-as-is			UEPBX	USAC2		0.102	0.102				11.90				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -	-	1													
400	Switch with change		<u> </u>	UEPBX	USACC		0.102	0.102				11.90				
ADDIT	2-Wire Voice Grade Loop/Line Port Combination - Subsequent															
	2-vvire voice Grade Loop/Line Port Combination - Subsequent Activity	1	1	UEPBX	USAS2		0.00	0.00				11.90				
2-WIR	E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)				00402		0.00	0.00				11.90				
	Port/Loop Combination Rates	1	1													
	2-Wire VG Loop/Port Combo - Zone 1		1			10.94										
	2-Wire VG Loop/Port Combo - Zone 2	1	2			15.05					İ					
	2-Wire VG Loop/Port Combo - Zone 3		3			25.80										
	oop Rates															

INBUNDLE	ED NETWORK ELEMENTS - Florida			•								1	Attachment:			bit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEPRG	UEPLX	9.77										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEPRG	UEPLX	13.88										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEPRG	UEPLX	24.63										
2-Wire	Voice Grade Line Port Rates (RES - PBX)	-			-											
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Res			UEPRG	UEPRD	1.17	174.81	100.65	75.88	10.70		11.90				
1.004				UEPRG	UEPRD	1.17	174.81	100.65	75.88	12.73		11.90				
LOCA	Local Number Portability (1 per port)			UEPRG	LNPCP	0.00	0.00	0.00			1	11.90				
FFAT	URES			ULFKG	LINFOF	0.00	0.00	0.00				11.90				
	All Features Offered			UEPRG	UEPVF	2.26	0.00	0.00				11.90				
NONE	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED			0EI NO	021 11	2.20	0.00	0.00	-		1	11.00				1
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -	1		1									1		1	
	Conversion - Switch-As-Is	1	1	UEPRG	USAC2		8.45	1.91				11.90				
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -	1	1								1		İ	ĺ	İ	1
	Conversion - Switch with Change			UEPRG	USACC		8.45	1.91				11.90				
ADDI	TIONAL NRCs															
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -															
	Subsequent Activity			UEPRG	USAS2	0.00	0.00	0.00				11.90				
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt															
	Group						7.86	7.86				11.90				
	E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)															
UNE F	Port/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			10.94										
	2-Wire VG Loop/Port Combo - Zone 2		2			15.05										
	2-Wire VG Loop/Port Combo - Zone 3		3			25.80										
UNE L	oop Rates															
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEPPX	UEPLX	9.77										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEPPX	UEPLX	13.88										
0.14/	2-Wire Voice Grade Loop (SL 1) - Zone 3	-	3	UEPPX	UEPLX	24.63										
2-Wire	• Voice Grade Line Port Rates (BUS - PBX)				_											
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPPX	UEPPC	1.17	174.81	100.65	75.88	12.73		11.90				
	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPPX	UEPPO	1.17	174.81	100.65	75.88	12.73		11.90				
	Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPPX	UEPP1	1.17	174.81	100.65	75.88	12.73	1	11.90				1
	2-Wire Voice Unbundled PBX LD Terminal Ports	-		UEPPX	UEPLD	1.17	174.81	100.65	75.88	12.73	-	11.90				-
	2-Wire Voice Unbundled PBX LD Terminal Ports 2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPPX	UEPXA	1.17	174.81	100.65	75.88	12.73		11.90				
	2-Wire Voice Unbundled 2-Way combination 1 bx 0sage 1 of			UEPPX	UEPXB	1.17	174.81	100.65	75.88	12.73	t	11.90				
	2-Wire Voice Unbundled PBX LD DDD Terminals Port	1		UEPPX	UEPXC	1.17	174.81	100.65	75.88	12.73	t	11.90	1	1	1	1
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port	1		UEPPX	UEPXD	1.17	174.81	100.65	75.88	12.73		11.90	İ		İ	
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD	1	1			-					1		İ	ĺ	İ	1
	Capable Port			UEPPX	UEPXE	1.17	174.81	100.65	75.88	12.73		11.90				
1	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy	1	1													
	Administrative Calling Port			UEPPX	UEPXL	1.17	174.81	100.65	75.88	12.73		11.90				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy	1														
	Room Calling Port			UEPPX	UEPXM	1.17	174.81	100.65	75.88	12.73		11.90				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital															
	Discount Room Calling Port			UEPPX	UEPXO	1.17	174.81	100.65	75.88	12.73		11.90				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPPX	UEPXS	1.17	174.81	100.65	75.88	12.73		11.90				
LOCA	L NUMBER PORTABILITY															
	Local Number Portability (1 per port)	I	I	UEPPX	LNPCP	3.15	0.00	0.00				11.90				
FEAT		-	ļ	LIEBBY/		0.55										
	All Features Offered	I	I	UEPPX	UEPVF	2.26	0.00	0.00				11.90				L
NONE	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED	<u> </u>									L					L
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -	1	1		110.4.00		8.45					11.90				
														1		1
	Conversion - Switch-As-Is			UEPPX	USAC2		8.45	1.91				11.90				
	Conversion - Switch-As-Is 2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with Change				USAC2		8.45	1.91				11.90				

JNBUNDLE	D NETWORK ELEMENTS - Florida												Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -															
	Subsequent Activity			UEPPX	USAS2	0.00	0.00	0.00			-	11.90				
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group						7.86	7.86				11.90				
2-WIRE	E VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN POR	PT			-		7.00	7.00				11.90				
	ort/Loop Combination Rates	Ì			+ +						1					
	2-Wire VG Coin Port/Loop Combo – Zone 1		1			10.94										
	2-Wire VG Coin Port/Loop Combo – Zone 2		2			15.05										
	2-Wire VG Coin Port/Loop Combo – Zone 3		3			25.80										
UNE L	oop Rates															
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPCO	UEPLX	9.77										
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPCO	UEPLX	13.88										
0.14/1	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPCO	UEPLX	24.63					L					
2-wire	Voice Grade Line Ports (COIN) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (FL)			UEPCO	UEP2F	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking (FL)			UEPCO	UEPFA	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (FL)			UEPCO	UEPCG	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Coin Outward with Operator Screening and 011 Blocking (AL, FL)			UEPCO	UEPRK	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+ (FL) 2-Wire Coin Outward with Operator Screening and Blocking:			UEPCO	UEPOF	1.17	53.31	26.46	27.50	8.37		11.90				
	900/976, 1+DDD, 011+, and Local (FL, GA)			UEPCO	UEPCQ	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire 2-Way Smartline with 900/976 (all states except LA)			UEPCO	UEPCK	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Coin Outward Smartline with 900/976 (all states except			02.00	02.01		00.01	20.10	21.00	0.01	1	11.00				
	LA)			UEPCO	UEPCR	1.17	53.31	26.46	27.50	8.37		11.90				
ADDIT	IONAL UNE COIN PORT/LOOP (RC)															
	UNE Coin Port/Loop Combo Usage (Flat Rate)			UEPCO	URECU	1.86	53.31	26.46	27.50	8.37		11.90				
LOCAL	NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPCO	LNPCX	0.35										
NONRI	ECURRING CHARGES - CURRENTLY COMBINED				+ +						-					
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is			UEPCO	USAC2		0.102	0.102				11.90				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -			UEPCO	USACC		0.102	0.102				11.90				
דיחתא	Switch with change				USACC		0.102	0.102				11.90				
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent				+ +						-					
	Activity			UEPCO	USAS2		0.00	0.00				11.90				
2-WIRE	E VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE	LINE F	PORT (0.00	0.00								
	ort/Loop Combination Rates		Ľ	·		_										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			13.64										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			18.80										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3		-	32.27										
UNE L	oop Rates				115050	10.01										
	2-Wire Voice Grade Loop (SL2) - Zone 1		1		UECF2 UECF2	12.24 17.40						ļ				
	2-Wire Voice Grade Loop (SL2) - Zone 2 2-Wire Voice Grade Loop (SL2) - Zone 3		2	UEPFR UEPFR	UECF2 UECF2	30.87										
2-Wire	Voice Grade Line Port Rates (Res)		3	OLFEN		30.07					-					
2-1116	2-Wire voice unbundled port - residence			UEPFR	UEPRL	1.40	174.81	100.65	75.88	12.73	1	11.90				
	2-Wire voice unbundled port + residence 2-Wire voice unbundled port with Caller ID - res			UEPFR	UEPRC	1.40	174.81	100.65	75.88	12.73	1	11.90				
	2-Wire voice unbundled port outgoing only - res			UEPFR	UEPRO	1.40	174.81	100.65	75.88	12.73		11.90				
	2-Wire voice unbundled Florida Area Calling with Caller ID - res			UEPFR	UEPAF	1.40	174.81	100.65	75.88	12.73		11.90				ļ
	2-Wire voice unbundles res, low usage line port with Caller ID (LUM)			UEPFR	UEPAP	1.40	174.81	100.65	75.88	12.73		11.90				
				UEFFK	UEPAP	1.40	174.81	100.65	75.88	12.73		11.90				

UNBUNDL	ED NETWORK ELEMENTS - Florida			1								1	Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility Termination			UEPFR	U1TV2	25.32	47.35	04.70								
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile			UEPFR	01172	20.32	47.35	31.78								<u> </u>
	or Fraction Mile			UEPFR	1L5XX	0.0091										1
FEAT	TURES															
	All Features Offered			UEPFR	UEPVF	2.26	0.00	0.00				11.90				
LOC	AL NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPFR	LNPCX	0.35										
NON	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED 2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
	Combination - Conversion - Switch-as-is			UEPFR	USAC2		16.97	3.73				11.90				
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port			OLITIK	00/02		10.57	5.75				11.30				
	Combination - Conversion - Switch-With-Change			UEPFR	USACC		16.97	3.73				11.90				
2-WI	RE VOICE LOOP/ 2WIRE VOICE GRADE IO TRĂNSPORT/ 2-WIRI	LINE	ORT (BUS)												
UNE	Port/Loop Combination Rates															
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			13.64										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			18.80										l
LINE	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3 Loop Rates		3			32.27										l
UNE	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFB	UECF2	12.24										
	2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFB	UECF2	17.40										
	2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFB	UECF2	30.87										
2-Wii	e Voice Grade Line Port (Bus)		-													
	2-Wire voice unbundled port without Caller ID - bus			UEPFB	UEPBL	1.40	174.81	100.65	75.88	12.73		11.90				
	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPFB	UEPBC	1.40	174.81	100.65	75.88	12.73		11.90				
	2-Wire voice unbundled port outgoing only - bus			UEPFB	UEPBO	1.40	174.81	100.65	75.88	12.73		11.90				ļ
1.00	2-Wire voice unbundled incoming only port with Caller ID - Bus			UEPFB	UEPB1	1.40	174.81	100.65	75.88	12.73		11.90		-		
LUCA	AL NUMBER PORTABILITY Local Number Portability (1 per port)			UEPFB	LNPCX	0.35										<u> </u>
INTE	ROFFICE TRANSPORT			OLITE		0.55										
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility															
	Termination			UEPFB	U1TV2	25.32	47.35	31.78								
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile															
	or Fraction Mile			UEPFB	1L5XX	0.0091										
FEAT	TURES															
NON	All Features Offered			UEPFB	UEPVF	2.26	0.00	0.00				11.90				
NON	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED 2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
	Combination - Conversion - Switch-as-is			UEPFB	USAC2		16.97	3.73				11.90				
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port							00								<u></u>
	Combination - Conversion - Switch with change			UEPFB	USACC		16.97	3.73				11.90				
	RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)															
UNE	Port/Loop Combination Rates															
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1		+	13.64										<u> </u>
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2 2-Wire VG Loop/IO Tranport/Port Combo - Zone 3	-	2		+	18.80 32.27										
UNF	Loop Rates		3		+ +	32.21					1					+
0.1L	2-Wire Voice Grade Loop (SL2) - Zone 1	1	1	UEPFP	UECF2	12.24					1					<u> </u>
	2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFP	UECF2	17.40					1		1			<u> </u>
	2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFP	UECF2	30.87										
2-Wi	e Voice Grade Line Port Rates (BUS - PBX)															
										· · · · ·						1
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus Line Side Unbundled Outward PBX Trunk Port - Bus	ļ		UEPFP UEPFP	UEPPC UEPPO	1.40 1.40	174.81	100.65 100.65	75.88	12.73		11.90				
	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPFP	UEPPO UEPP1	1.40 1.40	174.81 174.81	100.65	75.88 75.88	12.73 12.73		11.90 11.90				
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPFP	UEPLD	1.40	174.81	100.65	75.88	12.73	1	11.90				+
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPFP	UEPXA	1.40	174.81	100.65	75.88	12.73	1	11.90		-		<u> </u>
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports	l		UEPFP	UEPXB	1.40	174.81	100.65	75.88	12.73	1	11.90	1		1	
	2-Wire Voice Unbundled PBX LD DDD Terminals Port	1	1	UEPFP	UEPXC	1.40	174.81	100.65	75.88	12.73	1	11.90				

UNBUNDLE	D NETWORK ELEMENTS - Florida													Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	S	USOC			RATES(\$)				Svc Order Submitted Manually per LSR		Incremental Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
<u> </u>								Nonrec	urring	Nonrecurring	Disconnect				Rates(\$)	Diac Tat	Disc Add I
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPFP		UEPXD	1.40	174.81	100.65	75.88	12.73		11.90				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD																
	Capable Port			UEPFP		UEPXE	1.40	174.81	100.65	75.88	12.73		11.90				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy			UEPFP		UEPXL	1.40	174.81	100.65	75.88	12.73		11.90				
	Administrative Calling Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy			UEPFP		UEPXL	1.40	174.81	100.001	75.88	12.73		11.90				
	Room Calling Port			UEPFP		UEPXM	1.40	174.81	100.65	75.88	12.73		11.90				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital																
	Discount Room Calling Port			UEPFP		UEPXO	1.40	174.81	100.65	75.88	12.73		11.90				
1.00.41	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPFP		UEPXS	1.40	174.81	100.65	75.88	12.73		11.90				
LOCAL	L NUMBER PORTABILITY Local Number Portability (1 per port)			UEPFP		LNPCP	3.15	0.00	0.00				11.90				
INTER	OFFICE TRANSPORT						5.15	0.00	0.00				11.30				
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility	1										1			1	1	1
	Termination			UEPFP		U1TV2	25.32	47.35	31.78								
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile																
EEATI	or Fraction Mile			UEPFP		1L5XX	0.0091										
FEATU	All Features Offered			UEPFP		UEPVF	2.26	0.00	0.00				11.90	-			
NONRI	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED					OLI VI	2.20	0.00	0.00				11.30				
1101111	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port																
	Combination - Conversion - Switch-as-is			UEPFP		USAC2		16.97	3.73				11.90				
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port																
	Combination - Conversion - Switch with change			UEPFP		USACC		16.97	3.73				11.90				
	PORT/LOOP COMBINATIONS - COST BASED RATES	(DODT															
	E VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK ort/Loop Combination Rates	PORT															
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1		1				20.95										
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2		2				26.11										
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3		3				39.58										
UNE L	oop Rates																
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1		1	UEPPX		UECD1	12.24						11.90			1.83	
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2		2	UEPPX		UECD1	17.40						11.90			1.83	
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3 ort Rate	-	3	UEPPX		UECD1	30.87						11.90			1.83	
UNE PO	Exchange Ports - 2-Wire DID Port			UEPPX		UEPD1	8.71	214.16	98.29				11.90			1.83	
NONRI	ECURRING CHARGES - CURRENTLY COMBINED	1		02117		02101	0.71	214.10	55.25				11.30			1.05	
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination -	·										1			1	1	1
	Switch-as-is			UEPPX		USAC1		7.85	1.87				11.90				
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion					10140											
	with BellSouth Allowable Changes			UEPPX		USA1C		7.85	1.87			<u> </u>	11.90				
	IONAL NRCs 2-Wire DID Subsequent Activity - Add Trunks, Per Trunk			UEPPX		USAS1		32.26	32.26			<u> </u>	11.90				
Teleph	none Number/Trunk Group Establisment Charges	1				00001		32.20	52.20				11.30				
	DID Trunk Termination (One Per Port)	1		UEPPX		NDT	0.00	0.00	0.00			1	11.90			1.83	
	DID Numbers, Establish Trunk Group and Provide First Group	1										1			İ		İ
	of 20 DID Numbers			UEPPX		NDZ	0.00	0.00	0.00				11.90			1.83	
	Additional DID Numbers for each Group of 20 DID Numbers		ļ	UEPPX		ND4	0.00	0.00	0.00			ļ	11.90			1.83	
├───┼────	DID Numbers, Non- consecutive DID Numbers, Per Number			UEPPX UEPPX		ND5	0.00	0.00	0.00			<u> </u>	11.90 11.90			1.83	
├──	Reserve Non-Consecutive DID numbers Reserve DID Numbers			UEPPX		ND6 NDV	0.00	0.00	0.00				11.90			1.83 1.83	
LOCAI	L NUMBER PORTABILITY						0.00	0.00	0.00				11.50			1.03	
	Local Number Portability (1 per port)	1		UEPPX		LNPCP	3.15	0.00	0.00			1					
	E ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LI	NE SIDE	PORT														
UNE P	ort/Loop Combination Rates																
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 1		1	UEPPB	UEPPR		22.63										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 2		2	UEPPB	UEPPR		29.05										

UNBUNDLE	D NETWORK ELEMENTS - Florida	1	1				1					1		Attachment:			ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	В	BCS	usoc			RATES(\$)				Submitted	Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Charge -	Charge - Manual Svo Order vs.
							Rec	Nonrec		Nonrecurring					Rates(\$)		
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	UNE Zone 3		3	UEPPB	UEPPR		45.84							1			
UNE L	oop Rates		Ŭ	OLITE	OLITIK	1	40.04					1		'	<u> </u>	<u> </u>	
-	2-Wire ISDN Digital Grade Loop - UNE Zone 1		1	UEPPB	UEPPR	USL2X	15.25						11.90			1.83	
	2-Wire ISDN Digital Grade Loop - UNE Zone 2		2	UEPPB	UEPPR	USL2X	21.67						11.90	ļ'		1.83	
	2-Wire ISDN Digital Grade Loop - UNE Zone 3		3	UEPPB	UEPPR	USL2X	38.46						11.90	ļ'		1.83	<u> </u>
UNE P	Port Rate Exchange Port - 2-Wire ISDN Line Side Port			UEPPB	UEPPR	UEPPB	7.38	194.52	145.09				11.09	 '	───	1.83	+
NONP	ECURRING CHARGES - CURRENTLY COMBINED			UEFFB	UEPPK	UEFFB	7.30	194.52	145.09			-	11.09	 '	ł	1.03	+
	2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port																1
1	Combination - Conversion	1	1	UEPPB	UEPPR	USACB	0.00	25.22	17.00			1	11.90	1		1.83	
	IONAL NRCs	1		İ													1
LOCAL	L NUMBER PORTABILITY																
	Local Number Portability (1 per port)			UEPPB	UEPPR	LNPCX	0.35	0.00	0.00					ļ'	L	'	<u> </u>
B-CHA	NNEL USER PROFILE ACCESS:													ļ'			
	CVS/CSD (DMS/5ESS) CVS (EWSD)			UEPPB UEPPB	UEPPR UEPPR	U1UCA U1UCB	0.00	0.00	0.00			-		 '			+
	CSD			UEPPB	UEPPR	U1UCC	0.00	0.00	0.00					<u> </u>			
B-CHA	ANNEL AREA PLUS USER PROFILE ACCESS: (AL.KY.LA.MS S		TN)	ULFFD	ULFFK	01000	0.00	0.00	0.00			1		<u> </u> '			+
	TERMINAL PROFILE	0,1410, 0				1						1		'	<u> </u>	<u> </u>	1
	User Terminal Profile (EWSD only)			UEPPB	UEPPR	U1UMA	0.00	0.00	0.00								1
VERTI	CAL FEATURES																1
	All Vertical Features - One per Channel B User Profile			UEPPB	UEPPR	UEPVF	2.26	0.00	0.00				11.90				
INTER	OFFICE CHANNEL MILEAGE													'			
	Interoffice Channel mileage each, including first mile and							17.07		10.01							
	facilities termination Interoffice Channel mileage each, additional mile			UEPPB UEPPB	UEPPR UEPPR	M1GNC M1GNM	25.3291 0.0091	47.35 0.00	31.78 0.00	18.31	7.03	-	11.90 11.90	 '		1.83 1.83	
4-WIRI	E DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK	POPT	-	UEFFD	UEPPK	IVITGINIVI	0.0091	0.00	0.00			-	11.90	<u> </u> '		1.03	+
	Port/Loop Combination Rates		1											'			
0.12	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE					1											1
	Zone 1		1	UEPPP			153.48										
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE																1
	Zone 2		2	UEPPP			183.28										
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE																
	Zone 3		3	UEPPP			261.12							ļ'			
UNE L	oop Rates 4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPPP		USL4P	70.74					-	11.90	 '		1.83	+
	4-Wire DS1 Digital Loop - UNE Zone 1		2	UEPPP		USL4P USL4P	100.54					-	11.90	 '	ł	1.83	
	4-Wire DS1 Digital Loop - UNE Zone 3			UEPPP		USL4P	178.38					1	11.90	'	<u> </u>	1.83	
UNE P	Port Rate		Ů	02		002.1	110.00						11.00				1
	Exchange Ports - 4-Wire ISDN DS1 Port			UEPPP		UEPPP	82.74	488.36	276.65				11.90			1.83	1
NONR	ECURRING CHARGES - CURRENTLY COMBINED																
	4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port																
	Combination - Conversion -Switch-as-is	<u> </u>		UEPPP		USACP	0.00	84.17	61.38			<u> </u>	11.90	 '	───	1.83	+
ADDIT	IONAL NRCs 4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy-											+		├ ────'	<u> </u>	<u> </u> '	+
	Inward/two way Tel Nos. (except NC)		1	UEPPP		PR7TF		0.5412				1	11.90	1		1.83	
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port -	1						0.0412				1	11.50	'	<u> </u>	1.00	+
1	Outward Tel Numbers (All States except NC)	1	1	UEPPP		PR7TO		12.71	12.71			1	11.90	1		1.83	
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port -	1										1					
	Subsequent Inward Tel Numbers			UEPPP		PR7ZT		25.42	25.42				11.90	<u> </u>		1.83	
LOCAL	L NUMBER PORTABILITY																
	Local Number Portability (1 per port)	I	Ļ	UEPPP		LNPCN	1.75							 '	L	ļ'	
INTER	FACE (Provsioning Only)		<u> </u>	UEPPP		PR71V	0.00	0.00	0.00					 '	├ ───	├ ────'	
1	Voice/Data			UEPPP		PR71V PR71D	0.00	0.00	0.00					 '	───	<u> </u> '	
	Digital Data Inward Data			UEPPP		PR71E	0.00	0.00	0.00								+

	TWORK ELEMENTS - Florida	1	1	[1						Svc Order	Suc Order	Attachment:			bit: B Incremen
TEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				SVC Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge Manual S Order vs Electron Disc Ade
						Rec	Nonrecu	urring	Nonrecurring	J Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	or Additional - Voice/Data B Channel			UEPPP	PR7BV	0.00	15.48					11.90			1.83	
	or Additional - Digital Data B Channel			UEPPP	PR7BF	0.00	15.48					11.90			1.83	
	or Additional Inward Data B Channel			UEPPP	PR7BD	0.00	15.48					11.90			1.83	
CALL TYPES																
Inward				UEPPP	PR7C1	0.00	0.00	0.00								
Outwa				UEPPP	PR7C0	0.00	0.00	0.00								
Two-w				UEPPP	PR7CC	0.00	0.00	0.00								
	nannel Mileage Each Including First Mile			UEPPP	1LN1A	88.6256	105.54	98.47	21.47	19.05		11.90			1.93	
	Airline-Fractional Additional Mile		-	UEPPP	1LN1A	0.1856	105.54	90.47	21.47	19.05		11.90			1.93	
	DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT	+	+		ILINID	0.1000					t					
	op Combination Rates	1	1													
	S1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1	+	1	UEPDC	+	125.69						11.90			1.83	<u> </u>
	S1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2	1	2	UEPDC	1	155.49					1	11.90			1.83	1
	S1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3		3	UEPDC	1	233.33						11.90			1.83	
UNE Loop Ra		1	Ť		1	_00.00										<u> </u>
	e DS1 Digital Loop - UNE Zone 1	1	1	UEPDC	USLDC	70.74					1	11.90			1.83	1
	e DS1 Digital Loop - UNE Zone 2		2	UEPDC	USLDC	100.54						11.90			1.83	
	e DS1 Digital Loop - UNE Zone 3		3	UEPDC	USLDC	178.38						11.90			1.83	
UNE Port Rat	te															
4-Wire	e DDITS Digital Trunk Port			UEPDC	UDD1T	54.95	464.86	259.23				11.90			1.83	
NONRECURR	RING CHARGES - CURRENTLY COMBINED															
4-Wire	e DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
- Swite	ch-as-is			UEPDC	USAC4		95.31	46.71				11.90			1.83	
	e DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
	version with DS1 Changes			UEPDC	USAWA		95.31	46.71				11.90			1.83	
	e DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
	version with Change - Trunk			UEPDC	USAWB		95.31	46.71				11.90			1.83	
ADDITIONAL																
	e DS1 Loop / 4-Wire DDITS Trunk Port - NRC -															
	equent Channel Activation/Chan - 2-Way Trunk			UEPDC	UDTTA		15.69	15.69				11.90			1.83	
	e DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent															
	nel Activation/Chan - 1-Way Outward Trunk			UEPDC	UDTTB		15.69	15.69				11.90			1.83	
	e DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel															
	ation/Chan Inward Trunk w/out DID			UEPDC	UDTTC		15.69	15.69				11.90			1.83	
	e DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan				UDTTD		45.00	45.00				44.00			1.00	
	ation Per Chan - Inward Trunk with DID			UEPDC	UDTTD		15.69	15.69				11.90			1.83	
	e DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan			UEPDC	UDTTE		15.69	15.69				11.90			1.83	
	ation / Chan - 2-Way DID w User Trans		-	UEFDC	ODITE		15.69	15.69				11.90			1.03	
	-Superframe Format		-	UEPDC	CCOSF		0.00	655.00				11.90			1.83	
	- Extended Superframe Format			UEPDC	CCOEF		0.00	655.00				11.90			1.83	
Alternate Mar				OLFDC	CCOLI		0.00	055.00			1	11.50			1.03	
	Superframe Format			UEPDC	MCOSF		0.00	0.00								
	Extended SuperFrame Format			UEPDC	MCOPO		0.00	0.00								
	umber/Trunk Group Establisment Charges	1	1	02100			0.00	0.00			1					1
	hone Number for 2-Way Trunk Group	1	1	UEPDC	UDTGX	0.00					t	11.90			1.83	
	hone Number for 1-Way Outward Trunk Group			UEPDC	UDTGY	0.00					1	11.90			1.83	
	hone Number for 1-Way Inward Trunk Group Without DID	1	1	UEPDC	UDTGZ	0.00						11.90			1.83	<u> </u>
	lumbers, Establish Trunk Group and Provide First Group	1									1					1
	DID Numbers	1	1	UEPDC	NDZ	0.00	0.00	0.00				11.90			1.83	
	lumbers for each Group of 20 DID Numbers	1	1	UEPDC	ND4	0.00						11.90			1.83	
	lumbers, Non- consecutive DID Numbers , Per Number	1		UEPDC	ND5	0.00						11.90			1.83	1
	rve Non-Consecutive DID Nos.			UEPDC	ND6	0.00	0.00	0.00				11.90			1.83	
	rve DID Numbers			UEPDC	NDV	0.00	0.00	0.00				11.90			1.83	
	S1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS	1 Digita	l Loop	with 4-Wire DDITS	Trunk Port											
	ffice Channel Mileage - Fixed rate 0-8 miles (Facilities															
Termi	nation)	1	1	UEPDC	1LNO1	88.44	105.54	98.47	21.47	19.05	1	11.90			1.83	1

IRONDLE	ED NETWORK ELEMENTS - Florida	1		1	1							-	Attachment:			bit: B
TEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremen Charge Manual S Order ve Electron Disc Ade
						Rec	Nonrec	urring	Nonrecurring	J Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMA
	Interoffice Channel Mileage - Additional rate per mile - 0-8 miles			UEPDC	1LNOA	0.1856	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities Termination)			UEPDC	1LNO2	0.00	0.00	0.00								
	Interoffice Channel Mileage - Additional rate per mile - 9-25			OLFDC	ILINO2	0.00	0.00	0.00								
	miles			UEPDC	1LNOB	0.1856	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities															
	Termination)			UEPDC	1LNO3	0.00	0.00	0.00	0.00							
	Interoffice Channel Mileage - Additional rate per mile - 25+ miles			UEPDC	1LNOC	0.1856	0.00	0.00								
	Local Number Portability, per DS0 Activated			UEPDC	LNPCP	3.15	0.00	0.00	0.00							
	Central Office Termininating Point			UEPDC	CTG	0.00										
	E DS1 LOOP WITH CHANNELIZATION WITH PORT m is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Acti					-					-					
	System can have up to 24 combinations of rates depending on			her of north word												
	System can have up to 24 combinations of rates depending on DS1 Loop	type ar	na num	iber of ports used												
UNEL	4-Wire DS1 Loop - UNE Zone 1		1	UEPMG	USLDC	70.74	0.00	0.00								
	4-Wire DS1 Loop - UNE Zone 2		2	UEPMG	USLDC	100.54	0.00	0.00								
	4-Wire DS1 Loop - UNE Zone 3		3	UEPMG	USLDC	178.38	0.00	0.00								
UNE D	DSO Channelization Capacities (D4 Channel Bank Configuration	ns)	-													
	24 DSO Channel Capacity - 1 per DS1	,		UEPMG	VUM24	118.06	0.00	0.00				11.90			1.83	
	48 DSO Channel Capacity - 1 per 2 DS1s			UEPMG	VUM48	236.12	0.00	0.00				11.90			1.83	
	96 DSO Channel Capacity -1per 4 DS1s			UEPMG	VUM96	472.24	0.00	0.00				11.90			1.83	
	144 DS0 Channel Capacity - 1 per 6 DS1s			UEPMG	VUM14	708.36	0.00	0.00				11.90			1.83	
	192 DS0 Channel Capacity -1 per 8 DS1s			UEPMG	VUM19	944.48	0.00	0.00				11.90			1.83	
	240 DS0 Channel Capacity - 1 per 10 DS1s			UEPMG	VUM20	1,180.60	0.00	0.00				11.90			1.83	
	288 DS0 Channel Capacity - 1 per 12 DS1s			UEPMG	VUM28	1,416.72	0.00	0.00				11.90			1.83	
	384 DS0 Channel Capacity - 1 per 16 DS1s			UEPMG	VUM38	1,888.96	0.00	0.00				11.90			1.83	
	480 DS0 Channel Capacity - 1 per 20 DS1s			UEPMG	VUM40	2,361.20	0.00	0.00			-	11.90			1.83	
	576 DS0 Channel Capacity -1 per 24 DS1s			UEPMG UEPMG	VUM57 VUM67	2,833.44 3.305.68	0.00	0.00				11.90			1.83 1.83	
Non B	672 DS0 Channel Capacity - 1 per 28 DS1s Recurring Charges (NRC) Associated with 4-Wire DS1 Loop with	Chong	olistio					0.00				11.90			1.83	
	imum System configuration is One (1) DS1, One (1) D4 Channe						stem									
	bles of this configuration functioning as one are considered Ac															
	NRC - Conversion (Currently Combined) with or without		1		guianen ie	oouniou			-		1					
	BellSouth Allowed Changes			UEPMG	USAC4	0.00	96.77	4.24				11.90				
Syster	m Additions at End User Locations Where 4-Wire DS1 Loop wit	th Chan	nelizat	ion with Port Comb	ination Curre	ntly Exists and										
New (Not Currently Combined) in all states, except in Density Zone 1	of Top	8 MSA	\'s												
	1 DS1/D4 Channel Bank - Additionally Add NRC for each Port															
	and Assoc Fea Activation			UEPMG	VUMD4	0.00	726.11	468.21	145.32	17.24		11.90				
Bipola	ar 8 Zero Substitution				-											
	Clear Channel Capability Format, superframe - Subsequent	1	1		00005	0.00	0.00	055.00				44.00				
_	Activity Only			UEPMG	CCOSF	0.00	0.00	655.00				11.90				
	Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only	1	1	UEPMG	CCOEF	0.00	0.00	655.00				11.90				
Altorn	ate Mark Inversion (AMI)			ULFING	CCOLI	0.00	0.00	055.00				11.50				
Altern	Superframe Format			UEPMG	MCOSF	0.00	0.00	0.00								
	Extended Superframe Format		1	UEPMG	MCOPO	0.00	0.00	0.00								
Excha	inge Ports Associated with 4-Wire DS1 Loop with Channelization	on with	Port													
	inge Ports				1											
			ľ													
	Line Side Combination Channelized PBX Trunk Port - Business			UEPPX	UEPCX	1.38	0.00	0.00	0.00	0.00		11.90			1.83	
	Line Side Outward Channelized PBX Trunk Port - Business			UEPPX	UEPOX	1.38	0.00	0.00	0.00	0.00		11.90			1.83	
			1													
	Line Side Inward Only Channelized PBX Trunk Port without DID			UEPPX	UEP1X	1.38	0.00	0.00	0.00	0.00	1	11.90			1.83	<u> </u>
	2-Wire Trunk Side Unbundled Channelized DID Trunk Port re Activations - Unbundled Loop Concentration		<u> </u>	UEPPX	UEPDM	8.71	0.00	0.00	0.00	0.00	ļ	11.90			1.83	<u> </u>
E a a t			1	1	1						1		1	1	1	1
Featu	Feature (Service) Activation for each Line Port Terminated in D4															

	ORK ELEMENTS - Florida					-							Attachment:	2	Exhil	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Feature (Se D4 Bank	Service) Activation for each Trunk Port Terminated in			UEPPX	1PQWU	0.66	70.40	18.42	56.03	10.95		11.90			1.83	
	er/ Group Establishment Charges for DID Service			UEPPX	IPQWU	0.00	78.16	18.42	56.03	10.95		11.90			1.83	
	Termination (1 per Port)			UEPPX	NDT	0.00	0.00	0.00				11.90				
	Grp and Provide 1st 20 DID Nos. (FL,GA, NC,& SC)			UEPPX	NDZ	0.00	0.00	0.00				11.90				
	pers - groups of 20 - Valid all States			UEPPX	ND4	0.00	0.00	0.00				11.90				
	ecutive DID Numbers - per number			UEPPX	ND5	0.00	0.00	0.00				11.90				
	Ion-Consecutive DID Numbers			UEPPX	ND6	0.00	0.00	0.00				11.90				
	DD Numbers			UEPPX	NDV	0.00	0.00	0.00				11.90				
Local Number Por	hber Portability - 1 per port			UEPPX	LNPCP	3.15	0.00	0.00								
	ical and Optional			ULFFA	LINFOF	3.13	0.00	0.00								
	Features Offered with Line Side Ports Only				1											
	es Available			UEPPX	UEPVF	2.26	0.00	0.00				11.90			1.83	
	P COMBINATIONS - MARKET RATES															
	Il apply where BellSouth is not required to provide	unbund	dled loo	cal switching or swi	tch ports pe	r FCC and/or St	ate Commissio	n rules.								
This includes:					L											
	oop combinations that are Currently Combined or N															
I ne Top 8 MSAS I BollSouth current	in BellSouth's region are: FL (Orlando, Ft. Lauderda tly is developing the billing capability to mechanica	ale, Mia	mi); G/	A (Atlanta); LA (New	Orleans); No	Batos in this s	Winston Salem	-Highpoint/Ch	arlotte-Gaston	a-ROCK HIII);	N (NashVIII	e). El and NC	In the interi	m whore Boll	South connot	hill Market
	shall bill the rates in the Cost-Based section preced								ig charges for	not currently t	ombineu m		. In the interi	III WHEre bein	South carmot	
	for unbundled ports includes all available features i			the warket rates an	iu reserves ti	le right to true-	up the binning t	interence.								
(USOC: URECU).	andem Switching Usage and Common Transport Us	-								-						
(USOC: URECU). For Not Currently Additional NRCs r		-								-						
(USOC: URECU). For Not Currently Additional NRCs r 2-WIRE VOICE GR UNE Port/Loop Co	Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates	-				as for each Port				-						
(USOC: URECU). For Not Currently Additional NRCs r 2-WIRE VOICE GR UNE Port/Loop Co 2-Wire VG	Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates Loop/Port Combo - Zone 1	-	in the F			23.77				-						
(USOC: URECU). For Not Currently Additional NRCs r 2-WIRE VOICE GR UNE Port/Loop Cc 2-Wire VG 2-Wire VG	Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates Loop/Port Combo - Zone 1 Loop/Port Combo - Zone 2	-	in the F			23.77 27.88				-						
(USOC: URECU). For Not Currently Additional NRCs or 2-WIRE VOICE GR UNE Port/Loop Cc 2-Wire VG 2-Wire VG 2-Wire VG	Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates Loop/Port Combo - Zone 1	-	in the F			23.77				-						
USOC: URECU). For Not Currently Additional NRCs r 2-WIRE VOICE GR UNE Port/Loop Cc 2-Wire VG 2-Wire VG 2-Wire VG 2-Wire VG UNE Loop Rates	Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates Loop/Port Combo - Zone 1 Loop/Port Combo - Zone 2 Loop/Port Combo - Zone 3	-	in the F	First and Additional	NRC columr	23.77 27.88 38.63				-						
(USOC: URECU). For Not Currently Additional NRCs r 2-WIRE VOICE GR UNE Port/Loop Cc 2-Wire VG 2-Wire VG 2-Wire VG 2-Wire Voic UNE Loop Rates 2-Wire Voic	Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates Loop/Port Combo - Zone 1 Loop/Port Combo - Zone 2 Loop/Port Combo - Zone 3 icce Grade Loop (SL1) - Zone 1	-	in the F	First and Additional	NRC column	23.77 27.88 38.63 9.77				-						
(USOC: URECU). For Not Currently Additional NRCs r 2-WIRE VOICE GR UNE Port/Loop CC 2-Wire VG 2-Wire VG UNE Loop Rates 2-Wire Voi 2-Wire Voi 2-Wire Voi	Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates Loop/Port Combo - Zone 1 Loop/Port Combo - Zone 2 Loop/Port Combo - Zone 3 ice Grade Loop (SL1) - Zone 1 ce Grade Loop (SL1) - Zone 2	-	in the F	First and Additional	NRC column	23.77 27.88 38.63 9.77 13.88				-						
(USOC: URECU). For Not Currently Additional NRCs or 2-Wire VOICE GR UNE Port/Loop Co 2-Wire VG 2-Wire VG 2-Wire VG UNE Loop Rates 2-Wire Voic 2-Wire Voic 2-Wire Voic	Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates Loop/Port Combo - Zone 1 Loop/Port Combo - Zone 2 Loop/Port Combo - Zone 3 icce Grade Loop (SL1) - Zone 1	-	in the F	First and Additional	NRC column	23.77 27.88 38.63 9.77				-						
(USOC: URECU). For Not Currently Additional NRCs r 2-WiRE VOICE GR UNE Port/Loop Cc 2-Wire VG 2-Wire VG UNE Loop Rates 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire Voic	/ Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates Loop/Port Combo - Zone 1 Loop/Port Combo - Zone 2 Loop/Port Combo - Zone 3 ice Grade Loop (SL1) - Zone 1 ice Grade Loop (SL1) - Zone 2 ice Grade Loop (SL1) - Zone 3	-	in the F	First and Additional	NRC column	23.77 27.88 38.63 9.77 13.88 24.63 14.00	USOC. For Co			-						
(USOC: URECU). For Not Currently Additional NRCs in 2-Wire VOICE GR UNE Port/Loop Cc 2-Wire VG 2-Wire VG 2-Wire VG 2-Wire VOIC 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire voic 2-Wire voic	Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates Loop/Port Combo - Zone 1 Loop/Port Combo - Zone 2 Loop/Port Combo - Zone 3 ice Grade Loop (SL1) - Zone 1 ice Grade Loop (SL1) - Zone 1 ice Grade Loop (SL1) - Zone 3 de Line Port (Res) ce unbundled port - residence ce unbundled port with Caller ID - res	-	in the F	irst and Additional UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	NRC column	s for each Port	USOC. For Co	urrently Combi		-		11.90 11.90				
(USOC: URECU). For Not Currently Additional NRCs or 2-WIRE VOICE GR UNE Port/Loop Cc 2-Wire VG 2-Wire VG 2-Wire VG 2-Wire VG 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire voic 2-Wire voic	Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates Loop/Port Combo - Zone 1 Loop/Port Combo - Zone 2 Loop/Port Combo - Zone 3 ice Grade Loop (SL1) - Zone 1 ice Grade Loop (SL1) - Zone 2 ice Grade Loop (SL1) - Zone 3 de Line Port (Res) ce unbundled port - residence	-	in the F	First and Additional	NRC column	23.77 27.88 38.63 9.77 13.88 24.63 14.00	USOC. For Co	urrently Combi		-		s are listed				
(USOC: URECU). For Not Currently Additional NRCs r 2-WIRE VOICE GR UNE Port/Loop CC 2-Wire VG 2-Wire VG UNE Loop Rates 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire voic 2-Wire voic 2-Wire voic	Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates Loop/Port Combo - Zone 1 Loop/Port Combo - Zone 2 Loop/Port Combo - Zone 3 ice Grade Loop (SL1) - Zone 1 ice Grade Loop (SL1) - Zone 2 ice Grade Loop (SL1) - Zone 2 ice Grade Loop (SL1) - Zone 3 de Line Port (Res) ce unbundled port - residence ce unbundled port with Caller ID - res ce unbundled port outgoing only - res	-	in the F	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	NRC column	23.77 27.88 38.63 9.77 13.88 24.63 14.00 14.00 14.00	USOC. For Cu	90.00 90.00 90.00		-		11.90 11.90				
(USOC: URECU). For Not Currently Additional NRCs ir 2-WIRE VOICE GR UNE Port/Loop CC 2-Wire VG 2-Wire VG UNE Loop Rates 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic	Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates Loop/Port Combo - Zone 1 Loop/Port Combo - Zone 2 Loop/Port Combo - Zone 3 Ge Grade Loop (SL1) - Zone 1 Ge Grade Loop (SL1) - Zone 1 Ge Grade Loop (SL1) - Zone 3 de Line Port (Res) ce unbundled port - residence ce unbundled port outgoing only - res ce unbundled port outgoing only - res ce unbundled Florida Area Calling with Caller ID - res	-	in the F	irst and Additional UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	NRC column	s for each Port	USOC. For Co	urrently Combi		-		11.90 11.90				
(USOC: URECU). For Not Currently Additional NRCs r 2-Wire VOICE GR UNE Port/Loop Cc 2-Wire VG 2-Wire VG UNE Loop Rates 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic	Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates Loop/Port Combo - Zone 1 Loop/Port Combo - Zone 2 Loop/Port Combo - Zone 3 ice Grade Loop (SL1) - Zone 1 ice Grade Loop (SL1) - Zone 2 ice Grade Loop (SL1) - Zone 2 ice Grade Loop (SL1) - Zone 3 de Line Port (Res) ce unbundled port - residence ce unbundled port with Caller ID - res ce unbundled port outgoing only - res	-	in the F	First and Additional	NRC column	23.77 27.88 38.63 9.77 13.88 24.63 14.00 14.00 14.00 14.00	USOC. For Co	90.00 90.00 90.00 90.00		-		11.90 11.90 11.90 11.90				
(USOC: URECU). For Not Currently Additional NRCs r 2-WIRE VOICE GR UNE Port/Loop CC 2-Wire VG 2-Wire VG UNE Loop Rates 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire voic 2-Wire	Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates i Loop/Port Combo - Zone 1 Loop/Port Combo - Zone 2 i Loop/Port Combo - Zone 3 ice Grade Loop (SL1) - Zone 1 ice Grade Loop (SL1) - Zone 1 ice Grade Loop (SL1) - Zone 2 ice Grade Loop (SL1) - Zone 2 ice Grade Loop (SL1) - Zone 3 de Line Port (Res) ce unbundled port - residence ce unbundled port outgoing only - res ce unbundled Port dutgoing only - res ce unbundled Florida Area Calling with Caller ID - res ce unbundles res, low usage line port with Caller ID	-	in the F	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	NRC column	23.77 27.88 38.63 9.77 13.88 24.63 14.00 14.00 14.00	USOC. For Cu	90.00 90.00 90.00		-		11.90 11.90				
(USOC: URECU). For Not Currently Additional NRCs r 2-WIRE VOICE GR UNE Port/Loop CC 2-Wire VG 2-Wire VG UNE Loop Rates 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire voic 2-Wire	Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates Loop/Port Combo - Zone 1 Loop/Port Combo - Zone 2 Loop/Port Combo - Zone 3 Ge Grade Loop (SL1) - Zone 1 Ge Grade Loop (SL1) - Zone 1 Ge Grade Loop (SL1) - Zone 3 de Line Port (Res) ce unbundled port - residence ce unbundled port outgoing only - res ce unbundled port outgoing only - res ce unbundled Florida Area Calling with Caller ID - res	-	in the F	irst and Additional UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	NRC column	23.77 27.88 38.63 9.77 13.88 24.63 14.00 14.00 14.00 14.00	USOC. For Co	90.00 90.00 90.00 90.00 90.00 90.00 90.00		-		11.90 11.90 11.90 11.90				
(USOC: URECU). For Not Currently Additional NRCs r 2-Wire VOCE GR UNE Port/Loop Cc 2-Wire VG 2-Wire VG 2-Wire VG 2-Wire VOC 2-Wire VOC 2-Wire VOC 2-Wire VOC 2-Wire VOC 2-Wire v	Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates Loop/Port Combo - Zone 1 Loop/Port Combo - Zone 2 Loop/Port Combo - Zone 3 ice Grade Loop (SL1) - Zone 1 ice Grade Loop (SL1) - Zone 2 ice Grade Loop (SL1) - Zone 3 de Line Port (Res) ce unbundled port - residence ce unbundled port outgoing only - res ce unbundled port outgoing only - res ce unbundled port outgoing only - res ce unbundled port with Caller ID - res ce unbundled port with Caller ID - res ce unbundled Florida Area Calling with Caller ID ce unbundled Florida Area Calling with Caller ID ce unbundled Florida extended dialing port for use	-	in the F	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	NRC column	23.77 27.88 38.63 9.77 13.88 24.63 14.00 14.00 14.00 14.00 14.00 14.00 14.00	USOC. For Co	90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00		-		11.90 11.90 11.90 11.90 11.90 11.90 11.90				
(USOC: URECU). For Not Currently Additional NRCs r 2-WIRE VOICE GR UNE Port/Loop CC 2-Wire VG 2-Wire VG UNE Loop Rates 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire voic 2-Wire	Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates Loop/Port Combo - Zone 1 Loop/Port Combo - Zone 2 Loop/Port Combo - Zone 3 de Grade Loop (SL1) - Zone 1 de Grade Loop (SL1) - Zone 1 de Line Port (Res) de Line Port (Res) de unbundled port - residence ce unbundled port outgoing only - res ce unbundled Port outgoing only - res ce unbundled Florida Area Calling with Caller ID - res ce unbundled Florida Area Calling with Caller ID - res ce unbundled Florida Area Calling with Caller ID ce unbundled Florida Area Calling with Caller ID ce unbundled Florida extended dialing port for use (7 and Caller ID	-	in the F	irst and Additional UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	NRC column	23.77 27.88 38.63 9.77 13.88 24.63 14.00 14.00 14.00 14.00 14.00	USOC. For Co	90.00 90.00 90.00 90.00 90.00 90.00 90.00		-		11.90 11.90 11.90 11.90 11.90 11.90				
(USOC: URECU). For Not Currently Additional NRCs ir 2-WIRE VOICE GR UNE Port/Loop CC 2-Wire VG 2-Wire VG UNE Loop Rates 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire voic 2-Wire	Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates Loop/Port Combo - Zone 1 Loop/Port Combo - Zone 2 Loop/Port Combo - Zone 3 Ge Grade Loop (SL1) - Zone 1 Ge Grade Loop (SL1) - Zone 1 Ge Grade Loop (SL1) - Zone 3 Ge Line Port (Res) ce unbundled port - residence ce unbundled port - residence ce unbundled port outgoing only - res ce unbundled port outgoing only - res ce unbundled Florida Area Calling with Caller ID - res ce unbundled Low Usage Line Port without Caller ID ce unbundled Florida extended dialing port for use (7 and Caller ID	-	in the F	irst and Additional UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	NRC column	s for each Port 23.77 27.88 38.63 9.77 13.88 24.63 14.00 14.	USOC. For Co	90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00		-		11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90				
(USOC: URECU). For Not Currently Additional NRCs rf 2-Wire VOCE GR UNE Port/Loop Cc 2-Wire VG 2-Wire VG 2-Wire VG 2-Wire VG 2-Wire VOCE 2-Wire VOCE 2-Wire VOCE 2-Wire VOCE 2-Wire VOCE 2-Wire VOCE 2-Wire VOCE 2-Wire vOCE 2-Wire VCE 2-WIRE VCE 2-WIRE	Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates Loop/Port Combo - Zone 1 Loop/Port Combo - Zone 2 Loop/Port Combo - Zone 3 Loop/Port (Res) ce unbundled port - residence ce unbundled port outgoing only - res ce unbundled Florida Area Calling with Caller ID - res ce unbundled Florida Area Calling with Caller ID ce unbundled Florida extended dialing port for use (7 and Caller ID ce unbundled Florida extended dialing port for use (7, without Caller ID capability	-	in the F	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	NRC column	23.77 27.88 38.63 9.77 13.88 24.63 14.00 14.00 14.00 14.00 14.00 14.00 14.00	USOC. For Co	90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00		-		11.90 11.90 11.90 11.90 11.90 11.90 11.90				
(USOC: URECU). For Not Currently Additional NRCs r 2-WiRE VOICE GR UNE Port/Loop CC 2-Wire VG 2-Wire VG UNE Loop Rates 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire voic 2-Wire	Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates i Loop/Port Combo - Zone 1 i.cop/Port Combo - Zone 2 i.cop/Port Combo - Zone 3 ice Grade Loop (SL1) - Zone 1 ice Grade Loop (SL1) - Zone 2 ice Grade Loop (SL1) - Zone 3 de Line Port (Res) ce unbundled port - residence ce unbundled port - residence ce unbundled port outgoing only - res ce unbundled port outgoing only - res ce unbundled Florida Area Calling with Caller ID - res ce unbundled Florida extended dialing port for use {7 and Caller ID ce unbundled Florida Area Calling Port for use {7, without Caller ID capability ce unbundled Florida Area Calling Port without Caller	-	in the F	irst and Additional UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	NRC column	23.77 27.88 38.63 9.77 13.88 24.63 14.00 14.00 14.00 14.00 14.00 14.00 14.00 14.00	USOC. For Co	90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00		-		11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90				
(USOC: URECU). For Not Currently Additional NRCs r 2-WIRE VOICE GR UNE Port/Loop CC 2-Wire VG 2-Wire VG UNE Loop Rates 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic Capability 2-Wire voic With CREX 2-Wire voic With CREX 2-Wire voic 1D Capability	Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates Loop/Port Combo - Zone 1 Loop/Port Combo - Zone 2 Loop/Port Combo - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combon - Zone 3 Combon - Zone 4	-	in the F	irst and Additional UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	NRC column	s for each Port 23.77 27.88 38.63 9.77 13.88 24.63 14.00 14.	USOC. For Co	90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00		-		11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90				
(USOC: URECU). For Not Currently Additional NRCs ir 2-WIRE VOICE GR UNE Port/Loop CC 2-Wire VG 2-Wire VG 2-Wire VG 2-Wire VG 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 1-2-Wire voic 2-Wire voic 1-2-Wire voic 2-Wire voic 1-2-Wire voic 2-Wire voic 1-2-Wire voic 1-2-Wire voic 2-Wire voic 1-2-Wire voic 1-2-Wire voic 1-2-Wire voic 2-Wire voic 1-2-	Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates Loop/Port Combo - Zone 1 Loop/Port Combo - Zone 2 Loop/Port Combo - Zone 3 Combined Comboned Combined Comboned	-	in the F	irst and Additional UEPRX	NRC column	s for each Port	USOC. For Co	90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00		-		11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90				
(USOC: URECU). For Not Currently Additional NRCs ir 2-WIRE VOICE GR UNE Port/Loop CC 2-Wire VG 2-Wire VG 2-Wire VG 2-Wire VG 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 1-2-Wire voic 2-Wire voic 1-2-Wire voic 2-Wire voic 1-2-Wire voic 2-Wire voic 1-2-Wire voic 1-2-Wire voic 2-Wire voic 1-2-Wire voic 1-2-Wire voic 1-2-Wire voic 2-Wire voic 1-2-	Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates Loop/Port Combo - Zone 1 Loop/Port Combo - Zone 2 Loop/Port Combo - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combined Combon - Zone 3 Combon - Zone 3 Combon - Zone 4	-	in the F	irst and Additional UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	NRC column	23.77 27.88 38.63 9.77 13.88 24.63 14.00 14.00 14.00 14.00 14.00 14.00 14.00 14.00	USOC. For Co	90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00		-		11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90				
(USOC: URECU). For Not Currently Additional NRCs r 2-WIRE VOICE GR UNE Port/Loop CC 2-Wire VG 2-Wire VG UNE Loop Rates 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire voic 1-D Capabili LOCAL NUMBER	Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates Loop/Port Combo - Zone 1 Loop/Port Combo - Zone 2 Loop/Port Combo - Zone 3 Combined Scenario (SL1) - Zone 1 Combored Scenario (SL1) - Zone 3 Combined Sc	-	in the F	irst and Additional UEPRX	NRC column	s for each Port	USOC. For Co	90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00		-		11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90				
(USOC: URECU). For Not Currently Additional NRCs ir 2-WIRE VOICE GR UNE Port/Loop CC 2-Wire VG 2-Wire VG 2-Wire VG 2-Wire VG 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 1-2-Wire voic 2-Wire voic 1-2-Wire voic 2-Wire voic 1-2-Wire voic 2-Wire voic 1-2-Wire voic 2-Wire voic 1-2-Wire vo	Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates Loop/Port Combo - Zone 1 Loop/Port Combo - Zone 2 Loop/Port Combo - Zone 3 Combined Scenario (SL1) - Zone 1 Combored Scenario (SL1) - Zone 3 Combined Sc	-	in the F	irst and Additional UEPRX	NRC column	s for each Port 23.77 27.88 38.63 9.77 13.88 24.63 14.00 14.	USOC. For Co	90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00		-		11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90				
(USOC: URECU). For Not Currently Additional NRCs r 2-WiRE VOICE GR UNE Port/Loop Cc 2-Wire VG 2-Wire VG 2-Wire VG 2-Wire VG 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire Voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 2-Wire voic 0 2-Wire voic 0 2-Wire voic 0 2-Wire voic 0 0 0 0 0 0 0 0	Combined scenarios the Nonrecurring charges are may apply also and are categorized accordingly. RADE LOOP WITH 2-WIRE LINE PORT (RES) ombination Rates Loop/Port Combo - Zone 1 Loop/Port Combo - Zone 2 Loop/Port Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 4 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 4 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 3 Combo - Zone 4 Combo - Zone 3 Combo - Zone 4 Combo - Zone 3 Combo - Zone 4 Combo - Zone 3 Combo - Zone 4 Combo	-	in the F	irst and Additional UEPRX	NRC column	s for each Port 23.77 27.88 38.63 9.77 13.88 24.63 14.00 14.	USOC. For Co	90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00		-		11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90				

JNBUNDLED N	NETWORK ELEMENTS - Florida											-	Attachment:			bit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Charge -	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Nire Voice Grade Loop / Line Port Combination - Switch with															
	ange			UEPRX	USACC		41.50	41.50				11.90				L
ADDITION	C - 2-Wire Voice Grade Loop/Line Port Combination -				_				-							<u> </u>
	bsequent			UEPRX	USAS2		0.00	0.00				11.90				
	DICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)			OLINA	00402		0.00	0.00				11.30				
	Loop Combination Rates															
	Wire VG Loop/Port Combo - Zone 1		1			23.77										
	Wire VG Loop/Port Combo - Zone 2		2			27.88										
	Nire VG Loop/Port Combo - Zone 3		3			38.63										
UNE Loop																ļ
	Nire Voice Grade Loop (SL1) - Zone 1		1	UEPBX	UEPLX	9.77			├ ───┤							<u> </u>
	Nire Voice Grade Loop (SL1) - Zone 2 Nire Voice Grade Loop (SL1) - Zone 3		2	UEPBX UEPBX	UEPLX UEPLX	13.88 24.63										I
	ice Grade Line Port (Bus)		3	UEFDA	UEPLA	24.03										
	Wire voice unbundled port without Caller ID - bus			UEPBX	UEPBL	14.00	90.00	90.00				11.90				
	Wire voice unbundled port with Caller + E484 ID - bus			UEPBX	UEPBC	14.00	90.00	90.00				11.90				
	Vire voice unbundled port outgoing only - bus			UEPBX	UEPBO	14.00	90.00	90.00				11.90				
2-V	Wire voice unbundled Incoming Only Port without Caller ID															
	pability			UEPBX	UEPBE	14.00	90.00	90.00				11.90				
	JMBER PORTABILITY															
	cal Number Portability (1 per port)			UEPBX	LNPCX	0.35										
NONRECU	IRRING CHARGES - CURRENTLY COMBINED															ļ
	Nire Voice Grade Loop / Line Port Combination - Switch-as-is			UEPBX	USAC2		41.50	41.50				11.90				
	Nire Voice Grade Loop / Line Port Combination - Switch with			UEPBX	USACC		44.50	44.50				44.00				
ADDITION	ange			UEPBA	USACC		41.50	41.50				11.90				
	C - 2-Wire Voice Grade Loop/Line Port Combination -															
	bsequent			UEPBX	USAS2		0.00	0.00				11.90				
	DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)															
	Loop Combination Rates															
	Vire VG Loop/Port Combo - Zone 1		1			23.77										
	Wire VG Loop/Port Combo - Zone 2		2			27.88										
	Vire VG Loop/Port Combo - Zone 3		3			38.63										
UNE Loop																<u> </u>
	Nire Voice Grade Loop (SL1) - Zone 1		1	UEPRG	UEPLX	9.77										
	Nire Voice Grade Loop (SL1) - Zone 2 Nire Voice Grade Loop (SL1) - Zone 3		2	UEPRG UEPRG	UEPLX UEPLX	13.88 24.63										
	ice Grade Line Port Rates (RES - PBX)		5	OLFKG	ULFLA	24.03										-
	Wire VG Unbundled Combination 2-Way PBX Trunk Port -				1				<u>├</u> ───┼							<u> </u>
Re				UEPRG	UEPRD	14.00	90.00	90.00				11.90				1
LOCAL NU	JMBER PORTABILITY	1			1										1	<u> </u>
	cal Number Portability (1 per port)			UEPRG	LNPCP	3.15	0.00	0.00								
FEATURE																
	Features Offered			UEPRG	UEPVF	0.00	0.00	0.00				11.90				Ļ
NONRECU	IRRING CHARGES - CURRENTLY COMBINED															
	Nire Voice Grade Loop/ Line Port Combination - Switch-As-Is			UEPRG	USAC2		41.50	41.50				11.90				
	Nire Voice Grade Loop/ Line Port Combination - Switch with				110400		11 50	11 50				44.00				
ADDITION				UEPRG	USACC		41.50	41.50	├			11.90				
	Nire Loop/Line Side Port Combination - Non feature -								<u> </u>							<u> </u>
	bsequent Activity- Nonrecurring				1		0.00	0.00				11.90				1
	BX Subsequent Activity - Change/Rearrange Multiline Hunt						0.00	0.00	<u>├</u>			11.30				
	Oup				1		7.09	7.09				11.90				1
	DICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)	1			1										1	<u> </u>
	Loop Combination Rates															
2-V	Wire VG Loop/Port Combo - Zone 1		1			23.77										

	NETWORK ELEMENTS - Florida	1		1							-	-	Attachment:			bit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
							Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Wire VG Loop/Port Combo - Zone 2		2			27.88										
	Wire VG Loop/Port Combo - Zone 3		3			38.63										
UNE Loop	p Rates -Wire Voice Grade Loop (SL1) - Zone 1		1	UEPPX	UEPLX	9.77										
	Wire Voice Grade Loop (SL1) - Zone 2		2	UEPPX	UEPLX	13.88										
	Wire Voice Grade Loop (SL1) - Zone 2		3		UEPLX	24.63										
	bice Grade Line Port Rates (BUS - PBX)															
	ne Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPPX	UEPPC	14.00	90.00	90.00				11.90				
	ne Side Unbundled Outward PBX Trunk Port - Bus			UEPPX	UEPPO	14.00	90.00	90.00				11.90				
	ne Side Unbundled Incoming PBX Trunk Port - Bus	l		UEPPX	UEPP1	14.00	90.00	90.00			ļ	11.90				
	Wire Voice Unbundled PBX LD Terminal Ports			UEPPX	UEPLD	14.00	90.00	90.00				11.90				
	Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPPX	UEPXA	14.00	90.00	90.00			I	11.90				
	Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPPX	UEPXB	14.00	90.00	90.00				11.90				
	Wire Voice Unbundled PBX LD DDD Terminals Port Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPPX UEPPX	UEPXC UEPXD	14.00 14.00	90.00 90.00	90.00 90.00				11.90 11.90				
	Wire Voice Unbundled PBX LD Terminal Switchboard IDD			UEFFA	UEFAD	14.00	90.00	90.00				11.90				
	apable Port			UEPPX	UEPXE	14.00	90.00	90.00				11.90				
	-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy			OLITX		14.00	30.00	30.00				11.30				
	dministrative Calling Port			UEPPX	UEPXL	14.00	90.00	90.00				11.90				
	Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy			0EI I X	021742	1 1100	00.00	00.00				11.00				
	oom Calling Port			UEPPX	UEPXM	14.00	90.00	90.00				11.90				
	Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital															
	iscount Room Calling Port			UEPPX	UEPXO	14.00	90.00	90.00				11.90				
	Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPPX	UEPXS	14.00	90.00	90.00				11.90				
	UMBER PORTABILITY															
	ocal Number Portability (1 per port)			UEPPX	LNPCP	3.15	0.00	0.00								
FEATURE																
	Il Features Offered			UEPPX	UEPVF	0.00	0.00	0.00				11.90				
NONRECO	URRING CHARGES - CURRENTLY COMBINED															
2	Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is			UEPPX	USAC2		41.50	41.50				11.90				
	Wire Voice Grade Loop/ Line Port Combination - Switch with			ULFFX	USACZ		41.50	41.50				11.90				
	hange			UEPPX	USACC		41.50	41.50				11.90				
	VAL NRCs															
2-	Wire Voice Grade Loop/ Line Port Combination - Subsequent			UEPPX	USAS2	0.00	0.00	0.00				11.90				
	Wire Loop/Line Side Port Combination - Non feature -															
	ubsequent Activity- Nonrecurring						0.00	0.00				11.90				
	BX Subsequent Activity - Change/Rearrange Multiline Hunt															
	roup						7.09	7.09				11.90				
	OICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN POR	RT														
	/Loop Combination Rates Wire VG Coin Port/Loop Combo – Zone 1		1			23.77										
	-Wire VG Coin Port/Loop Combo – Zone 1 -Wire VG Coin Port/Loop Combo – Zone 2		1			23.77				1						
	Wire VG Coin Port/Loop Combo – Zone 2		2			38.63										
UNE Loop			5			30.03										
	-Wire Voice Grade Loop (SL1) - Zone 1	1	1	UEPCO	UEPLX	9.77					1					
	Wire Voice Grade Loop (SL1) - Zone 2	1	2	UEPCO	UEPLX	13.88			1		1				1	
	Wire Voice Grade Loop (SL1) - Zone 3	1		UEPCO	UEPLX	24.63					1				ĺ	
2-Wire Vo	pice Grade Line Port Rates (Coin)															
	Wire Coin 2-Way with Operator Screening and Blocking: 011,															
90	00/976, 1+DDD (FL)			UEPCO	UEP2F	14.00	90.00	90.00				11.90				
	Wire Coin 2-Way with Operator Screening and 011 Blocking										1					
	FL)			UEPCO	UEPFA	14.00	90.00	90.00				11.90				
	Wire Coin 2-Way with Operator Screening and Blocking:															
	00/976, 1+DDD, 011+, and Local (FL)			UEPCO	UEPCG	14.00	90.00	90.00				11.90				
	Wire Coin Outward with Operator Screening and 011 Blocking	1	1	1							1	1			1	1

JNBUNDLE	ED NETWORK ELEMENTS - Florida											-	Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add ⁴
						Rec	Nonrec		Nonrecurring					Rates(\$)	_	-
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+ (FL)			UEPCO	UEPOF	14.00	90.00	90.00				11.90				
	2-Wire Coin Outward with Operator Screening and Blocking:			UEFCO	UEFUF	14.00	90.00	90.00				11.90				
	900/976, 1+DDD, 011+, and Local (FL, GA)			UEPCO	UEPCQ	14.00	90.00	90.00				11.90				
LOCA	L NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPCO	LNPCX	0.35										
NONR	ECURRING CHARGES - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is			UEPCO	USAC2		41.50	41.50				11.90				
	2-Wire Voice Grade Loop/ Line Port Combination - Switch with															
40017	Change			UEPCO	USACC		41.50	41.50			-					
ADDI					+											
	2-Wire Voice Grade Loop/ Line Port Combination - Subsequent			UEPCO	USAS2		0.00	0.00				11.90				
2-WIR	E VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE				03432		0.00	0.00				11.90				
	Port/Loop Combination Rates		5 (1 1						1					
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1		1	26.24					1					
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			31.40										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			44.87										
UNE L	oop Rates															
	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFR	UECF2	12.24										
	2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFR	UECF2	17.40										
	2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFR	UECF2	30.87										
2-Wire	e Voice Grade Line Port Rates (Res)															
	2-Wire voice unbundled port - residence			UEPFR	UEPRL	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire voice unbundled port with Caller ID - res			UEPFR	UEPRC	14.00	180.00	110.00	85.00	20.00	-	11.90				
	2-Wire voice unbundled port outgoing only - res			UEPFR	UEPRO	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire voice unbundled Florida Area Calling with Caller ID - res			UEPFR	UEPAF	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire voice unbuilding right and a real caring with Caller ID - res			OLITIK		14.00	100.00	110.00	05.00	20.00		11.30				
	(LUM)			UEPFR	UEPAP	14.00	180.00	110.00	85.00	20.00		11.90				
INTER	ROFFICE TRANSPORT				0											
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility															
	Termination			UEPFR	U1TV2	25.32	47.35	31.78								
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile															
	or Fraction Mile			UEPFR	1L5XX	0.0091										
FEAT																
	All Features Offered			UEPFR	UEPVF	0.00	0.00	0.00				11.90				
LOCA					LNDOX											
NOND	Local Number Portability (1 per port)			UEPFR	LNPCX	0.35										
NONR	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port				+											
	Combination - Conversion - Switch-as-is			UEPFR	USAC2		16.97	3.73				11.90				
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port				00,02		10.37	5.15				11.30				
	Combination - Conversion - Switch-With-Change			UEPFR	USACC		16.97	3.73				11.90				
2-WIR	E VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE	LINEF	PORT (1				ĺ	
	Port/Loop Combination Rates															
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			26.24										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			31.40										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			44.87										
UNEL	Loop Rates					10.07										
	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFB	UECF2	12.24					ļ	L				
	2-Wire Voice Grade Loop (SL2) - Zone 2 2-Wire Voice Grade Loop (SL2) - Zone 3		2	UEPFB UEPFB	UECF2 UECF2	17.40 30.87					-					
2-Wire	e Voice Grade Line Port (Bus)		5			50.67					<u> </u>					
2-10116	2-Wire voice unbundled port without Caller ID - bus			UEPFB	UEPBL	14.00	180.00	110.00	85.00	20.00		11.90				
-	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPFB	UEPBC	14.00	180.00	110.00	85.00	20.00	1	11.90			1	
	2-Wire voice unbundled port outgoing only - bus			UEPFB	UEPBO	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire voice unbundled incoming only port with Caller ID - Bus			UEPFB	UEPB1	14.00	180.00	110.00	85.00	20.00	1	11.90				

CATEGORY RATE ELEMENTS Intering m Zone BCS USOC RATES(5) Soc Order Submitted Submitted Decision Soc Order Part S(5) Soc Order Part S(5) Soc Order Submitted Submitted Decision CATEGORY RATE ELEMENTS Intering m Rec Norrecurring Print Norrecurring Disconnect First SoleCC	VC Order ubmitted Aanually per LSR SOMAN 11.90 11.90	ubmitted Submitte Elec Manually per LSR per LSR SOMEC SOMAN	omitted Charge - inually Manual Svc I or LSR Order vs. Electronic- 1st OSS F	Incremental Charge -		Charge -
CATEGORY RATE ELEMENTS Intering m Zore BCS USOC FATES(S) RATES(S) Lesson per LSR Manually per LSR <th>Annually per LSR SOMAN 11.90 11.90</th> <th>Elec Manually per LSR per LSR</th> <th>nually Manual Svc I or LSR Order vs. Electronic- 1st OSS F</th> <th>Manual Svc Order vs. Electronic- Add'l Rates(\$)</th> <th>Manual Svc Order vs. Electronic- Disc 1st</th> <th>Manual Svc Order vs. Electronic- Disc Add'l</th>	Annually per LSR SOMAN 11.90 11.90	Elec Manually per LSR per LSR	nually Manual Svc I or LSR Order vs. Electronic- 1st OSS F	Manual Svc Order vs. Electronic- Add'l Rates(\$)	Manual Svc Order vs. Electronic- Disc 1st	Manual Svc Order vs. Electronic- Disc Add'l
CATEGORY RATE ELEMENTS m Zone BCS USOC RATES(6) per LSR per LS	SOMAN 5000000000000000000000000000000000000	SOMEC SOMAN	r LSR Order vs. Electronic- 1st OSS F	Order vs. Electronic- Add'l Rates(\$)	Order vs. Electronic- Disc 1st	Order vs. Electronic- Disc Add'l
CATEGORY RATE ELEMENTS Im Zone BCS USOC RATES(5) per LSR per L	SOMAN 5000000000000000000000000000000000000	SOMEC SOMAN	r LSR Order vs. Electronic- 1st OSS F	Electronic- Add'l Rates(\$)	Electronic- Disc 1st	Electronic- Disc Add'l
Image: Image:	SOMAN	SOMEC SOMAN	Electronic- 1st OSS F	Add'l Rates(\$)	Disc 1st	Disc Add'l
Image: Constraint of the second sec	11.90		OSS F	Rates(\$)		
Image: Constraint of the second sec	11.90				SOMAN	
Image: Constraint of the second sec	11.90				SOMAN	SOMAN
LOCAL NUMBER PORTABILITY UEPF UEPC 0.35 INTEROFFICE TRANSPORT UEPFB ULPCX 0.35 INTEROFICE TRANSPORT UEPFB ULPCX 0.35 Interdifie Transport - Dedicated - 2 Wire Voice Grade - Per Mile UEPFB ULPFB 0.001 FlattureS UEPFB UEPFB UEPFB 0.001 NONRECURRING CHARGES (NRC9) - CURRENTLY COMBINED UEPFB UEPFB UEPFB 0.001	11.90		JMAN SOMAN	SOMAN	SOMAN	SOMAN
Local Number Portability (1 per port) UEPFB UNPCX 0.35 Interdifice Transport - Dedicated - 2 Wire Voice Grade - Facility UEPFB U1TV2 25.32 47.35 31.76 Interdifice Transport - Dedicated - 2 Wire Voice Grade - Facility UEPFB U1TV2 25.32 47.35 31.78	11.90	11.90				<u> </u>
INTEROFFICE TRANSPORT Image: Control Transport - Dedicated - 2 Wire Voice Grade - Per Mile UEPFB UTV2 25.32 47.35 31.78 Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile UEPFB UEPFB	11.90	11.90				
Interdifice Transport - Dedicated - 2 Wire Voice Grade - Facility UEPFB U1V/2 25.2 47.35 31.78 Interdifice Transport - Dedicated - 2 Wire Voice Grade - Per Mile or Fraction Mile UEPFB U1V/2 25.32 47.35 31.78 If adatures Offreid UEPFB UEPFB UEPFB 0 <td>11.90</td> <td>11.90</td> <td></td> <td></td> <td></td> <td>1</td>	11.90	11.90				1
Image: Image:	11.90	11.90				
Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile UEPFB 1L5XX 0.0091 FEATURES UEPFB L5XX 0.0091 0 0 0 NONRECURRING CHARGES (MRCs) - CURRENTLY COMBINED UEPFB UEPFB UEPFB 0.000 0.00 0.00 0.00 2:Wire Loop / Dedicated IO Transport / 2 Wire Line Port UEPFB USAC2 16.97 3.73 11.90 2:Wire Loop / Dedicated IO Transport / 2 Wire Line Port UEPFB USAC2 16.97 3.73 11.90 2:Wire Voice GRADE LOOP WIT1 2-WIRE LINE PORT (BUS - PBX) UEPFB USACC 16.97 3.73 11.90 UNE Port/Loop Combination Rates UEPFB USACC 16.97 3.73 11.90 UNE Port/Loop Combination Rates UEPFB USACC 16.97 3.73 11.90 UNE Port/Loop Combination Rates UEPFB UEAPC 14.97 11.90 11.90 11.90 UWE Port/Loop Combination Rates UEPFB UEAPC 14.47 11.90 11.90 11.90 11.90 11.90 11.90	11.90	11.90				
or Fraction Mile UEPFB 1L5XX 0.0091 FEATURES III Features Offered UEPVF 0.00 0.00 0.00 11.80 NONRECURRING CHARGES (NRCs): CURRENTLY COMBINED UEPFB UEPVF 0.00 0.00 0.00 0.00 2-Wire Loop / Dedicated f0 Transport /2 Wire Line Port UEPFB USAC2 16.97 3.73 11.90 2-Wire Loop / Dedicated f0 Transport /2 Wire Line Port UEPFB USAC2 16.97 3.73 11.90 2-Wire Voice GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX) UEPFB USACC 16.97 3.73 11.90 2-Wire Voice GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX) UEPFB USACC 16.97 3.73 11.90 2-Wire Voice GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX) UEPFB USACC 16.97 3.73 11.90 2-Wire Voice Grade Loop (SL2) - Combo 2 zone 2 2 31.40 - - - 2-Wire Voice Grade Loop (SL2) - Zone 2 2 UEPFP UECF2 17.40 - - 2-Wire Voice Gra	11.90	11.90				<u> </u>
FEATURES U U U U U U NONRECURRING CHARGES (MRCg) - CURRENTLY COMBINED UEPPB UEPVF 0.00 0.00 0.00 0.11.00 2.Wire Loop Chadcadd IO Transport / 2 Wire Line Port UEPFB USAC2 16.97 3.73 11.90 2.Wire Loop Checkradt IO Transport / 2 Wire Line Port UEPFB USACC 16.97 3.73 11.90 2.Wire Voice Grabe LOOP WITH 2 Wine Line Port (BUS - PBX) UEPFB USACC 16.97 3.73 11.90 2.Wire Voice Grabe LOOP WITH 2 WIRE LINE PORT (BUS - PBX) UEPFB USACC 16.97 3.73 11.90 2.Wire Voice Grabe Loop NIT 2 WIRE LINE PORT (BUS - PBX) UEPFB UEPFB USACC 16.97 3.73 11.90 2.Wire Voice Grabe Loop NITA proprived Combo - Zone 3 3 44.87 2.Wire Voice Grade Loop (32) - Zone 1 1 UEPFP UEPFP 2.Wire Voice Grade Loop (32) - Zone 3 3 UEPFP UEPFP 10.00 85.00	11.90	11.90				
IMI Features Offered UEPFB UEPVF 0.00 0.00 0.00 0.00 NONRECERNING CHARGES (INCE) - CURRENTLY COMBINED Image: Combination - Conversion - Switch-as-is UEPFB USAC2 16.97 3.73 11.90 2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port UEPFB USAC2 16.97 3.73 11.90 2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port UEPFB USACC 16.97 3.73 11.90 2-Wire Voice GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX) UEPFB USACC 16.97 3.73 11.90 2-Wire Voice DRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX) UEPFB USACC 16.97 3.73 11.90 2-Wire Voice DRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX) UEPFB USACC 16.97 3.73 11.90 2-Wire Voice Grade Loop (SL2) - Cane 1 1 1 E2.84 2-Wire Voice Grade Loop (SL2) - Zone 3 3 44.87 2-Wire Voice Grade Loop (SL2) - Zone 3 3 UEPFP UECF2 17.40	11.90	11.90				
NONRECURRING CHARGES (NRCs) - CURRENTLY COMBINED Image: Combination and control of the Port Combination - Switch-as-is Image: Combination - Switch-as-is<	11.90		11.90			
Image: Construction Switch-askie LUEPFB USAC2 16.97 3.73 11.90 2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port UEPFB USAC2 16.97 3.73 11.90 2-Wire Voice GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX) UEPFB USACC 16.97 3.73 11.90 2-Wire Voice GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX) UEPFB USACC 16.97 3.73 11.90 2-Wire Voice GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX) UEPFB USACC 16.97 3.73 11.90 2-Wire Voice Combination - Zone 1 1 26.24 11.90 2-Wire Voice Loop/IO TranportPort Combo - Zone 2 2 31.40 URE Loop Rates 2-Wire Voice Grade Loop (SL2) - Zone 1 1 UEPFP UECF2 17.40 2-Wire Voice Grade Loop (SL2) - Zone 3 3 UEPFP UECF2 30.87 2-Wire Voice Grade Loop (SL2						
Combination - Conversion - Switch-as-is UEPFB USAC2 16.97 3.73 11.90 2-Wire Loop Dedicated D Transport / 2Wire Line Port Combination - Conversion - Switch with change UEPFB USACC 16.97 3.73 11.90 2-Wire Voice GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX) 11.90 2-Wire Voice GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX) 11.90 2-Wire Voice Orabination Rates						
Combination - Conversion - Switch with change UEPFB USACC 16.97 3.73 11.90 2-Wire VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)	11.90	11.90	11.90			
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX) UNP Port/Loop Combination Rates	11.90					
UNE PortLoop Combination Rates Image: Combination Rates I		11.90	11.90			
2-Wire VG Loop/IO Tranport/Port Combo - Zone 1 1 26.24 2-Wire VG Loop/IO Tranport/Port Combo - Zone 2 2 31.40 2-Wire VG Loop/IO Tranport/Port Combo - Zone 3 3 44.87 UNE Loop Rates 2-Wire Voice Grade Loop (SL2) - Zone 1 1 UEPFP UECF2 17.40 2-Wire Voice Grade Loop (SL2) - Zone 3 3 UEPFP UECF2 30.87 2-Wire Voice Grade Loop (SL2) - Xone 3 3 UEPFP UECF2 30.87						
2-Wire VG Loop/IO Tranport/Port Combo - Zone 2 2 31.40 2-Wire VG Loop/IO Tranport/Port Combo - Zone 3 3 44.87 UNE Loop Rates 1 1 1 UEPFP UECF2 12.24 2-Wire Voice Grade Loop (SL2) - Zone 1 1 1 UEPFP UECF2 17.40 2-Wire Voice Grade Loop (SL2) - Zone 2 2 UEPFP UECF2 30.87 2-Wire Voice Grade Loop (SL2) - Zone 3 3 UEPFP UECF2 30.87 2-Wire Voice Grade Loop (SL2) - Zone 3 3 UEPFP UEPFP						
2-Wire VG Loop/IO Tranport/Port Combo - Zone 3 3 44.87 UNE Loop Rates - <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>						
UNE Loop Rates Image: Control of SL2) - Zone 1 Image: Control of SL2) - Zone 1 Image: Control of SL2) - Zone 2 Image: Control of SL2) - Zone 2 Image: Control of SL2) - Zone 2 Image: Control of SL2) - Zone 2 Image: Control of SL2) - Zone 2 Image: Control of SL2) - Zone 2 Image: Control of SL2) - Zone 3 Image: Contro of SL2) - Zone 3<						
Image: Provide Grade Loop (SL2) - Zone 1 1 UEPFP UECF2 12.24 Image: Provide Grade Loop (SL2) - Zone 2 2 UEPFP UECF2 17.40 Image: Provide Grade Loop (SL2) - Zone 3 3 UEPFP UECF2 17.40 Image: Provide Grade Loop (SL2) - Zone 3 3 UEPFP UECF2 30.87 Image: Provide Grade Loop (SL2) - Zone 3 3 UEPFP UECF2 30.87 Image: Provide Grade Loop (SL2) - Zone 3 3 UEPFP UECF2 30.87 Image: Provide Grade Loop (SL2) - Zone 3 3 UEPFP UECF2 30.87 Image: Provide Grade Loop (SL2) - Zone 3 3 UEPFP UECF2 30.87 Image: Provide Grade Loop (SL2) - Zone 3 3 UEPFP UECF2 30.87 Image: Provide Grade Loop (SL2) - Zone 3 3 UEPFP UECF2 30.87 Image: Provide Grade Loop (SL2) - Zone 3 3 UEPFP UEP						
2-Wire Voice Grade Loop (SL2) - Zone 2 2 UEPFP UECF2 17.40 2-Wire Voice Grade Loop (SL2) - Zone 3 3 UEPFP UECF2 30.87 2-Wire Voice Grade Loop (SL2) - Zone 3 3 UEPFP UECF2 30.87 2-Wire Voice Grade Loop (SL2) - Zone 3 3 UEPFP UECF2 30.87 Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus UEPFP UEPFP UEPPO 14.00 180.00 110.00 85.00 20.00 11.90 Line Side Unbundled Incoming PBX Trunk Port - Bus UEPFP UEPFP UEPPO 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Ports UEPFP UEPFP UEPXA 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Ports UEPFP UEPXB 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD DDD T						
2-Wire Voice Grade Loop (SL2) - Zone 3 3 UEPFP UECF2 30.87 2-Wire Voice Grade Line Port Rates (BUS - PBX) Image: Constraint of the state (BUS -						
2-Wire Voice Grade Line Port Rates (BUS - PBX) UEPFP UEPFP UEPPC 14.00 180.00 110.00 85.00 20.00 11.90 Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus UEPFP UEPPP UEPPO 14.00 180.00 110.00 85.00 20.00 11.90 Line Side Unbundled Incoming PBX Trunk Port - Bus UEPFP UEPPP UEPP1 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Ports UEPFP UEPFP UEPLD 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Ports UEPFP UEPFP UEPXA 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Ports UEPFP UEPXB 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD DDD Terminals Port UEPFP UEPXB 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice U						
Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus UEPFP UEPPC 14.00 180.00 110.00 85.00 20.00 11.90 Line Side Unbundled Outward PBX Trunk Port - Bus UEPFP UEPPC 14.00 180.00 110.00 85.00 20.00 11.90 Line Side Unbundled Incoming PBX Trunk Port - Bus UEPFP UEPPP UEPP1 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Ports UEPFP UEPPP UEPLD 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Ports UEPFP UEPFP UEPXA 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD DDD Terminal Ports UEPFP UEPKA 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD DDD Terminal Sort UEPFP UEPKA 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port						
Line Side Unbundled Outward PBX Trunk Port - Bus UEPFP UEPPO 14.00 180.00 110.00 85.00 20.00 11.90 Line Side Unbundled Incoming PBX Trunk Port - Bus UEPFP UEPP1 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Ports UEPFP UEPLD 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 2-Way Combination PBX Usage Port UEPFP UEPPA 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports UEPFP UEPXA 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD DDD Terminals Port UEPFP UEPXB 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port UEPFP UEPXD 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD UEPFP UEPXE						
Line Side Unbundled Outward PBX Trunk Port - Bus UEPFP UEPPO 14.00 180.00 110.00 85.00 20.00 11.90 Line Side Unbundled Incoming PBX Trunk Port - Bus UEPFP UEPP1 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Ports UEPFP UEPLD 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 2-Way Combination PBX Usage Port UEPFP UEPPA 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports UEPFP UEPXA 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD DDD Terminal Sort UEPFP UEPXB 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port UEPFP UEPXD 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD UEPFP UEPXE	11.00	11.00	11.00			
Line Side Unbundled Incoming PBX Trunk Port - Bus UEPFP UEPP1 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Ports UEPFP UEPLD 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX Toll Terminal Ports UEPFP UEPXA 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports UEPFP UEPXB 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX TD Terminal Sort UEPFP UEPXB 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD DDD Terminal Switchboard Port UEPFP UEPXD 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD UEPFP UEPKD 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 2-Way PBX Hote//Hospital Economy UEPFP UEPKL						
2-Wire Voice Unbundled PBX LD Terminal Ports UEPFP UEPLD 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 2-Way Combination PBX Usage Port UEPFP UEPXA 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports UEPFP UEPXB 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX Toll Terminal Soft UEPFP UEPXB 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD DDD Terminals Port UEPFP UEPXD 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port UEPFP UEPXD 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD UEPFP UEPKE 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy UEPFP UEPKL						
2-Wire Voice Unbundled 2-Way Combination PBX Usage Port UEPFP UEPXA 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports UEPFP UEPXB 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD DDD Terminal Fort UEPFP UEPXC 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port UEPFP UEPXD 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port UEPFP UEPXD 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD UEPFP UEPXE 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy UEPFP UEPXL 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy UEPFP						
2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports UEPFP UEPXB 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD DDD Terminals Port UEPFP UEPXC 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port UEPFP UEPXD 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD UEPFP UEPXD 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD UEPFP UEPKE 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy UEPFP UEPXL 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy UEPFP UEPXL 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy UEPFP						<u> </u>
2-Wire Voice Unbundled PBX LD DDD Terminals Port UEPFP UEPXC 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port UEPFP UEPXD 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD UEPFP UEPXE 14.00 180.00 110.00 85.00 20.00 11.90 Capable Port UEPFP UEPFP UEPXE 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port UEPFP UEPKL 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port UEPFP UEPKM 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital UEPFP UEPKM 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital </td <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>						
2-Wire Voice Unbundled PBX LD Terminal Switchboard Port UEPFP UEPXD 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port UEPFP UEPKE 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port UEPFP UEPKL 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port UEPFP UEPKL 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port UEPFP UEPKM 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital UEPFP UEPKM 14.00 180.00 110.00 85.00 20.00 11.90						
2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port UEPFP UEPXE 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port UEPFP UEPFP UEPXL 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port UEPFP UEPFP UEPXL 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital UEPFP UEPXM 14.00 180.00 110.00 85.00 20.00 11.90						
Capable Port UEPFP UEPXE 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port UEPFP UEPKL 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port UEPFP UEPKL 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital UEPFP UEPKM 14.00 180.00 110.00 85.00 20.00 11.90						
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port UEPFP UEPXL 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port UEPFP UEPFP UEPXM 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital UEPFP UEPXM 14.00 180.00 110.00 85.00 20.00 11.90	11.90	11.90	11.90			
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port UEPFP UEPXM 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital UEPFP UEPXM 14.00 180.00 110.00 85.00 20.00 11.90			1			
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port UEPFP UEPXM 14.00 180.00 110.00 85.00 20.00 11.90 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital UEPFP UEPXM 14.00 180.00 110.00 85.00 20.00 11.90	11.90	11.90	11.90			
2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital						
	11.90	11.90	11.90			
	T					
Discount Room Calling Port UEPFP UEPXO 14.00 180.00 110.00 85.00 20.00 11.90	11.90					L
2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port UEPFP UEPXS 14.00 180.00 110.00 85.00 20.00 11.90	11.90	11.90	11.90			L
						
Local Number Portability (1 per port) UEPFP LNPCP 3.15 0.00 0.00 11.90	11.90	11.90	11.90			───
						───
Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility						
Termination UEPFP U1TV2 25.32 47.35 31.78	ł					───
Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile or Fraction Mile UEPFP 1L5XX 0.0091						
OF FRACION Mile DEPPP LSXX 0.0091					ł	t
PEATORES UEPFP UEPVF 0.00 0.00 11.90	11 90	11 0/	11.90			
NONRECURRING CHARGES (NRCs) - CURRENTLY COMBINED	11.00	11.90	11.00			<u> </u>
Indiractor (integrated by Transport / 2 Wire Line Port	\rightarrow					<u> </u>
		11.90	11.90			
2-Vire Loop / Dedicated IO Transport / 2 Wire Line Port	11.90				1	t
	11.90	11.90	11.90			
UNBUNDLED PORT/LOOP COMBINATIONS - MARKET BASED RATES						1
2-WIRE VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK PORT	11.90 11.90		<u> </u>			1
UNE Port/Loop Combination Rates			ł		İ	İ

UNBUNDLE	D NETWORK ELEMENTS - Florida													Attachment:			bit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	В	cs	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
							Rec	Nonrec	urring	Nonrecurring	g Disconnect			OSS	Rates(\$)		
								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1		1				67.24										
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2		2				72.40										
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3		3				85.87										L
UNE L	oop Rates																L
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1		1	UEPPX		UECD1	12.24					-	11.90			1.83	ł
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2		2	UEPPX		UECD1	17.40 30.87						11.90			1.83	<u> </u>
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3 ort Rate		3	UEPPX		UECD1	30.87						11.90			1.83	<u> </u>
UNE P	Exchange Ports - 2-Wire DID Port			UEPPX		UEPD1	55.00	850.00	75.00				11.90			1.83	
NONP	ECURRING CHARGES - CURRENTLY COMBINED			ULFFA		OLFDI	55.00	850.00	75.00			1	11.90			1.03	ł
NONKI	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination -																-
1	Switch-As-Is Top 8 MSAs only	1		UEPPX		USAC1		850.00	75.00			1	11.90				1
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion	1						000.00	. 0.00	1		1		1	1	1	
	with BellSouth Allowable Changes Top 8 MSAs only	1		UEPPX		USA1C		850.00	75.00			1	11.90				1
ADDIT	IONAL NRCs	1		<u> </u>				,				1					<u> </u>
	2-Wire DID Subsequent Activity - Add Trunks, Per Trunk			UEPPX		USAS1		32.26	32.26				11.90				
Teleph	one Number/Trunk Group Establisment Charges																
	DID Trunk Termination (One Per Port)			UEPPX		NDT	0.00	0.00	0.00				11.90			1.83	
	DID Numbers, Establish Trunk Group and Provide First Group																
	of 20 DID Numbers			UEPPX		NDZ	0.00	0.00	0.00				11.90			1.83	
	Additional DID Numbers for each Group of 20 DID Numbers			UEPPX		ND4	0.00	0.00	0.00				11.90			1.83	
	DID Numbers, Non- consecutive DID Numbers , Per Number			UEPPX		ND5	0.00	0.00	0.00				11.90			1.83	
	Reserve Non-Consecutive DID numbers			UEPPX		ND6	0.00	0.00	0.00				11.90			1.83	
	Reserve DID Numbers			UEPPX		NDV	0.00	0.00	0.00				11.90			1.83	
LOCAL	NUMBER PORTABILITY																
	Local Number Portability (1 per port)			UEPPX		LNPCP	3.15	0.00	0.00			-					L
	E ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LI	NE SIDE	PORI	1								-					L
UNE P	ort/Loop Combination Rates 2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -											-					
	UNE Zone 1		1	UEPPB	UEPPR		85.25										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -			UEFFD	UEFFR		65.25					1	1				ł
	UNE Zone 2		2	UEPPB	UEPPR		91.67										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -	-	2	OLITD	OLITIK		31.07										
	UNE Zone 3		3	UEPPB	UEPPR		108.46										
UNE L	oop Rates		-														
	2-Wire ISDN Digital Grade Loop - UNE Zone 1		1	UEPPB	UEPPR	USL2X	15.25						11.90			1.83	
		1		1								1					
	2-Wire ISDN Digital Grade Loop - UNE Zone 2		2	UEPPB	UEPPR	USL2X	21.67					1	11.90			1.83	1
	2-Wire ISDN Digital Grade Loop - UNE Zone 3		3	UEPPB	UEPPR	USL2X	38.46						11.90			1.83	
UNE P	ort Rate																
	Exchange Port - 2-Wire ISDN Line Side Port			UEPPB	UEPPR	UEPPB	70.00	525.00	400.00				11.09			1.83	
NONRI	ECURRING CHARGES - CURRENTLY COMBINED	I															I
	2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port	1										1					1
	Combination - Conversion - Top 8 MSAs only	ļ		UEPPB	UEPPR	USACB	0.00	215.00	215.00			ļ	11.90			1.83	l
		<u> </u>		<u> </u>								<u> </u>	L				l
LOCAL	- NUMBER PORTABILITY					LNDOY	0.05	0.00	0.00				ļ				───
B CUA	Local Number Portability (1 per port) NNEL USER PROFILE ACCESS:	 		UEPPB	UEPPR	LNPCX	0.35	0.00	0.00		-		ļ				ł
B-CHA	INNEL USER PROFILE ACCESS: CVS/CSD (DMS/5ESS)			UEPPB	UEPPR	U1UCA	0.00	0.00	0.00								l
	CVS/CSD (DMS/SESS) CVS (EWSD)			UEPPB	UEPPR	U1UCA U1UCB	0.00	0.00	0.00	łł		+		1	ł	ł	t
	CSD			UEPPB	UEPPR	U1UCC	0.00	0.00	0.00			+					t
B-CHA	NNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS S	C.MS. &	TN)	50110	JEITIN	51000	0.00	0.00	0.00			1	1				<u> </u>
	TERMINAL PROFILE	_,0, 0	· · · · ·									1	1				<u> </u>
	User Terminal Profile (EWSD only)	1		UEPPB	UEPPR	U1UMA	0.00	0.00	0.00	1		1	1	1	1	1	l
VERTI	CAL FEATURES	1				-						1	1				
	All Vertical Features - One per Channel B User Profile	1		UEPPB	UEPPR	UEPVF	2.26	0.00	0.00			1	11.90	İ	ĺ	ĺ	İ.
INTER	OFFICE CHANNEL MILEAGE																
İ	Interoffice Channel mileage each, including first mile and																
	facilities termination	1	1	UEPPB	UEPPR	M1GNC	18.4491	47.35	31.78	18.31	7.03	1	11.90			1.83	1

UNBUND	LED NETWORK ELEMENTS - Florida	1	1	1							-	-	Attachment:			bit: B
CATEGORY	Y RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
							Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		
		1	1	İ		Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Interoffice Channel mileage each, additional mile			UEPPB UEPPR	M1GNM	0.0091	0.00	0.00				11.90			1.83	
	/IRE DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUN	K PORT														
UNE	E Port/Loop Combination Rates															
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 1		1	UEPPP		970.74										
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE			ULFFF		570.74										
	Zone 2		2	UEPPP		1,000.54										
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE															
	Zone 3		3	UEPPP		1,078.39										
UNE	E Loop Rates															
	4-Wire DS1 Digital Loop - UNE Zone 1			UEPPP	USL4P	70.74						11.90			1.83	
	4-Wire DS1 Digital Loop - UNE Zone 2			UEPPP	USL4P	100.54						11.90			1.83	
1 16.17	4-Wire DS1 Digital Loop - UNE Zone 3 E Port Rate	-	3	UEPPP	USL4P	178.39			l		1	11.90			1.83	
UNE	E Port Rate Exchange Ports - 4-Wire ISDN DS1 Port			UEPPP	UEPPP	900.00	1,150.00	1,150.00				11.90			1.83	
NOP	NRECURRING CHARGES - CURRENTLY COMBINED	1			ULFPP	900.00	1,150.00	1,150.00			1	11.90			1.03	
	4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port															
	Combination - Conversion -Switch-As-Is Top 8 MSAs only			UEPPP	USACP	0.00	925.00	925.00				11.90			1.83	
ADD	DITIONAL NRCs			-												
	4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy-															
	Inward/two way Telephone Numbers (except NC)			UEPPP	PR7TF		0.5412					11.90			1.83	
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port -															
	Outward Tel Numbers (All States except NC)			UEPPP	PR7TO		12.71	12.71				11.90			1.83	
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port -															
	Subsequent Inward Telephone Numbers			UEPPP	PR7ZT		25.42	25.42				11.90			1.83	
LOC	CAL NUMBER PORTABILITY Local Number Portability (1 per port)	-		UEPPP	LNPCN	1.75										
INT	ERFACE (Provsioning Only)	-		UEFFF	LINFON	1.75										
	Voice/Data			UEPPP	PR71V	0.00	0.00	0.00								
	Digital Data			UEPPP	PR71D	0.00	0.00	0.00								
	Inward Data			UEPPP	PR71E	0.00	0.00	0.00								
New	w or Additional "B" Channel															
	New or Additional - Voice/Data B Channel			UEPPP	PR7BV	0.00	20.00					11.90			1.83	
	New or Additional - Digital Data B Channel			UEPPP	PR7BF	0.00	20.00					11.90			1.83	
	New or Additional Inward Data B Channel			UEPPP	PR7BD	0.00	20.00					11.90			1.83	
CAL	LL TYPES				00704	0.00	0.00	0.00								
	Inward Outward			UEPPP UEPPP	PR7C1 PR7C0	0.00	0.00	0.00								
	Two-way	-		UEPPP	PR7CC	0.00	0.00	0.00								
Inte	eroffice Channel Mileage			0EITT	11000	0.00	0.00	0.00								
	Fixed Each Including First Mile			UEPPP	1LN1A	88.6256	105.54	98.47	21.47	19.05		11.90			1.93	
	Each Airline-Fractional Additional Mile			UEPPP	1LN1B	0.1856										
4-W	IRE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT															
UNE	E Port/Loop Combination Rates															
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1			UEPDC		820.74						11.90			1.83	
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2		2	UEPDC		850.54						11.90			1.83	L
1.16.17	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3 E Loop Rates		3	UEPDC	<u>├</u> ───	928.39						11.90		L	1.83	
UNE	4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPDC	USLDC	70.74			ł			11.90			1.83	
	4-Wire DS1 Digital Loop - UNE Zone 1	1		UEPDC	USLDC	100.54			<u> </u>		1	11.90			1.83	
	4-Wire DS1 Digital Loop - UNE Zone 3			UEPDC	USLDC	178.39						11.90			1.83	
UNE	E Port Rate	1	<u> </u>	-							1					1
	4-Wire DDITS Digital Trunk Port	1		UEPDC	UDD1T	750.00	1,019.56	479.87	204.92	20.10		11.90			1.83	
NO	NRECURRING CHARGES - CURRENTLY COMBINED															
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
	- Switch-As-Is Top 8 MSAs only	<u> </u>		UEPDC	USAC4		95.31	46.71				11.90			1.83	<u> </u>
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															

NBUNDLE	D NETWORK ELEMENTS - Florida												Attachment:			bit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge - Manual Sy Order vs. Electronic Disc Add
						Rec	Nonrec		Nonrecurring			•		Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with Change - Trunk Top 8 MSAs only			UEPDC	USAWB		95.31	46.71				11.90			1.83	
ADDIT	IONAL NRCs															
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC - Subsequent Channel Activation/Chan - 2-Way Trunk			UEPDC	UDTTA		15.69	15.69				11.90			1.83	
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent						45.00	45.00				44.00			4.02	
	Channel Activation/Chan - 1-Way Outward Trunk 4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel			UEPDC	UDTTB		15.69	15.69				11.90			1.83	
	Activation/Chan Inward Trunk w/out DID 4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan			UEPDC	UDTTC		15.69	15.69				11.90			1.83	
	Activation Per Chan - Inward Trunk with DID			UEPDC	UDTTD		15.69	15.69				11.90			1.83	
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan							.0.00		1						
	Activation / Chan - 2-Way DID w User Trans			UEPDC	UDTTE		15.69	15.69				11.90			1.83	
BIPOL	AR 8 ZERO SUBSTITUTION															
	B8ZS -Superframe Format			UEPDC	CCOSF		0.00	655.00				11.90			1.83	
	B8ZS - Extended Superframe Format			UEPDC	CCOEF		0.00	655.00				11.90			1.83	
Alterna	ate Mark Inversion				1400.05		0.00	0.00								
	AMI -Superframe Format AMI - Extended SuperFrame Format			UEPDC UEPDC	MCOSF MCOPO		0.00	0.00								
Telenh	one Number/Trunk Group Establisment Charges			UEFDC	MCOPO		0.00	0.00								
relepi	Telephone Number for 2-Way Trunk Group			UEPDC	UDTGX	0.00						11.90			1.83	
	Telephone Number for 1-Way Outward Trunk Group			UEPDC	UDTGY	0.00						11.90			1.83	
	Telephone Number for 1-Way Inward Trunk Group Without DID			UEPDC	UDTGZ	0.00						11.90			1.83	
	DID Numbers, Establish Trunk Group and Provide First Group															
	of 20 DID Numbers			UEPDC	NDZ	0.00	0.00	0.00				11.90			1.83	
	DID Numbers for each Group of 20 DID Numbers			UEPDC	ND4	0.00						11.90			1.83	
	DID Numbers, Non- consecutive DID Numbers , Per Number			UEPDC	ND5	0.00	0.00	0.00				11.90			1.83	
	Reserve Non-Consecutive DID Nos. Reserve DID Numbers			UEPDC UEPDC	ND6 NDV	0.00	0.00	0.00				11.90 11.90			1.83 1.83	
Dedica	ted DS1 (Interoffice Channel Mileage) -			UEFDC	NDV	0.00	0.00	0.00				11.90			1.03	
	O for 4-Wire DS1 Digital Loop with 4-Wire DDITS Trunk Port															
	Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities Termination)			UEPDC	1LNO1	88.44	105.54	98.47	21.47	19.05		11.90			1.83	
	Interoffice Channel Mileage - Additional rate per mile - 0-8 miles			UEPDC	1LNOA	0.1856	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities Termination)			UEPDC	1LNO2	0.00	0.00	0.00								
	Interoffice Channel Mileage - Additional rate per mile - 9-25 miles			UEPDC	1LNOB	0.1856	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Termination)		_	UEPDC	1LNO3	0.00	0.00	0.00	0.00							
	Interoffice Channel Mileage - Additional rate per mile - 25+ miles			UEPDC	1LNOC	0.1856	0.00	0.00								
	Local Number Portability, per DS0 Activated			UEPDC	LNPCP	3.15	0.00	0.00	0.00	1			1		1	
	Central Office Termininating Point			UEPDC	CTG	0.00										
	E DS1 LOOP WITH CHANNELIZATION WITH PORT															
	n is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Acti															
	em can have various rate combinations based on type and nu	nber of	ports	used	_						ļ					
UNE D	S1 Loop		1			70 74	0.00	0.00								
	4-Wire DS1 Loop - UNE Zone 1 4-Wire DS1 Loop - UNE Zone 2		1	UEPMG UEPMG	USLDC USLDC	70.74 100.54	0.00	0.00								
	4-Wire DST Loop - UNE Zone 3		2	UEPMG	USLDC	178.39	0.00	0.00	-	1	+		ł	-	ł	
UNE D	SO Channelization Capacities (D4 Channel Bank Configuration	ns)	5		OOLDO	110.59	0.00	0.00			1					-
0	24 DSO Channel Capacity - 1 per DS1	/		UEPMG	VUM24	118.06	0.00	0.00		1		11.90	1		1.83	1
	48 DSO Channel Capacity - 1 per 2 DS1s			UEPMG	VUM48	236.12	0.00	0.00		İ	1	11.90			1.83	
	96 DSO Channel Capacity -1per 4 DS1s			UEPMG	VUM96	472.24	0.00	0.00				11.90			1.83	
	144 DS0 Channel Capacity - 1 per 6 DS1s			UEPMG	VUM14	708.36	0.00	0.00				11.90			1.83	
	192 DS0 Channel Capacity -1 per 8 DS1s			UEPMG	VUM19	944.48	0.00	0.00				11.90			1.83	

CATEGORY RATE ELEMENTS Interm m 240 DS0 Channel Capacity - 1 per 10 DS1s 280 DS0 Channel Capacity - 1 per 10 DS1s 280 DS0 Channel Capacity - 1 per 10 DS1s 384 DS0 Channel Capacity - 1 per 10 DS1s 384 DS0 Channel Capacity - 1 per 10 DS1s 576 DS0 Channel Capacity - 1 per 20 DS1s 672 DS0 Channel Capacity - 1 per 20 DS1s 576 DS0 Channel Capacity - 1 per 24 DS1s 576 DS0 Channel Capacity - 1 per 24 DS1s Multiples of this configuration is One (1) DS1, One (1) D4 Channel Banl Multiples of this configuration functioning as one are considered Add'I af Multiples of this configuration functioning as one are considered Add'I af NRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes - Top 8 MSAs Only System Additions Where Currently Combined and New (Not Currently Corr In Density Zone 1 Top 8 MSAs 1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc Fea Activation - Bipolar 8 Zero Substitution Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only Alternate Mark Inversion (AMI) Superframe Format Extended Superframe Format Extended Superframe Format Exchange Ports Associated with 4-Wire DS1 Loop with Channelization with Exchange Ports Superframe Format	2001	UEPMG UEPMG UEPMG UEPMG UEPMG ion with Port - Cor Up To 24 DSO Por minimum system	ts with Feature / configuration is USAC4 VUMD4 CCOSF CCOEF MCOSF	Activations.	450.00 950.00 0.00 0.00	RATES(\$) urring Add'I 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.	Nonrecurring I First	Disconnect Add'l 30.00	Submitted Elec per LSR	Submitted Manually per LSR 11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90	Incremental Charge - Manual Svc Order vs. Electronic- 1st OSS SOMAN	Incremental Charge - Manual Svc Order vs. Electronic- Add'l Rates(\$) SOMAN	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st SOMAN 1.83 1.83 1.83 1.83 1.83 1.83 1.83 1.83	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'l SOMAN
288 DS0 Channel Capacity - 1 per 12 DS1s 384 DS0 Channel Capacity - 1 per 20 DS1s 480 DS0 Channel Capacity - 1 per 20 DS1s 576 DS0 Channel Capacity - 1 per 20 DS1s 672 DS0 Channel Capacity - 1 per 20 DS1s Non-Recurring Charges (NRC) Associated with 4-Wire DS1 Loop with Cha A Minimum System configuration is One (1) DS1, One (1) D4 Channel Band Multiples of this configuration functioning as one are considered Add'I af NRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes - Top 8 MSAs Only System Additions Where Currently Combined and New (Not Currently Com In Density Zone 1 Top 8 MSAs I DS1/D4 Channel Bank - Add NRC for each Port and Assoc Fea Activation - Bipolar 8 Zero Substitution Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only Alternate Mark Inversion (AMI) Superframe Format Extended Superframe Format Extended Ports Associated with 4-Wire DS1 Loop with Channelization wit Exchange Ports Line Side Combination Channelized PBX Trunk Pot - Business	k, and l fter the mbined	UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UPMG UEPMG UEPMG UEPMG UEPMG UEPMG	VUM28 VUM38 VUM40 VUM67 VUM67 nversion Charge configuration is USAC4 VUMD4 CCOSF CCOEF MCOSF	1,180.60 1,416.72 1,888.96 2,361.20 2,833.44 3,305.68 Based on a Sy activations. counted. 0.00 0.00 0.00	First 0.00 0.00 0.00 0.00 0.00 0.00 stem 450.00 950.00 0.00 0.00	Add'l 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 600.00 655.00	First	Add'l	SOMEC	11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90	OSS	Rates(\$)	SOMAN 1.83 1.83 1.83 1.83 1.83 1.83	
288 DS0 Channel Capacity - 1 per 12 DS1s 384 DS0 Channel Capacity - 1 per 20 DS1s 480 DS0 Channel Capacity - 1 per 20 DS1s 576 DS0 Channel Capacity - 1 per 20 DS1s 672 DS0 Channel Capacity - 1 per 20 DS1s Non-Recurring Charges (NRC) Associated with 4-Wire DS1 Loop with Cha A Minimum System configuration is One (1) DS1, One (1) D4 Channel Banl Multiples of this configuration functioning as one are considered Add'I af NRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes - Top 8 MSAs Only System Additions Where Currently Combined and New (Not Currently Cor In Density Zone 1 Top 8 MSAs 1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc Fea Activation - Bipolar 8 Zero Substitution Clear Channel Capability Format, superframe - Subsequent Activity Only Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only Alternate Mark Inversion (AMI) Superframe Format Extended Superframe Format Extended Superframe Format Extended Superframe Format Line Side Combination Channelized PBX Trunk Port - Business	k, and l fter the mbined	UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UPMG UEPMG UEPMG UEPMG UEPMG UEPMG	VUM28 VUM38 VUM40 VUM67 VUM67 nversion Charge configuration is USAC4 VUMD4 CCOSF CCOEF MCOSF	1,180.60 1,416.72 1,888.96 2,361.20 2,833.44 3,305.68 Based on a Sy activations. counted. 0.00 0.00 0.00	First 0.00 0.00 0.00 0.00 0.00 0.00 stem 450.00 950.00 0.00 0.00	Add'l 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 600.00 655.00	First	Add'l	SOMEC	11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90			1.83 1.83 1.83 1.83 1.83 1.83	SOMAN
288 DS0 Channel Capacity - 1 per 12 DS1s 384 DS0 Channel Capacity - 1 per 20 DS1s 480 DS0 Channel Capacity - 1 per 20 DS1s 576 DS0 Channel Capacity - 1 per 20 DS1s 672 DS0 Channel Capacity - 1 per 20 DS1s Non-Recurring Charges (NRC) Associated with 4-Wire DS1 Loop with Cha A Minimum System configuration is One (1) DS1, One (1) D4 Channel Banl Multiples of this configuration functioning as one are considered Add'I af NRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes - Top 8 MSAs Only System Additions Where Currently Combined and New (Not Currently Cor In Density Zone 1 Top 8 MSAs 1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc Fea Activation - Bipolar 8 Zero Substitution Clear Channel Capability Format, superframe - Subsequent Activity Only Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only Alternate Mark Inversion (AMI) Superframe Format Extended Superframe Format Extended Superframe Format Extended Superframe Format Line Side Combination Channelized PBX Trunk Port - Business	k, and l fter the mbined	UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UPMG UEPMG UEPMG UEPMG UEPMG UEPMG	VUM28 VUM38 VUM40 VUM67 VUM67 nversion Charge configuration is USAC4 VUMD4 CCOSF CCOEF MCOSF	1,416.72 1,888.96 2,361.20 2,833.44 3,305.68 Based on a Sy Activations. counted. 0.00 0.00 0.00	0.00 0.00 0.00 0.00 0.00 stem 450.00 950.00 0.00	0.00 0.00 0.00 0.00 0.00 50.00 600.00 655.00				11.90 11.90 11.90 11.90 11.90 11.90 11.90 11.90			1.83 1.83 1.83 1.83 1.83 1.83	
384 DS0 Channel Capacity - 1 per 16 DS1s 480 DS0 Channel Capacity - 1 per 20 DS1s 576 DS0 Channel Capacity - 1 per 28 DS1s 672 DS0 Channel Capacity - 1 per 28 DS1s Non-Recurring Charges (NRC) Associated with 4-Wire DS1 Loop with Cha A Minimum System configuration is One (1) DS1, One (1) D4 Channel Ban Multiples of this configuration functioning as one are considered Add'I af NRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes - Top 8 MSAs Only System Additions Where Currently Combined and New (Not Currently Corr In Density Zone 1 Top 8 MSAs 1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc Fea Activation - Bipolar 8 Zero Substitution Clear Channel Capability Format, superframe - Subsequent Activity Only Alternate Mark Inversion (AMI) Superframe Format Extended Superframe Format Extended Superframe Format Extended Superframe Format Extended Superframe Format Line Side Combination Channelized PBX Trunk Port - Business	k, and l fter the mbined	UEPMG UEPMG UEPMG UEPMG UPT 24 DSO Por minimum system UEPMG UEPMG UEPMG UEPMG UEPMG	VUM38 VUM40 VUM67 VUM67 vUM67 ts with Feature / configuration is USAC4 VUMD4 CCOSF CCOEF MCOSF	1,888.96 2,361.20 2,833.44 3,305.68 Based on a Sy Activations. counted. 0.00 0.00 0.00 0.00 0.00	0.00 0.00 0.00 0.00 stem 450.00 950.00 0.00	0.00 0.00 0.00 50.00 600.00 655.00	200.00	30.00		11.90 11.90 11.90 11.90 11.90 11.90			1.83 1.83 1.83	
480 DS0 Channel Capacity - 1 per 20 DS1s 576 DS0 Channel Capacity - 1 per 24 DS1s 672 DS0 Channel Capacity - 1 per 28 DS1s Non-Recurring Charges (NRC) Associated with 4-Wire DS1 Loop with Cha A Minimum System configuration is One (1) DS1, One (1) D4 Channel Banl Multiples of this configuration functioning as one are considered Add'I af NRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes - Top 8 MSAs Only System Additions Where Currently Combined and New (Not Currently Cor In Density Zone 1 Top 8 MSAs 1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc Fea Activation - Bipolar 8 Zero Substitution Clear Channel Capability Format, superframe - Subsequent Activity Only Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only Alternate Mark Inversion (AMI) Superframe Format Extended Superframe Format Extended Superframe Format Extended Superframe Format Line Side Combination Channelized PBX Trunk Port - Business	k, and l fter the mbined	UEPMG UEPMG UEPMG ion with Port - Coi Up To 24 DSO Por minimum system UEPMG UEPMG UEPMG UEPMG UEPMG	VUM40 VUM57 VUM67 nversion Charge ts with Feature / configuration is USAC4 VUMD4 CCOSF CCOEF CCOEF	2,361.20 2,833.44 3,305.68 Based on a Sy Activations. counted. 0.00 0.00 0.00	0.00 0.00 stem 450.00 950.00 0.00	0.00 0.00 50.00 600.00 655.00	200.00	30.00		11.90 11.90 11.90 11.90 11.90			1.83 1.83	
576 DS0 Channel Capacity -1 per 24 DS1s 672 DS0 Channel Capacity - 1 per 28 DS1s Non-Recurring Charges (NRC) Associated with 4-Wire DS1 Loop with Cha A Minimum System configuration is One (1) DS1, One (1) D4 Channel Banl Multiples of this configuration functioning as one are considered Add'I af NRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes - Top 8 MSAs Only System Additions Where Currently Combined and New (Not Currently Com In Density Zone 1 Top 8 MSAs 1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc Fea Activation - Bipolar 8 Zero Substitution Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only Alternate Mark Inversion (AMI) Superframe Format Extended Superframe Format Extended Superframe Format Extended Superframe Format Extended Superframe Format Extended Superframe Format Line Side Combination Channelized PBX Trunk Port - Business	k, and l fter the mbined	UEPMG UEPMG ion with Port - Cor Up To 24 DSO Por minimum system UEPMG) UEPMG UEPMG UEPMG UEPMG	VUM57 VUM67 nversion Charge configuration is USAC4 VUMD4 CCOSF CCOEF MCOSF	2,833.44 3,305.68 Based on a Sy Activations. counted. 0.00 0.00 0.00	0.00 0.00 stem 450.00 950.00 0.00 0.00	0.00 0.00 50.00 600.00 655.00	200.00	30.00		11.90 11.90 11.90 11.90			1.83	
672 DS0 Channel Capacity - 1 per 28 DS1s Non-Recurring Charges (NRC) Associated with 4-Wire DS1 Loop with Cha A Minimum System configuration is One (1) DS1, One (1) D4 Channel Band Multiples of this configuration functioning as one are considered Add'l af NRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes - Top 8 MSAs Only System Additions Where Currently Combined and New (Not Currently Com In Density Zone 1 Top 8 MSAs 1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc Fea Activation - Bipolar 8 Zero Substitution Clear Channel Capability Format, superframe - Subsequent Activity Only Alternate Mark Inversion (AMI) Superframe Format Extended Superframe Format Extended Superframe Format Extended Superframe Format Extended Superframe Format Line Side Combination Channelized PBX Trunk Port - Business	k, and l fter the mbined	UEPMG ion with Port - Coi Up To 24 DSO Por minimum system UEPMG UEPMG UEPMG UEPMG UEPMG	VUM67 nversion Charge ts with Feature / configuration is USAC4 VUMD4 CCOSF CCOEF MCOSF	3,305.68 Based on a Sy Activations. counted. 0.00 0.00 0.00 0.00 0.00 0.00	0.00 stem 450.00 950.00 0.00 0.00	0.00 50.00 600.00 655.00	200.00	30.00		11.90 11.90 11.90				
Non-Recurring Charges (NRC) Associated with 4-Wire DS1 Loop with Cha A Minimum System configuration is One (1) DS1, One (1) D4 Channel Ban Multiples of this configuration functioning as one are considered Add'I af NRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes - Top 8 MSAs Only System Additions Where Currently Combined and New (Not Currently Corr In Density Zone 1 Top 8 MSAs 1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc Fea Activation - Bipolar 8 Zero Substitution Clear Channel Capability Format, superframe - Subsequent Activity Only Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only Alternate Mark Inversion (AMI) Extended Superframe Format Extended Superframe Format Extended Superframe Format Line Side Combination Channelized PBX Trunk Port - Business	k, and l fter the mbined	ion with Port - Coi Up To 24 DSO Por minimum system UEPMG UEPMG UEPMG UEPMG	version Charge ts with Feature / configuration is USAC4 VUMD4 CCOSF CCOEF MCOSF	Based on a Sy Activations. counted. 0.00 0.00 0.00 0.00	450.00 950.00 0.00 0.00	50.00 600.00 655.00	200.00	30.00		11.90				
A Minimum System configuration is One (1) DS1, One (1) D4 Channel Banl Multiples of this configuration functioning as one are considered Add'l af NRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes - Top 8 MSAs Only System Additions Where Currently Combined and New (Not Currently Com In Density Zone 1 Top 8 MSAs 1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc Fea Activation - Bipolar 8 Zero Substitution Clear Channel Capability Format, superframe - Subsequent Activity Only Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only Alternate Mark Inversion (AMI) Superframe Format Extended Superframe Format Extended Superframe Format Exchange Ports Associated with 4-Wire DS1 Loop with Channelization wil Exchange Ports Line Side Combination Channelized PBX Trunk Port - Business	k, and l fter the mbined	Up To 24 DSO Por minimum system UEPMG UEPMG UEPMG UEPMG UEPMG	ts with Feature / configuration is USAC4 VUMD4 CCOSF CCOEF MCOSF	Activations. counted. 0.00 0.00 0.00 0.00 0.00	450.00 950.00 0.00 0.00	600.00	200.00	30.00		11.90				
Multiples of this configuration functioning as one are considered Add'I af NRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes - Top 8 MSAS Only System Additions Where Currently Combined and New (Not Currently Com In Density Zone 1 Top 8 MSAs 1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc Fea Activation - Bipolar 8 Zero Substitution Clear Channel Capability Format , superframe - Subsequent Activity Only Superframe Format Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only Alternate Mark Inversion (AMI) Superframe Format Extended Superframe Format Extended Superframe Format Extended Superframe Format Exchange Ports Associated with 4-Wire DS1 Loop with Channelization wite Exchange Ports Line Side Combination Channelized PBX Trunk Port - Business	fter the mbined	Minimum system UEPMG) UEPMG UEPMG UEPMG UEPMG	COOSE CCOSE CCOSE MCOSE	counted. 0.00 0.00 0.00 0.00	950.00 0.00 0.00	600.00	200.00	30.00		11.90				
NRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes - Top 8 MSAs Only System Additions Where Currently Combined and New (Not Currently Com In Density Zone 1 Top 8 MSAs 1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc Fea Activation - Bipolar 8 Zero Substitution Clear Channel Capability Format, superframe - Subsequent Activity Only Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only Alternate Mark Inversion (AMI) Extended Superframe Format Extended Superframe Format Extended Superframe Format Extended Superframe Format Line Side Combination Channelized PBX Trunk Port - Business	mbined	UEPMG UEPMG UEPMG UEPMG UEPMG	USAC4 VUMD4 CCOSF CCOEF MCOSF	0.00	950.00 0.00 0.00	600.00	200.00	30.00		11.90				
BellSouth Allowed Changes - Top 8 MSAs Only System Additions Where Currently Combined and New (Not Currently Com In Density Zone 1 Top 8 MSAs 1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc Fea Activation - Bipolar 8 Zero Substitution Clear Channel Capability Format, superframe - Subsequent Activity Only Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only Alternate Mark Inversion (AMI) Superframe Format Extended Superframe Format Extended Superframe Format Extended Superframe Format Line Side Combination Channelized PBX Trunk Port - Business		UEPMG UEPMG UEPMG UEPMG	VUMD4 CCOSF CCOEF MCOSF	0.00	950.00 0.00 0.00	600.00	200.00	30.00		11.90				
System Additions Where Currently Combined and New (Not Currently Con In Density Zone 1 Top 8 MSAs 1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc Fea Activation - Bipolar 8 Zero Substitution Clear Channel Capability Format, superframe - Subsequent Activity Only Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only Alternate Mark Inversion (AMI) Superframe Format Extended Superframe Format Extended Superframe Format Exchange Ports Line Side Combination Channelized PBX Trunk Port - Business		UEPMG UEPMG UEPMG UEPMG	VUMD4 CCOSF CCOEF MCOSF	0.00	950.00 0.00 0.00	600.00	200.00	30.00		11.90				
In Density Zone 1 Top 8 MSAs 1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc Fea Activation Bipolar 8 Zero Substitution Clear Channel Capability Format, superframe - Subsequent Activity Only Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only Alternate Mark Inversion (AMI) Superframe Format Extended Superframe Format Extended Superframe Format Extended Superframe Format Line Side Combination Channelized PBX Trunk Port - Business		UEPMG UEPMG UEPMG UEPMG	CCOSF CCOEF MCOSF	0.00	0.00	655.00	200.00	30.00						
1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc Fea Activation - Bipolar 8 Zero Substitution Clear Channel Capability Format, superframe - Subsequent Activity Only Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only Alternate Mark Inversion (AMI) Superframe Format Extended Superframe Format	th Port	UEPMG UEPMG UEPMG	CCOSF CCOEF MCOSF	0.00	0.00	655.00	200.00	30.00						
Bipolar 8 Zero Substitution Clear Channel Capability Format, superframe - Subsequent Activity Only Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only Alternate Mark Inversion (AMI) Superframe Format Extended Superframe Format Extended Superframe Format Exchange Ports Line Side Combination Channelized PBX Trunk Port - Business	th Port	UEPMG UEPMG UEPMG	CCOSF CCOEF MCOSF	0.00	0.00	655.00	200.00	30.00						
Clear Channel Capability Format, superframe - Subsequent Activity Only Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only Alternate Mark Inversion (AMI) Superframe Format Extended Superframe Format Exchange Ports Associated with 4-Wire DS1 Loop with Channelization with Exchange Ports Line Side Combination Channelized PBX Trunk Port - Business	th Port	UEPMG UEPMG	CCOEF	0.00	0.00									
Activity Only Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only Alternate Mark Inversion (AMI) Extended Superframe Format Exchange Ports Associated with 4-Wire DS1 Loop with Channelization wite Exchange Ports Line Side Combination Channelized PBX Trunk Port - Business	th Port	UEPMG UEPMG	CCOEF	0.00	0.00									
Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only Alternate Mark Inversion (AMI) Superframe Format Extended Superframe Format Exchange Ports Associated with 4-Wire DS1 Loop with Channelization with Exchange Ports Line Side Combination Channelized PBX Trunk Port - Business	th Port	UEPMG UEPMG	CCOEF	0.00	0.00									
Subsequent Activity Only Alternate Mark Inversion (AMI) Superframe Format Extended Superframe Format Exchange Ports Associated with 4-Wire DS1 Loop with Channelization with Exchange Ports Line Side Combination Channelized PBX Trunk Port - Business	th Port	UEPMG	MCOSF			655.00				11.90				
Alternate Mark Inversion (AMI) Superframe Format Extended Superframe Format Exchange Ports Associated with 4-Wire DS1 Loop with Channelization with Exchange Ports Line Side Combination Channelized PBX Trunk Port - Business	th Port	UEPMG	MCOSF			655.00								
Superframe Format Extended Superframe Format Exchange Ports Associated with 4-Wire DS1 Loop with Channelization with Exchange Ports Line Side Combination Channelized PBX Trunk Port - Business	th Port			0.00						11.90				
Extended Superframe Format Exchange Ports Associated with 4-Wire DS1 Loop with Channelization wit Exchange Ports Line Side Combination Channelized PBX Trunk Port - Business	th Port			0.00										
Exchange Ports Associated with 4-Wire DS1 Loop with Channelization wit Exchange Ports Line Side Combination Channelized PBX Trunk Port - Business	th Port	UEPMG			0.00	0.00								
Exchange Ports Line Side Combination Channelized PBX Trunk Port - Business	th Port		MCOPO	0.00	0.00	0.00								
Line Side Combination Channelized PBX Trunk Port - Business														
		UEPPX	UEPCX	14.00	0.00	0.00	0.00	0.00		11.90			1.83	
Line Side Outward Channelized PBX Trunk Port - Business		UEPPX	UEPOX	14.00	0.00	0.00	0.00	0.00		11.90			1.83	
Line Side Inward Only Channelized PBX Trunk Port without DID		UEPPX	UEP1X	14.00	0.00	0.00	0.00	0.00		11.90			1.83	
2-Wire Trunk Side Unbundled Channelized DID Trunk Port		UEPPX	UEPDM	55.00	0.00	0.00	0.00	0.00		11.90			1.83	
Feature Activations - Unbundled Loop Concentration														
Feature (Service) Activation for each Line Port Terminated in D4														
Bank		UEPPX	1PQWM	0.66	40.00	20.00	6.00	5.00		11.90			1.83	
Feature (Service) Activation for each Trunk Port Terminated in														
D4 Bank		UEPPX	1PQWU	0.66	110.00	30.00	65.00	20.00		11.90			1.83	
Telephone Number/ Group Establishment Charges for DID Service		UEPPX	NDT	0.00	0.00	0.00				11.00				
DID Trunk Termination (1 per Port) Estab Trk Grp and Provide 1st 20 DID Nos. (FL,GA, NC,& SC)	_	UEPPX	NDZ	0.00	0.00	0.00				11.90 11.90				
DID Numbers - groups of 20 - Valid all States		UEPPX	ND2 ND4	0.00	0.00	0.00				11.90				
Non-Consecutive DID Numbers - per number		UEPPX	ND5	0.00	0.00	0.00				11.90				
Reserve Non-Consecutive DID Numbers		UEPPX	ND6	0.00	0.00	0.00				11.90				
Reserve DID Numbers		UEPPX	NDV	0.00	0.00	0.00				11.90				
Local Number Portability														
Local Number Portability - 1 per port		UEPPX	LNPCP	3.15	0.00	0.00								
FEATURES - Vertical and Optional														
Local Switching Features Offered with Line Side Ports Only														
All Features Available		UEPPX	UEPVF	2.26	0.00	0.00				11.90			1.83	
UNBUNDLED CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES				<u> </u>	deal de la C		├ ──── ├						ļ	
1. Cost Based Rates are applied where BellSouth is required by FCC and/o							diad Davi		Early 19-14					
2. Features shall apply to the Unbundled Port/Loop Combination - Cost B										oin Bomt" -	on Combined	l		
3. End Office and Tandem Switching Usage and Common Transport Usage						-					-			
4. The first and additional Port nonrecurring charges apply to Not Current	tly Com	bined Combos. F	For Currently Co	mbined Combo	s, the nonrecu	rring charges	shall be those id	dentified in t	he Nonrecu	rring - Curre	ently Combine	ed sections.	Additional NR	Cs may
apply also and are categorized accordingly.												•		
5. Market Rates for Unbundled Centrex Port/Loop Combination will be ne	egotiate	ed on an Individua	l Case Basis, un	til further notic	e.									
UNE-P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only)														
2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Combo				ļ			├ ──── ├		L	L				
UNE Port/Loop Combination Rates (Non-Design)	1	1		1					1	1	1	1		

UNBUNDL	ED NETWORK ELEMENTS - Florida			-									Attachment:			bit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrec		Nonrecurring			r		Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design		1	UEP91		10.94										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			UEF91	-	10.94			-							
	Non-Design		2	UEP91		15.05										1
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		_													
	Non-Design		3	UEP91		25.80										
UNE	Port/Loop Combination Rates (Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	Design		1	UEP91		13.41										l
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		2	UEP91		18.57										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		2	UEP91	-	18.57			-							
	Design		3	UEP91		32.04										1
UNE	Loop Rate		Ű	02.01		02.01										
-	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP91	UECS1	9.77										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP91	UECS1	13.88										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP91	UECS1	24.63										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP91	UECS2	12.24										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP91	UECS2	17.40										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP91	UECS2	30.87										l
	Ports				_											ļ
All S	tates (Except North Carolina and Sout Carolina)					4 47	50.04	00.40	07.50	8.37		44.00				ļ
	2-Wire Voice Grade Port (Centrex) Basic Local Area 2-Wire Voice Grade Port (Centrex 800 termination)Basic Local			UEP91	UEPYA	1.17	53.31	26.46	27.50	8.37		11.90				l
	Area			UEP91	UEPYB	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local			OLF91	ULFID	1.17	55.51	20.40	21.30	0.57		11.90				
	Area			UEP91	UEPYH	1.17	53.31	26.46	27.50	8.37		11.90				1
	2-Wire Voice Grade Port (Centrex from diff Serving Wire			02.01	02		00.01	20.10	21.00	0.01		11.00				
	Center)2 Basic Local Area			UEP91	UEPYM	1.17	139.49	86.10	65.41	13.81		11.90				1
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term - Basic Local Area			UEP91	UEPYZ	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent															
	- Basic Local Area			UEP91	UEPY9	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port Terminated on 800 Service Term -															1
	Basic Local Area			UEP91	UEPY2	1.17	53.31	26.46	27.50	8.37		11.90				ļ
Geor	gia and Florida Only			UEP91	UEPHA	4 47	50.04	00.40	07.50	0.07		11.90				l
	2-Wire Voice Grade Port (Centrex) 2-Wire Voice Grade Port (Centrex 800 termination)			UEP91 UEP91	UEPHA	1.17 1.17	53.31	26.46 26.46	27.50 27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex 800 termination) 2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP91 UEP91	UEPHB	1.17	53.31 53.31	26.46		8.37 8.37	+	11.90	ł		{	
	2-Wire Voice Grade Port (Centrex from diff Serving Wire			OLI 01	0EI IIII		00.01	20.40	21.00	0.07		11.00				
	Center)2			UEP91	UEPHM	1.17	139.49	86.10	65.41	13.81		11.90				i i
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service										1			İ		[
	Term			UEP91	UEPHZ	1.17	139.49	86.10	65.41	13.81		11.90				
																1
	2-Wire Voice Grade Port terminated in on Megalink or equivalent	l		UEP91	UEPH9	1.17	53.31	26.46	27.50	8.37		11.90				ļ
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP91	UEPH2	1.17	53.31	26.46	27.50	8.37		11.90				ļ
Loca	I Switching				LIDECS	0.7384										
1 000	Centrex Intercom Funtionality, per port			UEP91	URECS	0.7384				1						<u> </u>
LUCA	Local Number Portability (1 per port)			UEP91	LNPCC	0.35										
Featu		1		02101	2.11.00	0.00					1					i
. car	All Standard Features Offered, per port			UEP91	UEPVF	2.26			1		1	11.90			İ	
	All Select Features Offered, per port			UEP91	UEPVS	0.00	370.70		1		1	11.90			İ	
	All Centrex Control Features Offered, per port	1		UEP91	UEPVC	2.26					1	11.90				[
NAR																
	Unbundled Network Access Register - Combination			UEP91	UARCX	0.00	0.00	0.00				11.90				
	Unbundled Network Access Register - Indial			UEP91	UAR1X	0.00	0.00	0.00				11.90				
	Unbundled Network Access Register - Outdial			UEP91	UAROX	0.00	0.00	0.00				11.90				L
Misc	ellaneous Terminations		1								1					L

BUNDLE	ED NETWORK ELEMENTS - Florida		1										Attachment:			bit: B
EGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
						D	Nonrec	urring	Nonrecurring	g Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
2-Wire	Trunk Side															
	Trunk Side Terminations, each			UEP91	CENA6	8.73										
Intero	ffice Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination - Voice Grade			UEP91	M1GBC	25.32										
	Interoffice Channel mileage, per mile or fraction of mile			UEP91	M1GBM	0.0091										
	re Activations (DS0) Centrex Loops on Channelized DS1 Servic	e														
D4 Ch	Pannel Bank Feature Activations Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP91	1PQWS	0.66										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP91	IFQWS	0.00										
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot Feature Activation on D-4 Channel Bank FX Trunk Side Loop			UEP91	1PQW6	0.66										
	Slot			UEP91	1PQW7	0.66										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP91	1PQWP	0.66										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP91	1PQWV											
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop					0.66										
	Slot			UEP91	1PQWQ	0.66										
Non F	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP91	1PQWA	0.66										
Non-F	Recurring Charges (NRC) Associated with UNE-P Centrex Conversion - Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP91	USAC2		21.50	8.42				11.90				
	Conversion of Existing Centrex Common Block			UEP91	USACN		5.17	8.32				11.90				
	New Centrex Standard Common Block			UEP91	MIACS	0.00	618.82	0.02				11.90				
	New Centrex Customized Common Block			UEP91	M1ACC	0.00	618.82					11.90				
	Secondary Block, per Block			UEP91	M2CC1	0.00	71.31					11.90				
	NAR Establishment Charge, Per Occasion			UEP91	URECA	0.00	66.48					11.90				
	P CENTREX - 5ESS (Valid in All States)															
	e VG Loop/2-Wire Voice Grade Port (Centrex) Combo															
UNE F	Port/Loop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design		1	UEP95		10.94										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		~			45.05										
	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		2	UEP95		15.05										
	Non-Design		3	UEP95		25.80										
UNE F	Port/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -					10.11										
+	Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		1	UEP95		13.41										
	Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		2	UEP95		18.57										
	Design		3	UEP95		32.04										
UNE L	oop Rate															
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP95	UECS1	9.77										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP95	UECS1	13.88										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP95	UECS1 UECS2	24.63 12.24										
_	2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 2		1	UEP95 UEP95	UECS2 UECS2	12.24								<u> </u>		
	2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3		2	UEP95	UECS2	30.87										
	Port Rate		5	01 30	01002	50.07					1					
All St					1 1					1	1				1	
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP95	UEPYA	1.17	53.31	26.46	27.50	8.37	1	11.90				
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPYB	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area			UEP95	UEPYH	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 Basic Local Area			UEP95	UEPYM	1.17	139.49	86.10	65.41	13.81		11.90				

IBUNDLE	ED NETWORK ELEMENTS - Florida											-	Attachment:			bit: B
TEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Charge -	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrec	urring	Nonrecurring	J Disconnect				Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term - Basic Local Area			UEP95	UEPYZ	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent - Basic Local Area			UEP95	UEPY9	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port Terminated on 800 Service Term -															
	Basic Local Area			UEP95	UEPY2	1.17	53.31	26.46	27.50	8.37		11.90				
	Y, LA, MS, SC, & TN Only															
FL & (GA Only			115545			50.01		07.50							
	2-Wire Voice Grade Port (Centrex)			UEP95 UEP95	UEPHA UEPHB	1.17	53.31 53.31	26.46	27.50	8.37		11.90 11.90				
	2-Wire Voice Grade Port (Centrex 800 termination)				UEPHB	1.17		26.46	27.50	8.37						
	2-Wire Voice Grade Port (Centrex with Caller ID)1 2-Wire Voice Grade Port (Centrex from diff Serving Wire			UEP95	UEPHH	1.17	53.31	26.46	27.50	8.37		11.90				
	Center)2			UEP95	UEPHM	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
_	Term			UEP95	UEPHZ	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP95	UEPH9	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP95	UEPH2	1.17	53.31	26.46	27.50	8.37		11.90				
Local	Switching															
	Centrex Intercom Funtionality, per port			UEP95	URECS	0.7384										
Local	Number Portability															
	Local Number Portability (1 per port)			UEP95	LNPCC	0.35										
Featu																
	All Standard Features Offered, per port			UEP95	UEPVF	2.26										
	All Select Features Offered, per port			UEP95	UEPVS	0.00	370.70					11.90				
	All Centrex Control Features Offered, per port			UEP95	UEPVC	2.26										
NARS						0.00	0.00	0.00				44.00				
_	Unbundled Network Access Register - Combination Unbundled Network Access Register - Indial			UEP95 UEP95	UARCX UAR1X	0.00	0.00	0.00				11.90 11.90				
_	Unbundled Network Access Register - Indial			UEP95	UAROX	0.00	0.00	0.00				11.90				
Misco	ellaneous Terminations			ULF 95	UAROA	0.00	0.00	0.00				11.50				
	e Trunk Side															
	Trunk Side Terminations, each			UEP95	CEND6	8.73										
4-Wire	e Digital (1.544 Megabits)															
	DS1 Circuit Terminations, each			UEP95	M1HD1	54.95										
	DS0 Channels Activated, each			UEP95	M1HDO	0.00	15.69					11.90				
Intero	office Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination			UEP95	MIGBC	25.32										
	Interoffice Channel mileage, per mile or fraction of mile			UEP95	MIGBM	0.0091										
	re Activations (DS0) Centrex Loops on Channelized DS1 Servic	e														
D4 Ch	nannel Bank Feature Activations				10000											
	Feature Activation on D-4 Channel Bank Centrex Loop Slot	ļ		UEP95	1PQWS	0.66						L				
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP95	1PQW6	0.66										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Stot			01 30	IF QWU	0.00										
	Slot			UEP95	1PQW7	0.66										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -	<u> </u>			11 64 11 /	0.00										
	Different Wire Center		ļ	UEP95	1PQWP	0.66										
	Facture Activation on D.4 Observed Dark Drivets Line Loss Obst				1001407	0.00										
_	Feature Activation on D-4 Channel Bank Private Line Loop Slot Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop	<u> </u>		UEP95	1PQWV	0.66				1				1		
	Slot			UEP95	1PQWQ	0.66										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP95 UEP95	1PQWQ 1PQWA	0.66									{	
Non-F	Recurring Charges (NRC) Associated with UNE-P Centrex	ł – –		01 30	II QWA	0.00										
	NRC Conversion Currently Combined Switch-As-Is with allowed	ł –			1 1											
	changes, per port			UEP95	USAC2	0.00	21.50	8.42				11.90				
	Conversion of Existing Centrex Common Block, each	<u> </u>		UEP95	USACN	0.00	5.17	8.32				11.90			İ	
	New Centrex Standard Common Block	<u> </u>		UEP95	MIACS	0.00	618.82					11.90			İ	
	New Centrex Customized Common Block			UEP95	M1ACC	0.00	618.82					11.90			1	1

UNBUNDL	ED NETWORK ELEMENTS - Florida	r	r	1								-	Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
				1150.05	110501		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	NAR Establishment Charge, Per Occasion			UEP95	URECA	0.00	66.48					11.90				───
	P CENTREX - DMS100 (Valid in All States) e VG Loop/2-Wire Voice Grade Port (Centrex) Combo															<u> </u>
	Port/Loop Combination Rates (Non-Design)															<u> </u>
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	-														
	Non-Design		1	UEP9D		10.94										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Non-Design		2	UEP9D		15.05										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Non-Design		3	UEP9D		25.80					-					ł
UNE	Port/Loop Combination Rates (Design)															<u> </u>
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design		1	UEP9D		13.41										1
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			OLI 3D		13.41										
	Design		2	UEP9D		18.57										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		_													
	Design		3	UEP9D		32.04										
UNE	Loop Rate															
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9D	UECS1	9.77										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9D	UECS1	13.88										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP9D	UECS1	24.63					-					ł
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9D UEP9D	UECS2 UECS2	12.24 17.40										ł
	2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9D	UECS2	30.87										ł
UNE	Port Rate		5	OLI 3D	02002	30.07										
	STATES										1					
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9D	UEPYA	1.17						11.90				
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local															
	Area			UEP9D	UEPYB	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3Basic Local															
	Area			UEP9D	UEPYC	1.17	53.31	26.46	27.50	8.37		11.90				L
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3Basic Local				UEPYD	4 47	50.04	00.40	07.50	0.07		44.00				
	Area 2-Wire Voice Grade Port (Centrex / EBS-M5209))3 Basic Local			UEP9D	UEPTD	1.17	53.31	26.46	27.50	8.37		11.90				
	Area			UEP9D	UEPYE	1.17	53.31	26.46	27.50	8.37		11.90				1
	2-Wire Voice Grade Port (Centrex / EBS-M5112))3 Basic Local			OEI OD	ULI IL		00.01	20.40	27.00	0.07	1	11.00				
	Area			UEP9D	UEPYF	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5312))3Basic Local															
	Area			UEP9D	UEPYG	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5008))3 Basic Local															1
	Area			UEP9D	UEPYT	1.17	53.31	26.46	27.50	8.37		11.90				L
	2-Wire Voice Grade Port (Centrex / EBS-M5208))3 Basic Local Area					1.17	E2 24	26.46	27.50	0.27		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5216))3 Basic Local			UEP9D	UEPYU	1.17	53.31	26.46	27.50	8.37		11.90				<u> </u>
	Area			UEP9D	UEPYV	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5316))3 Basic Local				ULI IV	1.17	00.01	20.40	27.00	0.01		11.00				
	Area			UEP9D	UEPY3	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex with Caller ID) Basic Local															
	Area			UEP9D	UEPYH	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp															
	Indication))3 Basic Local Area			UEP9D	UEPYW	1.17	53.31	26.46	27.50	8.37		11.90				<u> </u>
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication))3				UEPYJ	4.47	50.04	00.40	07.50	0.07		44.00				1
┝───┤───	Basic Local Area 2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)			UEP9D	UEPYJ	1.17	53.31	26.46	27.50	8.37	-	11.90				+
	2 Basic Local Area	1	1	UEP9D	UEPYM	1.17	53.31	26.46	27.50	8.37		11.90				1
	2-Basic Local Area 2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			561 30		1.17	00.01	20.40	21.50	0.37		11.50				<u> </u>
	Basic Local Area	1	1	UEP9D	UEPYO	1.17	53.31	26.46	27.50	8.37		11.90				1
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3										İ 👘					
1	Basic Local Area			UEP9D	UEPYP	1.17	53.31	26.46	27.50	8.37	1	11.90				1

UNBUNDL	ED NETWORK ELEMENTS - Florida												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonred	curring	Nonrecurring	g Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3															
	Basic Local Area			UEP9D	UEPYQ	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3 Basic Local Area			UEP9D	UEPYR	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3			021 00	0EI IIX		100.40	00.10	00.41	10.01		11.00				
	Basic Local Area			UEP9D	UEPYS	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3															
	Basic Local Area			UEP9D	UEPY4	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3					4.47	100.10	00.40	05.44	10.01		44.00				
	Basic Local Area 2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3			UEP9D	UEPY5	1.17	139.49	86.10	65.41	13.81		11.90				
	Basic Local Area			UEP9D	UEPY6	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			021 00	02110		100.40	00.10	00.41	10.01		11.00				
	Basic Local Area			UEP9D	UEPY7	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term			UEP9D	UEPYZ	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent															
	Basic Local Area 2-Wire Voice Grade Port Terminated on 800 Service Term Basic			UEP9D	UEPY9	1.17	53.31	26.46	27.50	8.37		11.90				
	Local Area			UEP9D	UEPY2	1.17	53.31	26.46	27.50	8.37		11.90				
FI &	GA Only			OLF 9D	ULF 12	1.17	55.51	20.40	21.50	0.57		11.90				
	2-Wire Voice Grade Port (Centrex)			UEP9D	UEPHA	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP9D	UEPHB	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3			UEP9D	UEPHC	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3			UEP9D	UEPHD	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5209)3			UEP9D	UEPHE	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5112)3			UEP9D	UEPHF	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5312)3			UEP9D	UEPHG	1.17	53.31	26.46		8.37		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5008)3			UEP9D	UEPHT	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5208)3			UEP9D	UEPHU	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5216)3 2-Wire Voice Grade Port (Centrex / EBS-M5316)3			UEP9D UEP9D	UEPHV UEPH3	1.17 1.17	53.31 53.31	26.46 26.46	27.50 27.50	8.37 8.37		11.90 11.90				
	2-Wire Voice Grade Port (Centrex vith Caller ID)			UEP9D	UEPHH	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp			021 00	021111	1.17	00.01	20.40	21.00	0.07		11.00				
	Indication)3			UEP9D	UEPHW	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication)3			UEP9D	UEPHJ	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)															
	2			UEP9D	UEPHM	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			UEP9D	UEPHO	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3			UEP9D	UEPHP	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-105009)2, 3 2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3			UEP9D	UEPHQ	1.17	139.49	86.10	65.41	13.81		11.90				
				021 00	0LI HQ		100.40	00.10	00.41	10.01		11.00				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			UEP9D	UEPHR	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3			UEP9D	UEPHS	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3			UEP9D	UEPH4	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3			UEP9D	UEPH5	1,17	139.49	86.10	65.41	13.81		11.90				
	2-WING VOICE GRAUE FUIL (CEITLIEX/UIITEL SWC /EDS-WD208)2, 3					1.17	139.49	00.10	00.41	13.81		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3			UEP9D	UEPH6	1.17	139.49	86.10	65.41	13.81		11.90				
				-				22.10								
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			UEP9D	UEPH7	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term			UEP9D	UEPHZ	1.17	139.49	86.10	65.41	13.81		11.90				
	2 Wire Voice Crode Port terminated in an Manaliak and in the			UEP9D	UEPH9	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent 2-Wire Voice Grade Port Terminated on 800 Service Term			UEP9D UEP9D	UEPH9 UEPH2	1.17	53.31	26.46	27.50	8.37		11.90			ļ	

NATE ELDENTS Norm	Exhibit: B			Attachment: 2		.										DLED NETWORK ELEMENTS - Florida	IRONDLE
Image: Notion of Price Name Add1 First Add1 First Add1 First Add1 First Add1 SOME SOMA	harge - Charge nual Svc Manual S der vs. Order v ctronic- Electron	Charge -	Charge - Manual Svc Order vs. Electronic-	Charge - Manual Svc Order vs. Electronic-	Submitted Manually	Submitted Elec			RATES(\$)			USOC	BCS	Zone		RY RATE ELEMENTS	TEGORY
Interface Image Addr Print Addr Print Addr Stoke <t< th=""><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th>Rec</th><th></th><th></th><th></th><th></th><th></th><th></th></t<>											Rec						
Decknown harmaning, and point DEPRO DEECS D.754 Desk	OMAN SOMA	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	Add'l	First	Add'l	First	Nec						
Local Mucher Portability Image: Constrainty I		───	!	↓ ↓							0.7004	110500			!		Local
Local Number Potential (1) (1 pr. pr. nf) UEPBO UEPPO 0.01		───		↓ ↓							0.7384	URECS	UEP9D		4		l and l
Petutine Id Description DEPAD <thdepad< th=""> DEPAD DEPAD</thdepad<>				├──── ┦							0.25	LNPCC					Local I
HS Bundar Frazence Oliferal, per part UPPAD				i – – – – – – – – – – – – – – – – – – –							0.35	LINFCC	UEF9D				Foatur
Image: Second Participal Calculation of prime provided and prime provided and prime provided and prime provided and prime provided and prime provided and prime prima prime prima prime prime prime prime prime prime prime prima pri		<u> </u>									2.26	LIED//E			+		reatur
M Derive Contor France Optierd, par port UEPPO UEPPO 2.88 Image: Contor France Optierd, par port		1			11.90					370.70					+		
NARS Image: Contranton UPPO <td></td> <td>1</td> <td></td> <td> </td> <td>11100</td> <td></td> <td></td> <td></td> <td></td> <td>010110</td> <td></td> <td></td> <td></td> <td></td> <td>+</td> <td></td> <td></td>		1			11100					010110					+		
Ubbundle Network Access Register - Unstall UFPRID UARX 0.00 0.00 0.00 11.00 Millocal Metwork Access Register - Outside UFPRID UARXX 0.00 0.00 0.00 0.00 Millocal Metwork Access Register - Outside UFPRID UARXX 0.00 0.00 0.00 0.00 0.00 Millocal Metwork Access Register - Outside UFPRID Millocal Metwork Access Register - Outside UFPRID Millocal Metwork Access Register - Outside																	NARS
Ubundled Network Access Register - Outcial UEPBD UARX 0.00 0.00 11.00 11.00 2Win Final KG I Image: Construction of the construction					11.90				0.00	0.00	0.00	UARCX	UEP9D			Unbundled Network Access Register - Combination	
Miscellaneous Terminations Image I																Unbundled Network Access Register - Inward	
34/10 Trank Side Image: Side Image: Side Termination, each Image: Side Term					11.90				0.00	0.00	0.00	UAROX	UEP9D				
Truck Side Terrinations, each UEPPO CENDs 8.73 No No No No No DS1 Circuit Terminations, each UEPPO MH101 54.85 <t< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></t<>																	
4Viro Digital (1.54 Megabia) E		───	!	ļ]											<u> </u>		2-Wire
Diff Circuit Terminations, each UEPRO MH101 54.85 Image: Control of Channel Image: Control of Channel Image: Control of Channel Image: Control of Channel Image: Control of Channel Image: Control of Channel Image: Control of C		───	!	ļ							8.73	CEND6	UEP9D		—		
BSD Channels Andiado per Channel UEPB0 MHDO 0.00 15.8 11.90 11.90 Interoffice Channel Millago - 2Wire UEPB0 MGBC 2.52 Image: Channel Status - 2000 <		───	!	├──── ┤							51.05	MALIDA			───′		4-Wire
Interofice Channel Bailesge - 2-Wire UEP80 MGEC 2.52 Image: Constraint of the const	 	───	!	↓ ↓	44.00					45.00				\vdash	──′		
Interdifice Channel Facilities Termination UEP80 MG8C 25.32 Image: Control of Control of Control of Control of Control of Control of Control of Control of Control of Control of Control of Control of Control of Control of Control of Control of Control of Control of Control Of Control		───		↓	11.90					15.69	0.00	MIHDO	UEP9D		J		Interef
Interdifice Channel mileage, per mile or fraction of mile UEP8D MGBM 0.001 Image: Constraint of Constraints of		┥────		├ ───┤							05.00	MICDO			───┘		Interor
Feature Activations (DS) Centrex Loops on Channelized DS1 Service Image: Control of Control Service Image: Co				i – – – – – – – – – – – – – – – – – – –													
DA Channel Bank Feature Activations UP90 Power				i – – – †							0.0091	IVIIGDIVI	UEF9D				Foatur
Feature Activation on D-4 Channel Bank Centrex Loop Slot UEP9D 1POWS 0.66 Image: Constraint on D-4 Channel Bank FX Tunk Side Loop Slot UEP9D 1POWS 0.66 Feature Activation on D-4 Channel Bank Centrex Loop Slot - Defenter Vinc Center UEP9D 1POW7 0.66 Image: Constraint on D-4 Channel Bank Centrex Loop Slot - UEP9D UEP9D 1POW7 0.66 Image: Constraint on D-4 Channel Bank Centrex Loop Slot - UEP9D UEP9D 1POW7 0.66 Image: Constraint on D-4 Channel Bank Tyle Line Loop Slot - UEP9D UEP9D 1POW7 0.66 Image: Constraint on D-4 Channel Bank Tyle Line Loop Slot - UEP9D UEP9D 1POW7 0.66 Image: Constraint on D-4 Channel Bank WITE Loop Slot - UEP9D UEP9D 1POW7 0.66 Image: Constraint on D-4 Channel Bank WITE Loop Slot - UEP9D UEP9D 1POW0 0.66 Image: Constraint on D-4 Channel Bank WITE Loop Slot - UEP9D UEP9D 1POW0 0.66 Image: Constraint on D-4 Channel Bank WITE Loop Slot - UEP9D UEP9D UEP9D 1POW0 0.66 Image: Constraint on D-4 Channel Bank WITE Loop Slot - UEP9D UEP9D USAC2 21.50 8.42 11.90 Image: Constraint on D-4 Channel Bank WITE Loop Slot - WITE Conversion of orbiting Centrex Centranton Block, each UEP9D <t< td=""><td></td><td></td><td></td><td>i – – – †</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>-</td><td></td><td></td><td>-e</td><td></td><td></td></t<>				i – – – †								-			-e		
Feature Activation on D-4 Channel Bank FX line Side Loop Sid UEP9D 1PQW6 0.66 Image: Side Side Side Side Side Side Side Side				 							0.66	1PQWS	UEP9D				04 0110
Feature Activation on D-4 Channel Bank FX Tunk Side Loop UEP90 IPOW7 0.66 Feature Activation on D-4 Channel Bank Centrex Loop Siot - Olfferent Wire Center UEP90 IPOW7 0.66 Feature Activation on D-4 Channel Bank Centrex Loop Siot - Olfferent Wire Center UEP90 IPOW7 0.66 Feature Activation on D-4 Channel Bank Private Line Loop Siot UEP90 IPOW7 0.66 IPOW7 0.66 Feature Activation on D-4 Channel Bank Tije LifeTunk Loop Siot UEP90 IPOW7 0.66 IPOW7 0.66 Feature Activation on D-4 Channel Bank Tije LifeTunk Loop Siot UEP90 IPOWA 0.66 IPOWA 0.66 IPOWA 0.66 RomeReuring Charges (RNG Associated with UNE-P Centrex IPOWA 0.66 IPOWA 0.66 IPOWA 0.66 IPOWA 0.66 IPOWA 0.66 IPOWA 0.66 IPOWA 0.66 IPOWA 0.66 IPOWA 0.66 IPOWA 0.66 IPOWA 0.66 IPOWA 0.66 IPOWA 0.66 IPOWA 0.66 IPOWA 0.66 IPOWA 0.66 IPOWA <															++		
Feature Activation on D-4 Channel Bank FX Trunk Side Loop UEP9D IPOW7 0.66 Feature Activation on D-4 Channel Bank Centrex Loop Sid - Different Wire Center UEP9D IPOW7 0.66 Feature Activation on D-4 Channel Bank Private Line Loop Sid UEP9D IPOW7 0.66 Feature Activation on D-4 Channel Bank Private Line Loop Sid UEP9D IPOW7 0.66 Feature Activation on D-4 Channel Bank Private Line Loop Sid UEP9D IPOW7 0.66 Feature Activation on D-4 Channel Bank Private Line Loop Sid UEP9D IPOW4 0.66 Feature Activation on D-4 Channel Bank WATS Loop Sid UEP9D IPOW4 0.66 Non-Recentring Charges (NRC) Associated with UNE-P Centrex Otheres Standard Centrex Common Biok, each UEP9D USAC2 21.50 8.42 11.90 New Centrex Standard Centrex Common Biok, each UEP9D UMACA 0.00 66.48 11.90 UNEP CENTREX - EWBD (Valid in AL, FL, KY, LA, MS & TN) <				1							0.66	1PQW6	UEP9D			Feature Activation on D-4 Channel Bank FX line Side Loop Slot	
Different Wire Center UEP9D 1PQWP 0.66 Image: Control of Contro											0.66	1PQW7	UEP9D				
Feature Activation on D-4 Channel Bank Tije Line/Trunk Loop Slot UPP9D 1PQWQ 0.66 Image: NRC / Associated with UNEP Centrex Image: NRC / Associated with VIEP Centrex Image: NRC / Associated with VIEP Centrex Image: NRC / Associated with VIEP Centrex <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>0.66</td> <td>1PQWP</td> <td>UEP9D</td> <td></td> <td></td> <td></td> <td></td>											0.66	1PQWP	UEP9D				
Stot UEP9D 1PQWQ 0.66 Image: Control of C											0.66	1PQWV	UEP9D				
Feature Activation on D-4 Channel Bank WATS Loop Stot UEP9D IPQWA 0.66			,													Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop	
Non-Recurring Charges (NRC) Associated with UNE-P Centrex Image: Description of existing Contrently Combined Switch-As-Is with allowed changes, per port UEPD USAC2 21.50 8.42 11.90 Image: Description of existing Centrex Common Block, each UEPD USAC2 21.50 8.42 11.90 Image: Description of existing Centrex Common Block, each UEPD USAC2 21.50 8.42 11.90 Image: Description of existing Centrex Common Block, each UEPDD USAC2 21.50 8.42 11.90 Image: Description of existing Centrex Common Block, each UEPDD USACN 5.17 8.32 11.90 Image: Description of existing Centrex Costomized Common Block UEPPDD USACN 6.17 8.32 11.90 Image: Description of existing Centrex Costomized Common Block UEPPDD URECA 0.00 618.82 11.90 Image: Description of existing Centrex Costomized Common Block UEPPDD URECA 0.00 66.48 Image: Description of existing Centrex Costomized Common Block Image: Description of existing Centrex Costomized Common Block Image: Description of existing Centrex Costomized Common Block Image: Description of existing Centrex Costomized Common Block Image: Description of existing Centrex Costomized Common Block Image: D			ļ														
NRC Conversion Currently Combined Switch-As-Is with allowed changes, per port UEP9D USAC2 21.50 8.42 11.90 1 Conversion of existing Centrex Common Block, each UEP9D USACN 5.17 8.32 11.90 1 New Centrex Standard Common Block UEP9D MACS 0.00 618.82 11.90 1 New Centrex Standard Common Block UEP9D MACC 0.00 618.82 11.90 1 NAR Establishment Charge, Per Occasion UEP9D UREAC 0.00 66.48 11.90 1 UNEP CENTREX - EWSD (Vaid in AL, FL, KY, LA, MS & TN) Image: Comparison of existing Centrex Combo Image: Comparison of existing Centrex Combo 1		<u> </u>	!								0.66	1PQWA	UEP9D				
changes.per port UEP9D USAC2 21:50 8.42 11:90 1 Conversion of existing Centrex Common Block, each UEP9D USACN 5:17 8.32 11:90 1 New Centrex Standard Common Block UEP9D MIACS 0.00 618.82 11:90 1 New Centrex Sustomized Common Block UEP9D MIACS 0.00 618.82 11:90 1 New Centrex Sustomized Common Block UEP9D MIACC 0.00 664.82 11:90 1 UNE Port/Loop Combination Rates (Non-Design Image: Centrex Customized Controp Image: Centrex Customized Centrex Customized Centrex Port Combo Image: Centrex Customized Centrex Port Centrex Port Combo Image: Centrex Customized Centrex Customized Centrex Port Centrex Port Combo Image: Centrex Customized Centrex Port Cent			!	L													Non-R
Conversion of existing Centrex Common Block, each UEP9D USACN 5.17 8.32 11.90 Image: Centrex Standard Common Block UEP9D M1ACS 0.00 618.82 11.90 Image: Centrex Standard Common Block 11.90 Image: Centrex Standard Common Block UEP9D M1ACS 0.00 618.82 Image: Centrex Standard Common Block 11.90 Image: Centrex Standard Common Block 11.90 Image: Centrex Standard Common Block I			, ,	1					a (a								
New Centrex Standard Common Block UEP9D M1ACS 0.00 618.82 11.90 New Centrex Customized Common Block UEP9D M1ACC 0.00 618.82 11.90 11.90 New Centrex Customized Common Block UEP9D MACC 0.00 618.82 11.90 11.90 NAR Establishment Charge, Per Occasion UEP9D URECA 0.00 66.88 11.90 11.90 11.90 UNE-P CENTREX - EWSD (Valid in AL, FL, KY, LA, MS & TN) 11.90 <td< td=""><td></td><td>───</td><td>!</td><td>┟────┤</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>───┘</td><td></td><td></td></td<>		───	!	┟────┤											───┘		
New Centrex Customized Common Block UEP9D MACC 0.00 618.82 11.90 NAR Establishment Charge, Per Occasion UEP9D UREA 0.00 66.48 11.90 UNE-P CENTREX - EWSD (Valid in AL, FL, KY, LA, MS & TN)		───		├──── ┤					8.32		0.00				──′		
NAR Establishment Charge, Per Occasion UEP9D URECA 0.00 66.48 11.90 1 UNE-P CENTREX - EWSD (Valid in AL, FL, KY, LA, MS & TN)		┥────		⊢													
UNE-P CENTREX - EWSD (Valid in AL, FL, KY, LA, MS & TN) Image: Constraint of the set o		 		 											┽───┤		
2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Combo Image: Combination Rates (Non-Design) Image: Centrex) Port Combo Im		<u> </u>		┌─── ┤	11.00					50.40	0.00	0			+		UNE-P
UNE Port/Loop Combination Rates (Non-Design) I Image: Complexity of the compl		<u> </u>											1		1		
2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo- Non-Design 1 UEP9E 10.94		1											İ		1		
Non-Design 2 UEP9E 15.05 Image: Constraint of the second s											10.94		UEP9E	1	-	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo	
2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 3 UEP9E 25.80 Image: Comparison of the comparis														2		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo	
UNE Port/Loop Combination Rates (Design) I Image: Combination Rates (Design) Image: Combination Rates (Desig										ľ				3		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo	
2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 1 UEP9E 13.41 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2 UEP9E 18.57 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2 UEP9E 18.57		1													1		UNE P
Design 2 UEP9E 18.57 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - <											13.41		UEP9E	1	1	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo	
2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -											18.57		UEP9E	2		Design	
Design 3 UEP9E 32.04											32.04		UEP9E	3		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo	

NDUNDLE	D NETWORK ELEMENTS - Florida	1		1	- <u> </u>						0	0 O	Attachment:			bit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrec	curring	Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9E	UECS1	9.77										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9E	UECS1	13.88										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP9E	UECS1	24.63										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9E	UECS2	12.24										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9E	UECS2	17.40										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9E	UECS2	30.87										
	ort Rate															
AL, FL	, KY, LA, MS, & TN only 2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9E	UEPYA	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex) Basic Local Alea 2-Wire Voice Grade Port (Centrex 800 termination)Basic Local			UEF9E	UEPTA	1.17	03.31	20.40	27.50	0.37		11.90				
	Area			UEP9E	UEPYB	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local					4 47	50.04	00.40	07.50	0.07		44.00				
	Area			UEP9E	UEPYH	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 Basic Local Area			UEP9E	UEPYM	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			ULF JL		1.17	139.49	00.10	00.41	13.01	t	11.90				
	Term - Basic Local Area			UEP9E	UEPYZ	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			OEI OE	02112	1.17	100.40	00.10	00.41	10.01		11.00				
	- Basic Local Area			UEP9E	UEPY9	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port Terminated on 800 Service Term -			02.02	02.10		00.01	20.10	21.00	0.01	1	11.00				
	Basic Local Area			UEP9E	UEPY2	1.17	53.31	26.46	27.50	8.37		11.90				
Florida																
	2-Wire Voice Grade Port (Centrex)			UEP9E	UEPHA	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP9E	UEPHB	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP9E	UEPHH	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP9E	UEPHM	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP9E	UEPHZ	1.17	139.49	86.10	65.41	13.81		11.90				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9E	UEPH9	1.17	53.31	26.46	27.50	8.37		11.90				
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP9E	UEPH2	1.17	53.31	26.46	27.50	8.37		11.90				
Local S	Switching															
	Centrex Intercom Funtionality, per port			UEP9E	URECS	0.7384										
Local N	umber Portability															
	Local Number Portability (1 per port)			UEP9E	LNPCC	0.35										
Feature																
	All Standard Features Offered, per port			UEP9E	UEPVF	2.26	0.00.00									
	All Select Features Offered, per port			UEP9E	UEPVS	0.00	370.70		-		-	11.90				
NARS	All Centrex Control Features Offered, per port			UEP9E	UEPVC	2.26										
NARS	Unbundled Network Assess Register Combination				LIADCY	0.00	0.00	0.00				11.90				
	Unbundled Network Access Register - Combination Unbundled Network Access Register - Indial			UEP9E UEP9E	UARCX UAR1X	0.00	0.00	0.00				11.90				
	Unbundled Network Access Register - Indial			UEP9E UEP9E	UAR1X	0.00	0.00	0.00		1		11.90				
Miscoll	aneous Terminations			OLI SL	UAILOA	0.00	0.00	0.00			<u> </u>	11.90				
	Trunk Side															
2 1110	Trunk Side Terminations, each			UEP9E	CEND6	8.73					t					
4-Wire	Digital (1.544 Megabits)												1			
	DS1 Circuit Terminations, each		1	UEP9E	M1HD1	54.95					1		l		ĺ	
	DS0 Channel Activated Per Channel	1		UEP9E	M1HDO	0.00	15.69					11.90				
Interof	ice Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination			UEP9E	MIGBC	25.32										
	Interoffice Channel mileage, per mile or fraction of mile			UEP9E	MIGBM	0.0091										
	e Activations (DS0) Centrex Loops on Channelized DS1 Servic	e														
D4 Cha	nnel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot		L	UEP9E	1PQWS	0.66										
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9E	1PQW6	0.66										

	D NETWORK ELEMENTS - Florida	-		1							-	-	Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonree		Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP9E	1PQW7	0.66										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -			ULF9L	IFQWI	0.00										
	Different Wire Center			UEP9E	1PQWP	0.66										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9E	1PQWV	0.66										
	Feature Activation on D-4 Channel Bank Tivate Line/Trunk Loop				ii qiii	0.00										
	Slot			UEP9E	1PQWQ	0.66										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9E	1PQWA	0.66										
Non-R	Recurring Charges (NRC) Associated with UNE-P Centrex															
1	NRC Conversion Currently Combined Switch-As-Is with allowed		1		110.000		04 -	0.10				44.00				
 	changes, per port Conversion of Existing Centrex Common Block, each	ļ	ļ	UEP9E UEP9E	USAC2 USACN		21.50 5.17	8.42 8.32				11.90 11.90				
	New Centrex Standard Common Block			UEP9E	M1ACS	0.00	618.82	8.32				11.90				
<u> </u>	New Centrex Standard Common Block	ł		UEP9E	MIACS	0.00	618.82					11.90				
<u> </u>	NAR Establishment Charge, Per Occasion	1	1	UEP9E	URECA	0.00	66.48					11.90				
Note 1	I - Required Port for Centrex Control in 1AESS, 5ESS & EWSD	1	1			1.00										
	2 - Requres Interoffice Channel Mileage															
	3 - Requires Specific Customer Premises Equipment															
	CENTREX PORT/LOOP COMBINATIONS - MARKET RATES															
	ket Rates are applied where BellSouth is not required by FCC					ndled Local Sw	itching or Sw	tch Ports.								
	curring Charges for all Standard Centrex and Centrex Conrol Fe															
	I Office and Tandem Switching Usage and Common Transport															
	first and additional Port nonrecurring charges apply to Not Co	urrently	Comb	ined Combos. Fo	or Currently Co	mbined Combo	s, the nonrecu	urring charges	shall be those	identified in t	he Nonrecu	rring - Curre	ently Combine	ed sections.	Additional NR	Cs may
	also and are categorized accordingly.				1			-	1	1						
UNE-P	P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only)														
UNE-P 2-Wire	P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only P VG Loop/2-Wire Voice Grade Port (Centrex) Combo)														
UNE-P 2-Wire	P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only P VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design))														
UNE-P 2-Wire	P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only P VG Loop/2-Wire Voice Grade Port (Centrex) Combo)	1	UEP91		26.94										
UNE-P 2-Wire	CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -)														
UNE-P 2-Wire	CENTREX - 1AËSS - (Valid in AL, FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Combo 'ort/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design)	1	UEP91 UEP91		26.94 31.06										
UNE-P 2-Wire	CENTREX - 1AËSS - (Valid in AL, FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire VOICe Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire VOICe Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire VG Loop/2-W)	2	UEP91		31.06										
UNE-P	CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design															
UNE-P	CENTREX - 1AËSS - (Valid in AL, FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire VOICe Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire VOICe Grade Port (Centrex)Port Combo - 2-Wire VG Loop/2-Wire VG Loop/2-W		2	UEP91		31.06										
UNE-P	CENTREX - 1AËSS - (Valid in AL, FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Combo 'ort/Loop Combination Rates (Mon-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Combination Rates (Design))	2	UEP91		31.06										
UNE-P	CENTREX - 1AËSS - (Valid in AL,FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Don-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		2 3 1	UEP91 UEP91 UEP91		31.06 45.87 29.36										
UNE-P	ZCENTREX - 1AËSS - (Valid in AL, FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Combo Ort/Loop Combination Rates (Non-Design) Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Jon-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Jon-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Jon-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design		2	UEP91 UEP91		31.06 45.87										
UNE-P	ZCENTREX - 1AËSS - (Valid in AL, FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Combo Zont/Loop Combination Rates (Non-Design) Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port		2 3 1 2	UEP91 UEP91 UEP91 UEP91		31.06 45.87 29.36 34.43										
UNE-P	CENTREX - 1AËSS - (Valid in AL,FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Combo Ort/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Don-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Don-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design		2 3 1	UEP91 UEP91 UEP91		31.06 45.87 29.36										
UNE-P	CENTREX - 1AËSS - (Valid in AL, FL,GA,KY,LA,MS,&TN only OS Loop/2-Wire Voice Grade Port (Centrex) Combo 'ort/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)		2 3 1 2 3	UEP91 UEP91 UEP91 UEP91 UEP91		31.06 45.87 29.36 34.43 50.68										
UNE-P	ZCENTREX - 1AËSS - (Valid in AL, FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Combo Ort/Loop Combination Rates (Non-Design) Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Son-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire Voice Grade Loop (SL 1) - Zone 1 Z-Wire Voice Grade Port Z-Wire Voice Grade Loop (SL 1) - Zone 1		2 3 1 2 3 3	UEP91 UEP91 UEP91 UEP91 UEP91 UEP91		31.06 45.87 29.36 34.43 50.68 12.94										
UNE-P	ZCENTREX - 1AËSS - (Valid in AL,FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Combo Zort/Loop Combination Rates (Non-Design) Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Zort/Loop Combination Rates (Design) Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Z-Wire VG Loop/2-Wire Voice Grade Loop (SL 1) - Zone 1 Z-Wire Voice Grade Loop (SL 1) - Zone 2		2 3 1 2 3 3 1 2 3	UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1	31.06 45.87 29.36 34.43 50.68 12.94 17.06										
UNE-P	ZENTREX - 1AËSS - (Valid in AL, FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Combo */vLoop Combination Rates (Non-Design) Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Sign Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire Voice Grade Loop (SL 1) - Zone 1 Z-Wire Voice Grade Loop (SL 1) - Zone 3		2 3 1 2 3 3	UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1	31.06 45.87 29.36 34.43 50.68 12.94 17.06 31.87										
UNE-P	CENTREX - 1AËSS - (Valid in AL, FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Combo 'ort/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 1		2 3 1 2 3 3 1 2 3	UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS2	31.06 45.87 29.36 34.43 50.68 12.94 17.06										
UNE-P	ZENTREX - 1AËSS - (Valid in AL, FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Combo */vLoop Combination Rates (Non-Design) Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Dosign Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design Z-Wire Voice Grade Loop (SL 1) - Zone 1 Z-Wire Voice Grade Loop (SL 1) - Zone 2 Z-Wire Voice Grade Loop (SL 1) - Zone 3		2 3 1 2 3 3 1 2 3 1	UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1	31.06 45.87 29.36 34.43 50.68 12.94 17.06 31.87 15.36										
UNE P	ZENTREX - 1AËSS - (Valid in AL, FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Combo Ovr/Loop Combination Rates (Non-Design) Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire Voice Grade Loop (SL 1) - Zone 1 Z-Wire Voice Grade Loop (SL 1) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 1 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (2 3 1 2 3 1 2 3 1 2 3 1 2	UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS2 UECS2	31.06 45.87 29.36 34.43 50.68 12.94 17.06 31.87 15.36 20.43										
UNE P	ZCENTREX - 1AËSS - (Valid in AL, FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Combo Yor/Loop Combination Rates (Non-Design) Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire Voice Grade Loop (SL 1) - Zone 1 Z-Wire Voice Grade Loop (SL 1) - Zone 2 Z-Wire Voice Grade Loop (SL 1) - Zone 2 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop		2 3 1 2 3 1 2 3 1 2 3 1 2	UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS2 UECS2 UECS2	31.06 45.87 29.36 34.43 50.68 12.94 17.06 31.87 15.36 20.43 36.68										
UNE P	CENTREX - 1AËSS - (Valid in AL, FL,GA,KY,LA,MS,&TN only Os Loop/2-Wire Voice Grade Port (Centrex) Combo Ort/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo 2-Wire VGice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3 3-Wire Voice Grade Loop (SL 2) - Zone 3 3-Wire Voice Grade Loop (SL 2) - Zone 3 3-Wire Voice Grade Loop (SL 2) - Zone 3 3-Wire Voice Grade Loop (SL 2) - Zone 3 3-Wire Voice Grade Loop (SL 2) - Zone 3 3-Wire Voice Grade Loop (SL 2) - Zone 3 3-Wire Voice Grade Loop (SL 2) - Zone 3 3-Wire Voice Grade Loop (SL 2) - Zone 3 3-Wire Voice Grade Loop (SL 2) - Zone 3 3-Wire Voice Grade Loop (SL 2) - Zone 3 3-Wire Voice Grade Loop (SL 2) - Zone 3 3-Wire Voice Grade Loop (SL 2) - Zone 4 3-Wire Voice Grade Loop (SL 2) - Zone 4 3-Wire Voice Grade Loop (SL 2) - Zone 4 3-Wire Voice Grade Loop (SL 2) - Zone 4 3-Wi		2 3 1 2 3 1 2 3 1 2 3 1 2	UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS2 UECS2	31.06 45.87 29.36 34.43 50.68 12.94 17.06 31.87 15.36 20.43	70.00	35.00	35.00	10.00		11.90				
UNE P	ZENTREX - 1AËSS - (Valid in AL, FL,GA,KY,LA,MS,&TN only SVG Loop/2-Wire Voice Grade Port (Centrex) Combo Yort/Loop Combination Rates (Non-Design) Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Z-Wire Voice Grade Loop (SL 1) - Zone 1 Z-Wire Voice Grade Loop (SL 1) - Zone 2 Z-Wire Voice Grade Loop (SL 2) - Zone 1 Z-Wire Voice Grade Loop (SL 2) - Zone 1 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (Centrex) Basic Local Area Z-Wire Voice Grade Port (Centrex 800 termination)Basic Local		2 3 1 2 3 1 2 3 1 2 3 1 2	UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS2 UECS2 UECS2 UECS2 UECS2	31.06 45.87 29.36 34.43 50.68 12.94 17.06 31.87 15.36 20.43 36.68 30.68										
UNE P	ZENTREX - 1AËSS - (Valid in AL, FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Combo Ovr/Loop Combination Rates (Non-Design) Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire Voice Grade Loop (SL 1) - Zone 1 Z-Wire Voice Grade Loop (SL 1) - Zone 2 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Port (Centrex) Basic Local Area Z-Wire Voice Grade Port (Centrex) Basic Local Area Z-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area		2 3 1 2 3 1 2 3 1 2 3 1 2	UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS2 UECS2 UECS2	31.06 45.87 29.36 34.43 50.68 12.94 17.06 31.87 15.36 20.43 36.68	70.00	35.00	35.00	10.00		11.90				
UNE P	ZCENTREX - 1AËSS - (Valid in AL, FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Combo Yor/Loop Combination Rates (Non-Design) Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire Voice Grade Loop (SL 1) - Zone 1 Z-Wire Voice Grade Loop (SL 1) - Zone 2 Z-Wire Voice Grade Loop (SL 1) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Port (Centrex N) Basic Local Area Z-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area Z-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Z-Wire Voice Grade Port (Centrex with Caller ID)		2 3 1 2 3 1 2 3 1 2 3 1 2	UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS2 UECS2 UECS2 UECS2 UEPYA UEPYB	31.06 45.87 29.36 34.43 50.68 12.94 17.06 31.87 15.36 20.43 36.68 	70.00	35.00	35.00	10.00		11.90				
UNE P	CENTREX - 1AËSS - (Valid in AL, FL,GA,KY,LA,MS,&TN only Os Loop/2-Wire Voice Grade Port (Centrex) Combo ort/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo 2-Besign 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo 2-Besign 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo 2-Besign 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 4 2-Wire Voice Grade Loop (SL 2) - Zone 4 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 4 2-Wire Voice Grade Loop (SL 2) - Zone 4 2-Wire Voice Grade Loop (SL 2) - Zone 4 2-Wire Voice Grade Loop (SL 2) - Zone 5 2-Wire Voice Grade Loop (SL 2) - Zone 4 2-Wire Voice Grad		2 3 1 2 3 1 2 3 1 2 3 1 2	UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS2 UECS2 UECS2 UECS2 UECS2	31.06 45.87 29.36 34.43 50.68 12.94 17.06 31.87 15.36 20.43 36.68 30.68										
UNE P	ZCENTREX - 1AËSS - (Valid in AL, FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Combo Yor/Loop Combination Rates (Non-Design) Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Design Z-Wire Voice Grade Loop (SL 1) - Zone 1 Z-Wire Voice Grade Loop (SL 1) - Zone 2 Z-Wire Voice Grade Loop (SL 1) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Loop (SL 2) - Zone 3 Z-Wire Voice Grade Port (Centrex N) Basic Local Area Z-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area Z-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Z-Wire Voice Grade Port (Centrex with Caller ID)		2 3 1 2 3 1 2 3 1 2 3 1 2	UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS2 UECS2 UECS2 UECS2 UEPYA UEPYB	31.06 45.87 29.36 34.43 50.68 12.94 17.06 31.87 15.36 20.43 36.68 	70.00	35.00	35.00	10.00		11.90				
ONRONDE	ED NETWORK ELEMENTS - Florida	1	1	I	1						-		Attachment:			ibit: B
----------	--	-------------	------	----------------	----------------	--------	--------	-----------	--------------	------------	---	-----------------------	-------------	-----------	---	----------
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Submitted Manually		Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		-
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port terminated in on Megalink or equivalent															
	- Basic Local Area			UEP91	UEPY9	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port Terminated on 800 Service Term -					11.00	70.00	05.00	05.00	10.00		44.00				
0	Basic Local Area gia and Florida Only			UEP91	UEPY2	14.00	70.00	35.00	35.00	10.00		11.90				
Geor	2-Wire Voice Grade Port (Centrex)	-	-	UEP91	UEPHA	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex 800 termination)		-	UEP91	UEPHB	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP91	UEPHH	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire		1	OEI OI	0EI IIII	14.00	70.00	00.00	00.00	10.00		11.00				
	Center)2			UEP91	UEPHM	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term			UEP91	UEPHZ	14.00	180.00	110.00	85.00	20.00		11.90				
			1				1									
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP91	UEPH9	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP91	UEPH2	14.00	70.00	35.00	35.00	10.00		11.90				
Loca	I Switching															
	Centrex Intercom Funtionality, per port			UEP91	URECS	0.7384										
Loca	I Number Portability															
	Local Number Portability (1 per port)			UEP91	LNPCC	0.35										
Featu																
	All Standard Features Offered, per port			UEP91	UEPVF	0.00						11.90				
	All Select Features Offered, per port			UEP91	UEPVS	0.00	370.70					11.90				
NAD	All Centrex Control Features Offered, per port			UEP91	UEPVC	0.00						11.90				
NARS				UEP91	UARCX	0.00	0.00	0.00				11.90				
	Unbundled Network Access Register - Combination			UEP91 UEP91	UARCX UAR1X	0.00	0.00	0.00				11.90				
	Unbundled Network Access Register - Indial Unbundled Network Access Register - Outdial			UEP91	UAROX	0.00	0.00	0.00				11.90				
Misc	ellaneous Terminations			OLF91	UAROA	0.00	0.00	0.00				11.90				
	e Trunk Side		-													1
2 000	Trunk Side Terminations, each		1	UEP91	CENA6	8.81										
Inter	office Channel Mileage - 2-Wire			02.01	021010	0.01										1
	Interoffice Channel Facilities Termination - Voice Grade		1	UEP91	M1GBC	25.32										
	Interoffice Channel mileage, per mile or fraction of mile			UEP91	M1GBM	0.0091										
Featu	ire Activations (DS0) Centrex Loops on Channelized DS1 Service	e														
D4 C	hannel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP91	1PQWS	0.66										
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP91	1PQW6	0.66										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop															
	Slot			UEP91	1PQW7	0.66										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -					0.00										
	Different Wire Center		-	UEP91	1PQWP	0.66										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP91	1PQWV	0.66										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop			025.91	IFQVV	0.00										
	Slot		1	UEP91	1PQWQ	0.66										1
	Feature Activation on D-4 Channel Bank WATS Loop Slot	t	1	UEP91	1PQWQ 1PQWA	0.66					1					t
Non-	Recurring Charges (NRC) Associated with UNE-P Centrex	t	1			0.00					1					<u> </u>
	Conversion - Currently Combined Switch-As-Is with allowed		1										İ	İ	İ	<u> </u>
	changes, per port		1	UEP91	USAC2		21.50	8.42				11.90				1
	Conversion of Existing Centrex Common Block	I	1	UEP91	USACN		5.17	8.32			1	11.90				
	New Centrex Standard Common Block			UEP91	M1ACS	0.00	618.82					11.90				
	New Centrex Customized Common Block			UEP91	M1ACC	0.00	618.82					11.90				
	Secondary Block, per Block			UEP91	M2CC1	0.00	71.31					11.90				
	NAR Establishment Charge, Per Occasion			UEP91	URECA	0.00	66.48					11.90				
	P CENTREX - 5ESS (Valid in All States)		I													
	e VG Loop/2-Wire Voice Grade Port (Centrex) Combo															ļ
UNE	Port/Loop Combination Rates (Non-Design)										1					1

UNBUNDLI	ED NETWORK ELEMENTS - Florida												Attachment:	2		bit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															i i
	Non-Design		1	UEP95	_	26.94										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		2	UEP95		31.06										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															i
	Non-Design		3	UEP95	_	45.87										L
UNE	Port/Loop Combination Rates (Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -		1			20.20										i -
	Design		1	UEP95		29.36					-					i
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		2	UEP95		34.43										i i
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			025,90	+	34.43					+					
	Design		3	UEP95		50.68					1					i
UNF	Loop Rate		3	021 30		50.00										<u> </u>
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP95	UECS1	12.94										
	2-Wire Voice Grade Loop (SL 1) - Zone 1		2	UEP95	UECS1	17.06					1					<u> </u>
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP95	UECS1	31.87					1				İ	<u> </u>
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP95	UECS2	15.36										-
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP95	UECS2	20.43										[
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP95	UECS2	36.68										
UNE	Port Rate															(
All St	ates															ſ
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP95	UEPYA	14.00	70.00	35.00	35.00	10.00		11.90				(
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPYB	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local															1
	Area			UEP95	UEPYH	14.00	70.00	35.00	35.00	10.00		11.90				I
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 Basic Local Area			UEP95	UEPYM	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															(
	Term - Basic Local Area			UEP95	UEPYZ	14.00	180.00	110.00	85.00	20.00		11.90				1
	2-Wire Voice Grade Port terminated in on Megalink or equivalent															1
	- Basic Local Area			UEP95	UEPY9	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port Terminated on 800 Service Term -															i
	Basic Local Area			UEP95	UEPY2	14.00	70.00	35.00	35.00	10.00		11.90				L
	Y, LA, MS, SC, & TN Only				_											I
FL &	GA Only									10.00	-					ļ
	2-Wire Voice Grade Port (Centrex)			UEP95	UEPHA	14.00	70.00	35.00	35.00	10.00		11.90				ļ
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPHB	14.00	70.00	35.00	35.00	10.00		11.90				<u> </u>
_	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP95	UEPHH	14.00	70.00	35.00	35.00	10.00	+	11.90				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2	1		UEP95	UEPHM	14.00	180.00	110.00	85.00	20.00	1	11.90				1
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			0LF 30		14.00	100.00	110.00	00.00	20.00	+	11.90				
	Term			UEP95	UEPHZ	14.00	180.00	110.00	85.00	20.00	1	11.90				i i
				52.00	001112	14.00	100.00	110.00	00.00	20.00	1	11.30				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP95	UEPH9	14.00	70.00	35.00	35.00	10.00	1	11.90				i i
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP95	UEPH2	14.00	70.00	35.00	35.00	10.00		11.90			İ	[
Local	Switching	1	1								1				İ	
	Centrex Intercom Funtionality, per port	1	1	UEP95	URECS	0.7384					1				İ	
Local	Number Portability										1					
	Local Number Portability (1 per port)			UEP95	LNPCC	0.35										
Featu																
	All Standard Features Offered, per port			UEP95	UEPVF	0.00										
	All Select Features Offered, per port			UEP95	UEPVS	0.00	370.70					11.90				ļ
	All Centrex Control Features Offered, per port			UEP95	UEPVC	0.00										
NARS											-					
	Unbundled Network Access Register - Combination		L	UEP95	UARCX	0.00	0.00	0.00				11.90				───
	Unbundled Network Access Register - Indial		L	UEP95	UAR1X	0.00	0.00	0.00				11.90				───
	Unbundled Network Access Register - Outdial			UEP95	UAROX	0.00	0.00	0.00				11.90				└───
MISCE	ellaneous Terminations										1					L

CATEGORY RATE ELEMENTS Interi Zone BCS USOC RATES(\$) Submitted Submitted Charge - Manual Svc Charge - Manual Svc Charge - Manual Svc Manual Svc </th <th></th> <th>ORK ELEMENTS - Florida</th> <th></th> <th>1</th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th> <th> ·</th> <th>Attachment:</th> <th></th> <th></th> <th>bit: B</th>		ORK ELEMENTS - Florida		1									·	Attachment:			bit: B
Image: Sign: 2007 Image: Sign: 2007 <thimage: 2007<="" sign:="" th=""> Image: Sign: 2007</thimage:>		RATE ELEMENTS		Zone	BCS	USOC			RATES(\$)			Submitted Elec	Submitted Manually	Manual Svc Order vs. Electronic-	Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge - Manual Sv Order vs. Electronic Disc Add
Jown Town State Print April Print April Print April Diractic Diract							Rec										
Inter Star Terms Star Terms Star Star Star Star Star Star Star Star		-					100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Attract State Life/Bit					LIEDOC		0.04										
BS 10 Oral Termstoor, ach UEPBe NH1401 64.86 </td <td></td> <td></td> <td></td> <td></td> <td>UEP95</td> <td>CEND6</td> <td>8.81</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>					UEP95	CEND6	8.81										
BSD Charmels Acoustics and Interface Charmel Millage 2 Milling CEP30 MIHOD 0.00 15.60 MIHOD 15.60 MIHOD					LIEP95	M1HD1	54.95										
Interface Channel Millages - Wine Like PM ACGAC PA D <thd< t+<="" td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>15.69</td><td></td><td></td><td></td><td></td><td>11.90</td><td></td><td></td><td></td><td></td></thd<>								15.69					11.90				
Interfactor Interfactor Idea <td>Chann</td> <td>nel Mileage - 2-Wire</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	Chann	nel Mileage - 2-Wire															
Feature Activations (050) Centres Loops on Channelized DS1 Service Image: Centre Loops	eroffice	e Channel Facilities Termination															
Dit Channel Bark Parture Activations Import I					UEP95	MIGBM	0.0091										
Image: Planue Activation on D-4 Channel Bank FX into Side Loop Side UEP66 1PQW5 0.66 Image: Planue Activation on D-4 Channel Bank FX into Side Loop Side Image: Planue Activation on D-4 Channel Bank FX into Side Loop Side Image: Planue Activation on D-4 Channel Bank FX into Side Loop Side Image: Planue Activation on D-4 Channel Bank FX into Side Loop Side Image: Planue Activation on D-4 Channel Bank Finate Line Loop Side Image: Planue Activation on D-4 Channel Bank Finate Line Loop Side UEP66 1PQW7 0.66 Image: Planue Activation on D-4 Channel Bank Finate Line Loop Side UEP66 1PQW7 0.66 Image: Planue Activation on D-4 Channel Bank Finate Line Loop Side UEP66 1PQW7 0.66 Image: Planue Activation on D-4 Channel Bank Finate Line Loop Side UEP66 1PQW7 0.66 Image: Planue Activation on D-4 Channel Bank Materia UEP66 1PQW7 0.66 Image: Planue Activation on D-4 Channel Bank Materia UEP66 Image: Planue Activation on D-4 Channel Bank Materia UEP66 Image: Planue Activation on D-4 Channel Bank Materia UEP66 Image: Planue Activation on D-4 Channel Bank Bank Bank Bank Bank UEP66 Image: Planue Activation on D-4 Channel Bank Bank Bank Bank Bank Bank Bank Bank			ce														
Peature Activation on D-4 Channel Bank FX Tunk Side Loop Sixt UEP95 1PQW6 0.66 Peature Activation on D-4 Channel Bank FX Tunk Side Loop Sixt UEP95 IPQW7 0.66 IPQW7 0.66 Peature Activation on D-4 Channel Bank FX Tunk Side Loop Sixt UEP95 IPQW7 0.66 IPQW7 0.66 Peature Activation on D-4 Channel Bank Typi LimPTurk Loop UEP95 IPQW7 0.66 IPQW7 0.66 Feature Activation on D-4 Channel Bank Typi LimPTurk Loop UEP95 IPQW7 0.66 IPQW7 0.66 Feature Activation on D-4 Channel Bank Typi LimPTurk Loop UEP95 IPQW7 0.66 IPQW7 0.66 IPQW7 0.66 IPQW7 0.66 IPQW7 0.66 IPQW7 0.66 IPQW7 0.66 IPQW7 0.66 IPQW7 0.66 IPQW7 0.66 IPQW7 0.66 IPQW7 0.66 IPQW7 0.66 IPQW7 0.66 IPQW7 0.66 IPQW7 0.66 IPQW7 0.66 IPQW7 0.66 IPQW7 0.66 IPQW7 0.67 IPQW7					115045	100110											
Feature Activation on D4 Channel Bank FAT Turk Side Loop UEP96 1PQW7 0.66 Feature Activation on D4 Channel Bank Centres Loop Sidt UEP96 1PQW9 0.66	ature Ad	Activation on D-4 Channel Bank Centrex Loop Slot			UEP95	1PQWS	0.66										
Feature Activation on D-A Channel Bank Centres Loop Sidt UEP9 1PQW7 0.66 Feature Activation on D-A Channel Bank Centres Loop Sidt UEP96 IPQWP 0.66 IPQW2 IPPQW2 IPQW2	ature A	Activation on D-4 Channel Bank FX line Side Loop Slot		1	LIEP95	1POW6	0.66										
Std. UP289 IPQW7 0.66 UP280 IPQW7<			1		021 30	11 2000	0.00					1					
Flature Activation on D-4 Channel Bank Controx Loop Stot LEPos IPQu/P 0.66 Image: Control C					UEP95	1PQW7	0.66										
Different Vine Canter UEP85 IPQWP 0.66 Image: Control of Control Bank Tipe Long Stot Image: Control Bank Tipe Long Sto		Activation on D-4 Channel Bank Centrex Loop Slot -															
Feature Activation on D-4 Channel Bank Tije LineTrunk (Loop Stot LineTrunk (Loop Stot LinePois IPOWQ 0.66 IPOWQ 0.66 Feature Activation on D-4 Channel Bank WATS Loop Stot UEP95 IPOWQ 0.66 IPOWQ 0.66 IPOWQ IPOWQ 0.66 IPOWQ IPOWQ 0.66 IPOWQ <td< td=""><td>ferent V</td><td>Wire Center</td><td></td><td></td><td>UEP95</td><td>1PQWP</td><td>0.66</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></td<>	ferent V	Wire Center			UEP95	1PQWP	0.66										
Feature Activation on D-4 Channel Bank Tije LineFTunk Loop UE P95 IPOWQ 0.66 IPOWQ 0.66 Feature Activation on D-4 Channel Bank WATS Loop Site UE P95 IPOWQ 0.66 IPOWQ 0.66 IPOWQ																	
Std UEP9S 1POWQ 0.68 Image: Control of Co					UEP95	1PQWV	0.66										
Peature Activation on D-4 Channel Bark WATS Loop Sixt UEP96 IPOWA 0.66 IPOWA 0.66 IPOWA 0.66 IPOWA 0.66 IPOWA 0.66 IPOWA 0.66 IPOWA 0.66 IPOWA 0.66 IPOWA 0.67 <th0.67< th=""> 0.67<td></td><td>Activation on D-4 Channel Bank Tjie Line/Trunk Loop</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></th0.67<>		Activation on D-4 Channel Bank Tjie Line/Trunk Loop															
Non-Recurring Charges (NRC) Associated with UNE-P Centrex Image		ativation on D.4 Channel Deals WATC Loop Clat															
NRC Conversion Currently Combined Switch-As-Is with allowed changes, per port UEP95 USAC2 0.00 21.50 8.42 11.90 Conversion of Existing Carters Common Block UEP95 USACA 0.00 51.77 8.32 11.90 New Centres Standard Common Block UEP95 MIACS 0.00 618.82 11.90 New Centres Customized Common Block UEP95 MIACS 0.00 618.82 11.90 VINE Conversion UEP95 MIACS 0.00 66.48 11.90 UNE-POTLoop Combination Rates (Non-Design) 2-Wine VG Loop/2-Wire Voice Grade Port (Centres) Port Combo 1 UEP9D 26.94 2-Wire VG Loop/2-Wire Voice Grade Port (Centres) Port Combo 1 UEP9D 31.06			-		UEP95	IPQWA	0.66										
changes, per port UEP95 USACN 5.17 8.42 11.90 11.90 Conversion of Existing Centrex Common Block, each UEP95 MIACS 0.00 618.82 11.90 11.90 11.90 New Centrex Standard Common Block UEP95 MIACS 0.00 618.82 11.90																	
Conversion of Existing Centrex Common Block, each UEP95 UBAR 5.77 8.32 11.80 New Centrex Standard Common Block UEP95 MIACS 0.00 618.82 11.90 New Centrex Standard Common Block UEP96 MIACS 0.00 618.82 11.90 NRAE Existing Centrex Common Block, each UEP96 MIACC 0.00 66.48 11.90 UNE PORTLOOP Centrination Rates (Non-Design) UEP96 WECA 0.00 66.48					UEP95	USAC2	0.00	21.50	8.42				11.90				
New Centres Customized Common Block UEP9S MACC 0.00 618.2 11.90 11.90 NAR Establishmer Charge, Per Occasion UEP9S URECA 0.00 66.48 11.90																	
NARE stabilishment Charge. Per Occasion UPP65 URECA 0.00 66.48 11.90 UNEP Contracts OMBIG Net Point																	
UNE-PCENTREX - DMS100 (Valid in All States) Image: Monopole (Context) Combo Image: Monopole (Context) Context) Image: Monopole (Context) Context)																	
EWire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo- Non-Design Image: Centre Voice Grade Port (Centrex) Port Combo- Non-Design Image: Centre Voice Grade Port (Centrex) Port Combo- Non-Design Image: Centre Voice Grade Port (Centrex) Port Combo- Non-Design Image: Centre Voice Grade Port (Centrex) Port Combo- Non-Design Image: Centre Voice Grade Port (Centrex) Port Combo- Non-Design Image: Centre Voice Grade Port (Centrex) Port Combo- Non-Design Image: Centre Voice Grade Port (Centrex) Port Combo- Non-Design Image: Centre Voice Grade Port (Centrex) Port Combo- Non-Design Image: Centre Voice Grade Port (Centrex) Port Combo- Design Image: Centre Voice Grade Port (Centrex) Port Combo- Design Image: Centre Voice Grade Port (Centrex) Port Combo- Design Image: Centre Voice Grade Port (Centrex) Port Combo- Design Image: Centre Voice Grade Port (Centrex) Port Combo- Design Image: Centre Voice Grade Port (Centrex) Port Combo- Design Image: Centre Voice Grade Port (Centrex) Port Combo- Design Image: Centre Voice Grade Port (Centrex) Port Combo- Design Image: Centre Voice Grade Port (Centrex) Port Combo- Design Image: Centre Voice Grade Port (Centrex) Port Combo- Design Image: Centre Voice Grade Port (Centrex) Port Combo- Design Image: Centre Voice Grade Port (Centrex) Port Combo- Design Image: Centre Voice Grade Port (Centrex) Port Combo- Design Image: Centre Voice Grade Port (Centrex) Port Combo- Design Image: Centre Voice Grade Port (Centrex) Port Combo- Design Image: Centre Voice Grade Port (Centrex) Port Combo- Design Image: Centre Voice Grade Port (Centrex) Port Combo- De					UEP95	URECA	0.00	66.48					11.90				
UNE Port/Loop Combination Rates (Non-Design) Image: Constraint of the second seco																	
2-Wire Vice Corp/2-Wire Voice Grade Port (Centrex)Port Combo- Non-Design 1 UEP9D 26.94																	
Non-Design 1 UEP9D 26.94 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo- Non-Design 2 UEP9D 31.06			_														
2-Wire Volice Grade Port (Centrex)Port Combo - Non-Design 2 UEP9D 31.06				1	UEP9D		26.94										
2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo- Non-Design 3 UEP9D 45.87 <t< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></t<>																	
Non-Design 3 UEP9D 45.87 <td></td> <td></td> <td></td> <td>2</td> <td>UEP9D</td> <td></td> <td>31.06</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>				2	UEP9D		31.06										
UNE Port/Loop Combination Rates (Design) I																	
2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo- Design 1 UEP9D 29.36 1 1 UEP9D 29.36 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo- Design 2 UEP9D 34.43 1				3	UEP9D		45.87										
Design 1 UEP9D 29.36 Image: Contrast of the state of the						_											
2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo- Design 2 UEP9D 34.43 </td <td></td> <td>3 Loop/2-Wire Voice Grade Port (Centrex) Port Combo</td> <td>-</td> <td></td> <td></td> <td></td> <td>00.00</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>		3 Loop/2-Wire Voice Grade Port (Centrex) Port Combo	-				00.00										
Design2UEP9D34.43<		2 Loop/2 Wire Visice Grade Bart (Contrax) Bart Comba		1	UEP9D		29.36										
2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo- Design 3 UEP9D 50.68 Image: Composition of the state of the st		5 Loop/2-wile voice Glade Folt (Centrex)Folt Combo -		2			34 43										
Design 3 UEP9D 50.68		G Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		~			04.40										
UNE Loop Rate Image: Constraint of the const				3	UEP9D		50.68										
2-Wire Voice Grade Loop (SL 1) - Zone 2 2 UEP9D UECS1 17.06																	
2-Wire Voice Grade Loop (SL 1) - Zone 3 3 UEP9D UECS1 31.87																	
2-Wire Voice Grade Loop (SL 2) - Zone 1 1 UEP9D UECS2 15.36																	
2-Wire Voice Grade Loop (SL 2) - Zone 2 2 UEP9D UECS2 20.43			-	-													
2-Wire Voice Grade Loop (SL 2) - Zone 3 3 UEP9D UECS2 36.68 Image: Constraint of the second																	
UNE Port Rate Image: Constraint of the system Image: Consystem <t< td=""><td></td><td></td><td>-</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td> </td><td></td></t<>			-														
ALL STATES Image: Constraint of the state o		100 Graue Loop (SL 2) - 20118 S	-	3	027.90	06032	30.08										
2-Wire Voice Grade Port (Centrex) Basic Local Area UEP9D UEPYA 14.00 11.00 11.00 2-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area UEP9D UEPYB 14.00 70.00 35.00 35.00 10.00 11.90 11.90			-	1													
2-Wire Voice Grade Port (Centrex 800 termination)Basic Local UEP9D UEPYB 14.00 70.00 35.00 35.00 10.00 11.90		pice Grade Port (Centrex) Basic Local Area	1	1	UEP9D	UEPYA	14.00						11.90			İ	
Area UEP9D UEPYB 14.00 70.00 35.00 35.00 11.00 11.90			1									1					1
2-Wire Voice Grade Port (Centrex / EBS-PSET)3Basic Local	ea				UEP9D	UEPYB	14.00	70.00	35.00	35.00	10.00		11.90				
Area UEP9D UEPYC 14.00 70.00 35.00 10.00 11.90		pice Grade Port (Centrex / EBS-PSET)3Basic Local															

UNBUNDL	ED NETWORK ELEMENTS - Florida				-							-	Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring					Rates(\$)	•	
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3Basic Local					1.00	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Area			UEP9D	UEPYD	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5209))3 Basic Local Area			UEP9D	UEPYE	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5112))3 Basic Local Area			UEP9D	UEPYF	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5312))3Basic Local					44.00	70.00	05.00	05.00	40.00		44.00				
	Area 2-Wire Voice Grade Port (Centrex / EBS-M5008))3 Basic Local			UEP9D	UEPYG	14.00	70.00	35.00	35.00	10.00		11.90				
	Area			UEP9D	UEPYT	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5208))3 Basic Local Area			UEP9D	UEPYU	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5216))3 Basic Local Area			UEP9D	UEPYV	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5316))3 Basic Local Area			UEP9D	UEPY3	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex with Caller ID) Basic Local Area			UEP9D	UEPYH	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp Indication))3 Basic Local Area			UEP9D	UEPYW	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication))3 Basic Local Area			UEP9D	UEPYJ	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center) 2 Basic Local Area			UEP9D	UEPYM	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3 Basic Local Area			UEP9D	UEPYO	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3 Basic Local Area			UEP9D	UEPYP	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3 Basic Local Area			UEP9D	UEPYQ	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3 Basic Local Area			UEP9D	UEPYR	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3 Basic Local Area			UEP9D	UEPYS	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3 Basic Local Area			UEP9D	UEPY4	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3 Basic Local Area			UEP9D	UEPY5	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3 Basic Local Area			UEP9D	UEPY6	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3 Basic Local Area			UEP9D	UEPY7	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP9D	UEPYZ	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent Basic Local Area			UEP9D	UEPY9	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port Terminated on 800 Service Term Basic Local Area			UEP9D	UEPY2	14.00	70.00	35.00	35.00	10.00		11.90				
FL &	GA Only					11.00	70.00	05.00	05.00	10.00		44.00				
	2-Wire Voice Grade Port (Centrex) 2-Wire Voice Grade Port (Centrex 800 termination)			UEP9D UEP9D	UEPHA UEPHB	14.00 14.00	70.00 70.00	35.00 35.00	35.00 35.00	10.00 10.00		11.90 11.90				
	2-Wire Voice Grade Port (Centrex 800 termination) 2-Wire Voice Grade Port (Centrex / EBS-PSET)3			UEP9D	UEPHC	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3			UEP9D	UEPHD	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5209)3			UEP9D	UEPHE	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5112)3	<u> </u>		UEP9D	UEPHF	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5312)3 2-Wire Voice Grade Port (Centrex / EBS-M5008)3			UEP9D UEP9D	UEPHG UEPHT	14.00 14.00	70.00 70.00	35.00 35.00	35.00 35.00	10.00		11.90 11.90				
<u>├</u>	2-Wire Voice Grade Port (Centrex / EBS-M5008)3 2-Wire Voice Grade Port (Centrex / EBS-M5208)3			UEP9D UEP9D	UEPHI	14.00	70.00	35.00	35.00	10.00	+	11.90				
	2-Wire Voice Grade Port (Centrex / EBS-M5208)3 2-Wire Voice Grade Port (Centrex / EBS-M5216)3	l		UEP9D	UEPHU	14.00	70.00	35.00	35.00	10.00	1	11.90			<u> </u>	

NBUNDLE	D NETWORK ELEMENTS - Florida										1		Attachment:			bit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrec	urring	Nonrecurring	Disconnect		•	OSS	Rates(\$)	•	•
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port (Centrex / EBS-M5316)3			UEP9D	UEPH3	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex with Caller ID)			UEP9D	UEPHH	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp															
	Indication)3	-		UEP9D UEP9D	UEPHW UEPHJ	14.00 14.00	70.00 70.00	35.00 35.00	35.00 35.00	10.00		11.90 11.90				
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication)3 2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)			UEP9D	UEPHJ	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wile Voice Grade Port (Centrex from diff Serving Wile Center)			UEP9D	UEPHM	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			UEP9D	UEPHO	14.00	180.00	110.00	85.00	20.00		11.90				
				02.05	02.110	1.00	100.00	110.00	00.00	20.00					1	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3			UEP9D	UEPHP	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3	1		UEP9D	UEPHQ	14.00	180.00	110.00	85.00	20.00		11.90				
	· · · · · · · · · · · · · · · · · · ·															
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			UEP9D	UEPHR	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3			UEP9D	UEPHS	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3			UEP9D	UEPH4	14.00	180.00	110.00	85.00	20.00		11.90				
	O Miles Miles Ore to Dist (October / Fifter OMO /EDO MEDDO)O. O					44.00	100.00	110.00	05.00			44.00				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3			UEP9D	UEPH5	14.00	180.00	110.00	85.00	20.00		11.90			-	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3			UEP9D	UEPH6	14.00	180.00	110.00	85.00	20.00		11.90				
	2-wire voice Grade Port (Centrex/diller SWC /EBS-M5216)2, 3			UEP9D	UEPHo	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			UEP9D	UEPH7	14.00	180.00	110.00	85.00	20.00		11.90				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			021 30	OLITI!	14.00	100.00	110.00	05.00	20.00		11.30				
	Term			UEP9D	UEPHZ	14.00	180.00	110.00	85.00	20.00		11.90				
				02.05	02.112	1 1100	100.00	110.00	00.00	20.00		11100				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9D	UEPH9	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP9D	UEPH2	14.00	70.00	35.00	35.00	10.00		11.90				
Local S	witching															
	Centrex Intercom Funtionality, per port			UEP9D	URECS	0.7384										
Local N	lumber Portability															
	Local Number Portability (1 per port)			UEP9D	LNPCC	0.35										
Feature						0.00										
	All Standard Features Offered, per port	-		UEP9D UEP9D	UEPVF UEPVS	0.00	370.70					11.90				
	All Select Features Offered, per port All Centrex Control Features Offered, per port			UEP9D UEP9D	UEPVS	0.00	370.70					11.90				
NARS	All Centrex Control Teatures Offered, per port			ULF3D	OLFVC	0.00										
	Unbundled Network Access Register - Combination			UEP9D	UARCX	0.00	0.00	0.00				11.90				
	Unbundled Network Access Register - Inward			UEP9D	UAR1X	0.00	0.00	0.00				11.90				
	Unbundled Network Access Register - Outdial			UEP9D	UAROX	0.00	0.00	0.00				11.90				
Miscell	aneous Terminations															
2-Wire	Trunk Side															
	Trunk Side Terminations, each			UEP9D	CEND6	8.81										
4-Wire	Digital (1.544 Megabits)															
	DS1 Circuit Terminations, each			UEP9D	M1HD1	54.95										
later *	DS0 Channels Activiated per Channel	<u> </u>		UEP9D	M1HDO	0.00	15.69				<u> </u>	11.90			ļ	<u> </u>
interof	ice Channel Mileage - 2-Wire	 			MIGBC	0E 00			├ ────			ļ			ł	
	Interoffice Channel Facilities Termination Interoffice Channel mileage, per mile or fraction of mile			UEP9D UEP9D	MIGBC	25.32 0.0091			<u>├</u> ────┤							<u> </u>
Feature	Activations (DS0) Centrex Loops on Channelized DS1 Service	-				0.0091			 			t			 	
	nnel Bank Feature Activations	1														
54 5114	Feature Activation on D-4 Channel Bank Centrex Loop Slot	1		UEP9D	1PQWS	0.66						t			1	
		1				0.00						t	1		1	
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot	1		UEP9D	1PQW6	0.66									1	
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop								I I							
	Slot			UEP9D	1PQW7	0.66										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -															
1	Different Wire Center	1	1	UEP9D	1PQWP	0.66					1	1	1			1

UNDLE	D NETWORK ELEMENTS - Florida	-											Attachment:			bit: B
												Svc Order		Incremental		
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
		Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Sv
GORY	RATE ELEMENTS		Zone	BCS	USOC			RATES(\$)			per LSR		Order vs.	Order vs.	Order vs.	Order vs.
	-	m						- (1)			percon	percon	Electronic-	Electronic-	Electronic-	Electronic
													1st	Add'l	Disc 1st	Disc Add'
					+ +		Nonrec	urring	Nonrecurring	Disconnect			220	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
-					+ +		FIISL	Auu i	FIISL	Auui	SOWIEC	JOWAN	SOWAN	JOWAN	SOWAN	JOWAN
	Facture Activation on D.4 Channel Deals Drivets Line Lean Clat				4001407	0.00										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9D	1PQWV	0.66										
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop															
	Slot			UEP9D	1PQWQ	0.66										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9D	1PQWA	0.66										
Non-R	ecurring Charges (NRC) Associated with UNE-P Centrex															
	NRC Conversion Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP9D	USAC2		21.50	8.42				11.90				
	Conversion of existing Centrex Common Block, each			UEP9D	USACN		5.17	8.32				11.90				
	New Centrex Standard Common Block			UEP9D	M1ACS	0.00	618.82					11.90				
+	New Centrex Standard Common Block		1	UEP9D	MIACS	0.00	618.82		ł – – – – – – – – – – – – – – – – – – –		t	11.90				ł
-					URECA	0.00						11.90				
	NAR Establishment Charge, Per Occasion	L	I	UEP9D	URECA	0.00	66.48				l	11.90				ļ
	CENTREX - EWSD (Valid in AL, FL, KY, LA, MS & TN)	ļ														
	VG Loop/2-Wire Voice Grade Port (Centrex) Combo															
UNE P	ort/Loop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	Non-Design		1	UEP9E		26.94										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Non-Design		2	UEP9E		31.06										
-			~	OEI SE	+ +	01.00										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		~			45.07										
	Non-Design		3	UEP9E	_	45.87										
UNE P	ort/Loop Combination Rates (Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	Design		1	UEP9E		29.36										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design		2	UEP9E		34.43										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -				1 1											
	Design		3	UEP9E		50.68										
	oop Rate		3	ULF9L		50.00										
UNEL					115004	40.04										
_	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9E	UECS1	12.94										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9E	UECS1	17.06										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP9E	UECS1	31.87										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9E	UECS2	15.36										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9E	UECS2	20.43										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9E	UECS2	36.68					1	1				1
UNF P	ort Rate		<u> </u>			22.50					1	1				1
	, KY, LA, MS, & TN only		1		+ +						1					
	2-Wire Voice Grade Port (Centrex) Basic Local Area		1	UEP9E	UEPYA	14.00	70.00	35.00	35.00	10.00	t	11.90				ł
-			 	021 02		14.00	70.00	55.00	35.00	10.00		11.90				
1	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local		1	UEDOE												1
-	Area			UEP9E	UEPYB	14.00	70.00	35.00	35.00	10.00	L	11.90				
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local										1	1				1
	Area			UEP9E	UEPYH	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire															
	Center)2 Basic Local Area			UEP9E	UEPYM	14.00	180.00	110.00	85.00	20.00		11.90				1
1	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service		1		1							1				1
1	Term - Basic Local Area		1	UEP9E	UEPYZ	14.00	180.00	110.00	85.00	20.00		11.90				1
	2-Wire Voice Grade Port terminated in on Megalink or equivalent								00.00	20.00	1					1
1	- Basic Local Area		1	UEP9E	UEPY9	14.00	70.00	35.00	35.00	10.00		11.90				1
+	2-Wire Voice Grade Port Terminated on 800 Service Term -		<u> </u>		JEFIS	14.00	70.00	55.00	55.00	10.00		11.90				<u> </u>
1										10						1
	Basic Local Area			UEP9E	UEPY2	14.00	70.00	35.00	35.00	10.00		11.90				
Florid	a Only										L					
	2-Wire Voice Grade Port (Centrex)			UEP9E	UEPHA	14.00	70.00	35.00	35.00	10.00		11.90				
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP9E	UEPHB	14.00	70.00	35.00	35.00	10.00		11.90				1
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP9E	UEPHH	14.00	70.00	35.00	35.00	10.00		11.90				
1	2-Wire Voice Grade Port (Centrex from diff Serving Wire		1		1 1						İ					1
	Center)2		1	UEP9E	UEPHM	14.00	180.00	110.00	85.00	20.00		11.90				1
					JEFTIN	14.00	100.00	110.00	05.00	20.00	1	11.90		1		L
_	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service				1											

UNBUNDLED N	ETWORK ELEMENTS - Florida												Attachment:	2	Exhi	bit: B
											Svc Order	Svc Order	Incremental	Incremental	Incremental	Incrementa
												Submitted Manually	Charge - Manual Svc	Charge -	Charge - Manual Svc	Charge - Manual Svo
ATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			Elec					
		m	20116	500	0000			ππτεο(ψ)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
					_		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
2-W	/ire Voice Grade Port terminated in on Megalink or equivalent			UEP9E	UEPH9	14.00	70.00	35.00	35.00	10.00		11.90				
	/ire Voice Grade Port Terminated on 800 Service Term			UEP9E	UEPH2	14.00	70.00	35.00	35.00	10.00		11.90				
Local Swite																
Cen	trex Intercom Funtionality, per port			UEP9E	URECS	0.7384										
	ber Portability															
Loca	al Number Portability (1 per port)			UEP9E	LNPCC	0.35										
Features																
	Standard Features Offered, per port			UEP9E	UEPVF	0.00										
	Select Features Offered, per port			UEP9E	UEPVS	0.00	370.70					11.90				
	Centrex Control Features Offered, per port			UEP9E	UEPVC	0.00										
NARS																
	oundled Network Access Register - Combination			UEP9E	UARCX	0.00	0.00	0.00				11.90				1
	oundled Network Access Register - Indial			UEP9E	UAR1X	0.00	0.00	0.00				11.90				
Unb	oundled Network Access Register - Outdial			UEP9E	UAROX	0.00	0.00	0.00				11.90				
Miscellaneo	ous Terminations															
2-Wire Trur	nk Side															
	nk Side Terminations, each			UEP9E	CEND6	8.81										
4-Wire Digi	tal (1.544 Megabits)															
	1 Circuit Terminations, each			UEP9E	M1HD1	54.95										
DSC	Channel Activated Per Channel			UEP9E	M1HDO	0.00	15.69					11.90				
	Channel Mileage - 2-Wire															
Inte	roffice Channel Facilities Termination			UEP9E	MIGBC	25.32										
	roffice Channel mileage, per mile or fraction of mile			UEP9E	MIGBM	0.0091										
Feature Act	tivations (DS0) Centrex Loops on Channelized DS1 Servic	e														
	I Bank Feature Activations															
Fea	ture Activation on D-4 Channel Bank Centrex Loop Slot			UEP9E	1PQWS	0.66										
Fea	ture Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9E	1PQW6	0.66										
	ture Activation on D-4 Channel Bank FX Trunk Side Loop		1						1					1		
Slot	t ·			UEP9E	1PQW7	0.66										
Fea	ture Activation on D-4 Channel Bank Centrex Loop Slot -															
Diffe	erent Wire Center			UEP9E	1PQWP	0.66										
Fea	ture Activation on D-4 Channel Bank Private Line Loop Slot			UEP9E	1PQWV	0.66										
Fea	ture Activation on D-4 Channel Bank Tjie Line/Trunk Loop															
Slot				UEP9E	1PQWQ	0.66										
Fea	ture Activation on D-4 Channel Bank WATS Loop Slot			UEP9E	1PQWA	0.66										
Non-Recurr	ring Charges (NRC) Associated with UNE-P Centrex															
NRO	C Conversion Currently Combined Switch-As-Is with allowed															
	nges, per port			UEP9E	USAC2		21.50	8.42				11.90				
	version of Existing Centrex Common Block, each			UEP9E	USACN		5.17	8.32				11.90				
	v Centrex Standard Common Block			UEP9E	M1ACS	0.00	618.82					11.90				
	v Centrex Customized Common Block			UEP9E	M1ACC	0.00	618.82					11.90				
	R Establishment Charge, Per Occasion			UEP9E	URECA	0.00	66.48					11.90				
	quired Port for Centrex Control in 1AESS, 5ESS & EWSD															1
	equres Interoffice Channel Mileage															
	quires Specific Customer Premises Equipment															
Note: Rate:	s displaying an "R" in Interim column are interim and sub	ject to	rate tru	e-up as set forth in	n General Term	s and Conditio	ons.		r i							

UNBUN	IDLE	D NETWORK ELEMENTS - Georgia												Attachment:	2	Exhi	bit: B
												Svc Order	Svc Order	Incremental	Incremental	Incremental	Incrementa
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svo
CATEGO	DRY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
												• • •	••••	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
																2.00 .01	Diotriau
							Rec	Nonree			g Disconnect				Rates(\$)		
								First	Add'l	First	Add'l		SOMAN		SOMAN	SOMAN	SOMAN
Т	The "Z	one" shown in the sections for stand-alone loops or loops as	part of	a com	bination refers to Ge	eographicall	y Deaveraged U	NE Zones. To	view Georgrap	phically Deave	raged UNE Zor	ne Desiganti	ons by C O	, refer to Inter	net Website:		
		www.interconnection.bellsouth.com/become_a_clec/html/inter	connec	tion.h	tm												
		_ SUPPORT SYSTEMS															
		(1) Electronic Service Order: CLEC should contact its contract															
e	exhibit	is the BellSouth regional electronic service ordering charge. (2) Any element that can be ordered electronically will be bill	CLEC I	may el	ect either the state s	pecific Com	mission ordered	d rates for the	electronic serv	ice ordering c	harges, or CLE	C may elec	t the region	al electronic s	service orderi	ng charge.	
N	NOTE:	(2) Any element that can be ordered electronically will be bill	ed acco	ording	to the SOMEC rate I	isted in this	category. Pleas	se refer to Bell	South's Busine	ess Rules for I	_ocal Ordering	(BBR-LO) to	o determine	if a product of	can be ordere	d electronical	ly. For
th	hose e	elements that cannot be ordered electronically at present per t	he BBR	l-LO, tl	ne listed SOMEC rate	e in this cate	egory reflects th	e charge that v	would be billed	to a CLEC or	nce electronic	ordering cap	abilities co	me on-line fo	r that elemen	t. Otherwise,	the manual
0	orderin	g charge, SOMAN, will be applied to a CLECs bill when it sub	omits ar	LSR	to BellSouth.												
		Electronic OSS Charge, per LSR, submitted via BST's OSS															
		interactive interfaces (Regional)				SOMEC		3.50									
UNE SER	RVICE	DATE ADVANCEMENT CHARGE															
		The Expedite charge will be maintained commensurate with	BellSou	th's F	CC No.1 Tariff. Section	on 5 as appl	icable.										
l l'		UNE Expedite Charge per Circuit or Line Assignable USOC, per			,,,,,					l	İ	1		l	1	l	l I
		Dav	1	1	ALL UNE	SDASP		200.00			1						1
		EXCHANGE ACCESS LOOP		1	<u> </u>		1	200.00			1	1	1		1		<u> </u>
		ANALOG VOICE GRADE LOOP												-		-	
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 1		1	UEANL	UEAL2	14.21	42.54	31.33		1	1		18.94	8.42		<u> </u>
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 2		2	UEANL	UEAL2	16.41	42.54	31.33					18.94	8.42		
\vdash		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 2 2-Wire Analog Voice Grade Loop - Service Level 1- Zone 3		3	UEANL	UEAL2	26.08	42.54	31.33		1	1		18.94	8.42	t	1
		Loop Testing - Basic 1st Half Hour		5	UEANL	URET1	20.00	78.92	78.92			1		18.94	8.42		
		Loop Testing - Basic Additional Half Hour			UEANL	URETA		23.33	23.33			1		18.94	8.42		
		CLEC to CLEC Conversion Charge Without Outside Dispatch			UEANL	UREIA		23.33	23.33			-		10.94	0.42	-	<u> </u>
		(UVL-SL1)			UEANL	UREWO		15.75	8.92								
			-		UEANL	UREWO		15.75	8.92								ł
		Unbundled Voice Loop, Unbundled Non-Design Voice Loop,						00 70	00.70								
		billing for BST providing make-up	-		UEANL	UEANM		28.72	28.72								ł
		Manual Order Coordiantion for UVL-SL1s (per loop)			UEANL	UEAMC		16.11	16.11								4
		Order Coordination for Specified Conversion Time for UVL-SL1															
		(per LSR)			UEANL	OCOSL		35.74	35.74		=			10.01	0.40		L
		2 Wire Unbundled Copper Loop Non-Designed- Zone 1		1	UEQ	UEQ2X		11.02	44.69	25.65				18.94	8.42		
		2 Wire Unbundled Copper Loop Non-Designed- Zone 2		2	UEQ	UEQ2X		12.72	44.69	25.65	7.06			18.94	8.42		1
		2 Wire Unbundled Copper Loop Non-Designed-Zone 3		3	UEQ	UEQ2X		20.22	44.69	25.65	7.06			18.94	8.42		
		Order Coordination 2 Wire Unbundled Copper Loop - Non-															
		Designed (per loop)			UEQ	USBMC		16.11	16.11					18.94	8.42		
		Unbundled Copper Loop, Non-Designed Billing for BST															
		providing make-up			UEQ	UEQMU		28.72	28.72		1			18.94	8.42		ļ
		Loop Testing - Basic 1st Half Hour			UEQ	URET1		78.92	78.92					18.94	8.42		
		Loop Testing - Basic Additional Half Hour			UEQ	URETA		23.33	23.33					18.94	8.42		
		CLEC to CLEC Conversion Charge Without Outside Dispatch															1
		(UCL-ND)			UEQ	UREWO		14.25	7.42		I			18.94	8.42		
UNBUND	DLED E	EXCHANGE ACCESS LOOP															
		ANALOG VOICE GRADE LOOP															
U	JNE Lo	pop Rates for Line Splitting (In Ga. PSC ordered the line split	tting lo	op US	OCs match the lower	r port- loop o	combo rates UE	PLX)									
		2-Wire Voice Grade Loop (SL1) for Line Splitting - Zone 1	Ĩ	1	UEPSR, UEPSB	UEALS,	12.59										
		2-Wire Voice Grade Loop (SL1) for Line Splitting - Zone 1	I	1	UEPSR, UEPSB	UEABS	12.59										
		2-Wire Voice Grade Loop (SL1) for Line Splitting - Zone 2	I	2	UEPSR, UEPSB	UEALS,	14.26					1					
		2-Wire Voice Grade Loop (SL1) for Line Splitting - Zone 2	I	2	UEPSR, UEPSB	UEABS	14.26					1					
		2-Wire Voice Grade Loop (SL1) for Line Splitting - Zone 3	i	3	UEPSR, UEPSB	UEALS	21.62			l	İ	1		l	1	l	
		2-Wire Voice Grade Loop (SL1)for Line Splitting - Zone 3		3	UEPSR, UEPSB	UEABS	21.62			İ	1	1		l I	İ	l I	İ
UNBUND	DLED F	EXCHANGE ACCESS LOOP		Ť			252				1	1		1	1	1	
		ANALOG VOICE GRADE LOOP		1	1	1	1				1	1		1	1	1	
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or		1	1	1	1				1	1	1		1		i
		Ground Start Signaling - Zone 1	1	1	UEA	UEAL2	16.84	104.17	78.10		1			18.94	8.42		1
\vdash		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or				JEALZ	10.04	104.17	70.10		1	1		10.94	0.42	1	<u> </u>
		Ground Start Signaling - Zone 2		2	UEA	UEAL2	19.45	104.17	78.10					18.94	8.42		
\vdash		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or		2		ULALZ	19.40	104.17	70.10			ł		10.94	0.42	ł	ł
		Ground Start Signaling - Zone 3	1	з	UEA	UEAL2	30.92	104.17	78.10		1			18.94	8.42		1
\vdash		Order Coordination for Specified Conversion Time (per LSR)		3	UEA	OCOSL	30.92	104.17 35.74	78.10					18.94	8.42	<u> </u>	<u> </u>
\vdash					UEA	ULUSL		35.74									───
1 1		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse	1				10.01	1011-	70.40		1	1		10.01	0.10		1
		Battery Signaling - Zone 1	1	1	UEA	UEAR2	16.84	104.17	78.10	1	1	1	1	18.94	8.42	1	1

UNBUNDLE	D NETWORK ELEMENTS - Georgia											-	Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse					10.15							10.01			
	Battery Signaling - Zone 2 2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse		2	UEA	UEAR2	19.45	104.17	78.10					18.94	8.42		
	Battery Signaling - Zone 3		3	UEA	UEAR2	30.92	104.17	78.10					18.94	8.42		
	Order Coordination for Specified Conversion Time (per LSR)		5	UEA	OCOSL	30.32	35.74	70.10					10.34	0.42		
	CLEC to CLEC Conversion Charge without outside dispatch			UEA	UREWO		87.72	36.36					18.94	8.42		
4-WIR	E ANALOG VOICE GRADE LOOP			-												
	4-Wire Analog Voice Grade Loop - Zone 1		1	UEA	UEAL4	22.26	206.95	170.57					18.94	8.42		
	4-Wire Analog Voice Grade Loop - Zone 2		2	UEA	UEAL4	25.70	206.95	170.57					18.94	8.42		
	4-Wire Analog Voice Grade Loop - Zone 3		3	UEA	UEAL4	40.86	206.95	170.57					18.94	8.42		
	Order Coordination for Specified Conversion Time (per LSR)			UEA	OCOSL		35.74	00.00					40.04	0.40		
2 14/10	CLEC to CLEC Conversion Charge without outside dispatch E ISDN DIGITAL GRADE LOOP			UEA	UREWO		87.72	36.36			ł	ļ	18.94	8.42		
2-WIR	2-Wire ISDN Digital Grade Loop - Zone 1		1	UDN	U1L2X	21.89	233.38	180.35					18.94	8.42		
	2-Wire ISDN Digital Grade Loop - Zone 2		2	UDN	U1L2X	25.27	233.38	180.35			1		18.94	8.42		
1	2-Wire ISDN Digital Grade Loop - Zone 2		3	UDN	U1L2X	40.17	233.38	180.35			1		18.94	8.42		
	Order Coordination For Specified Conversion Time (per LSR)			UDN	OCOSL		35.74									
	CLEC to CLEC Conversion Charge without outside dispatch			UDN	UREWO		120.98	33.04					18.94	8.42		
2-WIR	E Universal Digital Channel (UDC) COMPATIBLE LOOP															
	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone															
			1	UDC	UDC2X	21.89	44.69	31.55	25.65	7.06			18.94	8.42		
	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone		2	UDC	UDC2X	25.27	44.69	31.55	25.65	7.06			18.94	8.42		
	2 2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone	I	2	UDC	UDC2X	25.27	44.69	31.55	25.65	7.06			18.94	8.42		
			3	UDC	UDC2X	40.17	44.69	31.55	25.65	7.06			18.94	8.42		
	CLEC to CLEC Conversion Charge without outside dispatch		5	UDC	UREWO	40.17	44.69	31.55	25.05	7.00			18.94	8.42		
2-WIR	E ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMP	ATIBLE	LOOF		UNLENTO		11100	01100					10.01	0.12		
	2 Wire Unbundled ADSL Loop including manual service inquiry															
	& facility reservation - Zone 1	1	1	UAL	UAL2X	11.23	44.69	31.55	25.65	7.06			18.94	8.42		
	2 Wire Unbundled ADSL Loop including manual service inquiry															
	& facility reservation - Zone 2		2	UAL	UAL2X	12.97	44.69	31.55	25.65	7.06			18.94	8.42		
	2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation - Zone 3		~	UAL	UAL2X	00.00	44.69	04.55	05.05	7.00			10.01	0.40		
	& facility reservation - Zone 3 Order Coordination for Specified Conversion Time (per LSR)	1	3	UAL	OCOSL	20.62	44.69 35.74	31.55	25.65	7.06			18.94	8.42		
	2 Wire Unbundled ADSL Loop without manual service inquiry &			UAL	00031		33.74									
	facility reservaton - Zone 1	1	1	UAL	UAL2W	11.23	44.69	31.55	25.65	7.06			18.94	8.42		
	2 Wire Unbundled ADSL Loop without manual service inquiry &			0,12	0/12211	11120	11100	01100	20.00	1100			10.01	0.12		
	facility reservaton - Zone 2	1	2	UAL	UAL2W	12.97	44.69	31.55	25.65	7.06			18.94	8.42		
	2 Wire Unbundled ADSL Loop without manual service inquiry &		ľ													
	facility reservaton - Zone 3	I	3	UAL	UAL2W	20.62	44.69	31.55	25.65	7.06			18.94	8.42		
	Order Coordination for Specified Conversion Time (per LSR)			UAL	OCOSL		35.74	00.02			L		10.01	0.40		
2 14/10	CLEC to CLEC Conversion Charge without outside dispatch E HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPA			UAL	UREWO		44.69	29.29					18.94	8.42		
2-11/R	2 Wire Unbundled HDSL Loop including manual service inquiry	IDLE			+											
	& facility reservation - Zone 1	1	1	UHL	UHL2X	7.88	44.69	31.55	25.65	7.06			18.94	8.42		
1	2 Wire Unbundled HDSL Loop including manual service inquiry			0.12		,	44.55	01.00	20.00	1.00			10.04	0.42		
	& facility reservation - Zone 2	1	2	UHL	UHL2X	9.09	44.69	31.55	25.65	7.06			18.94	8.42		
	2 Wire Unbundled HDSL Loop including manual service inquiry															
	& facility reservation - Zone 3	1	3	UHL	UHL2X	14.46	44.69	31.55	25.65	7.06			18.94	8.42		
	Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		35.74									
	2 Wire Unbundled HDSL Loop without manual service inquiry							o						a		
	and facility reservation - Zone 1 2 Wire Unbundled HDSL Loop without manual service inquiry		1	UHL	UHL2W	7.88	44.69	31.55	25.65	7.06			18.94	8.42		
	2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 2		2	UHL	UHL2W	9.09	44.69	31.55	25.65	7.06			18.94	8.42		
	2 Wire Unbundled HDSL Loop without manual service inquiry		2	UTIL		5.09	44.09	31.00	20.00	7.00			10.94	0.42		
	and facility reservation - Zone 3	1	3	UHL	UHL2W	14.46	44.69	31.55	25.65	7.06			18.94	8.42		
<u> </u>	Order Coordination for Specified Conversion Time (per LSR)		Ť	UHL	OCOSL		35.74	000	20.00		1		10.04	0.12		
	CLEC to CLEC Conversion Charge without outside dispatch	I		UHL	UREWO		44.69	31.55					18.94	8.42		
4 14/10	E HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPA	TIBLE	LOOP													

CATEGORY RATE BLEMENTS Infall Zer BCD Discrete State Discrete Sta	JNBUNDLED	NETWORK ELEMENTS - Georgia												Attachment:	2	Exhi	bit: B
Image: Control Line Control (Contro) (Control (Control (Control (Contro) (Control (Contr				Zone	BCS	USOC			RATES(\$)			Submitted Elec	Submitted Manually	Charge - Manual Svc Order vs. Electronic-	Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
- - - - - - - - PMI Addr DMRC DMMA DM							Rec										
end naity reservator in I In Unit							Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Attive Unbundle HDSI. Loop incluing manual service inquiry I UHL UHLAX 12:00 44.69 31:05 25:05 7:05 16:34 6.42 Charles interaction - Zong and manual service inquiry 1 UHL UHLAX 19:07 44.69 31:05 25:05 7:05 18:34 6.42 Charles Constratute Signated Conversion Time get LSB UHL UHLAX 19:07 44.69 31:05 25:05 7:06 18:34 8.42 Attive Unbundler HDSL Loop without nanual service inquiry 1 UHL UHLAX 19:07 44.69 31:05 25:05 7:06 18:34 8.42 Attive Unbundler HDSL Loop without nanual service inquiry 1 2 UHL UHLAX 19:07 44.60 31:05 25:06 7:06 18:34 8.42 Attive Unbundler HDSL Loop without nanual service inquiry 1 3 UHL UHLAX 10:07 44.60 31:05 25:06 7:06 16:38 8:42 Attive Unbundler HDSL Loop without nanual service inquiry 1 20:04 <td< td=""><td></td><td></td><td></td><td>1</td><td></td><td></td><td>10.20</td><td>44.60</td><td>21 55</td><td>25.65</td><td>7.06</td><td></td><td></td><td>19.04</td><td>9.42</td><td></td><td></td></td<>				1			10.20	44.60	21 55	25.65	7.06			19.04	9.42		
and bally networks. Ture 2 1 2 UR. URLAX 12.00 44.490 31.55 20.66 7.00 18.84 6.02 4.Wire Under HSEL consistion for generation for generation of spectram Constraint and service requires and the spectram Constram Constraint and service requires and the spectram Constraint and service requires and the spectram Constraint and service requires and the spectram Constraint and the spectram Constraint and service requires and the spectram Constraint and service requires and the spectram Constraint and the spectram Constraint and service requires and the spectram Constraint and service requires and the spectram Constraint and service requires and the spectram Constram Conspectram Constraint and service requires and the sp				1	UHL	UHL4X	10.39	44.69	31.00	25.65	7.06			18.94	8.42		-
All Mile labeled MSL. Logs including named arrow inputs I			1	2	UHL	UHL4X	12.00	44.69	31.55	25.65	7.06			18.94	8.42		
Odde Continuitor for Specified Convention Time (per LSR) UHL OCOSIL 38.74 UN UN UN UN 4 Web build with Structures Intrust derive inplay 1 1 UHL UHLAW 10.39 44.48 31.55 25.68 7.06 18.94 8.42 end locitly interview Intrust derive inplay 1 1 UHLAW 10.29 44.40 31.55 25.65 7.06 18.94 8.42 end locitly interview Inplay 1 0 UHLAW 11.09 44.40 31.55 25.65 7.06 18.94 8.42 end locitly interview Inplay Structures Inp					-	-											
4-Wire Unbunded HSBL Loop without manual service mapping 1 0			I	3			19.07		31.55	25.65	7.06			18.94	8.42		
and nainy security. Zure 1 1 </td <td></td> <td></td> <td></td> <td></td> <td>UHL</td> <td>OCOSL</td> <td></td> <td>35.74</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>					UHL	OCOSL		35.74									
4/We Unbundled HDSL toge what manual service range? 0 <				4			10.20	44.60	21 55	25.65	7.06			19.04	9.42		
Bed failty reservation Jone 2 UHLW ULW U200 44.69 31.65 Z6.65 7.06 18.4 8.42 4 Week bubwinded FDL (so without namula service input) 1 0 UHL 00.000 44.69 31.65 26.65 7.06 18.4 8.42 4 UPL 0.010 0.0208 45.71 44.69 31.65 26.65 7.06 18.4 8.42 4 MERE DST Digital (up - 2m - 1) 1 0.11 0.01 0.010 26.95 <td></td> <td></td> <td></td> <td></td> <td>UHL</td> <td>UHL4VV</td> <td>10.39</td> <td>44.09</td> <td>31.55</td> <td>25.65</td> <td>7.06</td> <td></td> <td></td> <td>10.94</td> <td>0.42</td> <td></td> <td></td>					UHL	UHL4VV	10.39	44.09	31.55	25.65	7.06			10.94	0.42		
EVent bilander HDSL Log without manual service inger LSR) I			1	2	UHL	UHL4W	12.00	44.69	31.55	25.65	7.06			18.94	8.42		
Order Coordination for Specified Conversion Time (per LSR) IDE UPL OCSU 35.74 IDE IDE IDE 4 WIRE DST DIGITAL LOOP IDE	4	4-Wire Unbundled HDSL Loop without manual service inquiry										1					
CLE C GLEC Conversion Charge without outside steppide 1 Diff. UREWO 44.49 3155 C 1 6.42 4 WINE SD IDIGIAL LOP 1 1 USL USL USL 0.435.8 286.1 - - 16.84 8.42 4 WINE SD IDIGIAL LOP 2 USL USL USL 0.153 286.1 - - 16.84 8.42 - Drade Contrastant of Standing Convertion Time (ptr. LSR) 2 USL USL USL 0.153 287.4 287.1 - 16.84 8.42 4 WINE 193, 50 CH 4 K495 DIGTAL CADDE LOOP - - - - - 16.84 8.42 4 Wine Unbundle Digtal Loop 5 K0 FGTAL CADDE LOOP - - - - 16.84 8.42 4 Wine Unbundle Digtal Loop 5 K0 FGTAL CADDE LOOP - - - 18.44 8.42 4 Wine Unbundle Digtal Loop 5 K0 FGTAL CADDE LOOP - - - 18.44 8.42 4 Wine Unbundle Digtal Loop 5 K0 FGTAL CADDE LOOP - -			1	3			19.07		31.55	25.65	7.06			18.94	8.42		
4-WHE 0S1 Digital Loop - Zone 1 1 USL US														10.01			
Image: Heat Non-Zone 1 Image: Heat Non-Zone 2 Image: Heat Non-Zone 2 Image: Heat Non-Zone 3 <thimage: 3<="" heat="" non-zone="" th=""> <thimage: 3<="" heat="" non-zone="" t<="" td=""><td></td><td></td><td>1</td><td></td><td>UHL</td><td>UREWO</td><td></td><td>44.69</td><td>31.55</td><td></td><td></td><td></td><td></td><td>18.94</td><td>8.42</td><td></td><td></td></thimage:></thimage:>			1		UHL	UREWO		44.69	31.55					18.94	8.42		
44We DS1 Digital Logo - Zone 2 2 USL				1	USI	USI XX	55 53	429.98	268 18					18 94	8 42		
44wre DS1 bigital Logo - Zone 3 3 USL USL OCX 42838 288.1 18.4 8.42 CLEC to Cle Conversion Time (per LSR) USL OCX 35.74 19.94 8.42 CLEC to Cle Conversion Charge without outside dispatch USL UREWO 100.91 42.97 19.94 8.42 4 Wire Unbundled Digital 192 Ktyps 1 UDL UDL 0.01.19 25.75 348.55 241.20 18.94 8.42 4 Wire Unbundled Digital 192 Ktyps 2 10DL UDL 0.01.56 25.71 348.55 241.20 18.94 8.42 4 Wire Unbundled Digital Logo Styps - Zone 1 1 10DL UDLS 25.75 348.55 241.20 18.94 8.42 4 Wire Unbundled Digital Logo Styps - Zone 1 1 UDL UDLS 25.75 348.55 241.20 18.94 8.42 4 Wire Unbundled Digital Logo Styps - Zone 1 1 UDL UDLS 25.75 <																	
CLEC to CLEC Conversion Charge without outside dispat V USL UREW 100.91 42.97 C E 16.84 8.42 4 Wire Unbundled Digital 192 Xhps 1 UDL UDL1 25.75 344.65 241.00 18.94 8.42 4 Wire Unbundled Digital 192 Xhps 2 UDL UDL1 25.75 344.65 241.00 18.94 8.42 4 Wire Unbundled Digital 192 Xhps 3 UDL UDL5 25.75 344.55 241.20 18.94 8.42 4 Wire Unbundled Digital Loop 5 (Kpps - Zone 1 1 UDL UDL5 25.75 344.55 241.20 18.94 8.42 4 Wire Unbundled Digital Loop 5 (Kpps - Zone 1 1 UDL UDL4 25.75 344.55 241.20 18.94 8.42 4 Wire Unbundled Digital Loop 6 (Kpps - Zone 1 1 UDL UDL4 25.75 344.55 241.00 18.94 8.42 4 Wire Unbundled Digital Loop 6 (Kpps - Zone 2 1 UDL UDL4 27.74 344.55 241.00 18.94				3	USL		101.93	429.98	268.18					18.94	8.42		
4-WIRE 192, 56 OR 64 KBPS NOTAL GRADE LOOP Image: constraint of the second																	
4 4 Wire Unbundled Digital 192 Kbps 1 UDL UDL UDL 2374 34455 24120 18.94 8.42 4 4 Wire Unbundled Digital 192 Kbps 3 UDL UDL 2374 34455 24120 18.94 8.42 4 Wire Unbundled Digital Lops 56 Kbps - Zone 2 2 UDL UDLS6 25.75 344555 24120 18.94 8.42 4 Wire Unbundled Digital Lops 56 Kbps - Zone 2 2 UDL UDLS6 27.47 344555 24120 18.94 8.42 4 Wire Unbundled Digital Lops 56 Kbps - Zone 1 1 UDL OCGSL 35.74 - 18.94 8.42 4 Wire Unbundled Digital Lops 64 Kbps - Zone 2 2 UDL UDL 62.57 344.55 241.20 18.94 8.42 4 Wire Unbundled Digital Lops 64 Kbps - Zone 2 2 UDL UDL 02.62 35.74 - 18.94 8.42 0 Order Coordination for Specified Conversion Time (per LSR) UDL UDL 02.62 35.74 - 18.94 8.42					USL	UREWO		100.91	42.97					18.94	8.42		
4 4 Wre Unbundled Digital 122 Rops 2 UDL UDL19 28.74 348.55 241.20 (a) (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< (b)< <th< td=""><td></td><td></td><td></td><td>1</td><td></td><td></td><td>25.75</td><td>249 EE</td><td>241.20</td><td></td><td></td><td></td><td></td><td>19.04</td><td>9.42</td><td></td><td></td></th<>				1			25.75	249 EE	241.20					19.04	9.42		
4 Wire Unbundled Digital Logs 65 Kbps - Zone 1 1 UDL 19 47.27 348.55 241.20 18.94 6.42 4 Wire Unbundled Digital Logs 65 Kbps - Zone 2 2 UDL UDL66 25.75 348.55 241.20 18.94 6.42 4 Wire Unbundled Digital Logs 65 Kbps - Zone 2 2 UDL UDL66 25.74 348.55 241.20 18.94 6.42 4 Wire Unbundled Digital Logs 64 Kbps - Zone 3 3 UDL UDL6 25.74 348.55 241.20 18.94 6.42 4 Wire Unbundled Digital Logs 64 Kbps - Zone 2 1 UDL UDL4 25.74 348.55 241.20 18.84 8.42 4 Wire Unbundled Digital Logs 64 Kbps - Zone 2 2 UDL UDL64 25.74 348.55 241.20 18.84 8.42 Order Coordination for Specified Conversion Time (per LSR) UDL UDL 0CDSL 35.74 10.84 8.42 18.84 8.42 Wire Unbundled CopPER LogP UDP UDL UCL UCLPB 12.02 44.69 31.55 25.65 7.06 18															-		
4 Wre Unbundled Digital Loop 56 Ktps - Zone 1 1 UDL UDL56 25.75 348.55 241.20 18.94 8.42 4 Wre Unbundled Digital Loop 56 Ktps - Zone 3 3 UDL UDL56 47.77 348.55 241.20 18.94 8.42 6 Wre Unbundled Digital Loop 56 Ktps - Zone 3 3 UDL UDC56 47.77 348.55 241.20 18.94 8.42 6 Wre Unbundled Digital Loop 64 Ktps - Zone 1 1 UDL UDL64 25.75 348.55 241.20 18.94 8.42 4 Wre Unbundled Digital Loop 64 Ktps - Zone 2 2 UDL UDL64 25.75 348.55 241.20 18.94 8.42 4 Wre Unbundled Digital Loop 64 Ktps - Zone 3 3 UDL UDL64 47.77 348.55 241.20 18.94 8.42 0 Order Coordination for Specified Conversion Time (per LSR) UDL UDCL UDCL 35.74 -															-		
4 Wrie Unbundled Digital Loop 56 Kbps - Zone 3 3 UDL UDL6 0C74 348.55 241.20 18.94 8.42 4 Wrie Unbundled Digital Loop 64 Kbps - Zone 1 1 UDL UDL4 25.75 348.55 241.20 18.94 8.42 4 Wrie Unbundled Digital Loop 64 Kbps - Zone 2 2 UDL UDL4 25.75 348.55 241.20 18.94 8.42 4 Wrie Unbundled Digital Loop 64 Kbps - Zone 3 3 UDL UDL4 47.27 348.55 241.20 18.94 8.42 0 order Coordination for Specified Conversion Time (per LSR) UDL 0C05L 35.74 -					-												
Order Coordination for Specified Conversion Time (per LSR) UDL OCOSL 35.74 Image: Conversion Time (per LSR) Image: Conversion Time (per LS																	
4 Wire Unbundled Digital Loop 64 Kbps - Zone 1 1 UDL UDL64 2575 348.55 241.20 18.94 8.42 4 Wire Unbundled Digital Loop 64 Kbps - Zone 3 3 UDL UDL64 2974 348.65 241.20 18.94 8.42 0 Order Coordination for Specified Conversion Transfer (sper LSR) 10DL OCGSL 35.74 18.94 8.42 2WIRe Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 1 10DL UCL UREWO 101.95 49.66 18.94 8.42 2WIRe Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 1 1 UCL UCLPB 12.02 44.69 31.55 25.65 7.06 18.94 8.42 2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 2 1 2 UCL UCLPB 12.02 44.69 31.55 25.65 7.06 18.94 8.42 2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 3 1 3 UCL UCLPB 2.07 44.69 31.55				3			47.27		241.20					18.94	8.42		
4 Wire Unbundled Digital Loop 64 Kbps - Zone 2 2 UDL UDL64 29.74 348.85 241.20 1 18.94 8.42 Order Coordination for Specified Conversion Time (per LSR) UDL UDL64 47.27 348.85 241.20 1 18.94 8.42 CLEC to CLEC Conversion Charge without outside dispatch UDL UDC 107.95 49.86 1 18.94 8.42 2WIRE Unbundled COPPER LOOP UDL UDL UREWO 101.95 49.86 1 18.94 8.42 2WIRE Unbundled COPPER LOOP UDL UCL UCLPB 12.02 44.69 31.55 25.65 7.06 18.94 8.42 2WIRE Unbundled COPPER LOOP 1 1 UCL UCLPB 12.02 44.69 31.55 25.65 7.06 18.94 8.42 2WIRe Unbundled Copper Loop/Short including manual service inclusing manual service inclusing manual service inclusing manual service inclusing manual service inclusing manual service inclusing manual service inclusing manual service inclusing manual service inclusing manual service inclusing manual service inclusing manual service inclusing manual service inclusing manual service inclusing manual service inclusing manual service inclusing manual service inclusing manual service inclusing manual							05.75		044.00					40.04	0.40		
4 Wire Unbundled Digital Loop 64 Kbps - Zone 3 3 UDL UDL64 47.27 348.55 241.20 18.94 8.42 CLEC to CLEC Conversion Charge without outside dispate h UDL OCGSL 35.74 1 18.94 8.42 2WIRE Unbundled CopPErs Loop CLEC to CLEC Conversion Charge without outside dispate h UDL UDL UREWO 101.95 49.66 18.94 8.42 2WIRE Unbundled CopPE Loop/Short including manual service inquiry & facility reservation - Zone 1 1 UCL ULCL ULCL ULCL 44.69 31.55 25.65 7.06 18.94 8.42 2Wire Unbundled CopPE Loop/Short including manual service inquiry & facility reservation - Zone 2 I 2 UCL UCLPB 13.88 44.69 31.55 25.65 7.06 18.94 8.42 2 Wire Unbundled CopPE Loop/Short including manual service inquiry & facility reservation - Zone 3 I 3 UCL UCLPB 22.07 44.69 31.55 25.65 7.06 18.94 8.42 2 Wire Unbundled CopPE Loop/Short including manual service inquiry & facility reservation - Zone 3 I 3 UCL UCLPB 18.94 8.42 <td< td=""><td></td><td></td><td></td><td>2</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></td<>				2													
Order Coordination for Specified Conversion Time (per LSR) UDL OCOSL 35.74 Image: Conversion Charge without outside dispate h Image: Conversion Charge without outside dis																	
2-WIRE Unbundled COPPER LOOP Image: Control (Copy of Copy (Copy (Copy of Copy)) Image: Copy (Copy (Copy)) Image: Copy (Copy) Image: Copy) Image: Copy (Copy) <thi< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></thi<>																	
2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 2 I 1 UCL UCLPB 12.02 44.69 31.55 25.65 7.06 18.94 8.42 2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 2 I 2 UCL UCLPB 13.88 44.69 31.55 25.65 7.06 18.94 8.42 2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 3 I 3 UCL UCLPB 22.07 44.69 31.55 25.65 7.06 18.94 8.42 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 3 I 3 UCL UCLPW 12.02 44.69 31.55 25.65 7.06 18.94 8.42 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 1 I 1 UCLPW 12.02 44.69 31.55 25.65 7.06 18.94 8.42 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 3 I 2 UCL UCLPW 12.02 44.69 31.55 25.65					UDL	UREWO		101.95	49.66					18.94	8.42		
inquiry & facility reservation - Zone 1 I I UCL UCLPB 12.02 44.69 31.55 25.65 7.06 18.94 8.42 2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 2 I 2 UCL UCLPB 13.88 44.69 31.55 25.65 7.06 18.94 8.42 2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 3 I 2 UCL UCLPB 22.07 44.69 31.55 25.65 7.06 18.94 8.42 0rder Coordination for Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 1 I UCL UCLW 12.02 44.69 31.55 25.65 7.06 18.94 8.42 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 1 I UCL UCLW 13.88 44.69 31.55 25.65 7.06 18.94 8.42 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 1 I UCL UCLW 13.88 44.69 31.55 25.65 7.06 18.94 8.42																	
2.Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 2 I 2 UCL UCL/PB 13.88 44.69 31.55 25.65 7.06 18.94 8.42 2.Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 3 I 3 UCL UCLPB 22.07 44.69 31.55 25.65 7.06 18.94 8.42 Order Coordination for Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 1 I 1 UCL UCLPW 12.02 44.69 31.55 25.65 7.06 18.94 8.42 2.Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 1 I 1 UCL UCLPW 12.02 44.69 31.55 25.65 7.06 18.94 8.42 2.Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 2 I 2 UCL UCLPW 13.88 44.69 31.55 25.65 7.06 18.94 8.42 2.Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 3 I <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>10.00</td> <td>44.00</td> <td>24.55</td> <td>05.05</td> <td>7.00</td> <td></td> <td></td> <td>40.04</td> <td>0.40</td> <td></td> <td></td>							10.00	44.00	24.55	05.05	7.00			40.04	0.40		
inquiry & facility reservation - Zone 2 I 2 UCL UCLPB 13.88 44.69 31.55 25.65 7.06 18.94 8.42 2 Wire Unbundled Copper Loop/Short without manual service inquiry & facility reservation - Zone 3 I 3 UCL UCLPB 22.07 44.69 31.55 25.65 7.06 18.94 8.42 Order Coordination for Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 1 I 1 UCL UCLWC 16.11 16.11 16.11 16.11 18.94 8.42 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 1 I 1 UCL UCLPW 12.02 44.69 31.55 25.65 7.06 18.94 8.42 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 2 I 2 UCL UCLPW 13.88 44.69 31.55 25.65 7.06 18.94 8.42 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 3 I 3 UCL UCLPW 22.07 44.69 31.55 25.65 7.06 18.94 8.				1	UCL	UCLPB	12.02	44.69	31.00	20.00	7.06			18.94	8.42		
2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 3 1 3 UCL ULPB 22.07 44.69 31.55 25.65 7.06 18.94 8.42 Order Coordination for Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 1 1 1 UCL UCLWC 16.11 18.94 8.42 16.11 16.11 16.11 18.94 8.42 16.11 16.11 16.11 16.11 16.11 16.11 16.11 16.11 16.11 16.11 16.11 16.11 16.11 16.11 16.11 16.11 <td></td> <td></td> <td>1</td> <td>2</td> <td>UCL</td> <td>UCLPB</td> <td>13.88</td> <td>44.69</td> <td>31.55</td> <td>25.65</td> <td>7.06</td> <td></td> <td></td> <td>18.94</td> <td>8.42</td> <td></td> <td></td>			1	2	UCL	UCLPB	13.88	44.69	31.55	25.65	7.06			18.94	8.42		
Order Coordination for Unbundled Copper Loops (per loop) UCL UCLMC 16.11 18.94 8.42 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 3 I 3 UCL UCLPW 22.07 44.69 31.55 25.65 7.06 18.94 8.42 2-Wire Unbundled Copper Loop/Long includes manual structure I 3 UCL UCLWC 16.11 16.11				1		-											
2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 1 I 1 UCL UCLPW 12.02 44.69 31.55 25.65 7.06 18.94 8.42 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 2 I 2 UCL UCLPW 13.88 44.69 31.55 25.65 7.06 18.94 8.42 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 3 I 3 UCL UCLPW 20.07 44.69 31.55 25.65 7.06 18.94 8.42 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 3 I 3 UCL UCLPW 20.07 44.69 31.55 25.65 7.06 18.94 8.42 2-Wire Unbundled Copper Loop/Long - includes manual srvc. inquiry and facility reservation - Zone 1 I UCL UCLUL UCLL 16.11 16.11 I I 9 8.42 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 2 I 2 UCL			1	3			22.07			25.65	7.06			18.94	8.42		
inquiry and facility reservation - Zone 1 I I UCLPW 12.02 44.69 31.55 25.65 7.06 18.94 8.42 2 Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 2 I 2 UCL UCLPW 13.88 44.69 31.55 25.65 7.06 18.94 8.42 2 Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 3 I 3 UCL UCLPW 2.07 44.69 31.55 25.65 7.06 18.94 8.42 2 Order Coordination for Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 3 I 3 UCL UCLPW 22.07 44.69 31.55 25.65 7.06 18.94 8.42 2 Order Coordination for Unbundled Copper Loop/Long - Includes manual service inquiry and facility reservation - Zone 1 I UCL UCL2L 35.56 44.69 31.55 25.65 7.06 18.94 8.42 2 Wire Unbundled Copper Loop/Long - Includes manual service inquiry and facility reservation - Zone 2 I 1 UCL2L 35.56 44.69 31.55 <				<u> </u>	UCL	UCLMC		16.11	16.11			ļ	ļ				
2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 2 I 2 UCL UCLPW 13.88 44.69 31.55 25.65 7.06 18.94 8.42 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 3 I 3 UCL UCLPW 22.07 44.69 31.55 25.65 7.06 18.94 8.42 Order Coordination for Unbundled Copper Loop/Long includes manual struc. inquiry and facility reservation - Zone 1 I 3 UCL UCLWC 16.11 <				1			12.02	44.60	21 55	25.65	7.06			19.04	0 40		
inquiry and facility reservation - Zone 2 I 2 UCL UCLPW 13.88 44.69 31.55 25.65 7.06 18.94 8.42 2-Wire Unbundled Copper Loop/Short without manual service - <t< td=""><td></td><td></td><td></td><td></td><td>UCL</td><td>UCLEVV</td><td>12.02</td><td>44.09</td><td>31.55</td><td>20.00</td><td>7.06</td><td><u> </u></td><td></td><td>10.94</td><td>0.42</td><td></td><td></td></t<>					UCL	UCLEVV	12.02	44.09	31.55	20.00	7.06	<u> </u>		10.94	0.42		
2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 3 I 3 UCL UCLPW 22.07 44.69 31.55 25.65 7.06 18.94 8.42 Order Coordination for Unbundled Copper Loop/Long - includes manual srv. inquiry and facility reservation - Zone 1 I 1 UCL UCLWC 16.11 <td></td> <td></td> <td></td> <td>2</td> <td>UCL</td> <td>UCLPW</td> <td>13.88</td> <td>44,69</td> <td>31,55</td> <td>25,65</td> <td>7.06</td> <td></td> <td> </td> <td>18,94</td> <td>8,42</td> <td></td> <td></td>				2	UCL	UCLPW	13.88	44,69	31,55	25,65	7.06			18,94	8,42		
Order Coordination for Unbundled Copper Loops (per loop) VUCL UCLMC 16.11				<u> </u>													
2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 1 I I UCL UCL2L 35.56 44.69 31.55 25.65 7.06 18.94 8.42 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 2 I 2 UCL UCL2L 41.07 44.69 31.55 25.65 7.06 18.94 8.42 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 2 I 2 UCL UCL2L 41.07 44.69 31.55 25.65 7.06 18.94 8.42 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 3 I 3 UCL UCL2L 65.28 44.69 31.55 25.65 7.06 18.94 8.42			1	3			22.07			25.65	7.06			18.94	8.42		
inquiry and facility reservation - Zone 1 I I UCL2L 35.56 44.69 31.55 25.65 7.06 18.94 8.42 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 2 I 2 UCL UCL2L 41.07 44.69 31.55 25.65 7.06 18.94 8.42 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 3 I 2 UCL UCL2L 41.07 44.69 31.55 25.65 7.06 18.94 8.42					UCL	UCLMC		16.11	16.11								
2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 2 I 2 UCL UCL2L 41.07 44.69 31.55 25.65 7.06 18.94 8.42 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 3 I 3 UCL UCL2L 65.28 44.69 31.55 25.65 7.06 18.94 8.42							05 50	44.00	04.55	05.05	7.00			40.04	0.40		
inquiry and facility reservation - Zone 2 I 2 UCL UCL2L 41.07 44.69 31.55 25.65 7.06 18.94 8.42 2-Wire Unbundled Copper Loop/Loog - includes manual svc. inquiry and facility reservation - Zone 3 I 3 UCL UCL2L 65.28 44.69 31.55 25.65 7.06 18.94 8.42				1	UUL	UUL2L	35.56	44.69	31.55	25.65	7.06	<u> </u>		18.94	8.42		
2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 3 I 3 UCL UCL2L 65.28 44.69 31.55 25.65 7.06 18.94 8.42			1	2	UCL	UCL2L	41.07	44.69	31.55	25.65	7.06			18.94	8.42		
inquiry and facility reservation - Zone 3 I 3 UCL UCL2L 65.28 44.69 31.55 25.65 7.06 18.94 8.42				1													
Order Coordination for Unbundled Copper Loops (per loop) UCL UCLMC 16.11 16.11	ir	nquiry and facility reservation - Zone 3		3			65.28			25.65	7.06			18.94	8.42		
		Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		16.11	16.11								
2-Wire Unbundled Copper Loop/Long - without manual service I UCL2W 35.56 44.69 31.55 25.65 7.06 18.94 8.42							05 50	44.00	04.55	05.05	7.00			40.04	0.40		

UNBUNDL	ED NETWORK ELEMENTS - Georgia												Attachment:	2	Exhi	bit: B
											Svc Order	Svc Order	Incremental			Incremental
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORI		m	20110	500	0000			KATEO(ψ)			per LSR	per LSR	Order vs. Electronic-	Order vs.	Order vs.	Order vs.
													Electronic- 1st	Electronic- Add'l	Electronic- Disc 1st	Electronic- Disc Add'l
															Disc ist	DISC AUU I
						Rec	Nonrec		Nonrecurring			_		Rates(\$)	_	
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility reservation - Zone 2		2	UCL	UCL2W	41.07	44.69	31.55	25.65	7.06			18.94	8.42		1 1
	2-Wire Unbundled Copper Loop/Long - without manual service		2	UCL	UCL2VV	41.07	44.05	31.33	23.03	7.00			10.94	0.42		┟────┦
	inquiry and facility reservation - Zone 3	1	3	UCL	UCL2W	65.28	44.69	31.55	25.65	7.06			18.94	8.42		1
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		16.11	16.11								
	CLEC to CLEC Conversion Charge without outside dispatch															l
	(UCL-Des)	I		UCL	UREWO		44.69	31.55					18.94	8.42		
4-WI	RE COPPER LOOP															ł
	4-Wire Copper Loop/Short - including manual service inquiry and facility reservation - Zone 1		1	UCL	UCL4S	12.02	44.69	31.55	25.65	7.06			18.94	8.42		1
	4-Wire Copper Loop/Short - including manual service inquiry		-		00140	12.02	44.09	31.00	20.00	1.00			10.94	0.42		1
	and facility reservation - Zone 2	1	2	UCL	UCL4S	13.88	44.69	31.55	25.65	7.06			18.94	8.42		
	4-Wire Copper Loop/Short - including manual service inquiry										1					
	and facility reservation - Zone 3		3	UCL	UCL4S	22.07	44.69	31.55	25.65	7.06			18.94	8.42		1
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		16.11	16.11								ļ
	4-Wire Copper Loop/Short - without manual service inquiry and					10.00	11.00	04 55	05.05	7.00			10.01	0.40		
	facility reservation - Zone 1		1	UCL	UCL4W	12.02	44.69	31.55	25.65	7.06	<u> </u>		18.94	8.42		l
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation - Zone 2		2	UCL	UCL4W	13.88	44.69	31.55	25.65	7.06			18.94	8.42		
	4-Wire Copper Loop/Short - without manual service inquiry and		-	UUL	OOLHII	10.00	44.00	01.00	20.00	1.00			10.04	0.42		
	facility reservation - Zone 3	1	3	UCL	UCL4W	22.07	44.69	31.55	25.65	7.06			18.94	8.42		1
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		16.11	16.11								ſ
	4-Wire Unbundled Copper Loop/Long - includes manual svc.															ĺ
	inquiry and facility reservation - Zone 1	I	1	UCL	UCL4L	35.56	44.69	31.55	25.65	7.06			18.94	8.42		l
	4-Wire Unbundled Copper Loop/Long - includes manual svc.		~			44.07	44.00	04.55	05.05	7.00			10.04	0.40		1
	inquiry and facility reservation - Zone 2 4-Wire Unbundled Copper Loop/Long - includes manual svc.	- 1	2	UCL	UCL4L	41.07	44.69	31.55	25.65	7.06			18.94	8.42		
	inquiry and facility reservation - Zone 3		3	UCL	UCL4L	65.28	44.69	31.55	25.65	7.06			18.94	8.42		1
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC	00.20	16.11	16.11	20.00	1.00			10.01	0.12		l .
	4-Wire Unbundled Copper Loop/Long - without manual svc.															1
	inquiry and facility reservation - Zone 1	I	1	UCL	UCL4O	35.56	44.69	31.55	25.65	7.06			18.94	8.42		
	4-Wire Unbundled Copper Loop/Long - without manual svc.															1
	inquiry and facility reservation - Zone 2	1	2	UCL	UCL40	41.07	44.69	31.55	25.65	7.06			18.94	8.42		ł
	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone 3		3	UCL	UCL4O	65.28	44.69	31.55	25.65	7.06			18.94	8.42		1
	Order Coordination for Unbundled Copper Loops (per loop)		5	UCL	UCLMC	05.20	16.11	16.11	23.03	7.00			10.94	0.42		
	CLEC to CLEC conversion Charge without outside dispatch	I		UCL	UREWO		44.69	31.55					18.94	8.42		l
LOOP MODI																
			T	UAL, UHL, UCL,												
			1	UEQ, ULS, UEA,												1
	Unbundled Loop Modification, Removal of Load Coils - 2 Wire		1	UEANL, UDL, UDC,	ULM2L		0.00	0.00					18.94	8.42		1
├ ── ├ ──	pair less than or equal to 18k ft Unbundled Loop Modification, Removal of Load Coils - 2 wire			UDN, UDL, USL	ULIVIZL		0.00	0.00					18.94	ö.42		
	greater than 18k ft		1	UCL, ULS, UEQ	ULM2G		0.00	0.00					18.94	8.42		1
	Unbundled Loop Modification Removal of Load Coils - 4 Wire		1				2.00	2.00								1
	less than or equal to 18K ft			UHL, UCL	ULM4L		0.00	0.00					18.94	8.42		
	Unbundled Loop Modification Removal of Load Coils - 4 Wire															
	pair greater than 18k ft	1	ļ	UCL	ULM4G		0.00	0.00			ļ		18.94	8.42		l
			1	UAL, UHL, UCL, UEQ, UEF, ULS,												
			1	UEQ, UEF, ULS, UEA, UEANL, UDL,												1
	Unbundled Loop Modification Removal of Bridged Tap Removal,	1	1	UDC, UDN, UDL,												1
	per unbundled loop	1	1	USL	ULMBT		0.00	0.00					18.94	8.42		1
SUB-LOOPS																[
Sub-	_oop Distribution															
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-	l .	1													1
├ ── ├ ──	Up			UEANL	USBSA		421.08	421.08					18.94	8.42		
	Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-Up		1	UEANL	USBSB		67.10	67.10					18.94	8.42		1
	Dub Loop 1 of 01035 Dox Location - 1 of 201 all Fallet Set-Op		I		00000		07.10	07.10	I	L	1		10.34	0.42	L	<u>.</u>

UNBUNDL	ED NETWORK ELEMENTS - Georgia												Attachment:	2	Exhi	ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Sub-Loop - Per Building Equipment Room - CLEC Feeder				110000		00474	00474					10.01	0.40		
	Facility Set-Up Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel	- 1	-	UEANL	USBSC		394.74	394.74					18.94	8.42		<u> </u>
	Sub-Loop - Fei Building Equipment Room - Fei 25 Fail Failei Set-Up	1		UEANL	USBSD		154.57	154.57					18.94	8.42		
	Unbundled Sub-Loops, Riser Cable, 2-Wire per Loop, Working			-												
	and Spare Loop Activation			UEANL	USBRC	1.37	2.48	2.48	1.74	1.74			18.94	8.42		
	Unbundled Sub-Loops, Riser Cable, 4-Wire per Loop, Working															
	and Spare Loop Activation			UEANL	USBRD	2.74	4.96	4.96	1.74	1.74			18.94	8.42		
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Statewide			UEANL	USBN2	9.12	207.01	171.32					18.94	8.42		
	Statewide		SW	UEANL	USBINZ	9.12	207.01	171.32					18.94	8.42		
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair		1	UEANL	USBMC		34.22	34.22								
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop -	1	1	1			022	0			1		1	1		1
	Statewide		SW	UEANL	USBN4	8.32	219.35	72.99	123.72	28.77			18.94	8.42		
			1													
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair	<u> </u>	<u> </u>	UEANL	USBMC		34.22	34.22			ļ	ļ				
	Sub-Loop 2-Wire Intrabuilding Network Cable (INC)	I		UEANL	USBR2	1.37	2.48	41.59	115.85	19.17			18.94	8.42		L
	Sub-Loop 2-Wire Intrabuilding Network Cable (INC) - Intermediary Access Terminal (IAT)			UEANL	USBRC	1.37	2.48	2.48	1.74	1.74			18.94	8.42		
				UEANL	USBRC	1.37	2.40	2.40	1.74	1.74			10.94	0.42		<u> </u>
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		34.22	34.22								
	Sub-Loop 4-Wire Intrabuilding Network Cable (INC) -			02/112	CODING		01.22	01122								
	Intermediary Access Terminal (IAT)			UEANL	USBRD	2.74	4.96	4.96	1.74	1.74			18.94	8.42		
	Sub-Loop 4-Wire Intrabuilding Network Cable (INC)	I		UEANL	USBR4	2.96	176.46	55.11	122.17	19.57			18.94	8.42		
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair		4	UEANL	USBMC	5.54	34.22 175.16	34.22 55.50	108.86	24.53			18.84	0.40		┣────
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 1 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2		1	UEF UEF	UCS2X UCS2X	5.54	175.16	55.50 55.50	108.86	24.53			18.84	8.42 8.42		
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2	l i	3	UEF	UCS2X	5.54	175.16	55.50	108.86	24.53			18.94	8.42		
			Ŭ	02.	0002/	0.01		00.00	100.00	2.100			10101	0.1.2		
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC		34.22	34.22								
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 1	I	1	UEF	UCS4X	6.89	219.35	72.99	123.72	28.77			18.94	8.42		
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 2	I	2	UEF	UCS4X	6.89	219.35	72.99	123.72	28.77			18.94	8.42		
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 3	I	3	UEF	UCS4X	6.89	219.35	72.99	123.72	28.77			18.94	8.42		
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC		34.22	34.22								
Unb	undled Network Terminating Wire (UNTW)			UEF	USDIVIC		34.22	34.22								
0110	Unbundled Network Terminating Wire (UNTW) per Pair			UENTW	UENPP	1.37	2.48	2.48	1.74	1.74			18.94	8.42		
Netw	vork Interface Device (NID)			-	-	-	-							-		
	Network Interface Device (NID) - 1-2 lines	I		UENTW	UND12		86.37	56.69					18.94	8.42		
	Network Interface Device (NID) - 1-6 lines	I		UENTW	UND16		127.93	98.21					18.94	8.42		
┝──┤──	Network Interface Device Cross Connect - 2 W		 	UENTW	UNDC2		6.15	6.15					18.94	8.42		───
SUB-LOOPS	Network Interface Device Cross Connect - 4W			UENTW	UNDC4		6.15	6.15			<u> </u>					───
	Loop Feeder		-	+	├ -						<u> </u>					╂─────
Jub	USL-Feeder, DS0 Set-up per Cross Box location - CLEC		1	UEA,												ł
	Distribution Facility set-up	1	1	UDN,UCL,UDL,UDC	USBFW		421.08						18.94	8.42		
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair	1	1	UEA,												
	set-up			UDN,UCL,UDL,UDC			67.10	67.10					18.94	8.42		<u> </u>
	USL Feeder DS1 Set-up at DSX location, per DS1 termination		L	USL	USBFZ		521.57	11.30			ļ		18.94	8.42		L
	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice	1				0.50	200 44	170.05					40.04	0.40		
	Grade- Statewide Order Coordination for Specified Conversion Time, per LSR		SW	UEA UEA	USBFA OCOSL	8.58	206.44 35.74	170.05					18.94	8.42		
├──┼──	Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice			ULA	JUUJL		33.74				<u> </u>		{	ł	-	ł
	Grade - Statewide		sw	UEA	USBFB	8.58	206.44	170.05					18.94	8.42		
	Order Coordination for Specified Time Conversion, per LSR	1	1	UEA	OCOSL	0.00	35.74							0.72		1
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery,															
	Voice Grade Loop - Statewide		SW	UEA	USBFC	8.58	206.44	170.05					18.94	8.42		<u> </u>
1 1	Order Coordination For Specified Conversion Time, per LSR			UEA	OCOSL		35.74									

UNBUNDLE	D NETWORK ELEMENTS - Georgia										.	a - ·	Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
					+		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Statewide		sw	UEA	USBFD	19.91	243.41	81.32	134.77	33.93			18.94	8.42		1
	Order Coordination For Specified Conversion Time, Per LSR		SW	UEA	OCOSL	19.91	35.74	01.32	134.77	33.93			10.94	0.42		1
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice			OEN	CCCCL		00.74									[
	Grade - Statewide		SW	UEA	USBFE	19.91	243.41	81.32	134.77	33.93			18.94	8.42		ł
	Order Coordination For Specified Conversion Time, Per LSR			UEA	OCOSL		35.74									í
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI -															l
	Statewide		SW	UDN	USBFF	17.73	208.50	62.31	119.68	29.58			18.94	8.42		1
	Order Coordination For Specified Conversion Time, Per LSR			UDN	OCOSL	17.70	35.74		110.00				10.00	10.00	10.00	10.00
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)		SW	UDC	USBFS	17.73	208.50	62.31	119.68	29.58			19.99	19.99 19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Statewide Order Coordination For Specified Conversion Time, Per LSR		SW	USL USL	USBFG OCOSL	79.30	203.69 35.74	128.76	124.09	34.80	<u> </u>		19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop -			00L	5003L	I	33.74						1			[
	Statewide		sw	UCL	USBFH	7.22	195.38	63.15	119.68	29.58			18.94	8.42		1
	Order Coordination For Specified Conversion Time, per LSR			UCL	OCOSL		35.74									
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Statewide		SW	UCL	USBFJ	13.72	243.41	81.32	134.77	33.93			18.94	8.42		i
	Order Coordination For Specified Conversion Time, per LSR			UCL	OCOSL		35.74									<u> </u>
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		SW	UDL	USBFN	24.50	243.41	81.32	134.77	33.93			19.99	19.99	19.99	19.99
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -												10.00	10.00	10.00	
	Statewide		SW	UDL	USBFO	24.50	243.41	81.32	134.77	33.93			19.99	19.99	19.99	19.99
	Order Coordination For Specified Time Conversion, per LSR Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -			UDL	OCOSL		35.74									i
	Statewide		sw	UDL	USBFP	24.50	243.41	81.32	134.77	33.93			19.99	19.99	19.99	19.99
-	Order Coordination For Specified Conversion Time, per LSR		0	UDL	OCOSL	2	35.74	01102		00.00	1		10.00	10100	10.00	
SUB-LOOPS				-												
Sub-Lo	pop Feeder															
	Sub Loop Feeder - DS3 - Per Mile Per Month	1		UE3	1L5SL	12.80										L
	Sub Loop Feeder - DS3 - Facility Termination Per Month	1		UE3	USBF1	329.94	3,396.56	406.50	163.61	92.75			18.94	8.42		I
	Sub Loop Feeder – STS-1 – Per Mile Per Month Sub Loop Feeder - STS-1 - Facility Termination Per Month	-		UDLSX	1L5SL USBF7	12.80 372.78	0 000 50	100 50	100.01	92.75			18.94	8.42		ł
	Sub Loop Feeder - STS-1 - Facility Termination Per Month Sub Loop Feeder - OC-3 - Per Mile Per Month			UDLSX UDLO3	1L5SL	3/2.78 9.71	3,396.56	406.50	163.61	92.75			18.94	8.42		
	Sub Loop Feeder - OC-3 - Facility Termination Protection Per			ODLO3	ILJOL	5.71										1
	Month	1		UDLO3	USBF5	57.79										1
	Sub Loop Feeder - OC-3 - Facility Termination Per Month	I		UDLO3	USBF2	524.13	3,396.56	406.50	163.61	92.75			18.94	8.42		l
	Sub Loop Feeder - OC-12 - Per Mile Per Month	I		UDL12	1L5SL	11.95										i
	Sub Loop Feeder - OC-12 - Facility Termination Protection Per															l
	Month	I		UDL12	USBF6	519.09										1
	Sub Loop Feeder - OC-12 - Facility Termination Per Month			UDL12	USBF3	1,570.00	3,396.56	406.50	163.61	92.75			18.94	8.42		l
	Sub Loop Feeder - OC-48 - Per Mile Per Month	1		UDL48	1L5SL	39.20										
	Sub Loop Feeder - OC-48 - Facility Termination Protection Per Month			UDL48	USBF9	259.99							1			ł
	Sub Loop Feeder - OC-48 - Facility Termination Per Month			UDL48	USBF4	1,505.00	3,582.56	406.50	163.61	92.75	1		18.94	8.42		1
	Sub Loop Feeder - OC-12 Interface On OC-48	i		UDL48	USBF8	323.43	803.69	406.50	163.61	92.75			18.94	8.42		
UNBUNDLED I	LOOP CONCENTRATION															
	Unbundled Loop Concentration - System A (TR008)			ULC	UCT8A	441.42	650.81	650.81					19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - System B (TR008)			ULC	UCT8B	52.97	271.17	271.17					19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - System A (TR303)			ULC	UCT3A	478.93	650.81	650.81			ļ		19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - System B (TR303) Unbundled Loop Concentration - DS1 Loop Interface Card			ULC ULC	UCT3B UCTCO	89.26 5.04	271.17 126.57	271.17 92.14	33.57	9.40			19.99 19.99	19.99 19.99	19.99 19.99	19.99 19.99
	Unbundled Loop Concentration - DS1 Loop Interface Card Unbundled Loop Concentration - ISDN Loop Interface (Brite				00100	5.04	120.07	92.14	33.37	9.40			19.99	19.99	19.99	19.99
	Card)			UDN	ULCC1	8.00	21.07	20.96	10.78	10.71			19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - UDC Loop Interface (Brite					0.00	21.01	20.00	10.70	10.71	1		10.00	10.00	10.00	10.00
	Card)			UDC	ULCCU	8.00	21.07	20.96	10.78	10.71			19.99	19.99	19.99	19.99
	Unbundled Loop Concentration2 Wire Voice-Loop Start or												1			i
	Ground Start Loop Interface (POTS Card)			UEA	ULCC2	2.00	21.07	20.96	10.78	10.71			19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery Loop Interface (SPOTS Card)			UEA	ULCCR	11.89	21.07	20.96	10.78	10.71			19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - 4 Wire Voice Loop Interface (Specials Card)			UEA	ULCC4	7.09	21.07	20.96	10.78	10.71			19.99	19.99	19.99	19.99

UNBUND	LED NETWORK ELEMENTS - Georgia												Attachment:	2	Exhi	bit: B
CATEGORY	Y RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonred		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Unbundled Loop Concentration - TEST CIRCUIT Card			ULC	UCTTC	34.67	21.07	20.96	10.78	10.71			19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop Interface			UDL	ULCC7	10.51	21.07	20.96	10.78	10.71			19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - Digital 56 Kbps Data Loop			ODL	01007	10.51	21.07	20.90	10.78	10.71			15.55	19.99	19.99	15.55
	Interface			UDL	ULCC5	10.51	21.07	20.96	10.78	10.71			19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - Digital 64 Kbps Data Loop						-									
	Interface			UDL	ULCC6	10.51	21.07	20.96	10.78	10.71			19.99	19.99	19.99	19.99
UNE OTHE	R, PROVISIONING ONLY - NO RATE															
	NID - Dispatch and Service Order for NID installation			UENTW	UNDBX	0.00	0.00									
	UNTW Circuit Id Establishment, Provisioning Only - No Rate		-	UENTW UEANL,UEF,UEQ,U	UENCE	0.00	0.00									
	Unbundled Contract Name, Provisioning Only - No Rate			ENTW	UNECN	0.00	0.00									1
UNE OTHE	R, PROVISIONING ONLY - NO RATE				0.12011	0.00	0.00				<u> </u>					
	Unbundled Contact Name, Provisioning Only - no rate			UAL,UCL,UDC,UDL, UDN,UEA,UHL,ULC		0.00	0.00									
	Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no			UDIN, UEA, UHL, ULC	UNECIN	0.00	0.00									<u> </u>
	rate			UEA,UDN,UCL,UDC	USBFQ	0.00	0.00									
	Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no															
	rate			UEA,USL,UCL,UDL	USBFR	0.00	0.00									
	Unbundled DS1 Loop - Superframe Format Option - no rate			USL	CCOSF	0.00	0.00									
	Unbundled DS1 Loop - Expanded Superframe Format option -															
	no rate ACITY UNBUNDLED LOCAL LOOP			USL	CCOEF	0.00	0.00									ł
HIGH CAPA																<u> </u>
	High Capacity Unbundled Local Loop - DS3 - Per Mile per month			UE3	1L5ND	8.90										
	High Capacity Unbundled Local Loop - DS3 - Facility Termination per month			UE3	UE3PX	390.34	639.50	426.40					37.55	37.55	18.03	18.03
	High Capacity Unbundled Local Loop - STS-1 - Per Mile per															
	month			UDLSX	1L5ND	8.90										
	High Capacity Unbundled Local Loop - STS-1 - Facility					404 50	C20 F0	400 40					07.55	07.55	40.02	40.00
LOOP MAK	Termination per month			UDLSX	UDLS1	421.59	639.50	426.40					37.55	37.55	18.03	18.03
LOOP WAR	Loop Makeup - Preordering Without Reservation, per working or				-				-							ł
	spare facility queried (Manual).			UMK	UMKLW		35.00	35.00								1
	Loop Makeup - Preordering With Reservation, per spare facility				C.I.I.L		00.00	00.00								
	queried (Manual).			UMK	UMKLP		45.00	45.00								1
	Loop MakeupWith or Without Reservation, per working or															
	spare facility queried (Mechanized)			UMK	PSUMK		0.075	0.075								l
					-											ł
	E SHARING LITTERS-CENTRAL OFFICE BASED		-													
381	Line Sharing Splitter, per System 96 Line Capacity	+		ULS	ULSDA	131.00	0.00	0.00	0.00	0.00	<u> </u>		18.94	8.42	ł	<u> </u>
\vdash	Line Sharing Splitter, per System 24 Line Capacity	1		ULS	ULSDA	32.00	0.00	0.00	0.00	0.00	<u> </u>		18.94	8.42		
	Line Sharing Splitter, Per System, 8 Line Capacity	1		ULS	ULSD8	11.00	0.00	0.00	0.00	0.00			18.94	8.42		l
	Line Sharing-DLEC Owned Splitter in CO-CFA activaton-	1														
	deactivation (per LSOD)			ULS	ULSDG		0.00	0.00	0.00	0.00			18.94	8.42		
ENI	D USER ORDERING-CENTRAL OFFICE BASED-HIGH FREQUENC	Y SPEC	TRUM				10						10-1			ļ
\vdash	Line Sharing - per Line Activation (BST Owned Splitter)			ULS	ULSDC	0.61	10.51	7.70	0.00	0.00			18.94	8.42		ł
	Line Sharing - per Subsequent Activity per Line Rearrangement(BST Owned Splitter	1		ULS	ULSDS		36.23	13.23					18.94	8.42		1
\vdash	Line Sharing - per Subsequent Activity per Line	1	+	010	01000		30.23	13.23			1		10.94	0.42		t
	Rearrangement(DLEC Owned Splitter			ULS	ULSCS		36.23	13.23					18.94	8.42		1
	Line Sharing - per Line Activation (DLEC owned Splitter)	1		ULS	ULSCC	0.61	47.44	19.31	0.00	0.00			18.94	8.42		
	E SPLITTING															
EN	D USER ORDERING-CENTRAL OFFICE BASED															
\vdash	Line Splitting - per line activation DLEC owned splitter			UEPSR UEPSB	UREOS	0.61	=0.1-		10.1-		ļ		10.0		10.0-	
\vdash	Line Splitting - per line activation BST owned - physical			UEPSR UEPSB	UREBP	0.61	53.48	34.48	16.45	12.75			18.94	8.42	19.99	19.99 19.99
1	Line Splitting - per line activation BST owned - virtual			UEPSR UEPSB	UREBV	0.61	53.48	34.48	16.45	12.75	1		18.94	8.42	19.99	19.99

UNBL	INDLE	D NETWORK ELEMENTS - Georgia												Attachment:	2	Exhi	bit: B
CATEC		RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
							Rec	Nonred		Nonrecurring		001150	001111		Rates(\$)	001111	0.000
	DEMO	I TE SITE HIGH FREQUENCY SPECTRUM						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		ERS-REMOTE SITE															<u> </u>
	0	Remote Site Line Share BellSouth Owned Splitter, 24 Port	I		ULS	ULSRB	32.00	0.00	0.00	0.00	0.00			18.94	8.42	19.99	19.99
		Remote Site Line Share Cable Pair Activation CLEC Owned at				1	1										
		RS and Deactivation	1		ULS	ULSTG		74.38	0.00	46.77	0.00			18.94	8.42	19.99	19.99
	END U	SER ORDERING-REMOTE SITE HIGH FREQUENCY SPECTRUM	M AKA I	REMOT	E SITE LINE SHAR	NG											ļ
		Remote Site Line Share Line Activationfor End User Served at RS, BST Splitter			ULS	ULSRC	0.61	10.51	7.70	0.00	0.00			18.94	8.42	19.99	19.99
		RS Line Share Line Activation for End User served at RS, CLEC	1		ULS	ULSKU	0.01	10.51	7.70	0.00	0.00			10.94	0.42	19.99	19.99
		Splitter Remote Site Line Share Subsequent Activity-RS BST Owned	1		ULS	ULSTC	0.61	10.51	7.70	0.00	0.00			18.94	8.42	19.99	19.99
		Splitter	Т		ULS	ULSRS		2.00	3.00					18.94	8.42	19.99	19.99
		Remote Site Line Share Subsequent Activity-RS CLEC Owned Splitter			ULS	ULSTS	1.00	2.00	3.00	4.00	5.00			18.94	8.42	19.99	19.99
UNBU					010	01010	1.00	2.00	5.00	4.00	5.00			10.34	0.42	13.33	13.33
0.120.		INTEROFFICE CHANNEL DEDICATED TRANSPORT - minimu	m billin	g perio	d - below DS3=one	month, DS3/	STS-1=four mo	nths									
		DFFICE CHANNEL - DEDICATED TRANSPORT															
		Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade - Per Mile per month			U1TVX	1L5XX	0.0222										
		Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade - Facility Termination			U1TVX	U1TV2	17.07	79.61	36.08					18.94	18.94		
		Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade Rev Bat Per Mile per month			U1TVX	1L5XX	0.0222										
		Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat Facility Termination			U1TVX	U1TR2	17.07	79.61	36.08					18.94	18.94		
		Interoffice Channel - Dedicated Transport - 56 kbps - per mile per month			U1TDX	1L5XX	0.0222										
		Interoffice Channel - Dedicated Transport - 56 kbps - Facility Termination			U1TDX	U1TD5	16.45	79.61	36.08					18.94	18.94		
		Interoffice Channel - Dedicated Transport - 64 kbps - per mile per month			U1TDX	1L5XX	0.0222										
		Interoffice Channel - Dedicated Transport - 64 kbps - Facility Termination Interoffice Channel - Dedicated Channel - DS1 - Per Mile per			U1TDX	U1TD6	16.45	79.61	36.08					18.94	18.94		
		meronice Channel - Dedicated Channel - DST - Per Mile per month Interoffice Channel - Dedicated Tranport - DST - Facility			U1TD1	1L5XX	0.4523										
		Termination Interoffice Channel - Dedicated Transport - DSI - Pacinty Interoffice Channel - Dedicated Transport - DS3 - Per Mile per			U1TD1	U1TF1	78.47	147.07	111.75					18.94	18.94		
		month Interoffice Channel - Dedicated Transport - DG3 - Fel Nille per Interoffice Channel - Dedicated Transport - DS3 - Facility			U1TD3	1L5XX	2.72										
		Termination per month Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per			U1TD3	U1TF3	788.00	511.10	330.77					37.55	37.55	18.03	18.03
		Interoffice Channel - Dedicated Transport - STS-1 - Facility			U1TS1	1L5XX	2.72										
i i	1	Termination		1	U1TS1	U1TFS	783.63	511.10	449.91					61.19	61.19	3.17	3.17
		CHANNEL - DEDICATED TRANSPORT															
	NOTE:	LOCAL CHANNEL DEDICATED TRANSPORT - minimum billin	g perio	d - belo													L
		Local Channel - Dedicated - 2-Wire Voice Grade Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat			ULDVX ULDVX	ULDV2 ULDR2	13.91 13.91	382.95 382.95	62.40 62.40					18.94 18.94	8.42 18.94		
		Local Channel - Dedicated - 2-Wile Voice Grade Rev Bat		1	UNDVX	ULDR2	14.99	368.44	64.05					18.94	8.42		<u> </u>
		Local Channel - Dedicated - DS1		1	ULDD1	ULDF1	38.36	356.15	312.89	1				44.22	44.22	18.03	18.03
		Local Channel - Dedicated - DS3 - Per Mile per month			ULDD3	1L5NC	6.92										
		Local Channel - Dedicated - DS3 - Facility Termination			ULDD3	ULDF3	515.91	639.50	426.31					37.55	37.55	18.03	18.03
L	I	Local Channel - Dedicated - STS-1- Per Mile per month			ULDS1	1L5NC	6.92	000 50	100.01					10.01	10.01		
DARK	FIREP	Local Channel - Dedicated - STS-1 - Facility Termination			ULDS1	ULDFS	517.56	639.50	426.31					18.94	18.94		<u> </u>
DAKK		Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month - Local Channel			UDF	1L5DC	44.22										
	-	NRC Dark Fiber - Local Channel		1	UDF	UDFC4	44.22	1,355.29	273.69					18.94	18.94		t
L	1		I	L		00104	1	1,000.29	215.09				I	10.94	10.94	l	L

UNBUNDLE	D NETWORK ELEMENTS - Georgia												Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring Dis					Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction															
	Thereof per month - Interoffice Channel			UDF UDF	1L5DF UDF14	44.22	4 055 00	070.00					40.04	40.04		
	NRC Dark Fiber - Interoffice Channel Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction			UDF	UDF14		1,355.29	273.69					18.94	18.94		
	Thereof per month - Local Loop			UDF	1L5DL	44.22										
	NRC Dark Fiber - Local Loop			UDF	UDFL4	77.22	1,355.29	273.69					18.94	18.94		
8XX ACCESS	TEN DIGIT SCREENING			05.	00.21		1,000.20	210.00					10.01	10.01		
	8XX Access Ten Digit Screening, Per Call			OHD		0.0004868										
	8XX Access Ten Digit Screening, Reservation Charge Per 8XX															
	Number Reserved			OHD	N8R1X		6.57	0.76					18.94	18.94		
	8XX Access Ten Digit Screening, Per 8XX No. Established W/O															
	POTS Translations	<u> </u>		OHD			12.81	1.45					18.94	18.94		
	8XX Access Ten Digit Screening, Per 8XX No. Established With	1	1	OHD	N8FTX		12.81	1.45					18.94	10.04		
<u> </u>	POTS Translations 8XX Access Ten Digit Screening, Customized Area of Service			UNU	NOFIA		12.81	1.45					18.94	18.94		
	Per 8XX Number			OHD	N8FCX		4.46	2.23					18.94	18.94		
	8XX Access Ten Digit Screening, Multiple InterLATA CXR	1	+	0.10			. т.	2.20					10.34	10.94		
	Routing Per CXR Requested Per 8XX No.			OHD	N8FMX		5.22	2.99					18.94	18.94		
	8XX Access Ten Digit Screening, Change Charge Per Request			OHD	N8FAX		7.33	0.76					18.94	18.94		
	8XX Access Ten Digit Screening, Call Handling and Destination															
	Features			OHD	N8FDX		4.72	4.46					18.94	18.94		
LINE INFORM	ATION DATA BASE ACCESS (LIDB)															
	LIDB Common Transport Per Query			OQT		0.0000338										
	LIDB Validation Per Query				NDDDY	0.0105974	50.00						18.94	18.94		
SIGNALING (C	LIDB Originating Point Code Establishment or Change			OQT, OQU	NRPBX		50.30						18.94	18.94		
SIGNALING (C	CCS7 Signaling Termination, Per STP Port			UDB	PT8SX	133.99										
	CCS7 Signaling Usage, Per TCAP Message			UDB	1100/	0.000087										
	CCS7 Signaling Connection, Per link (A link)			UDB	TPP++	17.05	131.96	131.96					18.94	18.94		
	CCS7 Signaling Connection, Per link (B link) (also known as D															
	link)			UDB	TPP++	17.05	131.96	131.96					18.94	18.94		
	CCS7 Signaling Usage, Per ISUP Message			UDB		0.0000354										
	CCS7 Signaling Usage Surrogate, per link per LATA			UDB	STU56	340.67										
	CCS7 Signaling Point Code, per Originating Point Code															
	Establishment or Change, per STP affected			UDB	CCAPO		40.00	40.00					18.94	18.94		
	CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected			UDB	CCAPD		8.00	8.00					18.94	18.94		
CALLING NAM	IE (CNAM) SERVICE			000	COAI D		0.00	0.00					10.34	10.34		
	CNAM for DB Owners, Per Query			OQV		0.01										
	CNAM for Non DB Owners, Per Query	1		OQV		0.01							l			l
	CNAM (Non-Databs Owner), NRC, applies when using the	T														
	Character Based User Interface (CHUI)			OQV	CDDCH		595.00	595.00					18.94	18.94		
OPERATOR C	ALL PROCESSING	I	<u> </u>													
	Oper. Call Processing - Oper. Provided, Per Min Using BST LIDB					1.00										
	Oper. Call Processing - Oper. Provided, Per Min Using					1.20										
	Foreign LIDB	1	1			1.24										
	Oper. Call Processing - Fully Automated, per Call - Using BST		1			1.24										
	LIDB	1	1			0.20										
	Oper. Call Processing - Fully Automated, per Call - Using	1	1													
	Foreign LIDB					0.20										
INWARD OPER	RATOR SERVICES															
	Inward Operator Svcs - Verification, Per Minute					1.15										
	Inward Operator Services - Verification and Emergency Interrupt	1	1													
	- Per Minute	1	1	1		1.15						1				
						1	1									
	PERATOR CALL PROCESSING															

CATE UNITY RATE ELEMENTS Image Base Base Base Use Submit	Exhibit: B			Attachment: 2											<u> </u>	BUNDLED NETWORK ELEMENTS - Georgia
Image: Control of Colum Banded OA Annoncommant part adeMW Image: Colum Banded OA Annoncommat part adeMW	Charge - Charge - anual Svc Manual Svc Order vs. Order vs. ectronic- Electronic-	Charge -	Charge - Manual Svc Order vs. Electronic- Add'l	Charge - Manual Svc Order vs. Electronic- 1st	Submitted Manually	Submitted Elec						USOC	BCS	Zone		EGORY RATE ELEMENTS
Lands of Costs Bandod CA Announcement per shellAVV CBVCL Fract Add1 Price Add1 Diffe GUADE SUMAL SU											Rec					
June CCH CDADL Scolo Scolo 18.88 11.98 11.98 11.98 IMPERATION International Control Control Branded CA Announcement per steelhaw International Control Branded CA Announcement per steelhaw International Control Control Branded CA Announcement per steelhaw International Control Branded CA Announcement per steelhaw International Control Branded CA Announcement per steelhaw International Control Branded CA Announcement per steelhaw International Control Branded CA Branded	SOMAN SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	Add'l	First	Add'l	First						
UNEP CLEC Production Printed OA Announcement per shallNAW Production Printed OA Announcement per shall Printed OA Announcement per shallNAW Production Printed OA Announcement per shall Printed OA Announcement per shall Printed OA Announcement per shall Printed OA Announcement per shall Printed OA Announcement per shall Printed OA Announcement per shall Printed OA Announcement per shall Printed OA Announcement per shall Printed OA Announcement per shall Printed OA Announcem			10.00	10.00					500.00	500.00		00401				
Necoring of Cutom Binded DA Announcement per shelfNM/ In the Out Announcement per shelfNM/ In the Out Network States Call Converting of Call Data Database States Call Convert States Call Convert States Call Data Database Call Convert States Call Convert States Call Database Call Databas			19.99	19.99					500.00	500.00		CBAUL				
Lossing of Lasting of Lasting Status Image: Construction Status Image: Constatus Image: Construction Status <th< td=""><td>19.99 19.99</td><td>19.9</td><td>19.99</td><td>19.99</td><td></td><td></td><td></td><td></td><td>7 000 00</td><td>7 000 00</td><td></td><td></td><td></td><td></td><td>┼───┦</td><td></td></th<>	19.99 19.99	19.9	19.99	19.99					7 000 00	7 000 00					┼───┦	
Iber CCN Iber CCN Iber CCN Iber CON	10.00	10.00	10.00	10.00					7,000.00	1,000.00					<u>├</u> ──┤	
Unbranding via OLS OF UNEP CLEC Image: Clear Classifier of UNE			19.99	19.99					500.00	500.00						
DIRECTORY ASSISTANCE SERVICES Image: Control Directory Assistance (Control Directory Directory Assistance (Control Directory Directory Assistance (Control Directory Directory Assistance (Control Directory Directory Assistance (Control Directory Directory Assistance (Control Directory Assistance																Unbranding via OLNS for UNEP CLEC
DIRECTORY ASSISTANCE ACCESS SERVICE Image: Charge Par Call Image: Cha			19.99	19.99					1,200.00	1,200.00					1	
Directory Assistance Access Service CodeC Other Control Assistance Access Service (Code Service OACC) Image: Control Assistance Code Service (Code Service) Image: Code Service Code Service (Code Service) Image: Code Service Code Service (Code Service) Image: Code Service (C																
UNECTORY ASSISTANCE CALL COMPLETION ACCESS SERVICE (DACC). Image: Complexity Assistance Data Service (DACC). Imag																
Directory Assistance Call Completion Access Service (DACC). Image: Complex Com											0.275					
Byer Call Attempt Image: Call Attempt		L	<u>لـــــا</u>	ļ]								ļ			JACC)	
Directory Asistance Services Image: Control Asistance Data Bases Service (DADS) Image			1 1								a				1 '	
IDEECTORY ASSISTANCE DATA BASE SERVICE (DADS) Image: Control of the service of the ser			ļļ	<u> </u>							0.10				\downarrow	
Directory Assistance Data Base Service, Darge Per Litting 0.04 0			├──── ┦	┟────┤											\vdash	
Directory Assistance Date Base Swriter, per month DBSOF 150.00 Image: Control of Control Contro Contero Control Control Control Control Control Cont		———		├ ──── │							0.04					
BRANDRS - DiRECTORY ASSISTANCE Image: Control of the con				┝────┦								DBSOE				
Feating Based CLEC Provisioning of DA Custom Branded AMT CBADA 6,000.00 6,000.00 16.94 16.94 Recording and Provisioning of DA Custom Branded Announcement per Switch AMT CBADA 6,000.00 6,000.00 16.94 8.42 UNRP CLEC INRP CLEC AMT CBADC 11,710.00 118.94 8.42 Instructure Instructure Instructure 3,000.00 3,000.00 3,000.00 3,000.00 118.94 8.42 Instructure Instructure Instructure 11,700.00 11,710.00 118.94 8.42 Ubbranding of DA per CON (1 COR) per Order) Instructure 420.00 420.00 18.94 8.42 Stelective Routing of DA per CON (1 cOR) per Order) Instructure 16.00 16.00 18.94 8.42 Stelective Routing Per Unique Line Class Code Per Request Per Stele Instructure 18.94 18.94 8.42 Virtual Collocation - Application Cost AMTFS ESPX 2,750.00 2,750.00 2,750.00 19.99 19.99 19.99			├ ────┦	├ ───┤							130.00	DB301			┼───┦	
Recording and Provisioning UD A Custom Branded Announcement pr Switch ANT CBADA 6,000.00 6,000.00 5,000.00 115.94 8.42 Loading Of Custom Branded Announcement PS Witch ANT CBADC 1,170.00 115.94 8.42 Nee Cell E 3000.00 1,170.00 115.94 8.42 Loading Of DA Custom Branded Announcement PS Witch per Cell Costom Branded Announcement PS Witch per Coll Costom Branded Announcement PS Witch per Coll Costom Branded Announcement PS Witch per Coll Costom Branded Announcement PS Witch per Coll Costom Branded Announcement PS Witch per Coll Costom Branded Announcement PS Witch per Coll Costom Branded Announcement PS Witch per Coll Costom Branded Announcement PS Witch per Coll Costom Branded Announcement PS Witch per Coll Costom Branded Announcement PS Witch per Coll Costom Branded Announcement PS Witch per Coll Costom Branded Announcement PS Witch Per Coll Costom Branded Announcement PS Witch Per Coll Costom Cost Per Request Per Coll Costom Cost Per Request Per Coll Costom Cost Per Request Per Coll Costom Cost Per Request Per Coll Costom Cost Per Request Per Coll Costom Cost Per Request Per Coll Costom Cost Per Request Per Coll Costom Cost Per Request Per Coll Costom Cost Per Request Per Coll Costom Cost Per Request Per Coll Costom Cost Per Request Per Coll Costom Cost Per Request Per Coll Costom Cost Per Request Per Cost Per Cost Per Cost Per Request Per Cost Per Cost Per Cost Per Request Per Co				 												
Announcement ANT CBADA 6,000.00 6,000.00 18.94 8.42 UNEP CLEC ANT CBADC 1,170.00 1.70.00 1.70.00 1.84 8.42 UNEP CLEC ANT CBADC 1,170.00 1.70.00 1.84 8.42 Loading of DX Gustom Branded Announcement per Switch per OCN 1 1.70.00 1.70.00 1.84 8.42 Unbranding via OLNS for UNEP CLEC 1 1.170.00 1.170.00 1.86.44 8.42 Electrive Routing of DA per CON (1 COX per Order) 1 4.420.00 420.00 18.94 8.42 SELECTIVE ROuting Per Unique Line Class Code Per Request Per Electrive Routing Per Unique Line Class Code Per Request Per 1.90.62 180.62 180.62 33.67 7.88 WITHUL COLLOCATION 2.970.00 2.970.00 19.99 19.99 19.99 Virtual Collocation - Application Cost, per cable AMTFS ESPVX 3.20 2.970.00 19.99 19.99 Virtual Collocation - Cable Installation Cost, per cable AMTFS ESPXX 3.20 19.99 <				├ ─── /											<u>├</u> ──┤	
Loading of Custom Branded Announcement per Switch AMT CBADC 1,170.00 1,170.00 1,170.00 1,170.00 1,170.00 1,170.00 1,170.00 1,170.00 1,170.00 1,170.00 1,170.00 1,18,94 8.42 International of DA Custom Branded Announcement per Switch per OCN 0.000.00 3,000.00 3,000.00 1,170.00 1,18,94 8.42 Usading of DA Stor UNEP CLEC 1,170.00 1,170.00 1,170.00 1,170.00 1,18,94 8.42 Usading of DA per OCN (1 GCN per Order) 1 420.00 420.00 18,94 8.42 SELECTIVE ROUTING 11,070.00 18,00 18,94 8.42 18,94 8.42 Selective Routing Per Unique Line Class Code Per Request Per Switch per OCN 18,00 18,00 18,94 8.42 Virtual Collocation - Application Cost AMITFS EAF 2,848.30 19,99 19,99 19,99 19,99 19,99 19,99 19,99 19,99 19,99 19,99 19,99 19,99 19,99 19,99 19,99 19,99 19,99			8.42	18.94					6.000.00	6.000.00		CBADA	AMT			
UNEP CLEC Image: Constraint of DA Custom Branded Announcement Image: Constraint of DA Custom Branded Announcement per Switch per Usation Branded Announcement per Switch per Devent per Inset and Branded Brande Branded Announcement per Switch per Per Per Per Per Per Per Per Per Per P		1													++	
Loading of DA Custom Branded Announcement per Switch per OCN Image: Constraint of the constraint o																UNEP CLEC
OCN OCN Introduction I			8.42	18.94					3,000.00	3,000.00					1	Recording of DA Custom Branded Announcement
Unbranding via OLNS for UNEP CLEC Image: Construction of the const																Loading of DA Custom Branded Announcement per Switch per
Loading of DA per OCN (1 OCN per Order) Image: Constraint of the CN <thimage: cn<="" constraint="" of="" td="" the="" tht<=""><td></td><td></td><td>8.42</td><td>18.94</td><td></td><td></td><td></td><td></td><td>1,170.00</td><td>1,170.00</td><td></td><td></td><td></td><td></td><td></td><td></td></thimage:>			8.42	18.94					1,170.00	1,170.00						
Loading of DA per Switch per OCN Image: Constraint of the cons																
SELECTVE ROUTING Image: Control of the second																
Stelective Routing Per Unique Line Class Code Per Request Per Switch USRCR 180.62 180.62 33.67 7.88 VIRTUAL COLLOCATION Image: Collocation - Application Cost Image: Collocation - Application Cost Image: Collocation - Application Cost Image: Collocation - Application Cost Image: Collocation - Collocation - Cost Image: Collocation - Cost<			8.42	18.94					16.00	16.00						
Switch USRCR 180.62 180.62 33.6 7.88 VIRTUAL COLLOCATION AMTFS EAF 2.848.30 180.62			ļļ	<u> </u>												
VIRTUAL ColLocation - Application Cost MTFS EAF 2,848,30 M </td <td></td> <td></td> <td>7.00</td> <td>00.07</td> <td></td> <td></td> <td></td> <td></td> <td>100.00</td> <td>100.00</td> <td></td> <td>10000</td> <td></td> <td></td> <td></td> <td></td>			7.00	00.07					100.00	100.00		10000				
Virtual Collocation - Application Cost AMTFS EAF 2,848.30 19.99 19.99 19.99 Virtual Collocation - Cable Installation Cost, per cable AMTFS ESPCX 2,750.00 19.99 19.99 19.99 Virtual Collocation - Cable Installation Cost, per cable AMTFS ESPX 3.20 19.99 19.99 Virtual Collocation - Power, per fused amp AMTFS ESPX 3.48 1000000000000000000000000000000000000		┟────	7.88	33.67					180.62	180.62		USRCR			\vdash	
Virtual Collocation - Cable Installation Cost, per cable AMTFS ESPCX 2,750.00 2,750.00 19.99 19.99 19.99 Virtual Collocation - Floor Space, per sq. ft. AMTFS ESPVX 3.20			10.00	10.00					2 848 20	2 949 20		EAE	AMTEC			
Virtual Collocation - Floor Space, per sq. ft. AMTFS ESPVX 3.20 Image: Constraint of the system of the syst															┼───┦	
Virtual Collocation - Power, per fused amp AMTFS ESPAX 3.48 Virtual Collocation - Cable Support Structure, per entrance cable AMTFS ESPSX 13.35		┝───	19.99	19.99					2,100.00	2,100.00	3 20				\vdash	
Virtual Collocation - Cable Support Structure, per entrance AMTFS ESPSX 13.35 Image: SepSX I		<u> </u>		├─── ┤											├ ───┦	
cable AMTFS ESPSX 13.35 Image: Constraint of the state of		<u> </u>	rł								0.10		1		<u>├</u> ──┦	
Image: space of the system			1 1								13.35	ESPSX	AMTFS		1 '	
Virtual Collocation - 2-wire Cross Connects (loop) UNCNX UEAC2 0.0283 24.56 23.56 9.20 8.30 19.99 19.99 19.99 Virtual Collocation - 2-wire Cross Connects (loop) UEA,UHL,UCL,UDL, AMTFS, UAL, UDN, UNCVX, UNCDX UEAC4 0.0566 24.75 23.70 9.03 8.10 19.99 19.99 19.99 Virtual Collocation - 4-wire Cross Connects (loop) AMTFS, UDL12, UDL03, U1748, U1712, U1703, ULD03, ULD12, ULD03, ULD12, ULD048, UDF AMTFS, UDL12, CNC2F 2.88 41.72 30.36 10.43 8.36 2.20 2.20													DC,UAL,UHL,UCL,U EQ, AMTFS, UDL,			
Virtual Collocation - 4-wire Cross Connects (loop) UEA,UHL,UCL,UDL, AMTFS, UAL, UDN, UNCVX, UNCDX UEAC4 0.0566 24.75 23.70 9.03 8.10 19.99 19.99 Virtual Collocation - 4-wire Cross Connects (loop) AMTFS, UAL, UDN, UNCVX, UNCDX UEAC4 0.0566 24.75 23.70 9.03 8.10 19.99 19.99 Virtual Collocation - 2-Fiber Cross Connects ULD03, ULT12, ULD03, ULD12, ULD04, UDF CNC2F 2.88 41.72 30.36 10.43 8.36 2.20 2.20	19.99 19.99	19.9	19.99	19.99			8.30	9.20	23.56	24.56	0.0283	UEAC2			1 '	Virtual Collocation - 2-wire Cross Connects (loop)
AMTFS, UDL12, UDL03, U1T48, U1T12, U1T03, ULD03, ULD12, ULD04, UDF AMTFS, UDL12, CNC2F AMTFS, UDL12, 2.88 AMTFS, UDL 12, 4.00 AMTFS, UDL12, ULD03, ULT12, UT103, ULD04, UDF AMTFS, UDL12, 2.88 AMTFS, UDL12, 4.00 AMTFS, UDL12, ULD04, UDF AMTFS, UDL12, 2.88 AMTFS, UDL12, 4.00 AMTFS, UDL12, <th< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>AMTFS, UAL, UDN,</td><td></td><td></td><td></td></th<>													AMTFS, UAL, UDN,			
AMTFS,UDL12,	19.99 19.99	19.99	19.99	19.99			8.10	9.03	23.70	24.75	0.0566	UEAC4	AMTFS,UDL12, UDLO3, U1T48, U1T12, U1T03,			Virtual Collocation - 4-wire Cross Connects (loop)
			2.20	2.20			8.36	10.43	30.36	41.72	2.88	CNC2F			1 '	Virtual Collocation - 2-Fiber Cross Connects
Virtual Collocation - 4-Fiber Cross Connects U1T12, U1T03, ULD03, ULD12, ULD03, ULD12, ULD03, ULD12, ULD03, ULD148, UDF 5.76 51.03 39.67 13.71 11.65 2.20 2.20			2.20	2.20			11 65	13 71	30.67	51.02	5 76		UDLO3, U1T48, U1T12, U1T03, ULDO3, ULD12,			Virtual Collocation - 4-Fiber Cross Connects

UNBUNDL	ED NETWORK ELEMENTS - Georgia												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Manually			Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge -
						_	Nonre	curring	Nonrecurring	g Disconnect			OSS	Rates(\$)		L
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN		SOMAN	SOMAN	SOMAN
	Virtual collocation - Special Access & UNE, cross-connect per DS1			JSL,ULC,AMTFS, JLR, UXTD1, JNC1X, ULDD1, J1TD1, USLEL, JNLD1	CNC1X	7.50	155.00	14.00					19.99	19.99		
	Virtual collocation - Special Access & UNE, cross-connect per DS3 Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable		E U U U	JSL,ULC,AMTFS,U E3, U1TD3, UXTS1, JXTD3, UNC3X, JNCSX, ULDD3, J1TS1, ULDS1, JDLSX, UNLD3	CND3X	56.25	151.90	11.83					19.99	19.99		
	Support Structure, per linear foot		А	MTFS	VE1CB	0.0023										
	Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per linear ft	1		AMTES	VE1CD	0.0034										
	Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per cable			AMTES	VE1CC	0.0004	553.43						19.99			
	Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per cable			AMTES	VE1CE		553.43						19.99			
	Virtual Collocation Cable Records - per request	-		AMTES	VE1CE VE1BA		1,706.00	1,706.00			-		19.99			
	Virtual Collocation Cable Records - VG/DS0 Cable, per cable record			AMTES	VE1BA		922.38	922.38								
	Virtual Collocation Cable Records - VG/DS0 Cable, per each 100 pair			AMTES	VE1BD		18.00	18.00								
	Virtual Collocation Cable Records - DS1, per T1TIE	-		MTFS	VE1BC		8.43	8.43			-					
	Virtual Collocation Cable Records - DS3, per T3TIE	-		AMTES	VE1BD		29.49	29.49								
	Virtual Collocation Cable Records - Fiber Cable, per 99 fiber records			AMTES	VE1BF		278.61	278.61								
	Virtual collocation - Security Escort - Basic, per half hour			MTFS	SPTBX		41.00	25.00					19.99	19.99		
	Virtual collocation - Security Escort - Overtime, per half hour			MTFS	SPTOX		48.00	30.00					19.99	19.99		
	Virtual collocation - Security Escort - Premium, per half hour			MTFS	SPTPX		55.00	35.00					19.99	19.99		
	Virtual collocation - Maintenance in CO - Basic, per half hour			MTFS	CTRLX		30.64	30.64					19.99	19.99		
	Virtual collocation - Maintenance in CO - Overtime, per half hour		А	MTFS	SPTOM		35.77	35.77					19.99	19.99		
	Virtual collocation - Maintenance in CO - Premium per half hour		A	AMTES	SPTPM		40.90	40.90					19.99	19.99		
VIRTUAL CO																
	Virtual Collocation - 2-wire Cross Connect, Exchange Port 2- Wire Analog - Res		U	JEPSR	VE1R2	0.30	12.60	12.60					18.94	8.42		
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2- Wire Line Side PBX Trunk - Bus		U	JEPSP	VE1R2	0.30	12.60	12.60					18.94	8.42		
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Voice Grade PBX Trunk - Res		U	JEPSE	VE1R2	0.30	12.60	12.60					18.94	8.42		
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Analog Bus		U	JEPSB	VE1R2	0.30	12.60	12.60					18.94	8.42		
	Virtual Collocation 2-Wire Cross Connect, Exchnage Port 2-Wire ISDN		U	JEPSX	VE1R2	0.30	12.60	12.60					18.94	8.42		
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire ISDN			JEPTX	VE1R2	0.30	12.60	12.60					18.94	8.42		
	Virtual Collocation 4-Wire Cross Connect, Exchange Port 4-Wire ISDN DS1			JEPEX	VE1R4	0.50	12.60	12.60					18.94	8.42		
VIRTUAL CO						0.00	.2.00	.2.00	İ	İ				5.72	İ	
	Virtual Collocation-2 Wire Cross Connects (Loop) for Line Splitting			JEPSR, UEPSB	VE1LS	0.03	24.56	23.56	9.20	8.30			19.99	19.99		
PHYSICAL C	OLLOCATION	1		,	-						1					
	Physical Collocation-2 Wire Cross Connects (Loop) for Line Splitting		U	JEPSR, UEPSB	PE1LS	0.0318	11.94	11.46					19.99	19.99		
AIN SELECT	VE CARRIER ROUTING			•											10	
	Regional Service Establishment		S	SRC	SRCEC		391,788.00				1		19.99	19.99	19.99	19.99

UNBUNDLE	D NETWORK ELEMENTS - Georgia		r		-	•						0	Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
					-		Nonrec	urring	Nonrecurring Dis	sconnect			OSS	Rates(\$)		L
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN		SOMAN	SOMAN	SOMAN
	End Office Establishment			SRC	SRCEO		320.53	320.53					19.99	19.99	19.99	19.99
	Line/Port NRC, per end user			SRC	SRCLP		2.06	2.06					19.99	19.99	19.99	19.99
	Query NRC, per query			SRC		0.000448										
AIN - BELLSO	UTH AIN SMS ACCESS SERVICE															
	AIN SMS Access Service - Service Establishment, Per State, Initial Setup			A1N	CAMSE		90.25	90.25					18.94	18.94		ļ
	AIN SMS Access Service - Port Connection - Dial/Shared Access			A1N	CAMDP		29.66	29.66					18.94	18.94		i
	AIN SMS Access Service - Port Connection - Dia/Shared Access			AIN	CAM1P		29.66	29.66					18.94	18.94		i
	AIN SMS Access Service - User Identification Codes - Per User				CANITI		23.00	23.00					10.34	10.34		l
	ID Code			A1N	CAMAU		84.43	84.43					18.94	18.94		i
1	AIN SMS Access Service - Security Card, Per User ID Code,			1		1 1	0.1.0	00	i				10.04			
	Initial or Replacement	1	1	A1N	CAMRC		35.44	35.44					18.94	18.94		1
	AIN SMS Access Service - Storage, Per Unit (100 Kilobytes)					0.0023										1
	AIN SMS Access Service - Session, Per Minute					0.0795604										
	AIN SMS Access Service - Company Performed Session, Per															(
	Minute					2.08										i
AIN - BELLSO	UTH AIN TOOLKIT SERVICE															
	AIN Toolkit Service - Service Establishment Charge, Per State,															Í
	Initial Setup			CAM	BAPSC		86.74	86.74					18.94	18.94		L
	AIN Toolkit Service - Training Session, Per Customer				BAPVX		8,348.00	8,348.00					18.94	18.94		
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per															i
	DN, Term. Attempt				BAPTT		19.13	19.13					18.94	18.94		
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per												10.01			i
	DN, Off-Hook Delay				BAPTD		114.80	114.80					18.94	18.94		
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate				BAPTM		19.13	19.13					18.94	18.94		
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP				BAPTO		70.06	70.06					18.94	18.94		
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP				BAPTC		70.06	70.06					18.94	18.94		
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code				BAPTF		70.06	70.06					18.94	18.94		
	AIN Toolkit Service - Query Charge, Per Query					0.0209223										
	AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit Subscription, Per Node, Per Query					0.0053137										
	AIN Toolkit Service - SCP Storage Charge, Per SMS Access															
	Account, Per 100 Kilobytes					1.46										L
	AIN Toolkit Service - Monthly report - Per AIN Toolkit Service Subscription			САМ	BAPMS	15.96	22.64	22.64					18.94	18.94		
	AIN Toolkit Service - Special Study - Per AIN Toolkit Service Subscription			CAM	BAPLS	0.0861109	22.64	22.64					18.94	18.94		
	AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service Subscription			CAM	BAPDS	15.87	22.64	22.64					18.94	18.94		
	AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit Service Subscription			CAM	BAPES	0.0028704	22.64	22.64					18.94	18.94		
	XTENDED LINK (EELs)															
	New Density Zone 1 EELs are available in the following MSA					Atlanta, Ga; Nev	v Orleans, LA,									
	Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-															
	In all states, EEL network elements shown below also apply to												UNEs.(Non-re	curring rates	do not apply.	.)
	In All States the EEL network elements apply to ordinarily con				vitch As Is Cha	arge.) When or	dering ordinari	ly combined r	etwork elements,	Non-recurr	ring rates do	o apply.				
2-WIR	E VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INT	EROFF		ANSPORT (EEL)		├ ──── │										
	First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1		1	UNCVX	UEAL2	16.84	104.14	78.10					18.94	8.42		ļ
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2		2	UNCVX	UEAL2	19.45	104.14	78.10					18.94	8.42		ļ
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 3		3	UNCVX	UEAL2	30.92	104.14	78.10					18.94	8.42		1

UNBUNDLE	D NETWORK ELEMENTS - Georgia											-	Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonree			g Disconnect				Rates(\$)		·
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Interoffice Transport - Dedicated - DS1 combination - Per Mile per month			UNC1X	1L5XX	0.4523										
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination per month			UNC1X	U1TF1	78.47	194.63	141.51					33.63	27.49	19.88	11.85
	DS1 Channelization System Per Month			UNC1X	MQ1	126.22	104.00	141.01					00.00	21.45	10.00	11.00
	Voice Grade COCI - DS1 To Ds0 Interface - Per Month			UNCVX	1D1VG	1.17	12.02	8.66					18.94	8.42		
	Each Additional 2-Wire VG Loop(SL 2) in the same DS1															
	Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL2	16.84	104.14	78.10					18.94	8.42		
	Each Additional 2-Wire VG Loop(SL2) in the same DS1															
	Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL2	19.45	104.14	78.10					18.94	8.42		L
	Each Additional 2-Wire VG Loop(SL2) in the same DS1		3			20.00	404.44	70.40					40.04	0.40		
	Interoffice Transport Combination - Zone 3 Voice Grade COCI - DS1 to DS0 Channel System combination -		3	UNCVX	UEAL2	30.92	104.14	78.10					18.94	8.42		
	per month			UNCVX	1D1VG	1.17	12.02	8.66					18.94	8.42		
	Nonrecurring Currently Combined Network Elements Switch -As-															
	Is Charge			UNC1X	UNCCC		12.97	11.27					45.46	15.72		
4-WIRE	E VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INT	EROFF	ICE TR	ANSPORT (EEL)												L
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice															
	Transport Combination - Zone 1		1	UNCVX	UEAL4	22.26	206.95	170.57					18.94	8.42		
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice															
	Transport Combination - Zone 2		2	UNCVX	UEAL4	25.70	206.95	170.57					18.94	8.42		L
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL4	40.86	206.95	170.57					18.94	8.42		
	Interoffice Transport - Dedicated - DS1 combination - Per Mile		Ŭ			40.00	200.00	110.01					10.04	0.42		
	Per Month			UNC1X	1L5XX	0.4523										
	Interoffice Transport - Dedicated - DS1 - Facility Termination Per				120/01	0.1020										
	Month			UNC1X	U1TF1	78.47	194.63	141.51					33.63	27.49	19.88	11.85
	Channelization - Channel System DS1 to DS0 combination Per															
	Month			UNC1X	MQ1	126.22										
	Voice Grade COCI - DS1 to DS0 Channel System combination -															
	per month			UNCVX	1D1VG	1.17	12.02	8.66								
	Additional 4-Wire Analog Voice Grade Loop in same DS1															
	Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL4	22.26	206.95	170.57					18.94	8.42		
	Additional 4-Wire Analog Voice Grade Loop in same DS1		~		115 11 1	05 70	000.05	470 57					40.04	0.40		
	Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL4	25.70	206.95	170.57					18.94	8.42		
	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL4	40.86	206.95	170.57					18.94	8.42		
	Voice Grade COCI - DS1 to DS0 Channel System combination -		3	UNCVA	ULAL4	40.00	200.95	170.57					10.94	0.42		ł
	per month			UNCVX	1D1VG	1.17	12.02	8.66					18.94	8.42		
	Nonrecurring Currently Combined Network Elements Switch -As-															
	Is Charge			UNC1X	UNCCC		12.97	11.27					45.46	15.72		
4-WIRE	56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1	INTERC	FFICE	TRANSPORT (EEL))											
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice															
	Transport Combination - Zone 1		1	UNCDX	UDL56	25.75	384.56	241.20					18.94	8.42		
	First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice		_													
	Transport Combination - Zone 2		2	UNCDX	UDL56	29.74	384.56	241.20					18.94	8.42		
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice		~			47.07	004 50	0.44.00					18.94	0.40		
	Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile		3	UNCDX	UDL56	47.27	384.56	241.20					18.94	8.42		ł
	Per Month			UNC1X	1L5XX	0.4523										
	Interoffice Transport - Dedicated - DS1 - combination Facility				123//	0.4525				1						ł
	Termination Per Month			UNC1X	U1TF1	78.47	194.63	141.51					33.63	27.49	19.88	11.85
	Channelization - Channel System DS1 to DS0 combination Per			001/		10.41	104.00	141.01			<u> </u>		00.00	21.43	10.00	11.00
	Month			UNC1X	MQ1	126.22										1
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per				1 1	.20.22			1	1	1		1	1	1	l
	month (2.4-64kbs)			UNCDX	1D1DD	1.86	12.02	8.66					18.94	8.42		
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1															
	Interoffice Transport Combination - Zone 1		1	UNCDX	UDL56	25.75	384.56	241.20		1	1		18.94	8.42		1

UNBUNDLE	D NETWORK ELEMENTS - Georgia	1	1	1	· · · ·						0		Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 2		2	UNCDX	UDL56	29.74	384.56	241.20					18.94	8.42		
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1		2	UNCDA	UDL36	29.74	364.30	241.20		1			10.94	0.42		1
	Interoffice Transport Combination - Zone 3		3	UNCDX	UDL56	47.27	384.56	241.20					18.94	8.42		
	OCU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-64kbs)			UNCDX	1D1DD	1.86	12.02	8.66					18.94	8.42		
	Nonrecurring Currently Combined Network Elements Switch -As-						10.07									
4.W/B	IS Charge E 64 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1			UNC1X	UNCCC		12.97	11.27					18.94	8.42		1
4-111	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice			TRANSPORT (EEE)												
	Transport Combination - Zone 1		1	UNCDX	UDL64	25.75	348.55	241.20					18.94	8.42		
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2		2	UNCDX	UDL64	29.74	348.55	241.20					18.94	8.42		
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice		-	0.100/	50104	23.14	5-10.00	271.20			1		10.34	0.42		
	Transport Combination - Zone 3		3	UNCDX	UDL64	47.27	348.55	241.20					18.94	8.42		
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month			UNC1X	1L5XX	0.4523										
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Month			UNC1X	U1TF1	78.47	194.63	141.51					33.63	27.49	19.88	11.85
	Channelization - Channel System DS1 to DS0 combination Per Month			UNC1X	MQ1	126.22										
	OCU-DP COCI (data) - DS1 to DS0 Channel System			UNCIX		120.22										
	combination - per month (2.4-64kbs)			UNCDX	1D1DD	1.86	12.02	8.66					18.94	8.42		
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 1		1	UNCDX	UDL64	25.75	348.55	241.20					18.94	8.42		
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 2		2	UNCDX	UDL64	29.74	348.55	241.20					18.94	8.42		
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 3		3	UNCDX	UDL64	47.27	348.55	241.20					18.94	8.42		
	OCU-DP COCI (data) - DS1 to DS0 Channel System		Ū													
	combination - per month (2.4-64kbs) Nonrecurring Currently Combined Network Elements Switch -As-			UNCDX	1D1DD	1.86	12.02	8.66					18.94	8.42		
	Is Charge			UNC1X	UNCCC		12.97	11.27					45.46	15.72		
4-WIR	E DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTE	ROFFI	CE TR/	ANSPORT (EEL)												
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1		1	UNC1X	USLXX	55.53	443.20	138.69					18.94	8.42		
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2		2	UNC1X	USLXX	64.13	443.20	138.69					18.94	8.42		
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice										1					
	Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile		3	UNC1X	USLXX	101.93	443.20	138.69	<u> </u>				18.94	8.42		
	Per Month Interoffice Transport - Dedicated - DS1 combination - Facility		<u> </u>	UNC1X	1L5XX	0.4523					+					
	Termination Per Month		<u> </u>	UNC1X	U1TF1	78.47	194.63	141.51					33.63	27.49	19.88	11.85
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge		1	UNC1X	UNCCC		12.97	11.27					45.46	15.72		
4-WIR	E DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INT	ROFFI	CE TR	ANSPORT (EEL)												
	First DS1Loop in DS3 Interoffice Transport Combination - Zone		1	UNC1X	USLXX	55.53	443.20	138.69					18.94	8.42		
	First DS1Loop in DS3 Interoffice Transport Combination - Zone									İ	1					ĺ
	2 First DS1Loop in DS3 Interoffice Transport Combination - Zone		2	UNC1X	USLXX	64.13	443.20	138.69			1		18.94	8.42		
	3 Interoffice Transport - Dedicated - DS3 combination - Per Mile		3	UNC1X	USLXX	101.93	443.20	138.69					18.94	8.42		
	Per Month			UNC3X	1L5XX	2.72					 					
	Interoffice Transport - Dedicated - DS3 - Facility Termination per month			UNC3X	U1TF3	788.00	198.45	153.15					37.55	37.55	18.03	18.03
	DS3 to DS1 Channel System combination per month			UNC3X	MQ3	137.73	196.66	204.61					18.94	8.42		

CATEGORY											Svc Order	Svc Order	Incremental	Incromontal		
	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 1		1	UNC1X	USLXX	55.53	443.20	138.69					18.94	8.42		1
	Additional DS1Loop in DS3 Interoffice Transport Combination -			UNCIX	USLAA	55.55	443.20	130.09					10.94	0.42		
	Zone 2		2	UNC1X	USLXX	64.13	443.20	138.69					18.94	8.42		1
	Additional DS1Loop in DS3 Interoffice Transport Combination -															
	Zone 3		3	UNC1X	USLXX	101.93	443.20	138.69					18.94	8.42		
	DS3 Interface Unit (DS1 COCI) combination per month Nonrecurring Currently Combined Network Elements Switch -As-			UNC1X	UC1D1	11.02	12.02	8.66					18.94	8.42		┣────
	Is Charge			UNC3X	UNCCC		12.97	11.27					45.46	15.72		1
2-WIRE	VOICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INT	EROFF	ICE TR													
	2-WireVG Loop used with 2-wire VG Interoffice Transport															
	Combination - Zone 1		1	UNCVX	UEAL2	16.84	104.14	78.10					18.94	8.42		L
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL2	19.45	104.14	78.10					18.94	8.42		
<u> </u>	2-WireVG Loop used with 2-wire VG Interoffice Transport		2	UNCVA	UEALZ	19.45	104.14	76.10					10.94	0.42		
	Combination - Zone 3		3	UNCVX	UEAL2	30.92	104.14	78.10					18.94	8.42		
	Interoffice Transport - Dedicated - 2-wire VG combination - Per															
	Mile Per Month			UNCVX	1L5XX	0.0222										L
	Interoffice Transport - Dedicated - 2- Wire Voice Grade															
	combination - Facility Termination per month Nonrecurring Currently Combined Network Elements Switch -As-			UNCVX	U1TV2	17.07	79.61	36.08					18.94	18.94		
	Is Charge			UNCVX	UNCCC		12.97	11.27					45.46	15.72		
4-WIRE	E VOICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INT	EROFF	ICE TR		0.1000		12.01						10.10	10.112		
	4-WireVG Loop used with 4-wire VG Interoffice Transport															
	Combination - Zone 1		1	UNCVX	UEAL4	22.26	206.95	170.57					18.94	8.42		L
	4-WireVG Loop used with 4-wire VG Interoffice Transport		~			05 70	2000.05	470.57					10.04	0.40		
	Combination - Zone 2 4-WireVG Loop used with 4-wire VG Interoffice Transport		2	UNCVX	UEAL4	25.70	206.95	170.57		-			18.94	8.42	-	
	Combination - Zone 3		3	UNCVX	UEAL4	40.86	206.95	170.57					18.94	8.42		
	Interoffice Transport - Dedicated - 4-wire VG combination - Per				-											
	Mile Per Month			UNCVX	1L5XX	0.0222										L
	Interoffice Transport - Dedicated - 4- Wire Voice Grade					47.07	70.04	00.00					10.01	40.04		
	combination - Facility Termination per month Nonrecurring Currently Combined Network Elements Switch -As-			UNCVX	U1TV4	17.07	79.61	36.08					18.94	18.94		
	Is Charge			UNCVX	UNCCC		12.97	11.27					45.46	15.72		
DS3 DI	GITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFIC	E TRA	SPOR		0.1000		12.01						10.10	10112		
	High Capacity Unbundled Local Loop - DS3 combination - Per															
	Mile per month			UNC3X	1L5ND	8.90										L
	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month			UNC3X	UE3PX	390.34	639.50	426.40					37.55	37.55	18.03	18.03
	Interoffice Transport - Dedicated - DS3 - Per Mile per month			UNC3X	1L5XX	2.72	639.50	420.40					37.55	37.55	16.03	16.03
	Interoffice Transport - Dedicated - DS3 combination - Facility				120/01	2.72										
	Termination per per month			UNC3X	U1TF3	788.00	198.45	153.15					37.55	37.55	18.03	18.03
	Nonrecurring Currently Combined Network Elements Switch -As-															
ete4 r	Is Charge DIGITAL EXTENDED LOOP WITH DEDICATED STS1 INTEROFF			UNC3X	UNCCC		12.97	11.27					45.46	15.72		┣────
51511	High Capacity Unbundled Local Loop - STS1 combination - Per		ANSP													
	Mile per month			UNCSX	1L5ND	8.90										
	High Capacity Unbundled Local Loop - STS1 combination -															
	Facility Termination per month			UNCSX	UDLS1	421.59	639.50	426.40					37.55	37.55	18.03	18.03
	Interoffice Transport - Dedicated - STS1 combination - Per Mile					0.70										1
	per month Interoffice Transport - Dedicated - STS1 combination - Facility			UNCSX	1L5XX	2.72					-					
	Termination per month			UNCSX	U1TFS	783.63	198.45	449.91					37.55	37.55	18.03	18.03
	Nonrecurring Currently Combined Network Elements Switch -As-												200	200		
	Is Charge			UNCSX	UNCCC		12.97	11.27					45.46	15.72		
	E ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPOR First 2-Wire ISDN Loop in a DS1 Interoffice Combination	RT (EEL)													
2-WIRE					1					1	1			1		1

UNBUNDLE	D NETWORK ELEMENTS - Georgia												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)		
	First 0 Wire ICDN Loss is a DC4 Interation Combination				-		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 2		2	UNCNX	U1L2X	25.27	233.38	180.38					18.94	8.42		1
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination		2	ONONA	UTLEX	23.21	200.00	100.00					10.34	0.42		
	Transport - Zone 3		3	UNCNX	U1L2X	40.17	233.38	180.38					18.94	8.42		1
	Interoffice Transport - Dedicated - DS1 combination - Per Mile			UNC1X	1L5XX	0.4523										
	Interoffice Transport - Dedicated - DS1 combintion - Facility Termination per month			UNC1X	U1TF1	78.47	194.63	141.51					33.63	27.49	19.88	11.85
	Channelization - Channel System DS1 to DS0 combination - per month			UNC1X	MQ1	126.22										
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System															1
	combination - per month Additional 2-wire ISDN Loop in same DS1Interoffice Transport			UNCNX	UC1CA	3.37	12.02	8.66	├ ─── ├				33.63	27.49	19.88	11.85
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1 Additional 2-wire ISDN Loop in some DS1Interoffice Transport		1	UNCNX	U1L2X	21.89	233.38	180.38	ļļ				18.94	8.42		
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2		2	UNCNX	U1L2X	25.27	233.38	180.38					18.94	8.42		ļ'
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3		3	UNCNX	U1L2X	40.17	233.38	180.38					18.94	8.42		
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combintaion- per month			UNCNX	UC1CA	3.37	12.02	8.66					33.63	27.49	19.88	11.85
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNC1X	UNCCC		12.97	11.27					45.46	15.72		
4-WIRE	DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 IN	TEROF	FICE T	RANSPORT (EEL)			-									1
	First DS1 Loop in STS1 Interoffice Transport Combination -															Í
	Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination -		1	UNC1X	USLXX	55.53	443.20	138.69					18.94	8.42		
	Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination -		2	UNC1X	USLXX	64.13	443.20	138.69					18.94	8.42		
	Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile		3	UNC1X	USLXX	101.93	443.20	138.69					18.94	8.42		
	Per Month Interoffice Transport - Dedicated - STS1 combination - Facility			UNCSX	1L5XX	2.72										
	Termination			UNCSX	U1TFS	783.63	198.45	449.91					37.55	37.55	18.08	18.03
	STS1 to DS1 Channel System conbination per month			UNCSX	MQ3	182.04	196.66	204.61					37.55	37.55	18.08	18.03
	DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	11.02	12.02	8.66					37.55	37.55	18.08	18.03
	Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1		1	UNC1X	USLXX	55.53	443.20	138.69					18.94	8.42		
	Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 2		2	UNC1X	USLXX	64.13	443.20	138.69					18.94	8.42		
	Additional DS1Loop in STS1 Interoffice Transport Combination -															1
	Zone 3		3	UNC1X UNC1X	USLXX UC1D1	101.93 11.02	443.20 12.02	138.69	<u>├</u>				18.94 18.94	8.42 8.42		
	DS3 Interface Unit (DS1 COCI) combination per month Nonrecurring Currently Combined Network Elements Switch -As-				00101	11.02	12.02	8.66					10.94	0.42		
	Is Charge			UNCSX	UNCCC		12.97	11.27					45.46	15.72		1
4-WIRE	56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTERO	FFICE 1	RANS	PORT (EEL)												[
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport															
	Combination - Zone 1 4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport		1	UNCDX	UDL56	25.75	384.56	241.20					18.94	8.42		
	Combination - Zone 2 4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport		2	UNCDX	UDL56	29.74	384.56	241.20					18.94	8.42		
	Combination - Zone 3 Interoffice Transport - Dedicated - 4-wire 56 kbps combination -		3	UNCDX	UDL56	47.27	384.56	241.20					18.94	8.42		
	Per Mile Interoffice Transport - Dedicated - 4-wire 56 kbps combination -		<u> </u>	UNCDX	1L5XX	0.0222			-							
	Facility Termination		ļ	UNCDX	U1TD5	16.45	147.07	111.75					33.63	27.49	19.88	11.85
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge				UNCCC		12.97	11.27					45.46	15.72		ļ
4-WIRE	64 KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTERO	FFICE 1	KANS	PORT (EEL)												l
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 1		1	UNCDX	UDL64	25.75	348.55	241.20					18.94	8.42		

UNBUN	IDLE	D NETWORK ELEMENTS - Georgia												Attachment:			ibit: B
CATEGO	RY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
							Rec	Nonrec		Nonrecurring					Rates(\$)		
		4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Combination - Zone 2		2	UNCDX	UDL64	29.74	348.55	241.20					18.94	8.42		
		4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport		-	011027	00201	2011 1	010.00	211120					10.01	0.12		
		Combination - Zone 3		3	UNCDX	UDL64	47.27	348.55	241.20					18.94	8.42		
		Interoffice Transport - Dedicated - 4-wire 64 kbps combination -															
		Per Mile Interoffice Transport - Dedicated - 4-wire 64 kbps combination -			UNCDX	1L5XX	0.0222										
		Facility Termination			UNCDX	U1TD6	16.45	147.07	111.75					33.63	27.49	19.88	11.85
		Nonrecurring Currently Combined Network Elements Switch -As-															
		Is Charge			UNCDX	UNCCC		12.97	11.27					45.46	15.72		
		ETWORK ELEMENTS															
		used as a part of a currently combined facility, the non-recurrules as ordinarily combined network elements in All States, the															
		urring Currently Combined Network Elements III All States, in						063 1101.									
		Nonrecurring Currently Combined Network Elements Switch -As-	g-														
		Is Charge - 2 wire/4-Wire VG			UNCVX	UNCCC		12.97	11.27					18.94	18.94		
		Nonrecurring Currently Combined Network Elements Switch -As-						10.07						10.01	10.01		
		Is Charge - 56/64 kbps Nonrecurring Currently Combined Network Elements Switch -As-			UNCDX	UNCCC		12.97	11.27					18.94	18.94		
		Is Charge - DS1			UNC1X	UNCCC		12.97	11.27					18.94	18.94		
		Nonrecurring Currently Combined Network Elements Switch -As-				0110000		12.57	11.27					10.34	10.34		
		Is Charge - DS3			UNC3X	UNCCC		12.97	11.27					18.94	18.94		
		Nonrecurring Currently Combined Network Elements Switch -As-															
⊢		Is Charge - STS1		_	UNCSX	UNCCC		12.97	11.27					18.94	18.94		
N	IOTE:	Local Channel - Dedicated Transport - minimum billing perioc Local Channel - Dedicated - 2-Wire Voice Grade	d - Belo	w DS3:	=one month, DS3 a UNCXV	ULDV2	months 13.91	272.07	60.43					18.94	18.94		
+		Local Channel - Dedicated - 2-Wire Voice Grade			UNCXV	ULDV2	14.99	272.07	60.43					18.94	18.94		
		Local Channel - Dedicated - DS1			UNC1X	ULDF1	38.36	356.15	312.89					10.04	10.04		
		Local Channel - Dedicated - DS3 - Per Mile per month			UNC3X	1L5NC	6.92										
		Local Channel - Dedicated - DS3 - Facility Termination			UNC3X	ULDF3	515.91	639.50	426.31					18.94	18.94		
		Local Channel - Dedicated - STS-1- Per Mile per month			UNCSX	1L5NC	6.92		100.01					10.01	10.01		
	Intion	Local Channel - Dedicated - STS-1 - Facility Termination al Features & Functions:			UNCSX	ULDFS	517.56	639.50	426.31					18.94	18.94		
		PLEXERS															
		Channelization - DS1 to DS0 Channel System			UXTD1	MQ1	126.22	198.22	123.59					14.75	6.55	10.70	
		OCU-DP COCI (data) - DS1 to DS0 Channel System - per															
		month (2.4-64kbs)			UDL	1D1DD	1.86	12.02	8.66					14.75	6.55	10.70	
		2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per month			UDN	UC1CA	3.37	12.02	8.66					14.75	6.55	10.70	
-+		Voice Grade COCI - DS1 to DS0 Channel System - per month			UEA	1D1VG	3.37	12.02	8.66					14.75	6.55	10.70	
-+		DS3 to DS1 Channel System per month			UXTD3	MQ3	182.04	265.91	188.78					14.75	6.55	10.70	
		STS1 to DS1 Channel System per month			UXTS1	MQ3	182.04	265.91	188.78					14.75	6.55	10.70	
		DS3 Interface Unit (DS1 COCI) used with Loop per month			USL	UC1D1	11.02	12.02	8.66					14.75	6.55	10.70	
		DS3 Interface Unit (DS1 COCI) used with Local Channel per						10.00	0.00						0.55	10 70	
 		month DS3 Interface Unit (DS1 COCI) used with Interoffice Channel			ULDD1	UC1D1	11.02	12.02	8.66					14.75	6.55	10.70	
		per month			U1TD1	UC1D1	11.02	12.02	8.66					14.75	6.55	10.70	
s		op Feeder													2.00		
		Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Statewide			UNC1X	USBFG	79.30	203.69	128.76	124.09	34.80						
		Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 1		1	UNC1X	USBFG											
\rightarrow		Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 2 Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 3		2	UNC1X UNC1X	USBFG USBFG											
\rightarrow		Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 3 Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 4		3	UNC1X UNC1X	USBFG											-
UNBUND		OCAL EXCHANGE SWITCHING(PORTS)		-							1						
E	xchan	ge Ports															
IN IN	OTE:	Although the Port Rate includes all available features in GA, H VOICE GRADE LINE PORT RATES (RES)	Y, LA	& TN, t	he desired features	will need to b	e ordered usin	g retail USOCs	6				_				
				1	1	1											1

UNBUNDL	ED NETWORK ELEMENTS - Georgia												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring		0.01450	001111		Rates(\$)	001111	001111
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Exchange Ports - 2-Wire Analog Line Port with Caller ID - Res.			UEPSR	UEPRC	1.85	17.16	17.16					18.94	8.42		1 1
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Res.			UEPSR	UEPRO	1.85	17.16	17.16					18.94	8.42		
	Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID (LUM)			UEPSR	UEPAP	1.85		17.16					18.94	8.42		
	Exchange Ports - 2-Wire Voice Georgia basic dialing port															í l
	without Caller ID 2-Wire voice unbundled Georgia basic dialing port for use with			UEPSR	UEPWC	1.85	17.16	17.16					18.94	8.42		
	Caller ID - res			UEPSR	UEPWQ	1.85	17.16	17.16					18.94	8.42		ļ′
	2-Wire voice unbundled Georgia basic dialing port - outgoing only			UEPSR	UEPWR	1.85	17.16	17.16					18.94	8.42		
	2-Wire voice unbundled Low Usage Line Port without Caller ID Capability			UEPSR	UEPRT	1.85	17.16	17.16					18.94	8.42		1 1
	Subsequent Activity			UEPSR	USASC	0.00	0.00	0.00					18.94	8.42		ll
FEAT	URES															1
	All Available Vertical Features			UEPSR	UEPVF	0.00	0.00	0.00					18.94	8.42		ļ'
2-WI	RE VOICE GRADE LINE PORT RATES (BUS) Exchange Ports - 2-Wire Analog Line Port without Caller ID -	-														j/
	Bus Exchange Ports - 2-Wire VG unbundled Line Port with			UEPSB	UEPBL	1.85	17.16	17.16					18.94	8.42		ļ!
	unbundled port with Caller+E484 ID - Bus. Exchange Ports - 2-Wire Voice Georgia Business Basic Dialing			UEPSB	UEPBC	1.85	17.16	17.16					18.94	8.42		
	Port, with Caller ID capability			UEPSB	UEPWP	1.85	17.16	17.16					18.94	8.42		ļ
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Bus.			UEPSB	UEPBO	1.85	17.16	17.16					18.94	8.42		
	Exhange Ports - 2-Wire VG unbundled incoming only port with Caller ID - Bus			UEPSB	UEPB1	1.85	17.16	17.16					18.94	8.42		
	Exchange Ports - 2-Wire Voice Georgia Business Dialing Plan without Caller ID			UEPSB	UEPWD	1.85	17.16	17.16					18.94	8.42		
	2-Wire voice unbundled Incoming Only Port without Caller ID					4.05	17.10	47.40					40.04	0.40		
	Capability Subsequent Activity			UEPSB UEPSB	UEPBE USASC	1.85 0.00	17.16 0.00	17.16					18.94 18.94	8.42 8.42		1
FEAT	URES			UEF3B	USASC	0.00	0.00	0.00					10.94	0.42		i
	All Available Vertical Features			UEPSB	UEPVF	0.00	0.00	0.00					18.94	8.42		1
EXCH	IANGE PORT RATES (DID & PBX)															
	2-Wire VG Unbundled 2-Way PBX Trunk - Res			UEPSE	UEPRD	1.85	17.16	17.16					18.94	8.42		ļ
	2-Wire voice unbundled Georgia extended dialing port, PBX 1- Way Outdial Trunk			UEPSE	UEPPO	1.85	17.16	17.16					18.94	8.42		
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus			UEPSP	UEPPC	1.85	17.16	17.16					18.94	8.42		i
	2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus			UEPSP	UEPPO	1.85	17.16	17.16					18.94	8.42		
	2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus			UEPSP	UEPP1	1.85	17.16	17.16					18.94	8.42		I
	2-Wire Analog Long Distance Terminal PBX Trunk - Bus			UEPSP	UEPLD	1.85	17.16	17.16					18.94	8.42		
	2-Wire Voice Unbundled PBX LD Terminal Ports 2-Wire Vice Unbundled 2-Way PBX Usage Port			UEPSP UEPSP	UEPLD UEPXA	1.85 1.85	17.16 17.16	17.16 17.16					18.94 18.94	8.42 8.42		l
	2-Wire Vice Unbundled 2-Way PBX Usage Port 2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPSP	UEPXA	1.85	17.16	17.16					18.94	8.42		i
	2-Wire Voice Unbundled PBX Toll Terminal Totel Ports			UEPSP	UEPXC	1.85	17.16	17.16					18.94	8.42		<u> </u>
	2-Wire Voice Unbundled PBX LD DDD Terminal Switchboard Port	1		UEPSP	UEPXD	1.85	17.16	17.16					18.94	8.42		
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port			UEPSP	UEPXE	1.85	17.16	17.16					18.94	8.42		
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port		1	UEPSP	UEPXL	1.85	17.16	17.16			İ		18.94	8.42		
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port			UEPSP	UEPXM	1.85	17.16	17.16					18.94	8.42		
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port			UEPSP	UEPXO	1.85	17.16	17.16					18.94	8.42		
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPSP	UEPXO	1.85	17.16	17.16					18.94	8.42		
	2-Wire voice unbundled Georgia basic dialing port - 1-Way	1			02.70	1.00	17.10	17.10	1		t		10.04	0.42		
	Oudial Trunk			UEPSP	UEPWS	1.85	17.16	17.16					18.94	8.42		<u> </u>

	LED NETWORK ELEMENTS - Georgia												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrec			g Disconnect		0		Rates(\$)		
						1.00	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire voice unbundled Georgia basic dialing port - 2-Way Trunk			UEPSP	UEPWT	1.85	17.16	17.16					18.94	8.42		
	2-Wire voice unbundled Georgia basic dialing port - 2-way PBX			ULFSF	OLFWI	1.00	17.10	17.10			-		10.54	0.42		
	Trunk			UEPSP	UEPPQ	1.85	17.16	17.16					18.94	8.42		
	2-Wire voice unbundled Georgia basic dialing port - PBX LD															
	Terminal Ports			UEPSP	UEPPS	1.85	17.16	17.16					18.94	8.42		
	2-Wire voice unbundled Georgia basic dialing port - PBX Toll Terminal Ports			UEPSP	UEPPT	1.85	17.16	17.16					18.94	8.42		
	2-Wire voice unbundled Georgia basic dialing port - PBX LD			UEFSF	UEPPI	1.00	17.10	17.10		+	-		10.94	0.42		
	DDD Terminal Port			UEPSP	UEPPU	1.85	17.16	17.16					18.94	8.42		
	2-Wire voice unbundled Georgia basic dialing port - PBX LD															
	Terminal Switchboard Port		<u> </u>	UEPSP	UEPPV	1.85	17.16	17.16		1			18.94	8.42		
	2-Wire voice unbundled Georgia basic dialing port - PBX LD Terminal Switchboard DDD Capable Port		1	UEPSP	UEPPW	1.85	17.16	17.16		1			18.94	8.42		
	Subsequent Activity	-		UEPSP	USASC	0.00	0.00	0.00					18.94	8.42		
FEA	TURES				00400	0.00	0.00	0.00		1			10.34	0.42		
	All Available Vertical Features			UEPSP UEPSE	UEPVF	0.00	0.00	0.00					18.94	8.42		
EXC	HANGE PORT RATES (COIN)															
	Exchange Ports - Coin Port					2.05	17.16	17.16					18.94	8.42		
	E: Transmission/usage charges associated with POTS circuit sv E: Access to B Channel or D Channel Packet capabilities will be													Boguoot Bro		
	D LOCAL EXCHANGE SWITCHING(PORTS)	avallar	bie oni	y through BFR/New	Business Re	quest Process.	Rates for the	раскет сарар	lities will be d	letermined via	the Bona Fid	e Request/	New Business	s Request Pro	cess.	
	HANGE PORT RATES															
	Exchange Ports - 2-Wire DID Port			UEPEX	UEPP2	11.35	61.91	61.91					19.99	19.99	19.99	19.99
	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID															
	capability					100.00	400.00						10.00	10.00	40.00	10.01
				UEPDD	UEPDD	120.80	108.38	60.88					19.99	19.99	19.99	19.99
	Exchange Ports - 2-Wire ISDN Port (See Notes below.)			UEPTX UEPSX	U1PMA	13.47	47.37	47.37					39.98	39.98	19.99	19.9
	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered	vitchod	116200	UEPTX UEPSX UEPTX UEPSX	U1PMA UEPVF	13.47 0.00	47.37 0.00	47.37 0.00	nission by B.C	bannols assoc	isted with 2	wire ISDN :	39.98		19.99	19.9
	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered TE: Transmission/usage charges associated with POTS circuit sv			UEPTX UEPSX UEPTX UEPSX will also apply to c	U1PMA UEPVF ircuit switche	13.47 0.00 d voice and/or	47.37 0.00 circuit switche	47.37 0.00 ed data transn					39.98	39.98		19.9
	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit sv E: Access to B Channel or D Channel Packet capabilities will be			UEPTX UEPSX UEPTX UEPSX will also apply to ci y through BFR/New	U1PMA UEPVF ircuit switche	13.47 0.00 d voice and/or	47.37 0.00 circuit switche	47.37 0.00 ed data transn					39.98	39.98		19.9
NOT	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered 'E: Transmission/usage charges associated with POTS circuit sv TE: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port	e availat		UEPTX UEPSX UEPTX UEPSX will also apply to c	U1PMA UEPVF ircuit switche Business Re	13.47 0.00 d voice and/or quest Process.	47.37 0.00 circuit switche Rates for the	47.37 0.00 ed data transn packet capabi					39.98	39.98		19.9
	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit sv Te: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY	e availal		UEPTX UEPSX UEPTX UEPSX will also apply to ci y through BFR/New UEPTX UEPSX	U1PMA UEPVF ircuit switche Business Re U1UMA	13.47 0.00 d voice and/or quest Process. 0.00	47.37 0.00 circuit switche Rates for the 0.00	47.37 0.00 ed data transn packet capabi 0.00					39.98 ports. New Business	39.98 s Request Pro		19.9
	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered TE: Transmission/usage charges associated with POTS circuit sv TE: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port SUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE	e availal		UEPTX UEPSX UEPTX UEPSX will also apply to ci y through BFR/New UEPTX UEPSX UEPEX	U1PMA UEPVF ircuit switche Business Re U1UMA UEPEX	13.47 0.00 d voice and/or quest Process. 0.00 163.16	47.37 0.00 circuit switche Rates for the 0.00 186.80	47.37 0.00 ed data transn packet capabi 0.00 186.80					39.98 ports. New Business 37.88	39.98 s Request Pro 37.88		19.9
	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit sv Te: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY	e availal		UEPTX UEPSX UEPTX UEPSX will also apply to ci y through BFR/New UEPTX UEPSX	U1PMA UEPVF ircuit switche Business Re U1UMA	13.47 0.00 d voice and/or quest Process. 0.00	47.37 0.00 circuit switche Rates for the 0.00	47.37 0.00 ed data transn packet capabi 0.00					39.98 ports. New Business	39.98 s Request Pro		19.9
	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered TE: Transmission/usage charges associated with POTS circuit sv TE: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY SUNDLED PORT WITH REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res	e availal		UEPTX UEPSX UEPTX UEPSX will also apply to ci y through BFR/New UEPTX UEPSX UEPTX UEPTX UEPVR	U1PMA UEPVF incuit switche Business Re U1UMA UEPEX UEPAC	13.47 0.00 d voice and/or quest Process. 0.00 163.16 1.85	47.37 0.00 circuit switche Rates for the 0.00 186.80 17.16	47.37 0.00 ed data transn packet capabi 0.00 186.80 17.16					39.98 ports. New Business 37.88 18.94	39.98 Request Pro 37.88 8.42		19.9
	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit sv E: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port SUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res Unbundled Remote Call Forwarding Service, Local Calling - Res	e availal		UEPTX UEPSX UEPTX UEPSX will also apply to ci y through BFR/New UEPTX UEPSX UEPEX UEPEX UEPVR UEPVR	U1PMA UEPVF ircuit switche Business Re U1UMA UEPEX UEPEX UERAC UERAC	13.47 0.00 d voice and/or quest Process. 0.00 163.16 1.85 1.85	47.37 0.00 circuit switche Rates for the 0.00 186.80 17.16	47.37 0.00 ed data transm packet capabi 0.00 186.80 17.16 17.16					39.98 ports. New Business 37.88 18.94 18.94	39.98 Request Pro 37.88 8.42 8.42		
	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered TE: Transmission/usage charges associated with POTS circuit sv TE: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY SUNDLED PORT WITH REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling, Res	e availal		UEPTX UEPSX UEPTX UEPSX will also apply to ci y through BFR/New UEPTX UEPSX UEPTX UEPTX UEPVR	U1PMA UEPVF incuit switche Business Re U1UMA UEPEX UEPAC	13.47 0.00 d voice and/or quest Process. 0.00 163.16 1.85	47.37 0.00 circuit switche Rates for the 0.00 186.80 17.16	47.37 0.00 ed data transn packet capabi 0.00 186.80 17.16					39.98 ports. New Business 37.88 18.94	39.98 Request Pro 37.88 8.42		
	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit sv E: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port SUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res	e availal		UEPTX UEPSX UEPTX UEPSX will also apply to ci y through BFR/New UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR	U1PMA UEPVF ircuit switche Business Re U1UMA UEPEX UERAC UERAC UERLC UERTE	13.47 0.00 d voice and/or quest Process. 0.00 163.16 1.85 1.85 1.85	47.37 0.00 circuit switch Rates for the 0.00 186.80 17.16 17.16 17.16	47.37 0.00 ed data transn packet capabi 0.00 186.80 17.16 17.16 17.16					39.98 ports. New Business 37.88 18.94 18.94 18.94	39.98 Request Pro 37.88 8.42 8.42 8.42 8.42		
	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered Transmission/usage charges associated with POTS circuit sv Te: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port SUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY SUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY SUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Local Calling . Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res Unbundled Remote Call Forwarding Service - Conversion -	e availal		UEPTX UEPSX UEPTX UEPSX will also apply to ci y through BFR/New UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR	U1PMA UEPVF ircuit switche Business Re U1UMA UEPEX UERAC UERAC UERLC UERTE UERTR	13.47 0.00 d voice and/or quest Process. 0.00 163.16 1.85 1.85 1.85	47.37 0.00 circuit switcher Rates for the 0.00 186.80 17.16 17.16 17.16 17.16	47.37 0.00 ed data transm packet capabi 0.00 186.80 17.16 17.16 17.16 17.16					39.98 ports. New Business 37.88 18.94 18.94 18.94 18.94	39.98 s Request Pro 37.88 8.42 8.42 8.42 8.42 8.42	Cess.	
	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered Te: Transmission/usage charges associated with POTS circuit sy Te: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY SUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is	e availal		UEPTX UEPSX UEPTX UEPSX will also apply to ci y through BFR/New UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR	U1PMA UEPVF ircuit switche Business Re U1UMA UEPEX UERAC UERAC UERLC UERTE	13.47 0.00 d voice and/or quest Process. 0.00 163.16 1.85 1.85 1.85	47.37 0.00 circuit switch Rates for the 0.00 186.80 17.16 17.16 17.16	47.37 0.00 ed data transn packet capabi 0.00 186.80 17.16 17.16 17.16					39.98 ports. New Business 37.88 18.94 18.94 18.94	39.98 Request Pro 37.88 8.42 8.42 8.42 8.42		
	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered Transmission/usage charges associated with POTS circuit sv Te: Transmission/usage charges associated with POTS circuit sv Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY SUNDLED PORT with REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res Hobundled Remote Call Forwarding Service - Conversion - Switch-as-is	e availal		UEPTX UEPSX UEPTX UEPSX will also apply to ci y through BFR/New UEPTX UEPSX UEPTX UEPSX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	U1PMA UEPVF ircuit switche Business Re U1UMA UEPEX UERAC UERAC UERLC UERTE UERTR UERTR	13.47 0.00 d voice and/or quest Process. 0.00 163.16 1.85 1.85 1.85	47.37 0.00 circuit switche Rates for the 0.00 186.80 17.16 17.16 17.16 17.16 17.16 17.16 2.01	47.37 0.00 ad data transm packet capabi 0.00 186.80 17.16 17.16 17.16 17.16 17.16 0.31					39.98 ports. New Business 37.88 18.94 18.94 18.94 18.94	39.98 s Request Pro 37.88 8.42 8.42 8.42 8.42 8.42	Cess.	
NOT	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit sv E: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC)	e availal		UEPTX UEPSX UEPTX UEPSX will also apply to ci y through BFR/New UEPTX UEPSX UEPEX UEPVR UEPVR UEPVR UEPVR UEPVR	U1PMA UEPVF ircuit switche Business Re U1UMA UEPEX UERAC UERAC UERLC UERTE UERTR	13.47 0.00 d voice and/or quest Process. 0.00 163.16 1.85 1.85 1.85	47.37 0.00 circuit switcher Rates for the 0.00 186.80 17.16 17.16 17.16 17.16	47.37 0.00 ed data transm packet capabi 0.00 186.80 17.16 17.16 17.16 17.16					39.98 ports. New Business 37.88 18.94 18.94 18.94 18.94	39.98 s Request Pro 37.88 8.42 8.42 8.42 8.42 8.42	Cess.	
NOT	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered Transmission/usage charges associated with POTS circuit sv Te: Transmission/usage charges associated with POTS circuit sv Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY SUNDLED PORT with REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res Hobundled Remote Call Forwarding Service - Conversion - Switch-as-is	e availal		UEPTX UEPSX UEPTX UEPSX will also apply to ci y through BFR/New UEPTX UEPSX UEPTX UEPSX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	U1PMA UEPVF ircuit switche Business Re U1UMA UEPEX UERAC UERAC UERLC UERTE UERTR UERTR	13.47 0.00 d voice and/or quest Process. 0.00 163.16 1.85 1.85 1.85	47.37 0.00 circuit switche Rates for the 0.00 186.80 17.16 17.16 17.16 17.16 17.16 17.16 2.01	47.37 0.00 ad data transm packet capabi 0.00 186.80 17.16 17.16 17.16 17.16 17.16 0.31					39.98 ports. New Business 37.88 18.94 18.94 18.94 18.94	39.98 s Request Pro 37.88 8.42 8.42 8.42 8.42 8.42	Cess.	
NOT	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit sv E: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC)	e availal		UEPTX UEPSX UEPTX UEPSX will also apply to ci y through BFR/New UEPTX UEPSX UEPTX UEPSX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	U1PMA UEPVF ircuit switche Business Re U1UMA UEPEX UERAC UERAC UERLC UERTE UERTR UERTR	13.47 0.00 d voice and/or quest Process. 0.00 163.16 1.85 1.85 1.85	47.37 0.00 circuit switche Rates for the 0.00 186.80 17.16 17.16 17.16 17.16 17.16 17.16 2.01	47.37 0.00 ad data transm packet capabi 0.00 186.80 17.16 17.16 17.16 17.16 17.16 0.31					39.98 ports. New Business 37.88 18.94 18.94 18.94 18.94	39.98 s Request Pro 37.88 8.42 8.42 8.42 8.42 8.42	Cess.	
NOT	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered Transmission/usage charges associated with POTS circuit sv Te: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port SUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY SUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY SUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) SUNDLED REMOTE CALL FORWARDING - Bus	e availab		UEPTX UEPSX UEPTX UEPSX will also apply to ci y through BFR/New UEPTX UEPSX UEPTX UEPSX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	U1PMA UEPVF ircuit switche Business Re U1UMA UEPEX UERAC UERLC UERLC UERTE UERTR USAC2 USAC2 USACC	13.47 0.00 d voice and/or quest Process. 0.00 163.16 1.85 1.85 1.85 1.85 1.85 1.85	47.37 0.00 circuit switcher Rates for the 0.00 186.80 17.16 17.16 17.16 17.16 17.16 2.01 2.01	47.37 0.00 ad data transm packet capabi 0.00 186.80 17.16 17.16 17.16 17.16 0.31 0.31					39.98 oorts. New Business 37.88 18.94 18.94 18.94 33.67 18.94 18.94	39.98 Request Pro 37.88 37.88 8.42 8.42 8.42 7.88 8.42 8	Cess.	3.91
NOT UNB UNB NOT	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered Transmission/usage charges associated with POTS circuit sv Te: Transmission/usage charges associated with POTS circuit sv Te: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 2-Wire ISDN DS1 Port SUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY SUNDLED PORT with REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) SUNDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus	e availab		UEPTX UEPSX UEPTX UEPSX will also apply to ci y through BFR/New UEPTX UEPSX UEPTX UEPSX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	U1PMA UEPVF ircuit switche Business Re U1UMA UEPEX UERAC UERAC UERLC UERTE USAC2 USAC2 USACC UERAC UERAC UERAC	13.47 0.00 d voice and/or guest Process. 0.00 163.16 1.85 1.85 1.85 1.85 1.85 1.85 1.85 1.85	47.37 0.00 circuit switch- Rates for the 0.00 186.80 17.16 17.16 17.16 17.16 2.01 2.01 17.16 17.16	47.37 0.00 ad data transm packet capabi 0.00 186.80 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16					39.98 oorts. New Business 37.88 18.94 18.94 18.94 18.94 33.67 18.94 18.94 18.94 18.94 18.94 18.94 18.94 18.94	39.98 Request Pro 37.88 8.42 8.42 8.42 8.42 7.88 7.88 8.42	Cess.	
NOT UNB UNB NOT	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit sv E: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 2-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY SUNDLED PORT with REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus	e availab		UEPTX UEPSX UEPTX UEPSX will also apply to ci y through BFR/New UEPTX UEPSX UEPTX UEPSX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	U1PMA UEPVF ircuit switche Business Re U1UMA UEPEX UERAC UERAC UERLC UERTR USAC2 USAC2 USAC2 UERAC UERAC UERAC	13.47 0.00 d voice and/or quest Process. 0.00 163.16 1.85 1.85 1.85 1.85 1.85 1.85 1.85 1.85	47.37 0.00 circuit switche Rates for the 0.00 186.80 17.16 17.16 17.16 17.16 2.01 2.01 2.01 17.16 17.16	47.37 0.00 ad data transm packet capabi 0.00 186.80 17.16 17.16 17.16 17.16 0.31 0.31 17.16 17.16 17.16					39.98 oorts. New Business 37.88 18.94 18.94 18.94 18.94 33.67 18.94	39.98 Request Pro 37.88 8.42 8.42 8.42 8.42 7.88 7.88 8.42	Cess.	
NOT	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit sv E: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) BUNDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus	e availab		UEPTX UEPSX UEPTX UEPSX will also apply to ci y through BFR/New UEPTX UEPSX UEPTX UEPSX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	U1PMA UEPVF ircuit switche Business Re U1UMA UEPEX UERAC UERAC UERLC UERTE USAC2 USAC2 USACC UERAC UERAC UERAC	13.47 0.00 d voice and/or guest Process. 0.00 163.16 1.85 1.85 1.85 1.85 1.85 1.85 1.85 1.85	47.37 0.00 circuit switch- Rates for the 0.00 186.80 17.16 17.16 17.16 17.16 2.01 2.01 17.16 17.16	47.37 0.00 ad data transm packet capabi 0.00 186.80 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16					39.98 oorts. New Business 37.88 18.94 18.94 18.94 18.94 33.67 18.94 18.94 18.94 18.94 18.94 18.94 18.94 18.94	39.98 Request Pro 37.88 8.42 8.42 8.42 8.42 7.88 7.88 8.42	Cess.	
NOT UNB UNB NOT	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered Transmission/usage charges associated with POTS circuit sv Te: Transmission/usage charges associated with POTS circuit sv Te: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 2-Wire ISDN DS1 Port BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY SUNDLED PORT with REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) SUNDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote	e availab		UEPTX UEPSX UEPTX UEPSX will also apply to ci y through BFR/New UEPTX UEPSX UEPTX UEPSX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR	U1PMA UEPVF ircuit switche Business Re U1UMA UEPEX UERAC UERAC UERLC UERTR USAC2 USAC2 USAC2 UERAC UERAC UERAC	13.47 0.00 d voice and/or quest Process. 0.00 163.16 1.85 1.85 1.85 1.85 1.85 1.85 1.85 1.85	47.37 0.00 circuit switche Rates for the 0.00 186.80 17.16 17.16 17.16 17.16 2.01 2.01 2.01 17.16 17.16	47.37 0.00 ad data transm packet capabi 0.00 186.80 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16					39.98 oorts. New Business 37.88 18.94 18.94 18.94 18.94 33.67 18.94	39.98 Request Pro 37.88 8.42 8.42 8.42 8.42 7.88 7.88 8.42	Cess.	
NOT UNB UNB Non UNB	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit sv E: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) BUNDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus	e availab		UEPTX UEPSX UEPTX UEPSX will also apply to ci y through BFR/New UEPTX UEPSX UEPTX UEPSX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVB UEPVB UEPVB UEPVB	U1PMA UEPVF ircuit switche Business Re U1UMA UEPEX UERAC UERAC UERLC UERTE USAC2 USAC2 USAC2 UERAC UERAC UERAC UERAC UERAC	13.47 0.00 d voice and/or guest Process. 0.00 163.16 1.85 1.85 1.85 1.85 1.85 1.85 1.85 1.85	47.37 0.00 circuit switch- Rates for the 0.00 186.80 17.16 17.16 17.16 17.16 2.01 2.01 2.01 17.16 17.16 17.16 17.16	47.37 0.00 ad data transm packet capabi 0.00 186.80 17.16 17.16 17.16 17.16 0.31 0.31 17.16 17.16 17.16					39.98 oorts. New Business 37.88 18.94 18.94 18.94 18.94 33.67 18.94 18.94 18.94 18.94 18.94 18.94 18.94	39.98 39.98 37.88 37.88 8.42 8.42 8.42 7.88 7.88 8.42 8.42 8.42 8.42 8.42 8.42 8.42 8.42 8.42 8.42 8.42	Cess.	
NOT UNB UNB Non UNB	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered Transmission/usage charges associated with POTS circuit sv Transmission/usage charges associated with POTS circuit sv Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 2-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY SUNDLED PORT with REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Area Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) SUNDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remot	e availab		UEPTX UEPSX UEPTX UEPSX will also apply to ci ythrough BFR/New UEPTX UEPSX UEPTX UEPSX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVB UEPVB UEPVB UEPVB	U1PMA UEPVF ircuit switche Business Re U1UMA UEPEX UERAC UERLC UERTE UERTR USAC2 USAC2 USAC2 UERAC UERAC UERAC UERAC UERAC UERTE UERTR UERTR	13.47 0.00 d voice and/or guest Process. 0.00 163.16 1.85 1.85 1.85 1.85 1.85 1.85 1.85 1.85	47.37 0.00 circuit switch- Rates for the 0.00 186.80 17.16 17.16 17.16 17.16 2.01 2.01 2.01 17.16 17.16 17.16 17.16 17.16	47.37 0.00 ad data transm packet capabi 0.00 186.80 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16					39.98 39.98 30rts. New Business 18.94 18.94 18.94 18.94 18.94 18.94 18.94 18.94 18.94 18.94 18.94 18.94 18.94 18.94 18.94	39.98 39.98 37.88 37.88 8.42 8.42 8.42 7.88 7.88 8.42 8.42 8.42 8.42 8.42 8.42 8.42 8.42 8.42 8.42 8.42 8.42 8.42 8.42	CCESS.	3.9
NOT UNB UNB Non UNB	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit sv E: Access to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 2-Wire ISDN DS1 Port Unbundled Remote Call Forwarding Service, Area Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is	e availab		UEPTX UEPSX UEPTX UEPSX will also apply to ci y through BFR/New UEPTX UEPSX UEPTX UEPSX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVB UEPVB UEPVB UEPVB	U1PMA UEPVF ircuit switche Business Re U1UMA UEPEX UERAC UERAC UERLC UERTE USAC2 USAC2 USAC2 UERAC UERAC UERAC UERAC UERAC	13.47 0.00 d voice and/or guest Process. 0.00 163.16 1.85 1.85 1.85 1.85 1.85 1.85 1.85 1.85	47.37 0.00 circuit switch- Rates for the 0.00 186.80 17.16 17.16 17.16 17.16 2.01 2.01 2.01 17.16 17.16 17.16 17.16	47.37 0.00 ad data transm packet capabi 0.00 186.80 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16					39.98 oorts. New Business 37.88 18.94 18.94 18.94 18.94 33.67 18.94 18.94 18.94 18.94 18.94 18.94 18.94	39.98 39.98 37.88 37.88 8.42 8.42 8.42 7.88 7.88 8.42 8.42 8.42 8.42 8.42 8.42 8.42 8.42 8.42 8.42 8.42	Cess.	3.9
NOT UNB UNB Non UNB	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered E: Transmission/usage charges associated with POTS circuit sv Teases to B Channel or D Channel Packet capabilities will be Exchange Ports - 2-Wire ISDN Port - Channel Profiles Exchange Ports - 4-Wire ISDN DS1 Port UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY SUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED PORT with REMOTE CALL FORWARDING CAPABILITY BUNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion - Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC) BUNDLED REMOTE CALL FORWARDING - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Area Calling - Bus Unbundled Remote Call Forwarding Service, Local Calling - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service, InterLATA - Bus Unbundled Remote Call Forwarding Service - Conversion and Exception Local Calling - Recurring Unbundled Remote Call Forwarding Service - IntraLATA - Bus Unbundled Remote Call Forwarding Service - Conversion and Exception Local Calling	e availab		UEPTX UEPSX UEPTX UEPSX will also apply to ci ythrough BFR/New UEPTX UEPSX UEPTX UEPSX UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVR UEPVB UEPVB UEPVB UEPVB	U1PMA UEPVF ircuit switche Business Re U1UMA UEPEX UERAC UERLC UERTE UERTR USAC2 USAC2 USAC2 UERAC UERAC UERAC UERAC UERTE UERTR UERTR UERTR	13.47 0.00 d voice and/or guest Process. 0.00 163.16 1.85 1.85 1.85 1.85 1.85 1.85 1.85 1.85	47.37 0.00 circuit switch- Rates for the 0.00 186.80 17.16 17.16 17.16 17.16 2.01 2.01 2.01 17.16 17.16 17.16 17.16 17.16	47.37 0.00 ad data transm packet capabi 0.00 186.80 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16 17.16					39.98 39.98 30rts. New Business 18.94 18.94 18.94 18.94 18.94 18.94 18.94 18.94 18.94 18.94 18.94 18.94 18.94 18.94 18.94	39.98 39.98 37.88 37.88 8.42 8.42 8.42 7.88 7.88 8.42 8.42 8.42 8.42 8.42 8.42 8.42 8.42 8.42 8.42 8.42 8.42 8.42 8.42	CCESS.	

UNBUNDLED	D NETWORK ELEMENTS - Georgia					1						-	Attachment:			bit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge - Manual Sv Order vs. Electronic Disc Add
						D	Nonred	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	ice Switching (Port Usage)															
	End Office Switching Function, Per MOU					0.0016333										
	End Office Trunk Port - Shared, Per MOU					0.0001564										
	n Switching (Port Usage) (Local or Access Tandem) Tandem Switching Function Per MOU					0.0006757										
	Tandem Switching Function Per MOU Tandem Trunk Port - Shared, Per MOU					0.0008757										
	on Transport					0.0002120										
	Common Transport - Per Mile, Per MOU					0.000008										
	Common Transport - Facilities Termination Per MOU					0.0004152										
	ORT/LOOP COMBINATIONS - COST BASED RATES															
Cost Ba	ased Rates are applied where BellSouth is required by FCC ar	nd/or St	ate Co	mmission rule to p	rovide Unbun	dled Local Swit	ching or Swite	ch Ports.								
	s shall apply to the Unbundled Port/Loop Combination - Cos															
	fice and Tandem Switching Usage and Common Transport Us															
	at and additional Port nonrecurring charges apply to Not Curr	rently C	ombine	d Combos. For Cu	irrently Comb	ined Combos th	e nonrecurrin	g charges sha	II be those iden	tified in the N	lonrecurring	- Currently	/ Combined se	ections.		
	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)										<u> </u>		ļ			<u> </u>
	ort/Loop Combination Rates		_		-	40.50							-			
	2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2		1			12.59 14.26										
	2-Wire VG Loop/Port Combo - Zone 3		3			21.62										
	op Rates		5			21.02										
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPRX	UEPLX	10.80										
	2-Wire Voice Grade Loop (SL1) - Zone 2			UEPRX	UEPLX	12.47										
	2-Wire Voice Grade Loop (SL1) - Zone 3			UEPRX	UEPLX	19.83										
	Voice Grade Line Port Rates (Res)			-	-											
	2-Wire voice unbundled port - residence			UEPRX	UEPRL	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.9
	2-Wire voice unbundled port with Caller ID - res			UEPRX	UEPRC	1.79	22.14	15.25	8.45	3.91			37.06	7.88	11.17	3.9
	2-Wire voice unbundled port outgoing only - res			UEPRX	UEPRO	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.9
	2-Wire voice unbundles res, low usage line port with Caller ID (LUM)			UEPRX	UEPAP	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.9
	2-Wire voice unbundled Georgia basic dialing port without Caller ID capability - res			UEPRX	UEPWC	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.9
	2-Wire voice unbundled Georgia basic dialing port for use with Caller ID - res			UEPRX	UEPWQ	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.9
	2-Wire voice unbundled Georgia basic dialing port - outgoing only			UEPRX	UEPWR	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.9
	2-Wire voice unbundled Low Usage Line Port without Caller ID Capability			UEPRX	UEPRT	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.9
FEATU	RES															
	All Features Offered			UEPRX	UEPVF	0.00	0.00	0.00					33.67	7.88	11.17	3.9
		I	L		1.1.15.5								ļ			
	Local Number Portability (1 per port)			UEPRX	LNPCX	0.35							ļ			<u> </u>
	CURRING CHARGES (NRCs) - CURRENTLY COMBINED 2-Wire Voice Grade Loop / Line Port Combination - Conversion -	l			+								<u> </u>			
	2-wire voice Grade Loop / Line Port Combination - Conversion - Switch-as-is			UEPRX	USAC2		2.01	0.3108					33.67	7.88	11.17	3.9
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with change			UEPRX	USACC		2.01	0.3108					33.67	7.88	11.17	3.9
	ONAL NRCs				USACC		2.01	0.3108	 				33.67	7.88		
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent			UEPRX	USAS2	0.00	0.00	0.00					33.67	7.88	11.17	3.9
	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)				03/32	0.00	0.00	0.00					33.07	7.08	11.17	3.9
	voice GRADE LOOP WITH 2-WIRE LINE PORT (BUS) prt/Loop Combination Rates				+				 		1		 			
	2-Wire VG Loop/Port Combo - Zone 1	1	1		1	12.59							1			
	2-Wire VG Loop/Port Combo - Zone 1		2		-	14.26										
	2-Wire VG Loop/Port Combo - Zone 3		3		1	21.62							1		1	
	op Rates										1		1			1
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPBX	UEPLX	10.80							1			
	2-Wire Voice Grade Loop (SL1) - Zone 2			UEPBX	UEPLX	12.47										
	2-Wire Voice Grade Loop (SL1) - Zone 3	Γ	3	UEPBX	UEPLX	19.83										
2 Wire)	Voice Grade Line Port (Bus)	1	1													

UNBUNDLE	D NETWORK ELEMENTS - Georgia												Attachment:	2	Exhi	ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
	2-Wire voice unbundled port without Caller ID - bus			UEPBX	UEPBL	1.79	First 22.14	Add'l 15.25	First 8.45	Add'l 3.91	SOMEC	SOMAN	SOMAN 33.67	SOMAN 7.88	SOMAN 11.17	SOMAN 3.91
	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPBX	UEPBC	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled port with caller + 2404 lb - 543			UEPBX	UEPBO	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled incoming only port with Caller ID - Bus			UEPBX	UPEB1	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port, without Caller ID capability - bus			UEPBX	UEPWD	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port for use with															
	Caller ID - bus 2-Wire voice unbundled Incoming Only Port without Caller ID			UEPBX	UEPWP	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	Capability			UEPBX	UEPBE	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
LOCA	L NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPBX	LNPCX	0.35										┨
FEATU	All Features Offered			UEPBX	UEPVF	0.00	0.00	0.00					33.67	7.88	11.17	3.91
NONR	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED			OLFBA	OLF VI	0.00	0.00	0.00					33.07	7.00	11.17	3.91
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -				1 1							-	-		-	t
	Switch-as-is			UEPBX	USAC2		2.01	0.3108					33.67	7.88	11.17	3.91
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with change			UEPBX	USACC		2.01	0.3108								
ADDIT	TONAL NRCs			OLI DA	UUACC		2.01	0.5100								
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity			UEPBX	USAS2		0.00	0.00					33.67	7.88	11.17	3.91
2-WIR	E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)				00/102		0.00	0.00					00.07	7.00		0.01
	Port/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			12.59										
	2-Wire VG Loop/Port Combo - Zone 2		2			14.26										
	2-Wire VG Loop/Port Combo - Zone 3		3			21.62										
UNEL	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEPRG	UEPLX	10.80										
	2-Wire Voice Grade Loop (SL 1) - Zone 1		2	UEPRG	UEPLX	12.47										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEPRG	UEPLX	19.83										
2-Wir€	Voice Grade Line Port Rates (RES - PBX)															
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Res			UEPRG	UEPRD	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia extended dialing port, PBX 1-															
	Way Outdial Trunk			UEPRG	UEPPO	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
LOCA	L NUMBER PORTABILITY															
FEATU	Local Number Portability (1 per port)			UEPRG	LNPCP	3.15	0.00	0.00					33.67	7.88	11.17	3.91
FEAT	All Features Offered			UEPRG	UEPVF	0.00	0.00	0.00					33.67	7.88	11.17	3.91
NONR	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED			0EI NO	OEI VI	0.00	0.00	0.00					00.01	7.00		0.01
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As-Is			UEPRG	USAC2		2.01	0.3108					33.67	7.88	11.17	3.91
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -															
	Conversion - Switch with Change			UEPRG	USACC		2.01	0.3108					33.67	7.88	11.17	3.91
ADDIT	IONAL NRCs															
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activity			UEPRG	USAS2	0.00	0.00	0.00					33.67	7.88	11.17	3.91
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt			UEFRG	03A32	0.00	0.00	0.00								
	Group				_		14.64	14.64					19.99	19.99	19.99	19.99
	E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)															
	Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1		1			12.59										<u> </u>
	2-Wire VG Loop/Port Combo - Zone 1		2			14.26										+
	2-Wire VG Loop/Port Combo - Zone 3		3			21.62								1		1
UNE L	.oop Rates		Ĺ													
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEPPX	UEPLX	10.80										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEPPX	UEPLX	12.47										
1	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEPPX	UEPLX	19.83										<u> </u>

NBUNDLEI	D NETWORK ELEMENTS - Georgia												Attachment:	2	Exhi	ibit: B
TEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec	urring	Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
2-Wire	Voice Grade Line Port Rates (BUS - PBX)				-											
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPPX	UEPPC	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPPX	UEPPO	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	
	Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPPX	UEPP1	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPPX	UEPLD	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPPX	UEPXA	1.79	22.14	15.25	8.45	3.91			37.06	7.88	11.17	3.91
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPPX	UEPXB	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPPX	UEPXC	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD			UEPPX	UEPXD	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	Capable Port			UEPPX	UEPXE	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy							.0.20	0.40	0.01			00.07			5.0
	Administrative Calling Port			UEPPX	UEPXL	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.9
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy															
	Room Calling Port			UEPPX	UEPXM	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital													=		
	Discount Room Calling Port			UEPPX UEPPX	UEPXO UEPXS	1.79	22.14	15.25	8.45	3.91	-		33.67	7.88 7.88	11.17	3.91 3.91
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port 2-Wire voice unbundled Georgia basic dialing port - 1-Way			UEPPX	UEPAS	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.9
	Oudial Trunk			UEPPX	UEPWS	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port - 2-Way			0EI I X	OEI WO	1.75	22.14	10.20	0.40	0.01			00.07	7.00		0.01
	Trunk			UEPPX	UEPWT	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port - 2-way PBX															1
	Trunk			UEPPX	UEPPQ	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port - PBX LD															
	Terminal Ports			UEPPX	UEPPS	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port - PBX Toll			UEPPX	UEPPT	1.79	22.44	15.25	8.45	3.91			33.67	7.88	44.47	3.91
	Terminal Ports 2-Wire voice unbundled Georgia basic dialing port - PBX LD			UEPPX	UEPPT	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	DDD Terminal Port			UEPPX	UEPPU	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port - PBX LD			0EI I X	OEI I O	1.75	22.14	10.20	0.40	0.01			00.07	7.00		0.0
	Terminal Switchboard Port			UEPPX	UEPPV	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port - PBX LD															
	Terminal Switchboard DDD Capable Port			UEPPX	UEPPW	1.79	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
															11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port - PBX 2-Way Trunk			UEPPX	UEPPC	1.79	22.44	15.25	8.45	2.01			33.67	7.88	11.17	2.0
				UEPPX	UEPPC	1.79	22.14	15.25	8.40	3.91			33.67	7.88	11.17	3.9
LOOAL	Local Number Portability (1 per port)			UEPPX	LNPCP	3.15	0.00	0.00					33.67	7.88	11.17	3.9
FEATU				OEI TX		0.10	0.00	0.00					00.07	7.00		0.0
	All Features Offered			UEPPX	UEPVF	0.00	0.00	0.00					33.67	7.88	11.17	3.91
NONRE	CURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -															
	Conversion - Switch-As-Is			UEPPX	USAC2		2.01	0.3108					33.67	7.88	11.17	3.91
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with Change			UEPPX	USACC		2.01	0.3108					33.67	7.88	11.17	3.91
	ONAL NRCs			ULFFA	USACC		2.01	0.3108					33.0/	7.88	11.17	3.91
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -				+											
	Subsequent Activity			UEPPX	USAS2	0.00	0.00	0.00					33.67	7.88	11.17	3.91
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt		l													
	Group						14.64	14.64					19.99	19.99	19.99	19.99
	VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN POR	RT	L		l											
UNE Po	ort/Loop Combination Rates		1		+	10.00										
	2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2		1		+	12.69 14.36										
	2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3		2		+ +	21.72										<u> </u>
					1	21.12					-					
UNE Lo	oop Rates															

UNBUNDLE	D NETWORK ELEMENTS - Georgia	1		1								1	Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Charge -	Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)		
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPCO	UEPLX	12.47	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPCO	UEPLX	19.83										
2-Wire	Voice Grade Line Ports (COIN)															
	2-Wire Coin 2-Way with Operator Screening (GA)			UEPCO	UEPGC	1.89	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire Coin 2-Way with Operator Screening and Blocking: 011,															
	900/976, 1+DDD (GA)			UEPCO	UEP2G	1.89	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking (GA)			UEPCO	UEPGA	1.89	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire Coin 2-Way with Operator Screening and 900/976			ULFCO	ULFGA	1.09	22.14	13.23	0.45	3.91			33.07	7.00	11.17	3.91
	Blocking (GA)			UEPCO	UEPGB	1.89	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire Coin 2-Way with Operator Screening and Blocking:															
	900/976, 1+DDD, 011+, and Local (GA)			UEPCO	UEPCH	1.89	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire Coin Outward with Operator Screening and 011 Blocking (GA, KY, MS)					1.00	00.44	45.05	0.45	2.04			33.67	7.88	44.47	3.91
	2-Wire Coin Outward with Operator Screening and Blocking:			UEPCO	UEPRJ	1.89	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	900/976, 1+DDD, 011+, and Local (FL, GA)			UEPCO	UEPCQ	1.89	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire 2-Way Smartline with 900/976 (all states except LA)			UEPCO	UEPCK	1.89	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire Coin Outward Smartline with 900/976 (all states except															
	LA)			UEPCO	UEPCR	1.89	22.14	15.25	8.45	3.91			33.67	7.88	11.17	3.91
ADDIT	IONAL UNE COIN PORT/LOOP (RC)					0.50										
1.0004	UNE Coin Port/Loop Combo Usage (Flat Rate)			UEPCO	URECU	3.59	0.00	0.00					33.67	7.88	11.17	3.91
LUCA	Local Number Portability (1 per port)			UEPCO	LNPCX	0.35										
NONR	ECURRING CHARGES - CURRENTLY COMBINED					0.55										
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -															1
	Switch-as-is			UEPCO	USAC2		2.01	0.3108					33.67	7.88	11.17	3.91
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -															
	Switch with change			UEPCO	USACC		2.01	0.31					33.67	7.88	11.17	3.91
ADDIT	IONAL NRCs															
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity			UEPCO	USAS2		0.00	0.00					33.67	7.88	11.17	3.91
2-WIR	E VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE				USA52		0.00	0.00	-				33.67	7.88	11.17	3.91
	ort/Loop Combination Rates															
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			18.69										1
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			21.30										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			32.77										
UNE L	oop Rates				115050	10.01										
	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFR UEPFR	UECF2 UECF2	16.84 19.45										
	2-Wire Voice Grade Loop (SL2) - Zone 2 2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFR	UECF2	30.92			-							ł
2-Wire	e Voice Grade Line Port Rates (Res)		5	OLITIK		30.32										
	2-Wire voice unbundled port - residence			UEPFR	UEPRL	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled port with Caller ID - res			UEPFR	UEPRC	1.85	121.33	95.26	8.45	3.91			37.06	7.88	11.17	3.91
	2-Wire voice unbundled port outgoing only - res			UEPFR	UEPRO	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundles res, low usage line port with Caller ID					Τ	Τ									
	(LUM)			UEPFR	UEPAP	1.85	121.33	95.26	8.45	3.91	<u> </u>		33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port, without Caller ID capability - res			UEPFR	UEPWC	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port for use with			JEI I N		1.00	121.00	55.20	0.40	5.91			55.07	1.00	11.17	3.91
	Caller ID - res		1	UEPFR	UEPWQ	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
1	2-Wire voice unbundled Georgia basic dialing port - outgoing															
	only			UEPFR	UEPWR	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
INTER	OFFICE TRANSPORT										ļ					ļ
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility				11111/2	17.07	70.04	26.00								1
	Termination Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile			UEPFR	U1TV2	17.07	79.61	36.08			<u> </u>					ł
				UEPFR	1L5XX	0.0000								1		1
	or Fraction Mile					(1 (1222)										
FEATU	or Fraction Mile JRES			UEPFR	ILOXX	0.0222										1

UNR	UNDLE	D NETWORK ELEMENTS - Georgia	r –	1		- <u> </u>						Come Cont	China China	Attachment:			ibit: B
CATE	GORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
							Rec	Nonrec	curring	Nonrecurring	J Disconnect				Rates(\$)		
							Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		NUMBER PORTABILITY															
		Local Number Portability (1 per port)			UEPFR	LNPCX	0.35										
		CURRING CHARGES (NRCs) - CURRENTLY COMBINED															
		2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port													=		
		Combination - Conversion - Switch-as-is			UEPFR	USAC2		93.83	93.83					33.67	7.88	11.17	3.91
		2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port			UEPFR	USACC		02.92	02.02					33.67	7 00		
		Combination - Conversion - Switch-With-Change VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE				USACC		93.83	93.83					33.67	7.88		
		ort/Loop Combination Rates			803)							1			-	-	
		2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			18.69										
		2-Wire VG Loop/IO Tranport/Port Combo - Zone 1 2-Wire VG Loop/IO Tranport/Port Combo - Zone 2	t	2			21.30					1					<u> </u>
		2-Wire VG Loop/IO Tranport/Port Combo - Zone 2	1	3	1	1 1	32.77					t		1			1
		op Rates				1											1
		2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFB	UECF2	16.84										
		2-Wire Voice Grade Loop (SL2) - Zone 2	I	2	UEPFB	UECF2	19.45										
		2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFB	UECF2	30.92										
		Voice Grade Line Port (Bus)															
		2-Wire voice unbundled port without Caller ID - bus			UEPFB	UEPBL	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
		2-Wire voice unbundled port with Caller + E484 ID - bus			UEPFB	UEPBC	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
		2-Wire voice unbundled port outgoing only - bus			UEPFB	UEPBO	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
		2-Wire voice unbundled incoming only port with Caller ID - Bus			UEPFB	UEPB1	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
		2-Wire voice unbundled Georgia basic dialing port, without Caller ID capability - bus			UEPFB	UEPWD	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
		2-Wire voice unbundled Georgia basic dialing port for use with Caller ID - bus			UEPFB	UEPWP	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
	LOCAL	NUMBER PORTABILITY			OLFTB	OLFWF	1.05	121.33	93.20	0.45	3.91			55.07	7.00	11.17	3.91
		Local Number Portability (1 per port)			UEPFB	LNPCX	0.35										
		OFFICE TRANSPORT															
		Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility Termination			UEPFB	U1TV2	17.07	79.61	36.08								
		Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile or Fraction Mile			UEPFB	1L5XX	0.0222										
	FEATU	RES															
		All Features Offered			UEPFB	UEPVF	0.00	0.00	0.00					33.67	7.88	11.17	3.91
	NONRE	CURRING CHARGES (NRCs) - CURRENTLY COMBINED															
		2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
		Combination - Conversion - Switch-as-is			UEPFB	USAC2		93.83	93.83					33.67	7.88	11.17	3.91
		2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
		Combination - Conversion - Switch with change			UEPFB	USACC		93.83	93.83			L					ł
		VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)	ļ							<u> </u>		ļ					
		ort/Loop Combination Rates		1			18.60										
		2-Wire VG Loop/IO Tranport/Port Combo - Zone 1 2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		1			18.69 21.30		1		1						l
		2-Wire VG Loop/IO Tranport/Port Combo - Zone 2 2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		2			32.77		1		1			1			ł
		op Rates		5			52.11					t					1
		2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFP	UECF2	16.84										1
		2-Wire Voice Grade Loop (SL2) - Zone 1	1	2	UEPFP	UECF2	19.45					t		1			1
		2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFP	UECF2	30.92			1							1
		Voice Grade Line Port Rates (BUS - PBX)	I									İ					
		Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPFP	UEPPC	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
	1	Line Side Unbundled Outward PBX Trunk Port - Bus	1	1	UEPFP	UEPPO	1.85	121.33	95.26	8.45	3.91	1		33.67	7.88	11.17	3.91
	1	Line Side Unbundled Incoming PBX Trunk Port - Bus	1	1	UEPFP	UEPP1	1.85	121.33	95.26	8.45	3.91	1		33.67	7.88	11.17	
		2-Wire Voice Unbundled PBX LD Terminal Ports	I	1	UEPFP	UEPLD	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
		2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPFP	UEPXA	1.85	121.33	95.26	8.45	3.91			37.06	7.88	11.17	
		2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPFP	UEPXB	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
		2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPFP	UEPXC	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
		2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPFP	UEPXD	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91

	RATE ELEMENTS 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port 2-Wire voice Unbundled 1-Way Outgoing PBX Measured Port 2-Wire voice Unbundled 1-Way Outgoing PBX Measured Port 2-Wire voice Unbundled 1-Way Outgoing PBX Measured Port 2-Wire voice Unbundled Georgia basic dialing port - 1-Way Oudial Trunk	Interi m	Zone	BCS UEPFP UEPFP	USOC	Rec	Nonree First		Nonrecurring	Disconnect	Svc Order Submitted Elec per LSR		Charge - Manual Svc Order vs. Electronic- 1st OSS	Order vs. Electronic- Add'I Rates(\$)	Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
	Capable Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port 2-Wire voice unbundled Georgia basic dialing port - 1-Way Oudral Trunk			-	UEPXE	Rec				Disconnect	1					
	Capable Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port 2-Wire voice unbundled Georgia basic dialing port - 1-Way Oudral Trunk			-	UEPXE		First									
	Capable Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port 2-Wire voice unbundled Georgia basic dialing port - 1-Way Oudral Trunk			-	UEPXE	1		Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port 2-Wire voice unbundled Georgia basic dialing port - 1-Way Oudial Trunk			-		1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
	Administrative Calling Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port 2-Wire voice unbundled Georgia basic dialing port - 1-Way Oudial Trunk			UEPFP		1.00	121.00	33.20	0.45	5.51			55.07	1.00	11.17	3.31
	Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port 2-Wire voice unbundled Georgia basic dialing port - 1-Way Oudial Trunk				UEPXL	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port 2-Wire voice unbundled Georgia basic dialing port - 1-Way Oudial Trunk															Ī
	Discount Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port 2-Wire voice unbundled Georgia basic dialing port - 1-Way Oudial Trunk			UEPFP	UEPXM	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port 2-Wire voice unbundled Georgia basic dialing port - 1-Way Oudial Trunk			UEPFP	UEPXO	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port - 1-Way Oudial Trunk			UEPFP	UEPXS	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
-																1
-				UEPFP	UEPWS	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port - 2-Way													=		
	Trunk NUMBER PORTABILITY			UEPFP	UEPWT	1.85	121.33	95.26	8.45	3.91			33.67	7.88	11.17	3.91
	Local Number Portability (1 per port)			UEPFP	LNPCP	3.15	0.00	0.00					33.67	7.88	11.17	3.91
	DFFICE TRANSPORT			02.11	2.11 0.	0.10	0.00	0.00					00.01	1100		0.01
í Í ľ	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility				1											Ĩ
	Termination			UEPFP	U1TV2	17.07	79.61	36.08								1
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile															ł
FEATUR	or Fraction Mile			UEPFP	1L5XX	0.0222										
	All Features Offered			UEPFP	UEPVF	0.00	0.00	0.00					33.67	7.88	11.17	3.91
	CURRING CHARGES (NRCs) - CURRENTLY COMBINED			02	02.11	0.00	0.00	0.00					00.01	1.00		0.01
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port										1					ĺ
	Combination - Conversion - Switch-as-is			UEPFP	USAC2		93.83	93.83					33.67	7.88	11.17	3.91
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port													=		
	Combination - Conversion - Switch with change ORT/LOOP COMBINATIONS - COST BASED RATES		-	UEPFP	USACC		93.83	93.83					33.67	7.88	11.17	3.91
	VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK	PORT			+ +											1
	ort/Loop Combination Rates				<u> </u>											[
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1		1			28.19										1
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2		2			30.80										i
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3		3			42.27										Í
	op Rates															1
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1		1	UEPPX	UECD1	16.84	104.17	78.10								1
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2		2	UEPPX	UECD1	19.45	104.17	78.10								I
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3		3	UEPPX	UECD1	30.92	104.17	104.10								l
UNE Po				UEPPX	UEPD1	11.35	61.91	61.91					33.67	7.88		}
	Exchange Ports - 2-Wire DID Port CURRING CHARGES - CURRENTLY COMBINED				UCPDI	11.30	01.91	01.91					33.67	7.88		
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination -		1													[
	Switch-as-is			UEPPX	USAC1		93.38	93.38					33.67	7.88		l
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion															1
	with BellSouth Allowable Changes			UEPPX	USA1C		93.38	93.38					33.67	7.88		
	ONAL NRCs				┥───┤											
	one Number/Trunk Group Establisment Charges DID Trunk Termination (One Per Port)			UEPPX	NDT	0.00	0.00	0.00								
	DID Trunk Termination (One Per Port) DID Numbers, Establish Trunk Group and Provide First Group		<u> </u>	UEFPA	וטא	0.00	0.00	0.00								
	of 20 DID Numbers			UEPPX	NDZ	0.00	0.00	0.00								ł
	Additional DID Numbers for each Group of 20 DID Numbers		1	UEPPX	ND4	0.00	0.00	0.00								
	DID Numbers, Non- consecutive DID Numbers , Per Number			UEPPX	ND5	0.00	0.00	0.00								1
	Reserve Non-Consecutive DID numbers			UEPPX	ND6	0.00	0.00	0.00								
	Reserve DID Numbers			UEPPX	NDV	0.00	0.00	0.00								I
	Local Number Portability (1 per port)			UEPPX	LNPCP	3.15	0.00	0.00								
	ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LIN prt/Loop Combination Rates				┼───┤											
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -				+ +											
	UNE Zone 1		1	UEPPB UEPPR		35.36										ł

UNBUNDLE	D NETWORK ELEMENTS - Georgia													Attachment:			ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	E	SCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge - Manual Svo Order vs.
							Rec	Nonrec		Nonrecurring Di					Rates(\$)		.
								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 2		2	UEPPB	UEPPR		29.74										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -		2	UEPPB	UEPPR		38.74										+
	UNE Zone 3		3	UEPPB	UEPPR		53.64										
UNE L	oop Rates		Ŭ	02.1.0	021111		00.01										1
	2-Wire ISDN Digital Grade Loop - UNE Zone 1		1	UEPPB	UEPPR	USL2X	21.89	252.32	188.77					19.99	19.99		
	2-Wire ISDN Digital Grade Loop - UNE Zone 2		2	UEPPB	UEPPR	USL2X	25.27	252.32	188.77					19.99	19.99		
	2-Wire ISDN Digital Grade Loop - UNE Zone 3	-	3	UEPPB	UEPPR	USL2X	40.17	252.32	188.77					19.99	19.99		
UNE P	ort Rate				UEPPR	UEPPB	13.47	47.07	47.37					19.99	19.99		+
NOND	Exchange Port - 2-Wire ISDN Line Side Port ECURRING CHARGES - CURRENTLY COMBINED			UEPPB	UEPPR	UEPPB	13.47	47.37	47.37					19.99	19.99		
NONKI	2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port																+
	Combination - Conversion			UEPPB	UEPPR	USACB	0.00	93.38	93.38					19.99	19.99		
ADDIT	IONAL NRCs																
	2-Wire ISDN Loop / 2-Wire ISDN Port Combination - Sub Actvy																1
	Non Feature/Add Trunk			UEPPB	UEPPR	USASB		165.95						19.99	19.99		
LOCAL	L NUMBER PORTABILITY																
	Local Number Portability (1 per port)			UEPPB	UEPPR	LNPCX	0.35	0.00	0.00								
B-CHA	NNEL USER PROFILE ACCESS:																<u> </u>
	CVS/CSD (DMS/5ESS)			UEPPB	UEPPR	U1UCA	0.00	0.00	0.00								
	CVS (EWSD)			UEPPB	UEPPR	U1UCB	0.00	0.00	0.00								+
ВСЦА	CSD NNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS SC	MC		UEPPB	UEPPR	U1UCC	0.00	0.00	0.00								+
	TERMINAL PROFILE	, IVI 3, 0		-													+
USER	User Terminal Profile (EWSD only)			UEPPB	UEPPR	U1UMA	0.00	0.00	0.00								-
VERTI	CAL FEATURES			OLITE	OLITIK	010101	0.00	0.00	0.00								1
	All Vertical Features - One per Channel B User Profile			UEPPB	UEPPR	UEPVF	0.00	0.00	0.00					19.99	19.99		
INTER	OFFICE CHANNEL MILEAGE																1
	Interoffice Channel mileage each, including first mile and																
	facilities termination	-			UEPPR	M1GNC	16.47	79.61	36.08					19.99	19.99		
	Interoffice Channel mileage each, additional mile			UEPPB	UEPPR	M1GNM	0.0222	0.00	0.00				0.00				
	E DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK	PORT															
UNE P	ort/Loop Combination Rates																-
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 1		1	UEPPP			218.69										
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE		1	UEFFF			210.09										-
	Zone 2		2	UEPPP			227.29										
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE			1		1				<u>├</u>						1	1
1	Zone 3		3	UEPPP			265.09										
UNE L	oop Rates																
	4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPPP		USL4P	55.53	448.92	276.60					19.99	19.99		
	4-Wire DS1 Digital Loop - UNE Zone 2		2	UEPPP		USL4P	64.13	448.92	276.60	\square				19.99	19.99		\vdash
	4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPPP		USL4P	101.93	448.92	276.60	├ ─── ├				19.99	19.99		───
UNE P	ort Rate			UEPPP		UEPPP	163.16	186.80	186.80	├────┤─				19.99	19.99		──
NOND	Exchange Ports - 4-Wire ISDN DS1 Port ECURRING CHARGES - CURRENTLY COMBINED			UEPPP		UEPPP	163.16	186.80	186.80	├				19.99	19.99		+
NONRI	4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port			-		1				<u>├</u>							+
1	Combination - Conversion -Switch-as-is		1	UEPPP		USACP	0.00	269.96	269.96					19.99	19.99		1
ADDIT	IONAL NRCs			1			0.00	200.00	200.00								+
	4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy-		1	1		1				1						İ	1
	Inward/two way Tel Nos. (except NC)			UEPPP		PR7TF		0.9686									
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port -			T													
	Outward Tel Numbers (All States except NC)			UEPPP		PR7TO		22.75	22.75								
1 -	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port -			l							Т						
1	Subsequent Inward Tel Numbers			UEPPP		PR7ZT		45.49	45.49								
LOCAI	L NUMBER PORTABILITY Local Number Portability (1 per port)			UEPPP		LNPCN	1.75										

UNBUNDLED NETWORK ELEMENTS - Georgia		-										Attachment:			bit: B
ATEGORY RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
					Rec	Nonrec		Nonrecurring					Rates(\$)		
				55544		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Voice/Data Digital Data		-	UEPPP UEPPP	PR71V PR71D	0.00	0.00 0.00	0.00								<u> </u>
Inward Data			UEPPP	PR71E	0.00	0.00	0.00								ł
New or Additional "B" Channel			0EIII	110/12	0.00	0.00	0.00								
New or Additional - Voice/Data B Channel			UEPPP	PR7BV	0.00	28.71						19.99	19.99		
New or Additional - Digital Data B Channel			UEPPP	PR7BF	0.00	28.71						19.99	19.99		
New or Additional Inward Data B Channel			UEPPP	PR7BD	0.00	28.71						19.99	19.99		
CALL TYPES			115000	22201		0.00									
Inward Outward		_	UEPPP UEPPP	PR7C1 PR7C0	0.00	0.00	0.00								ł
Two-way		-	UEPPP	PR7C0 PR7CC	0.00	0.00	0.00								<u> </u>
Interoffice Channel Mileage		-	OLFFF	FILIO	0.00	0.00	0.00								<u> </u>
Fixed Each Including First Mile			UEPPP	1LN1A	78.9223	147.07	111.75	0.00				19.99	19.99		
Each Airline-Fractional Additional Mile			UEPPP	1LN1B	0.4523					1					L
4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE DDITS T	RUNK PORT									<u> </u>					
UNE Port/Loop Combination Rates															
4W DS1 Digital Loop/4W DDITS Trunk Port -		1	UEPDC		176.33										
4W DS1 Digital Loop/4W DDITS Trunk Port -		2	UEPDC		184.93										
4W DS1 Digital Loop/4W DDITS Trunk Port -	UNE Zone 3	3	UEPDC		222.73										ł
UNE Loop Rates 4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPDC	USLDC	55.53	448.92	276.00					19.99	19.99		<u> </u>
4-Wire DS1 Digital Loop - UNE Zone 1 4-Wire DS1 Digital Loop - UNE Zone 2		2	UEPDC	USLDC	64.13	448.92	276.00					19.99	19.99		
4-Wire DS1 Digital Loop - UNE Zone 2		3	UEPDC	USLDC	101.93	448.92	276.60					19.99	19.99		
UNE Port Rate		-													
4-Wire DDITS Digital Trunk Port			UEPDC	UDD1T	120.80	89.44	52.46					19.99	19.99		
NONRECURRING CHARGES - CURRENTLY COMB															
4-Wire DS1 Digital Loop / 4-Wire DDITS Truni - Switch-as-is			UEPDC	USAC4		269.96	269.96					19.99	19.99		
4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk - Conversion with DS1 Changes			UEPDC	USAWA		269.96	269.96					19.99	19.99		
4-Wire DS1 Digital Loop / 4-Wire DDITS Trun - Conversion with Change - Trunk	Port Combination		UEPDC	USAWB		269.96	269.96					19.99	19.99		
ADDITIONAL NRCs			1												
4-Wire DS1 Loop / 4-Wire DDITS Trunk Port -	Subsequent														
Service Activity Per Service Order			UEPDC	USAS4		147.47	147.47								ļ
4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent Channel Activation/Chan - 2-Way	/ Trunk		UEPDC	UDTTA		28.71	28.71					19.99	19.99		
4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Channel Activation/Chan - 1-Way Outward Tr			UEPDC	UDTTB		28.71	28.71					19.99	19.99		
4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Activation/Chan Inward Trunk w/out DID			UEPDC	UDTTC		28.71	28.71					19.99	19.99		
4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Activation Per Chan - Inward Trunk with DID			UEPDC	UDTTD		28.71	28.71					19.99	19.99		
4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Activation / Chan - 2-Way DID w User Trans	Subsqnt Chan		UEPDC	UDTTE		28.71	28.71					19.99	19.99		
BIPOLAR 8 ZERO SUBSTITUTION															
B8ZS - Superframe Format			UEPDC	CCOSF		0.00	600.00								
B8ZS - Extended Superframe Format Alternate Mark Inversion		-	UEPDC	CCOEF		0.00	600.00								<u> </u>
Alternate Mark Inversion AMI -Superframe Format		-	UEPDC	MCOSF		0.00	0.00								<u> </u>
AMI - Extended SuperFrame Format			UEPDC	MCOPO		0.00	0.00								
Telephone Number/Trunk Group Establisment Cha	irges					0.00	0.00			1					<u></u>
Telephone Number for 2-Way Trunk Group	-		UEPDC	UDTGX	0.00			i i		1		1	1		[
Telephone Number for 1-Way Outward Trunk			UEPDC	UDTGY	0.00										
Telephone Number for 1-Way Inward Trunk G			UEPDC	UDTGZ	0.00										
DID Numbers, Establish Trunk Group and Pro of 20 DID Numbers			UEPDC	NDZ	0.00	0.00	0.00								
DID Numbers for each Group of 20 DID Numb			UEPDC	ND4	0.00										L
DID Numbers, Non- consecutive DID Numbers	s, Per Number		UEPDC	ND5	0.00										L

DUNULE	D NETWORK ELEMENTS - Georgia	1	1	r	-r	1					0	A A	Attachment:			ibit: B
TEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR		Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
						_	Nonrec	urrina	Nonrecurring Di	isconnect			OSS	Rates(\$)		
			1			Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Reserve Non-Consecutive DID Nos.			UEPDC	ND6	0.00	0.00	0.00								
	Reserve DID Numbers		<u> </u>	UEPDC	NDV	0.00	0.00	0.00								
Dedicat	ted DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS' Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities	1 Digita	I Loop	with 4-Wire DDHS	Trunk Port											
	Termination)			UEPDC	1LNO1	78.47	147.07	111.75					19.99	19.99		
	Interoffice Channel Mileage - Additional rate per mile - 0-8 miles			UEPDC	1LNOA	0.4523	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities Termination)			UEPDC	1LNO2	0.00	0.00	0.00								
	Interoffice Channel Mileage - Additional rate per mile - 9-25 miles			UEPDC	1LNOB	0.4523	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities		<u> </u>		121100	0.4020	0.00	0.00	<u>├</u>							
	Termination)			UEPDC	1LNO3	0.00	0.00	0.00								
			T													
	Interoffice Channel Mileage - Additional rate per mile - 25+ miles			UEPDC	1LNOC	0.4523	0.00	0.00								
	Local Number Portability, per DS0 Activated Central Office Termininating Point			UEPDC UEPDC	LNPCP CTG	3.15 0.00										
	DS1 LOOP WITH CHANNELIZATION WITH PORT			UEPDC	CIG	0.00										
	is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Act	ivations														
	ystem can have up to 24 combinations of rates depending on			ber of ports used	-											
	S1 Loop	.,														
	4-Wire DS1 Loop - UNE Zone 1		1	UEPMG	USLDC	55.53	0.00	0.00								
	4-Wire DS1 Loop - UNE Zone 2		2	UEPMG	USLDC	64.13	0.00	0.00								
	4-Wire DS1 Loop - UNE Zone 3		3	UEPMG	USLDC	101.93	0.00	0.00								
UNE DS	SO Channelization Capacities (D4 Channel Bank Configuration	ns)			1.4.19.40.4	100.01							10.00	10.00		
	24 DSO Channel Capacity - 1 per DS1 48 DSO Channel Capacity - 1 per 2 DS1s			UEPMG UEPMG	VUM24 VUM48	102.64 205.28	0.00	0.00					19.99 19.99	19.99 19.99		
	96 DSO Channel Capacity - 1 per 2 DS1s 96 DSO Channel Capacity -1per 4 DS1s			UEPMG	VUM48 VUM96	410.56	0.00	0.00					19.99	19.99		
	144 DS0 Channel Capacity - 1 per 6 DS1s			UEPMG	VUM14	615.84	0.00	0.00					19.99	19.99		
	192 DS0 Channel Capacity -1 per 8 DS1s			UEPMG	VUM19	821.12	0.00	0.00					19.99	19.99		
	240 DS0 Channel Capacity - 1 per 10 DS1s			UEPMG	VUM20	1,026.40	0.00	0.00					19.99	19.99		
	288 DS0 Channel Capacity - 1 per 12 DS1s			UEPMG	VUM28	1,231.68	0.00	0.00					19.99	19.99		
	384 DS0 Channel Capacity - 1 per 16 DS1s			UEPMG	VUM38	1,642.24	0.00	0.00					19.99	19.99		
	480 DS0 Channel Capacity - 1 per 20 DS1s			UEPMG	VUM40	2,052.80	0.00	0.00					19.99	19.99		
	576 DS0 Channel Capacity -1 per 24 DS1s			UEPMG	VUM57	2,463.36	0.00	0.00					19.99	19.99		
	672 DS0 Channel Capacity - 1 per 28 DS1s			UEPMG	VUM67	2,873.92	0.00	0.00					19.99	19.99		
	curring Charges (NRC) Associated with 4-Wire DS1 Loop with						stem									
	num System configuration is One (1) DS1, One (1) D4 Channe es of this configuration functioning as one are considered Ac															
wunp	NRC - Conversion (Currently Combined) with or without			inininum system co	Iniguration is	counted.										
	BellSouth Allowed Changes			UEPMG	USAC4	0.00	328.35	16.52					19.99	19.99		
System	Additions at End User Locations Where 4-Wire DS1 Loop wi	th Char	neliza				220.00	10.02							İ	1
	ot Currently Combined) in all states, except in Density Zone 1												l	l	l	1
	1 DS1/D4 Channel Bank - Additionally Add NRC for each Port															
	and Assoc Fea Activation			UEPMG	VUMD4	0.00	738.61	462.53	144.05	17.09			19.99	19.99		
Bipolar	8 Zero Substitution	I	Ļ													ļ
	Clear Channel Capability Format, superframe - Subsequent	1	1	UEPMG	00005	0.00	0.00	C00 00								
	Activity Only Clear Channel Capability Format - Extended Superframe -			UEPING	CCOSF	0.00	0.00	600.00	<u>├</u>							
	Subsequent Activity Only		1	UEPMG	CCOEF	0.00	0.00	600.00								
Alterna	te Mark Inversion (AMI)		1			0.00	0.00	500.00								
	Superframe Format	1		UEPMG	MCOSF	0.00	0.00	0.00								1
	Extended Superframe Format			UEPMG	MCOPO	0.00	0.00	0.00								
Exchan	ge Ports Associated with 4-Wire DS1 Loop with Channelization	on with	Port													
	no Donto	1	1	1		1			1		1	1	1	1	1	1
Exchan	ge Ports		-			+ +										
	ge Ports			UEPPX	UEPCX	1.79	0.00	0.00	0.00	0.00			33.67	7.88		

NBUNDLE	D NETWORK ELEMENTS - Georgia					1							Attachment:	r	Exhi	
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC		Nama	RATES(\$)	Nanananina	Diagonat		Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonree		Nonrecurring		COMEC	COMAN		Rates(\$)	SOMAN	COMAN
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Line Side Inward Only Channelized PBX Trunk Port without DID			UEPPX	UEP1X	1.79	0.00	0.00	0.00	0.00			33.67	7.88		
	2-Wire Trunk Side Unbundled Channelized DID Trunk Port			UEPPX	UEPDM	11.35	0.00	0.00	0.00	0.00			33.67	7.88		
Featur	e Activations - Unbundled Loop Concentration			02.17	02. 5	11.00	0.00	0.00	0.00	0.00			00.01	1.00		
	Feature (Service) Activation for each Line Port Terminated in D4															
	Bank			UEPPX	1PQWM	0.62	25.09	13.25	3.99	3.97			33.67	7.88		
	Feature (Service) Activation for each Trunk Port Terminated in															
	D4 Bank			UEPPX	1PQWU	0.62	77.21	18.20	56.49	11.04			33.67	7.88		
Teleph	one Number/ Group Establishment Charges for DID Service															
	DID Trunk Termination (1 per Port)			UEPPX	NDT	0.00	0.00	0.00								
	Estab Trk Grp and Provide 1st 20 DID Nos. (FL,GA, NC,& SC)			UEPPX	NDZ	0.00	0.00	0.00								
	DID Numbers - groups of 20 - Valid all States			UEPPX	ND4	0.00	0.00	0.00	├							
	Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers			UEPPX UEPPX	ND5 ND6	0.00	0.00	0.00								
	Reserve Non-Consecutive DID Numbers			UEPPX	ND6 NDV	0.00	0.00	0.00								
Local	Number Portability			UEFFA	NDV	0.00	0.00	0.00					-			
LUCALI	Local Number Portability - 1 per port			UEPPX	LNPCP	3.15	0.00	0.00								
FEATU	IRES - Vertical and Optional			0EITX		0.10	0.00	0.00								
	Switching Features Offered with Line Side Ports Only				-											
	All Features Available			UEPPX	UEPVF	0.00	0.00	0.00								
BUNDLED I	PORT LOOP COMBINATIONS - MARKET RATES															
	Rates shall apply where BellSouth is not required to provide	unbun	died lo	cal switching or sy	witch ports pe	FCC and/or St	ate Commissio	on rules.								
Market	Rates shall apply where beiloouth is not required to provide															
This in Unbun The To BellSo	cludes: dled port/loop combinations that are Currently Combined or I p 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderd uth currently is developing the billing capability to mechanica	Not Cur ale, Mia ally bill	ami); G the rec	A (Atlanta); LA (Ne curring and non-re-	w Orleans); No curring Market	C (Greensboro-) Rates in this se	Winston Salem	-Highpoint/Ch or nonrecurrin	arlotte-Gastoni	a-Rock Hill); 1	N (Nashville		. In the interi	m where Bell	South cannot	bill Market
This in Unbun The To BellSo Rates, The Ma	cludes: dled port/loop combinations that are Currently Combined or I p 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderd uth currently is developing the billing capability to mechanicz BellSouth shall bill the rates in the Cost-Based section preced arket Rate for unbundled ports includes all available features in	Not Cur ale, Mia ally bill ding in in all st	ami); G the rec lieu of ates.	A (Atlanta); LA (Ne curring and non-re the Market Rates a	w Orleans); No curring Market and reserves th	C (Greensboro-) Rates in this some right to true-	Winston Salem ection except up the billing	-Highpoint/Ch or nonrecurrir difference.	arlotte-Gastoni ig charges for r	a-Rock Hill); 1 not currently c	N (Nashville ombined in	FL and NC				
This in Unbun The To BellSo Rates, The Ma End Of (USOC	cludes: died port/loop combinations that are Currently Combined or I p 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderd uth currently is developing the billing capability to mechanica BellSouth shall bill the rates in the Cost-Based section prece arket Rate for unbundled ports includes all available features i fice and Tandem Switching Usage and Common Transport Us : URECU).	Not Cur ale, Mia ally bill ding in in all st sage rat	ami); G. the rec lieu of ates. tes in t	A (Atlanta); LA (Ne curring and non-re- the Market Rates a he Port section of	w Orleans); NG curring Market and reserves th this rate exhib	C (Greensboro-\ Rates in this so he right to true- it shall apply to	Winston Salem ection except f up the billing of all combination	-Highpoint/Ch or nonrecurrin difference. ons of loop/po	arlotte-Gastoni ig charges for r rt network elem	a-Rock Hill); 1 not currently c	N (Nashville combined in or UNE Coin	FL and NC	Combination	ns which have	e a flat rate us	age charge
This in Unbun The To BellSo Rates, The Ma End Of (USOC For No	cludes: died port/loop combinations that are Currently Combined or I po 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdi uth currently is developing the billing capability to mechanice BellSouth shall bill the rates in the Cost-Based section preced arket Rate for unbundled ports includes all available features i fifice and Tandem Switching Usage and Common Transport Us URECU). t Currently Combined scenarios the Nonrecurring charges are	Not Cur ale, Mia ally bill ding in in all st sage rat	ami); G. the rec lieu of ates. tes in t	A (Atlanta); LA (Ne curring and non-re- the Market Rates a he Port section of	w Orleans); NG curring Market and reserves th this rate exhib	C (Greensboro-\ Rates in this so he right to true- it shall apply to	Winston Salem ection except f up the billing of all combination	-Highpoint/Ch or nonrecurrin difference. ons of loop/po	arlotte-Gastoni ig charges for r rt network elem	a-Rock Hill); 1 not currently c	N (Nashville combined in or UNE Coin	FL and NC	Combination	ns which have	e a flat rate us	age charge
This in Unbun The To BellSo Rates, The Ma End Of (USOC For No Additio	cludes: dled port/loop combinations that are Currently Combined or I pp 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderd uth currently is developing the billing capability to mechanica BellSouth shall bill the rates in the Cost-Based section preced arket Rate for unbundled ports includes all available features i fice and Tandem Switching Usage and Common Transport Us : URECU). t Currently Combined scenarios the Nonrecurring charges are onal NRCs may apply also and are categorized accordingly.	Not Cur ale, Mia ally bill ding in in all st sage rat	ami); G. the rec lieu of ates. tes in t	A (Atlanta); LA (Ne curring and non-re- the Market Rates a he Port section of	w Orleans); NG curring Market and reserves th this rate exhib	C (Greensboro-\ Rates in this so he right to true- it shall apply to	Winston Salem ection except f up the billing of all combination	-Highpoint/Ch or nonrecurrin difference. ons of loop/po	arlotte-Gastoni ig charges for r rt network elem	a-Rock Hill); 1 not currently c	N (Nashville combined in or UNE Coin	FL and NC	Combination	ns which have	e a flat rate us	age charge
This in Unbun The To BellSo Rates, The Ma End Of (USOC For No Additio 2-WIRE	icludes: died port/loop combinations that are Currently Combined or I op 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderd, uth currently is developing the billing capability to mechanica BellSouth shall bill the rates in the Cost-Based section prece arket Rate for unbundled ports includes all available features i fice and Tandem Switching Usage and Common Transport Us : URECU). t Currently Combined scenarios the Nonrecurring charges are onal NRCs may apply also and are categorized accordingly. E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)	Not Cur ale, Mia ally bill ding in in all st sage rat	ami); G. the rec lieu of ates. tes in t	A (Atlanta); LA (Ne curring and non-re- the Market Rates a he Port section of	w Orleans); NG curring Market and reserves th this rate exhib	C (Greensboro-\ Rates in this so he right to true- it shall apply to	Winston Salem ection except f up the billing of all combination	-Highpoint/Ch or nonrecurrin difference. ons of loop/po	arlotte-Gastoni ig charges for r rt network elem	a-Rock Hill); 1 not currently c	N (Nashville combined in or UNE Coin	FL and NC	Combination	ns which have	e a flat rate us	age charge
This in Unbun The To BellSo Rates, The Ma End Od (USOC For No Additio 2-WIRE	icludes: died port/loop combinations that are Currently Combined or I ps 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdi uth currently is developing the billing capability to mechanica BellSouth shall bill the rates in the Cost-Based section preced arket Rate for unbundled ports includes all available features i fice and Tandem Switching Usage and Common Transport Us : URECU). t Currently Combined scenarios the Nonrecurring charges are onal NRCs may apply also and are categorized accordingly. E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates	Not Cur ale, Mia ally bill ding in in all st sage rat	ami); G the rec lieu of ates. tes in t in the	A (Atlanta); LA (Ne curring and non-re- the Market Rates a he Port section of	w Orleans); NG curring Market and reserves th this rate exhib	C (Greensboro-1 Rates in this so re right to true- it shall apply to s for each Port	Winston Salem ection except f up the billing of all combination	-Highpoint/Ch or nonrecurrin difference. ons of loop/po	arlotte-Gastoni ig charges for r rt network elem	a-Rock Hill); 1 not currently c	N (Nashville combined in or UNE Coin	FL and NC	Combination	ns which have	e a flat rate us	age charge
This in Unbun The To BellSo Rates, The Ma End Of (USOC For No Additio 2-WIRE	cludes: died port/loop combinations that are Currently Combined or I pp 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdi uth currently is developing the billing capability to mechanice BellSouth shall bill the rates in the Cost-Based section preced arket Rate for unbundled ports includes all available features i fifice and Tandem Switching Usage and Common Transport Us : URECU). t Currently Combined scenarios the Nonrecurring charges are onal NRCs may apply also and are categorized accordingly. E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates [2-Wire VG Loop/Port Combo - Zone 1	Not Cur ale, Mia ally bill ding in in all st sage rat	ami); G the rec lieu of ates. tes in t in the 1	A (Atlanta); LA (Ne curring and non-re- the Market Rates a he Port section of	w Orleans); NG curring Market and reserves th this rate exhib	C (Greensboro-V Rates in this s- ne right to true- it shall apply to s for each Port	Winston Salem ection except f up the billing of all combination	-Highpoint/Ch or nonrecurrin difference. ons of loop/po	arlotte-Gastoni ig charges for r rt network elem	a-Rock Hill); 1 not currently c	N (Nashville combined in or UNE Coin	FL and NC	Combination	ns which have	e a flat rate us	age charge
This in Unbun The To BellSo Rates, The Ma End Of (USOC For No Additio 2-WIRE	icludes: died port/loop combinations that are Currently Combined or I op 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderd, uth currently is developing the billing capability to mechanica BellSouth shall bill the rates in the Cost-Based section prece arket Rate for unbundled ports includes all available features i fice and Tandem Switching Usage and Common Transport Us : URECU). t Currently Combined scenarios the Nonrecurring charges are onal NRCs may apply also and are categorized accordingly. E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2	Not Cur ale, Mia ally bill ding in in all st sage rat	ami); G the rec lieu of ates. tes in t in the 1 2	A (Atlanta); LA (Ne curring and non-re- the Market Rates a he Port section of	w Orleans); NG curring Market and reserves th this rate exhib	C (Greensboro-N Rates in this so he right to true- it shall apply to s for each Port 24.80 26.47	Winston Salem ection except f up the billing of all combination	-Highpoint/Ch or nonrecurrin difference. ons of loop/po	arlotte-Gastoni ig charges for r rt network elem	a-Rock Hill); 1 not currently c	N (Nashville combined in or UNE Coin	FL and NC	Combination	ns which have	e a flat rate us	age charge
This in Unbun The Tic BellSo Rates, The Mi End Oi (USOC For No Additic 2-WIRE UNE P	icludes: died port/loop combinations that are Currently Combined or I ps 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdi uth currently is developing the billing capability to mechanica BellSouth shall bill the rates in the Cost-Based section prece- arket Rate for unbundled ports includes all available features i fice and Tandem Switching Usage and Common Transport Us : URECU). t Currently Combined scenarios the Nonrecurring charges are onal NRCs may apply also and are categorized accordingly. E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates [2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 [2-Wire VG Loop/Port Combo - Zone 3	Not Cur ale, Mia ally bill ding in in all st sage rat	ami); G the rec lieu of ates. tes in t in the 1	A (Atlanta); LA (Ne curring and non-re- the Market Rates a he Port section of	w Orleans); NG curring Market and reserves th this rate exhib	C (Greensboro-V Rates in this s- ne right to true- it shall apply to s for each Port	Winston Salem ection except f up the billing of all combination	-Highpoint/Ch or nonrecurrin difference. ons of loop/po	arlotte-Gastoni ig charges for r rt network elem	a-Rock Hill); 1 not currently c	N (Nashville combined in or UNE Coin	FL and NC	Combination	ns which have	e a flat rate us	age charge
This in Unbun The Tc BellSo Rates, The Mi End Oi (USOC FOR No Additio 2-WIRE UNE Po	includes: died port/loop combinations that are Currently Combined or I po 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdi uth currently is developing the billing capability to mechanice BellSouth shall bill the rates in the Cost-Based section preced arket Rate for unbundled ports includes all available features i ffice and Tandem Switching Usage and Common Transport Us URECU). t Currently Combined scenarios the Nonrecurring charges are onal NRCs may apply also and are categorized accordingly. E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 oop Rates	Not Cur ale, Mia ally bill ding in in all st sage rat	ami); G the rec lieu of ates. tes in t in the 1 2	A (Atlanta); LA (Ne urring and non-re the Market Rates a he Port section of First and Addition:	w Orleans); NC curring Market and reserves th this rate exhib al NRC column	C (Greensboro-V Rates in this su- eright to true- it shall apply to s for each Port 24.80 26.47 33.83	Winston Salem ection except f up the billing of all combination	-Highpoint/Ch or nonrecurrin difference. ons of loop/po	arlotte-Gastoni ig charges for r rt network elem	a-Rock Hill); 1 not currently c	N (Nashville combined in or UNE Coin	FL and NC	Combination	ns which have	e a flat rate us	age charge
This in Unbun The Tc BellSo Rates, The Mi End Oi (USOC FOR No Additio 2-WIRE UNE P	cludes: 1 ded port/loop combinations that are Currently Combined or I pp 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdi uth currently is developing the billing capability to mechanica BellSouth shall bill the rates in the Cost-Based section preced arket Rate for unbundled ports includes all available features i fifice and Tandem Switching Usage and Common Transport Us URECU). t Currently Combined scenarios the Nonrecurring charges are onal NRCs may apply also and are categorized accordingly. E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 oop Rates 2-Wire Voice Grade Loop (SL1) - Zone 1	Not Cur ale, Mia ally bill ding in in all st sage rat	ami); G. the rec lieu of ates. tes in t in the 1 2 3 1	A (Atlanta); LA (Ne surring and non-re- the Market Rates a he Port section of First and Addition: UEPRX	w Orleans); NC curring Market and reserves th this rate exhib al NRC column	C (Greensboro-V Rates in this s- ne right to true- it shall apply to s for each Port 24.80 26.47 33.83 10.80	Winston Salem ection except f up the billing of all combination	-Highpoint/Ch or nonrecurrin difference. ons of loop/po	arlotte-Gastoni ig charges for r rt network elem	a-Rock Hill); 1 not currently c	N (Nashville combined in or UNE Coin	FL and NC	Combination	ns which have	e a flat rate us	age charge
This in Unbun The Tc BellSo Rates, The Mi End Oi (USOC For No Additio 2-WIRE UNE P	icludes: died port/loop combinations that are Currently Combined or I ps 8 MSAs in BellSouth's region are: FL (Orlando, FL Lauderd, uth currently is developing the billing capability to mechanica BellSouth shall bill the rates in the Cost-Based section prece arket Rate for unbundled ports includes all available features i fice and Tandem Switching Usage and Common Transport Us : URECU). t Currently Combined scenarios the Nonrecurring charges are bonal NRCs may apply also and are categorized accordingly. E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates [2-Wire VG Loop/Port Combo - Zone 1 [2-Wire VG Loop/Port Combo - Zone 2 [2-Wire VG Loop/Port Combo - Zone 3 oop Rates [2-Wire Voice Grade Loop (SL1) - Zone 1 [2-Wire Voice Grade Loop (SL1) - Zone 2	Not Cur ale, Mia ally bill ding in in all st sage rat	ami); G. the rec lieu of ates. tes in t in the 1 2 3 1 2 1 2	A (Atlanta); LA (Ne urring and non-re- the Market Rates a he Port section of First and Addition UEPRX UEPRX UEPRX	w Orleans); NC curring Market and reserves th this rate exhib al NRC column	2 (Greensboro-) Rates in this s he right to true- it shall apply to as for each Port 24.80 26.47 33.83 10.80 12.47	Winston Salem ection except f up the billing of all combination	-Highpoint/Ch or nonrecurrin difference. ons of loop/po	arlotte-Gastoni ig charges for r rt network elem	a-Rock Hill); 1 not currently c	N (Nashville combined in or UNE Coin	FL and NC	Combination	ns which have	e a flat rate us	age charge
This in Unbun The TC BellSo Rates, The M. End O(USOC For No Additic 2-WIRE UNE P UNE L	includes: died port/loop combinations that are Currently Combined or I pp 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdi uth currently is developing the billing capability to mechanice BellSouth shall bill the rates in the Cost-Based section preceder arket Rate for unbundled ports includes all available features is fice and Tandem Switching Usage and Common Transport Us : URECU). t Currently Combined scenarios the Nonrecurring charges are onal NRCs may apply also and are categorized accordingly. E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates [2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 3 oop Rates [2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3	Not Cur ale, Mia ally bill ding in in all st sage rat	ami); G. the rec lieu of ates. tes in t in the 1 2 3 1	A (Atlanta); LA (Ne surring and non-re- the Market Rates a he Port section of First and Addition: UEPRX	w Orleans); NC curring Market and reserves th this rate exhib al NRC column	C (Greensboro-V Rates in this s- ne right to true- it shall apply to s for each Port 24.80 26.47 33.83 10.80	Winston Salem ection except f up the billing of all combination	-Highpoint/Ch or nonrecurrin difference. ons of loop/po	arlotte-Gastoni ig charges for r rt network elem	a-Rock Hill); 1 not currently c	N (Nashville combined in or UNE Coin	FL and NC	Combination	ns which have	e a flat rate us	age charge
This in Unbun The Tc BellSo Rates, The M End O(USOC For No Additic 2-WIRE UNE P UNE L	cludes: 1 died port/loop combinations that are Currently Combined or I pp 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdi uth currently is developing the billing capability to mechanics BellSouth shall bill the rates in the Cost-Based section preced arket Rate for unbundled ports includes all available features i ffice and Tandem Switching Usage and Common Transport Us URECU). t Currently Combined scenarios the Nonrecurring charges are onal NRCs may apply also and are categorized accordingly. E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 3 oop Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 Voice Grade Loop (SL1) - Zone 3 Voice Grade Loop (SL1) - Zone 3	Not Cur ale, Mia ally bill ding in in all st sage rat	ami); G. the rec lieu of ates. tes in t in the 1 2 3 1 2 1 2	A (Atlanta); LA (Ne urring and non-re- the Market Rates a he Port section of First and Addition UEPRX UEPRX UEPRX	w Orleans); NC curring Market and reserves th this rate exhib al NRC column	2 (Greensboro-) Rates in this s he right to true- it shall apply to as for each Port 24.80 26.47 33.83 10.80 12.47	Winston Salem ection except f up the billing of all combination	-Highpoint/Ch or nonrecurrin difference. ons of loop/po	arlotte-Gastoni ig charges for r rt network elem	a-Rock Hill); 1 not currently c	N (Nashville combined in or UNE Coin	FL and NC	Combination	ns which have	e a flat rate us	age charge
This in Unbun The Tc BellSo Rates, The M End O(USOC For No Additic 2-WIRE UNE P UNE L	includes: died port/loop combinations that are Currently Combined or I pp 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdi uth currently is developing the billing capability to mechanice BellSouth shall bill the rates in the Cost-Based section preceder arket Rate for unbundled ports includes all available features is fice and Tandem Switching Usage and Common Transport Us : URECU). t Currently Combined scenarios the Nonrecurring charges are onal NRCs may apply also and are categorized accordingly. E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates [2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 3 oop Rates [2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3	Not Cur ale, Mia ally bill ding in in all st sage rat	ami); G. the rec lieu of ates. tes in t in the 1 2 3 1 2 1 2	A (Atlanta); LA (Ne surring and non-re- the Market Rates a he Port section of First and Addition: UEPRX UEPRX UEPRX UEPRX UEPRX	w Orleans); NC curring Market and reserves th this rate exhib al NRC column UEPLX UEPLX UEPLX UEPLX UEPLX	C (Greensboro-V Rates in this s- ne right to true- it shall apply to s for each Port 24.80 26.47 33.83 10.80 12.47 19.83	Winston Šalem ection except i up the billing i all combinati USOC. For C	-Highpoint/Ch or nonrecurrin lifference.	arlotte-Gastoni ig charges for r rt network elem	a-Rock Hill); 1 not currently c	N (Nashville combined in or UNE Coin	FL and NC	in the NRC - (Currently Con	a flat rate us	age charge h.
This in Unbun The Tc BellSo Rates, The M End O(USOC For No Additic 2-WIRE UNE P UNE L	icludes: died port/loop combinations that are Currently Combined or I p 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderd, uth currently is developing the billing capability to mechanica BellSouth shall bill the rates in the Cost-Based section prece arket Rate for unbundled ports includes all available features i ffice and Tandem Switching Usage and Common Transport Us : URECU). t Currently Combined scenarios the Nonrecurring charges are conal NRCs may apply also and are categorized accordingly. E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 oop Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 3 Voice Grade Line Port (Res) [2-Wire voice unbundled port - residence	Not Cur ale, Mia ally bill ding in in all st sage rat	ami); G. the rec lieu of ates. tes in t in the 1 2 3 1 2 1 2	A (Atlanta); LA (Ne surring and non-re- the Market Rates a he Port section of First and Addition: UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	w Orleans); NC curring Market and reserves th this rate exhib al NRC column UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX	C (Greensboro-) Rates in this s- ne right to true- it shall apply to s for each Port 24.80 26.47 33.83 10.80 12.47 19.83 14.00	Winston Šalem ection except i up the billing i o all combinati- USOC. For C	-Highpoint/Ch or nonrecurrin difference. Dons of loop/po urrently Comb	arlotte-Gastoni ig charges for r rt network elem	a-Rock Hill); 1 not currently c	N (Nashville combined in or UNE Coin	FL and NC	in the NRC - (Currently Con	a flat rate us	age charge n.
This in Unbun The Tc BellSo Rates, The Mi End OU (USOC For No Additic 2-WIRE UNE P UNE Li	cludes: 11.1 died port/loop combinations that are Currently Combined or I op 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderd, uth currently is developing the billing capability to mechanica BellSouth shall bill the rates in the Cost-Based section precederated the arter of unbundled ports includes all available features i ffice and Tandem Switching Usage and Common Transport Us: : URECU). t Currently Combined scenarios the Nonrecurring charges are onal NRCs may apply also and are categorized accordingly. E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 oop Rates 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire Voice Grade Loop (SL1) - Zone 3 Voice Grade Loop (SL1) - Zone 3 2-Wire voice unbundled port vesidence 2-Wire voice unbundled port vesidence 2-Wire voice unbundled port vesidence 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res	Not Cur ale, Mia ally bill ding in in all st sage rat	ami); G. the rec lieu of ates. tes in t in the 1 2 3 1 2 1 2	A (Atlanta); LA (Ne surring and non-re- the Market Rates a he Port section of First and Addition UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	w Orleans); NC curring Market and reserves the this rate exhibe al NRC column UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRL UEPRC	C (Greensboro-V Rates in this s he right to true- it shall apply to s for each Port 24.80 26.47 33.83 10.80 12.47 19.83 14.00 14.00	Winston Šalem ection except i up the billing i o all combinati USOC. For C	-Highpoint/Ch or nonrecurrin lifference. ons of loop/po urrently Comb	arlotte-Gastoni ig charges for r rt network elem	a-Rock Hill); 1 not currently c	N (Nashville combined in or UNE Coin	FL and NC	0 Combination in the NRC - (33.67 33.67	Currently Con	a flat rate us nbined section	age charge h. 3.9 3.9 3.9
This in Unbun The Tc BellSo Rates, The Mi End OU (USOC For No Additic 2-WIRE UNE P UNE Li	cludes: 11.1 died port/loop combinations that are Currently Combined or I gled port/loop combinations that are Currently Combined or I g MSAs in BellSouth's region are: FL (Orlando, FL Lauderd) uth currently is developing the billing capability to mechanica BellSouth shall bill the rates in the Cost-Based section preceder arket Rate for unbundled ports includes all available features i fice and Tandem Switching Usage and Common Transport Us : URECU). t Currently Combined scenarios the Nonrecurring charges are onal NRCs may apply also and are categorized accordingly. 2 VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port with Caller D - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port with Caller ED - res 2-Wire voice unbundled port with usage line port without Caller ID <	Not Cur ale, Mia ally bill ding in in all st sage rat	ami); G. the rec lieu of ates. tes in t in the 1 2 3 1 2 1 2	A (Atlanta); LA (Ne surring and non-re- the Market Rates a he Port section of First and Addition: UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	w Orleans); NC curring Market and reserves th this rate exhib al NRC column UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC	C (Greensboro-) Rates in this s- ne right to true- it shall apply to s for each Port 24.80 26.47 33.83 10.80 12.47 19.83 14.00 14.00 14.00	Winston Šalem ection except i up the billing i b all combinati- USOC. For C	-Highpoint/Ch or nonrecurrin lifference. Dons of loop/po urrently Comb 90.00 90.00 90.00 90.00	arlotte-Gastoni ig charges for r rt network elem	a-Rock Hill); 1 not currently c	N (Nashville combined in or UNE Coin	FL and NC	2 Combination in the NRC - (33.67 33.67 33.67	Currently Con Currently Con 7.88 7.88 7.88 7.88	a flat rate us nbined section 11.17 11.17 11.17 11.17	age charge n. 3.9 3.9 3.9 3.9
This in Unbun The Tc BellSo Rates, The Mi End OU (USOC For No Additic 2-WIRE UNE P UNE Li	cludes: 11 died port/loop combinations that are Currently Combined or I pp 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdiuth currently is developing the billing capability to mechanica BellSouth shall bill the rates in the Cost-Based section preceder arket Rate for unbundled ports includes all available features i ffice and Tandem Switching Usage and Common Transport Us : URECU). t Currently Combined scenarios the Nonrecurring charges are onal NRCs may apply also and are categorized accordingly. E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 oop Rates 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire voice unbundled port residence 2-Wire voice unbundled port residence 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled Georgia basic dialing port without Caller ID (LUM) 2-Wire voice unbundled Georgia basic dialing port fo	Not Cur ale, Mia ally bill ding in in all st sage rat	ami); G. the rec lieu of ates. tes in t in the 1 2 3 1 2 1 2	A (Atlanta); LA (Ne surring and non-re the Market Rates a he Port section of First and Addition: UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	w Orleans); NC curring Market and reserves th this rate exhib al NRC column UEPLX UEPLX UEPLX UEPLX UEPLX UEPRL UEPRC UEPRO UEPAP	C (Greensboro-V Rates in this su- eright to true- it shall apply to s for each Port 24.80 26.47 33.83 10.80 12.47 19.83 10.80 12.47 19.83 10.80 12.47 19.83 10.80 12.47 19.83	Winston Šalem ection except i up the billing . b all combinati USOC. For Co USOC. For Co 90.00 90.00 90.00 90.00	-Highpoint/Ch or nonrecurrin lifference. Dons of loop/pc urrently Comb 90.00 90.00 90.00 90.00	arlotte-Gastoni ig charges for r rt network elem	a-Rock Hill); 1 not currently c	N (Nashville combined in or UNE Coin	FL and NC	2 Combination in the NRC - (33.67 33.67 33.67	7.88 7.88 7.88	e a flat rate us nbined section	age charge n. 3.97 3.97 3.97 3.97 3.91 3.91
This in Unbun The TC BellSo Rates, The M. End O(USOC For No Additic 2-WIRE UNE P UNE L	In the second se	Not Cur ale, Mia ally bill ding in in all st sage rat	ami); G. the rec lieu of ates. tes in t in the 1 2 3 1 2 1 2	A (Atlanta); LA (Ne surring and non-re- the Market Rates a he Port section of First and Addition: UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	w Orleans); NC curring Market and reserves the this rate exhibe al NRC column UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP UEPWC	C (Greensboro-) Rates in this s- ne right to true- it shall apply to s for each Port 24.80 26.47 33.83 10.80 12.47 19.83 14.00 14.00 14.00 14.00	Winston Šalem ection except i up the billing i b all combinati- USOC. For C USOC. For C 90.00 90.00 90.00 90.00 90.00	-Highpoint/Ch or nonrecurrin lifference. Dons of loop/pc urrently Comb 90.00 90.00 90.00 90.00 90.00 90.00	arlotte-Gastoni ig charges for r rt network elem	a-Rock Hill); 1 not currently c	N (Nashville combined in or UNE Coin	FL and NC	2 Combination in the NRC - 0 33.67 33.67 33.67 33.67 33.67	7.88 7.88 7.88 7.88	a flat rate us nbined section 11.17 11.17 11.17 11.17 11.17 11.17	age charge 1. 3.97 3.97 3.97 3.97 3.97 3.97 3.97 3.97
This in Unbun The TC BellSo Rates, The M. End OI (USOC For No Additic 2-WIRE UNE P UNE P UNE L 2-Wire	cludes: 1 ded port/loop combinations that are Currently Combined or I pp 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdiuth currently is developing the billing capability to mechanica BellSouth shall bill the rates in the Cost-Based section preceder arket Rate for unbundled ports includes all available features i ffice and Tandem Switching Usage and Common Transport Use: uRECU). t Currently Combined scenarios the Nonrecurring charges are onal NRCs may apply also and are categorized accordingly. E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 poop Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire voice Grade Loop (SL1) - Zone 2 2-Wire voice unbundled port vith Caller ID res 2-Wire voice unbundled port vith Caller ID res 2-Wire voice unbundled port vith Caller ID res 2-Wire voice unbundled port vith Caller ID res 2-Wire voice unbundled port vith Caller ID res 2-Wire voice unbundled port with Caller ID res 2-Wire voice unbundled port vith Caller ID res 2-Wire voice unbundled port vith Caller ID res 2-Wire voice unbundled port	Not Cur ale, Mia ally bill ding in in all st sage rat	ami); G. the rec lieu of ates. tes in t in the 1 2 3 1 2 1 2	A (Atlanta); LA (Ne surring and non-re the Market Rates a he Port section of First and Addition: UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	w Orleans); NC curring Market and reserves the this rate exhibe al NRC column UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRC UEPAP UEPWC UEPWQ	C (Greensboro-V Rates in this s- ne right to true- it shall apply to s for each Port 24.80 26.47 33.83 10.80 12.47 19.83 14.00 14.00 14.00 14.00 14.00	Winston Salem ection except i up the billing i b all combinati USOC. For C 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00	-Highpoint/Ch or nonrecurrin lifference. 	arlotte-Gastoni ig charges for r rt network elem	a-Rock Hill); 1 not currently c	N (Nashville combined in or UNE Coin	FL and NC	2 Combination in the NRC - 0 33.67 33.67 33.67 33.67 33.67 33.67	Currently Con Currently Con 7.88 7.88 7.88 7.88 7.88 7.88 7.88 7.8	a flat rate us nbined section 11.17 11.17 11.17 11.17 11.17 11.17 11.17	age charge n. 3.91 3.91 3.91 3.91 3.91 3.91 3.91
This in Unbun The Tc BellSo Rates, The Mi End O (USOC For No Additic 2-WIRE UNE P UNE L 2-WIRE 2-WIRE	cludes: 111 died port/loop combinations that are Currently Combined or I ps MSAs in BellSouth's region are: FL (Orlando, FL Lauderd, uth currently is developing the billing capability to mechanica BellSouth shall bill the rates in the Cost-Based section precederate trate for unbundled ports includes all available features i ffice and Tandem Switching Usage and Common Transport Us: : URECU). t Currently Combined scenarios the Nonrecurring charges are sonal NRCs may apply also and are categorized accordingly. 2 VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 3 voice Grade Loop (SL1) - Zone 3 Voice Grade Loop (SL1) - Zone 3 2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled Georgia basic dialing port without Caller ID (LUM) 2-Wire voice unbundled Georgia basic dialing port - outgoing only 2-Wire voice unbundled Georgia basic dialing port - outgoing only </td <td>Not Cur ale, Mia ally bill ding in in all st sage rat</td> <td>ami); G. the rec lieu of ates. tes in t in the 1 2 3 1 2 1 2</td> <td>A (Atlanta); LA (Ne surring and non-re- the Market Rates a he Port section of First and Addition: UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX</td> <td>w Orleans); NC curring Market and reserves the this rate exhibe al NRC column UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRC UEPRC UEPRC UEPWC UEPWQ UEPWR</td> <td>C (Greensboro-) Rates in this s ne right to true- it shall apply to s for each Port 24.80 26.47 33.83 10.80 12.47 19.83 14.00 14.00 14.00 14.00 14.00 14.00</td> <td>Winston Salem ection except i up the billing i b all combinati USOC. For C 0 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00</td> <td>-Highpoint/Ch or nonrecurrin lifference. ons of loop/pc urrently Comb 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00</td> <td>arlotte-Gastoni ig charges for r rt network elem</td> <td>a-Rock Hill); 1 not currently c</td> <td>N (Nashville combined in or UNE Coin</td> <td>FL and NC</td> <td>2 Combination in the NRC - (33.67 33.67 33.67 33.67 33.67 33.67</td> <td>7.88 7.88 7.88 7.88 7.88 7.88 7.88</td> <td>11.17 11.17 11.17 11.17 11.17 11.17 11.17</td> <td>age charge</td>	Not Cur ale, Mia ally bill ding in in all st sage rat	ami); G. the rec lieu of ates. tes in t in the 1 2 3 1 2 1 2	A (Atlanta); LA (Ne surring and non-re- the Market Rates a he Port section of First and Addition: UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	w Orleans); NC curring Market and reserves the this rate exhibe al NRC column UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRC UEPRC UEPRC UEPWC UEPWQ UEPWR	C (Greensboro-) Rates in this s ne right to true- it shall apply to s for each Port 24.80 26.47 33.83 10.80 12.47 19.83 14.00 14.00 14.00 14.00 14.00 14.00	Winston Salem ection except i up the billing i b all combinati USOC. For C 0 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00	-Highpoint/Ch or nonrecurrin lifference. ons of loop/pc urrently Comb 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00	arlotte-Gastoni ig charges for r rt network elem	a-Rock Hill); 1 not currently c	N (Nashville combined in or UNE Coin	FL and NC	2 Combination in the NRC - (33.67 33.67 33.67 33.67 33.67 33.67	7.88 7.88 7.88 7.88 7.88 7.88 7.88	11.17 11.17 11.17 11.17 11.17 11.17 11.17	age charge
This in Unbun The Tc BellSo Rates, The Mi End O (USOC For No Additic 2-WIRE UNE P UNE L 2-WIRE 2-WIRE	cludes: 11 died port/loop combinations that are Currently Combined or I ps 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderd) uth currently is developing the billing capability to mechanica BellSouth shall bill the rates in the Cost-Based section preceder arket Rate for unbundled ports includes all available features i ffice and Tandem Switching Usage and Common Transport Us : URECU). t Currently Combined scenarios the Nonrecurring charges are ponal NRCs may apply also and are categorized accordingly. EVOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 3 Voice Grade Loop (SL1) - Zone 3 Voice Grade Loop (SL1) - Zone 3 2-Wire voice unbundled port residence 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled Georgia basic dialing port without Caller ID 2-Wire voice unbundled Georgia basic dialing port or use with Caller ID - res 2-Wire voice unbundled Georgia basic dialing port - outgoing only 2-Wire voice unbundled Georgia basic dialing port - outgoing only	Not Cur ale, Mia ally bill ding in in all st sage rat	ami); G. the rec lieu of ates. tes in t in the 1 2 3 1 2 1 2	A (Atlanta); LA (Ne surring and non-re the Market Rates a he Port section of First and Addition UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	w Orleans); NC curring Market and reserves th this rate exhib al NRC column UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRC UEPRC UEPAP UEPWC UEPWC UEPWR	C (Greensboro-V Rates in this s- ne right to true- it shall apply to s for each Port 24.80 26.47 33.83 10.80 12.47 19.83 14.00 14.00 14.00 14.00 14.00 14.00 14.00	Winston Salem ection except i up the billing i b all combinati USOC. For C 0 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00	-Highpoint/Ch or nonrecurrin lifference. ons of loop/pc urrently Comb 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00 90.00	arlotte-Gastoni ig charges for r rt network elem	a-Rock Hill); 1 not currently c	N (Nashville combined in or UNE Coin	FL and NC	2 Combination in the NRC - (33.67 33.67 33.67 33.67 33.67 33.67	7.88 7.88 7.88 7.88 7.88 7.88 7.88	11.17 11.17 11.17 11.17 11.17 11.17 11.17	age charge n. 3.91 3.91 3.91 3.91 3.91 3.91 3.91 3.9
BUNDLED	D NETWORK ELEMENTS - Georgia			1								r -	Attachment:			bit: B
---------	---	-------------	----------	----------------	----------------	--------	--------	-----------	-------	---	----------	---	---	--	---	---
EGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge - Manual Sy Order vs. Electronic Disc Add
						Rec	Nonrec			g Disconnect				Rates(\$)		-
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
NONRE	CURRING CHARGES - CURRENTLY COMBINED															
														=		
	2-Wire Voice Grade Loop / Line Port Combination - Switch-as-is			UEPRX	USAC2		41.50	41.50					33.67	7.88	11.17	3.9
	2-Wire Voice Grade Loop / Line Port Combination - Switch with change			UEPRX	USACC		41.50	41.50					33.67	7.88	11.17	3.9
	ONAL NRCs			UEFRA	USACC		41.50	41.50					33.07	1.00	11.17	3.8
	NRC - 2-Wire Voice Grade Loop/Line Port Combination -															
	Subsequent			UEPRX	USAS2	0.00	0.00	0.00					33.67	7.88	11.17	3.9
	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)			-												
UNE Po	ort/Loop Combination Rates	1														1
	2-Wire VG Loop/Port Combo - Zone 1		1			24.80										
	2-Wire VG Loop/Port Combo - Zone 2		2			26.47										
	2-Wire VG Loop/Port Combo - Zone 3		3			33.83										
	op Rates	I	<u> </u>													L
	2-Wire Voice Grade Loop (SL1) - Zone 1	<u> </u>	1	UEPBX	UEPLX	10.80					<u> </u>					L
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPBX UEPBX	UEPLX UEPLX	12.47										
	2-Wire Voice Grade Loop (SL1) - Zone 3 Voice Grade Line Port (Bus)		3	UEPBA	UEPLA	19.83										
	2-Wire voice unbundled port without Caller ID - bus	-		UEPBX	UEPBL	14.00	90.00	90.00					33.67	7.88	11.17	3.
	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPBX	UEPBC	14.00	90.00	90.00					33.67	7.88	11.17	3.
	2-Wire voice unbundled port outgoing only - bus			UEPBX	UEPBO	14.00	90.00	90.00		1			33.67	7.88	11.17	3.
	2-Wire voice unbundled Georgia basic dialing port, without			02. 8/	02.00	1 1100	00.00	00.00					00.01	1.00		0.
	Caller ID capability - bus			UEPBX	UEPWD	14.00	90.00	90.00					33.67	7.88	11.17	3.9
	2-Wire voice unbundled Incoming Only Port without Caller ID			-	-											
	Capability			UEPBX	UEPBE	14.00	90.00	90.00					33.67	7.88	11.17	3.9
	2-Wire voice unbundled Georgia basic dialing port for use with															
	Caller ID - bus			UEPBX	UEPWP	14.00	90.00	90.00					33.67	7.88	11.17	3.9
	NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPBX	LNPCX	0.35										
FEATU														=		
	All Features Offered			UEPBX	UEPVF	0.00	0.00	0.00					33.67	7.88	11.17	3.
NUNKE	CURRING CHARGES - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop / Line Port Combination - Switch-as-is			UEPBX	USAC2		41.50	41.50					33.67	7.88	11.17	3.
	2-Wire Voice Grade Loop / Line Port Combination - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Switch with				00/102		41.50	41.50					33.07	7.00	11.17	5.
	change			UEPBX	USACC		41.50	41.50					33.67	7.88	11.17	3.
	ONAL NRCs	1								1			00.01			. 0.
	NRC - 2-Wire Voice Grade Loop/Line Port Combination -	1								1	İ		l			1
	Subsequent			UEPBX	USAS2		0.00	0.00					33.67	7.88	11.17	3.
	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)															
	ort/Loop Combination Rates	ļ	<u> </u>													ļ
	2-Wire VG Loop/Port Combo - Zone 1	 	1		_	24.80				ļ	ļ					
	2-Wire VG Loop/Port Combo - Zone 2	<u> </u>	2	+		26.47				ł – – – – – – – – – – – – – – – – – – –				-	-	ļ
	2-Wire VG Loop/Port Combo - Zone 3	+	3			33.83										
	op Rates 2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPRG	UEPLX	10.80				<u> </u>						
	2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2	+	2	UEPRG	UEPLX	12.47				<u> </u>						t
	2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3	-	3	UEPRG	UEPLX	12.47				1						
	Voice Grade Line Port Rates (RES - PBX)	1	Ť	1					1	ł			1			t
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port -	1									1					
	Res			UEPRG	UEPRD	14.00	90.00	90.00					33.67	7.88	11.17	3.
	2-Wire voice unbundled Georgia extended dialing port, PBX 1-	T														Γ
	Way Outdial Trunk			UEPRG	UEPPO	14.00	90.00	90.00					33.67	7.88	11.17	3.
	2-Wire voice unbundled Low Usage Line Port without Caller ID															
	Capability			UEPRX	UEPRT	14.00	90.00	90.00					33.67	7.88	11.17	3.
	NUMBER PORTABILITY	1	1													L
						a ·										
	Local Number Portability (1 per port)			UEPRG	LNPCP	3.15	0.00	0.00								

BUNDLE	D NETWORK ELEMENTS - Georgia	_											Attachment:			ibit: B
TEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge
						Rec	Nonrec			g Disconnect				Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
NONR	ECURRING CHARGES - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is			UEPRG	USAC2		41.50	41.50					33.67	7.88	11.17	3.9
	2-Wire Voice Grade Loop/ Line Port Combination - Switch with															
	Change			UEPRG	USACC		41.50	41.50					33.67	7.88	11.17	3.9
ADDIT	IONAL NRCs															
	2 Wire Loop/Line Side Port Combination - Non feature -															
	Subsequent Activity- Nonrecurring						0.00	0.00					33.67	7.88	11.17	3.9
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt															
	Group						14.64	14.64					19.99	19.99	19.99	19.9
	E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)															
UNE P	ort/Loop Combination Rates					0.1.00										
	2-Wire VG Loop/Port Combo - Zone 1		1			24.80										
	2-Wire VG Loop/Port Combo - Zone 2		2			26.47				ł						L
	2-Wire VG Loop/Port Combo - Zone 3		3			33.83										
UNEL	oop Rates					40.00				-						
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPPX	UEPLX	10.80										
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPPX	UEPLX	12.47										
0.14/	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPPX	UEPLX	19.83				-						
2-Wire	Voice Grade Line Port Rates (BUS - PBX)															
														=		
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPPX	UEPPC	14.00	90.00	90.00					33.67	7.88	11.17	3.
	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPPX	UEPPO	14.00	90.00	90.00					33.67	7.88	11.17	3
	Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPPX	UEPP1	14.00	90.00	90.00					33.67	7.88	11.17	3.
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPPX	UEPLD	14.00	90.00	90.00		-			33.67	7.88	11.17	3.
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPPX	UEPXA	14.00	90.00	90.00		-			33.67	7.88	11.17	
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPPX	UEPXB	14.00	90.00	90.00					33.67	7.88	11.17	
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPPX	UEPXC	14.00	90.00	90.00					33.67	7.88	11.17	3.
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPPX	UEPXD	14.00	90.00	90.00		-			33.67	7.88	11.17	3
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD				UEPXE	44.00	00.00	00.00					33.67	7.00		
	Capable Port		-	UEPPX	UEPXE	14.00	90.00	90.00					33.67	7.88	11.17	3.
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port			UEPPX	UEPXL	14.00	90.00	90.00					33.67	7.88	11.17	3
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy		-	UEPPX	UEPAL	14.00	90.00	90.00					33.67	7.88	11.17	3
						44.00	00.00	00.00					00.07	7.00		
	Room Calling Port			UEPPX	UEPXM	14.00	90.00	90.00		-			33.67	7.88	11.17	3
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port			UEPPX	UEPXO	14.00	90.00	90.00					33.67	7.88	11.17	з
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPPX	UEPXS	14.00	90.00	90.00					33.67	7.88	11.17	3
	2-Wire voice onbundled 1-Way Outgoing PBA Measured Port 2-Wire voice unbundled Georgia basic dialing port - 1-Way			UEFFA	UEFAS	14.00	90.00	90.00					33.07	1.00	11.17	3
	Oudial Trunk			UEPPX	UEPWS	14.00	90.00	90.00					33.67	7.88	11.17	3
	2-Wire voice unbundled Georgia basic dialing port - 2-Way			UEFFA	UEPW3	14.00	90.00	90.00		ł	1		33.07	1.00	11.17	3
	Trunk			UEPPX	UEPWT	14.00	90.00	00.00					33.67	7.88	11.17	3
	2-Wire voice unbundled Georgia basic dialing port - 2-way PBX				JEPWI	14.00	90.00	90.00					33.67	7.88	11.17	3
	Trunk			UEPPX	UEPPQ	14.00	90.00	90.00					33.67	7.88	11.17	3
	2-Wire voice unbundled Georgia basic dialing port - PBX LD			ULFFA	ULFFQ	14.00	90.00	90.00	1	<u> </u>			33.07	1.00	11.17	3
	Terminal Ports			UEPPX	UEPPS	14.00	90.00	90.00		1			33.67	7.88	11.17	3
	2-Wire voice unbundled Georgia basic dialing port - PBX Toll			ULFFA	ULFFO	14.00	90.00	90.00	ł	ł	<u> </u>		33.07	1.00	11.17	3
	Terminal Ports			UEPPX	UEPPT	14.00	90.00	90.00					33.67	7.88	11.17	3
	2-Wire voice unbundled Georgia basic dialing port - PBX LD				JEFFI	14.00	90.00	50.00		ł	<u> </u>		55.67	1.00	11.17	3
	DDD Terminal Port			UEPPX	UEPPU	14.00	90.00	90.00		1			33.67	7.88	11.17	3
	2-Wire voice unbundled Georgia basic dialing port - PBX LD				JEITO	14.00	30.00	30.00					55.07	1.00	11.17	
	Terminal Switchboard Port			UEPPX	UEPPV	14.00	90.00	90.00					33.67	7.88	11.17	3
	2-Wire voice unbundled Georgia basic dialing port - PBX LD				JLIIV	14.00	30.00	30.00		 	-		55.07	1.00	11.17	
	Terminal Switchboard DDD Capable Port			UEPPX	UEPPW	14.00	90.00	90.00					33.67	7.88	11.17	3
1004	L NUMBER PORTABILITY				0211 11	14.00	30.00	30.00		1	-		55.07	1.00	11.17	
LUCA	Local Number Portability (1 per port)			UEPPX	LNPCP	3.15	0.00	0.00		 	-					1
						5.15	0.00	0.00		1	-					1
FEATI																1
FEAT	All Features Offered			UEPPX	UEPVF	0.00	0.00	0.00					33.67	7.88	11.17	3.

SUNDLED	NETWORK ELEMENTS - Georgia	-	1	1									Attachment:			ibit: B
EGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Charge Manual S Order v
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMA
	Wire Maine Crede Lean / Line Bert Combination - Cuitab As Is			UEPPX	USAC2		41.50	44 50					33.67	7.00	44.47	2
	-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is -Wire Voice Grade Loop/ Line Port Combination - Switch with			UEPPX	USACZ		41.50	41.50					33.67	7.88	11.17	3.
	Change			UEPPX	USACC		41.50	41.50					33.67	7.88	11.17	3.
	NAL NRCs			0EITX	00/100		41.00	41.00					00.01	1.00		0.
2	-Wire Voice Grade Loop/ Line Port Combination - Subsequent			UEPPX	USAS2	0.00	0.00	0.00					33.67	7.88	11.17	3.
	Wire Loop/Line Side Port Combination - Non feature -															
	Subsequent Activity- Nonrecurring						0.00	0.00					33.67	7.88	11.17	3
	BX Subsequent Activity - Change/Rearrange Multiline Hunt															
	Group						14.64	14.64					19.99	19.99	19.99	19
	OICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PO	RT														
	t/Loop Combination Rates -Wire VG Coin Port/Loop Combo – Zone 1		1			24.80										
	-Wire VG Coin Port/Loop Combo – Zone 1		2	-		24.80										
	-Wire VG Coin Port/Loop Combo – Zone 2	-	2	1		33.83										
UNE Loo			5			55.65										1
	-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPCO	UEPLX	10.80										
	-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPCO	UEPLX	12.47										
2	-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPCO	UEPLX	19.83										
2-Wire Vo	oice Grade Line Port Rates (Coin)															
	-Wire Coin 2-Way with Operator Screening (GA)			UEPCO	UEPGC	14.00	90.00	90.00					33.67	7.88	11.17	
	-Wire Coin 2-Way with Operator Screening and Blocking: 011,															
	00/976, 1+DDD (GA)			UEPCO	UEP2G	14.00	90.00	90.00					33.67	7.88	11.17	:
	-Wire Coin 2-Way with Operator Screening and 011 Blocking													=		
	GA)			UEPCO	UEPGA	14.00	90.00	90.00					33.67	7.88	11.17	3
	-Wire Coin 2-Way with Operator Screening and 900/976 Blocking (GA)			UEPCO	UEPGB	14.00	90.00	90.00					33.67	7.88	11.17	:
	-Wire Coin 2-Way with Operator Screening and Blocking:			ULFCO	ULFGB	14.00	90.00	90.00					33.07	7.00	11.17	· ·
	00/976, 1+DDD, 011+, and Local (GA)			UEPCO	UEPCH	14.00	90.00	90.00					33.67	7.88	11.17	
	-Wire Coin Outward with Operator Screening and 011Blocking			02100	OEI OII	14.00	50.00	00.00					00.07	7.00		
	GA, KY, MS)			UEPCO	UEPRJ	14.00	90.00	90.00					33.67	7.88	11.17	:
	-Wire Coin Outward with Operator Screening and Blocking:															
	00/976, 1+DDD, 011+, and Local (FL, GA)			UEPCO	UEPCQ	14.00	90.00	90.00					33.67	7.88	11.17	
	IUMBER PORTABILITY															
	ocal Number Portability (1 per port)			UEPCO	LNPCX	0.35										
NONREC	URRING CHARGES - CURRENTLY COMBINED															
	Wire Voice Crade Loop/Line Bat Carthingting Curity has be		1				44 50	44 50					22.07	7.00	44.47	1
	-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is -Wire Voice Grade Loop/ Line Port Combination - Switch with	+		UEPCO	USAC2		41.50	41.50					33.67	7.88	11.17	
	-wire voice Grade Loop/ Line Port Combination - Switch with Change			UEPCO	USACC		41.50	41.50					33.67	7.88	11.17	
	NAL NRCs	1			00/00		41.50	41.30					55.07	1.00	11.17	<u> </u>
		1		1												1
2	-Wire Voice Grade Loop/ Line Port Combination - Subsequent		1	UEPCO	USAS2		0.00	0.00					33.67	7.88	11.17	
2-WIRE V	/OICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRI	E LINE I	PORT (RES)												
	t/Loop Combination Rates															
	-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1	ļ		30.84										
	-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			33.45										
	-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			44.92										<u> </u>
UNE Loo			1		UECF2	40.04										+
	-Wire Voice Grade Loop (SL2) - Zone 1 -Wire Voice Grade Loop (SL2) - Zone 2		1	UEPFR UEPFR	UECF2	16.84 19.45									ł	+
	-Wire Voice Grade Loop (SL2) - Zone 2 -Wire Voice Grade Loop (SL2) - Zone 3	+		UEPFR	UECF2 UECF2	30.92			1	1				-	ł	
	oice Grade Line Port Rates (Res)	+				50.52										+
	-Wire voice unbundled port - residence	1		UEPFR	UEPRL	14.00	160.00	125.00	1	1			33.67	7.88	11.17	1
	-Wire voice unbundled port with Caller ID - res			UEPFR	UEPRC	14.00	160.00	125.00					37.06	7.88	11.17	
	-Wire voice unbundled port outgoing only - res	1	1	UEPFR	UEPRO	14.00	160.00	125.00					33.67	7.88	11.17	
2	-whe voice unbundled port outgoing only - res															

INBUNDLED NE	ETWORK ELEMENTS - Georgia	1									-	-	Attachment:			ibit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						Bee	Nonrec	urring	Nonrecurring	g Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	ire voice unbundled Georgia basic dialing port, without															
Calle	er ID capability - res			UEPFR	UEPWC	14.00	160.00	125.00					33.67	7.88	11.17	3.91
	ire voice unbundled Georgia basic dialing port for use with er ID - res			UEPFR	UEPWQ	14.00	160.00	125.00					33.67	7.88	11.17	3.9
2-Wi	ire voice unbundled Georgia basic dialing port - outgoing			UEPFR	UEPWR	14.00	160.00	125.00					33.67	7.88	11.17	3.9
INTEROFFIC	CE TRANSPORT			-												
Intere	roffice Transport - Dedicated - 2 Wire Voice Grade - Facility															
Term	nination roffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile			UEPFR	U1TV2	17.07	79.61	36.08								
	raction Mile			UEPFR	1L5XX	0.0222										
FEATURES									1		1		İ	1		1
	eatures Offered			UEPFR	UEPVF	0.00	0.00	0.00					33.67	7.88	11.17	3.9
	MBER PORTABILITY										1					
	al Number Portability (1 per port)	1		UEPFR	LNPCX	0.35					1		İ	l	ĺ	
	RING CHARGES (NRCs) - CURRENTLY COMBINED		İ													1
	ire Loop / Dedicated IO Transport / 2 Wire Line Port	1														
	nbination - Conversion - Switch-as-is			UEPFR	USAC2		93.83	93.83					33.67	7.88	11.17	3.9
2-Wi	ire Loop / Dedicated IO Transport / 2 Wire Line Port															
Com	bination - Conversion - Switch-With-Change			UEPFR	USACC		93.83	93.83					33.67	7.88		
2-WIRE VOI	CE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE	E LINE F	PORT (I	BUS)												
	oop Combination Rates															
	ire VG Loop/IO Tranport/Port Combo - Zone 1		1			30.84										
	ire VG Loop/IO Tranport/Port Combo - Zone 2		2			33.45										
	ire VG Loop/IO Tranport/Port Combo - Zone 3		3			44.92										
UNE Loop R																
	ire Voice Grade Loop (SL2) - Zone 1		1	UEPFB	UECF2	16.84										
	ire Voice Grade Loop (SL2) - Zone 2		2	UEPFB	UECF2	19.45										
	ire Voice Grade Loop (SL2) - Zone 3		3	UEPFB	UECF2	30.92										
	e Grade Line Port (Bus)															
	ire voice unbundled port without Caller ID - bus			UEPFB	UEPBL	14.00	160.00	125.00					33.67	7.88	11.17	3.
	ire voice unbundled port with Caller + E484 ID - bus			UEPFB	UEPBC	14.00	160.00	125.00					33.67	7.88	11.17	3.
	ire voice unbundled port outgoing only - bus			UEPFB	UEPBO	14.00	160.00	125.00					33.67	7.88	11.17	3.
	ire voice unbundled incoming only port with Caller ID - Bus			UEPFB	UEPB1	14.00	160.00	125.00					33.67	7.88	11.17	3.
Calle	ire voice unbundled Georgia basic dialing port, without er ID capability - bus			UEPFB	UEPWD	14.00	160.00	125.00					33.67	7.88	11.17	3.
	ire voice unbundled Georgia basic dialing port for use with															
	er ID - bus			UEPFB	UEPWP	14.00	160.00	125.00					33.67	7.88	11.17	3.
	MBER PORTABILITY															
	al Number Portability (1 per port)			UEPFB	LNPCX	0.35										
	CE TRANSPORT															
Term	roffice Transport - Dedicated - 2 Wire Voice Grade - Facility nination			UEPFB	U1TV2	17.07	79.61	36.08								
	roffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile raction Mile			UEPFB	1L5XX	0.0222										
FEATURES			l						1		1		l	l	ĺ	
All F	eatures Offered			UEPFB	UEPVF	0.00	0.00	0.00					33.67	7.88	11.17	3.
NONRECUR	RING CHARGES (NRCs) - CURRENTLY COMBINED															
	ire Loop / Dedicated IO Transport / 2 Wire Line Port															
	bination - Conversion - Switch-as-is			UEPFB	USAC2		93.83	93.83					33.67	7.88	11.17	3.
	ire Loop / Dedicated IO Transport / 2 Wire Line Port															
Com	bination - Conversion - Switch with change	1		UEPFB	USACC		93.83	93.83								1
	CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)															
UNE Port/Lo	oop Combination Rates					_										
	ire VG Loop/IO Tranport/Port Combo - Zone 1		1			30.84										
	ire VG Loop/IO Tranport/Port Combo - Zone 2		2			33.45										
2-Wi	ire VG Loop/IO Tranport/Port Combo - Zone 3		3			44.92										
UNE Loop R	Rates															
	ire Voice Grade Loop (SL2) - Zone 1		1	UEPFP	UECF2	16.84										

UNBUNDL	ED NETWORK ELEMENTS - Georgia												Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring D					Rates(\$)		
			0		115050		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Loop (SL2) - Zone 2 2-Wire Voice Grade Loop (SL2) - Zone 3		2	UEPFP UEPFP	UECF2 UECF2	19.45 30.92										
2-Wir	e Voice Grade Line Port Rates (BUS - PBX)		3	UEFFF	UECF2	30.92										i
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPFP	UEPPC	14.00	160.00	125.00					33.67	7.88	11.17	3.91
	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPFP	UEPPO	14.00	160.00	125.00					33.67	7.88	11.17	3.91
	Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPFP	UEPP1	14.00	160.00	125.00					33.67	7.88	11.17	3.91
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPFP	UEPLD	14.00	160.00	125.00					33.67	7.88	11.17	3.91
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPFP	UEPXA	14.00	160.00	125.00					37.06	7.88	11.17	3.91
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPFP	UEPXB	14.00	160.00	125.00					33.67	7.88	11.17	3.91
├ ── ├ ──	2-Wire Voice Unbundled PBX LD DDD Terminals Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPFP UEPFP	UEPXC UEPXD	14.00 14.00	160.00 160.00	125.00 125.00	<u>├</u>				33.67 33.67	7.88 7.88	<u>11.17</u> 11.17	3.91 3.91
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD				ULFAD	14.00	100.00	125.00	├				33.07	1.08	11.17	3.91
	Capable Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy		<u> </u>	UEPFP	UEPXE	14.00	160.00	125.00					33.67	7.88	11.17	3.91
	Administrative Calling Port			UEPFP	UEPXL	14.00	160.00	125.00					33.67	7.88	11.17	3.91
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port			UEPFP	UEPXM	14.00	160.00	125.00					33.67	7.88	11.17	3.91
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital					11.00	100.00	405.00					33.67	7.00	44.47	2.01
	Discount Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPFP UEPFP	UEPXO UEPXS	14.00 14.00	160.00 160.00	125.00 125.00					33.67	7.88 7.88	11.17 11.17	3.91 3.91
	2-Wire voice unbundled Georgia basic dialing port - 1-Way			ULFIF	ULF X3	14.00	100.00	123.00					33.07	7.00	11.17	3.91
	Oudial Trunk			UEPFP	UEPWS	14.00	160.00	125.00					33.67	7.88	11.17	3.91
	2-Wire voice unbundled Georgia basic dialing port - 2-Way Trunk			UEPFP	UEPWT	14.00	160.00	125.00					33.67	7.88	11.17	3.91
LOCA	AL NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPFP	LNPCP	3.15	0.00	0.00					33.67	7.88	11.17	3.91
INTE	ROFFICE TRANSPORT															l
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility Termination			UEPFP	U1TV2	17.07	79.61	36.08								
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile or Fraction Mile			UEPFP	1L5XX	0.0222										
FEAT	URES													=		
NON	All Features Offered			UEPFP	UEPVF	0.00	0.00	0.00					33.67	7.88	11.17	3.91
NON	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED 2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
	Combination - Conversion - Switch-as-is			UEPFP	USAC2		93.83	93.83					33.67	7.88	11.17	3.91
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port			OLITI	00/02		33.03	33.05					33.07	7.00	11.17	3.31
	Combination - Conversion - Switch with change			UEPFP	USACC		93.83	93.83					33.67	7.88	11.17	3.91
	PORT/LOOP COMBINATIONS - MARKET BASED RATES															
	RE VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK	PORT														1
UNE	Port/Loop Combination Rates					00.01										l
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1		1			99.84										ł
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2 2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3		2			102.45 113.92										
UNE	Loop Rates		3			113.92										İ
UNL	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1		1	UEPPX	UECD1	16.84	104.78	78.10								
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2		2	UEPPX	UECD1	19.45	104.78	78.10	<u> </u>							1
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3		3	UEPPX	UECD1	30.92	104.78	104.10								
UNE	Port Rate															
	Exchange Ports - 2-Wire DID Port			UEPPX	UEPD1	83.00	850.00	75.00					33.67	7.88		I
NON	RECURRING CHARGES - CURRENTLY COMBINED		I								ļ	ļ				I
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination - Switch-As-Is Top 8 MSAs only			UEPPX	USAC1		850.00	75.00					33.67	7.88		
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth Allowable Changes Top 8 MSAs only			UEPPX	USA1C		850.00	75.00					33.67	7.88		
	TIONAL NRCs															
Telep	hone Number/Trunk Group Establisment Charges															L
	DID Trunk Termination (One Per Port)		1	UEPPX	NDT	0.00	0.00	0.00								 '

UNBUNDLI	ED NETWORK ELEMENTS - Georgia	1	1				1							Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	E	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'I
							Rec	Nonrec	urring	Nonrecurring	g Disconnect			OSS	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	DID Numbers, Establish Trunk Group and Provide First Group																
	of 20 DID Numbers			UEPPX		NDZ	0.00	0.00	0.00								
	Additional DID Numbers for each Group of 20 DID Numbers			UEPPX		ND4	0.00	0.00	0.00								
	DID Numbers, Non- consecutive DID Numbers , Per Number			UEPPX		ND5	0.00	0.00	0.00								
	Reserve Non-Consecutive DID numbers			UEPPX		ND6	0.00	0.00	0.00								
	Reserve DID Numbers			UEPPX		NDV	0.00	0.00	0.00								
LOCA						INDOD	0.45	0.00	0.00								
0 14/15	Local Number Portability (1 per port) RE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LI		DODI	UEPPX		LNPCP	3.15	0.00	0.00								
		NE SIDE	POR	1													
UNE	Port/Loop Combination Rates 2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -																
	UNE Zone 1		1	UEPPB	UEPPR		81.89										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -		-	UEFFD	UEFFR		01.09										
	UNE Zone 2		2	UEPPB	UEPPR		85.27										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -		~	ULFFD	ULFER		05.27										ł
	UNE Zone 3		3	UEPPB	UEPPR		100.17										
	Loop Rate		3	ULITU	OLITIK		100.17										
0.112	2-Wire ISDN Digital Grade Loop - UNE Zone 1		1	UEPPB	UEPPR	USL2X	21.89	252.32	188.77					19.99	19.99		
				02.10	02.111	00LL/	200	202.02	100.11					10.00	10.00		<u> </u>
	2-Wire ISDN Digital Grade Loop - UNE Zone 2		2	UEPPB	UEPPR	USL2X	25.27	252.32	188.77					19.99	19.99		
	2-Wire ISDN Digital Grade Loop - UNE Zone 3		3	UEPPB	UEPPR	USL2X	40.17	252.32	188.77					19.99	19.99		
UNE I	Port Rate				-												
	Exchange Port - 2-Wire ISDN Line Side Port			UEPPB	UEPPR	UEPPB	60.00	525.00	400.00					19.99	19.99		
NONF	RECURRING CHARGES - CURRENTLY COMBINED																
	2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port																
	Combination - Conversion - Top 8 MSAs only			UEPPB	UEPPR	USACB	0.00	215.00	215.00					19.99	19.99		
ADDI	TIONAL NRCs																
	2-Wire ISDN Loop / 2-Wire ISDN Port Combination - Sub Actvy	ł															
	Non Feature/Add Trunk			UEPPB	UEPPR	USASB		165.95						19.99	19.99		
LOCA	L NUMBER PORTABILITY																
	Local Number Portability (1 per port)			UEPPB	UEPPR	LNPCX	0.35	0.00	0.00								
B-CH	ANNEL USER PROFILE ACCESS:																
	CVS/CSD (DMS/5ESS)			UEPPB	UEPPR	U1UCA	0.00	0.00	0.00								
	CVS (EWSD)			UEPPB	UEPPR	U1UCB	0.00	0.00	0.00								
D CI	CSD ANNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS S		TNI	UEPPB	UEPPR	U1UCC	0.00	0.00	0.00								
	REAL PLUS USER PROFILE ACCESS: (AL, KT, LA, MS S	C,IVIS, 6	(T N)														
USER	User Terminal Profile (EWSD only)			UEPPB	UEPPR	U1UMA	0.00	0.00	0.00								
VERT	ICAL FEATURES			ULFFD	OLFER	OTOWA	0.00	0.00	0.00								ł
VENT	All Vertical Features - One per Channel B User Profile			UEPPB	UEPPR	UEPVF	0.00	0.00	0.00					19.99	19.99		
INTER	ROFFICE CHANNEL MILEAGE			OLITO	OLITIK		0.00	0.00	0.00					13.33	13.33		
	Interoffice Channel mileage each, including first mile and																
	facilities termination			UEPPB	UEPPR	M1GNC	16.47	79.61	36.08					19.99	19.99		
	Interoffice Channel mileage each, additional mile				UEPPR	M1GNM	0.0222	0.00	0.00								
4-WIF	RE DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUN	K PORT															
	Port/Loop Combination Rates																
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE	1				1											
	Zone 1		1	UEPPP		I	955.53										
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE	T															
	Zone 2		2	UEPPP			964.13										
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE	1															
	Zone 3		3	UEPPP			1,001.93										
UNE I	Loop Rates																
	4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPPP		USL4P	55.53	448.92	276.60					19.99	19.99		
	4-Wire DS1 Digital Loop - UNE Zone 2		2	UEPPP		USL4P	64.13	448.92	276.60					19.99	19.99		I
	4-Wire DS1 Digital Loop - UNE Zone 3	I	3	UEPPP		USL4P	101.93	448.92	276.60					19.99	19.99		
UNE	Port Rate	I															
	Exchange Ports - 4-Wire ISDN DS1 Port			UEPPP		UEPPP	900.00	1,200.00	1,200.00					19.99	19.99		<u> </u>
NONF	RECURRING CHARGES - CURRENTLY COMBINED	1		1		1						1					L

UNBUNDLED N	IETWORK ELEMENTS - Georgia											1	Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'l
						Rec	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Vire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port															
	mbination - Conversion -Switch-As-Is Top 8 MSAs only			UEPPP	USACP	0.00	925.00	925.00					19.99	19.99		
ADDITION																
	Vire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy-				PR7TF		0.0000									
INW 4 V	vard/two way Telephone Numbers (except NC) Vire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port -			UEPPP	PR/IF		0.9686									
	itward Tel Numbers (All States except NC)			UEPPP	PR7TO		22.75	22.75								
	Vire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port -			OLITI	110/10		22.15	22.15								
	bsequent Inward Telephone Numbers			UEPPP	PR7ZT		45.49	45.49								
	IMBER PORTABILITY															
	cal Number Portability (1 per port)			UEPPP	LNPCN	1.75										
INTERFAC	E (Provsioning Only)															
	ice/Data			UEPPP	PR71V	0.00	0.00	0.00								
	gital Data			UEPPP	PR71D	0.00	0.00	0.00								
	vard Data			UEPPP	PR71E	0.00	0.00	0.00								
	ditional "B" Channel															
	w or Additional - Voice/Data B Channel			UEPPP	PR7BV	0.00	28.71						19.99	19.99		
	w or Additional - Digital Data B Channel			UEPPP	PR7BF	0.00	28.71						19.99	19.99		
	w or Additional Inward Data B Channel			UEPPP	PR7BD	0.00	28.71						19.99	19.99		
CALL TYP					00704	0.00	0.00	0.00								
	vard Itward			UEPPP UEPPP	PR7C1 PR7C0	0.00	0.00	0.00								
	itward o-way			UEPPP	PR7C0 PR7CC	0.00	0.00	0.00								
	Channel Mileage			ULFFF	FRICC	0.00	0.00	0.00								
	ed Each Including First Mile			UEPPP	1LN1A	78.9223	147.07	111.75	0.00				19.99	19.99		
	ch Airline-Fractional Additional Mile			UEPPP	1LN1B	0.4523	147.07	111.70	0.00				10.00	10.00		
	S1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT					0.10-0										
	Loop Combination Rates															
4W	/ DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1		1	UEPDC		176.33										
	/ DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2		2	UEPDC		184.93										
	/ DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3		3	UEPDC		222.73										
UNE Loop																
	Vire DS1 Digital Loop - UNE Zone 1		1	UEPDC	USLDC	55.53	448.92	276.00					19.99	19.99		
	Vire DS1 Digital Loop - UNE Zone 2		2	UEPDC	USLDC	64.13	448.92	276.60					19.99	19.99		
	Vire DS1 Digital Loop - UNE Zone 3		3	UEPDC	USLDC	101.93	448.92	276.60					19.99	19.99		
UNE Port F					UDD1T	750.00	4 044 42	477.07	206.70	20.70			19.99	19.99		
	Vire DDITS Digital Trunk Port RRING CHARGES - CURRENTLY COMBINED			UEPDC	UDDII	750.00	1,011.43	477.87	206.70	20.70			19.99	19.99		
	Vire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
	witch-As-Is Top 8 MSAs only			UEPDC	USAC4		269.96	269.96					19.99	19.99		
				021 00	00/104		200.00	200.00					10.00	10.00		
4-V	Vire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
	conversion with DS1 Changes Top 8 MSAs only			UEPDC	USAWA		269.96	269.96					19.99	19.99		
	0, , ,															
4-V	Vire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
	conversion with Change - Trunk Top 8 MSAs only			UEPDC	USAWB		269.96	269.96					19.99	19.99		
ADDITION																
	Vire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent															
	rvice Activity Per Service Order			UEPDC	USAS4		147.47	147.47								
	Vire DS1 Loop / 4-Wire DDITS Trunk Port - NRC -	1					aa = :						10	10		
	bsequent Channel Activation/Chan - 2-Way Trunk			UEPDC	UDTTA		28.71	28.71					19.99	19.99		
	Vire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent	1					00.71	00.71					10.00	10.00		
	annel Activation/Chan - 1-Way Outward Trunk Vire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel			UEPDC	UDTTB		28.71	28.71					19.99	19.99		
	Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel tivation/Chan Inward Trunk w/out DID				UDTTC		00 74	20 74					19.99	19.99		
	tivation/Chan Inward Trunk w/out DID Vire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan			UEPDC	UDITC		28.71	28.71					19.99	19.99		
	tivation Per Chan - Inward Trunk with DID			UEPDC	UDTTD		28.71	28.71					19.99	19.99		
	Vire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan			561 00			20.71	20.71			1		15.39	19.99		
	tivation / Chan - 2-Way DID w User Trans	1		UEPDC	UDTTE		28.71	28.71					19.99	19.99		

INBUNDLI	ED NETWORK ELEMENTS - Georgia												Attachment:			bit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
							Nonrec	urring	Nonrecurring	Disconnect			055	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN		SOMAN	SOMAN	SOMAN
BIPO	LAR 8 ZERO SUBSTITUTION															
	B8ZS -Superframe Format			UEPDC	CCOSF		0.00	600.00								
	B8ZS - Extended Superframe Format			UEPDC	CCOEF		0.00	600.00								
Alterr	nate Mark Inversion															
	AMI -Superframe Format			UEPDC	MCOSF		0.00	0.00								
	AMI - Extended SuperFrame Format			UEPDC	MCOPO		0.00	0.00								
Telep	hone Number/Trunk Group Establisment Charges															
	Telephone Number for 2-Way Trunk Group			UEPDC	UDTGX	0.00										
	Telephone Number for 1-Way Outward Trunk Group			UEPDC	UDTGY	0.00										
	Telephone Number for 1-Way Inward Trunk Group Without DID			UEPDC	UDTGZ	0.00										
	DID Numbers, Establish Trunk Group and Provide First Group															
	of 20 DID Numbers			UEPDC	NDZ	0.00	0.00	0.00								
	DID Numbers for each Group of 20 DID Numbers			UEPDC	ND4	0.00										
	DID Numbers, Non- consecutive DID Numbers, Per Number			UEPDC	ND5	0.00										
	Reserve Non-Consecutive DID Nos.			UEPDC	ND6	0.00	0.00	0.00								
	Reserve DID Numbers			UEPDC	NDV	0.00	0.00	0.00								
	cated DS1 (Interoffice Channel Mileage) -															
FX/FC	CO for 4-Wire DS1 Digital Loop with 4-Wire DDITS Trunk Port															
	Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities															
	Termination)			UEPDC	1LNO1	78.47	147.07	111.75					19.99	19.99		
	Interoffice Channel Mileage - Additional rate per mile - 0-8 miles			UEPDC	1LNOA	0.4523	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities															
	Termination)			UEPDC	1LNO2	0.00	0.00	0.00								
	Interoffice Channel Mileage - Additional rate per mile - 9-25															
	miles			UEPDC	1LNOB	0.4523	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities															
	Termination)			UEPDC	1LNO3	0.00	0.00	0.00								
	Interoffice Channel Mileage - Additional rate per mile - 25+ miles	i		UEPDC	1LNOC	0.4523	0.00	0.00								
	Local Number Portability, per DS0 Activated			UEPDC	LNPCP	3.15										
	Central Office Termininating Point			UEPDC	CTG	0.00										
	RE DS1 LOOP WITH CHANNELIZATION WITH PORT															
	m is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Act															
	tem can have various rate combinations based on type and nu	mber of	ports	used												
UNE	DS1 Loop		<u> </u>													
	4-Wire DS1 Loop - UNE Zone 1		1	UEPMG	USLDC	55.53	0.00	0.00								
	4-Wire DS1 Loop - UNE Zone 2		2	UEPMG UEPMG	USLDC USLDC	64.13	0.00	0.00								<u> </u>
	4-Wire DS1 Loop - UNE Zone 3	1	3	UEPMG	USLDC	101.93	0.00	0.00								<u> </u>
UNEI	DSO Channelization Capacities (D4 Channel Bank Configuratio	ons)	-	UEPMG	VUM24	102.64	0.00	0.00					10.00	10.00		
	24 DSO Channel Capacity - 1 per DS1				VUM24 VUM48		0.00	0.00					19.99 19.99	19.99 19.99		
	48 DSO Channel Capacity - 1 per 2 DS1s			UEPMG	VUM48 VUM96	205.28	0.00	0.00					19.99	19.99		
	96 DSO Channel Capacity -1per 4 DS1s 144 DS0 Channel Capacity - 1 per 6 DS1s		-		VUM96 VUM14	410.56 615.84	0.00	0.00				-	19.99	19.99		
	144 DS0 Channel Capacity - 1 per 6 DS1s 192 DS0 Channel Capacity -1 per 8 DS1s	+	<u> </u>		VUM14 VUM19	615.84 821.12	0.00	0.00					19.99	19.99		
	240 DS0 Channel Capacity - 1 per 8 DS1s 240 DS0 Channel Capacity - 1 per 10 DS1s			UEPMG	VUM19 VUM20	1,026.40	0.00	0.00					19.99	19.99		
	288 DS0 Channel Capacity - 1 per 10 DS1s 288 DS0 Channel Capacity - 1 per 12 DS1s	-	ł	UEPMG	VUM20	1,026.40	0.00	0.00	-				19.99	19.99		ł
	384 DS0 Channel Capacity - 1 per 16 DS1s	1	ł	UEPMG	VUM38	1,231.00	0.00	0.00			<u> </u>		19.99	19.99	ł	1
	480 DS0 Channel Capacity - 1 per 20 DS1s	-	ł	UEPMG	VUM40	2,052.80	0.00	0.00	-				19.99	19.99		ł
	576 DS0 Channel Capacity -1 per 24 DS1s	+	1	UEPMG	VUM57	2,052.80	0.00	0.00					19.99	19.99		1
	672 DS0 Channel Capacity - 1 per 28 DS1s	1	1	UEPMG	VUM67	2,463.36	0.00	0.00			1		19.99	19.99		1
Non-F	Recurring Charges (NRC) Associated with 4-Wire DS1 Loop wit	h Chan	neliztio					0.00					10.00	13.33		
	nimum System configuration is One (1) DS1, One (1) D4 Channel															
	ples of this configuration functioning as one are considered A										1					1
Multi											1	ł	1			
Multi	NRC - Conversion (Currently Combined) with or without															
Multi	NRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes - Top 8 MSAs Only			UEPMG	USAC4	0.00	450.00	50.00					19,99	19,99		
	NRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes - Top 8 MSAs Only m Additions Where Currently Combined and New (Not Current	ly Com	pined)	UEPMG	USAC4	0.00	450.00	50.00					19.99	19.99		

UNBUNDL	ED NETWORK ELEMENTS - Georgia	-	_										Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonred	curring	Nonrecurring	Disconnect				Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc															
Dine	Fea Activation -			UEPMG	VUMD4	0.00	950.00	600.00	200.00	30.00			19.99	19.99		
віро	Clear Channel Capability Format, superframe - Subsequent															
	Activity Only			UEPMG	CCOSF	0.00	0.00	600.00								
	Clear Channel Capability Format - Extended Superframe -				00001	0.00	0.00	000.00								
	Subsequent Activity Only			UEPMG	CCOEF	0.00	0.00	600.00								
Alter	mate Mark Inversion (AMI)															
	Superframe Format			UEPMG	MCOSF	0.00	0.00	0.00								
	Extended Superframe Format			UEPMG	MCOPO	0.00	0.00	0.00								
	nange Ports Associated with 4-Wire DS1 Loop with Channelization	on with	Port													
Exch	nange Ports		Ļ		_											
i 1	Line Cide Combination Channelined DDV Truck Durth Durth		1	UEPPX		44.00	0.00	0.00	0.00	0.00	1		33.67	7.00		1
┢───┤───	Line Side Combination Channelized PBX Trunk Port - Business				UEPCX	14.00	0.00	0.00		0.00				7.88		
·	Line Side Outward Channelized PBX Trunk Port - Business			UEPPX	UEPOX	14.00	0.00	0.00	0.00	0.00			33.67	7.88		
1	Line Side Inward Only Channelized PBX Trunk Port without DID		1	UEPPX	UEP1X	14.00	0.00	0.00	0.00	0.00			33.67	7.88		
r	2-Wire Trunk Side Unbundled Channelized DD Trunk Port			UEPPX	UEPDM	83.00	0.00	0.00	0.00	0.00			33.67	7.88		
Feat	ure Activations - Unbundled Loop Concentration		1	OEITX	OEI DIVI	00.00	0.00	0.00	0.00	0.00			00.07	1.00		
(·	Feature (Service) Activation for each Line Port Terminated in D4															
1	Bank			UEPPX	1PQWM	0.62	40.00	20.00	6.00	5.00			33.67	7.88		
í T	Feature (Service) Activation for each Trunk Port Terminated in															
ı	D4 Bank			UEPPX	1PQWU	0.62	110.00	30.00	65.00	20.00			33.67	7.88		
Teler	phone Number/ Group Establishment Charges for DID Service															
	DID Trunk Termination (1 per Port)			UEPPX	NDT	0.00	0.00	0.00								
└──┤ ──	Estab Trk Grp and Provide 1st 20 DID Nos. (FL,GA, NC,& SC)			UEPPX	NDZ	0.00	0.00	0.00								
⊢	DID Numbers - groups of 20 - Valid all States			UEPPX	ND4	0.00	0.00	0.00								
⊢−−−	Non-Consecutive DID Numbers - per number			UEPPX	ND5	0.00	0.00	0.00								
┢───┤───	Reserve Non-Consecutive DID Numbers			UEPPX	ND6	0.00	0.00	0.00								
	Reserve DID Numbers al Number Portability		-	UEPPX	NDV	0.00	0.00	0.00								
LUCA	Local Number Portability - 1 per port			UEPPX	LNPCP	3.15	0.00	0.00								
FFA.	TURES - Vertical and Optional		-			5.15	0.00	0.00								
	al Switching Features Offered with Line Side Ports Only															
	All Features Available			UEPPX	UEPVF	0.00	0.00	0.00								
UNBUNDLE	D CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES	S			-											
	ost Based Rates are applied where BellSouth is required by FCC															
	eatures shall apply to the Unbundled Port/Loop Combination - C															
	nd Office and Tandem Switching Usage and Common Transport															
	ne first and additional Port nonrecurring charges apply to Not Co	urrently	Comb	ined Combos. Fo	r Currently Co	mbined Combo	s, the nonrecu	urring charges	shall be those	identified in t	he Nonrecu	rring - Curre	ently Combine	ed sections.	Additional NR	Cs may
	y also and are categorized accordingly.										1					1
	larket Rates for Unbundled Centrex Port/Loop Combination will		otiated	on an Individual C	ase Basis, un	il further notice).									
	-P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only		<u> </u>		+											
	re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design)				-											
UNE	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
i	Non-Design		1	UEP91		12.59										
• •	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		<u> </u>		1	12.00							1	1	1	
└───┼───	2-WIE VG LOOD/2-WIE VOICE GIAGE FOIL (CEILLEXFOIL COMOD -			UEP91		14.26										
┝──┼──	Non-Design		2								İ		İ	1	1	i
			2	OLI UI												
	Non-Design		3	UEP91		21.62										
UNE	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design Port/Loop Combination Rates (Design)					21.62										
UNE	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design Port/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -			UEP91												
	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design Port/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design					21.62										
	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design Port/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		3	UEP91 UEP91		18.63										
	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design Port/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design			UEP91												
	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design Port/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		3	UEP91 UEP91		18.63										

NBUNDLED N	NETWORK ELEMENTS - Georgia		r									1	Attachment:	1		bit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP91	UECS1	10.80										
	Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP91 UEP91	UECS1 UECS1	12.47 19.83										
	Wire Voice Grade Loop (SL 1) - Zone 3 Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP91	UECS1	19.83										
	Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP91	UECS2	19.45										
	Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP91	UECS2	30.92					1					
UNE Ports																
All States	(Except North Carolina and Sout Carolina)															
	Wire Voice Grade Port (Centrex) Basic Local Area			UEP91	UEPYA	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	Wire Voice Grade Port (Centrex 800 termination)Basic Local															
Are				UEP91	UEPYB	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	Wire Voice Grade Port (Centrex with Caller ID)1Basic Local ea		1	UEP91	UEPYH	1.79	22.4.4	15.25	8.45	2.04			33.67	7.88		
	ea Wire Voice Grade Port (Centrex from diff Serving Wire			ULFSI		1.79	22.14	15.25	8.45	3.91			33.07	7.88		
	enter)2 Basic Local Area			UEP91	UEPYM	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	Wire Voice Grade Port, Diff Serving Wire Center - 800 Service					1.73	22.14	10.20	0.43	5.91	1		55.07	7.00		
	erm - Basic Local Area			UEP91	UEPYZ	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
2-\	Wire Voice Grade Port terminated in on Megalink or equivalent															
	Basic Local Area			UEP91	UEPY9	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	Wire Voice Grade Port Terminated on 800 Service Term -															
	asic Local Area			UEP91	UEPY2	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	nd Florida Only Wire Voice Grade Port (Centrex)			UEP91	UEPHA	1.79	00.44	45.05	0.45	2.04			33.67	7.88		
	Wire Voice Grade Port (Centrex) Wire Voice Grade Port (Centrex 800 termination)			UEP91 UEP91	UEPHA	1.79	22.14 22.14	15.25 15.25	8.45 8.45	3.91 3.91			33.67	7.88		
	Wire Voice Grade Port (Centrex with Caller ID)1			UEP91	UEPHH	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	Wire Voice Grade Port (Centrex from diff Serving Wire			OLF 91	OLFTIIT	1.75	22.14	15.25	0.45	3.91			33.07	7.00		
	enter)2			UEP91	UEPHM	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
2-\	Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
Te	erm			UEP91	UEPHZ	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	Wire Voice Grade Port terminated in on Megalink or equivalent			UEP91	UEPH9	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	Wire Voice Grade Port Terminated on 800 Service Term			UEP91	UEPH2	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
Local Swi	entrex Intercom Funtionality, per port			UEP91	URECS	0.5554										
	nber Portability			UEF91	URECS	0.5554										
	cal Number Portability (1 per port)			UEP91	LNPCC	0.35										
Features						0.00							İ	İ		
	Standard Features Offered, per port			UEP91	UEPVF	0.00										
All	Select Features Offered, per port			UEP91	UEPVS	0.00	454.69									
	Centrex Control Features Offered, per port			UEP91	UEPVC	0.00										
NARS						0.00	0.00	0.00			1		00.07	7.00		
	hbundled Network Access Register - Combination			UEP91 UEP91	UARCX UAR1X	0.00	0.00	0.00			<u> </u>		33.67 33.67	7.88 7.88		
	nbundled Network Access Register - Indial nbundled Network Access Register - Outdial			UEP91 UEP91	UAR1X	0.00	0.00	0.00					33.67	7.88		
	eous Terminations		<u> </u>	02101	UNION	0.00	0.00	0.00			-		33.07	1.00		
2-Wire Tru					1 1						1		1	1		
	unk Side Terminations, each			UEP91	CENA6	11.35	61.91	61.91			1		33.67	7.88	l	
Interoffice	e Channel Mileage - 2-Wire															
	teroffice Channel Facilities Termination - Voice Grade			UEP91	M1GBC	17.07										
	teroffice Channel mileage, per mile or fraction of mile			UEP91	M1GBM	0.0222										
	ctivations (DS0) Centrex Loops on Channelized DS1 Servic	e			+						1					
	el Bank Feature Activations				1000/0	0.00										
Fe	eature Activation on D-4 Channel Bank Centrex Loop Slot			UEP91	1PQWS	0.62										
Fo	eature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP91	1PQW6	0.62										
	eature Activation on D-4 Channel Bank FX Trunk Side Loop Side				11 52,000	0.02					1					
Slo			1	UEP91	1PQW7	0.62										
Fe	eature Activation on D-4 Channel Bank Centrex Loop Slot -			İ							1		l	l	l	
	fferent Wire Center	1	1	UEP91	1PQWP	0.62										

NBUNDLE	D NETWORK ELEMENTS - Georgia	1		1	<u> </u>						-	-	Attachment:			bit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge Manual S Order vs Electroni Disc Add
						Rec	Nonrec		Nonrecurring					Rates(\$)		·
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP91	1PQWV	0.62										
	Feature Activation on D-4 Channel Bank Filvate Line/Trunk Loop			ULF91	TFQVVV	0.02										
	Slot			UEP91	1PQWQ	0.62										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP91	1PQWA	0.62										
Non-Re	curring Charges (NRC) Associated with UNE-P Centrex															
	Conversion - Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP91	USAC2	0.00	2.01	0.3108					33.67	7.88		l
	New Centrex Standard Common Block New Centrex Customized Common Block			UEP91 UEP91	M1ACS M1ACC	0.00	659.41 659.41						33.67 33.67	7.88 7.88		
	Secondary Block, per Block			UEP91 UEP91	M1ACC M2CC1	0.00	77.10				1		33.67	7.88		+
	NAR Establishment Charge, Per Occasion	1		UEP91	URECA	0.00	71.88		1		1		33.67	7.88	1	
UNE-P	CENTREX - 5ESS (Valid in All States)										1				1	<u> </u>
	VG Loop/2-Wire Voice Grade Port (Centrex) Combo															
UNE Po	ort/Loop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	Non-Design		1	UEP95		12.59										<u> </u>
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		2	UEP95		14.26										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		2	UEF95	+ +	14.20			-							ł
	Non-Design		3	UEP95		21.62										
UNE Po	ort/Loop Combination Rates (Design)		Ŭ	02.00		21102										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	Design		1	UEP95		18.63										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
			2	UEP95		21.24										ļ
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		3	UEP95		32.71										
UNELO	Design pop Rate		3	UEF95	+ +	32.71			-							ł
ONL L	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP95	UECS1	10.80										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP95	UECS1	12.47										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP95	UECS1	19.83										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP95	UECS2	16.84										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP95	UECS2	19.45										L
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP95	UECS2	30.92										ł
All Stat	ort Rate															
All Sta	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP95	UEPYA	1.79	22.14	15.25	8.45	3.91			33.67	7.88		ł
	2-Wire Voice Grade Port (Centrex 9) Dasie Escal 7 red			UEP95	UEPYB	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local				-											
	Area			UEP95	UEPYH	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex from diff Serving Wire															
	Center)2 Basic Local Area			UEP95	UEPYM	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term - Basic Local Area			UEP95	UEPYZ	1.79	22.14	15.25	8.45	3.91			33.67	7.88		ł
	2-Wire Voice Grade Port terminated in on Megalink or equivalent - Basic Local Area			UEP95	UEPY9	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port Terminated on 800 Service Term -			0LF 95	OLF 19	1.75	22.14	13.25	0.43	3.91			33.07	7.00		
	Basic Local Area			UEP95	UEPY2	1.79	22.14	15.25	8.45	3.91			33.67	7.88		1
FL & G	A Only									2.01	1				1	<u> </u>
	2-Wire Voice Grade Port (Centrex)			UEP95	UEPHA	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPHB	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex with Caller ID)1	l		UEP95	UEPHH	1.79	22.14	15.25	8.45	3.91			33.67	7.88		Ļ
	2-Wire Voice Grade Port (Centrex from diff Serving Wire					4 70	00.44	45.05	0.45	0.01			22.07	7.00		
	Center)2 2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			UEP95	UEPHM	1.79	22.14	15.25	8.45	3.91			33.67	7.88		<u> </u>
	Term			UEP95	UEPHZ	1.79	22.14	15.25	8.45	3.91			33.67	7.88		1
		1			02.112	1.75	22.17	10.20	0.40	0.01	1		00.07	7.00		├ ───
1	2-Wire Voice Grade Port terminated in on Megalink or equivalent	1		UEP95	UEPH9	1.79	22.14	15.25	8.45	3.91			33.67	7.88		1

IBUNDLE	D NETWORK ELEMENTS - Georgia												Attachment:			ibit: B
TEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge
						Rec	Nonrec		Nonrecurring			_		Rates(\$)		
	0 Wire Vision Crede Dert Termineted on 900 Certine Term			UEP95	UEPH2	1.79	First 22.14	Add'l 15.25	First 8.45	Add'l 3.91	SOMEC	SOMAN	SOMAN 33.67	SOMAN 7.88	SOMAN	SOMA
Local	2-Wire Voice Grade Port Terminated on 800 Service Term Switching		-	UEP95	UEPHZ	1.79	ZZ.14	15.25	8.45	3.91			33.67	7.88		
LUCal	Centrex Intercom Funtionality, per port			UEP95	URECS	0.5554										
Local	Number Portability		-	01 33	UNECO	0.0004										
	Local Number Portability (1 per port)			UEP95	LNPCC	0.35										
Featur																
	All Standard Features Offered, per port			UEP95	UEPVF	0.00							33.67	7.88		
	All Select Features Offered, per port			UEP95	UEPVS	0.00	454.69						33.67	7.88		
	All Centrex Control Features Offered, per port			UEP95	UEPVC	0.00							33.67	7.88		
NARS																
_	Unbundled Network Access Register - Combination	ļ	L	UEP95	UARCX	0.00	0.00	0.00					33.67	7.88		
_	Unbundled Network Access Register - Indial		<u> </u>	UEP95	UAR1X	0.00	0.00	0.00					33.67	7.88		
Mincel	Unbundled Network Access Register - Outdial	<u> </u>		UEP95	UAROX	0.00	0.00	0.00					33.67	7.88		<u> </u>
	laneous Terminations Trunk Side	 														
2-wire	Trunk Side Trunk Side Terminations, each			UEP95	CEND6	11.35	61.91	61.91					33.67	7.88		
4-Wire	Digital (1.544 Megabits)	1		025,90	CENDO	11.35	01.91	01.91					33.67	7.88		
4-44116	DS1 Circuit Terminations, each			UEP95	M1HD1	120.80	89.44	52.46					33.67	7.88		
	DS0 Channels Activated, each		-	UEP95	M1HDO	0.00	28.71	52.40					33.67	7.88		
Interof	fice Channel Mileage - 2-Wire			02.00		0.00	2011 1						00.01	1.00		
	Interoffice Channel Facilities Termination			UEP95	MIGBC	17.07										
	Interoffice Channel mileage, per mile or fraction of mile			UEP95	MIGBM	0.0222										
Featur	e Activations (DS0) Centrex Loops on Channelized DS1 Service	e			-											
D4 Cha	annel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP95	1PQWS	0.62										
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP95	1PQW6	0.62										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop															
	Slot			UEP95	1PQW7	0.62										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -					0.00										
_	Different Wire Center			UEP95	1PQWP	0.62										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP95	1PQWV	0.62										
-	Feature Activation on D-4 Channel Bank Tile Line/Trunk Loop			0LF 93	IFQVVV	0.02										
	Slot			UEP95	1PQWQ	0.62										
	Feature Activation on D-4 Channel Bank WATS Loop Slot	1	1	UEP95	1PQWQ	0.62					1					1
Non-R	ecurring Charges (NRC) Associated with UNE-P Centrex	1	1			0.02					1				İ	1
1	NRC Conversion Currently Combined Switch-As-Is with allowed	1	1								1					1
	changes, per port			UEP95	USAC2		2.01	0.3108					33.67	7.88		
	New Centrex Standard Common Block			UEP95	M1ACS	0.00	659.41						33.67	7.88		
	New Centrex Customized Common Block			UEP95	M1ACC	0.00	659.41						33.67	7.88		
	NAR Establishment Charge, Per Occasion	<u> </u>	L	UEP95	URECA	0.00	71.88						33.67	7.88		\vdash
	CENTREX - DMS100 (Valid in All States)		<u> </u>													
	VG Loop/2-Wire Voice Grade Port (Centrex) Combo	<u> </u>	<u> </u>													ļ
UNE P	ort/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo	<u> </u>														<u> </u>
	Non-Design	1	4	UEP9D		12.59										
-	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	+	<u> </u>			12.09										
	Non-Design	1	2	UEP9D		14.26										1
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	1	1	-	1 1						1				İ	1
	Non-Design	1	3	UEP9D		21.62										1
UNE P	ort/Loop Combination Rates (Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo	-			l											
	Design		1	UEP9D		18.63										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
_	Design	ļ	2	UEP9D	_	21.24										ļ
1	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	1	1								1			1	1	1
	Design		3	UEP9D		32.71										

UNBUNDLE	D NETWORK ELEMENTS - Georgia	1	1	1								-	Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonreo		Nonrecurring					Rates(\$)		
			<u> </u>		115001		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9D	UECS1	10.80										l
	2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3		2	UEP9D UEP9D	UECS1 UECS1	12.47 19.83										i
	2-Wire Voice Grade Loop (SL 1) - Zone 3		1	UEP9D	UECS2	16.84										i
	2-Wire Voice Grade Loop (SL 2) - Zone 1		2	UEP9D	UECS2	19.45										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9D	UECS2	30.92										i
UNE P	ort Rate		-													i
	TATES															1
	2-Wire Voice Grade Port (Centrex) Basic Local Area	1		UEP9D	UEPYA	1.79	22.14	15.25	8.45	3.91			33.67	7.88		(
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local															í
	Area		<u> </u>	UEP9D	UEPYB	1.79	22.14	15.25	8.45	3.91			33.67	7.88		ļ
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3Basic Local Area			UEP9D	UEPYC	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3Basic Local Area			UEP9D	UEPYD	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5209))3 Basic Local Area			UEP9D	UEPYE	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5112))3 Basic Local Area			UEP9D	UEPYF	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5312))3Basic Local Area			UEP9D	UEPYG	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5008))3 Basic Local Area			UEP9D	UEPYT	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5208))3 Basic Local Area			UEP9D	UEPYU	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5216))3 Basic Local Area			UEP9D	UEPYV	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5316))3 Basic Local Area			UEP9D	UEPY3	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex with Caller ID) Basic Local Area			UEP9D	UEPYH	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp Indication))3 Basic Local Area			UEP9D	UEPYW	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication))3 Basic Local Area			UEP9D	UEPYJ	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center) 2 Basic Local Area			UEP9D	UEPYM	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3 Basic Local Area			UEP9D	UEPYO	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3 Basic Local Area			UEP9D	UEPYP	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3 Basic Local Area			UEP9D	UEPYQ	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3 Basic Local Area			UEP9D	UEPYR	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3 Basic Local Area			UEP9D	UEPYS	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3 Basic Local Area			UEP9D	UEPY4	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3 Basic Local Area			UEP9D	UEPY5	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3 Basic Local Area 2. Wire Vision Crade Dert (Centrex/differ SWC /EBS M5216)2, 2			UEP9D	UEPY6	1.79	22.14	15.25	8.45	3.91			33.67	7.88		ļ
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3 Basic Local Area			UEP9D	UEPY7	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP9D	UEPYZ	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port terminated in on Megalink or equivalent Basic Local Area			UEP9D	UEPY9	1.79	22.14	15.25	8.45	3.91			33.67	7.88		

UNBUNDLE	D NETWORK ELEMENTS - Georgia												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR		Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port Terminated on 800 Service Term Basic					1.70	00.44	45.05	0.45	0.01			00.07	7.00		1 1
EL @ C	Local Area			UEP9D	UEPY2	1.79	22.14	15.25	8.45	3.91			33.67	7.88		↓ ′
FL & G	2-Wire Voice Grade Port (Centrex)			UEP9D	UEPHA	1.79	22.14	15.25	8.45	3.91			33.67	7.88		j/
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP9D	UEPHB	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3			UEP9D	UEPHC	1.79	22.14	15.25	8.45	3.91			33.67	7.88		1
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3			UEP9D	UEPHD	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5209)3			UEP9D	UEPHE	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5112)3			UEP9D	UEPHF	1.79	22.14	15.25	8.45	3.91			33.67	7.88		1
	2-Wire Voice Grade Port (Centrex / EBS-M5312)3			UEP9D	UEPHG	1.79	22.14	15.25	8.45	3.91			33.67	7.88		l
	2-Wire Voice Grade Port (Centrex / EBS-M5008)3		<u> </u>	UEP9D	UEPHT	1.79	22.14	15.25	8.45	3.91	ļ		33.67	7.88		I
	2-Wire Voice Grade Port (Centrex / EBS-M5208)3 2-Wire Voice Grade Port (Centrex / EBS-M5216)3			UEP9D UEP9D	UEPHU UEPHV	1.79 1.79	22.14 22.14	15.25 15.25	8.45 8.45	3.91 3.91	<u>├</u> ──		33.67 33.67	7.88 7.88		l
 	2-Wire Voice Grade Port (Centrex / EBS-N5216)3 2-Wire Voice Grade Port (Centrex / EBS-M5316)3			UEP9D UEP9D	UEPHV UEPH3	1.79	22.14	15.25	8.45	3.91	<u> </u>	<u> </u>	33.67	7.88		i
<u> </u>	2-Wire Voice Grade Port (Centrex vith Caller ID)		1	UEP9D	UEPHH	1.79	22.14	15.25	8.45	3.91			33.67	7.88		1
<u> </u>	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp		1					.0.20	00	0.01			00.01			(
	Indication)3			UEP9D	UEPHW	1.79	22.14	15.25	8.45	3.91			33.67	7.88		1
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication)3			UEP9D	UEPHJ	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)															1
	2			UEP9D	UEPHM	1.79	22.14	15.25	8.45	3.91			33.67	7.88		ļ'
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			UEP9D	UEPHO	1.79	22.14	15.25	8.45	3.91			33.67	7.88		ļ!
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3			UEP9D	UEPHP	1.79	22.14	15.25	8.45	3.91			33.67	7.88		1 1
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-105009)2, 3 2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3			UEP9D UEP9D	UEPHQ	1.79	22.14	15.25	8.45	3.91			33.67	7.88		j/
	2-Wire Voice Grade Fort (Centrex diner GWO /EBC-3203)2, 3			OEI 3D	OLITIQ	1.75	22.14	10.20	0.45	5.51			55.07	7.00		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			UEP9D	UEPHR	1.79	22.14	15.25	8.45	3.91			33.67	7.88		1 1
																(
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3			UEP9D	UEPHS	1.79	22.14	15.25	8.45	3.91			33.67	7.88		1 1
																í T
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3			UEP9D	UEPH4	1.79	22.14	15.25	8.45	3.91			33.67	7.88		ļ!
														=		1 1
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3			UEP9D	UEPH5	1.79	22.14	15.25	8.45	3.91			33.67	7.88		↓ ′
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3			UEP9D	UEPH6	1.79	22.14	15.25	8.45	3.91			33.67	7.88		1 1
				OLI 3D	OLITIO	1.75	22.14	15.25	0.45	5.51			55.07	7.00		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			UEP9D	UEPH7	1.79	22.14	15.25	8.45	3.91			33.67	7.88		1
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term			UEP9D	UEPHZ	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
																1
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9D	UEPH9	1.79	22.14	15.25	8.45	3.91			33.67	7.88		 '
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP9D	UEPH2	1.79	22.14	15.25	8.45	3.91			33.67	7.88		ļ'
Local	Switching Centrex Intercom Funtionality, per port			UEP9D	URECS	0.5554										
Local	Number Portability				UNLOG	0.0004					-	-				i
	Local Number Portability (1 per port)		1	UEP9D	LNPCC	0.35						1	1	1	1	i
Featur									i i		1	1	1	1	1	ſ'
	All Standard Features Offered, per port			UEP9D	UEPVF	0.00										(<u> </u>
	All Select Features Offered, per port			UEP9D	UEPVS	0.00	454.69						33.67	7.88		
	All Centrex Control Features Offered, per port			UEP9D	UEPVC	0.00										ļ
NARS						0.00	0.00	0.00					00.07	7.00		I'
	Unbundled Network Access Register - Combination			UEP9D UEP9D	UARCX UAR1X	0.00	0.00	0.00					33.67 33.67	7.88 7.88		¹
<u> </u>	Unbundled Network Access Register - Inward Unbundled Network Access Register - Outdial			UEP9D UEP9D	UAR1X	0.00	0.00	0.00					33.67	7.88		
Miscel	Ianeous Terminations		+			0.00	0.00	0.00					55.07	1.00		
	Trunk Side		1		1						İ					1
	Trunk Side Terminations, each			UEP9D	CEND6	11.35										
4-Wire	Digital (1.544 Megabits)															
	DS1 Circuit Terminations, each			UEP9D	M1HD1	120.80	89.44	52.46					33.67	7.88		└─── ─
	DS0 Channels Activiated per Channel			UEP9D	M1HDO	0.00	28.71				I	l	33.67	7.88		L!

CATEGORY) NETWORK ELEMENTS - Georgia												Attachment:	2	Exhi	bit: B
CATEGORY											Svc Order	Svc Order			Incremental	
CATEGORY			1								Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
CATEGORY		Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svo
	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
											-		Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
							Nonrec	urring	Nonrocurrin	g Disconnect			220	Rates(\$)		
					-	Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Interoffi	ice Channel Mileage - 2-Wire		1				1 11 30	Add I	11130	Auui	JOMILO	JOINIAN	JONIAN	JONIAN	JONIAN	JONIAN
	Interoffice Channel Facilities Termination			UEP9D	MIGBC	17.07										
	Interoffice Channel mileage, per mile or fraction of mile			UEP9D	MIGBM	0.0222										
Feature	Activations (DS0) Centrex Loops on Channelized DS1 Service	e														
	nnel Bank Feature Activations															
F	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9D	1PQWS	0.62										
					40014/0	0.00										
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot Feature Activation on D-4 Channel Bank FX Trunk Side Loop			UEP9D	1PQW6	0.62										
	Slot		1	UEP9D	1PQW7	0.62										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -		1	001 00	11 04 99 /	0.02										
	Different Wire Center		1	UEP9D	1PQWP	0.62										
			1								1					
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9D	1PQWV	0.62										
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop															
	Slot		<u> </u>	UEP9D	1PQWQ	0.62				ļ						
	Feature Activation on D-4 Channel Bank WATS Loop Slot		Ļ	UEP9D	1PQWA	0.62				ļ	ļ					
	curring Charges (NRC) Associated with UNE-P Centrex		<u> </u>		+					l						
	NRC Conversion Currently Combined Switch-As-Is with allowed changes, per port			UEP9D	USAC2		2.01	0.3108					33.67	7.88		
	New Centrex Standard Common Block			UEP9D UEP9D	M1ACS	0.00	659.41	0.3106					33.67	7.88		
	New Centrex Customized Common Block		1	UEP9D	MIACC	0.00	659.41						33.67	7.88		
	NAR Establishment Charge, Per Occasion			UEP9D	URECA	0.00	71.88						33.67	7.88		
Note 1 -	Required Port for Centrex Control in 1AESS, 5ESS & EWSD															
	- Requres Interoffice Channel Mileage															
	Requires Specific Customer Premises Equipment															
	ENTREX PORT/LOOP COMBINATIONS - MARKET RATES															
	et Rates are applied where BellSouth is not required by FCC					ndled Local Sw	itching or Swi	tch Ports.		-						
	rring Charges for all Standard Centrex and Centrex Conrol Fe Office and Tandem Switching Usage and Common Transport					ibit chall apply	to all combina	tions of loon/	oort notwork o	lomonte ovcor	t for LINE C	oin Port/Lo	on Combinat	ions		
		U U														A
	irst and additional Port nonrecurring charges apply to Not Co	urrently	Comp	ined Compos. For	Currently Co	mbined Combo	s, the nonrecu	irring charges	snall be those	e identified in t	ne Nonrecu	ring - Curre	ently Combine	ed sections.	Additional NR	Cs may
	lso and are categorized accordingly. CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only	•								1	1					
	/G Loop/2-Wire Voice Grade Port (Centrex) Combo	<u>/</u>								1						
	rt/Loop Combination Rates (Non-Design)		1													
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -		1	1	1	+ +				ł	1			1	1	
	Non-Design		1	UEP91	1	24.80										
1	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -					İ										
1 I F	Non-Design		2	UEP91		26.47										
1	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		1													
2	Non-Design															
2			3	UEP91		33.83										
	rt/Loop Combination Rates (Design)		3	UEP91		33.83										
UNE Por	rt/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	rt/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design		3	UEP91 UEP91		33.83 30.84										
UNE Por 2 2 2 2 2 2 2 2	rt/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			UEP91		30.84										
UNE Por 2 2 2 2 2 2 2 2 2 2	rt/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design		1													
UNE Por 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	rt/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		1	UEP91		30.84										
	rt/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design op Rate		1 2 3	UEP91 UEP91 UEP91		30.84 33.45 44.92										
UNE Por 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	rt/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design op Rate 2-Wire Voice Grade Loop (SL 1) - Zone 1		1 2 3 1	UEP91 UEP91 UEP91 UEP91	UECS1	30.84 33.45 44.92 10.80										
UNE Loc	tr/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design op Rate 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2		1 2 3 1 2	UEP91 UEP91 UEP91 UEP91 UEP91	UECS1	30.84 33.45 44.92 10.80 12.47										
UNE Por UNE Por C C C C C C C C C C C C C C C C C C C	rt/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design op Rate 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3		1 2 3 1 2 3	UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1	30.84 33.45 44.92 10.80 12.47 19.83										
	rt/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design op Rate 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1		1 2 3 1 2 3 1 2 3 1	UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS2	30.84 33.45 44.92 10.80 12.47 19.83 16.84										
UNE Por 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	rt/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design op Rate 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 1		1 2 3 1 2 3 1 2 3 1 2	UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS2 UECS2	30.84 33.45 44.92 10.80 12.47 19.83 16.84 19.45										
	rt/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design op Rate 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3		1 2 3 1 2 3 1 2 3 1	UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS2	30.84 33.45 44.92 10.80 12.47 19.83 16.84										
UNE POR UNE POR 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	rt/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design op Rate 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3		1 2 3 1 2 3 1 2 3 1 2	UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS2 UECS2	30.84 33.45 44.92 10.80 12.47 19.83 16.84 19.45										

UNBUNDLE	D NETWORK ELEMENTS - Georgia												Attachment:	2	Exhi	ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge -	Incremental Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area			UEP91	UEPYB	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area			UEP91	UEPYH	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 Basic Local Area			UEP91	UEPYM	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term - Basic Local Area 2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP91	UEPYZ	14.00	90.00	45.00	20.00	10.00			33.67	7.88		<u> </u>
				UEP91	UEPY9	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
0	Basic Local Area			UEP91	UEPY2	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
Georg	ia and Florida Only 2-Wire Voice Grade Port (Centrex)			UEP91	UEPHA	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex) 2-Wire Voice Grade Port (Centrex 800 termination)			UEP91	UEPHA	14.00	90.00	45.00	20.00	10.00			33.67	7.88		1
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP91	UEPHH	14.00	90.00	45.00	20.00	10.00	1		33.67	7.88	-	<u> </u>
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP91	UEPHM	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP91	UEPHZ	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP91	UEPH9	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP91	UEPH2	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
Local	Switching															
	Centrex Intercom Funtionality, per port			UEP91	URECS	0.5554										
Local	Number Portability Local Number Portability (1 per port)			UEP91	LNPCC	0.35										
Featu				UEF91	LINFCC	0.35										
· outu	All Standard Features Offered, per port			UEP91	UEPVF	0.00										
	All Select Features Offered, per port			UEP91	UEPVS	0.00	454.69									1
	All Centrex Control Features Offered, per port			UEP91	UEPVC	0.00										
NARS																
	Unbundled Network Access Register - Combination			UEP91	UARCX	0.00	0.00	0.00					33.67	7.88		
	Unbundled Network Access Register - Indial			UEP91	UAR1X UAROX	0.00	0.00	0.00					33.67	7.88 7.88		
Micco	Unbundled Network Access Register - Outdial Ilaneous Terminations			UEP91	UARUX	0.00	0.00	0.00					33.67	7.88		
	e Trunk Side				-											
2 1110	Trunk Side Terminations, each			UEP91	CENA6	11.35	61.91	61.91					33.67	7.88		
Intero	ffice Channel Mileage - 2-Wire			-												
	Interoffice Channel Facilities Termination - Voice Grade			UEP91	M1GBC	17.07										
	Interoffice Channel mileage, per mile or fraction of mile			UEP91	M1GBM	0.0222										
	re Activations (DS0) Centrex Loops on Channelized DS1 Servic	е														
D4 Ch	annel Bank Feature Activations Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP91	1PQWS	0.62									-	<u> </u>
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot Feature Activation on D-4 Channel Bank FX Trunk Side Loop			UEP91	1PQW6	0.62										<u> </u>
	Slot Feature Activation on D-4 Channel Bank Centrex Loop Slot -			UEP91	1PQW7	0.62										<u> </u>
	Different Wire Center			UEP91	1PQWP	0.62								<u> </u>	<u> </u>	<u> </u>
	Feature Activation on D-4 Channel Bank Private Line Loop Slot Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot			UEP91 UEP91	1PQWV 1PQWQ	0.62										<u> </u>
<u> </u>	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP91 UEP91	1PQWQ 1PQWA	0.62					<u> </u>		{	-	-	ł
Non-F	Lecurring Charges (NRC) Associated with UNE-P Centrex				IFQIVA	0.02										<u> </u>
	Conversion - Currently Combined Switch-As-Is with allowed				1 1				1		1		1			1
	changes, per port			UEP91	USAC2		2.01	0.3108					33.67	7.88		
	New Centrex Standard Common Block			UEP91	M1ACS	0.00	659.41						33.67	7.88		

UNBUNDL	ED NETWORK ELEMENTS - Georgia	-	1										Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
	New October October 1 October Division				144.000		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	New Centrex Customized Common Block Secondary Block, per Block			UEP91 UEP91	M1ACC M2CC1	0.00	659.41 77.10						33.67 33.67	7.88 7.88		
	NAR Establishment Charge, Per Occasion			UEP91	URECA	0.00	71.10						33.67	7.88		
LINE-	P CENTREX - 5ESS (Valid in All States)		-	UEP91	URECA	0.00	71.00						33.07	7.00		
	e VG Loop/2-Wire Voice Grade Port (Centrex) Combo		-													
	Port/Loop Combination Rates (Non-Design)								1 1							
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	-														
	Non-Design		1	UEP95		24.80										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Non-Design		2	UEP95		26.47										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Non-Design		3	UEP95		33.83										
UNE	Port/Loop Combination Rates (Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	1	l .													
	Design		1	UEP95		30.84										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			LIEDOE		00.45										
	Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		2	UEP95		33.45										
	Design		3	UEP95		44.92										
	Loop Rate		3	UEP95		44.92			<u> </u>							
UNE	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP95	UECS1	10.80										
	2-Wire Voice Grade Loop (SL 1) - Zone 1		2	UEP95	UECS1	12.47										
	2-Wire Voice Grade Loop (SL 1) - Zone 2			UEP95	UECS1	19.83										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP95	UECS2	16.84										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP95	UECS2	19.45										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP95	UECS2	30.92										
UNE	Port Rate															
All St	ates															
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP95	UEPYA	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPYB	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local															
	Area			UEP95	UEPYH	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex from diff Serving Wire															
	Center)2 Basic Local Area			UEP95	UEPYM	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			UEDOE		44.00	00.00	45.00	00.00	40.00			00.07	7.00		
	Term - Basic Local Area			UEP95	UEPYZ	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port terminated in on Megalink or equivalent - Basic Local Area			UEP95	UEPY9	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port Terminated on 800 Service Term -		-	UEP95	UEP19	14.00	90.00	45.00	20.00	10.00			33.07	7.00		
	Basic Local Area			UEP95	UEPY2	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
FI &	GA Only		1	021 33	01112	14.00	30.00	40.00	20.00	10.00			55.07	7.00		
	2-Wire Voice Grade Port (Centrex)			UEP95	UEPHA	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPHB	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP95	UEPHH	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex from diff Serving Wire															
	Center)2	<u> </u>		UEP95	UEPHM	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term	I		UEP95	UEPHZ	14.00	90.00	45.00	20.00	10.00			33.67	7.88		ļ
		1	1						I							
	2-Wire Voice Grade Port terminated in on Megalink or equivalent	<u> </u>	 	UEP95	UEPH9	14.00	90.00	45.00	20.00	10.00			33.67	7.88		l
1	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP95	UEPH2	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
Loca	Switching Centrex Intercom Funtionality, per port	 		UEP95	URECS	0.5554										
1.000	Number Portability			024.80	URECO	0.0004			├							
LOCA	Local Number Portability (1 per port)	1		UEP95	LNPCC	0.35			 							ł
				01.30	LINFOO	0.55										
Featu				1							1			i		l
Featu			1	UEP95	UEPVE	0.00							33 67	7 88		
Featu	All Standard Features Offered, per port All Select Features Offered, per port			UEP95 UEP95	UEPVF UEPVS	0.00	454.69						33.67 33.67	7.88 7.88		

IBUNDLE	D NETWORK ELEMENTS - Georgia											1	Attachment:			ibit: B
TEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec	urring	Nonrecurring	g Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
NARS				LIEDOC	UARCX	0.00	0.00			-			33.67	7.00		
<u> </u>	Unbundled Network Access Register - Combination Unbundled Network Access Register - Indial			UEP95 UEP95	UARCX UAR1X	0.00	0.00	0.00					33.67	7.88 7.88		
	Unbundled Network Access Register - Outdial			UEP95	UAROX	0.00	0.00	0.00		ł			33.67	7.88		
Miscel	Inneous Terminations			021 00	0/4(0)/	0.00	0.00	0.00		1			00.07	7.00		
	Trunk Side															
	Trunk Side Terminations, each			UEP95	CEND6	11.35	61.91	61.91					33.67	7.88		
	Digital (1.544 Megabits)															
	DS1 Circuit Terminations, each			UEP95	M1HD1	120.80	89.44	52.46					33.67	7.88		
	DS0 Channels Activated, each			UEP95	M1HDO	0.00	28.71			-			33.67	7.88		
	fice Channel Mileage - 2-Wire Interoffice Channel Facilities Termination			UEP95	MIGBC	17.07										
	Interoffice Channel mileage, per mile or fraction of mile			UEP95	MIGBC	0.0222										
Featur	e Activations (DS0) Centrex Loops on Channelized DS1 Servic	e		021 33		0.0222										
	annel Bank Feature Activations				+ +											
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP95	1PQWS	0.62										
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP95	1PQW6	0.62										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop															
_	Slot			UEP95	1PQW7	0.62										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -				1PQWP	0.00										
_	Different Wire Center			UEP95	TPQWP	0.62										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP95	1PQWV	0.62										
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop			02.00		0.02				1	1					
	Slot			UEP95	1PQWQ	0.62										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP95	1PQWA	0.62										
Non-Re	ecurring Charges (NRC) Associated with UNE-P Centrex															
	NRC Conversion Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP95	USAC2		2.01	0.3108					33.67	7.88		
—	New Centrex Standard Common Block New Centrex Customized Common Block			UEP95	M1ACS M1ACC	0.00	659.41						33.67	7.88		
_	New Centrex Customized Common Block NAR Establishment Charge, Per Occasion			UEP95 UEP95	URECA	0.00	659.41 71.88						33.67 33.67	7.88 7.88		
	CENTREX - DMS100 (Valid in All States)			UEF95	URECA	0.00	/ 1.00			ł			33.07	7.00		
	VG Loop/2-Wire Voice Grade Port (Centrex) Combo									1						
	ort/Loop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -				1 1					1						
	Non-Design		1	UEP9D		24.80										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
_	Non-Design		2	UEP9D	┥───┤	26.47				 						
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		3	UEP9D		33.83										
	ort/Loop Combination Rates (Design)		3	02-90	+	33.83				<u> </u>						
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -				+ +					1						
	Design		1	UEP9D		30.84										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	1			1 1					1	1		l	l	1	1
	Design		2	UEP9D		33.45										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design		3	UEP9D		44.92										
	oop Rate				115004	10.00										
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9D	UECS1	10.80										
_	2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3		2	UEP9D UEP9D	UECS1 UECS1	12.47 19.83					-					
	2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1		3	UEP9D UEP9D	UECS1	19.83			-	ł	<u> </u>		ł	ł	ł	+
+	2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 2	1	2	UEP9D	UECS2	19.45				<u> </u>	1					1
	2-Wire Voice Grade Loop (SL 2) - Zone 3			UEP9D	UECS2	30.92				1						1
	ort Rate															

UNBUNDLE	ED NETWORK ELEMENTS - Georgia												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Area			UEP9D	UEPYB	14.00	90.00	45.00	20.00	10.00			33.67	7.88		1 1
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3Basic Local Area			UEP9D	UEPYC	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3Basic Local Area			UEP9D	UEPYD	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5209))3 Basic Local															
	Area 2-Wire Voice Grade Port (Centrex / EBS-M5112))3 Basic Local			UEP9D	UEPYE	14.00	90.00	45.00	20.00	10.00			33.67	7.88		!
	Area			UEP9D	UEPYF	14.00	90.00	45.00	20.00	10.00			33.67	7.88		ļ!
	2-Wire Voice Grade Port (Centrex / EBS-M5312))3Basic Local Area			UEP9D	UEPYG	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5008))3 Basic Local Area			UEP9D	UEPYT	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5208))3 Basic Local Area			UEP9D	UEPYU	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5216))3 Basic Local Area			UEP9D	UEPYV	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5316))3 Basic Local Area			UEP9D	UEPY3	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex with Caller ID) Basic Local Area			UEP9D	UEPYH	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp Indication))3 Basic Local Area			UEP9D	UEPYW	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication))3				UEPYJ									7.88		
	Basic Local Area 2-Wire Voice Grade Port (Centrex from diff Serving Wire Center) 2 Basic Local Area			UEP9D UEP9D	UEPYJ	14.00 14.00	90.00 90.00	45.00	20.00	10.00			33.67 33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			UEP9D	UEPYO	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	Basic Local Area 2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3															
	Basic Local Area 2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3			UEP9D	UEPYP	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	Basic Local Area 2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			UEP9D	UEPYQ	14.00	90.00	45.00	20.00	10.00			33.67	7.88		!
	Basic Local Area			UEP9D	UEPYR	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3 Basic Local Area			UEP9D	UEPYS	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3 Basic Local Area			UEP9D	UEPY4	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3 Basic Local Area			UEP9D	UEPY5	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3 Basic Local Area			UEP9D	UEPY6	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3 Basic Local Area			UEP9D	UEPY7	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP9D	UEPYZ	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port terminated in on Megalink or equivalent Basic Local Area			UEP9D	UEPY9	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port Terminated on 800 Service Term Basic Local Area			UEP9D	UEPY2	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
FL & (GA Only															
	2-Wire Voice Grade Port (Centrex)			UEP9D	UEPHA	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
┝──┼──	2-Wire Voice Grade Port (Centrex 800 termination) 2-Wire Voice Grade Port (Centrex / EBS-PSET)3			UEP9D UEP9D	UEPHB UEPHC	14.00 14.00	90.00 90.00	45.00 45.00	20.00 20.00	10.00			33.67 33.67	7.88 7.88		├ ────
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3 2-Wire Voice Grade Port (Centrex / EBS-M5009)3			UEP9D UEP9D	UEPHC	14.00	90.00	45.00	20.00	10.00			33.67	7.88		├ ────┘
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3			UEP9D	UEPHE	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5112)3	1		UEP9D	UEPHF	14.00	90.00	45.00	20.00	10.00	1		33.67	7.88		r 1

NRONDLE	D NETWORK ELEMENTS - Georgia				· ·						-	-	Attachment:			ibit: B
TEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
					_										DISC ISL	DISC Add
						Rec	Nonrec		Nonrecurring		0.01150	001111		Rates(\$)	001111	0.011411
	2-Wire Voice Grade Port (Centrex / EBS-M5312)3			UEP9D	UEPHG	14.00	First 90.00	Add'l 45.00	First 20.00	Add'l 10.00	SOMEC	SOMAN	SOMAN 33.67	SOMAN 7.88	SOMAN	SOMAN
	2-Wire Voice Grade Port (Centrex / EBS-M5512)5 2-Wire Voice Grade Port (Centrex / EBS-M5008)3			UEP9D	UEPHT	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5008)3			UEP9D	UEPHU	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5206)3			UEP9D	UEPHV	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex / EBS-M5216)3			UEP9D	UEPH3	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex vith Caller ID)			UEP9D	UEPHH	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp			02.05	02.1	1.00	00.00	10100	20.00	10.00			00.01	1.00		
	Indication)3			UEP9D	UEPHW	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication)3			UEP9D	UEPHJ	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)															
	2			UEP9D	UEPHM	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			UEP9D	UEPHO	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
					1				1							
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3			UEP9D	UEPHP	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3			UEP9D	UEPHQ	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			UEP9D	UEPHR	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3			UEP9D	UEPHS	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3			UEP9D	UEPH4	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3			UEP9D	UEPH5	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
														=		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3			UEP9D	UEPH6	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	ONE MELO AL DEL COLLES (ESCONO / FROM SOLO)O O					44.00	00.00	45.00	00.00	10.00			00.07	7.00		
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			UEP9D	UEPH7	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service					44.00	00.00	45.00	20.00	40.00			20.07	7.00		
	Term			UEP9D	UEPHZ	14.00	90.00	45.00	20.00	10.00			33.67	7.88		4
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9D	UEPH9	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	2-Wire Voice Grade Port Terminated in on Negalink of equivalent			UEP9D	UEPH9	14.00	90.00	45.00	20.00	10.00			33.67	7.88		
	Switching			OLF 9D	ULFIIZ	14.00	90.00	45.00	20.00	10.00			33.07	7.00		
Local C	Centrex Intercom Funtionality, per port			UEP9D	URECS	0.5554										
Local	Number Portability			OEI OD	ONLOG	0.0004										
Loodin	Local Number Portability (1 per port)			UEP9D	LNPCC	0.35										-
Feature				02.05	2.11 00	0.00										
	All Standard Features Offered, per port			UEP9D	UEPVF	0.00										
	All Select Features Offered, per port			UEP9D	UEPVS	0.00	454.69						33.67	7.88		
	All Centrex Control Features Offered, per port			UEP9D	UEPVC	0.00										
NARS																1
	Unbundled Network Access Register - Combination			UEP9D	UARCX	0.00	0.00	0.00					33.67	7.88		
	Unbundled Network Access Register - Inward			UEP9D	UAR1X	0.00	0.00	0.00					33.67	7.88		1
	Unbundled Network Access Register - Outdial			UEP9D	UAROX	0.00	0.00	0.00					33.67	7.88		
Miscell	aneous Terminations															
	Trunk Side															
	Trunk Side Terminations, each			UEP9D	CEND6	11.35										
4-Wire	Digital (1.544 Megabits)															
	DS1 Circuit Terminations, each		L	UEP9D	M1HD1	120.80	89.44	52.46					33.67	7.88		<u> </u>
	DS0 Channels Activiated per Channel	ļ	ļ	UEP9D	M1HDO	0.00	28.71		ļļ				33.67	7.88		<u> </u>
	fice Channel Mileage - 2-Wire	L			14050	1										<u> </u>
	Interoffice Channel Facilities Termination			UEP9D	MIGBC	17.07										+
F 4-	Interoffice Channel mileage, per mile or fraction of mile	<u> </u>		UEP9D	MIGBM	0.0222										<u> </u>
	e Activations (DS0) Centrex Loops on Channelized DS1 Servic	e							├ ────							
D4 Cha	nnel Bank Feature Activations				400000	0.00										┥────
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9D	1PQWS	0.62			├────							───
			1						1 1		i i		1		1	1

UNBUNDLE	D NETWORK ELEMENTS - Georgia												Attachment:	2	Exhi	ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Submitted Elec	Submitted	Charge -	Charge - Manual Svc Order vs.	Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec	urring	Nonrecurrin	g Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP9D	1PQW7	0.62										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP9D	1PQWP	0.62										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9D	1PQWV	0.62										
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot			UEP9D	1PQWQ	0.62										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9D	1PQWA	0.62										
Non-Re	ecurring Charges (NRC) Associated with UNE-P Centrex															
	NRC Conversion Currently Combined Switch-As-Is with allowed changes, per port			UEP9D	USAC2		2.01	0.3108					33.67	7.88		
	New Centrex Standard Common Block			UEP9D	M1ACS	0.00	659.41						33.67	7.88		
	New Centrex Customized Common Block			UEP9D	M1ACC	0.00	659.41				1		33.67	7.88		
	NAR Establishment Charge, Per Occasion			UEP9D	URECA	0.00	71.88						33.67	7.88		
Note 1	- Required Port for Centrex Control in 1AESS, 5ESS & EWSD															
Note 2	2 - Requres Interoffice Channel Mileage															
	- Requires Specific Customer Premises Equipment															
Note:	Rates displaying an "R" in Interim column are interim and sub	ject to	rate tru	e-up as set forth in	General Tern	ns and Conditio	ons.									

JNBUNDLE	D NETWORK ELEMENTS - Kentucky												Attachment: 2	2	Exhi	oit: B	
ATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st		Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I	
							Nonrec	urring	Nonrecurring	Disconnect			0990	Rates(\$)			├──
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	<u> </u>
	ne" shown in the sections for stand-alone loops or loops as pa			tion refers to Geogra	phically Deav	eraged UNE Zor	nes. To view G	eorgraphically	Deaveraged U	NE Zone Desiga	antions by C	O, refer to I	nternet Websi	te:			
	ww.interconnection.bellsouth.com/become_a_clec/html/interco SUPPORT SYSTEMS	onnection	n.htm	1	1	1 1				1							├──
NOTE:	(1) Electronic Service Order: CLEC should contact its contract th regional electronic service ordering charge. CLEC may elect													y contained in	this rate exhit	it is the	
that can	(2) Any element that can be ordered electronically will be billed not be ordered electronically at present per the BBR-LO, the lis ed to a CLECs bill when it submits an LSR to BellSouth.																
be appli	Manual Service Order Charge, per LSR, Disconnect Only (KY)	1			SOMAN	1			0.99								\vdash
	Electronic OSS Charge, per LSR, submitted via BST's OSS	1															
	interactive interfaces (Regional) DATE ADVANCEMENT CHARGE	+			SOMEC	<u> </u>	3.50										├──
	The Expedite charge will be maintained commensurate with Be	ellSouth's	s FCC	No.1 Tariff. Section 5	as applicabl	e.											<u> </u>
	UNE Expedite Charge per Circuit or Line Assignable USOC, per									1							
	Day			ALL UNE	SDASP		200.00										<u> </u>
	XCHANGE ACCESS LOOP	<u> </u>				↓				L							—
2-WIRE	ANALOG VOICE GRADE LOOP 2-Wire Analog Voice Grade Loop - Service Level 1- Zone 1		1	UEANL	UEAL2	10.56	46.66	22.57	26.65	7.65		7.86					—
	2-Wire Analog Voice Grade Loop - Service Level 1- Zone 1 2-Wire Analog Voice Grade Loop - Service Level 1- Zone 2	1	2	UEANL	UEAL2	15.34	46.66	22.57	26.65	7.65		7.86					<u> </u>
	2-Wire Analog Voice Grade Loop - Service Level 1- Zone 3	1	3	UEANL	UEAL2	31.11	46.66	22.57	26.65	7.65		7.86					
	Loop Testing - Basic 1st Half Hour			UEANL	URET1		46.88	46.88				7.86					
	Loop Testing - Basic Additional Half Hour	-		UEANL	URETA	┥───┤	24.16	24.16		ļ		7.86					<u> </u>
	CLEC to CLEC Conversion Charge Without Outside Dispatch (UVL-SL1)			UEANL	UREWO		15.78	8.94				7.86					
	(UVL-SL1) Unbundled Voice Loop, Unbundled Non-Design Voice Loop, billing	1		OLANL	UNEWU		13.78	0.94				00.1					├──
	for BST providing make-up	·		UEANL	UEANM		13.49	13.49									
	Manual Order Coordination for UVL-SL1s (per loop)			UEANL	UEAMC		9.00	9.00									
	Order Coordination for Specified Conversion Time for UVL-SL1																i i
2 MIDE	(per LSR) Unbundled COPPER LOOP			UEANL	OCOSL		23.01	23.01									—
2-WIRE	2-Wire Unbundled Copper Loop - Non-Designed Zone 1		1	UEQ	UEQ2X	10.58	44.97	20.89	25.64	6.65		7.86					┝──
	2 Wire Unbundled Copper Loop - Non-Designed - Zone 2	i	2	UEQ	UEQ2X	11.51	44.97	20.89	25.64	6.65		7.86					
	2 Wire Unbundled Copper Loop - Non-Designed - Zone 3		3	UEQ	UEQ2X	13.19	44.97	20.89	25.64	6.65		7.86					
	Order Coordination 2 Wire Unbundled Copper Loop - Non-																Ì
	Designed (per loop) Unbundled Copper Loop, Non-Designed Billing for BST providing			UEQ	USBMC		9.00	9.00									
	make-up			UEQ	UEQMU		13.49	13.49									i i
	Loop Testing - Basic 1st Half Hour			UEQ	URET1		46.88	46.88				7.86					L
	Loop Testing - Basic Additional Half Hour			UEQ	URETA		24.16	24.16				7.86					
	CLEC to CLEC Conversion Charge Without Outside Dispatch			1150		T						=					1
	(UCL-ND) XCHANGE ACCESS LOOP			UEQ	UREWO	<u>├</u> ───┤	14.27	7.43				7.86					
	ANALOG VOICE GRADE LOOP	1		1	1	1 1			-	1							<u> </u>
	2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zone 1		1	UEPSR UEPSB	UEALS	10.56	46.66	22.57	26.65	7.65		7.86					
	2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zone 1 2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting-		1	UEPSR UEPSB	UEABS	10.56	46.66	22.57	26.65	7.65		7.86					\vdash
	2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting- Zone 2		2	UEPSR UEPSB	UEALS	15.34	46.66	22.57	26.65	7.65		7.86					1
	2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting-	1															
	Zone 2 2 Wire Anales Vision Crade Lean Service Level 4 Line Spitting	<u> </u>	2	UEPSR UEPSB	UEABS	15.34	46.66	22.57	26.65	7.65		7.86					—
	2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zone 3	1	3	UEPSR UEPSB	UEALS	31.11	46.66	22.57	26.65	7.65		7.86					1
	2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-	1					40.00		20.00	1.00		7.00					<u> </u>
	Zone 3		3	UEPSR UEPSB	UEABS	31.11	46.66	22.57	26.65	7.65		7.86					L
UNE Lo	op Rates for Line Splitting					10 =-				l							_
	2-Wire Voice Grade Loop (SL1) for Line Splitting - Zone 1 2-Wire Voice Grade Loop (SL1) for Line Splitting - Zone 2	<u> </u>	1	UEPRX UEPRX	UEPLX UEPLX	10.79 15.52											
	2-Wire Voice Grade Loop (SL1) for Line Splitting - Zone 2 2-Wire Voice Grade Loop (SL1) for Line Splitting - Zone 3	1	3	UEPRX	UEPLX	31.74				<u> </u>							—
NBUNDLED E	XCHANGE ACCESS LOOP																
2-WIRE	ANALOG VOICE GRADE LOOP																
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or					10	101-5-	a	=0								1
	Ground Start Signaling - Zone 1		1	UEA	UEAL2	12.67	134.89	81.87	73.65	14.88		7.86					L

NDUNULE	D NETWORK ELEMENTS - Kentucky	1			1	r							Attachment: 2			bit: B	
TEGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l	
															DISC 1St	DISC Add I	
						Rec	Nonreo First	urring Add'l	Nonrecurring First	Disconnect Add'l	SOMEC	SOMAN	OSS SOMAN	Rates(\$) SOMAN	SOMAN	SOMAN	╞
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling - Zone 2		2	UEA	UEAL2	17.45	134.89	81.87	73.65	14.88	SOWIEC	7.86	SOWAN	SOWAN	SOMAN	SOWAN	
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling - Zone 3		3	UEA	UEAL2	33.22	134.89	81.87	73.65	14.88		7.86					
	Order Coordination for Specified Conversion Time (per LSR) 2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse			UEA	OCOSL		23.01										-
	Battery Signaling - Zone 1 2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse		1	UEA	UEAR2	12.67	134.89	81.87	73.65	14.88		7.86					-
	Battery Signaling - Zone 2 2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse		2	UEA	UEAR2	17.45	134.89	81.87	73.65	14.88		7.86					_
	Battery Signaling - Zone 3 Order Coordination for Specified Conversion Time (per LSR)		3	UEA UEA	UEAR2 OCOSL	33.22	134.89 23.01	81.87	73.65	14.88		7.86					_
	CLEC to CLEC Conversion Charge without outside dispatch			UEA	UREWO		87.72	36.36				7.86					t
	ANALOG VOICE GRADE LOOP																L
	4-Wire Analog Voice Grade Loop - Zone 1		1	UEA UEA	UEAL4 UEAL4	29.26 34.25	164.11 164.11	112.36 112.36	78.91 78.91	18.66 18.66		7.86 7.86					╀
	4-Wire Analog Voice Grade Loop - Zone 2 4-Wire Analog Voice Grade Loop - Zone 3	1	3	UEA	UEAL4 UEAL4	34.25	164.11 164.11	112.36	78.91	18.66		7.86					+
	Order Coordination for Specified Conversion Time (per LSR)			UEA	OCOSL		23.01										t
	CLEC to CLEC Conversion Charge without outside dispatch ISDN DIGITAL GRADE LOOP			UEA	UREWO		87.72	36.36				7.86					F
	2-Wire ISDN Digital Grade Loop - Zone 1		1	UDN	U1L2X	18.44	146.77	95.02	71.38	13.83		7.86					╈
	2-Wire ISDN Digital Grade Loop - Zone 2			UDN	U1L2X	25.08	146.77	95.02	71.38	13.83		7.86					T
	2-Wire ISDN Digital Grade Loop - Zone 3			UDN	U1L2X	42.87	146.77	95.02	71.38	13.83		7.86					L
	Order Coordination For Specified Conversion Time (per LSR) CLEC to CLEC Conversion Charge without outside dispatch			UDN UDN	OCOSL UREWO		23.01 91.63	44.16				7.86					┢
	Universal Digital Channel (UDC) COMPATIBLE LOOP			ODN	UKEWO		91.03	44.10				7.00					$^+$
	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone 1		1	UDC	UDC2X	18.44	146.77	95.02	71.38	13.83		7.86					
	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone 2		2	UDC	UDC2X	25.08	146.77	95.02	71.38	13.83		7.86					
	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone 3		3	UDC	UDC2X	42.87	146.77	95.02	71.38	13.83		7.86					
	CLEC to CLEC Conversion Charge without outside dispatch			UDC	UREWO		91.63	44.16				7.86					T
2-WIRE	ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPA		OOP														+
	2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation - Zone 1 2 Wire Unbundled ADSL Loop including manual service inquiry &		1	UAL	UAL2X	10.82	141.98	79.73	69.02	11.47		7.86					
	facility reservation - Zone 2 Wire Unbundled ADSL Loop including manual service inquiry &		2	UAL	UAL2X	11.79	141.98	79.73	69.02	11.47		7.86					_
	facility reservation - Zone 3		3	UAL	UAL2X	12.87	141.98	79.73	69.02	11.47		7.86					
	Order Coordination for Specified Conversion Time (per LSR) 2 Wire Unbundled ADSL Loop without manual service inquiry &			UAL	OCOSL		23.01										╞
	facility reservaton - Zone 1 2 Wire Unbundled ADSL Loop without manual service inquiry &		1	UAL	UAL2W	10.82	121.18	69.00	69.09	11.54		7.86					┢
-	facility reservaton - Zone 2 2 Wire Unbundled ADSL Loop without manual service inquiry &		2	UAL	UAL2W	11.79	121.18	69.00	69.09	11.54		7.86					┢
_	facility reservaton - Zone 3		3	UAL	UAL2W	12.87	121.18	69.00	69.09	11.54		7.86					1
-	Order Coordination for Specified Conversion Time (per LSR) CLEC to CLEC Conversion Charge without outside dispatch			UAL UAL	OCOSL UREWO		23.01 86.20	40.40				7.86					┢
2-WIRE	HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPAT	TIBLE LO	OOP		0		00.20	0+.0+				7.00					t
	2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation - Zone 1		1	UHL	UHL2X	8.75	151.54	89.29	69.09	11.54		7.86					Γ
	2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation - Zone 2		2	UHL	UHL2X	9.56	151.54	89.29	69.09	11.54		7.86					
	2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation - Zone 3		3	UHL	UHL2X	10.61	151.54	89.29	69.09	11.54		7.86					
	Order Coordination for Specified Conversion Time (per LSR) 2 Wire Unbundled HDSL Loop without manual service inquiry and			UHL	OCOSL		23.01										F
	facility reservation - Zone 1 2 Wire Unbundled HDSL Loop without manual service inquiry and		1	UHL	UHL2W	8.75	130.74	78.56	69.09	11.54		7.86					┢
_	facility reservation - Zone 2 2 Wire Unbundled HDSL Loop without manual service inquiry and	-	2	UHL	UHL2W	9.56	130.74	78.56	69.09	11.54		7.86					┢
	facility reservation - Zone 3		3	UHL	UHL2W	10.61	130.74	78.56	69.09	11.54		7.86					

	NETWORK ELEMENTS - Kentucky												Attachment: 2	2	Exhi	bit: B	T
CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge -	
					-	Rec	Nonrec		Nonrecurring		001150	00141		Rates(\$)	001111	00000	
				UHL	UREWO		First	Add'l	First	Add'l	SOMEC	SOMAN 7.86	SOMAN	SOMAN	SOMAN	SOMAN	
	CLEC to CLEC Conversion Charge without outside dispatch HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPAT			UHL	UREWO		86.14	40.40				7.00					
	4 Wire Unbundled HDSL Loop including manual service inquiry and																+
	acility reservation - Zone 1		1	UHL	UHL4X	13.95	185.75	123.50	74.95	14.69		7.86					
	4-Wire Unbundled HDSL Loop including manual service inquiry and			0112	OTIL IX	10.00	100.10	120.00	1 1.00	1 1.00		1.00					1
	acility reservation - Zone 2	1	2	UHL	UHL4X	15.68	185.75	123.50	74.95	14.69		7.86					
4	4-Wire Unbundled HDSL Loop including manual service inquiry and																T
	acility reservation - Zone 3		3	UHL	UHL4X	16.98	185.75	123.50	74.95	14.69		7.86					
	Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		23.01										_
	4-Wire Unbundled HDSL Loop without manual service inquiry and																
	acility reservation - Zone 1		1	UHL	UHL4W	13.95	164.95	114.04	77.32	15.80		7.86					_
	4-Wire Unbundled HDSL Loop without manual service inquiry and	1	2	UHL	UHL4W	15.68	464.05	444.04	77.32	15.80		7.00			1		
	acility reservation - Zone 2 4-Wire Unbundled HDSL Loop without manual service inquiry and		2	UTL		15.68	164.95	114.04	11.32	15.80		7.86		1			+
	acility reservation - Zone 3	1	3	UHL	UHL4W	16.98	164.95	114.04	77.32	15.80		7.86					
	Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL	10.30	23.01	114.04	11.52	10.00		7.50					+
	CLEC to CLEC Conversion Charge without outside dispatch			UHL	UREWO		86.14	40.40				7.86					1
	DS1 DIGITAL LOOP																L
	4-Wire DS1 Digital Loop - Zone 1			USL	USLXX	86.47	306.69	174.44	65.83	14.55		7.86					I.
	4-Wire DS1 Digital Loop - Zone 2			USL	USLXX	114.10	306.69	174.44	65.83	14.55		7.86					
	4-Wire DS1 Digital Loop - Zone 3		3	USL	USLXX	297.76	306.69	174.44	65.83	14.55		7.86					_
	Order Coordination for Specified Conversion Time (per LSR)			USL	OCOSL		23.01										_
	CLEC to CLEC Conversion Charge without outside dispatch			USL	UREWO		101.09	43.04									_
	19.2, 56 OR 64 KBPS DIGITAL GRADE LOOP					07.50	457.04	100.00	70.04	40.00		7.00					_
	4 Wire Unbundled Digital 19.2 Kbps			UDL UDL	UDL19 UDL19	27.59 32.48	157.81 157.81	106.06 106.06	78.91 78.91	18.66 18.66		7.86					+
	4 Wire Unbundled Digital 19.2 Kbps 4 Wire Unbundled Digital 19.2 Kbps			UDL	UDL19	32.46	157.81	106.06	78.91	18.66		7.86				-	+
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 1			UDL	UDL56	27.59	157.81	106.06	78.91	18.66		7.86					+
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 2			UDL	UDL56	32.48	157.81	106.06	78.91	18.66		7.86					+
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 3			UDL	UDL56	36.37	157.81	106.06	78.91	18.66		7.86					T
C	Order Coordination for Specified Conversion Time (per LSR)			UDL	OCOSL		23.01										T
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 1		1	UDL	UDL64	27.59	157.81	106.06	78.91	18.66		7.86					
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 2		2	UDL	UDL64	32.48	157.81	106.06	78.91	18.66		7.86					_
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 3		3	UDL	UDL64	36.37	157.81	106.06	78.91	18.66		7.86					4
	Order Coordination for Specified Conversion Time (per LSR)			UDL	OCOSL		23.01	10.75				7.00					_
	CLEC to CLEC Conversion Charge without outside dispatch Unbundled COPPER LOOP			UDL	UREWO		102.13	49.75				7.86					+
	2-Wire Unbundled Copper Loop/Short including manual service				-											-	+
i i	nguiry & facility reservation - Zone 1	1	1	UCL	UCLPB	10.82	140.95	78.70	69.09	11.54		7.86			1		1
	2-Wire Unbundled Copper Loop/Short including manual service	1	<u> </u>		552.0	10.02	140.00	10.10	00.09	11.04		7.00					+
ir	nquiry & facility reservation - Zone 2		2	UCL	UCLPB	11.79	140.95	78.70	69.09	11.54		7.86					1
	2 Wire Unbundled Copper Loop/Short including manual service																T
	nquiry & facility reservation - Zone 3		3	UCL	UCLPB	12.87	140.95	78.70	69.09	11.54		7.86					1
	Order Coordination for Unbundled Copper Loops (per loop)		ļ	UCL	UCLMC		9.00	9.00									+
	2-Wire Unbundled Copper Loop/Short without manual service		1	UCL		10.00	100.15	67.07	60.00	44.54		7.00					
	nquiry and facility reservation - Zone 1 2-Wire Unbundled Copper Loop/Short without manual service		1	UCL	UCLPW	10.82	120.15	67.97	69.09	11.54		7.86					+
	nquiry and facility reservation - Zone 2		2	UCL	UCLPW	11.79	120.15	67.97	69.09	11.54		7.86					1
	2-Wire Unbundled Copper Loop/Short without manual service	1	~	552		11.79	120.13	01.31	03.09	11.34		1.00					+
	nquiry and facility reservation - Zone 3		3	UCL	UCLPW	12.87	120.15	67.97	69.09	11.54		7.86					1
	Order Coordination for Unbundled Copper Loops (per loop)	1		UCL	UCLMC	1	9.00	9.00									1
	2-Wire Unbundled Copper Loop/Long - includes manual srvc.																T
	nguiry and facility reservation - Zone 1		1	UCL	UCL2L	24.91	140.95	78.70	69.09	11.54		7.86					4
	2-Wire Unbundled Copper Loop/Long - includes manual svc.																1
	nquiry and facility reservation - Zone 2	I	2	UCL	UCL2L	36.94	140.95	78.70	69.09	11.54	L	7.86					+
	2-Wire Unbundled Copper Loop/Long - includes manual svc. nguiry and facility reservation - Zone 3		3	UCL	UCL2L	69.95	140.95	78.70	69.09	11.54		7.86					1
	nquiry and facility reservation - Zone 3 Order Coordination for Unbundled Copper Loops (per loop)	l	3	UCL	UCL2L	69.95	9.00	9.00	69.09	11.54		00.1			1	-	+
	2-Wire Unbundled Copper Loop/Long - without manual service	1		JUL .	JOLINO	1	9.00	9.00									+
	nguiry and facility reservation - Zone 1	1	1	UCL	UCL2W	24.91	120.15	67.97	69.09	11.54		7.86			1		1
ir	and the second sec	1	+ · ·	1	1									-	1		1
	2-Wire Unbundled Copper Loop/Long - without manual service																
2	2-Wire Unbundled Copper Loop/Long - without manual service nquiry and facility reservation - Zone 2		2	UCL	UCL2W	36.94	120.15	67.97	69.09	11.54		7.86					

NBUNDLE	D NETWORK ELEMENTS - Kentucky												Attachment: 2			bit: B	Ť
TEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l	
						Rec	Nonrec		Nonrecurring					Rates(\$)			_
				1101			First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	+-
_	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		9.00	9.00									+
	CLEC to CLEC Conversion Charge without outside dispatch (UCL-			1101			07.00	40.40				7.00					
4 14/10/	Des)			UCL	UREWO		97.23	42.48				7.86					+
4-WIRE	4-Wire Copper Loop/Short - including manual service inquiry and																+
	facility reservation - Zone 1		4	UCL	UCL4S	16.92	170.31	108.06	74.95	14.69		7.86					
_	4-Wire Copper Loop/Short - including manual service inquiry and			UCL	UCL45	10.92	170.31	106.06	74.95	14.09		1.00					+
			2	UCL	UCL4S	17.36	170.31	108.06	74.95	14.69		7.00					
	facility reservation - Zone 2		2	UCL	UCL45	17.30	170.31	106.06	74.95	14.09		7.86					+
	4-Wire Copper Loop/Short - including manual service inquiry and		3	UCL	UCL4S	28.10	170.31	108.06	74.95	14.69		7.86					
	facility reservation - Zone 3		3	UCL	UCL45 UCLMC	28.10		9.00	74.95	14.69		7.86					+
_	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLINC		9.00	9.00									_
	4-Wire Copper Loop/Short - without manual service inquiry and		1	UCL	UCL4W	16.92	149.52	97.33	74.05	14.69		7.86					
_	facility reservation - Zone 1			UCL	UCL4W	10.92	149.52	97.33	74.95	14.09		1.00					+
	4-Wire Copper Loop/Short - without manual service inquiry and		2	UCL	UCL4W	17.36	149.52	97.33	74.95	14.69		7.86					
+	facility reservation - Zone 2			UUL	UUL4VV	17.36	149.52	97.33	74.95	14.69		7.86					+
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation - Zone 3		3	UCL	UCL4W	28.10	149.52	97.33	74.95	14.69		7.86					1
+	Order Coordination for Unbundled Copper Loops (per loop)		3	UCL	UCL4W UCLMC	20.10	9.00	97.33	74.95	14.09		00.1					+
+	4-Wire Unbundled Copper Loop/Long - includes manual svc.			UUL	UCLIVIC	1 1	9.00	9.00									+
	inquiry and facility reservation - Zone 1		1	UCL	UCL4L	46.91	170.31	108.06	74.95	14.69		7.86					
_	4-Wire Unbundled Copper Loop/Long - includes manual svc.			UCL	UCL4L	40.91	170.31	106.06	74.95	14.09		1.00					+
	inquiry and facility reservation - Zone 2		2	UCL	UCL4L	45.78	170.31	108.06	74.95	14.69		7.86					
_	4-Wire Unbundled Copper Loop/Long - includes manual svc.		2	UCL	UCL4L	45.76	170.31	106.06	74.95	14.09		1.00					+
	inquiry and facility reservation - Zone 3		3	UCL	UCL4L	171.34	170.31	108.06	74.95	14.69		7.86					
_			3	UCL	UCL4L UCLMC	171.34		9.00	74.95	14.69		7.86					+
_	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLINC		9.00	9.00									+
	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone 1		1	UCL	UCL4O	46.91	149.52	97.33	74.95	14.69		7.86					
_			1	UCL	UCL40	46.91	149.52	97.33	74.95	14.69		7.86					+
	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone 2		2	UCL	UCL4O	45.78	149.52	97.33	74.95	14.69		7.86					
_			2	UCL	UCL40	45.78	149.52	97.33	74.95	14.69		7.86					+
	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry		3	UCL		171.34	149.52	07.00	74.05	14.69		7.00					
_	and facility reservation - Zone 3		3		UCL40	171.34		97.33	74.95	14.69		7.86					_
_	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		9.00	9.00									_
	CLEC to CLEC Conversion Charge without outside dispatch (UCL	î		1101			07.00	42.48				7.00					
P MODIFIC	Des)			UCL	UREWO		97.23	42.48				7.86					_
	ATION																_
				UAL, UHL, UCL,													
	Unbundled Loop Modification, Removal of Loop Calls, CAM-			UEQ, ULS, UEA,								1					
	Unbundled Loop Modification, Removal of Load Coils - 2 Wire			UEANL, UDL, UDC,	LIL MOI		0.04	0.04				7.00					1
<u> </u>	pair less than or equal to 18k ft		<u> </u>	UDN, UDL, USL	ULM2L		9.24	9.24				7.86					+
	Unbundled Loop Modification, Removal of Load Coils - 2 wire			UCL, ULS, UEQ	LIL MOC		342.24	342.24				7.86					
—	greater than 18k ft Unbundled Loop Modification Removal of Load Coils - 4 Wire less	<u> </u>		UUL, ULS, UEQ	ULM2G		342.24	342.24				7.86					+
	than or equal to 18K ft			UHL, UCL	ULM4L		9.24	9.24				7.86					
_	Unbundled Loop Modification Removal of Load Coils - 4 Wire			UTIL, UUL	ULIVI4L		9.24	3.24				1.00					+
	pair greater than 18k ft			UCL	ULM4G		342.24	342.24				7.86					
		<u> </u>		UCL UAL, UHL, UCL,	ULIVI4G		342.24	342.24				1.80					+
				UEQ. UEF. ULS.													
				UEA, UEANL, UDL.													
	Unbundled Loop Modification Removal of Bridged Tap Removal,			UEA, UEANL, UDL, UDC, UDN, UDL,								1					
	per unbundled loop			USL	ULMBT		10.47	10.47				7.86					
LOOPS				UUL		1 1	10.47	10.47				00.1					+
	op Distribution																+
Sub-LO	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-				ł	1											+
	Un			UEANL	USBSA		207.91	207.91				7.86					
					USBOA		207.91	207.91				1.00					+
	Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-Up			UEANL	USBSB		12.50	12.50				7.86					
_	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility	- '		ULANL	03030	1	12.50	12.30				1.00					+
	Set-Up			UEANL	USBSC		80.87	80.87				7.86					
				OLANL	03030	1 1	00.07	00.07				00.1					+
	Sub-Loop - Per Building Equipment Boom - Dor 25 Dair Densi Cot				1	1						1			1		
	Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-				LIGBOD		45.04	AE 0.4									
<u> </u>	Up	I		UEANL	USBSD		45.04	45.04				7.86					┿
<u> </u>	Up Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop -	1	4			6.04			E0 04	7.00							╈
<u> </u>	Up	1	1	UEANL UEANL	USBSD USBN2	6.34	45.04 85.03	45.04 39.05	59.81	7.90		7.86 7.86					t

UNBUNDLI	ED NETWORK ELEMENTS - Kentucky												Attachment: 2	2	Exhi	bit: B	(
CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-	
							Manua		Neuroeuring	Discourset			1st	Add'l	Electronic- Disc 1st	Disc Add'l	<u> </u>
						Rec	Nonree First	Add'l	Nonrecurring First	Add'l	SOMEC	SOMAN	SOMAN	Rates(\$) SOMAN	SOMAN	SOMAN	
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 3	I	3	UEANL	USBN2	14.82	85.03	39.05	59.81	7.90	0011120	7.86		0011211	00112 11	00112 11	
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		9.00	9.00									
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 1		1	UEANL	USBN4	8,14	102.31	56.32	65.24	10.88		7.86					1
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 2		2	UEANL	USBN4	8.63	102.31	56.32	65.24	10.88		7.86					
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 3		3	UEANL	USBN4	25.60	102.31	56.32	65.24	10.88		7.86					
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		9.00	9.00									l
	Sub-Loop 2-Wire Intrabuilding Network Cable (INC)	I		UEANL	USBR2	2.57	68.35	22.36	59.81	7.90		7.86					
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		9.00	9.00									i i
	Sub-Loop 4-Wire Intrabuilding Network Cable (INC)	I		UEANL	USBR4	4.98	76.49	30.51	65.24	10.88		7.86					\square
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		9.00	9.00									l
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 1	I		UEF	UCS2X	5.45	85.03	39.05	59.81	7.90		7.86					
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2	1		UEF	UCS2X	7.06	85.03	39.05	59.81	7.90		7.86					
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 3 Order Coordination for Unbundled Sub-Loops, per sub-loop pair		3	UEF	UCS2X USBMC	9.67	85.03 9.00	39.05 9.00	59.81	7.90		7.86					
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 1	I	1	UEF	UCS4X	7.09	102.31	56.32	65.24	10.88		7.86					<u> </u>
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 2	I	2	UEF	UCS4X	8.66	102.31	56.32	65.24	10.88		7.86					
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 3	I	3	UEF	UCS4X	19.40	102.31	56.32	65.24	10.88		7.86					
l la han	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC		9.00	9.00									
Undru	dled Sub-Loop Modification Unbundled Sub-Loop Modification - 2-W Copper Dist Load																
	Coil/Equip Removal per 2-W PR Unbundled Sub-loop Modification - 4-W Copper Dist Load			UEF	ULM2X		5.23	5.23				7.86					─
	Coil/Equip Removal per 4-W PR Unbundled Sub-loop Modification - 2-w/4-w Copper Dist Bridged			UEF	ULM4X		5.23	5.23				7.86					
	Tap Removal, per PR unloaded			UEF	ULM4T		7.97	7.97				7.86					L
Unbur	dled Network Terminating Wire (UNTW) Unbundled Network Terminating Wire (UNTW) per Pair			UENTW	UENPP	0.53	23.51	23.51				7.86					┝───
Netwo	rk Interface Device (NID)					0.55		20.01				7.00					
	Network Interface Device (NID) - 1-2 lines			UENTW	UND12		73.53	49.47				7.86					Ĺ
	Network Interface Device (NID) - 1-6 lines Network Interface Device Cross Connect - 2 W			UENTW UENTW	UND16 UNDC2	-	115.96 8.56	91.91 8.56	-	-		7.86 7.86					
	Network Interface Device Cross Connect - 4W			UENTW	UNDC4		8.56	8.56				7.86					<u> </u>
SUB-LOOPS																	
Sub-L	Dop Feeder USL-Feeder, DS0 Set-up per Cross Box location - CLEC			UEA,													
	Distribution Facility set-up USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-			UDN,UCL,UDL,UDC UEA,			207.91					7.86					
	up USL Feeder DS1 Set-up at DSX location, per DS1 termination	<u> </u>		UDN,UCL,UDL,UDC	USBFX USBFZ		12.50 527.98	12.50 11.32				7.86 7.86					─
	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1		1	UEA	USBFA	7.67	114.83	64.61	72.34	17.21		7.86					<u> </u>
	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice			-													<u> </u>
	Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start,		2		USBFA	9.70	114.83	64.61	72.34	17.21		7.86					<u> </u>
	Voice Grade - Zone 3 Order Coordination for Specified Conversion Time, per LSR		3	UEA UEA	USBFA OCOSL	19.53	114.83 23.01	64.61	72.34	17.21		7.86					
	Unbundlde Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 1		1	UEA	USBFB	7.67	114.83	64.61	72.34	17.21		7.86					
	Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2		2	UEA	USBFB	9.70	114.83	64.61	72.34	17.21		7.86					
	Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice Grade - Zone 3		3	UEA	USBFB	19.53	114.83	64.61	72.34	17.21		7.86					
	Order Coordination for Specified Time Conversion, per LSR			UEA	OCOSL		23.01										

UNBUNDLE	D NETWORK ELEMENTS - Kentucky												Attachment: 2	2	Exh	bit: B	
CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs.	
													Electronic- 1st	Electronic- Add'l	Electronic- Disc 1st	Electronic- Disc Add'l	
						Rec	Nonree	curring Add'l	Nonrecurring First	Disconnect Add'l	SOMEC	SOMAN	OSS SOMAN	Rates(\$) SOMAN	SOMAN	SOMAN	<u> </u>
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1		1	UEA	USBFC	7.67	114.83	64.61	72.34	17.21	SOMEC	7.86	SOWAN	SOMAN	SOMAN	SOWAN	
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2		2	UEA	USBFC	9.70	114.83	64.61	72.34	17.21		7.86					
	Unbundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse Battery, Voice Grade - Zone 3		3	UEA	USBEC	19.53	114.83	64.61	72.34	17.21		7.86					1
	Order Coordination For Specified Conversion Time, per LSR		0	UEA	OCOSL	10.00	23.01	04.01	72.04	17.21		7.00					
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice						404 70	70.00		54.50		7.00					
	Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice		1	UEA	USBFD	22.82	131.73	79.98	81.82	51.56		7.86					
	Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice		2	UEA	USBFD	27.24	131.73	79.98	81.82	51.56		7.86					<u> </u>
	Grade - Zone 3		3	UEA	USBFD	61.41	131.73	79.98	81.82	51.56		7.86					L
	Order Coordination For Specified Conversion Time, Per LSR Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice			UEA	OCOSL		23.01										┝───
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice		1	UEA	USBFE	22.82	131.73	79.98	81.82	51.56		7.86					
	Grade - Zone 2		2	UEA	USBFE	27.24	131.73	79.98	81.82	51.56		7.86					<u> </u>
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 3		3	UEA UEA	USBFE	61.41	131.73	79.98	81.82	51.56		7.86					
	Order Coordination For Specified Conversion Time, Per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 1		1	UEA	OCOSL USBFF	13.00	23.01 131.79	80.04	74.16	16.60		7.86					<u> </u>
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 2			UDN	USBFF	16.95	131.79	80.04	74.16	16.60		7.86					
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 3		3	UDN	USBFF	28.95	131.79	80.04	74.16	16.60		7.86					
	Order Coordination For Specified Conversion Time, Per LSR Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)		1	UDN UDC	OCOSL USBFS	13.00	23.01 131.79	80.04	74.16	16.60		7.86					<u> </u>
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible) Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)			UDC	USBFS	13.00	131.79	80.04	74.16	16.60		7.86					
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)			UDC	USBFS	28.95	131.79	80.04	74.16	16.60		7.86					
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 1			USL	USBFG	62.57	125.43	73.68	81.82	21.56		7.86					
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 2			USL	USBFG	87.71	125.43	73.68	81.82	21.56		7.86					
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 3 Order Coordination For Specified Conversion Time, Per LSR			USL USL	USBFG OCOSL	273.33	125.43 23.01	73.68	81.82	21.56		7.86					
	Unbundled Sub-Loop Feeder, 2-Wire Copper Loop - Zone 1			UCL	USBFH	6.44	105.31	53.57	71.16	13.61		7.86					
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone 2			UCL	USBFH	5.78	105.31	53.57	71.16	13.61		7.86					
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone 3		3	UCL	USBFH	4.25	105.31	53.57	71.16	13.61		7.86					
	Order Coordination For Specified Conversion Time, per LSR			UCL	OCOSL		23.01										
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 1			UCL	USBFJ	11.33	125.55	73.80	77.12	16.86		7.86					└───
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 2 Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 3			UCL	USBFJ USBFJ	10.18 10.32	125.55 125.55	73.80 73.80	77.12	16.86 16.86		7.86					<u> </u>
	Order Coordination For Specified Conversion Time, per LSR	1	3	UCL	OCOSL	10.32	23.01	13.60	11.12	10.00		1.00	1				<u> </u>
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop			UDL	USBFN	20.78	125.43	73.68	81.82	21.56		7.86					
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop			UDL	USBFN	26.41	125.43	73.68	81.82	21.56		7.86					I
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone		3	UDL	USBFN	23.10	125.43	73.68	81.82	21.56	<u> </u>	7.86	<u> </u>				
	1		1	UDL	USBFO	20.78	125.43	73.68	81.82	21.56		7.86					
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone 2		2	UDL	USBFO	26.41	125.43	73.68	81.82	21.56		7.86					I
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone 3		3	UDL	USBFO	23.10	125.43	73.68	81.82	21.56		7.86					<u> </u>
	Order Coordination For Specified Time Conversion, per LSR			UDL	OCOSL		23.01										I
<u> </u>	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 1 Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone	<u> </u>	1	UDL	USBFP	20.78	125.43	73.68	81.82	21.56		7.86					I
<u> </u>	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 2	<u> </u>	2	UDL	USBFP	26.41	125.43	73.68	81.82	21.56		7.86					I
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 3		3	UDL	USBFP	23.10	125.43	73.68	81.82	21.56		7.86					
SUB-LOOPS	Order Coordination For Specified Conversion Time, per LSR			UDL	OCOSL		23.01										┝───
	op Feeder				1	1				 			1				
	Sub Loop Feeder - DS3 - Per Mile Per Month	1		UE3	1L5SL	15.38				1							[
	Sub Loop Feeder - DS3 - Facility Termination Per Month	1		UE3	USBF1	346.30	3,402.59	407.14	160.86	91.19		7.86					
	Sub Loop Feeder – STS-1 – Per Mile Per Month			UDLSX	1L5SL	15.38											

UNBUNDL	ED NETWORK ELEMENTS - Kentucky												Attachment: 2	2	Exhi	bit: B	
													Incremental	Incremental	Incremental	Incremental	
						1					Submitted		Charge -	Charge -	Charge -	Charge -	
			_	200							Elec	Manually	Manual Svc	Manual Svc	Manual Svc		
CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.	
													Electronic-	Electronic-	Electronic-	Electronic-	
													1st	Add'l	Disc 1st	Disc Add'l	
						_	Nonrec	urring	Nonrecurring	Disconnect			055	Rates(\$)			<u> </u>
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	
	Sub Loop Feeder - STS-1 - Facility Termination Per Month	-			USBF7	372.80	3,402.59	407.14	160.86	91.19		7.86					
	Sub Loop Feeder – OC-3 – Per Mile Per Month			UDLO3	1L5SL	11.67											
	Sub Loop Feeder - OC-3 - Facility Termination Protection Per																
	Month Sub Loop Feeder - OC-3 - Facility Termination Per Month			UDLO3 UDLO3	USBF5 USBF2	58.27 564.68	3,402.59	407.14	160.86	91.19		7.86					
	Sub Loop Feeder - OC-3 - Pacinty Termination Per Month Sub Loop Feeder - OC-12 - Per Mile Per Month	1			1L5SL	14.36	3,402.59	407.14	100.00	91.19		7.00					
	Sub Loop Feeder - OC-12 - Facility Termination Protection Per			ODE12	TLUGE	14.50											
	Month	1		UDL12	USBF6	658.35											
	Sub Loop Feeder - OC-12 - Facility Termination Per Month	1		UDL12	USBF3	1,778.00	3,402.59	407.14	160.86	91.19		7.86					
	Sub Loop Feeder - OC-48 - Per Mile Per Month			UDL48	1L5SL	47.11											
	Sub Loop Feeder - OC-48 - Facility Termination Protection Per																
├──-┼──-	Month	+		UDL48	USBF9	330.39	0 507 50	107.1.1	100.00	04.40		7.00					───
├ ── ├ ──	Sub Loop Feeder - OC-48 - Facility Termination Per Month Sub Loop Feeder - OC-12 Interface On OC-48	1			USBF4 USBF8	1,533.00 372.76	3,587.59 804.96	407.14 407.14	160.86 160.86	91.19 91.19		7.86 7.86					──
	Sub Loop Feeder - OC-12 Interface On OC-48			00140	USDFÖ	312.76	804.96	407.14	160.86	91.19		1.86					──
ONBONDEED	Unbundled Loop Concentration - System A (TR008)			ULC	UCT8A	423.72	359.34	359.34				7.86					
	Unbundled Loop Concentration - System B (TR008)				UCT8B	51.60	149.72	149.72				7.86					
	Unbundled Loop Concentration - System A (TR303)				UCT3A	460.27	359.34	359.34		<u> </u>		7.86					
	Unbundled Loop Concentration - System B (TR303)				UCT3B	86.95	149.72	149.72				7.86					
	Unbundled Loop Concentration - DS1 Loop Interface Card	1		ULC	UCTCO	4.90	71.69	51.51	22.99	6.00		7.86					\square
						7.70	10.50	10.50		0.07		7.00					
	Unbundled Loop Concentration - ISDN Loop Interface (Brite Card)	1		UDN	ULCC1	7.78	16.59	16.50	8.42	8.37		7.86					
	Unbundled Loop Concentration - UDC Loop Interface (Brite Card)			UDC	ULCCU	7.78	16.59	16.50	8.42	8.37		7.86					
	Unbundled Loop Concentration - 2 Wire Voice-Loop Start or			ODC	ULCCU	1.10	10.59	10.50	0.42	0.37		7.00					
	Ground Start Loop Interface (POTS Card)			UEA	ULCC2	1.95	16.59	16.50	8.42	8.37		7.86					
	Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery																
	Loop Interface (SPOTS Card)			UEA	ULCCR	11.58	16.59	16.50	8.42	8.37		7.86					
	Unbundled Loop Concentration - 4 Wire Voice Loop Interface																
	(Specials Card)			UEA	ULCC4	6.90	16.59	16.50	8.42	8.37		7.86					
	Unbundled Loop Concentration - TEST CIRCUIT Card			ULC	UCTTC	33.74	16.59	16.50	8.42	8.37		7.86					
	Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop Interface			UDL	ULCC7	10.23	16.59	16.50	8.42	8.37		7.86					
	Unbundled Loop Concentration - Digital 56 Kbps Data Loop			ODL	ULCC/	10.23	10.59	10.50	0.42	0.37		7.00					
	Interface			UDL	ULCC5	10.23	16.59	16.50	8.42	8.37		7.86					
	Unbundled Loop Concentration - Digital 64 Kbps Data Loop																
	Interface			UDL	ULCC6	10.23	16.59	16.50	8.42	8.37		7.86					
UNE OTHER,	PROVISIONING ONLY - NO RATE																
	NID - Dispatch and Service Order for NID installation			UENTW	UNDBX	0.00	0.00										
	UNTW Circuit Id Establishment, Provisioning Only - No Rate			UENTW	UENCE	0.00	0.00										<u> </u>
	Unbundled Contract Name, Provisioning Only - No Rate			UEANL,UEF,UEQ,U ENTW	UNECN	0.00	0.00										
UNE OTHER	PROVISIONING ONLY - NO RATE	1			UNEUN	0.00	0.00					<u> </u>					
							-										
				UAL,UCL,UDC,UDL,		1											
	Unbundled Contact Name, Provisioning Only - no rate			UDN,UEA,UHL,ULC	UNECN	0.00	0.00										
	Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no rate			UEA,UDN,UCL,UDC	USBFQ	0.00	0.00										<u> </u>
	Links and a d Oak Loope Freedox (1977) O. D. D. J.				10055												
├──	Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no rate Unbundled DS1 Loop - Superframe Format Option - no rate				USBFR CCOSF	0.00	0.00										──
├ ── ┼──	Unbundled DS1 Loop - Superframe Format Option - no rate Unbundled DS1 Loop - Expanded Superframe Format option - no	-		USL	CCOSF	0.00	0.00						-		-		<u> </u>
	rate	1		USL	CCOEF	0.00	0.00				1						1
HIGH CAPAC	ITY UNBUNDLED LOCAL LOOP	1				0.50	0.00			ĺ		1					
			l			1											
	High Capacity Unbundled Local Loop - DS3 - Per Mile per month			UE3	1L5ND	9.25											
	High Capacity Unbundled Local Loop - DS3 - Facility Termination																
	per month			UE3	UE3PX	308.31	551.38	338.08	173.00	120.42		7.86					└───
	Link Operative lakerediation and CTO (CD, 117, 17, 17, 17, 17, 17, 17, 17, 17, 1																
	High Capacity Unbundled Local Loop - STS-1 - Per Mile per month	n	I	UDLSX	1L5ND	9.25											──
	Llick Consolity Unkundled Lesel Leen CTC 4 Franktig																1
	High Capacity Unbundled Local Loop - STS-1 - Facility Termination per month			UDLSX	UDLS1	320.51	551.38	338.08	173.00	120.42		7.86					

UNBUNDI	LED NETWORK ELEMENTS - Kentucky												Attachment: 2	2	Exhi	bit: B	
											Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental	
		1									Submitted	Submitted	Charge -	Charge -	Charge -	Charge -	1
											Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc	
ATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.	
													Electronic-	Electronic-	Electronic-	Electronic-	
													1st	Add'l	Disc 1st	Disc Add'l	
														,	2.00 101	Dictriau	
						Rec	Nonred	curring	Nonrecurring	Disconnect				Rates(\$)			
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	
	Loop Makeup - Preordering Without Reservation, per working or																
	spare facility queried (Manual).			UMK	UMKLW		23.40	23.40									
	Loop Makeup - Preordering With Reservation, per spare facility																
	queried (Manual).			UMK	UMKLP		24.85	24.85									
	Loop MakeupWith or Without Reservation, per working or spare																
	facility queried (Mechanized)			UMK	PSUMK		0.67	0.67									
	UENCY SPECTRUM																
	E SHARING																
SPLI	ITTERS-CENTRAL OFFICE BASED																
	Line Sharing Splitter, per System 96 Line Capacity			ULS	ULSDA	198.83	379.05	0.00	358.55	0.00		7.86					
	Line Sharing Splitter, per System 24 Line Capacity			ULS	ULSDB	49.71	379.05	0.00	358.55	0.00		7.86					
	Line Sharing Splitter, Per System, 8 Line Capacity			ULS	ULSD8	16.94	377.71	0.00	357.29	0.00		7.86					
	Line Sharing-DLEC Owned Splitter in CO-CFA activaton-																1
	deactivation (per LSOD)			ULS	ULSDG		173.62	0.00	100.40	0.00		7.86					L
END	USER ORDERING-CENTRAL OFFICE BASED-HIGH FREQUENC	Y SPECT	RUM														
	Line Sharing - per Line Activation (BST Owned Splitter)	L		ULS	ULSDC	0.61	37.16	21.28	20.17	9.90		7.86					L
	Line Sharing - per Subsequent Activity per Line	1									1						1
	Rearrangement(BST Owned Splitter)			ULS	ULSDS		32.90	16.43				7.86					
	Line Sharing - per Subsequent Activity per Line																
	Rearrangement(DLEC Owned Splitter)			ULS	ULSCS		32.90	16.43				7.86					L
	Line Sharing - per Line Activation (DLEC owned Splitter)	-		ULS	ULSCC	0.61	47.44	19.31	20.67	12.74		7.86					
	E SPLITTING																
END	USER ORDERING-CENTRAL OFFICE BASED																
	Line Splitting - per line activation DLEC owned splitter	Ī		UEPSR UEPSB	UREOS	0.61											
	Line Splitting - per line activation BST owned - physical	Ī		UEPSR UEPSB	UREBP	0.61	37.02	21.20	21.10	9.87		7.86					
	Line Splitting - per line activation BST owned - virtual			UEPSR UEPSB	UREBV	0.61	37.02	21.20	21.10	9.87		7.86					
	NOTE SITE HIGH FREQUENCY SPECTRUM																
SPLI	ITTERS-REMOTE SITE																
	Remote Site Line Share BellSouth Owned Splitter, 24 Port			ULS	ULSRB	50.83	377.71	0.00	357.29	0.00		7.86					
	Remote Site Line Share Cable Pair Activation CLEC Owned at RS	5															1
	and Deactivation			ULS	ULSTG		74.38	0.00	46.77	0.00		7.86					
END	USER ORDERING-REMOTE SITE HIGH FREQUENCY SPECTRU	MAKAF	REMOT	E SITE LINE SHARIN	IG												
	Remote Site Line Share Line Activationfor End User Served at																1
	RS, BST Splitter			ULS	ULSRC	0.61	37.16	21.28	20.17	9.90		7.86					
	RS Line Share Line Activation for End User served at RS, CLEC																
	Splitter			ULS	ULSTC	0.61	37.16	21.28	20.17	9.90		7.86					L
	D DEDICATED TRANSPORT																
	TE: INTEROFFICE CHANNEL DEDICATED TRANSPORT - minimu	m billing	period	- below DS3=one mo	onth, DS3/ST	S-1=four month	S										
	EROFFICE CHANNEL - DEDICATED TRANSPORT																
	Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade -	T															
	Per Mile per month			U1TVX	1L5XX	0.01											
	Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade -																1
	Facility Termination			U1TVX	U1TV2	29.11	47.34	31.78	22.77	8.75		7.86					
	Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade																1
	Rev Bat Per Mile per month			U1TVX	1L5XX	0.01											
	Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat																
	Facility Termination			U1TVX	U1TR2	29.11	47.34	31.78	22.77	8.75		7.86					
	Interoffice Channel - Dedicated Transport - 4-Wire Voice Grade -																
	Per Mile per month			U1TVX	1L5XX	0.01											
	Interoffice Channel - Dedicated Transport - 4- Wire Voice Grade -																
	Facility Termination			U1TVX	U1TV4	25.86	47.34	31.78	22.77	8.75		7.86					
	Interoffice Channel - Dedicated Transport - 56 kbps - per mile per																
	month			U1TDX	1L5XX	0.0115											
	Interoffice Channel - Dedicated Transport - 56 kbps - Facility																1
	Termination	1		U1TDX	U1TD5	20.97	47.35	31.78	22.77	8.75		7.86					
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile per																
	month			U1TDX	1L5XX	0.0115											
	Interoffice Channel - Dedicated Transport - 64 kbps - Facility	T															Γ
	Termination			U1TDX	U1TD6	20.97	47.35	31.78	22.77	8.75		7.86					L
	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per																
	month	1		U1TD1	1L5XX	0.23					1						1
	monun																1
	Interoffice Channel - Dedicated Tranport - DS1 - Facility																

NRONDL	ED NETWORK ELEMENTS - Kentucky		r		1						a - ·	a - ·	Attachment: 2			bit: B	┢
ATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I	
						Rec	Nonrec		Nonrecurring					Rates(\$)		-	
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	
	Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month			U1TD3	1L5XX	4.97											
	Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month			U1TD3	U1TF3	1,175.15	335.40	219.24	89.57	87.75		7.86					
	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per month			U1TS1	1L5XX	4.97											
	Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination			U1TS1	U1TFS	1,149.51	335.40	219.24	89.57	87.75		7.86					
LOCA	L CHANNEL - DEDICATED TRANSPORT																1
	: LOCAL CHANNEL DEDICATED TRANSPORT - minimum billing	period -	below	DS3=one month, DS	S3/STS-1=fou	ir months											1
	Local Channel - Dedicated - 2-Wire Voice Grade			ULDVX	ULDV2	18.57	265.78	46.96	46.79	4.98		7.86					1
	Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat			ULDVX	ULDR2	18.57	265.78	46.96	46.79	4.98		7.86					T
	Local Channel - Dedicated - 4-Wire Voice Grade	1		UNDVX	ULDV4	19.86	266.48	47.65	47.54	5.73		7.86					1
	Local Channel - Dedicated - DS1 - Zone 1	1	1	ULDD1	ULDF1	40.46	209.60	176.51	30.21	21.07		7.86					T
	Local Channel - Dedicated - DS1 - Zone 2		2	ULDD1	ULDF1	43.39	209.60	176.51	30.21	21.07		7.86	1				1
	Local Channel - Dedicated - DS1 - Zone 3		3	ULDD1	ULDF1	164.50	209.60	176.51	30.21	21.07		7.86	1				1
	Local Channel - Dedicated - DS3 - Per Mile per month		Ŭ	ULDD3	1L5NC	8.74	200.00		00.21	257			1				1
	Local Channel - Dedicated - DS3 - Facility Termination		1	ULDD3	ULDF3	576.05	551.38	338.08	173.00	120.42		7.86	1				1
	Local Channel - Dedicated - STS-1- Per Mile per month	1		ULDS1	1L5NC	8.74	001.00	500.00		120.12			1		1		t - t
1	Local Channel - Dedicated - STS-1 - Facility Termination	1		ULDS1	ULDFS	543.24	551.38	338.08	173.00	120.42		7.86	1		1		+
RK FIBER		1			522.0	0-10.24	001.00	300.00	170.00	120.42		7.00					+
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereo	f															+
	per month - Local Channel			UDF	1L5DC	47.01											
-	NRC Dark Fiber - Local Channel	1		UDF	UDFC4	47.01	732.53	192.67	377.27	241.67		7.86					+
_		4		UDF	00FC4		132.53	192.67	311.21	241.07		1.86					+
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereo per month - Interoffice Channel	'		UDF	1L5DF	30.74											1
	NRC Dark Fiber - Interoffice Channel			UDF	UDF14		732.53	192.67	377.27	241.67		7.86					t
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereo	f		00.	05.11		102.00	102.01	011.21	211101		1.00					+
	per month - Local Loop			UDF	1L5DL	47.01											
	NRC Dark Fiber - Local Loop			UDF	UDFL4	47.01	732.53	192.67	377.27	241.67		7.86					+
ACCESS	TEN DIGIT SCREENING			001	ODI L4		102.00	102.01	011.21	241.07		1.00					+
(AUDEDD	8XX Access Ten Digit Screening, Per Call			OHD		0.0006478											+
	8XX Access Ten Digit Screening, Reservation Charge Per 8XX			OND		0.0000470											+
	Number Reserved			OHD	N8R1X		4.14	0.70				7.86					
	8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translations			ОНД			8.78	1.18	7.08	0.86		7.86					
	8XX Access Ten Digit Screening, Per 8XX No. Established With POTS Translations			OHD	N8FTX		8.78	1.18	7.08	0.86		7.86					
	8XX Access Ten Digit Screening, Customized Area of Service Pe 8XX Number	r		OHD	N8FCX		4.14	2.07				7.86					I
	8XX Access Ten Digit Screening, Multiple InterLATA CXR Routing	1						2.57	İ						İ		1
	Per CXR Requested Per 8XX No.	1		OHD	N8FMX		4.85	2.78				7.86					1
	8XX Access Ten Digit Screening, Change Charge Per Request			OHD	N8FAX		4.85	0.70				7.86					Г
	8XX Access Ten Digit Screening, Call Handling and Destination																Γ
	Features		L	OHD	N8FDX		4.14	4.14				7.86					+
	8XX Access Ten Digit Screening w/ 8FL No. Delivery,		L	OHD		0.0006478											+
	8XX Access Ten Digit Screening, w/ POTS No. Delivery,			OHD		0.0006478											4
E INFORM	ATION DATA BASE ACCESS (LIDB)																4
	LIDB Common Transport Per Query	1		OQT	1	0.000023											┶
	LIDB Validation Per Query			OQU		0.0137322											1
	LIDB Originating Point Code Establishment or Change			OQT, OQU	NRPBX		55.12		67.59			7.86					1
NALING (C		1															┶
	CCS7 Signaling Connection, Per 56 Kbps Facility		L	UDB	TPP++	20.71	43.56	43.56	22.45	22.45							+
_	CCS7 Signaling Termination, Per STP Port	1		UDB	PT8SX	151.39											┶
	CCS7 Signaling Usage, Per TCAP Message			UDB		0.0000656											┶
	CCS7 Signaling Connection, Per link (A link)			UDB	TPP++	20.71	43.56	43.56	22.45	22.45		7.86					1
	CCS7 Signaling Connection, Per link (B link) (also known as D link)			UDB	TPP++	20.71	43.56	43.56	22.45	22.45		7.86					1
	CCS7 Signaling Usage, Per ISUP Message			UDB		0.0000164											Т
	CCS7 Signaling Usage Surrogate, per link per LATA			UDB	STU56	751.08											Г
	CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affected			UDB	CCAPO		46.02	46.02	56.43	56.43		7.86					Γ
	CCS7 Signaling Point Code, per Orn Destination Point Code Establishment or Change, Per Stp Affected			UDB	CCAPD		46.02	46.02	56.43	56.43		7.86					t

UNBUND	DLE	O NETWORK ELEMENTS - Kentucky												Attachment: 2	2	Exhi	bit: B	
												Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental	
CATEGOR	ŧ۲	RATE ELEMENTS	Interim	Zone	BCS	usoc			RATES(\$)			Submitted Elec per LSR	Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge - Manual Svc Order vs. Electronic- Disc Add'l	
							Rec	Nonrec		Nonrecurring					Rates(\$)			
E911 SER	VICE		-					First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	<u> </u>
ESTISER	VICE	Local Channel - Dedicated - 2-wr Voice Grade					18.57	265.78	46.96	46.79	4.98		7.86					<u> </u>
		Interoffice Transport - Dedicated - 2-wr Voice Grade Per Mile					0.0115											
		Interoffice Transport - Dedicated - 2-wr Voice Grade Per Facility Termination					29.11	47.34	31.78	22.77	8.75		7.86					
		Local Channel - Dedicated - DS1 - Zone 1					40.46	209.60	176.51	30.21	21.07		7.86					
		Local Channel - Dedicated - DS1 - Zone 2					43.39	209.60	176.51	30.21	21.07		7.86					<u> </u>
		Local Channel - Dedicated - DS1 - Zone 3 Interoffice Transport - Dedicated - DS1 Per Mile					164.50 0.23	209.60	176.51	30.21	21.07		7.86					
		Interonice Transport - Dedicated - DST Per Mile					0.23											
		Interoffice Transport - Dedicated - DS1 Per Facility Termination					96.04	105.52	98.46	23.09	20.49		7.86					
CALLING N		(CNAM) SERVICE	İ			<u> </u>						İ						
		CNAM For DB Owners - Service Establishment			OQV			25.34	25.34	23.30	23.30		7.86					
\vdash		CNAM For Non DB Owners - Service Establishment			OQV			25.34	25.34	23.30	23.30		7.86					
		CNAM For DB Owners - Service Provisioning With Point Code	1		OQV			1.591.54	1,177.08	431.95	317.61		7.86					1
		Establishment CNAM For Non DB Owners - Service Provisioning With Point																
		Code Establishment			OQV OQV		0.0010348	546.40	393.74	438.93	317.61		7.86					
		CNAM for DB Owners, Per Query CNAM for Non DB Owners, Per Query			OQV		0.0010348											
		CNAM (Non-Databs Owner), NRC, applies when using the			OQV		0.0010346											
		Character Based User Interface (CHUI)			OQV	CDDCH		595.00	595.00				7.86					
LNP Query																		
		LNP Charge Per query					0.0008695											
		LNP Service Establishment Manual						13.82	13.82	12.71	12.71		7.86					ļ
0050470		LNP Service Provisioning with Point Code Establishment						953.27	487.00	431.95	317.61		7.86					
OPERATO		LL PROCESSING Oper. Call Processing - Oper. Provided, Per Min Using BST																
		LIDB					1.20											ļ
		Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIDB					1.24											
		Oper. Call Processing - Fully Automated, per Call - Using BST LIDB					0.20											
		Oper. Call Processing - Fully Automated, per Call - Using Foreign					0.20											
INWARD O	PER.	ATOR SERVICES					0.20											
		Inward Operator Services - Verification, Per Call					1.00											
		Inward Operator Services - Verification and Emergency Interrupt - Per Call					1.95											
		ERATOR CALL PROCESSING																Ē
Fac	cility	based CLEC	<u> </u>			00463		7	7				=					<u> </u>
<u> </u>		Recording of Custom Branded OA Announcement				CBAOS		7,000.00	7,000.00				7.86					
		Loading of Custom Branded OA Announcement per shelf/NAV per OCN	1			CBAOL		500.00	500.00			1	7.86	1				1
UN	NEP C		1				İ	000.00	000.00									<u> </u>
		Recording of Custom Branded OA Announcement						7,000.00	7,000.00				7.86					
		Loading of Custom Branded OA Announcement per shelf/NAV per																1
<u>├</u>			<u> </u>					500.00	500.00				7.86					<u> </u>
Un		ling via OLNS for UNEP CLEC Loading of OA per OCN (Regional)						1,200.00	1,200.00				7.86					+
DIRECTOR		SISTANCE SERVICES	1			1		1,200.00	1,200.00				1.00					<u> </u>
		ORY ASSISTANCE ACCESS SERVICE	1			1	İ											<u> </u>
		Directory Assistance Access Service Calls, Charge Per Call					0.275											
DIF		ORY ASSISTANCE CALL COMPLETION ACCESS SERVICE (E	DACC)															1
		Directory Assistance Call Completion Access Service (DACC),					0.10											1
DIRECTOR		Per Call Attempt SISTANCE SERVICES					0.10											<u> </u>
		ORY ASSISTANCE DATA BASE SERVICE (DADS)																├ ──
		Directory Assistance Data Base Service Charge Per Listing	1			1	0.04											<u> </u>
		Directory Assistance Data Base Service, per month				DBSOF	150.00			<u> </u>								
		RECTORY ASSISTANCE																
Fac	cility	Based CLEC	I	L														
		Recording and Provisioning of DA Custom Branded Announcement	1		AMT	CBADA		6,000.00	6,000.00				7.86					
		· · · · · • • · · • • · · · • • • • • •	1			55.51		0,000.00	0,000.00	1			1.00					·

UNBUNDL	ED NETWORK ELEMENTS - Kentucky												Attachment:	2	Exhi	bit: B	
											Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental	
											Submitted			Charge -	Charge -	Charge -	
CATEGORY	RATE ELEMENTS	Interim	7	BCS	USOC			RATES(\$)			Elec	Manually	Manual Svc	Manual Svc	Manual Svc		
CATEGORY	RATE ELEMENTS	interim	Zone	BCS	0500			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.	
													Electronic-	Electronic-	Electronic-	Electronic-	
													1st	Add'l	Disc 1st	Disc Add'l	
						_	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)			
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	
	Loading of Custom Branded Announcement per Switch			AMT	CBADC		1,170.00	1,170.00				7.86					
UNEP	CLEC																
	Recording of DA Custom Branded Announcement						3,000.00	3,000.00				7.86					
	Loading of DA Custom Branded Announcement per Switch per OCN						1,170.00	1,170.00				7.86					
Unbra	Inding via OLNS for UNEP CLEC						1,170.00	1,170.00				1.00					
	Loading of DA per OCN (1 OCN per Order)						420.00	420.00				7.86					
	Loading of DA per Switch per OCN						16.00	16.00				7.86					
SELECTIVE R																	
	Selective Routing Per Unique Line Class Code Per Request Per																
VIRTUAL COL	Switch				USRCR		93.53	93.53	15.58	15.58		7.86					
VIR I UAL COL	Virtual Collocation - Application Cost	1	+	AMTFS	EAF		2,419,86	2.419.86	1.01	1.01		7.86	<u> </u>			ł	
	Virtual Collocation - Cable Installation Cost, per cable	1	1	AMTES	ESPCX		1,729.11	1,729.11	45.16	45.16		7.86	t			1	
	Virtual Collocation - Floor Space, per sq. ft.			AMTFS	ESPVX	7.99		,									
	Virtual Collocation - Power, per fused amp			AMTFS	ESPAX	8.06											
		1					Т										
\vdash	Virtual Collocation - Cable Support Structure, per entrance cable	<u> </u>		AMTES	ESPSX	17.38						<u> </u>	<u> </u>			l	<u> </u>
				UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U													
				EQ, AMTFS, UDL,													
				UNCVX, UNCDX,													
	Virtual Collocation - 2-wire Cross Connects (loop)			UNCNX	UEAC2	0.0309	24.68	23.68	12.14	10.95		7.86					
				UEA,UHL,UCL,UDL,													
				AMTFS, UAL, UDN,					10.77			=					
	Virtual Collocation - 4-wire Cross Connects (loop)			UNCVX, UNCDX AMTFS,UDL12,	UEAC4	0.0619	24.88	23.82	12.77	11.46		7.86					
				UDLO3, U1T48,													
				U1T12, U1T03,													
				ULDO3, ULD12,													
	Virtual Collocation - 2-Fiber Cross Connects			ULD48, UDF	CNC2F	3.80	41.94	30.51	14.76	11.84		7.86					
				AMTFS,UDL12,													
				UDLO3, U1T48, U1T12, U1T03,													
				ULDO3, ULD12,													
	Virtual Collocation - 4-Fiber Cross Connects				CNC4F	7.59	51.29	39.87	19.41	16.49		7.86					
				USL,ULC,AMTFS,													
		1	1	ULR, UXTD1,													
		1	1	UNC1X, ULDD1,										1			
	Virtual collocation - Special Access & UNE, cross-connect per DS1			U1TD1, USLEL, UNLD1	CNC1X	1.48	44.23	31.98	12.81	11.57						1	
		1	+	UNLD1 USL,ULC,AMTFS,U		1.48	44.23	31.98	12.81	11.57			<u> </u>			ł	
		1	1	E3, U1TD3, UXTS1,													
		1	1	UXTD3, UNC3X,													
				UNCSX, ULDD3,												1	
	Virtual collocation - Special Access & UNE, cross-connect per	1	1	U1TS1, ULDS1,										1			
	DS3	<u> </u>		UDLSX, UNLD3	CND3X	18.89	41.93	30.51	14.75	11.83		<u> </u>	<u> </u>			l	<u> </u>
	Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear foot	1	1	AMTFS	VE1CB	0.003								1			
	Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax		+	AWIT FO	VEIGD	0.003						<u> </u>					
	Cable Support Structure, per linear ft			AMTFS	VE1CD	0.0045										1	
	Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable	1	1									1	İ.				
	Support Structure, per cable			AMTFS	VE1CC		535.55									ļ	
	Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax	1	1											1			
	Cable Support Structure, per cable	 		AMTFS AMTFS	VE1CE	l	535.55	000.01	007.00	007.00			ļ			ł	
┣──┤──	Virtual Collocation Cable Records - per request Virtual Collocation Cable Records - VG/DS0 Cable, per cable	<u> </u>	+	AWITES	VE1BA		1,524.45	980.01	267.02	267.02						ł	
	record			AMTES	VE1BB		656.37	656.37	379.70	379.70						1	
	Virtual Collocation Cable Records - VG/DS0 Cable, per each 100	1	1				500.01	566.51	0.0.10	0.00			<u> </u>			1	
	pair			AMTFS	VE1BC		9.65	9.65	11.84	11.84							
	Virtual Collocation Cable Records -DS1, per T1TIE			AMTFS	VE1BD		4.52	4.52	5.54	5.54							
	Virtual Collocation Cable Records - DS3, per T3TIE			AMTFS	VE1BE		15.81	15.81	19.39	19.39							

UNBUNDL	ED NETWORK ELEMENTS - Kentucky												Attachment: 2	2	Exhi	bit: B	
•											Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental	
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -	
			_								Elec	Manually	Manual Svc	Manual Svc	Manual Svc		
CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.	
													Electronic-	Electronic-	Electronic-	Electronic-	
													1st	Add'l	Disc 1st	Disc Add'l	
							Nonrec	urring	Nonrecurring	Disconnect			055	Rates(\$)			<u> </u>
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	
	Virtual Collocation Cable Records - Fiber Cable, per 99 fiber																
	records			AMTFS	VE1BF		169.63	169.63	154.85	154.85							
	Virtual collocation - Security Escort - Basic, per half hour			AMTFS	SPTBX		33.98	21.53									L
	Virtual collocation - Security Escort - Overtime, per half hour			AMTFS AMTFS	SPTOX		44.26 54.54	27.81 34.09									
	Virtual collocation - Security Escort - Premium, per half hour Virtual collocation - Maintenance in CO - Basic, per half hour			AMTES	SPTPX CTRLX		54.54	21.53	-	-							l
				/	OTTLEX		00.07	21.00									
	Virtual collocation - Maintenance in CO - Overtime, per half hour			AMTFS	SPTOM		73.23	27.81									
	Virtual collocation - Maintenance in CO - Premium per half hour			AMTFS	SPTPM		90.39	34.09									
VIRTUAL COL																	
	Virtual Collocation - 2-wire Cross Connect, Exchange Port 2-Wire			115000	1/5400	0.0000	04.00	00.00	10.11	10.05		7.00					
	Analog - Res Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire			UEPSR	VE1R2	0.0309	24.68	23.68	12.14	10.95	<u> </u>	7.86					1
	Line Side PBX Trunk - Bus		1	UEPSP	VE1R2	0.0309	24.68	23.68	12.14	10.95		7.86					1
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire		1					20.00	2.74	.0.00	1						
	Voice Grade PBX Trunk - Res			UEPSE	VE1R2	0.0309	24.68	23.68	12.14	10.95		7.86					
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire																
	Analog Bus			UEPSB	VE1R2	0.0309	24.68	23.68	12.14	10.95		7.86					<u> </u>
	Virtual Collocation 2-Wire Cross Connect, Exchnage Port 2-Wire ISDN			UEPSX	VE1R2	0.0309	24.68	23.68	12.14	10.95		7.86					
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire			UEPSA	VEIRZ	0.0309	24.00	23.00	12.14	10.95		7.00					
	ISDN			UEPTX	VE1R2	0.0309	24.68	23.68	12.14	10.95		7.86					
	Virtual Collocation 4-Wire Cross Connect, Exchange Port 4-Wire																
	ISDN DS1			UEPEX	VE1R4	1.48	44.23	31.98	12.81	11.57		7.86					
VIRTUAL COL	LOCATION																
										10.05		=					
PHYSICAL CO	Virtual Collocation-2 Wire Cross Connects (Loop) for Line Splitting			UEPSR, UEPSB	VE1LS	0.309	24.68	23.68	12.14	10.95		7.86					
FITSICAL CO	Physical Collocation-2 Wire Cross Connects (Loop) for Line										1						
	Splitting			UEPSR, UEPSB	PE1LS	0.0333	24.68	23.68	12.14	10.95		7.86					
AIN SELECTIV	E CARRIER ROUTING																
	Regional Service Establishment			SRC	SRCEC		193,401.00	193,401.00	9,483.34	9,483.34		7.86					
	End Office Establishment			SRC SRC	SRCEO SRCLP		194.09 2.06	194.09 2.06	0.85	0.85		7.86 7.86					
	Line/Port NRC, per end user Query NRC, per query			SRC	SRCLP	0.0037502	2.06	2.06				7.86					
AIN - BELLSO	UTH AIN SMS ACCESS SERVICE			SKC		0.0037302											
	AIN SMS Access Service - Service Establishment, Per State,																
	Initial Setup			A1N	CAMSE		43.55	43.55	44.93	44.93		7.86					
																	1
┝──┤──	AIN SMS Access Service - Port Connection - Dial/Shared Access			A1N	CAMDP		8.64	8.64	10.03	10.03		7.86					<u> </u>
	AIN SMS Access Service - Port Connection - ISDN Access AIN SMS Access Service - User Identification Codes - Per User			A1N	CAM1P		8.64	8.64	10.03	10.03		7.86					+
	ID Code		1	A1N	CAMAU		38.65	38.65	29.88	29.88		7.86					1
	AIN SMS Access Service - Security Card, Per User ID Code,	1	1		5	1	00.00	00.00	20.00	20.00	t	7.00	1				<u> </u>
	Initial or Replacement			A1N	CAMRC		75.08	75.08	12.93	12.93		7.86					
	AIN SMS Access Service - Storage, Per Unit (100 Kilobytes)					0.0025											
	AIN SMS Access Service - Session, Per Minute					0.666											
	AIN SMS Access Service - Company Performed Session, Per					o 100-											
AIN - RELLSO	Minute UTH AIN TOOLKIT SERVICE		<u> </u>		+	0.4608						<u> </u>					<u> </u>
ANN - DELLOU	AIN Toolkit Service - Service Establishment Charge, Per State,				+							<u> </u>					1
	Initial Setup		1	CAM	BAPSC		43.55	43.55	44.93	44.93		7.86					1
	AIN Toolkit Service - Training Session, Per Customer				BAPVX		8,436.93	8,436.93				7.86					
_	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN,																1
┝──┼──	Term. Attempt				BAPTT	+	8.64	8.64	10.03	10.03	ļ	7.86				L	
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay		1		BAPTD		8.64	8.64	10.03	10.03		7.86					1
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN,				DAFID		0.04	0.04	10.03	10.03		1.00					+
	Off-Hook Immediate		1		BAPTM		8.64	8.64	10.03	10.03		7.86					1
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN,		1			1											
	10-Digit PODP				BAPTO		51.01	51.01	18.50	18.50		7.86					L
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN,		1														1
. 1	CDP	I			BAPTC		51.01	51.01	18.50	18.50	1	7.86					<u> </u>

UNBUNDL	ED NETWORK ELEMENTS - Kentucky												Attachment: 2	2	Exhi	bit: B	
													Incremental	Incremental	Incremental	Incremental	
											Submitted		Charge -	Charge -	Charge -	Charge -	
CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			Elec	Manually	Manual Svc	Manual Svc	Manual Svc		
CATEGORT	RATE ELEMENTS	merm	Zone	603	0300			KATES(\$)			per LSR	per LSR	Order vs. Electronic-	Order vs. Electronic-	Order vs. Electronic-	Order vs. Electronic-	
													1st	Add'l	Disc 1st	Disc Add'l	
									-						0130 130	DISC Add 1	
						Rec	Nonred		Nonrecurring					Rates(\$)			L
┝──┤──	AIN Teellik Centing Trigger Assess Charge Der Trigger Der DN						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	──
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN Feature Code	۹,			BAPTF		51.01	51.01	18.50	18.50		7.86					
	AIN Toolkit Service - Query Charge, Per Query				D/(I II	0.0549207	01.01	01.01	10.00	10.00		7.00					
	AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit																
	Subscription, Per Node, Per Query					0.0066492											L
	AIN Toolkit Service - SCP Storage Charge, Per SMS Access Account, Per 100 Kilobytes					0.07											
├──	ACCOUNT, Per 100 Kilobytes AIN Toolkit Service - Monthly report - Per AIN Toolkit Service					0.07											
	Subscription			CAM	BAPMS	7.87	8.64	8.64	6.08	6.08		7.86					
	AIN Toolkit Service - Special Study - Per AIN Toolkit Service																
	Subscription			CAM	BAPLS	3.26	9.56	9.56				7.86					L
	AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service			0.004	04000	4.70	0.04	0.04	0.00	0.00		7.00					
┝──┼──	Subscription AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit	+		CAM	BAPDS	4.72	8.64	8.64	6.08	6.08		7.86	<u> </u>				├───
1	Service Subscription	1		CAM	BAPES	0.11	9.56	9.56				7.86					1
	EXTENDED LINK (EELs)	1						2.30									
	E: New Density Zone 1 EELs are available in the following MSAs:				ale, FL; Atlan	ta, Ga; New Orl	eans, LA,										\square
	E: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-H																<u> </u>
	E: In all states, EEL network elements shown below also apply to E: In All States the EEL network elements apply to ordinarily comb											to UNES.(N	on-recurring ra	ites do not ap	oly.)		┝───
	RE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 IN				AS IS Charge.) when orderin	ig or unarity con	IDITIEU TIELWOIK	elements, Non	-recurring rates	s uo appiy.						
	First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport		1														
	Combination - Zone 1		1	UNCVX	UEAL2	12.67	125.22	60.48	59.69	7.84		7.86					
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport																
┝──┤──	Combination - Zone 2		2	UNCVX	UEAL2	17.45	125.22	60.48	59.69	7.84		7.86					──
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 3		3	UNCVX	UEAL2	33.22	125.22	60.48	59.69	7.84		7.86					
	Interoffice Transport - Dedicated - DS1 combination - Per Mile per	r	Ŭ	UNUTA	OLALL	00.22	120.22	00.40	00.00	1.04		7.00					
	month			UNC1X	1L5XX	0.19											
	Interoffice Transport - Dedicated - DS1 combination - Facility																
	Termination per month			UNC1X	U1TF1	79.02	181.24	123.53	56.72	22.32		7.86	-				
├ ── ├ ──	DS1 Channelization System Per Month Voice Grade COCI - DS1 To Ds0 Interface - Per Month			UNC1X UNCVX	MQ1 1D1VG	113.33 0.62	57.26 6.71	14.74 4.84	1.86	1.67		7.86 7.86					
	Each Additional 2-Wire VG Loop(SL 2) in the same DS1			UNCVA	IDIVG	0.02	0.71	4.04				7.00					
	Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL2	12.67	125.22	60.48	59.69	7.84		7.86					
	Each Additional 2-Wire VG Loop(SL2) in the same DS1																
	Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL2	17.45	125.22	60.48	59.69	7.84		7.86					
	Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice	е	3			00.00	105.00	00.40	50.00	7.04		7.00					
├── ├──	Transport Combination - Zone 3 Voice Grade COCI - DS1 to DS0 Channel System combination -		3	UNCVX	UEAL2	33.22	125.22	60.48	59.69	7.84		7.86					
	per month			UNCVX	1D1VG	0.62	6.71	4.84				7.86					
	Nonrecurring Currently Combined Network Elements Switch -As-Is	6	Î														
	Charge	<u> </u>		UNC1X	UNCCC		8.98	8.98	11.17	11.17		7.86					L
4-WIF	RE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INT	TEROFFI	CE TR	ANSPORT (EEL)	+												──
1	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 1	1	1	UNCVX	UEAL4	29.26	125.22	60.48	59.69	7.84		7.86					1
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice	1		0017	52/124	23.20	120.22	00.40	00.00	,.04		7.00	1				
	Transport Combination - Zone 2		2	UNCVX	UEAL4	34.25	125.22	60.48	59.69	7.84		7.86					
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice																1
\vdash	Transport Combination - Zone 3		3	UNCVX	UEAL4	85.06	125.22	60.48	59.69	7.84		7.86					<u> </u>
1	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month			UNC1X	1L5XX	0.19											1
	Interoffice Transport - Dedicated - DS1 - Facility Termination Per	1			LOAA	0.19							<u> </u>				
	Month			UNC1X	U1TF1	79.02	181.24	123.53	56.72	22.32		7.86					
	Channelization - Channel System DS1 to DS0 combination Per																
\vdash	Month			UNC1X	MQ1	113.33	57.26	14.74	1.86	1.67		7.86					──
1	Voice Grade COCI - DS1 to DS0 Channel System combination - per month	1			1D1VG	0.60	6.71	4 0 4				7 00					1
	Additional 4-Wire Analog Voice Grade Loop in same DS1	+		UNCVX	UIVG	0.62	6.71	4.84	ł	ł		7.86	<u> </u>				t
		1	ι.		UEAL4	29.26	125.22	60.48	59.69	7.84	1	7.86					1
	Interoffice Transport Combination - Zone 1		1	UNCVX		29.20	123.22										
	Interoffice Transport Combination - Zone 1 Additional 4-Wire Analog Voice Grade Loop in same DS1		1	UNCVX	ULAL4	29.20	125.22	00.40	00.00	1.01							
			1 2		UEAL4	34.25	125.22	60.48	59.69	7.84		7.86					

BUNDLE	D NETWORK ELEMENTS - Kentucky	1	1								0	0	Attachment: 2			oit: B	┢
EGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l	
						Rec	Nonree		Nonrecurring					Rates(\$)	_		┶
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	╇
	Voice Grade COCI - DS1 to DS0 Channel System combination - per month			UNCVX	1D1VG	0.62	6.71	4.84				7.86					
	Nonrecurring Currently Combined Network Elements Switch -As-Is			UNCVA	IDIVG	0.02	0.71	4.04				7.00					+
	Charge			UNC1X	UNCCC		8.98	8.98	11.17	11.17		7.86					
4-WIRE	56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1	INTERO	FFICE	TRANSPORT (EEL)													
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice																
	Transport Combination - Zone 1		1	UNCDX	UDL56	27.59	125.22	60.48	59.69	7.84		7.86					+
	First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2		2	UNCDX	UDL56	32.48	125.22	60.48	59.69	7.84		7.86					
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice		-	CITOD/T	00200	02.10	120.22	00.10	00.00	1.01		1.00					+
	Transport Combination - Zone 3		3	UNCDX	UDL56	36.37	125.22	60.48	59.69	7.84		7.86					
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per																Г
	Month	<u> </u>		UNC1X	1L5XX	0.19											╀
	Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Month	1		UNC1X	U1TF1	79.02	181.24	123.53	56.72	22.32		7.86					1
	Channelization - Channel System DS1 to DS0 combination Per	1	<u> </u>	UNUTA	31111	19.02	101.24	123.33	00.72	22.32		1.00					t
	Month			UNC1X	MQ1	113.33	57.26	14.74	1.86	1.67		7.86					
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per month	1															Γ
	(2.4-64kbs)			UNCDX	1D1DD	1.32	6.71	4.84				7.86					+
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1		1			07.50	105.00	60 40	50.00	7.04		7.00					
_	Interoffice Transport Combination - Zone 1 Additional 4-Wire 56Kbps Digital Grade Loopin same DS1		1	UNCDX	UDL56	27.59	125.22	60.48	59.69	7.84		7.86					╈
	Interoffice Transport Combination - Zone 2		2	UNCDX	UDL56	32.48	125.22	60.48	59.69	7.84		7.86					
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1																t
	Interoffice Transport Combination - Zone 3		3	UNCDX	UDL56	36.37	125.22	60.48	59.69	7.84		7.86					
	OCU-DP COCI (data) - DS1 to DS0 Channel System -																
_	combination per month (2.4-64kbs)			UNCDX	1D1DD	1.32	6.71	4.84				7.86					+
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge			UNC1X	UNCCC		8.98	8.98	11.17	11.17		7.86					
4-WIRE	64 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1	INTERO	FFICE				0.00	0.00	11.17			7.00					t
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice																T
	Transport Combination - Zone 1		1	UNCDX	UDL64	27.59	125.22	60.48	59.69	7.84		7.86					_
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice		0	UNIODY		00.40	105.00	60 40	50.00	7.04		7.00					
_	Transport Combination - Zone 2 First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice		2	UNCDX	UDL64	32.48	125.22	60.48	59.69	7.84		7.86					┿
	Transport Combination - Zone 3		3	UNCDX	UDL64	36.37	125.22	60.48	59.69	7.84		7.86					
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per		Ŭ	011027	00201	00.07	120.22	00.10	00.00	1.01		1.00					Ť
	Month			UNC1X	1L5XX	0.19											
	Interoffice Transport - Dedicated - DS1 combination - Facility																
	Termination Per Month			UNC1X	U1TF1	79.02	181.24	123.53	56.72	22.32		7.86					┢
	Channelization - Channel System DS1 to DS0 combination Per Month	1		UNC1X	MQ1	113.33	57.26	14.74	1.86	1.67		7.86					1
	OCU-DP COCI (data) - DS1 to DS0 Channel System combination	1	+			113.33	51.20	14.74	1.00	1.07		7.00					+
	per month (2.4-64kbs)			UNCDX	1D1DD	1.32	6.71	4.84				7.86					1
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1	1															Г
	Interoffice Transport Combination - Zone 1	 	1	UNCDX	UDL64	27.59	125.22	60.48	59.69	7.84		7.86					╄
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1	1	2			22.40	105.00	60.40	50.60	7.04		7.00					1
	Interoffice Transport Combination - Zone 2 Additional 4-Wire 64Kbps Digital Grade Loopin same DS1	ł	2	UNCDX	UDL64	32.48	125.22	60.48	59.69	7.84		7.86					+
	Interoffice Transport Combination - Zone 3	1	3	UNCDX	UDL64	36.37	125.22	60.48	59.69	7.84		7.86					1
	OCU-DP COCI (data) - DS1 to DS0 Channel System combination	ł			T												T
	per month (2.4-64kbs)			UNCDX	1D1DD	1.32	6.71	4.84				7.86					L
	Nonrecurring Currently Combined Network Elements Switch -As-Is																Г
4-WIP 5	Charge DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTE	POFFIC	E TP/		UNCCC		8.98	8.98	11.17	11.17		7.86					+
4-11116	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice																+
	Transport - Zone 1		1	UNC1X	USLXX	86.47	210.70	114.60	63.96	17.97		7.86					1
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice																T
	Transport - Zone 2	L	2	UNC1X	USLXX	114.10	210.70	114.60	63.96	17.97		7.86					1
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice		_	LINCAY		207 70	210 70	444.00	62.00	47.07		7.00					1
-	Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per		3	UNC1X	USLXX	297.76	210.70	114.60	63.96	17.97		7.86					+
1	Month	1	1	UNC1X	1L5XX	0.19			1	1	1						1

SONDEE	D NETWORK ELEMENTS - Kentucky	1	1	1		1					Suo Orden	Svo Order	Attachment: 2		Exhi		
EGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l	
						Rec	Nonrec		Nonrecurring					Rates(\$)			
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	
	Interoffice Transport - Dedicated - DS1 combination - Facility					70.00		100 50	50 70			7.00					
-	Termination Per Month			UNC1X	U1TF1	79.02	181.24	123.53	56.72	22.32		7.86			<u> </u>		
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge			UNC1X	UNCCC		8.98	8.98	11.17	11.17		7.86					
4-WIRF	DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTE		E TRA		UNCCC		0.30	0.30	11.17	11.17		7.00					
			1														
	First DS1Loop in DS3 Interoffice Transport Combination - Zone 1		1	UNC1X	USLXX	86.47	210.70	114.60	63.96	17.97		7.86					
_																	
	First DS1Loop in DS3 Interoffice Transport Combination - Zone 2		2	UNC1X	USLXX	114.10	210.70	114.60	63.96	17.97		7.86					
	First DS1Loop in DS3 Interoffice Transport Combination - Zone 3		3	UNC1X	USLXX	297.76	210.70	114.60	63.96	17.97		7.86					
	Interoffice Transport - Dedicated - DS3 combination - Per Mile Per														1		
_	Month	I		UNC3X	1L5XX	4.09									 		
	Interoffice Transport - Dedicated - DS3 - Facility Termination per			LINCOV		000.00	050 50	444.50	10.00	00.00		7.00			1		
	month DS3 to DS1 Channel System combination per month		+	UNC3X UNC3X	U1TF3 MQ3	966.89 158.20	350.56 115.48	141.58 56.53	48.00 15.12	23.39 5.30		7.86 7.86			t		
-	DS3 to DS1 Channel System combination per month DS3 Interface Unit (DS1 COCI) combination per month	1	+	UNC3X UNC1X	UC1D1	158.20	6.71	4.84	15.12	5.30		7.86			<u> </u>		
	Additional DS1Loop in DS3 Interoffice Transport Combination -		1		50101	11.00	0.71	4.04				1.00			<u> </u>		
	Zone 1	1	1	UNC1X	USLXX	86.47	210.70	114.60	63.96	17.97	1	7.86			1		
	Additional DS1Loop in DS3 Interoffice Transport Combination -																
	Zone 2		2	UNC1X	USLXX	114.10	210.70	114.60	63.96	17.97		7.86					
	Additional DS1Loop in DS3 Interoffice Transport Combination -																
	Zone 3		3	UNC1X	USLXX	297.76	210.70	114.60	63.96	17.97		7.86					
	DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	11.80	6.71	4.84				7.86					
	Nonrecurring Currently Combined Network Elements Switch -As-Is																
	Charge			UNC3X	UNCCC		8.98	8.98	11.17	11.17		7.86			L		
2-WIRE	VOICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INT	EROFF	ICE TR	ANSPORT (EEL)											<u> </u>		
	2-WireVG Loop used with 2-wire VG Interoffice Transport					10.07	105.00	00.40	50.00	7.04		7.00					
	Combination - Zone 1 2-WireVG Loop used with 2-wire VG Interoffice Transport		1	UNCVX	UEAL2	12.67	125.22	60.48	59.69	7.84		7.86			<u> </u>		
	Combination - Zone 2		2	UNCVX	UEAL2	17.45	125.22	60.48	59.69	7.84		7.86					
	2-WireVG Loop used with 2-wire VG Interoffice Transport		2	UNCVA	ULALZ	17.45	125.22	00.40	33.03	7.04		7.00			<u> </u>		
	Combination - Zone 3		3	UNCVX	UEAL2	33.22	125.22	60.48	59.69	7.84		7.86					
	Interoffice Transport - Dedicated - 2-wire VG combination - Per									-							
	Mile Per Month			UNCVX	1L5XX	0.01											
	Interoffice Transport - Dedicated - 2- Wire Voice Grade																
	combination - Facility Termination per month			UNCVX	U1TV2	23.95	98.09	53.67	56.31	22.42		7.86					
	Nonrecurring Currently Combined Network Elements Switch -As-Is																
	Charge	1		UNCVX	UNCCC		8.98	8.98	11.17	11.17		7.86					
4-WIRE	VOICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INT	EROFF	ICE TR	ANSPORT (EEL)											───		
1	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL4	29.26	125.22	60.48	59.69	7.84		7.86			1		
_	4-WireVG Loop used with 4-wire VG Interoffice Transport	-			JLAL4	23.20	120.22	00.40	59.09	1.04		1.00			<u> </u>		
	Combination - Zone 2		2	UNCVX	UEAL4	34.25	125.22	60.48	59.69	7.84		7.86			1		
	4-WireVG Loop used with 4-wire VG Interoffice Transport	1	<u> </u>	1		020	120.22	00.10	00.00	1.54			1		()		
	Combination - Zone 3	1	3	UNCVX	UEAL4	85.06	125.22	60.48	59.69	7.84	1	7.86			1		
	Interoffice Transport - Dedicated - 4-wire VG combination - Per			1													
	Mile Per Month			UNCVX	1L5XX	0.01											
	Interoffice Transport - Dedicated - 4- Wire Voice Grade														1		
_	combination - Facility Termination per month			UNCVX	U1TV4	21.28	98.09	53.67	56.31	22.42		7.86			 		
	Nonrecurring Currently Combined Network Elements Switch -As-Is		1		110000			0.6-			1	= 0-			1		
062.0	Charge GITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFIC	E TDA	L	UNCVX	UNCCC		8.98	8.98	11.17	11.17		7.86			───		
DS3 DI	High Capacity Unbundled Local Loop - DS3 combination - Per		VOR		1	+	-								t		
1	Mile per month			UNC3X	1L5ND	9.25									1		
1	High Capacity Unbundled Local Loop - DS3 combination - Facility		1	0.1.00/		5.20									<u> </u>		
1	Termination per month			UNC3X	UE3PX	308.31	237.36	147.69	83.43	32.67		7.86			1		
	Interoffice Transport - Dedicated - DS3 - Per Mile per month	1		UNC3X	1L5XX	4.09							l				
	Interoffice Transport - Dedicated - DS3 combination - Facility			1													
	interonnee manaport bedieated boo combination radiity							141.58	48.00	23.39	1	7.86	1		1		
	Termination per per month			UNC3X	U1TF3	966.89	350.56	141.50	40.00	23.33		1.00					
				UNC3X UNC3X	U1TF3	966.89	350.56 8.98	8.98	11.17	11.17		7.86			<u> </u>		
IRONDLE	D NETWORK ELEMENTS - Kentucky			1	-						<u> </u>	<u> </u>	Attachment: 2			bit: B	╄
---------	---	---------	--------	----------------	----------------	--------	--------	----------------	--------------	------------	---	---	--	--	---	---	---
EGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l	
						Rec	Nonree	curring	Nonrecurring	Disconnect				Rates(\$)			
						Ree	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	┶
	High Capacity Unbundled Local Loop - STS1 combination - Per Mile per month			UNCSX	1L5ND	9.25											
	High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per month			UNCSX	UDLS1	320.51	237.36	147.69	83.43	32.67		7.86					
	Interoffice Transport - Dedicated - STS1 combination - Per Mile per month			UNCSX	1L5XX	4.09											
	Interoffice Transport - Dedicated - STS1 combination - Facility Termination per month			UNCSX	U1TFS	945.79	350.56	141.58	48.00	23.39		7.86					
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge			UNCSX	UNCCC		8.98	8.98	11.17	11.17		7.86					
2-WIRE	ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPOR	RT (EEL)														
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1		1	UNCNX	U1L2X	18.44	125.22	60.48	59.69	7.84		7.86					
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination		2														
-	Transport - Zone 2 First 2-Wire ISDN Loop in a DS1 Interoffice Combination		2		U1L2X U1L2X	42.87	125.22	60.48		7.84		7.86					t
	Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile		3	UNCIX UNC1X	1L5XX	42.87	125.22	60.48	59.69	7.84		7.86					╈
	Interoffice Transport - Dedicated - DS1 combination - Facility			UNCIX	ILJAA	0.13											+
	Termination per month Channelization - Channel System DS1 to DS0 combination - per			UNC1X	U1TF1	79.02	181.24	123.53	56.72	22.32		7.86					+
	month 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System			UNC1X	MQ1	113.33	57.26	14.74	1.86	1.67		7.86					+
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport			UNCNX	UC1CA	2.84	6.71	4.84				7.86					_
_	Combination - Zone 1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport		1	UNCNX	U1L2X	18.44	125.22	60.48	59.69	7.84		7.86					_
	Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport		2	UNCNX	U1L2X	25.08	125.22	60.48	59.69	7.84		7.86					_
	Combination - Zone 3 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System		3	UNCNX	U1L2X	42.87	125.22	60.48	59.69	7.84		7.86					+
-	Combination-per month Nonrecurring Currently Combined Network Elements Switch -As-Is			UNCNX	UC1CA	2.84	6.71	4.84				7.86					+
	Charge			UNC1X	UNCCC		8.98	8.98	11.17	11.17		7.86					
4-WIRE	DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 IN	TEROF			0.1000		0.00	0.00				1.00					t
	First DS1 Loop in STS1 Interoffice Transport Combination - Zone																t
	1 First DS1 Loop in STS1 Interoffice Transport Combination - Zone		1	UNC1X	USLXX	86.47	210.70	114.60	63.96	17.97		7.86					┝
	2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone		2	UNC1X	USLXX	114.10	210.70	114.60	63.96	17.97		7.86					╀
+	3 Interoffice Transport - Dedicated - STS1 combination - Per Mile		3	UNC1X	USLXX	297.76	210.70	114.60	63.96	17.97		7.86					┝
+	Per Month Interoffice Transport - Dedicated - STS1 combination - Facility			UNCSX	1L5XX	4.09											╀
	Termination			UNCSX	U1TFS	945.79	350.56	141.58	48.00	23.39		7.86					
_	STS1 to DS1 Channel System conbination per month			UNCSX	MQ3	158.20	115.48	56.53	15.12	5.30		7.86					Ļ
	DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination -			UNC1X	UC1D1	11.80	6.71	4.84		47.07		7.86					┢
	Zone 1 Additional DS1Loop in STS1 Interoffice Transport Combination -		1	UNC1X	USLXX	86.47	210.70	114.60		17.97		7.86					t
	Zone 2 Additional DS1Loop in STS1 Interoffice Transport Combination -		2	UNC1X	USLXX	114.10	210.70	114.60	63.96	17.97		7.86					t
+	Zone 3 DS3 Interface Unit (DS1 COCI) combination per month		3	UNC1X UNC1X	USLXX UC1D1	297.76	210.70	114.60 4.84	63.96	17.97		7.86					╋
	Nonrecurring Currently Combined Network Elements Switch -As-Is		1							1	1		1	1		1	t
4.\//ID	Charge 56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROF		RANSP		UNCCC	ł	8.98	8.98	11.17	11.17		7.86					╀
4-1116	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport		INAN3P			1											+
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport 4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport		1	UNCDX	UDL56	27.59	125.22	60.48	59.69	7.84		7.86					╞
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport 4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport		2	UNCDX	UDL56	32.48	125.22	60.48	59.69	7.84		7.86					╞
1	Combination - Zone 3		3	UNCDX	UDL56	36.37	125.22	60.48	59.69	7.84		7.86					1

UNBUNDLE	ED NETWORK ELEMENTS - Kentucky												Attachment: 2	2	Exhi	bit: B	[
			[Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental	
CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			Submitted Elec per LSR	Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge - Manual Svc Order vs. Electronic- Disc Add'l	
						-	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)			<u> </u>
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Per Mile			UNCDX	1L5XX	0.01											
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Termination			UNCDX	U1TD5	17.25	98.09	53.67	56.31	22.42		7.86					<u> </u>
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge			UNCDX	UNCCC		8.98	8,98	11.17	11.17		7.86					1
4-WIR	E 64 KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTERO	FFICE T	RANSF		011000		0.00	0.00				7.00					<u> </u>
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport																
	Combination - Zone 1 4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport			UNCDX	UDL64	27.59	125.22	60.48	59.69	7.84		7.86					<u> </u>
	Combination - Zone 2		2	UNCDX	UDL64	32.48	125.22	60.48	59.69	7.84		7.86					<u> </u>
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 3		3	UNCDX	UDL64	36.37	125.22	60.48	59.69	7.84		7.86					
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per Mile			UNCDX	1L5XX	0.01											┝──
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility Termination			UNCDX	U1TD6	17.25	98.09	53.67	56.31	22.42		7.86					┝──
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge			UNCDX	UNCCC		8.98	8.98	11.17	11.17		7.86					
	IETWORK ELEMENTS used as a part of a currently combined facility, the non-recurrng	charges	do not	apply but a Switch	As Is charge	does annly											┝───
When	used as ordinarily combined network elements in All States, the	non-recu	Irring c	harges apply and the	e Switch As la	s Charge does n	ot.										
Nonre	curring Currently Combined Network Elements "Switch As Is" Ch	harge (O	ne app	lies to each combina	ition)												
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge - 2 wire/4-Wire VG			UNCVX	UNCCC		8.98	8.98	11.17	11.17		7.86					
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge - 56/64 kbps			UNCDX	UNCCC		8.98	8.98	11.17	11.17		7.86					ĺ
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge - DS1			UNC1X	UNCCC		8.98	8.98	11.17	11.17		7.86					1
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge - DS3			UNC3X	UNCCC		8.98	8.98	11.17	11.17		7.86					1
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge - STS1			UNCSX	UNCCC		8.98	8.98	11.17	11.17		7.86					
NOTE	Local Channel - Dedicated Transport - minimum billing period -	Below D	S3=on														L
	Local Channel - Dedicated - 2-Wire Voice Grade Local Channel - Dedicated - 4-Wire Voice Grade			UNCXV UNCXV	ULDV2 ULDV4	18.57 19.86	265.78 266.48	46.96 47.65	46.79 47.54	4.98 5.73		7.86 7.86					
	Local Channel - Dedicated - 4-wire Voice Grade		1	UNCXV UNC1X	ULDV4	40.46	266.48	47.65	47.54 30.21	21.07		7.86				-	I
	Local Channel - Dedicated - DS1 Per Month Zone 1		2	UNC1X	ULDF1	43.39	209.60	176.51	30.21	21.07		7.86					
	Local Channel - Dedicated - DS1- Per Month Zone 3		3	UNC1X	ULDF1	164.50	209.60	176.51	30.21	21.07		7.86					
	Local Channel - Dedicated - DS3 - Per Mile per month Local Channel - Dedicated - DS3 - Facility Termination			UNC3X UNC3X	1L5NC ULDF3	8.74 576.05	551.38	338.08	173.00	120.42		7.86					
	Local Channel - Dedicated - STS-1- Per Mile per month			UNCSX	1L5NC	8.74											
MUT	Local Channel - Dedicated - STS-1 - Facility Termination PLEXERS			UNCSX	ULDFS	543.24	551.38	338.08	173.00	120.42		7.86				L	┝───
WULT	Channelization - DS1 to DS0 Channel System			UXTD1	MQ1	113.33	101.40	71.60	13.79	13.04		7.86					<u> </u>
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kbs)			UDL	1D1DD	1.32	10.07	7.08	10.10	10.01		7.86					
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per month	1		UDN	UC1CA	2.84	10.07	7.08				7.86					(
	Voice Grade COCI - DS1 to DS0 Channel System - per month	1	1	UEA	1D1VG	0.6228	10.07	7.08				7.86					<u> </u>
	DS3 to DS1 Channel System per month			UXTD3	MQ3	158.20	199.23	118.62	50.16	48.59		7.86					
	STS1 to DS1 Channel System per month			UXTS1	MQ3	158.20	199.23	118.62	50.16	48.59		7.86					Ļ
	DS3 Interface Unit (DS1 COCI) used with Loop per month			USL	UC1D1	11.80	10.07	7.08				7.86					
	DS3 Interface Unit (DS1 COCI) used with Local Channel per month DS3 Interface Unit (DS1 COCI) used with Interoffice Channel per	1		ULDD1	UC1D1	11.80	10.07	7.08				7.86					
Sub L	month sop Feeder		<u> </u>	U1TD1	UC1D1	11.80	10.07	7.08				7.86					I
Sub-Li	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Statewide	1	SW	UNC1X	USBFG	1											1
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Statewide	1		UNC1X	USBFG	62.57	125.43	73.68	81.82	21.56							1
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 2		2	UNC1X	USBFG	87.71	125.43	73.68	81.82	21.56							
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 3		3	UNC1X	USBFG	273.33	125.43	73.68	81.82	21.56							L
NIDUN: 51 5-	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 4	<u> </u>	4	UNC1X	USBFG	1						<u> </u>					<u> </u>
JNBUNDLED	LOCAL EXCHANGE SWITCHING(PORTS)			l	1	1			l		L	l				l	ł

UNBUNDLE	D NETWORK ELEMENTS - Kentucky												Attachment: 2	2	Exhi	bit: B	
											Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental	
CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			Submitted Elec per LSR		Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge - Manual Svc Order vs. Electronic- Disc Add'l	
						Rec	Nonree		Nonrecurring					Rates(\$)			
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	L
	nge Ports Although the Port Rate includes all available features in GA, KY		44.0	desired feetures will a		daved uplane set											┝───
	VOICE GRADE LINE PORT RATES (RES)	, LA α Π	N, LIE			dered using ret											<u> </u>
	Exchange Ports - 2-Wire Analog Line Port- Res.			UEPSR	UEPRL	1.49	3.74	3.63	2.23	2.13		7.86					
	Exchange Ports - 2-Wire Analog Line Port with Caller ID - Res.			UEPSR	UEPRC	1.49	3.74	3.63	2.23	2.13		7.86					┝───
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Res.			UEPSR	UEPRO	1.49	3.74	3.63	2.23	2.13		7.86					1
	Exchange Ports - 2-Wire VG unbundled KY extended local dialing				-												
	parity Port with Caller ID - Res.			UEPSR	UEPRM	1.49	3.74	3.63	2.23	2.13		7.86					
	Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID (LUM)			UEPSR	UEPAP	1.49	3.74	3.63	2.23	2.13		7.86					1
-+-	Exchange Ports - 2-Wire Voice Kentucky Residence Dialing Plan	+				1.49	3.74	3.03	2.23	2.13		1.00					<u> </u>
	without Caller ID			UEPSR	UEPWE	1.49	3.74	3.63	2.23	2.13		7.86					L
	2-Wire voice unbundled Low Usage Line Port without Caller ID											= 0 -					1
<u> </u>	Capability Subsequent Activity	+		UEPSR UEPSR	UEPRT USASC	1.49 0.00	3.74 0.00	3.63	2.23	2.13		7.86 7.86					i
FEATU				02.00	00/100	0.00	0.00	0.00				7.00					<u> </u>
	All Available Vertical Features			UEPSR	UEPVF	0.00	0.00	0.00				7.86					
2-WIRE	VOICE GRADE LINE PORT RATES (BUS)																└───
	Exchange Ports - 2-Wire Analog Line Port without Caller ID - Bus			UEPSB	UEPBL	1.49	3.74	3.63	2.23	2.13		7.86					1
	Exchange Ports - 2-Wire VG unbundled Line Port without Caller ID - Bus			UEF3B	UEFBL	1.49	3.74	3.03	2.23	2.13		7.00					
	port with Caller+E484 ID - Bus.			UEPSB	UEPBC	1.49	3.74	3.63	2.23	2.13		7.86					1
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Bus.			UEPSB	UEPBO	1.49	3.74	3.63	2.23	2.13		7.86					┢────
	Exchange Ports - 2-Wire VG unbundled KY extended local dialing parity Port with Caller ID - Bus.			UEPSB	UEPBM	1.49	3.74	3.63	2.23	2.13		7.86					1
	Exhange Ports - 2-Wire VG unbundled incoming only port with			02.05	OL: Dill		0.11	0.00	LIEG	2.10		1.00					
	Caller ID - Bus			UEPSB	UEPB1	1.49	3.74	3.63	2.23	2.13		7.86					L
	Exchange Ports - 2-Wire Voice Kentucky Business Dialing Plan					1.40	0.74	2.62	2.02	0.10		7.00					1
	without Caller ID 2-Wire voice unbundled Incoming Only Port without Caller ID			UEPSB	UEPWF	1.49	3.74	3.63	2.23	2.13		7.86					
	Capability			UEPSB	UEPBE	1.49	3.74	3.63	2.23	2.13		7.86					1
	Subsequent Activity			UEPSB	USASC	0.00	0.00	0.00				7.86					
FEATU				115000								=					I
EXCH/	All Available Vertical Features NGE PORT RATES (DID & PBX)			UEPSB	UEPVF	0.00	0.00	0.00				7.86					
EXONA	2-Wire VG Unbundled 2-Way PBX Trunk - Res			UEPSE	UEPRD	1.49	39.05	18.17	15.38	0.89		7.86					<u> </u>
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus			UEPSP	UEPPC	1.49	39.05	18.17	15.38	0.89		7.86					
	2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus			UEPSP	UEPPO	1.49	39.05	18.17	15.38	0.89		7.86					I
	2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus 2-Wire Analog Long Distance Terminal PBX Trunk - Bus	+		UEPSP UEPSP	UEPP1 UEPLD	1.49 1.49	39.05 39.05	18.17 18.17	15.38 15.38	0.89		7.86 7.86					
	2-Wire Voice Unbundled PBX LD Terminal Ports	1		UEPSP	UEPLD	1.49	39.05	18.17	15.38	0.89		7.86					<u> </u>
	2-Wire Vice Unbundled 2-Way PBX Usage Port			UEPSP	UEPXA	1.49	39.05	18.17	15.38	0.89		7.86					
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPSP	UEPXB	1.49	39.05	18.17	15.38	0.89		7.86					
	2-Wire Voice Unbundled PBX LD DDD Terminals Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port	-		UEPSP UEPSP	UEPXC UEPXD	1.49 1.49	39.05 39.05	18.17 18.17	15.38 15.38	0.89		7.86 7.86					
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD			UEFOP	UPAD	1.49	39.05	10.17	10.38	0.89		1.80					<u> </u>
	Capable Port			UEPSP	UEPXE	1.49	39.05	18.17	15.38	0.89		7.86					<u> </u>
	2-Wire Voice Unbundled 2-Way PBX Kentucky Room Area Calling	1															[
	Port Without LUD		<u> </u>	UEPSP UEPSP	UEPXF UEPXG	1.49	39.05 39.05	18.17 18.17	15.38 15.38	0.89		7.86 7.86					┝───
	2-Wire Voice Unbundled PBX Kentucky LUD Area Calling Port 2-Wire Voice Unbundled PBX Kentucky Premium Callling Port	1		UEPSP	UEPXG	1.49		18.17	15.38	0.89		7.86					<u> </u>
	2-Wire Voice Unbundled 2-Way PBX Kentucky Area Calling Port Without LUD	1		UEPSP	UEPXJ	1.49		18.17	15.38	0.89		7.86					
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy						00.00	10.17	.0.00								(
	Administrative Calling Port			UEPSP	UEPXL	1.49	39.05	18.17	15.38	0.89		7.86					L
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy				UEPXM	1.40	20.05	18.17	45.00	0.00		7.00					i –
	Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital	+		UEPSP	UEPXM	1.49	39.05	18.17	15.38	0.89		7.86					
1	Discount Room Calling Port			UEPSP	UEPXO	1.49	39.05	18.17	15.38	0.89		7.86					i -
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPSP	UEPXS	1.49	39.05	18.17	15.38	0.89		7.86					
	Subsequent Activity	1		UEPSP	USASC	0.00	0.00	0.00				7.86					

UNBUNDL	ED NETWORK ELEMENTS - Kentucky												Attachment:	2	Exhi	bit: B	(
CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	1	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l	
							Nonree	urring	Nonrecurring	Disconnect			089	Rates(\$)			<u> </u>
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	<u> </u>
FEAT	URES																
	All Available Vertical Features			UEPSP UEPSE	UEPVF	0.00	0.00	0.00				7.86					
EXCH	IANGE PORT RATES (COIN)	-				4.40	0.74	0.00	0.00	2.13		7.00					┣───
Local	Exchange Ports - Coin Port Switching Features offered with Port					1.49	3.74	3.63	2.23	2.13		7.86			-	-	├───
	E: Transmission/usage charges associated with POTS circuit swi	itched u	isade w	ill also apply to circu	it switched ve	oice and/or circu	uit switched dat	a transmission	by B-Channels	s associated wit	h 2-wire ISD	N ports.					<u> </u>
NOTE	E: Access to B Channel or D Channel Packet capabilities will be a	vailable	only th	rough BFR/New Bus	iness Reque	st Process. Rat	es for the pack	et capabilities w	vill be determin	ed via the Bona	Fide Reque	st/New Bus	siness Reques	t Process.			
	Exchange port - 4-wire ISDN trunk port -all available features																1
	included I LOCAL EXCHANGE SWITCHING(PORTS)	-			UEPEX	101.60	188.36	95.15	61.92	22.67		7.86					┝───
	IANGE PORT RATES																<u> </u>
	Exchange Ports - 2-Wire DID Port			UEPEX	UEPP2	10.51	92.18	15.82	52.16	5.30		7.86					
	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID																(
	capability	ļ		UEPDD	UEPDD	74.77	164.86	77.74	60.69		L	7.86					
-+-	Exchange Ports - 2-Wire ISDN Port (See Notes below.) All Features Offered			UEPTX UEPSX UEPTX UEPSX	U1PMA UEPVF	13.46	60.60 0.00	50.67 0.00	32.83	14.17		7.86					<u> </u>
NOTE	Transmission/usage charges associated with POTS circuit swi	itched u	isade w						by B-Channels	s associated wit	h 2-wire ISC	N ports.					
	E: Access to B Channel or D Channel Packet capabilities will be a			rough BFR/New Bus	iness Reque								iness Reques	t Process.			
	Exchange Ports - 2-Wire ISDN Port Channel Profiles			UEPTX UEPSX	U1UMA	0.00	0.00	0.00									
	Exchange Ports - 4-Wire ISDN DS1 Port			UEPEX	UEPEX	101.60	188.36	95.15	61.92	22.67		7.86					
	JNDLED PORT with REMOTE CALL FORWARDING CAPABILITY JNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE																
UNBO	Unbundled Remote Call Forwarding Service - Residence			UEPVR	UERAC	1.49	3.74	3.63				7.86					
				02. 11	021010		0.11	0.00				1.00					
	Unbundled Remote Call Forwarding Service, Local Calling - Res			UEPVR	UERLC	1.49	3.74	3.63				7.86					
	Unbundled Remote Call Forwarding Service, InterLATA - Res			UEPVR	UERTE	1.49	3.74	3.63				7.86					
Non F	Unbundled Remote Call Forwarding Service, IntraLATA - Res	-		UEPVR	UERTR	1.49	3.74	3.63				7.86					
NON-R	Recurring Unbundled Remote Call Forwarding Service - Conversion - Switch-																
	as-is			UEPVR	USAC2		0.10	0.10				7.86					1
	Unbundled Remote Call Forwarding Service - Conversion with																
	allowed change (PIC and LPIC)			UEPVR	USACC		0.10	0.10									┝───
UNBU	JNDLED REMOTE CALL FORWARDING - Bus	-															
1	Unbundled Remote Call Forwarding Service, Area Calling - Bus			UEPVB	UERAC	1.49	3.74	3.63				7.86					1
	onbundiba Romoto Cain Fornarding Corrico, rica Caining Bao			02.10	021010		0.11	0.00				1.00					
	Unbundled Remote Call Forwarding Service, Local Calling - Bus			UEPVB	UERLC	1.49	3.74	3.63				7.86					
	Unbundled Remote Call Forwarding Service, InterLATA - Bus			UEPVB	UERTE	1.49	3.74	3.63				7.86					I
	Unbundled Remote Call Forwarding Service, IntraLATA - Bus Unbundled Remote Call Forwarding Service Expanded and	-		UEPVB	UERTR	1.49	3.74	3.63				7.86					I
1	Exception Local Calling			UEPVB	UERVJ	1.49	3.74	3.63				7.86					1
Non-R	Recurring			02110	021110		0.11	0.00				1.00					
	Unbundled Remote Call Forwarding Service - Conversion - Switch-																1
 	as-is	<u> </u>	-	UEPVB	USAC2		0.10	0.10				7.86					<u> </u>
	Unbundled Remote Call Forwarding Service - Conversion with allowed change (PIC and LPIC)	1	1	UEPVB	USACC		0.10	0.10		1							l l
UNBUNDLED					00/100		0.10	0.10									<u> </u>
	Office Switching (Port Usage)																
	End Office Switching Function, Per MOU					0.0011971											L
Tende	End Office Trunk Port - Shared, Per MOU em Switching (Port Usage) (Local or Access Tandem)				l	0.0002112					ļ	ļ			L	L	
	Tandem Switching Function Per MOU		1	1	1	0.000194				1							<u> </u>
$ \longrightarrow $	Tandem Trunk Port - Shared, Per MOU	1	1	ł	1	0.0002416				1							
۱ <u> </u>	non Transport																
Comm		1	1			0.000003											──
Comm	Common Transport - Per Mile, Per MOU	1							1	1							I
	Common Transport - Facilities Termination Per MOU					0.0007466											
UNBUNDLED	Common Transport - Facilities Termination Per MOU PORT/LOOP COMBINATIONS - COST BASED RATES	or State	e Comm	hission rule to provide	e Unbundled		or Switch Por	s.									
UNBUNDLED Cost E	Common Transport - Facilities Termination Per MOU					Local Switching			section of this I	Rate Exhibit.							<u> </u>
UNBUNDLED Cost E Featur End O	Common Transport - Facilities Termination Per MOU PORT/LOOP COMBINATIONS - COST BASED RATES Based Rates are applied where BellSouth is required by FCC and/ res shall apply to the Unbundled Port/Loop Combination - Cost Ba Office and Tandem Switching Usage and Common Transport Usa	ased Ra ge rates	ite secti s in the	ion in the same mann Port section of this ra	er as they are	Local Switching e applied to the all apply to all co	Stand-Alone Ur ombinations of	bundled Port s	ork elements ex	cept for UNE C							
UNBUNDLED Cost E Featur End O The fir	Common Transport - Facilities Termination Per MOU PORT/LOOP COMBINATIONS - COST BASED RATES Based Rates are applied where BellSouth is required by FCC and/ res shall apply to the Unbundled Port/Loop Combination - Cost B- Office and Tandem Switching Usage and Common Transport Usage rst and additional Port nonrecurring charges apply to Not Current	ased Ra ge rates	ite secti s in the	ion in the same mann Port section of this ra	er as they are	Local Switching e applied to the all apply to all co	Stand-Alone Ur ombinations of	bundled Port s	ork elements ex	cept for UNE C							
UNBUNDLED Cost E Featur End O The fir 2-WIR	Common Transport - Facilities Termination Per MOU PORT/LOOP COMBINATIONS - COST BASED RATES Based Rates are applied where BellSouth is required by FCC and/ res shall apply to the Unbundled Port/Loop Combination - Cost Ba Office and Tandem Switching Usage and Common Transport Usa	ased Ra ge rates	ite secti s in the	ion in the same mann Port section of this ra	er as they are	Local Switching e applied to the all apply to all co	Stand-Alone Ur ombinations of	bundled Port s	ork elements ex	cept for UNE C							

INBUNDLE	D NETWORK ELEMENTS - Kentucky	T		1		1							Attachment: 2		Exhi		—
ATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l	
						Rec	Nonrec		Nonrecurring					Rates(\$)			
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	
	2-Wire VG Loop/Port Combo - Zone 2		2			15.52											<u> </u>
	2-Wire VG Loop/Port Combo - Zone 3		3		-	31.74											<u> </u>
UNE LO	op Rates 2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPRX	UEPLX	9.64											
	2-Wire Voice Grade Loop (SL1) - Zone 1		2	UEPRX	UEPLX	14.37											
	2-Wire Voice Grade Loop (SL1) - Zone 3			UEPRX	UEPLX	30.59											1
2-Wire	/oice Grade Line Port Rates (Res)																
	2-Wire voice unbundled port - residence			UEPRX	UEPRL	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire voice unbundled port with Caller ID - res			UEPRX UEPRX	UEPRC UEPRO	1.15 1.15	21.29 21.29	15.49 15.49	2.85 2.85	2.67 2.67		7.86 7.86					
	2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade unbundled Kentucky extended local dialing			UEPKA	UEPRO	1.15	21.29	15.49	2.00	2.07		7.00					+
	parity port with Caller ID - res			UEPRX	UEPRM	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire voice unbundles res, low usage line port with Caller ID						20		2.00								1
	(LUM)			UEPRX	UEPAP	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Unbundled Kentucky Residence Dialing Plan without																1
	Caller ID			UEPRX	UEPWE	1.15	21.29	15.49	2.85	2.67		7.86	L				—
	2-Wire voice unbundled Low Usage Line Port without Caller ID Capability			UEPRX	UEPRT	1.15	21.29	15.49	2.85	2.67		7.86					
FEATU		1			JEINI	1.15	21.29	10.49	2.00	2.07		1.00					+
	All Features Offered		1	UEPRX	UEPVF	0.00	0.00	0.00				7.86					1
LOCAL	NUMBER PORTABILITY																
	Local Number Portability (1 per port)			UEPRX	LNPCX	0.35											
NONRE	CURRING CHARGES (NRCs) - CURRENTLY COMBINED																
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -				110.4.00		0.40	0.40				7.00					
	Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Conversion -			UEPRX	USAC2		0.10	0.10				7.86					-
	Switch with change			UEPRX	USACC		0.10	0.10				7.86					
ADDITI	DNAL NRCs																1
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent																
	Activity			UEPRX	USAS2	0.00	0.00	0.00				7.86					
	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS) rt/Loop Combination Rates																—
	2-Wire VG Loop/Port Combo - Zone 1		1			10.79											
	2-Wire VG Loop/Port Combo - Zone 2		2			15.52											-
	2-Wire VG Loop/Port Combo - Zone 3		3			31.74											
UNE Lo	op Rates																
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPBX	UEPLX	9.64											
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPBX UEPBX	UEPLX UEPLX	14.37 30.59											
2-Wire	2-Wire Voice Grade Loop (SL1) - Zone 3 /oice Grade Line Port (Bus)	1	3	ULFDA	UEFLA	30.59											┼──
2 1110	2-Wire voice unbundled port without Caller ID - bus			UEPBX	UEPBL	1.15	21.29	15.49	2.85	2.67		7.86					1
	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPBX	UEPBC	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire voice unbundled port outgoing only - bus			UEPBX	UEPBO	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire voice Grade unbundled Kentucky extended local dialing						o., o-		o o-			=					1
	parity port with Caller ID - bus 2-Wire voice unbundled incoming only port with Caller ID - Bus	+		UEPBX UEPBX	UEPBM UPEB1	1.15 1.15	21.29 21.29	15.49 15.49	2.85 2.85	2.67 2.67		7.86 7.86					
	2-Wire Voice unbundled incoming only port with Caller ID - Bus 2-Wire Voice Unbundled Kentucky Business Dialing Plan without	1		ULFDA	UFEDI	1.15	21.29	15.49	2.65	2.07		1.00					+
	Caller ID			UEPBX	UEPWF	1.15	21.29	15.49	2.85	2.67		7.86					1
	2-Wire voice unbundled Incoming Only Port without Caller ID	1															1
	Capability			UEPBX	UEPBE	1.15	21.29	15.49	2.85	2.67		7.86					
				UEDDY	L NDC::												\bot
	Local Number Portability (1 per port)		ļ	UEPBX	LNPCX	0.35							L				
FEATU	RES All Features Offered			UEPBX	UEPVF	0.00	0.00	0.00				7.86					+
	CURRING CHARGES (NRCs) - CURRENTLY COMBINED	1			J	0.00	0.00	0.00		1		7.00					1
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -		[1												1
	Switch-as-is			UEPBX	USAC2		0.10	0.10				7.86					
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -																
4007	Switch with change			UEPBX	USACC		0.10	0.10				7.86					
ADDITI	DNAL NRCs 2-Wire Voice Grade Loop/Line Port Combination - Subsequent	+			-				1								
	2-wire voice Grade Loop/Line Port Combination - Subsequent Activity			UEPBX	USAS2		0.00	0.00				7.86					1
2-WIRF	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)	1		021 0/	30,102		0.00	0.00				1.00					\vdash
	rt/Loop Combination Rates	1		1	+												+

JNBUNDLE	ED NETWORK ELEMENTS - Kentucky				-						-		Attachment: 2			bit: B	
												Svc Order	Incremental	Incremental	Incremental	Incremental	
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -	
			_								Elec	Manually	Manual Svc	Manual Svc	Manual Svc		
ATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.	
													Electronic-	Electronic-	Electronic-	Electronic-	
													1st	Add'l	Disc 1st	Disc Add'l	
						Rec	Nonrec		Nonrecurring			_		Rates(\$)			
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	
	2-Wire VG Loop/Port Combo - Zone 1		1			10.79											
	2-Wire VG Loop/Port Combo - Zone 2		2			15.52											
	2-Wire VG Loop/Port Combo - Zone 3		3		_	31.74											
UNEL	oop Rates			UEPRG	UEPLX	9.64											
	2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2	-	1	UEPRG	UEPLX	9.64											
		-		UEPRG	UEPLX	30.59											
0.14/1	2-Wire Voice Grade Loop (SL 1) - Zone 3 Voice Grade Line Port Rates (RES - PBX)		3	UEPRG	UEPLX	30.59											
2-wire	Voice Grade Line Port Rates (RES - PBX)	-															
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Res			UEPRG	UEPRD	1.15	21.29	15.49	2.85	2.67		7.86					
1.004	- NUMBER PORTABILITY			UEPRG	UEPRD	1.15	21.29	15.49	2.60	2.07		7.00					
LUCAL	Local Number Portability (1 per port)			UEPRG	LNPCP	3.15	0.00	0.00				7.86					
FEATU				UEPKG	LINPUP	3.15	0.00	0.00				7.86					
FEATU	All Features Offered			UEPRG	UEPVF	0.00	0.00	0.00				7.86					<u> </u>
NOND	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED			UEPRG	UEPVF	0.00	0.00	0.00				1.00					<u> </u>
NONRI	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -	+			+	+ +			l								
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As-Is	1		UEPRG	USAC2		8.45	1.91				7.86					
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -	+		ULFING	03402	+ +	0.45	1.91	l			00.1					
	Conversion - Switch with Change	1		UEPRG	USACC		8.45	1.91				7.86					
דיחתא	IONAL NRCs	+	<u> </u>	ULING	USACC	<u>↓ </u>	0.40	1.91				1.00					
ADDIT	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -	+			+	<u>↓ </u>											<u> </u>
	Subsequent Activity			UEPRG	USAS2	0.00	0.00	0.00				7.86					
	Subsequent Activity			UEFKG	03832	0.00	0.00	0.00				7.00					
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group						7.86	7.86				7.86					
2 WIDE	E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)	,				+ +	7.00	7.00				7.00					
	ort/Loop Combination Rates					+ +											
UNEF	2-Wire VG Loop/Port Combo - Zone 1		1		-	10.79											
	2-Wire VG Loop/Port Combo - Zone 1		2			15.52											
	2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3	-	3			31.74											
	oop Rates	-	3			31.74											
	2-Wire Voice Grade Loop (SL 1) - Zone 1	-	1	UEPPX	UEPLX	9.64											
	2-Wire Voice Grade Loop (SL 1) - Zone 2	-	2	UEPPX	UEPLX	14.37											
	2-Wire Voice Grade Loop (SL 1) - Zone 2	-	3	UEPPX	UEPLX	30.59											
2-Wire	Voice Grade Line Port Rates (BUS - PBX)		Ŭ	0ETTX	OLIEX	00.00											
2-1116	Voice Grade Line Fort Rates (DOG - FDX)																
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPPX	UEPPC	1.15	21.29	15,49	2.85	2.67		7.86					
	Line Side Unbuildied Outward PBX Trunk Port - Bus			UEPPX	UEPPO	1.15	21.29	15.49	2.85	2.67		7.86					
	Line Side Unbuilded Incoming PBX Trunk Port - Bus			UEPPX	UEPP1	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPPX	UEPLD	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Unbuildied PBX LD Terminal Ports 2-Wire Voice Unbuildied 2-Way Combination PBX Usage Port			UEPPX	UEPXA	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Unbuildied 2-Way Combination PBX Osage Port 2-Wire Voice Unbuildied PBX Toll Terminal Hotel Ports	1	1	UEPPX	UEPXB	1.15	21.29	15.49	2.85	2.67		7.86					1
		1	l	UEPPX	UEPXC		21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Unbundled PBX LD DDD Terminals Port																
	2-Wire Voice Unbundled PBX LD DDD Terminals Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port					1.15						7 86					
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPPX	UEPXC	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD			UEPPX	UEPXD	1.15	21.29	15.49	2.85	2.67							
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port											7.86 7.86					
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port 2-Wire Voice Unbundled 2-Way PBX Kentucky Room Area Calling			UEPPX UEPPX	UEPXD UEPXE	1.15 1.15	21.29 21.29	15.49 15.49	2.85 2.85	2.67 2.67		7.86					
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port 2-Wire Voice Unbundled 2-Way PBX Kentucky Room Area Calling Port without LUD			UEPPX UEPPX UEPPX	UEPXD UEPXE UEPXF	1.15 1.15 1.15	21.29 21.29 21.29	15.49 15.49 15.49	2.85 2.85 2.85	2.67 2.67 2.67		7.86 7.86					
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port 2-Wire Voice Unbundled 2-Way PBX Kentucky Room Area Calling Port without LUD 2-Wire Voice Unbundled PBX Kentucky LUD Area Calling Port			UEPPX UEPPX UEPPX UEPPX	UEPXD UEPXE UEPXF UEPXG	1.15 1.15 1.15 1.15 1.15	21.29 21.29 21.29 21.29 21.29	15.49 15.49 15.49 15.49	2.85 2.85 2.85 2.85	2.67 2.67 2.67 2.67		7.86 7.86 7.86					
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port 2-Wire Voice Unbundled 2-Way PBX Kentucky Room Area Calling Port without LUD 2-Wire Voice Unbundled PBX Kentucky LUD Area Calling Port 2-Wire Voice Unbundled PBX Kentucky Premium Calling Port 2-Wire Voice Unbundled PBX Kentucky Premium Calling Port			UEPPX UEPPX UEPPX	UEPXD UEPXE UEPXF	1.15 1.15 1.15	21.29 21.29 21.29	15.49 15.49 15.49	2.85 2.85 2.85	2.67 2.67 2.67		7.86 7.86					
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port 2-Wire Voice Unbundled 2-Way PBX Kentucky Room Area Calling Port without LUD 2-Wire Voice Unbundled PBX Kentucky LUD Area Calling Port			UEPPX UEPPX UEPPX UEPPX UEPPX	UEPXD UEPXE UEPXF UEPXG UEPXH	1.15 1.15 1.15 1.15 1.15 1.15	21.29 21.29 21.29 21.29 21.29 21.29	15.49 15.49 15.49 15.49 15.49	2.85 2.85 2.85 2.85 2.85 2.85	2.67 2.67 2.67 2.67 2.67		7.86 7.86 7.86 7.86					
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port 2-Wire Voice Unbundled 2-Way PBX Kentucky Room Area Calling Port without LUD 2-Wire Voice Unbundled PBX Kentucky LUD Area Calling Port 2-Wire Voice Unbundled PBX Kentucky Premium Calling Port 2-Wire Voice Unbundled PBX Kentucky Area Calling Port 2-Wire Voice Unbundled 2-Way Kentucky Area Calling Port without LUD			UEPPX UEPPX UEPPX UEPPX	UEPXD UEPXE UEPXF UEPXG	1.15 1.15 1.15 1.15 1.15	21.29 21.29 21.29 21.29 21.29	15.49 15.49 15.49 15.49	2.85 2.85 2.85 2.85	2.67 2.67 2.67 2.67		7.86 7.86 7.86					
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port 2-Wire Voice Unbundled 2-Way PBX Kentucky Room Area Calling Port without LUD 2-Wire Voice Unbundled PBX Kentucky LUD Area Calling Port 2-Wire Voice Unbundled PBX Kentucky Premium Calling Port 2-Wire Voice Unbundled 2-Way Kentucky Area Calling Port UD 2-Wire Voice Unbundled 2-Way Kentucky Area Calling Port 2-Wire Voice Unbundled 2-Way BX Hotel/Hospital Economy			UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPXD UEPXE UEPXF UEPXG UEPXH UEPXJ	1.15 1.15 1.15 1.15 1.15 1.15 1.15	21.29 21.29 21.29 21.29 21.29 21.29 21.29	15.49 15.49 15.49 15.49 15.49 15.49 15.49	2.85 2.85 2.85 2.85 2.85 2.85 2.85	2.67 2.67 2.67 2.67 2.67 2.67		7.86 7.86 7.86 7.86 7.86					
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port 2-Wire Voice Unbundled 2-Way PBX Kentucky Room Area Calling Port without LUD 2-Wire Voice Unbundled PBX Kentucky LUD Area Calling Port 2-Wire Voice Unbundled PBX Kentucky Premium Calling Port 2-Wire Voice Unbundled PBX Kentucky Area Calling Port without LUD 2-Wire Voice Unbundled 2-Way Kentucky Area Calling Port without LUD 2-Wire Voice Unbundled 2-Way Kentucky Area Calling Port without LUD 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port			UEPPX UEPPX UEPPX UEPPX UEPPX	UEPXD UEPXE UEPXF UEPXG UEPXH	1.15 1.15 1.15 1.15 1.15 1.15	21.29 21.29 21.29 21.29 21.29 21.29	15.49 15.49 15.49 15.49 15.49	2.85 2.85 2.85 2.85 2.85 2.85	2.67 2.67 2.67 2.67 2.67		7.86 7.86 7.86 7.86					
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port 2-Wire Voice Unbundled PBX Kentucky Room Area Calling Port without LUD 2-Wire Voice Unbundled PBX Kentucky LUD Area Calling Port 2-Wire Voice Unbundled PBX Kentucky Area Calling Port 2-Wire Voice Unbundled 2-Way Kentucky Area Calling Port UD 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy			UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPXD UEPXE UEPXF UEPXG UEPXH UEPXJ UEPXL	1.15 1.15 1.15 1.15 1.15 1.15 1.15 1.15	21.29 21.29 21.29 21.29 21.29 21.29 21.29 21.29	15.49 15.49 15.49 15.49 15.49 15.49 15.49	2.85 2.85 2.85 2.85 2.85 2.85 2.85	2.67 2.67 2.67 2.67 2.67 2.67 2.67		7.86 7.86 7.86 7.86 7.86 7.86					
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port 2-Wire Voice Unbundled P2Way PBX Kentucky Room Area Calling Port without LUD 2-Wire Voice Unbundled PBX Kentucky LUD Area Calling Port 2-Wire Voice Unbundled PBX Kentucky Premium Calling Port 2-Wire Voice Unbundled 2-Way Kentucky Area Calling Port 2-Wire Voice Unbundled 2-Way Kentucky Area Calling Port without LUD 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port			UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPXD UEPXE UEPXF UEPXG UEPXH UEPXJ	1.15 1.15 1.15 1.15 1.15 1.15 1.15	21.29 21.29 21.29 21.29 21.29 21.29 21.29	15.49 15.49 15.49 15.49 15.49 15.49 15.49	2.85 2.85 2.85 2.85 2.85 2.85 2.85	2.67 2.67 2.67 2.67 2.67 2.67		7.86 7.86 7.86 7.86 7.86					
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD 2-Wire Voice Unbundled 2-Way PBX Kentucky Room Area Calling Port without LUD 2-Wire Voice Unbundled PBX Kentucky LUD Area Calling Port 2-Wire Voice Unbundled PBX Kentucky Premium Calling Port 2-Wire Voice Unbundled 2-Way Kentucky Area Calling Port without LUD 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital 2-Wire 4			UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPXD UEPXE UEPXF UEPXG UEPXH UEPXJ UEPXL UEPXL	1.15 1.15 1.15 1.15 1.15 1.15 1.15 1.15	21.29 21.29 21.29 21.29 21.29 21.29 21.29 21.29 21.29 21.29	15.49 15.49 15.49 15.49 15.49 15.49 15.49 15.49 15.49	2.85 2.85 2.85 2.85 2.85 2.85 2.85 2.85	2.67 2.67 2.67 2.67 2.67 2.67 2.67 2.67		7.86 7.86 7.86 7.86 7.86 7.86 7.86 7.86					
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port 2-Wire Voice Unbundled PBX Kentucky Room Area Calling Port without LUD 2-Wire Voice Unbundled PBX Kentucky Premium Calling Port 2-Wire Voice Unbundled PBX Kentucky Premium Calling Port 2-Wire Voice Unbundled 2-Way Kentucky Area Calling Port without LUD 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port			UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPXD UEPXF UEPXF UEPXG UEPXH UEPXJ UEPXL UEPXM UEPXO	1.15 1.15 1.15 1.15 1.15 1.15 1.15 1.15	21.29 21.29 21.29 21.29 21.29 21.29 21.29 21.29 21.29 21.29 21.29	15.49 15.49 15.49 15.49 15.49 15.49 15.49 15.49 15.49 15.49	2.85 2.85 2.85 2.85 2.85 2.85 2.85 2.85	2.67 2.67 2.67 2.67 2.67 2.67 2.67 2.67		7.86 7.86 7.86 7.86 7.86 7.86 7.86 7.86					
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port 2-Wire Voice Unbundled PBX Kentucky Room Area Calling Port without LUD 2-Wire Voice Unbundled PBX Kentucky Premium Calling Port 2-Wire Voice Unbundled PBX Kentucky Premium Calling Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port 2-Wire Voice Unbundled 2-Way VBX Hotel/Hospital Economy Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPXD UEPXE UEPXF UEPXG UEPXH UEPXJ UEPXL UEPXL	1.15 1.15 1.15 1.15 1.15 1.15 1.15 1.15	21.29 21.29 21.29 21.29 21.29 21.29 21.29 21.29 21.29 21.29	15.49 15.49 15.49 15.49 15.49 15.49 15.49 15.49 15.49	2.85 2.85 2.85 2.85 2.85 2.85 2.85 2.85	2.67 2.67 2.67 2.67 2.67 2.67 2.67 2.67		7.86 7.86 7.86 7.86 7.86 7.86 7.86 7.86					
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port 2-Wire Voice Unbundled 2-Way PBX Kentucky Room Area Calling Port without LUD 2-Wire Voice Unbundled PBX Kentucky LUD Area Calling Port 2-Wire Voice Unbundled PBX Kentucky Premium Calling Port 2-Wire Voice Unbundled PBX Kentucky Area Calling Port without LUD 2-Wire Voice Unbundled 2-Way Kentucky Area Calling Port without LUD 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port NUMBER PORT ABILITY			UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPXD UEPXF UEPXF UEPXG UEPXH UEPXJ UEPXL UEPXM UEPXM UEPXS	1.15 1.15 1.15 1.15 1.15 1.15 1.15 1.15	21.29 21.29 21.29 21.29 21.29 21.29 21.29 21.29 21.29 21.29 21.29 21.29	15.49 15.49 15.49 15.49 15.49 15.49 15.49 15.49 15.49 15.49	2.85 2.85 2.85 2.85 2.85 2.85 2.85 2.85	2.67 2.67 2.67 2.67 2.67 2.67 2.67 2.67		7.86 7.86 7.86 7.86 7.86 7.86 7.86 7.86					
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port 2-Wire Voice Unbundled PBX Kentucky Room Area Calling Port without LUD 2-Wire Voice Unbundled PBX Kentucky Premium Calling Port 2-Wire Voice Unbundled PBX Kentucky Premium Calling Port 2-Wire Voice Unbundled 2-Way Kentucky Area Calling Port UD 2-Wire Voice Unbundled 2-Way Kentucky Area Calling Port without UD 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port NUMBER PORTABILITY Local Number Portability (1 per port)			UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPXD UEPXF UEPXF UEPXG UEPXH UEPXJ UEPXL UEPXM UEPXO	1.15 1.15 1.15 1.15 1.15 1.15 1.15 1.15	21.29 21.29 21.29 21.29 21.29 21.29 21.29 21.29 21.29 21.29 21.29	15.49 15.49 15.49 15.49 15.49 15.49 15.49 15.49 15.49 15.49	2.85 2.85 2.85 2.85 2.85 2.85 2.85 2.85	2.67 2.67 2.67 2.67 2.67 2.67 2.67 2.67		7.86 7.86 7.86 7.86 7.86 7.86 7.86 7.86					
LOCAL	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port 2-Wire Voice Unbundled PBX Kentucky Room Area Calling Port without LUD 2-Wire Voice Unbundled PBX Kentucky Premium Calling Port 2-Wire Voice Unbundled PBX Kentucky Premium Calling Port 2-Wire Voice Unbundled 2-Way Kentucky Area Calling Port UD 2-Wire Voice Unbundled 2-Way Kentucky Area Calling Port without UD 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port NUMBER PORTABILITY Local Number Portability (1 per port)			UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPXD UEPXF UEPXF UEPXG UEPXH UEPXJ UEPXL UEPXM UEPXM UEPXS	1.15 1.15 1.15 1.15 1.15 1.15 1.15 1.15	21.29 21.29 21.29 21.29 21.29 21.29 21.29 21.29 21.29 21.29 21.29 21.29	15.49 15.49 15.49 15.49 15.49 15.49 15.49 15.49 15.49 15.49	2.85 2.85 2.85 2.85 2.85 2.85 2.85 2.85	2.67 2.67 2.67 2.67 2.67 2.67 2.67 2.67		7.86 7.86 7.86 7.86 7.86 7.86 7.86 7.86					

JUNDLE	D NETWORK ELEMENTS - Kentucky					1					Sue Order	Sue Order	Attachment: 2			bit: B	+
EGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Order vs. Electronic-	
													1st	Add'l	Disc 1st	Disc Add'l	
						Rec	Nonrec		Nonrecurring					Rates(\$)			t
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	┶
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As-Is			UEPPX	USAC2		8.45	1.91				7.86					
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -			OLITX	00402		0.43	1.51				7.00					+
	Conversion - Switch with Change			UEPPX	USACC		8.45	1.91				7.86					
ADDIT	ONAL NRCs																_
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activity			UEPPX	USAS2	0.00	0.00	0.00				7.86					
	Subsequent Activity			UEPPA	05452	0.00	0.00	0.00				7.00					╈
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group						7.86	7.86				7.86					
	VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT	Г															
UNE P	ort/Loop Combination Rates																4
_	2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2		1			10.79 15.52											+
	2-Wile VG Coin Port/Loop Combo – Zone 3		3		1	31.74											+
UNE L	oop Rates																1
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPCO	UEPLX	9.64											1
	2-Wire Voice Grade Loop (SL1) - Zone 2			UEPCO	UEPLX	14.37											_
2-Wire	2-Wire Voice Grade Loop (SL1) - Zone 3 Voice Grade Line Ports (COIN)		3	UEPCO	UEPLX	30.59											+
2 1110	2-Wire Coin 2-Way without Operator Screening and without																t
	Blocking (AL, KY, LA, MS)			UEPCO	UEPRF	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Coin 2-Way with Operator Screening (AL, KY)			UEPCO	UEPRE	1.15	21.29	15.49	2.85	2.67		7.86					Ι
	2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, KY, LA, MS)			UEPCO	UEPRA	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking (KY)			UEPCO	UEPKA	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Coin 2-Way with Operator Screening & Blocking: 900/976, 1+DDD, 011+, & Local (AL, KY, LA, MS)			UEPCO	UEPCD	1.15	21.29	15.49	2.85	2.67		7.86					_
	2-Wire Coin Outward without Blocking and without Operator Screening (KY, LA, MS) 2-Wire Coin Outward with Operator Screening and 011 Blocking			UEPCO	UEPRN	1.15	21.29	15.49	2.85	2.67		7.86					_
	(GA, KY, MS) 2-Wire Coin Outward with Operator Screening and Blocking: 011,			UEPCO	UEPRJ	1.15	21.29	15.49	2.85	2.67		7.86					╞
	2-Wire Coin Outward Wirk Operator Screening & Blocking: 900/976, 2-Wire Coin Outward Operator Screening & Blocking: 900/976,			UEPCO	UEPRH	1.15	21.29	15.49	2.85	2.67		7.86					╞
	1+DDD, 011+, and Local (AL, KY, LA, MS)			UEPCO	UEPCN	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire 2-Way Smartline with 900/976 (all states except LA)			UEPCO	UEPCK	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Coin Outward Smartline with 900/976 (all states except LA)			UEPCO	UEPCR	1.15	21.29	15.49	2.85	2.67		7.86					
ADDIT	ONAL UNE COIN PORT/LOOP (RC) UNE Coin Port/Loop Combo Usage (Flat Rate)			UEPCO	URECU	2.57	21.29	15.49	2.85	2.67							+
LOCAL	NUMBER PORTABILITY			02100	JALOU	2.37	21.29	10.49	2.00	2.07							$^{+}$
	Local Number Portability (1 per port)			UEPCO	LNPCX	0.35											T
NONRI	CURRING CHARGES - CURRENTLY COMBINED				-												╇
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Conversion -			UEPCO	USAC2		0.10	0.10				7.86					1
	2-wire voice Grade Loop / Line Port Combination - Conversion - Switch with change ONAL NRCs			UEPCO	USACC		0.10	0.10				7.86					+
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity			UEPCO	USAS2	1	0.00	0.00				7.86					t
2-WIRE	VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE	LINE PO	ORT (R				0.00	0.00									t
UNE P	prt/Loop Combination Rates																T
_	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1		-	13.90											+
-	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2 2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		2		+	18.68 34.45											+
UNE L	pop Rates		5		1	34.43											$^{+}$
	2-Wire Voice Grade Loop (SL2) - Zone 1			UEPFR	UECF2	12.67											T
	2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFR	UECF2	17.45											T
0.14/1-	2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFR	UECF2	33.22						L					╇
∠-Wire	Voice Grade Line Port Rates (Res) 2-Wire voice unbundled port - residence			UEPFR	UEPRL	1.23	128.96	64.11	61.92	9.97		7.86					+
	2-Wire voice unbundled port vith Caller ID - res			UEPFR	UEPRC	1.23	128.96	64.11	61.92	9.97		7.86					+
	2-Wire voice unbundled port outgoing only - res			UEPFR	UEPRO	1.23	128.96	64.11	61.92	9.97		7.86					1

DUNDLE	D NETWORK ELEMENTS - Kentucky	1	1			1					Svo Orden	Sup Order	Attachment: 2			bit: B	+
EGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l	
					-		Nonrec	urring	Nonrecurring	Disconnect			055	Rates(\$)		1	+
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	+
	2-Wire voice Grade unbundled Kentucky extended local dialing																1
	parity port with Caller ID - res			UEPFR	UEPRM	1.23	128.96	64.11	61.92	9.97		7.86					
	2-Wire voice unbundles res, low usage line port with Caller ID																
	(LUM)			UEPFR	UEPAP	1.23	128.96	64.11	61.92	9.97		7.86					+
	2-Wire Voice Unbundled Kentucky Residence Dialing Plan without Caller ID			UEPFR	UEPWE	1.23	128.96	64.11	61.92	9.97		7.86					
INTERC	DFFICE TRANSPORT			02.111	02.112	1.20	120.00	0	01102	0.07		1.00					+
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility																T
	Termination			UEPFR	U1TV2	23.95	98.09	53.67	56.31	22.42		7.86					_
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile				1L5XX	0.0005											
FEATU	or Fraction Mile			UEPFR	1L5XX	0.0095											+
	All Features Offered			UEPFR	UEPVF	0.00	0.00	0.00				7.86					+
	NUMBER PORTABILITY	1				2.00		2100									1
	Local Number Portability (1 per port)			UEPFR	LNPCX	0.35											Т
NONRE	CURRING CHARGES (NRCs) - CURRENTLY COMBINED	<u> </u>															╇
1	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port Combination - Conversion - Switch-as-is			UEPFR	USAC2		9.03	1.87				7.86					1
-	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port	1			UGAGZ		9.03	1.07				1.00					+
	Combination - Conversion - Switch-With-Change	1		UEPFR	USACC		9.03	1.87				7.86					I
	VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE	LINE P	ORT (B	US)													Τ
	ort/Loop Combination Rates																_
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			13.90											┿
-	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2 2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		2			18.68 34.45											+
	op Rates		5			34.43											t
	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFB	UECF2	12.67											1
	2-Wire Voice Grade Loop (SL2) - Zone 2			UEPFB	UECF2	17.45											L
	2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFB	UECF2	33.22											+
z-wile	Voice Grade Line Port (Bus) 2-Wire voice unbundled port without Caller ID - bus			UEPFB	UEPBL	1.23	128.96	64.11	61.92	9.97		7.86					+
	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPFB	UEPBC	1.23	128.96	64.11	61.92	9.97		7.86					T
	2-Wire voice unbundled port outgoing only - bus			UEPFB	UEPBO	1.23	128.96	64.11	61.92	9.97		7.86					
	2-Wire voice Grade unbundled Kentucky extended local dialing																
	parity port with Caller ID - bus			UEPFB UEPFB	UEPBM UEPB1	1.23 1.23	128.96 128.96	64.11 64.11	61.92 61.92	9.97 9.97		7.86 7.86					+
	2-Wire voice unbundled incoming only port with Caller ID - Bus 2-Wire Voice Unbundled Kentucky Business Dialing Plan without			UEPFB	UEPDI	1.23	120.90	64.11	01.92	9.97		7.00					+
	Caller ID			UEPFB	UEPWF	1.23	128.96	64.11	61.92	9.97		7.86					
	NUMBER PORTABILITY																
	Local Number Portability (1 per port)			UEPFB	LNPCX	0.35											F
INTERC	DFFICE TRANSPORT Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility	 			+						ļ						╋
	Termination	1		UEPFB	U1TV2	23.95	98.09	53.67	56.31	22.42		7.86					I
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile	1				20.00	00.00	00.01	00.01	T2							t
	or Fraction Mile			UEPFB	1L5XX	0.0095											
FEATU		<u> </u>			11501/5												Ļ
	All Features Offered CURRING CHARGES (NRCs) - CURRENTLY COMBINED			UEPFB	UEPVF	0.00	0.00	0.00				7.86					+
NUNRE	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port	1															+
	Combination - Conversion - Switch-as-is			UEPFB	USAC2		9.03	1.87				7.86					1
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port																Т
	Combination - Conversion - Switch with change	I		UEPFB	USACC	ļ	9.03	1.87			L	7.86					╇
	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX) ort/Loop Combination Rates	<u> </u>			+												╋
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1	1	1		+	13.90											+
1	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2	1	2		1	18.68											\uparrow
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			34.45											Т
	op Rates	<u> </u>	<u> </u>		115.055												Ļ
	2-Wire Voice Grade Loop (SL2) - Zone 1 2-Wire Voice Grade Loop (SL2) - Zone 2		1	UEPFP UEPFP	UECF2 UECF2	12.67 17.45											╋
+	2-Wire Voice Grade Loop (SL2) - Zone 2 2-Wire Voice Grade Loop (SL2) - Zone 3	1	2	UEPFP	UECF2 UECF2	33.22											╋
2-Wire	Voice Grade Line Port Rates (BUS - PBX)	1			520.2	00.22							1				$^{+}$
			-			-											+-

IRONDLE	D NETWORK ELEMENTS - Kentucky			1	T								Attachment: 2			bit: B	⊢
TEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l	
						Rec	Nonrec	urring	Nonrecurring			-		Rates(\$)			
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	
	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPFP	UEPPO	1.23	164.27	78.65	75.05	8.73		7.86					
	Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPFP	UEPP1	1.23	164.27	78.65	75.05	8.73		7.86					
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPFP	UEPLD	1.23	164.27	78.65	75.05	8.73		7.86					
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPFP	UEPXA	1.23	164.27	78.65	75.05	8.73		7.86					
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPFP	UEPXB	1.23	164.27	78.65	75.05	8.73		7.86					
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPFP	UEPXC	1.23	164.27	78.65	75.05	8.73		7.86					
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPFP	UEPXD	1.23	164.27	78.65	75.05	8.73		7.86					
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD																
	Capable Port			UEPFP	UEPXE	1.23	164.27	78.65	75.05	8.73		7.86					
	2-Wire Voice Unbundled 2-Way PBX Kentucky Room Area Calling																
	Port without LUD			UEPFP	UEPXF	1.23	164.27	78.65	75.05	8.73		7.86					
	2-Wire Voice Unbundled PBX Kentucky LUD Area Calling Port			UEPFP	UEPXG	1.23	164.27	78.65	75.05	8.73		7.86					
	2-Wire Voice Unbundled PBX Kentucky Premium Calling Port			UEPFP	UEPXH	1.23	164.27	78.65	75.05	8.73		7.86					
	2-Wire Voice Unbundled 2-Way Kentucky Area Calling Port without																
	LUD			UEPFP	UEPXJ	1.23	164.27	78.65	75.05	8.73		7.86					L
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy	_															1
	Administrative Calling Port			UEPFP	UEPXL	1.23	164.27	78.65	75.05	8.73		7.86					⊢
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy																1
	Room Calling Port			UEPFP	UEPXM	1.23	164.27	78.65	75.05	8.73		7.86					
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital																1
	Discount Room Calling Port			UEPFP	UEPXO	1.23	164.27	78.65	75.05	8.73		7.86					
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPFP	UEPXS	1.23	164.27	78.65	75.05	8.73		7.86					
LOCAL	NUMBER PORTABILITY																
	Local Number Portability (1 per port)			UEPFP	LNPCP	3.15	0.00	0.00									
INTER	OFFICE TRANSPORT																1
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility																1
	Termination			UEPFP	U1TV2	23.95	98.09	53.67	56.31	22.42		7.86					
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile																1
	or Fraction Mile			UEPFP	1L5XX	0.0095											
FEATU																	1
	All Features Offered			UEPFP	UEPVF	0.00	0.00	0.00				7.86					
NONR	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED			-	-												
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port																
	Combination - Conversion - Switch-as-is			UEPFP	USAC2		9.03	1.87				7.86					
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port																1
	Combination - Conversion - Switch with change			UEPFP	USACC		9.03	1.87				7.86					
	PORT/LOOP COMBINATIONS - COST BASED RATES																1
	VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK	PORT															t
	ort/Loop Combination Rates																t
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1		1			21.30				İ							t
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2		2		1	26.08				İ							
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3		3		1	41.85				İ							
UNE L	oop Rates	l			1												T
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1		1	UEPPX	UECD1	12.67				İ		7.86					
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2	l	2	UEPPX	UECD1	17.45						7.86					T
1	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3	l	3	UEPPX	UECD1	33.22						7.86					t
UNE P	ort Rate				1					İ							
-	Exchange Ports - 2-Wire DID Port	l	İ	UEPPX	UEPD1	8.63	336.11	27.75	132.37	9.31		7.86					T
NONR	ECURRING CHARGES - CURRENTLY COMBINED		1		1			-									
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with		İ		1												T
1	BellSouth Allowable Changes			UEPPX	USA1C		7.85	1.87			1	7.86					1
ADDIT	IONAL NRCs		1														
	2-Wire DID Subsequent Activity - Add Trunks, Per Trunk		1	UEPPX	USAS1		32.25	32.25		İ		7.86					
Teleph	one Number/Trunk Group Establisment Charges		Ι		-												
	DID Trunk Termination (One Per Port)		1	UEPPX	NDT	0.00	0.00	0.00				7.86					
	Additional DID Numbers for each Group of 20 DID Numbers		1	UEPPX	ND4	0.00	0.00	0.00				7.86					
	DID Numbers, Non- consecutive DID Numbers , Per Number		1	UEPPX	ND5	0.00	0.00	0.00		İ		7.86					
+	Reserve Non-Consecutive DID numbers		1	UEPPX	ND6	0.00	0.00	0.00		İ		7.86				İ	t
	Reserve DID Numbers			UEPPX	NDV	0.00	0.00	0.00		1		7.86				1	F
LOCAI					1	0.00	0.00	0.00		1						1	F
	Local Number Portability (1 per port)			UEPPX	LNPCP	3.15	0.00	0.00									t
2-WIRE	E ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINI		ORT			0.10	0.00	0.00		1	1						1
	ort/Loop Combination Rates				+	1						-					+

UNBUNDLE	D NETWORK ELEMENTS - Kentucky													Attachment: 2	2	Exhi	bit: B	
												Svc Order	Svc Order		Incremental	Incremental	Incremental	[
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -	
												Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc	
ATEGORY	RATE ELEMENTS	Interim	Zone	BC	cs	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.	ĺ
												-	-	Electronic-	Electronic-	Electronic-	Electronic-	ĺ
														1st	Add'l	Disc 1st	Disc Add'l	
																		
							Rec	Nonrec		Nonrecurring					Rates(\$)			L
	OW JORN Divited Oresta Laws (OW JORN Divited Line Olds Dest							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 1		4	UEPPB	UEPPR		25.69											
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -	-		UEPPD	UEPPR		25.69						-					<u> </u>
	UNE Zone 2		2	UEPPB	UEPPR		31.92											
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -		2	UEFFB	UEFFK		31.92											<u> </u>
	UNE Zone 3		3	UEPPB	UEPPR		50.21											ĺ
UNE L	pop Rates			02.110	021111		00.21											
	2-Wire ISDN Digital Grade Loop - UNE Zone 1		1	UEPPB	UEPPR	USL2X	16.10						7.86					
	2-Wire ISDN Digital Grade Loop - UNE Zone 2		2	UEPPB	UEPPR	USL2X	22.33						7.86					ĺ
	2-Wire ISDN Digital Grade Loop - UNE Zone 3		3	UEPPB	UEPPR	USL2X	40.63						7.86					
UNE P	ort Rate																	
	Exchange Port - 2-Wire ISDN Line Side Port			UEPPB	UEPPR	UEPPB	9.59	320.53	289.13	92.19	17.56		7.86					
NONR	ECURRING CHARGES - CURRENTLY COMBINED																	
	2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port																	1
	Combination - Conversion	<u> </u>		UEPPB	UEPPR	USACB	0.00	22.77	17.00				7.86					L
	IONAL NRCs	 	L			ļ						L						
LOCAL	NUMBER PORTABILITY																	
	Local Number Portability (1 per port)			UEPPB	UEPPR	LNPCX	0.35	0.00	0.00									L
B-CHA	NNEL USER PROFILE ACCESS:																	ļ
	CVS/CSD (DMS/5ESS)				UEPPR	U1UCA	0.00	0.00	0.00									ļ
	CVS (EWSD) CSD				UEPPR UEPPR	U1UCB	0.00	0.00	0.00									
B CUA	NNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS SC			UEPPB	UEPPR	U1UCC	0.00	0.00	0.00									
Б-СПА	CVS/CSD (DMS/5ESS)	∠,IVI⊃, α	N)	UEPPB	UEPPR	U1UCD	0.00	0.00	0.00									
	CVS/CSD (DMS/SESS) CVS (EWSD)	-		UEPPB	UEPPR	U1UCE	0.00	0.00	0.00				-					
	CSD			UEPPB	UEPPR	U1UCF	0.00	0.00	0.00									
USER	TERMINAL PROFILE			OLITO	ULITIK	01001	0.00	0.00	0.00									
OULK	User Terminal Profile (EWSD only)			UEPPB	UEPPR	U1UMA	0.00	0.00	0.00									
VERTI	CAL FEATURES	1		02110	02111	0.0101	0.00	0.00	0.00									
	All Vertical Features - One per Channel B User Profile			UEPPB	UEPPR	UEPVF	0.00	0.00	0.00									
INTER	OFFICE CHANNEL MILEAGE																	
	Interoffice Channel mileage each, including first mile and facilities																	
	termination			UEPPB		M1GNC	29.12	47.34	31.78	22.77	8.75		7.86					1
	Interoffice Channel mileage each, additional mile			UEPPB	UEPPR	M1GNM	0.01	0.00	0.00				7.86					
	E DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK	PORT																
UNE P	ort/Loop Combination Rates																	
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE																	ĺ
	Zone 1		1	UEPPP		ļ	170.06					ļ						
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE	1																1
	Zone 2	<u> </u>	2	UEPPP		ļ	197.70								-	-		
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE		~				004.05											ĺ
UNE	Zone 3 pop Rates	<u> </u>	3	UEPPP			381.35											<u> </u>
UNE L	4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPPP		USL4P	86.47					+	7.86					
	4-Wire DS1 Digital Loop - UNE Zone 2	+		UEPPP		USL4P USL4P	114.10					<u> </u>	7.86					
	4-Wire DS1 Digital Loop - UNE Zone 3			UEPPP		USL4P	297.76						7.86					
LINF P	ort Rate	1	5	36111		302-1	231.10					1	1.00					<u> </u>
	Exchange Ports - 4-Wire ISDN DS1 Port	1		UEPPP		UEPPP	83.59	736.16	382.74	159.48	48.82	1	7.86					i
NONR		1					00.00		002.14		.0.02			1				-
	4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port	1				1	1			İ		1		l			1	<u> </u>
	Combination - Conversion -Switch-as-is			UEPPP		USACP	0.00	81.70	61.37				7.86					ĺ
ADDIT	ONAL NRCs																	
	4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy-																	1
	Inward/two way Tel Nos. (except NC)			UEPPP		PR7TF		0.54					7.86					
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward																	ĺ
	Tel Numbers (All States except NC)	<u> </u>		UEPPP		PR7TO		12.71	12.71				7.86					L
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port -	1											_					1
	Subsequent Inward Tel Numbers	I		UEPPP		PR7ZT		25.41	25.41			L	7.86					I
LOCAL	NUMBER PORTABILITY	<u> </u>				LNDCY						L	L					
INTER S	Local Number Portability (1 per port)	 		UEPPP		LNPCN	1.75					ł						I
INTER	FACE (Provsioning Only)						0.00	0.00	0.00									──
	Voice/Data	 		UEPPP		PR71V	0.00	0.00	0.00			l						I
	Digital Data	1		UEPPP		PR71D	0.00	0.00	0.00	I		L	I	1	l			L

	ED NETWORK ELEMENTS - Kentucky	1			1	1					0	0	Attachment: 2			oit: B	──
ATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l	
						Rec	Nonrec	urring	Nonrecurring	Disconnect				Rates(\$)			
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	
	Inward Data			UEPPP	PR71E	0.00	0.00	0.00									
New o	r Additional "B" Channel																
	New or Additional - Voice/Data B Channel			UEPPP	PR7BV	0.00	15.48					7.86					<u> </u>
	New or Additional - Digital Data B Channel	-		UEPPP	PR7BF	0.00	15.48					7.86					
CALL	New or Additional Inward Data B Channel TYPES			UEPPP	PR7BD	0.00	15.48					7.86					+
CALL	Inward			UEPPP	PR7C1	0.00	0.00	0.00									+
	Outward			UEPPP	PR7C0	0.00	0.00	0.00									+
	Two-way			UEPPP	PR7CC	0.00	0.00	0.00									
Intero	fice Channel Mileage																1
	Fixed Each Including First Mile			UEPPP	1LN1A	96.27	105.52	98.46	23.09	20.49		7.86					
	Each Airline-Fractional Additional Mile			UEPPP	1LN1B	0.23											
	E DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT																
UNE F	ort/Loop Combination Rates	-	4			147.00											+
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2	-	1	UEPDC UEPDC		147.99 175.62											+
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3	1	2	UEPDC	1	359.28											+
UNE L	oop Rates	1			1	000.20											+
	4-Wire DS1 Digital Loop - UNE Zone 1	1	1	UEPDC	USLDC	86.47						7.86					
	4-Wire DS1 Digital Loop - UNE Zone 2	1	2	UEPDC	USLDC	114.10						7.86					L
	4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPDC	USLDC	297.76						7.86					
UNE F	ort Rate																
	4-Wire DDITS Digital Trunk Port			UEPDC	UDD1T	61.52	780.61	375.52	176.19	16.98		7.86					
NONR	ECURRING CHARGES - CURRENTLY COMBINED																
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination	-			110404		00.04	10 70				7.00					
	Switch-as-is 4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination	-		UEPDC	USAC4		92.84	46.70				7.86					-
	Conversion with DS1 Changes	-		UEPDC	USAWA		92.84	46.70				7.86					
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination	-			00/11/		32.04	40.70				7.00					+
	Conversion with Change - Trunk			UEPDC	USAWB		92.84	46.70				7.86					
ADDIT	IONAL NRCs																
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC - Subsequent																
	Channel Activation/Chan - 2-Way Trunk	_		UEPDC	UDTTA		15.09	15.09				7.86					
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent						45.00	45.00				7.00					
	Channel Activation/Chan - 1-Way Outward Trunk 4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel			UEPDC	UDTTB		15.09	15.09		-		7.86					-
	Activation/Chan Inward Trunk w/out DID			UEPDC	UDTTC		15.09	15.09				7.86					
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan			02100	00110		10.00	10.00				1.00					+
	Activation Per Chan - Inward Trunk with DID			UEPDC	UDTTD		15.09	15.09				7.86					
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan																Γ
	Activation / Chan - 2-Way DID w User Trans			UEPDC	UDTTE		15.09	15.09				7.86					
BIPOL	AR 8 ZERO SUBSTITUTION																<u> </u>
	B8ZS -Superframe Format			UEPDC	CCOSF		0.00	730.00			L	7.86	ļ				<u> </u>
Altor	B8ZS - Extended Superframe Format ate Mark Inversion			UEPDC	CCOEF		0.00	730.00	1			7.86	-				+
Allem	AMI - Superframe Format	1		UEPDC	MCOSF		0.00	0.00									+
	AMI - Extended SuperFrame Format	1		UEPDC	MCOPO		0.00	0.00									+
Telepi	none Number/Trunk Group Establisment Charges	1					1.00	2.00									1
	Telephone Number for 2-Way Trunk Group			UEPDC	UDTGX	0.00	0.00	0.00				7.86					
	Telephone Number for 1-Way Outward Trunk Group			UEPDC	UDTGY	0.00		0.00				7.86					
	Telephone Number for 1-Way Inward Trunk Group Without DID			UEPDC	UDTGZ	0.00	0.00	0.00				7.86					
	DID Numbers for each Group of 20 DID Numbers		L	UEPDC	ND4	0.00	0.00	0.00				7.86					_
	DID Numbers, Non- consecutive DID Numbers , Per Number	-		UEPDC	ND5	0.00	0.00	0.00				7.86					<u> </u>
	Reserve Non-Consecutive DID Nos. Reserve DID Numbers			UEPDC UEPDC	ND6 NDV	0.00	0.00	0.00				7.86 7.86					+
Dedic	ted DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1 [)igital Lo			Port	0.00	0.00	0.00				1.00					⊢
Jealo	Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities					1											\vdash
	Termination)	1		UEPDC	1LNO1	96.04	105.52	98.46	23.09	20.49		7.86					1
	,																Γ
	Interoffice Channel Mileage - Additional rate per mile - 0-8 miles			UEPDC	1LNOA	0.23	0.00	0.00									
	Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities	1															1
	Termination)			UEPDC	1LNO2	0.00	0.00	0.00									
				1	1	1											1

	ED NETWORK ELEMENTS - Kentucky	r –	1			1					Our Our	Our Out	Attachment: 2			bit: B
					1							Svc Order	Incremental	Incremental	Incremental	Incremental
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			_								Elec	Manually	Manual Svc	Manual Svc	Manual Svc	
EGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
					-	Rec	Nonrec	urring Add'l	Nonrecurring First	Disconnect Add'l	SOMEC	SOMAN	OSS SOMAN	Rates(\$) SOMAN	SOMAN	SOMAN
	Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities				-		First	Add1	First	Add1	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOWAN
	Termination)			UEPDC	1LNO3	0.00	0.00	0.00								
	,															
_	Interoffice Channel Mileage - Additional rate per mile - 25+ miles			UEPDC	1LNOC	0.45	0.00	0.00								
	Local Number Portability, per DS0 Activated Central Office Termininating Point			UEPDC UEPDC	LNPCP CTG	3.15	0.00	0.00								
4-WIRF	E DS1 LOOP WITH CHANNELIZATION WITH PORT			UEFDC	010	0.00										
	n is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activa	ations														
Each S	System can have up to 24 combinations of rates depending on ty		number	of ports used												
UNE D	S1 Loop															
	4-Wire DS1 Loop - UNE Zone 1		1	UEPMG	USLDC	86.47	0.00	0.00								
_	4-Wire DS1 Loop - UNE Zone 2		2	UEPMG UEPMG	USLDC USLDC	114.10	0.00	0.00								
	4-Wire DS1 Loop - UNE Zone 3 SO Channelization Capacities (D4 Channel Bank Configurations)		3	UEPING	USLUC	297.76	0.00	0.00								
	24 DSO Channel Capacity - 1 per DS1	-		UEPMG	VUM24	111.16	0.00	0.00				7.86				
1	48 DSO Channel Capacity - 1 per 2 DS1s	1	1	UEPMG	VUM48	222.32	0.00	0.00				7.86				1
	96 DSO Channel Capacity -1per 4 DS1s			UEPMG	VUM96	444.64	0.00	0.00				7.86				
	144 DS0 Channel Capacity - 1 per 6 DS1s			UEPMG	VUM14	666.96	0.00	0.00				7.86				
	192 DS0 Channel Capacity -1 per 8 DS1s			UEPMG	VUM19	889.28	0.00	0.00				7.86				
	240 DS0 Channel Capacity - 1 per 10 DS1s			UEPMG	VUM20	1,111.60	0.00	0.00				7.86				
_	288 DS0 Channel Capacity - 1 per 12 DS1s			UEPMG UEPMG	VUM28 VUM38	1,333.92	0.00	0.00				7.86				
	384 DS0 Channel Capacity - 1 per 16 DS1s 480 DS0 Channel Capacity - 1 per 20 DS1s			UEPMG	VUM38 VUM40	1,778.56 2,223.20	0.00	0.00				7.86 7.86				
	576 DS0 Channel Capacity -1 per 24 DS1s			UEPMG	VUM57	2,223.20	0.00	0.00				7.86				
	672 DS0 Channel Capacity - 1 per 28 DS1s			UEPMG	VUM67	3,112.48	0.00	0.00				7.86				
Non-Re	ecurring Charges (NRC) Associated with 4-Wire DS1 Loop with	Channel	iztion w		on Charge Bas		0.00									
	num System configuration is One (1) DS1, One (1) D4 Channel B															
Multiple	es of this configuration functioning as one are considered Add'l a	after the	minimu	Im system configur	ation is counte	ed.										
	NRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes			UEPMG	USAC4	0.00	94.30	4.24				7.86				
Systen	n Additions at End User Locations Where 4-Wire DS1 Loop with	Channe	ization				01.00					1.00				
New (N	Not Currently Combined) in all states, except in Density Zone 1 of	f Top 8 M	/ISA's													
	1 DS1/D4 Channel Bank - Additionally Add NRC for each Port and						740.00	100.00				7.00				
Pinolo	Assoc Fea Activation r 8 Zero Substitution			UEPMG	VUMD4	0.00	718.89	469.86	149.83	17.77		7.86				
ыроа	Clear Channel Capability Format, superframe - Subsequent Activity	/			-											
	Only	'		UEPMG	CCOSF	0.00	0.00	730.00				7.86				
	Clear Channel Capability Format - Extended Superframe -															
	Subsequent Activity Only			UEPMG	CCOEF	0.00	0.00	730.00				7.86				
Alterna	ate Mark Inversion (AMI)															
	Superframe Format															
				UEPMG	MCOSF	0.00	0.00	0.00								
Evolo	Extended Superframe Format	with D-	rt	UEPMG UEPMG	MCOSF MCOPO	0.00	0.00 0.00	0.00								
	Extended Superframe Format nge Ports Associated with 4-Wire DS1 Loop with Channelization	with Po	rt													
	Extended Superframe Format	with Po	rt													
	Extended Superframe Format nge Ports Associated with 4-Wire DS1 Loop with Channelization	with Po	rt						0.00	0.00		7.86				
	Extended Superframe Format nge Ports Associated with 4-Wire DS1 Loop with Channelization nge Ports	with Pc	rt	UEPMG	MCOPO	0.00	0.00	0.00	0.00	0.00		7.86				
	Extended Superframe Format nge Ports Associated with 4-Wire DS1 Loop with Channelization nge Ports Line Side Combination Channelized PBX Trunk Port - Business Line Side Outward Channelized PBX Trunk Port - Business	with Po	rt	UEPMG UEPPX UEPPX	MCOPO UEPCX UEPOX	0.00 1.15 1.15	0.00	0.00	0.00	0.00		7.86				
	Extended Superframe Format nge Ports Associated with 4-Wire DS1 Loop with Channelization nge Ports Line Side Combination Channelized PBX Trunk Port - Business Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID	with Po	rt	UEPMG UEPPX UEPPX UEPPX	MCOPO UEPCX UEPOX UEP1X	0.00 1.15 1.15 1.15	0.00	0.00 0.00 0.00 0.00	0.00	0.00		7.86				
Exchar	Extended Superframe Format nge Ports Associated with 4-Wire DS1 Loop with Channelization nge Ports Line Side Combination Channelized PBX Trunk Port - Business Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port	with Po	rt	UEPMG UEPPX UEPPX	MCOPO UEPCX UEPOX	0.00 1.15 1.15	0.00	0.00	0.00	0.00		7.86				
Exchar	Extended Superframe Format nge Ports Associated with 4-Wire DS1 Loop with Channelization nge Ports Line Side Combination Channelized PBX Trunk Port - Business Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port a Activations - Unbundled Loop Concentration			UEPMG UEPPX UEPPX UEPPX	MCOPO UEPCX UEPOX UEP1X	0.00 1.15 1.15 1.15	0.00	0.00 0.00 0.00 0.00	0.00	0.00		7.86				
Exchar	Extended Superframe Format nge Ports Associated with 4-Wire DS1 Loop with Channelization nge Ports Line Side Combination Channelized PBX Trunk Port - Business Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port e Activations - Unbundled Loop Concentration Feature (Service) Activation for each Line Port Terminated in D4			UEPPX UEPPX UEPPX UEPPX UEPPX	MCOPO UEPCX UEPOX UEP1X UEPDM	1.15 1.15 1.15 1.15 8.65	0.00 0.00 0.00 0.00 0.00	0.00 0.00 0.00 0.00 0.00	0.00 0.00 0.00	0.00 0.00 0.00		7.86 7.86 7.86				
Exchar	Extended Superframe Format nge Ports Associated with 4-Wire DS1 Loop with Channelization nge Ports Line Side Combination Channelized PBX Trunk Port - Business Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port e Activations - Unbundled Loop Concentration Feature (Service) Activation for each Line Port Terminated in D4 Bank	with Po		UEPMG UEPPX UEPPX UEPPX	MCOPO UEPCX UEPOX UEP1X	0.00 1.15 1.15 1.15	0.00	0.00 0.00 0.00 0.00	0.00	0.00		7.86				
Exchar	Extended Superframe Format nge Ports Associated with 4-Wire DS1 Loop with Channelization nge Ports Line Side Combination Channelized PBX Trunk Port - Business Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port e Activations - Unbundled Loop Concentration Feature (Service) Activation for each Line Port Terminated in D4	with Pc		UEPPX UEPPX UEPPX UEPPX UEPPX	MCOPO UEPCX UEPOX UEP1X UEPDM	1.15 1.15 1.15 1.15 8.65	0.00 0.00 0.00 0.00 0.00	0.00 0.00 0.00 0.00 0.00	0.00 0.00 0.00	0.00 0.00 0.00		7.86 7.86 7.86				
Exchar Featur	Extended Superframe Format nge Ports Associated with 4-Wire DS1 Loop with Channelization nge Ports Line Side Combination Channelized PBX Trunk Port - Business Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port e Activations - Unbundled Loop Concentration Feature (Service) Activation for each Line Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Port Terminated in D4 Bank			UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	MCOPO UEPCX UEPOX UEP1X UEPDM 1PQWM	0.00 1.15 1.15 1.15 8.65 0.62 0.62	0.00 0.00 0.00 0.00 25.40 78.15	0.00 0.00 0.00 0.00 13.41 19.68	0.00 0.00 4.17	0.00 0.00 4.15		7.86 7.86 7.86 7.86				
Feature	Extended Superframe Format nge Ports Associated with 4-Wire DS1 Loop with Channelization nge Ports Line Side Combination Channelized PBX Trunk Port - Business Line Side Outward Channelized PBX Trunk Port - Business Line Side Unward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port e Activations - Unbundled Loop Concentration Feature (Service) Activation for each Line Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Port Terminated in D4 Bank One Number/ Group Establishment Charges for DID Service DID Trunk Termination (1 per Port)			UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	MCOPO UEPCX UEPOX UEP1X UEP1X UEP1M 1PQWM 1PQWU NDT	0.00 1.15 1.15 1.15 8.65 0.62 0.62 0.00	0.00 0.00 0.00 0.00 25.40 78.15 0.00	0.00 0.00 0.00 0.00 13.41 19.68 0.00	0.00 0.00 4.17	0.00 0.00 4.15		7.86 7.86 7.86 7.86 7.86 7.86				
Exchar	Extended Superframe Format nge Ports Associated with 4-Wire DS1 Loop with Channelization nge Ports Line Side Combination Channelized PBX Trunk Port - Business Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port - Business Peature (Service) Activation for each Line Port Terminated in D4 Bank Peature (Service) Activation for each Trunk Port Terminated in D4 Bank Din Bumber' Group Establishment Charges for DID Service DID Trunk Termination (1 per Port) DID Numbers - groups of 20 - Valid all States	with Pc		UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	MCOPO UEPCX UEPOX UEP1X UEPDM 1PQWM 1PQWU NDT ND4	0.00 1.15 1.15 1.15 0.62 0.62 0.62 0.00 0.00	0.00 0.00 0.00 0.00 25.40 78.15 0.00 0.00	0.00 0.00 0.00 0.00 13.41 19.68 0.00 0.00	0.00 0.00 4.17	0.00 0.00 4.15		7.86 7.86 7.86 7.86 7.86 7.86 7.86 7.86				
Exchar	Extended Superframe Format nge Ports Associated with 4-Wire DS1 Loop with Channelization nge Ports Line Side Combination Channelized PBX Trunk Port - Business Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port e Activations - Unbundled Loop Concentration Feature (Service) Activation for each Line Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Port Terminated in D4 Bank Forture (Service) Activation for each Trunk Port Terminated in D4 Bank None Number/ Group Establishment Charges for DID Service DID Trunk Termination (1 per Port) DID Numbers - groups of 20 - Valid all States Non-Consecutive DID Numbers - per number			UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	MCOPO UEPCX UEPOX UEP1X UEPDM 1PQWM 1PQWU NDT ND4 ND5	0.00 1.15 1.15 1.15 0.62 0.62 0.62 0.00 0.00 0.00	0.00 0.00 0.00 0.00 25.40 78.15 0.00 0.00 0.00	0.00 0.00 0.00 0.00 13.41 19.68 0.00 0.00 0.00	0.00 0.00 4.17	0.00 0.00 4.15		7.86 7.86 7.86 7.86 7.86 7.86 7.86 7.86				
Feature	Extended Superframe Format nge Ports Associated with 4-Wire DS1 Loop with Channelization nge Ports Line Side Combination Channelized PBX Trunk Port - Business Line Side Outward Channelized PBX Trunk Port - Business Line Side Untward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port e Activations - Unbundled Loop Concentration Feature (Service) Activation for each Line Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Port Terminated in D4 Bank Fore Number/ Group Establishment Charges for DID Service DID Trunk Termination (1 per Port) DID Numbers - groups of 20 - Valid all States Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers			UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	MCOPO UEPCX UEPCX UEPDX UEP1X UEPDM 1PQWM 1PQWU NDT ND4 ND5 ND6	0.00 1.15 1.15 1.15 0.62 0.62 0.62 0.00 0.00 0.00 0.00	0.00 0.00 0.00 0.00 25.40 78.15 0.00 0.00 0.00 0.00	0.00 0.00 0.00 0.00 13.41 19.68 0.00 0.00 0.00 0.00 0.00	0.00 0.00 4.17	0.00 0.00 4.15		7.86 7.86 7.86 7.86 7.86 7.86 7.86 7.86				
Exchai	Extended Superframe Format nge Ports Associated with 4-Wire DS1 Loop with Channelization nge Ports Line Side Combination Channelized PBX Trunk Port - Business Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port a Activations - Unbundled Loop Concentration Feature (Service) Activation for each Line Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Port Terminated in D4 Bank One Number/ Group Establishment Charges for DID Service DID Trunk Termination (1 per Port) DID Numbers - goups of 20 - Valid all States Non-Consecutive DID Numbers Reserve Non-Consecutive DID Numbers Reserve DID Numbers	with Pc		UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	MCOPO UEPCX UEPOX UEP1X UEPDM 1PQWM 1PQWU NDT ND4 ND5	0.00 1.15 1.15 1.15 0.62 0.62 0.62 0.00 0.00 0.00	0.00 0.00 0.00 0.00 25.40 78.15 0.00 0.00 0.00	0.00 0.00 0.00 0.00 13.41 19.68 0.00 0.00 0.00	0.00 0.00 4.17	0.00 0.00 4.15		7.86 7.86 7.86 7.86 7.86 7.86 7.86 7.86				
Exchai	Extended Superframe Format nge Ports Associated with 4-Wire DS1 Loop with Channelization nge Ports Line Side Combination Channelized PBX Trunk Port - Business Line Side Outward Channelized PBX Trunk Port - Business Line Side Untward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port e Activations - Unbundled Loop Concentration Feature (Service) Activation for each Line Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Port Terminated in D4 Bank Fore Number/ Group Establishment Charges for DID Service DID Trunk Termination (1 per Port) DID Numbers - groups of 20 - Valid all States Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers	with Pc		UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	MCOPO UEPCX UEPCX UEPDX UEP1X UEPDM 1PQWM 1PQWU NDT ND4 ND5 ND6	0.00 1.15 1.15 1.15 0.62 0.62 0.62 0.00 0.00 0.00 0.00	0.00 0.00 0.00 0.00 25.40 78.15 0.00 0.00 0.00 0.00	0.00 0.00 0.00 0.00 13.41 19.68 0.00 0.00 0.00 0.00 0.00	0.00 0.00 4.17	0.00 0.00 4.15		7.86 7.86 7.86 7.86 7.86 7.86 7.86 7.86				

DONDEL	D NETWORK ELEMENTS - Kentucky					T					Svo Order	Svo Order	Attachment: 2		Exhi	
EGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
													ist	Add I	DISC 1St	DISC Add I
						Rec	Nonre		Nonrecurring		0.0117.0			Rates(\$)		
L ocal S	witching Features Offered with Line Side Ports Only						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	All Features Available			UEPPX	UEPVF	0.00	0.00	0.00								
	ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE	s														
	Based Rates are applied where BellSouth is required by FCC an															
	res shall apply to the Unbundled Port/Loop Combination - Cost															
	Office and Tandem Switching Usage and Common Transport Us															
	rst and additional Port nonrecurring charges apply to Not Curre	ently Cor	nbined	Combos. For Curre	ntly Combine	ed Combos, the	nonrecurring c	narges shall be	those identified	I in the Nonrec	urring - Curr	ently Combi	ned sections.	Additional NR	Cs may apply	also and are
	zed accordingly.												-		-	
5. Mark	et Rates for Unbundled Centrex Port/Loop Combination will be CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only)	negotiat	ed on a	in Individual Case Ba	sis, until furt	her notice.										
	/G Loop/2-Wire Voice Grade Port (Centrex) Combo															
	rt/Loop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	Non-Design		1	UEP91		10.79										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		~													
	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		2	UEP91		15.52										
	2-wire VG Loop/2-wire voice Grade Port (Centrex)Port Combo - Non-Design		3	UEP91		31.74										
UNE Po	rt/Loop Combination Rates (Design)		5	02101		51.74										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -				1											
	Design		1	UEP91		13.82			1							
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design		2	UEP91		18.60										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		3	UEP91		34.37										
	Design op Rate		3	UEP91		34.37										
UNE LO	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP91	UECS1	9.64						7.86				
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP91	UECS1	14.37						7.86				
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP91	UECS1	30.59						7.86				
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP91	UECS2	12.67						7.86				
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP91	UECS2	17.45						7.86				
	2-Wire Voice Grade Loop (SL 2) - Zone 3	ļ	3	UEP91	UECS2	33.22						7.86				
UNE Po	rts es (Except North Carolina and Sout Carolina)	-														
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP91	UEPYA	1.15	21.29	15.49	2.85	2.67		7.86		1		
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local						2.1.20		2.50	2.57						
	Area			UEP91	UEPYB	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local															
	Area			UEP91	UEPYH	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 Basic Local Area			UEP91	UEPYM	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service	-		06191	UCPTIVI	1.15	21.29	15.49	2.85	2.67		00.1				
	Term - Basic Local Area			UEP91	UEPYZ	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent -						220									
	Basic Local Area			UEP91	UEPY9	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port Terminated on 800 Service Term - Basic			115504			a		0	a						
	Local Area			UEP91	UEPY2	1.15	21.29	15.49	2.85	2.67		7.86	L		L	
AL, KY,	LA, MS, & TN Only 2-Wire Voice Grade Port (Centrex)			UEP91	UEPQA	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex 800 termination)	<u> </u>		UEP91	UEPQB	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex with Caller ID)1	1		UEP91	UEPQH	1.15	21.29	15.49	2.85	2.67		7.86		İ		
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP91	UEPQM	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service						o		o o-	a (=		= ==				
	Term	ļ		UEP91	UEPQZ	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP91	UEPQ9	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent 2-Wire Voice Grade Port Terminated on 800 Service Term			UEP91	UEPQ9 UEPQ2	1.15	21.29	15.49	2.85	2.67		7.86				
	witching					1.13	21.23	10.49	2.00	2.01		1.00				
	Centrex Intercom Funtionality, per port			UEP91	URECS	0.8873						7.86		<u> </u>		
	umber Portability															
	Local Number Portability (1 per port)			UEP91	LNPCC	0.35										
Feature																

BUNDLE	ED NETWORK ELEMENTS - Kentucky										-	-	Attachment: 2			bit: B	_
EGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l	
																	_
						Rec	Nonrec		Nonrecurring		0.0115.0	001111		Rates(\$)	001111	SOMAN	—
	All Select Features Offered, per port			UEP91	UEPVS	0.00	First 405.66	Add'l	First	Add'l	SOMEC	SOMAN 7.86	SOMAN	SOMAN	SOMAN	SOMAN	├──
	All Centrex Control Features Offered, per port			UEP91	UEPVC	0.00	403.00					7.86					<u> </u>
NARS																	1
	Unbundled Network Access Register - Combination			UEP91	UARCX	0.00	0.00	0.00				7.86					
	Unbundled Network Access Register - Indial			UEP91	UAR1X	0.00	0.00	0.00				7.86					I
M' I	Unbundled Network Access Register - Outdial	_		UEP91	UAROX	0.00	0.00	0.00				7.86					—
	laneous Terminations Trunk Side	-				-											<u> </u>
2-1116	Trunk Side Terminations, each			UEP91	CENA6	10.51	92.18	15.82	52.16	5.30		7.86					<u> </u>
Interof	fice Channel Mileage - 2-Wire			02.01	021010	10.01	02.10	10.02	02.10	0.00		1.00					<u> </u>
	Interoffice Channel Facilities Termination - Voice Grade			UEP91	M1GBC	29.11						7.86					
	Interoffice Channel mileage, per mile or fraction of mile			UEP91	M1GBM	0.01						7.86					<u> </u>
	e Activations (DS0) Centrex Loops on Channelized DS1 Service	<u> </u>											L				
U4 Ch	annel Bank Feature Activations Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP91	1PQWS	0.62						7.86					┣──
		1		01 31	11 00/10	0.02						1.00					<u> </u>
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP91	1PQW6	0.62						7.86					┣—
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP91	1PQW7	0.62						7.86					L
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center				1PQWP	0.62						7.00					l
-		-		UEP91	TPQWP	0.62						7.86					⊢
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP91	1PQWV	0.62						7.86					
	Feature Activation on D-4 Channel Bank Tije Line/Trunk Loop Slot			UEP91	1PQWQ	0.62						7.86					ł
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP91	1PQWA	0.62						7.86					
Non-R	ecurring Charges (NRC) Associated with UNE-P Centrex																
	Conversion - Currently Combined Switch-As-Is with allowed changes, per port			UEP91	USAC2		0.102	0.102				7.86					
	Conversion of Existing Centrex Common Block			UEP91	USACN		18.95	8.32									
	New Centrex Standard Common Block			UEP91	M1ACS	0.00	669.80	78.32	111.05	13.27		7.86					
	New Centrex Customized Common Block			UEP91	M1ACC	0.00	669.80	78.32	111.05	13.27		7.86					└──
	Secondary Block, per Block NAR Establishment Charge, Per Occasion			UEP91 UEP91	M2CC1 URECA	0.00	78.32 72.75	78.32	13.27	13.27		7.86 7.86					⊢
UNE-P	P CENTREX - 5ESS (Valid in All States)			DEF91	URECA	0.00	12.15					7.00					<u> </u>
	VG Loop/2-Wire Voice Grade Port (Centrex) Combo																1
UNE P	ort/Loop Combination Rates (Non-Design)																
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design		1	UEP95		10.79											
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		2	UEP95		15.52											
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		3	UEP95		31.74	7										
UNE P	ort/Loop Combination Rates (Design)																
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	·															1
_	Design	<u> </u>	1	UEP95		13.82					<u> </u>	<u> </u>					
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		2	UEP95		18.60											
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		3	UEP95		34.37											Ĺ
UNE L	oop Rate																Ļ
_	2-Wire Voice Grade Loop (SL 1) - Zone 1	<u> </u>	1	UEP95	UECS1	9.64					<u> </u>	7.86					
-	2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3	+	2	UEP95 UEP95	UECS1 UECS1	14.37 30.59			ł			7.86 7.86					<u> </u>
	2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1	1	1	UEP95	UECS2	12.67						7.86					-
	2-Wire Voice Grade Loop (SL 2) - Zone 2	1	2	UEP95	UECS2	17.45				İ		7.86					
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP95	UECS2	33.22			<u> </u>			7.86					
	ort Rate																Ē
All Sta		<u> </u>		UE DOS		4.15	01.00	45.10	0.05	0.07		7.00					_
	2-Wire Voice Grade Port (Centrex) Basic Local Area 2-Wire Voice Grade Port (Centrex 800 termination)			UEP95 UEP95	UEPYA UEPYB	1.15	21.29 21.29	15.49 15.49	2.85 2.85	2.67 2.67		7.86 7.86					
_	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local	1		01 30	JEFID	1.15	21.29	15.49	2.00	2.07		1.00					<u> </u>
	Area			UEP95	UEPYH	1.15	21.29	15.49	2.85	2.67		7.86				L	_

	D NETWORK ELEMENTS - Kentucky	r	1								Suo Orden	Svo Order	Attachment: 2			bit: B	+
GORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs.	Order vs.	;
										_			Electronic- 1st	Electronic- Add'l	Electronic- Disc 1st	Electronic- Disc Add'l	
						Rec	Nonrec First	urring Add'l	Nonrecurring First	Add'l	SOMEC	SOMAN	SOMAN	Rates(\$) SOMAN	SOMAN	SOMAN	╇
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2						First	Add I	First	Add I	SOMEC	SUMAN	SUMAN	SOWAN	SOWAN	SOWAN	+
	Basic Local Area			UEP95	UEPYM	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			02.00	021111		21.20	10.10	2.00	2.07		1.00					+
	Term - Basic Local Area			UEP95	UEPYZ	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Grade Port terminated in on Megalink or equivalent -																Т
	Basic Local Area			UEP95	UEPY9	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Grade Port Terminated on 800 Service Term - Basic																T
	Local Area			UEP95	UEPY2	1.15	21.29	15.49	2.85	2.67		7.86					_
AL, KY,	LA, MS, SC, & TN Only			115005			04.00	15.10	0.05	0.07		7.00					4
	2-Wire Voice Grade Port (Centrex) 2-Wire Voice Grade Port (Centrex 800 termination)	-		UEP95	UEPQA UEPQB	1.15	21.29	15.49 15.49	2.85	2.67		7.86					+
	2-Wire Voice Grade Port (Centrex 800 termination) 2-Wire Voice Grade Port (Centrex with Caller ID)1		<u> </u>	UEP95 UEP95	UEPQB	1.15 1.15	21.29 21.29	15.49 15.49	2.85 2.85	2.67		7.86 7.86					+
				01190	JEFQH	1.15	21.29	15.49	2.00	2.07		1.00					+
1	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP95	UEPQM	1.15	21.29	15.49	2.85	2.67		7.86			1	1	1
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service	1					0		2.00						1	1	t
	Term			UEP95	UEPQZ	1.15	21.29	15.49	2.85	2.67		7.86					
1																	Т
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP95	UEPQ9	1.15	21.29	15.49	2.85	2.67		7.86					
<u> </u>	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP95	UEPQ2	1.15	21.29	15.49	2.85	2.67		7.86					4
	witching				110500	0.0070						7.00					+
	Centrex Intercom Funtionality, per port			UEP95	URECS	0.8873						7.86					+
	Local Number Portability (1 per port)			UEP95	LNPCC	0.35											+
Feature				UEP95	LINPCC	0.35				-							+
	All Standard Features Offered, per port			UEP95	UEPVF	0.00						7.86					+
	All Select Features Offered, per port			UEP95	UEPVS	0.00	405.66					7.86					+
	All Centrex Control Features Offered, per port			UEP95	UEPVC	0.00	100.00					7.86					Ŧ
NARS																	T
	Unbundled Network Access Register - Combination			UEP95	UARCX	0.00	0.00	0.00				7.86					T
	Unbundled Network Access Register - Indial			UEP95	UAR1X	0.00	0.00	0.00				7.86					
	Unbundled Network Access Register - Outdial			UEP95	UAROX	0.00	0.00	0.00				7.86					+
	neous Terminations																+
	T runk Side Trunk Side Terminations, each			UEP95	CEND6	10.51	92.18	15.82	52.16	5.30		7.86					╇
	Digital (1.544 Megabits)			UEP95	CENDO	10.51	92.10	15.62	52.10	5.30		7.00					+
	DS1 Circuit Terminations, each			UEP95	M1HD1	74.77	164.86	77.74	60.69	3.86		7.86					+
	DS0 Channels Activated, each			UEP95	M1HDO	0.00	15.09					7.86					Ť
Interoff	ce Channel Mileage - 2-Wire																Т
	Interoffice Channel Facilities Termination			UEP95	MIGBC	29.11						7.86					Ι
	Interoffice Channel mileage, per mile or fraction of mile			UEP95	MIGBM	0.01						7.86					T
	Activations (DS0) Centrex Loops on Channelized DS1 Service				+							<u> </u>					+
	nnel Bank Feature Activations	-		UEP95	1PQWS	0.62						7.00					+
<u> </u>	Feature Activation on D-4 Channel Bank Centrex Loop Slot			066,90	IPQVI3	0.62						7.86					+
'	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP95	1PQW6	0.62						7.86			1	1	1
	- catale ristration on D + onamici Dank i A line olde Loop Slot	<u> </u>		02100	11 52/10	0.02						1.00					+
'	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP95	1PQW7	0.62						7.86					
1	Feature Activation on D-4 Channel Bank Centrex Loop Slot -	1													1	1	T
	Different Wire Center			UEP95	1PQWP	0.62						7.86					
																	Γ
ļ'	Feature Activation on D-4 Channel Bank Private Line Loop Slot	ļ		UEP95	1PQWV	0.62						7.86					4
'				LIEDOE	100110							=		1			
'	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot	-	-	UEP95	1PQWQ	0.62						7.86			l	l	+
Non-Pr	Feature Activation on D-4 Channel Bank WATS Loop Slot curring Charges (NRC) Associated with UNE-P Centrex			UEP95	1PQWA	0.62						7.86					+
Non-Ke	NRC Conversion Currently Combined Switch-As-Is with allowed				1										1		+
'	changes, per port			UEP95	USAC2		0.102	0.102				7.86			1	1	
	Conversion of Existing Centrex Common Block, each			UEP95	USACN		18.95	8.32				7.86	1		1	1	$^{+}$
	New Centrex Standard Common Block	1		UEP95	MIACS	0.00	669.80	78.32	111.05	13.27		7.86	l	İ	1	1	t
	New Centrex Customized Common Block			UEP95	M1ACC	0.00	669.80	78.32	111.05	13.27		7.86					t
	NAR Establishment Charge, Per Occasion			UEP95	URECA	0.00	72.75					7.86					T
	CENTREX - DMS100 (Valid in All States)	1	1														Т
	/G Loop/2-Wire Voice Grade Port (Centrex) Combo																

INBUNDLE	D NETWORK ELEMENTS - Kentucky										-		Attachment: 2		Exhi	pit: B	<u> </u>
TEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I	
						Rec	Nonred		Nonrecurring					Rates(\$)			
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	<u> </u>
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design		1	UEP9D		10.79											l
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		2	UEP9D		15.52											
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		3	UEP9D		31.74											
UNE P	ort/Loop Combination Rates (Design)		Ŭ	OEI 3D		01.74											·
0.121	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -																
	Design		1	UEP9D	_	13.82											<u> </u>
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		2	UEP9D		18.60											
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		3	UEP9D		34.37											l
UNE L	pop Rate	1			1	0			1		1						
	2-Wire Voice Grade Loop (SL 1) - Zone 1	1	1	UEP9D	UECS1	9.64	İ		1			7.86					
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9D	UECS1	14.37						7.86					
	2-Wire Voice Grade Loop (SL 1) - Zone 3	1	3	UEP9D	UECS1	30.59						7.86					
	2-Wire Voice Grade Loop (SL 2) - Zone 1	1	1	UEP9D	UECS2	12.67						7.86					
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9D	UECS2	17.45						7.86					
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9D	UECS2	33.22						7.86					
UNE P	ort Rate																
ALL S	ATES																
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9D	UEPYA	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area			UEP9D	UEPYB	1.15	21.29	15.49	2.85	2.67		7.86					1
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3Basic Local Area			UEP9D	UEPYC	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3Basic Local Area			UEP9D	UEPYD	1.15	21.29	15.49	2.85	2.67		7.86					 !
	2-Wire Voice Grade Port (Centrex / EBS-M5209))3 Basic Local Area			UEP9D	UEPYE	1.15	21.29	15.49	2.85	2.67		7.86					
-	2-Wire Voice Grade Port (Centrex / EBS-M5112))3 Basic Local Area			UEP9D	UEPYF	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Grade Port (Centrex / EBS-M5312))3Basic Local Area			UEP9D	UEPYG	1.15		15.49	2.85	2.67		7.86					1
	2-Wire Voice Grade Port (Centrex / EBS-M5008))3 Basic Local						21.29										1
	Area 2-Wire Voice Grade Port (Centrex / EBS-M5208))3 Basic Local			UEP9D	UEPYT	1.15	21.29	15.49	2.85	2.67		7.86					
-	Area 2-Wire Voice Grade Port (Centrex / EBS-M5216))3 Basic Local			UEP9D	UEPYU	1.15	21.29	15.49	2.85	2.67		7.86					
_	Area			UEP9D	UEPYV	1.15	21.29	15.49	2.85	2.67		7.86					┣—
	2-Wire Voice Grade Port (Centrex / EBS-M5316))3 Basic Local Area			UEP9D	UEPY3	1.15	21.29	15.49	2.85	2.67		7.86					J
	2-Wire Voice Grade Port (Centrex with Caller ID) Basic Local Area			UEP9D	UEPYH	1.15	21.29	15.49	2.85	2.67		7.86					1
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp Indication))3 Basic Local Area			UEP9D	UEPYW	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication))3 Basic Local Area			UEP9D	UEPYJ	1.15	21.29	15.49	2.85	2.67		7.86					_
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center) 2 Basic Local Area	2		UEP9D	UEPYM	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3 Basic Local Area			UEP9D	UEPYO	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3 Basic Local Area			UEP9D	UEPYP	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3 Basic Local Area			UEP9D	UEPYQ	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3 Basic Local Area			UEP9D	UEPYR	1.15	21.29	15.49	2.85	2.67		7.86					1
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3 Basic Local Area			UEP9D	UEPYS	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3 Basic Local Area			UEP9D	UEPY4	1.15	21.29	15.49	2.85	2.67		7.86					

DONDLE	D NETWORK ELEMENTS - Kentucky	-									0	0	Attachment: 2		Exhi		+
EGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	
													1st	Add'l	Disc 1st	Disc Add'l	
						Rec	Nonrec		Nonrecurring					Rates(\$)			T
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	╄
	Basic Local Area			UEP9D	UEPY5	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3																Τ
_	Basic Local Area 2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			UEP9D	UEPY6	1.15	21.29	15.49	2.85	2.67		7.86					+
	Basic Local Area			UEP9D	UEPY7	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP9D	UEPYZ	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			OLI 3D					2.00			7.00					t
	Basic Local Area			UEP9D	UEPY9	1.15	21.29	15.49	2.85	2.67		7.86					∔
	2-Wire Voice Grade Port Terminated on 800 Service Term Basic Local Area			UEP9D	UEPY2	1.15	21.29	15.49	2.85	2.67		7.86					1
AL, KY	LA, MS, SC, & TN Only																t
	2-Wire Voice Grade Port (Centrex)			UEP9D	UEPQA	1.15	21.29	15.49	2.85	2.67		7.86					Ŧ
-	2-Wire Voice Grade Port (Centrex 800 termination) 2-Wire Voice Grade Port (Centrex / EBS-PSET)3			UEP9D UEP9D	UEPQB UEPQC	1.15 1.15	21.29 21.29	15.49 15.49	2.85	2.67		7.86					╀
-	2-Wire Voice Grade Port (Centrex / EBS-PSET)3 2-Wire Voice Grade Port (Centrex / EBS-M5009)3			UEP9D UEP9D	UEPQC	1.15	21.29	15.49 15.49	2.85	2.67		7.86					+
1	2-Wire Voice Grade Port (Centrex / EBS-M5009)3			UEP9D	UEPQE	1.15	21.29	15.49	2.85	2.67		7.86	1				t
	2-Wire Voice Grade Port (Centrex / EBS-M5112)3	1		UEP9D	UEPQF	1.15	21.29	15.49	2.85			7.86	l				t
	2-Wire Voice Grade Port (Centrex / EBS-M5312)3			UEP9D	UEPQG	1.15	21.29	15.49	2.85	2.67		7.86					T
	2-Wire Voice Grade Port (Centrex / EBS-M5008)3			UEP9D	UEPQT	1.15	21.29	15.49	2.85			7.86					Γ
	2-Wire Voice Grade Port (Centrex / EBS-M5208)3			UEP9D	UEPQU	1.15	21.29	15.49	2.85	2.67		7.86					_
	2-Wire Voice Grade Port (Centrex / EBS-M5216)3			UEP9D	UEPQV	1.15	21.29	15.49	2.85	2.67		7.86					╇
-	2-Wire Voice Grade Port (Centrex / EBS-M5316)3			UEP9D UEP9D	UEPQ3	1.15 1.15	21.29	15.49 15.49	2.85	2.67		7.86 7.86					+
+	2-Wire Voice Grade Port (Centrex with Caller ID) 2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp			UEPSD	UEPQH	1.15	21.29	15.49	2.85	2.67		7.86					+
	Indication)3			UEP9D	UEPQW	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication)3			UEP9D	UEPQJ	1.15	21.29	15.49	2.85	2.67		7.86					Ť
																	Т
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center) 2			UEP9D	UEPQM	1.15	21.29	15.49	2.85	2.67		7.86					+
_	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			UEP9D	UEPQO	1.15	21.29	15.49	2.85	2.67		7.86					+
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3			UEP9D	UEPQP	1.15	21.29	15.49	2.85	2.67		7.86					
-	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M0003/2, 3			UEP9D	UEPQQ	1.15	21.29	15.49	2.85	2.67		7.86					+
				OEI 3D	OLI QQ	1.10	21.25	10.40	2.00	2.07		1.00					t
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			UEP9D	UEPQR	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3			UEP9D	UEPQS	1.15	21.29	15.49	2.85	2.67		7.86					
																	T
+	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3			UEP9D	UEPQ4	1.15	21.29	15.49	2.85	2.67		7.86		1	1		+
—	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3			UEP9D	UEPQ5	1.15	21.29	15.49	2.85	2.67		7.86					╀
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3			UEP9D	UEPQ6	1.15	21.29	15.49	2.85	2.67		7.86					L
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			UEP9D	UEPQ7	1.15	21.29	15.49	2.85	2.67		7.86					ľ
1	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service																t
	Term			UEP9D	UEPQZ	1.15	21.29	15.49	2.85	2.67		7.86					╀
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9D	UEPQ9	1.15	21.29	15.49	2.85	2.67		7.86					1
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP9D	UEPQ2	1.15	21.29	15.49	2.85	2.67		7.86					T
Local S	witching																L
-	Centrex Intercom Funtionality, per port			UEP9D	URECS	0.8873						7.86					╀
Local N	lumber Portability			UEP9D	LNPCC	0.35											+
Feature	Local Number Portability (1 per port)				LINECC	0.35											+
. such	All Standard Features Offered, per port			UEP9D	UEPVF	0.00				1		7.86	1				t
	All Select Features Offered, per port	1		UEP9D	UEPVS	0.00	405.66			ĺ		7.86	l				t
	All Centrex Control Features Offered, per port			UEP9D	UEPVC	0.00						7.86					T
NARS																	Ţ
	Unbundled Network Access Register - Combination			UEP9D	UARCX	0.00	0.00	0.00				7.86					+
-	Unbundled Network Access Register - Inward Unbundled Network Access Register - Outdial			UEP9D UEP9D	UAR1X UAROX	0.00	0.00	0.00				7.86 7.86					╀
	Unbundled Network Access Register - Outdial aneous Terminations			UEP9D	UARUX	0.00	0.00	0.00			-	1.86					╀

BUNDLE	D NETWORK ELEMENTS - Kentucky					1						.	Attachment: 2			bit: B
EGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
					-	Rec	Nonrec		Nonrecurring		0.0115.0	001111		Rates(\$)	001111	00000
O Mire	Frunk Side						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Trunk Side Terminations, each			UEP9D	CEND6	10.51	92.18	15.82	52.16	5.30		7.86				
	Digital (1.544 Megabits)				CLINDO	10.51	32.10	10.02	52.10	5.50		7.00				
4 1110	DS1 Circuit Terminations, each			UEP9D	M1HD1	74.77	164.86	77.74	60.69	3.86		7.86				
	DS0 Channels Activiated per Channel			UEP9D	M1HDO	0.00	15.09		00.00	0.00		7.86				
Interoff	ce Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination			UEP9D	MIGBC	29.11						7.86				
	Interoffice Channel mileage, per mile or fraction of mile			UEP9D	MIGBM	0.01						7.86				
	Activations (DS0) Centrex Loops on Channelized DS1 Service															
D4 Cha	nnel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9D	1PQWS	0.62						7.86				
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9D	1PQW6	0.62						7.86				
_	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot	-		UEP9D	1PQW7	0.62						7.86				
_	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP9D	1PQWP	0.62						7.86				
_	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9D	1PQWV	0.62						7.86				
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot			UEP9D	1PQWQ	0.62					1	7.86				
	Feature Activation on D-4 Channel Bank Tile Line/Trunk Loop Slot			UEP9D UEP9D	1PQWQ 1PQWA	0.62						7.86				
Non-Re	curring Charges (NRC) Associated with UNE-P Centrex			UEF9D	IFQWA	0.02						7.00				
NON-INC	NRC Conversion Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP9D	USAC2		0.102	0.102				7.86				
	Conversion of existing Centrex Common Block, each			UEP9D	USACN		18.95	8.32				7.86				
	New Centrex Standard Common Block			UEP9D	M1ACS	0.00	669.80	78.32	111.05	13.27		7.86				
	New Centrex Customized Common Block			UEP9D	M1ACC	0.00	669.80	78.32	111.05	13.27		7.86				
	NAR Establishment Charge, Per Occasion			UEP9D	URECA	0.00	72.75					7.86				
	CENTREX - EWSD (Valid in AL, FL, KY, LA, MS & TN)															
	/G Loop/2-Wire Voice Grade Port (Centrex) Combo															
UNE PO	ort/Loop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design		1	UEP9E		10.79										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		2	UEP9E		15.52										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		3	UEP9E		31.74										
	ort/Loop Combination Rates (Design)		3	UEP9E	-	31.74									-	-
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	Desian		1	UEP9E		13.82					1					
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -				1											
	Design		2	UEP9E		18.60										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		3	UEP9E		34.37										
UNE Lo	op Rate															
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9E	UECS1	9.64						7.86				
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9E	UECS1	14.37						7.86				
	2-Wire Voice Grade Loop (SL 1) - Zone 3	1	3	UEP9E	UECS1	30.59						7.86				
	2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 2	+	1	UEP9E UEP9E	UECS2 UECS2	12.67 17.45						7.86 7.86				
-	2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3	+	2	UEP9E	UECS2	33.22	├					7.86			-	-
UNE P			5	01.31	01002	33.22						1.00				
	KY, LA, MS, & TN only				i	l				l						
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9E	UEPYA	1.15	21.29	15.49	2.85	2.67		7.86				
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area			UEP9E	UEPYB	1.15	21.29	15.49	2.85	2.67		7.86				
1	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local				1											
	Area 2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP9E	UEPYH	1.15	21.29	15.49	2.85	2.67		7.86				
	Basic Local Area 2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			UEP9E	UEPYM	1.15	21.29	15.49	2.85	2.67		7.86				
	Term - Basic Local Area	1		UEP9E	UEPYZ	1.15	21.29	15.49	2.85	2.67	1	7.86				

BUNDLE	D NETWORK ELEMENTS - Kentucky											1	Attachment: 2		Exhi		⊢
EGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs.	
I											P	P	Electronic- 1st	Electronic- Add'l	Electronic- Disc 1st	Electronic- Disc Add'l	
						Rec	Nonreo First	urring Add'l	Nonrecurring First	Add'l	SOMEC	SOMAN	SOMAN	Rates(\$) SOMAN	SOMAN	SOMAN	┢
	2-Wire Voice Grade Port terminated in on Megalink or equivalent -						FIISL	Add I	FIISL	Add1	SOWEC	JOWAN	SOWAN	SOMAN	SOWAN	SOWAN	-
	Basic Local Area			UEP9E	UEPY9	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Grade Port Terminated on 800 Service Term - Basic																
	Local Area			UEP9E	UEPY2	1.15	21.29	15.49	2.85	2.67		7.86					
AL, KY	, LA, MS, & TN Only																_
	2-Wire Voice Grade Port (Centrex)	-		UEP9E	UEPQA	1.15 1.15	21.29	15.49	2.85	2.67		7.86					-
_	2-Wire Voice Grade Port (Centrex 800 termination) 2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP9E UEP9E	UEPQB UEPQH	1.15	21.29 21.29	15.49 15.49	2.85 2.85	2.67 2.67		7.86 7.86					+
-	2-Wile Voice Grade Fort (Centrex with Caller ID)1			UEF9E	UEFQH	1.15	21.29	15.49	2.00	2.07		7.00					+
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP9E	UEPQM	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			OEI SE	OEI GIN	1.10	21.25	10.40	2.00	2.07		1.00					1
	Term			UEP9E	UEPQZ	1.15	21.29	15.49	2.85	2.67		7.86					1
1		1		-													T
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9E	UEPQ9	1.15	21.29	15.49	2.85	2.67		7.86					L
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP9E	UEPQ2	1.15	21.29	15.49	2.85	2.67		7.86					Γ
Local S	Switching	<u> </u>															L
-	Centrex Intercom Funtionality, per port	I		UEP9E	URECS	0.8873						7.86					Ļ
Local	lumber Portability	<u> </u>			INDCC						L	=			l	l	⊢
Fration	Local Number Portability (1 per port)	-		UEP9E	LNPCC	0.35						7.86					+
Featur				UEP9E	UEPVF	0.00						7.86					┢
	All Standard Features Offered, per port All Select Features Offered, per port	-		UEP9E	UEPVF	0.00	405.66				-	7.86					╈
_	All Centrex Control Features Offered, per port			UEP9E	UEPVC	0.00	405.00					7.86					╈
NARS	All Centrex Control realtales Onered, per port			OLI 3L	ULI VC	0.00						1.00					⊢
NAILO	Unbundled Network Access Register - Combination			UEP9E	UARCX	0.00	0.00	0.00			-						t
	Unbundled Network Access Register - Indial	-		UEP9E	UAR1X	0.00	0.00	0.00									t
	Unbundled Network Access Register - Outdial			UEP9E	UAROX	0.00	0.00	0.00									T
Miscell	aneous Terminations																T
2-Wire	Trunk Side																
	Trunk Side Terminations, each			UEP9E	CEND6	10.51	92.18	15.82	52.16	5.30		7.86					
4-Wire	Digital (1.544 Megabits)																_
	DS1 Circuit Terminations, each			UEP9E	M1HD1	74.77	164.86	77.74	60.69	3.86		7.86					_
h	DS0 Channel Activated Per Channel			UEP9E	M1HDO	0.00	15.09					7.86					+
interor	ice Channel Mileage - 2-Wire Interoffice Channel Facilities Termination	-		UEP9E	MIGBC	29.11						7.86					┢
_	Interoffice Channel mileage, per mile or fraction of mile	-		UEP9E UEP9E	MIGBC	0.01					-	7.86					┢
Featur	Activations (DS0) Centrex Loops on Channelized DS1 Service			OLI SL	WIGDW	0.01						7.00					⊢
	annel Bank Feature Activations				1												t
	Feature Activation on D-4 Channel Bank Centrex Loop Slot	1		UEP9E	1PQWS	0.62						7.86					t
																	T
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9E	1PQW6	0.62						7.86					
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP9E	1PQW7	0.62						7.86					L
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -	1			100110												
_	Different Wire Center	<u> </u>		UEP9E	1PQWP	0.62					L	7.86			l	l	⊢
	Feature Activation on D-4 Channel Bank Private Line Loop Slot	1		UEP9E	1PQWV	0.62						7.86					1
+	reature Activation on D-4 Channel Bank Private Line Loop Slot			DELAE	PQVV	0.62						1.80					⊢
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot	1		UEP9E	1PQWQ	0.62						7.86					1
	Feature Activation on D-4 Channel Bank WATS Loop Slot	1		UEP9E	1PQWA	0.62						7.86					t
Non-R	ecurring Charges (NRC) Associated with UNE-P Centrex	1								1							t
1	NRC Conversion Currently Combined Switch-As-Is with allowed	1															Г
	changes, per port			UEP9E	USAC2		0.102	0.102				7.86					
	Conversion of Existing Centrex Common Block, each			UEP9E	USACN		18.95	8.32									
	New Centrex Standard Common Block	<u> </u>		UEP9E	M1ACS	0.00	669.80	78.32	111.05	13.27		7.86					L
_	New Centrex Customized Common Block	I		UEP9E	M1ACC	0.00	669.80	78.32	111.05	13.27		7.86					1
	NAR Establishment Charge, Per Occasion	<u> </u>		UEP9E	URECA	0.00	72.75				L	7.86			l	l	⊢
	CENTREX - DCO - Valid in AL, KY, LA, MS, & TN)																╀
	VG Loop/2-Wire Voice Grade Port (Centrex) Combo ort/Loop Combination Rates (Non-Design)																┢
UNE P	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	1		-	+	ł											⊢
	Non-Design	1	1	UEP93		10.79						1					1
1	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	1			1	10.73					1						t
1	Non-Design	1	2	UEP93	1	15.52				1	1	1	1				1

BUNDLE	D NETWORK ELEMENTS - Kentucky	1	r									-	Attachment: 2			bit: B	╞
													Incremental	Incremental	Incremental		1
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -	
EGORY	RATE ELEMENTS	Interim	Zono	BCS	USOC			RATES(\$)			Elec	Manually	Manual Svc	Manual Svc	Manual Svc		
EGORT	RATE ELEMENTS	interim	Zone	BCS	0300			KAIES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.	
													Electronic-	Electronic-	Electronic-	Electronic-	
													1st	Add'l	Disc 1st	Disc Add'l	
						Rec	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)			\mathbf{T}
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		3	UEP93		31.74											
	Non-Design ort/Loop Combination Rates (Design)		3	UEP93	-	31.74											+
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -																+
	Design		1	UEP93		13.82											
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -																
_	Design		2	UEP93	-	18.60										-	╇
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		3	UEP93		34.37											
UNEL	pop Rate		3	UEF93		34.37											+
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP93	UECS1	9.64											+
	2-Wire Voice Grade Loop (SL 1) - Zone 2			UEP93	UECS1	14.37											1
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP93	UECS1	30.59											Γ
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP93	UECS2	12.67											
	2-Wire Voice Grade Loop (SL 2) - Zone 2			UEP93	UECS2	17.45											Г
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP93	UECS2	33.22											+
	ort Rate				-											-	_
AL, KY	, LA, MS, & TN only 2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP93	UEPYA	1.15	21.29	15.49	2.85	2.67		7.86					+
-	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local			UEF 93	UEFTA	1.15	21.29	15.49	2.00	2.07		7.00					+
	Area			UEP93	UEPYB	1.15	21.29	15.49	2.85	2.67		7.86					1
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local																T
	Area			UEP93	UEPYH	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2																
	Basic Local Area			UEP93	UEPYM	1.15	21.29	15.49	2.85	2.67		7.86					+
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term - Basic Local Area			UEP93	UEPYZ	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Grade Port terminated in on Megalink or equivalent -			UEP93	UEPTZ	1.15	21.29	15.49	2.65	2.07		7.00		-		-	┿
	Basic Local Area			UEP93	UEPY9	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Grade Port Terminated on 800 Service Term - Basic			021.00	ULI IU		21120	10.10	2.00	2.01		1.00					t
	Local Area			UEP93	UEPY2	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Grade Port (Centrex)			UEP93	UEPQA	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP93	UEPQB	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP93	UEPQH	1.15	21.29	15.49	2.85	2.67		7.86					1
	2 Wire Vales Crade Dart (Centrey from diff Serving Wire Center)2			UEP93	UEPQM	1.15	21.29	15.49	2.85	2.67		7.00					
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			UEP93	UEPQIM	1.15	21.29	15.49	2.65	2.07		7.86		-		-	+
	Term			UEP93	UEPQZ	1.15	21.29	15.49	2.85	2.67		7.86					
																	t
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP93	UEPQ9	1.15	21.29	15.49	2.85	2.67		7.86					
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP93	UEPQ2	1.15	21.29	15.49	2.85	2.67		7.86					+
Local S	witching Centrex Intercom Funtionality, per port			UEP93	URECS	0.8873						7.86					+
	lumber Portability			UEP93	URECS	0.0073						1.00					+
LUCAIN	Local Number Portability (1 per port)			UEP93	LNCCC	0.35											+
Feature				011 33	LINCCC	0.55											+
	All Standard Features Offered, per port			UEP93	UEPVF	0.00			1			7.86				<u> </u>	t
	All Centrex Control Features Offered, per port			UEP93	UEPVC	0.00						7.86					ſ
NARS																	Γ
	Unbundled Network Access Register - Combination			UEP93	UARCX	0.00	0.00	0.00									Ĩ
	Unbundled Network Access Register - Indial			UEP93	UAR1X	0.00	0.00	0.00									1
Miner	Unbundled Network Access Register - Outdial			UEP93	UAROX	0.00	0.00	0.00								l	╋
	aneous Terminations Trunk Side	-			+											<u> </u>	╋
2-wire	Trunk Side Terminations, each		<u> </u>	UEP93	CEND6	10.51	92.18	15.82	52.16	5.30		7.86				<u> </u>	+
4-Wire	Digital (1.544 Megabits)						02.10	10.02	02.10	0.00						<u> </u>	t
	DS1 Circuit Terminations, each			UEP93	M1HD1	74.77	164.86	77.74	60.69	3.86		7.86					T
	DS0 Channels Activated, Per Channel			UEP93	M1HDO	0.00	15.09					7.86					L
Interoff	ice Channel Mileage - 2-Wire				MIOFO							= 0-				<u> </u>	╇
-	Interoffice Channel Facilities Termination			UEP93 UEP93	MIGBC	29.11						7.86 7.86					╀
Feature	Interoffice Channel mileage, per mile or fraction of mile Activations (DS0) Centrex Loops on Channelized DS1 Service			02130	IVIGOIVI	0.01						1.00				1	+
	nnel Bank Feature Activations	1			1	1									t	<u> </u>	+
	Feature Activation on D-4 Channel Bank Centrex Loop Slot	1	1	UEP93	1PQWS	0.62						7.86				1	+

CHIGON RATE LEMENTS Res Res USCC Second 2 Second	JNBUNDLE	ED NETWORK ELEMENTS - Kentucky												Attachment: 2	2	Exhi	oit: B	
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$			Interim	Zone	BCS	usoc			RATES(\$)			Submitted Elec	Submitted Manually	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs.	
Image: constraint of the state of the st																Disc 1st	Disc Add'l	
Feature Activation on D-4 Channel Bank FX Tunk Side Loop Slot UEP93 1PQW7 0.62 7.86 Different Wire Center UEP93 1PQWP 0.62 7.86 1 Feature Activation on D-4 Channel Bank Centrex Loop Slot UEP93 1PQWP 0.62 7.86 1 Feature Activation on D-4 Channel Bank Private Line Loop Slot UEP93 1PQWV 0.62 7.86 1 Feature Activation on D-4 Channel Bank Tie Line/Trunk Loop Slot UEP93 1PQWQ 0.62 7.86 1 Non-Recurring Charges (NRC) Associated with UNE-P Centrex UEP93 1PQWA 0.62 7.86 1 Non-Recurring Charges (NRC) Associated with UNE-P Centrex UEP93 1PQWA 0.62 7.86 1 Non-Recurring Charges (NRC) Associated with UNE-P Centrex UEP93 USAC2 0.102 7.86 1 New Centrex Standard Common Block, each UEP93 USAC2 0.102 7.86 1 1 New Centrex Standard Common Block UEP93 W1ACS 0.00 68.90 78.32 111.05 13.27 7.86 1 New Centrex Standard Common Block UEP93 W1ACS							Rec					SOMEC	SOMAN			SOMAN	SOMAN	
Feature Activation on D-4 Channel Bank Centrex Loop Slot - UEP93 1PQWP 0.62 7.86 Image: Contract State Control in AESS, 5ESS & EWSD UEP93 1PQWP 0.62 7.86 Image: Contract State Control in AESS, 5ESS & EWSD UEP93 1PQWQ 0.62 7.86		Feature Activation on D-4 Channel Bank FX Line Side Loop Slot			UEP93	1PQW6	0.62						7.86					
Different Wire CenterUEP33IPQWP0.62OO7.86OOOFeature Activation on D-4 Channel Bank Private Line Loop SlotUEP931PQWQ0.62OSSS					UEP93	1PQW7	0.62						7.86					
Feature Activation on D-4 Channel Bank Tie Line/Trunk Loop Slot UEP93 1PQWQ 0.62 0 0 7.86 0 0 0 Feature Activation on D-4 Channel Bank WATS Loop Slot UEP93 1PQWA 0.62 0 0 7.86 0 0 0 Non-Recurring Charges (NRC) Associated with UNE-P Centrex UEP93 1PQWA 0.62 0		Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP93	1PQWP	0.62						7.86					
Feature Activation on D-4 Channel Bank WATS Loop Slot UEP93 1PQWA 0.62 0 7.86 0 0 Non-Recurring Charges (NRC) Associated with URE-P Centrex 0 <td></td> <td>Feature Activation on D-4 Channel Bank Private Line Loop Slot</td> <td></td> <td></td> <td>UEP93</td> <td>1PQWV</td> <td>0.62</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>7.86</td> <td></td> <td></td> <td></td> <td></td> <td></td>		Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP93	1PQWV	0.62						7.86					
Non-Recurring Charges (NRC) Associated with UNE-P Centrex Image: Second Currently Combined Switch-As-Is with allowed changes, per port Image: Second Currently Combined Switch-As-Is with allowed UEP93 USAC2 0.102 0.102 0.102 7.86 Image: Second Currently Combined Switch-As-Is with allowed UEP93 USAC2 0.102 0.102 0.102 7.86 Image: Second Currently Combined Switch-As-Is with allowed UEP93 USACN 18.95 8.32 Image: Second Currently Combined Switch-As-Is with allowed UEP93 Image: Second Currently Combined Switch-As-Is with allowed UEP93 Image: Second Currently Combined Switch-As-Is with allowed UEP93 Image: Second Currently Combined Switch-As-Is with allowed UEP93 Image: Second Currently Combined Switch-As-Is with allowed UEP93 Image: Second Currently Combined Switch-As-Is with allowed UEP93 Image: Second Currently Combined Switch-As-Is with allowed UEP93 Image: Second Currently Combined Switch-As-Is with allowed UEP93 Image: Second Currently Combined Switch-As-Is with allowed UEP93 Image: Second Current Status Switch As-Is with allowed UEP93 Image: Second Current Status Switch As-Is with allowed UEP93 Image: Second Current Status Switch As-Is with allowed UEP93 Image: Second Current Status Switch As-Is with allowed UEP93 Image: Second Current Status Switch As-Is with allowed UEP93 Image: Second Current Status Switch As-Is with allowed UEP93 Image: Second Current Status Switch As-Is with allowed UEP93 Image: Second Current Status Switch As-Is with allowed UEP93 Image: Secon																		
NRC Conversion Currently Combined Switch-As-Is with allowed changes, per port UEP93 USAC2 0.102 0.102 7.86 Conversion of Existing Centrex Common Block, each UEP93 USACN 18.95 8.32 7.86					UEP93	1PQWA	0.62						7.86					
changes, per port UEP93 USAC2 0.102 0.102 7.86 0 0 Conversion of Existing Centrex Common Block, each UEP93 USACN 18.95 8.32 7.86 0 0 New Centrex Standard Common Block UEP93 M1ACS 0.00 669.80 78.32 111.05 13.27 7.86 0 0 NAR Establishment Charge, Per Occasion UEP93 M1ACS 0.00 669.80 78.32 111.05 13.27 7.86 0 0 NAR Establishment Charge, Per Occasion UEP93 URECA 0.00 72.75 0 0 7.86 0 0 0 Note 1 - Required Port for Centrex Control in 1AESS, 5ESS & EWSD UEP3 URECA 0.00 72.75 0 <td>Non-R</td> <td>ecurring Charges (NRC) Associated with UNE-P Centrex</td> <td></td> <td><u> </u></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>└───</td>	Non-R	ecurring Charges (NRC) Associated with UNE-P Centrex		<u> </u>														└───
Conversion of Existing Centrex Common Block, each UEP93 USACN 18.95 8.32 7.86 New Centrex Standard Common Block UEP93 M1ACS 0.00 669.80 78.32 111.05 13.27 7.86 New Centrex Customized Common Block UEP93 M1ACC 0.00 669.80 78.32 111.05 13.27 7.86 <		NRC Conversion Currently Combined Switch-As-Is with allowed	1	1	115 000	110465		o	o				=					1
New Centrex Standard Common Block UEP93 M1ACS 0.00 669.80 78.32 111.05 13.27 7.86 0 0 New Centrex Customized Common Block UEP93 M1ACC 0.00 669.80 78.32 111.05 13.27 7.86 0 0 NAR Establishment Charge, Per Occasion UEP93 URECA 0.00 78.32 111.05 13.27 7.86 0 0 Note 1 - Required Port for Centrex Control in 1AESS, 5ESS & EWSD 72.75 0 <			+															
New Centrex Customized Common Block UEP93 M1ACC 0.00 669.80 78.32 111.05 13.27 7.86 Image: Common Block Image: Com			+				0.00			444.05	40.07					1		<u> </u>
NAR Establishment Charge, Per Occasion UEP93 URECA 0.00 72.75 7.86 Note 1 - Required Port for Centrex Control in 14ESS, 5ESS & EWSD Image:			+															
Note 1 - Required Port for Centrex Control in 1AESS, 5ESS & EWSD			+						10.32	111.05	13.27							
Note 2 - Requires Interoffice Channel Mileage	Note 1		1	<u> </u>	01 30	UNLOA	0.00	12.15					1.00					
Note 3 - Requires Specific Customer Premises Equipment																		
Note Rates displaying in "1" interim and tablet. Under the denominance of the denomina																		
Image: state	Note:	Rates displaying an "R" in Interim column are interim and subject	t to rate	true-up	as set forth in Gene	ral Terms and	d Conditions.											
1 1 <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>																		
Image by the sector Image by the sec																		
1 1 <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>																		
Image: Section of the section of t																		
Image: Problem Imade: Problem Image: Problem Image: Proble																		
Image: Sector																		
I I <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>																		
I I <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>																		
I I <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>																		
I I <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>																		
Image: sector																		
I I <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>-</td> <td>-</td> <td></td> <td></td> <td></td> <td></td> <td></td>												-	-					
1 1 <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>																		
I I <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>																		
1 1 <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>																		
- - <td></td> <td></td> <td>1</td> <td>1</td> <td>1</td> <td>1</td> <td></td> <td></td> <td></td> <td>1</td> <td>1</td> <td>1</td> <td>1</td> <td>l</td> <td></td> <td></td> <td></td> <td></td>			1	1	1	1				1	1	1	1	l				
- - <td></td> <td>İ.</td> <td>1</td> <td>1</td> <td></td> <td>1</td> <td></td> <td></td> <td></td> <td>l</td> <td>ĺ</td> <td></td> <td></td> <td>İ</td> <td></td> <td></td> <td></td> <td><u> </u></td>		İ.	1	1		1				l	ĺ			İ				<u> </u>
Image: state I																		
1 1 <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>																		
I I <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>L</td>																		L
Image: Problem intermediate			ļ	I														
Image: serie																		
Image: state in the state			+															ł
NNN						l												<u> </u>
III			+															<u> </u>
Image: Constraint of the sector of the sec			1															
III			1			1												
Image: space spac			1	1	1	1					1			1				
Image: state of the state		İ.	1	1				l										
Image: state in the state																		
Image: state in the state																		
Image: state of the state																		
Image: state of the state																		
- -																		L
			ļ	I														
			ļ	I														
												L	L					└───
					l	L	ł					L	L					└───
						+	ł											
		L	1	1		I	I					1	1	I				<u> </u>

UN	BUNDL	ED NETWORK ELEMENTS - Kentucky												Attachment: 2	2	Exhi	bit: B	
															Incremental		Incremental	
												Submitted			Charge -	Charge -	Charge -	
				_								Elec		Manual Svc	Manual Svc			
CAT	EGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.	
														Electronic-	Electronic-	Electronic-	Electronic-	
														1st	Add'l	Disc 1st	Disc Add'l	
							_	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)			<u> </u>
-							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	
																		
																		
																		
																		1

UNBUN	DLED	NETWORK ELEMENTS - South Carolina												Attachment:	2	Exhi	bit: B
												Svc Order	Svc Order	Incremental			
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svo
CATEGOR	RY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													-	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
								Nonrec		Nonrecurring	Disconnect		<u> </u>		Rates(\$)	J	
							Rec	First		First	Add'l	COMEC	COMAN	SOMAN	SOMAN	SOMAN	SOMAN
ти	ho "70	ne" shown in the sections for stand-alone loops or loops as	nart of	2.00m	hination refers to G	ographically	Dogworagod II		Add'l							SOWAN	SOMAN
		ww.interconnection.bellsouth.com/become_a_clec/html/inter				eographically	Deaveraged Of	NE Zones. To	view Georgrap	fincally Deaver	aged ONE 201	le Desiganti		, refer to filter	het website.		
		SUPPORT SYSTEMS	connec	uon.n	T	1	,					1			r		r
		1) Electronic Service Order: CLEC should contact its contract	t negot	iator i	f it prefers the state	specific elect	tronic service o	rdering charge	es as ordered h	w the State Co	mmissions T	he electron	ic service o	rdering charg	e currently cr	ntained in th	is rate
		is the BellSouth regional electronic service ordering charge.															15 rule
		2) Any element that can be ordered electronically will be bille															lly For
		lements that cannot be ordered electronically at present per the															
		g charge, SOMAN, will be applied to a CLECs bill when it sub				e in this cate	gory reflects the	s charge that w	70ulu be billeu			nuening cap	abilities co	me on-me to	inat element	Otherwise,	ine manual
01	uenn	Manual Service Order Charge, per LSR, Disconnect Only (SC)	iiiits ai	LOK	lo Belisoutii.	SOMAN		I		1.97		1					r
		Electronic OSS Charge, per LSR, submitted via BST's OSS				0011/041	<u>├</u>			1.07			<u> </u>	<u> </u>	<u> </u>		
		interactive interfaces (Regional)				SOMEC		3.50	, ,						1		1
UNE SERV	VICE	DATE ADVANCEMENT CHARGE					1 1	0.00						1	<u> </u>		t
		The Expedite charge will be maintained commensurate with	BellSou	th's F	CC No.1 Tariff, Secti	on 5 as appli	cable.	ł							1		t
		UNE Expedite Charge per Circuit or Line Assignable USOC, per			1			I				1	1	1	İ.	†	İ
		Day			ALL UNE	SDASP		200.00	, ,						1	1	1
		XCHANGE ACCESS LOOP															
2-	WIRE	ANALOG VOICE GRADE LOOP															
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 1		1	UEANL	UEAL2	14.94	37.92	17.62	23.56	5.32		15.69				
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 2		2	UEANL	UEAL2	21.39	37.92	17.62	23.56	5.32		15.69				
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 3		3	UEANL	UEAL2	26.72	37.92	17.62	23.56	5.32		15.69	\vdash	Ļ		ļ
		Loop Testing - Basic 1st Half Hour		L	UEANL	URET1	<u> </u>	34.23	34.23				15.69	 	 	ļ!	
		Loop Testing - Basic Additional Half Hour		L	UEANL	URETA	<u> </u>	19.90	19.90				15.69	───		Ļ'	
		CLEC to CLEC Conversion Charge Without Outside Dispatch						45.55	0.00				45.00		1	1	1
\vdash		(UVL-SL1)			UEANL	UREWO	┥────┤	15.81	8.96				15.69	───	 	↓ ′	<u> </u>
		Unbundled Voice Loop, Unbundled Non-Design Voice Loop,						40.47	40.47						1	1	1
\vdash		billing for BST providing make-up Manual Order Coordination for UVL-SL1s (per loop)			UEANL UEANL	UEANM UEAMC	╂────┤	13.47 8.17	13.47 8.17				┝────	ł	 	┢────┘	
\vdash		Order Coordination for OVL-SL1S (per loop) Order Coordination for Specified Conversion Time for UVL-SL1			OLAINL	JEAIVIC	├──── ┤	8.17	8.17		1		┝───	───	<u> </u>	───┘	
		(per LSR)			UEANL	OCOSL		18.13	18.13								
2-1	WIRE	Unbundled COPPER LOOP			OLANE	OCOOL	+	10.15	10.15						<u> </u>		
2-	WIILL	2-Wire Unbundled Copper Loop - Non-Designed Zone 1	1	1	UEQ	UEQ2X	12.94	36.40	16.10	22.66	4.42		15.69	<u> </u>	<u> </u>		
		2 Wire Unbundled Copper Loop - Non-Designed - Zone 2	-i-	2	UEQ	UEQ2X	14.51	36.40	16.10	22.66	4.42		15.69				
		2 Wire Unbundled Copper Loop - Non-Designed - Zone 3	i	3	UEQ	UEQ2X	15.02	36.40	16.10	22.66	4.42		15.69				
		Order Coordination 2 Wire Unbundled Copper Loop - Non-	-	-													
		Designed (per loop)			UEQ	USBMC		8.17	8.17								
		Unbundled Copper Loop, Non-Designed Billing for BST										1					1
		providing make-up			UEQ	UEQMU		13.47	13.47				15.69		1	1	1
		Loop Testing - Basic 1st Half Hour			UEQ	URET1		34.23	34.23				15.69				
		Loop Testing - Basic Additional Half Hour			UEQ	URETA		19.90	19.90				15.69				
		CLEC to CLEC Conversion Charge Without Outside Dispatch													1		
		(UCL-ND)			UEQ	UREWO		14.30	7.45				15.69	L		↓ '	ļ
		XCHANGE ACCESS LOOP			1	1		I	,'				\square	L		Ļ'	L
2-	WIRE	ANALOG VOICE GRADE LOOP		L			<u> </u>		!				L	 	 	ļ!	
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-													1		1
\vdash		Zone 1 2 Wire Apples Vains Crade Loop Capital Loup 4 Line Califying		1	UEPSR UEPSB	UEALS	14.94	37.92	17.62	23.56	5.32		15.69	───	───	┟────┘	ł
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-						07.00	17.00	00 50	F 60		15.00		1	1	1
\vdash		Zone 1 2 Wire Apples Voice Crode Leap, Service Level 1 Line Splitting		1	UEPSR UEPSB	UEABS	14.94	37.92	17.62	23.56	5.32		15.69	───	 	┟────┘	ł
		2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting- Zone 2		2	UEPSR UEPSB	UEALS	21.39	37.92	17.62	23.56	5.32		15.69		1	1	1
\vdash		2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting-		- 4	ULF ON ULFOD	JLALO	21.39	51.92	17.02	23.00	J.32	1	15.09	t	t	┟────┘	<u> </u>
		Z wire Analog voice Grade Loop- Service Level 1-Line Splitting- Zone 2		2	UEPSR UEPSB	UEABS	21.39	37.92	17.62	23.56	5.32		15.69		1		1
\vdash		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-		- 4	ULF ON ULFOD	JLABO	21.39	31.92	17.02	23.00	J.32		15.09	<u> </u>	t	┢────┘	
		Z while Analog voice Grade Loop-Service Lever 1-Line Splitting- Zone 3		3	UEPSR UEPSB	UEALS	26.72	37.92	17.62	23.56	5.32		15.69		1	1	1
\vdash		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-		5	OLI ON OLF OD	JEALO	20.72	31.92	17.02	20.00	5.52		13.09	<u> </u>	t	┢────┘	<u> </u>
1 1		Z whe Analog voice Grade Loop-Service Level 1-Line Splitting- Zone 3		3	UEPSR UEPSB	UEABS	26.72	37.92	17.62	23.56	5.32		15.69		1	1	
		op Rates for Line Splitting				SLADO	20.72	51.92	17.02	20.00	5.52		10.09	<u> </u>	<u> </u>		<u> </u>
LIN				<u> </u>			+	0.40				1	<u> </u>	H	<u> </u>	ļ/	
U		2-Wire Voice Grade Loop (SL1) for Line Splitting - Zone 1		1	UEPRX	UEPLX	14.89	0.10	0.10								
10		2-Wire Voice Grade Loop (SL1) for Line Splitting - Zone 1 2-Wire Voice Grade Loop (SL1) for Line Splitting - Zone 2		1	UEPRX	UEPLX	14.89 21.52	0.10	0.10				<u> </u>		<u> </u>	<u>├</u> i	

UNBUNDL	ED NETWORK ELEMENTS - South Carolina		r										Attachment:			ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonred		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	EXCHANGE ACCESS LOOP															
2-WI	RE ANALOG VOICE GRADE LOOP															
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling - Zone 1		1	UEA	UEAL2	16.68	105.98	68.43	53.05	10.61		15.69				
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling - Zone 2		2	UEA	UEAL2	23.13	105.98	68.43	53.05	10.61		15.69				
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling - Zone 3		3	UEA	UEAL2	28.46	105.98	68.43	53.05	10.61		15.69				
	Order Coordination for Specified Conversion Time (per LSR)			UEA	OCOSL		18.13									
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zone 1		1	UEA	UEAR2	16.68	105.98	68.43	53.05	10.61		15.69				
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zone 2		2	UEA	UEAR2	23.13	105.98	68.43	53.05	10.61		15.69				
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zone 3		3	UEA	UEAR2	28.46	105.98	68.43	53.05	10.61		15.69				
	Order Coordination for Specified Conversion Time (per LSR)		3	UEA	OCOSL	20.40	105.98	00.43	55.05	10.01		15.69				
	CLEC to CLEC Conversion Charge without outside dispatch			UEA	UREWO		87.90	36.44				15.69				-
4-WI	RE ANALOG VOICE GRADE LOOP			0E/(UNEWO		07.00	00.11				10.00				
	4-Wire Analog Voice Grade Loop - Zone 1		1	UEA	UEAL4	32.59	132.38	94.83	59.35	14.61		15.69				
	4-Wire Analog Voice Grade Loop - Zone 2		2	UEA	UEAL4	43.89	132.38	94.83	59.35	14.61		15.69				
	4-Wire Analog Voice Grade Loop - Zone 3			UEA	UEAL4	43.38	132.38	94.83	59.35	14.61		15.69				
	Order Coordination for Specified Conversion Time (per LSR)		Ű	UEA	OCOSL	10.00	18.13	0 1100	00.00	11101		10.00				
	CLEC to CLEC Conversion Charge without outside dispatch			UEA	UREWO		87.90	36.44				15.69				
2-WI	RE ISDN DIGITAL GRADE LOOP			-												
	2-Wire ISDN Digital Grade Loop - Zone 1		1	UDN	U1L2X	25.21	117.58	80.03	53.05	10.61		15.69				
	2-Wire ISDN Digital Grade Loop - Zone 2		2	UDN	U1L2X	32.76	117.58	80.03	53.05	10.61		15.69				
	2-Wire ISDN Digital Grade Loop - Zone 3		3	UDN	U1L2X	37.70	117.58	80.03	53.05	10.61		15.69				
	Order Coordination For Specified Conversion Time (per LSR)			UDN	OCOSL		18.13									
	CLEC to CLEC Conversion Charge without outside dispatch			UDN	UREWO		91.82	44.25				15.69				
2-WI	RE Universal Digital Channel (UDC) COMPATIBLE LOOP															
	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone 1		1	UDC	UDC2X	25.21	117.58	80.03	53.05	10.61		15.69				
	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone 2		2	UDC	UDC2X	32.76	117.58	80.03	53.05	10.61		15.69				
	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone		3	UDC	UDC2X	37.70	117.58	80.03	53.05	10.61		15.69				
	CLEC to CLEC Conversion Charge without outside dispatch		5	UDC	UREWO	51.10	91.82	44.25	55.05	10.01		15.69				
2-WI	RE ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMP		LOOP		0.110		01.02					10.00				<u> </u>
	2 Wire Unbundled ADSL Loop including manual service inquiry				1 1										1	1
	& facility reservation - Zone 1 2 Wire Unbundled ADSL Loop including manual service inquiry		1	UAL	UAL2X	12.19	120.84	70.56	50.37	7.93		15.69				
	& facility reservation - Zone 2	<u> </u>	2	UAL	UAL2X	13.71	120.84	70.56	50.37	7.93		15.69				
	2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation - Zone 3		3	UAL	UAL2X	14.14	120.84	70.56	50.37	7.93		15.69				
	Order Coordination for Specified Conversion Time (per LSR)	ļ	I	UAL	OCOSL		18.13									4
	2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 1		1	UAL	UAL2W	12.19	95.81	57.82	50.37	7.93		15.69				
	2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 2		2	UAL	UAL2W	13.71	95.81	57.82	50.37	7.93		15.69				
	2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 3		3	UAL	UAL2W	14.14	95.81	57.82	50.37	7.93		15.69				
	Order Coordination for Specified Conversion Time (per LSR)			UAL	OCOSL		18.13									1
	CLEC to CLEC Conversion Charge without outside dispatch			UAL	UREWO		86.38	40.48				15.69				1
2-WI	RE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPA	TIBLE	LOOP													
	2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation - Zone 1		1	UHL	UHL2X	9.58	129.52	79.24	50.37	7.93		15.69				
	2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation - Zone 2		2	UHL	UHL2X	10.92	129.52	79.24	50.37	7.93		15.69				

	ED NETWORK ELEMENTS - South Carolina		1								-	-	Attachment:			ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonred		Nonrecurring					Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2 Wire Unbundled HDSL Loop including manual service inquiry		_													
	& facility reservation - Zone 3		3	UHL	UHL2X	11.40	129.52	79.24	50.37	7.93		15.69				<u> </u>
	Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		18.13									
	2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 1		1	UHL	UHL2W	9.58	104.49	66.50	50.37	7.93		15.69				
	2 Wire Unbundled HDSL Loop without manual service inquiry		1	UHL	UHLZW	9.56	104.49	00.30	50.57	7.93		15.69				<u> </u>
	and facility reservation - Zone 2		2	UHL	UHL2W	10.92	104.49	66.50	50.37	7.93		15.69				
	2 Wire Unbundled HDSL Loop without manual service inquiry		_	0.12	01.122.11	10.02	101110	00.00	00.0.	1.00		10.00				+
	and facility reservation - Zone 3		3	UHL	UHL2W	11.40	104.49	66.50	50.37	7.93		15.69				
	Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		18.13									
	CLEC to CLEC Conversion Charge without outside dispatch			UHL	UREWO		86.32	40.48				15.69				
4-WIR	RE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPA	TIBLE	-00P													
	4 Wire Unbundled HDSL Loop including manual service inquiry		Ι. Ι		Ι											
	and facility reservation - Zone 1		1	UHL	UHL4X	16.02	158.18	107.89	55.12	10.38		15.69				───
	4-Wire Unbundled HDSL Loop including manual service inquiry		2	UHL		14.00	450.40	407.00	FF 10	40.00		45.00				1
	and facility reservation - Zone 2		2	UHL	UHL4X	14.33	158.18	107.89	55.12	10.38		15.69				
	4-Wire Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 3		3	UHL	UHL4X	16.84	158.18	107.89	55.12	10.38		15.69				
	Order Coordination for Specified Conversion Time (per LSR)		3	UHL	OCOSL	10.04	136.16	107.69	55.1Z	10.30		15.69				<u> </u>
	4-Wire Unbundled HDSL Loop without manual service inquiry			OTIL	OCCOL		10.15									+
	and facility reservation - Zone 1		1	UHL	UHL4W	16.02	133.14	95.16	55.12	10.38		15.69				
	4-Wire Unbundled HDSL Loop without manual service inquiry				OTIL	10.02	100.14	55.10	00.12	10.00		10.00				+
	and facility reservation - Zone 2		2	UHL	UHL4W	14.33	133.14	95.16	55.12	10.38		15.69				
	4-Wire Unbundled HDSL Loop without manual service inquiry			-	-											1
	and facility reservation - Zone 3		3	UHL	UHL4W	16.84	133.14	95.16	55.12	10.38		15.69				
	Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		18.13									
	CLEC to CLEC Conversion Charge without outside dispatch			UHL	UREWO		86.32	40.48				15.69				
4-WIR	RE DS1 DIGITAL LOOP															<u> </u>
	4-Wire DS1 Digital Loop - Zone 1			USL	USLXX	79.51	253.03	157.89	44.80	11.73		15.69				
	4-Wire DS1 Digital Loop - Zone 2		2	USL	USLXX	136.00	253.03	157.89	44.80	11.73		15.69				───
	4-Wire DS1 Digital Loop - Zone 3		3	USL USL	USLXX OCOSL	229.15	253.03 18.13	157.89	44.80	11.73		15.69				<u> </u>
	Order Coordination for Specified Conversion Time (per LSR) CLEC to CLEC Conversion Charge without outside dispatch			USL	UREWO		101.30	43.13				15.69				
4-WIF	RE 19.2, 56 OR 64 KBPS DIGITAL GRADE LOOP			UUL	UNEWO		101.50	45.15				15.05				+
	4 Wire Unbundled Digital 19.2 Kbps		1	UDL	UDL19	29.93	126.66	89.12	59.35	14.61		15.69				<u> </u>
	4 Wire Unbundled Digital 19.2 Kbps			UDL	UDL19	33.99	126.66	89.12	59.35	14.61		15.69				1
· · · · ·	4 Wire Unbundled Digital 19.2 Kbps			UDL	UDL19	34.74	126.66	89.12	59.35	14.61		15.69				1
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 1		1	UDL	UDL56	29.93	126.66	89.12	59.35	14.61		15.69				
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 2			UDL	UDL56	33.99	126.66	89.12	59.35	14.61		15.69				
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 3		3	UDL	UDL56	34.74	126.66	89.12	59.35	14.61		15.69				
	Order Coordination for Specified Conversion Time (per LSR)			UDL	OCOSL		18.13									
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 1			UDL	UDL64	29.93	126.66	89.12	59.35	14.61		15.69				
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 2 4 Wire Unbundled Digital Loop 64 Kbps - Zone 3			UDL UDL	UDL64 UDL64	33.99 34.74	126.66 126.66	89.12 89.12	59.35 59.35	14.61		15.69				
	Order Coordination for Specified Conversion Time (per LSR)		3	UDL	OCOSL	34.74	120.00	89.12	59.35	14.61		15.69				
	CLEC to CLEC Conversion Charge without outside dispatch			UDL	UREWO		102.34	49.85				15.69				<u> </u>
2-WIF	RE Unbundled COPPER LOOP	1			0112400		102.04	+0.00				10.09				+
	2-Wire Unbundled Copper Loop/Short including manual service				+ +								1			t
	inquiry & facility reservation - Zone 1		1	UCL	UCLPB	12.19	119.91	69.62	50.37	7.93		15.69				1
- 1	2-Wire Unbundled Copper Loop/Short including manual service															1
	inquiry & facility reservation - Zone 2		2	UCL	UCLPB	13.71	119.91	69.62	50.37	7.93		15.69				
	2 Wire Unbundled Copper Loop/Short including manual service															
	inquiry & facility reservation - Zone 3		3	UCL	UCLPB	14.14	119.91	69.62	50.37	7.93		15.69				<u> </u>
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.17	8.17								───
	2-Wire Unbundled Copper Loop/Short without manual service	1	Ι.			10.1-	a	=0								1
	inquiry and facility reservation - Zone 1 2-Wire Unbundled Copper Loop/Short without manual service		1	UCL	UCLPW	12.19	94.87	56.89	50.37	7.93		15.69				───
	K-WORE DODUDOJEO COODELLOOD/SDOT WITDOUT MADUAL SEV/CE	1	2	UCL	UCLPW	13.71	94.87	56.89	50.37	7.93	1	15.69	1		1	1

UNBUNDLE	D NETWORK ELEMENTS - South Carolina												Attachment:	2	Exhi	bit: B
											Svc Order	Svc Order			Incremental	
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
		Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
							Nonrec	urring	Nonrecurring	Disconnect			055	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Unbundled Copper Loop/Short without manual service				1											
	inquiry and facility reservation - Zone 3		3	UCL	UCLPW	14.14	94.87	56.89	50.37	7.93		15.69				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.17	8.17								
	2-Wire Unbundled Copper Loop/Long - includes manual srvc.			UCL		00.00	110.01	00.00	50.07	7.00		45.00				
	inquiry and facility reservation - Zone 1 2-Wire Unbundled Copper Loop/Long - includes manual svc.		1	UCL	UCL2L	38.22	119.91	69.62	50.37	7.93		15.69				
	inquiry and facility reservation - Zone 2		2	UCL	UCL2L	55.33	119.91	69.62	50.37	7.93		15.69				
	2-Wire Unbundled Copper Loop/Long - includes manual svc.															
	inquiry and facility reservation - Zone 3		3	UCL	UCL2L	67.95	119.91	69.62	50.37	7.93		15.69				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.17	8.17								
	2-Wire Unbundled Copper Loop/Long - without manual service					00.00	04.67	50.00	50.07	7.00		45.00				
├ ── ├ ──	inquiry and facility reservation - Zone 1 2-Wire Unbundled Copper Loop/Long - without manual service		1	UCL	UCL2W	38.22	94.87	56.89	50.37	7.93		15.69				
	inquiry and facility reservation - Zone 2	1	2	UCL	UCL2W	55.33	94.87	56.89	50.37	7.93		15.69				
	2-Wire Unbundled Copper Loop/Long - without manual service	1			301211	00.00	54.67	00.00	00.07	1.33		10.00		-		
	inquiry and facility reservation - Zone 3		3	UCL	UCL2W	67.95	94.87	56.89	50.37	7.93		15.69				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.17	8.17								
	CLEC to CLEC Conversion Charge without outside dispatch															
	(UCL-Des)			UCL	UREWO		94.87	42.57				15.69				
4-WIR	E COPPER LOOP	-														
	4-Wire Copper Loop/Short - including manual service inquiry and facility reservation - Zone 1		1	UCL	UCL4S	19.64	144.17	93.88	55.12	10.38		15.69				
	4-Wire Copper Loop/Short - including manual service inquiry			002	00L-10	10.04	144.17	00.00	00.12	10.00		10.00				
	and facility reservation - Zone 2		2	UCL	UCL4S	20.90	144.17	93.88	55.12	10.38		15.69				
	4-Wire Copper Loop/Short - including manual service inquiry															
	and facility reservation - Zone 3		3	UCL	UCL4S	19.34	144.17	93.88	55.12	10.38		15.69				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.17	8.17								
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation - Zone 1		1	UCL	UCL4W	19.64	119.13	81.15	55.12	10.38		15.69				
	4-Wire Copper Loop/Short - without manual service inquiry and			UCL	OCL4VV	13.04	113.15	01.15	55.12	10.50		15.05				
	facility reservation - Zone 2		2	UCL	UCL4W	20.90	119.13	81.15	55.12	10.38		15.69				
	4-Wire Copper Loop/Short - without manual service inquiry and															
	facility reservation - Zone 3		3	UCL	UCL4W	19.34	119.13	81.15	55.12	10.38		15.69				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.17	8.17								
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 1		1	UCL	UCL4L	77.29	144.17	93.88	55.12	10.38		15.69				
	4-Wire Unbundled Copper Loop/Long - includes manual svc.			002	UUL4L	11.29	144.17	30.00	JJ. 12	10.30		15.09				
	inquiry and facility reservation - Zone 2	1	2	UCL	UCL4L	118.78	144.17	93.88	55.12	10.38		15.69				
	4-Wire Unbundled Copper Loop/Long - includes manual svc.															
	inquiry and facility reservation - Zone 3		3	UCL	UCL4L	144.10	144.17	93.88	55.12	10.38		15.69				
┝──┤──	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.17	8.17								
	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone 1	1	1	UCL	UCL4O	77.29	119.44	81.45	55.12	10.38		15.69				
<u>├── </u>	4-Wire Unbundled Copper Loop/Long - without manual svc.				301-10	11.29	110.44	01.43	55.12	10.30		13.09				
	inquiry and facility reservation - Zone 2	1	2	UCL	UCL4O	118.78	119.44	81.45	55.12	10.38		15.69				
	4-Wire Unbundled Copper Loop/Long - without manual svc.															
	inquiry and facility reservation - Zone 3		3	UCL	UCL40	144.10	119.44	81.45	55.12	10.38		15.69				
├──-	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.17	8.17								
	CLEC to CLEC Conversion Charge without outside dispatch (UCL-Des)	1		UCL	UREWO		94.87	42.57				15.69				
LOOP MODIFI				50L	JILL WO		34.07	42.37				13.09				
				UAL, UHL, UCL,												
				UEQ, ULS, UEA,												
	Unbundled Loop Modification, Removal of Load Coils - 2 Wire	1		UEANL, UDL, UDC,												
└── └──	pair less than or equal to 18k ft	ļ		UDN, UDL, USL	ULM2L		32.46	32.46				15.69				
	Unbundled Loop Modification, Removal of Load Coils - 2 wire greater than 18k ft	1		UCL, ULS, UEQ	ULM2G		170.89	170.89				15.69				
		<u> </u>		UUL, ULO, UEQ	ULIVIZG		170.89	170.89			l	15.69				
	Unbundled Loop Modification Removal of Load Coils - 4 Wire						1									

ONRONDL	ED NETWORK ELEMENTS - South Carolina		1		1						. - ·		Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrec		Nonrecurring			_		Rates(\$)	_	
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Unbundled Loop Modification Removal of Load Coils - 4 Wire			UCL	ULM4G		470.00	170.00				45.00				
	pair greater than 18k ft Unbundled Loop Modification Removal of Bridged Tap Removal, per unbundled loop			UAL, UHL, UCL, UEQ, UEF, ULS, UEA, UEANL, UDL, UDC, UDN, UDL, USL	ULMBT		170.89 32.48	170.89 32.48				15.69				
SUB-LOOPS																l
Sub-	Loop Distribution															L
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set- Up	I		UEANL	USBSA		241.42	241.42				15.69				
	Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-Up Sub-Loop - Per Building Equipment Room - CLEC Feeder	I		UEANL	USBSB		22.69	22.69				15.69				
	Facility Set-Up	1		UEANL	USBSC		177.84	177.84				15.69				
	Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-Up	Ι		UEANL	USBSD		55.58	55.58				15.69				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 1	I	1	UEANL	USBN2	8.87	65.94	31.03	45.35	6.71		15.69				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 2	I	2	UEANL	USBN2	12.58	65.94	31.03	45.35	6.71		15.69				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 3	I	3	UEANL	USBN2	14.79	65.94	31.03	45.35	6.71		15.69				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		8.17	8.17								
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 1		1	UEANL	USBN4	14.11	79.21	44.29	49.82	9.09		15.69				
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 2		2	UEANL	USBN4	19.40	79.21	44.29	49.82	9.09		15.69				
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 3		3	UEANL	USBN4	18.90	79.21	44.29	49.82	9.09		15.69				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		8.17	8.17								
	Sub-Loop 2-Wire Intrabuilding Network Cable (INC)			UEANL	USBR2	2.41	53.13	18.21	45.35	6.71		15.69				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		8.17	8.17								
	Sub-Loop 4-Wire Intrabuilding Network Cable (INC)	- 1		UEANL	USBR4	5.36	59.38	24.47	49.82	9.09		15.69				ļ
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		8.17	8.17								
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 1	1	1	UEF	UCS2X	7.11	65.94	31.03	45.35	6.71		15.69				
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2	i		UEF	UCS2X	9.83	65.94	31.03	45.35	6.71		15.69				
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 3	I	3	UEF	UCS2X	10.48	65.94	31.03	45.35	6.71		15.69				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC		8.17	8.17								
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 1	1	1	UEF	UCS4X	7.85	79.21	44.29	49.82	9.09		15.69				
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 2		2	UEF	UCS4X	14.17	79.21	44.29	49.82	9.09		15.69				
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 3		3	UEF	UCS4X USBMC	12.64	79.21	44.29 8.17	49.82	9.09		15.69				
Unbi	Order Coordination for Unbundled Sub-Loops, per sub-loop pair andled Sub-Loop Modification	<u> </u>	+		USDIVIC		8.17	ö.17								<u> </u>
	Unbundled Sub-Loop Modification - 2-W Copper Dist Load Coil/Equip Removal per 2-W PR			UEF	ULM2X		176.17	5.11				15.69				
	Unbundled Sub-loop Modification - 4-W Copper Dist Load Coil/Equip Removal per 4-W PR			UEF	ULM4X		176.17	5.11				15.69				
	Unbundled Sub-loop Modification - 2-w/4-w Copper Dist Bridged Tap Removal, per PR unloaded			UEF	ULM4T		278.82	6.13				15.69				
Unb	undled Network Terminating Wire (UNTW)										1					
	Unbundled Network Terminating Wire (UNTW) per Pair			UENTW	UENPP	0.3303	30.20	30.20				15.69				
Netw	ork Interface Device (NID)															

UNBUNDLE	D NETWORK ELEMENTS - South Carolina			1									Attachment:			ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)					Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec	urring	Nonrecurring	J Disconnect			OSS	Rates(\$)	•	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Network Interface Device (NID) - 1-2 lines			UENTW	UND12		43.68	28.79				15.69				
	Network Interface Device (NID) - 1-6 lines			UENTW	UND16		64.42	49.53				15.69				
	Network Interface Device Cross Connect - 2 W			UENTW	UNDC2		5.92	5.92				15.69				
	Network Interface Device Cross Connect - 4W			UENTW	UNDC4		5.92	5.92				15.69				
SUB-LOOPS																───
Sub-L	oop Feeder USL-Feeder, DS0 Set-up per Cross Box location - CLEC			UEA,												
	Distribution Facility set-up			UEA, UDN,UCL,UDL,UDC			241.42					15.69				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair			UEA,	036170		241.42					15.09				
	set-up			UDN,UCL,UDL,UDC	USBFX		22.69	22.69				15.69				
	USL Feeder DS1 Set-up at DSX location, per DS1 termination			USL	USBFZ		523.87	11.34			1	15.69				
	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice			İ							1				İ	1
	Grade - Zone 1		1	UEA	USBFA	8.93	93.28	56.69	54.68	13.74		15.69				
1	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice															
	Grade - Zone 2		2	UEA	USBFA	11.74	93.28	56.69	54.68	13.74		15.69				
	Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start,															
	Voice Grade - Zone 3		3	UEA	USBFA	14.74	93.28	56.69	54.68	13.74		15.69				
	Order Coordination for Specified Conversion Time, per LSR			UEA	OCOSL		18.13									
	Unbundlde Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice															
	Grade - Zone 1		1	UEA	USBFB	8.93	93.28	56.69	54.68	13.74		15.69				
	Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice							=0.00				15.00				
	Grade - Zone 2		2	UEA	USBFB	11.74	93.28	56.69	54.68	13.74		15.69				<u> </u>
	Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice Grade - Zone 3		3	UEA	USBFB	14.74	93.28	56.69	54.68	13.74		15.69				
			3	UEA	OCOSL	14.74	93.28	56.69	54.68	13.74		15.69				
	Order Coordination for Specified Time Conversion, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery,			UEA	OCOSL		10.13									
	Voice Grade - Zone 1		1	UEA	USBFC	8.93	93.28	56.69	54.68	13.74		15.69				
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery,		· ·	0E/(00010	0.00	00.20	00.00	04.00	10.14		10.00				
	Voice Grade - Zone 2		2	UEA	USBFC	11.74	93.28	56.69	54.68	13.74		15.69				
	Unbundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse			-												
	Battery, Voice Grade - Zone 3		3	UEA	USBFC	14.74	93.28	56.69	54.68	13.74		15.69				
	Order Coordination For Specified Conversion Time, per LSR			UEA	OCOSL		18.13									
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice															
	Grade - Zone 1		1	UEA	USBFD	21.63	107.91	70.36	62.26	17.52		15.69				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice															
	Grade - Zone 2		2	UEA	USBFD	27.57	107.91	70.36	62.26	17.52		15.69				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice		~			00.04	107.01	70.00	00.00	47.50		45.00				
	Grade - Zone 3		3	UEA	USBFD	26.04	107.91	70.36	62.26	17.52	-	15.69				
	Order Coordination For Specified Conversion Time, Per LSR Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice			UEA	OCOSL		18.13									
	Grade - Zone 1		1	UEA	USBFE	21.63	107.91	70.36	62.26	17.52		15.69				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice			UEA	USBFE	21.03	107.91	70.30	02.20	17.52		15.69				
	Grade - Zone 2		2	UEA	USBFE	27.57	107.91	70.36	62.26	17.52		15.69				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice		~	0E/(OOD! L	21.01	107.01	10.00	02.20	17.02		10.00				
	Grade - Zone 3		3	UEA	USBFE	26.04	107.91	70.36	62.26	17.52		15.69				
	Order Coordination For Specified Conversion Time, Per LSR		-	UEA	OCOSL		18.13									
	Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 1		1	UDN	USBFF	17.05	106.47	68.92	55.81	13.37		15.69				
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 2		2	UDN	USBFF	20.92	106.47	68.92	55.81	13.37		15.69				
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 3		3	UDN	USBFF	23.49	106.47	68.92	55.81	13.37		15.69				
	Order Coordination For Specified Conversion Time, Per LSR			UDN	OCOSL		18.13									L
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)	I	1	UDC	USBFS	17.05	106.47	68.92	55.81	13.37		15.69				───
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)	I		UDC	USBFS	20.92	106.47	68.92	55.81	13.37		15.69				
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)		3	UDC	USBFS	23.49	106.47	68.92	55.81	13.37	L	15.69				───
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 1		1	USL	USBFG	55.85	102.19	64.64	62.26	17.52	ļ	15.69				───
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 2 Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 3	l	2	USL USL	USBFG USBFG	109.16 203.35	102.19 102.19	64.64 64.64	62.26 62.26	17.52 17.52		15.69				┨──────
	10000000e0 300-L000 Feeder L000 4-Wire UST - Z006 3	1	3		USBEG	203.35	102.19	04.64	02.26	17.52		15.69			1	────
	Order Coordination For Specified Conversion Time, Per LSR			USL	OCOSL		18.13									

JUNDONUL	ED NETWORK ELEMENTS - South Carolina												Attachment:	2	Exhi	bit: B
											Svc Order	Svc Order	Incremental		Incremental	
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
		Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svo
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		m										•	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
							N		N	<u></u>						
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone		2	UCL		4.00	00.07	40,40	50.44	40.00		45.00				
	Linhundlad Sub Loop Fooder Loop, 2 Wire Conner Loop, Zope		2	UCL	USBFH	4.80	83.97	46.42	53.14	10.69		15.69				
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone		3	UCL	USBFH	4.59	83.97	46.42	53.14	10.69		15.69				
	Order Coordination For Specified Conversion Time, per LSR		3	UCL	OCOSL	4.55	18.13	40.42	55.14	10.09	1	13.09				-
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 1		1	UCL	USBFJ	13.21	101.22	63.67	58.03	13.29		15.69				
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 2			UCL	USBFJ	8.28	101.22	63.67	58.03	13.29	1	15.69				
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 3			UCL	USBFJ	8.42	101.22	63.67	58.03	13.29		15.69				
	Order Coordination For Specified Conversion Time, per LSR		Ů	UCL	OCOSL	0.12	18.13	00.01	00.00	10.20	1	10.00				
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		1	UDL	USBFN	21.02	102.19	64.64	62.26	17.52		15.69				
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		2	UDL	USBFN	21.30	102.19	64.64	62.26	17.52		15.69				
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop	1		UDL	USBFN	20.17	102.19	64.64	62.26	17.52	1	15.69	ĺ	ĺ	İ	1
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -		1													
	Zone 1	1	1	UDL	USBFO	21.02	102.19	64.64	62.26	17.52		15.69				
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -															
	Zone 2		2	UDL	USBFO	21.30	102.19	64.64	62.26	17.52		15.69				
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -															
	Zone 3		3	UDL	USBFO	20.17	102.19	64.64	62.26	17.52		15.69				
	Order Coordination For Specified Time Conversion, per LSR			UDL	OCOSL		18.13									
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -															
	Zone 1		1	UDL	USBFP	21.02	102.19	64.64	62.26	17.52		15.69				
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -															
	Zone 2		2	UDL	USBFP	21.30	102.19	64.64	62.26	17.52		15.69				
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -		_													
	Zone 3		3	UDL	USBFP	20.17	102.19	64.64	62.26	17.52		15.69				
	Order Coordination For Specified Conversion Time, per LSR			UDL	OCOSL		18.13									
SUB-LOOPS	Loop Feeder		-													
Sub-	Sub Loop Feeder - DS3 - Per Mile Per Month		-	UE3	1L5SL	20.44										
	Sub Loop Feeder - DS3 - Fei Mile Fei Month Sub Loop Feeder - DS3 - Facility Termination Per Month			UE3	USBF1	348.12	3,408.62	407.90	160.83	91.17	1	15.69				1
	Sub Loop Feeder – STS-1 – Per Mile Per Month			UDLSX	1L5SL	20.44	3,400.02	407.90	100.03	91.17		15.09				
	Sub Loop Feeder - STS-1 - Facility Termination Per Month		-	UDLSX	USBF7	369.07	3,408.62	407.90	160.83	91.17		15.69				
	Sub Loop Feeder – OC-3 – Per Mile Per Month	- i		UDLO3	1L5SL	15.51	3,400.02	407.30	100.00	31.17		15.05				
	Sub Loop Feeder - OC-3 - Facility Termination Protection Per		-	ODLOG	TEOOL	10.01										
	Month	1		UDLO3	USBF5	56.04										
	Sub Loop Feeder - OC-3 - Facility Termination Per Month	i		UDLO3	USBF2	565.50	3,408.62	407.90	160.83	91.17		15.69				
	Sub Loop Feeder - OC-12 - Per Mile Per Month			UDL12	1L5SL	19.08										
	Sub Loop Feeder - OC-12 - Facility Termination Protection Per			-												
	Month	1		UDL12	USBF6	669.82										
	Sub Loop Feeder - OC-12 - Facility Termination Per Month	1		UDL12	USBF3	1,840.00	3,408.62	407.90	160.83	91.17		15.69				
	Sub Loop Feeder - OC-48 - Per Mile Per Month			UDL48	1L5SL	62.60										
	Sub Loop Feeder - OC-48 - Facility Termination Protection Per															
	Month			UDL48	USBF9	326.16										
	Sub Loop Feeder - OC-48 - Facility Termination Per Month	I		UDL48	USBF4	1,560.00	3,594.62	407.90	160.83	91.17		15.69				
	Sub Loop Feeder - OC-12 Interface On OC-48	1		UDL48	USBF8	366.86	806.47	407.90	160.83	91.17		15.69				
UNBUNDLE	LOOP CONCENTRATION															ļ
	Unbundled Loop Concentration - System A (TR008)			ULC	UCT8A	318.73	326.13	326.13				15.69				ļ
	Unbundled Loop Concentration - System B (TR008)		L	ULC	UCT8B	46.69	135.89	135.89			L	15.69				
	Unbundled Loop Concentration - System A (TR303)		<u> </u>	ULC	UCT3A	351.78	326.13	326.13				15.69				
\vdash	Unbundled Loop Concentration - System B (TR303)		<u> </u>	ULC	UCT3B	78.67	135.89	135.89	10.00	4	ļ	15.69				ł
\vdash	Unbundled Loop Concentration - DS1 Loop Interface Card		<u> </u>	ULC	UCTCO	4.42	63.43	46.18	16.83	4.71	ļ	15.69				ł
	Unbundled Loop Concentration - ISDN Loop Interface (Brite		1		111.004	7.00	10.50	40.50	F 44	E 07		45.00				1
\vdash	Card)		<u> </u>	UDN	ULCC1	7.02	10.56	10.50	5.41	5.37	ļ	15.69				ł
	Unbundled Loop Concentration - UDC Loop Interface (Brite	1	1			7.00	40.50	40.50	F 44	F 07		45.00				
	Card) Unbundled Loop Concentration2 Wire Voice-Loop Start or	<u> </u>		UDC	ULCCU	7.02	10.56	10.50	5.41	5.37		15.69				
· · -	Unbunuled LOOP Concentration2 Wire Voice-Loop Start or	1	1	1							1	1	1	1	1	1
					111 CC2	1 75	10.56	10 50	E / 1	E 27		15 60				
	Ground Start Loop Interface (POTS Card) Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery			UEA	ULCC2	1.75	10.56	10.50	5.41	5.37		15.69				1

CATEGONT RATE ELEMENTS Into Do BCO BCO	UNBUNDLED	ONETWORK ELEMENTS - South Carolina												Attachment:	2	Exhi	bit: B
Impund Loss Constration - Wei Yook Log Nichtag UBA ULCX4 27 165 6.41 5.37 6.50 5.50 5.50 Instruction - Test ColCUT Card UEA ULCX4 10.50 15.60 15.60 5.41 5.37 15.60 - Instruction - Test ColCUT Card UEA ULCX 15.60 15.60 5.41 5.37 15.60 - - Instruction - Test ColCUT Card UDL ULCS 0.21 15.60 15.61 5.27 15.60 -	CATEGORY	RATE ELEMENTS		Zone	BCS	USOC						Submitted Elec	Submitted Manually	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
Ubbuilded Log Ubbuilded Log Ubbuilded Log ULCAL 6.27 10.60 6.41 6.37 11.60 Head ULCA 0.21 0.06 0.00 6.41 6.37 11.60 Head ULCA 0.017 0.03 10.00 6.41 6.37 11.60 Head ULCA 0.21 10.66 10.00 6.41 6.37 11.60 Head ULCA ULCA 0.21 10.66 10.00 6.41 6.37 11.60 Head ULCA ULCA 0.21 10.66 10.00 6.41 6.37 11.60 Head ULCA ULCA ULCA 0.00							Rec										
Special Cash UBC ULC2 6.22 1050 5.41 6.57 15.69 Without Casp Control 12.8 (bp Contr		Linbundled Loop Concentration - 4 Wire Voice Loop Interface						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Ubbindie Loop Concentration - Diguti 19:2 Kops Data Loop UDL ULCC A.21 10.46 10.40 6.44 6.37 11.66 Ubbindie Loop Concentration - Diguti 6 Kops Data Loop UDL ULCCS 0.21 10.56 10.50 6.44 6.37 115.66 10.50 Ubbindie Loop Concentration - Diguti 6 Kops Data Loop UDL ULCCS 0.21 10.56 10.50 6.44 6.37 15.66 10.50 UNE OTHER Participation of Service Distribution UDL ULCCS 0.21 10.56 10.50 6.44 6.37 15.66 10.50 UNE OTHER Participation of Service Distribution UDL UNECT 0.00 0.00 0.00 10.50					UEA	ULCC4	6.22	10.56	10.50	5.41	5.37		15.69				
Ubbindie Loop Concentration - Diguti 19:2 Kops Data Loop UDL ULCC A.21 10.46 10.40 6.44 6.37 11.66 Ubbindie Loop Concentration - Diguti 6 Kops Data Loop UDL ULCCS 0.21 10.56 10.50 6.44 6.37 115.66 10.50 Ubbindie Loop Concentration - Diguti 6 Kops Data Loop UDL ULCCS 0.21 10.56 10.50 6.44 6.37 15.66 10.50 UNE OTHER Participation of Service Distribution UDL ULCCS 0.21 10.56 10.50 6.44 6.37 15.66 10.50 UNE OTHER Participation of Service Distribution UDL UNECT 0.00 0.00 0.00 10.50	1	Unbundled Loop Concentration - TEST CIRCUIT Card			ULC	UCTTC	30.38	10.56	10.50	5.41	5.37		15.69				
Unbindle Loc Critochinsten - Digiti S Kipp Dial Loop ULC 9.21 10.56 15.61 5.77 15.68 UNDURING Concentration - Digiti S Kipp Dial Loop ULL ULC COR 2.21 10.56 0.50 5.41 5.37 15.68 WE OTHER, FROMOSING ON, Y-N RA PATE ULL ULC COR 2.21 10.56 0.50 5.41 5.37 15.68 ME OTHER, FROMOSING ON, Y-N RA PATE UERCW UERC OTHER, FROMOSING ON, Y-N RA PATE <		Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop															
Interface UPL UPC 0.20 0.64 0.05 5.41 5.37 15.69 UPE OTHER, PROVISIONE ONLY: NO RATE UP UPL 0.06 0.05 5.41 5.37 15.69 NEW DEPART, PROVISIONE ONLY: NO RATE UPL UPL UPL 0.06 0.00					UDL	ULCC7	9.21	10.56	10.50	5.41	5.37		15.69				┟─────┤
United of Concentration - Dipit 64 Rips Data Loop UPL UPL UPC 10.68 10.59 5.41 5.57 15.69 UNE OTHER PROF. Dipit And Start Office from 10 Institution UPR YM VADBX 0.00					UDL	ULCC5	9.21	10.56	10.50	5.41	5.37		15.69				
UNE OFFICE, PROVISIONNO ONLY - NO RATE Image: Control of the Monitoria Statution UNIX																	
NNO Degrate and Served Order IN ND Installation UENTY UNDEX 0.00 0.00 UNIT OF Degrate And Served Order IN ND Installation UERNULFER, DOI 30.00 0.00 0.00 0.00 0.00 UNIT OF CARL IS Establishment, Proteinaning Only - No Rate UERNULFER, DOI 30.00 0.00 0.00 0.00 0.00 UNIT OF THE Proteinang Only - No Rate UDNUELAUELUELUELUELUELUELUELUELUELUELUELUELUELU					UDL	ULCC6	9.21	10.56	10.50	5.41	5.37		15.69				
UNTW Circuit B Establishment, Provisioning Only - No Rate UENTW UERCE 0.00 0.00 0.00 UNE OTHER, ROWSING ONLY - No Rate UEALUFFLICEU UNE CM 0.00 0.00 0.00 0.00 UNE OTHER, ROWSING ONLY - No RATE UNLUFLUEUUL UNE CM 0.00 0.00 0.00 0.00 UNE OTHER, ROWSING ONLY - No RATE UNLUFLUEUUL UNLUFLUEUUL 0.00 0.00 0.00 0.00 UNDURED COLLOG, Superframe Format Option - no rate UNLUFLUEUUL UNE CCCEF 0.00 <t< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td>0.00</td><td>0.00</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td><u> </u></td></t<>							0.00	0.00									<u> </u>
Unter Unter Provisioning Only - No Rate UEANLUFF. VIEW UNECY 0.00 0.00 0.00 UNE CYTER, PROVISIONING ONLY - NO RATE H </td <td></td> <td></td> <td></td> <td></td> <td></td>																	
UNE OTHER, PROVISIONING ONLY - NO RATE Image: Control Name, Provisioning Only - no rate Image: Control Name, Provision Provisin Provisi Provision Provision Provision Provision Provision Provi					UEANL,UEF,UEQ,U												
Unbundled Contact Name, Provisioning Only - no rate UALUCL, UDC, UDL UDN, UEA, UUL, UDC, UDL UDN, UEA, UUL, UDL, UDC 0.00 0.00 Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no the UEA, UDN, UEA, UUL, UDL, UDC 0.00 0.00 0.00 Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no the UEA, UDN, UEA, UUL, UDL, UDE 0.00 0.00 0.00 Unbundled DS1 Loop - Superforma Format Option - no rate USL CCOSE 0.00 0.00 Uhbundled DS1 Loop - Superforma Format Option - no rate USL CCOSE 0.00 0.00 Heft CAPACTY WORKVLED LCCAL LOOP UEA JUSL/UCL, UDL, UDL USL CCOSE 0.00 0.00 Heft CAPACTY WORKVLED LCCAL LOOP UEA 1LSMD 12.26 0.00 0.00 0.00 Heft CAPACTY WORKVLED LCCAL LOOP - STS-1 - Per Mile per month UESX 1LSMD 12.26 0.00 0.00 0.00 Heft Capacity Unbundled Local Loop - STS-1 - Facility UDLSX ULSX 1LSMD 12.26 0.00 0.00 0.00 0.00 0.00 Heft Capacity Unbundled Local Loop - STS-1 - Facility UDLSX ULSX <					ENTW	UNECN	0.00	0.00									
Ubundled Contact Name, Provisioning Only - no rate UDPUELUNULUCU UNEX 0.00 0.00 0.00 Ubundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no UEAUDNUCLUCC USER 0.00 0.00 0.00 0.00 Ubundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no UEAUDNUCLUCC USER 0.00	UNE OTHER, PR	ROVISIONING ONLY - NO RATE															L
Ubundled Contact Name, Provisioning Only - no rate UDPULELUNLULUD UNEX 0.00 0.00 0.00 Ubbundled Sub-Loop Feeder-X Wire Cross Box Jumper - no gate UEAUDNUCLUDC USEFD 0.00 0.00 0.00 0.00 Ubbundled Sub-Loop Feeder-X Wire Cross Box Jumper - no gate UEAUDNUCLUDC USEFD 0.00 0.00 0.00 0.00 Ubbundled DS1 Logo - Superframe Format Option - no title (Bit CAPACITY UNBUNDLE DLOCAL LOAD) USE 0.00 0																	
Unbundled Sub-Loop Feeder 2 Wire Cross Box Jumper - no rate UEAUDN_UCL_UC USR 0.00 0.00 Whounded Sub-Loop Feeder4 Wire Cross Box Jumper - no rate UEAUDN_UCL_UC USR 0.00 0.00 0.00 0.00 Whounded Sub-Loop Feeder4 Wire Cross Box Jumper - no rate UEAUSL_UCL_UL USR CCOSF 0.00 0.00 0.00 0.00 Whounded Data Losp - Expanded Sub-frame Format option - no rate USR CCOSF 0.00		Unbundled Contact Name, Provisioning Only - no rate				UNECN	0.00	0.00									
Unbuilded Sub-Loop Feeder-4 Wire Cross Box Jumper - no rate UEA USLUC, UUL USBFR 0.00 0.00 Unbuilded DS1 Loop - Supardame Format Option - no rate USL CCOSF 0.00 0.00 0.00 0.00 Unbuilded DS1 Loop - Supardad Supertrane Format Option - no rate USL CCOSF 0.00 0.00 0.00 0.00 High Capacity Unbundled Local Loop - DS3 - Per Mile per month UE3 11.5ND 12.26 0.00 0.00 0.00 0.00 High Capacity Unbundled Local Loop - DS3 - Faoility UE3 11.5ND 12.26 0.00																	
Interview UPBA USL UCLUEU USBRF 0.00 0.00 Uhbundled DS1 Loop - Superframe Format Option - no rate USL CCOSF 0.00 0.00 0.00 HGH CAPACITY UNROULED LOCAL LOOP USL CCOSF 0.00 0.00 0.00 0.00 HGH CAPACITY UNROULED LOCAL LOOP USL CCOSF 0.00 0.00 0.00 0.00 HGH CAPACITY UNROULED LOCAL LOOP USL CCOSF 0.00 0.00 0.00 0.00 0.00 HGH CAPACITY UNROULED LOCAL LOOP USL ST 1LSND 12.28 0.00 0.00 0.00 0.00 HGH Capacity Unbundled Local Loop - DS3 - Facility US ST - Facility US ST - Facility 115.80 119.75 83.77 15.69 0.00 HGH Capacity Unbundled Local Loop - STS - Facility UD LSX UD LSX ULD ST - Facility 119.75 83.77 15.69 0.00 LOOP MAKE-UP Preordering Withourles Reservation, per working or pager facility queried (Menual). UM KUW 24.04 24.04 0.01 0.00 0.00 0.00 0					UEA,UDN,UCL,UDC	USBFQ	0.00	0.00									
Unbundled DS1 Loop - Superframe Format option - no rate USL CCOSF 0.00 0.00 0.00 Withbundled DS1 Loop - Expanded Superframe Format option - no rate USL CCOSF 0.00 0.00 0.00 0.00 High Capacity Unbundled Local Loop - DS3 - Per Mile per month UE3 12.26 0.00 0.00 0.00 High Capacity Unbundled Local Loop - DS3 - Facility UE3 12.36 0.03 0.00 0.00 0.00 High Capacity Unbundled Local Loop - DS3 - Facility UE3 UE3 0.03 4.62.52 284.53 110.75 83.77 15.60 High Capacity Unbundled Local Loop - STS-1 - Per Mile per month UDLSX 11.5ND 12.26 110.75 83.77 15.60 LOOP MAKEUP Procedeng WithoutReservation, per working or spare facility queried (Manual). UDLSX UDLSX 25.44 53 119.75 83.77 15.60 LOOP MAKEUP Procedeng With Reservation, per working or loop Makeup - Procedeng With Reservation, per working or loop Makeup - Neordeng With Reservation, per working or loop Makeup - Neordeng With Reservation, per working or loop Makeup - Neordeng With Reservation, per working or loop Makeup - Neordeng With Reservation, per wor							0.00	0.00									
Unbundled DS1 Loop - Expanded Superframe Format option - Ino rate USL CCOEF 0.00 0.00 0.00 0.00 HIGH CAPACITY UNBUNDLED LOCAL LOOP USL CCOEF 0.00 <td< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>-</td><td></td><td></td><td></td><td></td><td></td><td>├────</td></td<>											-						├ ────
Hield CaPACITY UNBUNDED LOCAL LOOP Image: Constraint of the co					001	0000.	0.00	0.00									
High Capacity Unbundled Local Loop - DS3 - Per Mile per month UE3 1LSND 12.26 Image: Comparison of the comparison of th					USL	CCOEF	0.00	0.00									
month UE3 ILSND 12.26 ILSND 12.66 ILSND ILSND ILSND ILSND ILSND ILSND I																	
High Capacity Unbundled Local Loop - DS3 - Facility UE3 U					1152		12.26										
Termination per month UE3 UE3X 306.36 452.52 264.53 119.75 83.77 15.69 High Capacity Unbundled Local Loop - STS-1 - Paolity Termination per month UDLSX 1LBND 12.26 15.69 15.69 LOOP MAKE-UP UDLSX UDLSX UDLSX UDLSX 131.49 452.52 264.53 119.75 83.77 15.69 15.69 LOOP MAKE-UP UDLSX UDLSX UDLSX UDLSX 244.52 264.53 119.75 83.77 15.69 15.69 Loop Makeup - Preordering Without Reservation, per working or gener facility queried (Manual). UMK UMKLW 24.04 16.04					013	TLOND	12.20										<u> </u>
month UDLSX 1LSND 12.26 15.69 High Capacity Unbundled Local Loop - STS-1 - Facility Termination per month UDLSX UDLSX UDLSX 12.06 15.69 15.69 LOOP MAKE-UP Image: Comp Makeup - Preordering Without Reservation, per working or spare facility queried (Manual). UMK UMK 24.04 26.04 15.69 15.69 Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual). UMK UMK 26.04 26.40 16.00					UE3	UE3PX	306.36	452.52	264.53	119.75	83.77		15.69				
High Capacity Unbundled Local Loop - STS-1 - Facility UDLSX UDLS1 313.49 452.52 264.53 119.75 83.77 15.69 LOOP MAKE-UP Image: Comparison of the servation, per working or spare facility queried (Manual). Image: Comparison of the servation, per working or gueried (Manual). Image: Comparison of the servation, per spare facility queried (Manual). Image: Comparison of the servation, per spare facility queried (Manual). Image: Comparison of the servation, per working or gueried (Manual). Image: Comparison of the servation, per working or gueried (Manual). Image: Comparison of the servation, per working or gueried (Manual). Image: Comparison of the servation, per working or gueried (Manual). Image: Comparison of the servation, per working or gueried (Machanized). Image: Comparison of the servation, per working or gueried (Machanized). Image: Comparison of the servation of the																	
Termination per month UDLSX UDLS1 313.49 452.52 264.53 119.75 83.77 15.69 LOOP MAKE-UP Image: Coop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual). UMK UMK 24.04 <td< td=""><td></td><td></td><td></td><td></td><td>UDLSX</td><td>1L5ND</td><td>12.26</td><td></td><td></td><td></td><td></td><td></td><td>15.69</td><td></td><td></td><td></td><td>L</td></td<>					UDLSX	1L5ND	12.26						15.69				L
LOOP MAKE-UP Image: Constraint of the servation per working or spare facility queried (Manual). Image: Constraint of the servation per working or spare facility queried (Manual). Image: Constraint of the servation per working or spare facility queried (Manual). Image: Constraint of the servation per working or spare facility queried (Manual). Image: Constraint of the servation per working or spare facility queried (Manual). Image: Constraint of the servation per working or spare facility queried (Manual). Image: Constraint of the servation per working or spare facility queried (Manual). Image: Constraint of the servation per working or spare facility queried (Manual). Image: Constraint of the servation per working or spare facility queried (Manual). Image: Constraint of the servation per working or spare facility queried (Manual). Image: Constraint of the servation per working or spare facility queried (Manual). Image: Constraint of the servation per working or spare facility queried (Manual). Image: Constraint of the servation per working or spare facility queried (Manual). Image: Constraint of the servation per working or spare facility queried (Manual). Image: Constraint of the servation per working or spare facility queried (Manual). Image: Constraint of the servation per working or spare facility queried (Manual). Image: Constraint of the servation per working or spare facility queried (Manual). Image: Constraint of the servation per working or spare facility queried (Manual). Image: Constraint of the servation per working or servation per working or servation per system 24 Line Spare facility queried (Manual). Image: Constrainter per system 24 Line Spare facility queried (Manual).							313 /0	452 52	264 53	110 75	83 77		15 60				
Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual). UMK UMKK UMKLW 24.04 <					ODEOX	ODLOT	515.45	402.02	204.33	113.75	00.11		10.00				
Loop Makeup - Preordering With Reservation, per spare facility queried (Manual). UMK UMK UMKLP 25.49 25.49 25.49																	
queried (Manual).UMKUMKUMKLP25.4925.49Loop MakeupWith or Without Reservation, per working or spare facility queried (Mechanized)UMKPSUMK0.340.34 <td< td=""><td></td><td></td><td></td><td></td><td>UMK</td><td>UMKLW</td><td></td><td>24.04</td><td>24.04</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></td<>					UMK	UMKLW		24.04	24.04								
Loop MakeupWith or Without Reservation, per working or spare facility queried (Mechanized) UMK PSUMK 0.34 0.34 0.34 0.34 HIGH FREQUENCY SPECTRUM Image: Comparison of the comp								25.40	25 40								
spare facility queried (Mechanized) UMK PSUMK 0.34 0.34 0.34 0.34 HIGH FREQUENCY SPECTRUM Image: Constraint of the constra					UWIN	UNINLE		25.49	25.49								
LINE SHARING Image: Constraint of the starting splitter, per System 96 Line Capacity ULS ULSDA 216.22 189.21 0.00 178.38 0.00 15.69 Line Sharing Splitter, per System 24 Line Capacity ULS ULSDA 216.22 189.21 0.00 178.38 0.00 15.69 Line Sharing Splitter, per System 24 Line Capacity ULS ULSDB 54.05 189.21 0.00 178.38 0.00 15.69 Line Sharing Splitter, Per System 8 Line Capacity I ULS ULSDB 18.02 189.21 0.00 178.38 0.00 15.69 Line Sharing Splitter, Per System 8 Line Capacity I ULS ULSDB 18.02 189.21 0.00 178.38 0.00 15.69 Line Sharing OLEC Owned Splitter in CO-CFA activaton- deactivation (pt LSOD) ULS ULSDG 86.67 0.00 49.95 0.00 15.69 END USER ORDERING-CENTRAL OFFICE BASED-HIGH FREQUENCY SPECTRUM AKA LINE SHARING Image: Constraint of the starting - per Line Activation (BST owned Splitter) ULS ULSDC 0.61 18.55 10.62 10.04		spare facility queried (Mechanized)			UMK	PSUMK		0.34	0.34								
SPLITTERS-CENTRAL OFFICE BASED ULS ULSDA 216.22 189.21 0.00 178.38 0.00 15.69 Line Sharing Splitter, per System 94 Line Capacity ULS ULSDB 54.05 189.21 0.00 178.38 0.00 15.69 Line Sharing Splitter, per System 34 Line Capacity ULS ULSDB 54.05 189.21 0.00 178.38 0.00 15.69 Line Sharing Splitter, per System, 8 Line Capacity I ULS ULSDB 54.05 189.21 0.00 178.38 0.00 15.69 Line Sharing-DLEC Owned Splitter in CO-CFA activaton- deactivation (per LSOD) I ULS ULSDG 86.67 0.00 49.95 0.00 15.69 END USER ORDERING-CENTRAL OFFICE BASED-HIGH FREQUENCY SPECTRUM AKA LINE SHARING ILS ULSDC 0.61 18.55 10.62 10.04 4.93 15.69 Line Sharing - per Subsequent Activity per Line Rearrangement(BST Owned Splitter) ULS ULSCS 16.42 8.21 15.69 15.69 Line Sharing - per Subsequent Activity per Line Rearrangement(DLEC Owned Splitter) ULS																	
Line Sharing Splitter, per System 96 Line Capacity ULS ULSDA 216.22 189.21 0.00 178.38 0.00 15.69 Line Sharing Splitter, per System 74 Line Capacity ULS ULSDB 54.05 189.21 0.00 178.38 0.00 15.69 Line Sharing Splitter, Per System 74 Line Capacity I ULS ULSDB 54.05 189.21 0.00 178.38 0.00 15.69 Line Sharing Splitter, Per System 74 Line Capacity I ULS ULSDB 18.02 189.21 0.00 178.38 0.00 15.69 Line Sharing DLEC Owned Splitter in CO-CFA activaton- deactivation (per LSOD) ULS ULSDG 86.67 0.00 49.95 0.00 15.69 END USER ORDERING-CENTRAL OFFICE BASED-HIGH FREQUENCY SPECTRUM AKA LINE SHARING ULSDC 0.61 18.55 10.62 10.04 4.93 15.69 Line Sharing - per Subsequent Activity per Line Rearrangement(BST Owned Splitter) ULS ULSDS 16.42 8.21 15.69 15.69 Line Sharing - per Subsequent Activity per Line Rearrangement(DLEC Owned Splitter) ULS																	↓]
Line Sharing Splitter, per System 24 Line Capacity ULS ULSDB 54.05 189.21 0.00 178.38 0.00 15.69 Line Sharing Splitter, Per System, 8 Line Capacity I ULS ULSDB 18.02 189.21 0.00 178.38 0.00 15.69 Line Sharing Splitter, Per System, 8 Line Capacity I ULS ULSDB 18.02 189.21 0.00 178.38 0.00 15.69 Line Sharing OLEC Owned Splitter in CO-CFA activaton- deactivation (per LSOD) ULS ULSDG 86.67 0.00 49.95 0.00 15.69 END USER ORDERING-CENTRAL OFFICE BASED-HIGH FREQUENCY SPECTRUM AKA LINE SHARING					111.5		216.22	180.21	0.00	178.20	0.00		15.60				├──── │
Line Sharing Splitter, Per System, 8 Line Capacity I ULS ULSDB 18.02 189.21 0.00 178.38 0.00 15.69 Line Sharing -DLEC Owned Splitter in CO-CFA activaton- deactivation (pt LSOD) ULS ULSDG 86.67 0.00 49.95 0.00 15.69 END USER ORDERING-CENTRAL OFFICE BASED-HIGH FREQUENCY SPECTRUM AKA LINE SHARING ULS ULSC 0.61 18.55 10.62 10.04 4.93 15.69 Line Sharing - per Line Activation (BST owned Splitter) ULS ULSC 0.61 18.55 10.62 10.04 4.93 15.69 Line Sharing - per Subsequent Activity per Line Rearrangement(BST Owned Splitter) ULS ULSCS 16.42 8.21 15.69 <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>1</td> <td></td> <td></td> <td></td> <td></td> <td> </td>												1					
deactivation (per LSOD) ULS ULSDG 86.67 0.00 49.95 0.00 15.69 END USER ORDERING-CENTRAL OFFICE BASED-HIGH FREQUENCY SPECTRUM AKA LINE SHARING <th< td=""><td></td><td>Line Sharing Splitter, Per System, 8 Line Capacity</td><td>I</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></th<>		Line Sharing Splitter, Per System, 8 Line Capacity	I														
END USER ORDERING-CENTRAL OFFICE BASED-HIGH FREQUENCY SPECTRUM AKA LINE SHARING <th<< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></th<<>																	
Line Sharing - per Line Activation (BST owned Splitter) ULS ULSDC 0.61 18.55 10.62 10.04 4.93 15.69 Line Sharing - per Subsequent Activity per Line Rearrangement(BST Owned Splitter) ULS ULSDS 16.42 8.21 16.49 15.69 Line Sharing - per Subsequent Activity per Line Rearrangement(DLEC Owned Splitter) ULS ULSCS 16.42 8.21 15.69 15.69 Line Sharing - per Subsequent Activity per Line Rearrangement(DLEC Owned Splitter) ULS ULSCS 16.42 8.21 15.69 15.69 Line Sharing - per Line Activation (DLEC owned Splitter) ULS ULSCS 16.42 8.21 15.69 15.69				TDUM		ULSDG		86.67	0.00	49.95	0.00		15.69				┟────┤
Line Sharing - per Subsequent Activity per Line Rearrangement(BST Owned Splitter) ULS ULSDS 16.42 8.21 15.69 Line Sharing - per Subsequent Activity per Line Rearrangement(DLEC Owned Splitter) ULS ULSCS 16.42 8.21 15.69 Line Sharing - per Line Activition (DLEC Owned Splitter) ULS ULSCS 16.42 8.21 15.69 Line Sharing - per Line Activation (DLEC owned Splitter) I ULS ULSCC 0.61 47.44 19.31 20.67 12.74 15.69			SPEC				0.61	18 55	10.62	10.04	1 03	ł	15.60				┟────┤
Rearrangement(BST Owned Splitter) ULS ULSDS 16.42 8.21 15.69 Line Sharing - per Subsequent Activity per Line Rearrangement(DLEC Owned Splitter) ULS ULSCS 16.42 8.21 15.69 15.69 Line Sharing - per Line Activation (DLEC owned Splitter) I ULS ULSCS 16.42 8.21 15.69					010	51000	0.01	10.00	10.02	10.04	7.33		13.09				
Rearrangement(DLEC Owned Splitter) ULS ULSCS 16.42 8.21 15.69 Line Sharing - per Line Activation (DLEC owned Splitter) I ULS ULSCC 0.61 47.44 19.31 20.67 12.74 15.69	1	Rearrangement(BST Owned Splitter)			ULS	ULSDS		16.42	8.21				15.69				
Line Sharing - per Line Activation (DLEC owned Splitter) I ULS ULSCC 0.61 47.44 19.31 20.67 12.74 15.69																	
				<u> </u>			0.01			20.07	40.74						
					ULO	ULSUU	0.61	47.44	19.31	20.67	12.74	ł	15.69				<u> </u>
END USER ORDERING-CENTRAL OFFICE BASED																	
Line Splitting - per line activation DLEC owned splitter I UEPSR UEPSB UREOS 0.61			I		UEPSR UEPSB	UREOS	0.61										

UNBUN	NDLEI	O NETWORK ELEMENTS - South Carolina			1									Attachment:			ibit: B
CATEGO	DRY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Charge -
							Rec	Nonrec		Nonrecurring					Rates(\$)	•	·
								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Line Splitting - per line activation BST owned - physical Line Splitting - per line activation BST owned - virtual			UEPSR UEPSB UEPSR UEPSB	UREBP UREBV	0.61 0.61	37.09 37.09	21.24 21.24	20.07 20.07	9.85 9.85		15.69 15.69				<u> </u>
F	REMOT	E SITE HIGH FREQUENCY SPECTRUM			UEFSK UEFSD	UKEDV	0.01	37.09	21.24	20.07	9.65		15.69				
		ERS-REMOTE SITE															
	-	Remote Site Line Share BellSouth Owned Splitter, 24 Port	I		ULS	ULSRB	54.05	378.42	0.00	356.76	0.00		15.69				
		Remote Site Line Share Cable Pair Activation CLEC Owned at															
		RS and Deactivation	1		ULS	ULSTG		74.38	0.00	46.77	0.00		15.69				
E	END US	SER ORDERING-REMOTE SITE HIGH FREQUENCY SPECTRU	MAKA	REMOT	E SITE LINE SHAR	NG											
		Remote Site Line Share Line Activationfor End User Served at RS, BST Splitter			ULS	ULSRC	0.61	37.09	21.24	20.07	9.85		15.69				
		RS Line Share Line Activation for End User served at RS, CLEC			ULS	ULSRC	0.01	37.09	21.24	20.07	9.65		15.69				
		Splitter	1		ULS	ULSTC	0.61	37.09	21.24	20.07	9.85		15.69				
UNBUND	DLED D	EDICATED TRANSPORT	-														
		INTEROFFICE CHANNEL DEDICATED TRANSPORT - minimu	m billin	g perio	od - below DS3=one	month, DS3/S	STS-1=four mor	nths									
	INTERC	OFFICE CHANNEL - DEDICATED TRANSPORT															
		Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade -			147.04	41 5207											
\vdash		Per Mile per month	ļ		U1TVX	1L5XX	0.0167										
		Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade - Facility Termination			U1TVX	U1TV2	24.30	40.63	27.47	16.77	6.91		15.69				
		Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade			UTIVA	01172	24.30	40.03	27.47	10.77	0.91		15.09				
		Rev Bat Per Mile per month			U1TVX	1L5XX	0.0167										
		Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat			01117	120/01	0.0101										
		Facility Termination			U1TVX	U1TR2	24.30	40.63	27.47	16.77	6.91		15.69				
		Interoffice Channel - Dedicated Transport - 4-Wire Voice Grade -	-														
		Per Mile per month			U1TVX	1L5XX	0.0167										
		Interoffice Channel - Dedicated Transport - 4- Wire Voice Grade						10.00									
		- Facility Termination Interoffice Channel - Dedicated Transport - 56 kbps - per mile			U1TVX	U1TV4	21.29	40.63	27.47	16.77	6.91		15.69				
		per month			U1TDX	1L5XX	0.0167										
		Interoffice Channel - Dedicated Transport - 56 kbps - Facility			OTIDA	TE3/01	0.0107										
		Termination			U1TDX	U1TD5	16.76	40.63	27.47	16.77	6.91		15.69				
		Interoffice Channel - Dedicated Transport - 64 kbps - per mile			-												
		per month			U1TDX	1L5XX	0.0167										
		Interoffice Channel - Dedicated Transport - 64 kbps - Facility															
		Termination			U1TDX	U1TD6	16.76	40.63	27.47	16.77	6.91		15.69				
		Interoffice Channel - Dedicated Channel - DS1 - Per Mile per			U1TD1	1L5XX	0.3415										
		month Interoffice Channel - Dedicated Tranport - DS1 - Facility			UTIDI	ILSAA	0.3415										
		Termination			U1TD1	U1TF1	77.14	89.47	81.99	16.39	14.48		15.69				
		Interoffice Channel - Dedicated Transport - DS3 - Per Mile per															
		month			U1TD3	1L5XX	8.02										
		Interoffice Channel - Dedicated Transport - DS3 - Facility															
		Termination per month			U1TD3	U1TF3	880.65	279.37	163.12	60.33	58.59		15.69				
		Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per															
		month Interoffice Channel - Dedicated Transport - STS-1 - Facility			U1TS1	1L5XX	8.02										
		Termination			U1TS1	U1TFS	880.55	279.37	163.12	60.33	58.59		15.69				
		CHANNEL - DEDICATED TRANSPORT			01101	01110	000.00	213.31	103.12	00.00	50.55		10.00				
		LOCAL CHANNEL DEDICATED TRANSPORT - minimum billin	g perio	d - belo	ow DS3=one month	DS3/STS-1=f	our months										
		Local Channel - Dedicated - 2-Wire Voice Grade			ULDVX	ULDV2	15.33	193.53	33.24	36.72	3.21		15.69				
		Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat			ULDVX	ULDR2	15.33	193.53	33.24	36.72	3.21		15.69				
		Local Channel - Dedicated - 4-Wire Voice Grade	L	<u> </u>	UNDVX	ULDV4	16.54	193.97	33.68	37.19	3.68	ļ	15.69				L
		Local Channel - Dedicated - DS1 - Zone 1 Local Channel - Dedicated - DS1 - Zone 2	ļ	1	ULDD1	ULDF1	42.62	177.87	154.06	22.24	15.30		15.69				
		Local Channel - Dedicated - DS1 - Zone 2 Local Channel - Dedicated - DS1 - Zone 3		2	ULDD1 ULDD1	ULDF1 ULDF1	70.32 190.68	177.87 177.87	154.06 154.06	22.24 22.24	15.30 15.30		15.69 15.69				
<u>├</u>		Local Channel - Dedicated - DS1 - 2016 3 Local Channel - Dedicated - DS3 - Per Mile per month		3	ULDD3	1L5NC	11.93	111.01	104.06	22.24	15.30	-	15.09				+
		Local Channel - Dedicated - DS3 - Fei Mile per month			ULDD3	ULDF3	446.00	452.52	264.53	119.75	83.77	1	15.69		-		
		Local Channel - Dedicated - STS-1- Per Mile per month	l		ULDS1	1L5NC	11.93	.02.02	201.00		00.77	1	.0.00	1		1	1
		Local Channel - Dedicated - STS-1 - Facility Termination	1	1	ULDS1	ULDFS	435.10	452.52	264.53	119.75	83.77	1	15.69	ĺ		l	

UNBUND	DLED	NETWORK ELEMENTS - South Carolina		-	-		-							Attachment:			bit: B
CATEGOR	RΥ	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'I
							Rec	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
DARK FIB																	
		Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction				41.500	07.05										
		Thereof per month - Local Channel NRC Dark Fiber - Local Channel			UDF UDF	1L5DC UDFC4	97.65	640.51	138.17	317.76	198.11		15.69				
		Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction			001	00104		040.51	130.17	317.70	190.11		13.09				
		Thereof per month - Interoffice Channel			UDF	1L5DF	36.41										
		NRC Dark Fiber - Interoffice Channel			UDF	UDF14		640.51	138.17	317.76	198.11		15.69				
		Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction															
		Thereof per month - Local Loop			UDF	1L5DL	97.65										
		NRC Dark Fiber - Local Loop			UDF	UDFL4		640.51	138.17	317.76	198.11		15.69				
8XX ACCE		EN DIGIT SCREENING															
		8XX Access Ten Digit Screening, Per Call 8XX Access Ten Digit Screening, Reservation Charge Per 8XX		-	OHD		0.0006673										
		Number Reserved			OHD	N8R1X		2.59	0.44				15.69				
		BXX Access Ten Digit Screening, Per 8XX No. Established W/O						2.59	0.44				15.09				ł
		POTS Translations			OHD			5.95	0.81	4.58	0.54		15.69				
		BXX Access Ten Digit Screening, Per 8XX No. Established With		1		1	i i	0.00	0.01		0.04	1	.0.00	İ			
		POTS Translations			OHD	N8FTX		5.95	0.81	4.58	0.54		15.69				
	1	BXX Access Ten Digit Screening, Customized Area of Service															
		Per 8XX Number			OHD	N8FCX		2.59	1.30				15.69				
		BXX Access Ten Digit Screening, Multiple InterLATA CXR															
		Routing Per CXR Requested Per 8XX No.			OHD	N8FMX		3.03	1.74				15.69				
		8XX Access Ten Digit Screening, Change Charge Per Request			OHD	N8FAX		3.03	0.44				15.69				
		BXX Access Ten Digit Screening, Call Handling and Destination Features			OHD	N8FDX		2.59	2.59				15 60				
		BXX Access Ten Digit Screening, w/ 8XX No. Delivery			OHD	NOFDA	0.0006673	2.59	2.59				15.69				-
		BXX Access Ten Digit Screening, w/ POTS No. Delivery		1	OHD		0.0006673										
LINE INFO		TION DATA BASE ACCESS (LIDB)			0.1.5		0.0000010										
	1	LIDB Common Transport Per Query			OQT		0.0000246										
		LIDB Validation Per Query			OQU		0.0138158										
		LIDB Originating Point Code Establishment or Change			OQT, OQU	NRPBX		34.40		42.18			15.69				
SIGNALIN																	
		CCS7 Signaling Connection, Per 56 Kbps Facility			UDB	TPP++	16.93	35.61	35.61	16.48	16.48						
		CCS7 Signaling Termination, Per STP Port			UDB	PT8SX	163.49										-
		CCS7 Signaling Usage, Per TCAP Message CCS7 Signaling Connection, Per link (A link)		-	UDB UDB	TPP++	0.0000692 16.93	35.61	35.61	16.48	16.48		15.69				
		CCS7 Signaling Connection, Per link (B link) (also known as D		-	008	16677	10.95	35.01	35.01	10.40	10.40		15.09				
		ink)			UDB	TPP++	16.93	35.61	35.61	16.48	16.48		15.69				
	(CCS7 Signaling Usage, Per ISUP Message			UDB		0.0000173										
		CCS7 Signaling Usage Surrogate, per link per LATA			UDB	STU56	791.37										
	(CCS7 Signaling Point Code, per Originating Point Code															
		Establishment or Change, per STP affected			UDB	CCAPO		29.08	29.08	35.65	35.65		15.69				
		CCS7 Signaling Point Code, per Destination Point Code															
5044 055		Establishment or Change, Per Stp Affected			UDB	CCAPD		29.08	29.08	35.65	35.65		15.69				-
E911 SER		Local Channel - Dedicated - 2-wr Voice Grade					15.33	193.53	33.24	36.72	3.21		15.69				
		Interoffice Transport - Dedicated - 2-wr Voice Grade Interoffice Transport - Dedicated - 2-wr Voice Grade Per Mile	ł			1	0.0167	193.03	əə.24	30.72	3.21	1	15.09				1
		Interoffice Transport - Dedicated - 2-wr Voice Grade Per Facility		1			5.0107					1					
		Termination		1			24.30	40.63	27.47	16.77	6.91		15.69				
		Local Channel - Dedicated - DS1 - Zone 1				1	42.62	177.87	154.06	22.24	15.30		15.69				
		Local Channel - Dedicated - DS1 - Zone 2					70.32	177.87	154.06	22.24	15.30		15.69				
		Local Channel - Dedicated - DS1 - Zone 3					190.68	177.87	154.06	22.24	15.30		15.69				
		nteroffice Transport - Dedicated - DS1 Per Mile	ļ	L			0.3415										
				1			77.44	00.17	04.00	40.00	44.40		45.00				
		Interoffice Transport - Dedicated - DS1 Per Facility Termination E (CNAM) SERVICE				-	77.14	89.47	81.99	16.39	14.48		15.69				-
CALLING		CNAM) SERVICE		-	OQV	+	├	23.00	23.00	21.15	21.15	+	15.69	{		1	<u> </u>

	RATE ELEMENTS	Interi									Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
		m	Zone	BCS	USOC			RATES(\$)			Submitted Elec per LSR	Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec	urring	Nonrecurring	Disconnect				Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	CNAM For DB Owners - Service Provisioning With Point Code															1 1
	Establishment			OQV			993.09	734.47	269.53	198.18		15.69				ļ!
	CNAM For Non DB Owners - Service Provisioning With Point Code Establishment			OQV			343.09	245.69	275.87	198.18		15.69				1 1
	CNAM for DB Owners, Per Query			OQV		0.0010433	343.09	245.09	213.01	190.10		13.09				┟────┦
	CNAM for Non DB Owners, Per Query			OQV		0.0010433										l
LNP Query Ser											1					1
	LNP Charge Per query					0.0008837										
	LNP Service Establishment Manual						25.09	25.09	23.07	23.07		15.69				ļ!
	LNP Service Provisioning with Point Code Establishment						594.82	303.88	269.53	198.18		15.69				└──── ┘
OPERATOR C	ALL PROCESSING															↓ ′
	Oper. Call Processing - Oper. Provided, Per Min Using BST LIDB					1.20										
	Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIDB					1.24										
	Oper. Call Processing - Fully Automated, per Call - Using BST LIDB					0.20										1
	Oper. Call Processing - Fully Automated, per Call - Using Foreign LIDB					0.20										
INWARD OPER	ATOR SERVICES															
	Inward Operator Services - Verification, Per Minute					1.15										
	Inward Operator Services - Verification and Emergency Interrupt															1
	- Per Minute					1.15										ļ!
	PERATOR CALL PROCESSING				_											ļ!
Facility	r based CLEC Recording of Custom Branded OA Announcement				CBAOS		7,000.00	7,000.00				15.69				ļ/
	Loading of Custom Branded OA Announcement per shelf/NAV				CBAUS		7,000.00	7,000.00				15.69				┟────┦
	per OCN				CBAOL		500.00	500.00				15.69				1 1
UNEP																l – – I
	Recording of Custom Branded OA Announcement						7,000.00	7,000.00				15.69				
	Loading of Custom Branded OA Announcement per shelf/NAV per OCN						500.00	500.00				15.69				
Unbrar	nding via OLNS for UNEP CLEC															
	Loading of OA per OCN (Regional)						1,200.00	1,200.00				15.69				
	SSISTANCE SERVICES															└──── ┘
DIREC	TORY ASSISTANCE ACCESS SERVICE				-	0.075										ا
DIREC	Directory Assistance Access Service Calls, Charge Per Call TORY ASSISTANCE CALL COMPLETION ACCESS SERVICE (D					0.275										ļ/
DIREC	Directory Assistance Call Completion Access Service (DACC),	JACC)			-											i
	Per Call Attempt					0.10										1 1
DIRECTORY A	SSISTANCE SERVICES															l – – I
DIREC	TORY ASSISTANCE DATA BASE SERVICE (DADS)															(
	Directory Assistance Data Base Service Charge Per Listing					0.04										
	Directory Assistance Data Base Service, per month				DBSOF	150.00										ļ!
BRANDING - D	IRECTORY ASSISTANCE															ļ!
Facility	/ Based CLEC				_											└────┘
	Recording and Provisioning of DA Custom Branded Announcement			AMT	CBADA		6,000.00	6,000.00				15.69				
	Loading of Custom Branded Announcement per Switch			AMT	CBADC		1,170.00	1,170.00				15.69				└──── ┘
UNEP					+	┟───┤	2 000 00	2 000 00			├ ───┥	45.00				└──── [/]
<u> </u>	Recording of DA Custom Branded Announcement Loading of DA Custom Branded Announcement per Switch per				+	├	3,000.00	3,000.00	┝───┤		-	15.69				
	OCN						1,170.00	1,170.00				15.69				1 1
Unbra	nding via OLNS for UNEP CLEC				1	<u> </u>	1,170.00	1,170.00				10.00				
	Loading of DA per OCN (1 OCN per Order)					† †	420.00	420.00				15.69				l
	Loading of DA per Switch per OCN						16.00	16.00				15.69				<u> </u>
SELECTIVE R																
	Selective Routing Per Unique Line Class Code Per Request Per Switch				USRCR		84.89	84.89	14.14	14.14		15.69				1 7

UNBU	NDLE	D NETWORK ELEMENTS - South Carolina												Attachment:	2	Exhi	bit: B
CATEG	ORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
							Rec	Nonrec		Nonrecurring					Rates(\$)		
							Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
VIRTU/	L COL	LOCATION															
		Virtual Collocation - Application Cost			AMTES	EAF		1,207.95	1,207.95	0.51	0.51		15.69				
		Virtual Collocation - Cable Installation Cost, per cable			AMTES	ESPCX	0.05	794.22	794.22	22.54	22.54		15.69				
		Virtual Collocation - Floor Space, per sq. ft.			AMTFS AMTFS	ESPVX ESPAX	3.95 9.19										
		Virtual Collocation - Power, per fused amp Virtual Collocation - Cable Support Structure, per entrance	-		AIVITES	ESPAX	9.19										
		cable			AMTES	ESPSX	18.66										
		Cable			UEANL,UEA,UDN,U	LOI OX	10.00										
					DC,UAL,UHL,UCL,U EQ, AMTFS, UDL, UNCVX, UNCDX,												
		Virtual Collocation - 2-wire Cross Connects (loop)			UNCNX	UEAC2	0.0317	12.32	11.83	6.04	5.45		15.69				
		Virtual Collocation - 4-wire Cross Connects (loop)			UEA,UHL,UCL,UDL, AMTFS, UAL, UDN, UNCVX, UNCDX AMTFS,UDL12, UDL03, U1T48,	UEAC4	0.0634	12.42	11.90	6.40	5.74		15.69				
					U1T12, U1T03, ULDO3, ULD12,	01005	0.00	00.04	45.00	7.40	5.00		45.00				
		Virtual Collocation - 2-Fiber Cross Connects			ULD48, UDF AMTFS,UDL12,	CNC2F	2.86	20.94	15.23	7.40	5.93		15.69				
		Virtual Collocation - 4-Fiber Cross Connects			UDLO3, U1T48, U1T12, U1T03, ULDO3, ULD12, ULD48, UDF	CNC4F	5.71	25.61	19.90	9.73	8.26		15.69				
		Virtual collocation - Special Access & UNE,cross-connect per DS1			USL,ULC,AMTFS, ULR, UXTD1, UNC1X, ULDD1, U1TD1, USLEL, UNLD1	CNC1X	1.12	22.08	15.96	6.42	5.80		15.69				
		Virtual collocation - Special Access & UNE, cross-connect per DS3			USL, ULC, AMTFS, U E3, U1TD3, UXTS1, UXTD3, UNC3X, UNCSX, ULDD3, U1TS1, ULDS1, UDLSX, UNLD3	CND3X	14.21	20.94	15.23	7.39	5.93		15.69				
		Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear foot			AMTFS	VE1CB	0.0022										
		Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per linear ft			AMTFS	VE1CD	0.0033										
		Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure,per cable Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax			AMTES	VE1CC		536.56									
		Cable Support Structure, per cable	1		AMTFS	VE1CE		536.56									
		Virtual Collocation Cable Records - per request			AMTFS	VE1BA		760.98	489.20	133.29	133.29						
		Virtual Collocation Cable Records - VG/DS0 Cable, per cable record			AMTFS	VE1BB		327.65	327.65	189.54	189.54						
		Virtual Collocation Cable Records - VG/DS0 Cable, per each 100 pair			AMTES	VE1BC		4.82	4.82	5.91	5.91						
		Virtual Collocation Cable Records - DS1, per T1TIE	1		AMTES	VE1BD		2.26	2.26	2.77	2.77			1		1	1
		Virtual Collocation Cable Records - DS3, per T3TIE	1		AMTES	VE1BE		7.90	7.90	9.68	9.68					1	
		Virtual Collocation Cable Records - Fiber Cable, per 99 fiber records			AMTES	VE1BF		84.68	84.68	77.30	77.30						
		Virtual collocation - Security Escort - Basic, per half hour			AMTFS	SPTBX		16.96	10.75				15.69				
		Virtual collocation - Security Escort - Overtime, per half hour			AMTFS	SPTOX		22.10	13.89				15.69				
		Virtual collocation - Security Escort - Premium, per half hour			AMTFS	SPTPX		27.23	17.02				15.69				
		Virtual collocation - Maintenance in CO - Basic, per half hour			AMTFS	CTRLX		27.99	10.75				15.69				

UNBUNDI	ED NETWORK ELEMENTS - South Carolina												Attachment:			oit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
						Rec	Nonred		Nonrecurring					Rates(\$)		
					-		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Virtual collocation - Maintenance in CO - Overtime, per half hour			AMTFS	SPTOM		36.56	13.89				15.69				
	Virtual conocation - Maintenance in OO - Overtime, per han nour			AWITO			50.50	13.03				10.00				
	Virtual collocation - Maintenance in CO - Premium per half hour			AMTFS	SPTPM		45.12	17.02				15.69				
VIRTUAL C	DLLOCATION															
	Virtual Collocation - 2-wire Cross Connect, Exchange Port 2- Wire Analog - Res			UEPSR	VE1R2	0.0317	12.32	11.83	6.04	5.45		15.69				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2- Wire Line Side PBX Trunk - Bus			UEPSP	VE1R2	0.0317	12.32	11.83	6.04	5.45		15.69				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire															
	Voice Grade PBX Trunk - Res			UEPSE	VE1R2	0.0317	12.32	11.83	6.04	5.45	<u> </u>	15.69				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Analog Bus			UEPSB	VE1R2	0.0317	12.32	11.83	6.04	5.45		15.69				
	Virtual Collocation 2-Wire Cross Connect, Exchnage Port 2-Wire ISDN			UEPSX	VE1R2	0.0317	12.32	11.83	6.04	5.45		15.69				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire ISDN			UEPTX	VE1R2	0.0317	12.32	11.83	6.04	5.45	ĺ	15.69				
	Virtual Collocation 4-Wire Cross Connect, Exchange Port 4-Wire ISDN DS1			UEPEX	VE1R4	1.12	22.08	15.96	6.42	5.80	1	15.69				
	DLLOCATION				VC 1R4	1.12	22.08	15.96	0.42	5.80	<u> </u>	15.69				
VINTOAL C	Virtual Collocation-2 Wire Cross Connects (Loop) for Line Splitting			UEPSR, UEPSB	VE1LS	0.0317	12.32	11.83	6.04	5.45		15.69				
PHYSICAL (COLLOCATION			OEI OR, OEI OD	VEILO	0.0317	12.52	11.00	0.04	3.43		10.00				
	Physical Collocation-2 Wire Cross Connects (Loop) for Line Splitting			UEPSR, UEPSB	PE1LS	0.0341	12.32	11.83	6.04	5.45		15.69				
AIN SELEC	IVE CARRIER ROUTING				1 2 1 2 0	0.0011	12.02	11100	0.01	0.10		10.00				
	Regional Service Establishment			SRC	SRCEC		101,324.34	101,324.34	8,609.85	8,609.85		15.69				
	End Office Establishment			SRC	SRCEO		175.66	175.66	1.70	1.70		15.69				
	Query NRC, per query			SRC		0.0035036										
AIN - BELL	AIN SMS ACCESS SERVICE AIN SMS Access Service - Service Establishment, Per State,															
	Initial Setup			A1N	CAMSE		39.53	39.53	40.78	40.78		15.69				
	AIN SMS Access Service - Port Connection - Dial/Shared Access			A1N	CAMDP		7.85	7.85	9.11	9.11		15.69				
	AIN SMS Access Service - Port Connection - ISDN Access			A1N	CAM1P		7.85	7.85	9.11	9.11		15.69				
	AIN SMS Access Service - User Identification Codes - Per User ID Code			A1N	CAMAU		35.08	35.08	27.12	27.12		15.69				
	AIN SMS Access Service - Security Card, Per User ID Code,															
	Initial or Replacement			A1N	CAMRC		41.98	41.98	11.74	11.74		15.69				
	AIN SMS Access Service - Storage, Per Unit (100 Kilobytes)					0.0027					<u> </u>					
	AIN SMS Access Service - Session, Per Minute AIN SMS Access Service - Company Performed Session, Per				-	0.7121										
	Minute					0.8364										
AIN - BELLS	OUTH AIN TOOLKIT SERVICE				İ											
	AIN Toolkit Service - Service Establishment Charge, Per State,															
	Initial Setup			CAM	BAPSC BAPVX		39.53 4,211.54	39.53	40.78	40.78	ļ	15.69	ļ		ļ	
	AIN Toolkit Service - Training Session, Per Customer AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per				DAMVX		4,211.54	4,211.54	0.00	0.00		15.69				
	DN, Term. Attempt				BAPTT		7.85	7.85	9.11	9.11		15.69				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay				BAPTD		7.85	7.85	9.11	9.11		15.69				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate				BAPTM		7.85	7.85	9.11	9.11		15.69				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per															
	DN, 10-Digit PODP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per				BAPTO		34.54	34.54	14.39	14.39		15.69				
	DN, CDP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per				BAPTC		34.54	34.54	14.39	14.39		15.69				
	DN, Feature Code				BAPTF		34.54	34.54	14.39	14.39		15.69				

JNBUNDLE	D NETWORK ELEMENTS - South Carolina		_		-								Attachment:			ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	AIN Toolkit Service - Query Charge, Per Query					0.0558238										
	AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit Subscription, Per Node, Per Query					0.0069214										
	AIN Toolkit Service - SCP Storage Charge, Per SMS Access Account, Per 100 Kilobytes					0.07										
	AIN Toolkit Service - Monthly report - Per AIN Toolkit Service Subscription			САМ	BAPMS	11.87	7.85	7.85	5.52	5.52		15.69				
	AIN Toolkit Service - Special Study - Per AIN Toolkit Service Subscription			CAM	BAPLS	3.51	8.68	8.68				15.69				
	AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service Subscription			САМ	BAPDS	8.48	7.85	7.85	5.52	5.52		15.69				
	AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit Service Subscription			САМ	BAPES	0.12	8.68	8.68				15.69				
ENHANCED E	XTENDED LINK (EELs)			-	-	-										
NOTE	New Density Zone 1 EELs are available in the following MSAs	s: Orlan	do, FL	; Miami, FL; Ft. Lau	derdale, FL; A	Atlanta, Ga; Nev	v Orleans, LA,									
	Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-															
	In all states, EEL network elements shown below also apply to												UNEs.(Non-re	curring rates	do not apply	r.)
	In All States the EEL network elements apply to ordinarily cor				itch As Is Cha	arge.) When or	dering ordinar	ily combined i	network elemer	nts, Non-recur	ing rates d	o apply.				
2-WIR	E VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INT First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport	EROFF	ICE TR	ANSPORT (EEL)												
	Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed		1	UNCVX	UEAL2	16.68	105.98	68.43	53.05	10.61		15.69				
	Transport Combination - Zone 2		2	UNCVX	UEAL2	23.13	105.98	68.43	53.05	10.61		15.69				
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 3		3	UNCVX	UEAL2	28.46	105.98	68.43	53.05	10.61		15.69				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile per month			UNC1X	1L5XX	0.27										
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination per month			UNC1X	U1TF1	61.71	89.47	81.99	16.39	14.48		15.69				
	DS1 Channelization System Per Month			UNC1X	MQ1	107.57	91.24	62.71	10.56	9.81		15.69				
	Voice Grade COCI - DS1 To Ds0 Interface - Per Month			UNCVX	1D1VG	0.56	6.59	4.73				15.69				
	Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL2	16.68	105.98	68.43	53.05	10.61		15.69				
	Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL2	23.13	105.98	68.43	53.05	10.61		15.69				
	Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL2	28.46	105.98	68.43	53.05	10.61		15.69				
	Voice Grade COCI - DS1 to DS0 Channel System combination - per month			UNCVX	1D1VG	0.56	6.59	4.73				15.69				
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNC1X	UNCCC		5.61	5.61	7.00	7.00		15.69				
4-WIR	E VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INT	EROFF	ICE TR													1
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL4	32.59	132.38	94.83	59.35	14.61		15.69				
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL4	43.89	132.38	94.83	59.35	14.61		15.69				
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL4	43.38	132.38	94.83	59.35	14.61		15.69				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month			UNC1X	1L5XX	0.27	102.00	04.00	00.00	14.01		10.00				
	Interoffice Transport - Dedicated - DS1 - Facility Termination Per Month			UNC1X	U1TF1	61.71	89.47	81.99	16.39	14.48		15.69				
	Channelization - Channel System DS1 to DS0 combination Per Month			UNC1X	MQ1	107.57	91.24	62.71	10.59	9.81		15.69				
	Voice Grade COCI - DS1 to DS0 Channel System combination - per month			UNCVX	1D1VG	0.56	6.59	4.73	10.56	3.01		15.69				
	per month Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone 1				1D1VG UEAL4	0.56 32.59	132.38	94.83	59.35	14.61		15.69				1
UNBUNDLE	D NETWORK ELEMENTS - South Carolina												Attachment:	2	Exhi	bit: B
----------	---	-------------	-------	-----------------	----------------	---------------	--------------	--------------	--------------	-------	-------	---	--	---	----------	---
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonred		Nonrecurring					Rates(\$)		
	Additional 4-Wire Analog Voice Grade Loop in same DS1						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL4	43.89	132.38	94.83	59.35	14.61		15.69				
	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL4	43.38	132.38	94.83	59.35	14.61		15.69				
	Voice Grade COCI - DS1 to DS0 Channel System combination - per month			UNCVX	1D1VG	0.56	6.59	4.73				15.69				
	Nonrecurring Currently Combined Network Elements Switch -As-									=						
4.WID	Is Charge 56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1		EEICE	UNC1X	UNCCC		5.61	5.61	7.00	7.00		15.69				
4-99160	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice	INTERC		TRANSPORT (EEL)												
	Transport Combination - Zone 1		1	UNCDX	UDL56	29.93	126.66	89.12	59.35	14.61		15.69				
	First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2		2	UNCDX	UDL56	33.99	126.66	89.12	59.35	14.61		15.69				
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3		3	UNCDX	UDL56	34.74	126.66	89.12	59.35	14.61		15.69				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month			UNC1X	1L5XX	0.27										
	Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Month Channelization - Channel System DS1 to DS0 combination Per			UNC1X	U1TF1	61.71	89.47	81.99	16.39	14.48		15.69				
	Month OCU-DP COCI (data) - DS1 to DS0 Channel System - per			UNC1X	MQ1	107.57	91.24	62.71	10.56	9.81		15.69				
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1			UNCDX	1D1DD	1.19	6.59	4.73				15.69				
	Interoffice Transport Combination - Zone 1 Additional 4-Wire 56Kbps Digital Grade Loopin same DS1		1	UNCDX	UDL56	29.93	126.66	89.12	59.35	14.61		15.69				
	Interoffice Transport Combination - Zone 2 Additional 4-Wire 56Kbps Digital Grade Loopin same DS1		2	UNCDX	UDL56	33.99	126.66	89.12	59.35	14.61		15.69				
	Interoffice Transport Combination - Zone 3 OCU-DP COCI (data) - DS1 to DS0 Channel System -		3	UNCDX	UDL56	34.74	126.66	89.12	59.35	14.61		15.69				
	combination per month (2.4-64kbs) Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNCDX UNC1X	1D1DD UNCCC	1.19	6.59 5.61	4.73 5.61	7.00	7.00		15.69 15.69				
4-WIRE	64 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1	INTERC	FFICE				5.61	5.01	7.00	7.00		15.69				
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice		1													
	Transport Combination - Zone 1 First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice		1	UNCDX	UDL64	29.93	126.66	89.12	59.35	14.61		15.69				
	Transport Combination - Zone 2 First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice		2	UNCDX	UDL64	33.99	126.66	89.12	59.35	14.61		15.69				
	Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month		3	UNCDX UNC1X	UDL64 1L5XX	34.74 0.27	126.66	89.12	59.35	14.61		15.69				
	Per Wonth Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Month			UNC1X	U1TF1	61.71	89.47	81.99	16.39	14.48		15.69				
	Channelization - Channel System DS1 to DS0 combination Per Month			UNC1X	MQ1	107.57	91.24	62.71	10.55	9.81		15.69				
	OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs)			UNCDX	1D1DD	1.19	6.59	4.73				15.69				
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 1		1	UNCDX	UDL64	29.93	126.66	89.12	59.35	14.61		15.69				
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 2		2	UNCDX	UDL64	33.99	126.66	89.12	59.35	14.61		15.69				
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 3		3	UNCDX	UDL64	34.74	126.66	89.12	59.35	14.61		15.69				
	OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs) Nonrecurring Currently Combined Network Elements Switch -As-			UNCDX	1D1DD	1.19	6.59	4.73				15.69				
	Is Charge 5 S1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTE 5 S1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTE				UNCCC		5.61	5.61	7.00	7.00		15.69				

UNBUNDLE	D NETWORK ELEMENTS - South Carolina		-	n									Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)		
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice	-					First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Transport - Zone 1		1	UNC1X	USLXX	90.87	253.03	157.89	44.80	11.73		15.69				1
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2		2	UNC1X	USLXX	155.43	253.03	157.89	44.80	11.73		15.69				
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3		3	UNC1X	USLXX	261.89	253.03	157.89	44.80	11.73		15.69				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month			UNC1X	1L5XX	0.27										
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Month			UNC1X	U1TF1	61.71	89.47	81.99	16.39	14.48		15.69				
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge	-		UNC1X	UNCCC		5.61	5.61	7.00	7.00		15.69				
4-WIRE	DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTI	EROFFI	CE TR/	ANSPORT (EEL)												↓
	First DS1Loop in DS3 Interoffice Transport Combination - Zone		1	UNC1X	USLXX	90.87	253.03	157.89	44.80	11.73		15.69				
	First DS1Loop in DS3 Interoffice Transport Combination - Zone 2		2	UNC1X	USLXX	155.43	253.03	157.89	44.80	11.73		15.69				
	First DS1Loop in DS3 Interoffice Transport Combination - Zone		3	UNC1X	USLXX	261.89	253.03	157.89	44.80	11.73		15.69				
	Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Month			UNC3X	1L5XX	6.42										
	Interoffice Transport - Dedicated - DS3 - Facility Termination per month			UNC3X	U1TF3	704.52	279.37	163.12	60.33	58.59		15.69				
	DS3 to DS1 Channel System combination per month			UNC3X	MQ3	144.02 8.64	178.54 6.59	94.18	33.33	31.90		15.69				ł
	DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 1		1	UNC1X UNC1X	UC1D1 USLXX	90.87	253.03	4.73	44.80	11.73		15.69 15.69				
	Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2		2	UNC1X	USLXX	155.43	253.03	157.89	44.80	11.73		15.69				
	Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 3		3	UNC1X	USLXX	261.89	253.03	157.89	44.80	11.73		15.69				
	DS3 Interface Unit (DS1 COCI) combination per month		-	UNC1X	UC1D1	8.64	6.59	4.73				15.69				
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge	-		UNC3X	UNCCC		5.61	5.61	7.00	7.00		15.69				
2-WIRE	VOICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE IN	TEROFF	ICE TR													
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL2	16.68	105.98	68.43	53.05	10.61		15.69				
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL2	23.13	105.98	68.43	53.05	10.61		15.69				
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL2	28.46	105.98	68.43	53.05	10.61		15.69				
	Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month			UNCVX	1L5XX	0.0134										
	Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility Termination per month			UNCVX	U1TV2	19.44	40.63	27.47	16.77	6.91		15.69				
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNCVX	UNCCC		5.61	5.61	7.00	7.00		15.69				
4-WIRE	VOICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INT	TEROFF	FICE TR	ANSPORT (EEL)												ļ
	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL4	32.59	132.38	94.83	59.35	14.61		15.69				
	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL4	43.89	132.38	94.83	59.35	14.61		15.69				
	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL4	43.38	132.38	94.83	59.35	14.61		15.69				
	Interoffice Transport - Dedicated - 4-wire VG combination - Per Mile Per Month			UNCVX	1L5XX	0.0134										
	Interoffice Transport - Dedicated - 4- Wire Voice Grade combination - Facility Termination per month			UNCVX	U1TV4	17.03	40.63	27.47	16.77	6.91		15.69				

UNBUNDL	ED NETWORK ELEMENTS - South Carolina		1										Attachment:			ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec	urring	Nonrecurring	J Disconnect			OSS	Rates(\$)	•	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Nonrecurring Currently Combined Network Elements Switch -As-									=						
Dear	IS Charge				UNCCC		5.61	5.61	7.00	7.00		15.69				───
D33 L	High Capacity Unbundled Local Loop - DS3 combination - Per		NSFUR													
	Mile per month			UNC3X	1L5ND	12.26										
	High Capacity Unbundled Local Loop - DS3 combination -															
	Facility Termination per month			UNC3X	UE3PX	306.36	452.52	264.53	119.75	83.77		15.69				
	Interoffice Transport - Dedicated - DS3 - Per Mile per month			UNC3X	1L5XX	6.42										
	Interoffice Transport - Dedicated - DS3 combination - Facility					704.50										
	Termination per per month Nonrecurring Currently Combined Network Elements Switch -As-			UNC3X	U1TF3	704.52	279.37	163.12	60.33	58.59		15.69				
	Is Charge			UNC3X	UNCCC		5.61	5.61	7.00	7.00		15.69				
STS1	DIGITAL EXTENDED LOOP WITH DEDICATED STS1 INTEROFI	FICE TF	RANSPO		311000		0.01	5.01	7.00	7.00		10.09				<u> </u>
	High Capacity Unbundled Local Loop - STS1 combination - Per			. ,	1	i i										
	Mile per month			UNCSX	1L5ND	12.26										
	High Capacity Unbundled Local Loop - STS1 combination -															
	Facility Termination per month			UNCSX	UDLS1	313.49	452.52	264.53	119.75	83.77		15.69				
	Interoffice Transport - Dedicated - STS1 combination - Per Mile			UNCSX	1L5XX	6.42										
	per month Interoffice Transport - Dedicated - STS1 combination - Facility			UNCSX	ILSAA	6.42										
	Termination per month			UNCSX	U1TFS	704.44	279.37	163.12	60.33	58.59		15.69				
	Nonrecurring Currently Combined Network Elements Switch -As-			ONCOX	01110	704.44	213.51	105.12	00.00	50.55		15.05				
	Is Charge			UNCSX	UNCCC		5.61	5.61	7.00	7.00		15.69				
2-WIF	E ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPOR	RT (EEL)													1
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination															
	Transport - Zone 1		1	UNCNX	U1L2X	25.21	117.58	80.03	53.05	10.61		15.69				
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination		2		U1L2X	20.70	447.50	00.00	52.05	10.01		45.00				
	Transport - Zone 2 First 2-Wire ISDN Loop in a DS1 Interoffice Combination		2	UNCNX	UILZX	32.76	117.58	80.03	53.05	10.61		15.69				
	Transport - Zone 3		3	UNCNX	U1L2X	37.70	117.58	80.03	53.05	10.61		15.69				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile		Ŭ	UNC1X	1L5XX	0.27	117.00	00.00	00.00	10.01		10.00				
	Interoffice Transport - Dedicated - DS1 combiniton - Facility															1
	Termination per month			UNC1X	U1TF1	61.71	89.47	81.99	16.39	14.48		15.69				
	Channelization - Channel System DS1 to DS0 combination -															
	per month			UNC1X	MQ1	107.57	91.24	62.71	10.56	9.81		15.69				
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per month			UNCNX	UC1CA	2.56	6.59	4.73				15.69				
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport			UNCINA	UCICA	2.50	0.59	4.73	-			15.69				
	Combination - Zone 1		1	UNCNX	U1L2X	25.21	117.58	80.03	53.05	10.61		15.69				
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport				-											
	Combination - Zone 2		2	UNCNX	U1L2X	32.76	117.58	80.03	53.05	10.61		15.69				
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport															
	Combination - Zone 3		3	UNCNX	U1L2X	37.70	117.58	80.03	53.05	10.61		15.69				
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System			UNCNX	UC1CA	2.56	6 50	4.73				15.69				
├── ┼──	combintaion- per month Nonrecurring Currently Combined Network Elements Switch -As-				UCICA	2.30	6.59	4.73				69.61				<u> </u>
	Is Charge			UNC1X	UNCCC		5.61	5.61	7.00	7.00		15.69				
4-WIR	E DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 IN	TEROF	FICE T				2.01	2.01							İ	
	First DS1 Loop in STS1 Interoffice Transport Combination -				1											
	Zone 1		1	UNC1X	USLXX	90.87	253.03	157.89	44.80	11.73		15.69				
	First DS1 Loop in STS1 Interoffice Transport Combination -		~			455.40	050.00	457.00		44		45.00				
├───┼───	Zone 2		2	UNC1X	USLXX	155.43	253.03	157.89	44.80	11.73		15.69				l
	First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3		3	UNC1X	USLXX	261.89	253.03	157.89	44.80	11.73		15.69				
	Interoffice Transport - Dedicated - STS1 combination - Per Mile		5		30L/M	201.09	200.00	157.09	44.00	11.73		15.09				<u> </u>
	Per Month			UNCSX	1L5XX	6.42										
	Interoffice Transport - Dedicated - STS1 combination - Facility															
	Termination			UNCSX	U1TFS	704.44	279.37	163.12	60.33	58.59		15.69				
	STS1 to DS1 Channel System conbination per month			UNCSX	MQ3	144.02	178.54	94.18	33.33	31.90		15.69				

CATEGORY RATE ELEMENTS Interi m Zone BCS USOC RATES(\$) Svc Order Elec per LSR Svc Order Manually per LSR Svc Order m Svc Order Manually per LSR	Attachment: 2 Incremental Incremental Charge - Charge - Manual Svc Order vs. Order vs. Electronic- 1st Add'l OSS Rates(\$) SOMAN SOMAN SOMAN	Charge - Charge - Manual Svc Manual S Order vs. Order vs
CATEGORY RATE ELEMENTS Intering Zone BCS USOC RATES(\$) Kates(\$) Electors Manually per LSR Manually per LS	Manual Svc Order vs. Electronic- 1st OSS Rates(\$)	Manual Svc Manual S Order vs. Order vs Electronic- Disc 1st Disc Add
CATEGORY RATE ELEMENTS Intell m Zone BCS USOC RATES(\$) Nonrecurring Disconnect per LSR <th< td=""><td>Order vs. Electronic- 1st OSS Rates(\$)</td><td>Order vs. Electronic- Disc 1st Disc Add</td></th<>	Order vs. Electronic- 1st OSS Rates(\$)	Order vs. Electronic- Disc 1st Disc Add
m m	Electronic- 1st Add'l OSS Rates(\$)	Electronic- Electronic Disc 1st Disc Add
Image: Nonsecuring Control in the image: Nonsecuring Disconnect Nonsecuring Disconnect Nonsecuring Disconnect Image: Nonsecuring Disconnect Image: Nonsecuring Disconnect First Add'I First Add'I SOMEC SOMAN Image: Disconnect Image: Disconnect Image: Disconnect First Add'I First Add'I SOMEC SOMEC SOMAN Image: Disconnect <td>1st Add'l OSS Rates(\$)</td> <td>Disc 1st Disc Add</td>	1st Add'l OSS Rates(\$)	Disc 1st Disc Add
Network Rec First Add'l First Add'l SOMEC SOMAN 0 DS3 Interface Unit (DS1 COCI) combination per month UNC1X UC1D1 8.64 6.59 4.73 15.69 15.69 Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1 1 UNC1X USLXX 90.87 253.03 157.89 44.80 11.73 15.69 Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 2 2 UNC1X USLXX 155.43 253.03 157.89 44.80 11.73 15.69 Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 3 2 UNC1X USLXX 155.43 253.03 157.89 44.80 11.73 15.69 DS3 Interface Unit (DS1 COCI) combination per month UNC1X USLXX 261.89 253.03 157.89 44.80 11.73 15.69 Nonrecurring Currently Combination per month UNC1X UC1D1 8.64 6.59 4.73 15.69 VNC1X UC1X UC1D1 8.64 6.59 4.73	OSS Rates(\$)	
Net Rec First Add'l First Add'l SOMEC SOMAN DS3 Interface Unit (DS1 COCI) combination per month UNC1X UC1D1 8.64 6.59 4.73 15.69 15.69 Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1 1 UNC1X USLXX 90.87 253.03 157.89 44.80 11.73 15.69 Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 2 2 UNC1X USLXX 155.43 253.03 157.89 44.80 11.73 15.69 Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 2 2 UNC1X USLXX 155.43 253.03 157.89 44.80 11.73 15.69 DS3 Interface Unit (DS1 COCI) combination per month UNC1X USLXX 261.89 253.03 157.89 44.80 11.73 15.69 DS3 Interface Unit (DS1 COCI) combination per month UNC1X UC1D1 8.64 6.59 4.73 15.69 Nonrecurring Currently Combined Network Elements Switch -As- Is Charge UNCSX UNCSX UNCCC		SOMAN SOMAN
Image: Construction of the second state of the second s	SOMAN SOMAN	SOMAN SOMAN
Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1 1 UNC1X USLXX 90.87 253.03 157.89 44.80 11.73 15.69 Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 2 2 UNC1X USLXX 155.43 253.03 157.89 44.80 11.73 15.69 Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 2 2 UNC1X USLXX 155.43 253.03 157.89 44.80 11.73 15.69 Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 3 3 UNC1X USLXX 261.89 253.03 157.89 44.80 11.73 15.69 DS3 Interface Unit (DS1 COCI) combination per month UNC1X UC1D1 8.64 6.59 4.73 15.69 Nonrecurring Currently Combined Network Elements Switch -As- Is Charge UNCSX UNCCC 5.61 7.00 7.00 15.69 4-WIRE 56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROFFICE TRANSPORT (EEL) 1 1 1 15.69 Combination - Zone 1 1		
Zone 1 Zone 1 UNC1X USLXX 90.87 253.03 157.89 44.80 11.73 15.69 Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 2 2 UNC1X USLXX 155.43 253.03 157.89 44.80 11.73 15.69 Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 3 2 UNC1X USLXX 155.43 253.03 157.89 44.80 11.73 15.69 DS3 Interface Unit (DS1 COCI) combination per month UNC1X USLXX 261.89 253.03 157.89 44.80 11.73 15.69 Nonrecurring Currently Combined Network Elements Switch -As- Is Charge UNC1X UC1D1 8.64 6.59 4.73 15.69 4-WIRE 56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROFFICE TRANSPORT (EEL) UNC2X UNC2X 5.61 5.61 7.00 7.00 15.69 4-WIRE 56 KbpS Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 1 1 UNC2X UDL56 29.93 126.66 89.12 59.35 14.61 15.69		
Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 2 2 UNC1X USLXX 155.43 253.03 157.89 44.80 11.73 15.69 Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 3 3 UNC1X USLXX 155.43 253.03 157.89 44.80 11.73 15.69 DS3 Interface Unit (DS1 COCI) combination per month UNC1X UC1D1 8.64 6.59 4.73 15.69 Nonrecurring Currently Combined Network Elements Switch -As- Is Charge UNCSX UNCCC 5.61 5.61 7.00 7.00 15.69 4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 1 1 UNCX UDL56 29.93 126.66 89.12 59.35 14.61 15.69		
Zone 2 Zone 2 UNC1X USLXX 155.43 253.03 157.89 44.80 11.73 15.69 Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 3 3 UNC1X USLXX 261.89 253.03 157.89 44.80 11.73 15.69 DS3 Interface Unit (DS1 COCI) combination per month UNC1X UC1D1 8.64 6.59 4.73 15.69 Nonrecurring Currently Combined Network Elements Switch -As- Is Charge UNC1X UNCCX UNCCC 5.61 7.00 7.00 15.69 4-WIRE 56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROFFICE TRANSPORT (EEL) 15.69 4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 1 1 UNCX UDL56 29.93 126.66 89.12 59.35 14.61 15.69		
Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 3 3 UNC1X USLXX 261.89 253.03 157.89 44.80 11.73 15.69 DS3 Interface Unit (DS1 COCI) combination per month UNC1X UC1D1 8.64 6.59 4.73 15.69 Nonrecurring Currently Combined Network Elements Switch - As- Is Charge UNC1X UC1D1 8.64 6.59 4.73 15.69 4-WIRE 56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROFFICE TRANSPORT (EEL) UNCSX UNCCC 5.61 7.00 7.00 15.69 4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 1 1 UNCDX UDL56 29.93 126.66 89.12 59.35 14.61 15.69		
DS3 Interface Unit (DS1 COCI) combination per month UNC1X UC1D1 8.64 6.59 4.73 15.69 Nonrecurring Currently Combined Network Elements Switch -As- Is Charge UNC5X UNCCC 5.61 7.00 7.00 15.69 4-WIRE 56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROFFICE TRANSPORT (EEL)		
Nonrecurring Currently Combined Network Elements Switch -As- Is Charge UNCSX UNCCC 5.61 7.00 7.00 15.69 4-WIRE 56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROFFICE TRANSPORT (EEL) 15.69 15.69 15.69 15.69 15.69 15.69 15.69 15.69 15.69 15.69 15.69 15.69 15.69 15.69 15.69 15.69 15.69 15.69 15.69 15.69 15.69 15.69 15.69 15.69 15.69 15.69 15.69 <td></td> <td></td>		
Is Charge UNCSX UNCCC 5.61 7.00 7.00 15.69 4-WIRE 56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROFFICE TRANSPORT (EEL) <		
4-WIRE 56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROFFICE TRANSPORT (EEL)		
4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport 1 UNCDX UDL56 29.93 126.66 89.12 59.35 14.61 15.69		
Combination - Zone 1 1 UNCDX UDL56 29.93 126.66 89.12 59.35 14.61 15.69		
Combination - Zone 2 2 UNCDX UDL56 33.99 126.66 89.12 59.35 14.61 15.69		
4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport		
Combination - Zone 3 3 UNCDX UDL56 34.74 126.66 89.12 59.35 14.61 15.69		
Interoffice Transport - Dedicated - 4-wire 56 kbps combination -		
Per Mile UNCDX 1L5XX 0.0134 Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Interoffice Transport - Dedicated - 4-wire 56 kbps combination -		
Facility Termination - UNCDX U1TD5 13.41 40.63 27.47 16.77 6.91 15.69		
Nonrecurring Currently Combined Network Elements Switch -As-		
Is Charge UNCDX UNCCC 5.61 5.61 7.00 7.00 15.69		
4-WIRE 64 KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEL)		
4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport		
Combination - Zone 1 1 UNCDX UDL64 29.93 126.66 89.12 59.35 14.61 15.69		
4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport 2 UNCDX UDL64 33.99 126.66 89.12 59.35 14.61 15.69		
Combination - 2016 2 01/2017 - 2017 - 2017 -		
Combination - Zone 3 UNCDX UDL64 34.74 126.66 89.12 59.35 14.61 15.69		
Interoffice Transport - Dedicated - 4-wire 64 kbps combination -		
Per Mile UNCDX 1L5XX 0.0134		
Interoffice Transport - Dedicated - 4-wire 64 kbps combination -		
Facility Termination UNCDX U1TD6 13.41 40.63 27.47 16.77 6.91 15.69		
Nonrecurring Currently Combined Network Elements Switch -As- Is Charge UNCDX UNCCC 5.61 7.00 7.00 15.69		
Is Charge UNCDX UNCCC 5.61 7.00 7.00 15.69 ADDITIONAL NETWORK ELEMENTS Image: Comparison of the second s		
When used as a part of a currently combined facility, the non-recurring charges do not apply, but a Switch As Is charge does apply.		
When used as ordinarily combined network elements in All States, the non-recurring charges apply and the Switch As Is Charge does not.		
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to each combination)		
Nonrecurring Currently Combined Network Elements Switch -As-		
Is Charge - 2 wire/4-Wire VG UNCVX UNCCC 5.61 7.00 15.69 Networking Currently Combined Network Elements Switch An UNCVX UNCCC 5.61 7.00 15.69		↓ ↓ ↓ ↓
Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - 56/64 kbps UNCDX UNCCC 5.61 5.01 7.00 15.69		
Is Charge - 56/64 Kbps UNCDX UNCCC 5.61 5.61 7.00 7.00 15.69		<u>↓ </u>
Is charge - DS1 UNC1X UNCCC 5.61 5.61 7.00 7.00 15.69		
Nonreguring Currently Combined Network Elements Switch -As-		
Is Charge - DS3 UNC3X UNCCC 5.61 5.61 7.00 7.00 15.69		
Nonrecurring Currently Combined Network Elements Switch -As-		
Is Charge - STS1 UNCSX UNCCC 5.61 7.00 7.00 15.69		
NOTE: Local Channel - Dedicated Transport - minimum billing period - Below DS3=one month, DS3 and above=four months		┨────┤────
Local Channel - Dedicated - 2-Wire Voice Grade UNCXV ULDV2 15.33 193.53 33.24 36.72 3.21 15.69 Local Channel - Dedicated - 4-Wire Voice Grade UNCXV ULDV4 16.54 193.97 33.68 37.19 3.68 15.69		<u> </u>
Local Channel - Dedicated - 4-Wite Volte Grade UNC/V ULDV4 10.54 193.97 33.66 37.19 3.66 15.69 Local Channel - Dedicated - DS1 per month Zone 1 1 UNC/X ULDF1 42.62 177.87 154.06 22.24 15.30 15.69		
Local channel - Dedicated - Dol Per Month Zone 2 2 UNC1X ULDF1 70.32 177.87 154.06 22.24 15.30 156.9		
Local Channel - Dedicated - DS1- Per Month Zone 3 3 UNC1X ULDF1 190.68 177.87 154.06 22.24 15.30 15.69		1 1
Local Channel - Dedicated - DS3 - Per Mile per month UNC3X 1L5NC 11.93		
Local Channel - Dedicated - DS3 - Facility Termination UNC3X ULDF3 446.00 452.52 264.53 119.75 83.77 15.69		
Local Channel - Dedicated - STS-1- Per Mile per month UNCSX 1L5NC 11.93		

UNBUNDLE	ED NETWORK ELEMENTS - South Carolina												Attachment:	2	Exhi	bit: B
											Svc Order	Svc Order	Incremental		Incremental	Incrementa
												Submitted	Charge -	Charge -	Charge -	Charge -
		Interi	_								Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svo
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
<u> </u>		-		-			Nonrec	urring	Nonrecurring	Disconnect			220	Rates(\$)	l	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Local Channel - Dedicated - STS-1 - Facility Termination			UNCSX	ULDFS	435.10	452.52	264.53	119.75	83.77	JOWIEC	15.69	JONIAN	JONIAN	JONIAN	JONIAN
Optior	nal Features & Functions:			01100/	OLDI O	400.10	402.02	204.00	110.70	00.11		10.00				
	IPLEXERS															
	Channelization - DS1 to DS0 Channel System			UXTD1	MQ1	107.57	91.24	62.71	10.56	9.81		15.69				
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per															
	month (2.4-64kbs)			UDL	1D1DD	1.19	6.59	4.73				15.69				
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per															
	month			UDN	UC1CA	2.56	6.59	4.73				15.69				
	Voice Grade COCI - DS1 to DS0 Channel System - per month			UEA	1D1VG	0.56	6.59	4.73				15.69				
	DS3 to DS1 Channel System per month			UXTD3	MQ3	144.02	178.54	94.18	33.33	31.90		15.69				
	STS1 to DS1 Channel System per month	ļ	<u> </u>	UXTS1	MQ3	144.02	178.54	94.18	33.33	31.90		15.69				
├ ── ├ ──	DS3 Interface Unit (DS1 COCI) used with Loop per month DS3 Interface Unit (DS1 COCI) used with Local Channel per			USL	UC1D1	8.64	6.59	4.73				15.69				
	month		1	ULDD1	UC1D1	8.64	6.59	4.73				15.69				
├ ── ├ ──	DS3 Interface Unit (DS1 COCI) used with Interoffice Channel	-	+		00101	0.04	0.59	4.73				15.69				
	per month		1	U1TD1	UC1D1	8.64	6.59	4.73				15.69				
Sub-L	oop Feeder			OTIDI	00101	0.04	0.00	4.70				10.00				
04.0 1	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Statewide		SW	UNC1X	USBFG											
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 1		1	UNC1X	USBFG	55.85	102.19	64.64	62.26	17.52						
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 2		2	UNC1X	USBFG	109.16	102.19	64.64	62.26	17.52						
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 3		3	UNC1X	USBFG	203.35	102.19	64.64	62.26	17.52						
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 4		4	UNC1X	USBFG											
	LOCAL EXCHANGE SWITCHING(PORTS)															
	inge Ports															
	: Although the Port Rate includes all available features in GA, I	KY, LA	& TN, t	he desired features	will need to b	e ordered usin	g retail USOCs	5								
2-WIR	E VOICE GRADE LINE PORT RATES (RES)					1.05			1.10			15.00				
	Exchange Ports - 2-Wire Analog Line Port- Res.			UEPSR	UEPRL	1.65	2.38	2.28	1.42	1.33		15.69				
	Exchange Ports - 2-Wire Analog Line Port with Caller ID - Res.			UEPSR	UEPRC	1.65	2.38	2.28	1.42	1.33		15.69				
	Exchange Forts - 2-Wile Analog Line Fort with Caller ID - Res.	-		UEPSK	UEFRC	1.05	2.30	2.20	1.42	1.33		15.69				
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Res.			UEPSR	UEPRO	1.65	2.38	2.28	1.42	1.33		15.69				
	Exchange Ports - 2-Wire VG unbundled SC extended local			DEFOR	OLINO	1.00	2.00	2.20	1.44	1.00		10.00				
	dialing parity Port with Caller ID - Res.			UEPSR	UEPAU	1.65	2.38	2.28	1.42	1.33		15.69				
	Exchange Ports - 2-Wire VG unbundled South Carolina Area															
	Calling port with Caller ID - Res (LW8)			UEPSR	UEPAJ	1.65	2.38	2.28	1.42	1.33		15.69				
	Exchange Ports - 2-Wire VG unbundled res, low usage line port															
	with Caller ID (LUM)			UEPSR	UEPAP	1.65	2.38	2.28	1.42	1.33		15.69				
	Exchange Ports - 2-Wire VG South Carolina Residence Dialing															
	Plan without Caller ID			UEPSR	UEPWL	1.65	2.38	2.28	1.42	1.33		15.69				
	Exchange Ports - 2-Wire VG South Carolina Residence Area		1													
	Calling Plan without Caller ID capability			UEPSR	UEPRS	1.65	2.38	2.28	1.42	1.33	<u> </u>	15.69				
	2-Wire voice unbundled Low Usage Line Port without Caller ID		1			4.05	0.00	0.00	4.40	4.00		45.00				
┝──┼───	Capability Subsequent Activity			UEPSR UEPSR	UEPRT USASC	1.65 0.00	2.38 0.00	2.28	1.42	1.33		15.69 15.69				
FEATU			+	ULPOR	USAGU	0.00	0.00	0.00				15.69				
FEATU	All Available Vertical Features		+	UEPSR	UEPVF	3.04	0.00	0.00				15.69				1
2-WIR	E VOICE GRADE LINE PORT RATES (BUS)	t	1		521 11	0.04	0.00	0.00				10.09				
<u> </u>	Exchange Ports - 2-Wire Analog Line Port without Caller ID -	1	1	1	1				1						1	1
	Bus		1	UEPSB	UEPBL	1.65	2.38	2.28	1.42	1.33		15.69				
	Exchange Ports - 2-Wire VG unbundled Line Port with	I	1				-				1					1
	unbundled port with Caller+E484 ID - Bus.			UEPSB	UEPBC	1.65	2.38	2.28	1.42	1.33		15.69				
		1														
					UEPBO	1.65	2.38	2.28	1.42	1.33		15.69				
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Bus.			UEPSB	UEPBU	1.00										
	Exchange Ports - 2-Wire VG unbundled SC extended local				1											
	Exchange Ports - 2-Wire VG unbundled SC extended local dialing parity Port with Caller ID - Bus.			UEPSB	UEPAZ	1.65	2.38	2.28	1.42	1.33		15.69				
	Exchange Ports - 2-Wire VG unbundled SC extended local dialing parity Port with Caller ID - Bus. Exhange Ports - 2-Wire VG unbundled incoming only port with			UEPSB	UEPAZ	1.65										
	Exchange Ports - 2-Wire VG unbundled SC extended local dialing parity Port with Caller ID - Bus.				1		2.38 2.38	2.28 2.28	1.42 1.42	1.33		15.69 15.69				

JNBUNDLE	D NETWORK ELEMENTS - South Carolina												Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						Rec	Nonree	curring	Nonrecurring					Rates(\$)	•	
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Exchange Ports - 2-Wire Voice South Carolina Business Dialing										1					
	Plan without Caller ID			UEPSB	UEPWM	1.65	2.38	2.28	1.42	1.33	$ \longrightarrow $	15.69				
	Exchange Ports - 2-Wire Voice South Carolina Business Area Calling Port without Caller ID			UEPSB	UEPBB	1.65	2.38	2.28	1.42	1.33		15.69				
	2-Wire voice unbundled Incoming Only Port without Caller ID							1			1					
	Capability			UEPSB	UEPBE	1.65	2.38	2.28	1.42	1.33		15.69				
	Subsequent Activity			UEPSB	USASC	0.00	0.00	0.00	ļ'	ļ'		15.69				
FEATU						0.04	0.00	0.00	ļ'	'	\square	45.00				
	All Available Vertical Features All Available Vertical Features			UEPSB	UEPVF UEPVF	3.04 3.04	0.00	0.00	 '	'	┟───┤	15.69				
EVOUA	NGE PORT RATES (DID & PBX)				UEPVF	3.04	0.00	0.00	<u> </u>	<u> </u>	├─── ┤	15.69				
EACHA	2-Wire VG Unbundled 2-Way PBX Trunk - Res			UEPSE	UEPRD	1.65	31.34	14.88	13.97	0.90	┝───┤	15.69				
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Res			UEPSP	UEPPC	1.65	31.34	14.88	13.97	0.90	┝───┤	15.69				
	2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus	ł		UEPSP	UEPPO	1.65	31.34	14.88	13.97	0.90	┝───┤	15.69				
	2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus			UEPSP	UEPP1	1.65	31.34	14.88	13.97	0.90	├─── ┤	15.69				
	2-Wire Analog Long Distance Terminal PBX Trunk - Bus	ł –		UEPSP	UEPLD	1.65	31.34	14.88	13.97	0.90	├───┤	15.69				
	2-Wire Voice Unbundled PBX LD Terminal Ports	1		UEPSP	UEPLD	1.65	31.34		13.97	0.90		15.69	1		1	
	2-Wire Vice Unbundled 2-Way PBX Usage Port			UEPSP	UEPXA	1.65	31.34	14.88	13.97	0.90	├ ──┤	15.69				
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPSP	UEPXB	1.65	31.34	14.88	13.97	0.90		15.69				
	2-Wire Voice Unbundled PBX LD DDD Terminals Port	l		UEPSP	UEPXC	1.65	31.34	14.88	13.97	0.90		15.69	l		1	l
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPSP	UEPXD	1.65	31.34	14.88	13.97	0.90		15.69				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD	Γ						([[
	Capable Port			UEPSP	UEPXE	1.65	31.34	14.88	13.97	0.90		15.69				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port			UEPSP	UEPXL	1.65	31.34	14.88	13.97	0.90		15.69				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port			UEPSP	UEPXM	1.65	31.34	14.88	13.97	0.90		15.69				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port			UEPSP		1.65	31.34	14.88	13.97	0.90		15.69				
<u> </u>	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPSP	UEPXO	1.65	31.34	14.88	13.97	0.90	┝───┦	15.69				┝────
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port 2-Wire Voice Unbundled 2-Way PBX South Carolina Area Plus			ULFOF	UEFAS	1.05	31.34	14.88	13.97	0.90	┢───┤	10.09	ł		ł	
	Calling Port			UEPSP	UEPXT	1.65	31.34	14.88	13.97	0.90		15.69				
	Subsequent Activity			UEPSP	USASC	0.00	0.00	0.00	13.37	0.30	┝───┤	15.69				
FEATU		<u> </u>		02.01	00,000	0.00	0.00	0.00			├ ──┤	10.00				
-	All Available Vertical Features	l		UEPSP UEPSE	UEPVF	3.04	0.00	0.00	i'	i'	<u>├</u>	15.69	1		1	<u> </u>
	NGE PORT RATES (COIN)	1							[]	[]			İ		ĺ	1
	Exchange Ports - Coin Port	1				1.65	2.38	2.28	1.42	1.33		15.69				1
Local S	Switching Features offered with Port															
	Transmission/usage charges associated with POTS circuit sy															
	Access to B Channel or D Channel Packet capabilities will be	e availat	ole only	y through BFR/New	Business Ree	quest Process.	Rates for the	packet capabi	lities will be de	etermined via t	ne Bona Fid	le Request/	New Business	Request Pro	cess.	
	OCAL EXCHANGE SWITCHING(PORTS)	ļ	L					 '	 '	 '						L
EXCHA	NGE PORT RATES	ļ	L			0.55		L	L	L		18.00				
	Exchange Ports - 2-Wire DID Port			UEPEX	UEPP2	8.86	119.57	18.78	60.03	3.77	↓ ↓	15.69				<u> </u>
	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability			UEPDD	UEPDD	73.62	202.47	95.90	72.75	2.47		15.69				
	Exchange Ports - 2-Wire ISDN Port (See Notes below.)			UEPTX UEPSX	U1PMA	13.38	72.93	53.11	47.90			15.69				
	All Features Offered			UEPTX UEPSX	UEPVF	3.04	0.00	0.00								
	Transmission/usage charges associated with POTS circuit su															
NOTE:	Access to B Channel or D Channel Packet capabilities will be	e availat	ole only						lities will be de	stermined via t	ne Bona Fid	le Request/	New Business	Request Pro	cess.	
	Exchange Ports - 2-Wire ISDN Port Channel Profiles	ļ		UEPTX UEPSX	U1UMA	0.00	0.00	0.00	 '	 '						
1	Exchange Ports - 4-Wire ISDN DS1 Port	ļ		UEPEX	UEPEX	107.44	204.27	101.78	79.35	20.10	↓	15.69				
	IDLED PORT with REMOTE CALL FORWARDING CAPABILITY							ļ'	 '	 '	↓					<u> </u>
UNBUN	IDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE				UERAC	4.05	0.00	0.00	4.10	4.00	↓	45.00				<u> </u>
	Unbundled Remote Call Forwarding Service, Area Calling, Res			UEPVR	UERAC	1.65	2.38	2.28	1.42	1.33	\vdash	15.69				
				UEPVR	UERLC	1.65	2.38	2.28	1.42	1.33		15.69				
					UERLU	1.05	2.38	L.28	1.42	1.33	1 1	10.09	1			L
	Unbundled Remote Call Forwarding Service, Local Calling - Res								1 / 1	1 22	1	15 60				
	Unbundled Remote Call Forwarding Service, Local Calling - Kes Unbundled Remote Call Forwarding Service, InterLATA - Res Unbundled Remote Call Forwarding Service, IntraLATA - Res			UEPVR	UERTE	1.65 1.65	2.38 2.38	2.28 2.28	1.42 1.42			15.69 15.69				

UNBUNDL	ED NETWORK ELEMENTS - South Carolina												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
					_		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Unbundled Remote Call Forwarding Service - Conversion -				110.000		0.40	0.40				45.00				
	Switch-as-is Unbundled Remote Call Forwarding Service - Conversion with			UEPVR	USAC2		0.10	0.10				15.69			-	-
	allowed change (PIC and LPIC)			UEPVR	USACC		0.10	0.10								
UNB	UNDLED REMOTE CALL FORWARDING - Bus				00400		0.10	0.10								
0.12																
	Unbundled Remote Call Forwarding Service, Area Calling - Bus			UEPVB	UERAC	1.65	2.38	2.28	1.42	1.33		15.69				
	Unbundled Remote Call Forwarding Service, Local Calling - Bus			UEPVB	UERLC	1.65	2.38	2.28	1.42	1.33		15.69				
	Unbundled Remote Call Forwarding Service, InterLATA - Bus			UEPVB	UERTE	1.65	2.38	2.28	1.42	1.33		15.69				
	Unbundled Remote Call Forwarding Service, IntraLATA - Bus			UEPVB	UERTR	1.65	2.38	2.28	1.42	1.33		15.69				
	Unbundled Remote Call Forwarding Service Expanded and	1	1			1.0-	0.00	0.00		1.00		45.00			1	1
No.	Exception Local Calling	<u> </u>	<u> </u>	UEPVB	UERVJ	1.65	2.38	2.28	1.42	1.33		15.69	ļ		ł	ł
Non-	Recurring Unbundled Remote Call Forwarding Service - Conversion -															
	Switch-as-is			UEPVB	USAC2		0.10	0.10				15.69			1	1
	Unbundled Remote Call Forwarding Service - Conversion with	<u> </u>	1		00/102		0.10	0.10				15.09	·			
	allowed change (PIC and LPIC)			UEPVB	USACC		0.10	0.10								
UNBUNDLED	D LOCAL SWITCHING, PORT USAGE															
	Office Switching (Port Usage)															
	End Office Switching Function, Per MOU					0.0010519										
	End Office Trunk Port - Shared, Per MOU					0.0002136										
Tand	lem Switching (Port Usage) (Local or Access Tandem)															
	Tandem Switching Function Per MOU					0.0001634										
	Tandem Trunk Port - Shared, Per MOU				_	0.0002863										
Com	mon Transport				-	0.00000.45									-	-
	Common Transport - Per Mile, Per MOU Common Transport - Facilities Termination Per MOU				_	0.0000045										
	D PORT/LOOP COMBINATIONS - COST BASED RATES				-	0.0004095										
	Based Rates are applied where BellSouth is required by FCC ar	nd/or St	tate Co	mmission rule to n	rovide Unbun	dled Local Swi	tching or Swite	h Ports								
	ures shall apply to the Unbundled Port/Loop Combination - Cos								ed Port section	of this Rate E	xhibit.					
End	Office and Tandem Switching Usage and Common Transport Us	sage rat	tes in t	he Port section of t	his rate exhib	it shall apply to	all combination	ons of loop/po	rt network elen	nents except f	or UNE Coi	n Port/Loop	Combination	is.		
	first and additional Port nonrecurring charges apply to Not Curr	rently C	ombin	ed Combos. For Cu	rrently Combi	ined Combos th	he nonrecurrin	g charges sha	II be those iden	tified in the N	onrecurring	- Currently	Combined se	ections.		
	RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)															
UNE	Port/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			14.89										
	2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3		2		_	21.52 27.17										
UNE	Loop Rates		3		+	21.17									<u> </u>	<u> </u>
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPRX	UEPLX	13.76							· · · · · · · · · · · · · · · · · · ·			
	2-Wire Voice Grade Loop (SL1) - Zone 1	1	2	UEPRX	UEPLX	20.38	-								ł	ł
	2-Wire Voice Grade Loop (SL1) - Zone 3	1	3	UEPRX	UEPLX	26.04										
2-Wi	re Voice Grade Line Port Rates (Res)															
	2-Wire voice unbundled port - residence			UEPRX	UEPRL	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire voice unbundled port with Caller ID - res			UEPRX	UEPRC	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire voice unbundled port outgoing only - res	I	ļ	UEPRX	UEPRO	1.13	40.30	19.90	24.98	6.65		15.69			ļ	
	2-Wire voice Grade unbundled South Carolina extended local	1	1			4.40	40.00	40.00	04.00	0.05		45.00			1	1
├ ── ├ ──	dialing parity port with Caller ID - res 2-Wire voice unbundled South Carolina Area Calling port with	<u> </u>	+	UEPRX	UEPAU	1.13	40.30	19.90	24.98	6.65		15.69	1		<u> </u>	
	Caller ID - res (LW8)	1	1	UEPRX	UEPAJ	1.13	40.30	19.90	24.98	6.65		15.69			1	1
	2-Wire voice unbundles res, low usage line port with Caller ID		+		ULFAU	1.13	40.30	19.90	24.90	0.05		15.09			1	1
	(LUM)	1	1	UEPRX	UEPAP	1.13	37.93	16.72				15.69			1	1
	2-Wire Voice Unbundled South Carolina Residence Dialing Plan	1	1		02.74	1.13	01.00	10.72				10.00			<u> </u>	<u> </u>
		1	1	UEPRX	UEPWL	1.13	40.30	19.90	24.98	6.65		15.69			1	1
	without Caller ID															
	without Caller ID 2-Wire voice unbundled South Carolina Area Calling Port			0EI TOX												
	2-Wire voice unbundled South Carolina Area Calling Port without Caller ID Capability			UEPRX	UEPRS	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire voice unbundled South Carolina Area Calling Port without Caller ID Capability 2-Wire voice unbundled Low Usage Line Port without Caller ID			UEPRX	UEPRS	1.13										
	2-Wire voice unbundled South Carolina Area Calling Port without Caller ID Capability						40.30 40.30	19.90 19.90	24.98 24.98	6.65 6.65		15.69 15.69				

UNBUNDL	ED NETWORK ELEMENTS - South Carolina	-	r	1							1	-	Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
1.00	All Features Offered			UEPRX	UEPVF	3.04	0.00	0.00				15.69				┢────
LOCA	AL NUMBER PORTABILITY			UEPRX	LNPCX	0.35										┢────
NON	Local Number Portability (1 per port) RECURRING CHARGES (NRCs) - CURRENTLY COMBINED			UEFRA	LINFGA	0.55										<u> </u>
NON	2-Wire Voice Grade Loop / Line Port Combination - Conversion -															ł
	Switch-as-is			UEPRX	USAC2		0.10	0.10				15.69				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -	-		021101	00/102		0.110	0.10				10.00				
	Switch with change			UEPRX	USACC		0.10	0.10				15.69				
ADDI	TIONAL NRCs															
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent															
	Activity			UEPRX	USAS2	0.00	0.00	0.00				15.69				
	RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)															
UNE	Port/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			14.89										ļ
	2-Wire VG Loop/Port Combo - Zone 2		2			21.52										I
	2-Wire VG Loop/Port Combo - Zone 3		3			27.17										<u> </u>
UNE	Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPBX	UEPLX	13.76										ł
																ł
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPBX	UEPLX	20.38										I
2 .Wi	2-Wire Voice Grade Loop (SL1) - Zone 3 re Voice Grade Line Port (Bus)		3	UEPBX	UEPLX	26.04										
2-001	2-Wire voice unbundled port without Caller ID - bus			UEPBX	UEPBL	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPBX	UEPBC	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire voice unbundled port outgoing only - bus			UEPBX	UEPBO	1.13	40.30	19.90	24.98	6.65		15.69				1
	2-Wire voice Grade unbundled South Carolina extended local			OEI DA	OLI DO	1.10	40.00	10.00	24.00	0.00		10.00				
	dialing parity port with Caller ID - bus			UEPBX	UEPAZ	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire voice unbundled incoming only port with Caller ID - Bus			UEPBX	UPEB1	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire voice unbundled South Carolina Bus Area Calling Port			-	-	-										
	with Caller ID (LMB)			UEPBX	UEPAB	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Unbundled South Carolina Business Dialing Plan															
	without Caller ID			UEPBX	UEPWM	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire voice unbundled South Carolina Business Area Calling															
	Port without Caller ID Capability			UEPBX	UEPBB	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire voice unbundled Incoming Only Port without Caller ID															
	Capability			UEPBX	UEPBE	1.13	40.30	19.90	24.98	6.65		15.69				
LOC					111201/											L
	Local Number Portability (1 per port)			UEPBX	LNPCX	0.35										L
FEA	All Features Offered			UEPBX	UEPVF	3.04	0.00	0.00				45.00				ł
NON	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED			UEPBA	UEPVF	3.04	0.00	0.00				15.69				
NON	2-Wire Voice Grade Loop / Line Port Combination - Conversion -															
	Switch-as-is			UEPBX	USAC2		0.10	0.10				15.69				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -	-		OEI DA	00/102		0.10	0.10				10.00				
	Switch with change			UEPBX	USACC		0.10	0.10				15.69				
ADD	TIONAL NRCs			-												
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent															
	Activity			UEPBX	USAS2		0.00	0.00				15.69				
	RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)															
UNE	Port/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			14.89										
	2-Wire VG Loop/Port Combo - Zone 2		2			21.52										
	2-Wire VG Loop/Port Combo - Zone 3		3			27.17										
UNE	Loop Rates															ļ
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEPRG	UEPLX	13.76										I
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEPRG	UEPLX	20.38										<u> </u>
0.147	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEPRG	UEPLX	26.04										<u> </u>
2-WI	e Voice Grade Line Port Rates (RES - PBX) 2-Wire VG Unbundled Combination 2-Way PBX Trunk Port -	-														
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Res	1	1	UEPRG	UEPRD	1.13	69.26	32.50	37.53	6.22		15.69				1

NRONDL	ED NETWORK ELEMENTS - South Carolina	-	-	1	· ·							-	Attachment:			bit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
LOC	AL NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPRG	LNPCP	3.15	0.00	0.00				15.69				
FEAT	TURES					0.01						1 = 0.0				
NON	All Features Offered	_		UEPRG	UEPVF	3.04	0.00	0.00			-	15.69				
NON	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As-Is			UEPRG	USAC2		7.93	1.91				15.69				
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -	_		UEFRG	USACZ		7.95	1.91				15.69				
	Conversion - Switch with Change			UEPRG	USACC		7.93	1.91				15.69				
	TIONAL NRCs	-		OLING	UUACC		1.55	1.31				15.05				
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -															
	Subsequent Activity			UEPRG	USAS2	0.00	0.00	0.00				15.69				
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt			02.110	00,102	0.00	0.00	0.00	-		1	10.00				
	Group						7.34	7.34				15.69				
2-WI	RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)															
	Port/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			14.89										
	2-Wire VG Loop/Port Combo - Zone 2		2			21.52										
	2-Wire VG Loop/Port Combo - Zone 3		3			27.17										
UNE	Loop Rates															
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEPPX	UEPLX	13.76										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEPPX	UEPLX	20.38										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEPPX	UEPLX	26.04										
2-Wir	e Voice Grade Line Port Rates (BUS - PBX)															
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus	i		UEPPX	UEPPC	1.13	69.26	32.50	37.53	6.22		15.69				
	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPPX	UEPPO	1.13	69.26	32.50	37.53	6.22		15.69				
	Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPPX	UEPP1	1.13	69.26	32.50	37.53	6.22		15.69				
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPPX	UEPLD	1.13	69.26	32.50	37.53	6.22		15.69				
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port	_		UEPPX	UEPXA	1.13	69.26	32.50	37.53	6.22		15.69				
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPPX	UEPXB	1.13	69.26	32.50	37.53	6.22		15.69				
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPPX	UEPXC UEPXD	1.13	69.26	32.50	37.53	6.22		15.69				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port	-		UEPPX	UEPAD	1.13	69.26	32.50	37.53	6.22		15.69				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD				UEPXE	1.13	co. oc	22.50	27.52	c 22		45.00				
	Capable Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy	_		UEPPX	UEPXE	1.13	69.26	32.50	37.53	6.22		15.69				
	Administrative Calling Port	1	1	UEPPX	UEPXL	1.13	69.26	32.50	37.53	6.22		15.69				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy	1	+		JLFAL	1.15	03.20	52.50	51.55	0.22	<u> </u>	15.09				
	Room Calling Port	1	1	UEPPX	UEPXM	1.13	69.26	32.50	37.53	6.22		15.69				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital	1	1				00.20	02.00	000	0.22	t	.0.00			1	
	Discount Room Calling Port			UEPPX	UEPXO	1.13	69.26	32.50	37.53	6.22		15.69				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port		1	UEPPX	UEPXS	1.13	69.26	32.50	37.53	6.22		15.69				
	2-Wire Voice Unbundled 2-Way PBX South Carolina Area Plus	1		İ							1				İ	
	Calling Port	1	1	UEPPX	UEPXT	1.13	69.26	32.50	37.53	6.22		15.69				
LOC	AL NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPPX	LNPCP	3.15	0.00	0.00				15.69				
FEAT	TURES															
	All Features Offered			UEPPX	UEPVF	3.04	0.00	0.00				15.69				
NON	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -															
	Conversion - Switch-As-Is	1	<u> </u>	UEPPX	USAC2		7.93	1.91				15.69				
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -	1	1													
	Conversion - Switch with Change	1	<u> </u>	UEPPX	USACC		7.93	1.91				15.69				
ADD	TIONAL NRCs	1									L					
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -	1	1		110.4.00	0.00	0.00	0.00				45.00				
<u> </u>	Subsequent Activity Change/Rearrange Multiline Hunt	+	-	UEPPX	USAS2	0.00	0.00	0.00				15.69	<u> </u>			┝────
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt	1	1				7.24	7 34				15 60				
1	Group	RT					7.34	7.34				15.69				L

IRONDL	ED NETWORK ELEMENTS - South Carolina			I							-	-	Attachment:			bit: B
TEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)					Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incremen Charge Manual S Order vs Electroni Disc Ado
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
UNE	Port/Loop Combination Rates															
	2-Wire VG Coin Port/Loop Combo – Zone 1		1			14.89										
	2-Wire VG Coin Port/Loop Combo – Zone 2		2			21.52										
	2-Wire VG Coin Port/Loop Combo – Zone 3		3		-	27.17										
UNEI	Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPCO	UEPLX	13.76										
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPCO	UEPLX	20.38										
-	2-Wire Voice Grade Loop (SL1) - Zone 3			UEPCO	UEPLX	26.04										
2-Wir	e Voice Grade Line Ports (COIN)		5		OLILA	20.04										
	2-Wire Coin 2-Way without Operator Screening and without															
	Blocking (SC)			UEPCO	UEPSD	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (SC)			UEPCO	UEPSA	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking (SC)			UEPCO	UEPSH	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking; with Dialing Parity (SC)			UEPCO	UEPSC	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Coin 2-Way with Operator Screening and: 900 Blocking: 900/976, 1+DDD, 011+, and Local (SC)			UEPCO	UEPCC	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Coin 2-W Operator Screen: 900 Block: 900/976, 1+DDD, 011+, Local; Enhanced Call OPT 3YV (SC)	1		UEPCO	UEPCE	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Coin 2-W Operator Screen: 900 Block: 900/976, 1+DDD, 011+, Local; Enhanced Call OPT AP7 (SC)			UEPCO	UEPCF	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Coin Outward without Blocking and without Operator Screening (SC)			UEPCO	UEPSG	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Coin Outward with Operator Screening and 011 Blocking (SC)			UEPCO	UEPSF	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Coin Outward with Operator Screening and Blocking: 011, 900/976, 1+DDD (SC)			UEPCO	UEPSJ	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (SC)			UEPCO	UEPCM	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Coin Out Operator Screen & Block: 900/976, 1+DDD, 011+, Local; Enhanced Calling OPT 3YW (SC)			UEPCO	UEPCP	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire 2-Way Smartline with 900/976 (all states except LA)			UEPCO	UEPCK	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Coin Outward Smartline with 900/976 (all states except LA)			UEPCO	UEPCR	1.13	40.30	19.90	24.98	6.65		15.69				
ADDI	TIONAL UNE COIN PORT/LOOP (RC)															
	UNE Coin Port/Loop Combo Usage (Flat Rate)			UEPCO	URECU	4.05	40.30	19.90	24.98	6.65		15.69				
LOCA					L NIDOY											
NONE	Local Number Portability (1 per port) RECURRING CHARGES - CURRENTLY COMBINED			UEPCO	LNPCX	0.35										
NONF	2-Wire Voice Grade Loop / Line Port Combination - Conversion -	<u> </u>			+											
	Switch-as-is			UEPCO	USAC2		0.10	0.10				15.69				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -				00,02		0.10	0.10				10.00				
	Switch with change			UEPCO	USACC		0.10	0.10				15.69				
ADDI	TIONAL NRCs															
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity			UEPCO	USAS2		0.00	0.00				15.69				
	RE VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE	E LINE F	PORT (RES)												
UNE	Port/Loop Combination Rates				1 1											
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1	ļ	1		+	22.50			-							
_	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2 2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		2			30.56 37.22										
	Loop Rates		3		+	31.22										
JNEI	2-Wire Voice Grade Loop (SL2) - Zone 1	ł	1	UEPFR	UECF2	20.85										
	2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFR	UECF2	20.85										
	2-Wire Voice Grade Loop (SL2) - Zone 2 2-Wire Voice Grade Loop (SL2) - Zone 3	1		UEPFR	UECF2	35.57										1
2-Wir	e Voice Grade Line Port Rates (Res)	ł –			520.2	00.07										t
	2-Wire voice unbundled port - residence		t	UEPFR	UEPRL	1.65	108.36	70.71	1.42	1.33		15.69				<u> </u>

JNBUNDLE	D NETWORK ELEMENTS - South Carolina												Attachment:	2	Exhi	bit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrec	urring	Nonrecurring	Disconnect				Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire voice unbundled port with Caller ID - res			UEPFR	UEPRC	1.65	108.36	70.71	1.42	1.33		15.69				
	2-Wire voice unbundled port outgoing only - res			UEPFR	UEPRO	1.65	108.36	70.71	1.42	1.33		15.69				
	2-Wire voice Grade unbundled South Carolina extended local dialing parity port with Caller ID - res			UEPFR	UEPAU	1.65	108.36	70.71	1.42	1.33		15.69				
	2-Wire voice unbundled South Carolina Area Calling port with Caller ID - res (LW8)			UEPFR	UEPAJ	1.65	108.36	70.71	1.42	1.33		15.69				
	2-Wire voice unbundles res, low usage line port with Caller ID (LUM)			UEPFR	UEPAP	1.65	108.36	70.71	1.42	1.33		15.69				
	2-Wire Voice Unbundled South Carolina Residence Dialing Plan without Caller ID			UEPFR	UEPWL		108.36	70.71	1.42			15.69				
INTED	OFFICE TRANSPORT		-	UEPFR	UEPVVL	1.65	108.36	70.71	1.42	1.33	-	15.69				
INTER	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility	1														
	Termination			UEPFR	U1TV2	24.30	40.63	27.47	16.77	6.91						
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile or Fraction Mile			UEPFR	1L5XX	0.0167										
FEATU																
	All Features Offered			UEPFR	UEPVF	3.04	0.00	0.00				15.69				
LOCAL	NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPFR	LNPCX	0.35										
NONR	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED				_									-		
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port Combination - Conversion - Switch-as-is			UEPFR	USAC2		17.00	3.74				15.69				
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port Combination - Conversion - Switch-With-Change			UEPFR	USACC		17.00	3.74				15.69				
2-WIRI	E VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE		PORT (
	ort/Loop Combination Rates															
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			22.50										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			30.56										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			37.22										
UNE L	oop Rates															
	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFB	UECF2	20.85										
	2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFB	UECF2	28.91										
0.14/5	2-Wire Voice Grade Loop (SL2) - Zone 3 Voice Grade Line Port (Bus)		3	UEPFB	UECF2	35.57										
2-wire	2-Wire voice unbundled port without Caller ID - bus			UEPFB	UEPBL	1.65	108.36	70.71	1.42	1.33		15.69				
	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPFB	UEPBC	1.65	108.36	70.71	1.42	1.33		15.69				
	2-Wire voice unbundled port with Caller + L+0+ 10 - 503			UEPFB	UEPBO	1.65	108.36	70.71	1.42	1.33		15.69				
	2-Wire voice Grade unbundled South Carolina extended local		1	OLITE	OLI DO	1.00	100.00	70.71	1.44	1.00		10.00				
	dialing parity port with Caller ID - bus	1	1	UEPFB	UEPAZ	1.65	108.36	70.71	1.42	1.33		15.69				
	2-Wire voice unbundled incoming only port with Caller ID - Bus			UEPFB	UEPB1	1.65	108.36	70.71	1.42	1.33		15.69				
	2-Wire voice unbundled South Carolina Bus Area Calling Port with Caller ID (LMB)			UEPFB	UEPAB	1.65	108.36	70.71	1.42	1.33		15.69				
	2-Wire Voice Unbundled South Carolina Business Dialing Plan without Caller ID			UEPFB	UEPWM	1.65	108.36	70.71	1.42	1.33		15.69				
LOCAI		1	1		02. 000	1.00	100.00	70.71	1.72	1.00	t	10.00				
	Local Number Portability (1 per port)	1	1	UEPFB	LNPCX	0.35										
INTER	OFFICE TRANSPORT		1													
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility Termination			UEPFB	U1TV2	24.30	40.63	27.47	16.77	6.91						
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile or Fraction Mile			UEPFB	1L5XX	0.0167	-10.00	21.41	10.17	0.01						
FEATU			1		ILJAA	0.0107										
	All Features Offered	1	1	UEPFB	UEPVF	3.04	0.00	0.00			t	15.69				
NONR	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED	1	1													
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port			UEPFB	USAC2		17.00	3.74				15.69				
	Combination - Conversion - Switch-as-is															
	Combination - Conversion - Switch-as-is 2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port Combination - Conversion - Switch with change			UEPFB	USACC		17.00	3.74				15.69				

UNBUNDL	ED NETWORK ELEMENTS - South Carolina	1	1	1	- 1								Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring					Rates(\$)	•	
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
UNE	Port/Loop Combination Rates		4			00.50										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1 2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		1			22.50 30.56										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		3			37.22										
UNE	Loop Rates					01.22										
	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFP	UECF2	20.85										
	2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFP	UECF2	28.91										
	2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFP	UECF2	35.57										
2-Wir	re Voice Grade Line Port Rates (BUS - PBX)															
	Line Side Liebundled Combination 2 Way DBY Truck Dart Dur		1	UEPFP	UEPPC	1.65	137.32	83.31	67.02	11 54		15.00				
├──	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus Line Side Unbundled Outward PBX Trunk Port - Bus				UEPPC	1.65 1.65	137.32 137.32	83.31	67.02	<u>11.51</u> 11.51		15.69 15.69				
	Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPFP	UEPP1	1.65	137.32	83.31	67.02	11.51	1	15.69				
	2-Wire Voice Unbundled PBX LD Terminal Ports		1	UEPFP	UEPLD	1.65	137.32	83.31	67.02	11.51	1	15.69				
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port		1	UEPFP	UEPXA	1.65	137.32	83.31	67.02	11.51	1	15.69	İ		İ	
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPFP	UEPXB	1.65	137.32	83.31	67.02	11.51	<u> </u>	15.69				
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPFP	UEPXC	1.65	137.32	83.31	67.02	11.51		15.69				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPFP	UEPXD	1.65	137.32	83.31	67.02	11.51		15.69				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD															
	Capable Port			UEPFP	UEPXE	1.65	137.32	83.31	67.02	11.51		15.69				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy					4.05	407.00	00.04	07.00	44.54		45.00				
	Administrative Calling Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy			UEPFP	UEPXL	1.65	137.32	83.31	67.02	11.51		15.69				
	Room Calling Port			UEPFP	UEPXM	1.65	137.32	83.31	67.02	11.51		15.69				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital			OLITI		1.05	137.32	00.01	07.02	11.01		15.05				
	Discount Room Calling Port			UEPFP	UEPXO	1.65	137.32	83.31	67.02	11.51		15.69				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPFP	UEPXS	1.65	137.32	83.31	67.02	11.51		15.69				
	2-Wire Voice Unbundled 2-Way PBX South Carolina Area Plus															
	Calling Port			UEPFP	UEPXT	1.65	137.32	83.31	67.02	11.51		15.69				
LOCA	AL NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPFP	LNPCP	3.15	0.00	0.00				15.69				
INTE	ROFFICE TRANSPORT				_											
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility Termination			UEPFP		24.20	40.63	07.47	40.77	C 01						
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile			UEPFP	U1TV2	24.30	40.63	27.47	16.77	6.91						
	or Fraction Mile			UEPFP	1L5XX	0.0167										
FFAT	TURES		1		TLOXX	0.0107										
	All Features Offered			UEPFP	UEPVF	3.04	0.00	0.00				15.69				
NONE	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
	Combination - Conversion - Switch-as-is			UEPFP	USAC2		17.00	3.74				15.69				
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
	Combination - Conversion - Switch with change O PORT/LOOP COMBINATIONS - COST BASED RATES			UEPFP	USACC		17.00	3.74				15.69				
	RE VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK	DODT			-											
	Port/Loop Combination Rates	FURI														
ONL	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1		1			23.75										
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2		2	İ		30.20					1		İ		İ	
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3		3			35.52										
UNE	Loop Rates															
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1		1	UEPPX	UECD1	16.68					ļ					
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2		2	UEPPX	UECD1	23.13										
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3		3	UEPPX	UECD1	28.46										
UNE	Port Rate Exchange Ports - 2-Wire DID Port			UEPPX	UEPD1	7.06	225.55	87.21	113.08	14.38			15.69			
NON	RECURRING CHARGES - CURRENTLY COMBINED			UEPPA	JEPUI	7.06	220.05	87.21	113.08	14.38			15.69			
NON	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination -		1													
	Switch-as-is	1	1	UEPPX	USAC1		7.32	1.87			1	1	15.69	1	1	1

UNBUNDLE	D NETWORK ELEMENTS - South Carolina													Attachment:			ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	В	cs	USOC			RATES(\$)				Submitted	Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
							Rec	Nonrec		Nonrecurring					Rates(\$)		
							1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth Allowable Changes			UEPPX		USA1C		7.32	1.87					15.69			
ADDITI	ONAL NRCs			ULFFA		USAIC		1.52	1.07					15.09			
7.000	2-Wire DID Subsequent Activity - Add Trunks, Per Trunk			UEPPX		USAS1		26.84						15.69			
Teleph	one Number/Trunk Group Establisment Charges																
	DID Trunk Termination (One Per Port)			UEPPX		NDT	0.00	0.00	0.00					15.69			
	DID Numbers, Establish Trunk Group and Provide First Group of 20 DID Numbers			UEPPX		NDZ	0.00	0.00	0.00					15.69			
	Additional DID Numbers for each Group of 20 DID Numbers			UEPPX		NDZ ND4	0.00	0.00	0.00					15.69			
	DID Numbers, Non- consecutive DID Numbers . Per Number			UEPPX		ND5	0.00	0.00	0.00					15.69			<u> </u>
	Reserve Non-Consecutive DID numbers			UEPPX		ND6	0.00	0.00	0.00					15.69			
	Reserve DID Numbers			UEPPX		NDV	0.00	0.00	0.00					15.69			
LOCAL						LNDCD											<u> </u>
2 14/100	Local Number Portability (1 per port) SDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LI			UEPPX		LNPCP	3.15	0.00	0.00			+					├ ───
	ort/Loop Combination Rates	NE SIDE	PURI														
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 1		1	UEPPB	UEPPR		30.86										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 2		2	UEPPB	UEPPR		38.60										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 3		3	UEPPB	UEPPR		44.23										
UNE Lo	pop Rates											1			1		
	2-Wire ISDN Digital Grade Loop - UNE Zone 1		1	UEPPB	UEPPR	USL2X	21.90							15.69			
	2-Wire ISDN Digital Grade Loop - UNE Zone 2		2	UEPPB	UEPPR	USL2X	29.64							15.69			
	2-Wire ISDN Digital Grade Loop - UNE Zone 3		3	UEPPB	UEPPR	USL2X	35.27							15.69			L
UNE Po	ort Rate Exchange Port - 2-Wire ISDN Line Side Port			UEPPB	UEPPR	UEPPB	8.96	190.51	133.14	100.95	21.37	-		15.69	-		L
NONRE	ECURRING CHARGES - CURRENTLY COMBINED			UEFFB	UEFFR	UEPPB	0.90	190.51	155.14	100.95	21.37			15.69			
North	2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port														1		
	Combination - Conversion			UEPPB	UEPPR	USACB	0.00	38.59	27.08					15.69			
	ONAL NRCs																
LOCAL																	
	Local Number Portability (1 per port) NNEL USER PROFILE ACCESS:			UEPPB	UEPPR	LNPCX	0.35	0.00	0.00								┣────
В-СПА	CVS/CSD (DMS/5ESS)			UEPPB	UEPPR	U1UCA	0.00	0.00	0.00								
	CVS (EWSD)			UEPPB	UEPPR	U1UCB	0.00	0.00	0.00								<u> </u>
	CSD			UEPPB	UEPPR	U1UCC	0.00	0.00	0.00								
B-CHA	NNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS S	C,MS, &	TN)														
	CVS/CSD (DMS/5ESS)			UEPPB	UEPPR	U1UCD	0.00	0.00	0.00								
	CVS (EWSD)			UEPPB	UEPPR	U1UCE	0.00	0.00	0.00								L
USER	CSD TERMINAL PROFILE	<u> </u>		UEPPB	UEPPR	U1UCF	0.00	0.00	0.00			+			<u> </u>		<u>├</u> ───
USER	User Terminal Profile (EWSD only)	<u> </u>		UEPPB	UEPPR	U1UMA	0.00	0.00	0.00			1					<u> </u>
VERTIC	CAL FEATURES	L															
	All Vertical Features - One per Channel B User Profile			UEPPB	UEPPR	UEPVF	3.04	0.00	0.00					15.69			
INTER						<u> </u>											
	Interoffice Channel mileage each, including first mile and facilities termination				UEPPR	M1GNC	24.30	40.63	27.47	16.77	6.91			15.69			ļ
4 14/1	Interoffice Channel mileage each, additional mile			UEPPB	UEPPR	M1GNM	0.0167	0.00	0.00						ł		───
	E DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK prt/Loop Combination Rates	THORE				+						-					
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE											1					<u> </u>
1	Zone 1		1	UEPPP			176.82					1					
	ZW DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 2	l	2	UEPPP			241.38										
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 3		3	UEPPP			347.84										

INBUNDLE	D NETWORK ELEMENTS - South Carolina												Attachment:	2	Exhi	ibit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR		Incremental Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec	urring	Nonrecurring	g Disconnect				Rates(\$)		-
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
UNE L	oop Rates												15.00			L
	4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPPP	USL4P	90.87							15.69			
	4-Wire DS1 Digital Loop - UNE Zone 2 4-Wire DS1 Digital Loop - UNE Zone 3			UEPPP UEPPP	USL4P USL4P	155.43 261.89							15.69 15.69			
	ort Rate		3	UEFFF	USL4P	201.09							15.69			
	Exchange Ports - 4-Wire ISDN DS1 Port			UEPPP	UEPPP	85.95	457.30	259.67	124.15	31.83			15.69			
NONR	ECURRING CHARGES - CURRENTLY COMBINED			0EI I I	OLITI	00.00	407.00	200.01	124.10	01.00			10.00			
	4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port															
	Combination - Conversion -Switch-as-is			UEPPP	USACP	0.00	119.34	78.73					15.69			
ADDIT	IONAL NRCs															
	4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy-															
	Inward/two way Tel Nos. (except NC)			UEPPP	PR7TF		0.49	0.49					15.69			L
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port -															
	Outward Tel Numbers (All States except NC)			UEPPP	PR7TO		11.54	11.54					15.69			<u> </u>
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port -				DD777		00.07	00.07					15.00			
1.000	Subsequent Inward Tel Numbers			UEPPP	PR7ZT		23.07	23.07					15.69			
LUCA	Local Number Portability (1 per port)			UEPPP	LNPCN	1.75										
	Voice/Data			UEPPP	PR71V	0.00	0.00	0.00								
	Digital Data			UEPPP	PR71D	0.00	0.00	0.00								
	Inward Data			UEPPP	PR71E	0.00	0.00	0.00								
New o	r Additional "B" Channel															
	New or Additional - Voice/Data B Channel			UEPPP	PR7BV	0.00	14.56						15.69			
	New or Additional - Digital Data B Channel			UEPPP	PR7BF	0.00	14.56						15.69			
	New or Additional Inward Data B Channel			UEPPP	PR7BD	0.00	14.56						15.69			
CALL	TYPES															
	Inward			UEPPP	PR7C1	0.00	0.00	0.00								
	Outward			UEPPP UEPPP	PR7C0 PR7CC	0.00	0.00	0.00								
Intoro	Two-way Ifice Channel Mileage			UEFFF	PR/CC	0.00	0.00	0.00								
Intero	Fixed Each Including First Mile			UEPPP	1LN1A	77.4815	89.47	81.99	16.39	14.48			15.69			
-	Each Airline-Fractional Additional Mile			UEPPP	1LN1B	0.3415	00.47	01.00	10.00	14.40			10.00			
4-WIR	E DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT															
	ort/Loop Combination Rates															
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1		1	UEPDC		149.77										
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2		2	UEPDC		214.33										
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3		3	UEPDC		320.78										
UNE L	oop Rates					00.07							15.00			
	4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPDC UEPDC	USLDC USLDC	90.87					ł	ļ	15.69 15.69			───
	4-Wire DS1 Digital Loop - UNE Zone 2 4-Wire DS1 Digital Loop - UNE Zone 3		2	UEPDC UEPDC	USLDC	155.43 261.89							15.69 15.69			
UNF P	ort Rate		5		JJLDC	201.09							15.09			1
ф <u></u> Г	4-Wire DDITS Digital Trunk Port			UEPDC	UDD1T	58.90	455.50	253.79	117.55	14.20			15.69	1	1	1
NONR	ECURRING CHARGES - CURRENTLY COMBINED			-						20	1					
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination										1					
	- Switch-as-is			UEPDC	USAC4		129.78	67.17					15.69			
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
	- Conversion with DS1 Changes			UEPDC	USAWA		129.78	67.17					15.69			L
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination				110 0110											
400-	- Conversion with Change - Trunk			UEPDC	USAWB		129.78	67.17					15.69			───
ADDIT	IONAL NRCs 4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent															
	Channel Activation/Chan - 1-Way Outward Trunk			UEPDC	UDTTB		14.51	14.51					15.69			
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel				00110		14.31	14.31					15.09			1
	Activation/Chan Inward Trunk w/out DID			UEPDC	UDTTC		14.51	14.51					15.69			
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan				520			1								1
	Activation Per Chan - Inward Trunk with DID			UEPDC	UDTTD		14.51	14.51					15.69			
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan															
1	Activation / Chan - 2-Way DID w User Trans	1		UEPDC	UDTTE		14.51	14.51			1		15.69			1

UNBUNDLE	D NETWORK ELEMENTS - South Carolina												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st		Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	
						Rec	Nonrec	urring	Nonrecurring	g Disconnect				Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
BIPOL	AR 8 ZERO SUBSTITUTION			UEPDC	CCOSF		0.00	605.00					15.69			
	B8ZS -Superframe Format B8ZS - Extended Superframe Format			UEPDC	CCOEF		0.00	605.00					15.69	-		
Alterna	ate Mark Inversion				CCOLI		0.00	005.00					15.05			
	AMI -Superframe Format			UEPDC	MCOSF		0.00	0.00								
	AMI - Extended SuperFrame Format			UEPDC	MCOPO		0.00	0.00								
Teleph	none Number/Trunk Group Establisment Charges				UBTOV								15.00			
	Telephone Number for 2-Way Trunk Group Telephone Number for 1-Way Outward Trunk Group			UEPDC UEPDC	UDTGX UDTGY	0.00							15.69 15.69			
	Telephone Number for 1-Way Dutward Trunk Group Without DID			UEPDC	UDTGZ	0.00							15.69			
	DID Numbers, Establish Trunk Group and Provide First Group			OEI DO	00102	0.00							10.00			
	of 20 DID Numbers			UEPDC	NDZ	0.00	0.00	0.00					15.69			
	DID Numbers for each Group of 20 DID Numbers			UEPDC	ND4	0.00							15.69			
	DID Numbers, Non- consecutive DID Numbers , Per Number			UEPDC	ND5	0.00	0.00	0.00					15.69			
	Reserve Non-Consecutive DID Nos. Reserve DID Numbers			UEPDC UEPDC	ND6 NDV	0.00	0.00	0.00					15.69 15.69			
Dedica	ated DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1	Digita	lloon			0.00	0.00	0.00					15.69			
	Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities															
	Termination)			UEPDC	1LNO1	77.14	89.47	81.99	16.39	14.48			15.69			
	Interoffice Channel Mileage - Additional rate per mile - 0-8 miles			UEPDC	1LNOA	0.3415	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities Termination)			UEPDC	1LNO2	0.00	0.00	0.00								
	Interoffice Channel Mileage - Additional rate per mile - 9-25				TENOZ	0.00	0.00	0.00								
	miles			UEPDC	1LNOB	0.3415	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities															
	Termination)			UEPDC	1LNO3	0.00	0.00	0.00								
	lateraffine Channel Milanes Additional rate non-mile OF amiles			UEPDC	1LNOC	0.3415	0.00	0.00								
	Interoffice Channel Mileage - Additional rate per mile - 25+ miles Local Number Portability, per DS0 Activated			UEPDC	LNPCP	3.15	0.00	0.00								
	Central Office Termininating Point			UEPDC	CTG	0.00	0.00	0.00								
4-WIRE	E DS1 LOOP WITH CHANNELIZATION WITH PORT															
	n is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Act															
	System can have up to 24 combinations of rates depending on	type ar	nd num	ber of ports used												
UNE D	S1 Loop 4-Wire DS1 Loop - UNE Zone 1		1	UEPMG	USLDC	90.87	0.00	0.00								
	4-Wire DST Loop - UNE Zone 2			UEPMG	USLDC	155.43	0.00	0.00								
	4-Wire DS1 Loop - UNE Zone 3		3	UEPMG	USLDC	261.89	0.00	0.00								
UNE D	SO Channelization Capacities (D4 Channel Bank Configuration	ns)	Ē													
	24 DSO Channel Capacity - 1 per DS1			UEPMG	VUM24	82.78	0.00	0.00					15.69			
	48 DSO Channel Capacity - 1 per 2 DS1s			UEPMG	VUM48	165.56	0.00	0.00					15.69			
	96 DSO Channel Capacity -1per 4 DS1s		<u> </u>	UEPMG UEPMG	VUM96 VUM14	331.12 496.68	0.00	0.00					15.69 15.69		ļ	
<u> </u>	144 DS0 Channel Capacity - 1 per 6 DS1s 192 DS0 Channel Capacity -1 per 8 DS1s			UEPMG	VUM14 VUM19	496.68	0.00	0.00		1	+		15.69			
<u> </u>	240 DS0 Channel Capacity - 1 per 10 DS1s			UEPMG	VUM20	827.80	0.00	0.00			1		15.69			
	288 DS0 Channel Capacity - 1 per 12 DS1s			UEPMG	VUM28	993.36	0.00	0.00			<u> </u>		15.69			
	384 DS0 Channel Capacity - 1 per 16 DS1s			UEPMG	VUM38	1,324.48	0.00	0.00					15.69			
	480 DS0 Channel Capacity - 1 per 20 DS1s			UEPMG	VUM40	1,655.60	0.00	0.00					15.69			
	576 DS0 Channel Capacity -1 per 24 DS1s			UEPMG UEPMG	VUM57 VUM67	1,986.72	0.00	0.00					15.69			
Non-P	672 DS0 Channel Capacity - 1 per 28 DS1s ecurring Charges (NRC) Associated with 4-Wire DS1 Loop with	h Chany	l			2,317.84 Based on a Sv	0.00 stem	0.00		1	+		15.69			
	mum System configuration is One (1) DS1, One (1) D4 Channe						otom									
	les of this configuration functioning as one are considered Ac															
·	NRC - Conversion (Currently Combined) with or without															
	BellSouth Allowed Changes			UEPMG	USAC4	0.00	150.81	8.38					15.69			
-	n Additions at End User Locations Where 4-Wire DS1 Loop with	th Chan	nelizat	ion with Port Com	bination Curre	ntly Exists and										
			0 140	10							1					
	Not Currently Combined) in all states, except in Density Zone 1 1 DS1/D4 Channel Bank - Additionally Add NRC for each Port		8 MSA	\'s												

UNBUNDLEI	D NETWORK ELEMENTS - South Carolina		1		-	•						-	Attachment:			ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Charge -
						Rec	Nonred		Nonrecurring					Rates(\$)		
						100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Bipolar	8 Zero Substitution															
	Clear Channel Capability Format, superframe - Subsequent Activity Only			UEPMG	CCOSF	0.00	0.00	605.00								
	Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only			UEPMG	CCOEF	0.00	0.00	605.00								
Altorna	te Mark Inversion (AMI)			UEPING	CCOEF	0.00	0.00	605.00								
	Superframe Format			UEPMG	MCOSE	0.00	0.00	0.00								
	Extended Superframe Format			UEPMG	MCOPO	0.00	0.00	0.00								
	ge Ports Associated with 4-Wire DS1 Loop with Channelization	on with	Port			0.00	0.00	0.00								
	ige Ports															
_Addition	3 01 010															
	Line Side Combination Channelized PBX Trunk Port - Business			UEPPX	UEPCX	1.13	0.00	0.00	0.00	0.00			15.69			1
	Line Side Outward Channelized PBX Trunk Port - Business			UEPPX	UEPOX	1.13	0.00	0.00	0.00	0.00			15.69		1	
1					1 2			2.50		2.30					1	
	Line Side Inward Only Channelized PBX Trunk Port without DID			UEPPX	UEP1X	1.13	0.00	0.00	0.00	0.00			15.69			
	2-Wire Trunk Side Unbundled Channelized DID Trunk Port			UEPPX	UEPDM	7.09	0.00	0.00	0.00	0.00			15.69			1
Feature	Activations - Unbundled Loop Concentration															1
	Feature (Service) Activation for each Line Port Terminated in D4															
	Bank Feature (Service) Activation for each Trunk Port Terminated in			UEPPX	1PQWM	0.56	25.45	13.44	4.20	4.17			15.69			
	D4 Bank			UEPPX	1PQWU	0.56	78.31	18.46	59.37	11.60			15.69			
	one Number/ Group Establishment Charges for DID Service															
	DID Trunk Termination (1 per Port)			UEPPX	NDT	0.00	0.00	0.00								
	Estab Trk Grp and Provide 1st 20 DID Nos. (FL,GA, NC,& SC)			UEPPX	NDZ	0.00	0.00	0.00								
	DID Numbers - groups of 20 - Valid all States			UEPPX	ND4	0.00	0.00	0.00								
	Non-Consecutive DID Numbers - per number			UEPPX	ND5	0.00	0.00	0.00								
	Reserve Non-Consecutive DID Numbers			UEPPX	ND6	0.00	0.00	0.00								
	Reserve DID Numbers			UEPPX	NDV	0.00	0.00	0.00								
Local N	lumber Portability					0.15										
FEATU	Local Number Portability - 1 per port			UEPPX	LNPCP	3.15	0.00	0.00								
	RES - Vertical and Optional Switching Features Offered with Line Side Ports Only															
	All Features Available			UEPPX	UEPVF	3.04	0.00	0.00					15.69			
	PORT LOOP COMBINATIONS - MARKET RATES			UEFFA	UEFVF	3.04	0.00	0.00					15.69			
	Rates shall apply where BellSouth is not required to provide	unhund		al switching or swi	itch ports por	ECC and/or St	to Commissio	n rulos								
	cludes:	unbunu	lieu iot	al switching of swi	nen ports per	FCC and/or Sta		in rules.								
	dled port/loop combinations that are Currently Combined or N	lot Curr	ently (combined in Zone 1	of the Top 8	MSAS in BellS	outh's region	for end users	with 4 or more	DS0 equivalen	t lines					
	p 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderda											e).				
BellSou	uth currently is developing the billing capability to mechanica	lly bill t	he rec	urring and non-recu	urring Market	Rates in this se	ection except f	or nonrecurrin	ng charges for I	not currently o	ombined in	FL and NC	. In the interi	m where Bell	South cannot	bill Market
Rates.	BellSouth shall bill the rates in the Cost-Based section preced	lina in l	ieu of	the Market Rates ar	nd reserves th	ne riaht to true-u	Ip the billing o	difference.								
	rket Rate for unbundled ports includes all available features i															
	fice and Tandem Switching Usage and Common Transport Us			e Port section of th	nis rate exhib	it shall apply to	all combination	ons of loop/po	ort network elen	nents except	or UNE Coi	n Port/Loor	Combination	s which hav	e a flat rate us	sage charge
	URECU).															5
	t Currently Combined scenarios the Nonrecurring charges are	listed i	n the F	irst and Additional	NRC column	s for each Port	USOC. For C	Irrently Comb	ined scenarios	the Nonrecur	ring charge	s are listed	in the NRC - 0	Currently Con	nbined sectio	n.
	onal NRCs may apply also and are categorized accordingly.										g onarge	o are noteu				
	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)															
	ort/Loop Combination Rates				1	 			 				1		1	1
	2-Wire VG Loop/Port Combo - Zone 1		1			27.76									1	1
	2-Wire VG Loop/Port Combo - Zone 2		2		1	34.38					1		İ		İ	1
	2-Wire VG Loop/Port Combo - Zone 3		3		1	40.04										1
	pop Rates															
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPRX	UEPLX	13.76			1							1
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPRX	UEPLX	20.38			1							
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPRX	UEPLX	26.04										
2-Wire	Voice Grade Line Port (Res)															
	2-Wire voice unbundled port - residence			UEPRX	UEPRL	14.00	90.00	90.00				15.69				
	2-Wire voice unbundled port with Caller ID - res			UEPRX	UEPRC	14.00	90.00	90.00				15.69				
	2-Wire voice unbundled port outgoing only - res			UEPRX	UEPRO	14.00	90.00	90.00				15.69				

Checkers Rate Libbers North Res Use Libbers Libbers State Diam Dia	UNBUNDLE	D NETWORK ELEMENTS - South Carolina												Attachment:	2	Exhi	bit: B
Image: Note of control of set or subject in part with Caller D Image: Note of control of set or subject in part with Caller D Image: Note of control of set or subject in part with Caller D Image: Note of control of set or subject in part with Caller D Image: Note of control of set or subject in part with Caller D Image: Note of control of set or subject in part with Caller D Image: Note of control of set or subject in part with Caller D Image: Note of control of set or subject in part with Caller D Image: Note of control of set or subject in part with Caller D Image: Note of control of set or subject in part with Caller D Image: Note of control of set or subject in part with Caller D Image: Note of control of set or subject in part with Caller D Image: Note of control of set or subject in part with Caller D Image: Note of control of set or subject in part with Caller D Image: Note of control of set or subject in part with Caller D Image: Note of control of set or subject in part with Caller D Image: Note of control of set or subject in part with Caller D Image: Note of control of set or subject in part with Caller D Image: Note of control of set or subject in part with Caller D Image: Note of control of set or subject in part with Caller D Image: Note of control of set or subject in part with Caller D Image: Note of control of set or subject in part with Caller D Image: Note of control of set or subject in part with Caller D Image: Note of control of set or subject in part with Caller D Image: Note of control of set or subject in part with Caller D Image: Note of control				Zone	BCS	usoc			RATES(\$)			Submitted Elec	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge -
Description Description Description Description Description Books BOOks							Boc	Nonred	urring	Nonrecurring	g Disconnect						
Image: Control Image: Contro Image: Control Image: C							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
SWI: SWI: <th< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></th<>																	
Image: Departing to control con					UEPRX	UEPAP	14.00	90.00	90.00				15.69				
sense Cater Jo LUPP X UEPX		Capability			UEPRX	UEPRT	14.00	90.00	90.00				15.69				
2.NWs vaca unstanded Sont Caraha Na Calling Port UEP2X UEP3X 14.00 90					UEPRX	UEPWL	14.00	90.00	90.00				15.69				
LOCAL NUMBER PORTABULY LEPRX LPC Desc Des																	
Loop Loop					UEPRX	UEPRS	14.00	90.00	90.00				15.69				
FURTURESII<	LOCA																L
MP Feature Order MP Feature Order<					UEPRX	LNPCX	0.35										L
ADDRIÑOML NRCA INC INC INC <	FEATU		 				0.00	0.00	0.00			+	45.00				┟────
NHC - 2-We Vote Grinds Loop Im 2 wing Loop				-	UEPKX	UEPVF	0.00	0.00	0.00			+	15.69				┣────
Babaseyard ULPRX USAS2 0.00 0.00 15.8 0 0 VME Ford.cg. Conduction Rass 1 0 277 0	ADDIT																ł
UHE PortLoop Conditionation Rates Image: Conditionation Rates Image: Conditionation Rates Image: Conditionationationationationationationation		Subsequent			UEPRX	USAS2		0.00	0.00				15.69				
Description Description <thdescription< th=""> <thdescription< th=""></thdescription<></thdescription<>												-					───
Byter VG LoopPart Comba - Zona 2 2 C 6.400 C <thc< th=""> C C</thc<>	UNE P			4			07.70					-					ł
EVMIN VGL LogPAR EVMIN VGL Compton Combo - Zone 1 1 UPE Log Parks Image: Compton Co												-					
UNE Loop Rate Image: Control of the state o						_											<u> </u>
Image: Solution Vision Grande Loop (Sk1) - Zoone 1 2 UEPRX				3			40.04					-					<u> </u>
B-VMe Value Grade Loop (SL1) - Zone 2 2 UEPRX UEPLX 20.03	UNE L			1			13.76					1					ł
E-Vite Vice Grade Loop (SL) - Zone 3 3 UEPBX UEPLX 20.04												1					ł
2 Wire Volze Grade Line Port (Biss) UEPBX UEPAX 00.00 00.00 15.69 Image: Control South																	t
Image: Constraint of a point without Caller ID - bus UEPBX UEPAX <t< td=""><td>2-Wire</td><td></td><td></td><td>Ű</td><td>02. 5/</td><td>02.2.</td><td>20.01</td><td></td><td></td><td></td><td></td><td>1</td><td></td><td></td><td></td><td></td><td></td></t<>	2-Wire			Ű	02. 5/	02.2.	20.01					1					
1 2-Wire visce unbundled port with Caller ± 644 10-bus UEPBX UEPBC 14.00 90.00 90.00 15.69 2-Wire visce unbundled port with Caller D-bus UEPBX UEPBX UEPAZ 14.00 90.00 90.00 15.69 2-Wire visce unbundled South Carolina Bus Accolling Drot UEPBX UEPAZ 14.00 90.00 90.00 15.69 2-Wire visce unbundled South Carolina Bus Accolling Drot UEPBX UEPBX UEPBX UEPBX 90.00 90.00 15.69 2-Wire visce unbundled South Carolina Business Daling Pian UEPBX UEPBX UEPBX UEPBX 90.00 90.00 15.69 </td <td></td> <td></td> <td></td> <td></td> <td>UEPBX</td> <td>UEPBL</td> <td>14.00</td> <td>90.00</td> <td>90.00</td> <td></td> <td></td> <td></td> <td>15.69</td> <td></td> <td></td> <td></td> <td></td>					UEPBX	UEPBL	14.00	90.00	90.00				15.69				
2-Wire volce unbundled port outgoing only- bus UEPBX UEPBX UEPBX 14.00 90.00 90.00 15.69 Image: Constraint of the constraint of the																	
dialing party port with caller D- bus UEPAX UEPAX 14.00 90.00 90.00 16.00 15.60 16.00 <td></td> <td></td> <td></td> <td></td> <td>UEPBX</td> <td>UEPBO</td> <td>14.00</td> <td>90.00</td> <td>90.00</td> <td></td> <td></td> <td></td> <td>15.69</td> <td></td> <td></td> <td></td> <td></td>					UEPBX	UEPBO	14.00	90.00	90.00				15.69				
E-Wire voice unbundled South Carolina Busie Area Calling Port Inclaire ID (LMB) UPPAX UEPAX		2-Wire voice Grade unbundled South Carolina extended local															
with Caller D (LMB) UEP8X UEP8X UEP8X UEP8X 90.00 90.00 90.00 15.69 Image: Comparison of the compa					UEPBX	UEPAZ	14.00	90.00	90.00				15.69				
2-Wire voice unbundled incoming Only Port without Caller ID UEPBX UEPBE 14.00 90.00 15.69 16.69 2-Wire Voice Unbundled South Carolina Business Dialing Plan without Caller ID UEPBX UEPBX UEPWM 14.00 90.00 90.00 15.69 15.69 16.69 15.69 15.69 15.69 15.69 16.69					UEPBX	UEPAB	14.00	90.00	90.00				15.69				
2-Wire Voice Unbundled South Carolina Business Dialing Plan without Caller ID UEPBX UEPWX UEPWX 14.00 90.00 90.00 90.00 15.89 Image: Constraints and the constraints and t		2-Wire voice unbundled Incoming Only Port without Caller ID															
withou Caller ID withou Caller ID UEPSX UEPSX UEPBX UEPSX <t< td=""><td></td><td></td><td></td><td></td><td>UEPBX</td><td>UEPBE</td><td>14.00</td><td>90.00</td><td>90.00</td><td></td><td></td><td></td><td>15.69</td><td></td><td></td><td></td><td></td></t<>					UEPBX	UEPBE	14.00	90.00	90.00				15.69				
Image: Port without Caller ID Capability Image: Port without Caller ID Capability Image: Port without Caller ID Capability Image: Port without Caller ID Capability Image: Port without Caller ID Capability Image: Port without Caller ID Capability Image: Port without Caller ID Capability Image: Port without Caller ID Capability Image: Port without Caller ID Capability Image: Port without Caller ID Capability Image: Port without Caller ID Capability Image: Port without Caller ID Capability Image: Port without Caller ID Capability Image: Port without Caller ID Capability Image: Port without Caller ID Capability Image: Port without Caller ID Capability Image: Port without Caller ID Capability Image: Port without Caller ID Capability Image: Port Without Caller ID Capability Image: Port Without Caller ID Capability Image: Port Without Caller ID Capability Image: Port Without Caller ID Capability Image: Port Without Caller ID Capability Image: Port Without Caller ID Capability Image: Port Without Caller ID Capability Image: Port Without Caller ID Capability Image: Port Without Caller ID Capability Image: Port Without Caller ID Capability Image: Port Without Caller ID Capability Image: Port Without Capability Image: Port Without Capability Image: Port Without Capability Image: Port Without Capability Image: Port Without Capability Image: Port Without Capability Image: Port Without Capability Image					UEPBX	UEPWM	14.00	90.00	90.00				15.69				
LOCAL NUMBER PORTABILITY LUPPA LUPPX 0.35 LUPPX 0.36 LUPPX LUPXX LUPXX LUPXX LUPXX LUPXX LUPXX <thlupxx< th=""> <thlupxx< th=""> LUPXX</thlupxx<></thlupxx<>																	
Image: Decal Number Portability (1 per port) Image: Decal Number Portability Image: Decal Number Portability Image: Decal Number Portability Image: Decal Number Portability Image: Decal Number Portability Image: Decal Number Portability Image: Decal Number Portability Image:					UEPBX	UEPBB	14.00	90.00	90.00				15.69				
FEATURES Image: Constraint of the cons	LOCA																ļ
All Features Offered UEPBX UEPVF 0.00 0.00 0.00 15.69 0 0 0 ADDITIONAL NRCs Image: Constraint of the second s			<u> </u>		UEPBX	LNPCX	0.35					<u> </u>					
ADDITIONAL NRCs Image: Constraint of the second secon	FEATU		 				0.00	0.00	0.00				45.00				ł
NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent UEPBX USAS2 0.00 0.00 0.00 15.69 0.00			<u> </u>	-	UEPBA	UEPVF	0.00	0.00	0.00			+	15.69				ł
Subsequent UEPBX USAS2 0.00 0.00 15.69 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX) </td <td>ADDIT</td> <td></td> <td></td> <td>+</td> <td>ł</td> <td></td> <td></td> <td></td> <td></td> <td>1</td> <td>1</td> <td>+</td> <td></td> <td></td> <td>ł</td> <td></td> <td>t</td>	ADDIT			+	ł					1	1	+			ł		t
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX) Image: Constraint of the second					UEPBX	USAS2		0.00	0.00			1	15.60				1
UNE Port/Loop Combination Rates Image: C	2-WIR		1	1		00,02		0.00	0.00			1	10.09				
2-Wire VG Loop/Port Combo - Zone 1 1 2 27.76 1			1	1								1		İ		İ	<u> </u>
2-Wire VG Loop/Port Combo - Zone 2 2 0 34.38 0		2-Wire VG Loop/Port Combo - Zone 1	1	1			27.76					1					
UNE Loop Rates Image: Constraint of the system of the				2													
2-Wire Voice Grade Loop (SL1) - Zone 1 1 UEPRG UEPLX 13.76				3			40.04										
2-Wire Voice Grade Loop (SL1) - Zone 2 2 UEPRG UEPLX 20.38	UNE L																L
2-Wire Voice Grade Loop (SL1) - Zone 3 3 UEPRG UEPLX 26.04			ļ									1					
2-Wire Voice Grade Line Port Rates (RES - PBX) <			I														
2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Res 2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - UEPRG 14.00 90.00 90.00 15.69 15.69 LOCAL NUMBER PORTABILITY 0<	0.14/1		 	3	UEPKG	UEPLX	26.04					+					ł
Res UEPRG UEPRD 14.00 90.00 90.00 15.69 LOCAL NUMBER PORTABILITY Image: Constraint of the second	2-wire																<u> </u>
		Res			UEPRG	UEPRD	14.00	90.00	90.00				15.69				
	LOCA	L NUMBER PORTABILITY Local Number Portability (1 per port)	ļ	1	UEPRG	LNPCP	3.15	0.00	0.00			1					

NRONDLE	D NETWORK ELEMENTS - South Carolina	r	1	1									Attachment:			ibit: B
TEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
FEATU																
	All Features Offered			UEPRG	UEPVF	0.00	0.00	0.00				15.69				
	ECURRING CHARGES - CURRENTLY COMBINED															
ADDIT	IONAL NRCs															
	2 Wire Loop/Line Side Port Combination - Non feature -						0.00	0.00				15.69				
	Subsequent Activity- Nonrecurring PBX Subsequent Activity - Change/Rearrange Multiline Hunt	-					0.00	0.00				15.69				
	Group						14.64	14.64				15.69				
2-WIR	E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)				-		14.04	14.04				13.09				
	ort/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1	ł	1		+ +	27.76			1		t		1			1
	2-Wire VG Loop/Port Combo - Zone 2	<u> </u>	2			34.38			1						İ	1
	2-Wire VG Loop/Port Combo - Zone 3	<u> </u>	3			40.04			1						İ	1
UNE L	oop Rates	l I	İ													
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPPX	UEPLX	13.76										
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPPX	UEPLX	20.38										
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPPX	UEPLX	26.04										
2-Wire	Voice Grade Line Port Rates (BUS - PBX)															
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPPX	UEPPC	14.00	90.00	90.00				15.69				
	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPPX	UEPPO	14.00	90.00	90.00				15.69				
	Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPPX	UEPP1	14.00	90.00	90.00				15.69				
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPPX	UEPLD	14.00	90.00	90.00				15.69				
_	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPPX UEPPX	UEPXA UEPXB	14.00 14.00	90.00	90.00 90.00				15.69				
_	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPPX	UEPXB	14.00	90.00 90.00	90.00				15.69 15.69				
-	2-Wire Voice Unbundled PBX LD DDD Terminals Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port	-		UEPPX	UEPXC	14.00	90.00	90.00				15.69				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD			ULFFX	ULFAD	14.00	90.00	90.00	-		1	13.09		-		
	Capable Port			UEPPX	UEPXE	14.00	90.00	90.00				15.69				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy			02.17.7	02.72	1.000	00.00	00.00			1	10.00				
	Administrative Calling Port			UEPPX	UEPXL	14.00	90.00	90.00				15.69				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy															
	Room Calling Port			UEPPX	UEPXM	14.00	90.00	90.00				15.69				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital															
	Discount Room Calling Port			UEPPX	UEPXO	14.00	90.00	90.00				15.69				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPPX	UEPXS	14.00	90.00	90.00				15.69				
LOCA	NUMBER PORTABILITY															
	Local Number Portability (1 per port)	ļ		UEPPX	LNPCP	3.15	0.00	0.00			ļ					
FEATU			ļ								L	48.00				l
NOND	All Features Offered ECURRING CHARGES - CURRENTLY COMBINED	ļ		UEPPX	UEPVF	0.00	0.00	0.00			ļ	15.69				<u> </u>
	ECORRING CHARGES - CORRENTLY COMBINED															
ADDIT	IONAL NRCS															
	2-Wire Voice Grade Loop/ Line Port Combination - Subsequent			UEPPX	USAS2		0.00	0.00				15.69				1
_	2 Wire Loop/Line Side Port Combination - Subsequent 2 Wire Loop/Line Side Port Combination - Non feature -	<u> </u>			USASZ		0.00	0.00				15.69				
	Subsequent Activity- Nonrecurring						0.00	0.00				15.69				1
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt						0.00	0.00				10.09				
	Group						7.34	7.34				15.69				1
2-WIR	E VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN POR	λT							1						İ	1
	ort/Loop Combination Rates	[1		İ		İ	1
	2-Wire VG Coin Port/Loop Combo – Zone 1		1			27.76										
	2-Wire VG Coin Port/Loop Combo – Zone 2		2			34.38										
	2-Wire VG Coin Port/Loop Combo – Zone 3		3			40.04										
UNE L	oop Rates															
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPCO	UEPLX	13.76										
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPCO	UEPLX	20.38										l
-	2-Wire Voice Grade Loop (SL1) - Zone 3	ļ	3	UEPCO	UEPLX	26.04					L					ļ
2-Wire	Voice Grade Line Port Rates (Coin)		1								1					

UNBUNDLE	D NETWORK ELEMENTS - South Carolina												Attachment:	2	Exhi	ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)	_	
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Coin 2-Way without Operator Screening and without Blocking (SC)			UEPCO	UEPSD	14.00	90.00	90.00				15.69				
	2-Wire Coin 2-Way with Operator Screening and Blocking: 011,			ULFCO	ULF3D	14.00	90.00	90.00				13.09				
	200/976, 1+DDD (AL, KY, LA, MS, SC) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011,			UEPCO	UEPRA	14.00	90.00	90.00				15.69				<u> </u>
	900/976, 1+DDD (SC) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking			UEPCO	UEPSA	14.00	90.00	90.00				15.69				
	(SC)			UEPCO	UEPSH	14.00	90.00	90.00				15.69				
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking; with Dialing Parity (SC)			UEPCO	UEPSC	14.00	90.00	90.00				15.69				
	2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (SC)			UEPCO	UEPCC	14.00	90.00	90.00				15.69				
	2-Wire Coin 2-W Oper Screen & Blocking: 900/976, 1+DDD,															
	011+ & Local; Enhanced Calling OPT 3YV (SC)		<u> </u>	UEPCO	UEPCE	14.00	90.00	90.00				15.69				<u> </u>
	2-Wire Coin 2-W Oper Screen & Block: 900/976, 1+DDD, 011+, & Local; Enhanced Calling OPT AP7 (SC)			UEPCO	UEPCF	14.00	90.00	90.00				15.69				
	2-Wire Coin Outward without Blocking and without Operator Screening (SC)			UEPCO	UEPSG	14.00	90.00	90.00				15.69				
	2-Wire Coin Outward with Operator Screening and 011 Blocking (SC)			UEPCO	UEPSF	14.00	90.00	90.00				15.69				
	2-Wire Coin Outward with Operator Screening and Blocking: 011, 900/976, 1+DDD (SC)			UEPCO	UEPSJ	14.00	90.00	90.00				15.69				
	2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (SC)			UEPCO	UEPCM	14.00	90.00	90.00				15.69				
	2-Wire Coin Out Oper Screen & Block: 900/976, 1+DDD, 011+, & Local ; w/ Enhanced Call OPT 3YW (SC)			UEPCO	UEPCP	14.00	90.00	90.00				15.69				
LOCA	L NUMBER PORTABILITY															L
	Local Number Portability (1 per port)			UEPCO	LNPCX	0.35										<u> </u>
	2-Wire Voice Grade Loop/ Line Port Combination - Subsequent			UEPCO	USAS2		0.00	0.00				15.69				
UNBUNDLED	PORT/LOOP COMBINATIONS - MARKET BASED RATES			021 00	00/102		0.00	0.00				10.00				
2-WIR	E VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK	PORT														
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1		1			73.68										
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2		2			80.13										
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3		3			85.46										
UNEL	oop Rates 2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1		1	UEPPX	UECD1	16.68					-					
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1	-	2	UEPPX	UECD1	23.13										-
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2 2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3		3	UEPPX	UECD1	28.46										
UNE F	ort Rate		Ŭ	0EITX	OLOD!	20.40										
	Exchange Ports - 2-Wire DID Port			UEPPX	UEPD1	57.00	600.00	75.00				15.69				
NONR	ECURRING CHARGES - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination - Switch-As-Is Top 8 MSAs only			UEPPX	USAC1		125.00	75.00				15.69				
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth Allowable Changes Top 8 MSAs only			UEPPX	USA1C		125.00	75.00				15.69				
ADDIT	IONAL NRCs															
	2-Wire DID Subsequent Activity - Add Trunks, Per Trunk			UEPPX	USAS1		53.68					15.69				\vdash
Teleph	none Number/Trunk Group Establisment Charges		<u> </u>		NDT	0.00	0.00	0.00								┣────
	DID Trunk Termination (One Per Port) DID Numbers, Establish Trunk Group and Provide First Group			UEPPX	NDT	0.00	0.00	0.00								
	of 20 DID Numbers		1	UEPPX	NDZ	0.00	0.00	0.00								
1	Additional DID Numbers for each Group of 20 DID Numbers		1	UEPPX	ND4	0.00	0.00	0.00								1
	DID Numbers, Non- consecutive DID Numbers , Per Number	1	1	UEPPX	ND5	0.00	0.00	0.00		ĺ			İ			1
	Reserve Non-Consecutive DID numbers	I		UEPPX	ND6	0.00	0.00	0.00								
j	Reserve DID Numbers			UEPPX	NDV	0.00	0.00	0.00								
LOCA	L NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPPX	LNPCP	3.15	0.00	0.00								

UNBUNDL	ED NETWORK ELEMENTS - South Carolina													Attachment:	2	Exhi	bit: B
												Svc Order	Svc Order			Incremental	1
													Submitted	Charge -	Charge -	Charge -	Charge -
		Interi										Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	m	Zone	В	CS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
												•	•	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
		-															
		_					Rec	Nonrec		Nonrecurring					Rates(\$)		
0 14/15								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	RE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LI	INE SIDE	POR														<u> </u>
UNE P	Port/Loop Combination Rates 2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -	-															<u> </u>
	UNE Zone 1		1	UEPPB	UEPPR		76.90										i
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -			OLITO	OLITIK		10.50										<u> </u>
	UNE Zone 2		2	UEPPB	UEPPR		84.64										i
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -		-	02.10	02.111		0										1
	UNE Zone 3		3	UEPPB	UEPPR		90.27										i
UNE I	_oop Rates			-	-												[
	2-Wire ISDN Digital Grade Loop - UNE Zone 1		1	UEPPB	UEPPR	USL2X	21.90										
																	(
	2-Wire ISDN Digital Grade Loop - UNE Zone 2		2	UEPPB	UEPPR	USL2X	29.64										1
	2-Wire ISDN Digital Grade Loop - UNE Zone 3		3	UEPPB	UEPPR	USL2X	35.27										
UNE F	Port Rate																
	Exchange Port - 2-Wire ISDN Line Side Port			UEPPB	UEPPR	UEPPB	55.00	525.00	400.00				15.69				
NONR	ECURRING CHARGES - CURRENTLY COMBINED																Ļ
	2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port	1	1	l													1
	Combination - Conversion - Top 8 MSAs only			UEPPB	UEPPR	USACB	0.00	225.00	225.00				15.69				
	TIONAL NRCs																
LOCA	L NUMBER PORTABILITY																
	Local Number Portability (1 per port)	_		UEPPB	UEPPR	LNPCX	0.35	0.00	0.00								
B-CHA	ANNEL USER PROFILE ACCESS:	_				1141104	0.00	0.00	0.00								
	CVS/CSD (DMS/5ESS) CVS (EWSD)	-		UEPPB UEPPB	UEPPR UEPPR	U1UCA U1UCB	0.00	0.00	0.00								├ ────
	CSD CSD	-		UEPPB	UEPPR	U1UCC	0.00	0.00	0.00								<u> </u>
B CH	ANNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS S	C MG 0		UEPPB	UEPPR	01000	0.00	0.00	0.00								<u> </u>
B-CHA	CVS/CSD (DMS/5ESS)			UEPPB	UEPPR	U1UCD	0.00	0.00	0.00								<u> </u>
	CVS (EWSD)			UEPPB	UEPPR	U1UCE	0.00	0.00	0.00								<u> </u>
	CSD	1		UEPPB	UEPPR	U1UCF	0.00	0.00	0.00								<u> </u>
USER	TERMINAL PROFILE			02.10	02.111	0.00	0.00	0.00	0.00				1				i
002.1	User Terminal Profile (EWSD only)			UEPPB	UEPPR	U1UMA	0.00	0.00	0.00								1
VERT	ICAL FEATURES																
	All Vertical Features - One per Channel B User Profile			UEPPB	UEPPR	UEPVF	3.04	0.00	0.00								
INTEF	ROFFICE CHANNEL MILEAGE																1
	Interoffice Channel mileage each, including first mile and																[
	facilities termination				UEPPR	M1GNC	24.30	60.00	40.00	25.00	10.00		15.69				<u> </u>
	Interoffice Channel mileage each, additional mile			UEPPB	UEPPR	M1GNM	0.0167	0.00	0.00								
	E D\$1 DIGITAL LOOP WITH 4-WIRE ISDN D\$1 DIGITAL TRUN	K PORT															
UNE F	Port/Loop Combination Rates																Ļ
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE	1	1	l		1											1
	Zone 1		1	UEPPP			940.87										
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE	1	_			1	4 005 10										1
\vdash	Zone 2		2	UEPPP		-	1,005.43						ļ				
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE	1	2	UEPPP		1	1 444 00										1
	Zone 3	-	3	UEPPP			1,111.89						+				
UNEL	Loop Rates 4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPPP		USL4P	90.87						15.69				
	4-Wire DS1 Digital Loop - UNE Zone 1 4-Wire DS1 Digital Loop - UNE Zone 2		2	UEPPP		USL4P USL4P	155.43						15.69				
├──┼──	4-Wire DS1 Digital Loop - UNE Zone 3	-	2	UEPPP		USL4P USL4P	261.89						15.69				i
	Port Rate	1		5-11		302-1	201.09						10.09				
	Exchange Ports - 4-Wire ISDN DS1 Port			UEPPP		UEPPP	850.00	1,150.00	1,150.00				15.69				
NONF	RECURRING CHARGES - CURRENTLY COMBINED	1				1	000.00	.,	.,				.0.00				1
	4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port	1		1		1				1	1	1	t			1	
	Combination - Conversion -Switch-As-Is Top 8 MSAs only	1	1	UEPPP		USACP	0.00	950.00	950.00				15.69				1
ADDI	TIONAL NRCs	1	1	1		1				l	İ	1				ĺ	ſ
	4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy-	1	1	1		1						1					
	Inward/two way Telephone Numbers (except NC)	1	1	UEPPP		PR7TF		0.9822					15.69				1
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port -																1
	Outward Tel Numbers (All States except NC)	1	1	UEPPP		PR7TO		23.02	23.02	1	1	1	15.69				1

UNBUNDL	ED NETWORK ELEMENTS - South Carolina												Attachment:			ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Charge -
						Rec	Nonrec			g Disconnect				Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port -															
	Subsequent Inward Telephone Numbers			UEPPP	PR7ZT		46.05	46.05				15.69				
LOCA	L NUMBER PORTABILITY Local Number Portability (1 per port)	_		UEPPP	LNPCN	1.75				-						
INITE	RFACE (Provsioning Only)			UEPPP	LINPCIN	1.75										<u> </u>
	Voice/Data		-	UEPPP	PR71V	0.00	0.00	0.00								
	Digital Data			UEPPP	PR71D	0.00	0.00	0.00								
	Inward Data			UEPPP	PR71E	0.00	0.00	0.00								
New o	or Additional "B" Channel															1
	New or Additional - Voice/Data B Channel			UEPPP	PR7BV	0.00	40.00									
	New or Additional - Digital Data B Channel			UEPPP	PR7BF	0.00	40.00									
	New or Additional Inward Data B Channel	ļ	L	UEPPP	PR7BD	0.00	40.00				ļ					L
CALL	TYPES	<u> </u>			DD7C4	0.00	0.00	0.00		ļ	l					
├ ── │	Inward Outward	 		UEPPP UEPPP	PR7C1 PR7C0	0.00	0.00	0.00		<u> </u>		ļ				───
	Outward Two-way			UEPPP	PR7C0 PR7CC	0.00	0.00	0.00								
Interc	office Channel Mileage			UEFFF	PR/CC	0.00	0.00	0.00								
interc	Fixed Each Including First Mile	-		UEPPP	1LN1A	77.4815	89.47	81.99	16.39	14.48		15.69				
	Each Airline-Fractional Additional Mile			UEPPP	1LN1B	0.3415	00.11	01.00	10.00			10.00				
4-WIF	RE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT			-												
UNE	Port/Loop Combination Rates	1														
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1		1	UEPDC		840.87										
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2			UEPDC		905.43										
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3		3	UEPDC		1,011.89										
UNE	Loop Rates															
	4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPDC UEPDC	USLDC USLDC	90.87 155.43										
	4-Wire DS1 Digital Loop - UNE Zone 2 4-Wire DS1 Digital Loop - UNE Zone 3		2	UEPDC	USLDC	261.89										
UNE	Port Rate		3	OLFDC	USLDC	201.09										
0112	4-Wire DDITS Digital Trunk Port		1	UEPDC	UDD1T	750.00	1,005.07	478.99	213.53	20.94		15.69				
NONF	RECURRING CHARGES - CURRENTLY COMBINED			02. 50	00011	100.00	1,000.01	110.00	210.00	20101		10.00				
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
	- Switch-As-Is Top 8 MSAs only			UEPDC	USAC4		259.56	134.33				15.69				
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
	 Conversion with DS1 Changes Top 8 MSAs only 			UEPDC	USAWA		259.56	134.33				15.69				
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination	1	1													
	4-Wire DS1 Digital Loop / 4-Wire DD11S Trunk Port Combination - Conversion with Change - Trunk Top 8 MSAs only	1	1	UEPDC	USAWB		259.56	134.33				15.69				
	TIONAL NRCs	+		021 00	USAND		209.00	104.00	1	1	1	15.09				<u> </u>
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC -	1	1		1				1	ł					1	
	Subsequent Channel Activation/Chan - 2-Way Trunk	1	1	UEPDC	UDTTA		29.01	29.01				15.69				
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent	1								1	1					
	Channel Activation/Chan - 1-Way Outward Trunk			UEPDC	UDTTB		29.01	29.01				15.69				
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel	1														
	Activation/Chan Inward Trunk w/out DID		<u> </u>	UEPDC	UDTTC		29.01	29.01				15.69				
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan	1	1				00.01	00.04				45.00				
	Activation Per Chan - Inward Trunk with DID 4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan	+		UEPDC	UDTTD		29.01	29.01				15.69				
	Activation / Chan - 2-Way DID w User Trans		1	UEPDC	UDTTE		29.01	29.01				15.69				
BIPO	LAR 8 ZERO SUBSTITUTION	1	1		JUTTE		23.01	20.01		<u> </u>		15.08				<u> </u>
	B8ZS -Superframe Format	1	1	UEPDC	CCOSF		0.00	605.00	İ	İ	1				İ	
	B8ZS - Extended Superframe Format	1	1	UEPDC	CCOEF		0.00	605.00			1					
Alterr	nate Mark Inversion															
	AMI -Superframe Format			UEPDC	MCOSF		0.00	0.00								
	AMI - Extended SuperFrame Format	<u> </u>		UEPDC	MCOPO		0.00	0.00			ļ					L
Telep	hone Number/Trunk Group Establisment Charges	 	Ļ		LIDTOX	0.05				ļ	ļ					L
├ ── ├ ──	Telephone Number for 2-Way Trunk Group			UEPDC	UDTGX	0.00				l	l	15.69				┫
	Telephone Number for 1-Way Outward Trunk Group	1		UEPDC	UDTGY	0.00				L		15.69		1		L

INBUNDLE	D NETWORK ELEMENTS - South Carolina												Attachment:			bit: B
ATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			Submitted Elec	Submitted Manually	Manual Svc	Charge - Manual Svc	Incremental Charge - Manual Svc	
ATEGORT	RATE ELEMENTS	m	Zone	BC3	0300			KATE3(\$)			per LSR	per LSR	Order vs. Electronic- 1st	Order vs. Electronic- Add'l	Order vs. Electronic- Disc 1st	Order vs. Electronic- Disc Add'l
						Rec	Nonrec	urring	Nonrecurring	g Disconnect				Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Telephone Number for 1-Way Inward Trunk Group Without DID			UEPDC	UDTGZ	0.00						15.69				
	DID Numbers, Establish Trunk Group and Provide First Group of 20 DID Numbers			UEPDC	NDZ	0.00	0.00	0.00				15.69				
	DID Numbers for each Group of 20 DID Numbers	-	-	UEPDC	ND2 ND4	0.00	0.00	0.00				15.69				
	DID Numbers Non- consecutive DID Numbers . Per Number		-	UEPDC	ND5	0.00	0.00	0.00				15.69				
	Reserve Non-Consecutive DID Nos.			UEPDC	ND6	0.00	0.00	0.00				15.69				
	Reserve DID Numbers			UEPDC	NDV	0.00	0.00	0.00				15.69				
Dedica	ated DS1 (Interoffice Channel Mileage) -															
FX/FC	O for 4-Wire DS1 Digital Loop with 4-Wire DDITS Trunk Port															
	Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities															
	Termination)			UEPDC	1LNO1	77.14	89.47	81.99	16.39	14.48		15.69				
	Interoffice Channel Mileage - Additional rate per mile - 0-8 miles Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities		<u> </u>	UEPDC	1LNOA	0.3415	0.00	0.00								
	Termination)			UEPDC	1LNO2	0.00	0.00	0.00								
	Interoffice Channel Mileage - Additional rate per mile - 9-25 miles			UEPDC	1LNOB	0.7598	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Termination)			UEPDC	1LNO3	0.00	0.00	0.00								
	Interoffice Channel Mileage - Additional rate per mile - 25+ miles			UEPDC	1LNOC	0.7598	0.00	0.00								
	Local Number Portability, per DS0 Activated			UEPDC	LNPCP	3.15	0.00	0.00								
	Central Office Termininating Point			UEPDC	CTG	0.00										
	E DS1 LOOP WITH CHANNELIZATION WITH PORT															
	n is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Acti em can have various rate combinations based on type and nu			l												
	en can have various rate combinations based on type and hui S1 Loop		pons	useu	-											
	4-Wire DS1 Loop - UNE Zone 1		1	UEPMG	USLDC	90.87	0.00	0.00								
	4-Wire DS1 Loop - UNE Zone 2		2	UEPMG	USLDC	155.43	0.00	0.00				1				
	4-Wire DS1 Loop - UNE Zone 3		3	UEPMG	USLDC	261.89	0.00	0.00								
UNE D	SO Channelization Capacities (D4 Channel Bank Configuration	ns)														
	24 DSO Channel Capacity - 1 per DS1			UEPMG	VUM24	103.47	0.00	0.00				15.69				
	48 DSO Channel Capacity - 1 per 2 DS1s			UEPMG	VUM48	206.94	0.00	0.00				15.69				
	96 DSO Channel Capacity -1per 4 DS1s			UEPMG	VUM96	413.88	0.00	0.00				15.69				
	144 DS0 Channel Capacity - 1 per 6 DS1s			UEPMG	VUM14	620.82	0.00	0.00				15.69				
	192 DS0 Channel Capacity -1 per 8 DS1s		 	UEPMG	VUM19	827.76	0.00	0.00			L	15.69				<u> </u>
	240 DS0 Channel Capacity - 1 per 10 DS1s	ļ	<u> </u>	UEPMG UEPMG	VUM20 VUM28	1,034.70	0.00	0.00			ļ	15.69				ļ
	288 DS0 Channel Capacity - 1 per 12 DS1s 384 DS0 Channel Capacity - 1 per 16 DS1s			UEPMG	VUM28 VUM38	1,241.64 1,655.52	0.00	0.00				15.69 15.69				
	480 DS0 Channel Capacity - 1 per 20 DS1s			UEPMG	VUM40	2,069.40	0.00	0.00			-	15.69				-
	576 DS0 Channel Capacity -1 per 24 DS1s	İ	1	UEPMG	VUM57	2,483.28	0.00	0.00			1	15.69				1
	672 DS0 Channel Capacity - 1 per 28 DS1s	1	<u> </u>	UEPMG	VUM67	2,403.20	0.00	0.00			1	15.69	1	1	1	t
Non-R	ecurring Charges (NRC) Associated with 4-Wire DS1 Loop with	h Chanı	neliztio					2.00					İ		İ	
	mum System configuration is One (1) DS1, One (1) D4 Channe										1	1	İ	ĺ	İ	1
	les of this configuration functioning as one are considered Ac															
	NRC - Conversion (Currently Combined) with or without			UEPMG	USAC4	0.00	150.81	8.38				15.69				
System	BellSouth Allowed Changes - Top 8 MSAs Only n Additions Where Currently Combined and New (Not Currenti	v Com	l hined \		03464	0.00	18.001	ö.38				90.01				┝───
	sity Zone 1 Top 8 MSAs															
	1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc									17.00		15.00				
Pinele	Fea Activation -			UEPMG	VUMD4	0.00	717.71	425.81	149.08	17.69		15.69				
Bipola	Clear Channel Capability Format, superframe - Subsequent															
	Activity Only			UEPMG	CCOSF	0.00	0.00	605.00								
1	Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only			UEPMG	CCOEF	0.00	0.00	605.00								
	Subsequent notifity only															
Alterna	ate Mark Inversion (AMI)															
Alterna				UEPMG UEPMG	MCOSF MCOPO	0.00	0.00	0.00								

	ONETWORK ELEMENTS - South Carolina	-		-									Attachment:		Exhi	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrec		Nonrecurring					Rates(\$)	_	
		L	<u> </u>				First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	ge Ports Associated with 4-Wire DS1 Loop with Channelization	on with	Port		-											
Exchange	ge Ports	-	-		-											
	Line Side Combination Channelized PBX Trunk Port - Business			UEPPX	UEPCX	14.00	0.00	0.00	0.00	0.00		15.69				
	Line Side Combination Channelized PBX Trunk Port - Business			UEPPX	UEPOX	14.00	0.00	0.00	0.00	0.00		15.69	-			
				ULFFA	ULFUX	14.00	0.00	0.00	0.00	0.00		15.09				
	Line Side Inward Only Channelized PBX Trunk Port without DID			UEPPX	UEP1X	14.00	0.00	0.00	0.00	0.00		15.69				
	2-Wire Trunk Side Unbundled Channelized DID Trunk Port			UEPPX	UEPDM	57.00	0.00	0.00	0.00	0.00		15.69				
	Activations - Unbundled Loop Concentration			02.1.7	02.0.	01.00	0.00	0.00	0.00	0.00		10.00				
	Feature (Service) Activation for each Line Port Terminated in D4															
	Bank			UEPPX	1PQWM	0.70	40.00	20.00	6.00	5.00		15.69				
	Feature (Service) Activation for each Trunk Port Terminated in	1														
	D4 Bank	1		UEPPX	1PQWU	0.70	110.00	30.00	65.00	20.00		15.69				
	one Number/ Group Establishment Charges for DID Service	1	1													
	DID Trunk Termination (1 per Port)			UEPPX	NDT	0.00	0.00	0.00				15.69				
	Estab Trk Grp and Provide 1st 20 DID Nos. (FL,GA, NC,& SC)			UEPPX	NDZ	0.00	0.00	0.00				15.69				
	DID Numbers - groups of 20 - Valid all States			UEPPX	ND4	0.00	0.00	0.00				15.69				
	Non-Consecutive DID Numbers - per number			UEPPX	ND5	0.00	0.00	0.00				15.69				
	Reserve Non-Consecutive DID Numbers			UEPPX	ND6	0.00	0.00	0.00				15.69				
	Reserve DID Numbers			UEPPX	NDV	0.00	0.00	0.00				15.69				
	umber Portability															
	Local Number Portability - 1 per port			UEPPX	LNPCP	3.15	0.00	0.00								
	RES - Vertical and Optional															
Local S	witching Features Offered with Line Side Ports Only															
	All Features Available			UEPPX	UEPVF	3.04	0.00	0.00				15.69				
INBUNDLED C	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATES											15.69				
JNBUNDLED C	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATES Based Rates are applied where BellSouth is required by FCC	and/or		Commission rule to	provide Unb	undled Local Sv	witching or Sw	itch Ports.				15.69				
NBUNDLED C 1. Cost 2. Featu	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC res shall apply to the Unbundled Port/Loop Combination - C	and/or	sed Rat	Commission rule to te section in the sa	provide Unb me manner as	undled Local Sv they are applie	witching or Sw d to the Stand	itch Ports. -Alone Unbun								
NBUNDLED C 1. Cost 2. Featu 3. End C	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC ires shall apply to the Unbundled Port/Loop Combination - C Diffice and Tandem Switching Usage and Common Transport	and/or Cost Bas Usage	sed Rat rates ir	Commission rule to te section in the sa n the Port section o	provide Unb me manner as of this rate exh	undled Local Sv they are applie ibit shall apply	witching or Sw d to the Stand to all combina	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo				
NBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC res shall apply to the Unbundled Port/Loop Combination - C Office and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not C	and/or Cost Bas Usage	sed Rat rates ir	Commission rule to te section in the sa n the Port section o	provide Unb me manner as of this rate exh	undled Local Sv they are applie ibit shall apply	witching or Sw d to the Stand to all combina	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
NBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi apply al	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC rres shall apply to the Unbundled Port/Loop Combination - C Office and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not Co lso and are categorized accordingly.	and/or Cost Bas Usage urrently	sed Rat rates ir / Comb	Commission rule to te section in the sain the Port section of ined Combos. Fo	provide Unb me manner as of this rate exh r Currently Co	undled Local Sv they are applie ibit shall apply mbined Combo	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
NBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi apply al 5. Mark	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC ires shall apply to the Unbundled Port/Loop Combination - C Office and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not C iso and are categorized accordingly. et Rates for Unbundled Centrex Port/Loop Combination will	and/or Cost Bas Usage urrently	sed Rat rates ir / Comb	Commission rule to te section in the sain the Port section of ined Combos. Fo	provide Unb me manner as of this rate exh r Currently Co	undled Local Sv they are applie ibit shall apply mbined Combo	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
NBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi apply al 5. Mark UNE-P C	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC res shall apply to the Unbundled Port/Loop Combination - C Office and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not C iso and are categorized accordingly. et Rates for Unbundled Centrex Port/Loop Combination will ENTREX - 5ESS (Valid in All States)	and/or Cost Bas Usage urrently	sed Rat rates ir / Comb	Commission rule to te section in the sain the Port section of ined Combos. Fo	provide Unb me manner as of this rate exh r Currently Co	undled Local Sv they are applie ibit shall apply mbined Combo	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
NBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi apply al 5. Mark UNE-P C 2-Wire N	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC Diffice and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not Cr iso and are categorized accordingly. et Rates for Unbundled Centrex Port/Loop Combination will CENTREX - SESS (Valid in All States) /G Loop/2-Wire Voice Grade Port (Centrex) Combo	and/or Cost Bas Usage urrently	sed Rat rates ir / Comb	Commission rule to te section in the sain the Port section of ined Combos. Fo	provide Unb me manner as of this rate exh r Currently Co	undled Local Sv they are applie ibit shall apply mbined Combo	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
NBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi apply al 5. Mark UNE-P UNE Po	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC trees shall apply to the Unbundled Port/Loop Combination - C Office and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not C iso and are categorized accordingly. tet Rates for Unbundled Centrex Port/Loop Combination will CENTREX - 5ESS (Valid in All States) /G Loop/2-Wire Voice Grade Port (Centrex) Combo rt/Loop Combination Rates (Non-Design)	and/or Cost Bas Usage urrently	sed Rat rates ir / Comb	Commission rule to te section in the sain the Port section of ined Combos. Fo	provide Unb me manner as of this rate exh r Currently Co	undled Local Sv they are applie ibit shall apply mbined Combo	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
NBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi apply al 5. Mark UNE-P C 2-Wire N UNE PO	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC tres shall apply to the Unbundled Port/Loop Combination - C Difice and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not C Iso and are categorized accordingly. Let Rates for Unbundled Centrex Port/Loop Combination will CENTREX - 5ESS (Valid in All States) //Loop/2-Wire Voice Grade Port (Centrex) Combo rt/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo	and/or Cost Bas Usage urrently	sed Rat rates ir / Comb	Commission rule to commission rule to te section in the sa n the Port section o ined Combos. Fo on an Individual C	provide Unb me manner as of this rate exh r Currently Co	undled Local S they are applie ibit shall apply mbined Combo til further notice	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
NBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi apply al 5. Mark UNE-P C 2-Wire N UNE Po	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC tres shall apply to the Unbundled Port/Loop Combination - C Office and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not C iso and are categorized accordingly. et Rates for Unbundled Centrex Port/Loop Combination will CENTREX - 5ESS (Valid in All States) /G Loop/2-Wire Voice Grade Port (Centrex) Combo rt/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design	and/or Cost Bas Usage urrently	sed Rat rates ir / Comb	Commission rule to te section in the sain the Port section of ined Combos. Fo	provide Unb me manner as of this rate exh r Currently Co	undled Local Sv they are applie ibit shall apply mbined Combo	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
INBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi apply al 5. Mark UNE-PO UNE PO	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC ires shall apply to the Unbundled Por/Loop Combination - C Office and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not Cr iso and are categorized accordingly. Let Rates for Unbundled Centrex Port/Loop Combination will DENTREX - 5ESS (Valid in All States) //6 Loop/2-Wire Voice Grade Port (Centrex) Combo rt/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo	and/or Cost Bas Usage urrently	sed Rat rates ir / Comb otiated	Commission rule to te section in the sa inte Port section o ined Combos. Fo on an Individual C UEP95	provide Unb me manner as of this rate exh r Currently Co	undled Local Sv they are applie ibit shall apply mbined Combo til further notice 14.89	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
INBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi apply al 5. Mark UNE-P C 2-Wire N UNE PO	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC tres shall apply to the Unbundled Port/Loop Combination - C Difice and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not C Iso and are categorized accordingly. ter Rates for Unbundled Centrex Port/Loop Combination will CENTREX - 5ESS (Valid in All States) //G Loop/2-Wire Voice Grade Port (Centrex) Combor rt/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design	and/or Cost Bas Usage urrently	sed Rat rates ir / Comb	Commission rule to commission rule to te section in the sa n the Port section o ined Combos. Fo on an Individual C	provide Unb me manner as of this rate exh r Currently Co	undled Local S they are applie ibit shall apply mbined Combo til further notice	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
NBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi apply al 5. Mark UNE-P C 2-Wire V UNE PO	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC tres shall apply to the Unbundled Port/Loop Combination - C Difice and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not Cr Iso and are categorized accordingly. Let Rates for Unbundled Centrex Port/Loop Combination will CENTREX - 5ESS (Valid in All States) 76 Loop/2-Wire Voice Grade Port (Centrex) Combo rt/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Win-VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Win-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo	and/or Cost Bas Usage urrently	sed Rat rates ir rotiated 1 2	Commission rule to te section in the sa inte Port section o ined Combos. Fo on an Individual C UEP95 UEP95	provide Unb me manner as of this rate exh r Currently Co	undled Local Sv they are applie ibit shall apply mbined Combo til further notice 14.89 21.52	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
NBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi apply al 5. Mark UNE-PO UNE PO UNE PO	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC irres shall apply to the Unbundled Port/Loop Combination - C Office and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not Cr iso and are categorized accordingly. Let Rates for Unbundled Centrex Port/Loop Combination will DENTREX - 5ESS (Valid in All States) //G Loop/2-Wire Voice Grade Port (Centrex) Combo rt/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo	and/or Cost Bas Usage urrently	sed Rat rates ir / Comb otiated	Commission rule to te section in the sa inte Port section o ined Combos. Fo on an Individual C UEP95	provide Unb me manner as of this rate exh r Currently Co	undled Local Sv they are applie ibit shall apply mbined Combo til further notice 14.89	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
INBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi apply al 5. Mark UNE-P C 2-Wire V UNE PO UNE PO UNE PO	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC tres shall apply to the Unbundled Port/Loop Combination - C Difice and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not Cl iso and are categorized accordingly. tet Rates for Unbundled Centrex Port/Loop Combination will ZENTREX - 5ESS (Valid in All States) /G Loop/2-Wire Voice Grade Port (Centrex) Combo rt/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo	and/or Cost Bas Usage urrently	sed Rat rates ir rotiated 1 2	Commission rule to te section in the sa inte Port section o ined Combos. Fo on an Individual C UEP95 UEP95	provide Unb me manner as of this rate exh r Currently Co	undled Local Sv they are applie ibit shall apply mbined Combo til further notice 14.89 21.52	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
JNBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi apply al 5. Mark UNE-P C 2-Wire N UNE PO UNE PO UNE PO	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC tres shall apply to the Unbundled Port/Loop Combination - C Difice and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not C Iso and are categorized accordingly. ter Rates for Unbundled Centrex Port/Loop Combination will CENTREX - 5ESS (Valid in All States) VG Loop/2-Wire Voice Grade Port (Centrex) Combor rt/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo	and/or Cost Bas Usage urrently	sed Rat rates ir rotiated 1 2	Commission rule to te section in the sain inte Port section of ined Combos. Fo on an Individual C UEP95 UEP95 UEP95	provide Unb me manner as of this rate exh r Currently Co	undled Local Sv they are applie ibit shall apply mbined Combo til further notice 14.89 21.52 27.17	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
INBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi apply al 5. Mark UNE-PO UNE PO UNE PO UNE PO	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC tres shall apply to the Unbundled Por/Loop Combination - C Office and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not Cr iso and are categorized accordingly. tet Rates for Unbundled Centrex Port/Loop Combination will CENTREX - 5ESS (Valid in All States) // G Loop/2-Wire Voice Grade Port (Centrex) Combo rt/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design rt/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design rt/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design rt/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Don-Design rt/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Don-Design rt/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo	and/or Cost Bas Usage urrently	sed Rat rates ir rotiated 1 2	Commission rule to te section in the sa inte Port section o ined Combos. Fo on an Individual C UEP95 UEP95	provide Unb me manner as of this rate exh r Currently Co	undled Local Sv they are applie ibit shall apply mbined Combo til further notice 14.89 21.52	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
INBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi apply al 5. Mark UNE-P C 2-Wire N UNE PO UNE PO UNE PO	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC trees shall apply to the Unbundled Port/Loop Combination - C Diffice and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not Cl iso and are categorized accordingly. Set Rates for Unbundled Centrex Port/Loop Combination will ZENTREX - 5ESS (Valid in All States) //G Loop/2-Wire Voice Grade Port (Centrex) Combo- rt/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo- Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design rt/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design rt/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design rt/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design rt/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design rt/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design	and/or Cost Bas Usage urrently	ed Rat rates ir comb otiated 1 2 3 1	Commission rule to te section in the san in the Port section of ined Combos. Fo on an Individual C UEP95 UEP95 UEP95 UEP95	provide Unb me manner as of this rate exh r Currently Co	undled Local St they are applie ibit shall apply mbined Combo til further notice 14.89 21.52 27.17 17.81	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
NBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi apply al 5. Mark UNE-P C 2-Wire N UNE PO UNE PO UNE PO	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC tres shall apply to the Unbundled Port/Loop Combination - C Difice and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not C Iso and are categorized accordingly. ter Rates for Unbundled Centrex Port/Loop Combination will CENTREX - 5ESS (Valid in All States) // Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo	and/or Cost Bas Usage urrently	sed Rat rates ir rotiated 1 2	Commission rule to te section in the sain inte Port section of ined Combos. Fo on an Individual C UEP95 UEP95 UEP95	provide Unb me manner as of this rate exh r Currently Co	undled Local Sv they are applie ibit shall apply mbined Combo til further notice 14.89 21.52 27.17	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
NBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi apply al 5. Mark UNE-PO UNE PO UNE PO UNE PO	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC tres shall apply to the Unbundled Port/Loop Combination - C Dffice and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not Cr iso and are categorized accordingly. tet Rates for Unbundled Centrex Port/Loop Combination will CENTREX - 5ESS (Valid in All States) //G Loop/2-Wire Voice Grade Port (Centrex) Port Combo rt/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design rt/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design -Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design -Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design -Wire VG Loop/2-Wire Voice Grade	and/or Cost Bas Usage urrently	eed Rat rates ir / Comb otiated 1 2 3 1 2	Commission rule to te section in the sain inte Port section of ined Combos. Fo on an Individual C UEP95 UEP95 UEP95 UEP95 UEP95	provide Unb me manner as of this rate exh r Currently Co	undled Local Sy they are applie ibit shall apply mbined Combo til further notice 14.89 21.52 27.17 17.81 24.26	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
INBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi apply al 5. Mark UNE-PC UNE PO UNE PO UNE PO	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC tres shall apply to the Unbundled Port/Loop Combination - C Office and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not Cr Iso and are categorized accordingly. tet Rates for Unbundled Centrex Port/Loop Combination will DENTREX - 5ESS (Valid in All States) //6 Loop/2-Wire Voice Grade Port (Centrex) Combo //1/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design //Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design -Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design -Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design -Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design -Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design -Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design -Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design -Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design -Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design -Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design -Design -	and/or Cost Bas Usage urrently	ed Rat rates ir comb otiated 1 2 3 1	Commission rule to te section in the san in the Port section of ined Combos. Fo on an Individual C UEP95 UEP95 UEP95 UEP95	provide Unb me manner as of this rate exh r Currently Co	undled Local St they are applie ibit shall apply mbined Combo til further notice 14.89 21.52 27.17 17.81	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
NBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi apply al 5. Mark UNE-P C 2-Wire V UNE PO UNE PO UNE PO UNE PO	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC trees shall apply to the Unbundled Port/Loop Combination - C Difice and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not CI iso and are categorized accordingly. tet Rates for Unbundled Centrex Port/Loop Combination will EENTREX - 5ESS (Valid in All States) //G Loop/2-Wire Voice Grade Port (Centrex) Combo rt/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire	and/or Cost Bas Usage urrently	eed Rat rates ir / Comb otiated 1 2 3 1 2	Commission rule to te section in the sam in the Port section of inned Combos. Fo on an Individual C UEP95 UEP95 UEP95 UEP95 UEP95 UEP95	provide Unb me manner as f this rate exh r Currently Co ase Basis, un	undled Local Sv they are applie ibit shall apply mbined Combo til further notice 14.89 21.52 27.17 17.81 24.26 29.59	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
NBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi apply al 5. Mark UNE-P C 2-Wire N UNE PO UNE PO UNE PO UNE PO UNE PO	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC pres shall apply to the Unbundled Port/Loop Combination - C Difice and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not Cr Iso and are categorized accordingly. Let Rates for Unbundled Centrex Port/Loop Combination will CENTREX - 5ESS (Valid in All States) 7G Loop/2-Wire Voice Grade Port (Centrex) Combo rt/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Dosign 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design - Design - Design - Des	and/or Cost Bas Usage urrently	sed Rat rates ir / Comb otiated 1 2 3 3 1 2 3 1 1	Commission rule to te section in the sain inte Port section of ined Combos. Fo on an Individual C UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95	UECS1	undled Local Sy they are applie ibit shall apply mbined Combo til further notice 14.89 21.52 27.17 17.81 24.26 29.59 13.76	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
NBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi apply al 5. Mark UNE-PO UNE PO UNE PO UNE PO	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC tres shall apply to the Unbundled Port/Loop Combination - C Office and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not Cr Iso and are categorized accordingly. et Rates for Unbundled Centrex Port/Loop Combination will DENTREX - 5ESS (Valid in All States) //6 Loop/2-Wire Voice Grade Port (Centrex) Combo ///Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design //Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2	and/or Cost Bas Usage urrently	sed Rat rates ir comb otiated 1 1 2 3 1 2 3 1 2 1 1 2	Commission rule to te section in the sa inte Port section of ined Combos. Fo on an Individual C UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95	UECS1	undled Local Sy they are applie ibit shall apply mbined Combo til further notice 21.52 27.17 17.81 24.26 29.59 13.76 20.38	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
NBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi apply al 5. Mark UNE-P C 2-Wire V UNE PO UNE P	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC trees shall apply to the Unbundled Port/Loop Combination - C Difice and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not CI iso and are categorized accordingly. et Rates for Unbundled Centrex Port/Loop Combination will EENTREX - 5ESS (Valid in All States) //G Loop/2-Wire Voice Grade Port (Centrex) Combo //t/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3	and/or Cost Bas Usage urrently	sed Ratirates in a composition of the second	Commission rule to commission rule to te section in the sain n the Port section of ined Combos. Fo on an Individual C UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95	UECS1 UECS1 UECS1	undled Local Sv they are applie ibit shall apply mbined Combo til further notice 14.89 21.52 27.17 17.81 24.26 29.59 13.76 20.38 26.04	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
NBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi apply al 5. Mark UNE-P C 2-Wire N UNE PO UNE PO UNE PO UNE PO UNE PO	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC Based Rates are applied where BellSouth is required by FCC res shall apply to the Unbundled Port/Loop Combination - C Difice and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not C Iso and are categorized accordingly. Ext Rates for Unbundled Centrex Port/Loop Combination will EENTREX - 5ESS (Valid in All States) //G Loop/2-Wire Voice Grade Port (Centrex) Combo //tLoop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Combination Rates (Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Combined Comp (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1	and/or Cost Bas Usage urrently	sed Ratarates in v comb otiated 1 1 2 3 1 1 2 3 1 1 2 3 1 1 2 3 1 1 2 3 1 1 2 3 1 1 2 3 1 1 2 3 1 1 2 3 1 1 1 2 3 3 1 1 1 2 3 1 1 1 2 1 2	Commission rule to te section in the sat in the Port section of ined Combos. Fo on an Individual C UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95	UECS1 UECS1 UECS2	undled Local Sv they are applie ibit shall apply mbined Combo til further notice 21.52 27.17 17.81 24.26 29.59 13.76 20.38 26.04 16.68	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
NBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi apply al 5. Mark UNE-PO UNE PO UNE PO UNE PO UNE PO	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC tres shall apply to the Unbundled Por/Loop Combination - C Office and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not Cr iso and are categorized accordingly. tet Rates for Unbundled Centrex Port/Loop Combination will CENTREX - 5ESS (Valid in All States) //G Loop/2-Wire Voice Grade Port (Centrex) Combo rt/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design rt/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 2	and/or Cost Bas Usage urrently	sed Ratirates in a combination of the set of	Commission rule to te section in the sati inte Port section of ined Combos. Fo on an Individual C UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95	UECS1 UECS1 UECS1 UECS2 UECS2	undled Local Sv they are applie ibit shall apply mbined Combo til further notice 21.52 27.17 27.17 17.81 24.26 29.59 13.76 20.38 26.04 16.68 23.13	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
NBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi apply al 5. Mark UNE-P C 2-Wire N UNE PO UNE PO UNE PO UNE PO UNE PO UNE LO UNE LO	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC trees shall apply to the Unbundled Port/Loop Combination - C Diffice and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not CI iso and are categorized accordingly. et Rates for Unbundled Centrex Port/Loop Combination will EENTREX - 5ESS (Valid in All States) //G Loop/2-Wire Voice Grade Port (Centrex) Combo //t/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design //Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design //Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design //Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design //Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design //Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design //Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design //Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design //Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone	and/or Cost Bas Usage urrently	sed Ratarates in v comb otiated 1 1 2 3 1 1 2 3 1 1 2 3 1 1 2 3 1 1 2 3 1 1 2 3 1 1 2 3 1 1 2 3 1 1 2 3 1 1 1 2 3 3 1 1 1 2 3 1 1 1 2 1 2	Commission rule to te section in the sat in the Port section of ined Combos. Fo on an Individual C UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95	UECS1 UECS1 UECS2	undled Local Sv they are applie ibit shall apply mbined Combo til further notice 21.52 27.17 17.81 24.26 29.59 13.76 20.38 26.04 16.68	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
INBUNDLED C I. Cost 2. Featu 3. End C 4. The fi apply al 5. Mark UNE-P C 2-Wire N UNE PO UNE PO UNE PO UNE LO UNE LO	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC res shall apply to the Unbundled Port/Loop Combination - C Difice and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not C iso and are categorized accordingly. tet Rates for Unbundled Centrex Port/Loop Combination will CENTREX - 5ESS (Valid in All States) //G Loop/2-Wire Voice Grade Port (Centrex) Combo //t/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wi	and/or Cost Bas Usage urrently	sed Ratirates in a combination of the set of	Commission rule to te section in the sati inte Port section of ined Combos. Fo on an Individual C UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95	UECS1 UECS1 UECS1 UECS2 UECS2	undled Local Sv they are applie ibit shall apply mbined Combo til further notice 21.52 27.17 27.17 17.81 24.26 29.59 13.76 20.38 26.04 16.68 23.13	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may
JINBUNDLED C 1. Cost 2. Featu 3. End C 4. The fi apply al 5. Mark UNE-PO UNE PO UNE PO UNE PO UNE LO UNE LO UNE LO	All Features Available ENTREX PORT/LOOP COMBINATIONS - COST BASED RATE: Based Rates are applied where BellSouth is required by FCC res shall apply to the Unbundled Port/Loop Combination - C Difice and Tandem Switching Usage and Common Transport irst and additional Port nonrecurring charges apply to Not C iso and are categorized accordingly. tet Rates for Unbundled Centrex Port/Loop Combination will CENTREX - 5ESS (Valid in All States) //G Loop/2-Wire Voice Grade Port (Centrex) Combo //t/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wi	and/or Cost Bas Usage urrently	sed Ratirates in a combination of the set of	Commission rule to te section in the sati inte Port section of ined Combos. Fo on an Individual C UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95 UEP95	UECS1 UECS1 UECS1 UECS2 UECS2	undled Local Sv they are applie ibit shall apply mbined Combo til further notice 21.52 27.17 27.17 17.81 24.26 29.59 13.76 20.38 26.04 16.68 23.13	witching or Sw d to the Stand to all combina s, the nonrecu	itch Ports. Alone Unbun tions of loop/	port network el	ements excep	t for UNE C	oin Port/Lo			Additional NR	Cs may

BUNDLED	NETWORK ELEMENTS - South Carolina												Attachment:	2	Exhi	bit: B
FEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local			UEP95	UEPYH	1.13	40.30	19.90	24.98	6.65		15.69				
	Area 2-Wire Voice Grade Port (Centrex from diff Serving Wire			UEP95	UEPTH	1.13	40.30	19.90	24.98	6.63		15.69				
(Center)2 Basic Local Area			UEP95	UEPYM	1.13	108.36	70.71	54.47	11.94		15.69				
-	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term - Basic Local Area			UEP95	UEPYZ	1.13	108.36	70.71	54.47	11.94		15.69				
2	2-Wire Voice Grade Port terminated in on Megalink or equivalent															
	Basic Local Area			UEP95	UEPY9	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port Terminated on 800 Service Term - Basic Local Area			UEP95	UEPY2	1.13	40.30	19.90	24.98	6.65		15.69				
	LA, MS, SC, & TN Only			UEP95	UEP 12	1.13	40.30	19.90	24.90	0.03		15.69				
	2-Wire Voice Grade Port (Centrex)			UEP95	UEPQA	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex 9)	1	1	UEP95	UEPQB	1.13	40.30	19.90	24.98	6.65	1	15.69				
	2-Wire Voice Grade Port (Centrex with Caller ID)1	1	1	UEP95	UEPQH	1.13	40.30	19.90	24.98	6.65	1	15.69		1	1	
2	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP95	UEPQM	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			021 00		1.10	100.00	70.71	04.47	11.04		10.00				
	Term			UEP95	UEPQZ	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP95	UEPQ9	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP95	UEPQ2	1.13	40.30	19.90	24.98	6.65		15.69				
Local S																
	Centrex Intercom Funtionality, per port			UEP95	URECS	0.7996										
Local N	umber Portability															
l	Local Number Portability (1 per port)			UEP95	LNPCC	0.35										
Features																
	All Standard Features Offered, per port			UEP95	UEPVF	3.04						15.69				
	All Select Features Offered, per port			UEP95	UEPVS	0.00	406.42					15.69				
	All Centrex Control Features Offered, per port			UEP95	UEPVC	3.04						15.69				
NARS					LIADOX	0.00	0.00	0.00				15.00				
	Unbundled Network Access Register - Combination Unbundled Network Access Register - Indial	-		UEP95 UEP95	UARCX UAR1X	0.00	0.00	0.00				15.69 15.69				
	Unbundled Network Access Register - Indiai	-		UEP95	UARTX	0.00	0.00	0.00				15.69				
	ineous Terminations			0LF 93	UAROA	0.00	0.00	0.00			1	13.09	-			
	runk Side															
	Trunk Side Terminations, each			UEP95	CEND6	8.86	119.57	18.78	60.03	3.77		15.69				
	Digital (1.544 Megabits)			02.00	02.100	0.00	110.01	10.110	00.00	0.11		10.00				
	DS1 Circuit Terminations, each		1	UEP95	M1HD1	73.62	202.47	95.90	72.75	2.47		15.69				
1	DS0 Channels Activated, each			UEP95	M1HDO	0.00	14.51					15.69				
	ce Channel Mileage - 2-Wire															
	nteroffice Channel Facilities Termination			UEP95	MIGBC	24.30	40.63	27.47	16.77	6.91		15.69				
	nteroffice Channel mileage, per mile or fraction of mile			UEP95	MIGBM	0.0167										
	Activations (DS0) Centrex Loops on Channelized DS1 Service	ce														
	nel Bank Feature Activations	<u> </u>			100140	0.50						45.00				
+ +	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP95	1PQWS	0.56						15.69				
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot	1	1	UEP95	1PQW6	0.56						15.69				
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Side	1		021 35	11 02/10	0.00					t	10.09				
5	Slot			UEP95	1PQW7	0.56						15.69				
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP95	1PQWP	0.56						15.69				
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP95	1PQWV	0.56						15.69				
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop	1									l I					
	Slot			UEP95	1PQWQ	0.56						15.69				
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP95	1PQWA	0.56						15.69				
	curring Charges (NRC) Associated with UNE-P Centrex															
	NRC Conversion Currently Combined Switch-As-Is with allowed		1													
	changes, per port	1		UEP95	USAC2		37.93	16.72				15.69				

UNBUNDLE	D NETWORK ELEMENTS - South Carolina												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec	urring	Nonrecurring	g Disconnect			OSS	Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	New Centrex Standard Common Block			UEP95	M1ACS	0.00	668.70					15.69				-
	New Centrex Customized Common Block NAR Establishment Charge, Per Occasion			UEP95 UEP95	M1ACC URECA	0.00	668.70 72.89					15.69 15.69				
UNF-P	CENTREX - DMS100 (Valid in All States)			021 33	UNLOA	0.00	72.03					15.05				
	VG Loop/2-Wire Voice Grade Port (Centrex) Combo															
	ort/Loop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	Non-Design		1	UEP9D		14.89										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		~			04.50										
	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		2	UEP9D		21.52										
	Non-Design		3	UEP9D		27.17										
UNE P	ort/Loop Combination Rates (Design)				1 1	21.17									-	t
-	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															İ
	Design		1	UEP9D		17.81										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design		2	UEP9D		24.26										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		3	UEP9D		29.59										
	Design oop Rate		3	UEP9D		29.59										
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9D	UECS1	13.76										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9D	UECS1	20.38										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP9D	UECS1	26.04										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9D	UECS2	16.68										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9D	UECS2	23.13										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9D	UECS2	28.46										-
	ort Rate															
ALL 5	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9D	UEPYA	1.13	40.30	19.90	24.98	6.65		15.69				ł
	2-Wire Voice Grade Port (Centrex 90 termination)Basic Local			021 00	0EI IX	1.10	40.00	10.00	24.00	0.00		10.00				
	Area			UEP9D	UEPYB	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3Basic Local															
	Area			UEP9D	UEPYC	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3Basic Local															
	Area			UEP9D	UEPYD	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5209))3 Basic Local Area			UEP9D	UEPYE	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5112))3 Basic Local			OEI 3D	OLITE	1.15	40.50	13.30	24.30	0.05		15.05				
	Area			UEP9D	UEPYF	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5312))3Basic Local															
	Area			UEP9D	UEPYG	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5008))3 Basic Local					1.10	40.00	40.00	04.00	0.05		45.00				
	Area 2-Wire Voice Grade Port (Centrex / EBS-M5208))3 Basic Local			UEP9D	UEPYT	1.13	40.30	19.90	24.98	6.65		15.69				
	Area			UEP9D	UEPYU	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5216))3 Basic Local			021 00	OEI TO	1.10	40.00	10.00	24.00	0.00		10.00				
	Area			UEP9D	UEPYV	1.13	40.30	19.90	24.98	6.65		15.69				
1	2-Wire Voice Grade Port (Centrex / EBS-M5316))3 Basic Local															1
	Area			UEP9D	UEPY3	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex with Caller ID) Basic Local Area			UEP9D	UEPYH	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp					1.13	40.30	19.90	24.98	60.0		15.09				
	Indication))3 Basic Local Area			UEP9D	UEPYW	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication))3									2.50						1
	Basic Local Area			UEP9D	UEPYJ	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)				UEPYM	1 1 2	109.30	70.74	EA 47	11.04		15 60				
1	2 Basic Local Area	L		UEP9D	UEPTIVI	1.13	108.36	70.71	54.47	11.94		15.69	1			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3															

UNBUNDLE	ED NETWORK ELEMENTS - South Carolina												Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		-
						1.00	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3 Basic Local Area			UEP9D	UEPYP	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3			UEF9D	UEPTP	1.13	106.30	70.71	54.47	11.94		15.69				ł
	Basic Local Area			UEP9D	UEPYQ	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3															
	Basic Local Area			UEP9D	UEPYR	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3					1.10	100.00	70 74	54 47			45.00				
	Basic Local Area 2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3			UEP9D	UEPYS	1.13	108.36	70.71	54.47	11.94		15.69				
	Basic Local Area			UEP9D	UEPY4	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3			02.00	02.11		100.00	10.11	0			10.00				
	Basic Local Area			UEP9D	UEPY5	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3															
	Basic Local Area			UEP9D	UEPY6	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3 Basic Local Area			UEP9D	UEPY7	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			ULF 9D	ULF17	1.15	100.30	70.71	34.47	11.94		13.09				
	Term			UEP9D	UEPYZ	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent															
	Basic Local Area			UEP9D	UEPY9	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port Terminated on 800 Service Term Basic						10.00	10.00								
	Local Area Y, LA, MS, SC, & TN Only			UEP9D	UEPY2	1.13	40.30	19.90	24.98	6.65		15.69				
AL, K	2-Wire Voice Grade Port (Centrex)			UEP9D	UEPQA	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP9D	UEPQB	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3			UEP9D	UEPQC	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3			UEP9D	UEPQD	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5209)3			UEP9D	UEPQE	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5112)3 2-Wire Voice Grade Port (Centrex / EBS-M5312)3			UEP9D UEP9D	UEPQF UEPQG	1.13 1.13	40.30 40.30	19.90 19.90	24.98 24.98	6.65 6.65		15.69 15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5512)5 2-Wire Voice Grade Port (Centrex / EBS-M5008)3			UEP9D UEP9D	UEPQG	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5208)3			UEP9D	UEPQU	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5216)3			UEP9D	UEPQV	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5316)3			UEP9D	UEPQ3	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex with Caller ID)			UEP9D	UEPQH	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp Indication)3			UEP9D	UEPQW	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication)3			UEP9D UEP9D	UEPQV	1.13	40.30	19.90	24.98	6.65		15.69	-	-		ł
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)					1.10	40.00	10.00	24.00	0.00		10.00				1
	2			UEP9D	UEPQM	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			UEP9D	UEPQO	1.13	108.36	70.71	54.47	11.94		15.69				
						1.10	100.00	70 74	54 47			45.00				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3 2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3		-	UEP9D UEP9D	UEPQP UEPQQ	1.13 1.13	108.36 108.36	70.71	54.47 54.47	11.94 11.94		15.69 15.69				
	2-Wile Voice Glade Fort (Centrex differ GWC/EBG-5203)2, 5			OLI 3D		1.15	100.00	70.71	34.47	11.34		13.03				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			UEP9D	UEPQR	1.13	108.36	70.71	54.47	11.94		15.69				
İ																
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3		<u> </u>	UEP9D	UEPQS	1.13	108.36	70.71	54.47	11.94		15.69				
├ ── ├ ──	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3			UEP9D	UEPQ4	1.13	108.36	70.71	54.47	11.94		15.69				<u> </u>
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3		1	UEP9D	UEPQ5	1.13	108.36	70.71	54.47	11.94		15.69				
			1						0			.0.00				1
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3			UEP9D	UEPQ6	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3	1	1	UEP9D	UEPQ7	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															

INBUNDLED NE	ETWORK ELEMENTS - South Carolina												Attachment:			bit: B
											Svc Order	Svc Order	Incremental	Incremental	Incremental	Increment
												Submitted	Charge -	Charge -	Charge -	Charge -
		Interi	_								Elec	Manually	Manual Svc		Manual Svc	
ATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	Electronic
													1st	Add'l	Disc 1st	Disc Add'
					+ +		Nonrec	urring	Nonrecurring	n Disconnect			055	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
							11130	Auui	11130	Add I	JOWIEC	JONIAN	JONIAN	JONIAN	JONIAN	JONIAN
2-Wi	ire Voice Grade Port terminated in on Megalink or equivalent			UEP9D	UEPQ9	1.13	40.30	19.90	24.98	6.65		15.69				
	ire Voice Grade Port Terminated on 800 Service Term			UEP9D	UEPQ2	1.13	40.30	19.90	24.98	6.65	1	15.69				
Local Switcl																
	trex Intercom Funtionality, per port			UEP9D	URECS	0.7996						15.69				
	per Portability			02.05	0.1200	0.1000						10.00				
	al Number Portability (1 per port)			UEP9D	LNPCC	0.35										
Features																
	Standard Features Offered, per port	1	1	UEP9D	UEPVF	3.04				1	1	15.69				
	Select Features Offered, per port			UEP9D	UEPVS	0.00	406.42				1	15.69				
	Centrex Control Features Offered, per port			UEP9D	UEPVC	3.04	400.42					15.69				
NARS	service control i balando binorod, por port			02.00	521 10	0.04					1	10.00				
	undled Network Access Register - Combination	<u> </u>		UEP9D	UARCX	0.00	0.00	0.00				15.69				<u> </u>
	undled Network Access Register - Inward			UEP9D	UAR1X	0.00	0.00	0.00				15.69				
	undled Network Access Register - Inward			UEP9D	UAROX	0.00	0.00	0.00				15.69				
	bus Terminations			ULFBD	UAROA	0.00	0.00	0.00				15.09				
2-Wire Trun					1 1											
	hk Side Terminations, each			UEP9D	CEND6	8.86	119.57	18.78	60.03	3.77		15.69				
	tal (1.544 Megabits)			UEF9D	CENDO	0.00	119.57	10.70	00.03	3.11		15.69				
	Circuit Terminations, each			UEP9D	M1HD1	73.62	202.47	95.90	72.75	2.47		15.69				
	Channels Activiated per Channel			UEP9D UEP9D	M1HD1 M1HD0		14.51	95.90	12.15	2.47		15.69				
	Channels Activiated per Channel Channel Mileage - 2-Wire			UEP9D	MIHDU	0.00	14.51					15.69				
	roffice Channel Facilities Termination			UEP9D	MIGBC	24.30	40.63	27.47	40.77	6.91		45.00				
							40.63	27.47	16.77	6.91		15.69				
	roffice Channel mileage, per mile or fraction of mile			UEP9D	MIGBM	0.0167										
	ivations (DS0) Centrex Loops on Channelized DS1 Service	e														
	Bank Feature Activations				100110							15.00				
Feat	ture Activation on D-4 Channel Bank Centrex Loop Slot			UEP9D	1PQWS	0.56						15.69				
	ture Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9D	1PQW6	0.56						15.69				
	ture Activation on D-4 Channel Bank FX Trunk Side Loop															
Slot				UEP9D	1PQW7	0.56					l	15.69				
	ture Activation on D-4 Channel Bank Centrex Loop Slot - erent Wire Center			UEP9D	1PQWP	0.56						15.69				
	ture Activation on D-4 Channel Bank Private Line Loop Slot			UEP9D	1PQWV	0.56						15.69				
	ture Activation on D-4 Channel Bank Tjie Line/Trunk Loop															
Slot				UEP9D	1PQWQ	0.56						15.69				
	ture Activation on D-4 Channel Bank WATS Loop Slot			UEP9D	1PQWA	0.56						15.69				
	ing Charges (NRC) Associated with UNE-P Centrex										L					
	C Conversion Currently Combined Switch-As-Is with allowed										1					1
	nges, per port			UEP9D	USAC2		37.93	16.72			L	15.69				
	Centrex Standard Common Block			UEP9D	M1ACS	0.00	668.70					15.69				
	Centrex Customized Common Block			UEP9D	M1ACC	0.00	668.70					15.69				
	R Establishment Charge, Per Occasion			UEP9D	URECA	0.00	72.89					15.69				
	quired Port for Centrex Control in 1AESS, 5ESS & EWSD															
	qures Interoffice Channel Mileage															
Note 3 - Reg	quires Specific Customer Premises Equipment	Γ														
	s displaying an "R" in Interim column are interim and sub	iect to	rate tru	e-up as set forth in	General Term	s and Conditio	ns.			l		l		1	l	1

UNBUN	IDLE	D NETWORK ELEMENTS - Tennessee												Attachment:			bit: B
CATEGO	RY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
																DISCIS	DISC AUU I
							Rec	Nonrecurring	Add'l	Nonrecurrin First	g Disconnect	COMEC	SOMAN		Rates(\$) SOMAN	SOMAN	SOMAN
т	'he "7c	one" shown in the sections for stand-alone loops or loops as	part of	a com	bination refers to Ge	ographically	v Deaveraged U	First NF Zones, To			Add'l raged UNF Zor					SOWAN	SUMAN
		ww.interconnection.bellsouth.com/become a clec/html/inter				ographically	y Deaveragea o		new deorgrap	Sincary Deave		ie Desiguiti	0110 09 0 0		net website.		
OPERATI	IONAL	SUPPORT SYSTEMS															
N	IOTE: ((1) Electronic Service Order: CLEC should contact its contract	t nego	iator i	it prefers the state	specific elect	tronic service c	ordering charge	es as ordered b	by the State Co	ommissions. T	he electron	ic service o	rdering charg	e currently co	ntained in thi	is rate
		is the BellSouth regional electronic service ordering charge.															
		(2) Any element that can be ordered electronically will be bill															
		lements that cannot be ordered electronically at present per t				e in this cate	gory reflects th	e charge that v	would be billed	I to a CLEC or	nce electronic o	ordering cap	abilities co	me on-line fo	r that elemen	t. Otherwise,	the manual
01	rderin	g charge, SOMAN, will be applied to a CLECs bill when it sub	mits ar	LSR 1	o BellSouth.			1		1	1	r	1	1	1	1	r
		Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)				SOMEC		3.50									
LINE SEP	VICE	DATE ADVANCEMENT CHARGE				SOMEC		3.50									
		The Expedite charge will be maintained commensurate with I	BellSou	th's F(C No.1 Tariff, Section	on 5 as appli	icable.										
		UNE Expedite Charge per Circuit or Line Assignable USOC, per									1	1	1				1
		Day			ALL UNE	SDASP		200.00									
		XCHANGE ACCESS LOOP															
2-	-WIRE	ANALOG VOICE GRADE LOOP															
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 1			UEANL	UEAL2	13.19	31.99	20.02	10.65				20.35	10.54	13.32	13.3
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 2	l		UEANL	UEAL2	17.23	31.99	20.02	10.65				20.35	10.54	13.32	13.3
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 3		3	UEANL	UEAL2	22.53	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.3
⊢		Loop Testing - Basic 1st Half Hour				URET1	ł	78.92	78.92		+			20.35	10.54	13.32	13.3
\vdash		Loop Testing - Basic Additional Half Hour CLEC to CLEC Conversion Charge Without Outside Dispatch			UEANL	URETA		23.33	23.33					20.35	10.54	13.32	13.3
		(UVL-SL1)			UEANL	UREWO		15.80	8.95					20.35	10.54	13.32	13.3
\vdash		Unbundled Voice Loop, Unbundled Non-Design Voice Loop,				JILLINO		15.00	0.90			1		20.35	10.34	13.32	10.0
		billing for BST providing make-up			UEANL	UEANM		28.80	28.80								
		Manual Order Coordination for UVL-SL1s (per loop)			UEANL	UEAMC		36.52	36.52					İ		İ	
		Order Coordination for Specified Conversion Time for UVL-SL1		l		1							1				İ
		(per LSR)			UEANL	OCOSL		34.29	34.29								
2-	-WIRE	Unbundled COPPER LOOP															
\vdash		2-Wire Unbundled Copper Loop - Non-Designed Zone 1		1	UEQ	UEQ2X	13.19	31.99	20.02	10.65		ļ		20.35	10.54	13.32	13.3
⊢		2 Wire Unbundled Copper Loop - Non-Designed - Zone 2		2	UEQ UEQ	UEQ2X UEQ2X	17.23 22.53	31.99 31.99	20.02	10.65 10.65	1.41 1.41			20.35 20.35	10.54 10.54	13.32 13.32	13.3 13.3
⊢ – –		2 Wire Unbundled Copper Loop - Non-Designed - Zone 3 Order Coordination 2 Wire Unbundled Copper Loop - Non-	-	3			22.53	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.3
		Designed (per loop)			UEQ	USBMC		36.52	36.52								
		Unbundled Copper Loop, Non-Designed Billing for BST				00000	1	00.02	00.02		1			1		1	<u> </u>
i		providing make-up			UEQ	UEQMU		28.80	28.80					20.35	10.54	13.32	13.3
		Loop Testing - Basic 1st Half Hour			UEQ	URET1		78.92	78.92					20.35	10.54	13.32	13.3
		Loop Testing - Basic Additional Half Hour			UEQ	URETA		23.33	23.33					20.35	10.54	13.32	13.3
		CLEC to CLEC Conversion Charge Without Outside Dispatch															
					UEQ	UREWO		14.29	7.44		-			20.35	10.54	13.32	13.3
		ANALOG VOICE GRADE LOOP															
2-		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-					1				1						<u> </u>
		Zone 1		1	UEPSR UEPSB	UEALS	13.19	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.3
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-						000	20.02	. 5.66				20.00		10.02	.0.0
		Zone 1		1	UEPSR UEPSB	UEABS	13.19	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.3
		2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting-															
		Zone 2		2	UEPSR UEPSB	UEALS	17.23	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.3
		2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting-		_		115450										10.0-	
\vdash		Zone 2 2 Wire Apples Vision Crade Leon Service Level 1 Line Splitting		2	UEPSR UEPSB	UEABS	17.23	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.3
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zone 3		3	UEPSR UEPSB	UEALS	22.53	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.3
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-		3	OLF ON DEFOD	JLALO	22.03	31.99	20.02	10.00	1.41	+		20.35	10.54	13.32	13.3
		Zone 3		3	UEPSR UEPSB	UEABS	22.53	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.3
U	JNE Lo	oop Rates for Line Splitting		Ť			00	000	20.02	. 5.66				20.00		10.02	.0.0
		2-Wire Voice Grade Loop (SL1) for Line Splitting - Zone 1	1	1	UEPRX	UEPLX	14.18			l		1	1	l		l	1
		2-Wire Voice Grade Loop (SL1) for Line Splitting - Zone 2		2	UEPRX	UEPLX	18.01										
		2-Wire Voice Grade Loop (SL1)for Line Splitting - Zone 3		3	UEPRX	UEPLX	23.02										
		XCHANGE ACCESS LOOP															

UNBUNDL	ED NETWORK ELEMENTS - Tennessee												Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
						Rec	Nonrecurring		Nonrecurring	J Disconnect				Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
2-WIF	RE ANALOG VOICE GRADE LOOP															
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling - Zone 1		1	UEA	UEAL2	16.56	75.06	48.20	28.70	17.64			20.35	10.54	13.32	13.32
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or		'	ULA	ULALZ	10.50	75.00	40.20	20.70	17.04			20.33	10.54	13.32	13.32
	Ground Start Signaling - Zone 2		2	UEA	UEAL2	21.63	75.06	48.20	28.70	17.64			20.35	10.54	13.32	13.32
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or															
	Ground Start Signaling - Zone 3		3	UEA	UEAL2	28.28	75.06	48.20	28.70	17.64			20.35	10.54	13.32	13.32
	Order Coordination for Specified Conversion Time (per LSR)			UEA	OCOSL		34.29									
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse															
	Battery Signaling - Zone 1		1	UEA	UEAR2	16.56	75.06	48.20	28.70	17.64			20.35	10.54	13.32	13.32
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zone 2		2	UEA	UEAR2	21.63	75.06	48.20	28.70	17.64			20.35	10.54	13.32	13.32
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse		2	OLA	ULANZ	21.03	75.00	40.20	20.70	17.04			20.33	10.34	13.32	13.32
	Battery Signaling - Zone 3		3	UEA	UEAR2	28.28	75.06	48.20	28.70	17.64			20.35	10.54	13.32	13.32
	Order Coordination for Specified Conversion Time (per LSR)			UEA	OCOSL		34.29									
	CLEC to CLEC Conversion Charge without outside dispatch			UEA	UREWO		75.06	36.41					20.35	10.54	13.32	13.32
4-WIF	RE ANALOG VOICE GRADE LOOP															
	4-Wire Analog Voice Grade Loop - Zone 1		1	UEA	UEAL4	24.70	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13.32
	4-Wire Analog Voice Grade Loop - Zone 2			UEA UEA	UEAL4 UEAL4	32.25 42.17	122.76 122.76	85.57 85.57	76.35 76.35	39.16 39.16			20.35 20.35	10.54 10.54	13.32 13.32	13.32 13.32
	4-Wire Analog Voice Grade Loop - Zone 3 Order Coordination for Specified Conversion Time (per LSR)		3	UEA	OCOSL	42.17	34.29	85.57	76.35	39.16			20.35	10.54	13.32	13.32
	CLEC to CLEC Conversion Charge without outside dispatch			UEA	UREWO		75.06	36.41					20.35	10.54	13.32	13.32
2-WIR	RE ISDN DIGITAL GRADE LOOP			0E/(ONLEWO		10.00	00.41					20.00	10.04	10.02	10.02
	2-Wire ISDN Digital Grade Loop - Zone 1		1	UDN	U1L2X	22.22	142.76	88.88	76.35	39.16			20.35	10.54	13.32	13.32
	2-Wire ISDN Digital Grade Loop - Zone 2		2	UDN	U1L2X	29.02	142.76	88.88	76.35	39.16			20.35	10.54	13.32	13.32
	2-Wire ISDN Digital Grade Loop - Zone 3		3	UDN	U1L2X	37.95	142.76	88.88	76.35	39.16			20.35	10.54	13.32	13.32
	Order Coordination For Specified Conversion Time (per LSR)			UDN	OCOSL		34.29									L
0.14/15	CLEC to CLEC Conversion Charge without outside dispatch			UDN	UREWO		91.77	44.22					20.35	10.54	13.32	13.32
2-9915	RE Universal Digital Channel (UDC) COMPATIBLE LOOP 2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone															<u> </u>
			1	UDC	UDC2X	22.22	142.76	88.88	76.35	39.16			20.35	10.54	13.32	13.32
	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone			000	ODOLA		142.10	00.00	70.00	00.10			20.00	10.04	10.02	10.02
	2		2	UDC	UDC2X	29.02	142.76	88.88	76.35	39.16			20.35	10.54	13.32	13.32
	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone															
	3		3	UDC	UDC2X	37.95	142.76	88.88	76.35	39.16			20.35	10.54	13.32	13.32
	CLEC to CLEC Conversion Charge without outside dispatch			UDC	UREWO		91.77	44.22					20.35	10.54	13.32	13.32
2-WIR	RE ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMP	ATIBLE	LOOP													ł
	2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation - Zone 1		1	UAL	UAL2X	13.82	270.01	234.63	74.54	39.14			20.35	10.54	13.32	13.32
	2 Wire Unbundled ADSL Loop including manual service inquiry				UALZA	15.02	270.01	204.00	74.54	33.14			20.00	10.04	10.02	10.02
	& facility reservation - Zone 2		2	UAL	UAL2X	18.05	270.01	234.63	74.54	39.14			20.35	10.54	13.32	13.32
	2 Wire Unbundled ADSL Loop including manual service inquiry															
	& facility reservation - Zone 3		3	UAL	UAL2X	23.60	270.01	234.63	74.54	39.14			20.35	10.54	13.32	13.32
	Order Coordination for Specified Conversion Time (per LSR)			UAL	OCOSL		34.29									l
	2 Wire Unbundled ADSL Loop without manual service inquiry &	Ι.				40.00	24.00	00.00	10.05				20.35	10 51	40.00	40.00
	facility reservaton - Zone 1 2 Wire Unbundled ADSL Loop without manual service inquiry &			UAL	UAL2W	13.82	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
1	facility reservaton - Zone 2	1	2	UAL	UAL2W	18.05	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	2 Wire Unbundled ADSL Loop without manual service inquiry &		-				000	20.02			1		20.00			.0.02
1	facility reservaton - Zone 3	1	3	UAL	UAL2W	23.60	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	Order Coordination for Specified Conversion Time (per LSR)			UAL	OCOSL		34.29									
	CLEC to CLEC Conversion Charge without outside dispatch	I		UAL	UREWO		31.99	20.02					20.35	10.54	13.32	13.32
2-WIR	RE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPA	TIBLE I	LOOP								ļ					
	2 Wire Unbundled HDSL Loop including manual service inquiry		1	UHL		10.00	270.04	004.00	74.54	20.44			20.25	10.54	13.32	12.00
	& facility reservation - Zone 1 2 Wire Unbundled HDSL Loop including manual service inquiry		1		UHL2X	10.83	270.01	234.63	/4.54	39.14	<u> </u>		20.35	10.54	13.32	13.32
	& facility reservation - Zone 2		2	UHL	UHL2X	14.15	270.01	234.63	74.54	39.14			20.35	10.54	13.32	13.32
	2 Wire Unbundled HDSL Loop including manual service inquiry		-				2.0.01	20.00		00.14	1		20.00			
	& facility reservation - Zone 3		3	UHL	UHL2X	18.50	270.01	234.63	74.54	39.14			20.35	10.54	13.32	13.32

UNBUNDLE	D NETWORK ELEMENTS - Tennessee				_								Attachment:			ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
						Rec	Nonrecurring			g Disconnect				Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		34.29									<u> </u>
	2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 1		1	UHL	UHL2W	10.83	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	2 Wire Unbundled HDSL Loop without manual service inquiry			OTIL	UTILZW	10.00	51.55	20.02	10.05	1.41			20.00	10.04	10.02	10.02
	and facility reservation - Zone 2	1	2	UHL	UHL2W	14.15	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	2 Wire Unbundled HDSL Loop without manual service inquiry															
	and facility reservation - Zone 3	I	3	UHL	UHL2W	18.50	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		34.29	00.00					00.05	40.54	40.00	10.00
4 WID	CLEC to CLEC Conversion Charge without outside dispatch E HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPA			UHL	UREWO		31.99	20.02					20.35	10.54	13.32	13.32
4-9916	4 Wire Unbundled HDSL Loop including manual service inquiry				-											
	and facility reservation - Zone 1		1	UHL	UHL4X	13.93	279.60	244.22	74.54	39.14			20.35	10.54	13.32	13.32
	4-Wire Unbundled HDSL Loop including manual service inquiry		-													
	and facility reservation - Zone 2		2	UHL	UHL4X	18.20	279.60	244.22	74.54	39.14			20.35	10.54	13.32	13.32
	4-Wire Unbundled HDSL Loop including manual service inquiry															
	and facility reservation - Zone 3		3	UHL	UHL4X	23.80	279.60	244.22	74.54	39.14			20.35	10.54	13.32	13.32
	Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		34.29									
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 1		1	UHL	UHL4W	13.93	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	4-Wire Unbundled HDSL Loop without manual service inquiry	1	1	UNL	UHL4VV	13.93	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	and facility reservation - Zone 2		2	UHL	UHL4W	18.20	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	4-Wire Unbundled HDSL Loop without manual service inquiry		_	0112	0112111	10.20	01100	20102	10.00				20.00	10.01	10102	10.02
	and facility reservation - Zone 3	1	3	UHL	UHL4W	23.80	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		34.29									
	CLEC to CLEC Conversion Charge without outside dispatch	I		UHL	UREWO		31.99	20.02					20.35	10.54	13.32	13.32
4-WIR	E DS1 DIGITAL LOOP				1101.207	57.70	010.00	010 70	00.00	40.45			10.00	0.40	44.05	11.05
	4-Wire DS1 Digital Loop - Zone 1 4-Wire DS1 Digital Loop - Zone 2		1	USL USL	USLXX USLXX	57.73 75.40	313.08 313.08	219.72 219.72	96.86 96.86	40.45 40.45			18.98 18.98	8.43 8.43	11.95 11.95	11.95 11.95
	4-Wire DS1 Digital Loop - Zone 3		3	USL	USLXX	98.59	313.08	219.72	96.86	40.45			18.98	8.43	11.95	11.95
	Order Coordination for Specified Conversion Time (per LSR)		Ŭ	USL	OCOSL	00.00	34.59	210112	00.00	10.10	1		10.00	0.10		
	CLEC to CLEC Conversion Charge without outside dispatch			USL	UREWO		130.47	40.11					20.35	10.54	13.32	13.32
4-WIR	E 19.2, 56 OR 64 KBPS DIGITAL GRADE LOOP															
	4 Wire Unbundled Digital 19.2 Kbps		1	UDL	UDL19	31.10	207.01	141.38	90.70	44.18			20.35	10.54	13.32	13.32
	4 Wire Unbundled Digital 19.2 Kbps		2	UDL	UDL19	40.61	207.01	141.38	90.70	44.18			20.35	10.54	13.32	13.32
	4 Wire Unbundled Digital 19.2 Kbps			UDL	UDL19	53.11 31.10	207.01	141.38	90.70 90.70	44.18 44.18			20.35	10.54	13.32 13.32	13.32
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 1 4 Wire Unbundled Digital Loop 56 Kbps - Zone 2			UDL UDL	UDL56 UDL56	40.61	207.01 207.01	141.38 141.38	90.70	44.18			20.35 20.35	10.54 10.54	13.32	13.32 13.32
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 3			UDL	UDL56	53.11	207.01	141.38	90.70	44.18			20.35	10.54	13.32	13.32
	Order Coordination for Specified Conversion Time (per LSR)		-	UDL	OCOSL		34.29									
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 1		1	UDL	UDL64	31.10	207.01	141.38	90.70	44.18			20.35	10.54	13.32	13.32
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 2			UDL	UDL64	40.61	207.01	141.38	90.70	44.18			20.35	10.54	13.32	13.32
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 3		3	UDL	UDL64	53.11	207.01	141.38	90.70	44.18			20.35	10.54	13.32	13.32
	Order Coordination for Specified Conversion Time (per LSR)			UDL	OCOSL		34.29	40.00					20.25	10.54	40.00	10.00
2 WID	CLEC to CLEC Conversion Charge without outside dispatch E Unbundled COPPER LOOP			UDL	UREWO		102.28	49.82					20.35	10.54	13.32	13.32
2-9916	2-Wire Unbundled Copper Loop/Short including manual service				-											
	inquiry & facility reservation - Zone 1	1	1	UCL	UCLPB	13.19	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	2-Wire Unbundled Copper Loop/Short including manual service															
	inquiry & facility reservation - Zone 2	1	2	UCL	UCLPB	17.23	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	2 Wire Unbundled Copper Loop/Short including manual service															
	inquiry & facility reservation - Zone 3		3	UCL	UCLPB	22.53	31.99	20.02	10.65	1.41	ļ		20.35	10.54	13.32	13.32
	Order Coordination for Unbundled Copper Loops (per loop)		<u> </u>	UCL	UCLMC		36.52	36.52	ł		L					ł
	2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 1		1	UCL	UCLPW	13.19	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	2-Wire Unbundled Copper Loop/Short without manual service			UUL	UULPVV	13.19	31.99	20.02	C0.01	1.41	<u> </u>		20.35	10.54	13.32	13.32
	inquiry and facility reservation - Zone 2	1	2	UCL	UCLPW	17.23	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
l	2-Wire Unbundled Copper Loop/Short without manual service						200									
	inquiry and facility reservation - Zone 3	1	3	UCL	UCLPW	22.53	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		36.52	36.52								

UNBUNDLE	D NETWORK ELEMENTS - Tennessee												Attachment:	2	Exhi	ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		-
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Unbundled Copper Loop/Long - includes manual srvc.					10.10			10.05					10.51	10.00	10.00
	inquiry and facility reservation - Zone 1	- 1	1	UCL	UCL2L	13.19	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 2		2	UCL	UCL2L	17.23	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	2-Wire Unbundled Copper Loop/Long - includes manual svc.		2	UCL	UCL2L	17.23	31.99	20.02	10.05	1.41			20.33	10.34	13.32	13.32
	inquiry and facility reservation - Zone 3	1	3	UCL	UCL2L	22.53	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	Order Coordination for Unbundled Copper Loops (per loop)		-	UCL	UCLMC		36.52	36.52								
	2-Wire Unbundled Copper Loop/Long - without manual service															
	inquiry and facility reservation - Zone 1	1	1	UCL	UCL2W	13.19	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	2-Wire Unbundled Copper Loop/Long - without manual service															
	inquiry and facility reservation - Zone 2	I	2	UCL	UCL2W	17.23	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	2-Wire Unbundled Copper Loop/Long - without manual service								10.05					10.51	10.00	10.00
	inquiry and facility reservation - Zone 3	- 1	3	UCL	UCL2W	22.53	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13.32
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		36.52	36.52								
	CLEC to CLEC Conversion Charge without outside dispatch (UCL-Des)			UCL	UREWO		31.99	20.02					20.35	10.54	13.32	13.32
4-WIRI	E COPPER LOOP			UCL	UKLWO		51.99	20.02					20.33	10.54	13.32	15.52
4 1111	4-Wire Copper Loop/Short - including manual service inquiry															
	and facility reservation - Zone 1	1	1	UCL	UCL4S	24.70	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13.32
	4-Wire Copper Loop/Short - including manual service inquiry															
	and facility reservation - Zone 2	1	2	UCL	UCL4S	32.25	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13.32
	4-Wire Copper Loop/Short - including manual service inquiry															
	and facility reservation - Zone 3	I	3	UCL	UCL4S	42.17	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13.32
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		36.52	36.52								
	4-Wire Copper Loop/Short - without manual service inquiry and						100 70							10 51	10.00	10.00
	facility reservation - Zone 1	1	1	UCL	UCL4W	24.70	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13.32
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation - Zone 2		2	UCL	UCL4W	32.25	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13.32
-	4-Wire Copper Loop/Short - without manual service inquiry and	1	2	UCL	UCL4VV	32.25	122.70	65.57	76.33	39.10			20.35	10.54	13.32	13.32
	facility reservation - Zone 3		3	UCL	UCL4W	42.17	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13.32
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		36.52	36.52	10.00	00.10			20.00	10101	10102	10.02
	4-Wire Unbundled Copper Loop/Long - includes manual svc.															
	inquiry and facility reservation - Zone 1	1	1	UCL	UCL4L	24.70	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13.32
	4-Wire Unbundled Copper Loop/Long - includes manual svc.															
	inquiry and facility reservation - Zone 2	I	2	UCL	UCL4L	32.25	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13.32
	4-Wire Unbundled Copper Loop/Long - includes manual svc.		_													
	inquiry and facility reservation - Zone 3	1	3	UCL	UCL4L	42.17	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13.32
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		36.52	36.52								
	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone 1		1	UCL	UCL4O	24.70	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13.32
	4-Wire Unbundled Copper Loop/Long - without manual svc.	<u> </u>			001-0	27.70	122.10	00.07	10.00	55.10	1		20.00	10.04	10.02	13.32
	inquiry and facility reservation - Zone 2	1	2	UCL	UCL40	32.25	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13.32
	4-Wire Unbundled Copper Loop/Long - without manual svc.	1									1	1				
	inquiry and facility reservation - Zone 3	I	3	UCL	UCL40	42.17	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13.32
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		36.52	36.52								
	CLEC to CLEC Conversion Charge without outside dispatch															
	(UCL-Des)			UCL	UREWO		31.99	20.02			ļ		20.35	10.54	13.32	13.32
LOOP MODIFI						-	├						ł			┟────┘
				UAL, UHL, UCL, UEQ, ULS, UEA,												'
	Unbundled Loop Modification, Removal of Load Coils - 2 Wire	1	1	UEANL, UDL, UDC,									1			
	pair less than or equal to 18k ft	1	1	UDN, UDL, USL	ULM2L		65.40	65.40					20.35	10.54	13.32	13.32
	Unbundled Loop Modification, Removal of Load Coils - 2 wire	i .					00.40	00.40		1			20.00			
	greater than 18k ft	1	1	UCL, ULS, UEQ	ULM2G		710.71	23.77					20.35	10.54	13.32	13.32
	Unbundled Loop Modification Removal of Load Coils - 4 Wire	1	1								1	İ	1			
	less than or equal to 18K ft	1		UHL, UCL	ULM4L		65.40	65.40					20.35	10.54	13.32	13.32
	Unbundled Loop Modification Removal of Load Coils - 4 Wire												I			
1 1	pair greater than 18k ft			UCL	ULM4G		710.71	23.77					20.35	10.54	13.32	13.32

UNBUNDL	LED NETWORK ELEMENTS - Tennessee		1	1									Attachment:			ibit: B
CATEGORY	C RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Submitted Manually	Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
	Unbundled Loop Modification Removal of Bridged Tap Removal, per unbundled loop			UAL, UHL, UCL, UEQ, UEF, ULS, UEA, UEANL, UDL, UDC, UDN, UDL, USL	ULMBT		First 65.44	Add'l 65.44	First	Add'l	SOMEC	SOMAN	SOMAN 20.35	SOMAN 10.54	SOMAN 13.32	SOMAN 13.32
SUB-LOOPS																
Sub	o-Loop Distribution															
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set- Up	I		UEANL	USBSA		517.25	517.25					20.35	10.54	13.32	13.32
	Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-Up			UEANL	USBSB		42.68	42.68					20.35	10.54	13.32	13.32
	Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-Op Sub-Loop - Per Building Equipment Room - CLEC Feeder	1		UEANL	USBSB		42.08	42.08					20.35	10.54	13.32	13.32
	Facility Set-Up Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel	I		UEANL	USBSC		313.01	313.01					20.35	10.54	13.32	13.32
	Set-Up	I		UEANL	USBSD		108.06	108.06					20.35	10.54	13.32	13.32
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Statewide		sw	UEANL	USBN2	10.02	148.84	112.34	73.14	36.65			20.35	10.54	13.32	13.32
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair	r		UEANL	USBMC		34.29	34.29								
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 1 Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop -		1	UEANL	USBN4	7.30	147.93	75.11	99.96	16.98			20.35	10.54	13.32	13.32
	Zone 2 Sub-Loop Distribution Fer 4-Wire Analog Voice Grade Loop -		2	UEANL	USBN4	9.54	147.93	75.11	99.96	16.98			20.35	10.54	13.32	13.32
	Zone 3		3	UEANL	USBN4	12.47	147.93	75.11	99.96	16.98			20.35	10.54	13.32	13.32
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair	r		UEANL	USBMC		34.29	34.29								
	Sub-Loop 2-Wire Intrabuilding Network Cable (INC)	1		UEANL	USBR2	1.35	94.56	29.35					20.35	10.54	13.32	13.32
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 4-Wire Intrabuilding Network Cable (INC)	r	-	UEANL UEANL	USBMC USBR4	2.26	34.29 116.14	34.29 37.10					20.35	10.54	13.32	13.32
	Sub-Loop 4-Wire intrabuilding Network Cable (INC)	1		UEANL	USBR4	2.20	116.14	37.10					20.35	10.54	13.32	13.34
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair	r		UEANL	USBMC		34.29	34.29								
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 1	1	1	UEF	UCS2X	5.16	110.71	37.89	94.41	13.09			20.35	10.54	13.32	13.32
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2	1	2	UEF	UCS2X	6.74	110.71	37.89	94.41	13.09			20.35	10.54	13.32	13.32
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 3	1	3	UEF	UCS2X	8.81	110.71	37.89	94.41	13.09			20.35	10.54	13.32	13.32
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair	r		UEF	USBMC	0.50	34.29	34.29	00.00	10.00			00.05	10.51	10.00	40.00
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 1 4 Wire Copper Unbundled Sub-Loop Distribution - Zone 2		1	UEF UEF	UCS4X UCS4X	6.52 8.52	117.12 117.12	44.30 44.30	99.96 99.96	16.98 16.98			20.35 20.35	10.54 10.54	13.32 13.32	13.32
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 4 Wire Copper Unbundled Sub-Loop Distribution - Zone 3			UEF	UCS4X UCS4X	11.14	117.12	44.30	99.96	16.98			20.35	10.54	13.32	
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair		Ŭ	UEF	USBMC	11.14	34.29	34.29	00.00	10.00			20.00	10.04	10.02	10.02
Unb	oundled Sub-Loop Modification				1											
	Unbundled Sub-Loop Modification - 2-W Copper Dist Load Coil/Equip Removal per 2-W PR			UEF	ULM2X		335.36	7.82					20.34	10.54	13.32	13.32
	Unbundled Sub-loop Modification - 4-W Copper Dist Load Coil/Equip Removal per 4-W PR			UEF	ULM4X		335.36	7.82					20.35	10.54	13.32	13.32
	Unbundled Sub-loop Modification - 2-w/4-w Copper Dist Bridged Tap Removal, per PR unloaded	1		UEF	ULM4T		528.48	9.74					20.35	10.54	13.32	13.32
Unb	Dundled Network Terminating Wire (UNTW) Unbundled Network Terminating Wire (UNTW) per Pair	<u> </u>		UENTW	UENPP	0.4555	2.48	2.48		1			20.35	10.54	13.32	13.32
Netv	work Interface Device (NID)					0.4005	∠.48	2.48					20.35	10.54	13.32	13.32
	Network Interface Device (NID) - 1-2 lines	1	1	UENTW	UND12		89.69	54.56	0.6391	0.6391	1		20.35	10.54	13.32	13.32
	Network Interface Device (NID) - 1-6 lines	1	1	UENTW	UND16		129.65	94.51	0.6522	0.6522	1	1	20.35	10.54	13.32	13.32
	Network Interface Device Cross Connect - 2 W			UENTW	UNDC2		11.11	11.11					20.35	10.54	13.32	13.32
	Network Interface Device Cross Connect - 4W			UENTW	UNDC4		11.11	11.11					20.35	10.54	13.32	13.32
SUB-LOOPS																
Sub	o-Loop Feeder				1							1				

UNBUNDLE	D NETWORK ELEMENTS - Tennessee												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrecurring		Nonrecurring					Rates(\$)	-	
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	USL-Feeder, DS0 Set-up per Cross Box location - CLEC			UEA,										10 51	10.00	10.00
	Distribution Facility set-up			UDN,UCL,UDL,UDC UEA,	USBEW		517.25				-		20.35	10.54	13.32	13.32
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up			UDN,UCL,UDL,UDC	USBFX		42.68	42.68					20.35	10.54	13.32	13.32
	USL Feeder DS1 Set-up at DSX location, per DS1 termination			USL	USBFZ		531.04	11.34					20.35	10.54	13.32	13.32
	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice															
	Grade- Statewide		SW	UEA	USBFA	12.05	122.24	85.05	76.35	39.16			20.35	10.54	13.32	13.32
	Order Coordination for Specified Conversion Time, per LSR			UEA	OCOSL		34.29									
	Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice					10.05								10 51	10.00	10.00
	Grade - Statewide Order Coordination for Specified Time Conversion, per LSR		SW	UEA UEA	USBFB OCOSL	12.05	122.24 34.29	85.05	76.35	39.16			20.35	10.54	13.32	13.32
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery,			ULA	UCUSL	1	34.29									
	Voice Grade Loop - Statewide		sw	UEA	USBFC	12.05	122.24	85.05	76.35	39.16			20.35	10.54	13.32	13.32
	Order Coordination For Specified Conversion Time, per LSR			UEA	OCOSL		34.29									
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice															
	Grade - Zone 1		1	UEA	USBFD	21.52	137.31	61.93	118.04	30.13			20.35	10.54	13.32	13.32
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice		~			00.47	407.01	04.00		00.40			00.07	10 51	10.00	10.00
	Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice	-	2	UEA	USBFD	28.11	137.31	61.93	118.04	30.13			20.35	10.54	13.32	13.32
	Grade - Zone 3		3	UEA	USBFD	36.76	137.31	61.93	118.04	30.13			20.35	10.54	13.32	13.32
	Order Coordination For Specified Conversion Time, Per LSR		3	UEA	OCOSL	30.70	34.29	01.95	110.04	30.13			20.33	10.34	13.32	13.32
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice			02.1	00002		01120									
	Grade - Zone 1		1	UEA	USBFE	21.52	137.31	61.93	118.04	30.13			20.35	10.54	13.32	13.32
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice															
	Grade - Zone 2	-	2	UEA	USBFE	28.11	137.31	61.93	118.04	30.13			20.35	10.54	13.32	13.32
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice		~			00.70	407.04	C1 02	110.01	20.42			20.25	10.54	40.00	40.00
	Grade - Zone 3 Order Coordination For Specified Conversion Time, Per LSR		3	UEA UEA	USBFE OCOSL	36.76	137.31 34.29	61.93	118.04	30.13			20.35	10.54	13.32	13.32
	Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 1		1	UDN	USBFF	16.11	142.83	67.45	104.67	18.53			19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 2		2	UDN	USBFF	21.04	142.83	67.45	104.67	18.53			19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 3		3	UDN	USBFF	27.51	142.83	67.45	104.64	18.53			19.99	19.99	19.99	19.99
	Order Coordination For Specified Conversion Time, Per LSR			UDN	OCOSL		34.29									
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)		1	UDC	USBFS	16.11	142.83	67.45	104.67	18.53			19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)		2	UDC	USBFS	21.04	142.83	67.45	104.67	18.53			19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible) Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 1		3	UDC USL	USBFS USBFG	27.51 39.74	142.83 116.00	67.45 40.62	104.64 106.82	18.53 18.91			19.99 19.99	19.99 19.99	19.99 19.99	19.99 19.99
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 1			USL	USBFG	51.90	116.00	40.62	106.82	18.91			19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 3		3	USL	USBFG	67.86	116.00	40.62	106.82	18.91			19.99	19.99	19.99	19.99
	Order Coordination For Specified Conversion Time, Per LSR			USL	OCOSL		34.59									
	Unbundled Sub-Loop Feeder, 2-Wire Copper Loop - Zone 1		1	UCL	USBFH	9.52	114.27	38.89	104.64	18.53			19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone		_													
			2	UCL	USBFH	12.43	114.27	38.89	104.64	18.53			19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone		3	UCL	USBFH	16.26	114.27	38.89	104.64	18.53			19.99	19.99	19.99	19.99
├ ── ├ ──	Order Coordination For Specified Conversion Time, per LSR		5	UCL	OCOSL	10.20	34.29	30.09	104.04	10.33			19.99	13.39	13.39	13.39
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 1		1	UCL	USBFJ	14.37	123.41	48.03	110.44	22.53	1		19.99	19.99	19.99	19.99
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 2		2	UCL	USBFJ	18.76	123.41	48.03	110.44	22.53			19.99	19.99	19.99	19.99
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 3		3	UCL	USBFJ	24.53	123.41	48.03	110.44	22.53			19.99	19.99	19.99	19.99
	Order Coordination For Specified Conversion Time, per LSR		<u> </u>	UCL	OCOSL		34.29									
├── ┼──	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		1	UDL	USBEN	26.06	116.00	40.62	106.82	18.91			19.99 19.99	19.99	19.99	19.99
├── ┼──	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		2	UDL UDL	USBFN USBFN	34.03 44.50	116.00 116.00	40.62 40.62	106.82 106.82	18.91 18.91			19.99 19.99	19.99 19.99	19.99 19.99	19.99 19.99
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop -		5	ODL		44.50	110.00	40.02	100.62	10.91			15.99	13.99	13.99	13.99
	Zone 1		1	UDL	USBFO	26.06	116.00	40.62	106.82	18.91			19.99	19.99	19.99	19.99
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -															
	Zone 2		2	UDL	USBFO	34.03	116.00	40.62	106.82	18.91			19.99	19.99	19.99	19.99
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -				10050				100.0-				10.0-			
	Zone 3 Order Coordination For Specified Time Conversion, per LSP		3	UDL UDL	USBFO OCOSL	44.50	116.00 34.29	40.62	106.82	18.91			19.99	19.99	19.99	19.99
	Order Coordination For Specified Time Conversion, per LSR			UUL	ULUSL	1	34.29				1	l				I

UNBON	DLE	ONETWORK ELEMENTS - Tennessee		1		,								Attachment:			bit: B
CATEGO	RY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
							Rec	Nonrecurring		Nonrecurring					Rates(\$)	_	
						↓ →		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 1		1	UDL	USBFP	26.06	116.00	40.62	106.82	18.91			19.99	19.99	19.99	19.99
		Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -		<u> </u>	ODL	CODIT	20.00	110.00	40.02	100.02	10.01			10.00	10.00	10.00	10.00
		Zone 2		2	UDL	USBFP	34.03	116.00	40.62	106.82	18.91			19.99	19.99	19.99	19.99
		Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -															
		Zone 3		3	UDL	USBFP	44.50	116.00	40.62	106.82	18.91			19.99	19.99	19.99	19.99
SUB-LOC		Order Coordination For Specified Conversion Time, per LSR			UDL	OCOSL		34.29									
		op Feeder															
-		Sub Loop Feeder - DS3 - Per Mile Per Month	1		UE3	1L5SL	14.11										
		Sub Loop Feeder - DS3 - Facility Termination Per Month	i	1	UE3	USBF1	333.26	3,406.61	407.68	165.17	501.31			20.35	10.54	13.32	t
		Sub Loop Feeder – STS-1 – Per Mile Per Month	Ì		UDLSX	1L5SL	14.11										1
		Sub Loop Feeder - STS-1 - Facility Termination Per Month	I	1	UDLSX	USBF7	359.02	3,406.61	407.68	165.17	501.31	1		20.35	10.54	13.32	İ
		Sub Loop Feeder – OC-3 – Per Mile Per Month	_		UDLO3	1L5SL	10.71										
		Sub Loop Feeder - OC-3 - Facility Termination Protection Per															
		Month	I		UDLO3	USBF5	56.64										
		Sub Loop Feeder - OC-3 - Facility Termination Per Month	_		UDLO3	USBF2	546.31	3,406.61	407.68	165.17	501.31			20.35	10.54	13.32	
		Sub Loop Feeder - OC-12 - Per Mile Per Month	- 1		UDL12	1L5SL	13.18										
		Sub Loop Feeder - OC-12 - Facility Termination Protection Per															
		Month			UDL12	USBF6	639.98	0 400 04	407.00	405.47	504.04			00.05	10.54	40.00	-
		Sub Loop Feeder - OC-12 - Facility Termination Per Month			UDL12	USBF3	1,697.00 43.22	3,406.61	407.68	165.17	501.31			20.35	10.54	13.32	
		Sub Loop Feeder - OC-48 - Per Mile Per Month Sub Loop Feeder - OC-48 - Facility Termination Protection Per	I		UDL48	1L5SL	43.22										
		Month			UDL48	USBF9	320.36										
		Sub Loop Feeder - OC-48 - Facility Termination Per Month	· ·	1	UDL48	USBF4	1.457.00	3.592.61	407.68	165.17	501.31			20.35	10.54	13.32	
		Sub Loop Feeder - OC-12 Interface On OC-48	i		UDL48	USBF8	361.44	806.02	407.68	165.17	501.31			20.35	10.54	13.32	
UNBUND		OOP CONCENTRATION															
		Loop Channelization System			ULC	ULCCS	307.07	307.34	74.37	4.18				20.35	10.54	13.32	13.32
		CO Channel Interface - 2-Wire Voice Grade			ULC	ULCC2	1.20	9.57	9.52	8.66	8.60			20.35	10.54	13.32	13.32
		Unbundled Loop Concentration - System A (TR008)			ULC	UCT8A	500.18	613.60	613.60					20.35	10.54	13.32	13.32
		Unbundled Loop Concentration - System B (TR008)			ULC	UCT8B	54.82	255.67	255.67					20.35	10.54	13.32	13.32
		Unbundled Loop Concentration - System A (TR303)			ULC	UCT3A	539.00	613.60	613.60					20.35	10.54	13.32	13.32
		Unbundled Loop Concentration - System B (TR303)			ULC	UCT3B	92.37	255.67	255.67					20.35	10.54	13.32	13.32
		Unbundled Loop Concentration - DS1 Loop Interface Card			ULC	UCTCO	6.23	74.39	53.07	30.23	8.46			20.35	10.54	13.32	13.32
		Unbundled Loop Concentration - ISDN Loop Interface (Brite Card)			UDN	ULCC1	8.46	8.69	0.65	9.71	9.65			20.35	10.54	13.32	13.32
		Card) Unbundled Loop Concentration - UDC Loop Interface (Brite		-	ИЦО	ULCUT	ö.4b	8.09	8.65	9.71	9.05			20.35	10.54	13.32	13.32
		Card)		1	UDC	ULCCU	8.46	8.69	8.65	9.71	9.65			20.35	10.54	13.32	13.32
		Unbundled Loop Concentration2 Wire Voice-Loop Start or		1			0.40	0.09	0.00	0.11	0.00			20.00	10.04	10.02	10.02
		Ground Start Loop Interface (POTS Card)		1	UEA	ULCC2	2.32	8.69	8.65	9.71	9.65			20.35	10.54	13.32	13.32
		Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery										1	İ				
		Loop Interface (SPOTS Card)			UEA	ULCCR	12.45	8.69	8.65	9.71	9.65			20.35	10.54	13.32	13.32
		Unbundled Loop Concentration - 4 Wire Voice Loop Interface		ľ													
		(Specials Card)		<u> </u>	UEA	ULCC4	7.53	8.69	8.65	9.71	9.65			20.35	10.54	13.32	13.332
		Unbundled Loop Concentration - TEST CIRCUIT Card		Ļ	ULC	UCTTC	35.77	8.69	8.65	9.71	9.65			20.35	10.54	13.32	13.32
		Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop		1		111.007	11.00	0.00	0.07	0.71	0.07			00.07	10 5 1	10.00	10.00
		Interface Unbundled Loop Concentration - Digital 56 Kbps Data Loop			UDL	ULCC7	11.03	8.69	8.65	9.71	9.65			20.35	10.54	13.32	13.32
		Unbundled Loop Concentration - Digital 56 Kbps Data Loop Interface		1	UDL	ULCC5	11.03	8.69	8.65	9.71	9.65			20.35	10.54	13.32	13.32
		Unbundled Loop Concentration - Digital 64 Kbps Data Loop			55L	31000	11.03	0.09	0.05	5.11	9.00			20.35	10.34	13.32	13.32
		Interface		1	UDL	ULCC6	11.03	8.69	8.65	9.71	9.65			20.35	10.54	13.32	13.32
				1				0.00	0.00	9.71	0.00			20.00		10.02	
UNE OTH	IER, P	ROVISIONING ONLY - NO RATE		1	İ	1		i i				1		İ		ĺ	İ
		NID - Dispatch and Service Order for NID installation			UENTW	UNDBX	0.00	0.00									
		UNTW Circuit Id Establishment, Provisioning Only - No Rate			UENTW	UENCE	0.00	0.00									
	٦			1	UEANL,UEF,UEQ,U	UNECN											
1		Unbundled Contract Name, Provisioning Only - No Rate			ENTW		0.00	0.00									

	D NETWORK ELEMENTS - Tennessee		-	1	1	1					-	I -	Attachment:			bit: B
TEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incrementa Charge - Manual Sve Order vs. Electronic Disc Add'l
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
				UAL,UCL,UDC,UDL,												
	Unbundled Contact Name, Provisioning Only - no rate			UDN,UEA,UHL,ULC	UNECN	0.00	0.00									
	Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no															
	rate			UEA,UDN,UCL,UDC	USBFQ	0.00	0.00									
	Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no rate			UEA,USL,UCL,UDL	USBFR	0.00	0.00									
	Unbundled DS1 Loop - Superframe Format Option - no rate			USL	CCOSF	0.00	0.00									
	Unbundled DS1 Loop - Expanded Superframe Format option -			UGL	00031	0.00	0.00									
	no rate			USL	CCOEF	0.00	0.00									
	TY UNBUNDLED LOCAL LOOP			UCL .	COOLI	0.00	0.00									
	High Capacity Unbundled Local Loop - DS3 - Per Mile per	1	1	İ	1	İ			İ							
	month	1	1	UE3	1L5ND	9.19										
1	High Capacity Unbundled Local Loop - DS3 - Facility	1	1								1					
	Termination per month			UE3	UE3PX	374.24	595.37	304.50	234.83	170.16			36.84	36.84	19.01	19.0
	High Capacity Unbundled Local Loop - STS-1 - Per Mile per															
	month	<u> </u>	L	UDLSX	1L5ND	9.19										
	High Capacity Unbundled Local Loop - STS-1 - Facility															
	Termination per month			UDLSX	UDLS1	389.35	595.37	304.50	215.82	151.15			36.84	36.84	19.01	19.0
): Rates provided in TN for both electronic and manual Loop	Makeu	p are ir	iterim and subject to	o retro-active	true-up adjust	ments pending	a permanent	rate ruling on	hese rate elen	ients from t	he Tenness	ee Regulatory	Authority.		
OP MAKE-U	Loop Makeup - Preordering Without Reservation, per working or															
	spare facility queried (Manual).	R		имк	UMKLW		0.76	0.76								
	Loop Makeup - Preordering With Reservation, per spare facility	ĸ		UWIN	OWINEV		0.70	0.70								
	queried (Manual).	R		ИМК	UMKLP		0.76	0.76								
	Loop MakeupWith or Without Reservation, per working or	IX.		UNIT	OWINE		0.70	0.70								
	spare facility queried (Mechanized)	R		UMK	PSUMK		0.76	0.76								
SH FREQUE	NCY SPECTRUM															
	HARING															
SPLITT	ERS-CENTRAL OFFICE BASED															
	Line Sharing Splitter, per System 96 Line Capacity			ULS	ULSDA	100.00	150.00	0.00	0.00	0.00			20.35	10.54	13.32	13.3
	Line Sharing Splitter, per System 24 Line Capacity			ULS	ULSDB	25.00	150.00	0.00	0.00	0.00			20.35	10.54	13.32	13.3
	Line Sharing-DLEC Owned Splitter in CO-CFA activaton-						100.00	0.00	00.74	0.00			00.05	10 51	40.00	40.0
	deactivation (per LSOD) SER ORDERING-CENTRAL OFFICE BASED-HIGH FREQUENC	0050		ULS	ULSDG		163.06	0.00	92.71	0.00			20.35	10.54	13.32	13.3
END U	Line Sharing - per Line Activation (BST owned Splitter)	Y SPEC	TRUM	ULS	ULSDC	0.61	40.00	31.39	0.00	0.00			20.35	10.54	13.32	13.3
	Line Sharing - per Subsequent Activity per Line			ULS	ULSDC	0.61	40.00	31.39	0.00	0.00			20.35	10.54	13.32	13.3
	Rearrangement(BST Owned Splitter)			ULS	ULSDS		30.00	15.00					20.35	10.54	13.32	13.3
	Line Sharing - per Subsequent Activity per Line			020	02020		00.00	10.00					20.00	10.01	10.02	1010
	Rearrangement(DLEC Owned Splitter)			ULS	ULSCS		30.00	15.00					20.35	10.54	13.32	13.3
	Line Sharing - per Line Activation (DLEC owned Splitter)	1		ULS	ULSCC	0.61	47.44	19.31	0.00	0.00			20.35	10.54	13.32	13.3
	PLITTING															
END US	SER ORDERING-CENTRAL OFFICE BASED															
	Line Splitting - per line activation DLEC owned splitter	I		UEPSR UEPSB	UREOS	0.61										
	Line Splitting - per line activation BST owned - physical			UEPSR UEPSB	UREBP	0.61	48.96	21.39	35.06	10.79			20.35	10.54	13.32	13.3
	Line Splitting - per line activation BST owned - virtual	I		UEPSR UEPSB	UREBV	0.61	48.96	21.39	35.06	10.79			20.35	10.54	13.32	13.3
	TE SITE HIGH FREQUENCY SPECTRUM															
	ERS-REMOTE SITE	1	+	ULS	ULSRB	25.00	150.00	0.00	150.00	0.00			20.35	10.54	13.32	13.3
					OLOND	25.00	150.00	0.00	150.00	0.00			20.33	10.54	13.32	10.0
	Remote Site Line Share BellSouth Owned Splitter, 24 Port	I		010					1		1					
	Remote Site Line Share BellSouth Owned Splitter, 24 Port Remote Site Line Share Cable Pair Activation CLEC Owned at			ULS	ULSTG		74 38	0.00	46 77	0.00			20.35	10 54	13.32	13.5
SPLITT	Remote Site Line Share BellSouth Owned Splitter, 24 Port Remote Site Line Share Cable Pair Activation CLEC Owned at RS and Deactivation	i	REMO	ULS	ULSTG		74.38	0.00	46.77	0.00			20.35	10.54	13.32	13.3
SPLITT	Remote Site Line Share BellSouth Owned Splitter, 24 Port Remote Site Line Share Cable Pair Activation CLEC Owned at RS and Deactivation SER ORDERING-REMOTE SITE HIGH FREQUENCY SPECTRUI	i	REMOT	ULS			74.38	0.00	46.77	0.00			20.35	10.54	13.32	13.3
SPLITT	Remote Site Line Share BellSouth Owned Splitter, 24 Port Remote Site Line Share Cable Pair Activation CLEC Owned at RS and Deactivation	i	REMOT	ULS		0.61	74.38	0.00	46.77 35.06	0.00			20.35	10.54	13.32	
SPLITT	Remote Site Line Share BellSouth Owned Splitter, 24 Port Remote Site Line Share Cable Pair Activation CLEC Owned at RS and Deactivation SER ORDERING-REMOTE SITE HIGH FREQUENCY SPECTRUI Remote Site Line Share Line Activationfor End User Served at RS, BST Splitter RS Line Share Line Activation for End User served at RS, CLEC	i	REMO	ULS TE SITE LINE SHARI ULS	NG ULSRC		40.00	31.39	35.06	10.79			20.35	10.54	13.32	13.3
END US	Remote Site Line Share BellSouth Owned Splitter, 24 Port Remote Site Line Share Cable Pair Activation CLEC Owned at RS and Deactivation SER ORDERING-REMOTE SITE HIGH FREQUENCY SPECTRUI Remote Site Line Share Line Activationfor End User Served at RS, BST Splitter RS Line Share Line Activation for End User served at RS, CLEC Splitter	i	REMOT	ULS TE SITE LINE SHARI	NG	0.61										13.3
END US	Remote Site Line Share BellSouth Owned Splitter, 24 Port Remote Site Line Share Cable Pair Activation CLEC Owned at RS and Deactivation SER ORDERING-REMOTE SITE HIGH FREQUENCY SPECTRUI Remote Site Line Share Line Activationfor End User Served at RS, BST Splitter RS Line Share Line Activation for End User served at RS, CLEC	I M AKA I		ULS FE SITE LINE SHARI ULS ULS	NG ULSRC ULSTC	0.61	40.00	31.39	35.06	10.79			20.35	10.54	13.32	

UNBUNDL	ED NETWORK ELEMENTS - Tennessee												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade - Per Mile per month			U1TVX	1L5XX	0.0054										
	Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade -			01177	ILJAA	0.0054										
	Facility Termination			U1TVX	U1TV2	18.58	55.39	17.37	27.96	3.51			20.35	21.09	9.80	10.54
	Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade Rev Bat Per Mile per month			U1TVX	1L5XX	0.0054										
	Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat			OTTAX	TE0/01	0.0004										
	Facility Termination			U1TVX	U1TR2	18.58	55.39	17.37	27.96	3.51			20.35	21.09	9.80	10.54
	Interoffice Channel - Dedicated Transport - 4-Wire Voice Grade - Per Mile per month			U1TVX	1L5XX	0.0054										
	Interoffice Channel - Dedicated Transport - 4- Wire Voice Grade			01117	120701	0.0001										
	- Facility Termination			U1TVX	U1TV4	24.09	37.87	26.02	30.78	13.07			15.08	15.08	8.66	8.66
	Interoffice Channel - Dedicated Transport - 56 kbps - per mile per month			U1TDX	1L5XX	0.0174										
	Interoffice Channel - Dedicated Transport - 56 kbps - Facility			UTIDA	ILSAA	0.0174										
	Termination			U1TDX	U1TD5	17.98	55.39	17.37	27.96	3.51			20.35	21.09	9.80	10.54
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile															
	per month Interoffice Channel - Dedicated Transport - 64 kbps - Facility			U1TDX	1L5XX	0.0174										
	Termination			U1TDX	U1TD6	17.98	55.39	17.37	27.96	3.51			20.35	21.09	9.80	10.54
	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per															
	month			U1TD1	1L5XX	0.3562										
	Interoffice Channel - Dedicated Tranport - DS1 - Facility Termination			U1TD1	U1TF1	77.86	112.40	76.27	19.55	14.99			20.35	21.09	9.80	10.54
	Interoffice Channel - Dedicated Transport - DS3 - Per Mile per						1									
	month			U1TD3	1L5XX	2.34										
	Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month			U1TD3	U1TF3	848.99	395.29	176.56	109.04	105.91			36.84	36.84	19.01	19.01
	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per			01103	01113	040.33	333.23	170.50	103.04	103.31			50.04	30.04	13.01	13.01
	month			U1TS1	1L5XX	2.34										
	Interoffice Channel - Dedicated Transport - STS-1 - Facility							170 50								
1.00	Termination AL CHANNEL - DEDICATED TRANSPORT			U1TS1	U1TFS	849.30	395.29	176.56	109.04	105.91			36.84	36.84	19.01	19.01
	E: LOCAL CHANNEL DEDICATED TRANSPORT - minimum billing	a perio	d - belo	w DS3=one month.	DS3/STS-1=f	our months										
	Local Channel - Dedicated - 2-Wire Voice Grade - Zone 1	5		ULDVX	ULDV2	17.18	199.33	24.16	54.81	4.80						
	Local Channel - Dedicated - 2-Wire Voice Grade - Zone 2		2	ULDVX	ULDV2	22.44	199.33	24.16	54.81	4.80						
	Local Channel - Dedicated - 2-Wire Voice Grade - Zone 3		3	UNDVX	ULDV2	29.34	199.33	24.16	54.81	4.80						
	Local Channel - Dedicated - 2-Wire Voice Grade Rev. Bat Zone 1		1	ULDVX	ULDR2	17.18	199.33	24.16	54.81	4.80						
	Local Channel - Dedicated - 2-Wire Voice Grade Rev. Bat			OLDVA		17.18	199.33	24.10	54.61	4.80						
	Zone 2		2	ULDVX	ULDR2	22.44	199.33	24.16	54.81	4.80						
	Local Channel - Dedicated - 2-Wire Voice Grade Rev. Bat															
	Zone 3		3	ULDVX	ULDR2	29.34	199.33	24.16	54.81	4.80						
	Local Channel - Dedicated - 4-Wire Voice Grade - Zone 1		1	UNDVX	ULDV4	18.18	201.53	24.83	55.52	5.51						
	Local Channel - Dedicated - 4-Wire Voice Grade - Zone 2		2		ULDV4	23.74	201.53	24.83	55.52	5.51	-					
├ ── ├ ──	Local Channel - Dedicated - 4-Wire Voice Grade - Zone 3 Local Channel - Dedicated - DS1 - Zone 1		3	UNDVX ULDD1	ULDV4 ULDF1	31.05 36.24	201.53 277.35	24.83 233.26	55.52 33.18	5.51 22.30						<u> </u>
┣──┤──	Local Channel - Dedicated - DS1 - Zone 1		2	ULDD1	ULDF1	47.33	277.35	233.26	33.18	22.30	+		ł	ł	ł	<u> </u>
	Local Channel - Dedicated - DS1 - Zone 2		3	ULDD1	ULDF1	61.89	277.35	233.26	33.18	22.30						1
	Local Channel - Dedicated - DS1 - Zone S Local Channel - Dedicated - DS3 - Per Mile per month			ULDD3	1L5NC	7.15	211.00	200.20	00.10	22.00	1					
	Local Channel - Dedicated - DS3 - Facility Termination			ULDD3	ULDF3	611.30	595.37	304.50	215.82	151.15	1		36.84	36.84	19.01	19.01
	Local Channel - Dedicated - STS-1- Per Mile per month			ULDS1	1L5NC	7.15										
	Local Channel - Dedicated - STS-1 - Facility Termination			ULDS1	ULDFS	599.59	588.07	297.20	215.82	151.15			20.35	21.09	9.80	10.54
DARK FIBER																
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction				41.500											
	Thereof per month - Local Channel NRC Dark Fiber - Local Channel			UDF UDF	1L5DC UDFC4	58.83	1,121.00	153.19	580.26	357.17			20.35	21.09	9.80	10.54
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction				00104		1,121.00	155.19	300.20	337.17	+		20.35	21.09	9.60	10.34
			1	UDF	1L5DF				1		1		1		1	1
UNBUNDLE	D NETWORK ELEMENTS - Tennessee										a - ·		Attachment:			bit: B
--------------	---	-------------	----------	------------	----------------	-----------	--------------	-----------	--------------	---------	---	-------	--	--	---	---
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	NRC Dark Fiber - Interoffice Channel			UDF	UDF14		1,121.00	153.19	580.26	357.17			20.35	21.09	9.80	10.54
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction					50.00										
	Thereof per month - Local Loop			UDF UDF	1L5DL UDFL4	58.83	1,121.00	153.19	580.26	357.17			20.35	21.09	9.80	10.54
	NRC Dark Fiber - Local Loop TEN DIGIT SCREENING			UDF	UDFL4		1,121.00	153.19	580.26	357.17			20.35	21.09	9.80	10.54
ONN ACCESS	8XX Access Ten Digit Screening, Per Call			OHD	+	0.0005192								-		-
	8XX Access Ten Digit Screening, Reservation Charge Per 8XX					0.0003192										
	Number Reserved			OHD	N8R1X		5.21	0.76					20.35	20.35	13.28	13.28
	8XX Access Ten Digit Screening, Per 8XX No. Established W/O			0110			0.21	0.10			1		20.00	20.00	10.20	10.20
	POTS Translations			OHD			11.47	1.46	7.34	0.7602			20.35	20.35	13.28	13.28
i	8XX Access Ten Digit Screening, Per 8XX No. Established With	1			1			10		0.1.002			20.00	20.00		.0.20
	POTS Translations			OHD	N8FTX		11.47	1.46	7.34	0.7602			20.35	20.35	13.28	13.28
	8XX Access Ten Digit Screening, Customized Area of Service	1			1											
	Per 8XX Number			OHD	N8FCX		4.47	2.24					20.35	20.35	13.28	13.28
	8XX Access Ten Digit Screening, Multiple InterLATA CXR															
	Routing Per CXR Requested Per 8XX No.			OHD	N8FMX		5.23	3.00					20.35	20.35	13.28	13.28
	8XX Access Ten Digit Screening, Change Charge Per Request			OHD	N8FAX		5.97	0.76					20.35	20.35	13.28	13.28
	8XX Access Ten Digit Screening, Call Handling and Destination															
	Features			OHD	N8FDX		4.47						20.35	20.35	13.28	13.28
LINE INFORM	ATION DATA BASE ACCESS (LIDB)															
	LIDB Common Transport Per Query			OQT		0.0000354										
	LIDB Validation Per Query			OQU		0.0117403										
	LIDB Originating Point Code Establishment or Change			OQT, OQU	NRPBX		49.03						20.35	20.35	13.28	13.28
SIGNALING (C																
	CCS7 Signaling Termination, Per STP Port			UDB	PT8SX	138.41										
	CCS7 Signaling Usage, Per TCAP Message			UDB		0.0000916										
	CCS7 Signaling Connection, Per link (A link)			UDB	TPP++	17.84	130.84	130.84					20.35	20.35	13.32	13.32
	CCS7 Signaling Connection, Per link (B link) (also known as D					17.01	100.01								10.00	10.00
	link)			UDB	TPP++	17.84	130.84	130.84					20.35	20.35	13.32	13.32
	CCS7 Signaling Usage, Per ISUP Message			UDB	OTUER	0.0000373										
	CCS7 Signaling Usage Surrogate, per link per LATA Signaling Point Code, per Originating Point Code Establishment			UDB	STU56	352.30										
	or Change, per STP			UDB	CCAPO		121.77	121.77					20.35	20.35	13.32	13.32
	ME (CNAM) SERVICE			ODB	CCAPO		121.77	121.77			1		20.35	20.33	13.32	13.32
	CNAM for DB Owners, Per Query			OQV		0.0010541										
	CNAM for Non DB Owners, Per Query			OQV		0.0010541										
	CNAM (Non-Databs Owner), NRC, applies when using the				1	0.0010041			1		t				1	1
	Character Based User Interface (CHUI)			OQV	CDDCH		595.00	595.00					20.35	20.35	13.28	13.28
OPERATOR C	ALL PROCESSING	1														
	Oper. Call Processing - Oper. Provided, Per Min Using BST										İ					
	LIDB					1.08										
	Oper. Call Processing - Oper. Provided, Per Min Using															
	Foreign LIDB					1.13										
	Oper. Call Processing - Fully Automated, per Call - Using BST															
	LIDB					0.1010353										
	Oper. Call Processing - Fully Automated, per Call - Using					Т	Т									I
	Foreign LIDB					0.122818										ļ
INWARD OPE	RATOR SERVICES					1.00										ļ
	Inward Operator Services - Verification, Per Minute		<u> </u>		+	1.03										ł
	Inward Operator Services - Verification and Emergency Interrupt				1	4.00										1
	- Per Minute OPERATOR CALL PROCESSING					1.03										
	y based CLEC									1						
Facility	Recording of Custom Branded OA Announcement				CBAOS		1,555.00	1,553.00	7.03	7.03			19.99	19.99	19.99	19.99
	Loading of Custom Branded OA Announcement per shelf/NAV				SDAGO		1,000.00	1,000.00	7.03	1.03	-		13.33	13.33	13.33	13.35
	per OCN				CBAOL		240.71	240.71					19.99	19.99		1
							210.71	2.0.71								
UNEP	CLEC															

UNB	UNDLED	NETWORK ELEMENTS - Tennessee												Attachment:	2	Exhi	ibit: B
CATE	GORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Charge - Manual Svc Order vs.
							Rec	Nonrecurring		Nonrecurring			_		Rates(\$)	-	
		Loading of Custom Branded OA Announcement per shelf/NAV						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		per OCN						240.71	240.71					19.99	19.99		
		ding via OLNS for UNEP CLEC															
		Loading of OA per OCN (Regional)						1,200.00	1,200.00					19.99	19.99		
DIRE		SISTANCE SERVICES															
		ORY ASSISTANCE ACCESS SERVICE Directory Assistance Access Service Calls, Charge Per Call					0.0000707										
		ORY ASSISTANCE CALL COMPLETION ACCESS SERVICE (0.2286787										
		Directory Assistance Call Completion Access Service (DACC),	,,,,,,														
		Per Call Attempt					0.0364771										
	NUMBE	R SERVICES INTERCEPT ACCESS SERVICE															
		Number Services Intercept Per Query					0.017793										
		ORY TRANSPORT (DT)															
		DT-Local Channel DS1					40.99	277.35	233.26	33.18	22.30			20.35	10.54	13.32	1.40
		DT-DS1 Level Interoffice per mile DT-DS1 Level Interoffice per facility termination					0.3562 77.86	112.40	76.27	19.55	14.99			20.35	10.54	13.32	1.40
		SWA Common Transport per Directory Assistance Access					77.00	112.40	10.21	19.55	14.55			20.33	10.34	13.32	1.40
		Service Per Call					0.000271										
		SWA Common Transport per Directory Assistance Access				1						1					
		Service Per Call Per Mile					0.0000165										
		Access Tandem Switching Per Directory Assistance Access															
		Service Per Call					0.0001875									-	
		DT- Directory Assistance Interconnection Per Directory Assistance Service Call					0.00										
		DT-Installation NRC, Per Trunk or Signaling Connection					0.00	204.62	4.43	136.09	4.43			20.35	10.54	13.32	1.40
		DT Local Channel DS1-Incremental Cost-Manual Svc Order vs						201102		100.00				20100	10.01	10.02	
		Electronic						45.68	1.76	21.75	1.76						
		DT Interoffice DS1-Incremental Cost-Manual Svc Order vs															
		Electronic						20.35	21.09	9.80	10.54						
DIRE		SISTANCE SERVICES ORY ASSISTANCE DATA BASE SERVICE (DADS)															
		Directory Assistance Data Base Service (DADS)					0.0485										
-		Directory Assistance Data Base Service Charge Fer Listing				DBSOF	104.13										
BRAN		RECTORY ASSISTANCE				5500.	101110									1	
	Facility	Based CLEC				1						1					
		Recording and Provisioning of DA Custom Branded															
		Announcement			AMT	CBADA		1,555.00	1,553.00	7.03	7.03			20.35	10.54	13.32	1.40
	UNEP C	Loading of Custom Branded Announcement per Switch			AMT	CBADC		240.71	240.71					20.35	10.54		
-		Recording of DA Custom Branded Announcement						1.555.00	1,553.00	7.03	7.03			20.35	10.54	13.32	1.40
		Loading of DA Custom Branded Announcement per Switch per				1		1,000.00	1,000.00	1.05	,.05			20.00	10.04	10.02	1.40
		OCN						240.71	240.71					20.35	10.54	1	
		ding via OLNS for UNEP CLEC															
		Loading of DA per OCN (1 OCN per Order)						420.00	420.00					20.35	10.54		
051 5		Loading of DA per Switch per OCN	<u> </u>	+				16.00	16.00					20.35	10.54	L	<u> </u>
SELE	CTIVE RO	UTING Selective Routing Per Unique Line Class Code Per Request Per		$\left - \right $		+										<u> </u>	
		Selective Routing Per Unique Line Class Code Per Request Per Switch				USRCR		179.60	179.60					20.35	20.35		
VIRTU		OCATION	-			50.00			110.00	1	1			20.00	20.00	1	
		Virtual Collocation - Application Cost	1		AMTFS	EAF		2,633.00	2,633.00					2.07	2.81	0.67	1.41
		Virtual Collocation - Cable Installation Cost, per cable			AMTFS	ESPCX		1,749.00	1,749.00					2.07	2.81	0.67	1.41
		Virtual Collocation - Floor Space, per sq. ft.			AMTFS	ESPVX	3.91									ļ	
L		Virtual Collocation - Power, per fused amp		+	AMTFS	ESPAX	6.79									ļ	
		Virtual Collocation - Cable Support Structure, per entrance	1	1		1				1	1	1		1	1	1	1

UNBUN	IDLEI	D NETWORK ELEMENTS - Tennessee												Attachment:	2	Exhi	bit: B
CATEGO		RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
								Nonrecurring		Nonrecurring	Disconnect			055	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Virtual Collocation - 2-wire Cross Connects (loop)			UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ, AMTFS, UDL, UNCVX, UNCDX, UNCNX	UEAC2	0.57	11.62	9.90	10.38	8.66			2.07	2.81	0.67	1.41
					UEA,UHL,UCL,UDL, AMTFS, UAL, UDN,												
		Virtual Collocation - 4-wire Cross Connects (loop)			UNCVX, UNCDX	UEAC4	0.57	11.81	10.04	10.44	8.67			2.07	2.81	0.67	1.41
		Virtual Collocation - 2-Fiber Cross Connects			AMTFS,UDL12, UDLO3, U1T48, U1T12, U1T03, ULDO3, ULD12, ULD48, UDF	CNC2F	3.03	41.56	29.82	12.96	10.34			2.69	2.69	1.56	1.56
		Virtual Collocation - 4-Fiber Cross Connects			AMTFS,UDL12, UDLO3, U1T48, U1T12, U1T03, ULDO3, ULD12, ULD48, UDF	CNC4F	6.06	50.53	38.78	16.97	14.35			2.69	2.69	1.56	1.56
		Virtual collocation - Special Access & UNE, cross-connect per DS1			USL,ULC,AMTFS, ULR, UXTD1, UNC1X, ULDD1, U1TD1, USLEL, UNLD1	CNC1X	1.32	32.22	17.76	10.46	8.75			2.07	2.81	0.67	1.41
		Virtual collocation - Special Acess & UNE, cross-connect per DS3			USL, ULC, AMTFS, U E3, U1TD3, UXTS1, UXTD3, UNC3X, UNCSX, ULDD3, U1TS1, ULDS1, UDLSX, UNLD3	CND3X	12.32	29.97	16.30	12.03	8.99			2.07	2.81	0.67	1.41
		Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear foot			AMTES	VE1CB	0.0031										
		Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax			AMTFS	VE1CD	0.0045										
		Cable Support Structure, per linear ft Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure,per cable			AMTES	VE1CD	0.0045	555.03						2.07	2.81	0.67	1.41
		Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax			AIVIT 5	VLICC		555.05						2.07	2.01	0.07	1.41
		Cable Support Structure, per cable			AMTFS	VE1CE		555.03						2.07	2.81	0.67	1.41
		Virtual Collocation Cable Records - per request			AMTFS	VE1BA		1,711.00									
		Virtual Collocation Cable Records - VG/DS0 Cable, per cable record			AMTFS	VE1BB		925.06				ļ					
		Virtual Collocation Cable Records - VG/DS0 Cable, per each 100 pair			AMTFS	VE1BC		18.05	18.05								
\vdash		Virtual Collocation Cable Records - DS1, per T1TIE			AMTES	VE1BD		8.45	8.45								
		Virtual Collocation Cable Records - DS3, per T3TIE			AMTFS	VE1BE	İ	29.57	29.57								
		Virtual Collocation Cable Records - Fiber Cable, per 99 fiber records			AMTES	VE1BF		279.42	279.42								
		Virtual collocation - Security Escort - Basic, per half hour			AMTFS	SPTBX		33.15	20.44					2.07	2.81	0.67	1.41
		Virtual collocation - Security Escort - Overtime, per half hour			AMTFS	SPTOX		41.50	25.61					2.07	2.81	0.67	1.41
\vdash		Virtual collocation - Security Escort - Premium, per half hour			AMTES	SPTPX		49.86	30.79					2.07	2.81	0.67	1.41
		Virtual collocation - Maintenance in CO - Basic, per half hour			AMTFS AMTFS	CTRLX SPTOM		30.64 35.77	30.64					2.07	2.81	0.67	1.41
		Virtual collocation - Maintenance in CO - Overtime, per half hour Virtual collocation - Maintenance in CO - Premium per half hour			AMTES	SPTOM		40.90	40.90					2.07	2.81	0.67	1.41
VIRTUAI	COLI	COCATION			AWITES			40.90	40.90					2.07	2.81	0.67	1.41
		Virtual Collocation - 2-wire Cross Connect, Exchange Port 2- Wire Analog - Res			UEPSR	VE1R2	0.30	19.20	19.20					20.35	10.54	13.32	1.40

UNBU	NDLE	D NETWORK ELEMENTS - Tennessee												Attachment:	2	Exhi	bit: B
CATEG		RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
							Rec	Nonrecurring		Nonrecurring					Rates(\$)		
		Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-	-					First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Wire Line Side PBX Trunk - Bus			UEPSP	VE1R2	0.30	19.20	19.20					20.35	10.54	13.32	1.40
		Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Voice Grade PBX Trunk - Res			UEPSE	VE1R2	0.30	19.20	19.20					20.35	10.54	13.32	1.40
		Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Analog Bus			UEPSB	VE1R2	0.30	19.20	19.20					20.35	10.54	13.32	1.40
		Virtual Collocation 2-Wire Cross Connect, Exchnage Port 2-Wire															
		ISDN Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire			UEPSX	VE1R2	0.30	19.20	19.20					20.35	10.54	13.32	1.40
		ISDN			UEPTX	VE1R2	0.30	19.20	19.20	-				20.35	10.54	13.32	1.40
VIDTUA		Virtual Collocation 4-Wire Cross Connect, Exchange Port 4-Wire ISDN DS1			UEPEX	VE1R4	0.50	19.20	19.20					20.35	10.54	13.32	1.40
VIRTUA	L COLL	OCATION Virtual Collocation-2 Wire Cross Connects (Loop) for Line			-					-		-					
		Splitting LOCATION			UEPSR, UEPSB	VE1LS	0.57	11.62	9.90	10.38	8.66			19.99	19.99	19.99	19.99
FHISIC	AL COL	Physical Collocation-2 Wire Cross Connects (Loop) for Line															
AIN SEI	FCTIV	Splitting E CARRIER ROUTING			UEPSR, UEPSB	PE1LS	0.0318	11.94	11.46					19.99	19.99	19.99	19.99
		Regional Service Establishment			SRC	SRCEC		190,638.00						20.35	1		1
		End Office Establishment			SRC	SRCEO		317.55	317.55	3.19	3.19			20.35	20.35	13.28	13.28
		Query NRC, per query			SRC		0.0206047										
AIN - BE	LLSOL	ITH AIN SMS ACCESS SERVICE															
		AIN SMS Access Service - Service Establishment, Per State, Initial Setup			A1N	CAMSE		135.56	135.56					20.35	20.35	13.28	13.28
		AIN SMS Access Service - Port Connection - Dial/Shared Access			A1N	CAMDP		41.75	41.75					20.35	20.35	13.28	13.28
		AIN SMS Access Service - Port Connection - ISDN Access			A1N	CAM1P		41.75	41.75					20.35	20.35	13.28	13.28
		AIN SMS Access Service - User Identification Codes - Per User ID Code			A1N	CAMAU		96.63	96.63					20.35	20.35	13.28	13.28
		AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement			A1N	CAMRC		113.67	113.67					20.35	20.35	13.28	13.28
		AIN SMS Access Service - Storage, Per Unit (100 Kilobytes)					0.0024										
		AIN SMS Access Service - Session, Per Minute					0.0820123										
		AIN SMS Access Service - Company Performed Session, Per Minute					2.27										
AIN - BE	LLSOU	ITH AIN TOOLKIT SERVICE				1										1	
		AIN Toolkit Service - Service Establishment Charge, Per State, Initial Setup			CAM	BAPSC		132.04	132.04					20.35	20.35	13.28	13.28
		AIN Toolkit Service - Training Session, Per Customer				BAPVX		7,915.00	7,915.00					20.35	20.35	13.28	13.28
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt				BAPTT		31.21	31.21					20.35	20.35	13.28	13.28
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay				BAPTD		31.21	31.21					20.35	20.35	13.28	13.28
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate				BAPTM		31.21	31.21					20.35	20.35	13.28	13.28
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP				BAPTO		85.24	85.24					20.35	20.35	13.28	13.28
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP				BAPTC		85.24	85.24					20.35	20.35	13.28	13.28
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code				BAPTF		85.24	85.24					20.35	20.35	13.28	13.28
		AIN Toolkit Service - Query Charge, Per Query			<u> </u>		0.0211882				İ						
		AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit Subscription, Per Node, Per Query					0.0054774										
		AIN Toolkit Service - SCP Storage Charge, Per SMS Access Account, Per 100 Kilobytes				1	1.50									1	
		AIN Toolkit Service - Monthly report - Per AIN Toolkit Service Subscription			CAM	BAPMS	17.43	33.52	33.52					20.35	20.35	13.28	13.28

UNBUNDLE	D NETWORK ELEMENTS - Tennessee												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st		Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrecurring		Nonrecurring	g Disconnect			OSS	Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	AIN Toolkit Service - Special Study - Per AIN Toolkit Service Subscription			CAM	BAPLS	0.1321116	36.23	36.23					20.35	20.35	13.28	13.28
	AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service Subscription			CAM	BAPDS	17.35	33.52	33.52					20.35	20.35	13.28	13.28
	AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit Service Subscription			CAM	BAPES	0.0511435	36.23	36.23					20.35	20.35	13.28	13.28
	XTENDED LINK (EELs)															
	New Density Zone 1 EELs are available in the following MSA					Atlanta, Ga; Ne	w Orleans, LA,									
	Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem															
	In all states, EEL network elements shown below also apply t												UNEs.(Non-re	curring rates	do not apply	.)
	In All States the EEL network elements apply to ordinarily control of the second states and the second s				tch As is Cha	arge.) when or	dering ordinar	ily combined i	network elemen	its, Non-recur	ing rates d	o appiy.				
2-9916	First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport	EROFF		ANSPORT (EEL)												
	Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed		1	UNCVX	UEAL2	16.56	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed		2	UNCVX	UEAL2	21.63	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile		3	UNCVX	UEAL2	28.28	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	per month			UNC1X	1L5XX	0.3562										
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination per month			UNC1X	U1TF1	77.86	171.24	113.12	70.07	30.90			20.35	21.09	9.80	10.54
	DS1 Channelization System Per Month			UNC1X	MQ1 1D1VG	80.77	105.76	14.48	3.04	2.74						
	Voice Grade COCI - DS1 To Ds0 Interface - Per Month Each Additional 2-Wire VG Loop(SL 2) in the same DS1			UNCVX	IDIVG	0.91	5.70	4.42								
	Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL2	16.56	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL2	21.63	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL2	28.28	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Voice Grade COCI - DS1 to DS0 Channel System combination - per month			UNCVX	1D1VG	0.91	5.70	4.42								
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNC1X	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.54
4-WIR	E VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INT	EROFF	ICE TR	ANSPORT (EEL)												
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL4	24.70	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL4	32.26	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL4	42.18	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month			UNC1X	1L5XX	0.3562										
	Interoffice Transport - Dedicated - DS1 - Facility Termination Per Month			UNC1X	U1TF1	77.86	171.24	113.12	70.07	30.90			20.35	21.09	9.80	10.54
	Channelization - Channel System DS1 to DS0 combination Per Month			UNC1X	MQ1	80.77	105.76	14.48	3.04	2.74						
	Voice Grade COCI - DS1 to DS0 Channel System combination - per month			UNCVX	1D1VG	0.91	5.70	4.42								
	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL4	24.70	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL4	32.26	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL4	42.18	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Voice Grade COCI - DS1 to DS0 Channel System combination - per month			UNCVX	1D1VG	0.91	5.70	4.42	72.04	10.00			20.00	21.00	0.00	10.04
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNC1X	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.54

UNBUNDL	ED NETWORK ELEMENTS - Tennessee	r	r		1	1						-	Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
4 14/15	RE 56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1	INTERC					First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
4-9916	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice	INTERC	FFICE	TRANSPORT (EEL)	ł											i
	Transport Combination - Zone 1		1	UNCDX	UDL56	31.10	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice								-							
	Transport Combination - Zone 2		2	UNCDX	UDL56	40.61	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice					=0.44			70.04							
	Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile		3	UNCDX	UDL56	53.11	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Per Month			UNC1X	1L5XX	0.3562										i
	Interoffice Transport - Dedicated - DS1 - combination Facility			ononx	120/01	0.0002										
	Termination Per Month			UNC1X	U1TF1	77.86	171.24	113.12	70.07	30.90			20.35	21.09	9.80	10.54
	Channelization - Channel System DS1 to DS0 combination Per															[
	Month			UNC1X	MQ1	80.77	105.76	14.48	3.04	2.74						1
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per															1
	month (2.4-64kbs) Additional 4-Wire 56Kbps Digital Grade Loopin same DS1			UNCDX	1D1DD	0.91	5.70	4.42								
	Interoffice Transport Combination - Zone 1		1	UNCDX	UDL56	31.10	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1		-	ONODA	ODLOO	01.10	100.70	00.41	72.04	10.00			20.00	21.00	0.00	10.04
	Interoffice Transport Combination - Zone 2		2	UNCDX	UDL56	40.61	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1															ſ
	Interoffice Transport Combination - Zone 3		3	UNCDX	UDL56	53.11	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	OCU-DP COCI (data) - DS1 to DS0 Channel System -															1
	combination per month (2.4-64kbs) Nonrecurring Currently Combined Network Elements Switch -As-			UNCDX	1D1DD	0.91	5.70	4.42								l
	Is Charge			UNC1X	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.54
4-WI	RE 64 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1	INTERC	FFICE				52.15	24.02	3.12	3.12			20.33	21.03	3.00	10.54
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice															
	Transport Combination - Zone 1		1	UNCDX	UDL64	31.10	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice															1
	Transport Combination - Zone 2		2	UNCDX	UDL64	40.61	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3		3	UNCDX	UDL64	53.11	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Interoffice Transport - Dedicated - DS1 combination - Per Mile		5	ONODA	ODL04	55.11	100.70	55.47	72.34	10.00			20.33	21.03	3.00	10.54
	Per Month			UNC1X	1L5XX	0.3562										i
	Interoffice Transport - Dedicated - DS1 combination - Facility															ſ
	Termination Per Month			UNC1X	U1TF1	77.86	171.24	113.12	70.07	30.90			20.35	21.09	9.80	10.54
	Channelization - Channel System DS1 to DS0 combination Per															
	Month OCU-DP COCI (data) - DS1 to DS0 Channel System			UNC1X	MQ1	80.77	105.76	14.48	3.04	2.74			20.35	21.09	9.80	10.54
	combination - per month (2.4-64kbs)			UNCDX	1D1DD	0.91	5.70	4.42								1
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1			ONODA	10100	0.01	0.70	-112								l .
	Interoffice Transport Combination - Zone 1		1	UNCDX	UDL64	31.10	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1															ſ
	Interoffice Transport Combination - Zone 2		2	UNCDX	UDL64	40.61	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1		3		UDL64	50.44	400.70	35.47	70.04	40.00			20.35	21.09	9.80	10.54
	Interoffice Transport Combination - Zone 3 OCU-DP COCI (data) - DS1 to DS0 Channel System		3	UNCDX	UDL64	53.11	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	combination - per month (2.4-64kbs)			UNCDX	1D1DD	0.91	5.70	4.42								1
	Nonrecurring Currently Combined Network Elements Switch -As-			01102/1	10100	0.01	0.10									1
	Is Charge			UNC1X	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.54
4-WI	RE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTI	EROFFI	CE TR/	NSPORT (EEL)												
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice						600 1-						~~~~			
├ ── ├ ──	Transport - Zone 1		1	UNC1X	USLXX	57.73	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.54
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2		2	UNC1X	USLXX	75.40	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.54
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice		- 4		JOLAA	75.40	220.40	101.74	19.01	24.00			20.35	21.09	9.00	10.34
	Transport - Zone 3		3	UNC1X	USLXX	98.59	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.54
	Interoffice Transport - Dedicated - DS1 combination - Per Mile	1														(
1	Per Month			UNC1X	1L5XX	0.3562										L

UNBUNDLE	ED NETWORK ELEMENTS - Tennessee		-	r	· · ·						Γ	Г	Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
						neo	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Month			UNC1X	U1TF1	77.86	171.24	113.12	70.07	30.90			20.35	21.09	9.80	10.54
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNC1X	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.54
4-WIR	E DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTE	EROFE			UNCCC		52.75	24.02	9.12	9.12			20.35	21.09	9.60	10.54
4-111	First DS1Loop in DS3 Interoffice Transport Combination - Zone															
			1	UNC1X	USLXX	57.73	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.54
	First DS1Loop in DS3 Interoffice Transport Combination - Zone															
	2		2	UNC1X	USLXX	75.40	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.54
	First DS1Loop in DS3 Interoffice Transport Combination - Zone 3		3	UNC1X	USLXX	98.59	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.54
	Interoffice Transport - Dedicated - DS3 combination - Per Mile															
	Per Month			UNC3X	1L5XX	2.34										ļ
	Interoffice Transport - Dedicated - DS3 - Facility Termination per															
	month			UNC3X UNC3X	U1TF3 MQ3	854.97	482.01 156.02	153.81 49.41	64.43 17.12	35.43 6.77			20.35	21.09	9.80	10.54
	DS3 to DS1 Channel System combination per month DS3 Interface Unit (DS1 COCI) combination per month			UNC3X UNC1X	UC1D1	222.98 17.58	5.70	49.41	17.12	6.77						
	Additional DS1Loop in DS3 Interoffice Transport Combination -			UNCIA	00101	17.50	5.70	4.42								
	Zone 1		1	UNC1X	USLXX	57.73	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.54
	Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2		2	UNC1X	USLXX	75.40	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.54
	Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 3		3	UNC1X	USLXX	98.59	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.54
	DS3 Interface Unit (DS1 COCI) combination per month		5	UNC1X	UC1D1	17.58	5.70	4.42	19.01	24.00			20.33	21.09	9.00	10.34
	Nonrecurring Currently Combined Network Elements Switch -As-															
	Is Charge			UNC3X	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.54
2-WIR	E VOICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INT	FEROFF	ICE TR	ANSPORT (EEL)												
	2-WireVG Loop used with 2-wire VG Interoffice Transport															
	Combination - Zone 1		1	UNCVX	UEAL2	16.56	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	2-WireVG Loop used with 2-wire VG Interoffice Transport		2			04.00	400.70	05.47	70.04	10.00			20.25	04.00	0.00	40.54
	Combination - Zone 2 2-WireVG Loop used with 2-wire VG Interoffice Transport		2	UNCVX	UEAL2	21.63	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Combination - Zone 3		3	UNCVX	UEAL2	28.28	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month			UNCVX	1L5XX	0.0174										
	Interoffice Transport - Dedicated - 2- Wire Voice Grade			UNCVA	ILSAA	0.0174										
	combination - Facility Termination per month			UNCVX	U1TV2	21.79	79.83	44.08	69.32	31.00			20.35	21.09	9.80	10.54
	Nonrecurring Currently Combined Network Elements Switch -As-				01112	20	10.00	1100	00.02	01.00			20.00	21100	0.00	10101
	Is Charge			UNCVX	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.54
4-WIR	E VOICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INT	TEROFF	ICE TR	ANSPORT (EEL)												
	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL4	24.70	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL4	32.26	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL4	42.18	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Interoffice Transport - Dedicated - 4-wire VG combination - Per			0	52/124	72.10	100.70	00.47	72.04	10.00			20.00	21.03	0.00	10.04
	Mile Per Month			UNCVX	1L5XX	0.0174										
	Interoffice Transport - Dedicated - 4- Wire Voice Grade				1							ĺ			İ	1
	combination - Facility Termination per month			UNCVX	U1TV4	27.30	79.83	44.08	69.32	31.00			20.35	21.09	9.80	10.54
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNCVX	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.54
D\$3 D	IGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFIC	E TRAI	SPOR											0	2.20	
	High Capacity Unbundled Local Loop - DS3 combination - Per															1
	Mile per month			UNC3X	1L5ND	9.19										
	Ulinh Connector Unbrundled Level Level DC2 combination	1		1							1	1				
	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month			UNC3X	UE3PX	373.47	240.23	180.87	106.78	45.24			20.35	21.09	9.80	10.54

NBUNDLI	ED NETWORK ELEMENTS - Tennessee												Attachment:			bit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Svc Order Submitted Elec per LSR		Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
					_	Rec	Nonrecurring		Nonrecurring		COMEC	COMAN		Rates(\$)	COMAN	COMAN
	Interoffice Transport - Dedicated - DS3 combination - Facility						First	Add'l	First	Add'l	SOWIEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Termination per per month			UNC3X	U1TF3	854.97	482.01	153.81	64.43	35.43			20.35	21.09	9.80	10.54
	Nonrecurring Currently Combined Network Elements Switch -As-															
070/	Is Charge			UNC3X	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.54
5151	DIGITAL EXTENDED LOOP WITH DEDICATED STS1 INTEROF High Capacity Unbundled Local Loop - STS1 combination - Per		ANSPO	DRT (EEL)												
	Mile per month			UNCSX	1L5ND	9.19										
	High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per month			UNCSX	UDLS1	394.56	240.23	180.87	106.78	45.24			20.35	21.09	9.80	10.54
	Interoffice Transport - Dedicated - STS1 combination - Per Mile				1L5XX	2.34										
	per month Interoffice Transport - Dedicated - STS1 combination - Facility			UNCSX	ILSAA	2.34										ł
	Termination per month Nonrecurring Currently Combined Network Elements Switch -As-			UNCSX	U1TFS	849.30	482.01	153.81	64.43	35.43			20.35	21.09	9.80	10.54
	Is Charge			UNCSX	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.54
2-WIF	RE ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPOR	RT (EEL)		0.1000		02.10	2 1102	0.12	0.12			20.00	21100	0.00	10.01
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1		1	UNCNX	U1L2X	22.22	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 2		2	UNCNX	U1L2X	29.02	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination															
	Transport - Zone 3		3	UNCNX	U1L2X	37.95	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Interoffice Transport - Dedicated - DS1 combintion - Facility			UNC1X	1L5XX	0.3562										
	Termination per month			UNC1X	U1TF1	77.86	171.24	113.12	70.07	30.90			20.35	21.09	9.80	10.54
	Channelization - Channel System DS1 to DS0 combination - per month			UNC1X	MQ1	80.77	105.76	14.48	3.04	2.74			20.35	21.09	9.80	10.54
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per month			UNCNX	UC1CA	3.24	5.70	4.42					20.35	21.09	9.80	10.54
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1		1	UNCNX	U1L2X	22.22	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
_	Additional 2-wire ISDN Loop in same DS1Interoffice Transport															
_	Combination - Zone 2 Additional 2-wire ISDN Loop in same DS1Interoffice Transport		2	UNCNX	U1L2X	29.02	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Combination - Zone 3		3	UNCNX	U1L2X	37.95	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combintaion- per month			UNCNX	UC1CA	3.24	5.70	4.42					20.35	21.09	9.80	10.54
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNC1X	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.54
4-WIE	RE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 IN	TEROE	FICE T				52.75	24.02	5.12	9.12			20.33	21.09	9.00	10.34
	First DS1 Loop in STS1 Interoffice Transport Combination -															
_	Zone 1 First DS1 Loop in STS1 Interoffice Transport Combination -		1	UNC1X	USLXX	57.73	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.54
_	Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination -		2	UNC1X	USLXX	75.40	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.54
	Zone 3		3	UNC1X	USLXX	98.59	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.54
	Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month			UNCSX	1L5XX	2.34										
	Interoffice Transport - Dedicated - STS1 combination - Facility Termination			UNCSX	U1TFS	849.30	482.01	153.81	64.43	35.43			20.35	21.09	9.80	10.54
	STS1 to DS1 Channel System conbination per month			UNCSX	MQ3	222.98	156.02	49.41	17.12	6.77			20.35	21.09	9.80	10.54
_	DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination -			UNC1X	UC1D1	17.58	5.70	4.42					20.35	21.09	9.80	10.54
	Zone 1		1	UNC1X	USLXX	57.73	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.54
	Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 2		2	UNC1X	USLXX	75.40	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.54
	Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 3		3	UNC1X	USLXX	98.59	228.40	161.74	79.87	24.88			20.35	21.09	9.80	10.54
	DS3 Interface Unit (DS1 COCI) combination per month		-	UNC1X	UC1D1	17.58	5.70	4.42		2.100			20.35	21.09	9.80	10.54

UNBUNDL	ED NETWORK ELEMENTS - Tennessee												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
						Rec	Nonrecurring		Nonrecurring					Rates(\$)	_	
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge	-		UNCSX	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10 54
4-WI	RE 56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTERO	FFICE 1	RANS		UNCCC		52.75	24.02	9.12	9.12			20.35	21.09	9.60	10.54
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport															
	Combination - Zone 1		1	UNCDX	UDL56	31.10	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport															
	Combination - Zone 2		2	UNCDX	UDL56	40.61	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport															
	Combination - Zone 3		3	UNCDX	UDL56	53.11	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Per Mile			UNCDX	1L5XX	0.0174										
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination -	1			ILJAA	0.0174					-					
	Facility Termination			UNCDX	U1TD5	21.19	79.83	44.08	69.32	31.00			20.35	21.09	9.80	10.54
	Nonrecurring Currently Combined Network Elements Switch -As-	1			-											
	Is Charge			UNCDX	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.54
4-WI	RE 64 KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTERO	FFICE 1	RANS	PORT (EEL)												
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport		1			04.40	100 70	05.47	70.04	10.00			00.05	04.00	0.00	10.54
	Combination - Zone 1 4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport		1	UNCDX	UDL64	31.10	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Combination - Zone 2		2	UNCDX	UDL64	40.61	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport		2	UNCDA	UDL04	40.01	100.70	33.47	72.94	10.00			20.33	21.09	9.00	10.34
	Combination - Zone 3		3	UNCDX	UDL64	53.11	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination -															
	Per Mile			UNCDX	1L5XX	0.0174										
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination -															
	Facility Termination			UNCDX	U1TD6	21.19	79.83	44.08	69.32	31.00			20.35	21.09	9.80	10.54
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge	-		UNCDX	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.54
	NETWORK ELEMENTS			UNCDA	UNCCC		52.75	24.02	9.12	9.12			20.35	21.09	9.60	10.54
	n used as a part of a currently combined facility, the non-recur	rng cha	raes da	not apply, but a S	witch As Is c	harge does ap	olv.									
	n used as ordinarily combined network elements in All States, t															
Nonr	ecurring Currently Combined Network Elements "Switch As Is"		(One a	pplies to each com	bination)											
	Nonrecurring Currently Combined Network Elements Switch -As-	-														
	Is Charge - 2 wire/4-Wire VG			UNCVX	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.54
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge - 56/64 kbps	-		UNCDX	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.54
	Nonrecurring Currently Combined Network Elements Switch -As-				UNCCC		52.13	24.02	9.12	9.12			20.35	21.09	9.00	10.54
	Is Charge - DS1	1		UNC1X	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.54
	Nonrecurring Currently Combined Network Elements Switch -As-	-									İ					
	Is Charge - DS3			UNC3X	UNCCC		52.73	24.62	9.12	9.12			20.35	21.09	9.80	10.54
	Nonrecurring Currently Combined Network Elements Switch -As-	_														10
Not	Is Charge - STS1	d D-1-		UNCSX	UNCCC	r month -	52.73	24.62	9.12	9.12	L		20.35	21.09	9.80	10.54
NOT	E: Local Channel - Dedicated Transport - minimum billing perio Local Channel - Dedicated - 2-Wire Voice Grade Zone 1	a - Belo		eone month, DS3 a	ULDV2	r months 17.18	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Local Channel - Dedicated - 2-Wire Voice Grade Zone 1	1	2	UNCVX	ULDV2	22.44	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Local Channel - Dedicated - 2-Wire Voice Grade Zone 2	1	3	UNCXV	ULDV2	29.34	108.76	35.47	72.94	10.86	1		20.35	21.09	9.80	10.54
	Local Channel - Dedicated - 4-Wire Voice Grade Zone 1	1	1	UNCVX	ULDV4	18.18	108.76	35.47	72.94	10.86	İ		20.35	21.09	9.80	10.54
	Local Channel - Dedicated - 4-Wire Voice Grade Zone 2		2	UNCVX	ULDV4	23.74	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Local Channel - Dedicated - 4-Wire Voice Grade Zone 3		3	UNCXV	ULDV4	31.05	108.76	35.47	72.94	10.86			20.35	21.09	9.80	10.54
	Local Channel - Dedicated - DS1 per month Zone 1		1	UNC1X	ULDF1	36.24	228.40	161.74	79.87	24.88	L		20.35	21.09	9.80	10.54
	Local Channel - Dedicated -DS1 Per Month Zone 2 Local Channel - Dedicated - DS1- Per Month Zone 3		2	UNC1X UNC1X	ULDF1 ULDF1	47.33 61.89	228.40 228.40	161.74 161.74	79.87 79.87	24.88 24.88			20.35 20.35	21.09 21.09	9.80 9.80	10.54 10.54
	Local Channel - Dedicated - DS1- Per Month Zone 3		ى ا	UNC3X	1L5NC	7.15	220.40	101.74	19.67	24.08	<u> </u>		20.35	21.09	9.00	10.54
	Local Channel - Dedicated - DS3 - Facility Termination	1		UNC3X	ULDF3	611.30	595.37	304.50	215.82	151.15	<u> </u>		20.35	21.09	9.80	10.54
	Local Channel - Dedicated - STS-1- Per Mile per month			UNCSX	1L5NC	7.15										
	Local Channel - Dedicated - STS-1 - Facility Termination			UNCSX	ULDFS	599.59	588.07	297.20	215.82	151.15			20.35	21.09	9.80	10.54
MUL	TIPLEXERS															
	Channelization - DS1 to DS0 Channel System			UXTD1	MQ1	80.77	141.67	77.11	14.51	13.46			20.35	9.80	11.49	1.18

UNBUNDLF	ED NETWORK ELEMENTS - Tennessee												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
						Rec	Nonrecurring		Nonrecurring					Rates(\$)	-	
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per			UDL	1D1DD	1.82	6.07	4.66					20.35	9.80	11.49	4.40
	month (2.4-64kbs) 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per			UDL	סטוטו	1.82	6.07	4.00					20.35	9.80	11.49	1.18
	month			UDN	UC1CA	3.10	6.07	4.66					20.35	9.80	11.49	1.18
	Voice Grade COCI - DS1 to DS0 Channel System - per month			UEA	1D1VG	0.91	6.07	4.66					20.35	9.80	11.49	1.18
	DS3 to DS1 Channel System per month			UXTD3	MQ3	222.98	308.03	108.47	44.47	42.62			20.35	9.80	11.49	1.18
	STS1 to DS1 Channel System per month			UXTS1	MQ3	222.98	308.03	108.47	44.47	42.62			20.35	21.09	9.80	9.80
	DS3 Interface Unit (DS1 COCI) used with Loop per month			USL	UC1D1	17.58	6.07	4.66					20.35	9.80	11.49	1.18
	DS3 Interface Unit (DS1 COCI) used with Local Channel per				10101		0.07	1.00					00.05	0.00	44.40	
Sub-I	month oop Feeder			ULDD1	UC1D1		6.07	4.66					20.35	9.80	11.49	1.18
Sub-L	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Statewide		SW	UNC1X	USBFG											ł
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Statewide		1	UNC1X	USBFG	39.74	116.00	40.62	106.82	18.91	1					
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 2		2	UNC1X	USBFG	51.90	116.00	40.62	106.82	18.91	1			İ		<u> </u>
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 3		3	UNC1X	USBFG	67.86	116.00	40.62	106.82	18.91						
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 4		4	UNC1X	USBFG											
	LOCAL EXCHANGE SWITCHING(PORTS)															L
	inge Ports															
	: Although the Port Rate includes all available features in GA, I E VOICE GRADE LINE PORT RATES (RES)	KY, LA	54. IN, t	he desired features	will need to b	be ordered usin	ng retail USOCs									┣────
2-WIR	Exchange Ports - 2-Wire Analog Line Port- Res.			UEPSR	UEPRL	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Exchange Folts - 2-Wile Analog Line Folt- Res.			ULFOR	ULFKL	1.09	9.93	9.19	3.00	2.52			20.33	10.54	13.32	1.40
	Exchange Ports - 2-Wire Analog Line Port with Caller ID - Res.			UEPSR	UEPRC	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Res. Exchange Ports - 2-Wire VG unbundled TN extended local			UEPSR	UEPRO	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	dialing parity Port with Caller ID - Res. Exchange Ports - 2-Wire VG unbundled Tennessee Area Plus			UEPSR	UEPAQ	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	with Caller ID - Res (AC7) Exchange Ports - 2-Wire VG unbundled Tennessee Area Calling			UEPSR	UEPAH	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	port with Caller ID - Res (F2R) Exchange Ports - 2-Wire VG unbundled Tennessee Area Calling			UEPSR	UEPAK	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	port with Caller ID - Res (TACER) Exchange Ports - 2-Wire VG unbundled Tennessee Area Calling			UEPSR	UEPAL	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	port with Caller ID - Res (TACSR) Exchange Ports - 2-Wire VG unbundled Tennessee Area Calling			UEPSR	UEPAM	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	port with Caller ID - Res (1MF2X) Exchange Ports - 2-Wire VG unbundled Tennessee Area Calling			UEPSR	UEPAN	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	port with Caller ID - Res (2MR) Exchange Ports - 2-Wire VG unbundled res, low usage line port			UEPSR	UEPAO	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	with Caller ID (LUM) Exchange Port - 2-Wire VG Tennessee Residence Dialing Plan			UEPSR	UEPAP	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	without Caller ID Exchange Port - 2-Wire VG Tennessee Residence Dialing Flat Exchange Port - 2-Wire VG Tennessee Residence Area Plus			UEPSR	UEPWN	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	without Caller ID			UEPSR	UEPRR	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire voice unbundled Low Usage Line Port without Caller ID Capability			UEPSR	UEPRT	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
FEAT	Subsequent Activity			UEPSR	USASC	0.00	0.00	0.00				ļ	20.35	10.54	13.32	1.40
FEAT	All Available Vertical Features			UEPSR	UEPVF	0.00	0.00	0.00					20.35	10.54	13.32	1.40
2-WIR	E VOICE GRADE LINE PORT RATES (BUS) Exchange Ports - 2-Wire Analog Line Port without Caller ID -					0.00	0.00	0.00					20.35	10.54	13.32	1.40
	Bus Exchange Ports - 2-Wire VG unbundled Line Port with			UEPSB	UEPBL	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	unbundled port with Caller+E484 ID - Bus.			UEPSB	UEPBC	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Bus.			UEPSB	UEPBO	1.89	9.93	9.19	3.66	2.92	ļ		20.35	10.54	13.32	1.40
	Exchange Ports - 2-Wire VG unbundled TN extended local dialing parity Port with Caller ID - Bus.			UEPSB	UEPAV	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40

UNBUNDL	ED NETWORK ELEMENTS - Tennessee												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrecurring		Nonrecurring	g Disconnect				Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Exhange Ports - 2-Wire VG unbundled incoming only port with Caller ID - Bus			UEPSB	UEPB1	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Exchange Ports - 2-Wire VG unbundled TN Bus 2-Way Area			UEFSB	UEPBI	1.69	9.93	9.19	3.00	2.92			20.35	10.54	13.32	1.40
	Calling Port Economy Option - Bus (TACC1)			UEPSB	UEPAC	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Exchange Ports - 2-Wire VG unbundled TN Bus 2-Way Area Calling Port Standard Option - Bus (TACC2)			UEPSB	UEPAD	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Exchange Ports - 2-W VG unbundled TN Bus 2-Way Collierville				UEPAE	1.89	9.93	9.19	2.00	2.92			20.35	10.54	40.00	1.40
	& Memphis Local Calling Port - Bus (B2F) Exchange Ports - 2-W VG unbundled TN Bus 2-Way Collierville			UEPSB	UEPAE	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	& Memphis Local Calling Port			UEPSB	UEPB2	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Exchange Ports - 2-W VG unbundled TN, Business Line Inward,															
	Collierville & Memphis Local Calling Plan			UEPSB	UEPB3	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Exchange Ports - 2-Wire Voice Tennessee Business Dialing														10.00	
	Plan without Caller ID			UEPSB	UEPWO	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire voice unbundled Incoming Only Port without Caller ID Capability			UEPSB	UEPBE	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Subsequent Activity			UEPSB	USASC	0.00	0.00	0.00	0.00	2.02			20.35	10.54	13.32	1.40
FEAT	URES															
	All Available Vertical Features			UEPSB	UEPVF	0.00	0.00	0.00					20.35	10.54	13.32	1.40
EXCH	IANGE PORT RATES (DID & PBX)															
	2-Wire VG Unbundled 2-Way PBX Trunk - Res			UEPSE	UEPRD	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus			UEPSP	UEPPC	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus			UEPSP	UEPPO	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus			UEPSP	UEPP1	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire Analog Long Distance Terminal PBX Trunk - Bus			UEPSP	UEPLD	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire Analog TN 2-Way Calling Plan PBX Trunk - Bus			UEPSP	UEPT2	1.79	9.93	9.19		2.92			20.35	10.54	13.32	1.40
	2-Wire TN Outward Calling Plan PBX Trunk - Bus			UEPSP	UEPTO	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPSP	UEPLD	1.79	9.93	9.19		2.92			20.35	10.54	13.32	1.40
	2-Wire Voice Unbundled 2-Way PBX Tennessee Calling Port			UEPSP	UEPT2	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire Voice Unbundled 1-Way Outgoing PBX Tennessee															
	Calling Port			UEPSP	UEPTO	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire Vice Unbundled 2-Way PBX Usage Port			UEPSP	UEPXA	1.79	9.93	9.19		2.92			20.35	10.54	13.32	1.40
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPSP	UEPXB	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPSP	UEPXC	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPSP	UEPXD	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD					4 70	0.02	0.10	2.00	0.00			20.25	10.54	40.00	4.40
	Capable Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy		-	UEPSP	UEPXE	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Administrative Calling Port			UEPSP	UEPXL	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy					1.75	3.35	3.13	5.00	2.52			20.00	10.04	10.02	1.40
	Room Calling Port			UEPSP	UEPXM	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-W Voice Unbundled 1-Way Out PBX Hotel/Hospital Economy															
	Administrative Calling Port TN Calling Port			UEPSP	UEPXN	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital		1		UEPXO	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Discount Room Calling Port Unbundled Exchange Ports, PBX Trunk Combination,			UEPSP	JEPXU	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Collierville and Memphis Local Calling Plan			UEPSP	UEPA6	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Unbundled Exchange Ports, PBX Trunk Combination, first trunk,		1			1 70	0.00	0.75	0.00	0.00			00.07	10 5 1	10.00	
	Collierville and Memphis Local Calling Plan		<u> </u>	UEPSP	UEPA7	1.79	9.93	9.19		2.92	ļ		20.35	10.54	13.32	1.40
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port 2-Wire Voice Unbundled PBX Collierville and Memphis Calling			UEPSP	UEPXS	1.79	9.93	9.19	3.66	2.92	-		20.35	10.54	13.32	1.40
	2-Wire Voice Unbundled PBX Collierville and Memphis Calling Port		1	UEPSP	UEPXU	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	2-Wire Voice Unbundled 2-Way PBX Tennessee RegionServ		1				0.00	0.10	0.00	2.02	1		20.00			
	Calling Port		1	UEPSP	UEPXV	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Subsequent Activity			UEPSP	USASC	0.00	0.00	0.00					20.35	10.54	13.32	1.40
FEAT	URES															
	All Available Vertical Features			UEPSP UEPSE	UEPVF	0.00	0.00	0.00					20.35	10.54	13.32	1.40
EXCH	IANGE PORT RATES (COIN)															
	Exchange Ports - Coin Port					2.11	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40

UNBUNDL	ED NETWORK ELEMENTS - Tennessee												Attachment:	2	Exhi	ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
						Rec	Nonrecurring	Add'l		Disconnect	0.01150	001111		Rates(\$)	0.014.01	SOMAN
NOT	E: Transmission/usage charges associated with POTS circuit s	witched	116300	will also apply to c	sircuit switch	ed voice and/or	First		First	Add'l		SOMAN		SOMAN	SOMAN	SOMAN
NOT	E: Access to B Channel or D Channel Packet capabilities will be	e availal	ble only	v through BFR/New	/ Business Re	equest Process	. Rates for the	packet capabi	lities will be de	etermined via t	he Bona Fic	de Request/	New Business	s Request Pro	cess.	
UNBUNDLED	D LOCAL EXCHANGE SWITCHING(PORTS)					1										
EXCI	IANGE PORT RATES															
	Exchange Ports - 2-Wire DID Port			UEPEX	UEPP2	8.97	47.75	47.01	9.21	8.47			20.35	10.54	13.32	1.40
	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability			UEPDD	UEPDD	35.74	75.93	38.15	8.77	8.04			20.35	10.54	13.32	1.40
	Exchange Ports - 2-Wire ISDN Port (See Notes below.)			UEPTX UEPSX	U1PMA	16.26	30.23	29.49	4.10	4.10			20.35	10.54	13.32	1.40
NOT	E: Transmission/usage charges associated with POTS circuit s	witched	usage									wire ISDN p		10101	10.02	
	E: Access to B Channel or D Channel Packet capabilities will be													s Request Pro	cess.	
	Exchange Ports - 2-Wire ISDN Port Channel Profiles			UEPTX UEPSX	U1UMA	0.00	0.00	0.00								
	Exchange Ports - 4-Wire ISDN DS1 Port			UEPEX	UEPEX	75.04	148.66	147.18	38.46	36.98			20.35	10.54	13.32	1.40
	UNDLED PORT with REMOTE CALL FORWARDING CAPABILITY															
UNB	UNDLED REMOTE CALL FORWARDING SERVICE - RESIDENCE															
	Unbundled Remote Call Forwarding Service, Area Calling, Res	I		UEPVR	UERAC	1.89	9.93	9.19	3.66	2.92	ļ		20.35	10.54	13.32	1.40
	Halour diad Demote Cell Ferries Pro October Level Co					1.00	0.00	0.10	0.00	0.00			00.07	10.51	10.00	
	Unbundled Remote Call Forwarding Service, Local Calling - Res Unbundled Remote Call Forwarding Service, InterLATA - Res			UEPVR UEPVR	UERLC	1.89 1.89	9.93 9.93	9.19 9.19	3.66 3.66	2.92 2.92			20.35 20.35	10.54 10.54	13.32 13.32	1.40 1.40
	Unbundled Remote Call Forwarding Service, InterLATA - Res			UEPVR	UERTR	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
Non-	Recurring				OLKIK	1.09	9.93	5.15	3.00	2.52			20.33	10.34	13.32	1.40
Non	Unbundled Remote Call Forwarding Service - Conversion -														<u> </u>	
	Switch-as-is			UEPVR	USAC2		1.03	0.29					20.35	10.54	13.32	1.40
	Unbundled Remote Call Forwarding Service - Conversion with															
	allowed change (PIC and LPIC)			UEPVR	USACC		1.03	0.29								
UNB	UNDLED REMOTE CALL FORWARDING - Bus															
	Unbundled Remote Call Forwarding Service, Area Calling - Bus			UEPVB	UERAC	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Unbundled Remote Call Forwarding Service, Local Calling - Bus			UEPVB	UERLC	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Unbundled Remote Call Forwarding Service, Local Calling - Bus			UEPVB	UERTE	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Unbundled Remote Call Forwarding Service, IntelEATA - Bus			UEPVB	UERTR	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
	Unbundled Remote Call Forwarding Service Expanded and			02.10	0LILII		0.00	0.10	0.00	2.02			20100	10101	10.02	
	Exception Local Calling			UEPVB	UERVJ	1.89	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.40
Non-	Recurring															
	Unbundled Remote Call Forwarding Service - Conversion -															
	Switch-as-is	I		UEPVB	USAC2		1.03	0.29					20.35	10.54	13.32	1.40
	Unbundled Remote Call Forwarding Service - Conversion with				110400		4.00	0.00							1	
	allowed change (PIC and LPIC) DLOCAL SWITCHING, PORT USAGE	l		UEPVB	USACC		1.03	0.29			ł				 	
	Office Switching (Port Usage)				+	<u> </u>			<u> </u>						 	
Ena	End Office Switching Function, Per MOU	<u> </u>				0.0008041			1						<u> </u>	╂────┦
Tand	em Switching (Port Usage) (Local or Access Tandem)				1	0.0000041			ł	1						1
	Tandem Switching Function Per MOU	1				0.0009778	İ		1		1		İ	İ	<u> </u>	<u> </u> ────
Com	mon Transport	1														
	Common Transport - Per Mile, Per MOU					0.0000064										
	Common Transport - Facilities Termination Per MOU					0.0003871										
	PORT/LOOP COMBINATIONS - COST BASED RATES	Ļ	I <u>. </u>		1	<u> </u>									 	<u> </u>
	Based Rates are applied where BellSouth is required by FCC at									af this Date -					 	
	ures shall apply to the Unbundled Port/Loop Combination - Cos Office and Tandem Switching Usage and Common Transport Us											n Bert/l	Combinet	I	 	┨────┤
	Office and Tandem Switching Usage and Common Transport Us first and additional Port nonrecurring charges apply to Not Curr														ł	┨────┤
	RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)							y charges Slid			Chrecuning	- Surrently	Sombilied St		<u> </u>	<u>├</u>
	Port/Loop Combination Rates										1				<u> </u>	1
	2-Wire VG Loop/Port Combo - Zone 1	1	1			14.18			1				İ	İ	1	1
	2-Wire VG Loop/Port Combo - Zone 2	1	2			18.01										
	2-Wire VG Loop/Port Combo - Zone 3		3			23.02										
UNE	Loop Rates															
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPRX	UEPLX	12.48									<u> </u>	\square
	2-Wire Voice Grade Loop (SL1) - Zone 2	1	2	UEPRX	UEPLX	16.31									L	

UNBUNDLE	D NETWORK ELEMENTS - Tennessee												Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPRX	UEPLX	21.32	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
2-Wire	Voice Grade Line Port Rates (Res)		3	UEFRA	UEFLA	21.32										<u> </u>
2-1110	2-Wire voice unbundled port - residence			UEPRX	UEPRL	1.70	22.14	15.25	8.45	3.91		15.69				<u> </u>
	2-Wire voice unbundled port with Caller ID - res			UEPRX	UEPRC	1.70	22.14	15.25	8.45	3.91		15.69				1
	2-Wire voice unbundled port outgoing only - res			UEPRX	UEPRO	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire voice Grade unbundled Tennessee extended local dialing parity port with Caller ID - res			UEPRX	UEPAQ	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire voice unbundled Tennessee Area Plus with Caller ID - res (AC7)			UEPRX	UEPAH	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (F2R)			UEPRX	UEPAK	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACER)			UEPRX	UEPAL	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACSR)			UEPRX	UEPAM	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (1MF2X)			UEPRX	UEPAN	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (2MR)			UEPRX	UEPAO	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire voice unbundles res, low usage line port with Caller ID (LUM)			UEPRX	UEPAP	1.70	22.14	15.25	8.45	3.91		15.69				L
	2-Wire Voice Unbundled Tennessee Residence Dialing Plan without Caller ID 2-Wire voice unbundled Tennessee Area Plus Port without			UEPRX	UEPWN	1.70	22.14	15.25	8.45	3.91		15.69				
	Caller ID Capability 2-Wire voice unbundled Low Usage Line Port without Caller ID			UEPRX	UEPRR	1.70	22.14	15.25	8.45	3.91		15.69				
FEATU	Capability			UEPRX	UEPRT	1.70	22.14	15.25	8.45	3.91		15.69				
	All Features Offered			UEPRX	UEPVF	0.00	0.00	0.00				15.69				
LOCA	NUMBER PORTABILITY															l
NOND	Local Number Portability (1 per port)			UEPRX	LNPCX	0.35										l
NONR	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is			UEPRX	USAC2		1.03	0.29				15.69				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with change			UEPRX	USACC		1.03	0.29				15.69				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update						0.76					15.69				
ADDIT	IONAL NRCs															ļ
2 WID	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)			UEPRX	USAS2	0.00	0.00	0.00				15.69				
	ort/Loop Combination Rates				+							-				
	2-Wire VG Loop/Port Combo - Zone 1		1		1 1	14.18										i
	2-Wire VG Loop/Port Combo - Zone 2		2			18.01			İ							
	2-Wire VG Loop/Port Combo - Zone 3		3			23.02										
UNE L	oop Rates					10.10										l
├ ── ├ ──	2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2		1	UEPBX UEPBX	UEPLX UEPLX	12.48 16.31										
	2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3		2	UEPBX	UEPLX	21.32										
2-Wire	Voice Grade Line Port (Bus)		5			21.32						-				
	2-Wire voice unbundled port without Caller ID - bus			UEPBX	UEPBL	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPBX	UEPBC	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire voice unbundled port outgoing only - bus			UEPBX	UEPBO	1.70	22.14	15.25	8.45	3.91		15.69				ļ
	2-Wire voice Grade unbundled Tennessee extended local dialing parity port with Caller ID - bus 2 Wire voice unbundled incoming only port with Caller ID. Bus					1.70	22.14	15.25	8.45	3.91		15.69				
┝──-	2-Wire voice unbundled incoming only port with Caller ID - Bus 2-Wire voice unbundled Tennessee Bus 2-Way Area Calling			UEPBX	UPEB1	1.70	22.14	15.25	8.45	3.91		15.69				
	Port Economy Option (TACC1)			UEPBX	UEPAC	1.70	22.14	15.25	8.45	3.91		15.69				<u> </u>

UNB	JNDLE	D NETWORK ELEMENTS - Tennessee												Attachment:	2	Exhi	bit: B
														Incremental	Incremental		
													Submitted	Charge -	Charge -	Charge -	Charge -
CATE	OPV	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			Elec	Manually	Manual Svc	Manual Svc		
CATE	JONT		m	20116	603	0300			KATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
														Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
-	1						B	Nonrecurring		Nonrecurring	Disconnect			OSS	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		2-Wire voice unbundled Tennessee Bus 2-Way Area Calling															
		Port Standard Option (TACC2)			UEPBX	UEPAD	1.70	22.14	15.25	8.45	3.91		15.69				
		2-Wire voice unbundled Tennessee Bus 2-Way Collierville and Memphis Local Calling Port (B2F)			UEPBX	UEPAE	1.70	22.14	15.25	8.45	3.91		15.69				
		2-Wire Voice Unbundled Tennessee Business Dialing Plan without Caller ID			UEPBX	UEPWO	1.70	22.14	15.25	8.45	3.91		15.69				
		Tennessee Inward Collierville and Memphis Local Calling Plan															
		(BUS)			UEPBX	UEPB2	1.70	22.14	15.25	8.45	3.91		15.69				L
		Tennessee 2-Way Collierville and Memphis Local Calling Plan (BUS)			UEPBX	UEPB3	1.70	22.14	15.25	8.45	3.91		15.69				
		2-Wire voice unbundled Incoming Only Port without Caller ID															
	1.004	Capability NUMBER PORTABILITY	<u> </u>		UEPBX	UEPBE	1.70	22.14	15.25	8.45	3.91	<u> </u>	15.69				
	LOCAL	Local Number Portability (1 per port)			UEPBX	LNPCX	0.35										├ ────
	FEATU		1		OLFDA		0.35					<u> </u>	<u> </u>	-	ł	ł	ł
		All Features Offered			UEPBX	UEPVF	0.00	0.00	0.00				15.69				<u> </u>
-		ECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
		2-Wire Voice Grade Loop / Line Port Combination - Conversion -															
		Switch-as-is			UEPBX	USAC2		1.03	0.29				15.69				
		2-Wire Voice Grade Loop / Line Port Combination - Conversion -															
		Switch with change			UEPBX	USACC		1.03	0.29				15.69				
		2-Wire Voice Grade Loop / Line Port Combination - Conversion -						0.70					15.00				
		Subsequent Database Update ONAL NRCs						0.76					15.69				├ ────
	ADDITI	2-Wire Voice Grade Loop/Line Port Combination - Subsequent															<u> </u>
		Activity			UEPBX	USAS2	0.00	0.00	0.00				15.69				
	2-WIRE	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)															
	UNE Po	ort/Loop Combination Rates															
		2-Wire VG Loop/Port Combo - Zone 1		1			14.18										
		2-Wire VG Loop/Port Combo - Zone 2		2			18.01										
		2-Wire VG Loop/Port Combo - Zone 3		3			23.02						-				───
	-	2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2		1	UEPRG UEPRG	UEPLX UEPLX	12.48 16.31										<u> </u>
		2-Wire Voice Grade Loop (SL 1) - Zone 2		3	UEPRG	UEPLX	21.32					-					<u> </u>
	2-Wire	Voice Grade Line Port Rates (RES - PBX)		5	OEI IKO	OLIEX	21.02										<u> </u>
		2-Wire VG Unbundled Combination 2-Way PBX Trunk Port -															
		Res			UEPRG	UEPRD	1.70	22.14	15.25	8.45	3.91		15.69				
	LOCAL	NUMBER PORTABILITY															
		Local Number Portability (1 per port)			UEPRG	LNPCP	3.15	0.00	0.00				15.69				<u> </u>
<u> </u>	FEATU		<u> </u>				0.00					<u> </u>	45.00				
<u> </u>	NONE	All Features Offered CURRING CHARGES (NRCs) - CURRENTLY COMBINED			UEPRG	UEPVF	0.00	0.00	0.00		1	<u> </u>	15.69				<u> </u>
<u> </u>	NONRE	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -										<u> </u>	<u> </u>				<u> </u>
		Conversion - Switch-As-Is			UEPRG	USAC2		1.03	0.29				15.69				1
	1	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -	1		-					1							<u> </u>
		Conversion - Switch with Change			UEPRG	USACC		1.03	0.29				15.69				L
		2-Wire Voice Grade Loop / Line Port Combination - Conversion -															
		Subsequent Database Update						0.76					15.69				
	ADDITI	ONAL NRCs		<u> </u>		4											ł
		2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activity			UEPRG	USAS2	0.00	0.00	0.00				15.69				1
<u> </u>		PBX Subsequent Activity - Change/Rearrange Multiline Hunt				00402	0.00	0.00	0.00				10.09				ł
1	1	Group						14.64	14.64				15.69				1
	2-WIRE	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)															l
		ort/Loop Combination Rates															
		2-Wire VG Loop/Port Combo - Zone 1		1			14.18										
L	<u> </u>	2-Wire VG Loop/Port Combo - Zone 2		2			18.01					L	L				
 		2-Wire VG Loop/Port Combo - Zone 3		3			23.02					-					
L	UNE LO	pop Rates	I		1	1				1		L	L		1	1	L

UNBUNDLE	ED NETWORK ELEMENTS - Tennessee												Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'l
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEPPX UEPPX	UEPLX UEPLX	12.48 16.31										
	2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEPPX	UEPLX	21.32										
2-Wir	e Voice Grade Line Port Rates (BUS - PBX)		3	ULFFX	ULFLA	21.32										
2 000			1													
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus	5		UEPPX	UEPPC	1.70	22.14	15.25	8.45	3.91		15.69				
	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPPX	UEPPO	1.70	22.14	15.25	8.45	3.91		15.69				
	Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPPX	UEPP1	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPPX	UEPLD	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Voice Unbundled 2-Way Combination PBX Tennessee															
	Calling Port			UEPPX	UEPT2	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Tennessee		1													
	Calling Port	+	<u> </u>	UEPPX	UEPTO	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port	_		UEPPX	UEPXA	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports	+		UEPPX	UEPXB UEPXC	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Voice Unbundled PBX LD DDD Terminals Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPPX UEPPX	UEPXC	<u>1.70</u> 1.70	22.14 22.14	15.25	8.45 8.45	3.91 3.91		15.69 15.69				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD			ULFFA	UEFAD	1.70	22.14	15.25	8.45	3.91		69.01				
	Capable Port			UEPPX	UEPXE	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy					1.70	22.14	10.20	0.45	5.31		15.05				
	Administrative Calling Port			UEPPX	UEPXL	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy		1	OLITX	OLI XL	1.70	22.14	10.20	0.40	5.31		10.00				
	Room Calling Port			UEPPX	UEPXM	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Voice Unbundled 1W Out PBX Hotel/Hospital Economy															
	Administrative Calling Port TN Calling Port			UEPPX	UEPXN	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital															
	Discount Room Calling Port			UEPPX	UEPXO	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPPX	UEPXS	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Voice Unbundled PBX Collierville and Memphis Calling															
	Port			UEPPX	UEPXU	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Voice Unbundled 2-Way PBX Tennessee RegionServ															
	Callling Port			UEPPX	UEPXV	1.70	22.14	15.25	8.45	3.91		15.69				
	Tennessee PBX 2-Way Combo Each Additional Trunk					. =0						15.00				
	Collierville and Memphis Local Calling Plan		-	UEPPX	UEPA6	1.70	22.14	15.25	8.45	3.91		15.69				
	Tennessee PBX 2-Way Combo First Trunk Collierville and					4 70	22.44	45.05	0.45	2.04		45.00				
1.004	Memphis Local Calling Plan			UEPPX	UEPA7	1.70	22.14	15.25	8.45	3.91		15.69				
LUCA	Local Number Portability (1 per port)			UEPPX	LNPCP	3.15	0.00	0.00				15.69				
FFAT	URES	+				5.15	0.00	0.00				13.09				
	All Features Offered	+	1	UEPPX	UEPVF	0.00	0.00	0.00				15.69				
NONR	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED	1	1			0.00	0.00	0.00	1			10.00			1	
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -	1	1												ĺ	1
	Conversion - Switch-As-Is		1	UEPPX	USAC2		1.03	0.29				15.69				
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -	1	1													
	Conversion - Switch with Change			UEPPX	USACC		1.03	0.29				15.69				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion	-														
	Subsequent Database Update		<u> </u>				0.76					15.69				
ADDI	TIONAL NRCs	_	L													
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -		1		110465											
	Subsequent Activity		<u> </u>	UEPPX	USAS2	0.00	0.00	0.00	<u> </u>	-		15.69				
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt		1				14.04	14.04				15 00				
	Group Port/Loop Combination Rates	+					14.64	14.64				15.69				
	2-Wire VG Coin Port/Loop Combo – Zone 1		1		+	14.18										
	2-Wire VG Coin Por/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2	+	2			14.18	├			-			-		ł	
	2-Wire VG Coin Port/Loop Combo – Zone 3		3			23.02										
UNF I	Loop Rates	1			+ +	20.02									1	
	2-Wire Voice Grade Loop (SL1) - Zone 1	1	1	UEPCO	UEPLX	12.48									1	
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPCO	UEPLX	16.31			1		-					

- DONDEI	ED NETWORK ELEMENTS - Tennessee	1	1		- <u> </u>						Sup Order	Sup Order	Attachment:			bit: B
TEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremen Charge Manual S Order vs Electroni Disc Add
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
	2-Wire Voice Grade Loop (SL1) - Zone 3		2	UEPCO	UEPLX	21.32	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
2-Win	e Voice Grade Line Ports (COIN)		3	ULFCO	ULFLA	21.32										
2-111	2-Wire Coin 2-Way without Operator Screening and without															
	Blocking (TN)			UEPCO	UEPTB	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (NC, TN)			UEPCO	UEPRP	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking (TN)			UEPCO	UEPTA	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Coin 2-Way with Operator Screening: 900 Blocking:															
_	900/976, 1+DDD, 011+, and Local (NC, TN) 2-Wire Coin Outward with Operator Screening and 011 Blocking			UEPCO	UEPCA	1.70	22.14	15.25	8.45	3.91		15.69				
	(TN)			UEPCO	UEPTC	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (TN)			UEPCO	UEPOT	1.70	22.14	15.25	8.45	3.91		15.69				
	2-Wire 2-Way Smartline with 900/976 (all states except LA)			UEPCO	UEPCK	1.88						15.69				
	2-Wire Coin Outward Smartline with 900/976 (all states except LA)			UEPCO	UEPCR	1.88						15.69				
ADD				02.00	02.0.0							10.00				
	UNE Coin Port/Loop Combo Usage (Flat Rate)			UEPCO	URECU	3.45	0.00	0.00				15.69				
	Local Number Portability (1 per port)			UEPCO	LNPCX	0.35										
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -															
_	Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Conversion -			UEPCO	USAC2		1.03	0.29				15.69				
	Switch with change			UEPCO	USACC		1.03	0.29				15.69				
0.14/1	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity			UEPCO	USAS2	0.00	0.00	0.00				15.69				
	RE VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE Port/Loop Combination Rates			KE3)												
UNE	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			18.45										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			23.52										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			30.17										
UNE I	Loop Rates		-													
-	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFR	UECF2	16.56										
	2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFR	UECF2	21.63										
	2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFR	UECF2	28.28										
2-Wir	e Voice Grade Line Port Rates (Res)															
	2-Wire voice unbundled port - residence			UEPFR	UEPRL	1.89	84.99	57.39	32.36	20.56		15.69				
	2-Wire voice unbundled port with Caller ID - res			UEPFR	UEPRC	1.89	84.99	57.39	32.36	20.56		15.69				
	2-Wire voice unbundled port outgoing only - res			UEPFR	UEPRO	1.89	84.99	57.39	32.36	20.56		15.69				
	2-Wire voice Grade unbundled Tennessee extended local dialing parity port with Caller ID - res			UEPFR	UEPAQ	1.89	84.99	57.39	32.36	20.56		15.69				
	2-Wire voice unbundled Tennessee Area Plus with Caller ID - res (AC7)			UEPFR	UEPAH	1.89	84.99	57.39	32.36	20.56		15.69				
	2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (F2R)			UEPFR	UEPAK	1.89	84.99	57.39	32.36	20.56		15.69				
	2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACER)			UEPFR	UEPAL	1.89	84.99	57.39	32.36	20.56		15.69				
	2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACSR)	1	1	UEPFR	UEPAM	1.89	84.99	57.39	32.36	20.56		15.69				
	2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (1MF2X)			UEPFR	UEPAN	1.89	84.99	57.39	32.36	20.56		15.69				
	2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (2MR)			UEPFR	UEPAO	1.89	84.99	57.39	32.36	20.56		15.69				
1	2-Wire voice unbundles res, low usage line port with Caller ID (LUM)			UEPFR	UEPAP	1.89	84.99	57.39	32.36	20.56		15.69				
	2-Wire Voice Unbundled Tennessee Residence Dialing Plan without Caller ID			UEPFR	UEPWN	1.89	84.99	57.39	32.36	20.56		15.69				
	ROFFICE TRANSPORT	I		ULITR	OLEWIN	1.09	04.99	51.59	32.30	20.30		10.09				l

UNBUNDLE	ED NETWORK ELEMENTS - Tennessee										-	-	Attachment:			ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC		_	RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Charge - Manual Sv Order vs.
						Rec	Nonrecurring		Nonrecurring	J Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility															
	Termination			UEPFR	U1TV2	18.58	55.39	17.37	27.96	3.51						<u> </u>
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile			UEPFR	1L5XX	0.0174										
EEAT	or Fraction Mile URES			UEPFR	IL5XX	0.0174										
FEAT	All Features Offered			UEPFR	UEPVF	0.00	0.00	0.00				15.69				<u>+</u>
	L NUMBER PORTABILITY		1	OLITIK	OLI VI	0.00	0.00	0.00				10.03				
	Local Number Portability (1 per port)		1	UEPFR	LNPCX	0.35										
NONR	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED															1
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
	Combination - Conversion - Switch-as-is			UEPFR	USAC2		16.94	3.72				15.69				
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															T
	Combination - Conversion - Switch-With-Change			UEPFR	USACC		16.94	3.72				15.69				
	E VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE	E LINE I	PORT (BUS)												
UNE P	Port/Loop Combination Rates		<u> </u>			10.15										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			18.45										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2 2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		2			23.52 30.17										
	Loop Rates		5			30.17										+
	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFB	UECF2	16.56										
	2-Wire Voice Grade Loop (SL2) - Zone 1		2	UEPFB	UECF2	21.63										
	2-Wire Voice Grade Loop (SL2) - Zone 3			UEPFB	UECF2	28.28										-
2-Wire	e Voice Grade Line Port (Bus)		-													1
	2-Wire voice unbundled port without Caller ID - bus			UEPFB	UEPBL	1.89	84.99	57.39	32.36	20.56		15.69				1
	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPFB	UEPBC	1.89	84.99	57.39	32.36	20.56		15.69				1
	2-Wire voice unbundled port outgoing only - bus			UEPFB	UEPBO	1.89	84.99	57.39	32.36	20.56		15.69				
	2-Wire voice Grade unbundled Tennessee extended local															
	dialing parity port with Caller ID - bus			UEPFB	UEPAV	1.89	84.99	57.39	32.36	20.56		15.69				
	2-Wire voice unbundled incoming only port with Caller ID - Bus			UEPFB	UEPB1	1.89	84.99	57.39	32.36	20.56		15.69				
	2-Wire voice unbundled Tennessee Bus 2-Way Area Calling															
	Port Economy Option (TACC1)			UEPFB	UEPAC	1.89	84.99	57.39	32.36	20.56		15.69				
	2-Wire voice unbundled Tennessee Bus 2-Way Area Calling					1.00	04.00	57.00	22.20	00.50		45.00				
	Port Standard Option (TACC2) 2-Wire voice unbundled Tennessee Bus 2-Way Collierville and			UEPFB	UEPAD	1.89	84.99	57.39	32.36	20.56		15.69				
	Memphis Local Calling Port (B2F)			UEPFB	UEPAE	1.89	84.99	57.39	32.36	20.56		15.69				
	2-Wire Voice Unbundled Tennessee Business Dialing Plan		-	OLFIB	ULFAL	1.09	04.99	57.59	32.30	20.30		15.09				
	without Caller ID			UEPFB	UEPWO	1.89	84.99	57.39	32.36	20.56		15.69				
· · · · ·	Tennessee Inward Collierville and Memphis Local Calling Plan			02.1.5	02.00		0 1100	01100	02.00	20.00		10.00			1	1
	(BUS)	1	1	UEPFB	UEPB2	1.89	84.99	57.39	32.36	20.56		15.69			1	1
	Tennessee 2-Way Collierville and Memphis Local Calling Plan															1
	(BUS)			UEPFB	UEPB3	1.89	84.99	57.39	32.36	20.56		15.69				<u> </u>
LOCA	L NUMBER PORTABILITY															
	Local Number Portability (1 per port)		<u> </u>	UEPFB	LNPCX	0.35									ļ	+
	ROFFICE TRANSPORT		ļ				├								ł	───
. 1	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility Termination		1	UEPFB	U1TV2	18.58	55.39	17.37	27.96	3.51					1	
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile				01172	10.08	55.39	17.37	21.90	3.51					ł	+
	or Fraction Mile	1	1	UEPFB	1L5XX	0.0174									1	1
FEATU			<u> </u>		.20,01	0.0.14									1	+
	All Features Offered			UEPFB	UEPVF	0.00	0.00	0.00				15.69				1
NONR	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port										1					
	Combination - Conversion - Switch-as-is			UEPFB	USAC2		16.94	3.72				15.69				
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port		1													
<u>_</u>	Combination - Conversion - Switch with change	I	L	UEPFB	USACC		16.94	3.72				15.69				
	E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)														ļ	+
	Port/Loop Combination Rates 2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			18.45										───
		1	1	1		18.45					1		1	1	1	1

UNBUNDL	ED NETWORK ELEMENTS - Tennessee											-	Attachment:			ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrecurring		Nonrecurring					Rates(\$)	•	
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3			30.17										
UNE	Loop Rates 2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFP	UECF2	16.56										l
	2-Wire Voice Grade Loop (SL2) - Zone 1 2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFP	UECF2	21.63										<u> </u>
·	2-Wire Voice Grade Loop (SL2) - Zone 2 2-Wire Voice Grade Loop (SL2) - Zone 3			UEPFP	UECF2	21.03										
2-Wir	e Voice Grade Line Port Rates (BUS - PBX)		5	OLITI	02012	20.20										
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPFP	UEPPC	1.79	106.40	63.08	42.67	18.54		15.69				
í – – –	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPFP	UEPPO	1.79	106.40	63.08	42.67	18.54		15.69				
	Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPFP	UEPP1	1.79	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPFP	UEPLD	1.79	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled 2-Way Combination PBX Tennessee Calling Port			UEPFP	UEPT2	1.79	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Tennessee				561 12	1.75	100.40	03.00	-12.07	10.34		15.09				t
1	Calling Port			UEPFP	UEPTO	1.79	106.40	63.08	42.67	18.54		15.69				
i	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPFP	UEPXA	1.79	106.40	63.08	42.67	18.54		15.69				
í – – –	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPFP	UEPXB	1.79	106.40	63.08	42.67	18.54		15.69				
í – – –	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPFP	UEPXC	1.79	106.40	63.08	42.67	18.54		15.69				
í – – – – – – – – – – – – – – – – – – –	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPFP	UEPXD	1.79	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port			UEPFP	UEPXE	1.79	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy				1											
┢───┤───	Administrative Calling Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy			UEPFP	UEPXL	1.79	106.40	63.08	42.67	18.54		15.69				
	Room Calling Port			UEPFP	UEPXM	1.79	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled 1W Out PBX Hotel/Hospital Economy Administrative Calling Port TN Calling Port			UEPFP	UEPXN	1.79	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port			UEPFP	UEPXO	1.79	106.40	63.08	42.67	18.54		15.69				
(<u> </u>	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPFP	UEPXS	1.79	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled PBX Collierville and Memphis Calling				0											
	Port			UEPFP	UEPXU	1.79	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled 2-Way PBX Tennessee RegionServ Callling Port			UEPFP	UEPXV	1.79	106.40	63.08	42.67	18.54		15.69				
LOC/	L NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPFP	LNPCP	3.15	0.00	0.00				15.69				
INTER	ROFFICE TRANSPORT															
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility Termination			UEPFP	U1TV2	18.58	55.39	17.37	27.96	3.51						
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile or Fraction Mile			UEPFP	1L5XX	0.0174										
FEAT	URES			ULFIF	TLJAA	0.0174										
	All Features Offered			UEPFP	UEPVF	0.00	0.00	0.00				15.69				
NONF	RECURRING CHARGES (NRCs) - CURRENTLY COMBINED			02.11	02. 1.	0.00	0.00	0.00				10.00				
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
	Combination - Conversion - Switch-as-is			UEPFP	USAC2		16.94	3.72				15.69				
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port Combination - Conversion - Switch with change			UEPFP	USACC		16.94	3.72				15.69				
UNBUNDLED	PORT/LOOP COMBINATIONS - COST BASED RATES			02.11	00,000		10101	0.72				10.00				
	RE VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK	PORT														
	Port/Loop Combination Rates															
í – – –	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1		1			18.38										
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2		2			19.87										
<u> </u>			3			24.78										
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3		-													
UNE I	Loop Rates															
UNE	Loop Rates 2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1		1	UEPPX	UECD1	9.60										
	Loop Rates		1	UEPPX UEPPX UEPPX	UECD1 UECD1 UECD1 UECD1	9.60 11.09 16.00										

BUNDLEI	D NETWORK ELEMENTS - Tennessee	1	1	r								0	0	Attachment:			ibit: B
EGORY	RATE ELEMENTS	Interi m	Zone	в	cs	USOC			RATES(\$)				Manually	Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
							Rec	Nonrecurring		Nonrecurring					Rates(\$)		
								First	Add'l	First	Add'l		SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Exchange Ports - 2-Wire DID Port			UEPPX		UEPD1	8.78	45.44	29.94	8.45	3.91			30.89	7.03		
NONRE	CURRING CHARGES - CURRENTLY COMBINED																
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination -																
	Switch-as-is			UEPPX		USAC1		8.76	5.75					30.89	7.03		
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion																
	with BellSouth Allowable Changes			UEPPX		USA1C		8.76	5.75					30.89	7.03		
	one Number/Trunk Group Establisment Charges																
	DID Trunk Termination (One Per Port)			UEPPX		NDT	0.00	0.00	0.00								
	Additional DID Numbers for each Group of 20 DID Numbers			UEPPX		ND4	0.00	0.00	0.00								
	DID Numbers, Non- consecutive DID Numbers, Per Number			UEPPX		ND5	0.00	0.00	0.00								
	Reserve Non-Consecutive DID numbers			UEPPX		ND6	0.00	0.00	0.00								
	Reserve DID Numbers			UEPPX		NDV	0.00	0.00	0.00								
	NUMBER PORTABILITY																
	Local Number Portability (1 per port)			UEPPX		LNPCP	3.15	0.00	0.00								
	ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LI	NE SIDE	E PORT	[
	ort/Loop Combination Rates																
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -																
	UNE Zone 1		1	UEPPB	UEPPR		32.27										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -																
	UNE Zone 2		2	UEPPB	UEPPR		34.78										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -																
	UNE Zone 3		3	UEPPB	UEPPR		44.32										
UNE Lo	oop Rates																
	2-Wire ISDN Digital Grade Loop - UNE Zone 1		1	UEPPB	UEPPR	USL2X	16.20										
	-																
	2-Wire ISDN Digital Grade Loop - UNE Zone 2		2	UEPPB	UEPPR	USL2X	18.71										
	2-Wire ISDN Digital Grade Loop - UNE Zone 3		3	UEPPB	UEPPR	USL2X	28.25										
	ort Rate			-	-												
	Exchange Port - 2-Wire ISDN Line Side Port			UEPPB	UEPPR	UEPPB	16.07	141.75	118.37	49.20	43.26			19.99	19.99		
	CURRING CHARGES - CURRENTLY COMBINED			-	-												
	2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port																
	Combination - Conversion			UEPPB	UEPPR	USACB	0.00	117.23	117.23					19.99	19.99		
	ONAL NRCs																
	2-Wire ISDN Loop / 2-Wire ISDN Port Combination - Sub Actvy							1									
	Non Feature/Add Trunk			UEPPB	UEPPR	USASB		212.88						19.99	19.99		
I OCAL	NUMBER PORTABILITY			02.10	02.111	00/102		212.00						10.00	10.00		
	Local Number Portability (1 per port)			UEPPB	UEPPR	LNPCX	0.35	0.00	0.00								
	NNEL USER PROFILE ACCESS:				32 IX		0.00	0.00	0.00			1					1
	CVS/CSD (DMS/5ESS)			UEPPB	UEPPR	U1UCA	0.00	0.00	0.00								
	CVS (EWSD)			UEPPB	UEPPR	U1UCB	0.00	0.00	0.00								
	CSD			UEPPB	UEPPR	U1UCC	0.00	0.00	0.00								
	NNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS S	CMS &	TN)	OLITE	ULITIK	01000	0.00	0.00	0.00								
	CVS/CSD (DMS/5ESS)	c, mo, a	,	UEPPB	UEPPR	U1UCD	0.00	0.00	0.00			1					
	CVS (EWSD)			UEPPB	UEPPR	U1UCE	0.00	0.00	0.00			<u> </u>		ł		ł	
	CSD			UEPPB	UEPPR	U1UCF	0.00	0.00	0.00			<u> </u>		ł		ł	
				JLFPD	ULFFR	OTOUR	0.00	0.00	0.00			<u> </u>		ł		ł	
	User Terminal Profile (EWSD only)			UEPPB	UEPPR	U1UMA	0.00	0.00	0.00			<u> </u>		ł		ł	
	CAL FEATURES			ULFPD	ULFFR	OTOWA	0.00	0.00	0.00		1						
	All Vertical Features - One per Channel B User Profile			UEPPB	UEPPR	UEPVF	0.00	0.00	0.00		1						
	Interoffice Channel mileage each, including first mile and			ULFPD	ULFFR		0.00	0.00	0.00		1						
	facilities termination	1	1		UEPPR	M1GNC	47.04	53.99	17.37					19.99	19.99		1
	Interoffice Channel mileage each, additional mile				UEPPR	M1GNC M1GNM	17.91 0.173	0.00	0.00		1			19.99	19.99		
	Interoffice Channel mileage each, additional mile			UEPPB	UEPPK		0.173	0.00	0.00								
	DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRONP ort/Loop Combination Rates	FURI															
4-WIRE			1	1		1	1					1	1	1		1	I
4-WIRE UNE Po																	
4-WIRE UNE Po	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE						100 50										
4-WIRE UNE Po			1	UEPPP			132.58										

UNBUNDLE	ED NETWORK ELEMENTS - Tennessee												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE					170.11										i
	Zone 3		3	UEPPP		173.44										ł
	oop Rates 4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPPP	USL4P	57.73										l
	4-Wire DS1 Digital Loop - UNE Zone 1	-	2	UEPPP	USL4P USL4P	75.40			<u> </u>							i
	4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPPP	USL4P	98.59			-							
UNE F	Port Rate			0EITT	00L-II	00.00										
	Exchange Ports - 4-Wire ISDN DS1 Port			UEPPP	UEPPP	74.85	415.53	366.90	89.28	77.43			19.99	19.99		
NONR	ECURRING CHARGES - CURRENTLY COMBINED															
	4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port															Í
	Combination - Conversion -Switch-as-is	ļ		UEPPP	USACP	0.00	328.53	328.53			L	L	19.99	19.99		ļ
ADDIT											ļ	ļ				
	4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy-				PR7TF		0.01						40.00	40.00		1
	Inward/two way Tel Nos. (except NC) 4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port -	-		UEPPP	PR/IF		0.94		<u> </u>		-	-	19.99	19.99		
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All States except NC)			UEPPP	PR7TO		22.36	22.36					19.99	19.99		1
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port -			ULFFF	FRITO		22.30	22.30					19.99	19.99		
	Subsequent Inward Tel Numbers			UEPPP	PR7ZT		44.71	44.70					19.99	19.99		i
LOCA	L NUMBER PORTABILITY															1
	Local Number Portability (1 per port)			UEPPP	LNPCN	1.75										
INTER	RFACE (Provsioning Only)															
	Voice/Data			UEPPP	PR71V	0.00	0.00	0.00								1
	Digital Data			UEPPP	PR71D	0.00	0.00	0.00								1
	Inward Data			UEPPP	PR71E	0.00	0.00	0.00								l
New o	r Additional "B" Channel				00701/	0.00	00.00				-	-	40.00	40.00		ł
	New or Additional - Voice/Data B Channel New or Additional - Digital Data B Channel			UEPPP UEPPP	PR7BV PR7BF	0.00	28.39 29.11						19.99 19.99	19.99 19.99		
	New or Additional Inward Data B Channel	-		UEPPP	PR7BD	0.00	29.11		<u> </u>				19.99	19.99		i
CALL	TYPES			OLITI	TRIBD	0.00	23.33		-				13.33	13.33		<u> </u>
0,	Inward			UEPPP	PR7C1	0.00	0.00	0.00				1				i
	Outward			UEPPP	PR7C0	0.00	0.00	0.00								
	Two-way			UEPPP	PR7CC	0.00	0.00	0.00								
Intero	ffice Channel Mileage															
	Fixed Each Including First Mile			UEPPP	1LN1A	76.1825	145.98	109.85	19.55				19.99	19.99		
	Each Airline-Fractional Additional Mile			UEPPP	1LN1B	0.3525										l
	E DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT															
UNE P	Port/Loop Combination Rates		4		_	93.28					-	-	19.99	19.99		ł
<u> </u>	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2		1	UEPDC UEPDC		93.28			┼───┤				19.99	19.99		
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3		3	UEPDC		134.14							19.99	19.99		<u> </u>
UNE I	Loop Rates	1		521 00	1	104.14			<u> </u>		t	1	13.33	10.00		
0	4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPDC	USLDC	57.53			1							
	4-Wire DS1 Digital Loop - UNE Zone 2	l	2	UEPDC	USLDC	75.40	i i				1	1		l	l	
	4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPDC	USLDC	98.59										
UNE F	Port Rate															1
	4-Wire DDITS Digital Trunk Port			UEPDC	UDD1T	35.55	342.80	257.87	61.41	48.49	ļ	ļ	19.99	19.99		
NONR	ECURRING CHARGES - CURRENTLY COMBINED								├ ───┤			L				ł
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Switch-as-is			UEPDC	USAC4		312.91	312.91					19.99	19.99		1
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination			ULFDU	03704	ł	312.91	312.91	<u>∤ </u>		<u> </u>	<u> </u>	19.99	19.99	ł	
	- Conversion with DS1 Changes			UEPDC	USAWA		312.91	312.91					19.99	19.99		1
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination	l				1	512.01	512.01	1 1		1	1	10.00	10.00	1	1
	- Conversion with Change - Trunk			UEPDC	USAWB		312.91	312.91					19.99	19.99		l
ADDIT	TIONAL NRCs			<u> </u>	1											
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent															1
	Service Activity Per Service Order			UEPDC	USAS4		94.88	94.88								
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC -															1
	Subsequent Channel Activation/Chan - 2-Way Trunk			UEPDC	UDTTA	l	108.67	108.67			I	I	19.99	19.99		<u> </u>

UNB	UNDLE	D NETWORK ELEMENTS - Tennessee												Attachment:	2	Exhi	ibit: B
												Svc Order Submitted	Svc Order Submitted		Incremental Charge -		
			Interi	_								Elec	Manually	Manual Svc	Manual Svc		Manual Svc
CATE	GORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
														Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							Rec	Nonrecurring		Nonrecurring	g Disconnect				Rates(\$)		
							Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent			UEPDC	UDTTB		108.67	108.67					19.99	19.99		
		Channel Activation/Chan - 1-Way Outward Trunk 4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel			UEPDC	UDIIB		108.67	108.67					19.99	19.99		<u> </u>
		Activation/Chan Inward Trunk w/out DID			UEPDC	UDTTC		108.67	108.67					19.99	19.99		
		4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation Per Chan - Inward Trunk with DID			UEPDC	UDTTD		108.67	108.67					19.99	19.99		
		4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan						100.07						10.00	10.00		
	BIROI	Activation / Chan - 2-Way DID w User Trans AR 8 ZERO SUBSTITUTION			UEPDC	UDTTE		108.67	108.67					19.99	19.99		┣────
	BIPUL	B8ZS -Superframe Format			UEPDC	CCOSF		0.00	590.00		-			19.99	19.99		
		B8ZS - Extended Superframe Format			UEPDC	CCOEF		0.00	590.00					19.99	19.99		
	Alterna	ate Mark Inversion			02.00			0.00	000.00					10.00	10100		
		AMI -Superframe Format			UEPDC	MCOSF		0.00	0.00								
		AMI - Extended SuperFrame Format			UEPDC	MCOPO		0.00	0.00								
	Teleph	one Number/Trunk Group Establisment Charges															
		Telephone Number for 2-Way Trunk Group			UEPDC	UDTGX	0.00							19.99	19.99		
		Telephone Number for 1-Way Outward Trunk Group			UEPDC	UDTGY	0.00							19.99	19.99		
	_	Telephone Number for 1-Way Inward Trunk Group Without DID DID Numbers for each Group of 20 DID Numbers			UEPDC UEPDC	UDTGZ ND4	0.00							19.99 19.99	19.99 19.99		
	_	DID Numbers, Non- consecutive DID Numbers , Per Number			UEPDC	ND4 ND5	0.00							19.99	19.99		
		Reserve Non-Consecutive DID Nos.			UEPDC	ND6	0.00	0.00	0.00					19.99	19.99		
-		Reserve DID Numbers			UEPDC	NDV	0.00	0.00	0.00								
-	Dedica	ted DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1	Digita	Loop		Trunk Port											
		Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities															
		Termination)			UEPDC	1LNO1	75.83	145.98	109.85	19.66	14.99						
		Interoffice Channel Mileage - Additional rate per mile - 0-8 miles			UEPDC	1LNOA	0.3525	0.00	0.00								
		Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities Termination)			UEPDC	1LNO2	0.00	0.00	0.00								
	_	Interoffice Channel Mileage - Additional rate per mile - 9-25			UEFDC	TLINO2	0.00	0.00	0.00								
		miles			UEPDC	1LNOB	0.3525	0.00	0.00								
		Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities				TENOD	0.0020	0.00	0.00								
		Termination)			UEPDC	1LNO3	0.00	0.00	0.00								-
		Interoffice Channel Mileage - Additional rate per mile - 25+ miles			UEPDC	1LNOC	0.3525	0.00	0.00								
		Local Number Portability, per DS0 Activated			UEPDC	LNPCP	3.15	0.00	0.00								<u> </u>
	1	Central Office Termininating Point			UEPDC	CTG	0.00	0.00	0.00	1		1			İ		1
		DS1 LOOP WITH CHANNELIZATION WITH PORT															
		n is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Act															
i		system can have up to 24 combinations of rates depending on	type ar	nd num	ber of ports used												L
L	UNE D	S1 Loop		-		1101.00	F7 70	0.00	0.00								┣────
<u> </u>		4-Wire DS1 Loop - UNE Zone 1		1	UEPMG UEPMG	USLDC USLDC	57.73 75.40	0.00	0.00				<u>├</u> ───				───
	+	4-Wire DS1 Loop - UNE Zone 2 4-Wire DS1 Loop - UNE Zone 3			UEPMG	USLDC	75.40 98.59	0.00	0.00								<u> </u>
		SO Channelization Capacities (D4 Channel Bank Configuration	ns)	5		30200	30.39	0.00	0.00								ł
<u> </u>	0.12 0	24 DSO Channel Capacity - 1 per DS1			UEPMG	VUM24	131.87	0.00	0.00	1		1	1	19.99	19.99	1	1
	1	48 DSO Channel Capacity - 1 per 2 DS1s			UEPMG	VUM48	263.74	0.00	0.00			1	1	19.99	19.99	l	1
		96 DSO Channel Capacity -1per 4 DS1s			UEPMG	VUM96	527.48	0.00	0.00					19.99	19.99		
		144 DS0 Channel Capacity - 1 per 6 DS1s			UEPMG	VUM14	791.42	0.00	0.00					19.99	19.99		
	1	192 DS0 Channel Capacity -1 per 8 DS1s			UEPMG	VUM19	827.76	0.00	0.00					19.99	19.99		I
		240 DS0 Channel Capacity - 1 per 10 DS1s			UEPMG	VUM20	1,318.70	0.00	0.00				-	19.99	19.99		l
		288 DS0 Channel Capacity - 1 per 12 DS1s			UEPMG UEPMG	VUM28 VUM38	1,582.44	0.00	0.00				<u> </u>	19.99 19.99	19.99 19.99		
	-	384 DS0 Channel Capacity - 1 per 16 DS1s 480 DS0 Channel Capacity - 1 per 20 DS1s			UEPMG	VUM38 VUM40	2,109.92 2,637.40	0.00	0.00					19.99 19.99	19.99		
<u> </u>	+	576 DS0 Channel Capacity - 1 per 20 DS1s			UEPMG	VUM40 VUM57	2,637.40	0.00	0.00		-	+	<u> </u>	19.99	19.99	ł	ł
		672 DS0 Channel Capacity - 1 per 28 DS1s			UEPMG	VUM67	3,104.00	0.00	0.00					19.99	19.99		
	Non-Re	ecurring Charges (NRC) Associated with 4-Wire DS1 Loop with	h Chanı	neliztio					0.00								
		mum System configuration is One (1) DS1, One (1) D4 Channe						-		1		1			İ		1
		es of this configuration functioning as one are considered Ac										1					
_																	

UNBUNDLED N	IETWORK ELEMENTS - Tennessee				-							-	Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						Rec	Nonrecurring		Nonrecurring	g Disconnect				Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	C - Conversion (Currently Combined) with or without															
	ISouth Allowed Changes			UEPMG	USAC4	0.00	303.61	15.74					19.99	19.99		
	ditions at End User Locations Where 4-Wire DS1 Loop with				bination Curre	ently Exists and										
	Currently Combined) in all states, except in Density Zone 1	of I op	8 M S A	's												
	0S1/D4 Channel Bank - Additionally Add NRC for each Port			UEPMG	VUMD4	0.00	704.68	441.48	138.36	16.41			19.99			
	Zero Substitution			UEPING	VUIVID4	0.00	704.68	441.48	138.30	16.41			19.99			
	er Channel Capability Format, superframe - Subsequent				-											
	ivity Only			UEPMG	CCOSF	0.00	0.00	590.00								
	ear Channel Capability Format - Extended Superframe -				00001	0.00	0.00	550.00								
	bsequent Activity Only			UEPMG	CCOEF	0.00	0.00	590.00								
	Mark Inversion (AMI)					0.00	0.00	555.50			1	1	1	1	1	
	perframe Format			UEPMG	MCOSF	0.00	0.00	0.00			1	1	1	1	1	
	ended Superframe Format			UEPMG	MCOPO	0.00	0.00	0.00								
	Ports Associated with 4-Wire DS1 Loop with Channelization	on with	Port	-							1	1			1	
Exchange																
ľ																
Line	e Side Combination Channelized PBX Trunk Port - Business			UEPPX	UEPCX	1.70	0.00	0.00	0.00	0.00			30.89	7.03		
Line	e Side Outward Channelized PBX Trunk Port - Business			UEPPX	UEPOX	1.70	0.00	0.00	0.00	0.00			30.89	7.03		
	e Side Inward Only Channelized PBX Trunk Port without DID			UEPPX	UEP1X	1.70	0.00	0.00	0.00	0.00			30.89	7.03		
	Vire Trunk Side Unbundled Channelized DID Trunk Port			UEPPX	UEPDM	8.97	0.00	0.00	0.00	0.00			30.89	7.03		
	tivations - Unbundled Loop Concentration															
Bar	ature (Service) Activation for each Line Port Terminated in D4 nk (includes Q.1.4, P50.1, P.50.498)			UEPPX	1PQWM	2.02	23.94	12.64	3.82	3.80			30.89	7.03		
D4	ature (Service) Activation for each Trunk Port Terminated in Bank (includes Q.1.4, P50.1, P.50.498)			UEPPX	1PQWU	2.02	73.67	17.37	54.09	10.57			30.89	7.03		
	Number/ Group Establishment Charges for DID Service															
	D Trunk Termination (1 per Port)			UEPPX	NDT	0.00	0.00	0.00								
	O Numbers - groups of 20 - Valid all States			UEPPX	ND4	0.00	0.00	0.00								
	n-Consecutive DID Numbers - per number			UEPPX	ND5	0.00	0.00	0.00								
	serve Non-Consecutive DID Numbers serve DID Numbers			UEPPX UEPPX	ND6 NDV	0.00	0.00	0.00								
	ber Portability			UEFFA	NDV	0.00	0.00	0.00								
	cal Number Portability - 1 per port			UEPPX	LNPCP	3.15	0.00	0.00								
	S - Vertical and Optional			UEFFA	LINFOF	3.15	0.00	0.00								
	ching Features Offered with Line Side Ports Only															
	Features Available			UEPPX	UEPVF	0.00	0.00	0.00								
	T LOOP COMBINATIONS - MARKET RATES			02.1.7	02. 11	0.00	0.00	0.00			1	1				
	es shall apply where BellSouth is not required to provide	unbund	led lo	cal switching or sw	itch ports per	FCC and/or St	tate Commissio	n rules.								
This includ									1		1		İ		İ	
	port/loop combinations that are Currently Combined or M	lot Curr	entiv (Combined in Zone 1	of the Top 8	MSAS in BellS	South's region f	or end users v	with 4 or more	DS0 equivaler	nt lines.					
The Top 8	MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderda	ale, Miar	mi); G/	A (Atlanta); LA (New	/ Orleans); NC	(Greensboro-	Winston Salem-	Highpoint/Ch	arlotte-Gaston	ia-Rock Hill);	TN (Nashvill		İ	ĺ	İ	
	currently is developing the billing capability to mechanica												. In the interi	m where Bell	South cannot	bill Market
Rates, Bell	South shall bill the rates in the Cost-Based section precede	ling in li	ieu of	the Market Rates ar	nd reserves th	e right to true-	up the billing d	ifference.	-							
	t Rate for unbundled ports includes all available features i															
	and Tandem Switching Usage and Common Transport Us	age rate	es in th	ne Port section of the	his rate exhibi	it shall apply to	o all combinatio	ns of loop/po	ort network eler	nents except	for UNE Coi	in Port/Loop	o Combination	ns which have	e a flat rate us	age charge
(USOC: UR																
	irrently Combined scenarios the Nonrecurring charges are	listed in	n the F	irst and Additional	NRC column	s for each Port	USOC. For Cu	rrently Combi	ined scenarios	, the Nonrecu	ring charge	s are listed	in the NRC - 0	Currently Con	nbined sectio	n
	NRCs may apply also and are categorized accordingly.															
	DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)															
	Loop Combination Rates															
	Vire VG Loop/Port Combo - Zone 1		1			26.48										
	Vire VG Loop/Port Combo - Zone 2		2			30.31										
	Vire VG Loop/Port Combo - Zone 3		3			35.32										
UNE Loop																
	Vire Voice Grade Loop (SL1) - Zone 1		1	UEPRX	UEPLX	12.48										
2-W	Vire Voice Grade Loop (SL1) - Zone 2		2	UEPRX	UEPLX	16.31										
	Vire Voice Grade Loop (SL1) - Zone 3		3	UEPRX	UEPLX	21.32										

NBUNDLE	D NETWORK ELEMENTS - Tennessee												Attachment:	2	Exhi	bit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
						100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
2-Wire	Voice Grade Line Port (Res) 2-Wire voice unbundled port - residence			UEPRX	UEPRL	14.00	00.00	90.00					30.89	7.03		
	2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res			UEPRX	UEPRC	14.00	90.00 90.00	90.00					30.89	7.03		
	2-Wire voice unbundled port with Galler ID - res			UEPRX	UEPRO	14.00	90.00	90.00					30.89	7.03		
	2-Wire voice Grade unbundled Tennessee extended local			-												
	dialing parity port with Caller ID - res			UEPRX	UEPAQ	14.00	90.00	90.00					30.89	7.03		
	2-Wire voice unbundled Tennessee Area Calling port with Caller															
	ID - res (F2R)			UEPRX	UEPAK	14.00	90.00	90.00					30.89	7.03		
	2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACER)			UEPRX	UEPAL	14.00	90.00	90.00					30.89	7.03		
	2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACSR)			UEPRX	UEPAM	14.00	90.00	90.00					30.89	7.03		
	2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (1MF2X)			UEPRX	UEPAN	14.00	90.00	90.00					30.89	7.03		
	2-Wire voice unbundled Tennessee Area Calling port with Caller			UEPKA	UEPAN	14.00	90.00	90.00					30.69	7.03		
	ID - res (2MR)			UEPRX	UEPAO	14.00	90.00	90.00					30.89	7.03	ļ	
	2-Wire voice unbundles res, low usage line port with Caller ID (LUM)			UEPRX	UEPAP	14.00	90.00	90.00					30.89	7.03		
	2-Wire voice unbundled Low Usage Line Port without Caller ID Capability			UEPRX	UEPRT	14.00	90.00	90.00					30.89	7.03		
	2-Wire Voice Unbundled Tennessee Residence Dialing Plan without Caller ID			UEPRX	UEPWN	14.00	90.00	90.00					30.89	7.03		
	2-Wire voice unbundled Tennessee Area Plus Port without Caller ID Capability			UEPRX	UEPRR	14.00	90.00	90.00					30.89	7.03		
	L NUMBER PORTABILITY			UEPRA	UEPKK	14.00	90.00	90.00	ł				30.69	7.03		
LOCA	Local Number Portability (1 per port)			UEPRX	LNPCX	0.35	-		1							
FEATU				-												
	All Features Offered			UEPRX	UEPVF	0.00	0.00	0.00					30.89	7.03		
NONR	ECURRING CHARGES - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop / Line Port Combination - Switch-as-is			UEPRX	USAC2		41.50	41.50					30.89	7.03		
	2-Wire Voice Grade Loop / Line Port Combination - Switch as-is 2-Wire Voice Grade Loop / Line Port Combination - Switch with			OLFRA	03A02		41.50	41.50					30.89	7.03		
	change			UEPRX	USACC		41.50	41.50					30.89	7.03		
ADDIT	IONAL NRCs															
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent			UEPRX	USAS2	0.00	0.00	0.00					30.89	7.03		
2-WIR	E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)			ULFKA	03432	0.00	0.00	0.00					30.09	7.03		
	ort/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			26.48										
	2-Wire VG Loop/Port Combo - Zone 2		2			30.31										
	2-Wire VG Loop/Port Combo - Zone 3		3		_	35.32			-							
UNEL	oop Rates 2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPBX	UEPLX	12.48										
	2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPBX	UEPLX	12.48			ł	1	+		ł	{	-	
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPBX	UEPLX	21.32			ł	1			1	1		
2-Wire	Voice Grade Line Port (Bus)															
	2-Wire voice unbundled port without Caller ID - bus			UEPBX	UEPBL	14.00	90.00	90.00					30.89	7.03		
	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPBX	UEPBC	14.00	90.00	90.00					30.89	7.03		
	2-Wire voice unbundled port outgoing only - bus			UEPBX	UEPBO	14.00	90.00	90.00			L		30.89	7.03		
	2-Wire voice Grade unbundled Tennessee extended local dialing parity port with Caller ID - bus			UEPBX	UEPAV	14.00	90.00	90.00					30.89	7.03		
	2-Wire voice unbundled Tennessee Bus 2-Way Area Calling Port Economy Option (TACC1)			UEPBX	UEPAC	14.00	90.00	90.00					30.89	7.03		
	2-Wire voice unbundled Tennessee Bus 2-Way Area Calling															
	Port Standard Option (TACC2) 2-Wire voice unbundled Tennessee Bus 2-Way Collierville and		<u> </u>	UEPBX	UEPAD	14.00	90.00	90.00					30.89	7.03		
	Memphis Local Calling Port (B2F) 2-Wire voice unbundled Incoming Only Port without Caller ID		<u> </u>	UEPBX	UEPAE	14.00	90.00	90.00					30.89	7.03		
	Capability			UEPBX	UEPBE	14.00	90.00	90.00					30.89	7.03		

BUNDLED NETWORK ELEMENTS - Tennessee		1										Attachment:			bit: B
EGORY RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Manually	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge - Manual Sv Order vs. Electronic Disc Add
					Rec	Nonrecurring		Nonrecurring					Rates(\$)		
					100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
2-Wire Voice Unbundled Tennessee Business Dialing Plan			UEPBX	UEPWO	14.00	90.00	90.00					30.89	7.00		
without Caller ID LOCAL NUMBER PORTABILITY	_		UEPBA	UEPWU	14.00	90.00	90.00					30.89	7.03		
Local Number Portability (1 per port)			UEPBX	LNPCX	0.35										
FEATURES															
All Features Offered			UEPBX	UEPVF	0.00	0.00	0.00					30.89	7.03		
NONRECURRING CHARGES - CURRENTLY COMBINED															
2-Wire Voice Grade Loop / Line Port Combination - Switch-as-i			UEPBX	USAC2		41.50	41.50					30.89	7.03		
2-Wire Voice Grade Loop / Line Port Combination - Switch with				USACC		44.50	44.50					00.00	7.03		
change ADDITIONAL NRCs	-		UEPBX	USACC		41.50	41.50			-		30.89	7.03		
NRC - 2-Wire Voice Grade Loop/Line Port Combination -															
Subsequent			UEPBX	USAS2	0.00	0.00	0.00					30.89	7.03		
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX	1														
UNE Port/Loop Combination Rates															
2-Wire VG Loop/Port Combo - Zone 1		1			26.48										
2-Wire VG Loop/Port Combo - Zone 2		2			30.31										
2-Wire VG Loop/Port Combo - Zone 3		3			35.32										
UNE Loop Rates					10.10										
2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPRG UEPRG	UEPLX UEPLX	12.48 16.31										
2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3	-	2	UEPRG	UEPLX	21.32					-					
2-Wire Voice Grade Line Port Rates (RES - PBX)		5			21.02										
2-Wire VG Unbundled Combination 2-Way PBX Trunk Port -															
Res			UEPRG	UEPRD	14.00	90.00	90.00					30.89	7.03		
LOCAL NUMBER PORTABILITY															
Local Number Portability (1 per port)			UEPRG	LNPCP	3.15	0.00	0.00								
FEATURES															
All Features Offered			UEPRG	UEPVF	0.00	0.00	0.00					30.89	7.03		
NONRECURRING CHARGES - CURRENTLY COMBINED	_			_											
2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is			UEPRG	USAC2		41.50	41.50					30.89	7.03		
2-Wire Voice Grade Loop/ Line Port Combination - Switch vith		1	OEI NO	00/102		41.00	41.00					00.00	1.00		
Change			UEPRG	USACC		41.50	41.50					30.89	7.03		
ADDITIONAL NRCs															
2 Wire Loop/Line Side Port Combination - Non feature -															
Subsequent Activity- Nonrecurring						0.00	0.00					30.89	7.03		
PBX Subsequent Activity - Change/Rearrange Multiline Hunt															
Group						14.64	14.64					30.89	7.03		
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX UNE Port/Loop Combination Rates)			-											
2-Wire VG Loop/Port Combo - Zone 1	-	1			26.48					-					
2-Wire VG Loop/Port Combo - Zone 2		2			30.31										
2-Wire VG Loop/Port Combo - Zone 3		3			35.32						1				
UNE Loop Rates	1	Ť	1		00.0E					1			İ		
2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPPX	UEPLX	12.48										
2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPPX	UEPLX	16.31										
2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPPX	UEPLX	21.32										
2-Wire Voice Grade Line Port Rates (BUS - PBX)	1							├ ─── 		-	-				<u> </u>
Line Cide Liebundled Combinetize 0 May DDV Trust, Durt, D	.1	1			44.00	00.00	00.00					00.00	7.00		
Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus Line Side Unbundled Outward PBX Trunk Port - Bus			UEPPX UEPPX	UEPPC UEPPO	14.00 14.00	90.00 90.00	90.00 90.00			-		30.89 30.89	7.03 7.03		
Line Side Unbundled Outward PBX Trunk Port - Bus	+	-	UEPPX	UEPP0	14.00	90.00	90.00	<u>} </u> }		+	<u> </u>	30.89	7.03		<u> </u>
2-Wire Voice Unbundled PBX LD Terminal Ports	1	1	UEPPX	UEPLD	14.00	90.00	90.00			1	1	30.89	7.03		1
2-Wire Voice Unbundled 2-Way Combination PBX Tennessee	1	1	1			00.00	00.00			1		00.00			
Calling Port	1	1	UEPPX	UEPT2	14.00	90.00	90.00					30.89	7.03		
2-Wire Voice Unbundled 1-Way Outgoing PBX Tennessee										Γ					
Calling Port	1	1	UEPPX	UEPTO	14.00	90.00	90.00			1	L	30.89	7.03		

INBUNDLE	D NETWORK ELEMENTS - Tennessee												Attachment:			bit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrecurring		Nonrecurring	g Disconnect			OSS	Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPPX	UEPXA	14.00	90.00	90.00					30.89	7.03		
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPPX	UEPXB	14.00	90.00	90.00					30.89	7.03		
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPPX	UEPXC UEPXD	14.00	90.00	90.00 90.00					30.89 30.89	7.03		
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD			UEPPX	UEPAD	14.00	90.00	90.00					30.89	7.03		
	Capable Port			UEPPX	UEPXE	14.00	90.00	90.00					30.89	7.03		
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy			OLITX	OLIXE	14.00	30.00	30.00					50.05	1.05		
	Administrative Calling Port			UEPPX	UEPXL	14.00	90.00	90.00					30.89	7.03		
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy															
	Room Calling Port			UEPPX	UEPXM	14.00	90.00	90.00					30.89	7.03		
	2-Wire Voice Unbundled 1-W Out PBX Hotel/Hospital Economy															
	Administrative Calling Port TN			UEPPX	UEPXN	14.00	90.00	90.00					30.89	7.03		
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital															
	Discount Room Calling Port			UEPPX	UEPXO	14.00	90.00	90.00					30.89	7.03		
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPPX	UEPXS	14.00	90.00	90.00					30.89	7.03		
	2-Wire Voice Unbundled PBX Collierville and Memphis Calling					44.00	00.00	00.00					00.00	7.00		
	Port			UEPPX	UEPXU	14.00	90.00	90.00					30.89	7.03		
	2-Wire Voice Unbundled 2-Way PBX Tennessee RegionServ Callling Port			UEPPX	UEPXV	14.00	90.00	90.00					30.89	7.03		
	Tennessee PBX 2-Way Combo Each Additional Trunk			UEPPX	UEPXV	14.00	90.00	90.00					30.89	7.03		
	Collierville and Memphis Local Calling Plan			UEPPX	UEPA6	14.00	90.00	90.00					30.89	7.03		
	Tennessee PBX 2-Way Combo First Trunk Collierville and			ULFFA	ULFAU	14.00	90.00	90.00					30.09	7.05		
	Memphis Local Calling Plan			UEPPX	UEPA7	14.00	90.00	90.00					30.89	7.03		
LOCAL	NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPPX	LNPCP	3.15	0.00	0.00								
FEATU																
	All Features Offered			UEPPX	UEPVF	0.00	0.00	0.00					30.89	7.03		
NONRE	CURRING CHARGES - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is			UEPPX	USAC2		41.50	41.50					30.89	7.03		
	2-Wire Voice Grade Loop/ Line Port Combination - Switch with													=		
	Change			UEPPX	USACC		41.50	41.50			-		30.89	7.03		
AUUITI	ONAL NRCs															
	2-Wire Voice Grade Loop/ Line Port Combination - Subsequent			UEPPX	USAS2	0.00	0.00	0.00					30.89	7.03		
	2 Wire Loop/Line Side Port Combination - Non feature -			OLITX	00402	0.00	0.00	0.00					50.03	1.00		
	Subsequent Activity- Nonrecurring						0.00	0.00					30.89	7.03		
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt															
	Group						14.64	14.64					30.89	7.03		
	VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN POP	RT														
UNE Po	ort/Loop Combination Rates															
	2-Wire VG Coin Port/Loop Combo – Zone 1		1			26.48										
	2-Wire VG Coin Port/Loop Combo – Zone 2		2			30.31										
	2-Wire VG Coin Port/Loop Combo – Zone 3		3			35.32										
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPCO	UEPLX	12.48										
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPCO	UEPLX	12.40										
	2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3	ł –	3	UEPCO	UEPLX	21.32					1					
	Voice Grade Line Port Rates (Coin)	l				252					1					
	2-Wire Coin 2-Way without Operator Screening and without	l		İ							1					l
	Blocking (TN)			UEPCO	UEPTB	14.00	90.00	90.00					30.89	7.03		
	2-Wire Coin 2-Way with Operator Screening and Blocking: 011,	ſ			ĺ											
	900/976, 1+DDD (NC, TN)			UEPCO	UEPRP	14.00	90.00	90.00					30.89	7.03		
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking															
	(TN)	ļ		UEPCO	UEPTA	14.00	90.00	90.00					30.89	7.03		
	2-Wire Coin 2-Way with Operator Screening and Blocking:				UEDO			~~~~								
_	900/976, 1+DDD, 011+, and Local (NC, TN)			UEPCO	UEPCA	14.00	90.00	90.00			L		30.89	7.03		<u> </u>
	2-Wire Coin Outward with Operator Screening and 011 Blocking	1	I.	UEPCO	UEPTC	14.00	90.00	90.00			1		30.89	7.03		1

UNBUNDLE	D NETWORK ELEMENTS - Tennessee		1	1								-	Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Dee	Nonrecurring		Nonrecurring	Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Coin Outward with Operator Screening and Blocking:															
	900/976, 1+DDD, 011+, and Local (TN)			UEPCO	UEPOT	14.00	90.00	90.00					30.89	7.03		
LOCAL	NUMBER PORTABILITY Local Number Portability (1 per port)			UEPCO	LNPCX	0.35										
NONRE	CURRING CHARGES - CURRENTLY COMBINED			UEFCO	LINFUA	0.35			-							
i lointe	Solutio Charges Solution Solution															
	2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is			UEPCO	USAC2		41.50	41.50					30.89	7.03		
	2-Wire Voice Grade Loop/ Line Port Combination - Switch with															
	Change			UEPCO	USACC		41.50	41.50					30.89	7.03		
ADDITI	ONAL NRCs				-											
	2-Wire Voice Grade Loop/ Line Port Combination - Subsequent			UEPCO	USAS2	0.00	0.00	0.00					30.89	7.03		
2-WIRF	VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE	LINF F	PORT		03732	0.00	0.00	0.00				-	30.09	1.03		
	ort/Loop Combination Rates		<i></i> (1 1						1	-				
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			30.56										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2	-		35.63										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3		3		_	42.28										
	pop Rates				115050	10.50										
	2-Wire Voice Grade Loop (SL2) - Zone 1 2-Wire Voice Grade Loop (SL2) - Zone 2		1	UEPFR UEPFR	UECF2 UECF2	16.56 21.63										
<u> </u>	2-Wire Voice Grade Loop (SL2) - Zone 2 2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFR	UECF2	21.63										
	Voice Grade Line Port Rates (Res)			OLITIK	02012	20.20										
	2-Wire voice unbundled port - residence			UEPFR	UEPRL	14.00	115.00	75.00	40.00	30.00		15.69				
	2-Wire voice unbundled port with Caller ID - res			UEPFR	UEPRC	14.00	115.00	75.00	40.00	30.00		15.69				
	2-Wire voice unbundled port outgoing only - res			UEPFR	UEPRO	14.00	115.00	75.00	40.00	30.00		15.69				
	2-Wire voice Grade unbundled Tennessee extended local															
	dialing parity port with Caller ID - res			UEPFR	UEPAQ	14.00	115.00	75.00	40.00	30.00		15.69				
	2-Wire voice unbundled Tennessee Area Plus with Caller ID - res (AC7)			UEPFR	UEPAH	14.00	115.00	75.00	40.00	30.00		15.69				
	2-Wire voice unbundled Tennessee Area Calling port with Caller			OLITIK	OLIAN	14.00	113.00	75.00	40.00	50.00		10.00				
	ID - res (F2R)			UEPFR	UEPAK	14.00	115.00	75.00	40.00	30.00		15.69				
	2-Wire voice unbundled Tennessee Area Calling port with Caller															
	ID - res (TACER)			UEPFR	UEPAL	14.00	115.00	75.00	40.00	30.00		15.69				
	2-Wire voice unbundled Tennessee Area Calling port with Caller															
	ID - res (TACSR)			UEPFR	UEPAM	14.00	115.00	75.00	40.00	30.00		15.69				
	2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (1MF2X)			UEPFR	UEPAN	14.00	115.00	75.00	40.00	30.00		15.69				
	2-Wire voice unbundled Tennessee Area Calling port with Caller			OLI I N		14.00	113.00	75.00	40.00	50.00		13.09				
1	ID - res (2MR)			UEPFR	UEPAO	14.00	115.00	75.00	40.00	30.00		15.69				
	2-Wire voice unbundles res, low usage line port with Caller ID										1		İ	İ	İ	l
	(LUM)			UEPFR	UEPAP	14.00	115.00	75.00	40.00	30.00		15.69				
	2-Wire Voice Unbundled Tennessee Residence Dialing Plan					11.00		75.00	10.00	00.00		45.00				
INTER	without Caller ID FFICE TRANSPORT			UEPFR	UEPWN	14.00	115.00	75.00	40.00	30.00		15.69				
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility															
	Termination			UEPFR	U1TV2	18.58	55.39	17.37	27.96	3.51						
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile									2.01	1					
	or Fraction Mile			UEPFR	1L5XX	0.0174										
FEATU																
	All Features Offered			UEPFR	UEPVF	0.00	0.00	0.00				15.69				
	NUMBER PORTABILITY Local Number Portability (1 per port)			UEPFR	LNPCX	0.35										
	CURRING CHARGES (NRCs) - CURRENTLY COMBINED			OLFIN	LINF GA	0.35	├					-				
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port				1 1						1					
	Combination - Conversion - Switch-as-is			UEPFR	USAC2		16.94	3.72				15.69				
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
	Combination - Conversion - Switch-With-Change	<u> </u>		UEPFR	USACC		16.94	3.72				15.69				L
	VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIRE	LINEF	PORT (BUS)												ļ
	ort/Loop Combination Rates	I			1		l l		1		L	l	l	1	l	L

JNBUNDLE	D NETWORK ELEMENTS - Tennessee											1	Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'l
						Rec	Nonrecurring		Nonrecurring	Disconnect			OSS	Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			30.56										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2		2			35.63										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 3 oop Rates		3			42.28										
UNE LO	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFB	UECF2	16.56										
	2-Wire Voice Grade Loop (SL2) - Zone 1 2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFB	UECF2	21.63										
	2-Wire Voice Grade Loop (SL2) - Zone 2		3	UEPFB	UECF2	28.28										
2-Wire	Voice Grade Line Port (Bus)		Ű	02.10	020.2	20.20										
	2-Wire voice unbundled port without Caller ID - bus			UEPFB	UEPBL	14.00	115.00	75.00	40.00	30.00		15.69				
	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPFB	UEPBC	14.00	115.00	75.00	40.00	30.00		15.69				
	2-Wire voice unbundled port outgoing only - bus			UEPFB	UEPBO	14.00	115.00	75.00	40.00	30.00		15.69				
	2-Wire voice Grade unbundled Tennessee extended local															
	dialing parity port with Caller ID - bus			UEPFB	UEPAV	14.00	115.00	75.00	40.00	30.00		15.69				
	2-Wire voice unbundled incoming only port with Caller ID - Bus			UEPFB	UEPB1	14.00	115.00	75.00	40.00	30.00		15.69				
	2-Wire voice unbundled Tennessee Bus 2-Way Area Calling Port Economy Option (TACC1)			UEPFB	UEPAC	14.00	115.00	75.00	40.00	30.00		15.69				
	2-Wire voice unbundled Tennessee Bus 2-Way Area Calling			UEPFB	UEPAD	14.00	115.00	75.00	40.00	30.00		15.69				
	Port Standard Option (TACC2) 2-Wire voice unbundled Tennessee Bus 2-Way Collierville and			UEPFB	UEPAD	14.00	115.00	75.00	40.00	30.00		15.69				
	Memphis Local Calling Port (B2F)			UEPFB	UEPAE	14.00	115.00	75.00	40.00	30.00		15.69				
	2-Wire Voice Unbundled Tennessee Business Dialing Plan without Caller ID			UEPFB	UEPWO	14.00	115.00	75.00	40.00	30.00		15.69				
	Tennessee Inward Collierville and Memphis Local Calling Plan (BUS)			UEPFB	UEPB2	14.00	115.00	75.00	40.00	30.00		15.69				
	Tennessee 2-Way Collierville and Memphis Local Calling Plan (BUS)			UEPFB	UEPB3	14.00	115.00	75.00	40.00	30.00		15.69				
LOCAL	NUMBER PORTABILITY				LUDOV											
INTER	Local Number Portability (1 per port)			UEPFB	LNPCX	0.35										
INTER	OFFICE TRANSPORT Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility Termination			UEPFB	U1TV2	18.58	55.39	17.37	27.96	3.51						
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile or Fraction Mile			UEPFB	1L5XX	0.0174	00.00	11.01	21.00	0.01						
FEATU	RES															
	All Features Offered			UEPFB	UEPVF	0.00	0.00	0.00				15.69				
NONRE	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port Combination - Conversion - Switch-as-is			UEPFB	USAC2		16.94	3.72				15.69				
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port			UEPFB	110.000		10.01	0.70				45.00				
2 WIDE	Combination - Conversion - Switch with change VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)			UEPFB	USACC		16.94	3.72				15.69				
	ort/Loop Combination Rates															
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		1			30.56							-	-		
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 1		2		1	35.63										
	2-Wire VG Loop/IO Tranport/Port Combo - Zone 2	1	3			42.28					1					
UNE Lo	pop Rates		Ē				İ				1					
	2-Wire Voice Grade Loop (SL2) - Zone 1		1	UEPFP	UECF2	16.56										
	2-Wire Voice Grade Loop (SL2) - Zone 2		2	UEPFP	UECF2	21.63										
	2-Wire Voice Grade Loop (SL2) - Zone 3		3	UEPFP	UECF2	28.28										
2-Wire	Voice Grade Line Port Rates (BUS - PBX)	l														
			1					~~~~	· · · · ·			4= 0-				
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPFP	UEPPC	14.00	106.40	63.08	42.67	18.54		15.69				
	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPFP UEPFP	UEPPO UEPP1	14.00	106.40	63.08 63.08	42.67	18.54	ł	15.69		L		
	Line Side Unbundled Incoming PBX Trunk Port - Bus 2-Wire Voice Unbundled PBX LD Terminal Ports			UEPFP	UEPP1	14.00 14.00	106.40 106.40	63.08	42.67 42.67	18.54 18.54		15.69 15.69				
	2-Wire Voice Unbundled 2-Way Combination PBX Tennessee Calling Port			UEPFP	UEPLD	14.00	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Tennessee Calling Port			UEPFP	UEPTO	14.00	106.40	63.08	42.67	18.54	1	15.69				

UNBUNDLE	D NETWORK ELEMENTS - Tennessee												Attachment:			ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge - Manual Sv Order vs.
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC		SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPFP	UEPXA	14.00	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPFP	UEPXB	14.00	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled PBX LD DDD Terminals Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPFP UEPFP	UEPXC UEPXD	14.00 14.00	106.40 106.40	63.08 63.08	42.67 42.67	18.54 18.54		15.69 15.69				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard PDR 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD			UEFFF	UEFAD	14.00	106.40	03.00	42.07	10.34		15.69				
	Capable Port			UEPFP	UEPXE	14.00	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy															
	Administrative Calling Port			UEPFP	UEPXL	14.00	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy															
	Room Calling Port			UEPFP	UEPXM	14.00	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled 1W Out PBX Hotel/Hospital Economy											/= a-				
	Administrative Calling Port TN Calling Port			UEPFP	UEPXN	14.00	106.40	63.08	42.67	18.54		15.69				───
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port			UEPFP	UEPXO	14.00	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPFP	UEPXS	14.00	106.40	63.08	42.67	18.54		15.69				-
	2-Wire Voice Unbundled PBX Collierville and Memphis Calling			0EITT	OEI XO	14.00	100.40	00.00	42.07	10.04		10.00				1
	Port			UEPFP	UEPXU	14.00	106.40	63.08	42.67	18.54		15.69				
	2-Wire Voice Unbundled 2-Way PBX Tennessee RegionServ															
	Callling Port			UEPFP	UEPXV	14.00	106.40	63.08	42.67	18.54		15.69				
LOCA	L NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPFP	LNPCP	3.15	0.00	0.00				15.69				
INTER	OFFICE TRANSPORT															
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Facility Termination			UEPFP	U1TV2	18.58	55.39	17.37	27.96	3.51						
	Interoffice Transport - Dedicated - 2 Wire Voice Grade - Per Mile			OLFIF	01172	10.30	55.55	17.57	21.90	3.51						+
	or Fraction Mile			UEPFP	1L5XX	0.0174										
FEAT																
	All Features Offered			UEPFP	UEPVF	0.00	0.00	0.00				15.69				
NONR	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port															
	Combination - Conversion - Switch-as-is			UEPFP	USAC2		16.94	3.72				15.69				
	2-Wire Loop / Dedicated IO Transport / 2 Wire Line Port			UEPFP			16.94	0.70				45.00				
	Combination - Conversion - Switch with change PORT/LOOP COMBINATIONS - MARKET BASED RATES			UEPFP	USACC		16.94	3.72				15.69				
	E VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK	POPT														+
	Port/Loop Combination Rates				+ +											-
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1		1			49.60										
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2		2			51.09										1
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3		3			56.00										
UNE L	oop Rates															
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1		1	UEPPX	UECD1	9.60										
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2 2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3		2		UECD1 UECD1	11.09 16.00										<u> </u>
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3 Exchange Ports - 2-Wire DID Port		3	UEPPX UEPPX	UECD1 UEPD1	40.00	600.00	45.00	8.45	3.91			30.89	7.03		
	ECURRING CHARGES - CURRENTLY COMBINED		<u> </u>		JLFDI	40.00	000.00	45.00	0.45	3.91			50.69	1.03		+
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination -	1			+ +											<u> </u>
1	Switch-As-Is Top 8 MSAs only		1	UEPPX	USAC1		100.00	42.50					30.89	7.03		
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion															1
	with BellSouth Allowable Changes Top 8 MSAs only			UEPPX	USA1C		100.00	42.50					30.89	7.03		
Telep	none Number/Trunk Group Establisment Charges															
	DID Trunk Termination (One Per Port)			UEPPX	NDT	0.00	0.00	0.00								
	Additional DID Numbers for each Group of 20 DID Numbers			UEPPX	ND4	0.00	0.00	0.00								
	DID Numbers, Non- consecutive DID Numbers, Per Number Reserve Non-Consecutive DID numbers	l		UEPPX UEPPX	ND5 ND6	0.00	0.00	0.00								<u> </u>
	Reserve Non-Consecutive DID numbers Reserve DID Numbers			UEPPX	ND6 NDV	0.00	0.00	0.00								
	L NUMBER PORTABILITY					0.00	0.00	0.00								
2004	Local Number Portability (1 per port)	1		UEPPX	LNPCP	3.15	0.00	0.00	1					1	1	1
2-WIR	E ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LI	NE SIDE	PORT			20		2.50						İ	İ	1
	Port/Loop Combination Rates	1			1									1	1	1

BUNDLE	D NETWORK ELEMENTS - Tennessee		1											Attachment:			ibit: B
TEGORY	RATE ELEMENTS	Interi m	Zone	E	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge Manual S Order ve
							B	Nonrecurring		Nonrecurring	Disconnect			OSS	Rates(\$)	L	<u> </u>
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -																
	UNE Zone 1		1	UEPPB	UEPPR		32.27										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 2		2	UEPPB	UEPPR		34.78										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -																
	UNE Zone 3		3	UEPPB	UEPPR		44.32										
	2-Wire ISDN Digital Grade Loop - UNE Zone 1		1	UEPPB	UEPPR	USL2X	16.20										
	2-Wire ISDN Digital Grade Loop - UNE Zone 2		2	UEPPB	UEPPR	USL2X	18.71										
	2-Wire ISDN Digital Grade Loop - UNE Zone 3		3	UEPPB	UEPPR	USL2X	28.25	505.00	100.00	75.00	70.00			00.00	7.00		
NOND	Exchange Port - 2-Wire ISDN Line Side Port CURRING CHARGES - CURRENTLY COMBINED			UEPPB	UEPPR	UEPPB	80.00	525.00	400.00	75.00	70.00			30.89	7.03		
NONRE	2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port			<u> </u>													<u> </u>
	2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination - Conversion - Top 8 MSAs only	1	1	UEPPB	UEPPR	USACB	0.00	225.00	225.00					30.89	7.03		1
	ONAL NRCs			JLPPD	ULPPR	USAUD	0.00	223.00	223.00					30.69	7.03	ł	+
ADDIT	2-Wire ISDN Loop / 2-Wire ISDN Port Combination - Sub Actvy	<u> </u>		<u> </u>													
	Non Feature/Add Trunk		1	UEPPB	UEPPR	USASB		212.88						30.89	7.03		
LOCAL	NUMBER PORTABILITY			50.10	ULITI	20,00		212.00						50.09	1.05		
	Local Number Portability (1 per port)			UEPPB	UEPPR	LNPCX	0.35	0.00	0.00			1					-
B-CHA	NNEL USER PROFILE ACCESS:			02.1.0	021111	2.1. 0.1	0.00	0.00	0.00			1					-
	CVS/CSD (DMS/5ESS)			UEPPB	UEPPR	U1UCA	0.00	0.00	0.00								
	CVS (EWSD)			UEPPB	UEPPR	U1UCB	0.00	0.00	0.00								
	CSD			UEPPB	UEPPR	U1UCC	0.00	0.00	0.00								
B-CHA	NNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS SO	C,MS, 8	TN)														
	CVS/CSD (DMS/5ESS)		1	UEPPB	UEPPR	U1UCD	0.00	0.00	0.00								
	CVS (EWSD)			UEPPB	UEPPR	U1UCE	0.00	0.00	0.00								
	CSD			UEPPB	UEPPR	U1UCF	0.00	0.00	0.00								1
USER	TERMINAL PROFILE																1
	User Terminal Profile (EWSD only)			UEPPB	UEPPR	U1UMA	0.00	0.00	0.00								1
VERTI	CAL FEATURES																
	All Vertical Features - One per Channel B User Profile			UEPPB	UEPPR	UEPVF	0.00	0.00	0.00								
	Interoffice Channel mileage each, including first mile and																1
	facilities termination				UEPPR	M1GNC	17.91	53.99	17.37								
	Interoffice Channel mileage each, additional mile			UEPPB	UEPPR	M1GNM	0.173	0.00	0.00								
	DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK	PORT															
UNE P	ort/Loop Combination Rates																
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE						000 70										
_	Zone 1		1	UEPPP		├ ──┤	982.73					ļ					+
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE	1	2	UEPPP			1 000 40										1
-	Zone 2 4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE		2	UEPPP			1,000.40							ł		ł	+
	Zone 3		3	UEPPP			1,023.59										1
	4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPPP		USL4P	57.73										
-	4-Wire DS1 Digital Loop - UNE Zone 1 4-Wire DS1 Digital Loop - UNE Zone 2		2	UEPPP		USL4P USL4P	75.40							ł		ł	+
-	4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPPP		USL4P USL4P	98.59					t					+
-	Exchange Ports - 4-Wire ISDN DS1 Port		3	UEPPP		USL4P UEPPP	98.59	950.00	950.00	130.00	100.00	t		30.89	7.03		+
NONPE	ECURRING CHARGES - CURRENTLY COMBINED			ULFFP		ULFFF	920.00	500.00	930.00	130.00	100.00	-		30.69	7.03		+
NONK	4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port											-					+
	Combination - Conversion -Switch-As-Is Top 8 MSAs only	1	1	UEPPP		USACP	0.00	925.00	925.00					30.89	7.03		1
	ONAL NRCs		1			- 5/ 10.	0.00	020.00	020.00			t		00.00			t
	4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy-		1												1		1
	Inward/two way Telephone Numbers (except NC)	1		UEPPP		PR7TF		0.94									1
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port -		1					0.04				t					t
	Outward Tel Numbers (All States except NC)	1	1	UEPPP		PR7TO		22.36	22.36								1
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port -		1					00	22.00			t					t
	Subsequent Inward Telephone Numbers	1		UEPPP		PR7ZT		44.71	44.70								1
LOCAL	NUMBER PORTABILITY		1									t					t
LOOAL	Local Number Portability (1 per port)			UEPPP		LNPCN	1.75										<u> </u>
	FACE (Provsioning Only)		1				1.75					1					+

UNBUND	DLED	NETWORK ELEMENTS - Tennessee										1		Attachment:			bit: B
CATEGOR	Y	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
								Nonrecurring		Nonrecurring	Disconnect			OSS	Rates(\$)	L	
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN		SOMAN	SOMAN	SOMAN
		Voice/Data			UEPPP	PR71V	0.00	0.00	0.00								
		Digital Data			UEPPP	PR71D	0.00	0.00	0.00								
Mar		Inward Data Additional "B" Channel			UEPPP	PR71E	0.00	0.00	0.00								
Nev		New or Additional - Voice/Data B Channel			UEPPP	PR7BV	0.00	28.39		-							
		New of Additional - Voice Data B Channel			UEPPP	PR7BF	0.00	29.11									
		New or Additional Inward Data B Channel			UEPPP	PR7BD	0.00	29.39									
CA	LL T	YPES															
		Inward			UEPPP	PR7C1	0.00	0.00	0.00								
		Outward			UEPPP	PR7C0	0.00	0.00	0.00								
		Two-way			UEPPP	PR7CC	0.00	0.00	0.00								
Inte	erotti	ce Channel Mileage Fixed Each Including First Mile			UEPPP	1LN1A	70 4005	145.98	109.85	19.55							
		Each Airline-Fractional Additional Mile			UEPPP	1LN1A 1LN1B	76.1825 0.3525	145.98	109.85	19.55							
4-14		DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT			OLFFF	TLINID	0.3323										
		rt/Loop Combination Rates															
		4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1		1	UEPDC		93.28										
		4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2		2	UEPDC		110.95										
		4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3		3	UEPDC		134.14										
UN		op Rates															
		4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPDC	USLDC	57.53										
		4-Wire DS1 Digital Loop - UNE Zone 2			UEPDC	USLDC	75.40										
LIN		4-Wire DS1 Digital Loop - UNE Zone 3 rt Rate		3	UEPDC	USLDC	98.59										
UN		4-Wire DDITS Digital Trunk Port			UEPDC	UDD1T	750.00	982.57	450.10	196.09	19.23			30.89	7.03		
NO					OLFDC	ODDTT	730.00	502.57	430.10	190.09	19.23			30.09	7.05		
		4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Switch-As-Is Top 8 MSAs only			UEPDC	USAC4		312.91	312.91					30.89	7.03		
		4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with DS1 Changes Top 8 MSAs only			UEPDC	USAWA		312.91	312.91					30.89	7.03		
		4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with Change - Trunk Top 8 MSAs only			UEPDC	USAWB		312.91	312.91					30.89	7.03		
AD		DNAL NRCs															
		4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent Service Activity Per Service Order			UEPDC	USAS4		94.88	94.88								
		4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC - Subsequent Channel Activation/Chan - 2-Way Trunk			UEPDC	UDTTA		108.67	108.67					30.89	7.03		
	ŀ	AWire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent Channel Activation/Chan - 1-Way Outward Trunk			UEPDC	UDTTB		108.67	108.67					30.89	7.03		
		4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel Activation/Chan Inward Trunk w/out DID			UEPDC	UDTTC		108.67	108.67					30.89	7.03		
		4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation Per Chan - Inward Trunk with DID			UEPDC	UDTTD		108.67	108.67					30.89	7.03		
		4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation / Chan - 2-Way DID w User Trans			UEPDC	UDTTE		108.67	108.67					30.89	7.03		
BIP		R 8 ZERO SUBSTITUTION				00005		0.00	F00.00			-					
		B8ZS -Superframe Format			UEPDC	CCOSF		0.00	590.00			+					
A 14.		B8ZS - Extended Superframe Format e Mark Inversion			UEPDC	CCOEF		0.00	590.00		1	+					
Alte		AMI -Superframe Format			UEPDC	MCOSF		0.00	0.00			+					
		AMI - Extended SuperFrame Format			UEPDC	MCOPO		0.00	0.00			+					
Tel		one Number/Trunk Group Establisment Charges						0.00	0.00					1	1	1	
	ľ	Telephone Number for 2-Way Trunk Group			UEPDC	UDTGX	0.00										
	ŀ	Telephone Number for 1-Way Outward Trunk Group			UEPDC	UDTGY	0.00										
		Telephone Number for 1-Way Inward Trunk Group Without DID			UEPDC	UDTGZ	0.00										
		DID Numbers, Establish Trunk Group and Provide First Group of 20 DID Numbers			UEPDC	NDZ	0.00	0.00	0.00								

NRONDLE	D NETWORK ELEMENTS - Tennessee		1		-								Attachment:			bit: B
TEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic-	Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Order vs Electronic
													1st	Add'l	Disc 1st	Disc Add
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	DID Numbers for each Group of 20 DID Numbers			UEPDC	ND4	0.00										
	DID Numbers, Non- consecutive DID Numbers , Per Number			UEPDC	ND5	0.00	0.00	0.00								
	Reserve Non-Consecutive DID Nos. Reserve DID Numbers			UEPDC UEPDC	ND6 NDV	0.00	0.00	0.00								
Dedied	ted DS1 (Interoffice Channel Mileage) -			UEFDC	NDV	0.00	0.00	0.00								
	o for 4-Wire DS1 Digital Loop with 4-Wire DDITS Trunk Port															
	Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities															
	Termination)			UEPDC	1LNO1	75.83	145.98	109.85	19.66	14.99						
	Interoffice Channel Mileage - Additional rate per mile - 0-8 miles			UEPDC	1LNOA	0.3525	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities					0.0020	0.00	0.00					1		1	
	Termination)			UEPDC	1LNO2	0.00	0.00	0.00								
	Interoffice Channel Mileage - Additional rate per mile - 9-25															
	miles			UEPDC	1LNOB	0.3525	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities															
	Termination)			UEPDC	1LNO3	0.00	0.00	0.00				ļ				
	Interoffice Channel Mileage - Additional rate per mile - 25+ miles			UEPDC	1LNOC	0.3525	0.00	0.00								
	Local Number Portability, per DS0 Activated			UEPDC	LNPCP	3.15	0.00	0.00								
	Central Office Termininating Point			UEPDC	CTG	0.00	0.00	0.00								
4-WIRE	DS1 LOOP WITH CHANNELIZATION WITH PORT			021 00	010	0.00										
	is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Acti	ivations														
	em can have various rate combinations based on type and nu			used												
	S1 Loop		Ĺ													
	4-Wire DS1 Loop - UNE Zone 1		1	UEPMG	USLDC	57.73	0.00	0.00								1
	4-Wire DS1 Loop - UNE Zone 2		2	UEPMG	USLDC	75.40	0.00	0.00								
	4-Wire DS1 Loop - UNE Zone 3		3	UEPMG	USLDC	98.59	0.00	0.00								
UNE DS	SO Channelization Capacities (D4 Channel Bank Configuration	ns)														
	24 DSO Channel Capacity - 1 per DS1			UEPMG	VUM24	131.87	0.00	0.00					30.89	7.03		
	48 DSO Channel Capacity - 1 per 2 DS1s			UEPMG	VUM48	263.74	0.00	0.00					30.89	7.03		
	96 DSO Channel Capacity -1per 4 DS1s			UEPMG UEPMG	VUM96 VUM14	527.48 791.42	0.00	0.00					30.89 30.89	7.03		
	144 DS0 Channel Capacity - 1 per 6 DS1s 192 DS0 Channel Capacity -1 per 8 DS1s			UEPMG	VUM14 VUM19	827.76	0.00	0.00					30.89	7.03		
	240 DS0 Channel Capacity - 1 per 10 DS1s	-		UEPMG	VUM20	1,318.70	0.00	0.00					30.89	7.03		
_	288 DS0 Channel Capacity - 1 per 12 DS1s			UEPMG	VUM28	1,582.44	0.00	0.00					30.89	7.03		
	384 DS0 Channel Capacity - 1 per 16 DS1s			UEPMG	VUM38	2,109.92	0.00	0.00					30.89	7.03		
	480 DS0 Channel Capacity - 1 per 20 DS1s	1		UEPMG	VUM40	2,637.40	0.00	0.00					30.89	7.03	1	1
	576 DS0 Channel Capacity -1 per 24 DS1s	1		UEPMG	VUM57	3,164.88	0.00	0.00					30.89	7.03	1	1
	672 DS0 Channel Capacity - 1 per 28 DS1s	I		UEPMG	VUM67	3,692.36	0.00	0.00			l		30.89	7.03	1	1
	curring Charges (NRC) Associated with 4-Wire DS1 Loop with						stem									
	num System configuration is One (1) DS1, One (1) D4 Channe															
Multiple	es of this configuration functioning as one are considered Ac	dd'l afte	r the m	inimum system co	nfiguration is	counted.										
	NRC - Conversion (Currently Combined) with or without				110461											
	BellSouth Allowed Changes - Top 8 MSAs Only			UEPMG	USAC4	0.00	303.61	15.74					30.89	7.03		<u> </u>
	Additions Where Currently Combined and New (Not Currentl	y Comb	ined)		+											──
in Dens	sity Zone 1 Top 8 MSAs 1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc															<u> </u>
	Fea Activation -			UEPMG	VUMD4	0.00	704.68	441.48	138.36	16.41			30.89	7.03		
Bipolar	* 8 Zero Substitution					0.00		40					00.00		1	
	Clear Channel Capability Format, superframe - Subsequent					_										
	Activity Only Clear Channel Capability Format - Extended Superframe -			UEPMG	CCOSF	0.00	0.00	590.00								├
	Subsequent Activity Only			UEPMG	CCOEF	0.00	0.00	590.00								
Alterna	te Mark Inversion (AMI)															
	Superframe Format	1	1	UEPMG	MCOSF	0.00	0.00	0.00			1		1		1	L
	Extended Superframe Format ge Ports Associated with 4-Wire DS1 Loop with Channelizati			UEPMG	MCOPO	0.00	0.00	0.00								

UNBUNDLE	ED NETWORK ELEMENTS - Tennessee												Attachment:		Exhi	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
														=		
	Line Side Combination Channelized PBX Trunk Port - Business Line Side Outward Channelized PBX Trunk Port - Business			UEPPX UEPPX	UEPCX UEPOX	14.00 14.00	0.00	0.00	0.00	0.00			30.89 30.89	7.03		
	Line Side Outward Channelized PBA Trunk Port - Business			UEFFX	UEFUX	14.00	0.00	0.00	0.00	0.00			30.69	7.03		
	Line Side Inward Only Channelized PBX Trunk Port without DID			UEPPX	UEP1X	14.00	0.00	0.00	0.00	0.00			30.89	7.03		
	2-Wire Trunk Side Unbundled Channelized DID Trunk Port			UEPPX	UEPDM	40.00	0.00	0.00	0.00	0.00			30.89	7.03		
Featu	re Activations - Unbundled Loop Concentration															
	Feature (Service) Activation for each Line Port Terminated in D4															
	Bank (includes Q.1.4, P.50.1, & P.50.498)			UEPPX	1PQWM	2.02	40.00	20.00	6.00	5.00						
	Feature (Service) Activation for each Trunk Port Terminated in					0.00	110.00	20.00	75.00	45.00						
Telen	D4 Bank (includes Q.1.4, P.50.1, & P.50.498) hone Number/ Group Establishment Charges for DID Service			UEPPX	1PQWU	2.02	110.00	30.00	75.00	15.00						
reiep	DID Trunk Termination (1 per Port)			UEPPX	NDT	0.00	0.00	0.00								
	DID Numbers - groups of 20 - Valid all States	1	1	UEPPX	ND4	0.00	0.00	0.00			t					
	Non-Consecutive DID Numbers - per number		1	UEPPX	ND5	0.00	0.00	0.00								
	Reserve Non-Consecutive DID Numbers			UEPPX	ND6	0.00	0.00	0.00								
	Reserve DID Numbers			UEPPX	NDV	0.00	0.00	0.00								
Local	Number Portability															
	Local Number Portability - 1 per port			UEPPX	LNPCP	3.15	0.00	0.00								
	URES - Vertical and Optional Switching Features Offered with Line Side Ports Only															
	Switching realures Onered with Line Side Forts Only			UEPPX	UEPVF	0.00	0.00	0.00								
Local																
	All Features Available	s		OLITX	02. 11											
UNBUNDLED			State (Indled Local S	witching or Sw	itch Ports.								
UNBUNDLED	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES	and/or		Commission rule t	o provide Unbu				dled Port section	on of this Rate	e Exhibit.					
UNBUNDLED 1. Cos 2. Fea 3. End	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES st Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport	and/or ost Bas Usage	sed Rat rates ir	Commission rule to the section in the sa the Port section of	o provide Unbu me manner as of this rate exh	they are applie ibit shall apply	ed to the Stand to all combination	Alone Unbun	port network e	lements excep	t for UNE C					
UNBUNDLED 1. Cos 2. Fea 3. End 4. The	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES st Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not Co	and/or ost Bas Usage	sed Rat rates ir	Commission rule to the section in the sa the Port section of	o provide Unbu me manner as of this rate exh	they are applie ibit shall apply	ed to the Stand to all combination	Alone Unbun	port network e	lements excep	t for UNE C				Additional NR	Cs may
UNBUNDLED 1. Cos 2. Fea 3. End 4. The apply	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES St Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not Co also and are categorized accordingly.	and/or ost Bas Usage urrently	ed Rat rates ir Comb	Commission rule to the section in the sa the Port section of ined Combos. Fo	o provide Unbu ime manner as of this rate exh or Currently Co	they are applie ibit shall apply mbined Combe	ed to the Stand / to all combina os, the nonrecu	Alone Unbun	port network e	lements excep	t for UNE C				Additional NR	Cs may
UNBUNDLED 1. Cos 2. Fea 3. End 4. The apply 5. Ma	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES st Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not C also and are categorized accordingly. arket Rates for Unbundled Centrex Port/Loop Combination will	and/or ost Bas Usage urrently be neg	ed Rat rates ir Comb	Commission rule to the section in the sa the Port section of ined Combos. Fo	o provide Unbu ime manner as of this rate exh or Currently Co	they are applie ibit shall apply mbined Combe	ed to the Stand / to all combina os, the nonrecu	Alone Unbun	port network e	lements excep	t for UNE C				Additional NR	Cs may
UNBUNDLED 1. Cos 2. Fea 3. End 4. The apply 5. Ma UNE-F	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES st Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not Co also and are categorized accordingly. arket Rates for Unbundled Centrex Port/Loop Combination will P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only	and/or ost Bas Usage urrently be neg	ed Rat rates ir Comb	Commission rule to the section in the sa the Port section of ined Combos. Fo	o provide Unbu ime manner as of this rate exh or Currently Co	they are applie ibit shall apply mbined Combe	ed to the Stand / to all combina os, the nonrecu	Alone Unbun	port network e	lements excep	t for UNE C				Additional NR	Cs may
UNBUNDLED 1. Cos 2. Fea 3. End 4. The apply 5. Ma UNE-f 2-Wird	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES st Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not Cu also and are categorized accordingly. arket Rates for Unbundled Centrex Port/Loop Combination will P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only e VG Loop/2-Wire Voice Grade Port (Centrex) Combo	and/or ost Bas Usage urrently be neg	ed Rat rates ir Comb	Commission rule to the section in the sa the Port section of ined Combos. Fo	o provide Unbu ime manner as of this rate exh or Currently Co	they are applie ibit shall apply mbined Combe	ed to the Stand / to all combina os, the nonrecu	Alone Unbun	port network e	lements excep	t for UNE C				Additional NR	Cs may
UNBUNDLED 1. Cos 2. Fea 3. End 4. The apply 5. Ma UNE-f 2-Wird	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES st Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not Co also and are categorized accordingly. arket Rates for Unbundled Centrex Port/Loop Combination will P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only	and/or ost Bas Usage urrently be neg	ed Rat rates ir Comb	Commission rule to the section in the sa the Port section of ined Combos. Fo	o provide Unbu ime manner as of this rate exh or Currently Co	they are applie ibit shall apply mbined Combe	ed to the Stand / to all combina os, the nonrecu	Alone Unbun	port network e	lements excep	t for UNE C				Additional NR	Cs may
UNBUNDLED 1. Cos 2. Fea 3. End 4. The apply 5. Ma UNE-f 2-Wird	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES is Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not Ct also and are categorized accordingly. arket Rates for Unbundled Centrex Port/Loop Combination will P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only e VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design)	and/or ost Bas Usage urrently be neg	ed Rat rates ir Comb	Commission rule to the section in the sa the Port section of ined Combos. Fo	o provide Unbu ime manner as of this rate exh or Currently Co	they are applie ibit shall apply mbined Combe	ed to the Stand / to all combina os, the nonrecu	Alone Unbun	port network e	lements excep	t for UNE C				Additional NR	Cs may
UNBUNDLED 1. Cos 2. Fea 3. End 4. The apply 5. Ma UNE-f 2-Wird	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES st Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not Ct also and are categorized accordingly. arket Rates for Unbundled Centrex Port/Loop Combination will P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only e VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo	and/or ost Bas Usage urrently be neg	eed Rat rates ir Comb otiated	Commission rule to e section in the sa the Port section ined Combos. Fo on an Individual (UEP91	o provide Unbu ime manner as of this rate exh or Currently Co	they are applie ibit shall apply mbined Combo il further notic	ed to the Stand / to all combina os, the nonrecu	Alone Unbun	port network e	lements excep	t for UNE C				Additional NR	Cs may
UNBUNDLED 1. Cos 2. Fea 3. Enc 4. The apply 5. Ma UNE-F 2-Wird	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES st Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not C(also and are categorized accordingly. arket Rates for Unbundled Centrex Port/Loop Combination will P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only e VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design	and/or ost Bas Usage urrently be neg	ed Rat rates ir Comb	Commission rule to e section in the sa the Port section ined Combos. Fo on an Individual (o provide Unbu ime manner as of this rate exh or Currently Co	they are applie ibit shall apply mbined Combo il further notic	ed to the Stand / to all combina os, the nonrecu	Alone Unbun	port network e	lements excep	t for UNE C				Additional NR	Cs may
UNBUNDLED 1. Cos 2. Fea 3. Enc 4. The apply 5. Ma UNE-F 2-Wird	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES st Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not Cu valso and are categorized accordingly. arket Rates for Unbundled Centrex Port/Loop Combination will P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only e VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo	and/or ost Bas Usage urrently be neg	eed Rat rates ir comb otiated	Commission rule to e section in the sa ined Combos. Fo on an Individual (UEP91 UEP91	o provide Unbu ime manner as of this rate exh or Currently Co	they are applied ibit shall apply mbined Combo il further notic 14.18 18.01	ed to the Stand / to all combina os, the nonrecu	Alone Unbun	port network e	lements excep	t for UNE C				Additional NR	Cs may
UNBUNDLED 1. Cos 2. Fea 3. Enc 4. The apply 5. Ma UNE 1 2-Wirr UNE F	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES St Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not Cu also and are categorized accordingly. arket Rates for Unbundled Centrex Port/Loop Combination will P CENTREX - IAESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only e VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design	and/or ost Bas Usage urrently be neg	eed Rat rates ir Comb otiated	Commission rule to e section in the sa the Port section ined Combos. Fo on an Individual (UEP91	o provide Unbu ime manner as of this rate exh or Currently Co	they are applie ibit shall apply mbined Combo il further notic 14.18	ed to the Stand / to all combina os, the nonrecu	Alone Unbun	port network e	lements excep	t for UNE C				Additional NR	Cs may
UNBUNDLED 1. Cos 2. Fea 3. Enc 4. The apply 5. Ma UNE F UNE F	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES St Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not Ct also and are categorized accordingly. arket Rates for Unbundled Centrex Port/Loop Combination will P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only e VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Port/Loop Combination Rates (Design)	and/or ost Bas Usage urrently be neg	eed Rat rates ir comb otiated	Commission rule to e section in the sa ined Combos. Fo on an Individual (UEP91 UEP91	o provide Unbu ime manner as of this rate exh or Currently Co	they are applied ibit shall apply mbined Combo il further notic 14.18 18.01	ed to the Stand / to all combina os, the nonrecu	Alone Unbun	port network e	lements excep	t for UNE C				Additional NR	Cs may
UNBUNDLED 1. Cos 2. Fea 3. Enc 4. The apply 5. Ma UNE F UNE F	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES St Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not Ct also and are categorized accordingly. arket Rates for Unbundled Centrex Port/Loop Combination will P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only e VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design	and/or ost Bas Usage urrently be neg	ed Rat rates ir Comb otiated 1 2 3	Commission rule to e section in the sa the Port section of ined Combos. Fo on an Individual (UEP91 UEP91 UEP91	o provide Unbu ime manner as of this rate exh or Currently Co	they are applied ibit shall apply mbined Combo il further notic 14.18 18.01 23.02	ed to the Stand / to all combina os, the nonrecu	Alone Unbun	port network e	lements excep	t for UNE C				Additional NR	Cs may
UNBUNDLED 1. Cos 2. Fea 3. Enc 4. The apply 5. Ma UNE 1 2-Wirr UNE F	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES St Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport also and are categorized accordingly. arket Rates for Unbundled Centrex Port/Loop Combination will P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only e VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design	and/or ost Bas Usage urrently be neg	eed Rat rates ir comb otiated	Commission rule to e section in the sa ined Combos. Fo on an Individual (UEP91 UEP91	o provide Unbu ime manner as of this rate exh or Currently Co	they are applied ibit shall apply mbined Combo il further notic 14.18 18.01	ed to the Stand / to all combina os, the nonrecu	Alone Unbun	port network e	lements excep	t for UNE C				Additional NR	Cs may
UNBUNDLED 1. Cos 2. Fea 3. Enc 4. The apply 5. Ma UNE F UNE F	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES St Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not Ct also and are categorized accordingly. arket Rates for Unbundled Centrex Port/Loop Combination will P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only e VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design	and/or ost Bas Usage urrently be neg	ed Rat rates ir Comb otiated 1 2 3	Commission rule to e section in the sa the Port section of ined Combos. Fo on an Individual (UEP91 UEP91 UEP91	o provide Unbu ime manner as of this rate exh or Currently Co	they are applied ibit shall apply mbined Combo il further notic 14.18 18.01 23.02	ed to the Stand / to all combina os, the nonrecu	Alone Unbun	port network e	lements excep	t for UNE C				Additional NR	Cs may
UNBUNDLED 1. Cos 2. Fea 3. Enc 4. The apply 5. Ma UNE 1 2-Wirr UNE F	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES St Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not Ct also and are categorized accordingly. arket Rates for Unbundled Centrex Port/Loop Combination will P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only e VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design Port/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo	and/or ost Bas Usage urrently be neg	ted Rat rates ir Comb otiated 1 2 3 1	Commission rule to e section in the sa the Port section ined Combos. Fo on an Individual (UEP91 UEP91 UEP91 UEP91	o provide Unbu ime manner as of this rate exh or Currently Co	they are applied ibit shall apply mbined Combo il further notic 14.18 18.01 23.02 18.26	ed to the Stand / to all combina os, the nonrecu	Alone Unbun	port network e	lements excep	t for UNE C				Additional NR	Cs may
UNBUNDLED 1. Cos 2. Fea 3. Enc 4. The apply 5. Ma UNE F UNE F UNE F	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES St Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not Ct also and are categorized accordingly. arket Rates for Unbundled Centrex Port/Loop Combination will P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design	and/or ost Bas Usage urrently be neg	ted Rat rates ir Comb otiated 1 2 3 1	Commission rule to e section in the sa the Port section ined Combos. Fo on an Individual (UEP91 UEP91 UEP91 UEP91	o provide Unbu ime manner as of this rate exh or Currently Co	they are applied ibit shall apply mbined Combo il further notic 14.18 18.01 23.02 18.26	ed to the Stand / to all combina os, the nonrecu	Alone Unbun	port network e	lements excep	t for UNE C				Additional NR	Cs may
UNBUNDLED 1. Cos 2. Fea 3. Enc 4. The apply 5. Ma UNE F UNE F UNE F	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES St Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not CC also and are categorized accordingly. arket Rates for Unbundled Centrex Port/Loop Combination will P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only e VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design	and/or ost Bas Usage urrently be neg	ed Rat rates ir Comb otiated 1 2 3 1 2 3	Commission rule to commission rule to a the Port section of ined Combos. For on an Individual (UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	o provide Unbu me manner as of this rate exh or Currently Co Case Basis, unt	they are applied ibit shall apply mbined Combo il further notic 14.18 18.01 23.02 18.26 23.33 29.98	ed to the Stand / to all combina os, the nonrecu	Alone Unbun	port network e	lements excep	t for UNE C				Additional NR	Cs may
UNBUNDLED 1. Cos 2. Fea 3. Enc 4. The apply 5. Ma UNE F UNE F UNE F	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES St Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not Ct also and are categorized accordingly. arket Rates for Unbundled Centrex Port/Loop Combination will P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only e VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire Voice Grade Loop (SL 1) - Zone 1	and/or ost Bas Usage urrently be neg	ed Rat rates ir comb otiated 1 2 3 1 2 3 1 1	Commission rule to e section in the sa ined Combos. Fo on an Individual (UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1	they are applied ibit shall apply mbined Combo il further notice 14.18 18.01 23.02 18.26 23.33 29.98 12.48	ed to the Stand / to all combina os, the nonrecu	Alone Unbun	port network e	lements excep	t for UNE C				Additional NR	Cs may
UNBUNDLED 1. Cos 2. Fea 3. Enc 4. The apply 5. Ma UNE F UNE F UNE F	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES st Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not Cu also and are categorized accordingly. arket Rates for Unbundled Centrex Port/Loop Combination will P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2	and/or ost Bas Usage urrently be neg	eed Rati rates ir Comb tiated 1 1 2 3 3 1 1 2 3 1 1 2	Commission rule t e section in the sa ined Combos. Fc on an Individual (UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1	they are applied ibit shall apply mbined Combe il further notic 14.18 18.01 23.02 18.26 23.33 29.98 12.48 16.31	ed to the Stand / to all combina os, the nonrecu	Alone Unbun	port network e	lements excep	t for UNE C				Additional NR	Cs may
UNBUNDLED 1. Cos 2. Fea 3. Enc 4. The apply 5. Ma UNE F UNE F UNE F	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES St Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not Ct also and are categorized accordingly. arket Rates for Unbundled Centrex Port/Loop Combination will P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only e VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3	and/or ost Bas Usage urrently be neg	eed Ratirates in a combined of the second se	Commission rule to commission rule to e section in the sa the Port section of ined Combos. For on an Individual (UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS1 UECS1	they are applied ibit shall apply mbined Combo il further notic 14.18 18.01 23.02 18.26 23.33 29.98 12.48 16.31 21.32	ed to the Stand / to all combina os, the nonrecu	Alone Unbun	port network e	lements excep	t for UNE C				Additional NR	Cs may
UNBUNDLED 1. Cos 2. Fea 3. Enc 4. The apply 5. Ma UNE F UNE F UNE F	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES St Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not CC also and are categorized accordingly. arket Rates for Unbundled Centrex Port/Loop Combination will P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only e VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1	and/or ost Bas Usage urrently be neg	eed Ratirates ir rates ir Comb otiated 1 1 2 3 1 1 2 3 1	Commission rule to commission rule to e section in the sa the Port section of ined Combos. Fo on an Individual (UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS2	they are applied ibit shall apply mbined Combo il further notic 14.18 18.01 23.02 18.26 23.33 29.98 12.48 16.31 21.32 16.56	ed to the Stand / to all combina os, the nonrecu	Alone Unbun	port network e	lements excep	t for UNE C				Additional NR	Cs may
UNBUNDLED 1. Cos 2. Fea 3. Enc 4. The apply 5. Ma UNE F UNE F UNE F	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES st Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not Cu also and are categorized accordingly. arket Rates for Unbundled Centrex Port/Loop Combination will P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 2	and/or ost Bas Usage urrently be neg	eed Ratirates in a combination of the combination o	Commission rule to e section in the sa ined Combos. Fo on an Individual (UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS2 UECS2	they are applied ibit shall apply mbined Combo il further notic 14.18 18.01 23.02 18.26 23.33 29.98 12.48 16.31 21.32	ed to the Stand / to all combina os, the nonrecu	Alone Unbun	port network e	lements excep	t for UNE C				Additional NR	Cs may
UNBUNDLED 1. Cos 2. Fea 3. Enc 4. The apply 5. Ma UNE F UNE F UNE F	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES st Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not Ct also and are categorized accordingly. arket Rates for Unbundled Centrex Port/Loop Combination will P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only e VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3	and/or ost Bas Usage urrently be neg	eed Ratirates in a combination of the combination o	Commission rule to commission rule to e section in the sa the Port section of ined Combos. Fo on an Individual (UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS2	they are applied they are applied tibit shall apply mbined Combe il further notic 14.18 18.01 23.02 18.26 23.33 29.98 12.48 16.56 21.63	ed to the Stand / to all combina os, the nonrecu	Alone Unbun	port network e	lements excep	t for UNE C					Cs may
UNBUNDLED 1. Co: 2. Fee 3. Enc 4. The apply 5. Ma UNE-f 2-Wirf UNE f UNE f UNE f	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES st Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not Cu also and are categorized accordingly. arket Rates for Unbundled Centrex Port/Loop Combination will P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 Ports ates (Except North Carolina and Sout Carolina)	and/or ost Bas Usage urrently be neg	eed Ratirates in a combination of the combination o	Commission rule to e section in the sat inte Port section of ined Combos. Fo on an Individual (UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS1 UECS2 UECS2 UECS2	they are applied they are applied to the shall apply mbined Combo il further notice 14.18 18.01 23.02 18.26 23.33 29.98 12.48 16.31 21.32 16.56 21.63 28.28	ed to the Stand to all combina os, the nonrecu e.	Alone Unbun tions of loop rring charges	port network el shall be those	identified in t	t for UNE C	rring - Curre	ently Combine		Additional NR	Cs may
UNBUNDLED 1. Co: 2. Fee 3. Enc 4. The apply 5. Ma UNE-f 2-Wirf UNE f UNE f UNE f	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES st Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not Ct also and are categorized accordingly. arket Rates for Unbundled Centrex Port/Loop Combination will P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 Ports ates (Except North Carolina and Sout Carolina) 2-Wire Voice Grade Port (Centrex) Basic Local Area	and/or ost Bas Usage urrently be neg	eed Ratirates in a combination of the combination o	Commission rule to e section in the sa ined Combos. Fo on an Individual (UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS2 UECS2	they are applied they are applied tibit shall apply mbined Combe il further notic 14.18 18.01 23.02 18.26 23.33 29.98 12.48 16.56 21.63	ed to the Stand / to all combina os, the nonrecu	Alone Unbun	port network e	lements excep	t for UNE C					Cs may
UNBUNDLED 1. Co: 2. Fea 3. Enc 4. The apply 5. Ma UNE-1 2-Wirr UNE F	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES st Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not Ct also and are categorized accordingly. arket Rates for Unbundled Centrex Port/Loop Combination will P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only e VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3 Ports ates (Except North Carolina and Sout Carolina) 2-Wire Voice Grade Port (Centrex) Basic Local Area 2-Wire Voice Grade Port (Centrex 800 termination)Basic Local	and/or ost Bas Usage urrently be neg	eed Ratirates in rate	Commission rule to commission rule to e section in the sa the Port section of ined Combos. For on an Individual (UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS2 UECS2 UECS2 UECS2 UECS2	they are applied ibit shall apply mbined Combo il further notic 14.18 18.01 23.02 18.26 23.33 29.98 16.31 21.32 16.56 21.63 28.28 	ed to the Stand- r to all combina oos, the nonrecu e. 	Alone Unbun tions of loop rring charges	port network el shall be those	identified in t	t for UNE C	rring - Curre	27.03		Additional NR	Cs may
UNBUNDLED 1. Co: 2. Fee 3. Enc 4. The apply 5. Ma UNE-f 2-Wirf UNE f UNE f UNE f	All Features Available CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES CENTREX PORT/LOOP COMBINATIONS - COST BASED RATES st Based Rates are applied where BellSouth is required by FCC atures shall apply to the Unbundled Port/Loop Combination - C d Office and Tandem Switching Usage and Common Transport e first and additional Port nonrecurring charges apply to Not Ct also and are categorized accordingly. arket Rates for Unbundled Centrex Port/Loop Combination will P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 Ports ates (Except North Carolina and Sout Carolina) 2-Wire Voice Grade Port (Centrex) Basic Local Area	and/or ost Bas Usage urrently be neg	eed Ratirates in rate	Commission rule to e section in the sat inte Port section of ined Combos. Fo on an Individual (UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS1 UECS2 UECS2 UECS2	they are applied they are applied to the shall apply mbined Combo il further notice 14.18 18.01 23.02 18.26 23.33 29.98 12.48 16.31 21.32 16.56 21.63 28.28	ed to the Stand to all combina os, the nonrecu e.	Alone Unbun tions of loop rring charges	port network el shall be those	identified in t	t for UNE C	rring - Curre	ently Combine		Additional NR	Cs may

IRONDLED I	NETWORK ELEMENTS - Tennessee		1										Attachment:			bit: B
TEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
	Mine Maine One la Dest (October Grand 1970 October Mine						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMA
	-Wire Voice Grade Port (Centrex from diff Serving Wire center)2 Basic Local Area			UEP91	UEPYM	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			UEF91	UEPTIVI	1.70	22.14	15.25	0.43	3.91		30.69	7.03			
	erm - Basic Local Area			UEP91	UEPYZ	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	Wire Voice Grade Port terminated in on Megalink or equivalent															
	Basic Local Area			UEP91	UEPY9	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	-Wire Voice Grade Port Terminated on 800 Service Term -															
	asic Local Area			UEP91	UEPY2	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	A, MS, & TN Only															
	-Wire Voice Grade Port (Centrex)			UEP91	UEPQA	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	Wire Voice Grade Port (Centrex 800 termination)			UEP91	UEPQB	1.70	22.14	15.25	8.45	3.91	ļ	30.89	7.03			
	-Wire Voice Grade Port (Centrex with Caller ID)1 -Wire Voice Grade Port (Centrex from diff Serving Wire			UEP91	UEPQH	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	enter)2			UEP91	UEPQM	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			02101		1.70	22.14	15.25	0.40	5.91	-	30.09	1.03			-
	erm			UEP91	UEPQZ	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
								10.20	0.40	0.01		00.00			İ	
2-	Wire Voice Grade Port terminated in on Megalink or equivalent			UEP91	UEPQ9	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	Wire Voice Grade Port Terminated on 800 Service Term			UEP91	UEPQ2	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
Local Swi																
	entrex Intercom Funtionality, per port			UEP91	URECS	0.6381										
	mber Portability															
	ocal Number Portability (1 per port)			UEP91	LNPCC	0.35										
Features				115504									=			
	Il Standard Features Offered, per port			UEP91 UEP91	UEPVF UEPVS	0.00	433.78					30.89	7.03			
	Il Select Features Offered, per port Il Centrex Control Features Offered, per port			UEP91 UEP91	UEPVS	0.00	433.78					30.89 30.89	7.03			
NARS	il Centrex Control Features Offered, per port			UEF91	UEFVC	0.00						30.69	7.03			
	nbundled Network Access Register - Combination			UEP91	UARCX	0.00	0.00	0.00				30.89	7.03			
	nbundled Network Access Register - Indial			UEP91	UAR1X	0.00	0.00	0.00				30.89	7.03			
	nbundled Network Access Register - Outdial			UEP91	UAROX	0.00	0.00	0.00				30.89	7.03			
	neous Terminations															
2-Wire Tru																
	runk Side Terminations, each			UEP91	CENA6	8.78	22.14	15.25	8.45	3.91		30.89	7.03			
	e Channel Mileage - 2-Wire															
	teroffice Channel Facilities Termination - Voice Grade			UEP91	M1GBC	18.58	22.14	15.25	8.45	3.91		30.89	7.03			
	teroffice Channel mileage, per mile or fraction of mile			UEP91	M1GBM	0.0174										
	Activations (DS0) Centrex Loops on Channelized DS1 Servic	e									-					
	nel Bank Feature Activations eature Activation on D-4 Channel Bank Centrex Loop Slot			UEP91	1PQWS	0.66										
Fe	eature Activation on D=4 Channel Dank Centrex Loop SIO			02131	11 4103	0.00			 		t					
Fe	eature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP91	1PQW6	0.66										
	eature Activation on D-4 Channel Bank FX Trunk Side Loop					0.00	1				1				1	1
	lot			UEP91	1PQW7	0.66										
	eature Activation on D-4 Channel Bank Centrex Loop Slot -													İ		1
Di	ifferent Wire Center			UEP91	1PQWP	0.66										
	eature Activation on D-4 Channel Bank Private Line Loop Slot			UEP91	1PQWV	0.66					L					L
	eature Activation on D-4 Channel Bank Tjie Line/Trunk Loop				100110											
	lot			UEP91	1PQWQ	0.66					<u>├</u> ──					ł
	eature Activation on D-4 Channel Bank WATS Loop Slot urring Charges (NRC) Associated with UNE-P Centrex			UEP91	1PQWA	0.66			├							
	conversion - Currently Combined Switch-As-Is with allowed								<u>├</u>							
	hanges, per port			UEP91	USAC2		1.03	0.29				30.89	7.03			
	ew Centrex Standard Common Block			UEP91	MIACS	0.00	658.60	0.23			1	30.89	7.03			1
	ew Centrex Customized Common Block			UEP91	MIACC	0.00	658.60				1	30.89	7.03	1	1	1
	econdary Block, per Block			UEP91	M2CC1	0.00	73.55					30.89	7.03			1
	AR Establishment Charge, Per Occasion			UEP91	URECA		68.57				1	30.89	7.03	ĺ	İ	1
	ENTREX - 5ESS (Valid in All States)															

JNUL	ED NETWORK ELEMENTS - Tennessee					-							Attachment:			ibit: B
GORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Charge
						Rec	Nonrecurring		Nonrecurring					Rates(\$)	-	
						neo	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMA
	e VG Loop/2-Wire Voice Grade Port (Centrex) Combo															
UNE F	Port/Loop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design	-	1	UEP95		14.18										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		2	UEP95		18.01										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Non-Design		3	UEP95		23.02										
UNE F	Port/Loop Combination Rates (Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	Design		1	UEP95		18.26										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		2	UEP95		23.33										
1	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		4	0LF 90		20.00	├				1		ł		<u> </u>	
1	Design	1	3	UEP95		29.98										1
UNE	Loop Rate		3	0 L F 30		29.98	├					t			 	1
ONE L	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP95	UECS1	12.48	├					t			 	
	2-Wire Voice Grade Loop (SL 1) - Zone 1		2	UEP95	UECS1	12.40						1		-		
-	2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP95	UECS1	21.32									-	-
	2-Wire Voice Grade Loop (SL 2) - Zone 3		1	UEP95	UECS2	16.56									-	-
	2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP95	UECS2	21.63									-	-
	2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3		2	UEP95 UEP95	UECS2	21.63										
			3	UEP95	UEC52	28.28										
All Sta	Port Rate															-
All Sta						4 70	00.44	45.05	0.45	0.01		00.00	7.00			
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP95	UEPYA UEPYB	1.70	22.14	15.25	8.45	3.91		30.89 30.89	7.03			-
	2-Wire Voice Grade Port (Centrex 800 termination) 2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area			UEP95 UEP95	UEPYB	1.70 1.70	22.14 22.14	15.25	8.45 8.45	3.91 3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex from diff Serving Wire			UEF95		1.70	22.14	15.25		3.91			7.03			
	Center)2 Basic Local Area 2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			UEP95	UEPYM	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	Term - Basic Local Area			UEP95	UEPYZ	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port terminated in on Megalink or equivalent - Basic Local Area			UEP95	UEPY9	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port Terminated on 800 Service Term - Basic Local Area			UEP95	UEPY2	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
AL, K	Y, LA, MS, SC, & TN Only															1
	2-Wire Voice Grade Port (Centrex)			UEP95	UEPQA	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPQB	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP95	UEPQH	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP95	UEPQM	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP95	UEPQZ	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP95	UEPQ9	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
1	2-Wire Voice Grade Port Terminated on 800 Service Term	1	1	UEP95	UEPQ2	1.70	22.14	15.25	8.45	3.91	1	30.89	7.03		1	1
FL & (GA Only	1	1	İ							1				l	1
	Switching	1														1
1	Centrex Intercom Funtionality, per port	1	1	UEP95	URECS	0.6381	i i				1	1	ĺ		l	1
Local	Number Portability	1														1
	Local Number Portability (1 per port)	1	1	UEP95	LNPCC	0.35					1				1	1
Featu		1	1								1					1
	All Standard Features Offered, per port			UEP95	UEPVF	0.00					1	30.89	7.03			1
1	All Select Features Offered, per port	1	1	UEP95	UEPVS	0.00	433.78				1	30.89	7.03		1	1
1	All Centrex Control Features Offered, per port	1	1	UEP95	UEPVC	0.00					1	30.89	7.03		1	1
NARS		1	1								1				1	1
1	Unbundled Network Access Register - Combination	1	1	UEP95	UARCX	0.00	0.00	0.00			1	30.89	7.03		l	1
	Unbundled Network Access Register - Indial	1	1	UEP95	UAR1X	0.00	0.00	0.00			1	30.89	7.03		1	1

	ETWORK ELEMENTS - Tennessee	1	1								Cure Curt	0	Attachment:			bit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN		SOMAN	SOMAN	SOMAN
	bundled Network Access Register - Outdial			UEP95	UAROX	0.00	0.00	0.00				30.89	7.03			L
	ous Terminations															
2-Wire Trur					CEND6	8.78	47.75	47.01	9.21	8.47		30.89	7.03			l
	nk Side Terminations, each ital (1.544 Megabits)			UEP95	CEND6	8.78	47.75	47.01	9.21	8.47		30.89	7.03			l
	1 Circuit Terminations, each			UEP95	M1HD1	35.55	75.93	38.15				30.89	7.03			ł
	0 Channels Activated, each			UEP95	M1HD0	0.00	108.67	30.15				30.89	7.03			
	Channel Mileage - 2-Wire			021 00	NITIE O	0.00	100.07					00.00	7.00			
	eroffice Channel Facilities Termination			UEP95	MIGBC	18.58	22.14	15.25	8.45	3.91		30.89	7.03			
	eroffice Channel mileage, per mile or fraction of mile			UEP95	MIGBM	0.0174										
	tivations (DS0) Centrex Loops on Channelized DS1 Servic	e		02.00		0.0111										
	Bank Feature Activations	ľ.			1 1											
	ature Activation on D-4 Channel Bank Centrex Loop Slot			UEP95	1PQWS	0.66										
- 1 - f																
	ature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP95	1PQW6	0.66										i i
	ature Activation on D-4 Channel Bank FX Trunk Side Loop															
Slot	t			UEP95	1PQW7	0.66										Ì
Fea	ature Activation on D-4 Channel Bank Centrex Loop Slot -															
Diffe	erent Wire Center			UEP95	1PQWP	0.66										
Fea	ature Activation on D-4 Channel Bank Private Line Loop Slot			UEP95	1PQWV	0.66										
	ature Activation on D-4 Channel Bank Tjie Line/Trunk Loop															
Slot				UEP95	1PQWQ	0.66										
	ature Activation on D-4 Channel Bank WATS Loop Slot			UEP95	1PQWA	0.66										
	ring Charges (NRC) Associated with UNE-P Centrex															
	C Conversion Currently Combined Switch-As-Is with allowed															Ì
	anges, per port			UEP95	USAC2		1.03	0.29				30.89	7.03			l
	w Centrex Standard Common Block			UEP95	M1ACS	0.00	658.60					30.89	7.03			L
	w Centrex Customized Common Block			UEP95	M1ACC	0.00	658.60					30.89	7.03			
	R Establishment Charge, Per Occasion TREX - DMS100 (Valid in All States)			UEP95	URECA	0.00	68.57					30.89	7.03			l
	Loop/2-Wire Voice Grade Port (Centrex) Combo															<u> </u>
	Loop Combination Rates (Non-Design)															
	Vire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	n-Design		1	UEP9D		14.18										
	Vire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			01 30		14.10										
	n-Design		2	UEP9D		18.01										1
	Vire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		-	02.05		10.01										
	n-Design		3	UEP9D		23.02										Ì
	.oop Combination Rates (Design)															
	Vire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	-														
Des	sign		1	UEP9D		18.26										Ì
2-W	Vire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	sign		2	UEP9D		23.33										
	Vire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
Des			3	UEP9D		29.98										
UNE Loop																
	Vire Voice Grade Loop (SL 1) - Zone 1		1	UEP9D	UECS1	12.48										L
	Vire Voice Grade Loop (SL 1) - Zone 2	I	2	UEP9D	UECS1	16.31										ļ
	Vire Voice Grade Loop (SL 1) - Zone 3		3	UEP9D	UECS1	21.32										l
	Vire Voice Grade Loop (SL 2) - Zone 1		1	UEP9D	UECS2	16.56										ļ
	Vire Voice Grade Loop (SL 2) - Zone 2	l	2	UEP9D	UECS2	21.63										L
	Vire Voice Grade Loop (SL 2) - Zone 3		3	UEP9D	UECS2	28.28										l
UNE Port R ALL STATE																l
	LS Vire Voice Grade Port (Centrex) Basic Local Area			UEP9D	UEPYA	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	Vire Voice Grade Port (Centrex) Basic Local Area			ULFSD	UEPTA	1.70	22.14	15.25	8.45	3.91		30.89	7.03	ł	ł	
	a			UEP9D	UEPYB	1.70	22.14	15.25	8.45	3.91	1	30.89	7.03			i i

	ED NETWORK ELEMENTS - Tennessee												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrecurring		Nonrecurring		0.01150	001411		Rates(\$)	0.011.011	001141
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3Basic Local				-		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Area			UEP9D	UEPYC	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3Basic Local Area			UEP9D	UEPYD	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5209))3 Basic Local Area			UEP9D	UEPYE	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5112))3 Basic Local															
	Area 2-Wire Voice Grade Port (Centrex / EBS-M5312))3Basic Local			UEP9D	UEPYF	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	Area			UEP9D	UEPYG	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5008))3 Basic Local Area			UEP9D	UEPYT	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5208))3 Basic Local			UEP9D	UEPYU	1.70	22.44	15.25	8.45	2.04		30.89	7.03			
	Area 2-Wire Voice Grade Port (Centrex / EBS-M5216))3 Basic Local						22.14			3.91						
	Area 2-Wire Voice Grade Port (Centrex / EBS-M5316))3 Basic Local			UEP9D	UEPYV	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	Area 2-Wire Voice Grade Port (Centrex with Caller ID) Basic Local			UEP9D	UEPY3	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	Area 2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp			UEP9D	UEPYH	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	Indication))3 Basic Local Area			UEP9D	UEPYW	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication))3 Basic Local Area			UEP9D	UEPYJ	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center) 2 Basic Local Area			UEP9D	UEPYM	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3 Basic Local Area			UEP9D	UEPYO	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3 Basic Local Area			UEP9D	UEPYP	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3 Basic Local Area			UEP9D	UEPYQ	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3															
	Basic Local Area 2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3			UEP9D	UEPYR	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	Basic Local Area			UEP9D	UEPYS	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3 Basic Local Area			UEP9D	UEPY4	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3 Basic Local Area			UEP9D	UEPY5	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3 Basic Local Area			UEP9D	UEPY6	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3 Basic Local Area			UEP9D	UEPY7	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service				-											
├── ┼──	Term 2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9D	UEPYZ	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
├──	Basic Local Area 2-Wire Voice Grade Port Terminated on 800 Service Term Basic	$\left \right $		UEP9D	UEPY9	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	Local Area			UEP9D	UEPY2	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
AL, K	Y, LA, MS, SC, & TN Only 2-Wire Voice Grade Port (Centrex)			UEP9D	UEPQA	1.70	22.14	15.25	8.45	3.91	1	30.89	7.03			
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP9D	UEPQB	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3			UEP9D	UEPQC	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3			UEP9D	UEPQD	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5209)3			UEP9D	UEPQE	1.70	22.14	15.25	8.45	3.91	L	30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5112)3	\mid		UEP9D	UEPQF	1.70	22.14	15.25	8.45	3.91	ļ	30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5312)3			UEP9D UEP9D	UEPQG UEPQT	1.70	22.14	15.25	8.45 8.45	3.91		30.89 30.89	7.03 7.03			
NRONDLE	D NETWORK ELEMENTS - Tennessee		1								-	-	Attachment:			bit: B
---------	--	-------------	----------	----------------	-----------	-----------------	--------------	-----------	--------------	------------	---	-----------------------	-------------	--	---	--
TEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Submitted Manually		Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
						D	Nonrecurring		Nonrecurring	Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port (Centrex / EBS-M5208)3			UEP9D	UEPQU	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5216)3			UEP9D	UEPQV	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5316)3			UEP9D	UEPQ3	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex with Caller ID)			UEP9D	UEPQH	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp															
	Indication)3			UEP9D	UEPQW	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication)3			UEP9D	UEPQJ	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)															
	2			UEP9D	UEPQM	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			UEP9D	UEPQO	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3		ļ	UEP9D	UEPQP	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3			UEP9D	UEPQQ	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
						. =0							=			
_	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			UEP9D	UEPQR	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	O MEN MAIN OF A DIST (OF ANY A TWO ON O (EDO MEDIO)O O					4 70	00.44	15.05	0.45	0.04		00.00	7.00			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3			UEP9D	UEPQS	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2 Mine Maine Crede Dart (Contraw/differ CMC (EDC ME000)2, 2			UEP9D	UEPQ4	4 70	00.44	45.05	8.45	2.01		30.89	7.02			
_	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3			UEP9D	UEPQ4	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2 Wire Voice Crede Bert (Centrey/differ SWC /EBS ME208)2 2			UEP9D	UEPQ5	1.70	22.14	15.05	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3			UEP9D	UEPQ5	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3			UEP9D	UEPQ6	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wile Voice Glade Fort (Centrex differ SWC /LB3-W5210)2, 3			ULF 9D	ULFQU	1.70	22.14	15.25	0.45	3.91		30.09	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			UEP9D	UEPQ7	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			OLF9D	ULFQI	1.70	22.14	15.25	0.45	3.91		30.09	7.03			
	Term			UEP9D	UEPQZ	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
				OEI OD	ULI QL	1.70	22.14	10.20	0.40	0.01		00.00	1.00			
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9D	UEPQ9	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP9D	UEPQ2	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
Local S	Switching								0.10							
	Centrex Intercom Funtionality, per port			UEP9D	URECS	0.6381										
Local I	Number Portability															
	Local Number Portability (1 per port)			UEP9D	LNPCC	0.35										
Feature	es															
	All Standard Features Offered, per port			UEP9D	UEPVF	0.00						30.89	7.03			
	All Select Features Offered, per port			UEP9D	UEPVS	0.00	433.78					30.89	7.03			
	All Centrex Control Features Offered, per port			UEP9D	UEPVC	0.00						30.89	7.03			
NARS																
	Unbundled Network Access Register - Combination			UEP9D	UARCX	0.00	0.00	0.00				30.89	7.03			
	Unbundled Network Access Register - Inward			UEP9D	UAR1X	0.00	0.00	0.00				30.89	7.03			
	Unbundled Network Access Register - Outdial	-		UEP9D	UAROX	0.00	0.00	0.00				30.89	7.03			
	aneous Terminations															
2-Wire	Trunk Side															
	Trunk Side Terminations, each			UEP9D	CEND6	8.78	22.14	15.25	8.45	3.91		30.89	7.03			
4-Wire	Digital (1.544 Megabits)															
	DS1 Circuit Terminations, each			UEP9D	M1HD1	35.55	75.93	38.15			l	30.89	7.03			
Interes	DS0 Channels Activiated per Channel			UEP9D	M1HDO	0.00	108.67					30.89	7.03			
interof	fice Channel Mileage - 2-Wire				MICDO	40.50	00.4.4	45.05	8.45	0.01		20.00	7.00			
_	Interoffice Channel Facilities Termination Interoffice Channel mileage, per mile or fraction of mile			UEP9D UEP9D	MIGBC	18.58 0.0174	22.14	15.25	8.45	3.91		30.89	7.03			
Foature	Interoffice Channel mileage, per mile or fraction of mile e Activations (DS0) Centrex Loops on Channelized DS1 Servic	•		ULF9D	IVIIGBIVI	0.0174	├				1					
	annel Bank Feature Activations	e			+											
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9D	1PQWS	0.66	├				1					
			<u> </u>		IF QWO	0.00	├									
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9D	1PQW6	0.66										
					11 00110	0.00										<u> </u>
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop															

JNBUNDLE	D NETWORK ELEMENTS - Tennessee										_		Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge -	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
						Rec	Nonrecurring		Nonrecurring	g Disconnect				Rates(\$)	•	
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -															
	Different Wire Center			UEP9D	1PQWP	0.66										
	Fasture Astigation on D.4 Channel Bank Bringto Line Lang Clat			UEP9D	1PQWV	0.66										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot Feature Activation on D-4 Channel Bank Tije Line/Trunk Loop			UEP9D	IPQWV	0.66										
	Slot			UEP9D	1PQWQ	0.66										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9D	1PQWA	0.66										
Non-R	ecurring Charges (NRC) Associated with UNE-P Centrex															
	NRC Conversion Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP9D	USAC2		1.03	0.29				30.89	7.03			
	New Centrex Standard Common Block			UEP9D	M1ACS	0.00	658.60					30.89	7.03			
	New Centrex Customized Common Block		L	UEP9D	M1ACC	0.00	658.60				ļ	30.89	7.03			
	NAR Establishment Charge, Per Occasion			UEP9D	URECA		68.57					30.89	7.03			
	CENTREX - EWSD (Valid in AL, FL, KY, LA, MS & TN) VG Loop/2-Wire Voice Grade Port (Centrex) Combo										+					
	ort/Loop Combination Rates (Non-Design)															
UNE P	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -										1					
	Non-Design		1	UEP9E		14.18										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		· ·	OEI OE		14.10										
	Non-Design		2	UEP9E		18.01										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Non-Design		3	UEP9E		23.02										
UNE P	ort/Loop Combination Rates (Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	Design		1	UEP9E		18.26										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design		2	UEP9E		23.33					-					
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		3	UEP9E		29.98										
	oop Rate		3	ULFBL		29.90					1					
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9E	UECS1	12.48										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9E	UECS1	16.31										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP9E	UECS1	21.32										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9E	UECS2	16.56										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9E	UECS2	21.63										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9E	UECS2	28.28										
	ort Rate															
AL, FL	., KY, LA, MS, & TN only					1.70	00.44	45.05	0.45	2.04	-	20.00	7.00			
	2-Wire Voice Grade Port (Centrex) Basic Local Area 2-Wire Voice Grade Port (Centrex 800 termination)Basic Local			UEP9E	UEPYA	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	Area		1	UEP9E	UEPYB	1.70	22.14	15.25	8.45	3.91	1	30.89	7.03			
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local		1	OLI 3L	OLITE	1.70	22.14	15.25	0.43	5.91		30.03	7.05			
	Area			UEP9E	UEPYH	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex from diff Serving Wire															
	Center)2 Basic Local Area			UEP9E	UEPYM	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term - Basic Local Area		<u> </u>	UEP9E	UEPYZ	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port terminated in on Megalink or equivalent		1						a		1	cc c-				
	- Basic Local Area		<u> </u>	UEP9E	UEPY9	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port Terminated on 800 Service Term - Basic Local Area		1	UEP9E	UEPY2	1.70	22.14	15.25	8.45	3.91	1	30.89	7.03			
	/, LA, MS, & TN Only			OLL'AE	UEFTZ	1.70	22.14	15.25	0.45	3.91	+	30.89	7.03			
~ <u> </u>	2-Wire Voice Grade Port (Centrex)		1	UEP9E	UEPQA	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex 800 termination)		1	UEP9E	UEPQB	1.70	22.14	15.25	8.45	3.91	1	30.89	7.03	1	1	
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP9E	UEPQH	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex from diff Serving Wire															
	Center)2			UEP9E	UEPQM	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service						Ι Τ									
	Term			UEP9E	UEPQZ	1.70	22.14	15.25	8.45	3.91		30.89	7.03			I

BUNDLE	D NETWORK ELEMENTS - Tennessee			r								1	Attachment:			bit: B
EGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
						. =0							=			
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9E UEP9E	UEPQ9 UEPQ2	1.70	22.14 22.14	15.25 15.25	8.45 8.45	3.91 3.91	-	30.89 30.89	7.03 7.03			
Local	2-Wire Voice Grade Port Terminated on 800 Service Term Switching			UEP9E	UEPQ2	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
LUCAI	Centrex Intercom Funtionality, per port			UEP9E	URECS	0.6381										
l ocal	Number Portability		1	OLI 3L	UNLOG	0.0001										
	Local Number Portability (1 per port)			UEP9E	LNPCC	0.35										
Featu																
	All Standard Features Offered, per port			UEP9E	UEPVF	0.00						30.89	7.03			
	All Select Features Offered, per port			UEP9E	UEPVS	0.00	433.78					30.89	7.03			
1	All Centrex Control Features Offered, per port		1	UEP9E	UEPVC	0.00						30.89	7.03			
NARS																
	Unbundled Network Access Register - Combination			UEP9E	UARCX	0.00	0.00	0.00				30.89	7.03			
	Unbundled Network Access Register - Indial			UEP9E	UAR1X	0.00	0.00	0.00				30.89	7.03			
	Unbundled Network Access Register - Outdial			UEP9E	UAROX	0.00	0.00	0.00				30.89	7.03			
	Ilaneous Terminations															
2-Wire	Trunk Side															
	Trunk Side Terminations, each			UEP9E	CEND6	8.78	22.14	15.25	8.45	3.91		30.89	7.03			
4-Wire	e Digital (1.544 Megabits)															
	DS1 Circuit Terminations, each			UEP9E	M1HD1	35.55	75.93	38.15				30.89	7.03			
	DS0 Channel Activated Per Channel			UEP9E	M1HDO	0.00	108.67					30.89	7.03			
Intero	ffice Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination			UEP9E	MIGBC	18.58	22.14	15.25	8.45	3.91		30.89	7.03			
	Interoffice Channel mileage, per mile or fraction of mile			UEP9E	MIGBM	0.0174										
	re Activations (DS0) Centrex Loops on Channelized DS1 Service	e														
D4 Ch	annel Bank Feature Activations				1.50110											
_	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9E	1PQWS	0.66										
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9E	10014/0	0.66										
_	Feature Activation on D-4 Channel Bank FX Time Side Loop Stot			UEP9E	1PQW6	0.66										
	Slot			UEP9E	1PQW7	0.66										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -			UEP9E	TPQW7	0.66										
	Different Wire Center			UEP9E	1PQWP	0.66										
			-	ULFØL	IFQVF	0.00										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9E	1PQWV	0.66										
-	Feature Activation on D-4 Channel Bank Filvate Line Loop Slot		1		IFQVVV	0.00					-					
	Slot		1	UEP9E	1PQWQ	0.66										
-	Feature Activation on D-4 Channel Bank WATS Loop Slot		1	UEP9E	1PQWQ	0.66										
Non-R	ecurring Charges (NRC) Associated with UNE-P Centrex		1			0.00					t					
	NRC Conversion Currently Combined Switch-As-Is with allowed		1	1	1 1						1		1		1	
	changes, per port		1	UEP9E	USAC2		1.03	0.29				30.89	7.03			1
	New Centrex Standard Common Block		1	UEP9E	MIACS	0.00	658.60	0.20	i İ		1	30.89	7.03		ĺ	
	New Centrex Customized Common Block		1	UEP9E	M1ACC	0.00	658.60		i İ		1	30.89	7.03		ĺ	
1	NAR Establishment Charge, Per Occasion		1	UEP9E	URECA	0.00	68.57					30.89	7.03			
UNE-F	CENTREX - DCO - Valid in AL, KY, LA, MS, & TN)		1													
2-Wire	VG Loop/2-Wire Voice Grade Port (Centrex) Combo															
	ort/Loop Combination Rates (Non-Design)															
UNE F																
UNEF	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -				1	14.18										
UNEF	Non-Design		1	UEP93												
	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		1 2	UEP93 UEP93		18.01										
	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		2	UEP93		18.01										
	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design															
	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design ort/Loop Combination Rates (Design)		2	UEP93		18.01										
	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design ort/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -		2	UEP93 UEP93		18.01 23.02										
	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design ort/Loop Combination Rates (Design)		2	UEP93		18.01										

UNBUNDLE	D NETWORK ELEMENTS - Tennessee												Attachment:	2	Exhi	ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Charge -	Charge -
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Design		3	UEP93		29.98										
UNE L	oop Rate			021 00		20.00										
-	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP93	UECS1	12.48										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP93	UECS1	16.31										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP93	UECS1	21.32										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP93	UECS2	16.56										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP93	UECS2	21.63										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP93	UECS2	28.28										
	ort Rate ′, LA, MS, & TN only															
AL, K1	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP93	UEPYA	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex) Basic Local Alea 2-Wire Voice Grade Port (Centrex 800 termination)Basic Local			021 33	OLITA	1.70	22.14	15.25	0.40	5.91		30.09	1.03			
	Area			UEP93	UEPYB	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local															1
	Area			UEP93	UEPYH	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex from diff Serving Wire															
	Center)2 Basic Local Area			UEP93	UEPYM	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term - Basic Local Area			UEP93	UEPYZ	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			115000		4 70	00.44	45.05	0.45	0.04		00.00	7 00			
	- Basic Local Area 2-Wire Voice Grade Port Terminated on 800 Service Term -			UEP93	UEPY9	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	Basic Local Area			UEP93	UEPY2	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex)			UEP93	UEPQA	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP93	UEPQB	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP93	UEPQH	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port (Centrex from diff Serving Wire															
	Center)2			UEP93	UEPQM	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term			UEP93	UEPQZ	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP93	UEPQ9	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP93	UEPQ2	1.70	22.14	15.25	8.45	3.91		30.89	7.03			
Local	Switching			UEP93	URECS	0.6381										
Local	Centrex Intercom Funtionality, per port Number Portability			UEP93	UREUS	0.6381										
Local I	Local Number Portability (1 per port)			UEP93	LNCCC	0.35										
Feature				02.00	2.1000	0.00										
	All Standard Features Offered, per port			UEP93	UEPVF	0.00										
	All Centrex Control Features Offered, per port			UEP93	UEPVC	0.00										
NARS																
	Unbundled Network Access Register - Combination			UEP93	UARCX	0.00	0.00	0.00				30.89	7.03			
	Unbundled Network Access Register - Indial			UEP93	UAR1X	0.00	0.00	0.00				30.89	7.03			
	Unbundled Network Access Register - Outdial			UEP93	UAROX	0.00	0.00	0.00				30.89	7.03			l
	laneous Terminations Trunk Sido															
2-wire	Trunk Side Trunk Side Terminations, each			UEP93	CEND6	8.78	22.14	15.25	8.45	3.91		30.89	7.03			
4-Wire	Digital (1.544 Megabits)			021 33	OLINDO	0.70	22.14	15.25	0.40	5.91		30.09	1.03			
4 1110	DS1 Circuit Terminations, each	1		UEP93	M1HD1	35.55	75.93	38.15				30.89	7.03			1
	DS0 Channels Activated, Per Channel			UEP93	M1HDO	0.00	108.67	00.10	† †			30.89	7.03		İ	
Interof	fice Channel Mileage - 2-Wire															1
	Interoffice Channel Facilities Termination			UEP93	MIGBC	18.58	22.14	15.25	8.45	3.91		30.89	7.03			
	Interoffice Channel mileage, per mile or fraction of mile			UEP93	MIGBM	0.0174										
	e Activations (DS0) Centrex Loops on Channelized DS1 Servic	e														
D4 Cha	annel Bank Feature Activations		ļ	1.5500	100110											
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP93	1PQWS	0.66										l
		1	1	1	1				1		1				1	1

UNBUNDL	ED NETWORK ELEMENTS - Tennessee		1	1									Attachment:		Exhi	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Charge -	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
-							Nonrecurring		Nonrecurring	Disconnect			055	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop															
	Slot			UEP93	1PQW7	0.66										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP93	1PQWP	0.66										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP93	1PQWV	0.66										
	Feature Activation on D-4 Channel Bank Tie Line/Trunk Loop															
	Slot			UEP93	1PQWQ	0.66										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP93	1PQWA	0.66										
Non-	Recurring Charges (NRC) Associated with UNE-P Centrex															
	NRC Conversion Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP93	USAC2		1.03	0.29				30.89	7.03			
	New Centrex Standard Common Block			UEP93	M1ACS	0.00	658.60					30.89	7.03			
	New Centrex Customized Common Block	I	L	UEP93	M1ACC	0.00	658.60					30.89	7.03			
	NAR Establishment Charge, Per Occasion		L	UEP93	URECA		68.57					30.89	7.03			
	1 - Required Port for Centrex Control in 1AESS, 5ESS & EWSD	I	<u> </u>													
	2 - Requres Interoffice Channel Mileage	I	<u> </u>													
	3 - Requires Specific Customer Premises Equipment	<u> </u>	 													
	CENTREX PORT/LOOP COMBINATIONS - MARKET RATES															
	arket Rates are applied where BellSouth is not required by FCC					ndied Local Sv	vitching or Swi	tch Ports.								
2. KE	curring Charges for all Standard Centrex and Centrex Conrol Fe												0			
2 Er																
	d Office and Tandem Switching Usage and Common Transport														A daliti a mal ND	C
4. Th	e first and additional Port nonrecurring charges apply to Not Co														Additional NR	Cs may
4. Th appl	e first and additional Port nonrecurring charges apply to Not Co y also and are categorized accordingly.	urrently													Additional NR	Cs may
4. Th appl UNE	e first and additional Port nonrecurring charges apply to Not Ci y also and are categorized accordingly. •P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only	urrently													Additional NR	Cs may
4. Th appl UNE 2-Wi	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo	urrently													Additional NR	Cs may
4. Th appl UNE 2-Wi	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design)	urrently													Additional NR	Cs may
4. Th appl UNE 2-Wi	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. PF CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design	urrently													Additional NR	Cs may
4. Th appl UNE 2-Wi	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo	urrently	Comb	ined Combos. Fe		mbined Combo									Additional NR	Cs may
4. Th appl UNE 2-Wi	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	urrently	1 Comb	UEP91		26.48									Additional NR	Cs may
4. Th appl UNE 2-Wi UNE	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design	urrently	1 Comb	UEP91		26.48									Additional NR	Cs may
4. Th appl UNE 2-Wi UNE	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	urrently	2 Comb	UEP91 UEP91		26.48 30.31									Additional NR	Cs may
4. Th appl UNE 2-Wi UNE	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design	urrently	2 Comb	UEP91 UEP91 UEP91		26.48 30.31 35.32									Additional NR	Cs may
4. Th appl UNE 2-Wi UNE	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. PP CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design Port/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design	urrently	2 Comb	UEP91 UEP91		26.48 30.31									Additional NR	Cs may
4. Th appl UNE 2-Wi UNE	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design	urrently	2 Comb	UEP91 UEP91 UEP91		26.48 30.31 35.32									Additional NR	Cs may
4. Th appl UNE 2-Wi UNE	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design Port/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design	urrently	2 Comb	ined Combos. Fr UEP91 UEP91 UEP91 UEP91		26.48 20.48 30.31 35.32 30.56									Additional NR	Cs may
4. Th appl UNE 2-Wi UNE	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design	urrently	2 Comb	ined Combos. Fr UEP91 UEP91 UEP91 UEP91		26.48 20.48 30.31 35.32 30.56									Additional NR	Cs may
4. Th appl UNE 2-Wi UNE	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. PC ECNTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design	urrently	Comb	UEP91 UEP91 UEP91 UEP91 UEP91 UEP91		26.48 26.48 30.31 35.32 30.56 35.63 42.28									Additional NR	Cs may
4. Th appl UNE 2-Wi UNE	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. PCENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design Port/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design	urrently	Comb	UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1	26.48 26.48 30.31 35.32 30.56 35.63									Additional NR	Cs may
4. Th appl UNE 2-Wi UNE	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design Port/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design	urrently	Comb	ined Combos. Fr UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91		26.48 26.48 30.31 35.32 30.56 35.63 42.28									Additional NR	Cs may
4. Th appl UNE 2-Wi UNE	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design Port/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3	urrently	Comb	ined Combos. Fr UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS1	26.48 26.48 30.31 35.32 30.56 35.63 42.28 12.48 16.31 21.32									Additional NR	Cs may
4. Th appl UNE 2-Wi UNE	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. PP CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2	urrently	Comb	ined Combos. F(UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS1	26.48 26.48 30.31 35.32 30.56 35.63 42.28 12.48 16.31 21.32 16.56									Additional NR	Cs may
4. Th appl UNE 2-Wi UNE	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. PP CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 1	urrently	Comb	ined Combos. F(UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS1 UECS2 UECS2	26.48 26.48 30.31 35.32 30.56 35.63 42.28 12.48 16.31 21.32 16.56 21.63									Additional NR	Cs may
4. Th appl UNE 2-Wi UNE UNE UNE	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 2	urrently	Comb	ined Combos. F(UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS1	26.48 26.48 30.31 35.32 30.56 35.63 42.28 12.48 16.31 21.32 16.56									Additional NR	Cs may
4. Th appi UNE 2-Wi UNE UNE UNE	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. PCENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design Port/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3 Ports	urrently	Comb	ined Combos. F(UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS1 UECS2 UECS2	26.48 26.48 30.31 35.32 30.56 35.63 42.28 12.48 16.31 21.32 16.56 21.63									Additional NR	Cs may
4. Th appi UNE 2-Wi UNE UNE UNE	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. PC ENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 Ports tates (Except North Carolina and Sout Carolina)	urrently	Comb	ined Combos. Fr UEP91 1 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS1 UECS1 UECS2 UECS2 UECS2	26.48 26.48 30.31 35.32 30.56 35.63 42.28 12.48 16.31 21.32 16.56 21.63 28.28	Dis, the nonrecu	rring chargés	shall be those	identified in t		rring - Curre	ently Combine		Additional NR	Cs may
4. Th appi UNE 2-Wi UNE UNE UNE	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. PP CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice G	urrently	Comb	ined Combos. F(UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS1 UECS2 UECS2	26.48 26.48 30.31 35.32 30.56 35.63 42.28 12.48 16.31 21.32 16.56 21.63									Additional NR	Cs may
4. Th appi UNE 2-Wi UNE UNE UNE	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3 Ports tates (Except North Carolina and Sout Carolina) 2-Wire Voice Grade Port (Centrex) Basic Local Area 2-Wire Voice Grade Port (Centrex 800 termination)Basic Local	urrently	Comb	ined Combos. Fr	UECS1 UECS1 UECS1 UECS1 UECS2 UECS2 UECS2 UECS2 UECS2	26.48 30.31 35.32 30.56 35.63 42.28 16.31 21.32 16.56 21.63 28.28 14.00	205, the nonrecu	45.00	shall be those	identified in t		rring - Curre	7.03		Additional NR	Cs may
4. Th appi UNE 2-Wi UNE UNE UNE	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. PC ENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 Ports Tates (Except North Carolina and Sout Carolina) 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 Ports Tates (Except North Carolina and Sout Carolina) 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 Ports	urrently	Comb	ined Combos. Fr UEP91 1 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UECS1 UECS1 UECS1 UECS1 UECS2 UECS2 UECS2	26.48 26.48 30.31 35.32 30.56 35.63 42.28 12.48 16.31 21.32 16.56 21.63 28.28	Dis, the nonrecu	rring chargés	shall be those	identified in t		rring - Curre	ently Combine		Additional NR	Cs may
4. Th appi UNE 2-Wi UNE UNE UNE	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. PP CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 Ports 2-Wire Voice Grade Loop (SL 2) - Zone 3 Ports 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-	urrently	Comb	ined Combos. Fr	UECS1 UECS1 UECS1 UECS1 UECS2 UECS2 UECS2 UECS2 UECS2 UECS2 UECS2	26.48 30.31 35.32 30.56 35.63 42.28 12.48 16.31 21.32 16.56 21.63 28.28 14.00 14.00	ps, the nonrecu	45.00	shall be those	identified in t		rring - Curre	7.03		Additional NR	Cs may
4. Th appi UNE 2-Wi UNE UNE UNE	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. P CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 3 Ports tates (Except North Carolina and Sout Carolina) 2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (Centrex 800 termination)Basic Local Area 2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area	urrently	Comb	ined Combos. Fr	UECS1 UECS1 UECS1 UECS1 UECS2 UECS2 UECS2 UECS2 UECS2	26.48 30.31 35.32 30.56 35.63 42.28 16.31 21.32 16.56 21.63 28.28 14.00	205, the nonrecu	45.00	shall be those	identified in t		rring - Curre	7.03		Additional NR	Cs may
4. Th appi UNE 2-Wi UNE UNE UNE	e first and additional Port nonrecurring charges apply to Not Cr y also and are categorized accordingly. PP CENTREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only re VG Loop/2-Wire Voice Grade Port (Centrex) Combo Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 Ports 2-Wire Voice Grade Loop (SL 2) - Zone 3 Ports 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-	urrently	Comb	ined Combos. Fr	UECS1 UECS1 UECS1 UECS1 UECS2 UECS2 UECS2 UECS2 UECS2 UECS2 UECS2	26.48 30.31 35.32 30.56 35.63 42.28 12.48 16.31 21.32 16.56 21.63 28.28 14.00 14.00	ps, the nonrecu	45.00	shall be those	identified in t		rring - Curre	7.03		Additional NR	Cs may

UNDUNDL	ED NETWORK ELEMENTS - Tennessee												Attachment:	2	Exhi	ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrecurring		Nonrecurring			r -		Rates(\$)	_	-
·							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port terminated in on Megalink or equivalent							15.00					=			
·	- Basic Local Area			UEP91	UEPY9	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port Terminated on 800 Service Term - Basic Local Area			UEP91	UEPY2	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	Y, LA, MS, & TN Only			UEF91	UEPTZ	14.00	90.00	45.00	20.00	10.00		30.69	7.03			
	2-Wire Voice Grade Port (Centrex)			UEP91	UEPQA	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex 9)			UEP91	UEPQB	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP91	UEPQH	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
· · · · ·	2-Wire Voice Grade Port (Centrex from diff Serving Wire			02.0.	02. 0.1	1.000	00.00	10.00	20.00	10.00		00.00	1.00			
	Center)2			UEP91	UEPQM	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service															
	Term			UEP91	UEPQZ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
					l											
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP91	UEPQ9	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP91	UEPQ2	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
Local	Switching															
	Centrex Intercom Funtionality, per port			UEP91	URECS	0.6381										
Local	Number Portability			115504	1.1.500											
	Local Number Portability (1 per port)			UEP91	LNPCC	0.35										
Featu				115504												
	All Standard Features Offered, per port			UEP91 UEP91	UEPVF	0.00	100 70					30.89	7.03			
	All Select Features Offered, per port				UEPVS	0.00	433.78					30.89	7.03			
NARS	All Centrex Control Features Offered, per port			UEP91	UEPVC	0.00					-	30.89	7.03			
INARG	Unbundled Network Access Register - Combination			UEP91	UARCX	0.00	0.00	0.00				30.89	7.03			
	Unbundled Network Access Register - Indial			UEP91	UAR1X	0.00	0.00	0.00			1	30.89	7.03			
	Unbundled Network Access Register - Outdial			UEP91	UAROX	0.00	0.00	0.00				30.89	7.03			
Miscr	ellaneous Terminations			OEI 01	0/11/0//	0.00	0.00	0.00				00.00	1.00			
	e Trunk Side															
	Trunk Side Terminations, each			UEP91	CENA6	8.78	90.00	45.00	20.00	10.00		30.89	7.03			
Interc	office Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination - Voice Grade			UEP91	M1GBC	18.58	90.00	45.00	20.00	10.00		30.89	7.03			
	Interoffice Channel mileage, per mile or fraction of mile			UEP91	M1GBM	0.0174										
Featu	re Activations (DS0) Centrex Loops on Channelized DS1 Servic	e														
D4 Cl	nannel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP91	1PQWS	0.66										
.		1			1000											
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP91	1PQW6	0.66	├ ───┤				I					
. 1	Feature Activation on D-4 Channel Bank FX Trunk Side Loop				10014/7	0.00										
	Slot Feature Activation on D-4 Channel Bank Centrex Loop Slot -			UEP91	1PQW7	0.66	├ ────┤									
.	Different Wire Center	1		UEP91	1PQWP	0.66										
				02131	TOWF	0.00	<u>∤</u>									1
.	Feature Activation on D-4 Channel Bank Private Line Loop Slot	1		UEP91	1PQWV	0.66										
	Feature Activation on D-4 Channel Bank Thrate Line/Trunk Loop	1				0.00	 				1					1
. 1	Slot	1		UEP91	1PQWQ	0.66										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP91	1PQWA	0.66										
Non-	Recurring Charges (NRC) Associated with UNE-P Centrex	1					t t		i i		1		ĺ	İ	ĺ	1
	Conversion - Currently Combined Switch-As-Is with allowed						T T									
	changes, per port			UEP91	USAC2		1.03	0.29				30.89	7.03			
	New Centrex Standard Common Block			UEP91	M1ACS	0.00	658.60					30.89	7.03			
	New Centrex Customized Common Block			UEP91	M1ACC	0.00	658.60					30.89	7.03			
	Secondary Block, per Block			UEP91	M2CC1	0.00	73.55					30.89	7.03			
	NAR Establishment Charge, Per Occasion			UEP91	URECA		68.57					30.89	7.03			
UNE-	P CENTREX - 5ESS (Valid in All States)		ļ				ļ ļ									
. 12-Wir	e VG Loop/2-Wire Voice Grade Port (Centrex) Combo						├ ────									
							1				1			1		1
	Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -						<u>↓</u>									

UNBUNDLI	ED NETWORK ELEMENTS - Tennessee												Attachment:	2	Exhi	ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
┝───┤────	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
1	Non-Design		2	UEP95		30.31										
(2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		-	021 00		00.01										
	Non-Design		3	UEP95		35.32										
UNE F	Port/Loop Combination Rates (Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design		1	UEP95		30.56										
┌───┼───	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		1	UEP95	-	30.56										'
	Design		2	UEP95		35.63										
i l	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design		3	UEP95		42.28										
	Loop Rate		4			10.40										ļ'
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP95	UECS1	12.48	├									───
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP95 UEP95	UECS1	16.31										'
	2-Wire Voice Grade Loop (SL 1) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 1		3	UEP95 UEP95	UECS1	21.32 16.56	┝────┤							1		───
<u> </u>	2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP95 UEP95	UECS2 UECS2	21.63	├									<u> </u>
	2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP95	UECS2	28.28										
UNE F	Port Rate		Ű	02.00	02002	20.20										
All Sta																
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP95	UEPYA	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPYB	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local															
	Area			UEP95	UEPYH	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex from diff Serving Wire															
	Center)2 Basic Local Area			UEP95	UEPYM	14.00	90.00	45.00	20.00	10.00		30.89	7.03			L
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service					44.00	00.00	45.00	20.00	40.00		20.00	7.00			
	Term - Basic Local Area 2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP95	UEPYZ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			'
	- Basic Local Area			UEP95	UEPY9	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port Terminated on 800 Service Term -			011 33	OLI 13	14.00	30.00	43.00	20.00	10.00		50.03	7.05			
	Basic Local Area			UEP95	UEPY2	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
AL, K	Y, LA, MS, SC, & TN Only				-											
	2-Wire Voice Grade Port (Centrex)			UEP95	UEPQA	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPQB	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP95	UEPQH	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex from diff Serving Wire															
	Center)2			UEP95	UEPQM	14.00	90.00	45.00	20.00	10.00		30.89	7.03			 '
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP95	UEPQZ	14.00	00.00	45.00	20.00	10.00		20.00	7.03			
<u> </u>				02793	UEFQZ	14.00	90.00	45.00	20.00	10.00		30.89	7.03		}	ł
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP95	UEPQ9	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
<u> </u>	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP95	UEPQ2	14.00	90.00	45.00	20.00	10.00		30.89	7.03		1	t'
FL & (GA Only														İ	1
	Switching															
	Centrex Intercom Funtionality, per port			UEP95	URECS	0.6381										
Local	Number Portability															L
	Local Number Portability (1 per port)		ļ	UEP95	LNPCC	0.35										
Featu				UEP95	UEPVF	0.00						30.89	7.03			───
<u> </u>	All Standard Features Offered, per port All Select Features Offered, per port			UEP95 UEP95	UEPVF	0.00	433.78					30.89	7.03			<u> </u>
	All Centrex Control Features Offered, per port			UEP95	UEPVS	0.00	400.70					30.89	7.03			1
NARS				021 00	021.70	0.00						50.09	1.05			1
	Unbundled Network Access Register - Combination			UEP95	UARCX	0.00	0.00	0.00				30.89	7.03		1	1
1	Unbundled Network Access Register - Indial			UEP95	UAR1X	0.00	0.00	0.00	i i		l	30.89	7.03		l	1
	Unbundled Network Access Register - Outdial			UEP95	UAROX	0.00	0.00	0.00				30.89	7.03			
	Ilaneous Terminations															
2-Wire	e Trunk Side															
	Trunk Side Terminations, each		L	UEP95	CEND6	8.78	47.75	47.01	9.21	8.47		30.89	7.03			<u> </u>

UNBUNDL	-ED	NETWORK ELEMENTS - Tennessee				- <u>1</u>						Com Cont	Sup Col	Attachment:			ibit: B
CATEGORY		RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
							Rec	Nonrecurring		Nonrecurring					Rates(\$)		
							Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
4-Wi		igital (1.544 Megabits)															
		OS1 Circuit Terminations, each			UEP95	M1HD1	35.55	75.93	38.15				30.89	7.03			
		DS0 Channels Activated, each			UEP95	M1HDO	0.00	108.67					30.89	7.03			
Inter		ce Channel Mileage - 2-Wire			1.55.45		10.50				10.00			=			
		nteroffice Channel Facilities Termination			UEP95	MIGBC	18.58	90.00	45.00	20.00	10.00		30.89	7.03			
Fast		nteroffice Channel mileage, per mile or fraction of mile			UEP95	MIGBM	0.0174										
		Activations (DS0) Centrex Loops on Channelized DS1 Servic nel Bank Feature Activations	e			-											
D4 C		eature Activation on D-4 Channel Bank Centrex Loop Slot			UEP95	1PQWS	0.66								-		
	-	eardre Activation on B-4 Channel Bank Centrex Loop Glot			011 35	II QWO	0.00										
	F	eature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP95	1PQW6	0.66										
		eature Activation on D-4 Channel Bank FX Trunk Side Loop			00		0.00			1		t		1		1	1
		Slot			UEP95	1PQW7	0.66										
		eature Activation on D-4 Channel Bank Centrex Loop Slot -															1
		Different Wire Center	1		UEP95	1PQWP	0.66										
	F	eature Activation on D-4 Channel Bank Private Line Loop Slot			UEP95	1PQWV	0.66										
	F	eature Activation on D-4 Channel Bank Tjie Line/Trunk Loop															
		Slot			UEP95	1PQWQ	0.66										
		eature Activation on D-4 Channel Bank WATS Loop Slot			UEP95	1PQWA	0.66										
Non		urring Charges (NRC) Associated with UNE-P Centrex															
		IRC Conversion Currently Combined Switch-As-Is with allowed															
		hanges, per port			UEP95	USAC2		1.03	0.29				30.89	7.03			
		lew Centrex Standard Common Block			UEP95	M1ACS	0.00	658.60					30.89	7.03			
		New Centrex Customized Common Block			UEP95	M1ACC	0.00	658.60					30.89	7.03			
		IAR Establishment Charge, Per Occasion			UEP95	URECA	0.00	68.57					30.89	7.03			
		ENTREX - DMS100 (Valid in All States)				_											
		G Loop/2-Wire Voice Grade Port (Centrex) Combo															
UNE		t/Loop Combination Rates (Non-Design)				-											
		-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Ion-Design		1	UEP9D		26.48										
		-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		1	DEF9D		20.40								-		
		Non-Design		2	UEP9D		30.31										
		-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		2			50.51										
		Ion-Design		3	UEP9D		35.32										
UNE		t/Loop Combination Rates (Design)		0			00.02					1					
		-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
		Design		1	UEP9D		30.56										
		-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -										İ 👘					
	D	Design		2	UEP9D		35.63										
	2	-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	D	Design		3	UEP9D		42.28										
UNE		p Rate															
		-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9D	UECS1	12.48										
		-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9D	UECS1	16.31										L
		-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP9D	UECS1	21.32										<u> </u>
		-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9D	UECS2	16.56										<u> </u>
		-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9D	UECS2 UECS2	21.63										L
		-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9D	UEUS2	28.28										<u> </u>
		t Rate ITES				+ +						+					
ALL		-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9D	UEPYA	14.00	90.00	45.00	20.00	10.00		30.89	7.03			+
		-Wire Voice Grade Port (Centrex) Basic Local Area				UEPTA	14.00	90.00	45.00	20.00	10.00		30.89	1.03			
		vrea	1		UEP9D	UEPYB	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
		-Wire Voice Grade Port (Centrex / EBS-PSET)3Basic Local			021.00		14.00	30.00	+3.00	20.00	10.00		50.09	1.03			
		Area	1		UEP9D	UEPYC	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
		-Wire Voice Grade Port (Centrex / EBS-M5009)3Basic Local				02.70	14.00	00.00	40.00	20.00	10.00	1	50.00	1.00			t
		Vrea	1		UEP9D	UEPYD	14.00	90.00	45.00	20.00	10.00		30.89	7.03			

UNBUNDLE	ED NETWORK ELEMENTS - Tennessee												Attachment:	2	Exhi	ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR		Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
	2-Wire Voice Grade Port (Centrex / EBS-M5209))3 Basic Local						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Area			UEP9D	UEPYE	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5112))3 Basic Local Area			UEP9D	UEPYF	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5312))3Basic Local Area			UEP9D	UEPYG	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5008))3 Basic Local			02.00		1.000		10.00		10.00						
	Area 2-Wire Voice Grade Port (Centrex / EBS-M5208))3 Basic Local			UEP9D	UEPYT	14.00	90.00	45.00	20.00	10.00		30.89	7.03			<u> </u>
	Area			UEP9D	UEPYU	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5216))3 Basic Local Area			UEP9D	UEPYV	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5316))3 Basic Local Area			UEP9D	UEPY3	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex with Caller ID) Basic Local Area			UEP9D	UEPYH	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp Indication))3 Basic Local Area			UEP9D	UEPYW	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication))3															
	Basic Local Area 2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)			UEP9D	UEPYJ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			<u> </u>
	2 Basic Local Area 2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3		-	UEP9D	UEPYM	14.00	90.00	45.00	20.00	10.00		30.89	7.03			<u> </u>
	Basic Local Area			UEP9D	UEPYO	14.00	90.00	45.00	20.00	10.00		30.89	7.03			L
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3 Basic Local Area			UEP9D	UEPYP	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3 Basic Local Area			UEP9D	UEPYQ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3 Basic Local Area			UEP9D	UEPYR	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3 Basic Local Area			UEP9D	UEPYS	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3 Basic Local Area			UEP9D	UEPY4	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3 Basic Local Area			UEP9D	UEPY5	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3 Basic Local Area			UEP9D	UEPY6	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
i	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			UEP9D	UEPY7			45.00	20.00			30.89	7.03			
	Basic Local Area 2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service					14.00	90.00			10.00						
	Term 2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9D	UEPYZ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	Basic Local Area 2-Wire Voice Grade Port Terminated on 800 Service Term Basic			UEP9D	UEPY9	14.00	90.00	45.00	20.00	10.00		30.89	7.03			<u> </u>
AL. K	Local Area Y, LA, MS, SC, & TN Only			UEP9D	UEPY2	14.00	90.00	45.00	20.00	10.00		30.89	7.03			<u> </u>
,	2-Wire Voice Grade Port (Centrex)			UEP9D	UEPQA	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP9D	UEPQB	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3 2-Wire Voice Grade Port (Centrex / EBS-M5009)3			UEP9D UEP9D	UEPQC	14.00	90.00 90.00	45.00 45.00	20.00 20.00	10.00		30.89 30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3	1		UEP9D	UEPQE	14.00	90.00	45.00	20.00	10.00	†	30.89	7.03			<u> </u>
	2-Wire Voice Grade Port (Centrex / EBS-M5112)3			UEP9D	UEPQF	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5312)3			UEP9D	UEPQG	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
┝──- ┝───	2-Wire Voice Grade Port (Centrex / EBS-M5008)3		<u> </u>	UEP9D UEP9D	UEPQT UEPQU	14.00	90.00 90.00	45.00 45.00	20.00 20.00	10.00	ļ	30.89 30.89	7.03			<u> </u>
├── 	2-Wire Voice Grade Port (Centrex / EBS-M5208)3 2-Wire Voice Grade Port (Centrex / EBS-M5216)3			UEP9D UEP9D	UEPQU	14.00	90.00	45.00	20.00	10.00	<u> </u>	30.89	7.03			
	2-Wire Voice Grade Port (Centrex / EBS-M5216)3 2-Wire Voice Grade Port (Centrex / EBS-M5316)3			UEP9D	UEPQ3	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex with Caller ID)	1		UEP9D	UEPQH	14.00	90.00	45.00	20.00	10.00		30.89	7.03		İ	1

UNBUNDL	ED NETWORK ELEMENTS - Tennessee												Attachment:	2	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp Indication)3			UEP9D	UEPQW	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication)3			UEP9D	UEPQJ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)			02.02	02. 00	1.000	00.00	10.00	20.00	10100		00.00	1100			
	2			UEP9D	UEPQM	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			UEP9D	UEPQO	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3			UEP9D	UEPQP	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-0009)2, 3			UEP9D	UEPQQ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			UEP9D	UEPQR	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3			UEP9D	UEPQS	14.00	90.00	45.00	20.00	10.00	<u> </u>	30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3			UEP9D	UEPQ4	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
				021 00		14.00	50.00	40.00	20.00	10.00		00.00	1.00			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3			UEP9D	UEPQ5	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3			UEP9D	UEPQ6	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			UEP9D	UEPQ7	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			OEI 3D	ULI Q/	14.00	30.00	43.00	20.00	10.00		30.03	1.00			
	Term			UEP9D	UEPQZ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9D	UEPQ9	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
Local	2-Wire Voice Grade Port Terminated on 800 Service Term Switching			UEP9D	UEPQ2	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
LUCA	Centrex Intercom Funtionality, per port			UEP9D	URECS	0.6381										
Local	Number Portability															
	Local Number Portability (1 per port)			UEP9D	LNPCC	0.35										
Featu													=			
	All Standard Features Offered, per port All Select Features Offered, per port			UEP9D UEP9D	UEPVF UEPVS	0.00	433.78					30.89 30.89	7.03			
	All Centrex Control Features Offered, per port			UEP9D	UEPVC	0.00	433.70					30.89	7.03			
NARS																
	Unbundled Network Access Register - Combination			UEP9D	UARCX	0.00	0.00	0.00				30.89	7.03			
	Unbundled Network Access Register - Inward			UEP9D	UAR1X	0.00	0.00	0.00				30.89	7.03			
Micor	Unbundled Network Access Register - Outdial			UEP9D	UAROX	0.00	0.00	0.00			<u> </u>	30.89	7.03			
	e Trunk Side			<u> </u>	+ +											
	Trunk Side Terminations, each			UEP9D	CEND6	8.78	90.00	45.00	20.00	10.00		30.89	7.03			
4-Wir	e Digital (1.544 Megabits)															
	DS1 Circuit Terminations, each			UEP9D	M1HD1	35.55	75.93	38.15				30.89	7.03			
Intere	DS0 Channels Activiated per Channel office Channel Mileage - 2-Wire			UEP9D	M1HDO	0.00	108.67		<u> </u>			30.89	7.03			
merc	Interoffice Channel Facilities Termination			UEP9D	MIGBC	18.58	90.00	45.00	20.00	10.00		30.89	7.03			
	Interoffice Channel mileage, per mile or fraction of mile	1		UEP9D	MIGBM	0.0174	00.00	.0.00	20.00		1	50.00				
	re Activations (DS0) Centrex Loops on Channelized DS1 Service	ce														
D4 Cł	nannel Bank Feature Activations								ļ		ļ					
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9D	1PQWS	0.66			<u> </u>							
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot	1		UEP9D	1PQW6	0.66										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop	1				0.00			1	1	t			-		
	Slot			UEP9D	1PQW7	0.66										
i	Feature Activation on D-4 Channel Bank Centrex Loop Slot -															
├───	Different Wire Center			UEP9D	1PQWP	0.66										
	1	1	1	UEP9D	1PQWV	0.66			1	1	1	1	1		1	1

UNBUNDLE	D NETWORK ELEMENTS - Tennessee										1		Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop				100110											i
	Slot			UEP9D UEP9D	1PQWQ 1PQWA	0.66										l
Non-R	Feature Activation on D-4 Channel Bank WATS Loop Slot ecurring Charges (NRC) Associated with UNE-P Centrex			UEP9D	IPQWA	0.66										
Non-K	NRC Conversion Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP9D	USAC2		1.03	0.29				30.89	7.03			i
	New Centrex Standard Common Block			UEP9D	M1ACS	0.00	658.60					30.89	7.03			
	New Centrex Customized Common Block			UEP9D	M1ACC	0.00	658.60					30.89	7.03			
	NAR Establishment Charge, Per Occasion			UEP9D	URECA		68.57					30.89	7.03			
	CENTREX - EWSD (Valid in AL, FL, KY, LA, MS & TN)															1
	VG Loop/2-Wire Voice Grade Port (Centrex) Combo															1
UNE P	ort/Loop Combination Rates (Non-Design)						├ ───┤				l					
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design	1	1	UEP9E	1	26.48										1
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			UEFSE		20.48	├									
	Non-Design		2	UEP9E		30.31										i i
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -	1	-		1	50.51					1					
	Non-Design		3	UEP9E		35.32										i
UNE P	ort/Loop Combination Rates (Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	Design		1	UEP9E		30.56										1
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															i
	Design		2	UEP9E		35.63										ļ
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															i
	Design		3	UEP9E		42.28										l
UNE L	oop Rate 2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9E	UECS1	12.48										i
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9E	UECS1	12.40										i
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP9E	UECS1	21.32										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9E	UECS2	16.56										i
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9E	UECS2	21.63										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9E	UECS2	28.28										
	ort Rate															
AL, FL	, KY, LA, MS, & TN only															
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9E	UEPYA	14.00	90.00	45.00	20.00	10.00		30.89	7.03			l
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local					44.00	00.00	15.00	00.00	40.00		00.00	7.00			i
	Area			UEP9E	UEPYB	14.00	90.00	45.00	20.00	10.00		30.89	7.03			ł
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area			UEP9E	UEPYH	14.00	90.00	45.00	20.00	10.00		30.89	7.03			i i
	2-Wire Voice Grade Port (Centrex from diff Serving Wire			021 02		14.00	90.00	45.00	20.00	10.00	1	30.09	1.03			
	Center)2 Basic Local Area			UEP9E	UEPYM	14.00	90.00	45.00	20.00	10.00		30.89	7.03			i i
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			02.02	02.11	1.00	00.00	10100	20.00	10.00		00.00	1100			l .
	Term - Basic Local Area			UEP9E	UEPYZ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			i
	2-Wire Voice Grade Port terminated in on Megalink or equivalent		I				1									[
	- Basic Local Area			UEP9E	UEPY9	14.00	90.00	45.00	20.00	10.00		30.89	7.03			1
	2-Wire Voice Grade Port Terminated on 800 Service Term -															1
	Basic Local Area			UEP9E	UEPY2	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
AL, KY	(, LA, MS, & TN Only					11.00	00.00	18 00	00.00	10.00	l	00.00	7.00			I
	2-Wire Voice Grade Port (Centrex)			UEP9E	UEPQA	14.00	90.00	45.00	20.00	10.00		30.89	7.03			l
	2-Wire Voice Grade Port (Centrex 800 termination) 2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP9E UEP9E	UEPQB UEPQH	14.00 14.00	90.00 90.00	45.00 45.00	20.00 20.00	10.00		30.89 30.89	7.03			ł
	2-Wire Voice Grade Port (Centrex with Caller ID) 1 2-Wire Voice Grade Port (Centrex from diff Serving Wire					14.00	90.00	45.00	20.00	10.00	1	30.09	7.03			
	Center)2			UEP9E	UEPQM	14.00	90.00	45.00	20.00	10.00		30.89	7.03			1
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service				52		00.00	.0.00	20.00			00.00		1	1	r
	Term			UEP9E	UEPQZ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			1
																ſ
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9E	UEPQ9	14.00	90.00	45.00	20.00	10.00		30.89	7.03			I
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP9E	UEPQ2	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
Local	Switching															I

UNBUNDLE	D NETWORK ELEMENTS - Tennessee											1	Attachment:			bit: B
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
	Centrex Intercom Funtionality, per port			UEP9E	URECS	0.6381	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Local	Number Portability			UEP9E	URECS	0.6381										ł
Local	Local Number Portability (1 per port)			UEP9E	LNPCC	0.35										-
Featur				02.02	2.1. 00	0.00										
	All Standard Features Offered, per port			UEP9E	UEPVF	0.00						30.89	7.03			
	All Select Features Offered, per port			UEP9E	UEPVS	0.00	433.78					30.89	7.03			
	All Centrex Control Features Offered, per port			UEP9E	UEPVC	0.00						30.89	7.03			
NARS																
	Unbundled Network Access Register - Combination			UEP9E	UARCX	0.00	0.00	0.00				30.89	7.03			
	Unbundled Network Access Register - Indial Unbundled Network Access Register - Outdial			UEP9E UEP9E	UAR1X UAROX	0.00	0.00	0.00				30.89 30.89	7.03			
Miscol	Ilaneous Terminations			UEP9E	UARUX	0.00	0.00	0.00				30.89	7.03			
	Trunk Side															
2 0010	Trunk Side Terminations, each	1		UEP9E	CEND6	8.78	90.00	45.00	20.00	10.00	1	30.89	7.03			t
4-Wire	Digital (1.544 Megabits)	1		-		20									İ	<u> </u>
	DS1 Circuit Terminations, each			UEP9E	M1HD1	35.55	75.93	38.15				30.89	7.03			
	DS0 Channel Activated Per Channel			UEP9E	M1HDO	0.00	108.67					30.89	7.03			
Interof	ffice Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination			UEP9E	MIGBC	18.58	90.00	45.00	20.00	10.00		30.89	7.03			
	Interoffice Channel mileage, per mile or fraction of mile			UEP9E	MIGBM	0.0174										
Featur	e Activations (DS0) Centrex Loops on Channelized DS1 Service	e									-					
D4 Cha	annel Bank Feature Activations Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9E	1PQWS	0.66					-					
	Feature Activation on D-4 Channel Bank Centrex Loop Stot			UEP9E	1PQW6	0.66										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP9E	1PQW7	0.66										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP9E	1PQWP	0.66										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9E	1PQWV	0.66										
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop															
	Slot			UEP9E	1PQWQ	0.66										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9E	1PQWA	0.66										
Non-R	ecurring Charges (NRC) Associated with UNE-P Centrex				_											
	NRC Conversion Currently Combined Switch-As-Is with allowed changes, per port			UEP9E	USAC2		1.03	0.29				30.89	7.03			
	New Centrex Standard Common Block			UEP9E UEP9E	M1ACS	0.00	658.60	0.29				30.89	7.03			
	New Centrex Standard Common Block			UEP9E	MIACS	0.00	658.60				-	30.89	7.03			
	NAR Establishment Charge, Per Occasion			UEP9E	URECA	0.00	68.57				1	30.89	7.03		1	<u> </u>
UNE-P	CENTREX - DCO - Valid in AL, KY, LA, MS, & TN)	1		-					i i		1				l	<u> </u>
2-Wire	VG Loop/2-Wire Voice Grade Port (Centrex) Combo															
UNE P	ort/Loop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design	-	1	UEP93		26.48										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		2	UEP93		30.31										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		3	UEP93		35.32										
UNE P	ort/Loop Combination Rates (Design)															l
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design		1	UEP93		30.56										
	Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		1	UEP93		30.56										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		3	UEP93		42.28										
	oop Rate			01 30	1	72.20					<u> </u>					<u> </u>
	2-Wire Voice Grade Loop (SL 1) - Zone 1	1	1	UEP93	UECS1	12.48					1		1		1	1
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP93	UECS1	16.31	i t		i i		1		l	İ	l	

UNBUNDLE	D NETWORK ELEMENTS - Tennessee												Attachment:			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC	RATES(\$)					Svc Order Submitted Elec per LSR	Submitted Manually	Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP93	UECS1	21.32										L
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP93	UECS2	16.56										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP93	UECS2	21.63 28.28										ł
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP93	UECS2	28.28										
	ort Rate , LA, MS, & TN only										1					ł
AL, KI,	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP93	UEPYA	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex 9 Dasic Local Area 2-Wire Voice Grade Port (Centrex 800 termination)Basic Local			011 33	ULITA	14.00	30.00	43.00	20.00	10.00		50.03	7.05			-
	Area			UEP93	UEPYB	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area			UEP93	UEPYH	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 Basic Local Area			UEP93	UEPYM	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term - Basic Local Area			UEP93	UEPYZ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port terminated in on Megalink or equivalent															
	- Basic Local Area 2-Wire Voice Grade Port Terminated on 800 Service Term -			UEP93	UEPY9	14.00	90.00	45.00	20.00	10.00		30.89	7.03			<u> </u>
	Basic Local Area			UEP93	UEPY2	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex)			UEP93	UEPQA	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP93	UEPQB	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP93	UEPQH	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP93	UEPQM	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP93	UEPQZ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			
	2-Wire Voice Grade Port terminated in on Megalink or equivalent 2-Wire Voice Grade Port Terminated on 800 Service Term			UEP93 UEP93	UEPQ9 UEPQ2	14.00 14.00	90.00 90.00	45.00 45.00	20.00 20.00	10.00		30.89 30.89	7.03 7.03			ł
L angl S	Switching			UEP93	UEPQZ	14.00	90.00	45.00	20.00	10.00		30.89	7.03			<u> </u>
	Centrex Intercom Funtionality, per port			UEP93	URECS	0.6381										ł
	lumber Portability			011 33	UNLOG	0.0001										
Local N	Local Number Portability (1 per port)			UEP93	LNCCC	0.35										t
Feature				02.00	2.1000	0.00					1					
	All Standard Features Offered, per port			UEP93	UEPVF	0.00										
	All Centrex Control Features Offered, per port			UEP93	UEPVC	0.00					1	1	1	l	l	<u> </u>
NARS							l l									
	Unbundled Network Access Register - Combination			UEP93	UARCX	0.00	0.00	0.00				30.89	7.03			
	Unbundled Network Access Register - Indial			UEP93	UAR1X	0.00	0.00	0.00				30.89	7.03			
	Unbundled Network Access Register - Outdial			UEP93	UAROX	0.00	0.00	0.00			ļ	30.89	7.03			
	aneous Terminations											<u> </u>	ļ			
2-Wire	Trunk Side			UEP93	CEND6	0 70	00.00	45.00	20.00	10.00		20.00	7.00			ł
4-10/1-0	Trunk Side Terminations, each Digital (1.544 Megabits)			05793	CEINDO	8.78	90.00	45.00	20.00	10.00		30.89	7.03			ł
	DS1 Circuit Terminations, each			UEP93	M1HD1	35.55	75.93	38.15				30.89	7.03			ł
	DSI Circuit Terminations, each DS0 Channels Activated, Per Channel			UEP93 UEP93	M1HD1 M1HD0	0.00	108.67	30.15			1	30.89	7.03			t
	rice Channel Mileage - 2-Wire			01 30		0.00	100.07				<u> </u>	50.09	7.03			
	Interoffice Channel Facilities Termination			UEP93	MIGBC	18.58	90.00	45.00	20.00	10.00	1	30.89	7.03			
	Interoffice Channel mileage, per mile or fraction of mile			UEP93	MIGBM	0.0174									İ	<u> </u>
	Activations (DS0) Centrex Loops on Channelized DS1 Servic	e										1	1			
	nnel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP93	1PQWS	0.66										
	Feature Activation on D-4 Channel Bank FX Line Side Loop Slot			UEP93	1PQW6	0.66										<u> </u>
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP93	1PQW7	0.66										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP93	1PQWP	0.66										

UNBUNDLED NETWORK ELEMENTS - Tennessee									Attachment:	2		Exhibit: B				
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Submitted Elec	Submitted	Charge - Manual Svc Order vs.	Charge -	Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
	1		1	1	-	1 -	Nonrecurring		Nonrecurring Disconnect				OSS Rates(\$)		.4	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP93	1PQWV	0.66										
	Feature Activation on D-4 Channel Bank Tie Line/Trunk Loop Slot			UEP93	1PQWQ	0.66										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP93	1PQWA	0.66										
Non-I	Recurring Charges (NRC) Associated with UNE-P Centrex NRC Conversion Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP93	USAC2		1.03	0.29				30.89	7.03			
	New Centrex Standard Common Block			UEP93	M1ACS	0.00	658.60					30.89	7.03			
	New Centrex Customized Common Block			UEP93	M1ACC	0.00	658.60					30.89	7.03			
	NAR Establishment Charge, Per Occasion			UEP93	URECA		68.57					30.89	7.03			
Note	1 - Required Port for Centrex Control in 1AESS, 5ESS & EWSD															
	2 - Requres Interoffice Channel Mileage															
	3 - Requires Specific Customer Premises Equipment															
Note:	Rates displaying an "R" in Interim column are interim and sub	ject to	rate tru	e-up as set forth in	General Tern	ns and Condition	ons.									

Attachment 3

Network Interconnection

TABLE OF CONTENTS

1.	Network Interconnection and the Corresponding Bill and Keep Compensation Mechanism	3	
2.	Interconnection Trunk Group Architectures	7	
3.	Network Design and Management	14	ŀ
4.	Local Dialing Parity	17	!
5.	Interconnection Compensation for Traffic Other Than Local and ESP/ISP Traffic	18	;
6.	Frame Relay Service Interconnection	. 23	;
7.	Operational Support Systems (OSS) Rates	. 27	1
6.	Rates Exhibit	t A	

The Parties shall provide interconnection with each other's networks for the transmission and routing of telephone exchange service (local) and exchange access (intraLATA toll and switched access) on the following terms:

1. Network Interconnection and the Corresponding Bill and Keep Compensation Mechanism

All negotiated rates, terms and conditions set forth in this Attachment pertain only to the provision of network interconnection where Knology owns and provides its switch(es).

- 1.1 Network Interconnection for call transport and termination may be provided by the Parties at any technically feasible point. Requests to BellSouth for interconnection at points other than as set forth in this Attachment may be made through the Bona Fide Request/New Business Request process set out in the General Terms and Conditions. The Parties agree that Knology may establish interconnection through physical interconnection, through Virtual Interconnection as described in Section 1.2.2.1 below, or through a Fiber Meet as described in Section 8 below.
- 1.2 Interconnection Points
- 1.2.1 An Interconnection Point (IP) is the physical telecommunications interface that performs the interconnection function for BellSouth and Knology. The Parties agree that Knology has established physical interconnection arrangements with BellSouth as of the Effective Date of this Agreement at the following locations: Augusta, Ga.; Columbus, GA; Panama City, FL.; Montgomery, AL; Huntsville; AL; Charleston, S.C.; and Knoxville, Tenn. Each Party is responsible for providing the network on its side of the IP.
- 1.2.1.1 For all future IPs, Knology shall establish an IP at each BellSouth access tandem in the LATA to which Knology intends to serve and exchange traffic with BellSouth. Knology may establish physical IPs, including a Fiber Meet as described in Section 8 of this Agreement, at the BellSouth access tandem to satisfy this requirement. In lieu of establishing physical IPs at the BellSouth access tandem, Knology may satisfy this requirement by establishing a Virtual IP as described in Sections 1.2.2 and 1.2.2.1 below if BellSouth has existing facilities in place. In those LATAs where Knology has established the physical interconnection arrangements identified in Section 1.2.1 above, and where such physical interconnection arrangements do not include the establishment of a physical IP at each BellSouth access tandem in that LATA, the Parties will assume that a Virtual IP is established at each applicable access tandem. This interconnection shall provide for the exchange of traffic between BellSouth and Knology within the basic local calling areas served by such tandems. It

may also be used by Knology to exchange Transit Traffic with third Parties subtending such tandems, including Interexchange Carriers with whom Knology exchanges Switched Access Traffic.

- 1.2.1.2 <u>Geographically Relevant Interconnection Points ("GR-Ips)</u> In addition to the establishment of IPs at each BellSouth access tandem, BellSouth may require that Knology establish IPs that are geographically relevant to the NXXs (and associated rate centers) assigned by Knology to serve end users within a basic local calling area. Such geographically relevant IP(s), if required, shall be established by Knology at BellSouth local tandems, or if there is not a BellSouth local tandem, at a mutually acceptable BellSouth end office. This interconnection shall provide for the exchange of traffic between BellSouth and Knology end users within such basic local calling areas. Such GR-Ips may be physical interconnection points as described in 1.2.1 or Virtual Interconnection Points as described in Sections 1.2.2 and 1.2.2.1.
- 1.2.1.3 Knology shall notify BellSouth prior to the activation of new NXXs to determine if a geographically relevant IP shall be required, and such IPs shall be established prior to the activation of new NXXs by Knology. If Knology fails to establish such IP(s) as provided herein, then Knology shall be deemed to have established a Virtual Interconnection Point (as described in Sections 1.2.2 and 1.2.2.1 below), and BellSouth shall bill and Knology shall pay nonrecurring and monthly recurring transport charges based on the dedicated interoffice transport rates in Exhibit A. The charges shall be calculated from a designated BellSouth local tandem or end office, pursuant to the previous section and within the basic local calling area where Knology has assigned NXXs, to an existing Knology IP for BellSouth originated traffic from such basic local calling area that is delivered to such existing Knology IP.
- 1.2.2 <u>Virtual Interconnection Points</u> A Virtual IP, for purposes of this Attachment 3, allows Knology to establish an IP at a BellSouth tandem or end office without providing the physical facilities to, or establishing a collocation arrangement within, such BellSouth office. In lieu of providing an IP in a collocation arrangement and if existing BellSouth facilities are in place, Knology may choose to designate a Virtual IP and BellSouth shall charge and Knology shall pay the nonrecurring and monthly recurring costbased dedicated interoffice transport rates from the Virtual IP location to the physical Knology IP location designated by Knology at such rates as set forth in Exhibit A to this Attachment. A Virtual IP arrangement may be used if collocation within a BellSouth tandem or end office is not feasible. The Parties will implement Virtual IP to Physical IP Connections according

to BellSouth's customary ordering processes consistent with Knology's ASRs.

- 1.2.2.1 Effective in 2003 and upon BellSouth's notification, Knology will implement Virtual IP using the appropriate Virtual IP rate elements in Exhibit A.
- 1.2.3 At any time that Knology establishes a collocation arrangement at a BellSouth local tandem or end office, then either Party may request that such collocation arrangement be established as an IP for the exchange of Local Traffic between Knology's end users and BellSouth end users. Such request and approval shall not be unreasonably withheld or delayed.

1.2.4 [SECTION INTENTIONALLY OMITTED]

- 1.2.5 Furthermore, unless the Parties otherwise agree, Knology, in its capacity as the originating Party, must establish direct end office trunking facilities (the charges for which will be subject to the bill and keep provisions of this Agreement) to a BellSouth end office (which may have a tandem routed overflow) if the volume of Knology originated traffic destined for that end office exceeds the equivalent capacity of two (2) DS1 circuits connecting that end office ("overloaded end office") and the BellSouth tandem ("overloaded tandem") to which it is subtended.
- 1.2.5.1 Should Knology fail to order trunking facilities in a timely manner so as to comply with this end office trunking requirement, BellSouth will notify Knology that the bill and keep compensation arrangement set forth in this Agreement shall no longer apply for Knology originated traffic terminated by BellSouth to that overloaded end office, and Knology shall pay the appropriate call transport and termination rates set forth in Exhibit A to this Attachment 3 for delivering that Knology-originated traffic to the overloaded end office. In the event Knology has properly forecasted and ordered the required trunking from BellSouth and BellSouth has been unable to provision the ordered trunking, Knology shall not be obligated to pay such reciprocal compensation until BellSouth is able to provide the requested trunking.
- 1.2.6 Bill and Keep Compensation
- 1.2.6.1 As of the effective date of this Agreement, in accordance with subsections 1.2.6.1.1 and 1.2.6.1.2, the Parties' reciprocal compensation obligations pursuant to 47 USC Section 251(b)(5), with the exception of multiplexing, shall be subject to the bill and keep compensation plan (described herein) under which neither Party will charge the other Party for call transport and termination compensation for Local and ISP-Bound Traffic between the

Parties. The Parties shall also institute a bill and keep compensation plan under which neither Party will charge the other Party recurring and nonrecurring charges associated with trunks and facilities for the exchange of Local and ISP-bound Traffic.

- 1.2.6.1.1. For purposes of this Agreement, a "bill and keep" compensation plan refers to the Parties' mutual waiver of any and all transport and termination charges customarily recovered from one another (pursuant to 47 U.S.C. Section 251(b)(5) for the exchange of Local Traffic and Enhanced Service Provider/Information Service Provider Traffic between the Parties. Such waived charges include all recurring and non-recurring charges such as (i) transport charges to and from the access tandem or end office at which Knology has an IP (other than a Virtual IP), (ii) call completion charges (including end office switching), and (iii) trunks and associated facilities charges for facilities connecting the Parties' networks on the respective sides of the physical IP. The bill and keep compensation plan does not apply to charges (i) for access traffic, other toll traffic, and transit traffic between the Parties, (ii) Multiplexing, or (iii) for dedicated transport and applicable charges pursuant to Sections 1.2.2 and 1.2.2.1.
- 1.2.6.1.2. As designated on the spreadsheet attached to this Attachment 3 as "Exhibit A," the charges designated with a "bk" and which are subject to the Percent Local Facility ("PLF") and Percent Local Usage ("PLU") Factors are to be applied in accordance with the Bill and Keep Compensation provisions of this Section 1.2.6. Charges that are 1) not designated with a "bk" on the spreadsheet, or 2) charges that are not subject to PLF and PLU Factors are not subject to the Bill and Keep Compensation provisions of this Section 1.2.6. All rate elements not listed in the spreadsheet are subject to future inclusion in this Agreement, subject to mutual agreement by the Parties consistent with the terms of this Agreement.
- 1.2.6.2 For reciprocal compensation between the Parties pursuant to this Attachment, Local Traffic is defined as any circuit switched call that is originated by an end user of one Party and terminated to an end user of the other Party within a given LATA on that other Party's network, except for those calls that are originated or terminated through switched access arrangements as established by the ruling regulatory body.

Additionally, Local Traffic includes any cross boundary, voice-to-voice intrastate, interLATA or interstate, interLATA calls between specific wire centers established as a local call by the ruling regulatory body.

1.2.6.3 ISP-bound Traffic is defined as calls to an information service provider or Internet service provider ("ISP") that are dialed by using a local dialing pattern (7 or 10 digits) by a calling party in one LATA to an ISP server or modem in the same LATA. ISP-bound Traffic is not Local Traffic subject to reciprocal compensation, but instead is information access traffic subject to the FCC's jurisdiction.

- 1.2.6.4 Notwithstanding the definitions of Local Traffic and ISP-bound traffic above, and pursuant to the FCC's Order on Remand and Report and Order in CC Docket 99-68 released April 27, 2001 ("ISP Order on Remand"), BellSouth and Knology agree to the rebuttable presumption that all combined circuit switched Local and ISP-bound Traffic delivered to BellSouth or Knology that exceeds a 3:1 ratio of terminating to originating traffic on a statewide basis shall be considered ISP-bound traffic for compensation purposes. BellSouth and Knology further agree to the rebuttable presumption that all combined circuit switched Local and ISP-bound Traffic delivered to BellSouth or the presumption that all combined circuit switched Local and ISP-bound Traffic delivered to BellSouth or Knology further agree to the rebuttable presumption that all combined circuit switched Local and ISP-bound Traffic delivered to BellSouth or Knology that does not exceed a 3:1 ratio of terminating to originating traffic on a statewide basis shall be considered Local Traffic for compensation purposes.
- 1.2.6.4.1 Neither Party shall pay compensation to the other Party for per minute of use rate elements associated with the Call Transport and Termination of Local Traffic or ISP-bound Traffic.
- 1.2.6.5 Nothing in this Agreement shall be construed to limit either Party's ability to designate the areas within which that Party's end users may make calls which that Party rates as "local" in its end user tariffs.
- 1.2.6.6 Neither Party shall represent access services traffic as Local Traffic.
- 1.2.6.7 The jurisdiction of a call is determined by its originating and terminating (end-to-end) points. If Knology assigns NPA/NXXs to specific BellSouth rate centers within a LATA and assigns numbers from those NPA/NXXs to Knology end users physically located outside of that LATA, BellSouth traffic originating from within the BellSouth rate center where the NPA/NXX is assigned and terminating to a Knology customer physically located outside of that LATA shall not be deemed Local Traffic.
- 1.2.6.7.1 To the extent Knology utilizes its NPA/NXXs to collect traffic from BellSouth end users that appears to be Local Traffic, but then delivers that traffic to Knology's end users located outside the LATA in which the call originated, Knology shall identify such traffic to BellSouth and compensate BellSouth based on the applicable rates for originating intrastate or interstate network access service as reflected in BellSouth's Intrastate or Interstate Access Service Tariff.
- 1.2.6.7.2 If Knology does not identify such traffic to BellSouth, to the best of BellSouth's ability BellSouth will determine which whole Knology

NPA/NXXs on which to charge the applicable rates for originating intrastate or interstate network access service as reflected in BellSouth's Intrastate Interstate Access Service Tariff. BellSouth shall make appropriate billing adjustments if Knology can provide sufficient information for BellSouth to determine whether said traffic is Local Traffic.

1.2.6.8 BellSouth shall be compensated for Knology's ordering of trunks and facilities transporting Transit Traffic as well as the elemental functions BellSouth performs in the transport and termination of Knology's Transit Traffic in accordance with this Agreement.

1.2.6.9 [Section Intentionally Omitted]

2.0 Interconnection Trunk Group Architectures

- 2.1 BellSouth and Knology shall establish interconnecting trunk groups and trunk group configurations between networks including the establishment of one-way or two-way trunks in accordance with the following provisions set forth in this Agreement. For trunking purposes, traffic will be routed based on the digits dialed by the originating end user and in accordance with the Local Exchange Routing Guide (LERG).
- 2.1.1 Any Knology interconnection request that deviates from the trunking architectures as described in this Agreement that affects traffic delivered to Knology from a BellSouth switch that requires special BellSouth switch translations and other network modifications will require Knology to submit a Bona Fide Request/New Business Request via the Bona Fide Request/New Business Request process set forth in the General Terms and Conditions.
- 2.2 Knology shall establish an interconnection trunk group(s) to all BellSouth access and local tandems in the LATA where Knology has homed (i.e. assigned) its NPA/NXXs. Knology shall home its NPA/NXXs on the BellSouth tandems that serve the Exchange Rate Center Areas to which the NPA/NXXs are assigned. The specified association between BellSouth tandems and Exchange Rate Centers is defined in the LERG. Knology shall enter its NPA/NXX access and/or local tandem homing arrangement into the LERG.
- 2.2.1 Switched Access traffic will be delivered to and by IXCs based on Knology's NXX Access Tandem homing arrangement as specified by Knology in the national Local Exchange Routing Guide (LERG).
- 2.3 All trunk groups will be provisioned as Signaling System 7 (SS7) capable where technically feasible. If SS7 is not technically feasible multi-frequency (MF) protocol signaling shall be used.

2.4 In cases where Knology is also an IXC, the IXC's Feature Group D (FG D) trunk groups must remain separate from the local interconnection trunk groups.

2.5 Unless in response to a blocking situation or for a project, when either Party orders interconnection trunk group augmentations, a Firm Order Confirmation (FOC) shall be returned to the ordering Party within four (4) business days from receipt of a valid error free ASR. A project is defined as a new trunk group or the request of 96 or more trunks on a single or multiple trunk group(s) in a given local calling area. Blocking situations and projects shall be managed through the BellSouth Interconnection Trunking Project Management group and Knology's equivalent trunking group.

2.6 Interconnection Trunk Groups for Exchange of Local, IntraLATA Toll and Transit Traffic

2.6.1 If the Parties' originated local and/or intraLATA toll traffic is utilizing the same two-way trunk group, the Parties shall mutually agree to use this type of two-way interconnection trunk group with the quantity of trunks being mutually determined and the provisioning being jointly coordinated. Furthermore, the Interconnection Point(s) for two-way interconnection trunk groups transporting both Parties local and/or intraLATA toll shall be mutually agreed upon. Knology shall order such two-way trunks via the Access Service Request (ASR) process in place for Local Interconnection upon determination by the Parties, in a joint planning meeting, that such trunk groups shall be utilized. BellSouth will use the Trunk Group Service Request (TGSR) to request changes in trunking. Both Parties reserve the right to issue ASRs, if so required, in the normal course of business. Furthermore, the Parties shall jointly review such trunk performance and forecasts on a periodic basis. The Parties use of two-way interconnection trunk groups for the transport of local and/or intraLATA toll traffic between the Parties does not preclude either Party from establishing additional one-way interconnection trunks for the delivery of its originated local and/or intraLATA toll traffic to the other Party.

2.6.2 BellSouth Access Tandem Interconnection Architectures

2.6.2.1 BellSouth Access Tandem Interconnection provides intratandem access to subtending end offices. BellSouth Multiple Tandem Access (MTA), described later in this Agreement, may be ordered using any of the following access tandem architectures.

2.6.2.2 Basic Architecture

2.6.2.2.1 In this architecture, Knology's originating Local and IntraLATA Toll and originating and terminating Transit Traffic is transported on a single twoway trunk group between Knology and BellSouth access tandem(s) within a LATA. This group carries intratandem Transit Traffic between Knology and Independent Companies, Interexchange Carriers, other CLECs and other network providers with which Knology desires interconnection and has the proper contractual arrangements. This group also carries Knology originated intertandem traffic transiting a single BellSouth access tandem destined to third party tandems such as an Independent Company tandem or other CLEC tandem. BellSouth originated Local and IntraLATA Toll traffic is transported on a single one-way trunk group terminating to Knology. Other trunk groups for operator services, directory assistance, emergency services and intercept may be established if required. The LERG should be referenced for current routing and tandem serving arrangements. The Basic Architecture is illustrated in Exhibit B.

2.6.2.3 One-Way Trunk Group Architecture

2.6.2.3.1 In this architecture, the Parties interconnect using two one-way trunk groups. One one-way trunk group carries Knology-originated local and intraLATA toll traffic destined for BellSouth end-users. The other oneway trunk group carries BellSouth-originated local and intraLATA toll traffic destined for Knology end-users. A third two-way trunk group is established for Knology's originating and terminating Transit Traffic. This group carries intratandem Transit Traffic between Knology and Independent Companies, Interexchange Carriers, other CLECs and other network providers with which Knology desires interconnection and has the proper contractual arrangements. This group also carries Knology originated intertandem traffic transiting a single BellSouth access tandem destined to third party tandems such as an Independent Company tandem or other CLEC tandem. Other trunk groups for operator services, directory assistance, emergency services and intercept may be established if required. The LERG should be referenced for current routing and tandem serving arrangements. The One-Way Trunk Group Architecture is illustrated in Exhibit C.

2.6.2.4 **Two-Way Trunk Group Architecture**

2.6.2.4.1 The Two-Way Trunk Group Architecture establishes one two-way trunk group to carry local and intraLATA toll traffic between Knology and BellSouth. In addition, a two-way transit trunk group must be established for Knology's originating and terminating Transit Traffic. This group carries intratandem Transit Traffic between Knology and Independent Companies, Interexchange Carriers, other CLECs and other network providers with which Knology desires interconnection and has the proper contractual arrangements. This group also carries Knology originated intertandem traffic transiting a single BellSouth access tandem destined to third party tandems such as an Independent Company tandem or other CLEC tandem. Other trunk groups for operator services, directory assistance, emergency services and intercept may be established if required. The LERG should be referenced for current routing and tandem serving arrangements. The Two-Way Trunk Group Architecture is illustrated in Exhibit D.

2.6.2.5 **Supergroup Architecture**

2.6.2.5.1 In the Supergroup Architecture, the Parties Local and IntraLATA Toll and Knology's Transit Traffic are exchanged on a single two-way trunk group between Knology and BellSouth. This group carries intratandem Transit Traffic between Knology and Independent Companies, Interexchange Carriers, other CLECs and other network providers with which Knology desires interconnection and has the proper contractual arrangements. This group also carries Knology originated intertandem traffic transiting a single BellSouth access tandem destined to third party tandems such as an Independent Company tandem or other CLEC tandem. Other trunk groups for operator services, directory assistance, emergency services and intercept may be established if required. The LERG should be referenced for current routing and tandem serving arrangements. The Supergroup Architecture is illustrated in Exhibit E.

2.6.3 Local Tandem Interconnection

- 2.6.3.1 Local Tandem Interconnection arrangement allows Knology to establish an interconnection trunk group(s) at BellSouth local tandems for: (1) the delivery of Knology-originated Local Traffic transported and terminated by BellSouth to BellSouth end offices within the local calling area as defined in BellSouth's General Subscriber Services Tariff (GSST), section A3 served by those BellSouth local tandems, and (2) for local Transit Traffic transported by BellSouth for third party network providers who have also established an interconnection trunk group(s) at those BellSouth local tandems.
- 2.6.3.2 When a specified local calling area is served by more than one BellSouth local tandem, Knology must designate a "home" local tandem for each of its assigned NPA/NXXs and establish trunk connections to such local tandems. Additionally, Knology may choose to establish an interconnection trunk group(s) at the BellSouth local tandems where it has no codes homing but is not required to do so. Knology may deliver local traffic to a "home" BellSouth local tandem that is destined for other BellSouth or third party network provider end offices subtending other BellSouth local tandems in the same local calling area where Knology does not choose to establish an interconnection trunk group(s). It is Knology's responsibility to enter its own NPA/NXX local tandem homing arrangements into the Local Exchange Routing Guide (LERG) either directly or via a vendor in order for other third party network providers to determine appropriate traffic routing to Knology's codes. Likewise, Knology shall obtain its routing information from the LERG.

- 2.6.3.3 Notwithstanding establishing an interconnection trunk group(s) to BellSouth's local tandems, Knology must also establish an interconnection trunk group(s) to BellSouth access tandems within the LATA on which Knology has NPA/NXXs homed for the delivery of Interexchange Carrier Switched Access (SWA) and toll traffic, and traffic to Type 2A CMRS connections located at the access tandems. BellSouth shall not switch SWA traffic through more than one BellSouth access tandem. SWA, Type 2A CMRS or toll traffic routed to the local tandem in error will not be backhauled to the BellSouth access tandem for completion. (Type 2A CMRS interconnection is defined in BellSouth's A35 General Subscriber Services Tariff).
- 2.6.3.4 BellSouth's provisioning of local tandem interconnection assumes that Knology has executed the necessary local interconnection agreements with the other third party network providers subtending those local tandems as required by the Act.

2.6.4 **Direct End Office-to-End Office Interconnection**

- 2.6.4.1 Direct End Office-to-End Office one-way or two-way interconnection trunk groups allow for the delivery of a Party's originating local or intraLATA toll traffic to the terminating Party on a Direct End Office-to-End Office basis.
- 2.6.4.2 The Parties shall utilize Direct End Office-to-End Office trunk groups under the following conditions:
- 2.6.4.3 Tandem Exhaust If a tandem through which the Parties are interconnected is unable to, or is forecasted to be unable to support additional traffic loads for any period of time, the Parties will mutually agree on an end office trunking plan that will alleviate the tandem capacity shortage and ensure completion of traffic between Knology and BellSouth's subscribers.
- 2.6.4.4 Traffic Volume To the extent either Party has the capability to measure the amount of traffic between a Knology switching center and a BellSouth end office, either Party shall install and retain direct end office trunking sufficient to handle actual or reasonably forecasted traffic volumes, whichever is greater, between a Knology switching center and a BellSouth end office where the traffic exceeds or is forecasted to exceed a single DS1 of local traffic per month. Either Party will install additional capacity between such points when overflow traffic between Knology's switching center and BellSouth's end office exceeds or is forecasted to exceed a single DS1 of local traffic per month. In the case of one way trunking, additional trunking shall only be required by the Party whose trunking has achieved the preceding usage threshold.

2.6.4.5 Mutual Agreement - The Parties may install direct end office trunking upon mutual agreement in the absence of conditions (1) or (2) above and agreement will not unreasonably be withheld.

2.6.5 Transit Traffic Trunk Group

2.6.5.1 Transit Traffic trunks can either be two-way trunks or two one-way trunks ordered by Knology to deliver and receive local and intraLATA toll Transit Traffic from third parties, such as Independent Companies and other CLECs, via BellSouth access tandems (or BellSouth local tandems for local traffic), and Switched Access traffic to and from Interexchange Carriers via BellSouth access tandems pursuant to the Transit Traffic section of this Attachment. Establishing Transit Traffic trunks at BellSouth access and local tandems provides intratandem access to the third parties also interconnected at those tandems.

2.6.5.2Toll Free Traffic

- 2.6.5.2.1 If Knology chooses BellSouth to handle Toll Free database queries from its switches, all Knology originating Toll Free traffic will be routed over the Transit Traffic Trunk Group.
- 2.6.5.2.2 All originating Toll Free Service (Toll Free) calls for which Knology requests that BellSouth perform the Service Switching Point ("SSP") function (i.e., perform the database query) shall be delivered using GR-394 format over the Transit Traffic Trunk Group. Carrier Code "0110" and Circuit Code (to be determined for each LATA) shall be used for all such calls.
- 2.6.5.2.3 Knology may handle its own Toll Free database queries from its switch. If so, Knology will determine the nature (local/intraLATA/interLATA) of the Toll Free call based on the response from the database. If the query determines that the call is a BellSouth local or intraLATA Toll Free number, Knology will route the post-query local or IntraLATA converted ten-digit local number to BellSouth over the local or intraLATA trunk group. If the query determines that the call is a third party (ICO or other CLEC) local or intraLATA Toll Free number, Knology will route the postquery local or intraLATA converted ten-digit local number to BellSouth over the Transit Traffic Trunk Group. In such case, Knology is to provide a Toll Free billing record when appropriate. If the query reveals the call is an interLATA Toll Free number, Knology will route the post-query interLATA call (Toll Free number) directly from its switch for carriers interconnected with its network or over the Transit Traffic Trunk Group to carriers not directly connected to its network but are connected to BellSouth's access tandem. Calls will be routed to BellSouth over the

local/intraLATA and Transit Traffic Trunk Groups within the LATA in which the calls originate.

2.6.5.2.4 All post-query Toll Free Service (Toll Free) calls for which Knology performs the SSP function, if delivered to BellSouth, shall be delivered using GR-394 format for calls destined to IXCs, and GR-317 format for calls destined to end offices that directly subtend the BellSouth access tandem.

3. Network Design And Management For Interconnection

- 3.1 <u>Network Management and Changes</u>. Both Parties will work cooperatively with each other to install and maintain the most effective and reliable interconnected telecommunications networks, including but not limited to, the exchange of toll-free maintenance contact numbers and escalation procedures. Both Parties agree to provide public notice of changes in the information necessary for the transmission and routing of services using its local exchange facilities or networks, as well as of any other changes that would affect the interoperability of those facilities and networks.
- 3.2 <u>Interconnection Technical Standards</u>. The interconnection of all networks will be based upon accepted industry/national guidelines for transmission standards and traffic blocking criteria. Interconnecting facilities shall conform, at a minimum, to the telecommunications industry standard of DS-1 pursuant to Bellcore Standard No. TR-NWT-00499. Signal transfer point, Signaling System 7 ("SS7") connectivity is required at each interconnection point. BellSouth will provide out-of-band signaling using Common Channel Signaling Access Capability where technically and economically feasible, in accordance with the technical specifications set forth in the BellSouth Guidelines to Technical Publication, TR-TSV-000905. Facilities of each Party shall provide the necessary on-hook, offhook answer and disconnect supervision and shall hand off calling number ID (Calling Party Number) when technically feasible.
- 3.3 <u>Quality of Interconnection</u>. The local interconnection for the transmission and routing of telephone exchange service and exchange access that each Party provides to each other will be at least equal in quality to what it provides to itself and any subsidiary or affiliate, where technically feasible, or to any other Party to which each Party provides local interconnection.
- 3.4 <u>Network Management Controls</u>. Both Parties will work cooperatively with each other to apply sound network management principles by invoking appropriate network management controls (e.g., call gapping) to alleviate or prevent network congestion.

- 3.5 <u>Common Channel Signaling</u>. Both Parties will provide LEC-to-LEC Common Channel Signaling ("CCS") to each other, where available, in conjunction with all traffic in order to enable full interoperability of CLASS features and functions except for call return. All CCS signaling parameters will be provided, including automatic number identification ("ANI"), originating line information ("OLI") calling company category, charge number, etc. All privacy indicators will be honored, and each Party will cooperate with each other on the exchange of Transactional Capabilities Application Part ("TCAP") messages to facilitate full interoperability of CCS-based features between the respective networks. Neither Party shall alter the CCS parameters, or be a party to altering such parameters, or knowingly pass CCS parameters that have been altered in order to circumvent appropriate interconnection charges.
- 3.6 <u>Signaling Call Information</u>. BellSouth and Knology will send and receive ten (10) digits for local traffic. Additionally, BellSouth and Knology will exchange the proper call information, i.e. originated call company number and destination call company number, CIC, and OZZ, including all proper translations for routing between networks and any information necessary for billing.

3.7 Forecasting for Trunk Provisioning

- 3.7.1 Within six (6) months after execution of this agreement, Knology shall provide an initial interconnection trunk group forecast for each LATA that it shall provide service within BellSouth's region, except for LATAs for which Knology has previously provided such forecasts. Upon receipt of Knology's forecast, the Parties shall schedule and participate in a joint planning meeting to develop a joint interconnection trunk group forecast. Each forecast provided under this Section shall be deemed "Confidential Information" under the General Terms and Conditions Part A of this Agreement.
- 3.7.1.1 At a minimum, the forecast shall include the projected quantity of Transit Trunks, Knology-to-BellSouth one-way trunks ("Knology Trunks"), BellSouth-to-Knology one-way trunks ("Reciprocal Trunks") and/or twoway interconnection trunks, if the Parties have agreed to interconnect using two-way trunking to transport the Parties' local and intraLATA toll. The quantities shall be projected for a minimum of six (6) months in advance and shall include the current year plus next two (2) years total forecasted quantities. Considering Knology's provided forecast, the Parties shall mutually develop Reciprocal Trunk and/or two-way interconnection trunk forecast quantities for the time periods listed and to be included within the initial forecast.

- 3.7.1.2 Additionally all forecasts shall include, at a minimum, Access Carrier Terminal Location ("ACTL"), trunk group type (local/intraLATA toll, Transit, Operator Services, 911, etc.), A location/Z location (CLLI codes for Knology location and BellSouth location where the trunks shall terminate), interface type (e.g., DS1), Direction of Signaling, Trunk Group Number, if known, (commonly referred to as the 2-6 code) and forecasted trunks in service each year (cumulative).
- 3.7.2 Each Party shall exercise its commercially reasonable efforts to provide the quantity of interconnection trunks mutually forecasted. However, the provision of the forecasted quantity of interconnection trunks is subject to trunk terminations and facility capacity existing at the time the trunk order is submitted. Furthermore, the receipt and development of trunk forecasts does not imply any liability for failure to perform if capacity (trunk terminations or facilities) is not available for use at the forecasted time.
- 3.7.3 The submitting and development of interconnection trunk forecasts shall not replace the ordering process in place for local interconnection trunks.
- 3.7.4 Once initial interconnection trunk forecasts have been developed, Knology shall continue to provide interconnection trunk forecasts on a semiannual basis or at otherwise mutually agreeable intervals. Knology shall use its commercially reasonable efforts to make the forecasts as accurate as possible based on reasonable engineering criteria. Interconnection trunk forecasts shall be updated and provided to BellSouth on an as needed basis, but no less frequently than semiannually and no more frequently than monthly. Upon receipt of Knology's forecast, including forecast updates, the Parties shall confer to mutually develop BellSouth Reciprocal Trunk and/or two-way interconnection trunk forecasts.

3.8 **Trunk Utilization**

3.8.1 BellSouth and Knology shall monitor traffic on each interconnection trunk group that is installed pursuant to the initial interconnection trunk requirements and subsequent forecasts. At any time after the end of a calendar quarter, based on a review of the capacity utilization during such quarter for installed Reciprocal Trunk groups and/or two-way interconnection trunk groups, subject to the provision of the section following, after fifteen (15) business days advance written notice to Knology, BellSouth may disconnect any Non-utilized Reciprocal Trunk(s) and/or request Knology to disconnect any Non-utilized two-way interconnection trunk(s) if BellSouth has determined that the trunk group is not being utilized at eighty-five percent (85%) of the time consistent busy hour utilization level, provided that the Parties have not otherwise agreed. Non-utilized trunks are defined as the trunks not being utilized as a result of a time consistent busy hour utilization of less than eighty-five percent (85%).

- 3.8.1.1 Within ten (10) business days following the notice prescribed in the section above, Knology may request that BellSouth not disconnect or not request disconnection for some or all of the Non-utilized Trunks, in which event BellSouth shall keep the trunks in service and may invoice Knology for, and Knology shall pay, all applicable recurring and nonrecurring trunk and facility access tariff charges for the Non-utilized Trunks. The charges shall be applied retroactive to the date on which such trunks were installed and to continue until such trunks are disconnected, or to the extent Knology requests that such trunks remain in service, until the trunk group reaches an eighty-five percent (85%) time-consistent busy hour utilization level. In addition, Knology shall reimburse BellSouth for any nonrecurring and/or recurring charges BellSouth may have paid to Knology for the Non-utilized Trunks and for any trunk installation expense BellSouth incurred. This expense shall equal the nonrecurring installation charge for trunks in BellSouth's intrastate tariff. Furthermore, the Knology forecasts for each subsequent forecast period shall be automatically reduced by the number of Reciprocal Trunks and/or two-way interconnection trunks that have been determined to be subject to disconnection pursuant to the foregoing procedures.
- 3.8.2 To the extent Knology requests BellSouth and BellSouth agrees to install additional Reciprocal and/or two-way interconnection trunks in any forecast period following the initial forecasting period that are not included in the forecast for that period (as such forecast may be revised from time to time), such trunks may be provisioned by BellSouth subject to the conditions set forth in the preceding sections above, and all applicable recurring and nonrecurring charges for such trunks shall be billed to and paid by Knology until such trunk groups reach an eighty-five percent (85%) time-consistent busy hour utilization level for Local Traffic.
- 3.8.3 To the extent that any interconnection trunk group is utilized at a timeconsistent busy hour of ninety percent (90%) or greater, the Parties shall negotiate in good faith for the installation of augmented facilities.

4. Local Dialing Parity

4.1 BellSouth and Knology shall provide local and toll dialing parity to each other with no unreasonable dialing delays. Dialing parity shall be provided for all originating telecommunications services that require dialing to route a call. BellSouth and Knology shall permit similarly situated telephone

exchange service end users to dial the same number of digits to make a local telephone call notwithstanding the identity of the end user's or the called party's telecommunications service provider.

5 Interconnection Compensation for Traffic Other Than Local and ESP/ISP Traffic as Provided in Section 1.2.6

- 5.1 Billing Factors
- 5.1 <u>Percent Local Use</u>. Each Party will report to the other a Percent Local Usage ("PLU") factor. The application of the PLU will determine the amount of Local Traffic and ISP-bound Traffic minutes to be billed to the other Party. For purposes of developing the PLU, each Party shall consider every Local call, every ISP-bound call and every long distance call. Each Party shall update its PLU on the first of January, April, July and October of the year and shall send it to the other Party to be received no later than thirty (30) calendar days after the first of each such month based on Local Traffic and ISP-bound Traffic usage for the past three (3) months ending the last day of December, March, June, and September, respectively. Requirements associated with PLU calculation and reporting shall be as set forth in BellSouth's Jurisdictional Factors Reporting Guide, as it is amended from time to time.
- 5.2 <u>Percent Local Facility</u>. Each Party will report to the other a Percent Local Facility ("PLF") factor. The application of the PLF will determine the portion of switched dedicated transport to be billed per the local jurisdiction rates. For purposes of developing the PLF, each Party shall consider every Local call, every ISP-bound call and every long distance call. The PLF shall be applied to multiplexing, local channel and interoffice channel switched dedicated transport utilized in the provision of local interconnection trunks. Each Party shall update its PLF on the first of January, April, July and October of the year and shall send it to the other Party to be received no later than thirty (30) calendar days after the first of each such month to be effective the first bill period the following month, respectively. Requirements associated with PLU and PLF calculation and reporting shall be as set forth in BellSouth's Jurisdictional Factors Reporting Guide, as it is amended from time to time.
- 5.3 <u>Percentage Interstate Usage</u>. Each Party shall report to the other the projected Percent Interstate Usage ("PIU") factor. All jurisdictional report requirements, rules and regulations for Interexchange Carriers specified in BellSouth's Intrastate Access Services Tariff will apply to Knology. After interstate and intrastate traffic percentages have been determined by use of PIU procedures, the PLU and PLF factors will be used for application and

billing of local interconnection. Each Party shall update its PIU on the first of January, April, July and October of the year and shall send it to the other Party to be received no later than thirty (30) calendar days after the first of each such month, for all services showing the percentages of use for the past three (3) months ending the last day of December, March, June and September.

- 5.4 Notwithstanding the provisions in Section 5.1, 5.2, and 5.3 above, where the terminating Party has message recording technology that identifies the jurisdiction of traffic terminated as defined in this Agreement, such information shall, at the terminating Party's option, be utilized to determine the appropriate jurisdictional reporting factors (PLU, PIU, and/or PLF), in lieu of those provided by the originating Party. In the event that the terminating Party opts to utilize its own data to determine jurisdictional reporting factors, such terminating Party shall notify the originating Party at least fifteen (15) days prior to the beginning of the calendar quarter in which the terminating Party will begin to utilize its own data. Such factors shall subject to the Dispute Resolution provisions in this Agreement, as well as the Audit provisions set forth in 5.5 below.
- 5.5 Audits. On thirty (30) days written notice, each Party must provide the other the ability and opportunity to conduct an annual audit to ensure the proper billing of traffic. BellSouth and Knology shall retain records of call detail for a minimum of nine (9) months from which a PLU, PLF and/or PIU can be ascertained. The audit shall be accomplished during normal business hours at an office designated by the Party being audited. Audit requests shall not be submitted more frequently than one (1) time per calendar year. Audits shall be performed by a mutually acceptable independent auditory paid for by the Party requesting the audit. The PLF, PLU and/or PIU shall be adjusted based upon the audit results and shall apply for the quarter the audit was completed, for the quarter prior to the completion of the audit, and for the two (2) quarters following the completion of the audit. If, as a result of an audit, either Party is found to have overstated the PLF, PLU and/or PIU by twenty percentage (20%) points or more, that Party shall reimburse the auditing Party for the cost of the audit.

5.6 8XX Traffic

5.6.1 <u>Compensation for 8XX Traffic</u>. Each Party shall compensate the other pursuant to the appropriate switched access charges, including the database query charge as set forth in the BellSouth intrastate or interstate switched access tariffs.

- 5.6.2 <u>Records for 8XX Billing</u>. Each Party will provide to the other the appropriate records necessary for billing intraLATA 8XX customers. The records provided will be in a standard EMI format.
- 5.6.3 <u>8XX Access Screening</u>. BellSouth's provision of 8XX TFD to Knology requires interconnection from Knology to BellSouth 8XX SCP. Such interconnections shall be established pursuant to BellSouth's Common Channel Signaling Interconnection Guidelines and Bellcore's CCS Network Interface Specification document, TR-TSV-000905. Knology shall establish CCS7 interconnection at the BellSouth Local Signal Transfer Points serving the BellSouth 8XX SCPs that Knology desires to query. The terms and conditions for 8XX TFD are set out in BellSouth's Intrastate Access Services Tariff as amended.
- 5.7 Mutual Provision of Switched Access Service
- 5.7.1 <u>Switched Access Traffic</u>. Switched Access Traffic is described in the BellSouth Access Tariff. Additionally, any Public Switched Telephone Network interexchange telecommunications traffic, regardless of transport protocol method, where the physical location of the calling party and the physical location of the called party are in different LATAs, or are in the same LATA and the Parties' Switched Access services are used for the origination or termination of the call, shall be considered Switched Access Traffic.
- 5.7.2 When Knology's end office switch, subtending the BellSouth Access Tandem switch for receipt or delivery of switched access traffic, provides an access service connection to or from an interexchange carrier ("IXC") by either a direct trunk group to the IXC utilizing BellSouth facilities, or via BellSouth's tandem switch, each Party will provide its own access services to the IXC and bill on a multi-bill, multi-tariff meet-point basis. Each Party will bill its own access services rates to the IXC with the exception of the interconnection charge. The interconnection charge will be billed by the Party providing the end office function. Each party will use the Multiple Exchange Carrier Access Billing (MECAB) guidelines to establish meet point billing for all applicable traffic. Thirty (30)-day billing periods will be employed for these arrangements. For tandem routed traffic, the tandem company agrees to provide to the Initial Billing Company as defined in MECAB, at no charge, all the switched access detail usage data, recorded at the access tandem, within no more than sixty (60) days after the recording date. The Initial Billing Company will provide the switched access summary usage data, for all originating and terminating traffic, to all Subsequent Billing Companies as defined in MECAB within ten (10) days of rendering the initial bill to the IXC. Each Party will notify the other when it is not feasible to meet these requirements

so that the customers may be notified for any necessary revenue accrual associated with the significantly delayed recording or billing. As business requirements change data reporting requirements may be modified as necessary.

In the event that either Party fails to provide switched access detailed usage data to the other Party within ninety (90) days after the recording date and the receiving Party is unable to bill and/or collect access revenues due to the sending Party's failure to provide such data within said time period, then the Party failing to send the data as specified herein shall be liable to the other Party in an amount equal to the unbillable or uncollectible revenues. Each company will provide complete documentation to the other to substantiate any claim of unbillable access revenues. A negotiated settlement will be agreed upon between the companies.

- 5.7.3 Each Party will retain for a minimum period of sixty (60) days, access message detail sufficient to recreate any data which is lost or damaged by their company or any third party involved in processing or transporting data.
- 5.7.4 Each Party agrees to recreate the lost or damaged data within forty-eight (48) hours of notification by the other or by an authorized third party handling the data.
- 5.7.5 Each Party also agrees to process the recreated data within forty-eight (48) hours of receipt at its data processing center.
- 5.7.6 All claims should be filed with the other Party within one hundred and twenty (120) days of the receipt of the date of the unbillable usage.
- 5.7.7 The Initial Billing Company shall keep records of its billing activities relating to jointly-provided Intrastate and Interstate access services in sufficient detail to permit the Subsequent Billing Party to, by formal or informal review or audit, to verify the accuracy and reasonableness of the jointly-provided access billing data provided by the Initial billing Party. Each Party agrees to cooperate in such formal or informal reviews or audits and further agrees to jointly review the findings of such reviews or audits in order to resolve any differences concerning the findings thereof.
- 5.7.8 Knology agrees not to deliver switched access traffic to BellSouth for termination except over Knology ordered switched access trunks and facilities.
Transit Traffic Service. BellSouth shall provide tandem switching and transport services for Knology's Transit Traffic. Transit traffic is traffic originating on Knology's network that is switched and/or transported by BellSouth and delivered to a third party's network, or traffic originating on a third Party's network that is switched and/or transported by BellSouth and delivered to Knology's network. Rates for local and ISP-bound Transit Traffic shall be the applicable call transport and termination charges as set forth in Exhibit A to this Attachment. Rates for intraLATA toll and Switched Access Transit Traffic shall be the applicable call transport and termination charges as set forth in BellSouth Interstate or Intrastate Switched Access tariffs. Switched Access Transit Traffic presumes that Knology's end office is subtending the BellSouth Access Tandem for switched access traffic to and from Knology's end users utilizing BellSouth facilities, either by direct trunks with the IXC, or via the BellSouth Access Tandem. Billing associated with all Transit Traffic shall be pursuant to Multiple Exchange Carrier Access Billing (MECAB) guidelines. Traffic between Knology and Wireless Type 1 third parties shall not be treated as Transit Traffic from a routing or billing perspective. Traffic between Knology and Wireless Type 2A or a third party CLEC utilizing BellSouth switching shall not be treated as Transit Traffic from a routing or billing perspective until BellSouth and the Wireless carrier or a third party CLEC utilizing BellSouth switching have the capability to properly meet-point-bill in accordance with MECAB guidelines.

5.8.1 The delivery of traffic which transits the BellSouth network and is transported to another carrier's network is excluded from any BellSouth billing guarantees and will be delivered at the rates stipulated in this Agreement to a terminating carrier. BellSouth agrees to deliver this traffic to the terminating carrier; provided, however, that Knology is solely responsible for negotiating and executing any appropriate contractual agreements with the terminating carrier for the exchange of Transit Traffic through the BellSouth network. BellSouth will not be liable for any compensation to the terminating carrier or to Knology. Knology agrees to compensate BellSouth for any charges or costs for the delivery of Transit Traffic to a connecting carrier on behalf of Knology. Additionally, the Parties agree that any billing to a third party or other telecommunications carrier under this section shall be pursuant to MECAB procedures.

6. Frame Relay Service Interconnection

6.1 In addition to the Local Interconnection services set forth above, BellSouth will offer a network to network Interconnection arrangement between BellSouth's and Knology's frame relay switches as set forth below. The following provisions will apply only to Frame Relay Service and Exchange

Access Frame Relay Service in those states in which Knology is certified and providing Frame Relay Service as a Local Exchange Carrier and where traffic is being exchanged between Knology and BellSouth Frame Relay Switches in the same LATA.

- 6.2 The Parties agree to establish two-way Frame Relay facilities between their respective Frame Relay Switches to the mutually agreed upon Frame Relay Service point(s) of interconnection ("POI(s)") within the LATA. All POIs shall be within the same Frame Relay Network Serving Areas as defined in Section A40 of BellSouth's General Subscriber Service Tariff except as set forth in this Attachment.
- 6.3 Upon the request of either Party, such interconnection will be established where BellSouth and Knology have Frame Relay Switches in the same LATA. Where there are multiple Frame Relay switches in one central office, an interconnection with any one of the switches will be considered an interconnection with all of the switches at that central office for purposes of routing packet traffic.
- 6.4 The Parties agree to provision local and IntraLATA Frame Relay Service and Exchange Access Frame Relay Service (both intrastate and interstate) over Frame Relay interconnection facilities between the respective Frame Relay switches and the POIs.
- 6.5 The Parties agree to assess each other reciprocal charges for the facilities that each provides to the other according to the Percent Local Circuit Use Factor (PLCU), determined as follows:
- 6.5.1 If the data packets originate and terminate in locations in the same LATA, and consistent with the local definitions of the Agreement, the traffic is considered local. Frame Relay framed packet data is transported within Virtual Circuits (VC). For the purposes of this Agreement, if all the data packets transported within a VC remain within the LATA, then consistent with the local definitions in this Agreement, the traffic on that VC is local ("Local VC").
- 6.5.2 If the originating and terminating locations of the two way packet data traffic are not in the same LATA, the traffic on that VC is interLATA ("InterLATA VC").
- 6.5.3 The PLCU is determined by dividing the total number of Local VCs, by the total number of VCs on each Frame Relay facility. To facilitate implementation, Knology may determine its PLCU in aggregate, by dividing the total number of Local VCs in a given LATA by the total number VCs in that LATA. The Parties agree to renegotiate the method

for determining PLCU, at BellSouth's request, and within 90 days, if BellSouth notifies Knology that it has found that this method does not adequately represent the PLCU.

- 6.5.4 If there are no VCs on a facility when it is billed, the PLCU will be zero.
- 6.5.5 BellSouth will provide the circuit between the Parties' respective Frame Relay Switches. The Parties will be compensated as follows: BellSouth will invoice, and Knology will pay, the total non-recurring and recurring charges for the circuit based upon the rates set forth in BellSouth's Interstate Access Tariff, FCC No. 1. Knology will then invoice, and BellSouth will pay, an amount calculated by multiplying the BellSouth billed charges for the circuit by one-half of Knology's PLCU.
- 6.6 The Parties agree to compensate each other for Frame Relay network-tonetwork interface (NNI) ports based upon the NNI rates set forth in BellSouth's Interstate Access Tariff, FCC No. 1. Compensation for each pair of NNI ports will be calculated as follows: BellSouth will invoice, and Knology will pay, the total non-recurring and recurring charges for the NNI port. Knology will then invoice, and BellSouth will pay, an amount calculated by multiplying the BellSouth billed non-recurring and recurring charges for the NNI port by Knology's PLCU.
- 6.7 Each Party agrees that there will be no charges to the other Party for its own subscriber's Permanent Virtual Circuit (PVC) rate elements for the local PVC segment from its Frame Relay switch to its own subscriber's premises. PVC rate elements include the Data Link Connection Identifier (DLCI) and Committed Information Rate (CIR).
- 6.8 For the PVC segment between the Knology and BellSouth Frame Relay switches, compensation for the PVC charges is based upon the rates in BellSouth's Interstate Access Tariff, FCC No. 1.
- 6.9 Compensation for PVC rate elements will be calculated as follows:
- 6.9.1 If Knology orders a VC connection between a BellSouth subscriber's PVC segment and a PVC segment from the BellSouth Frame Relay switch to the Knology Frame Relay switch, BellSouth will invoice, and Knology will pay, the total non-recurring and recurring PVC charges for the PVC segment between the BellSouth and Knology Frame Relay switches. If the VC is a Local VC, Knology will then invoice and BellSouth will pay, the total nonrecurring and recurring PVC charges billed for that segment. If the VC is not local, no compensation will be paid to Knology for the PVC segment.

- 6.9.2 If BellSouth orders a Local VC connection between a Knology subscriber's PVC segment and a PVC segment from the Knology Frame Relay switch to the BellSouth Frame Relay switch, BellSouth will invoice, and Knology will pay, the total non-recurring and recurring PVC and CIR charges for the PVC segment between the BellSouth and Knology Frame Relay switches. If the VC is a Local VC, Knology will then invoice and BellSouth will pay the total non-recurring and recurring PVC and CIR charges billed for that segment. If the VC is not local, no compensation will be paid to Knology for the PVC segment.
- 6.9.3 The Parties agree to compensate each other for requests to change a PVC segment or PVC service order record, according to the Feature Change charge as set forth in the BellSouth access tariff BellSouth Tariff FCC No. 1.
- 6.9.4 If Knology requests a change, BellSouth will invoice and Knology will pay a Feature Change charge for each affected PVC segment.
- 6.9.4.1 If BellSouth requests a change to a Local VC, Knology will invoice and BellSouth will pay a Feature Change charge for each affected PVC segment.
- 6.9.5 The Parties agree to limit the sum of the CIR for the VCs on a DS1 NNI port to not more than three (3) times the port speed, or not more than six (6) times the port speed on a DS3 NNI port.
- 6.9.6 Except as expressly provided herein, this Agreement does not address or alter in any way either Party's provision of Exchange Access Frame Relay Service or interLATA Frame Relay Service. All charges by each Party to the other for carriage of Exchange Access Frame Relay Service or interLATA Frame Relay Service are included in the BellSouth access tariff BellSouth Tariff FCC No. 1.
- 6.10 Knology will identify and report quarterly to BellSouth the PLCU of the Frame Relay facilities it uses, per section 6.5.3 above.
- 6.11 Either Party may request a review or audit of the various service components, consistent with the provisions of section E2 of the BellSouth State Access Services tariffs or Section 2 of the BellSouth FCC No.1 Tariff.
- 6.12 If during the term of this Agreement, BellSouth obtains authority to provide interLATA Frame Relay in any State, the Parties agree to renegotiate this arrangement for the exchange of Frame Relay Service Traffic within one hundred eighty (180) days of the date BellSouth receives

interLATA authority. In the event the Parties fail to renegotiate this Section 6 within the one hundred eighty-day period, they will submit this matter to the appropriate State commission(s) for resolution.

7. Operational Support Systems (OSS) Rates

All Local Service Requests ("LSRs") or Access Service Requests ("ASRs") submitted for products and services under this Attachment will be subject to the OSS charges set forth in the General Terms and Conditions of this Agreement.

8. **Fiber Meet**

- 8.1 A "Fiber Meet" is an interconnection arrangement whereby the Parties physically interconnect their networks via an optical fiber interface at which one Party's facilities, provisioning, and maintenance responsibility begins and the other Party's responsibility ends.
- 8.2. If Knology elects to interconnect with Bellsouth pursuant to a fiber meet, Knology and Bellsouth shall jointly engineer, operate and maintain a synchronous optical network ("SONET") transmission system by which they shall interconnect their transmission and routing of local traffic via a local channel at either the DS1 or DS3 level. The Parties shall work jointly to determine the specific transmission system. However, Knology's SONET transmission system must be compatible with Bellsouth's equipment, and the data communications channel (DCC) must be turned off.
- 8.2.3 Each Party, at its own expense, shall procure, install and maintain the agreed upon SONET transmission system in its network.

- 8.2.4 The Parties shall agree to a fiber meet point between the BellSouth serving wire center and the Knology serving wire center. The Parties shall deliver their fiber optic facilities to the fiber meet point with sufficient spare length to reach the fusion splice point for the fiber meet point. Bellsouth shall, at its own expense, provide and maintain the fusion splice point for the fiber meet. A building type common language location identification ("CLLI") code will be established for each fiber meet point. All orders for interconnection facilities from the fiber meet point shall indicate the fiber meet point as the originating point for the facility.
- 8.2.5 Upon verbal request by Knology, BellSouth shall allow Knology access to the fusion splice point for the fiber meet point for maintenance purposes on Knology's side of the fiber meet point.
- 8.2.6 Neither Party shall charge the other for its local channel portion of the fiber meet facility used exclusively for local traffic. All other appropriate charges will apply. Knology shall be billed for a mixed use of the local channel as set forth in the appropriate tariff(s) using the PIU/PLF factors supplied by Knology. Charges for switched and special access services shall be billed in accordance with the applicable access service tariff.

LOCAL INT	ERCONNECTION - Alabama												Attach	ment: 3	Exhi	bit: A
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES (\$)					Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	I RCONNECTION (CALL TRANSPORT AND TERMINATION)															
	: "bk" beside a rate indicates that the Parties have agreed to bi	II and ke	eep for	that element purs	uant to the ter	rms and conditi	ons in Attachn	nent 3.								
	End Office Switching Function per MOU rate element is not sh								between Knol	ogy and BellS	outh					
TAND	EM SWITCHING															
	Tandem Switching Function Per MOU			OHD		0.000498bk										
	Multiple Tandem Switching, per MOU (applies to intial tandem only)			OHD		0.000498										
	Tandem Intermediary Charge, per MOU*			OHD		0.000498										
* This	charge is applicable only to transit traffic and is applied in ad	dition to	applio		d/or intercon											
TRUN	K CHARGE															
	Installation Trunk Side Service - per DS0			OHD	TPP++	0.00	333.69bk	56.91bk								
	Dedicated End Office Trunk Port Service-per DS0** Dedicated End Office Trunk Port Service-per DS1**			OHD 0H1 OH1MS	TDE0P TDE1P	0.00										<u> </u>
	Dedicated End Once Trunk Port Service-per DS1			OHD	TDE IP	0.00										
	Dedicated Tandem Trunk Port Service-per DS1**			OH1 OH1MS	TDW1P	0.00										
	s rate element is recovered on a per MOU basis and is included	in the	End Of	fice Switching and	Tandem Swi	tching, per MOL	J rate elements	3								
COMM	NON TRANSPORT (Shared)					0.00000011										
	Common Transport - Per Mile, Per MOU Common Transport - Facilities Termination Per MOU			OHD OHD		0.000023bk 0.0003224bk										
	RCONNECTION (DEDICATED TRANSPORT)			ОПО		0.0003224DK										
INTER	ROFFICE CHANNEL - DEDICATED TRANSPORT															
	Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade -															
	Per Mile per month			OHL, OHM	1L5NF	0.008838bk										
	Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade -			OHL, OHM	1L5NF	21.13bk	40.54bk	27.41bk	16.74bk	6.9bk						
	Facility Termination per month Interoffice Channel - Dedicated Transport - 56 kbps - per mile			UHL, UHM	ILDINF	21.13DK	40.54DK	27.41DK	16.74DK	6.9DK						
	per month			OHL, OHM	1L5NK	0.008838bk										
	Interoffice Channel - Dedicated Transport - 56 kbps - Facility															
	Termination per month			ohl, ohm	1L5NK	15.12bk	40.54bk	27.41bk	16.74bk	6.9bk						
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile					0.0000001.1										
	per month Interoffice Channel - Dedicated Transport - 64 kbps - Facility			OHL, OHM	1L5NK	0.008838bk										
	Termination per month			OHL, OHM	1L5NK	15.12bk	40.54bk	27.41bk	16.74bk	6.9bk						
	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per															
	month			OH1, OH1MS	1L5NL	0.18bk										
	Interoffice Channel - Dedicated Tranport - DS1 - Facility															
	Termination per month Interoffice Channel - Dedicated Transport - DS3 - Per Mile per			OH1, OH1MS	1L5NL	60.16bk	89.27bk	81.81bk	16.35bk	14.44bk						
	meronice channel - Dedicated Transport - DSS - Per Mile per month			OH3, OH3MS	1L5NM	4.09bk										
	Interoffice Channel - Dedicated Transport - DS3 - Facility				LOIN	4.0051										
	Termination per month			OH3, OH3MS	1L5NM	703.52k	278.75bk	162.76bk	60.2bk	58.46bk						
LOCA	L CHANNEL - DEDICATED TRANSPORT															
	Local Channel - Dedicated - 2-Wire Voice Grade per month			OHL, OHM	TEFV2	13.97bk	193.1bk	33.17bk	36.64bk	3.2bk						
	Local Channel - Dedicated - 4-Wire Voice Grade per month Local Channel - Dedicated - DS1 per month			OHL, OHM OH1	TEFV4 TEFHG	14.93bk 35.76k	193.53bk 177.47bk	33.6bk 153.72bk	37.11bk 22.19bk	3.67bk 15.26bk			-			
	Local Channel - Dedicated - DST per month			OITI	TEFIIG	33.70K	177.47DK	133.72DK	22.19DK	13.2008						
1.001	Local Channel - Dedicated - DS3 Facility Termination per month			OH3	TEFHJ	416.54bk	451.52bk	263.94bk	119.49bk	83.58bk						ļ
	L INTERCONNECTION MID-SPAN MEET : If Access service ride Mid-Span Meet, one-half the tariffed ser	viceLo	ral Chi	annel rate is annlic	ahle	├										
	Local Channel - Dedicated - DS1 per month	1.00 20		OH1MS	TEFHG	0.00	0.00									
	Local Channel - Dedicated - DS3 per month			OH3MS	TEFHJ	0.00	0.00							1	1	1
LOCA	L INTERCONNECTION - VIRTUAL INTERCONNECTION POINT															
	Local Channel-Dedicated- DS3 - per month			BP3	BP3LC	416.54	451.52	263.94	119.49	83.58						
	Interoffice Channel - Dedicated transport - DS3 per mile per															
	month Interoffice Channel - Dedicated transport - DS3 facility			BP3	1L5BP	4.09										
	unterouice Channel - Legicated transport - US3 facility		1		1	1					1	1		1		1

LOC	al IN	INTERCONNECTION - Alabama												Attachr	nent: 3	Exhi	oit: A				
												Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental				
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -				
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc				
CATE	EGOR	RY RATE ELEMENTS	m	Zone	BCS	USOC			RATES (\$)		per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.					
								Neuromine Neuromine Discourse t						Electronic-	Electronic-	Electronic-	Electronic-				
														1st	Add'l	Disc 1st	Disc Add'l				
							Dee	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)						
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN				
	MU	ULTIPLEXERS																			
		Channelization - DS1 to DS0 Channel System			OH1, OH1MS	SATN1	101.06	91.04	62.57	10.54	9.79										
		DS3 to DS1 Channel System per month			OH3, OH3MS	SATNS	166.13	178.14	93.97	33.26	31.63										
		DS3 Interface Unit (DS1 COCI) per month			OH1, OH1MS	SATCO	12.70	6.58	4.72												
	Not	otes: If no rate is identified in the contract, the rates, terms, and co	ondition	s for th	ne specific service	or function wi	ill be as set forth	in applicable	BellSouth tar	iff.											

LOCAL I	INTE	RCONNECTION - Florida												Attach	ment: 3	Exhi	bit: A
CATEGOR		RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES (\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge -
								Nonrec	urring	Nonrecurring	n Disconnect			055	Rates(\$)		
\vdash							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN		SOMAN	SOMAN	SOMAN
		CONNECTION (CALL TRANSPORT AND TERMINATION)															
		'bk" beside a rate indicates that the Parties have agreed to bil								h atwa an Kn a							
		nd Office Switching Function per MOU rate element is not she M SWITCHING	own be	low bu	t is subject to the B	in and keep	provision of At	tachment 3 of t	ne Agreement	between Knol	logy and Bells	outh		-			
		Tandem Switching Function Per MOU			OHD		0.0006019bk										
		Multiple Tandem Switching, per MOU (applies to intial tandem			-												
		only)			OHD		0.0006019										
		Tandem Intermediary Charge, per MOU*	1141 o m 4 d		OHD		0.0015										
		harge is applicable only to transit traffic and is applied in add CHARGE	attion to	appilo	cable switching and	/or interconr	lection charges	•									
		Installation Trunk Side Service - per DS0			OHD	TPP++		336.43bk	57.38bk								
		Dedicated End Office Trunk Port Service-per DS0**			OHD	TDE0P	0.00										
		Dedicated End Office Trunk Port Service-per DS1**			0H1 OH1MS	TDE1P	0.00										
		Dedicated Tandem Trunk Port Service-per DS0**			OHD	TDW0P	0.00										
**		Dedicated Tandem Trunk Port Service-per DS1** rate element is recovered on a per MOU basis and is included	in the	End Of	OH1 OH1MS	TDW1P	0.00	l rate elements						-			
		ON TRANSPORT (Shared)	in the		ince Switching and	Tandem Swi	ichning, per MOC		•								
		Common Transport - Per Mile, Per MOU			OHD		0.0000035bk										
		Common Transport - Facilities Termination Per MOU			OHD		0.0004372bk										
		CONNECTION (DEDICATED TRANSPORT)															
IN	ITERO	OFFICE CHANNEL - DEDICATED TRANSPORT															
		Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade - Per Mile per month			OHL, OHM	1L5NF	0.0091bk										
		Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade - Facility Termination per month			OHL, OHM	1L5NF	25.32bk	47.35bk	31.78bk	18.31bk	7.03bk						
		Interoffice Channel - Dedicated Transport - 56 kbps - per mile per month				1L5NK	0.0091bk	11100DIX	0111001	1010101	110051						
		Interoffice Channel - Dedicated Transport - 56 kbps - Facility Termination per month			OHL, OHM	1L5NK	18.44bk	47.35bk	31.78bk	18.31bk	7.03bk						
		Interoffice Channel - Dedicated Transport - 64 kbps - per mile						11.00DK	0111001	10.0101	110001						
\vdash		per month			OHL, OHM	1L5NK	0.0091bk										
		Interoffice Channel - Dedicated Transport - 64 kbps - Facility Termination per month			OHL, OHM	1L5NK	18.44bk	47.35bk	31.78bk	18.31bk	7.03bk	:					
		Interoffice Channel - Dedicated Channel - DS1 - Per Mile per month			OH1, OH1MS	1L5NL	0.1856bk										
		Interoffice Channel - Dedicated Tranport - DS1 - Facility Termination per month			OH1, OH1MS	1L5NL	88.44bk	105.54bk	98.47bk	21.47bk	19.05bk						
		Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month			OH3, OH3MS	1L5NM	3.87bk					ļ					
		Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month			OH3, OH3MS	1L5NM	1071bk	335.46bk	219.28bk	72.03bk	70.56bk						
		CHANNEL - DEDICATED TRANSPORT Local Channel - Dedicated - 2-Wire Voice Grade per month			OHL, OHM	TEFV2	19.66bk	265.84bk	46.97bk	37.63bk	4bk						
\vdash		Local Channel - Dedicated - 2-Wire Voice Grade per month			OHL, OHM	TEFV2	20.45bk	265.840k 266.54bk	46.97bk 47.67bk	44.22bk	40k 5.33bk	1					
\vdash		Local Channel - Dedicated - DS1 per month			OHL, OHM OH1	TEFHG	36.49bk	216.65bk	183.54bk	24.3bk	16.95bk						
		·															
\vdash		Local Channel - Dedicated - DS3 Facility Termination per month			OH3	TEFHJ	531.91bk	556.37bk	343.01bk	139.13bk	96.84bk						
LC	OCAL	INTERCONNECTION - VIRTUAL INTERCONNECTION POINT															
\vdash		Local Channel-Dedicated- DS3 - per month			BP3	BP3LC	531.91	556.37	343.01	139.13	96.84						
		Interoffice Channel - Dedicated transport - DS3 per mile per month			BP3	1L5BP	3.87										
		Interoffice Channel - Dedicated transport - DS3 facility termination per month			BP3	1L5BP	1,071.00	335.46	219.28	72.03	70.56						
		INTERCONNECTION MID-SPAN MEET															
	OTE: I	f Access service ride Mid-Span Meet, one-half the tariffed ser Local Channel - Dedicated - DS1 per month	vice Lo	cal Ch	annel rate is applica OH1MS	TEFHG	0.00	0.00									
\vdash		Local Channel - Dedicated - DS1 per month			OH TWS OH 3MS	TEFHG	0.00	0.00									
		Loour onamior Dedicated - Doo per month					0.00	0.00									

LOCAL IN	TERCONNECTION - Florida												Attach	ment: 3	Exhi	bit: A
											Svc Order	Svc Order	Incremental	Incremental	Incremental	Incrementa
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
1		Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES (\$)			per LSR	-	Order vs.	Order vs.		Order vs.
											•	•	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						Dee	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		1
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Channelization - DS1 to DS0 Channel System			OH1, OH1MS	SATN1	146.77	101.42	71.62	11.09	10.49						
	DS3 to DS1 Channel System per month			OH3, OH3MS	SATNS	211.19	199.28	118.64	40.34	39.07						
	DS3 Interface Unit (DS1 COCI) per month			OH1, OH1MS	SATCO	13.76	10.07	7.08								
Note	s: If no rate is identified in the contract, the rates, terms,	and condition	s for th	he specific service	or function wi	II be as set forth	n in applicable	BellSouth tar	iff.							1

LOCAL INT	ERCONNECTION - Georgia												Attach	ment: 3	Exhi	bit: A
											Svc Order	Svc Order	Incremental			Incremental
												Submitted	Charge -	Charge -	Charge -	Charge -
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES (\$)			Elec	-	Manual Svc		Manual Svc	
CATEGORT	RATE ELEMENTS	m	Zone	BC3	0300			KATES (\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	
									-				1st	Add'l	Disc 1st	Disc Add'l
						Rec	Nonrec			g Disconnect				Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
LOCAL INTER	I RCONNECTION (CALL TRANSPORT AND TERMINATION)															
	: "bk" beside a rate indicates that the Parties have agreed to bi	l and k	eep for	that element pursu	ant to the ter	rms and conditi	ions in Attachn	nent 3.								
Note:	End Office Switching Function per MOU rate element is not sh								between Kno	logy and BellS	outh					
TAND	EM SWITCHING			0.115												
	Tandem Switching Function Per MOU Multiple Tandem Switching, per MOU (applies to intial tandem			OHD		0.0011009bk										4
	only)			OHD		0.0011009										
	Tandem Intermediary Charge, per MOU*			OHD		0.0015										
	charge is applicable only to transit traffic and is applied in add	dition to	o appli	cable switching and	/or interconr	nection charges	i.									
TRUN	K CHARGE			0115	TDD	ļ	000.001.1	50.0								<u> </u>
├──-┼───	Installation Trunk Side Service - per DS0 Dedicated End Office Trunk Port Service-per DS0**			OHD OHD	TPP++ TDE0P	0.00	333.28bk	56.84bk			+					
├ ── ├ ──	Dedicated End Office Trunk Port Service-per DS0**			0HD 0H1 OH1MS	TDE0P	0.00				1	+					
	Dedicated Tandem Trunk Port Service-per DS0**			OHD	TDW0P	0.00				1						<u> </u>
	Dedicated Tandem Trunk Port Service-per DS1**			OH1 OH1MS	TDW1P	0.00										
	s rate element is recovered on a per MOU basis and is included	in the	End Of	fice Switching and	Tandem Swit	tching, per MOl	U rate elements	;			<u> </u>					\vdash
COMM	ION TRANSPORT (Shared) Common Transport - Per Mile, Per MOU			OHD		0.000080bk					-					-
├ ── ├ ──	Common Transport - Facilities Termination Per MOU			OHD	+	0.00000800k										
LOCAL INTEF	RCONNECTION (DEDICATED TRANSPORT)			OTID		0.00041320K										
	ROFFICE CHANNEL - DEDICATED TRANSPORT															1
	Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade -															
	Per Mile per month			OHL, OHM	1L5NF	0.0222bk										
	Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade - Facility Termination per month			OHL, OHM	1L5NF	17.07bk	79.61bk	36.08bk								
	Interoffice Channel - Dedicated Transport - 56 kbps - per mile				ILUNI	17.0758	73.0108	30.00DK								1
	per month			OHL, OHM	1L5NK	0.0222bk										
	Interoffice Channel - Dedicated Transport - 56 kbps - Facility															
	Termination per month			OHL, OHM	1L5NK	16.45bk	79.61bk	36.08bk								L
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile per month			OHL, OHM	1L5NK	0.0222bk										
	Interoffice Channel - Dedicated Transport - 64 kbps - Facility				TESINK	0.02220K										
	Termination per month			OHL, OHM	1L5NK	16.45bk	79.61bk	36.08bk								
	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per															
	month			OH1, OH1MS	1L5NL	0.4523bk										
	Interoffice Channel - Dedicated Tranport - DS1 - Facility Termination per month			OH1, OH1MS	1L5NL	78.47bk	147.07bk	111.75bk								
├ ── ├ ──	Interoffice Channel - Dedicated Transport - DS3 - Per Mile per				TLOINE	/0.4/DK	147.07DK	111.730K		1	+					
	month			OH3, OH3MS	1L5NM	2.72bk										
	Interoffice Channel - Dedicated Transport - DS3 - Facility															
<u></u>	Termination per month			OH3, OH3MS	1L5NM	788bk	511.1bk	330.77bk								L
LOCA	L CHANNEL - DEDICATED TRANSPORT			OHL, OHM	TEFV2	13.91bk	382.95bk	62.4bk								┝───
┣───╂────	Local Channel - Dedicated - 2-Wire Voice Grade per month Local Channel - Dedicated - 4-Wire Voice Grade per month			OHL, OHM OHL, OHM	TEFV2 TEFV4	13.91bk 14.99bk	382.950K 368.44bk	62.40K 64.05bk		1	+					├
	Local Channel - Dedicated - 4-Wire Voice Grade per month			OHL, OHM OH1	TEFHG	38.36bk	356.15bk	312.89bk								
	Local Channel - Dedicated - DS3 Facility Termination per month			ОНЗ	TEFHJ	515.91bk	639.5bk	426.31bk								Ļ
LOCA	L INTERCONNECTION - VIRTUAL INTERCONNECTION POINT															<u> </u>
	Local Channel-Dedicated- DS3 - per month		L	BP3	BP3LC	515.91	639.50	426.31								Ļ
	Interoffice Channel - Dedicated transport - DS3 per mile per			802	1L5BP	0.70										
┝──┼──	month Interoffice Channel - Dedicated transport - DS3 facility			BP3	ILOBP	2.72										
	termination per month			BP3	1L5BP	788.00	511.10	330.77								
LOCA	L INTERCONNECTION MID-SPAN MEET		1	-			50	500.11		1		1	-			
NOTE	: If Access service ride Mid-Span Meet, one-half the tariffed ser	vice Lo	cal Ch													
	Local Channel - Dedicated - DS1 per month			OH1MS	TEFHG	0.00	0.00									
	Local Channel - Dedicated - DS3 per month			OH3MS	TEFHJ	0.00	0.00				+					├ ───
MULT	IPLEXERS	1				1			I		1	I		1	1	L

LOCAL IN	TERCONNECTION - Georgia												Attach	ment: 3	Exhi	bit: A
											Svc Order	Svc Order	Incremental	Incremental	Incremental	Incrementa
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
		Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svo
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES (\$)			per LSR	-	Order vs.	Order vs.		Order vs.
												•	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						Dee	Nonrec	urring	Nonrecurrin	g Disconnect			oss	Rates(\$)		1
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Channelization - DS1 to DS0 Channel System			OH1, OH1MS	SATN1	126.22	198.22	123.59								
	DS3 to DS1 Channel System per month			OH3, OH3MS	SATNS	182.04	280.66	195.33								
	DS3 Interface Unit (DS1 COCI) per month			OH1, OH1MS	SATCO	11.02	12.02	8.66								1
Note	es: If no rate is identified in the contract, the rates, terms,	and condition	is for th	he specific service	or function wi	ill be as set forth	n in applicable	BellSouth tar	riff.							1

LOCAL I	NTE	RCONNECTION - Kentucky													ment: 3		bit: A
CATEGOR	۲Y	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES (\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
							Rec	Nonrec	urring	Nonrecurring	Disconnect				Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	TERC	ONNECTION (CALL TRANSPORT AND TERMINATION)															
		bk" beside a rate indicates that the Parties have agreed to bi	ll and k	een for	that element nursu	ant to the ter	rms and conditi	ons in Attachn	nent 3								
		nd Office Switching Function per MOU rate element is not sh								between Knold	ogy and BellS	outh					
	NDEN	I SWITCHING															
		Tandem Switching Function Per MOU			OHD		0.0006772bk										
		Multiple Tandem Switching, per MOU (applies to intial tandem only)			OHD		0.0006772										
		Tandem Intermediary Charge, per MOU*			OHD	-	0.0006772										
* T		harge is applicable only to transit traffic and is applied in ad	dition to	o appliq		d/or intercon											
		CHARGE			,												
		Installation Trunk Side Service - per DS0			OHD	TPP++		334.09bk	57.12bk								
		Dedicated End Office Trunk Port Service-per DS0**			OHD	TDE0P	0.00										
\vdash		Dedicated End Office Trunk Port Service-per DS1** Dedicated Tandem Trunk Port Service-per DS0**			0H1 OH1MS OHD	TDE1P TDW0P	0.00										
\vdash		Dedicated Tandem Trunk Port Service-per DS0			OHD OH1 OH1MS	TDW0P	0.00										
	This r	ate element is recovered on a per MOU basis and is included	in the					J rate elements	;			1					
	оммо	N TRANSPORT (Shared)															
		Common Transport - Per Mile, Per MOU			OHD		0.0000030bk										
		Common Transport - Facilities Termination Per MOU			OHD	-	0.0007466bk										
		ONNECTION (DEDICATED TRANSPORT) FFICE CHANNEL - DEDICATED TRANSPORT				-											
		Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade -															
		Per Mile per month			OHL, OHM	1L5NF	0.01bk										
		Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade -															
		Facility Termination per month			OHL, OHM	1L5NF	29.11bk	47.34bk	31.78bk	22.77bk	8.75bk						
		Interoffice Channel - Dedicated Transport - 56 kbps - per mile per month			OHL, OHM	1L5NK	0.0115bk										
		Interoffice Channel - Dedicated Transport - 56 kbps - Facility				LONK	0.01130K										
		Termination per month			OHL, OHM	1L5NK	20.97bk	47.35bk	31.78bk	22.77bk	8.75bk						
		Interoffice Channel - Dedicated Transport - 64 kbps - per mile															
		per month			OHL, OHM	1L5NK	0.0115bk										
		Interoffice Channel - Dedicated Transport - 64 kbps - Facility			OHL, OHM	1L5NK	00.0754	47 OF L	04 7051	00.7751	0.7551						
	-	Termination per month Interoffice Channel - Dedicated Channel - DS1 - Per Mile per			OHL, OHM	ILDINK	20.97bk	47.35bk	31.78bk	22.77bk	8.75bk						
		month			OH1, OH1MS	1L5NL	0.23bk										
		Interoffice Channel - Dedicated Tranport - DS1 - Facility										İ					
		Termination per month			OH1, OH1MS	1L5NL	96.04bk	105.52bk	98.46bk	23.09bk	20.49bk						
		Interoffice Channel - Dedicated Transport - DS3 - Per Mile per				41.55.154	4.071										
\vdash		month Interoffice Channel - Dedicated Transport - DS3 - Facility			OH3, OH3MS	1L5NM	4.97bk			├							
		Termination per month			OH3, OH3MS	1L5NM	1175.15bk	335.4bk	219.24bk	89.57bk	87.75bk						
LC		CHANNEL - DEDICATED TRANSPORT															
	l	Local Channel - Dedicated - 2-Wire Voice Grade per month			ohl, ohm	TEFV2	18.57bk	265.78bk	46.96bk	46.79bk	4.98bk						
\vdash		Local Channel - Dedicated - 4-Wire Voice Grade per month			OHL, OHM	TEFV4	19.86bk	266.48bk	47.65bk	47.54bk	5.73bk						
\vdash	l	Local Channel - Dedicated - DS1 per month			OH1	TEFHG	40.46bk	209.6bk	176.51bk	30.21bk	21.07bk						
		Local Channel - Dedicated - DS3 Facility Termination per month			ОНЗ	TEFHJ	576.05bk	551.38bk	338.08bk	173bk	120.42bk						
		INTERCONNECTION - VIRTUAL INTERCONNECTION POINT					0.0.000	00.10000	000.0000		.20. /20/						
		Local Channel-Dedicated- DS3 - per month			BP3	BP3LC	576.05	551.38	338.08	173.00	120.42						
	li	Interoffice Channel - Dedicated transport - DS3 per mile per			•	2. 020	0, 0.00	551.00	300.00	170.00	120.42	t					
	r	month			BP3	1L5BP	4.97										
		Interoffice Channel - Dedicated transport - DS3 facility															
		termination per month			BP3	1L5BP	1,175.15	335.40	219.24	89.57	87.75						
		INTERCONNECTION MID-SPAN MEET f Access service ride Mid-Span Meet, one-half the tariffed ser	vice	cal Ch	annel rate is annlis	able											
		Local Channel - Dedicated - DS1 per month	AICS LO	cai Uni	OH1MS	TEFHG	0.00	0.00									
		Local Channel - Dedicated - DS3 per month			OH3MS	TEFHJ	0.00	0.00									
		LEXERS	1	1		1				1		1		1	1	1	1

LOCAL IN	TERCONNECTION - Kentucky												Attachi	ment: 3	Exhi	bit: A
											Svc Order	Svc Order	Incremental	Incremental	Incremental	Incrementa
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
		Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES (\$)			per LSR	-	Order vs.	Order vs.		Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						Dee	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		1
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Channelization - DS1 to DS0 Channel System			OH1, OH1MS	SATN1	113.33	101.40	71.60	13.79	13.04						
	DS3 to DS1 Channel System per month			OH3, OH3MS	SATNS	158.20	199.23	118.62	50.16	48.59						
	DS3 Interface Unit (DS1 COCI) per month			OH1, OH1MS	SATCO	11.80	10.07	7.08								
Note	es: If no rate is identified in the contract, the rates, terms,	and condition	s for th	he specific service	or function wi	ill be as set forth	n in applicable	BellSouth tar	iff.							

LOCAL IN	NTERCONNECTION - South Carolina												Attach	nent: 3	Exhi	bit: A
CATEGORY	RY RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES (\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Order vs.
					-	Rec	Nonrec		Nonrecurring		0.01/150	001111		Rates(\$)	001111	
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	I TERCONNECTION (CALL TRANSPORT AND TERMINATION)															
	DTE: "bk" beside a rate indicates that the Parties have agreed to bil	I and k	eep for	that element pursu	ant to the ter	rms and conditi	ons in Attachn	nent 3.								
Not	te: End Office Switching Function per MOU rate element is not sho								between Knol	ogy and BellS	outh					
TAN	NDEM SWITCHING															
	Tandem Switching Function Per MOU Multiple Tandem Switching, per MOU (applies to intial tandem			OHD		0.0007360bk										<u> </u>
	only)			онр		0.000736										
	Tandem Intermediary Charge, per MOU*			OHD		0.0015										
	his charge is applicable only to transit traffic and is applied in add	dition to	o appli	cable switching and	l/or interconr	nection charges	•									
TRL	RUNK CHARGE			0.115	700			== (0)								
\vdash	Installation Trunk Side Service - per DS0 Dedicated End Office Trunk Port Service-per DS0**			OHD OHD	TPP++ TDE0P	0.00	335.14bk	57.16bk								├───
<u> </u>	Dedicated End Office Trunk Port Service-per DS0			0H1 OH1MS	TDE0P	0.00										
	Dedicated Tandem Trunk Port Service-per DS0**			OHD	TDW0P	0.00										<u> </u>
	Dedicated Tandem Trunk Port Service-per DS1**			OH1 OH1MS	TDW1P	0.00										
	This rate element is recovered on a per MOU basis and is included	in the	End Of	fice Switching and	Tandem Swit	tching, per MOL	J rate elements	5								L
CO	DMMON TRANSPORT (Shared) Common Transport - Per Mile, Per MOU			OHD		0.0000045bk										L
├ ── ├ ─	Common Transport - Per Mile, Per MOU Common Transport - Facilities Termination Per MOU			OHD	-	0.0004095bk										<u> </u>
LOCAL INT	TERCONNECTION (DEDICATED TRANSPORT)			OND		0.0004093DK										<u> </u>
	TEROFFICE CHANNEL - DEDICATED TRANSPORT															
	Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade -															
	Per Mile per month			OHL, OHM	1L5NF	0.0167bk										
	Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade -			OHL, OHM	1L5NF	24.3bk	40.63bk	27.47bk	16.77bk	6.91bk						
	Facility Termination per month Interoffice Channel - Dedicated Transport - 56 kbps - per mile			UHL, UHIVI	TLOINF	24.3DK	40.63DK	27.47DK	16.77DK	6.91DK						<u> </u>
	per month			OHL, OHM	1L5NK	0.0167bk										
	Interoffice Channel - Dedicated Transport - 56 kbps - Facility															
	Termination per month			OHL, OHM	1L5NK	16.76bk	40.63bk	27.47bk	16.77bk	6.91bk						
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile					0.040711										
	per month Interoffice Channel - Dedicated Transport - 64 kbps - Facility			OHL, OHM	1L5NK	0.0167bk										<u> </u>
	Termination per month			OHL, OHM	1L5NK	16.76bk	40.63bk	27.47bk	16.77bk	6.91bk						
	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per															
	month			OH1, OH1MS	1L5NL	0.3415bk										
	Interoffice Channel - Dedicated Tranport - DS1 - Facility				41.55.11	77.4411	00.4711	04.0011	40.0011	44.401.1						
	Termination per month Interoffice Channel - Dedicated Transport - DS3 - Per Mile per			OH1, OH1MS	1L5NL	77.14bk	89.47bk	81.99bk	16.39bk	14.48bk						
	month			OH3, OH3MS	1L5NM	8.02bk										
	Interoffice Channel - Dedicated Transport - DS3 - Facility				1201111	0.0LDIX										
	Termination per month			OH3, OH3MS	1L5NM	880.65bk	279.37bk	163.12bk	60.33bk	58.59bk						
LOC	DCAL CHANNEL - DEDICATED TRANSPORT				75510	15.0011										
	Local Channel - Dedicated - 2-Wire Voice Grade per month Local Channel - Dedicated - 4-Wire Voice Grade per month			OHL, OHM OHL, OHM	TEFV2	15.33bk 16.54bk	193.53bk 193.97bk	33.24bk 33.68bk	36.72bk	3.21bk 3.68bk						L
	Local Channel - Dedicated - 4-wire voice Grade per month			OHL, OHM OH1	TEFV4 TEFHG	42.62bk	193.970k 177.87bk	154.06k	37.19bk 22.24bk	3.680k						<u> </u>
	Escal Chaimer - Dedicated - Do't per month			om		42.0201	111.070K	104.00k	22.240K	10.004						
	Local Channel - Dedicated - DS3 Facility Termination per month			OH3	TEFHJ	446bk	452.52bk	264.53k	119.75bk	83.77bk						
LOC	CAL INTERCONNECTION - VIRTUAL INTERCONNECTION POINT															
	Local Channel-Dedicated- DS3 - per month			BP3	BP3LC	446.00	452.52	264.53	119.75	83.77						
	Interoffice Channel - Dedicated transport - DS3 per mile per			220												
\vdash	month			BP3	1L5BP	8.02										<u> </u>
	Interoffice Channel - Dedicated transport - DS3 facility termination per month			BP3	1L5BP	880.65	279.37	163.12	60.33	58.59						
LO	DCAL INTERCONNECTION MID-SPAN MEET			5.5	(LODF	000.00	213.31	103.12	00.33	30.39						<u> </u>
	DTE: If Access service ride Mid-Span Meet, one-half the tariffed service ride Mid-Span Meet, one-half the tariffed service ride service ride Mid-Span Meet, one-half the tariffed service ride service r	vice Lo	cal Ch	annel rate is applica	ble.	1										<u> </u>
	Local Channel - Dedicated - DS1 per month			OH1MS	TEFHG	0.00	0.00									
	Local Channel - Dedicated - DS3 per month			OH3MS	TEFHJ	0.00	0.00									
MU	ULTIPLEXERS		I													L

LOCAL INTE	RCONNECTION - South Carolina												Attach	ment: 3	Exhi	bit: A
											Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
		Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES (\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
											•	•	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						Bee	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		1
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Channelization - DS1 to DS0 Channel System			OH1, OH1MS	SATN1	107.57	91.24	62.71	10.56	9.81						
	DS3 to DS1 Channel System per month			OH3, OH3MS	SATNS	144.02	178.54	94.18	33.33	31.90						
	DS3 Interface Unit (DS1 COCI) per month			OH1, OH1MS	SATCO	8.64	6.59	4.73								
Notes:	If no rate is identified in the contract, the rates, terms, an	d condition	ns for th	ne specific service	or function w	ill be as set fort	h in applicable	e BellSouth tar	iff.							

LOCAL	INTE	RCONNECTION - Tennessee													ment: 3		bit: A
CATEGO	RY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES (\$)					Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
								Nonrecurring		Nonrecurring	Disconnect			OSS	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN		SOMAN	SOMAN	SOMAN
		CONNECTION (CALL TRANSPORT AND TERMINATION) bk" beside a rate indicates that the Parties have agreed to bi	llandk	ann far	that alamant nura	iont to the to	mo and conditi	ono in Attochn	nont 2								
		nd Office Switching Function per MOU rate element is not sh								between Knold	ogy and BellS	outh					ł
		M SWITCHING		1011 20					ine Agreement		gy and Done						1
		Tandem Switching Function Per MOU			OHD		0.0009778bk										
		Multiple Tandem Switching, per MOU (applies to intial tandem															
		only)			OHD OHD		0.0009778										
* .		Tandem Intermediary Charge, per MOU* harge is applicable only to transit traffic and is applied in add	dition to	o annli	÷··=	d/or intercon	0.0015										
		CHARGE			cable switching and	u/or intercom	lection charges	».									
		Installation Trunk Side Service - per DS0			OHD	TPP++		334.29bk	57.01bk								
		Dedicated End Office Trunk Port Service-per DS0**			OHD	TDE0P	0.00										
		Dedicated End Office Trunk Port Service-per DS1**			0H1 OH1MS	TDE1P	0.00										
		Dedicated Tandem Trunk Port Service-per DS0**			OHD	TDW0P TDW1P	0.00										
**		Dedicated Tandem Trunk Port Service-per DS1** rate element is recovered on a per MOU basis and is included	in the	End Of	OH1 OH1MS		0.00 tching, per MO	l rate elements				<u> </u>					<u> </u>
		IN TRANSPORT (Shared)			ince ownering and	Tandem Own	lenning, per mo	o rate elementa	,								
		Common Transport - Per Mile, Per MOU			OHD		0.0000064bk										
		Common Transport - Facilities Termination Per MOU			OHD		0.0003871bk										
		CONNECTION (DEDICATED TRANSPORT)															
IN		FFICE CHANNEL - DEDICATED TRANSPORT Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade -				-											-
		Per Mile per month			OHL, OHM	1L5NF	0.0174bk										
		Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade -				LOIN	0.01746K										
		Facility Termination per month			OHL, OHM	1L5NF	18.58bk	55.39bk	17.37bk	27.96bk	3.51bk						
		Interoffice Channel - Dedicated Transport - 56 kbps - per mile per month			OHL, OHM	1L5NK	0.0174bk										
		Interoffice Channel - Dedicated Transport - 56 kbps - Facility															
		Termination per month			OHL, OHM	1L5NK	17.98bk	55.39bk	17.37bk	27.96bk	3.51bk						
		Interoffice Channel - Dedicated Transport - 64 kbps - per mile					0.047411										
		per month Interoffice Channel - Dedicated Transport - 64 kbps - Facility			OHL, OHM	1L5NK	0.0174bk										
		Interoffice Channel - Dedicated Transport - 04 Kbps - Facility Termination per month Interoffice Channel - Dedicated Channel - DS1 - Per Mile per			OHL, OHM	1L5NK	17.98bk	55.39bk	17.37bk	27.96bk	3.51bk						
		month			OH1, OH1MS	1L5NL	0.3562bk										
		Interoffice Channel - Dedicated Tranport - DS1 - Facility		1													
		Termination per month		I	OH1, OH1MS	1L5NL	77.86bk	112.4bk	76.27bk	19.55bk	14.99bk						
		Interoffice Channel - Dedicated Transport - DS3 - Per Mile per		1			0.0411										
\vdash		month Interoffice Channel - Dedicated Transport - DS3 - Facility			OH3, OH3MS	1L5NM	2.34bk										
		Termination per month		1	OH3, OH3MS	1L5NM	848.99bk	395.29bk	176.56bk	109.04bk	105.91bk						
L		CHANNEL - DEDICATED TRANSPORT															
		Local Channel - Dedicated - 2-Wire Voice Grade per month			OHL, OHM	TEFV2	19.43bk	199.33bk	24.16bk	54.81bk	4.8bk						
		Local Channel - Dedicated - 4-Wire Voice Grade per month			OHL, OHM	TEFV4	20.56bk	201.53bk	24.83bk	55.52bk	5.51bk						
		Local Channel - Dedicated - DS1 per month			OH1	TEFHG	40.99bk	277.35bk	233.26bk	33.18bk	22.3bk						
		Local Channel - Dedicated - DS3 Facility Termination per month		1	ОНЗ	TEFHJ	611.3bk	595.37bk	304.5bk	215.82bk	151.15bk						
		INTERCONNECTION - VIRTUAL INTERCONNECTION POINT		1	0.10	121110	011.3DK	555.57 DK	504.5DK	210.020K	101.1000	+					
	JUAL	Local Channel-Dedicated- DS3 - per month		1	BP3	BP3LC	611.30	595.37	304.50	215.82	151.15	<u> </u>					
\vdash		Interoffice Channel - Dedicated transport - DS3 per mile per		1	5.5	01 320	011.30	393.37	304.30	210.02	101.10						
		month		1	BP3	1L5BP	2.34										
		Interoffice Channel - Dedicated transport - DS3 facility										l l	l				İ
		termination per month		<u> </u>	BP3	1L5BP	848.99	395.29	176.56	109.04	105.91						
		INTERCONNECTION MID-SPAN MEET	L		<u></u>	1						ļ					ļ
N		f Access service ride Mid-Span Meet, one-half the tariffed ser	rvice Lo	cal Ch			0.00	0.00				L					L
		Local Channel - Dedicated - DS1 per month Local Channel - Dedicated - DS3 per month			OH1MS OH3MS	TEFHG TEFHJ	0.00	0.00									
		Local Channel - Dedicated - DSS per month	<u> </u>	ł	CT IOWO		0.00	0.00				+				<u> </u>	

LOCAL IN	FERCONNECTION - Tennessee												Attachi	ment: 3	Exhil	bit: A
											Svc Order	Svc Order	Incremental	Incremental	Incremental	Incrementa
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
		Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svo
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES (\$)			per LSR	-	Order vs.	Order vs.		Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						D	Nonrecurring		Nonrecurring	Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Channelization - DS1 to DS0 Channel System			OH1, OH1MS	SATN1	80.77	141.87	77.11	44.47	42.62						Í
	DS3 to DS1 Channel System per month			OH3, OH3MS	SATNS	222.98	308.03	108.47	6.34	4.23						Í
	DS3 Interface Unit (DS1 COCI) per month			OH1, OH1MS	SATCO	17.58	6.07	4.66								í
Note	s: If no rate is identified in the contract, the rates, terms,	and condition	s for th	he specific service	or function wi	ill be as set for	th in applicable	e BellSouth ta	riff.							í

Attachment 4

Physical Collocation

BELLSOUTH

PHYSICAL COLLOCATION

1. <u>Scope of Attachment</u>

- 1.1 The rates, terms, and conditions contained within this Attachment shall only apply when Knology is physically collocated as a sole occupant or as a Host within a Premises location pursuant to this Attachment. BellSouth Premises include BellSouth Central Offices and Serving Wire Centers (hereinafter "Premises"). This Attachment is applicable to Premises owned or leased by BellSouth. However, if the Premises occupied by BellSouth is leased by BellSouth from a third party, special considerations and intervals may apply in addition to the terms and conditions of this Attachment.
- 1.2 <u>Right to Occupy</u>. BellSouth shall offer to Knology collocation on rates, terms, and conditions that are just, reasonable, non-discriminatory and consistent with the rules of the Federal Communications Commission ("FCC"). Subject to the rates, terms and conditions of this Attachment where space is available and it is technically feasible, BellSouth will allow Knology to occupy that certain area designated by BellSouth within a BellSouth Premises, or on BellSouth property upon which the BellSouth Premises is located, of a size which is specified by Knology and agreed to by BellSouth (hereinafter "Collocation Space"). The necessary rates, terms and conditions for BellSouth locations other than BellSouth Premises shall be negotiated upon request for collocation at such location(s).
- 1.2.1 Neither BellSouth nor any of BellSouth's affiliates may reserve space for future use on more preferential terms than those set forth below.
- 1.2.1.1 In all states other than Florida, the size specified by Knology may contemplate a request for space sufficient to accommodate Knology's growth within a two-year period.
- 1.2.1.2 In the state of Florida, the size specified by Knology may contemplate a request for space sufficient to accommodate Knology's growth within an eighteen (18) month period.
- 1.3 <u>Space Allocation</u>. BellSouth shall attempt to accommodate Knology's requested preferences if any. In allocating Collocation Space, BellSouth shall not materially increase Knology's cost or materially delay Knology's occupation and use of the Collocation Space, shall not assign Collocation Space that will impair the quality of service or otherwise limit the service Knology wishes to offer, and shall not reduce unreasonably the total space available for physical collocation or preclude unreasonably physical collocation within the Premises. Space shall not be available for collocation if it is: (a) physically occupied by non-obsolete equipment; (b) assigned to another collocator; (c) used to provide physical access to occupied space; (e) properly

reserved for future use, either by BellSouth or by another carrier; or (f) essential for the administration and proper functioning of BellSouth's Premises. BellSouth may segregate Collocation Space and require separate entrances in accordance with FCC rules.

- 1.4 <u>Space Reclamation.</u> In the event of space exhaust within a Central Office Premises, BellSouth may include in its documentation for the Petition for Waiver filing any unutilized space in the Central Office Premises. Knology will be responsible for any justification of unutilized space within its space, if the Commission requires such justification.
- 1.5 <u>Use of Space</u>. Knology shall use the Collocation Space for the purposes of installing, maintaining and operating Knology's equipment (to include testing and monitoring equipment) necessary for interconnection with BellSouth services and facilities or for accessing BellSouth unbundled network elements for the provision of telecommunications services, as specifically set forth in this Attachment. The Collocation Space may be used for no other purposes except as specifically described herein or in any amendment hereto.
- 1.6 <u>Rates and Charges</u>. Knology agrees to pay the rates and charges identified in Exhibit B attached hereto.
- 1.7 If any due date contained in this Attachment falls on a weekend or National holiday, then the due date will be the next business day thereafter. For intervals of ten (10) calendar days or less National holidays will be excluded.
- 1.8 The parties agree to comply with all applicable federal, state, county, local and administrative laws, rules, ordinances, regulations and codes in the performance of their obligations hereunder.

2. <u>Space Availability Report</u>

- 2.1 <u>Space Availability Report</u>. Upon request from Knology, BellSouth will provide a written report ("Space Availability Report") describing in detail the space that is available for collocation and specifying the amount of Collocation Space available at the Premises requested, the number of collocators present at the Premises, any modifications in the use of the space since the last report on the Premises requested and the measures BellSouth is taking to make additional space available for collocation arrangements. A Space Availability Report does not reserve space at the Premises.
- 2.1.1 The request from Knology for a Space Availability Report must be written and must include the Premises street address, as identified in the Local Exchange Routing Guide ("LERG"), and Common Language Location Identification ("CLLI") code of the Premises. CLLI code information is located in the National Exchange Carriers Association ("NECA") Tariff FCC No. 4.

2.1.2 BellSouth will respond to a request for a Space Availability Report for a particular Premises within ten (10) calendar days of receipt of such request. BellSouth will make best efforts to respond in ten (10) calendar days to such a request when the request includes from two (2) to five (5) Premises within the same state. The response time for requests of more than five (5) Premises shall be negotiated between the Parties. If BellSouth cannot meet the ten (10) calendar day response time, BellSouth shall notify Knology and inform Knology of the time frame under which it can respond.

3. Collocation Options

- 3.1 <u>Cageless</u>. BellSouth shall allow Knology to collocate Knology's equipment and facilities without requiring the construction of a cage or similar structure. BellSouth shall allow Knology to have direct access to Knology's equipment and facilities. BellSouth shall make cageless collocation available in single bay increments. Except where Knology's equipment requires special technical considerations (e.g., special cable racking, isolated ground plane, etc.), BellSouth shall assign cageless Collocation Space in conventional equipment rack lineups where feasible. For equipment requiring special technical considerations, Knology must provide the equipment layout, including spatial dimensions for such equipment pursuant to generic requirements contained in Telcordia GR-63-Core, and shall be responsible for compliance with all special technical requirements associated with such equipment.
- 3.2 Caged. At Knology's expense, Knology may arrange with a Supplier certified by BellSouth ("Certified Supplier") to construct a collocation arrangement enclosure in accordance with BellSouth's guidelines and specifications prior to starting equipment installation. BellSouth will provide guidelines and specifications upon request. Where local building codes require enclosure specifications more stringent than BellSouth's standard enclosure specification, Knology and Knology's Certified Supplier must comply with the more stringent local building code requirements. Knology's Certified Supplier shall be responsible for filing and receiving any and all necessary permits and/or licenses for such construction. BellSouth shall cooperate with Knology and provide, at Knology's expense, the documentation, including existing building architectural drawings, enclosure drawings, and specifications required and necessary for Knology to obtain the zoning, permits and/or other licenses. Knology's Certified Supplier shall bill Knology directly for all work performed for Knology pursuant to this Attachment and BellSouth shall have no liability for nor responsibility to pay such charges imposed by Knology's Certified Supplier. Knology must provide the local BellSouth building contact with two Access Keys used to enter the locked enclosure. Except in case of emergency, BellSouth will not access Knology's locked enclosure prior to notifying Knology. Upon request, BellSouth shall construct the enclosure for Knology.
- 3.2.1 BellSouth may elect to review Knology's plans and specifications prior to allowing construction to start to ensure compliance with BellSouth's guidelines and specifications. Notification to Knology indicating BellSouth's desire to execute this

review will be provided in BellSouth's response to the Initial Application, if Knology has indicated its desire to construct its own enclosure. If Knology's Initial Application does not indicate its desire to construct its own enclosure, but its subsequent firm order does indicate its desire to construct its own enclosure, then notification to review will be given within ten (10) calendar days after the Firm Order date. BellSouth shall complete its review within fifteen (15) calendar days after the receipt of the plans and specifications. Regardless of whether or not BellSouth elects to review Knology's plans and specifications, BellSouth reserves the right to inspect the enclosure after construction to make sure it is constructed according to the submitted plans and specifications and/or BellSouth's guidelines and specifications, as applicable. If BellSouth decides to inspect, BellSouth will complete its inspection within fifteen (15) calendar days after receipt of written notification of completion of the enclosure from Knology. BellSouth shall require Knology to remove or correct within seven (7) calendar days at Knology's expense any structure that does not meet these plans and specifications or, where applicable, BellSouth guidelines and specifications.

- 3.3 <u>Shared Caged Collocation</u>. Knology may allow other telecommunications carriers to share Knology's caged collocation arrangement pursuant to terms and conditions agreed to by Knology ("Host") and other telecommunications carriers ("Guests") and pursuant to this Section, except where the BellSouth Premises is located within a leased space and BellSouth is prohibited by said lease from offering such an option. Knology shall notify BellSouth in writing upon execution of any agreement between the Host and its Guest within ten (10) calendar days of its execution and prior to any Firm Order. Further, such notice shall include the name of the Guest(s) and the term of the agreement, and shall contain a certification by Knology that said agreement imposes upon the Guest(s) the same terms and conditions for Collocation Space as set forth in this Attachment between BellSouth and Knology.
- 3.3.1 Knology, as the Host, shall be the sole interface and responsible Party to BellSouth for the assessment and billing of rates and charges contained within this Attachment and for the purposes of ensuring that the safety and security requirements of this Attachment are fully complied with by the Guest(s), its employees and agents. BellSouth shall provide Knology with a proration of the costs of the Collocation Space based on the number of collocators and the space used by each with a minimum charge of one (1) bay/rack per Host/Guest. In all states other than Florida, and in addition to the foregoing, Knology shall be the responsible party to BellSouth for the purpose of submitting applications for initial and additional equipment placement of the Guest. In Florida the Guest may directly submit initial and additional equipment placement applications using the Host's access carrier name abbreviation (ACNA). A separate Guest application shall require the assessment of an Initial or Subsequent Application Fee, as set forth in Exhibit B, which will be billed to the Host on the date that BellSouth provides its written response ("Application Response").
- 3.3.2 Notwithstanding the foregoing, the Guest may arrange directly with BellSouth for the provision of the interconnecting facilities between BellSouth and the Guest and for the provision of the services and access to unbundled network elements. The bill for these

interconnecting facilities, services and access to UNEs will be charged to the Guest pursuant to the applicable tariff or the Guest's Interconnection Agreement with BellSouth.

- 3.3.3 Knology shall indemnify and hold harmless BellSouth from any and all claims, actions, causes of action, of whatever kind or nature arising out of the presence of Knology's Guests in the Collocation Space except to the extent caused by BellSouth's sole negligence, gross negligence, or willful misconduct.
- 3.4 <u>Adjacent Collocation</u>. Subject to technical feasibility and space availability, BellSouth will permit adjacent collocation arrangements ("Adjacent Arrangement") on the Premises' property, where the Adjacent Arrangement does not interfere with access to existing or planned structures or facilities on the Premises property. The Adjacent Arrangement shall be constructed or procured by Knology and in conformance with BellSouth's design and construction specifications. Further, Knology shall construct, procure, maintain and operate said Adjacent Arrangement(s) pursuant to all of the rates, terms and conditions set forth in this Attachment.
- 3.4.1 Should Knology elect Adjacent Collocation, Knology must arrange with a Certified Supplier to construct an Adjacent Arrangement structure in accordance with BellSouth's guidelines and specifications. BellSouth will provide guidelines and specifications upon request. Where local building codes require enclosure specifications more stringent than BellSouth's standard specification, Knology and Knology's Certified Supplier must comply with the more stringent local building code requirements. Knology's Certified Supplier shall be responsible for filing and receiving any and all necessary zoning, permits and/or licenses for such construction. Knology's Certified Supplier shall bill Knology directly for all work performed for Knology pursuant to this Attachment and BellSouth shall have no liability for nor responsibility to pay such charges imposed by Knology's Certified Supplier. Knology must provide the local BellSouth building contact with two cards, keys or other access device used to enter the locked enclosure. Except in cases of emergency, BellSouth shall not access Knology's locked enclosure prior to notifying Knology.
- 3.4.2 Knology must submit its plans and specifications to BellSouth with its Firm Order. BellSouth shall review Knology's plans and specifications prior to construction of an Adjacent Arrangement(s) to ensure compliance with BellSouth's guidelines and specifications. BellSouth shall complete its review within fifteen (15) calendar days after receipt of plans and specifications. BellSouth may inspect the Adjacent Arrangement during and after construction to confirm it is constructed according to the submitted plans and specifications. If BellSouth decides to inspect, BellSouth will complete its inspection within fifteen (15) calendar days after receipt of written notification of completion of the enclosure from Knology. BellSouth shall require Knology to remove or correct within seven (7) calendar days at Knology's expense any structure that does not meet these plans and specifications or, where applicable, BellSouth's guidelines and specifications.

- 3.4.3 Knology shall provide a concrete pad, the structure housing the arrangement, heating/ventilation/air conditioning ("HVAC"), lighting, and all facilities that connect the structure (i.e. racking, conduits, etc.) to the BellSouth point of demarcation. At Knology's option, and where the local authority having jurisdiction permits, BellSouth shall provide an AC power source and access to physical collocation services and facilities subject to the same nondiscriminatory requirements as applicable to any other physical collocation arrangement. In Alabama and Louisiana, BellSouth will provide DC power to Adjacent Collocation sites where technically feasible, as that term has been defined by the FCC, and subject to individual case basis pricing. Knology's Certified Supplier shall be responsible, at Knology's expense, for filing and receiving any and all necessary zoning, permits and/or licenses for such arrangement. BellSouth shall allow Shared Caged Collocation within an Adjacent Arrangement pursuant to the terms and conditions set forth herein.
- 3.5 <u>Co-Carrier Cross Connect (CCXC)</u>. The primary purpose of collocation is for a collocated telecommunications carrier to interconnect with BellSouth's network or to access BellSouth's unbundled network elements for the provision of telecommunications services within a BellSouth Premises. BellSouth will permit Knology to interconnect between its virtual or physical collocation arrangements and those of another collocated telecommunications carrier within the same central office. Both Knology's agreement and the other collocated telecommunications carrier's agreement must contain rates, terms and conditions for CCXC language. At no point in time shall Knology use the Collocated telecommunications carriers.
- 3.5.1 Knology must use a BellSouth Certified Supplier to place the CCXC. The CCXC shall be provisioned through facilities owned by Knology. Such connections to other carriers may be made using either optical or electrical facilities. In cases where Knology's equipment and the equipment of the other interconnector are located in contiguous caged Collocation Spaces, Knology will have the option of using Knology's own technicians to deploy co-carrier cross connects using either electrical or optical facilities between the sets of equipment and construct its own dedicated cable support structure. Knology may deploy such optical or electrical connections directly between its own facilities and the facilities of other collocated telecommunications carriers without being routed through BellSouth equipment. Knology may not self-provision CCXC on any BellSouth distribution frame, POT (Point of Termination) Bay, DSX (Digital System Cross-connect) or LGX (Light Guide Cross-connect). Knology is responsible for ensuring the integrity of the signal.
- 3.5.2 Knology shall be responsible for providing written authorization to BellSouth from the other collocated telecommunications carrier prior to installing the CCXC. Knology-provisioned CCXC shall utilize common cable support structure. There will be a recurring charge per linear foot, per cable, of common cable support structure used. In the case of two contiguous caged collocation arrangements, Knology will have the option of using Knology's own technicians to construct its own dedicated support structure.

3.5.3 To order CCXCs Knology must submit an Initial Application or Subsequent Application. If no modification to the Collocation Space is requested other than the placement of CCXCs, the Subsequent Application Fee for CCXC, as defined in Exhibit B, will apply. If modifications in addition to the placement of CCXCs are requested, the Initial Application or Subsequent Application Fee will apply. This nonrecurring fee will be billed by BellSouth on the date that BellSouth provides an Application Response.

4. <u>Occupancy</u>

- 4.1 Occupancy. BellSouth will notify Knology in writing that the Collocation Space is ready for occupancy ("Space Ready Date"). Knology will schedule and complete an acceptance walk-through of each Collocation Space with BellSouth within fifteen (15) calendar days of BellSouth's notifying Knology that the Collocation Space is ready for occupancy. BellSouth will correct any deviations to Knology's original or jointly amended requirements within seven (7) calendar days after the walk-through, unless the Parties jointly agree upon a different time frame, and BellSouth shall establish a new Space Ready Date. Another acceptance walk-through will then be scheduled and conducted within fifteen (15) calendar days of the new Space Ready Date. This follow-up acceptance walk-through will be limited to those items identified in the initial walk-through. If Knology has met the fifteen (15) calendar day interval(s), billing will begin upon the date of Knology's acceptance of the Collocation Space ("Space Acceptance Date"). In the event that Knology fails to complete an acceptance walk-through within this fifteen (15) calendar day interval, the Collocation Space shall be deemed accepted by Knology. Billing will commence on the Space Ready Date or on the Space Acceptance Date, whichever is sooner. Knology must notify BellSouth in writing that collocation equipment installation is complete and is operational with BellSouth's network. BellSouth may, at its option, not accept orders for cross connects until receipt of such notice. For purposes of this paragraph, Knology's telecommunications equipment will be deemed operational when cross-connected to BellSouth's network for the purpose of service provisioning.
- 4.2 <u>Termination of Occupancy</u>. In addition to any other provisions addressing termination of occupancy in this Agreement, Knology may terminate occupancy in a particular Collocation Space by submitting a Subsequent Application requesting termination of occupancy. A Subsequent Application Fee will not apply for termination of occupancy. BellSouth may terminate Knology's right to occupy the Collocation Space in the event Knology fails to comply with any provision of this Agreement including the payment of applicable fees.

Upon termination of occupancy, Knology at its expense shall remove its equipment and other property from the Collocation Space. Knology shall have thirty (30) calendar days from the termination date to complete such removal, including the removal of all equipment and facilities of Knology's Guests, unless Knology's Guest has assumed responsibility for the Collocation Space housing the Guest's equipment

and executed the documentation required by BellSouth prior to such removal date. Knology shall continue payment of monthly fees to BellSouth until such date as Knology, and if applicable Knology's Guest, has fully vacated the Collocation Space and the Space Relinquish Form has been accepted by BellSouth. Should Knology or Knology's Guest fail to vacate the Collocation Space within thirty (30) calendar days from the termination date, BellSouth shall have the right to remove the equipment and dispose of the equipment and other property of Knology or Knology's Guest(s), in any manner that BellSouth deems fit, at Knology's expense and with no liability whatsoever for Knology's property or Knology's Guest(s)'s property. Upon termination of Knology's right to occupy Collocation Space, the Collocation Space will revert back to BellSouth, and Knology shall surrender such Collocation Space to BellSouth in the same condition as when first occupied by Knology except for ordinary wear and tear, unless otherwise agreed to by the Parties. Knology's BellSouth Certified Supplier shall be responsible for updating and making any necessary changes to BellSouth's records as required by BellSouth's guidelines and specifications including but not limited to Central Office Record Drawings and ERMA Records. Knology shall be responsible for the cost of removing any Knology constructed enclosure, together with all support structures (e.g., racking, conduits, power cables, etc.), at the termination of occupancy and restoring the grounds to their original condition.

5. <u>Use of Collocation Space</u>

- 5.1 <u>Equipment Type</u>. BellSouth permits the collocation of any type of equipment necessary for interconnection to BellSouth's network or for access to BellSouth's unbundled network elements in the provision of telecommunications services, as the term "necessary" is defined by FCC 47 C.F.R. Section 51.323 (b). The primary purpose and function of any equipment collocated in a Premises must be for interconnection to BellSouth's network or for access to BellSouth's unbundled network elements in the provision of telecommunications services.
- 5.1.1 Examples of equipment that would not be considered necessary include but are not limited to: Traditional circuit switching equipment, equipment used exclusively for call-related databases, computer servers used exclusively for providing information services, operations support system (OSS) equipment used to support collocated telecommunications carrier network operations, equipment that generates customer orders, manages trouble tickets or inventory, or stores customer records in centralized databases, etc. BellSouth will determine upon receipt of an application if the requested equipment is necessary based on the criteria established by the FCC. Multifunctional equipment placed on BellSouth's Premises must not place any greater relative burden on BellSouth's property than comparable single-function equipment. BellSouth reserves the right to permit collocation of any equipment on a nondiscriminatory basis.
- 5.1.2 Such equipment must, at a minimum, meet the following Telcordia Network Equipment Building Systems (NEBS) General Equipment Requirements: Criteria

Level 1 requirements as outlined in the Telcordia Special Report SR-3580, Issue 1; equipment design spatial requirements per GR-63-CORE, Section 2; thermal heat dissipation per GR-063-CORE, Section 4, Criteria 77-79; acoustic noise per GR-063-CORE, Section 4, Criterion 128, and National Electric Code standards. Except where otherwise required by a Commission, BellSouth shall comply with the applicable FCC rules relating to denial of collocation based on Knology's failure to comply with this Section.

- 5.1.3 Knology shall not request more DS0, DS1, DS3 and optical terminations for a collocation arrangement than the total port or termination capacity of the equipment physically installed in the arrangement. The total capacity of the equipment collocated in the arrangement will include equipment contained in the application in question as well as equipment already placed in the arrangement. If full network termination capacity of the equipment being installed is not requested in the application, additional network terminations for the installed equipment will require the submission of another application. In the event that Knology submits an application for terminations that exceed the total capacity of the collocated equipment, Knology will be informed of the discrepancy and will be required to submit a revision to the application.
- 5.2 Knology shall identify to BellSouth whenever Knology submits a Method of Procedure ("MOP") adding equipment to Knology's Collocation Space all UCC-1 lien holders or other entities that have a financial interest, secured and otherwise, in the equipment in Knology's Collocation Space.
- 5.3 Knology shall not use the Collocation Space for marketing purposes nor shall it place any identifying signs or markings outside the Collocation Space or on the grounds of the Premises.
- 5.4 Knology shall place a plaque or other identification affixed to Knology's equipment necessary to identify Knology's equipment, including a list of emergency contacts with telephone numbers.
- 5.5 Entrance Facilities. Knology may elect to place Knology-owned or Knology-leased fiber entrance facilities into the Collocation Space. BellSouth will designate the point of interconnection in close proximity to the Premises building housing the Collocation Space, such as an entrance manhole or a cable vault, which are physically accessible by both Parties. Knology will provide and place fiber cable at the point of entrance of sufficient length to be pulled through conduit and into the splice location. Knology will provide and install a sufficient length of fire retardant riser cable, to which the entrance cable will be spliced by BellSouth, which will extend from the splice location to Knology's equipment in the Collocation Space. In the event Knology utilizes a non-metallic, riser-type entrance facility, a splice will not be required. Knology must contact BellSouth for instructions prior to placing the entrance facilities. At Knology's option BellSouth will accommodate where technically feasible a microwave entrance facility pursuant to separately negotiated terms and conditions. In the case of

adjacent collocation, unless BellSouth determines that limited space is available for the entrance facilities, copper facilities may be used between the adjacent collocation arrangement and the central office demarcation point.

- 5.5.1 <u>Dual Entrance</u>. BellSouth will provide at least two interconnection points at each Premises where there are at least two such interconnection points available and where capacity exists. Upon receipt of a request for physical collocation under this Attachment, BellSouth shall provide Knology with information regarding BellSouth's capacity to accommodate dual entrance facilities. If conduit in the serving manhole(s) is available and is not reserved for another purpose for utilization within 12 months of the receipt of an application for collocation, BellSouth will make the requested conduit space available for installing a second entrance facility to Knology's arrangement. The location of the serving manhole(s) will be determined at the sole discretion of BellSouth. Where dual entrance is not available due to lack of capacity, BellSouth will so state in the Application Response.
- 5.5.2 <u>Shared Use</u>. Knology may utilize spare capacity on an existing interconnector entrance facility for the purpose of providing an entrance facility to Knology's collocation arrangement within the same BellSouth Premises. BellSouth shall allow the splice, provided that the fiber is non-working fiber. Knology must arrange with BellSouth for BellSouth to splice the Knology provided riser cable to the spare capacity on the entrance facility. The rates set forth in Exhibit B will apply. If Knology desires to allow another telecommunications carrier to use its entrance facilities, additional rates, terms and conditions will apply and shall be negotiated between the Parties.
- 5.6 <u>Demarcation Point</u>. BellSouth will designate the point(s) of demarcation between Knology's equipment and/or network and BellSouth's network. Each Party will be responsible for maintenance and operation of all equipment/facilities on its side of the demarcation point. For 2-wire and 4-wire connections to BellSouth's network, the demarcation point shall be a common block on the BellSouth designated conventional distributing frame (CDF). Knology shall be responsible for providing, and a supplier certified by BellSouth ("BellSouth Certified Supplier") shall be responsible for installing and properly labeling/stenciling the common block and necessary cabling pursuant to Section 7. For all other terminations BellSouth shall designate a demarcation point on a per arrangement basis. Knology or its agent must perform all required maintenance to equipment/facilities on its side of the demarcation point, pursuant to Section 5.7, following, and may self-provision cross-connects that may be required within the Collocation Space to activate service requests.
- 5.6.1 In Tennessee, BellSouth will designate the point(s) of demarcation between Knology's equipment and/or network and BellSouth's network. Each Party will be responsible for maintenance and operation of all equipment/facilities on its side of the demarcation point. For connections to BellSouth's network, the demarcation point shall be a Knology provided Point of Termination Bay (POT Bay) in a common area within the Premises. Knology shall be responsible for providing, and a supplier certified by

BellSouth shall be responsible for installing and properly labeling/stenciling the POT Bay as well as installing the necessary cabling between Knology's Collocation Space and the demarcation point. Knology or its agent must perform all required maintenance to equipment/facilities on its side of the demarcation point, pursuant to Section 5.7, following, and may self-provision cross-connects that may be required within the Collocation Space to activate service requests. BellSouth will negotiate alternative rates, terms and conditions related to the demarcation point in Tennessee in the event that Knology desires to avoid the use of an intermediary device as contemplated by the Tennessee Regulatory Authority.

- 5.7 <u>Knology's Equipment and Facilities</u>. Knology, or if required by this Attachment, Knology's BellSouth Certified Supplier, is solely responsible for the design, engineering, installation, testing, provisioning, performance, monitoring, maintenance and repair of the equipment and facilities used by Knology which must be performed in compliance with all applicable BellSouth policies and guidelines. Such equipment and facilities may include but are not limited to cable(s), equipment, and point of termination connections. Knology and its selected BellSouth Certified Supplier must follow and comply with all BellSouth requirements outlined in BellSouth's TR 73503, TR 73519, TR 73572, and TR 73564.
- 5.8 <u>BellSouth's Access to Collocation Space</u>. From time to time BellSouth may require access to the Collocation Space. BellSouth retains the right to access such space for the purpose of making BellSouth equipment and building modifications (e.g., running, altering or removing racking, ducts, electrical wiring, HVAC, and cables). BellSouth will give notice to Knology at least forty-eight (48) hours before access to the Collocation Space is required. Knology may elect to be present whenever BellSouth performs work in the Collocation Space. The Parties agree that Knology will not bear any of the expense associated with this work.
- 5.9 <u>Access</u>. Pursuant to Section 12, Knology shall have access to the Collocation Space twenty-four (24) hours a day, seven (7) days a week. Knology agrees to provide the name and social security number or date of birth or driver's license number of each employee, supplier, or agent of Knology or Knology's Guests provided with access keys or devices ("Access Keys") prior to the issuance of said Access Keys. Key acknowledgement forms must be signed by Knology and returned to BellSouth Access Management within fifteen (15) calendar days of Knology's receipt. Failure to return properly acknowledged forms will result in the holding of subsequent requests until acknowledgements are current. Access Keys shall not be duplicated under any circumstances. Knology agrees to be responsible for all Access Keys and for the return of all said Access Keys in the possession of Knology's employees, suppliers, Guests, or agents after termination of the employment relationship, contractual obligation with Knology or upon the termination of this Attachment or the termination of occupancy of an individual collocation arrangement.
- 5.9.1 BellSouth will permit one accompanied site visit to Knology's designated collocation arrangement location after receipt of the Bona Fide Firm Order (BFFO) without

charge to Knology. Knology must submit to BellSouth the completed Access Control Request Form for all employees or agents requiring access to the BellSouth Premises a minimum of thirty (30) calendar days prior to the date Knology desires access to the Collocation Space. In order to permit reasonable access during construction of the Collocation Space, Knology may submit such a request at any time subsequent to BellSouth's receipt of the BFFO. In the event Knology desires access to the Collocation Space after submitting such a request but prior to access being approved, in addition to the first accompanied free visit, BellSouth shall permit Knology to access the Collocation Space accompanied by a security escort at Knology's expense. Knology must request escorted access at least three (3) business days prior to the date such access is desired.

- 5.10 <u>Lost or Stolen Access Keys</u>. Knology shall notify BellSouth in writing immediately in the case of lost or stolen Access Keys. Should it become necessary for BellSouth to re-key buildings or deactivate a card as a result of a lost Access Key(s) or for failure to return an Access Key(s), Knology shall pay for all reasonable costs associated with the re-keying or deactivating the card.
- 5.11 Interference or Impairment. Notwithstanding any other provisions of this Attachment, Knology shall not use any product or service provided under this Agreement, any other service related thereto or used in combination therewith, or place or use any equipment or facilities in any manner that 1) significantly degrades, interferes with or impairs service provided by BellSouth or by any other entity or any person's use of its telecommunications service; 2) endangers or damages the equipment, facilities or other property of BellSouth or of any other entity or person; 3) compromises the privacy of any communications; or 4) creates an unreasonable risk of injury or death to any individual or to the public. If BellSouth reasonably determines that any equipment or facilities of Knology violates the provisions of this paragraph, BellSouth shall give written notice to Knology, which notice shall direct Knology to cure the violation within forty-eight (48) hours of Knology's actual receipt of written notice or, at a minimum, to commence curative measures within twenty-four (24) hours and to exercise reasonable diligence to complete such measures as soon as possible thereafter. After receipt of the notice, the Parties agree to consult immediately and, if necessary, to inspect the arrangement.
- 5.11.1 Except in the case of the deployment of an advanced service which significantly degrades the performance of other advanced services or traditional voice band services, if Knology fails to take curative action within forty-eight (48) hours or if the violation is of a character which poses an immediate and substantial threat of damage to property, injury or death to any person, or any other significant degradation, interference or impairment of BellSouth's or another entity's service, then and only in that event BellSouth may take such action as it deems appropriate to correct the violation, including without limitation the interruption of electrical power to Knology's equipment. BellSouth will endeavor, but is not required, to provide notice to Knology prior to taking such action and shall have no liability to Knology for any damages

arising from such action, except to the extent that such action by BellSouth constitutes willful misconduct.

- For purposes of this Section, the term significantly degrade shall mean an action that 5.11.2 noticeably impairs a service from a user's perspective. In the case of the deployment of an advanced service which significantly degrades the performance of other advanced services or traditional voice band services and Knology fails to take curative action within forty-eight (48) hours then BellSouth will establish before the Commission that the technology deployment is causing the significant degradation. Any claims of network harm presented to Knology or, if subsequently necessary, the Commission must be supported with specific and verifiable information. Where BellSouth demonstrates that a deployed technology is significantly degrading the performance of other advanced services or traditional voice band services, Knology shall discontinue deployment of that technology and migrate its customers to technologies that will not significantly degrade the performance of other such services. Where the only degraded service itself is a known disturber, and the newly deployed technology satisfies at least one of the criteria for a presumption that is acceptable for deployment under Section 47 C.F.R. 51.230, the degraded service shall not prevail against the newly deployed technology.
- 5.12 <u>Personalty and its Removal</u>. Facilities and equipment placed by Knology in the Collocation Space shall not become a part of the Collocation Space, even if nailed, screwed or otherwise fastened to the Collocation Space, but shall retain their status as personal property and may be removed by Knology at any time. Any damage caused to the Collocation Space by Knology's employees, agents or representatives during the removal of such property shall be promptly repaired by Knology at its expense.
- 5.12.1 <u>If Knology decides to remove equipment from its Collocation Space and the removal</u> requires no physical changes, BellSouth will bill Knology an Administrative Only Application Fee as set forth in Exhibit B for these changes. This non-recurring fee will be billed on the date that BellSouth provides an Application Response.
- 5.13 <u>Alterations</u>. In no case shall Knology or any person acting on behalf of Knology make any rearrangement, modification, improvement, addition, or other alteration which could affect in any way space, power, HVAC, and/or safety considerations to the Collocation Space or the BellSouth Premises without the written consent of BellSouth, which consent shall not be unreasonably withheld. The cost of any such specialized alterations shall be paid by Knology. Any such material rearrangement, modification, improvement, addition, or other alteration shall require a Subsequent Application and Subsequent Application Fee, which will be billed by BellSouth on the date that BellSouth makes an Application Response.
- 5.14 Janitorial Service. Knology shall be responsible for the general upkeep of the Collocation Space. Knology shall arrange directly with a BellSouth Certified Supplier for janitorial services applicable to Caged Collocation Space. BellSouth shall provide a list of such suppliers on a site-specific basis upon request.

6. Ordering and Preparation of Collocation Space

- 6.1 Should any state or federal regulatory agency impose procedures or intervals applicable to Knology and BellSouth that are different from procedures or intervals set forth in this Section, whether now in effect or that become effective after execution of this Agreement, those procedures or intervals shall supersede the requirements set forth herein for that jurisdiction for all applications submitted for the first time after the effective date thereof.
- 6.2 <u>Initial Application</u>. For Knology or Knology's Guest(s) initial equipment placement, Knology shall submit to BellSouth a Physical Expanded Interconnection Application Document ("Initial Application"). The Initial Application is Bona Fide when it is complete and accurate, meaning that all required fields on the application are completed with the appropriate type of information. An application fee will apply which will be billed by BellSouth on the date that BellSouth makes an Application Response.
- 6.3 <u>Subsequent Application.</u> In the event Knology or Knology's Guest(s) desires to modify the use of the Collocation Space after a BFFO, Knology shall complete an application detailing all information regarding the modification to the Collocation Space ("Subsequent Application"). The Subsequent Application is Bona Fide when it is complete and accurate, meaning that all required fields on the Subsequent Application are completed with the appropriate type of information. BellSouth shall determine what modifications, if any, to the Premises are required to accommodate the change requested by Knology in the application. Such necessary modifications to the Premises may include, but are not limited to, floor loading changes, changes necessary to meet HVAC requirements, changes to power plant requirements, equipment additions, etc.
- 6.3.1 <u>Subsequent Application Fee.</u> The application fee paid by Knology for its request to modify the use of the Collocation Space shall be dependent upon the level of assessment needed for the modification requested. The fee for a Subsequent Application where the modification requested has limited effect (e.g., requires labor expenditure but no capital expenditure by BellSouth and where sufficient cable support structure, HVAC, power and terminations are available) shall be the Subsequent Application Fee as set forth in Exhibit B. If the modification requires capital expenditure, an Initial Application Fee shall apply. This non-recurring fee will be billed on the date that BellSouth makes an Application Response.
- 6.4 <u>Space Preferences</u>. If Knology has previously requested and received a Space Availability Report for the Premises, Knology may submit up to three (3) space preferences on its application identifying specific space identification numbers as referenced on the Space Availability Report. In the event that BellSouth can-not accommodate the Knology's preference(s), Knology may elect to accept the space allocated by BellSouth or may cancel its application and submit another application requesting additional preferences, which will be treated as a new application and an

application fee will apply which will be billed by BellSouth on the date that BellSouth makes an Application Response.

- 6.5 <u>Space Availability Notification</u>.
- 6.5.1 Unless otherwise specified, BellSouth will respond to an application within ten (10) calendar days as to whether space is available or not available within a BellSouth Premises. BellSouth will also respond as to whether the application is Bona Fide and if it is not Bona Fide, the items necessary to cause the application to become Bona Fide. If the amount of space requested is not available, BellSouth will notify Knology of the amount of space that is available and no application fee shall apply. When BellSouth's response includes an amount of space less than that requested by Knology or differently configured, Knology must resubmit its application to reflect the actual space available.
- 6.5.2 BellSouth will respond to a Florida application within fifteen (15) calendar days as to whether space is available or not available within a BellSouth Premises. BellSouth will also respond as to whether the application is Bona Fide and if it is not Bona Fide, the items necessary to cause the application to become Bona Fide. If a lesser amount of space than requested is available, BellSouth will provide an Application Response for the amount of space that is available and an application fee will be billed by BellSouth on the date that BellSouth makes an Application Response. When BellSouth's Application Response includes an amount of space less than that requested by Knology or differently configured, Knology must amend its application to reflect the actual space available prior to submitting a BFFO.
- 6.5.3 BellSouth will respond to a Louisiana application within ten (10) calendar days for space availability for one (1) to ten (10) applications; fifteen (15) calendar days for eleven (11) to twenty (20) applications; and for more than twenty (20) applications, the response interval is increased by five (5) calendar days for every five additional applications received within five (5) business days. If the amount of space requested is not available, BellSouth will notify Knology of the amount of space that is available and no application fee shall apply. When BellSouth's response includes an amount of space less than that requested by Knology or differently configured, Knology must resubmit its application to reflect the actual space available. BellSouth will also respond as to whether the application is Bona Fide and if it is not Bona Fide, the items necessary to cause the application to become Bona Fide.
- 6.6 <u>Denial of Application</u>. If BellSouth notifies Knology that no space is available ("Denial of Application"), BellSouth will not assess an Application Fee. After notifying Knology that BellSouth has no available space in the requested Premises, BellSouth will allow Knology, upon request, to tour the entire Premises within ten (10) calendar days of such Denial of Application. In order to schedule said tour within ten (10) calendar days, the request for a tour of the Premises must be received by BellSouth within five (5) calendar days of the Denial of Application.

- 6.7 <u>Filing of Petition for Waiver</u>. Upon Denial of Application, BellSouth will timely file a petition with the Commission pursuant to 47 U.S.C. § 251(c)(6). BellSouth shall provide to the Commission any information requested by that Commission. Such information shall include which space, if any, BellSouth or any of BellSouth's affiliates have reserved for future use and a detailed description of the specific future uses for which the space has been reserved. Subject to an appropriate nondisclosure agreement or provision, BellSouth shall permit Knology to inspect any floor plans or diagrams that BellSouth provides to the Commission.
- 6.8 <u>Waiting List.</u> On a first-come, first-served basis governed by the date of receipt of an application or Letter of Intent, BellSouth will maintain a waiting list of requesting carriers who have either received a Denial of Application or, where it is publicly known that the Premises is out of space, have submitted a Letter of Intent to collocate. BellSouth will notify the telecommunications carriers on the waiting list that can be accommodated by the amount of space that becomes available according to the position of the telecommunications carriers on said waiting list.
- 6.8.1 In Florida, on a first-come, first-served basis governed by the date of receipt of an application or Letter of Intent, BellSouth will maintain a waiting list of requesting carriers who have either received a Denial of Application or, where it is publicly known that the Premises is out of space, have submitted a Letter of Intent to collocate. Sixty (60) calendar days prior to space becoming available, if known, BellSouth will notify the Florida PSC and the telecommunications carriers on the waiting list by mail when space becomes available according to the position of the telecommunications carrier on said waiting list. If not known sixty (60) calendar days in advance, BellSouth shall notify the Florida PSC and the telecommunications carriers on the waiting list within two (2) business days of the determination that space is available. A telecommunications carrier that, upon denial of physical collocation, requests virtual collocation shall be automatically placed on the waiting list.
- 6.8.2 When space becomes available, Knology must submit an updated, complete, and correct application to BellSouth within thirty (30) calendar days of such notification. If Knology has originally requested caged Collocation Space and cageless Collocation Space becomes available, Knology may refuse such space and notify BellSouth in writing within that time that Knology wants to maintain its place on the waiting list without accepting such space. Knology may accept an amount of space less than its original request by submitting an application as set forth above, and upon request, may maintain its position on the waiting list for the remaining space that was initially requested. If Knology does not submit such an application or notify BellSouth in writing as described above, BellSouth will offer such space to the next telecommunications carrier on the waiting list and remove Knology from the waiting list. Upon request, BellSouth will advise Knology as to its position on the list.
- 6.9 <u>Public Notification</u>. BellSouth will maintain on its Interconnection Services website a notification document that will indicate all Central Offices that are without available space. BellSouth shall update such document within ten (10) calendar days of the date

BellSouth becomes aware that there is insufficient space to accommodate physical collocation. BellSouth will also post a document on its Interconnection Services website that contains a general notice where space has become available in a Central Office previously on the space exhaust list.

6.10 <u>Application Response.</u>

- 6.10.1 In Alabama, when space has been determined to be available, BellSouth will provide an Application Response within fifteen (15) calendar days of the receipt of a Bona Fide Application, which will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and any other applicable space preparation fees, described in Section 8.
- 6.10.2 In Florida, within fifteen (15) calendar days of receipt of a Bona Fide Application, when space has been determined to be available or when a lesser amount of space than that requested is available, then with respect to the space available, BellSouth will provide an Application Response including sufficient information to enable Knology to place a Firm Order. The Application Response will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 8. When Knology submits ten (10) or more applications within ten (10) calendar days, the initial fifteen (15) calendar day response period will increase by ten (10) calendar days for every additional ten (10) applications or fraction thereof.
- 6.10.3 In Georgia, Kentucky, Mississippi, North Carolina, South Carolina, and Tennessee, when space has been determined to be available for caged or cageless arrangements, BellSouth will provide an Application Response within twenty (20) calendar days of receipt of a Bona Fide application. The Application Response will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and any other applicable space preparation fees, as described in Section 8.
- 6.10.4 In Louisiana, when space has been determined to be available, BellSouth will provide an Application Response within thirty (30) calendar days for one (1) to ten (10) applications; thirty-five (35) calendar days for eleven (11) to twenty (20) applications; and for requests of more than twenty (20) applications, the Application Response interval will be increased by five (5) calendar days for every five (5) applications received within five (5) business days. The Application Response will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 8.

6.11 <u>Application Modifications</u>.

6.11.1 If a modification or revision is made to any information in the Bona Fide application prior to a BFFO, with the exception of modifications to Customer Information, Contact Information or Billing Contact Information, either at the request of Knology or necessitated by technical considerations, said application shall be considered a new
application and shall be handled as a new application with respect to response and provisioning intervals and BellSouth may charge Knology an additional application fee. The fee for an application modification where the modification requested has limited effect (e.g., requires labor expenditure but no capital expenditure by BellSouth and where sufficient cable support structure, HVAC, power and terminations are available) shall be the Subsequent Application Fee as set forth in Exhibit B. A modification involving a capital expenditure by BellSouth shall require Knology to submit the application with an Initial Application Fee. This non-recurring fee will be billed by BellSouth on the date that BellSouth provides an Application Response.

6.12 <u>Bona Fide Firm Order</u>.

- 6.12.1 Knology shall indicate its intent to proceed with equipment installation in a BellSouth Premises by submitting a Firm Order to BellSouth. The BFFO must be received by BellSouth no later than thirty (30) calendar days after BellSouth's Application Response to Knology's Bona Fide application or the application will expire.
- 6.12.2 BellSouth will establish a firm order date based upon the date BellSouth is in receipt of a BFFO. BellSouth will acknowledge the receipt of Knology's BFFO within seven (7) calendar days of receipt indicating that the BFFO has been received. A BellSouth response to a BFFO will include a Firm Order Confirmation containing the firm order date. No revisions will be made to a BFFO.

7. <u>Construction and Provisioning</u>

7.1 <u>Construction and Provisioning Intervals</u>

7.1.1 In Alabama, BellSouth will complete construction for caged collocation arrangements under ordinary conditions as soon as possible within a maximum of ninety (90) calendar days from receipt of a BFFO or as agreed to by the Parties. BellSouth will complete construction for cageless collocation arrangements when preconditioned space is available within thirty (30) calendar days from receipt of a BFFO (ordinary conditions) or as agreed to by the Parties. Under extraordinary conditions, BellSouth will complete construction for cageless collocation arrangements as soon as possible within a maximum of ninety (90) calendar days from receipt of a BFFO or as agreed to by the Parties. Preconditioned space is defined as when all infrastructure is in place and only a record change is required to show that the space has been assigned to Knology. Ordinary conditions are defined as space available with only minor changes to support systems required, such as, but not limited to HVAC, cabling and the power plant(s). Extraordinary conditions are defined to include, but are not limited to, major BellSouth equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement; and arrangements for which equipment shipping intervals are extraordinary in length. The Parties may mutually agree to renegotiate an alternative provisioning interval or BellSouth may seek a waiver from this interval from the Commission.

- 7.1.2 In Florida, BellSouth will complete construction for collocation arrangements as soon as possible and within a maximum of ninety (90) calendar days from receipt of a BFFO or as agreed to by the Parties. For changes to the Collocation Space after initial space completion ("Augmentation"), BellSouth will complete construction for collocation arrangements as soon as possible and within a maximum of forty-five (45) calendar days from receipt of a BFFO or as agreed to by the Parties. If BellSouth does not believe that construction will be completed within the relevant time frame and BellSouth and Knology cannot agree upon a completion date, within forty-five (45) calendar days of receipt of the BFFO for an initial request, and within thirty (30) calendar days for Augmentations, BellSouth may seek an extension from the Florida Commission.
- 7.1.3 In Georgia, Kentucky Mississippi, North Carolina, and Tennessee, BellSouth will complete construction for caged collocation arrangements under ordinary conditions as soon as possible and within a maximum of ninety (90) calendar days from receipt of a BFFO or as agreed to by the Parties. BellSouth will complete construction for cageless collocation arrangements under ordinary conditions as soon as possible and within a maximum of sixty (60) calendar days from receipt of a BFFO and ninety (90) calendar days from receipt of a BFFO for extraordinary conditions or as agreed to by the Parties. Ordinary conditions are defined as space available with only minor changes to support systems required, such as but not limited to, HVAC, cabling and the power plant(s). Extraordinary conditions are defined to include but are not limited to major BellSouth equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement; and arrangements for which equipment shipping intervals are extraordinary in length. The Parties may mutually agree to renegotiate an alternative provisioning interval or BellSouth may seek a waiver from this interval from the Commission.
- 7.1.4 In Louisiana, BellSouth will complete construction for collocation arrangements under ordinary conditions as soon as possible and within a maximum of ninety (90) calendar days for caged and sixty (60) calendar days for cageless from receipt of a BFFO for an initial request, and within sixty (60) calendar days for an Augmentation, or as agreed to by the Parties. Ordinary conditions are defined as space available with only minor changes to support systems required, such as but not limited to, HVAC, cabling and the power plant(s). BellSouth will complete construction of all other Collocation Space ("extraordinary conditions") within one hundred twenty (120) calendar days for caged and ninety (90) calendar days for cageless from the receipt of a BFFO. Examples of extraordinary conditions include but are not limited to, extended license or permitting intervals; major BellSouth equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement; and arrangements for which equipment shipping intervals are extraordinary in length. The Parties may mutually agree to renegotiate an alternative provisioning interval or BellSouth may seek a waiver from this interval from the Commission.

- 7.1.5 In South Carolina, BellSouth will complete construction for caged collocation arrangements as soon as possible and within a maximum of ninety (90) calendar days from receipt of a BFFO or as agreed to by the Parties. BellSouth will complete construction for cageless collocation arrangements under ordinary conditions as soon as possible and within a maximum of sixty (60) calendar days from receipt of the BFFO and within a maximum of ninety (90) calendar days from receipt of the BFFO under extraordinary conditions, or as agreed to by the Parties. Ordinary conditions are defined as space available with only minor changes to support systems required, such as but not limited to, HVAC, cabling and the power plant(s). Extraordinary conditions are defined to include, but not limited to, a major BellSouth equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement; and arrangements for which equipment shipping intervals are extraordinary in length. The Parties may mutually agree to renegotiate an alternative provisioning interval or BellSouth may seek a waiver from this interval from the Public Service Commission of South Carolina.
- 7.2 Joint Planning. Joint planning between BellSouth and Knology will commence within a maximum of twenty (20) calendar days from BellSouth's receipt of a BFFO. BellSouth will provide the preliminary design of the Collocation Space and the equipment configuration requirements as reflected in the Bona Fide application and affirmed in the BFFO. The Collocation Space completion time period will be provided to Knology during joint planning.
- 7.3 <u>Permits</u>. Each Party or its agents will diligently pursue filing for the permits required for the scope of work to be performed by that Party or its agents within ten (10) calendar days of the completion of finalized construction designs and specifications.
- 7.4 <u>Acceptance Walk-through</u>. Knology will schedule and complete an acceptance walkthrough of each Collocation Space with BellSouth within fifteen (15) calendar days of BellSouth's notifying Knology that the Collocation Space is ready for occupancy (Space Ready Date). In the event that Knology fails to complete an acceptance walkthrough within this fifteen (15) day interval, the Collocation Space shall be deemed accepted by Knology. BellSouth will correct any deviations to Knology's original or jointly amended requirements within seven (7) calendar days after the walk-through, unless the Parties jointly agree upon a different time frame.
- 7.5 <u>Circuit Facility Assignments (CFAs).</u> Unless otherwise specified, BellSouth will provide CFAs to Knology prior to the applicable provisioning interval set forth herein ("Provisioning Interval") for those Premises in which Knology has a physical collocation arrangement with no POT bay or with a POT bay provided by BellSouth prior to 6/1/99. BellSouth cannot provide CFAs to Knology prior to the Provisioning Interval for those Premises in which Knology has a physical collocation arrangement with a POT bay provided by Knology prior to 6/1/99 or a virtual collocation arrangement until Knology provides BellSouth with the following information:

For Knology-provided POT bay - a complete layout of the POT panels (equipment inventory update (EIU) form) showing locations, speeds, etc.

For virtual - a complete layout of Knology's equipment (equipment inventory update (EIU) form), including the locations of the low speed ports and the specific frame terminations to which the equipment will be wired by Knology's BellSouth Certified Supplier

BellSouth cannot begin work on the CFAs until the complete and accurate EIU form is received from Knology. If this EIU is provided ten (10) calendar days prior to the Provisioning Interval, then CFAs will be made available by the Provisioning Interval. If this EIU is not received ten (10) calendar days prior to the Provisioning Interval, then the CFAs will be provided within ten (10) calendar days of receipt of the EIU.

- 7.5.1 BellSouth will bill Knology a nonrecurring charge, as set forth in Exhibit B, each time Knology requests a resend of its CFAs for any reason other than a BellSouth error in the CFAs.
- Use of BellSouth Certified Supplier. Knology shall select a supplier which has been 7.6 approved as a BellSouth Certified Supplier to perform all engineering and installation work. Knology and Knology's BellSouth Certified Supplier must follow and comply with all BellSouth requirements outlined in BellSouth's TR 73503, TR 73519, TR 73572, and TR 73564. In some cases, Knology must select separate BellSouth Certified Suppliers for transmission equipment, switching equipment and power equipment. BellSouth shall provide Knology with a list of BellSouth Certified Suppliers upon request. The BellSouth Certified Supplier(s) shall be responsible for installing Knology's equipment and components, extending power cabling to the BellSouth power distribution frame, performing operational tests after installation is complete, and notifying BellSouth's equipment engineers and Knology upon successful completion of installation, etc. The BellSouth Certified Supplier shall bill Knology directly for all work performed for Knology pursuant to this Attachment, and BellSouth shall have no liability for nor responsibility to pay such charges imposed by the BellSouth Certified Supplier. BellSouth shall make available its supplier certification program to Knology or any supplier proposed by Knology and will not unreasonably withhold certification. All work performed by or for Knology shall conform to generally accepted industry guidelines and standards.
- 7.7 <u>Alarm and Monitoring</u>. BellSouth shall place environmental alarms in the Premises for the protection of BellSouth equipment and facilities. Knology shall be responsible for placement, monitoring and removal of environmental and equipment alarms used to service Knology's Collocation Space. Upon request, BellSouth will provide Knology with applicable tariffed service(s) to facilitate remote monitoring of collocated equipment by Knology. Both Parties shall use best efforts to notify the other of any verified environmental condition known to that Party.

- 7.8 Virtual to Physical Collocation Relocation. In the event physical Collocation Space was previously denied at a location due to technical reasons or space limitations, and physical Collocation Space has subsequently become available, Knology may relocate its virtual collocation arrangements to physical collocation arrangements and pay the appropriate fees for physical collocation and for the rearrangement or reconfiguration of services terminated in the virtual collocation arrangement, as outlined in the appropriate BellSouth tariffs. In the event that BellSouth knows when additional space for physical collocation may become available at the location requested by Knology, such information will be provided to Knology in BellSouth's written denial of physical collocation. To the extent that (i) physical Collocation Space becomes available to Knology within one hundred eighty (180) calendar days of BellSouth's written denial of Knology's request for physical collocation, (ii) BellSouth had knowledge that the space was going to become available, and (iii) Knology was not informed in the written denial that physical Collocation Space would become available within such one hundred eighty (180) calendar days, then Knology may relocate its virtual collocation arrangement to a physical collocation arrangement and will receive a credit for any nonrecurring charges previously paid for such virtual collocation. Knology must arrange with a BellSouth Certified Supplier for the relocation of equipment from its virtual Collocation Space to its physical Collocation Space and will bear the cost of such relocation.
- 7.8.1 In Alabama, BellSouth will complete a relocation from virtual collocation to cageless physical collocation within thirty (30) calendar days and from virtual collocation to caged physical collocation within ninety (90) calendar days.
- 7.9 <u>Virtual to Physical Conversion (In-Place)</u>. Virtual collocation arrangements may be converted to "in-place" physical arrangements if the potential conversion meets the following four criteria: 1) there is no change in the amount of equipment or the configuration of the equipment that was in the virtual collocation arrangement; 2) the conversion of the virtual collocation arrangement will not cause the equipment or the results of that conversion to be located in a space that BellSouth has reserved for its own future needs; 3) the converted arrangement does not limit BellSouth's ability to secure its own equipment and facilities due to the location of the virtual collocation arrangement; and 4) any changes to the arrangement can be accommodated by existing power, HVAC, and other requirements. Unless otherwise specified, BellSouth will complete virtual to in-place physical collocation conversions within sixty (60) calendar days from receipt of the BFFO. BellSouth will bill Knology an Administrative Only Application Fee as set forth in Exhibit B for these changes on the date that BellSouth provides an Application Response.
- 7.9.1 In Alabama and Tennessee, BellSouth will complete Virtual to Physical Conversions (In Place) within thirty (30) calendar days from receipt of the BFFO.
- 7.10 <u>Cancellation</u>. If, at any time prior to space acceptance, Knology cancels its order for the Collocation Space(s) ("Cancellation"), BellSouth will bill the applicable non-recurring rate for any and all work processes for which work has begun. In Georgia, if

Knology cancels its order for Collocation Space at any time prior to space acceptance, BellSouth will bill Knology for all costs incurred prior to the date of Cancellation and for any costs incurred as a direct result of the Cancellation, not to exceed the total amount that would have been due had the order not been cancelled.

- 7.11 <u>Licenses.</u> Knology, at its own expense, will be solely responsible for obtaining from governmental authorities, and any other appropriate agency, entity, or person, all rights, privileges, and licenses necessary or required to operate as a provider of telecommunications services to the public or to occupy the Collocation Space.
- 7.12 <u>Environmental Compliance.</u> The Parties agree to utilize and adhere to the Environmental Hazard Guidelines identified in Exhibit A attached hereto.

8. <u>Rates and Charges</u>

- 8.1 <u>Recurring Charges.</u> If Knology has met the applicable fifteen (15) calendar day walkthrough interval(s) specified in Section 4, billing for recurring charges will begin upon the Space Acceptance Date. In the event that Knology fails to complete an acceptance walk-through within the applicable fifteen (15) calendar day interval(s), billing for recurring charges will commence on the Space Ready Date or on the Space Acceptance Date, whichever is sooner.
- 8.2 <u>Application Fee</u>. BellSouth shall assess an application fee via a service order, which shall be issued at the time BellSouth responds that space is available pursuant to Section 6 (Application Response). Payment of said application fee will be due as dictated by Knology's current billing cycle and is non-refundable.
- 8.2.1 In Tennessee the applicable application fee is the planning fee for both Initial Applications and Subsequent Applications placed by Knology. This fee will be billed by Bellsouth on the date that BellSouth provides an Application Response.
- 8.3 <u>Space Preparation.</u> Space preparation fees consist of a nonrecurring charge for firm order processing and monthly recurring charges for central office modifications, assessed per arrangement, per square foot, and common systems modifications, assessed per arrangement, per square foot, for cageless collocation and per cage for caged collocation. Knology shall remit payment of the nonrecurring firm order-processing fee coincident with submission of a BFFO. The charges recover the costs associated with preparing the Collocation Space, which includes survey, engineering of the Collocation Space, design and modification costs for network, building and support systems. In the event Knology opts for cageless space, the space preparation fees will be assessed based on the total floor space dedicated to Knology as prescribed in this Section.
- 8.4 <u>Cable Installation</u>. Cable Installation Fee(s) are assessed per entrance cable placed. This non-recurring fee will be billed by BellSouth upon receipt of the Knology's BFFO.

- 8.5 Floor Space. The Floor Space Charge includes reasonable charges for lighting, HVAC, and other allocated expenses associated with maintenance of the Premises but does not include any power-related costs incurred by BellSouth. When the Collocation Space is enclosed, Knology shall pay floor space charges based upon the number of square feet so enclosed. When the Collocation Space is not enclosed, Knology shall pay floor space charges based upon the following floor space calculation: [(depth of the equipment lineup in which the rack is placed) + (0.5 x)maintenance aisle depth) + (0.5 x wiring aisle depth)] X (width of rack and spacers). For purposes of this calculation, the depth of the equipment lineup shall consider the footprint of equipment racks plus any equipment overhang. BellSouth will assign unenclosed Collocation Space in conventional equipment rack lineups where feasible. In the event Knology's collocated equipment requires special cable racking, isolated grounding or other treatment which prevents placement within conventional equipment rack lineups, Knology shall be required to request an amount of floor space sufficient to accommodate the total equipment arrangement.
- 8.6 <u>Power</u>. BellSouth shall make available –48 Volt (-48V) DC power for Knology's Collocation Space at a BellSouth Power Board or BellSouth Battery Distribution Fuse Bay (BDFB) at Knology's option within the Premises.
- 8.6.1 When obtaining power from a BDFB, fuses and power cables (A&B) must be engineered (sized), and installed by Knology's BellSouth Certified Supplier. When obtaining power from a BellSouth power board, power cables (A&B) must be engineered (sized), and installed by Knology's BellSouth Certified Supplier. Knology is responsible for contracting with a BellSouth Certified Supplier for power distribution feeder cable runs from a BellSouth BDFB or power board to Knology's equipment. The determination of the BellSouth BDFB or BellSouth power board as the power source will be made at BellSouth's sole, but reasonable, discretion. The BellSouth Certified Supplier contracted by Knology must provide BellSouth a copy of the engineering power specification prior to the day on which Knology's equipment becomes operational. BellSouth will provide the common power feeder cable support structure between the BellSouth BDFB or power board and Knology's arrangement area. Knology shall contract with a BellSouth Certified Supplier who will be responsible for the following: dedicated power cable support structure within Knology's arrangement, power cable feeds, and terminations of cable. Any terminations at a BellSouth power board must be performed by a BellSouth Certified Supplier. Knology shall comply with all applicable National Electric Code (NEC), BellSouth TR73503, Telcordia and ANSI Standards regarding power cabling.
- 8.6.2 If Knology elects to install its own DC Power Plant, BellSouth shall provide AC power to feed Knology's DC Power Plant. Charges for AC power will be assessed per breaker ampere per month. Rates include the provision of commercial and standby AC power. When obtaining power from a BellSouth service panel, protection devices and power cables must be engineered (sized), and installed by Knology's BellSouth Certified Supplier except that BellSouth shall engineer and install protection devices and power cables for Adjacent Collocation. Knology's BellSouth Certified Supplier

must also provide a copy of the engineering power specification prior to the equipment becoming operational. Charges for AC power shall be assessed pursuant to the rates specified in Exhibit B. AC power voltage and phase ratings shall be determined on a per location basis. At Knology's option, Knology may arrange for AC power in an Adjacent Collocation arrangement from a retail provider of electrical power.

- 8.6.3 In Tennessee, recurring charges for -48V DC power consumption will be assessed per ampere per month based upon the engineered and installed power feed fused ampere capacity. Rates include redundant feeder fuse positions (A&B) and common cable racks to Knology's equipment or space enclosure. Knology shall contract with a Certified Supplier who will be responsible for the following: dedicated power cable support structure within Knology's arrangement and terminations of cable within the Collocation Space.
- 8.6.3.1 In Tennessee, non-recurring charges for -48V DC power distribution will be based on the common power feeder cable support structure between the BellSouth BDFB and Knology's arrangement area.
- 8.6.4 In Alabama and Louisiana, Knology has the option to purchase power directly from an electric utility company. Under such an option, Knology is responsible for contracting with the electric utility company for its own power feed and meter, and is financially responsible for purchasing all equipment necessary to accomplish the arrangement, including inverters, batteries, power boards, bus bars, BDFBs, backup power supplies and cabling. The actual work to install this arrangement must be performed by a BellSouth Certified Supplier hired by Knology. Knology's BellSouth Certified Supplier must comply with all applicable safety codes, including the National Electric Safety Codes, in installing this power arrangement. If Knology previously had power supplied by BellSouth, Knology may request to change its arrangement to obtain power from an electric utility company by submitting a subsequent application. BellSouth will waive any application fee for this subsequent application if no other change was requested therein. Any floor space, cable racking, etc utilized by Knology in provisioning said power will be billed on an ICB basis.
- 8.6.5 In South Carolina, Knology has the option to purchase power directly from an electric utility company where technically feasible and where space is available in a requested BellSouth Premises. Under such an option, Knology is responsible for contracting with the electric utility company for its own power feed and meter, and is financially responsible for purchasing all equipment necessary to accomplish the arrangement, including inverters, batteries, power boards, bus bars, BDFBs, backup power supplies and power cabling. The actual work to install this arrangement must be performed by a BellSouth Certified Supplier hired by Knology. Knology's BellSouth Certified Supplier must comply with all applicable national, regional, state and local safety, electrical, fire and building codes, including the National Electric Safety Code standards, in installing this power arrangement, just as BellSouth for the appropriate amount of collocation space that Knology requires to install this type of power

arrangement. BellSouth will evaluate the request and determine if the appropriate amount of space is available within the office for the installation of Knology's power equipment and facilities. This type of power arrangement must be located in an appropriate area in the central office that has been properly conditioned for the installation of power equipment and conforms to the applicable national, regional, state and local safety, electrical, fire and building codes. BellSouth shall waive the application fee or any other non-recurring charge that would otherwise be due from a CLEC that decides to reconfigure an existing collocation power arrangement so as to purchase power directly from an electric utility company as provided herein. Knology shall be responsible for the recurring charges associated with the central office space needed for collocation of this type of power arrangement, including space required to place associated power-related equipment and facilities (i.e., batteries, generator, power meter, etc.). If there is no space available for this type of power arrangement in the requested central office, BellSouth may seek a waiver of these requirements from the Public Service Commission of South Carolina for the central office requested. Knology would still have the option to order its power needs directly from BellSouth.

- 8.6.6 If Knology requests a reduction in the amount of power that BellSouth is currently providing Knology must submit a Subsequent Application. If no modification to the Collocation Space is requested other than the reduction in power, the Subsequent Application Fee for Power Reduction as set forth in Exhibit B will apply. If modifications are requested in addition to the reduction of power the Subsequent Application Fee will apply. This non-recurring fee will be billed by BellSouth on the date that BellSouth provides an Application Response.
- 8.6.7 In Alabama and Louisiana, if Knology is currently served from the BellSouth main power board and requests that its power be reconfigured to connect to a BellSouth BDFB, in a specific central office, Knology must submit a Subsequent Application. BellSouth will respond to such application within seven (7) calendar days and no application fee will apply.
- 8.7 <u>Security Escort</u>. A security escort will be required whenever Knology or its approved agent desires access to the entrance manhole or must have access to the Premises after the one accompanied site visit allowed pursuant to Section 5 prior to completing BellSouth's Security Training requirements. Rates for a security escort are assessed according to the schedule appended hereto as Exhibit B beginning with the scheduled escort time. BellSouth will wait for one-half (1/2) hour after the scheduled time for such an escort and Knology shall pay for such half-hour charges in the event Knology fails to show up.
- 8.8 <u>Cable Record charges.</u> These charges apply for work required to build cable records in BellSouth systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records. These nonrecurring fees will be billed upon receipt of Knology's BFFO.
- 8.9 <u>Other</u>. If no rate is identified in the contract, the rate for the specific service or function will be negotiated by the Parties upon request by either Party.
 Version 3O02: 09/06/02

9. <u>Insurance</u>

- 9.1 Knology shall, at its sole cost and expense, procure, maintain, and keep in force insurance as specified in this Section and underwritten by insurance companies licensed to do business in the states applicable under this Attachment and having a Best's Insurance Rating of A-.
- 9.2 Knology shall maintain the following specific coverage:
- 9.2.1 Commercial General Liability coverage in the amount of ten million dollars (\$10,000,000.00) or a combination of Commercial General Liability and Excess/Umbrella coverage totaling not less than ten million dollars (\$10,000,000.00). BellSouth shall be named as an Additional Insured on the Commercial General Liability policy as specified herein.
- 9.2.2 Statutory Workers Compensation coverage and Employers Liability coverage in the amount of one hundred thousand dollars (\$100,000.00) each accident, one hundred thousand dollars (\$100,000.00) each employee by disease, and five hundred thousand dollars (\$500,000.00) policy limit by disease.
- 9.2.3 All Risk Property coverage on a full replacement cost basis insuring all of Knology's real and personal property situated on or within BellSouth's Central Office location(s).
- 9.2.4 Knology may elect to purchase business interruption and contingent business interruption insurance, having been advised that BellSouth assumes no liability for loss of profit or revenues should an interruption of service occur.
- 9.3 The limits set forth in Section 9.2 above may be increased by BellSouth from time to time during the term of this Attachment upon thirty (30) calendar days notice to Knology to at least such minimum limits as shall then be customary with respect to comparable occupancy of BellSouth structures.
- 9.4 All policies purchased by Knology shall be deemed to be primary and not contributing to or in excess of any similar coverage purchased by BellSouth. All insurance must be in effect on or before the date equipment is delivered to BellSouth's Premises and shall remain in effect for the term of this Attachment or until all Knology's property has been removed from BellSouth's Premises, whichever period is longer. If Knology fails to maintain required coverage, BellSouth may pay the premiums thereon and seek reimbursement of same from Knology.
- 9.5 Knology shall submit certificates of insurance reflecting the coverage required pursuant to this Section a minimum of ten (10) business days prior to the commencement of any work in the Collocation Space. Failure to meet this interval may result in construction and equipment installation delays. Knology shall arrange for BellSouth to receive thirty (30) business days' advance notice of cancellation from Knology's insurance company. Knology shall forward a certificate of insurance and notice of cancellation/non-renewal to BellSouth at the following address:

BellSouth Telecommunications, Inc. Attn.: Risk Management Coordinator 17H53 BellSouth Center 675 W. Peachtree Street Atlanta, Georgia 30375

- 9.6 Knology must conform to recommendations made by BellSouth's fire insurance company to the extent BellSouth has agreed to, or shall hereafter agree to, such recommendations.
- 9.7 <u>Self-Insurance</u>. If Knology's net worth exceeds five hundred million dollars (\$500,000,000), Knology may elect to request self-insurance status in lieu of obtaining any of the insurance required in Sections 9.2.1 and 9.2.2. Knology shall provide audited financial statements to BellSouth thirty (30) calendar days prior to the commencement of any work in the Collocation Space. BellSouth shall then review such audited financial statements and respond in writing to Knology in the event that self-insurance, Knology shall annually furnish to BellSouth, and keep current, evidence of such net worth that is attested to by one of Knology's corporate officers. The ability to self-insure shall continue so long as the Knology meets all of the requirements of this Section. If Knology subsequently no longer satisfies this Section, Knology is required to purchase insurance as indicated by Sections 9.2.1 and 9.2.2.
- 9.8 The net worth requirements set forth in Section 9.7 may be increased by BellSouth from time to time during the term of this Attachment upon thirty (30) calendar days' notice to Knology to at least such minimum limits as shall then be customary with respect to comparable occupancy of BellSouth structures.
- 9.9 Failure to comply with the provisions of this Section will be deemed a material breach of this Attachment.

10. <u>Mechanics Liens</u>

10.1 If any mechanics lien or other liens shall be filed against property of either Party (BellSouth or Knology), or any improvement thereon by reason of or arising out of any labor or materials furnished or alleged to have been furnished or to be furnished to or for the other Party or by reason of any changes, or additions to said property made at the request or under the direction of the other Party, the other Party directing or requesting those changes shall, within thirty (30) business days after receipt of written notice from the Party against whose property said lien has been filed, either pay such lien or cause the same to be bonded off the affected property in the manner provided by law. The Party causing said lien to be placed against the property of the other shall also defend, at its sole cost and expense, on behalf of the other, any action, suit or proceeding which may be brought for the enforcement of such liens and shall pay any damage and discharge any judgment entered thereon.

11. <u>Inspections</u>

11.1 BellSouth may conduct an inspection of Knology's equipment and facilities in the Collocation Space(s) prior to the activation of facilities between Knology's equipment and equipment of BellSouth. BellSouth may conduct an inspection if Knology adds equipment and may otherwise conduct routine inspections at reasonable intervals mutually agreed upon by the Parties. BellSouth shall provide Knology with a minimum of forty-eight (48) hours or two (2) business days, whichever is greater, advance notice of all such inspections. All costs of such inspection shall be borne by BellSouth.

12. <u>Security and Safety Requirements</u>

- 12.1 Unless otherwise specified, Knology will be required, at its own expense, to conduct a statewide investigation of criminal history records for each Knology employee hired in the past five years being considered for work on the BellSouth Premises, for the states/counties where the Knology employee has worked and lived for the past five years. Where state law does not permit statewide collection or reporting, an investigation of the applicable counties is acceptable. Knology has performed an investigation of the Knology employee seeking access, if such investigation meets the criteria set forth above. This requirement will not apply if Knology has performed a pre-employment statewide investigation of criminal history records of the Knology employee for the states/counties where the Knology employee has worked and lived for the past five years or, where state law does not permit a statewide investigation, an investigation of the applicable counties.
- 12.2 Knology will be required to administer to its personnel assigned to the BellSouth Premises security training either provided by BellSouth, or meeting criteria defined by BellSouth.
- 12.3 Knology shall provide its employees and agents with picture identification, which must be worn and visible at all times while in the Collocation Space or other areas in or around the Premises. The photo identification card shall bear, at a minimum, the employee's name and photo and Knology's name. BellSouth reserves the right to remove from its Premises any employee of Knology not possessing identification issued by Knology or who has violated any of BellSouth's policies as outlined in the CLEC Security Training documents. Knology shall hold BellSouth harmless for any damages resulting from such removal of its personnel from BellSouth Premises. Knology shall be solely responsible for ensuring that any Guest of Knology is in compliance with all subsections of this Section.
- 12.4 Knology shall not assign to the BellSouth Premises any personnel with records of felony criminal convictions. Knology shall not assign to the BellSouth Premises any personnel with records of misdemeanor convictions, except for misdemeanor traffic violations, without advising BellSouth of the nature and gravity of the offense(s).

BellSouth reserves the right to refuse building access to any Knology personnel who have been identified to have misdemeanor criminal convictions. Notwithstanding the foregoing, in the event that Knology chooses not to advise BellSouth of the nature and gravity of any misdemeanor conviction, Knology may, in the alternative, certify to BellSouth that it shall not assign to the BellSouth Premises any personnel with records of misdemeanor convictions (other than misdemeanor traffic violations).

- 12.4.1 Knology shall not knowingly assign to the BellSouth Premises any individual who was a former employee of BellSouth and whose employment with BellSouth was terminated for a criminal offense whether or not BellSouth sought prosecution of the individual for the criminal offense.
- 12.4.2 Knology shall not knowingly assign to the BellSouth Premises any individual who was a former supplier of BellSouth and whose access to a BellSouth Premises was revoked due to commission of a criminal offense whether or not BellSouth sought prosecution of the individual for the criminal offense.
- 12.5 For each Knology employee or agent hired by Knology within five years of being considered for work on the BellSouth Premises, who requires access to a BellSouth Premises pursuant to this Attachment, Knology shall furnish BellSouth, prior to an employee or agent gaining such access, a certification that the aforementioned background check and security training were completed. The certification will contain a statement that no felony convictions were found and certifying that the security training was completed by the employee. If the employee's criminal history includes misdemeanor convictions, Knology will disclose the nature of the convictions to BellSouth at that time. In the alternative, Knology may certify to BellSouth that it shall not assign to the BellSouth Premises any personnel with records of misdemeanor convictions other than misdemeanor traffic violations.
- 12.5.1 For all other Knology employees requiring access to a BellSouth Premises pursuant to this Attachment, Knology shall furnish BellSouth, prior to an employee gaining such access, a certification that the employee is not subject to the requirements of Section 12.5 above and that security training was completed by the employee.
- 12.6 At BellSouth's request, Knology shall promptly remove from BellSouth's Premises any employee of Knology BellSouth does not wish to grant access to its Premises 1) pursuant to any investigation conducted by BellSouth or 2) prior to the initiation of an investigation if an employee of Knology is found interfering with the property or personnel of BellSouth or another collocated telecommunications carrier, provided that an investigation shall promptly be commenced by BellSouth.
- 12.7 <u>Security Violations</u>. BellSouth reserves the right to interview Knology's employees, agents, or suppliers in the event of wrongdoing in or around BellSouth's property or involving BellSouth's or another telecommunications carrier's property or personnel, provided that BellSouth shall provide reasonable notice to Knology's Security contact of such interview. Knology and its suppliers shall reasonably cooperate with BellSouth's investigation into allegations of wrongdoing or criminal conduct

committed by, witnessed by, or involving Knology's employees, agents, or suppliers. Additionally, BellSouth reserves the right to bill Knology for all reasonable costs associated with investigations involving its employees, agents, or suppliers if it is established and mutually agreed in good faith that Knology's employees, agents, or suppliers are responsible for the alleged act. BellSouth shall bill Knology for BellSouth property, which is stolen or damaged where an investigation determines the culpability of Knology's employees, agents, or suppliers and where Knology agrees, in good faith, with the results of such investigation. Knology shall notify BellSouth in writing immediately in the event that Knology discovers one of its employees already working on the BellSouth Premises is a possible security risk. Upon request of the other Party, the Party who is the employer shall discipline consistent with its employment practices, up to and including removal from BellSouth Premises, any employee found to have violated the security and safety requirements of this Section. Knology shall hold BellSouth harmless for any damages resulting from such removal of its personnel from BellSouth Premises.

- 12.8 <u>Use of Supplies</u>. Unauthorized use of equipment, supplies or other property by either Party, whether or not used routinely to provide telephone service will be strictly prohibited and handled appropriately. Costs associated with such unauthorized use may be charged to the offending Party, as may be all associated investigative costs.
- 12.9 <u>Use of Official Lines</u>. Except for non-toll calls necessary in the performance of their work, neither Party shall use the telephones of the other Party on the BellSouth Premises. Charges for unauthorized telephone calls may be charged to the offending Party, as may be all associated investigative costs.
- 12.10 <u>Accountability</u>. Full compliance with the Security requirements of this Section shall in no way limit the accountability of either Party to the other for the improper actions of its employees.

13. <u>Destruction of Collocation Space</u>

13.1 In the event a Collocation Space is wholly or partially damaged by fire, windstorm, tornado, flood or by similar causes to such an extent as to be rendered wholly unsuitable for Knology's permitted use hereunder, then either Party may elect within ten (10) calendar days after such damage, to terminate occupancy of the damaged Collocation Space, and if either Party shall so elect, by giving the other written notice of termination, both Parties shall stand released of and from further liability under the terms hereof. If the Collocation Space shall suffer only minor damage and shall not be rendered wholly unsuitable for Knology's permitted use, or is damaged and the option to terminate is not exercised by either Party, BellSouth covenants and agrees to proceed promptly without expense to Knology, except for improvements not the property of BellSouth, to repair the damage. BellSouth shall have a reasonable time within which to rebuild or make any repairs, and such rebuilding and repairing shall be subject to delays caused by storms, shortages of labor and materials, government regulations, strikes, walkouts, and causes beyond the control of BellSouth, which

causes shall not be construed as limiting factors, but as exemplary only. Knology may, at its own expense, accelerate the rebuild of its collocated space and equipment provided however that a BellSouth Certified Supplier is used and the necessary space preparation has been completed. If Knology's acceleration of the project increases the cost of the project, then those additional charges will be incurred by Knology. Where allowed and where practical, Knology may erect a temporary facility while BellSouth rebuilds or makes repairs. In all cases where the Collocation Space shall be rebuilt or repaired, Knology shall be entitled to an equitable abatement of rent and other charges, depending upon the unsuitability of the Collocation Space for Knology's permitted use, until such Collocation Space is fully repaired and restored and Knology's equipment installed therein (but in no event later than thirty (30) calendar days after the Collocation Space is fully repaired and restored). Where Knology has placed an Adjacent Arrangement pursuant to Section 3, Knology shall have the sole responsibility to repair or replace said Adjacent Arrangement provided herein. Pursuant to this Section, BellSouth will restore the associated services to the Adjacent Arrangement.

14. <u>Eminent Domain</u>

14.1 If the whole of a Collocation Space or Adjacent Arrangement shall be taken by any public authority under the power of eminent domain, then this Attachment shall terminate with respect to such Collocation Space or Adjacent Arrangement as of the day possession shall be taken by such public authority and rent and other charges for the Collocation Space or Adjacent Arrangement shall be paid up to that day with proportionate refund by BellSouth of such rent and charges as may have been paid in advance for a period subsequent to the date of the taking. If any part of the Collocation Space or Adjacent Arrangement shall be taken under eminent domain, BellSouth and Knology shall each have the right to terminate this Attachment with respect to such Collocation Space or Adjacent Arrangement and declare the same null and void, by written notice of such intention to the other Party within ten (10) calendar days after such taking.

15. <u>Nonexclusivity</u>

15.1 Knology understands that this Attachment is not exclusive and that BellSouth may enter into similar agreements with other Parties. Assignment of space pursuant to all such agreements shall be determined by space availability and made on a first come, first served basis

ENVIRONMENTAL AND SAFETY PRINCIPLES

The following principles provide basic guidance on environmental and safety issues when applying for and establishing Physical Collocation arrangements.

1. GENERAL PRINCIPLES

- 1.1 <u>Compliance with Applicable Law</u>. BellSouth and Knology agree to comply with applicable federal, state, and local environmental and safety laws and regulations including U.S. Environmental Protection Agency (USEPA) regulations issued under the Clean Air Act (CAA), Clean Water Act (CWA), Resource Conservation and Recovery Act (RCRA), Comprehensive Environmental Response, Compensation and Liability Act (CERCLA), Superfund Amendments and Reauthorization Act (SARA), the Toxic Substances Control Act (TSCA), and OSHA regulations issued under the Occupational Safety and Health Act of 1970, as amended and NFPA and National Electrical Codes (NEC) and the NESC ("Applicable Laws"). Each Party shall notify the other if compliance inspections are conducted by regulatory agencies and/or citations are issued that relate to any aspect of this Attachment.
- 1.2 <u>Notice</u>. BellSouth and Knology shall provide notice to the other, including Material Safety Data Sheets (MSDSs), of known and recognized physical hazards or Hazardous Chemicals existing on site or brought on site. A Hazardous Chemical inventory list is posted on an OSHA Poster and updated annually at each Central Office. This Poster is normally located near the front entrance of the building or in the lounge area. Each Party is required to provide specific notice for known potential Imminent Danger conditions. Knology should contact 1-800-743-6737 for any BellSouth MSDS required.
- 1.3 <u>Practices/Procedures</u>. BellSouth may make available additional environmental control procedures for Knology to follow when working at a BellSouth Premises (See Section 2, below). These practices/procedures will represent the regular work practices required to be followed by the employees and suppliers of BellSouth for environmental protection. Knology will require its suppliers, agents and others accessing the BellSouth Premises to comply with these practices. Section 2 lists the Environmental categories where BST practices should be followed by Knology when operating in the BellSouth Premises.
- 1.4 <u>Environmental and Safety Inspections</u>. BellSouth reserves the right to inspect the Knology space with proper notification. BellSouth reserves the right to stop any Knology work operation that imposes Imminent Danger to the environment, employees or other persons in the area or Premises.
- 1.5 <u>Hazardous Materials Brought On Site</u>. Any hazardous materials brought into, used, stored or abandoned at the BellSouth Premises by Knology are owned by Knology. Knology will indemnify BellSouth for claims, lawsuits or damages to persons or property caused by these materials. Without prior written BellSouth approval, no substantial new safety or environmental hazards can be created by Knology or different hazardous materials used by Knology at BellSouth Premises. Knology must demonstrate adequate emergency response capabilities for its materials used or remaining at the BellSouth Premises.

- 1.6 <u>Spills and Releases</u>. When contamination is discovered at a BellSouth Premises, the Party discovering the condition must notify BellSouth. All Spills or Releases of regulated materials will immediately be reported by Knology to BellSouth.
- 1.7 <u>Coordinated Environmental Plans and Permits</u>. BellSouth and Knology will coordinate plans, permits or information required to be submitted to government agencies, such as emergency response plans, spill prevention control and countermeasures (SPCC) plans and community reporting. If fees are associated with filing, BellSouth and Knology will develop a cost sharing procedure. If BellSouth's permit or EPA identification number must be used, Knology must comply with all of BellSouth's permit conditions and environmental processes, including environmental "best management practices (BMP)" (see Section 2, below) and/or selection of BST disposition vendors and disposal sites.
- 1.8 <u>Environmental and Safety Indemnification</u>. BellSouth and Knology shall indemnify, defend and hold harmless the other Party from and against any claims (including, without limitation, third-party claims for personal injury or death or real or personal property damage), judgments, damages (including direct and indirect damages and punitive damages), penalties, fines, forfeitures, costs, liabilities, interest and losses arising in connection with the violation or alleged violation of any Applicable Law or contractual obligation or the presence or alleged presence of contamination arising out of the acts or omissions of the indemnifying Party, its agents, suppliers, or employees concerning its operations at the Premises.

2. CATEGORIES FOR CONSIDERATION OF ENVIRONMENTAL ISSUES

- 2.1 When performing functions that fall under the following Environmental categories on BellSouth's Premises, Knology agrees to comply with the applicable sections of the current issue of BellSouth's Environmental and Safety Methods and Procedures (M&Ps), incorporated herein by this reference. Knology further agrees to cooperate with BellSouth to ensure that Knology's employees, agents, and/or suppliers are knowledgeable of and satisfy those provisions of BellSouth's Environmental M&Ps which apply to the specific Environmental function being performed by Knology, its employees, agents and/or suppliers.
- 2.2 The most current version of the reference documentation must be requested from Knology's BellSouth Account Team Collocation Coordinator (ATCC) Representative.

ENVIRONMENTAL CATEGORIES	ENVIRONMENTAL ISSUES	ADDRESSED BY THE FOLLOWING DOCUMENTATION
Disposal of hazardous material or other regulated material	Compliance with all applicable local, state, & federal laws and regulations	Std T&C 450 Fact Sheet Series 17000
(e.g., batteries, fluorescent tubes, solvents & cleaning	Pollution liability insurance	Std T&C 660-3
materials)	EVET approval of supplier	Approved Environmental Vendor List (Contact ATCC

		Representative)
Emergency response	Hazmat/waste release/spill fire safety emergency	Fact Sheet Series 17000 Building Emergency Operations Plan (EOP) (specific to and located on Premises)
Contract labor/outsourcing for services with environmental implications to be performed on BellSouth Premises (e.g., disposition of hazardous material/waste; maintenance of storage tanks)	Compliance with all applicable local, state, & federal laws and regulations Performance of services in accordance with BST's environmental M&Ps Insurance	Std T&C 450 Std T&C 450-B (Contact ATCC Representative for copy of appropriate E/S M&Ps.) Std T&C 660
Transportation of hazardous material	Compliance with all applicable local, state, & federal laws and regulations Pollution liability insurance EVET approval of supplier	Std T&C 450 Fact Sheet Series 17000 Std T&C 660-3 Approved Environmental Vendor List (Contact ATCC Representative)
Maintenance/operations work which may produce a waste	Compliance with all applicable local, state, & federal laws and regulations	Std T&C 450 29CFR 1910.147 (OSHA
Other maintenance work	Protection of BST employees and equipment	Standard) 29CFR 1910 Subpart O (OSHA Standard)
Janitorial services	All waste removal and disposal must conform to all applicable federal, state and local regulations	Procurement Manager (CRES Related Matters)-BST Supply Chain Services
	All Hazardous Material and Waste	Fact Sheet Series 17000
	Asbestos notification and protection of employees and	GU-BTEN-001BT, Chapter 3 BSP 010-170-001BS (Hazcom)

	equipment	
Manhole cleaning	Compliance with all applicable local, state, & federal laws and regulations	Std T&C 450 Fact Sheet 14050 BSP 620-145-011PR Issue A, August 1996
	Pollution liability insurance	Std T&C 660-3
	EVET approval of supplier	Approved Environmental Vendor List (Contact ATCC Representative)
Removing or disturbing building materials that may contain asbestos	Asbestos work practices	GU-BTEN-001BT, Chapter 3 For questions regarding removing or disturbing materials that contain asbestos, call the BellSouth Building Service Center: AL, MS, TN, KY & LA (local area code) 557-6194 FL, GA, NC & SC (local area code) 780-2740

3. DEFINITIONS

<u>Generator</u>. Under RCRA, the person whose act produces a Hazardous Waste, as defined in 40 CFR 261, or whose act first causes a Hazardous Waste to become subject to regulation. The Generator is legally responsible for the proper management and disposal of Hazardous Wastes in accordance with regulations.

<u>Hazardous Chemical</u>. As defined in the U.S. Occupational Safety and Health (OSHA) hazard communication standard (29 CFR 1910.1200), any chemical which is a health hazard or physical hazard.

Hazardous Waste. As defined in Section 1004 of RCRA.

<u>Imminent Danger</u>. Any conditions or practices at a facility which are such that a danger exists which could reasonably be expected to cause immediate death or serious harm to people or immediate significant damage to the environment or natural resources.

Spill or Release. As defined in Section 101 of CERCLA.

4. ACRONYMS

ATCC - Account Team Collocation Coordinator

BST - BellSouth Telecommunications

- <u>CRES</u> Corporate Real Estate and Services (formerly PS&M)
- <u>DEC/LDEC</u> Department Environmental Coordinator/Local Department Environmental Coordinator
- $\underline{E/S}$ Environmental/Safety
- <u>EVET</u> Environmental Vendor Evaluation Team
- <u>GU-BTEN-001BT</u> BellSouth Environmental Methods and Procedures
- <u>NESC</u> National Electrical Safety Codes
- <u>P&SM</u> Property & Services Management
- Std T&C Standard Terms & Conditions

Attachment 4

Remote Site Physical Collocation

BELLSOUTH

REMOTE SITE PHYSICAL COLLOCATION

1. <u>Scope of Attachment</u>

- 1.1 <u>Scope of Attachment.</u> The rates, terms, and conditions contained within this Attachment shall only apply when Knology is occupying the Remote Collocation Space as a sole occupant or as a Host within a Remote Site Location pursuant to this Attachment.
- 1.2 <u>Right to occupy</u>. BellSouth shall offer to Knology Remote Site Collocation on rates, terms, and conditions that are just, reasonable, non-discriminatory and consistent with the rules of the Federal Communications Commission ("FCC"). Subject to the rates, terms, and conditions of this Attachment where space is available and collocation is technically feasible, BellSouth will allow Knology to occupy that certain area designated by BellSouth within a BellSouth Remote Site Location, or on BellSouth property upon which the BellSouth Remote Site Location is located, of a size, which is specified by Knology and agreed to by BellSouth (hereinafter "Remote Collocation Space"). BellSouth Remote Site Locations include cabinets, huts, and controlled environmental vaults owned or leased by BellSouth that house BellSouth Network Facilities. To the extent this Attachment does not include all the necessary rates, terms and conditions for BellSouth remote locations other than cabinets, huts and controlled environmental vaults, the Parties will negotiate said rates, terms, and conditions upon request for collocation at BellSouth remote locations other than those specified above.
- 1.3 <u>Space Reservation.</u>
- 1.3.1 In all states other than Florida, the number of racks/bays specified by Knology may contemplate a request for space sufficient to accommodate Knology's growth within a two year period.
- 1.3.2 In the state of Florida, the number of racks/bays specified by Knology may contemplate a request for space sufficient to accommodate Knology's growth within an eighteen (18) month period.
- 1.3.3 Neither BellSouth nor any of BellSouth's affiliates may reserve space for future use on more preferential terms than those set forth above.
- 1.4 <u>Third Party Property.</u> If the Premises, or the property on which it is located, is leased by BellSouth from a Third Party or otherwise controlled by a Third Party, special considerations and intervals may apply in addition to the terms and conditions of this

Attachment. Additionally, where BellSouth notifies Knology that BellSouth's agreement with a Third Party does not grant BellSouth the ability to provide access and use rights to others, upon Knology's request, BellSouth will use its best efforts to obtain the owner's consent and to otherwise secure such rights for Knology. Knology agrees to reimburse BellSouth for the reasonable and demonstrable costs incurred by BellSouth in obtaining such rights for Knology. In cases where a Third Party agreement does not grant BellSouth the right to provide access and use rights to others as contemplated by this Attachment and BellSouth, despite its best efforts, is unable to secure such access and use rights for Knology as above, Knology shall be responsible for obtaining such permission to access and use such property. BellSouth shall cooperate with Knology in obtaining such permission.

- 1.5 <u>Space Reclamation</u>. In the event of space exhaust within a Remote Site Location, BellSouth may include in its documentation for the Petition for Waiver filing any unutilized space in the Remote Site Location. Knology will be responsible for any justification of unutilized space within its Remote Collocation Space, if the Commission requires such justification.
- 1.6 <u>Use of Space.</u> Knology shall use the Remote Collocation Space for the purposes of installing, maintaining and operating Knology's equipment (to include testing and monitoring equipment) necessary for interconnection with BellSouth services and facilities or for accessing BellSouth unbundled network elements (UNEs) for the provision of telecommunications services, as specifically set forth in this Attachment. The Remote Collocation Space may be used for no other purposes except as specifically described herein or in any amendment hereto.
- 1.7 <u>Rates and charges</u>. Knology agrees to pay the rates and charges identified in Exhibit B attached hereto.
- 1.8 If any due date contained in this Attachment falls on a weekend or National holiday, then the due date will be the next business day thereafter. For intervals of ten (10) calendar days or less National holidays will be excluded.
- 1.9 The Parties agree to comply with all applicable federal, state, county, local and administrative laws, rules, ordinances, regulations and codes in the performance of their obligations hereunder.

2. <u>Space Availability Report</u>

2.1 <u>Space Availability Report</u>. Upon request from Knology, BellSouth will provide a written report ("Space Availability Report"), describing in detail the space that is available for collocation and specifying the amount of Remote Collocation Space available at the Remote Site Location requested, the number of collocators present at the Remote Site Location, any modifications in the use of the space since the last report on the Remote Site Location requested and the measures BellSouth is taking to

make additional space available for collocation arrangements. A Space Availability Report does not reserve space at the Remote Site Location.

- 2.1.1 The request from Knology for a Space Availability Report must be written and must include the Common Language Location Identification ("CLLI") code for both the Remote Site Location and the serving wire center. The CLLI code information for the serving wire center is located in the National Exchange Carriers Association (NECA) Tariff FCC No. 4. If Knology is unable to obtain the CLLI code for the Remote Site Location from, for example, a site visit to the remote site, Knology may request the CLLI code from BellSouth. To obtain a CLLI code for a Remote Site Location directly from BellSouth, Knology should submit to BellSouth a Remote Site Interconnection Request for the serving wire center CLLI code prior to submitting its request for a Space Availability Report. Knology should complete all the requested information and submit the Request to BellSouth. BellSouth will bill the applicable fee upon receipt of the request.
- 2.1.2 BellSouth will respond to a request for a Space Availability Report for a particular Remote Site Location within ten (10) calendar days of receipt of such request. BellSouth will make best efforts to respond in ten (10) calendar days to such a request when the request includes from two (2) to five (5) Remote Site Locations within the same state. The response time for requests of more than five (5) Remote Site Locations shall be negotiated between the Parties. If BellSouth cannot meet the ten (10) calendar day response time, BellSouth shall notify Knology and inform Knology of the time frame under which it can respond.
- 2.2 <u>Remote Terminal information.</u> Upon request, BellSouth will provide Knology with the following information concerning BellSouth's remote terminals: (i) the address of the remote terminal; (ii) the CLLI code of the remote terminal; (iii) the carrier serving area of the remote terminal; (iv) the designation of which remote terminals subtend a particular central office; and (v) the number and address of customers that are served by a particular remote terminal.
- 2.2.1 BellSouth will provide this information on a first come, first served basis within thirty (30) calendar days of a Knology request subject to the following conditions: (i) the information will only be provided on a CD in the same format in which it appears in BellSouth's systems; (ii) the information will only be provided for each serving wire center designated by Knology, up to a maximum of thirty (30) wire centers per Knology request per month per state, and up to for a maximum of 120 wire centers total per month per state for all CLECs; and (iii) Knology agrees to pay the costs incurred by BellSouth in providing the information.

3. <u>Collocation Options</u>

3.1 <u>Cageless</u>. BellSouth shall allow Knology to collocate Knology's equipment and facilities without requiring the construction of a cage or similar structure. BellSouth

shall allow Knology to have direct access to Knology's equipment and facilities. BellSouth shall make cageless collocation available in single rack/bay increments. Except where Knology's equipment requires special technical considerations (e.g., special cable racking, isolated ground plane, etc.), BellSouth shall assign cageless Remote Collocation Space in conventional equipment rack lineups where feasible. For equipment requiring special technical considerations, Knology must provide the equipment layout, including spatial dimensions for such equipment pursuant to generic requirements contained in Telcordia GR-63-Core, and shall be responsible for compliance with all special technical requirements associated with such equipment pursuant.

- 3.2 <u>Caged</u>. At Knology's expense, Knology may arrange with a Supplier certified by BellSouth ("Certified Supplier") to construct a collocation arrangement enclosure, where technically feasible as that term has been defined by the FCC, in accordance with BellSouth's guidelines and specifications prior to starting equipment installation. BellSouth will provide guidelines and specifications upon request. Knology's Certified Supplier shall be responsible for filing and receiving any and all necessary permits and/or licenses for such construction. BellSouth shall cooperate with Knology and provide, at Knology's expense, the documentation, including existing building architectural drawings, enclosure drawings, and specifications required and necessary for Knology to obtain the zoning, permits and/or other licenses. Knology's Certified Supplier shall bill Knology directly for all work performed for Knology pursuant to this Attachment and BellSouth shall have no liability for nor responsibility to pay such charges imposed by Knology's Certified Supplier. Knology must provide the local BellSouth Remote Site Location contact with two Access Keys used to enter the locked enclosure. Except in case of emergency, BellSouth will not access Knology's locked enclosure prior to notifying Knology. Upon request, BellSouth shall construct the enclosure for Knology.
- 3.2.1 BellSouth may elect to review Knology's plans and specifications prior to allowing construction to start to ensure compliance with BellSouth's guidelines and specifications. Notification to Knology indicating BellSouth's desire to execute this review will be provided in BellSouth's response to the Initial Application, if Knology has indicated their desire to construct their own enclosure. If Knology's Initial Application does not indicate their desire to construct their own enclosure, but their subsequent firm order does indicate their desire to construct their own enclosure, then notification to review will be given within ten (10) calendar days after the Firm Order date. BellSouth shall complete its review within fifteen (15) calendar days after the receipt of the plans and specifications. Regardless of whether or not BellSouth elects to review Knology's plans and specifications, BellSouth reserves the right to inspect the enclosure after construction to make sure it is constructed according to the submitted plans and specifications and/or BellSouth's guidelines and specifications, as applicable. BellSouth shall require Knology to remove or correct within seven (7) calendar days at Knology's expense any structure that does not meet these plans and specifications or, where applicable, BellSouth guidelines and specifications.

- 3.3 <u>Shared Collocation</u>. Knology may allow other telecommunications carriers to share Knology's Remote Collocation Space pursuant to terms and conditions agreed to by Knology ("Host") and other telecommunications carriers ("Guests") and pursuant to this Section, except where the BellSouth Remote Site Location is located within a leased space and BellSouth is prohibited by said lease from offering such an option or is located on property for which BellSouth holds an easement and such easement does not permit such an option. Knology shall notify BellSouth in writing upon execution of any agreement between the Host and its Guest within ten (10) calendar days of its execution and prior to any Firm Order. Further, such notice shall include the name of the Guest(s) and the term of the agreement, and shall contain a certification by Knology that said agreement imposes upon the Guest(s) the same terms and conditions for Remote Collocation Space as set forth in this Attachment between BellSouth and Knology.
- 3.3.1 Knology, as the Host, shall be the sole interface and responsible Party to BellSouth for assessment of rates and charges contained within this Attachment and for the purposes of ensuring that the safety and security requirements of this Attachment are fully complied with by the Guest, its employees and agents. BellSouth shall provide Knology with a proration of the costs of the Remote Collocation Space based on the number of collocators and the space used by each with a minimum charge of one (1) bay/rack per Host/Guest. In those instances where the Host permits a Guest to use a shelf within the Host's bay, BellSouth will not prorate the cost of the bay. In all states other than Florida, and in addition to the foregoing, Knology shall be the responsible party to BellSouth for the purpose of submitting applications for initial and additional equipment placement of Guest. In Florida the Guest may directly submit initial and additional equipment placement applications using the Host's access carrier name abbreviation (ACNA). A separate Guest application shall require the assessment of an Initial or Subsequent Application Fee, as set forth in Exhibit B, which will be charged to the Host. BellSouth shall bill this non-recurring fee on the date that BellSouth provides it written response ("Application Response").
- 3.3.2 Notwithstanding the foregoing, the Guest may arrange directly with BellSouth for the provision of the interconnecting facilities between BellSouth and the Guest and for the provision of the services and access to unbundled network elements. The bill for these interconnecting facilities, services and access to UNEs will be charged to the Guest pursuant to the applicable tariff or the Guest's Interconnection Agreement with BellSouth.
- 3.3.3 Knology shall indemnify and hold harmless BellSouth from any and all claims, actions, causes of action, of whatever kind or nature arising out of the presence of Knology's Guests in the Remote Collocation Space except to the extent caused by BellSouth's sole negligence, gross negligence, or willful misconduct.

- 3.4 <u>Adjacent Collocation</u>. Subject to technical feasibility and space availability, BellSouth will permit adjacent Remote Site collocation arrangements ("Remote Site Adjacent Arrangement") on the property on which the Remote Site is located, where the Remote Site Adjacent Arrangement does not interfere with access to existing or planned structures or facilities on the Remote Site Location property. The Remote Site Adjacent Arrangement shall be constructed or procured by Knology and in conformance with BellSouth's design and construction specifications. Further, Knology shall construct, procure, maintain and operate said Remote Site Adjacent Arrangement(s) pursuant to all of the terms and conditions set forth in this Attachment. Rates shall be negotiated at the time of the application for the Remote Site Adjacent Arrangement.
- 3.4.1 Should Knology elect Adjacent Collocation, Knology must arrange with a Certified Supplier to construct a Remote Site Adjacent Arrangement structure in accordance with BellSouth's guidelines and specifications. Where local building codes require enclosure specifications more stringent than BellSouth's standard specification, Knology and Knology's Certified Supplier must comply with local building code requirements. Knology's Certified Supplier shall be responsible for filing and receiving any and all necessary zoning, permits and/or licenses for such construction. Knology's Certified Supplier shall bill Knology directly for all work performed for Knology pursuant to this Attachment and BellSouth shall have no liability for nor responsibility to pay such charges imposed by Knology's Certified Supplier. Knology must provide the local BellSouth Remote Site Location contact with two cards, keys or other access device used to enter the locked enclosure. Except in cases of emergency, BellSouth shall not access Knology's locked enclosure prior to notifying Knology.
- 3.4.2 Knology must submit its plans and specifications to BellSouth with its Firm Order. BellSouth shall review Knology's plans and specifications prior to construction of a Remote Site Adjacent Arrangement(s) to ensure compliance with BellSouth's guidelines and specifications. BellSouth shall complete its review within fifteen (15) calendar days after receipt of plans and specifications. BellSouth may inspect the Remote Site Adjacent Arrangement(s) during and after construction to confirm it is constructed according to the submitted plans and specifications. BellSouth shall require Knology to remove or correct within seven (7) calendar days at Knology's expense any structure that does not meet these plans and specifications.
- 3.4.3 Knology shall provide a concrete pad, the structure housing the arrangement, heating/ventilation/air conditioning ("HVAC"), lighting, and all facilities that connect the structure (i.e. racking, conduits, etc.) to the BellSouth point of demarcation. At Knology's option, and where the local authority having jurisdiction permits, BellSouth shall provide an AC power source and access to physical collocation services and facilities subject to the same nondiscriminatory requirements as applicable to any other physical collocation arrangement. In Alabama and Louisiana, BellSouth will provide DC power to Adjacent Collocation sites where technically feasible, as that term has been defined by the FCC, and subject to individual case basis pricing. Knology's

Certified Supplier shall be responsible, at Knology's expense, for filing and receiving any and all necessary zoning, permits and/or licenses for such arrangement. BellSouth shall allow Shared Collocation within a Remote Site Adjacent Arrangement pursuant to the terms and conditions set forth herein.

- 3.5 <u>Co-carrier cross-connect (CCXC)</u>. The primary purpose of collocation is for a collocated telecommunications carrier to interconnect with BellSouth's network or to access BellSouth's unbundled network elements for the provision of telecommunications services within a BellSouth Premises. BellSouth will permit Knology to interconnect between its virtual or physical collocation arrangements and those of another collocated telecommunications carrier within the same remote site premises. Both Knology's agreement and the other collocated telecommunications carrier's agreement must contain rates, terms and conditions for CCXC language. At no point in time shall Knology use the Remote Collocated telecommunications carriers.
- 3.5.1 Knology must use a BellSouth Certified Supplier to place the CCXC. The CCXC shall be provisioned through facilities owned by Knology. Such connections to other carriers may be made using either optical or electrical facilities. In cases where Knology's equipment and the equipment of the other interconnector are located in contiguous caged Collocation Spaces, Knology will have the option of using Knology's own technicians to deploy co-carrier cross connects using either electrical or optical facilities between the sets of equipment and construct its own dedicated cable support structure. Knology may deploy such optical or electrical connections directly between its own facilities and the facilities of other collocated telecommunications carriers without being routed through BellSouth equipment. Knology may not self-provision CCXC on any BellSouth distribution frame, POT (Point of Termination) Bay, DSX (Digital System Cross-connect) or LGX (Light Guide Cross-connect). Knology is responsible for ensuring the integrity of the signal.
- 3.5.2 Knology shall be responsible for providing written authorization to BellSouth from the other collocated telecommunications carrier prior to installing the CCXC. Knology-provisioned CCXC shall utilize common cable support structure. There will be a recurring charge per linear foot, per cable, of common cable support structure used. In the case of two contiguous caged collocation arrangements, Knology will have the option of using Knology's own technicians to construct its own dedicated support structure.
- 3.5.3 To order CCXCs Knology must submit an Initial Application or Subsequent Application. If no modification to the Remote Collocation Space is requested other than the placement of CCXCs, the Subsequent Application Fee for CCXC, as defined in Exhibit B, will apply. If modifications in addition to the placement of CCXCs are requested, the Initial Application or Subsequent Application Fee will apply. This nonrecurring fee will be billed by BellSouth on the date that BellSouth provides an Application Response.

4. Occupancy

- 4.1 Occupancy. BellSouth will notify Knology in writing that the Remote Collocation Space is ready for occupancy ("Space Ready Date"). Knology will schedule and complete an acceptance walk-through of each Remote Collocation Space with BellSouth within fifteen (15) calendar days of BellSouth's notifying Knology that Remote Collocation Space is ready for occupancy ("Space Ready Date"). . BellSouth will correct any deviations to Knology's original or jointly amended requirements within seven (7) calendar days after the walk-through, unless the Parties jointly agree upon a different time frame, and BellSouth shall establish a new Space Ready Date. Another acceptance walk-through will then be scheduled and conducted within fifteen (15) calendar days of the new Space Ready Date. This follow-up acceptance walkthrough will be limited to those items identified in the initial walk-through. If Knology has met the fifteen (15) calendar day interval(s), billing will begin upon the date of Knology's acceptance of the Collocation Space ("Space Acceptance Date"). In the event that Knology fails to complete an acceptance walk-through within this fifteen (15) calendar day interval, the Remote Collocation Space shall be deemed accepted by Knology. Billing will commence on the Space Ready Date or on the Space Acceptance Date, whichever is sooner. Knology must notify BellSouth in writing that collocation equipment installation is complete and is operational with BellSouth's network. BellSouth may, at its option, not accept orders for cross connects until receipt of such notice. For purposes of this paragraph, Knology's telecommunications equipment will be deemed operational when cross-connected to BellSouth's network for the purpose of service provision.
- 4.2 <u>Termination of Occupancy</u>. In addition to any other provisions addressing termination of occupancy in this Attachment, Knology may terminate occupancy in a particular Remote Collocation Space by submitting a Subsequent Application requesting termination of occupancy. A Subsequent Application Fee will not apply for termination of occupancy. BellSouth may terminate Knology's right to occupy the Remote Collocation Space in the event Knology fails to comply with any provision of this Agreement.
- 4.2.1 Upon termination of occupancy, Knology at its expense shall remove its equipment and other property from the Remote Collocation Space. Knology shall have thirty (30) calendar days from the termination date to complete such removal, including the removal of all equipment and facilities of Knology's Guests, unless Knology's Guest has assumed responsibility for the Remote Collocation Space housing the Guest's equipment and executed the documentation required by BellSouth prior to such removal date. Knology shall continue payment of monthly fees to BellSouth until such date as Knology, and if applicable Knology's Guest, has fully vacated the Remote Collocation Space and the Space Relinquish Form has been accepted by BellSouth. Should Knology or Knology's Guest fail to vacate the Remote Collocation Space within thirty (30) calendar days from the termination date, BellSouth shall have the right to remove the equipment and dispose of the equipment and other property of

Knology or Knology's Guest, in any manner that BellSouth deems fit, at Knology's expense and with no liability whatsoever for Knology or Knology's Guest's property. Upon termination of Knology's right to occupy Remote Collocation Space, the Remote Collocation Space will revert back to BellSouth, and Knology shall surrender such Remote Collocation Space to BellSouth in the same condition as when first occupied by the Knology except for ordinary wear and tear unless otherwise agreed to by the Parties. For CEVs and huts Knology's BellSouth Certified Supplier shall be responsible for updating and making any necessary changes to BellSouth's records as required by BellSouth's guidelines and specifications including but not limited to Record Drawings and ERMA Records. Knology shall be responsible for the cost of removing any Knology constructed enclosure, together with all support structures (e.g., racking, conduits, power cables, etc.), at the termination of occupancy and restoring the grounds to their original condition.

5. <u>Use of Remote Collocation Space</u>

- 5.1 Equipment Type. BellSouth permits the collocation of any type of equipment necessary for interconnection to BellSouth's network or for access to BellSouth's unbundled network elements in the provision of telecommunications services, as the term "necessary" is defined by FCC 47 C.F.R. Section 51.323 (b). The primary purpose and function of any equipment collocated in a Remote Collocated Space must be for interconnection to BellSouth's network or for access to BellSouth's unbundled network elements in the provision of telecommunications services.
- 5.1.1 Examples of equipment that would not be considered necessary include but are not limited to: Traditional circuit switching equipment, equipment used exclusively for call-related databases, computer servers used exclusively for providing information services, operations support system (OSS) equipment used to support collocated telecommunications carrier network operations, equipment that generates customer orders, manages trouble tickets or inventory, or stores customer records in centralized databases, etc. BellSouth will determine upon receipt of an application if the requested equipment is necessary based on the criteria established by the FCC. Multifunctional equipment placed on BellSouth's Premises must not place any greater relative burden on BellSouth's property than comparable single-function equipment. BellSouth reserves the right to permit collocation of any equipment on a nondiscriminatory basis.
- 5.1.2 Such equipment must, at a minimum, meet the following Telcordia Network Equipment Building Systems (NEBS) General Equipment Requirements: Criteria Level 3 requirements as outlined in the Telcordia Special Report SR-3580, Issue 1; equipment design spatial requirements per GR-63-CORE, Section 2; thermal heat dissipation per GR-063-CORE, Section 4, Criteria 77-79; acoustic noise per GR-063-CORE, Section 4, Criterion 128, and National Electric Code standards. Except where otherwise required by a Commission, BellSouth shall comply with the applicable FCC rules relating to denial of collocation based on Knology's failure to comply with this Section.

- 5.1.2.1 All Knology equipment installation shall comply with BellSouth TR 73503-11h, "Grounding - Engineering Procedures". Metallic cable sheaths and metallic strength members of optical fiber cables as well as the metallic cable sheaths of all copper conductor cables shall be bonded to the designated grounding bus for the Remote Site Location. All copper conductor pairs, working and non-working, shall be equipped with a solid-state protector unit (over-voltage protection only) which has been listed by a nationally recognized testing laboratory.
- 5.1.3 Knology shall identify to BellSouth whenever Knology submits a Method of Procedure ("MOP") adding equipment to Knology's Remote Collocation Space all UCC-1 lien holders or other entities that have a financial interest, secured or otherwise, in the equipment in Knology's Remote Collocation Space.
- 5.2 Knology shall not use the Remote Collocation Space for marketing purposes nor shall it place any identifying signs or markings in the area surrounding the Remote Collocation Space or on the grounds of the Remote Site Location.
- 5.3 Knology shall place a plaque or other identification affixed to Knology's equipment to identify Knology's equipment, including a list of emergency contacts with telephone numbers.
- 5.4 <u>Entrance Facilities</u>. Knology may elect to place Knology-owned or Knology-leased fiber entrance facilities into the Remote Collocation Space. BellSouth will designate the point of interconnection at the Remote Site Location housing the Remote Collocation Space, which is physically accessible by both Parties. Knology will provide and place copper cable through conduit from the Remote Collocation Space to the Feeder Distribution Interface to the splice location of sufficient length for splicing by BellSouth. Knology must contact BellSouth for instructions prior to placing the entrance facility cable. Knology is responsible for maintenance of the entrance facilities.
- 5.4.1 <u>Shared Use</u>. Knology may utilize spare capacity on an existing interconnector entrance facility for the purpose of providing an entrance facility to Knology's collocation arrangement within the same BellSouth Remote Site Location. BellSouth shall allow splicing to the entrance facility, provided that the fiber is non-working fiber. The rates set forth in Exhibit B will apply. If Knology desires to allow another telecommunications carrier to use its entrance facilities, additional rates, terms and conditions will apply and shall be negotiated between the Parties.
- 5.5 <u>Demarcation Point</u>. BellSouth will designate the point(s) of demarcation between Knology's equipment and/or network and BellSouth's network. Each Party will be responsible for maintenance and operation of all equipment/facilities on its side of the demarcation point. Knology or its agent must perform all required maintenance to Knology equipment/facilities on its side of the demarcation point, pursuant to Section 5.6, following.

- 5.6 <u>Knology's Equipment and Facilities</u>. Knology, or if required by this Attachment, Knology's Certified Supplier, is solely responsible for the design, engineering, installation, testing, provisioning, performance, monitoring, maintenance and repair of the equipment and facilities used by Knology which must be performed in compliance with all applicable BellSouth policies and guidelines. Such equipment and facilities may include but are not limited to cable(s), equipment, and point of termination connections. Knology and its selected Certified Supplier must follow and comply with all BellSouth requirements outlined in BellSouth's TR 73503, TR 73519, TR 73572, and TR 73564.
- 5.7 <u>BellSouth's Access to Remote Collocation Space</u>. From time to time BellSouth may require access to the Remote Collocation Space. BellSouth retains the right to access the Remote Collocation Space for the purpose of making BellSouth equipment and Remote Site Location modifications.
- 5.8 <u>Access.</u> Pursuant to Section 12, Knology shall have access to the Remote Collocation Space twenty-four (24) hours a day, seven (7) days a week. Knology agrees to provide the name and social security number or date of birth or driver's license number of each employee, supplier, or agents of Knology or Knology's Guests provided with access keys or devices ("Access Keys") prior to the issuance of said Access Keys. Key acknowledgement forms must be signed by Knology and returned to BellSouth Access Management within fifteen (15) calendar days of Knology's receipt. Failure to return properly acknowledged forms will result in the holding of subsequent requests until acknowledgements are current. Access Keys shall not be duplicated under any circumstances. Knology agrees to be responsible for all Access Keys and for the return of all said Access Keys in the possession of Knology's employees, suppliers, Guests, or agents after termination of the employment relationship, contractual obligation with Knology or upon the termination of this Attachment or the termination of occupancy of an individual Remote Site collocation arrangement.
- 5.8.1 BellSouth will permit one accompanied site visit to Knology's designated collocation arrangement location after receipt of the Bona Fide Firm Order (BFFO) without charge to Knology. Knology must submit to BellSouth the completed Access Control Request Form for all employees or agents requiring access to the BellSouth Remote Site Location a minimum of thirty (30) calendar days prior to the date Knology desires access to the Remote Collocation Space. In order to permit reasonable access during construction of the Remote Collocation Space, Knology may submit such a request at any time subsequent to BellSouth's receipt of the BFFO. In the event Knology desires access being approved, in addition to the first accompanied free visit, BellSouth shall permit Knology to access the Remote Collocation Space accompanied by a security escort at Knology's expense. Knology must request escorted access at least three (3) business days prior to the date such access is desired.

- 5.9 <u>Lost or Stolen Access Keys</u>. Knology shall notify BellSouth in writing immediately in the case of lost or stolen Access Keys. Should it become necessary for BellSouth to re-key Remote Site Locations or deactivate a card as a result of a lost Access Key(s) or for failure to return an Access Key(s), Knology shall pay for all reasonable costs associated with the re-keying or deactivating the card.
- 5.10 Interference or Impairment. Notwithstanding any other provisions of this Attachment, Knology shall not use any product or service provided under this Agreement, any other service related thereto or used in combination therewith, or place or use any equipment and facilities in any manner that 1) significantly degrades, interferes with or impairs service provided by BellSouth or by any other entity or any person's use of its telecommunications service; 2) endangers or damages the equipment, facilities or other property of BellSouth or of any other entity or person; 3) compromises the privacy of any communications; or 4)creates an unreasonable risk of injury or death to any individual or to the public. If BellSouth reasonably determines that any equipment or facilities of Knology violates the provisions of this paragraph, BellSouth shall give written notice to Knology, which notice shall direct Knology to cure the violation within forty-eight (48) hours of Knology's actual receipt of written notice or, at a minimum, to commence curative measures within 24 hours and to exercise reasonable diligence to complete such measures as soon as possible thereafter. After receipt of the notice, the Parties agree to consult immediately and, if necessary, to inspect the arrangement.
- 5.10.1 Except in the case of the deployment of an advanced service which significantly degrades the performance of other advanced services or traditional voice band services, if Knology fails to take curative action within 48 hours or if the violation is of a character which poses an immediate and substantial threat of damage to property, injury or death to any person, or any other significant degradation, interference or impairment of BellSouth's or any other entity's service, then and only in that event BellSouth may take such action as it deems appropriate to correct the violation, including without limitation the interruption of electrical power to Knology's equipment. BellSouth will endeavor, but is not required, to provide notice to Knology prior to taking such action and shall have no liability to Knology for any damages arising from such action, except to the extent that such action by BellSouth constitutes willful misconduct.
- 5.10.2 For purposes of this section, the term significantly degrade shall mean an action that noticeably impairs a service from a user's perspective. In the case of the deployment of an advanced service which significantly degrades the performance of other advanced services or traditional voice band services and Knology fails to take curative action within 48 hours then BellSouth will establish before the Commission that the technology deployment is causing the significant degradation. Any claims of network harm presented to Knology or, if subsequently necessary, the Commission must be supported with specific and verifiable information. Where BellSouth demonstrates that a deployed technology is significantly degrading the performance of other

advanced services or traditional voice band services, Knology shall discontinue deployment of that technology and migrate its customers to technologies that will not significantly degrade the performance of other such services. Where the only degraded service itself is a known disturber, and the newly deployed technology satisfies at least one of the criteria for a presumption that is acceptable for deployment under Section 47 C.F.R. 51.230, the degraded service shall not prevail against the newly deployed technology.

- 5.11 <u>Personalty and its Removal</u>. Facilities and equipment placed by Knology in the Remote Collocation Space shall not become a part of the Remote Site Location, even if nailed, screwed or otherwise fastened to the Remote Collocation Space but shall retain their status as personalty and may be removed by Knology at any time. Any damage caused to the Remote Collocation Space by Knology's employees, agents or representatives shall be promptly repaired by Knology at its expense.
- 5.11.1 <u>If Knology decides to remove equipment from its Remote Collocation Space and the</u> removal requires no physical changes, BellSouth will bill Knology an Administrative Only Application Fee as set forth in Exhibit B for these changes. This non-recurring fee will be billed on the date that BellSouth provides an Application Response.
- 5.12 <u>Alterations</u>. In no case shall Knology or any person acting on behalf of Knology make any rearrangement, modification, improvement, addition, or other alteration which could affect in any way space, power, HVAC, and/or safety considerations to the Remote Collocation Space or the BellSouth Remote Site Location without the written consent of BellSouth, which consent shall not be unreasonably withheld. The cost of any specialized alterations shall be paid by Knology. Any such material rearrangement, modification, improvement, addition, or other alteration shall require an application and Application Fee. BellSouth will bill the non-recurring fee on the date that BellSouth provides an Application Response.
- 5.13 <u>Upkeep of Remote Collocation Space</u>. Knology shall be responsible for the general upkeep and cleaning of the Remote Collocation Space. Knology shall be responsible for removing any Knology debris from the Remote Collocation Space and from in and around the Remote Collocation Site on each visit.

6. Ordering and Preparation of Remote Collocation Space

- 6.1 Should any state or federal regulatory agency impose procedures or intervals applicable to Knology and BellSouth that are different from procedures or intervals set forth in this Section, whether now in effect or that become effective after execution of this Agreement, those procedures or intervals shall supersede the requirements set forth herein for that jurisdiction for all applications submitted for the first time after the effective date thereof
- 6.2 <u>Initial Application</u>. For Knology or Knology's Guest(s) initial equipment placement, Knology shall submit to BellSouth a Physical Expanded Interconnection Application Document ("Initial Application"). The application is Bona Fide when it is complete

and accurate, meaning that all required fields on the application are completed with the appropriate type of information. An application fee will apply which will be billed on the date that BellSouth provides an Application Response.

- 6.3 <u>Subsequent Application</u> In the event Knology or Knology's Guest(s) desires to modify the use of the Remote Collocation Space after a BFFO, Knology shall complete an application detailing all information regarding the modification to the Remote Collocation Space ("Subsequent Application"). BellSouth shall determine what modifications, if any, to the Remote Site Location are required to accommodate the change requested by Knology in the application. Such necessary modifications to the Remote Site Location may include, but are not limited to floor loading changes, changes necessary to meet HVAC requirements, changes to power plant requirements, equipment additions, etc.
- 6.3.1 <u>Application Fee for Subsequent Application.</u> The application fee paid by Knology for its request to modify the use of the Collocation Space shall be a full Application Fee as set forth in Exhibit B. The Subsequent Application is Bona Fide when it is complete and accurate, meaning that all required fields on the application are completed with the appropriate type of information. BellSouth will bill the non-recurring fee on the date that BellSouth provides an Application Response.
- 6.4 <u>Availability of Space.</u> Upon submission of an application, BellSouth will permit Knology to physically collocate, pursuant to the terms of this Attachment, at any BellSouth Remote Site Location, unless BellSouth has determined that there is no space available due to space limitations or that Remote Site Collocation is not practical for technical reasons. In the event space is not immediately available at a Remote Site Location, BellSouth reserves the right to make additional space available, in which case the conditions in Section 7 shall apply, or BellSouth may elect to deny space in accordance with this Section in which case virtual or adjacent collocation options may be available. If the amount of space requested is not available, BellSouth will notify Knology of the amount that is available.

6.5 <u>Space Availability Notification</u>.

- 6.5.1 Unless otherwise specified, BellSouth will respond to an application within ten (10) calendar days as to whether space is available or not available within a BellSouth Remote Site Location. BellSouth will also respond as to whether the application is Bona Fide and if it is not Bona Fide the items necessary to cause the application to become Bona Fide. If the amount of space requested is not available, BellSouth will notify Knology of the amount of space that is available and no Application Fee shall apply. When BellSouth's response includes an amount of space less than that requested by Knology or differently configured, Knology must resubmit its application to reflect the actual space available.
- 6.5.2 BellSouth will respond to a Florida application within fifteen (15) calendar days as to whether space is available or not available within a BellSouth Remote Site Location.

BellSouth will also respond as to whether the application is Bona Fide and if it is not Bona Fide the items necessary to cause the application to become Bona Fide. If a lesser amount of space than requested is available, BellSouth will provide an Application Response for the amount of space that is available and an Application Fee will be billed by BellSouth on the date that BellSouth provides an Application Response. When BellSouth's Application Response includes an amount of space less than that requested by Knology or differently configured, Knology must amend its application to reflect the actual space available prior to submitting a BFFO.

- 6.5.3 BellSouth will respond to a Louisiana application within ten (10) calendar days for space availability for one (1) to ten (10) applications; fifteen (15) calendar days for eleven (11) to twenty (20) applications; and for more than twenty (20) applications, the response interval is increased by five (5) calendar days for every five additional applications received within five (5) business days. If the amount of space requested is not available, BellSouth will notify Knology of the amount of space that is available and no Application Fee will apply. When BellSouth's response includes an amount of space less than that requested by Knology or differently configured, Knology must resubmit its application to reflect the actual space available. BellSouth will also respond as to whether the application is Bona Fide and if it is not Bona Fide the items necessary to cause the application to become Bona Fide.
- 6.6 <u>Denial of Application</u>. If BellSouth notifies Knology that no space is available ("Denial of Application"), BellSouth will not assess an Application Fee. After notifying Knology that BellSouth has no available space in the requested Remote Site Location, BellSouth will allow Knology, upon request, to tour the Remote Site Location within ten (10) calendar days of such Denial of Application. In order to schedule said tour within ten (10) calendar days, the request for a tour of the Remote Site Location must be received by BellSouth within five (5) calendar days of the Denial of Application.
- 6.7 <u>Filing of Petition for Waiver</u>. Upon Denial of Application BellSouth will timely file a petition with the Commission pursuant to 47 U.S.C. § 251(c)(6). BellSouth shall provide to the Commission any information requested by that Commission. Such information shall include which space, if any, BellSouth or any of BellSouth's affiliates have reserved for future use and a detailed description of the specific future uses for which the space has been reserved. Subject to an appropriate nondisclosure agreement or provision, BellSouth shall permit Knology to inspect any plans or diagrams that BellSouth provides to the Commission.
- 6.8 <u>Waiting List</u>. On a first-come, first-served basis governed by the date of receipt of an application or Letter of Intent, BellSouth will maintain a waiting list of requesting carriers who have either received a Denial of Application or, where it is publicly known that the Remote Site Location is out of space, have submitted a Letter of Intent to collocate. BellSouth will notify the telecommunications carriers on the waiting list
that can be accommodated by the amount of space that becomes available according to the position of the telecommunications carriers on said waiting list.

- 6.8.1 In Florida, on a first-come, first-served basis governed by the date of receipt of an application or Letter of Intent, BellSouth will maintain a waiting list of requesting carriers who have either received a Denial of Application or, where it is publicly known that the Remote Site Location is out of space, have submitted a Letter of Intent to collocate. Sixty (60) calendar days prior to space becoming available, if known, BellSouth will notify the Florida PSC and the telecommunications carriers on the waiting list by mail when space becomes available according to the position of the telecommunications carrier on said waiting list. If not known sixty (60) calendar days in advance, BellSouth shall notify the Florida PSC and the telecommunications carriers on the waiting list within two business days of the determination that space is available. A telecommunications carrier that, upon denial of physical collocation, requests virtual collocation shall be automatically placed on the waiting list.
- 6.8.2 When space becomes available, Knology must submit an updated, complete, and correct application to BellSouth within thirty (30) calendar days of such notification. If Knology has originally requested caged Remote Collocation Space and cageless Remote Collocation Space becomes available, Knology may refuse such space and notify BellSouth in writing within that time that Knology wants to maintain its place on the waiting list without accepting such space. Knology may accept an amount of space less than its original request by submitting an application as set forth above, and upon request, may maintain its position on the waiting list for the remaining space that was initially requested. If Knology does not submit such an application or notify BellSouth in writing as described above, BellSouth will offer such space to the next telecommunications carrier on the waiting list and remove Knology from the waiting list. Upon request, BellSouth will advise Knology as to its position on the list.
- 6.9 <u>Public Notification</u>. BellSouth will maintain on its Interconnection Services website a notification document that will indicate all Remote Site Locations that are without available space. BellSouth shall update such document within ten (10) calendar days of the date that BellSouth becomes aware that there is insufficient space to accommodate Remote Site Collocation. BellSouth will also post a document on its Interconnection Services website that contains a general notice where space has become available in a Remote Site Location previously on the space exhaust list.
- 6.10 <u>Application Response</u>.
- 6.10.1 In Alabama, when space has been determined to be available, BellSouth will provide an Application Response within fifteen (15) calendar days of the receipt of a Bona Fide Application, which will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and any other applicable space preparation fees, described in Section 8.

- 6.10.2 In Florida, within fifteen (15) calendar days of receipt of a Bona Fide Application, when space has been determined to be available or when a lesser amount of space than that requested is available, then with respect to the space available, BellSouth will provide an Application Response including sufficient information to enable Knology to place a Firm Order. The Application Response will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 8. When Knology submits ten (10) or more applications within ten (10) calendar days, the initial fifteen (15) calendar day response period will increase by ten (10) calendar days for every additional ten (10) applications or fraction thereof.
- 6.10.3 In Georgia, Kentucky, Mississippi, North Carolina, South Carolina, and Tennessee when space has been determined to be available, BellSouth will provide an Application Response within twenty (20) calendar days of receipt of a Bona Fide application. The Application Response will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 8.
- 6.10.4 In Louisiana, when space has been determined to be available, BellSouth will respond with an Application Response within thirty (30) calendar days for one (1) to ten (10) applications; thirty (35) calendar days for eleven (11) to twenty (20) applications; and for requests of more than twenty (20) applications, the Application Response interval will be increased by five (5) calendar days for every five (5) applications received within five (5) business days. The Application Response will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 8.

6.11 <u>Application Modifications</u>.

- 6.11.1 If a modification or revision is made to any information in the Bona Fide application prior to a BFFO, with the exception of modifications to Customer Information, Contact Information or Billing Contact Information, either at the request of Knology or necessitated by technical considerations, said application shall be considered a new application and shall be handled as a new application with respect to response and provisioning intervals and BellSouth will charge Knology a full application fee as set forth in Exhibit B. BellSouth will bill the non-recurring fee on the date that BellSouth provides an Application Response.
- 6.12 <u>Bona Fide Firm Order</u>.
- 6.12.1 Knology shall indicate its intent to proceed with equipment installation in a BellSouth Remote Site Location by submitting a Firm Order to BellSouth. The BFFO must be received by BellSouth no later than thirty (30) calendar days after BellSouth's Application Response to Knology's Bona Fide application or the application will expire.

6.12.2 BellSouth will establish a firm order date based upon the date BellSouth is in receipt of a BFFO. BellSouth will acknowledge the receipt of Knology's BFFO within seven (7) calendar days of receipt indicating that the BFFO has been received. A BellSouth response to a BFFO will include a Firm Order Confirmation containing the firm order date. No revisions will be made to a BFFO.

7. <u>Construction and Provisioning</u>

7.1 <u>Construction and Provisioning Intervals.</u>

- 7.1.1 In Alabama, BellSouth will complete construction for Remote Site collocation arrangements when preconditioned space is available within thirty (30) calendar days from receipt of a BFFO (ordinary conditions) or as agreed to by the Parties. Under extraordinary conditions, BellSouth will complete construction for Remote Site collocation arrangements as soon as possible within a maximum of ninety (90) calendar days from receipt of a BFFO or as agreed to by the Parties. Preconditioned space is defined as when all infrastructure is in place and only a record change is required to show that the space has been assigned to Knology. Ordinary conditions are defined as space available with only minor changes to support systems required, such as, but not limited to HVAC, cabling and the power plant(s). Extraordinary conditions are defined to include, but are not limited to, major BellSouth equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement; and arrangements for which equipment shipping intervals are extraordinary in length. The Parties may mutually agree to renegotiate an alternative provisioning interval or BellSouth may seek a waiver from this interval from the Commission.
- 7.1.2 In Florida, BellSouth will complete construction for collocation arrangements as soon as possible and within a maximum of ninety (90) calendar days from receipt of a BFFO or as agreed to by the Parties. For changes to Remote Collocation Space after initial space completion ("Augmentation"), BellSouth will complete construction for collocation arrangements as soon as possible and within a maximum of forty-five (45) calendar days from receipt of a BFFO or as agreed to by the Parties. If BellSouth does not believe that construction will be completed within the relevant time frame and BellSouth and Knology cannot agree upon a completion date, within forty-five (45) calendar days of receipt of the BFFO for an initial request, and within thirty (30) calendar days for Augmentations, BellSouth may seek an extension from the Florida Commission.
- 7.1.3 In Georgia, Kentucky, Mississippi, North Carolina, South Carolina, and Tennessee, BellSouth will complete construction for collocation arrangements under ordinary conditions as soon as possible and within a maximum of sixty (60) calendar days from receipt of a BFFO or as agreed to by the Parties. The Parties may mutually agree to renegotiate an alternative provisioning interval or BellSouth may seek a waiver from this interval from the Commission.

- 7.1.4 In Louisiana, BellSouth will complete construction for collocation arrangements under ordinary conditions as soon as possible and within a maximum of sixty (60) calendar days from receipt of a BFFO for an initial request, and within 60 calendar days for an Augmentation, or as agreed to by the Parties. The Parties may mutually agree to renegotiate an alternative provisioning interval or BellSouth may seek a waiver from this interval from the Commission.
- 7.2 In the event BellSouth does not have space immediately available at a Remote Site Location, BellSouth may elect to make additional space available by, for example but not limited to, rearranging BellSouth facilities or constructing additional capacity. In such cases, the above intervals shall not apply and BellSouth will provision the Remote Collocation Space in a nondiscriminatory manner and at parity with BellSouth and will provide Knology with the estimated completion date in its Response.
- Joint Planning. Joint planning between BellSouth and Knology will commence within a maximum of twenty (20) calendar days from BellSouth's receipt of a BFFO. BellSouth will provide the preliminary design of the Remote Collocation Space and the equipment configuration requirements as reflected in the Bona Fide application and affirmed in the BFFO. The Remote Collocation Space completion time period will be provided to Knology during joint planning.
- 7.4 <u>Permits</u>. Each Party or its agents will diligently pursue filing for the permits required for the scope of work to be performed by that Party or its agents within ten (10) calendar days of the completion of finalized construction designs and specifications.
- 7.5 <u>Acceptance Walk-through</u>. Knology will schedule and complete an acceptance walkthrough of each Remote Collocation Space with BellSouth within fifteen (15) calendar days of BellSouth's notifying Knology that the Remote Collocation Space is ready for occupancy ("Space Ready Date"). In the event that Knology fails to complete an acceptance walk-through within this fifteen (15) calendar day interval, the Remote Collocation Space shall be deemed accepted by Knology. BellSouth will correct any deviations to Knology's original or jointly amended requirements within seven (7) calendar days after the walk-through, unless the Parties jointly agree upon a different time frame.
- 7.6 Use of BellSouth Certified Supplier. Knology shall select a supplier which has been approved by BellSouth to perform all engineering and installation work Knology and Knology's BellSouth Certified Supplier must follow and comply with all BellSouth requirements outlined in BellSouth's TR 73503, TR 73519, TR 73572, and TR 73564. In some cases, Knology must select separate BellSouth Certified Suppliers for transmission equipment, switching equipment and power equipment. BellSouth shall provide Knology with a list of BellSouth Certified Suppliers upon request. The BellSouth Certified Supplier(s) shall be responsible for installing Knology's equipment and components, extending power cabling to the BellSouth power distribution frame,

performing operational tests after installation is complete, and notifying BellSouth's Outside Plant engineers and Knology upon successful completion of installation. The BellSouth Certified Supplier shall bill Knology directly for all work performed for Knology pursuant to this Attachment, and BellSouth shall have no liability for nor responsibility to pay such charges imposed by the BellSouth Certified Supplier. BellSouth shall make available its supplier certification program to Knology or any supplier proposed by Knology and will not unreasonably withhold certification. All work performed by or for Knology shall conform to generally accepted industry guidelines and standards.

- 7.7 <u>Alarm and Monitoring</u>. BellSouth may place alarms in the Remote Site Location for the protection of BellSouth equipment and facilities. Knology shall be responsible for placement, monitoring and removal of environmental and equipment alarms used to service Knology's Remote Collocation Space. Upon request, BellSouth will provide Knology with applicable tariffed service(s) to facilitate remote monitoring of collocated equipment by Knology. Both Parties shall use best efforts to notify the other of any verified hazardous conditions known to that Party.
- 7.8 Virtual Remote Site Collocation Relocation. In the event physical Remote Collocation Space was previously denied at a Remote Site Location due to technical reasons or space limitations, and physical Remote Collocation Space has subsequently become available, Knology may relocate its virtual Remote Site collocation arrangements to physical Remote Site collocation arrangements and pay the appropriate fees for physical Remote Site collocation and for the rearrangement or reconfiguration of services terminated in the virtual Remote Site collocation arrangement, as outlined in the appropriate BellSouth tariffs. In the event that BellSouth knows when additional space for physical Remote Site collocation may become available at the location requested by Knology, such information will be provided to Knology in BellSouth's written denial of physical Remote Site collocation. To the extent that (i) physical Remote Collocation Space becomes available to Knology within one hundred eighty 180 calendar days of BellSouth's written denial of Knology's request for physical collocation, (ii) BellSouth had knowledge that the space was going to become available, and (iii) Knology was not informed in the written denial that physical Remote Collocation Space would become available within such one hundred eighty 180 calendar days, then Knology may relocate its virtual Remote Site collocation arrangement to a physical Remote Site collocation arrangement and will receive a credit for any nonrecurring charges previously paid for such virtual Remote Site collocation. Knology must arrange with a BellSouth Certified Supplier for the relocation of equipment from its virtual Remote Collocation Space to its physical Remote Collocation Space and will bear the cost of such relocation.
- 7.8.1 In Alabama, BellSouth will complete a relocation from virtual collocation to physical collocation within ninety (90) calendar days.

- 7.9 <u>Virtual to Physical Conversion (In-Place).</u> Virtual collocation arrangements may be converted to "in-place" physical arrangements if the potential conversion meets the following four criteria: 1) there is no change in the amount of equipment or the configuration of the equipment that was in the virtual collocation arrangement; 2) the conversion of the virtual collocation arrangement will not cause the equipment or the results of that conversion to be located in a space that BellSouth has reserved for its own future needs; 3) the converted arrangement does not limit BellSouth's ability to secure its own equipment and facilities due to the location of the virtual collocation arrangement; and 4) any changes to the arrangement can be accommodated by existing power, HVAC, and other requirements. Unless otherwise specified, BellSouth will complete virtual to in-place physical collocation conversions within sixty (60) calendar days from receipt of the BFFO. BellSouth will bill Knology an Administrative Only Application Fee as set forth in Exhibit B for these changes on the date that BellSouth provides an Application Response.
- 7.9.1 In Alabama and Tennessee, BellSouth will complete Virtual to Physical Conversions (In Place) within thirty (30) calendar days from receipt of the BFFO.
- 7.10 <u>Cancellation</u>. If, at any time prior to space acceptance, Knology cancels its order for the Remote Collocation Space(s) ("Cancellation"), BellSouth will bill the applicable non-recurring rate for any and all work processes for which work has begun. In Georgia, if Knology cancels its order for Remote Collocation Space at any time prior to space acceptance, BellSouth will bill Knology for all costs incurred prior to the date of Cancellation and for any costs incurred as a direct result of the Cancellation, not to exceed the total amount that would have been due had the order not been cancelled.
- 7.11 <u>Licenses</u>. Knology, at its own expense, will be solely responsible for obtaining from governmental authorities, and any other appropriate agency, entity, or person, all rights, privileges, and licenses necessary or required to operate as a provider of telecommunications services to the public or to occupy the Remote Collocation Space.
- 7.12 <u>Environmental Hazard Guidelines</u>. The Parties agree to utilize and adhere to the Environmental Hazard Guidelines identified in Exhibit A attached hereto.

8. <u>Rates and Charges</u>

- 8.1 <u>Recurring Charges</u>. If Knology has met the applicable fifteen (15) calendar day walkthrough interval(s) specified in Section 4, billing for recurring charges will begin upon the Space Acceptance Date. In the event that Knology fails to complete an acceptance walk-through within the applicable fifteen (15) calendar day interval, billing for recurring charges will commence on the Space Ready Date or on the Space Acceptance Date, whichever is sooner.
- 8.2 <u>Application Fee</u>. BellSouth shall assess an Application Fee via a service order, which shall be issued at the time BellSouth responds that space is available pursuant to

Section 2. Payment of said Application Fee will be due as dictated by Knology's current billing cycle and is non-refundable.

- 8.2.1 In Tennessee the applicable Application Fee is the Planning Fee for both Initial Applications and Subsequent Applications placed by Knology. BellSouth will bill the non-recurring fee on the date that BellSouth provides an Application Response.
- 8.3 <u>Rack/Bay Space</u>. The rack/bay space charge includes reasonable charges for air conditioning, ventilation and other allocated expenses associated with maintenance of the Remote Site Location, and includes amperage necessary to power Knology's equipment. Knology shall pay rack/bay space charges based upon the number of racks/bays requested. BellSouth will assign Remote Collocation Space in conventional remote site rack/bay lineups where feasible.
- 8.4 <u>Power</u>. BellSouth shall make available –48 Volt (-48V) DC power for Knology's Remote Collocation Space at a BellSouth Power Board or BellSouth Battery Distribution Fuse Bay (BDFB) at Knology's option within the Remote Site Location. The charge for power shall be assessed as part of the recurring charge for rack/bay space. If the power requirements for Knology's equipment exceeds the capacity available, then such power requirements shall be assessed on an individual case basis.
- 8.4.1 <u>Adjacent Collocation Power.</u> Charges for AC power will be assessed per breaker ampere per month. Rates include the provision of commercial and standby AC power, where available. When obtaining power from a BellSouth service panel, protection devices and power cables must be engineered (sized), and installed by Knology's BellSouth Certified Supplier except that BellSouth shall engineer and install protection devices and power cables for Adjacent Collocation. Knology's BellSouth Certified Supplier must also provide a copy of the engineering power specification prior to the equipment becoming operational. Charges for AC power shall be assessed pursuant to the rates specified in Exhibit B. AC power voltage and phase ratings shall be determined on a per location basis. At Knology's option, Knology may arrange for AC power in an Adjacent Collocation arrangement from a retail provider of electrical power.
- 8.5 <u>Security Escort</u>. A security escort will be required whenever Knology or its approved agent desires access to the Remote Site Location after the one accompanied site visit allowed pursuant to Section 5 prior to completing BellSouth's Security Training requirements. Rates for a security escort are assessed according to the schedule appended hereto as Exhibit B beginning with the scheduled escort time. BellSouth will wait for one-half (1/2) hour after the scheduled time for such an escort and Knology shall pay for such half-hour charges in the event Knology fails to show up.
- 8.6 <u>Other</u>. If no rate is identified in the contract, the rate for the specific service or function will be negotiated by the Parties upon request by either Party.

9. <u>Insurance</u>

- 9.1 Knology shall, at its sole cost and expense, procure, maintain, and keep in force insurance as specified in this Section and underwritten by insurance companies licensed to do business in the states applicable under this Attachment and having a Best's Insurance Rating of A-.
- 9.2 Knology shall maintain the following specific coverage:
- 9.2.1 Commercial General Liability coverage in the amount of ten million dollars (\$10,000,000.00) or a combination of Commercial General Liability and Excess/Umbrella coverage totaling not less than ten million dollars (\$10,000,000.00). BellSouth shall be named as an Additional Insured on the Commercial General Liability policy as specified herein.
- 9.2.2 Statutory Workers Compensation coverage and Employers Liability coverage in the amount of one hundred thousand dollars (\$100,000.00) each accident, one hundred thousand dollars (\$100,000.00) each employee by disease, and five hundred thousand dollars (\$500,000.00) policy limit by disease.
- 9.2.3 All Risk Property coverage on a full replacement cost basis insuring all of Knology's real and personal property situated on or within BellSouth's Remote Site Location.
- 9.2.4 Knology may elect to purchase business interruption and contingent business interruption insurance, having been advised that BellSouth assumes no liability for loss of profit or revenues should an interruption of service occur.
- 9.3 The limits set forth in Section 9.2 above may be increased by BellSouth from time to time during the term of this Attachment upon thirty (30) calendar days notice to Knology to at least such minimum limits as shall then be customary with respect to comparable occupancy of BellSouth structures.
- 9.4 All policies purchased by Knology shall be deemed to be primary and not contributing to or in excess of any similar coverage purchased by BellSouth. All insurance must be in effect on or before the date equipment is delivered to BellSouth's Remote Site Location and shall remain in effect for the term of this Attachment or until all of Knology's property has been removed from BellSouth's Remote Site Location, whichever period is longer. If Knology fails to maintain required coverage, BellSouth may pay the premiums thereon and seek reimbursement of same from Knology.
- 9.5 Knology shall submit certificates of insurance reflecting the coverage required pursuant to this Section a minimum of ten (10) business days prior to the commencement of any work in the Remote Collocation Space. Failure to meet this interval may result in construction and equipment installation delays. Knology shall arrange for BellSouth to receive thirty (30) business days' advance notice of

cancellation from Knology's insurance company. Knology shall forward a certificate of insurance and notice of cancellation/non-renewal to BellSouth at the following address:

BellSouth Telecommunications, Inc. Attn.: Risk Management Coordinator 17H53 BellSouth Center 675 W. Peachtree Street Atlanta, Georgia 30375

- 9.6 Knology must conform to recommendations made by BellSouth's fire insurance company to the extent BellSouth has agreed to, or shall hereafter agree to, such recommendations.
- 9.7 <u>Self-Insurance</u>. If Knology's net worth exceeds five hundred million dollars (\$500,000,000), Knology may elect to request self-insurance status in lieu of obtaining any of the insurance required in Sections 9.2.1 and 9.2.2. Knology shall provide audited financial statements to BellSouth thirty (30) calendar days prior to the commencement of any work in the Remote Collocation Space. BellSouth shall then review such audited financial statements and respond in writing to Knology in the event that self-insurance status is not granted to Knology. If BellSouth approves Knology for self-insurance, Knology shall annually furnish to BellSouth, and keep current, evidence of such net worth that is attested to by one of Knology's corporate officers. The ability to self-insurance as indicated by Sections 9.2.1 and Section 9.2.2.
- 9.8 The net worth requirements set forth in Section 9.7 may be increased by BellSouth from time to time during the term of this Attachment upon thirty (30) calendar days' notice to Knology to at least such minimum limits as shall then be customary with respect to comparable occupancy of BellSouth structures.
- 9.9 Failure to comply with the provisions of this Section will be deemed a material breach of this Attachment.

10. <u>Mechanics Liens</u>

10.1 If any mechanics lien or other liens shall be filed against property of either Party (BellSouth or Knology), or any improvement thereon by reason of or arising out of any labor or materials furnished or alleged to have been furnished or to be furnished to or for the other Party or by reason of any changes, or additions to said property made at the request or under the direction of the other Party, the other Party directing or requesting those changes shall, within thirty (30) business days after receipt of written notice from the Party against whose property said lien has been filed, either pay such lien or cause the same to be bonded off the affected property in the manner provided

by law. The Party causing said lien to be placed against the property of the other shall also defend, at its sole cost and expense, on behalf of the other, any action, suit or proceeding which may be brought for the enforcement of such liens and shall pay any damage and discharge any judgment entered thereon.

11. <u>Inspections</u>

11.1 BellSouth may conduct an inspection of Knology's equipment and facilities in the Remote Collocation Space(s) prior to the activation of facilities between Knology's equipment and equipment of BellSouth. BellSouth may conduct an inspection if Knology adds equipment and may otherwise conduct routine inspections at reasonable intervals mutually agreed upon by the Parties. BellSouth shall provide Knology with a minimum of forty-eight (48) hours or two (2) business days, whichever is greater, advance notice of all such inspections. All costs of such inspection shall be borne by BellSouth.

12. <u>Security and Safety Requirements</u>

- 12.1 Unless otherwise specified, Knology will be required, at its own expense, to conduct a statewide investigation of criminal history records for each Knology employee hired in the past five years being considered for work on the BellSouth Remote Site Location, for the states/counties where the Knology employee has worked and lived for the past five years. Where state law does not permit statewide collection or reporting, an investigation of the applicable counties is acceptable. Knology has performed an investigation of the Knology employee seeking access, if such investigation meets the criteria set forth above. This requirement will not apply if Knology has performed a pre-employment statewide investigation of criminal history records of the Knology employee for the states/counties where the Knology employee has worked and lived for the past five years or, where state law does not permit a statewide investigation, an investigation of the applicable counties.
- 12.2 Knology will be required to administer to their personnel assigned to the BellSouth Premises security training either provided by BellSouth, or meeting criteria defined by BellSouth.
- 12.3 Knology shall provide its employees and agents with picture identification, which must be worn, and visible at all times while in the Remote Collocation Space or other areas in or around the Remote Site Location. The photo Identification card shall bear, at a minimum, the employee's name and photo, and Knology's name. BellSouth reserves the right to remove from its Remote Site Location any employee of Knology not possessing identification issued by Knology or who have violated any of BellSouth's policies as outlined in the CLEC Security Training documents. Knology shall hold BellSouth harmless for any damages resulting from such removal of its personnel from

BellSouth Remote Site Location. Knology shall be solely responsible for ensuring that any Guest of Knology is in compliance with all subsections of this Section 12.

- 12.4 Knology shall not assign to the BellSouth Remote Site Location any personnel with records of felony criminal convictions. Knology shall not assign to the BellSouth Remote Site Location any personnel with records of misdemeanor convictions, except for misdemeanor traffic violations, without advising BellSouth of the nature and gravity of the offense(s). BellSouth reserves the right to refuse access to any Knology personnel who have been identified to have misdemeanor criminal convictions. Notwithstanding the foregoing, in the event that Knology chooses not to advise BellSouth of the nature and gravity of any misdemeanor conviction, Knology may, in the alternative, certify to BellSouth that it shall not assign to the BellSouth Remote Site Location any personnel with records of misdemeanor convictions (other than misdemeanor traffic violations).
- 12.4.1 Knology shall not knowingly assign to the BellSouth Remote Site Location any individual who was a former employee of BellSouth and whose employment with BellSouth was terminated for a criminal offense whether or not BellSouth sought prosecution of the individual for the criminal offense.
- 12.4.2 Knology shall not knowingly assign to the BellSouth Remote Site Location any individual who was a former supplier of BellSouth and whose access to a BellSouth Remote Site Location was revoked due to commission of a criminal offense whether or not BellSouth sought prosecution of the individual for the criminal offense.
- 12.5 For each Knology employee or agent hired by Knology within five years of being considered for work on the BellSouth Remote Site Location, who requires access to a BellSouth Remote Site Location pursuant to this Attachment, Knology shall furnish BellSouth, prior to an employee gaining such access, a certification that the aforementioned background check and security training were completed. The certification will contain a statement that no felony convictions were found and certifying that the security training was completed by the employee. If the employee's criminal history includes misdemeanor convictions, Knology will disclose the nature of the convictions to BellSouth at that time. In the alternative, Knology may certify to BellSouth that it shall not assign to the BellSouth Remote Site Location any personnel with records of misdemeanor convictions other than misdemeanor traffic violations.
- 12.5.1 For all other Knology employees requiring access to a BellSouth Remote Site Location pursuant to this Attachment, Knology shall furnish BellSouth, prior to an employee gaining such access, a certification that the employee is not subject to the requirements of Section 12.5 above and that security training was completed by the employee.
- 12.6 At BellSouth's request, Knology shall promptly remove from BellSouth's Remote Site Location any employee of Knology BellSouth does not wish to grant access to its

Remote Site Location 1) pursuant to any investigation conducted by BellSouth or 2) prior to the initiation of an investigation if an employee of Knology is found interfering with the property or personnel of BellSouth or another collocated telecommunications carrier, provided that an investigation shall promptly be commenced by BellSouth.

- 12.7 Security Violations. BellSouth reserves the right to interview Knology's employees, agents, or suppliers in the event of wrongdoing in or around BellSouth's property or involving BellSouth's or another telecommunications carrier's property or personnel, provided that BellSouth shall provide reasonable notice to Knology's Security contact of such interview. Knology and its suppliers shall reasonably cooperate with BellSouth's investigation into allegations of wrongdoing or criminal conduct committed by, witnessed by, or involving Knology's employees, agents, or suppliers. Additionally, BellSouth reserves the right to bill Knology for all reasonable costs associated with investigations involving its employees, agents, or suppliers if it is established and mutually agreed in good faith that Knology's employees, agents, or suppliers are responsible for the alleged act. BellSouth shall bill Knology for BellSouth property, which is stolen or damaged where an investigation determines the culpability of Knology's employees, agents, or suppliers and where Knology agrees, in good faith, with the results of such investigation. Knology shall notify BellSouth in writing immediately in the event that the Knology discovers one of its employees already working on the BellSouth Remote Site Location is a possible security risk. Upon request of the other Party, the Party who is the employer shall discipline consistent with its employment practices, up to and including removal from BellSouth's Remote Site Location, any employee found to have violated the security and safety requirements of this section. Knology shall hold BellSouth harmless for any damages resulting from such removal of its personnel from BellSouth's Remote Site Location.
- 12.8 <u>Use of Supplies</u>. Unauthorized use of telecommunications equipment or supplies by either Party, whether or not used routinely to provide telephone service (e.g. plug-in cards,) will be strictly prohibited and handled appropriately. Costs associated with such unauthorized use may be charged to the offending Party, as may be all associated investigative costs.
- 12.9 <u>Use of Official Lines</u>. Except for non-toll calls necessary in the performance of their work, neither Party shall use the telephones of the other Party on the BellSouth Remote Site Location. Charges for unauthorized telephone calls may be charged to the offending Party, as may be all associated investigative costs.
- 12.10 <u>Accountability</u>. Full compliance with the Security requirements of this Section shall in no way limit the accountability of either Party to the other for the improper actions of its employees.

13. Destruction of Remote Collocation Space

13.1 In the event a Remote Collocation Space is wholly or partially damaged by fire, windstorm, tornado, flood or by similar causes to such an extent as to be rendered wholly unsuitable for Knology's permitted use hereunder, then either Party may elect within ten (10) calendar days after such damage, to terminate this Attachment with respect to the affected Remote Collocation Space, and if either Party shall so elect, by giving the other written notice of termination, both Parties shall stand released of and from further liability under the terms hereof with respect to such Remote Collocation Space. If the Remote Collocation Space shall suffer only minor damage and shall not be rendered wholly unsuitable for Knology's permitted use, or is damaged and the option to terminate is not exercised by either Party, BellSouth covenants and agrees to proceed promptly without expense to Knology, except for improvements not the property of BellSouth, to repair the damage. BellSouth shall have a reasonable time within which to rebuild or make any repairs, and such rebuilding and repairing shall be subject to delays caused by storms, shortages of labor and materials, government regulations, strikes, walkouts, and causes beyond the control of BellSouth, which causes shall not be construed as limiting factors, but as exemplary only. Knology may, at its own expense, accelerate the rebuild of its Remote Collocation Space and equipment provided however that a BellSouth Certified Supplier is used and the necessary space preparation has been completed. Rebuild of equipment must be performed by a BellSouth Certified Vendor. If Knology's acceleration of the project increases the cost of the project, then those additional charges will be incurred by Knology. Where allowed and where practical, Knology may erect a temporary facility while BellSouth rebuilds or makes repairs. In all cases where the Remote Collocation Space shall be rebuilt or repaired, Knology shall be entitled to an equitable abatement of rent and other charges, depending upon the unsuitability of the Remote Collocation Space for Knology's permitted use, until such Remote Collocation Space is fully repaired and restored and Knology's equipment installed therein (but in no event later than thirty (30) calendar days after the Remote Collocation Space is fully repaired and restored). Where Knology has placed a Remote Site Adjacent Arrangement pursuant to Section 3, Knology shall have the sole responsibility to repair or replace said Remote Site Adjacent Arrangement provided herein. Pursuant to this Section, BellSouth will restore the associated services to the Remote Site Adjacent Arrangement.

14. <u>Eminent Domain</u>

14.1 If the whole of a Remote Collocation Space or Remote Site Adjacent Arrangement shall be taken by any public authority under the power of eminent domain, then this Attachment shall terminate with respect to such Remote Collocation Space or Remote Site Adjacent Arrangement as of the day possession shall be taken by such public authority and rent and other charges for the Remote Collocation Space or Remote Site Adjacent Arrangement shall be paid up to that day with proportionate refund by BellSouth of such rent and charges as may have been paid in advance for a period subsequent to the date of the taking. If any part of the Remote Collocation Space or Remote Site Adjacent Arrangement shall be taken under eminent domain, BellSouth and Knology shall each have the right to terminate this Attachment with respect to such Remote Collocation Space or Remote Site Adjacent Arrangement and declare the same null and void, by written notice of such intention to the other Party within ten (10) calendar days after such taking.

15. <u>Nonexclusivity</u>

15.1 Knology understands that this Attachment is not exclusive and that BellSouth may enter into similar agreements with other Parties. Assignment of space pursuant to all such agreements shall be determined by space availability and made on a first come, first served basis.

ENVIRONMENTAL AND SAFETY PRINCIPLES

The following principles provide basic guidance on environmental and safety issues when applying for and establishing Physical Collocation arrangements.

1. GENERAL PRINCIPLES

- 1.1 <u>Compliance with Applicable Law</u>. BellSouth and Knology agree to comply with applicable federal, state, and local environmental and safety laws and regulations including U.S. Environmental Protection Agency (USEPA) regulations issued under the Clean Air Act (CAA), Clean Water Act (CWA), Resource Conservation and Recovery Act (RCRA), Comprehensive Environmental Response, Compensation and Liability Act (CERCLA), Superfund Amendments and Reauthorization Act (SARA), the Toxic Substances Control Act (TSCA), and OSHA regulations issued under the Occupational Safety and Health Act of 1970, as amended and NFPA and National Electrical Codes (NEC) and the NESC ("Applicable Laws"). Each Party shall notify the other if compliance inspections are conducted by regulatory agencies and/or citations are issued that relate to any aspect of this Attachment.
- 1.2 <u>Notice</u>. BellSouth and Knology shall provide notice to the other, including Material Safety Data Sheets (MSDSs), of known and recognized physical hazards or Hazardous Chemicals existing on site or brought on site. A Hazardous Chemical inventory list is posted on an OSHA Poster and updated annually at each Central Office. This Poster is normally located near the front entrance of the building or in the lounge area. Each Party is required to provide specific notice for known potential Imminent Danger conditions. Knology should contact 1-800-743-6737 for any BellSouth MSDS required.
- 1.3 <u>Practices/Procedures</u>. BellSouth may make available additional environmental control procedures for Knology to follow when working at a BellSouth Remote Site Location (See Section 2, below). These practices/procedures will represent the regular work practices required to be followed by the employees and suppliers of BellSouth for environmental protection. Knology will require its suppliers, agents and others accessing the BellSouth Remote Site Location to comply with these practices. Section 2 lists the Environmental categories where BST practices should be followed by Knology when operating in the BellSouth Remote Site Location.
- 1.4 <u>Environmental and Safety Inspections</u>. BellSouth reserves the right to inspect the Knology space with proper notification. BellSouth reserves the right to stop any Knology work operation that imposes Imminent Danger to the environment, employees or other persons in the area or Remote Site Location.
- 1.5 <u>Hazardous Materials Brought On Site</u>. Any hazardous materials brought into, used, stored or abandoned at the BellSouth Remote Site Location by Knology are owned by Knology. Knology will indemnify BellSouth for claims, lawsuits or damages to persons or property caused by these materials. Without prior written BellSouth approval, no substantial new safety or environmental hazards can be created by Knology or different hazardous materials used by Knology at the BellSouth Remote Site Location. Knology must demonstrate adequate emergency response capabilities for its materials used or remaining at the BellSouth Remote Site Location.

- 1.6 <u>Spills and Releases</u>. When contamination is discovered at a BellSouth Remote Site Location, the Party discovering the condition must notify BellSouth. All Spills or Releases of regulated materials will immediately be reported by Knology to BellSouth.
- 1.7 <u>Coordinated Environmental Plans and Permits</u>. BellSouth and Knology will coordinate plans, permits or information required to be submitted to government agencies, such as emergency response plans, spill prevention control and countermeasures (SPCC) plans and community reporting. If fees are associated with filing, BellSouth and Knology will develop a cost sharing procedure. If BellSouth's permit or EPA identification number must be used, Knology must comply with all of BellSouth's permit conditions and environmental processes, including environmental "best management practices (BMP)" (see Section 2, below) and/or selection of BST disposition vendors and disposal sites.
- 1.8 <u>Environmental and Safety Indemnification</u>. BellSouth and Knology shall indemnify, defend and hold harmless the other Party from and against any claims (including, without limitation, third-party claims for personal injury or death or real or personal property damage), judgments, damages, (including direct and indirect damages, and punitive damages), penalties, fines, forfeitures, costs, liabilities, interest and losses arising in connection with the violation or alleged violation of any Applicable Law or contractual obligation or the presence or alleged presence of contamination arising out of the acts or omissions of the indemnifying Party, its agents, suppliers, or employees concerning its operations at the Remote Site Location.

2. CATEGORIES FOR CONSIDERATION OF ENVIRONMENTAL ISSUES

- 2.1 When performing functions that fall under the following Environmental categories on BellSouth's Remote Site Location, Knology agrees to comply with the applicable sections of the current issue of BellSouth's Environmental and Safety Methods and Procedures (M&Ps), incorporated herein by this reference. Knology further agrees to cooperate with BellSouth to ensure that Knology's employees, agents, and/or suppliers are knowledgeable of and satisfy those provisions of BellSouth's Environmental M&Ps which apply to the specific Environmental function being performed by Knology, its employees, agents and/or suppliers.
- 2.1.1 The most current version of reference documentation must be requested from Knology's BellSouth Account Team Collocation Coordinator (ATCC) Representative.

ENVIRONMENTAL CATEGORIES	ENVIRONMENTAL ISSUES	ADDRESSED BY THE FOLLOWING DOCUMENTATION
Disposal of hazardous material	Compliance with all applicable	• Std T&C 450
or other regulated material (e.g., batteries, fluorescent	local, state, & federal laws and regulations	• Fact Sheet Series 17000
tubes, solvents & cleaning materials)	Pollution liability insurance	• Std T&C 660-3
	EVET approval of supplier	Approved Environmental Vendor List (Contact ATCC

		Representative)
Emergency response	Hazmat/waste release/spill fire safety emergency	 Fact Sheet Series 1700 Building Emergency Operations Plan (EOP) (specific to and located on Remote Site Location)
Contract labor/outsourcing for services with environmental implications to be performed on BellSouth Remote Site Location (e.g., disposition of hazardous material/waste; maintenance of storage tanks)	Compliance with all applicable local, state, & federal laws and regulations Performance of services in accordance with BST's environmental M&Ps Insurance	 Std T&C 450 Std T&C 450-B (Contact ATCC Representative for copy of appropriate E/S M&Ps.) Std T&C 660
Transportation of hazardous material	Compliance with all applicable local, state, & federal laws and regulations Pollution liability insurance	 Std T&C 450 Fact Sheet Series 17000 Std T&C 660-3
	EVET approval of supplier	• Approved Environmental Vendor List (Contact ATCC Representative)
Maintenance/operations work which may produce a waste Other maintenance work	Compliance with all applicable local, state, & federal laws and regulations Protection of BST employees and equipment	 Std T&C 450 29CFR 1910.147 (OSHA Standard) 29CFR 1910 Subpart O (OSHA Standard)
Janitorial services	All waste removal and disposal must conform to all applicable federal, state and local regulations All Hazardous Material and Waste Asbestos notification and protection of employees and equipment	 –Procurement Manager (CRES Related Matters)-BST Supply Chain Services Fact Sheet Series 17000 GU-BTEN-001BT, Chapter 3 BSP 010-170-001BS (Hazcom)

Exhibit A 34

Manhole cleaning	Compliance with all applicable local, state, & federal laws and regulations	 Std T&C 450 Fact Sheet 14050 BSP 620-145-011PR Issue A, August 1996
	Pollution liability insurance	• Std T&C 660-3
	EVET approval of supplier	• Approved Environmental Vendor List (Contact ATCC Representative)
Removing or disturbing building materials that may contain asbestos	Asbestos work practices	• GU-BTEN-001BT, Chapter 3 For questions regarding removing or disturbing materials that contain asbestos, call the BellSouth Building Service Center: AL, MS, TN, KY & LA (local area code) 557-6194 FL, GA, NC & SC (local area code) 780-2740

3. **DEFINITIONS**

<u>Generator</u>. Under RCRA, the person whose act produces a Hazardous Waste, as defined in 40 CFR 261, or whose act first causes a Hazardous Waste to become subject to regulation. The Generator is legally responsible for the proper management and disposal of Hazardous Wastes in accordance with regulations.

<u>Hazardous Chemical</u>. As defined in the U.S. Occupational Safety and Health (OSHA) hazard communication standard (29 CFR 1910.1200), any chemical which is a health hazard or physical hazard.

Hazardous Waste. As defined in section 1004 of RCRA.

<u>Imminent Danger</u>. Any conditions or practices at a facility which are such that a danger exists which could reasonably be expected to cause immediate death or serious harm to people or immediate significant damage to the environment or natural resources.

Spill or Release. As defined in Section 101 of CERCLA.

4. ACRONYMS

ATCC – Account Team Collocation Coordinator

<u>BST</u> – BellSouth Telecommunications

- <u>CRES</u> Corporate Real Estate and Services (formerly PS&M)
- DEC/LDEC Department Environmental Coordinator/Local Department Environmental Coordinator
- $\underline{E/S}$ Environmental/Safety
- <u>EVET</u> Environmental Vendor Evaluation Team
- **<u>GU-BTEN-001BT</u>** BellSouth Environmental Methods and Procedures
- NESC National Electrical Safety Codes
- <u>P&SM</u> Property & Services Management
- Std T&C Standard Terms & Conditions

PI	RATE ELEMENTS	Interi m	Zone	BCS								Svc Order Submitted	Incremental Charge -	Incremental Charge -	Incremental Charge -	Incremental
Pi Pi					USOC			RATES(\$)			Elec per LSR	Manually	Manual Svc Order vs. Electronic- 1st	Manual Svc Order vs. Electronic- Add'l	Manual Svc Order vs. Electronic- Disc 1st	Charge - Manual Svo Order vs. Electronic- Disc Add'I
Pi Pi						Rec	Nonrec		Nonrecurring					Rates(\$)		
Pi Pi							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Pi Pi	OCATION				1											(
	hysical Collocation - Application Fee - Initial			CLO	PE1BA		1,879.48	1,879.48	0.51	0.51						i
	hysical Collocation - Application Fee - Subsequent			CLO	PE1CA		1,566.60	1,566.60	0.51	0.51						l
	hysical Collocation - Cageless - Application Fee			CLO	PE1CH		1,205.26	1,205.26	0.51	0.51						l
	hysical Collocation Administrative Only - Application Fee			CLO	PE1BL		742.15									l
	hysical Collocation - Space Preparation - Firm Order rocessing			CLO	PE1SJ		600.71	600.71								ł
PI	hysical Collocation - Space Preparation - C.O. Modification per juare ft.			CLO	PEISK	1.96	000.71	000.71								
PI	hysical Collocation - Space Preparation - Common Systems odification per square ft Cageless			CLO	PE1SL	2.62										
PI	hysical Collocation - Space Preparation - Common Systems		1						1							
	odification per Cage			CLO	PE1SM	88.86										
	hysical Collocation - Cable Installation		 	CLO	PE1BD	0.02	859.71	859.71	22.49	22.49						
	hysical Collocation - Floor Space per Sq. Ft.			CLO CLO	PE1PJ	3.22 17.11										l
	hysical Collocation - Cable Support Structure hysical Collocation - Cageless - Cable Support Structure			CLO	PE1PM PE1CJ	14.97										
	hysical Collocation - Power -48V DC Power, per Fused Amp			CLO	PE1PL	7.83										1
	hysical Collocation - Power Reduction, Application Fee			CLO	PE1PR	1.00	399.51		1							[
	hysical Collocation - 120V, Single Phase Standby Power Rate			CLO	PE1FB	4.91										
PI	hysical Collocation - 240V, Single Phase Standby Power Rate			CLO	PE1FD	9.84										
Pi	hysical Collocation - 120V, Three Phase Standby Power Rate			CLO	PE1FE	14.74										
PI	hysical Collocation - 277V, Three Phase Standby Power Rate			CLO	PE1FG	34.06										
	hysical Collocation - 2-Wire Cross-Connects			UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ, UDL, UNCVX, UNLDX, UNCNX		0.03	12.30	11.80	6.03	5.44						
PI	Tysical Collocation - 2-Wile Closs-Collifects			CLO, UAL, UDL,	FEIFZ	0.03	12.30	11.00	0.03	5.44						1
P	hysical Collocation - 4-Wire Cross-Connects			UDN, UEA, UHL, UNCVX, UNCDX, UCL	PE1P4	0.05	12.39	11.87	6.39	5.73						
				CLO,UEANL,UEQ,W DS1L,WDS1S, USL, U1TD1, UXTD1, UNC1X, ULDD1, USLEL, UNLD1,	V											
PI	hysical Collocation - DS1 Cross-Connects			UDL CLO, UE3,U1TD3, UXTD3, UXTS1, UNC3X, UNCSX, ULDD3,	PE1P1	1.11	22.03	15.93	6.40	5.79						
PI	hysical Collocation - DS3 Cross-Connects			U1TS1,ULDS1, UNLD3, UDL	PE1P3	14.16	20.89	15.20	7.38	5.92						
PI	hysical Collocation - 2-Fiber Cross-Connect			CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3, UDL12, UDF	PE1F2	2.81	20.89	15.20	7.38	5.92						
	hysical Collocation - Cageless - 2 Fiber Cross Connect			CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3, UDL12, UDF	PE1CK	2.84	20.89	15.20	7.38	5.92						

OOLEOOM	ION - Alabama												Attach	ment: 4	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	
													1st	Add'l	Disc 1st	Disc Add'l
							Nonred	curring	Nonrecurrin	g Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Physical Collocation - 4-Fiber Cross-Connect			CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3, UDL12, UDF	PE1F4	4.99	25.55	19.86	9.71	8.25						
	Physical Collocation - 4-1 iber Closs-Collinect		-	CLO, ULDO3,	FLII4	4.55	23.33	19.00	5.71	0.25						
				ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3,		5.00	05 55	40.00	0.74	0.05						
	Physical Collocation - Cageless - 4-Fiber Cross-Connect			UDL12, UDF	PE1CL PE1BW	5.69	25.55	19.86	9.71	8.25						
	Physical Collocation - Welded Wire Cage - First 100 Sq. Ft. Physical Collocation - Welded Wire Cage - Add'I 50 Sq. Ft.			CLO CLO	PE1BW PE1CW	156.33 15.34		1								
┝───┤────	Physical Collocation - Welded Wire Cage - Add 50 Sq. Ft. Physical Collocation - Security Access System - Security System				FEIGW	15.34										
	per Central Office Physical Collocation - Security Access System - Security System Physical Collocation - Security Access System - New Access			CLO	PE1AX	45.70										
	Card Activation, per Card			CLO	PE1A1	0.05	27.79	27.79								
	Physical Collocation-Security Access System-Administrative				PETAT	0.05	27.79	21.19								
	Change, existing Access Card, per Request, per State, per Card Physical Collocation - Security Access System - Replace Lost or			CLO	PE1AA		7.79	7.79								
	Stolen Card, per Card			CLO	PE1AR		22.78	22.78								
	Physical Collocation - Security Access - Initial Key, per Key			CLO	PE1AK		13.10	13.10								
	Physical Collocation - Security Access - Key, Replace Lost or															
	Stolen Key, per Key			CLO	PE1AL		13.10	13.10								
	Physical Collocation - Space Availability Report per premises			CLO	PE1SR		1,075.17	1,075.17								
	POT Bay Arrangements prior to 6/1/99 - 2-Wire Cross-Connect, per cross-connect			UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ,CLO,UDL, UNCVX, UNCDX, UNCNX	PE1PE	0.08										
	POT Bay Arrangements prior to 6/1/99 - 4-Wire Cross-Connect, per cross-connect			UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ,CLO, USL, UNCVX, UNCDX	PE1PF	0.17										
	POT Bay Arrangements prior to 6/1/99 - DS1 Cross-Connect, per cross-connect			UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ,CLO,WDS1L,W DS1S, USL, U1TD1, UXTD1, UNC1X, ULDD1, USLEL, UNLD1	PE1PG	1.20										
	POT Bay Arrangements prior to 6/1/99 - DS3 Cross-Connect,			UEANL, UEA, UDN, U DC, UAL, UHL, UCL, U EQ, CLO, UE3, U1TD3, UXTD3, UXTS1, UNC3X, UNCSX, ULDD3, U1TS1, ULDS1, UNLD3, UDL,												
	per cross-connect			UDLSX	PE1PH	10.67					1				1	1
	POT Bay Arrangements prior to 6/1/99 - 2-Fiber Cross-Connect, per cross-connect			UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ,CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3, UDL12, UDF	PE1B2	36,40										

COLLOCAT	ION - Alabama				-									ment: 4		bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Submitted Elec	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	POT Bay Arrangements prior to 6/1/99 - 4-Fiber Cross-Connect, per cross-connect			UEANL, UEA, UDN, U DC, UAL, UHL, UCL, U EQ, CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3, UDL12, UDF	PE1B4	49.09										
	Physical Collocation - Request Resend of CFA Information, per CLLI			CLO	PE1C9		77.56									
	Nonrecurring Collocation Cable Records - per request			CLO	PE1CR		759.29	488.11	133.00	133.00						
	Nonrecurring Collocation Cable Records - VG/DS0 Cable, per															
	cable record			CLO	PE1CD		326.92	326.92	189.12	189.12						
	Nonrecurring Collocation Cable Records - VG/DS0 Cable, per each 100 pair			CLO	PE1CO		4.81	4.81	5.90	5.90						
	Nonrecurring Collocation Cable Records - DS1, per T1TIE			CLO	PE1C1		2.25	2.25	2.76	2.76						
	Nonrecurring Collocation Cable Records - DS3, per T3TIE			CLO	PE1C3		7.88	7.88	9.66	9.66						
	Nonrecurring Collocation Cable Records - Fiber Cable, per 99															
	fiber records			CLO	PE1CB		84.49	84.49	77.13	77.13						
	Physical Collocation - Security Escort - Basic, per Half Hour			CLO,CLORS	PE1BT		16.93	10.73								
	Physical Collocation - Security Escort - Overtime, per Half Hour			CLO,CLORS	PE1OT		22.05	13.86								
	Physical Collocation - Security Escort - Premium, per Half Hour			CLO,CLORS	PE1PT		27.17	16.98								
	V to P Conversion, Per Customer Request-Voice Grade			CLO	PE1BV	33.00										
	V to P Conversion, Per Customer Request-DS0			CLO	PE1BO	33.00										
	V to P Conversion, Per Customer Request-DS1 V to P Conversion, Per Customer request-DS3			CLO CLO	PE1B1 PE1B3	52.00 52.00										
	V to P Conversion, Per Customer Request per VG Circuit Reconfigured			CLO	PE1BR	23.00										
	V to P Conversion, Per Customer Request per DS0 Circuit Reconfigured			CLO	PE1BP	23.00										
	V to P Conversion, Per Customer Request per DS1 Circuit Reconfigured			CLO	PE1BS	33.00										
	V to P Conversion, Per Customer Request per DS3 Circuit Reconfigured			CLO	PE1BE	37.00										
	V to P Conversion, Cable Pairs Assigned to Collo Space per 700 prs or fraction thereof			CLO	PE1B7	592.00										
	Physical Collocation - Co-Carrier Cross Connects - Fiber Cable			010		002.00										
	Support Structure, per cable, per linear ft. Physical Collocation - Co-Carrier Cross Connects - Copper/Coax			CLO,UDF	PE1ES	0.0011										
	Cable Support Structure, per cable, per lin. ft. Physical Collocation - Co-Carrier Cross Connects - Application			CLO, UE3, USL	PE1DS	0.0016										
	Fee, per application			CLO	PE1DT		584.22									
PHYSICAL CC																
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2- Wire Analog - Res			UEPSR	PE1R2	0.03	12.30	11.80	6.03	5.44		15.66				
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2- Wire Line Side PBX Trunk - Bus			UEPSP	PE1R2	0.03	12.30	11.80	6.03	5.44		15.66				
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2- Wire Voice Grade PBX Trunk - Res			UEPSE	PE1R2	0.03	12.30	11.80	6.03	5.44	ļ	15.66				
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2- Wire Analog - Bus			UEPSB	PE1R2	0.03	12.30	11.80	6.03	5.44		15.66				
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2- Wire ISDN			UEPSX	PE1R2	0.03	12.30	11.80	6.03	5.44		15.66				
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2- Wire ISDN			UEPTX	PE1R2	0.03	12.30	11.80	6.03	5.44		15.66				
	Physical Collocation 4-Wire Cross Connect, Exchange Port 4- Wire ISDN DS1 DLLOCATION			UEPEX	PE1R4	0.05	12.39	11.87	6.39	5.73		15.66				

COLLOCAT	ION - Alabama													ment: 4		bit: B
											Svc Order	Svc Order	Incremental	Incremental	Incremental	Increment
												Submitted		Charge -	Charge -	Charge
											Elec	Manually			Manual Svc	
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			per LSR			Order vs.		Order vs.
		m		200				===(+)			perLSR	perLSR	Order vs.		Order vs.	
													Electronic-	Electronic-	Electronic-	Electronic
													1st	Add'l	Disc 1st	Disc Add
							Nonrec	urring	Nonrecurring	Disconnect			055	Rates(\$)	1	J
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Adjacent Collocation - Space Charge per Sq. Ft.			CLOAC	PE1JA	0.14		,		/1441	00			00		
	Adjacent Collocation - Electrical Facility Charge per Linear Ft.			CLOAC	PE1JC	5.41										1
	Adjacent Collocation - 2-Wire Cross-Connects			CLOAC	PE1P2	0.02	12.30	11.80	6.03	5.44						1
				UEA,UHL,UDL,UCL,		0.02	12.00	11100	0.00	0.111						1
	Adjacent Collocation - 4-Wire Cross-Connects			CLOAC	PE1P4	0.04	12.39	11.87	6.39	5.73						
	Adjacent Collocation - DS1 Cross-Connects			USL.CLOAC	PE1P1	1.03	22.03	15.93	6.40	5.79						
	Adjacent Collocation - DS1 Closs-Connects			CLOAC	PE1P3	13.95	22.03	15.20	7.38	5.92						
	Adjacent Collocation - 2-Fiber Cross-Connect			CLOAC	PE1F3 PE1F2	2.36	20.89	15.20	7.38	5.92						
	Adjacent Collocation - 2-Fiber Cross-Connect			CLOAC	PE1F2 PE1F4	4.52	20.89	19.86	9.71	8.25						4
						4.52		19.80		8.25						
	Adjacent Collocation - Application Fee			CLOAC	PE1JB		1,576.69		0.51							
	Adjacent Collocation - 120V, Single Phase Standby Power Rate															
	per AC Breaker Amp			CLOAC	PE1FB	4.91										
	Adjacent Collocation - 240V, Single Phase Standby Power Rate															
	per AC Breaker Amp			CLOAC	PE1FD	9.84										
	Adjacent Collocation - 120V, Three Phase Standby Power Rate															
	per AC Breaker Amp			CLOAC	PE1FE	14.74										
	Adjacent Collocation - 277V, Three Phase Standby Power Rate															
	per AC Breaker Amp			CLOAC	PE1FG	34.06										
	Adjacent Collocation - DC power provisioning			CLOAC			ICB									
	Note: ICB means Individual Case Basis															
PHYSICAL CO	LLOCATION IN THE REMOTE SITE															
	Physical Collocation in the Remote Site - Application Fee			CLORS	PE1RA		307.70	307.70	168.22	168.22						
	Cabinet Space in the Remote Site per Bay/ Rack			CLORS	PE1RB	201.42										
	Physical Collocation in the Remote Site - Security Access - Key			CLORS	PE1RD		13.10	13.10								
	Physical Collocation in the Remote Site - Space Availability															
	Report per Premises Requested			CLORS	PE1SR		115.87	115.87								
	Physical Collocation in the Remote Site - Remote Site CLLI															
	Code Request, per CLLI Code Requested			CLORS	PE1RE		37.56	37.56								
	Remote Site DLEC Data (BRSDD), per Compact Disk, per CO			CLORS	PE1RR		233.38									1
PHYSICAL CO	LLOCATION IN THE REMOTE SITE - ADJACENT			020110			200.00									1
				1							1	1		1		
	Remote Site-Adjacent Collocation - AC Power, per breaker amp			CLORS	PE1RS	6.27						1				
├ ── ├ ──	Nonote one najacent conocation - Ac i ower, per breaker amp			01010	1 2 110	0.27					1	+		1	1	╂─────
	Remote Site-Adjacent Collocation - Real Estate, per square foot			CLORS	PE1RT	0.134										
	Remote Site-Adjacent Collocation - Real Estate, per square root			CLORS	PE1RI PE1RU	0.134	755.62	755.62								
1 1				ICLORS										1	1	L

COLLOCAT	ON - Florida				-								Attachr			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
PHYSICAL CO	LLOCATION															ŀ
	Physical Collocation - Application Fee - Initial			CLO	PE1BA		2,597.00		1.01							
	Physical Collocation - Application Fee - Subsequent			CLO	PE1CA		2,236.00		1.01							
	Physical Collocation Administrative Only - Application Fee			CLO	PE1BL		742.00									
	Physical Collocation - Space Preparation - Firm Order Processing			CLO	PE1SJ		288.93									
	Physical Collocation - Space Preparation - C.O. Modification per square ft.			CLO	PE1SK	2.38										
	Physical Collocation - Space Preparation - Common Systems			CLO	PE1SM	92.55										
┝───┼────	Modification per Cage Physical Collocation - Cable Installation per Cable		<u> </u>		PE1SM PE1BD	92.00	1,750.00		45.16			-				t
	Physical Collocation - Floor Space per Sg. Ft.			CLO	PE1PJ	7.86	1,700.00					<u> </u>				
	Physical Collocation - Cable Support Structure			CLO	PE1PM	18.96			1					1		<u> </u>
	Physical Collocation - Power, per Fused Amp			CLO	PE1PL	7.80										
	Physical Collocation - Power Reduction, Application Fee	I		CLO	PE1PR		399.43									
	Physical Collocation - 120V, Single Phase Standby Power Rate			CLO	PE1FB	5.38										ļ
	Physical Collocation - 240V, Single Phase Standby Power Rate			CLO	PE1FD	10.77										
	Physical Collocation - 120V, Three Phase Standby Power Rate			CLO	PE1FE	16.15										
	Physical Collocation - 277V, Three Phase Standby Power Rate			CLO	PE1FG	37.30										
	Physical Collocation - 2-Wire Cross-Connects			UEANL, UEA, UDN, U DC, UAL, UHL, UCL, U EQ, UDL, UNCVX, UNLDX, UNCNX CLO, UAL, UDL,	PE1P2	0.0276	8.22	7.22	5.74	4.58						
				UDN, UEA, UHL, UNCVX, UNCDX,												
	Physical Collocation - 4-Wire Cross-Connects			UCL	PE1P4	0.0552	8.42	7.36	5.90	4.66						
				CLO,UEANL,UEQ,W DS1L,WDS1S, USL, U1TD1, UXTD1, UNC1X, ULDD1, USLEL, UNLD1,												
	Physical Collocation - DS1 Cross-Connects			UDL	PE1P1	1.32	27.77	15.52	5.93	4.77						
				CLO, UE3,U1TD3, UXTD3, UXTS1, UNC3X, UNCSX, ULDD3, U1TS1,ULDS1,												
┝──┼──	Physical Collocation - DS3 Cross-Connects		<u> </u>		PE1P3	16.81	25.48	14.05	7.77	5.01						
	Physical Collocation - 2-Fiber Cross-Connect			CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3, UDL12, UDF	PE1F2	3.34	41.94	30.52	13.91	11.16						
	Physical Collocation - 4-Fiber Cross-Connect			CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3, UDL12, UDF	PE1F4	5.92	51.30	39.87	18.29	15.54						
	Physical Collocation - Welded Wire Cage - First 100 Sq. Ft.	1			PE1BW	189.45	01.00	00.07	10.20	10.04		1		1	1	l
	Physical Collocation - Welded Wire Cage - Add'I 50 Sq. Ft.			CLO	PE1CW	18.58										
	Physical Collocation - Security System Per Central Office Per Assignable Sq. Ft.			CLO	PE1AY	0.0105										

COLLOCAT	FION - Florida												Attachr			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Physical Collocation - Security Access System - New Access															
	Card Activation, per Card			CLO	PE1A1	0.0577	55.80									
	Physical Collocation-Security Access System-Administrative			0.0	DEAAA		15.05									
	Change, existing Access Card, per Request, per State, per Card Physical Collocation - Security Access System - Replace Lost or			CLO	PE1AA		15.65									
	Stolen Card, per Card			CLO	PE1AR		45.75									
	Physical Collocation - Security Access - Initial Key, per Key			CLO	PE1AK		26.30									
	Physical Collocation - Security Access - Key, Replace Lost or															
	Stolen Key, per Key			CLO	PE1AL		26.30									
	Physical Collocation - Space Availability Report per premises			CLO	PE1SR		2,159.00									
				UEANL,UEA,UDN,U												
				DC,UAL,UHL,UCL,U												
				EQ,CLO,UDL,												
	POT Bay Arrangements prior to 6/1/99 - 2-Wire Cross-Connect,			UNCVX, UNCDX,												
	per cross-connect			UNCNX UEANL,UEA,UDN,U	PE1PE	0.00										
				DEANL, UEA, UDN, U DC. UAL, UHL, UCL, U												
	POT Bay Arrangements prior to 6/1/99 - 4-Wire Cross-Connect,			EQ,CLO, USL,												
	per cross-connect			UNCVX, UNCDX	PE1PF	0.00										
				UEANL,UEA,UDN,U	PEIPF	0.00										
				DC,UAL,UHL,UCL,U												
				EQ,CLO,WDS1L,W												
				DS1S, USL, U1TD1,												1
				UXTD1, UNC1X,												1
	POT Bay Arrangements prior to 6/1/99 - DS1 Cross-Connect,			ULDD1, USLEL,												1
	per cross-connect			UNLD1	PE1PG	0.00										
	Ť			UEANL,UEA,UDN,U							1					
				DC,UAL,UHL,UCL,U												
				EQ,CLO,UE3,												
				U1TD3, UXTD3,												
				UXTS1, UNC3X,												
				UNCSX, ULDD3,												
				U1TS1, ULDS1,												
	POT Bay Arrangements prior to 6/1/99 - DS3 Cross-Connect,			UNLD3, UDL,	DEADU	0.00										
	per cross-connect			UDLSX	PE1PH	0.00										
				UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U												
1					I											
						1									1	
				EQ,CLO, ULDO3,												
				EQ,CLO, ULDO3, ULD12, ULD48,												
	POT Bay Arrangements prior to 6/1/99 - 2-Fiber Cross-Connect			EQ,CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12,												
	POT Bay Arrangements prior to 6/1/99 - 2-Fiber Cross-Connect,			EQ,CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3,	PE1B2	0.00										
	POT Bay Arrangements prior to 6/1/99 - 2-Fiber Cross-Connect, per cross-connect			EQ,CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3, UDL12, UDF	PE1B2	0.00										
				EQ,CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3,	PE1B2	0.00										
				EQ,CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3, UDL12, UDF UEANL,UEA,UDN,U	PE1B2	0.00										
				EQ,CLO, ULDO3, ULD12, ULD48, U1T03, U1T12, U1T48, UDL03, UDL12, UDF UEANL,UEA,UDA,U DC,UAL,UHL,UCL,U EQ,CLO, ULD03, ULD12, ULD48,	PE1B2	0.00										
	per cross-connect			EQ,CLO, ULDO3, ULD12, ULD48, U1T03, U1T12, U1T03, U1T12, U1748, UDLO3, UDL12, UDF UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ,CLO, ULD03, ULD12, ULD48, U1T03, U1T12,	PE1B2	0.00										
	per cross-connect POT Bay Arrangements prior to 6/1/99 - 4-Fiber Cross-Connect,			EQ,CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3, UDL12, UDF UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ,CLO, ULDO3, ULD12, ULD48, U1T03, U1T12, U1T48, UDLO3,												
	POT Bay Arrangements prior to 6/1/99 - 4-Fiber Cross-Connect, per cross-connect			EQ,CLO, ULDO3, ULD12, ULD48, U1T03, U1T12, U1T03, U1T12, U1748, UDLO3, UDL12, UDF UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ,CLO, ULD03, ULD12, ULD48, U1T03, U1T12,	PE1B2 PE1B4	0.00										
	POT Bay Arrangements prior to 6/1/99 - 4-Fiber Cross-Connect, per cross-connect Physical Collocation - Request Resend of CFA Information, per			EQ,CLO, ULDO3, ULD12, ULD48, U1T03, UD112, U1T03, UD112, UT48, UDL03, UD24, UDF UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ,CLO, ULD03, ULD12, ULD48, U1T03, U1T12, U1T48, UDL03, UDL12, UDF	PE1B4											
	POT Bay Arrangements prior to 6/1/99 - 4-Fiber Cross-Connect, per cross-connect Physical Collocation - Request Resend of CFA Information, per CLLI			EQ,CLO, ULDO3, ULD12, ULD48, U1T03, U1T12, U1T48, UDL03, UDL12, UDF UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ,CLO, ULD03, ULD12, ULD48, U1T03, U1T12, U1T48, UDL03, UDL12, UDF CLO	PE1B4 PE1C9		77.54									
	POT Bay Arrangements prior to 6/1/99 - 4-Fiber Cross-Connect, per cross-connect Physical Collocation - Request Resend of CFA Information, per CLLI Nonrecurring Collocation Cable Records - per request			EQ,CLO, ULDO3, ULD12, ULD48, U1T03, UD112, U1T03, UD112, UT48, UDL03, UD24, UDF UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ,CLO, ULD03, ULD12, ULD48, U1T03, U1T12, U1T48, UDL03, UDL12, UDF	PE1B4		77.54 1,525.00	980.22	267.08							
	POT Bay Arrangements prior to 6/1/99 - 4-Fiber Cross-Connect, per cross-connect Physical Collocation - Request Resend of CFA Information, per CLLI Nonrecurring Collocation Cable Records - per request Nonrecurring Collocation Cable Records - VG/DS0 Cable, per			EQ,CLO, ULDO3, ULD12, ULD48, U1T03, U1T12, U1T48, UDLO3, UDL12, UDF UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ,CLO, ULD03, ULD12, ULD48, U1T03, U1T12, ULD148, UDLO3, UDL12, UDF CLO CLO	PE1B4 PE1C9 PE1CR		1,525.00	980.22								
	POT Bay Arrangements prior to 6/1/99 - 4-Fiber Cross-Connect, per cross-connect Physical Collocation - Request Resend of CFA Information, per CLLI Nonrecurring Collocation Cable Records - per request Nonrecurring Collocation Cable Records - VG/DS0 Cable, per cable record			EQ,CLO, ULDO3, ULD12, ULD48, U1T03, U1T12, U1T48, UDL03, UDL12, UDF UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ,CLO, ULD03, ULD12, ULD48, U1T03, U1T12, U1T48, UDL03, UDL12, UDF CLO	PE1B4 PE1C9			980.22	267.08 379.78							
	POT Bay Arrangements prior to 6/1/99 - 4-Fiber Cross-Connect, per cross-connect Physical Collocation - Request Resend of CFA Information, per CLLI Nonrecurring Collocation Cable Records - vG/DS0 Cable, per cable record Nonrecurring Collocation Cable Records - VG/DS0 Cable, per			EQ,CLO, ULDO3, ULD12, ULD48, U1T03, U1T12, U1T03, U1T12, U1T48, UDL03, UDL12, UDF UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ,CLO, ULD03, ULD12, ULD48, U1T03, U1T12, U1T48, UDL03, UDL12, UDF CLO CLO	PE1B4 PE1C9 PE1CR PE1CD		1,525.00 656.50		379.78							
	POT Bay Arrangements prior to 6/1/99 - 4-Fiber Cross-Connect, per cross-connect Physical Collocation - Request Resend of CFA Information, per CLLI Nonrecurring Collocation Cable Records - per request Nonrecurring Collocation Cable Records - VG/DS0 Cable, per cable record			EQ,CLO, ULDO3, ULD12, ULD48, U1T03, U1T12, U1T48, UDLO3, UDL12, UDF UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ,CLO, ULD03, ULD12, ULD48, U1T03, U1T12, ULD148, UDLO3, UDL12, UDF CLO CLO	PE1B4 PE1C9 PE1CR		1,525.00	980.22 9.66 4.52		11.84						

COLLOCAT	ION - Florida											-		ment: 4		bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Nonrecurring Collocation Cable Records - Fiber Cable, per 99 fiber records			CLO	PE1CB		169.67	169.67	154.89	154.89						
	liber records			CLO	PEICB		169.67	169.67	154.89	154.89						
	Physical Collocation - Security Escort - Basic, Per Quarter Hour			CLO	PE1BQ		10.89									1
	Physical Collocation - Security Escort - Overtime, Per Quarter			020	. 2.00		10.00									
	Hour			CLO	PE10Q		13.64									1
	Physical Collocation - Security Escort - Premium, Per Quarter															
	Hour			CLO	PE1PQ		16.40									
	Physical Collocation - Security Escort - Basic, per Half Hour			CLO,CLORS	PE1BT		33.99	21.54								
				CLO.CLORS	PE1OT		44.27	07.00								
	Physical Collocation - Security Escort - Overtime, per Half Hour			CLU,CLURS	PEIOI		44.27	27.82								<u> </u>
	Physical Collocation - Security Escort - Premium, per Half Hour			CLO,CLORS	PE1PT		54.55	34.10								1
	V to P Conversion, Per Customer Request-Voice Grade			CLO	PE1BV	33.00	04.00	04.10								
	V to P Conversion, Per Customer Request-DS0			CLO	PE1BO	33.00										
	V to P Conversion, Per Customer Request-DS1			CLO	PE1B1	52.00										
	V to P Conversion, Per Customer request-DS3			CLO	PE1B3	52.00										
	V to P Conversion, Per Customer Request per VG Circuit															
	Reconfigured			CLO	PE1BR	23.00										
	V to P Conversion, Per Customer Request per DS0 Circuit															
	Reconfigured			CLO	PE1BP	23.00										L
	V to P Conversion, Per Customer Request per DS1 Circuit			a. a	55450											
	Reconfigured V to P Conversion, Per Customer Request per DS3 Circuit			CLO	PE1BS	33.00										ł
	Reconfigured			CLO	PE1BE	37.00										
	V to P Conversion, Cable Pairs Assigned to Collo Space per 700			CLU	FEIDE	37.00										
	prs or fraction thereof			CLO	PE1B7	592.00										
	Physical Collocation - Co-Carrier Cross Connects - Fiber Cable			010	I EIDI	002.00										<u> </u>
	Support Structure, per cable, per linear ft.			CLO,UDF	PE1ES	0.001										
	Physical Collocation - Co-Carrier Cross Connects - Copper/Coax															
	Cable Support Structure, per cable, per lin. ft.			CLO, UE3, USL	PE1DS	0.0014										
	Physical Collocation - Co-Carrier Cross Connects - Application															
	Fee, per application			CLO	PE1DT		584.11									
PHYSICAL CO																
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2-				05400	0.074	04.50	00.54				44.00				
	Wire Analog - Res Physical Collocation 2-Wire Cross Connect, Exchange Port 2-			UEPSR	PE1R2	0.074	34.53	32.51				11.90				
	Wire Line Side PBX Trunk - Bus			UEPSP	PE1R2	0.074	34.53	32.51				11.90				
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2-				I LINZ	0.074	34.33	52.51				11.30				
	Wire Voice Grade PBX Trunk - Res			UEPSE	PE1R2	0.074	34.53	32.51				11.90				
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2-															
	Wire Analog - Bus			UEPSB	PE1R2	0.074	34.53	32.51				11.90				
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2-															
	Wire ISDN			UEPSX	PE1R2	0.074	34.53	32.51				11.90				L
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2-															
	Wire ISDN			UEPTX	PE1R2	0.074	34.53	32.51				11.90				<u> </u>
	Physical Collocation 4-Wire Cross Connect, Exchange Port 4- Wire ISDN DS1			UEPEX	PE1R4	0.148	34.54	32.53				11.90				
	OLLOCATION			UEPEA	PEIR4	0.140	34.34	32.33				11.90				
ADDAOLINI O	Adjacent Collocation - Space Charge per Sq. Ft.			CLOAC	PE1JA	0.1635										-
	Adjacent Collocation - Space Charge per Sq. 11. Adjacent Collocation - Electrical Facility Charge per Linear Ft.			CLOAC	PE1JC	5.11					1					
1	Adjacent Collocation - 2-Wire Cross-Connects			CLOAC	PE1P2	0.0213	24.69	23.69	11.77	10.62	1				İ	
l				UEA,UHL,UDL,UCL,							1					
	Adjacent Collocation - 4-Wire Cross-Connects			CLOAC	PE1P4	0.0426	24.88	23.83	12.04	10.80						L
	Adjacent Collocation - DS1 Cross-Connects			USL,CLOAC	PE1P1	1.22	44.24	31.98	12.07	10.91						
	Adjacent Collocation - DS3 Cross-Connects			CLOAC	PE1P3	16.56	41.94	30.52	13.91	11.15						
	Adjacent Collocation - 2-Fiber Cross-Connect			CLOAC	PE1F2	2.81	41.94	30.52	13.91	11.16						I
	Adjacent Collocation - 4-Fiber Cross-Connect		L	CLOAC	PE1F4	5.36	51.30	39.87	18.29	15.54						
	Adjacent Collocation - Application Fee			CLOAC	PE1JB		2,785.00		1.01		1					1

COLLOCAT	ION - Florida												Attach	ment: 4	Exhi	ibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted Manually	Charge -	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge - Manual Sv Order vs.
						Dee	Nonrec	urring	Nonrecurring	J Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Adjacent Collocation - 120V, Single Phase Standby Power Rate per AC Breaker Amp			CLOAC	PE1FB	5.38										
	Adjacent Collocation - 240V, Single Phase Standby Power Rate per AC Breaker Amp			CLOAC	PE1FD	10.77										
	Adjacent Collocation - 120V, Three Phase Standby Power Rate per AC Breaker Amp			CLOAC	PE1FE	16.15										
	Adjacent Collocation - 277V, Three Phase Standby Power Rate per AC Breaker Amp			CLOAC	PE1FG	37.30										
	Adjacent Collocation - Cable Support Structure per Entrance Cable			CLOAC	PE1PM	18.96										
PHYSICAL CO	LLOCATION IN THE REMOTE SITE															
	Physical Collocation in the Remote Site - Application Fee			CLORS	PE1RA		617.91		328.81							
	Cabinet Space in the Remote Site per Bay/ Rack			CLORS	PE1RB	219.49										
	Physical Collocation in the Remote Site - Security Access - Key			CLORS	PE1RD		26.30									
	Physical Collocation in the Remote Site - Space Availability Report per Premises Requested			CLORS	PE1SR		232.69									
	Physical Collocation in the Remote Site - Remote Site CLLI Code Request, per CLLI Code Requested			CLORS	PE1RE		75.41									
	Remote Site DLEC Data (BRSDD), per Compact Disk, per CO			CLORS	PE1RR		233.51									
PHYSICAL CO	LLOCATION IN THE REMOTE SITE - ADJACENT															───
	Remote Site-Adjacent Collocation - AC Power, per breaker amp			CLORS	PE1RS	6.27										
	Remote Site-Adjacent Collocation - Real Estate, per square foot			CLORS	PE1RT	0.134										
	Remote Site-Adjacent Collocation-Application Fee If Security Escort and/or Add'I Engineering Fees become nec			CLORS	PE1RU		755.62	755.62	2							

COLLOCAT	ION - Georgia												Attachr			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec First	urring Add'l	Nonrecurring First	Disconnect Add'l	SOMEC	SOMAN		Rates(\$) SOMAN	SOMAN	SOMAN
						1	11130	Add I	11130	Add I	JOWILO	JOINIAN	JOWAN	JONIAN	JONIAN	JOINAN
PHYSICAL CO																l .
	Physical Collocation - Application Fee - Initial			CLO	PE1BA		3,850.00									i
	Physical Collocation - Application Fee - Subsequent			CLO CLO	PE1CA PE1BL		3,130.00 740.83	3,130.00								I
	Physical Collocation Administrative Only - Application Fee Physical Collocation - Space Preparation Fee Per Square Ft.			CLO	PEIBL		100.00	100.00								i
	Physical Collocation - Space Preparation - Firm Order			OLO	1 2100		100.00	100.00								[
	Processing	1		CLO	PE1SJ		1,187.00									ł
	Physical Collocation - Space Preparation - C.O. Modification per square ft.	I		CLO	PE1SK	2.02										
	Physical Collocation - Space Preparation - Common Systems Modification per square ft Cageless	I		CLO	PE1SL	2.80										
	Physical Collocation - Space Preparation - Common Systems				DE 4011											i
┝──┤──	Modification per Cage		<u> </u>	CLO	PE1SM PE1BD	95.23	0.750.00	0 750 00	├────							┢─────
	Physical Collocation - Cable Installation Physical Collocation - Floor Space per Sq. Ft.			CLO CLO	PE1BD PE1PJ	7.50	2,750.00	2,750.00								J
	Physical Collocation - Floor Space - Zone B per Sq. Ft.			CLO	PE1PK	6.75										
	Physical Collocation - Cable Support Structure			CLO	PE1PM	13.35										[
	Physical Collocation - Power -48V DC Power, per Fused Amp	1		CLO	PE1PL	8.06										i
	Physical Collocation - Power Reduction, Application Fee	I		CLO	PE1PR		398.80									
	Physical Collocation - 120V, Single Phase Standby Power Rate	I		CLO	PE1FB	5.52										ļ
	Physical Collocation - 240V, Single Phase Standby Power Rate	I		CLO	PE1FD	11.05										
	Physical Collocation - 120V, Three Phase Standby Power Rate	I		CLO	PE1FE	16.58										
	Physical Collocation - 277V, Three Phase Standby Power Rate	Т		CLO	PE1FG	38.27										
				UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ, UDL, UNCVX,												
	Physical Collocation - 2-Wire Cross-Connects			UNLDX, UNCNX	PE1P2	0.30	12.60	12.60								
	Physical Collocation - 4-Wire Cross-Connects			CLO, UAL, UDL, UDN, UEA, UHL, UNCVX, UNCDX, UCL	PE1P4	0.50	12.60	12.60								
	Physical Collocation - DS1 Cross-Connects			CLO,UEANL,UEQ,W DS1L,WDS1S, USL, U1TD1, UXTD1, UNC1X, ULDD1, USLEL, UNLD1, UDL	PE1P1	8.00	155.00	27.00								
	Physical Collocation - DST Closs-Connects			UDL CLO, UE3,U1TD3, UXTD3, UXTS1, UNC3X, UNCSX, ULDD3, U1TS1,ULDS1,		8.00	155.00	27.00								
	Physical Collocation - DS3 Cross-Connects			UNLD3, UDL	PE1P3	72.00	155.00	27.00								l
	Physical Collocation - 2-Fiber Cross-Connect			CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3, UDL12, UDF	PE1F2	2.86	52.14	38.72								
				CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3,												
	Physical Collocation - 4-Fiber Cross-Connect			UDL12, UDF	PE1F4	5.08	64.74	51.31								<u> </u>

COLLOCAT	ION - Georgia												Attachr			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add ⁴
						Rec	Nonrec			g Disconnect		0		Rates(\$)		
				<u>a: a</u>	251211		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Physical Collocation - Welded Wire Cage - First 100 Sq. Ft. Physical Collocation - Welded Wire Cage - Add'l 50 Sq. Ft.			CLO CLO	PE1BW PE1CW	161.27 15.82										
	Physical Collocation - Welded Wife Cage - Add 150 Sq. Ft. Physical Collocation - Security System Per Central Office Per	1		CLU	PEICW	15.82										
	Assignable Sq. Ft. Physical Collocation - Security Access System - New Access			CLO	PE1AY	0.0172										
	Card Activation, per Card			CLO	PE1A1	0.0607	46.20	46.20								
	Physical Collocation - Security Access System - New Access Card Deactivation, per Card			CLO	PE1A4		8.72	8.72								
	Physical Collocation-Security Access System-Administrative Change, existing Access Card, per Request, per State, per Card Physical Collocation - Security Access System- Replace Lost or			CLO	PE1AA		15.40	15.40								
	Stolen Card, per Card			CLO	PE1AR		45.02	45.02								
	Physical Collocation - Security Access - Initial Key, per Key			CLO	PE1AK		26.16	26.16		ļ					ļ	
	Physical Collocation - Security Access - Key, Replace Lost or				DEAN		00.10	00.10							1	
┝──	Stolen Key, per Key Physical Collocation - Space Availability Report per premises			CLO CLO	PE1AL PE1SR		26.16 2,148.00	26.16		<u> </u>	───			L	<u> </u>	
		- 1		UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ,CLO,UDL,			2,148.00	2,146.00								
	POT Bay Arrangements prior to 6/1/99 - 2-Wire Cross-Connect, per cross-connect			UNCVX, UNCDX, UNCNX	PE1PE	0.40										
	POT Bay Arrangements prior to 6/1/99 - 4-Wire Cross-Connect, per cross-connect			UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ,CLO, USL, UNCVX, UNCDX		1.20										
	POT Bay Arrangements prior to 6/1/99 - DS1 Cross-Connect, per cross-connect			UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ,CLO,WDS1L,W DS1S, USL, U1TD1, UXTD1, UNC1X, ULDD1, USLEL, UNLD1	PE1PG	1.20										
	POT Bay Arrangements prior to 6/1/99 - DS3 Cross-Connect, per cross-connect			UEANL, UEA, UDN, U DC, UAL, UHL, UCL, U EQ, CLO, UE3, U1TD3, UXTD3, UXTS1, UNC3X, UNCSX, ULDD3, U1TS1, ULDS1, UNLD3, UDL, UDLSX		8.00										
	POT Bay Arrangements prior to 6/1/99 - 2-Fiber Cross-Connect, per cross-connect			UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ,CLO, ULDO3, ULD12, ULD48, U1T03, U112, U1T48, UDLO3, UDL12, UDF		38.79										
	POT Bay Arrangements prior to 6/1/99 - 4-Fiber Cross-Connect, per cross-connect			UDL12, UDF UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ,CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3, UDL12, UDF		52.31										
	Physical Collocation - Request Resend of CFA Information, per		1							1					1	
	CLLI		1	CLO	PE1C9		77.42		1	1		1			1	1

COLLOCAT	ION - Georgia												Attachr			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring					Rates(\$)	-	
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Nonrecurring Collocation Cable Records - VG/DS0 Cable, per			a . a												i
	cable record			CLO	PE1CD		922.38									
	Nonrecurring Collocation Cable Records - VG/DS0 Cable, per each 100 pair			CLO	PE1CO		18.00	18.00								i
	Nonrecurring Collocation Cable Records - DS1, per T1TIE			CLO	PE1C0		8.43	8.43			1				-	i
	Nonrecurring Collocation Cable Records - DS3, per T3TIE		1	CLO	PE1C3		29.49	29.49								
	Nonrecurring Collocation Cable Records - Fiber Cable, per 99			020	1 2100		20.40	20.40								
	fiber records			CLO	PE1CB		278.61	278.61								i
	Physical Collocation - Security Escort - Basic, per Half Hour			CLO,CLORS	PE1BT		41.00	25.00								
	Physical Collocation - Security Escort - Overtime, per Half Hour			CLO,CLORS	PE10T		48.00	30.00								1
						Т	Т									1
	Physical Collocation - Security Escort - Premium, per Half Hour	ļ		CLO,CLORS	PE1PT		55.00	35.00			L					ļ
	V to P Conversion, Per Customer Request-Voice Grade		<u> </u>	CLO	PE1BV	33.00										L
	V to P Conversion, Per Customer Request-DS0			CLO CLO	PE1BO PE1B1	33.00					-					ł
	V to P Conversion, Per Customer Request-DS1 V to P Conversion, Per Customer request-DS3			CLO	PE1B1 PE1B3	52.00 52.00										
	V to P Conversion, Per Customer Request per VG Circuit	-	-	CLO	PEIDS	52.00										i
	Reconfigured			CLO	PE1BR	23.00										i
	V to P Conversion, Per Customer Request per DS0 Circuit		-	010	I E IBR	20.00										
	Reconfigured			CLO	PE1BP	23.00										i
	V to P Conversion, Per Customer Request per DS1 Circuit			020		20.00										1
	Reconfigured			CLO	PE1BS	33.00										i
	V to P Conversion, Per Customer Request per DS3 Circuit															[
	Reconfigured			CLO	PE1BE	37.00										i
	V to P Conversion, Cable Pairs Assigned to Collo Space per 700															Í
	prs or fraction thereof			CLO	PE1B7	592.00										1
	Physical Collocation - Co-Carrier Cross Connects - Fiber Cable															i
	Support Structure, per cable, per linear ft.			CLO,UDF	PE1ES	0.001										
	Physical Collocation - Co-Carrier Cross Connects - Copper/Coax															i
	Cable Support Structure, per cable, per lin. ft.			CLO, UE3, USL	PE1DS	0.0015					-					ł
	Physical Collocation - Co-Carrier Cross Connects - Application Fee, per application			CLO	PE1DT		583.18									i
PHYSICAL CO		-	-	CLU	PEIDI		505.10									i
FHISICAL CO	Physical Collocation 2-Wire Cross Connect, Exchange Port 2-															i
	Wire Analog - Res			UEPSR	PE1R2	0.30	12.60	12.60					18.94	8.42		i
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2-			021011		0.00	12.00	12.00					10.01	0.12		1
	Wire Line Side PBX Trunk - Bus			UEPSP	PE1R2	0.30	12.60	12.60					18.94	8.42		i
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2-															
	Wire Voice Grade PBX Trunk - Res			UEPSE	PE1R2	0.30	12.60	12.60					18.94	8.42		
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2-															1
	Wire Analog - Bus			UEPSB	PE1R2	0.30	12.60	12.60					18.94	8.42		l
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2-		1		05400									a		1
├──-┼───	Wire ISDN	ļ	<u> </u>	UEPSX	PE1R2	0.30	12.60	12.60					18.94	8.42	-	
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2- Wire ISDN		1	UEPTX	PE1R2	0.30	12.60	12.60					18.94	8.42		1
	Wire ISDN Physical Collocation 4-Wire Cross Connect, Exchange Port 4-			UEPIX	PE1R2	0.30	12.60	12.60					18.94	8.42		
	Wire ISDN DS1		1	UEPEX	PE1R4	0.50	12.60	12.60					18.94	8.42		1
ADJACENT CO			1		1 - 1114	0.50	12.00	12.00					10.94	0.42		
	Adjacent Collocation - Space Charge per Sq. Ft.		<u> </u>	CLOAC	PE1JA	0.2542				1	t			1		r
	Adjacent Collocation - Electrical Facility Charge per Linear Ft.	1	1	CLOAC	PE1JC	5.44				ĺ	1	1		ĺ		(
	Adjacent Collocation - 2-Wire Cross-Connects		1	CLOAC	PE1P2	0.598	24.95	23.97	11.80	10.67						(
			1	UEA,UHL,UDL,UCL,	1											1
	Adjacent Collocation - 4-Wire Cross-Connects			CLOAC	PE1P4	0.1196	25.14	24.11	12.15	10.93						
	Adjacent Collocation - DS1 Cross-Connects			USL,CLOAC	PE1P1	1.04	44.19	32.13	11.93	10.81						
	Adjacent Collocation - DS3 Cross-Connects		I	CLOAC	PE1P3	14.12	41.93	30.69	13.71	11.04	ļ					ļ
	Adjacent Collocation - 2-Fiber Cross-Connect		ļ	CLOAC	PE1F2	2.39	41.93	30.69	13.71	11.05	ļ					
<u> </u>	Adjacent Collocation - 4-Fiber Cross-Connect		 	CLOAC	PE1F4	4.57	51.14	39.90	17.96	15.29						I
	Adjacent Collocation - Application Fee			CLOAC	PE1JB		1,555.00									i

COLLOCAT	ION - Georgia											Attach	ment: 4	Exhi	bit: B	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted Manually	Charge -	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						D	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Adjacent Collocation - 120V, Single Phase Standby Power Rate per AC Breaker Amp			CLOAC	PE1FB	5.39										
	Adjacent Collocation - 240V, Single Phase Standby Power Rate per AC Breaker Amp			CLOAC	PE1FD	10.79										
	Adjacent Collocation - 120V, Three Phase Standby Power Rate per AC Breaker Amp			CLOAC	PE1FE	16.18										
	Adjacent Collocation - 277V, Three Phase Standby Power Rate per AC Breaker Amp			CLOAC	PE1FG	38.27										
	Adjacent Collocation - 240V, Three Phase Standby Power Rate per AC Breaker Amp			CLOAC	PEIJD	37.37										
PHYSICAL CO	LLOCATION IN THE REMOTE SITE															
	Physical Collocation in the Remote Site - Application Fee			CLORS	PE1RA		608.18	608.17	323.63	323.63						
	Cabinet Space in the Remote Site per Bay/ Rack			CLORS	PE1RB	224.82										
	Physical Collocation in the Remote Site - Security Access - Key			CLORS	PE1RD		25.88	25.88								
	Physical Collocation in the Remote Site - Space Availability Report per Premises Requested			CLORS	PE1SR		229.02	229.02								
	Physical Collocation in the Remote Site - Remote Site CLLI Code Request, per CLLI Code Requested			CLORS	PE1RE		74.22	74.22								
	Remote Site DLEC Data (BRSDD), per Compact Disk, per CO			CLORS	PE1RR		232.88									
PHYSICAL CO	LLOCATION IN THE REMOTE SITE - ADJACENT															
	Remote Site-Adjacent Collocation - AC Power, per breaker amp			CLORS	PE1RS	6.27										
	Remote Site-Adjacent Collocation - Real Estate, per square foot			CLORS	PE1RT	0.134										
	Remote Site-Adjacent Collocation-Application Fee			CLORS ote site collocation	PE1RU		755.62	755.62			1					

COLLOCAT	ON - Kentucky												Attachi			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring					Rates(\$)	-	
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
PHYSICAL CO																ł
I III OIOAL OO	Physical Collocation - Application Fee - Initial		1	CLO	PE1BA		3,773.54	3,773.54	1.01	1.01						
	Physical Collocation - Application Fee - Subsequent			CLO	PE1CA		3,145.35	3,145.35	1.01	1.01						
	Physical Collocation Administrative Only - Application Fee			CLO	PE1BL		742.12									
	Physical Collocation - Space Preparation - Firm Order Processing			CLO	PE1SJ		1,206.07	1,206.07								
	Physical Collocation - Space Preparation - C.O. Modification per square ft.			CLO	PE1SK	2.32										
	Physical Collocation - Space Preparation - Common Systems															
	Modification per square ft Cageless Physical Collocation - Space Preparation - Common Systems			CLO	PE1SL	3.26										
	Modification per Cage			CLO	PE1SM	110.57	1 200 11		15.10							
┝──┼──	Physical Collocation - Cable Installation		<u> </u>	CLO CLO	PE1BD PE1PJ	7.99	1,729.11		45.16		<u> </u>					
<u> </u>	Physical Collocation - Floor Space per Sq. Ft. Physical Collocation - Cable Support Structure			CLO	PE1PJ PE1PM	7.99										
	Physical Collocation - Cable Support Structure Physical Collocation - Power -48V DC Power, per Fused Amp		-	CLO	PE1PM PE1PL	8.06										
	Physical Collocation - Power Reduction, Application Fee	1		CLO	PE1PR	0.00	399.50									1
	Physical Collocation - 120V, Single Phase Standby Power Rate			CLO	PE1FB	5.44										
	Physical Collocation - 240V, Single Phase Standby Power Rate			CLO	PE1FD	10.88										
	Physical Collocation - 120V, Three Phase Standby Power Rate			CLO	PE1FE	16.32										
	Physical Collocation - 277V, Three Phase Standby Power Rate			CLO	PE1FG	37.68										
	Physical Collocation - 2-Wire Cross-Connects				PE1P2	0.0333	24.68	23.68	12.14	10.95						
	Physical Collocation - 4-Wire Cross-Connects			CLO, UAL, UDL, UDN, UEA, UHL, UNCVX, UNCDX, UCL	PE1P4	0.0665	24.88	23.82	12.77	11.46						
	Physical Collocation - DS1 Cross-Connects			CLO,UEANL,UEQ,W DS1L,WDS1S, USL, U1TD1, UXTD1, UNC1X, ULDD1, USLEL, UNLD1, UDL	PE1P1	1.48	44.23	31.98	12.81	11.57						
				CLO, UE3,U1TD3, UXTD3, UXTS1, UNC3X, UNCSX, ULDD3, U1TS1,ULDS1,												
	Physical Collocation - DS3 Cross-Connects Physical Collocation - 2-Fiber Cross-Connect			UNLD3, UDL CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3, UDL12, UDF	PE1P3 PE1F2	3.75	41.93	30.51	14.75	11.83						
				CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3,												
├──	Physical Collocation - 4-Fiber Cross-Connect		<u> </u>	UDL12, UDF	PE1F4 PE1BW	6.65 184.97	51.29	39.87	19.41	16.49	<u> </u>					
┝──┼──	Physical Collocation - Welded Wire Cage - First 100 Sq. Ft. Physical Collocation - Welded Wire Cage - Add'l 50 Sq. Ft.	-		CLO CLO	PE1BW PE1CW	184.97 18.14					-					<u> </u>

COLLOCAT	ION - Kentucky									-				ment: 4		bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Physical Collocation - Security Access System - Security System per Central Office			CLO	PE1AX	76.10										
	Physical Collocation - Security Access System - New Access			010		70.10										
	Card Activation, per Card			CLO	PE1A1	0.058	55.79	55.79								
	Physical Collocation-Security Access System-Administrative Change, existing Access Card, per Request, per State, per Card Physical Collocation - Security Access System - Replace Lost or			CLO	PE1AA		15.64	15.64								
	Stolen Card, per Card			CLO	PE1AR		45.74	45.74								
	Physical Collocation - Security Access - Initial Key, per Key			CLO	PE1AK		26.29	26.29								
	Physical Collocation - Security Access - Key, Replace Lost or			CLO	PE1AL		26.29	26.29								
	Stolen Key, per Key Physical Collocation - Space Availability Report per premises			CLO	PETAL PE1SR		26.29	26.29								ł
	POT Bay Arrangements prior to 6/1/99 - 2-Wire Cross-Connect, per cross-connect			UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ,CLO,UDL, UNCVX, UNCDX, UNCNX UEANL,UEA,UDN,U	PE1PE	0.113		,								
	POT Bay Arrangements prior to 6/1/99 - 4-Wire Cross-Connect, per cross-connect			DC,UAL,UHL,UCL,U EQ,CLO, USL, UNCVX, UNCDX UEANL,UEA,UDN,U		0.23										
	POT Bay Arrangements prior to 6/1/99 - DS1 Cross-Connect, per cross-connect			DC,UAL,UHL,UCL,U EQ,CLO,WDS1L,W DS1S, USL, U1TD1, UXTD1, UNC1X, ULDD1, USLEL, UNLD1	PE1PG	1.60										
	POT Bay Arrangements prior to 6/1/99 - DS3 Cross-Connect, per cross-connect			UEANL, UEA, UDN, U DC, UAL, UHL, UCL, U EQ, CLO, UE3, U1TD3, UXTD3, UXTS1, UNC3X, UNCSX, ULDD3, U1TS1, ULDS1, UNLD3, UDL, UDLSX	PE1PH	14.23										
	POT Bay Arrangements prior to 6/1/99 - 2-Fiber Cross-Connect, per cross-connect			UEANL, UEA, UDN, U DC, UAL, UHL, UCL, U EQ, CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3, UDL12, UDF	PE1B2	48.57										
	POT Bay Arrangements prior to 6/1/99 - 4-Fiber Cross-Connect, per cross-connect			UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ,CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3, UDL12, UDF	PE1B4	65.50										
	Physical Collocation - Request Resend of CFA Information, per			01.0	05400											1
├──	CLLI Nonrecurring Collocation Cable Records - per request			CLO CLO	PE1C9 PE1CR		77.55 1,524.45	980.01	267.02							
	Nonrecurring Collocation Cable Records - per request Nonrecurring Collocation Cable Records - VG/DS0 Cable, per			010	FEIGR	1	1,024.45	980.01	207.02							<u> </u>
	cable record			CLO	PE1CD		656.37	656.37	379.70							<u> </u>
	Nonrecurring Collocation Cable Records - VG/DS0 Cable, per each 100 pair			CLO	PE1CO		9.65	9.65	11.84	11.84						

COLLOCAT	ION - Kentucky					r					1	1		ment: 4		bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Nonrecurring Collocation Cable Records - DS1, per T1TIE			CLO	PE1C1		4.52	4.52	5.54	5.54						
	Nonrecurring Collocation Cable Records - DS3, per T3TIE			CLO	PE1C3		15.81	15.81	19.39	19.39						
	Nonrecurring Collocation Cable Records - Fiber Cable, per 99			a: a	55405		100.00									
	fiber records			CLO	PE1CB		169.63	169.63	154.85	154.85				-	-	
	Physical Collocation - Security Escort - Basic, per Half Hour			CLO,CLORS	PE1BT	-	33.98	21.53								
	Physical Collocation - Security Escort - Overtime, per Half Hour			CLO,CLORS	PE1OT		44.26	27.81								
	Physical Collocation - Security Escort - Premium, per Half Hour			CLO,CLORS	PE1PT		54.54	34.09								
	V to P Conversion, Per Customer Request-Voice Grade			CLO	PE1BV	33.00	54.54	54.05	<u> </u>					1	1	
	V to P Conversion, Per Customer Request-DS0			CLO	PE1BO	33.00			1		1		1	1	1	1
	V to P Conversion, Per Customer Request-DS1			CLO	PE1B1	52.00			1		1			1	t	1
	V to P Conversion, Per Customer request-DS3			CLO	PE1B3	52.00			1		1			1	1	1
	V to P Conversion, Per Customer Request per VG Circuit															
	Reconfigured			CLO	PE1BR	23.00										
	V to P Conversion, Per Customer Request per DS0 Circuit															
	Reconfigured			CLO	PE1BP	23.00										
	V to P Conversion, Per Customer Request per DS1 Circuit															
	Reconfigured			CLO	PE1BS	33.00										
	V to P Conversion, Per Customer Request per DS3 Circuit Reconfigured			CLO	PE1BE	37.00										
	V to P Conversion, Cable Pairs Assigned to Collo Space per 700				1									1	1	
	prs or fraction thereof			CLO	PE1B7	592.00			ļ ļ		ļ					
	Physical Collocation - Co-Carrier Cross Connects - Fiber Cable															
	Support Structure, per cable, per linear ft.			CLO,UDF	PE1ES	0.0012								-	-	
	Physical Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per cable, per lin. ft.			CLO, UE3, USL	PE1DS	0.0018										
	Physical Collocation - Co-Carrier Cross Connects - Application				DEADT		504.00									
	Fee, per application			CLO	PE1DT		584.20									
PHYSICAL CO	Physical Collocation 2-Wire Cross Connect, Exchange Port 2-					-										
	Wire Analog - Res			UEPSR	PE1R2	0.0333	24.68	23.68	12.14	10.95		7.86				
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2-			UEFSK	PEIRZ	0.0333	24.00	23.00	12.14	10.95		7.00		ł	ł	
	Wire Line Side PBX Trunk - Bus			UEPSP	PE1R2	0.0333	24.68	23.68	12.14	10.95		7.86				
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2-															
	Wire Voice Grade PBX Trunk - Res			UEPSE	PE1R2	0.0333	24.68	23.68	12.14	10.95		7.86		1	1	
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2-			-												
	Wire Analog - Bus			UEPSB	PE1R2	0.0333	24.68	23.68	12.14	10.95		7.86				
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2-															
	Wire ISDN			UEPSX	PE1R2	0.0333	24.68	23.68	12.14	10.95		7.86		ļ	ļ	
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2-													1	1	
L	Wire ISDN		ļ	UEPTX	PE1R2	0.0333	24.68	23.68	12.14	10.95		7.86			ļ	
	Physical Collocation 4-Wire Cross Connect, Exchange Port 4-				DEADA	4.40	11.00	24.00	10.01	44 57		7.00		1	1	
ADJACENT C				UEPEX	PE1R4	1.48	44.23	31.98	12.81	11.57		7.86		<u> </u>	<u> </u>	ł
ADJACENT C	Adjacent Collocation - Space Charge per Sq. Ft.			CLOAC	PE1JA	0.0173			╂────┤					ł	<u> </u>	
<u>├──</u>	Adjacent Collocation - Space Charge per Sq. Ft. Adjacent Collocation - Electrical Facility Charge per Linear Ft.			CLOAC	PE1JA PE1JC	5.35			 					1	1	
	Adjacent Collocation - 2-Wire Cross-Connects			CLOAC	PE1P2	0.0258	24.68	23.68	12.14	10.95				1	1	
				UEA,UHL,UDL,UC		0.0200	200	20.00					1	1	1	
	Adjacent Collocation - 4-Wire Cross-Connects			CLOAC	PE1P4	0.0515	24.88	23.82	12.77	11.46				1	1	
	Adjacent Collocation - DS1 Cross-Connects			USL,CLOAC	PE1P1	1.37	44.23	31.98	12.81	11.57	1		1	1	1	1
İ	Adjacent Collocation - DS3 Cross-Connects			CLOAC	PE1P3	18.61	41.93	30.51	14.75	11.83						
	Adjacent Collocation - 2-Fiber Cross-Connect			CLOAC	PE1F2	3.15	41.93	30.51	14.76	11.84						
	Adjacent Collocation - 4-Fiber Cross-Connect			CLOAC	PE1F4	6.02	51.29	39.87	19.41	16.49						
	Adjacent Collocation - Application Fee			CLOAC	PE1JB		3,165.50		1.01							
	Adjacent Collocation - 120V, Single Phase Standby Power Rate per AC Breaker Amp			CLOAC	PE1FB	5.44										
	Adjacent Collocation - 240V, Single Phase Standby Power Rate per AC Breaker Amp			CLOAC	PE1FD	10.88										

COLLOCAT	ON - Kentucky												Attach	nent: 4	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Submitted Elec	Submitted	Charge -	Charge -	Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
						_	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)	•	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Adjacent Collocation - 120V, Three Phase Standby Power Rate per AC Breaker Amp			CLOAC	PE1FE	16.32										
	Adjacent Collocation - 277V, Three Phase Standby Power Rate per AC Breaker Amp			CLOAC	PE1FG	37.68										
PHYSICAL CO	LLOCATION IN THE REMOTE SITE															
	Physical Collocation in the Remote Site - Application Fee			CLORS	PE1RA		617.78		338.89							
	Cabinet Space in the Remote Site per Bay/ Rack			CLORS	PE1RB	219.67										
	Physical Collocation in the Remote Site - Security Access - Key			CLORS	PE1RD		26.29									
	Physical Collocation in the Remote Site - Space Availability Report per Premises Requested			CLORS	PE1SR		232.64									
	Physical Collocation in the Remote Site - Remote Site CLLI Code Request, per CLLI Code Requested			CLORS	PE1RE		75.40									
	Remote Site DLEC Data (BRSDD), per Compact Disk, per CO			CLORS	PE1RR		233.42									
PHYSICAL CO	LLOCATION IN THE REMOTE SITE - ADJACENT															
	Remote Site-Adjacent Collocation - AC Power, per breaker amp			CLORS	PE1RS	6.27										
	Remote Site-Adjacent Collocation - Real Estate, per square foot			CLORS	PE1RT	0.134										
	Remote Site-Adjacent Collocation-Application Fee			CLORS	PE1RU		755.62	755.62								
NOTE:	If Security Escort and/or Add'I Engineering Fees become nec	essary	for rem	ote site collocation,	, the Parties w	ill negotiate ap	propriate rate	s.								

COLLOCATI	ON - South Carolina			r									Attachr			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
PHYSICAL CO																
	Physical Collocation - Application Fee - Initial			CLO	PE1BA		1,883.67	1,883.67	0.51	0.51						
	Physical Collocation - Application Fee - Subsequent			CLO	PE1CA		1,570.10	1,570.10	0.51	0.51						
	Physical Collocation Administrative Only - Application Fee			CLO	PE1BL		743.66									
	Physical Collocation - Space Preparation - Firm Order Processing			CLO	PE1SJ		602.05	602.05								
	Physical Collocation - Space Preparation - C.O. Modification per square ft.			CLO	PE1SK	2.75										
	Physical Collocation - Space Preparation - Common Systems Modification per square ft Cageless			CLO	PE1SL	3.24										
	Physical Collocation - Space Preparation - Common Systems			CLU	PEISL	3.24										
	Modification per Cage			CLO	PE1SM	110.16										
	Physical Collocation - Cable Installation			CLO	PE1BD		794.22	794.22	22.54	22.54			-			
	Physical Collocation - Floor Space per Sq. Ft.			CLO	PE1PJ	3.95										
	Physical Collocation - Cable Support Structure			CLO	PE1PM	21.33										
	Physical Collocation - Power -48V DC Power, per Fused Amp Physical Collocation - Power Reduction, Application Fee			CLO CLO	PE1PL PE1PR	9.19	400.33									
				CLO		5.07	400.00									
	Physical Collocation - 120V, Single Phase Standby Power Rate				PE1FB	5.67										
	Physical Collocation - 240V, Single Phase Standby Power Rate			CLO	PE1FD	11.36										
	Physical Collocation - 120V, Three Phase Standby Power Rate			CLO	PE1FE	17.03										
	Physical Collocation - 277V, Three Phase Standby Power Rate			CLO	PE1FG	39.33					-					
	Physical Collocation - 2-Wire Cross-Connects			UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ, UDL, UNCVX, UNLDX, UNCNX	PE1P2	0.0341	12.32	11.83	6.04	5.45						
	Physical Collocation - 4-Wire Cross-Connects			CLO, UAL, UDL, UDN, UEA, UHL, UNCVX, UNCDX, UCL	PE1P4	0.0682	12.42	11.90	6.40	5.74						
	Physical Collocation - DS1 Cross-Connects			CLO,UEANL,UEQ,W DS1L,WDS1S, USL, U1TD1, UXTD1, UNC1X, ULDD1, USLEL, UNLD1, UDL	/ PE1P1	1.12	22.08	15.96	6.42	5.80						
	Physical Collocation - DS3 Cross-Connects			CLO, UE3,U1TD3, UXTD3, UXTS1, UNC3X, UNCSX, ULDD3, U1TS1,ULDS1, UNLD3, UDL	PE1P3	14.21	20.94	15.23	7.39	5.93						
	Physical Collocation - DSS Cross-Connects			CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3, UDL12, UDF	PE1P3	2.82	20.94	15.23	7.40	5.93						
				CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3,												
	Physical Collocation - 4-Fiber Cross-Connect		<u> </u>	UDL12, UDF	PE1F4	5.01	25.61	19.90	9.73	8.26	<u> </u>					
	Physical Collocation - Welded Wire Cage - First 100 Sq. Ft. Physical Collocation - Welded Wire Cage - Add'l 50 Sq. Ft.			CLO CLO	PE1BW PE1CW	219.19 21.50										
COLLOCAT	ON - South Carolina												Attachr			bit: B
-----------	---	-------------	------	---	----------------	---------	-------------------	-------------------	--------------	------------------	-------	---	--	--	---	---
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonrec	urring	Nonrecurring	g Disconnect				Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Physical Collocation - Security Access System - Security System per Central Office			CLO	PE1AX	74.72										
┟───┤────	Physical Collocation - Security Access System - New Access			CLU	PETAX	74.72										
	Card Activation, per Card			CLO	PE1A1	0.0601	27.85	27.85								
	Physical Collocation-Security Access System-Administrative Change, existing Access Card, per Request, per State, per Card Physical Collocation - Security Access System - Replace Lost or			CLO	PE1AA		7.81	7.81								
	Stolen Card, per Card			CLO	PE1AR		22.83	22.83								
	Physical Collocation - Security Access - Initial Key, per Key			CLO	PE1AK		13.13	13.13								
	Physical Collocation - Security Access - Key, Replace Lost or			a. a			10.10									
┢───┼────	Stolen Key, per Key Physical Collocation - Space Availability Report per premises			CLO CLO	PE1AL PE1SR	╂─────┤	13.13 1,077.57	13.13 1,077.57								
	POT Bay Arrangements prior to 6/1/99 - 2-Wire Cross-Connect, per cross-connect			UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ,CLO,UDL, UNCVX, UNCDX, UNCNX UEANL,UEA,UDN,U	PE1PE	0.085	1,011.01	1,011.01								
	POT Bay Arrangements prior to 6/1/99 - 4-Wire Cross-Connect, per cross-connect			DEANL, UEA, UDN, U DC, UAL, UHL, UCL, U EQ, CLO, USL, UNCVX, UNCDX UEANL, UEA, UDN, U	PE1PF	0.1701										
	POT Bay Arrangements prior to 6/1/99 - DS1 Cross-Connect, per cross-connect			DC,UAL,UHL,UCL,U EQ,CLO,WDS1L,W DS1S, USL, U1TD1, UXTD1, UNC1X, ULDD1, USLEL, UNLD1	PE1PG	1.20										
	POT Bay Arrangements prior to 6/1/99 - DS3 Cross-Connect, per cross-connect			UEANL, UEA, UDN, U DC, UAL, UHL, UCL, U EQ, CLO, UE3, U1TD3, UXTD3, UXTS1, UNC3X, UNCSX, ULDD3, U1TS1, ULDS1, UNLD3, UDL, UDLSX	PE1PH	10.71										
	POT Bay Arrangements prior to 6/1/99 - 2-Fiber Cross-Connect, per cross-connect			UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ,CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3, UDL12, UDF	PE1B2	36.55										
	POT Bay Arrangements prior to 6/1/99 - 4-Fiber Cross-Connect, per cross-connect			UEANL, UEA, UDN, U DC, UAL, UHL, UCL, U EQ, CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3, UDL12, UDF	PE1B4	49.29										
1	Physical Collocation - Request Resend of CFA Information, per		1													
┢───┤────	CLLI			CLO	PE1C9		77.71	400.00	400.00	400.00						
	Nonrecurring Collocation Cable Records - per request Nonrecurring Collocation Cable Records - VG/DS0 Cable, per cable record			CLO	PE1CR PE1CD		760.98 327.65	489.20 327.65	133.29	133.29 189.54						
	Nonrecurring Collocation Cable Records - VG/DS0 Cable, per each 100 pair			CLO	PE1CO		4.82	4.82	5.91	5.91						

COLLOCAT	ION - South Carolina													ment: 4		bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Nonrecurring Collocation Cable Records - DS1, per T1TIE			CLO	PE1C1		2.26	2.26	2.77	2.77						
	Nonrecurring Collocation Cable Records - DS3, per T3TIE			CLO	PE1C3		7.90	7.90	9.68	9.68						
	Nonrecurring Collocation Cable Records - Fiber Cable, per 99 fiber records			CLO	PE1CB		84.68	84.68	77.30	77.30						
	Physical Collocation - Security Escort - Basic, per Half Hour			CLO,CLORS	PE1CB PE1BT		16.96	10.75	11.30	77.30						
	Filysical Collocation - Security Escont - Basic, per hair riou			CLO, CLOKS	FLIDI		10.90	10.75								
	Physical Collocation - Security Escort - Overtime, per Half Hour			CLO,CLORS	PE1OT		22.10	13.89								
	Physical Collocation - Security Escort - Premium, per Half Hour			CLO.CLORS	PE1PT		27.23	17.02								
	V to P Conversion, Per Customer Request-Voice Grade			CLO	PE1BV	33.00	21.23	17.02	ł – – – – – – – – – – – – – – – – – – –							
	V to P Conversion, Per Customer Request-Voice Grade			CLO	PE1BO	33.00										
	V to P Conversion, Per Customer Request-DS0			CLO	PE1B1	52.00			† †		1		1	1	1	1
	V to P Conversion, Per Customer request-DS3			CLO	PE1B3	52.00			† †					1		
	V to P Conversion, Per Customer Request per VG Circuit				-						1		1	1		1
	Reconfigured			CLO	PE1BR	23.00										
	V to P Conversion, Per Customer Request per DS0 Circuit								İ							
	Reconfigured			CLO	PE1BP	23.00										
	V to P Conversion, Per Customer Request per DS1 Circuit															
	Reconfigured			CLO	PE1BS	33.00										
	V to P Conversion, Per Customer Request per DS3 Circuit Reconfigured			CLO	PE1BE	37.00										
	V to P Conversion, Cable Pairs Assigned to Collo Space per 700															
	prs or fraction thereof Physical Collocation - Co-Carrier Cross Connects - Fiber Cable			CLO	PE1B7	592.00										
	Support Structure, per cable, per linear ft.			CLO,UDF	PE1ES	0.001										
	Physical Collocation - Co-Carrier Cross Connects - Copper/Coax															
┝──┤──	Cable Support Structure, per cable, per lin. ft. Physical Collocation - Co-Carrier Cross Connects - Application			CLO, UE3, USL	PE1DS	0.0015								-		
	Figure 2010cation - Co-Carnel Cross Connects - Application Fee, per application			CLO	PE1DT		584.42							1		
PHYSICAL CO						1	0042		† †				1	1	1	
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2-				1				† †		1			1		1
	Wire Analog - Res			UEPSR	PE1R2	0.0341	12.32	11.83	6.04	5.45		15.69				
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2-															
	Wire Line Side PBX Trunk - Bus			UEPSP	PE1R2	0.0341	12.32	11.83	6.04	5.45		15.69				
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2- Wire Voice Grade PBX Trunk - Res			UEPSE	PE1R2	0.0341	12.32	11.83	6.04	5.45		15.69				
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2-			OLFOL	FLIKZ	0.0341	12.32	11.63	0.04	5.45	1	15.09		 		1
	Wire Analog - Bus			UEPSB	PE1R2	0.0341	12.32	11.83	6.04	5.45		15.69				
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2-				1						1		1	1		1
	Wire ISDN			UEPSX	PE1R2	0.0341	12.32	11.83	6.04	5.45		15.69				
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2-				05400	0.0044	10.02	11.00	0.01			15.00				
┝──┤──	Wire ISDN Physical Collocation 4-Wire Cross Connect, Exchange Port 4-			UEPTX	PE1R2	0.0341	12.32	11.83	6.04	5.45		15.69				
	Physical Collocation 4-Wire Cross Connect, Exchange Port 4- Wire ISDN DS1			UEPEX	PE1R4	1.12	22.08	15.96	6.42	5.80		15.69				
	OLLOCATION			OLFLA	F L 17(4	1.12	22.08	10.90	0.42	5.60	<u> </u>	15.09	ł	ł	1	ł
	Adjacent Collocation - Space Charge per Sq. Ft.		<u> </u>	CLOAC	PE1JA	0.0939					1				1	1
	Adjacent Collocation - Electrical Facility Charge per Linear Ft.			CLOAC	PE1JC	6.40			† †					1		
	Adjacent Collocation - 2-Wire Cross-Connects			CLOAC	PE1P2	0.0264	12.32	11.83	6.04	5.45	1			1		1
				UEA,UHL,UDL,UCL	L,											
	Adjacent Collocation - 4-Wire Cross-Connects			CLOAC	PE1P4	0.0527	12.42	11.90	6.40	5.74						
	Adjacent Collocation - DS1 Cross-Connects			USL,CLOAC	PE1P1	1.03	22.08	15.96	6.42	5.80						
	Adjacent Collocation - DS3 Cross-Connects			CLOAC	PE1P3	14.00	20.94	15.23	7.39	5.93				ļ		
	Adjacent Collocation - 2-Fiber Cross-Connect			CLOAC	PE1F2	2.37	20.94	15.23	7.40	5.93	ļ					
	Adjacent Collocation - 4-Fiber Cross-Connect		L	CLOAC	PE1F4	4.53	25.61	19.90	9.73	8.26				ļ		ļ
┝──┤──	Adjacent Collocation - Application Fee			CLOAC	PE1JB		1,580.20		0.51	0.51	<u> </u>			ļ		
	Adjacent Collocation - 120V, Single Phase Standby Power Rate per AC Breaker Amp			CLOAC	PE1FB	5.67										
	Adjacent Collocation - 240V, Single Phase Standby Power Rate								1					İ		
1	per AC Breaker Amp			CLOAC	PE1FD	11.36										

COLLOCAT	ION - South Carolina												Attach	ment: 4	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Submitted Elec	Submitted	Charge -	Charge -	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						D	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Adjacent Collocation - 120V, Three Phase Standby Power Rate per AC Breaker Amp			CLOAC	PE1FE	17.03										
	Adjacent Collocation - 277V, Three Phase Standby Power Rate per AC Breaker Amp			CLOAC	PE1FG	39.33										
PHYSICAL CO	LLOCATION IN THE REMOTE SITE															
	Physical Collocation in the Remote Site - Application Fee			CLORS	PE1RA		308.38	308.38	168.60	168.60						
	Cabinet Space in the Remote Site per Bay/ Rack			CLORS	PE1RB	246.44										
	Physical Collocation in the Remote Site - Security Access - Key			CLORS	PE1RD		13.13	13.13								
	Physical Collocation in the Remote Site - Space Availability Report per Premises Requested			CLORS	PE1SR		116.13	116.13								
	Physical Collocation in the Remote Site - Remote Site CLLI Code Request, per CLLI Code Requested			CLORS	PE1RE		37.64	37.64								
	Remote Site DLEC Data (BRSDD), per Compact Disk, per CO			CLORS	PE1RR		234.50									
PHYSICAL CO	LLOCATION IN THE REMOTE SITE - ADJACENT															
	Remote Site-Adjacent Collocation - AC Power, per breaker amp			CLORS	PE1RS	6.27										
	Remote Site-Adjacent Collocation - Real Estate, per square foot			CLORS	PE1RT	0.134										
	Remote Site-Adjacent Collocation-Application Fee			CLORS	PE1RU		755.62	755.62								
NOTE:	If Security Escort and/or Add'I Engineering Fees become nec	essary f	for rem	ote site collocation	, the Parties w	vill negotiate ap	opropriate rate	s.								

COLLOCATI	ON - Tennessee		-	n		1							Attachr			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrecurring First	Add'l	Nonrecurring First	Disconnect Add'l	CONFO	SOMAN		Rates(\$) SOMAN	SOMAN	SOMAN
							First	Add I	FIrst	Add I	SOMEC	SOMAN	SOWAN	SOWAN	SOWAN	SOWAN
PHYSICAL COL	LLOCATION															
	Physical Collocation - Application Fee - Initial			CLO	PE1BA		3,767.00	3,767.00								
	Physical Collocation - Application Fee - Subsequent			CLO	PE1CA		3,140.00	3,140.00								
	Physical Collocation Administrative Only - Application Fee			CLO	PE1BL	-	743.25									-
	Physical Collocation - Space Preparation - Firm Order Processing	I		CLO	PE1SJ		1,204.00	1,204.00								
	Physical Collocation - Space Preparation - C.O. Modification per square ft.	I		CLO	PE1SK	2.74										
	Physical Collocation - Space Preparation - Common Systems			CLO	PE1SL	0.05										
	Modification per square ft Cageless Physical Collocation - Space Preparation - Common Systems	1		CLO	PEISL	2.95										
	Modification per Cage	1	1	CLO	PE1SM	100.14										
	Physical Collocation - Cable Installation		1	CLO	PE1BD		1,757.00	1,757.00								
	Physical Collocation - Floor Space per Sq. Ft.			CLO	PE1PJ	6.75										
	Physical Collocation - Cable Support Structure			CLO	PE1PM	19.80										
	Physical Collocation - Power -48V DC Power, per Fused Amp			CLO	PE1PL	8.87										
	Physical Collocation - Power Reduction, Application Fee	I		CLO	PE1PR		400.10									
	Physical Collocation - 120V, Single Phase Standby Power Rate	I		CLO	PE1FB	5.60										
	Physical Collocation - 240V, Single Phase Standby Power Rate	I		CLO	PE1FD	11.22										
	Physical Collocation - 120V, Three Phase Standby Power Rate	I		CLO	PE1FE	16.82										
	Physical Collocation - 277V, Three Phase Standby Power Rate	I		CLO	PE1FG	38.84										
	Physical Collocation - 2-Wire Cross-Connects			UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ, UDL, UNCVX, UNLDX, UNCNX	PE1P2	0.033	33.82	31.92								
	Physical Collocation - 4-Wire Cross-Connects			CLO, UAL, UDL, UDN, UEA, UHL, UNCVX, UNCDX, UCL	PE1P4	0.066	33.94	31.95								
	Physical Collocation - DS1 Cross-Connects			CLO,UEANL,UEQ,W DS1L,WDS1S, USL, U1TD1, UXTD1, UNC1X, ULDD1, USLEL, UNLD1, UDL	/ PE1P1	1.51	53.27	40.16								
	Physical Collocation - DS3 Cross-Connects			CLO, UE3,U1TD3, UXTD3, UXTS1, UNC3X, UNCSX, ULDD3, U1TS1,ULDS1, UNLD3, UDL	PE1P3	19.26	52.37									
	Physical Collocation - DSS Cross-Connects			CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3, UDL12, UDF	PE1P3	19.26	41.56	38.89	12.96	10.34			2.69	2.69	1.56	1.56
				CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3,												
	Physical Collocation - 4-Fiber Cross-Connect		<u> </u>	UDL12, UDF	PE1F4	28.11	50.53	38.78	16.97	14.35		L	2.69	2.69	1.56	1.56
	Physical Collocation - Welded Wire Cage - First 100 Sq. Ft. Physical Collocation - Welded Wire Cage - Add'l 50 Sq. Ft.			CLO CLO	PE1BW PE1CW	218.53 21.44	┝────┤									ļ

COLLOCAT	ION - Tennessee		1		1	r					1	Γ		ment: 4		oit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC		_	RATES(\$)	_	_	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrecurring			g Disconnect				Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
1	Physical Collocation - Security Access System - Security System per Central Office			CLO	PE1AX	55.99										
	Physical Collocation - Security Access System - New Access Card Activation, per Card			CLO	PE1A1	0.059	55.67	55.67								
	Physical Collocation-Security Access System-Administrative Change, existing Access Card, per Request, per State, per Card Physical Collocation - Security Access System - Replace Lost or			CLO	PE1AA		15.61	15.61								
1	Stolen Card, per Card			CLO	PE1AR		45.64	45.64								
/	Physical Collocation - Security Access - Initial Key, per Key			CLO	PE1AK		26.24	26.24	1							
i i	Physical Collocation - Security Access - Key, Replace Lost or															
	Stolen Key, per Key	L		CLO	PE1AL		26.24	26.24								
├─── ┤────	Physical Collocation - Space Availability Report per premises		<u> </u>	CLO UEANL,UEA,UDN,U	PE1SR		2,027.00	2,154.00								
	POT Bay Arrangements prior to 6/1/99 - 2-Wire Cross-Connect,			DC,UAL,UHL,UCL,U EQ,CLO,UDL, UNCVX, UNCDX,												
└── ┟───	per cross-connect				PE1PE	0.40										
	POT Bay Arrangements prior to 6/1/99 - 4-Wire Cross-Connect, per cross-connect			UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ,CLO, USL, UNCVX, UNCDX UEANL,UEA,UDN,U	PE1PF	1.20										
	POT Bay Arrangements prior to 6/1/99 - DS1 Cross-Connect, per cross-connect			DC,UAL,UHL,UCL,U EQ,CLO,WDS1L,W DS1S, USL, U1TD1, UXTD1, UNC1X, ULDD1, USLEL, UNLD1	PE1PG	1.20										
	POT Bay Arrangements prior to 6/1/99 - DS3 Cross-Connect, per cross-connect			UEANL, UEA, UDN, U DC, UAL, UHL, UCL, U EQ, CLO, UE3, U1TD3, UXTD3, UXTS1, UNC3X, UNCSX, ULDD3, U1TS1, ULDS1, UNLD3, UDL, UDLSX	PE1PH	8.00										
	POT Bay Arrangements prior to 6/1/99 - 2-Fiber Cross-Connect, Per Cross-Connect			UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ,CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3, UDL12, UDF	PE1B2	38.79										
	POT Bay Arrangements prior to 6/1/99 - 4-Fiber Cross-Connect, per cross-connect			UEANL, UEA, UDN, U DE, UAL, UHL, UCL, U EQ, CLO, ULDO3, ULD12, ULD48, U1TO3, U1T12, U1T48, UDLO3, UDL12, UDF	PE1B4	52.31										
í Í	Physical Collocation - Request Resend of CFA Information, per															
	CLLI		<u> </u>	CLO	PE1C9		77.67				l					
	Nonrecurring Collocation Cable Records - per request Nonrecurring Collocation Cable Records - VG/DS0 Cable, per cable record			CLO	PE1CR PE1CD		1,711.00 925.06									
	Nonrecurring Collocation Cable Records - VG/DS0 Cable, per each 100 pair			CLO	PE1CD		18.05	18.05								

COLLOCAT	ION - Tennessee												Attachr			bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrecurring			g Disconnect				Rates(\$)		<u> </u>
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Nonrecurring Collocation Cable Records - DS1, per T1TIE			CLO CLO	PE1C1 PE1C3		8.45 29.57	8.45 29.57								ł
	Nonrecurring Collocation Cable Records - DS3, per T3TIE Nonrecurring Collocation Cable Records - Fiber Cable, per 99			CLO	PE1C3		29.57	29.57			-					
	fiber records			CLO	PE1CB		279.42	279.42								
	Physical Collocation - Security Escort - Basic, per Half Hour		1	CLO,CLORS	PE1BT		33.91	21.49								
	Physical Collocation - Security Escort - Overtime, per Half Hour			CLO,CLORS	PE1OT		44.17	27.76								
	Physical Collocation - Security Escort - Premium, per Half Hour			CLO,CLORS	PE1PT		54.42	34.02								ļ
	V to P Conversion, Per Customer Request-Voice Grade			CLO	PE1BV	33.00					-					
	V to P Conversion, Per Customer Request-DS0 V to P Conversion, Per Customer Request-DS1			CLO CLO	PE1BO PE1B1	33.00 52.00										<u> </u>
	V to P Conversion, Per Customer request-DS1		-	CLO	PE1B3	52.00										
	V to P Conversion, Per Customer Request per VG Circuit			010	T E I D S	52.00										t
	Reconfigured			CLO	PE1BR	23.00										
	V to P Conversion, Per Customer Request per DS0 Circuit															
	Reconfigured			CLO	PE1BP	23.00										
	V to P Conversion, Per Customer Request per DS1 Circuit Reconfigured			CLO	PE1BS	33.00										
	V to P Conversion, Per Customer Request per DS3 Circuit Reconfigured			CLO	PE1BE	37.00										
	V to P Conversion, Cable Pairs Assigned to Collo Space per 700 prs or fraction thereof			CLO	PE1B7	592.00										
	Physical Caged Collocation-App Cost(initial & sub)-Planning, per request			CLO	PEIAC	16.16	2,903.66	2,903.66								
	Physical Caged Collocation-Space Prep-Grounding, per location			CLO	PE1BB	4.32										
	Physical Caged Collocation-Space Prep-Power Delivery, per 40 amp Feed			CLO	PE1SN		142.40									
	Physical Caged Collocation-Space Prep-Power Delivery, per 100 amp Feed			CLO	PE1SO		185.72									
	Physical Caged Collocation-Space Prep-Power Delivery, per 200 amp Feed Physical Caged Collocation-Space Enclosure-Cage Preparation,			CLO	PEISP		242.05									
	per first 100 sq. ft. Phycical Caged Collocation-Space Enclosure-Cage			CLO	PE1S1	110.97										
	Preparation2, per add'I 50 sq. ft.			CLO	PE1S5	55.49										
	Physical Caged collocation-Cable Installation-Entrance Fiber Structure, interduct per ft.			CLO	PE1CP	0.0156										
	Phycical Caged Collocation-Cable Installation-Entrance Fiber, per cable			CLO	PE1CQ	2.56	944.27									
	Physical Caged Collocation-Floor Space-Land & Buildings, per sq. ft.			CLO	PE1FS	5.94										
	Physical Caged Collocation-Cable Support Structure-Cable Racking, per entrance cable			CLO	PE1CS	21.47										
	Physical Caged Collocation-Power-Power Construction, per amp DC plant			CLO	PE1PN	3.55										
	Physical Caged Collocation-Power-Power Consumption,per amp AC usage			CLO	PE1PO	2.03										
	Physical Caged Collocation-2-wire Cross Connects-Voice Grade ckts, per ckt.			CLO	PE12C	0.0475	7.68									
	Physical Caged Collocation-4-wire Cross Connects-Voice Grade Ckts, per ckt.			CLO	PE14C	0.0475	7.68									
	Physical Caged Collocation-DS1 Cross Connects-connection to DCS, per ckt.			CLO	PE11S	7.68	41.65									
	Physical Caged Collocation-DS1 Cross Connects-Connection to DSX, per ckt.			CLO	PE11X	0.38	41.65									
	Physical Caged Collocation-DS3 Cross Connects-Connection to DCS, per ckt.			CLO	PE13S	53.96	298.03									

COLLOCA	TION - Tennessee												Attach	ment: 4	Exhi	bit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge -	Incremental Charge -
						Rec	Nonrecurring		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Physical Caged Collocation-DS3 Cross Connects-Connection to DSX, per ckt.			CLO	PE13X	9.32	298.03									
	Physical Caged Collocation-Security Access-Access Cards, per			CLO	FLIJA	9.32	290.03									
	5 Cards			CLO	PE1A2		76.10									
	Physical Collocation - Co-Carrier Cross Connects - Fiber Cable															
	Support Structure, per cable, per linear ft.			CLO,UDF	PE1ES	0.0013										
	Physical Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per cable, per lin. ft.			CLO, UE3, USL	PE1DS	0.0019										
	Physical Collocation - Co-Carrier Cross Connects - Application			010, 010, 001	1 2120	0.0010										
	Fee, per application			CLO	PE1DT		585.09									
PHYSICAL CO	OLLOCATION															
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2-			UEPSR	PE1R2	0.30	19.20	19.20			1		20.35	10.54	13.32	1.40
	Wire Analog - Res Physical Collocation 2-Wire Cross Connect, Exchange Port 2-			ULFOR		0.30	19.20	19.20			+		20.35	10.54	13.32	1.40
	Wire Line Side PBX Trunk - Bus			UEPSP	PE1R2	0.30	19.20	19.20			1		20.35	10.54	13.32	1.40
-	Physical Collocation 2-Wire Cross Connect, Exchange Port 2-															
	Wire Voice Grade PBX Trunk - Res			UEPSE	PE1R2	0.30	19.20	19.20					20.35	10.54	13.32	1.40
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2- Wire Analog - Bus			UEPSB	PE1R2	0.30	19.20	19.20					20.35	10.54	13.32	1.40
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2-			OLF 3D	FLIKZ	0.30	19.20	19.20					20.33	10.54	13.32	1.40
	Wire ISDN			UEPSX	PE1R2	0.30	19.20	19.20					20.35	10.54	13.32	1.40
	Physical Collocation 2-Wire Cross Connect, Exchange Port 2-															
	Wire ISDN			UEPTX	PE1R2	0.30	19.20	19.20					20.35	10.54	13.32	1.40
	Physical Collocation 4-Wire Cross Connect, Exchange Port 4-					0.50	40.00	10.00					20.25	40.54	40.00	1.10
	Wire ISDN DS1			UEPEX	PE1R4	0.50	19.20	19.20					20.35	10.54	13.32	1.40
ADUAGENTIC	Adjacent Collocation - Space Charge per Sq. Ft.			CLOAC	PE1JA	0.0656										
-	Adjacent Collocation - Electrical Facility Charge per Linear Ft.			CLOAC	PE1JC	5.53										
	Adjacent Collocation - 2-Wire Cross-Connects			CLOAC	PE1P2	0.034	11.12	10.18	11.33	10.23			1.77	1.77	1.12	1.12
	Adjacent Collocation - 4-Wire Cross-Connects			UEA,UHL,UDL,UCL, CLOAC	PE1P4	0.33	11.30	10.31	11.62	10.44			1.77	1.77	1.12	1.12
	Adjacent Collocation - 4-Wile Cross-Connects			USL,CLOAC	PE1P4 PE1P1	1.70	28.39	16.88	11.65	10.44			1.77	1.77	1.12	1.12
	Adjacent Collocation - DS3 Cross-Connects			CLOAC	PE1P3	19.03	26.23	15.51	13.40	10.77			1.77	1.77	1.12	1.12
	Adjacent Collocation - 2-Fiber Cross-Connect			CLOAC	PE1F2	3.49	26.23	15.51	13.41	10.78			1.77	1.77	1.12	1.12
	Adjacent Collocation - 4-Fiber Cross-Connect			CLOAC	PE1F4	6.50	29.75	19.02	17.60	14.97			1.77	1.77	1.12	1.12
	Adjacent Collocation - Application Fee Adjacent Collocation - 120V, Single Phase Standby Power Rate			CLOAC	PE1JB		2,973.00		0.9475							
	per AC Breaker Amp			CLOAC	PE1FB	5.81										
	Adjacent Collocation - 240V, Single Phase Standby Power Rate			020/10		0.01										
	per AC Breaker Amp			CLOAC	PE1FD	11.64										
	Adjacent Collocation - 120V, Three Phase Standby Power Rate			01.040		47.45										
	per AC Breaker Amp Adjacent Collocation - 277V, Three Phase Standby Power Rate			CLOAC	PE1FE	17.45										
	per AC Breaker Amp			CLOAC	PE1FG	40.30										
PHYSICAL C	OLLOCATION IN THE REMOTE SITE															
	Physical Collocation in the Remote Site - Application Fee			CLORS	PE1RA		580.20		312.76							
	Cabinet Space in the Remote Site per Bay/ Rack			CLORS	PE1RB	220.41										
	Physical Collocation in the Remote Site - Security Access - Key			CLORS	PE1RD		24.69									
	Physical Collocation in the Remote Site - Space Availability	<u> </u>				-	24.00				1					
	Report per Premises Requested			CLORS	PE1SR		218.49									
	Physical Collocation in the Remote Site - Remote Site CLLI			0,000												
	Code Request, per CLLI Code Requested Remote Site DLEC Data (BRSDD), per Compact Disk, per CO	<u> </u>		CLORS CLORS	PE1RE PE1RR		70.81 234.15									
PHYSICAL	OLLOCATION IN THE REMOTE SITE - ADJACENT			OLONG			234.13				1					
									1		1					
	Remote Site-Adjacent Collocation - AC Power, per breaker amp	1		CLORS	PE1RS	6.27					1					
	Remote Site-Aujacent Conocation - AC Fower, per breaker amp															

COLLOCAT	ION - Tennessee												Attachr	nent: 4	Exhil	oit: B	
											Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental	
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -	
		Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc	
CATEGORY	RATE ELEMENTS	m	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.	
												-	-	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l	
						Dee	Nonrecurring		Nonrecurring	J Disconnect			OSS	Rates(\$)			
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	
	Remote Site-Adjacent Collocation-Application Fee			CLORS	PE1RU		755.62	755.62									
NOTE:	If Security Escort and/or Add'I Engineering Fees become nec	essary f	or remo	ote site collocation,	the Parties v	vill negotiate a	ppropriate rates	S.									

ATTACHMENT 5

ACCESS TO NUMBERS AND NUMBER PORTABILITY

TABLE OF CONTENTS

1.	NON-DISCRIMINATORY ACCESS TO TELEPHONE NUMBERS
	LOCAL SERVICE PROVIDER NUMBER PORTABILITY - PERMANENT DLUTION (LNP)
3.	OPERATIONAL SUPPORT SYSTEM (OSS) RATES4

ACCESS TO NUMBERS AND NUMBER PORTABILITY

1. NON-DISCRIMINATORY ACCESS TO TELEPHONE NUMBERS

- 1.1 During the term of this Agreement, where Knology is utilizing its own switch, Knology shall contact the North American Numbering Plan Administrator, NeuStar, for the assignment of numbering resources. In order to be assigned a Central Office Code, Knology will be required to complete the Central Office Code (NXX) Assignment Request and Confirmation Form (Code Request Form) in accordance with Industry Numbering Committee's Central Office Code (NXX) Assignment Guidelines (INC 95-0407-008).
- 1.2 Where BellSouth provides local switching or resold services to Knology, BellSouth will provide Knology with on-line access to intermediate telephone numbers as defined by applicable FCC rules and regulations on a first come first served basis. Knology acknowledges that such access to numbers shall be in accordance with the appropriate FCC rules and regulations. Knology acknowledges that there may be instances where there is a shortage of telephone numbers in a particular rate center; and in such instances, BellSouth may request that Knology return unused intermediate numbers to BellSouth. Knology shall return unused intermediate numbers to BellSouth upon BellSouth's request. BellSouth shall make all such requests on a nondiscriminatory basis.
- 1.3 BellSouth will allow Knology to designate up to 100 intermediate telephone numbers per rate center for Knology's sole use. Assignment, reservation and use of telephone numbers shall be governed by applicable FCC rules and regulations. Knology acknowledges that there may be instances where there is a shortage of telephone numbers in a particular rate center and BellSouth has the right to limit access to blocks of intermediate telephone numbers. These instances include: 1) where jeopardy status has been declared by the North American Numbering Plan (NANP) for a particular Numbering Plan Area (NPA); or 2) where a rate center has less than six months supply of numbering resources.

2. LOCAL SERVICE PROVIDER NUMBER PORTABILITY - PERMANENT SOLUTION (LNP)

- 2.1 The Parties will offer Number Portability in accordance with rules, regulations and guidelines adopted by the Commission, the FCC and industry fora.
- 2.2 <u>End User Line Charge</u>. Where Knology subscribes to BellSouth's local switching, BellSouth shall bill and Knology shall pay the end user line charge associated with implementing LNP as set forth in BellSouth's FCC Tariff No. 1. This charge is not subject to the resale discount set forth in Attachment 1 of this Agreement.

2.3	To limit service outage, BellSouth and Knology will adhere to the process flows
	and cutover guidelines for porting numbers as outlined in the LNP Reference
	Guide, as amended from time to time. The LNP Reference Guide, incorporated
	herein by reference, is accessible via the Internet at the following site:
	http://www.interconnection.bellsouth.com. All intervals referenced in the LNP
	Reference Guide shall apply to both BellSouth and Knology.

- 2.4 The Parties will set Location Routing Number (LRN) unconditional or 10-digit triggers where applicable. Where triggers are set, the porting Party will remove the ported number at the same time the trigger is removed.
- 2.5 A trigger order is a service order issued in advance of the porting of a number. A trigger order 1) initiates call queries to the AIN SS7 network in advance of the number being ported; and 2) provides for the new service provider to be in control of when a number ports.
- 2.6 Where triggers are not set, the Parties shall coordinate the porting of the number between service providers so as to minimize service interruptions to the end user.
- 2.7 BellSouth and Knology will work cooperatively to implement changes to LNP process flows ordered by the FCC or as recommended by standard industry forums addressing LNP.

3. OPERATIONAL SUPPORT SYSTEM (OSS) RATES

3.1 The terms, conditions and rates for OSS are as set forth in Attachment 2.

Attachment 6

Pre-Ordering, Ordering, Provisioning, Maintenance and Repair

TABLE OF CONTENTS

3.	MISCELLANEOUS	5
2.	ACCESS TO OPERATIONS SUPPORT SYSTEMS	3
1.	QUALITY OF PRE-ORDERING, ORDERING, PROVISIONING, MAINTENANCE AND REPAIR 3	3

PRE-ORDERING, ORDERING, PROVISIONING, MAINTENANCE AND REPAIR

1. QUALITY OF PRE-ORDERING, ORDERING, PROVISIONING, MAINTENANCE AND REPAIR

- 1.1 BellSouth shall provide pre-ordering, ordering, provisioning, and maintenance and repair services to Knology that are equivalent to the pre-ordering, ordering, provisioning, and maintenance and repair services BellSouth provides to itself or any other CLEC where technically feasible. The guidelines for pre-ordering, ordering, provisioning, and maintenance and repair are set forth in the various guides and business rules, as appropriate, and as they are amended from time to time during this Agreement. The guides and business rules are found at http://www.interconnection.bellsouth.com and are incorporated herein by reference.
- 1.2 For purposes of this Agreement, BellSouth's regular working hours for provisioning are defined as follows:

Monday - Friday - 8:00 a.m 5:00 p.m. (Excluding Holidays)							
	(Resale/UNE non-coordinated,						
	coordinated orders and order						
	coordinated-time specific)						
Saturday -	8:00 a.m. – 5:00 p.m. (Excluding Holidays)						
	(Resale/UNE non-coordinated						
	orders)						

- 1.2.1 The above hours represent the hours, either Eastern or Central Time, of the location where the physical work is being performed.
- 1.2.2 To the extent Knology requests provisioning of service to be performed outside BellSouth's regular working hours, or the work so requested requires BellSouth's technicians or Project Manager to work outside of regular working hours, overtime billing charges shall apply. Notwithstanding the foregoing, if such work is performed outside of regular working hours by a BellSouth technician or Project Manager during his or her scheduled shift and BellSouth does not incur any overtime charges in performing the work on behalf of Knology, BellSouth will not assess Knology additional charges beyond the rates and charges specified in this Agreement.

2. ACCESS TO OPERATIONS SUPPORT SYSTEMS

2.1 BellSouth shall provide Knology access to operations support systems ("OSS") functions for pre-ordering, ordering, provisioning, maintenance and repair, and billing. BellSouth shall provide access to the OSS through manual and/or electronic interfaces as described in this Attachment. It is the sole responsibility of

Version 3Q02: 09/06/02

Knology to obtain the technical capability to access and utilize BellSouth's OSS interfaces. Specifications for Knology's access and use of BellSouth's electronic interfaces are set forth at <u>www.interconnection.bellsouth.com</u> and are incorporated herein by reference.

- 2.1.1 Pre-Ordering. In accordance with FCC and Commission rules and orders, BellSouth will provide electronic access to the following pre-ordering functions: service address validation, telephone number selection, service and feature availability, due date information, customer record information and loop makeup information. Access is provided through the Local Exchange Navigation System (LENS) interface and the Telecommunications Access Gateway (TAG) interface. Customer record information includes customer specific information in CRIS and RSAG. Knology shall provide to BellSouth access to customer record information including circuit numbers associated with each telephone number where applicable. Knology shall provide such information within four (4) hours after request via electronic access where available. If electronic access is not available, Knology shall provide to BellSouth paper copies of customer record information including circuit numbers associated with each telephone number where applicable. If BellSouth requests the information before noon, the customer record information shall be provided the same day. If BellSouth requests the information after noon, the customer record information shall be provided by noon the following day.
- 2.1.2 The Parties agree not to view, copy, or otherwise obtain access to the customer record information of any customer without that customer's permission. Knology will obtain access to customer record information only in strict compliance with applicable laws, rules, or regulations of the state in which the service is provided. BellSouth reserves the right to audit Knology's access to customer record information. If a BellSouth audit of Knology's access to customer record information reveals that Knology is accessing customer record information without having obtained the proper End User authorization, BellSouth upon reasonable notice to Knology may take corrective action, including but not limited to suspending or terminating Knology's electronic access to BellSouth's OSS functionality. All such information obtained through an audit shall be deemed Information covered by the Proprietary and Confidential Information section in the General Terms and Conditions of this Agreement.
- 2.1.3 <u>Service Ordering</u>. BellSouth will make available the Electronic Data Interchange (EDI) interface and the TAG ordering interface for the purpose of exchanging order information, including order status and completion notification, for noncomplex and certain complex resale requests and certain network elements. Knology may integrate the EDI interface or the TAG ordering interface with the TAG pre-ordering interface. In addition, BellSouth will provide integrated preordering and ordering capability through the LENS interface for non-complex and certain complex resale service requests and certain network element requests.
- 2.1.4 <u>Maintenance and Repair</u>. Knology may report and monitor service troubles and obtain repair services from BellSouth via electronic interfaces. BellSouth provides

Version 3Q02: 09/06/02

several options for electronic trouble reporting. For exchange services, BellSouth will offer Knology non-discriminatory access to the Trouble Analysis Facilitation Interface (TAFI). In addition, BellSouth will offer an industry standard, machine-to-machine Electronic Communications Trouble Administration (ECTA) Gateway interface. For designed services, BellSouth will provide non-discriminatory trouble reporting via the ECTA Gateway. BellSouth will provide Knology an estimated time to repair, an appointment time or a commitment time, as appropriate, on trouble reports. Requests for trouble repair will be billed in accordance with the provisions of this Attachment. BellSouth and Knology agree to adhere to BellSouth's Operational Understanding, as amended from time to time during this Agreement and as incorporated herein by reference. The Operational Understanding may be accessed via the Internet at http://www.interconnection.bellsouth.com.

- 2.2 <u>Change Management</u>. BellSouth provides a collaborative process for change management of the electronic interfaces through the Change Control Process (CCP). Guidelines for this process are set forth in the CCP document as amended from time to time during this Agreement. The CCP document may be accessed via the Internet at <u>http://www.interconnection.bellsouth.com</u>.
- 2.3 <u>BellSouth's Versioning Policy for Electronic Interfaces.</u> BellSouth's Versioning Policy is part of the Change Control Process (CCP). Pursuant to the CCP, BellSouth will issue new software releases for new industry standards for its EDI and TAG electronic interfaces. The Versioning Policy, including the appropriate notification to Knology, is set forth in the CCP document as amended from time to time during this Agreement. The CCP document may be accessed via the Internet at http://www.interconnection.bellsouth.com.
- 2.4 <u>Rates.</u> Charges for use of OSS shall be as set forth in Attachments 1 and 2 of this Agreement and are incorporated herein by reference.

3. MISCELLANEOUS

- 3.1 <u>Pending Orders</u>. Orders placed in the hold or pending status by Knology will be held for a maximum of thirty (30) days from the date the order is placed on hold. After such time, Knology shall be required to submit a new service request. Incorrect or invalid requests returned to Knology for correction or clarification will be held for thirty (30) days. If Knology does not return a corrected request within thirty (30) days, BellSouth will cancel the request.
- 3.2 <u>Single Point of Contact</u>. Knology will be the single point of contact with BellSouth for ordering activity for network elements and other services used by Knology to provide services to its end users, except that BellSouth may accept a request directly from another CLEC, or BellSouth, acting with authorization of the affected end user. Knology and BellSouth shall each execute a blanket letter of authorization with respect to customer requests so that prior proof of end-user authorization will not be necessary with every request. The Parties shall each be

entitled to adopt their own internal processes for verification of customer authorization for requests, provided, however, that such processes shall comply with applicable state and federal law including, until superseded, the FCC guidelines and orders applicable to Presubscribed Interexchange Carrier (PIC) changes, including Un-PIC. Pursuant to a request from another carrier, BellSouth may disconnect any network element being used by Knology to provide service to that end user and may reuse such network elements or facilities to enable such other carrier to provide service to the end user. BellSouth will notify Knology that such a request has been processed but will not be required to notify Knology in advance of such processing.

- 3.2.1 Neither BellSouth nor Knology shall prevent or delay an end-user from migrating to another carrier because of unpaid bills, denied service, or contract terms.
- 3.2.2 BellSouth shall provide access to customer service records (CSRs), Firm Order Confirmations (FOCs) and Local Service Request rejects within the intervals set forth in Attachment 9 of this Agreement.
- 3.2.3 Knology shall return a FOC to BellSouth within thirty-six (36) hours after Knology's receipt from BellSouth of a valid LSR.
- 3.2.4 Knology shall provide a Reject Response to BellSouth within twenty-four (24) hours after BellSouth's submission of an LSR which is incomplete or incorrectly formatted.
- 3.3 <u>Use of Facilities</u>. When a customer of Knology elects to discontinue service and to transfer service to another local exchange carrier, including BellSouth, BellSouth shall have the right to reuse the facilities provided to Knology by BellSouth. In addition, where BellSouth provides local switching, BellSouth may disconnect and reuse facilities when the facility is in a denied state and BellSouth has received a request to establish new service or transfer of service from a customer or a customer's CLEC at the same address served by the denied facility. BellSouth will notify Knology that such a request has been processed after the disconnect order has been completed.
- 3.4 <u>Contact Numbers</u>. The Parties agree to provide one another with toll-free nationwide (50 states) contact numbers for the purpose of ordering, provisioning and maintenance of services.
- 3.5 <u>Subscription Functions</u>. In cases where BellSouth performs subscription functions for an interexchange carrier ("IXC") (i.e. PIC and LPIC changes via Customer Account Record Exchange (CARE)), BellSouth will provide the affected IXCs with the Operating Company Number (OCN) of the local provider for the purpose of obtaining end user billing account and other end user information required under subscription requirements.
- 3.6 <u>Cancellation Charges</u>. If Knology cancels a request for network elements or other services, any costs incurred by BellSouth in conjunction with the provisioning of Version 3Q02: 09/06/02

that request will be recovered in accordance with BellSouth's Private Line Tariff or BellSouth's FCC No. 1 Tariff, Section 5.4, as applicable. Notwithstanding the foregoing, if Knology places an LSR based upon BellSouth's loop makeup information, and such information is inaccurate resulting in the inability of BellSouth to provision the network elements or services requested in accordance with the transmission characteristics of the network elements or services requested, cancellation charges described in this Section shall not apply. Where Knology places a single LSR for multiple network elements or services based upon loop makeup information, and information as to some, but not all, of the network elements or services is inaccurate, if BellSouth cannot provision the network elements or services that were the subject of the inaccurate loop makeup information, Knology may cancel its request for those network elements or services without incurring cancellation charges as described in this Section. In such instance, should Knology elect to cancel the entire LSR, cancellation charges as described in this Section shall apply to those elements and services that were not the subject of inaccurate loop makeup.

3.7 <u>Service Date Advancement Charges (a.k.a. Expedites)</u>. For Service Date Advancement requests by Knology, Service Date Advancement charges will apply for intervals less than the standard interval as outlined in the BellSouth Product and Services Interval Guide. The charges as outlined in BellSouth's FCC No. 1 Tariff, Section 5, will apply as applicable.

Attachment 7 Page 1

Attachment 7

Billing

TABLE OF CONTENTS

1.	PAYMENT AND BILLING ARRANGEMENTS	3
2.	BILLING DISPUTES	7
3.	RAO HOSTING	8
4.	OPTIONAL DAILY USAGE FILE	11
5.	ACCESS DAILY USAGE FILE	14
Rat	tes	.Exhibit A

BILLING

1. PAYMENT AND BILLING ARRANGEMENTS

The terms and conditions set forth in this Attachment shall apply to all services ordered and provisioned pursuant to this Agreement.

- 1.1 <u>Billing</u>. BellSouth will bill through the Carrier Access Billing System (CABS) and through the Customer Records Information System (CRIS) depending on the particular service(s) provided to Knology under this Agreement. BellSouth will format all bills in CBOS Standard or CLUB/EDI format, depending on the type of service provided. For those services where standards have not yet been developed, BellSouth's billing format will change as necessary when standards are finalized by the applicable industry forum.
- 1.1.1 For any service(s) BellSouth orders and receives from Knology, Knology shall bill BellSouth in CABS format.
- 1.1.2 If either Party requests multiple billing media or additional copies of bills, the billing Party will provide these at a reasonable cost.
- 1.1.3 Any switched access charges associated with interexchange carrier access to the resold local exchange lines will be billed by, and due to BellSouth.
- 1.1.4 BellSouth will submit invoices to Knology each month for resold lines and for all other services for each of Knology's accounts.
- 1.1.5 BellSouth will calculate charges and credits on an individual End User account level, including, if applicable, any charge or credit for usage or usage allowances, but will apply relevant charges and credits to Knology's master accounts. BellSouth will also bill Knology, and Knology will be responsible for and remit to BellSouth, all undisputed charges directly applicable to resold services imposed by any governmental entity, including but not limited to 911 and E911 charges, End Users common line charges, federal subscriber line charges, telecommunications relay charges (TRS), and franchise fees. Bills will not be rendered for any charges which are incurred under this Agreement on or before one (1) year preceding the bill date. However, both Parties recognize that situations exists that would necessitate billing beyond the one year limit as permitted by law. These exceptions include:

- (i) charges connected with jointly provided services where by meet point billing guidelines require either Party to rely on records provided by a third Party.
- (ii) charges incorrectly billed due to error in or omission of customer provided data such as PLU or PIU factors or other ordering data.
- 1.1.6 BellSouth will not perform billing and collection services for Knology as a result of the execution of this Agreement. All requests for billing services should be referred to the appropriate entity or operational group within BellSouth.
- 1.2 <u>Establishing Accounts</u>. After receiving certification as a local exchange carrier from the appropriate regulatory agency, Knology will provide the appropriate BellSouth account manager the necessary documentation to enable BellSouth to establish accounts for Local Interconnection, Network Elements and Other Services, Collocation and/or resold services. Such documentation shall include the Application for Master Account, if applicable, proof of authority to provide telecommunications services, the appropriate Operating Company Number (OCN) assigned by the National Exchange Carriers Association (NECA), Carrier Identification Code (CIC), Group Access Code (GAC), Access Customer Name and Abbreviation (ACNA), as applicable, and a tax exemption certificate, if applicable.
- 1.2.1 <u>Payment Responsibility</u>. Payment of all charges owed by Knology to BellSouth under this Agreement will be the responsibility of Knology. Knology shall make payment to BellSouth for all charges for services billed. Payments made by Knology to BellSouth as payment on account will be credited to Knology's accounts receivable master accounts. BellSouth will not become involved in billing disputes that may arise between Knology and Knology's customer. Payment of all charges or credits owed by BellSouth to Knology will be the responsibility of BellSouth. BellSouth shall make payment to Knology for all charges for services billed. Knology will not become involved in billing disputes that may arise between BellSouth and BellSouth's customer.
- 1.3 <u>Payment Due</u>. Payment for services provided will be due on or before the next bill date (i.e., same date in the following month as the bill date) and is payable in immediately available funds. Payment is considered to have been made when received by BellSouth.
- 1.4 If the payment due date falls on a Sunday or on a Holiday that is observed on a Monday, the payment due date shall be the first non-Holiday day following such Sunday or Holiday. If the payment due date falls on a Saturday or on a Holiday which is observed on Tuesday, Wednesday, Thursday, or Friday, the payment due date shall be the last non-Holiday day preceding such Saturday or Holiday. If

payment is not received by the payment due date, a late payment charge, as set forth in Section 1.6, below, shall apply.

- 1.5 <u>Tax Exemption</u>. Upon BellSouth's receipt of a tax exemption certificate from Knology, the total amount billed to Knology will not include those taxes or fees from which Knology is exempt. Knology will be solely responsible for the computation, tracking, reporting and payment of all taxes and like fees associated with the services provided to the end user of Knology.
- 1.6 Late Payment. If any portion of the payment is received by either Party after the payment due date as set forth preceding, or if any portion of the payment is received by a Party in funds that are not immediately available to either Party, a late payment charge shall be due the receiving Party. The late payment charge shall be the portion of the payment not received by the payment due date multiplied by a late factor and will be applied on a per bill basis. For BellSouth, the late factor shall be as set forth in Section A2 of the General Subscriber Services Tariff, Section B2 of the Private Line Service Tariff or Section E2 of the Intrastate Access Tariff, as appropriate. For Knology, (The late payment shall be) a simple interest rate equal to the lesser of (i) 1 ½ % per month or (ii) the maximum rate permitted by law. In addition to any applicable late payment charges, either Party may be charged a fee for all returned checks as set forth in Section A2 of the General Subscriber state law.
- 1.7 <u>Discontinuing Service to Knology</u>. The procedures for discontinuing service to Knology are as follows:
- 1.7.1 BellSouth reserves the right to suspend or terminate service in the event of prohibited, unlawful or improper use of BellSouth facilities or service, abuse of BellSouth facilities, or any other violation or noncompliance by Knology of the rules and regulations of BellSouth's tariffs; provided, however, that BellSouth shall provide notice of such use, violation or noncompliance to Knology and provide Knology with a reasonable opportunity to cure, except in the event of imminent threat of damage to the network or security thereof, in which case BellSouth may provide notice immediately after or contemporaneously with denial of service.
- 1.7.2 BellSouth reserves the right to suspend or terminate service for nonpayment of undisputed charges after notice and Knology's opportunity to cure in accordance with the provisions hereof. If payment of amounts not subject to a billing dispute, as described in Section 2, is not received by the bill date in the month after the original bill date, BellSouth will provide written notice to Knology that additional applications for service may be refused, that any pending orders for service may not be completed, and/or that access to ordering systems may be suspended if payment is not received by the fifteenth day following the date of the notice. In

addition, BellSouth may, at the same time, provide written notice to the person designated by Knology to receive notices of noncompliance that BellSouth may discontinue the provision of existing services to Knology if payment of undisputed charges is not received by the thirtieth day following the date of the initial notice; provided, however, that in no event shall BellSouth terminate the provision of existing services to Knology for nonpayment of disputed charges during the pendency of any billing dispute in accordance with Section 2.1

- 1.7.3 In the case of such discontinuance, all billed charges, as well as applicable termination charges, shall become due.
- 1.7.4 If BellSouth does not discontinue the provision of the services involved on the date specified in the thirty days notice and Knology's noncompliance continues, nothing contained herein shall preclude BellSouth's right to discontinue the provision of the services to Knology for such noncompliance without further notice.
- 1.7.4.1 Upon discontinuance of service on Knology's account, service to Knology's end users will be denied. BellSouth will reestablish service for Knology upon payment of all past due charges and the appropriate connection fee subject to BellSouth's normal application procedures. Knology is solely responsible for notifying the end user of the proposed service disconnection. If within fifteen (15) days after Knology has been denied and no arrangements to reestablish service have been made consistent with this subsection, Knology's service will be disconnected.
- 1.8 Deposit Policy. Knology shall complete the BellSouth Credit Profile and provide information to BellSouth, if it has not already done so, regarding credit worthiness. Based on the results of the credit analysis, BellSouth, to the extent it has reasonable concerns as to Knology's creditworthiness, reserves the right to secure the account with a suitable form of security deposit. Such security deposit shall take the form of cash or at BellSouth's sole discretion, an Irrevocable Letter of Credit (BellSouth form), or Surety Bond (BellSouth form). Any such security deposit shall in no way release Knology from its obligation to make complete and timely payments of its undisputed invoices. Knology shall pay any applicable deposits prior to the inauguration of service. If, in the sole reasonable opinion of BellSouth, Knology's creditworthiness has deteriorated and/or gross monthly billing has substantially increased beyond the level initially used to determine the level of security deposit, BellSouth reserves the right to request additional security. Interest on a security deposit, if provided in cash, shall accrue and be paid in accordance with the terms in the appropriate BellSouth tariff. Security deposits collected under this Section shall not exceed two months' estimated billing. In the event Knology fails to remit to BellSouth any deposit requested pursuant to this Section, service to Knology may be terminated in accordance with the terms of Section 1.7 of this Attachment, and any security deposits will be applied to Knology's account(s).

- 1.9 Notices. Notwithstanding anything to the contrary in this Agreement, all bills and notices regarding billing matters, including notices relating to security deposits, disconnection of services for nonpayment of charges, and rejection of additional orders from Knology, shall be forwarded to the individual and/or address provided by Knology in establishment of its billing account(s) with BellSouth, or to the individual and/or address subsequently provided by Knology as the contact for billing information. All monthly bills and notices described in this Section shall be forwarded to the same individual and/or address. BellSouth shall send disconnection notices in writing both to the contact person and the individuals designated in the Notices provision of the General Terms and Conditions of this Agreement; provided, however, upon written notice from Knology to BellSouth's billing organization, a final notice of disconnection of services purchased by Knology under this Agreement shall be sent via certified mail to the individual(s) listed in the Notices provision of the General Terms and Conditions of this Agreement at least 30 days before BellSouth takes any action to terminate such services.
- 1.10 <u>Rates.</u> Rates for Optional Daily Usage File (ODUF), Access Daily Usage File (ADUF), and Centralized Message Distribution Service (CMDS) are set out in Exhibit A to this Attachment. If no rate is identified in this Attachment, the rate for the specific service or function will be as set forth in applicable BellSouth tariff or as negotiated by the Parties upon request by either Party.

2. BILLING DISPUTES

- 2.1 Each Party agrees to notify the other Party in writing upon the discovery of a billing dispute. Knology shall report all billing disputes to BellSouth using the Billing Adjustment Request Form (RF 1461) provided by BellSouth. In the event of a billing dispute, the Parties will endeavor to resolve the dispute within sixty (60) calendar days of the notification date. If the Parties are unable within the 60 day period to reach resolution, then the aggrieved Party may pursue dispute resolution in accordance with the General Terms and Conditions of this Agreement.
- 2.2 For purposes of this Section 2, a billing dispute means a reported dispute of a specific amount of money actually billed by either Party. The dispute must be explained by the disputing Party and supported by written documentation, which shows with specificity the basis for disputing charges. By way of example and not by limitation, a billing dispute will not include the refusal to pay all or part of a bill or bills when no written documentation is provided to support the dispute, nor shall a billing dispute include the refusal to pay other amounts owed by the billed Party until the dispute is resolved. Claims by the billed Party for damages of any kind will not be considered a billing dispute for purposes of this Section. If the billing dispute is resolved in favor of the billing Party, the disputing Party will make immediate payment of any of the dispute amount owed to the billing Party or the billing Party shall have the right to pursue normal notification and collection

procedures in Section 1.7. Any credits due to the disputing Party, pursuant to the billing dispute, will be applied to the disputing Party's account by the billing Party immediately upon resolution of the dispute.

- 2.3 Resolution of a dispute is expected to occur at the first level of management resulting in a recommendation for settlement of a dispute and closure of a specific billing period. If the issues are not resolved at the first level of management to a Party's satisfaction, the following resolution procedure will begin:
- 2.3.1 If the dispute is not resolved at the first level of management, the dispute will be escalated to the second level of management for each of the respective Parties for resolution. If the dispute is not resolved to a Party's satisfaction, the dispute will be escalated to the third level of management for each of the respective parties for resolution.
- 2.3.2 If the dispute is not resolved at the 3rd level of management, either Party may seek dispute resolution pursuant to Section 12 of the General Terms and Conditions of this Agreement.

3. RAO HOSTING

- 3.1 RAO Hosting, Calling Card and Third Number Settlement System (CATS) and Non-Intercompany Settlement System (NICS) services provided to Knology by BellSouth will be in accordance with the methods and practices regularly applied by BellSouth to its own operations during the term of this Agreement, including such revisions as may be made from time to time by BellSouth.
- 3.2 Knology shall furnish all relevant information required by BellSouth for the provision of RAO Hosting, CATS and NICS.
- 3.3 Charges or credits, as applicable, will be applied by BellSouth to Knology on a monthly basis in arrears. Amounts due (excluding adjustments) are payable within thirty (30) days of receipt of the billing statement.
- 3.4 Knology must have its own unique hosted RAO code. Where BellSouth is the selected CMDS interfacing host, Knology must request that BellSouth establish a unique hosted RAO code for Knology. Such request shall be in writing to the BellSouth RAO Hosting coordinator and must be submitted at least eight (8) weeks prior to provision of services pursuant to this Section. Services shall commence on a date mutually agreed by the Parties.

- 3.5 BellSouth will receive messages from Knology that are to be processed by BellSouth, another LEC in the BellSouth region or a LEC outside the BellSouth region. Knology shall send all messages to BellSouth no later than sixty (60) days after the message date.
- 3.6 BellSouth will perform invoice sequence checking, standard EMI format editing, and balancing of message data with the EMI trailer record counts on all data received from Knology.
- 3.7 All data received from Knology that is to be processed or billed by another LEC within the BellSouth region will be distributed to that LEC in accordance with the Agreement(s) in effect between BellSouth and the involved LEC.
- 3.8 All data received from Knology that is to be placed on the CMDS network for distribution outside the BellSouth region will be handled in accordance with the agreement(s) in effect between BellSouth and its connecting contractor.
- 3.9 BellSouth will receive messages from the CMDS network that are destined to be processed by Knology and will forward them to Knology on a daily basis for processing.
- 3.10 Transmission of message data between BellSouth and Knology will be via CONNECT:Direct.
- 3.10.1 Data circuits (private line or dial-up) will be required between BellSouth and Knology for the purpose of data transmission. Where a dedicated line is required, Knology will be responsible for ordering (or otherwise providing) the circuit and coordinating the installation with BellSouth or a third party, as applicable. Knology is responsible for any applicable BellSouth or third party charges associated with this line. Equipment required on the BellSouth end to attach the line to the mainframe computer and to transmit data will be negotiated on an individual case basis. Where a dial-up facility is required, dial circuits will be installed in the BellSouth data center by BellSouth and the associated charges assessed to Knology. Additionally, all message toll charges associated with the use of the dial circuit by Knology will be the responsibility of Knology. Associated equipment on the BellSouth end, including a modem, will be negotiated on an individual case basis between the Parties. All equipment, including modems and software, that is required on the Knology end for the purpose of data transmission will be the responsibility of Knology.
- 3.11 All messages and related data exchanged between BellSouth and Knology will be formatted for EMI formatted records and packed between appropriate EMI header and trailer records in accordance with accepted industry standards.

- 3.12 Knology will maintain recorded message detail necessary to recreate files provided to BellSouth for a period of three (3) calendar months beyond the related message dates.
- 3.13 Should it become necessary for Knology to send data to BellSouth more than sixty (60) days past the message date(s), Knology will notify BellSouth in advance of the transmission of the data. BellSouth will work with its connecting contractor and/or Knology, where necessary, to notify all affected LECs.
- 3.14 In the event that data to be exchanged between the two Parties should become lost or destroyed, the Party responsible for delivering the data will make every effort to restore and retransmit such data.
- 3.15 Should an error be detected by the EMI format edits performed by BellSouth on data received from Knology, the entire pack containing the affected data will not be processed by BellSouth. BellSouth will notify Knology of the error. Knology will correct the error(s) and will resend the entire pack to BellSouth for processing. In the event that an out-of-sequence condition occurs on subsequent packs, Knology will resend these packs to BellSouth after the pack containing the error has been successfully reprocessed by BellSouth.
- 3.16 In association with message distribution service, BellSouth will provide Knology with associated intercompany settlements reports (CATS and NICS) as appropriate.
- 3.17 Notwithstanding anything in this Agreement to the contrary, in no case shall either Party be liable to the other for any direct or consequential damages incurred as a result of the obligations set out in this Section 3.
- 3.18 Intercompany Settlements Messages
- 3.18.1 Intercompany Settlements Messages facilitate the settlement of revenues associated with traffic originated from or billed by Knology as a facilities based provider of local exchange telecommunications services outside the BellSouth region. Only traffic that originates in one Bell operating territory and bills in another Bell operating territory is included. Traffic that originates and bills within the same Bell operating territory will be settled on a local basis between Knology and the involved company(ies), unless that company is participating in NICS.
- 3.18.2 Both traffic that originates outside the BellSouth region by Knology and is billed within the BellSouth region, and traffic that originates within the BellSouth region and is billed outside the BellSouth region by Knology, is covered by CATS. Also covered is traffic that either is originated by or billed by Knology, involves a company other than Knology, qualifies for inclusion in the CATS settlement, and is not originated or billed within the BellSouth region (NICS).

- 3.18.3 Once Knology is operating within the BellSouth territory, revenues associated with calls originated and billed within the BellSouth region will be settled via NICS.
- 3.18.4 BellSouth will receive the monthly NICS reports from Telcordia on behalf of Knology. BellSouth will distribute copies of these reports to Knology on a monthly basis.
- 3.18.5 BellSouth will receive the monthly CATS reports from Telcordia on behalf of Knology. BellSouth will distribute copies of these reports to Knology on a monthly basis.
- 3.18.6 BellSouth will collect the revenue earned by Knology from the Bell operating company in whose territory the messages are billed via CATS, less a per message billing and collection fee of five cents (\$0.05), on behalf of Knology. BellSouth will remit the revenue billed by Knology to the Bell operating company in whose territory the messages originated, less a per message billing and collection fee of five cents (\$0.05), on behalf on Knology. These two amounts will be netted together by BellSouth and the resulting charge or credit issued to Knology via a monthly Carrier Access Billing System (CABS) miscellaneous bill.
- 3.18.7 BellSouth will collect the revenue earned by Knology within the BellSouth territory from another CLEC also within the BellSouth territory (NICS) where the messages are billed, less a per message billing and collection fee of five cents (\$0.05), on behalf of Knology. BellSouth will remit the revenue billed by Knology within the BellSouth region to the CLEC also within the BellSouth region, where the messages originated, less a per message billing and collection fee of five cents (\$0.05). These two amounts will be netted together by BellSouth and the resulting charge or credit issued to Knology via a monthly CABS miscellaneous bill.
- 3.18.8 BellSouth and Knology agree that monthly netted amounts of less than fifty dollars (\$50.00) will not be settled.

4. OPTIONAL DAILY USAGE FILE

- 4.1 Upon written request from Knology, BellSouth will provide the Optional Daily Usage File (ODUF) service to Knology pursuant to the terms and conditions set forth in this section.
- 4.2 Knology shall furnish all relevant information required by BellSouth for the provision of the ODUF.
- 4.3 The ODUF feed will contain billable messages that were carried over the BellSouth Network and processed in the BellSouth Billing System, but billed to a Knology customer.
- 4.4 Charges for the ODUF will appear on Knologys' monthly bills. The charges are as set forth in Exhibit A to this Attachment.

- 4.5 The ODUF feed will contain both rated and unrated messages. All messages will be in the standard Alliance for Telecommunications Industry Solutions (ATIS) EMI record format.
- 4.6 Messages that error in the billing system of Knology will be the responsibility of Knology. If, however, Knology should encounter significant volumes of errored messages that prevent processing by Knology within its systems, BellSouth will work with Knology to determine the source of the errors and the appropriate resolution.
- 4.7 The following specifications shall apply to the ODUF feed.
- 4.7.1 ODUF Messages to be Transmitted
- 4.7.1.1 The following messages recorded by BellSouth will be transmitted to Knology:
- 4.7.1.1.1 Message recording for per use/per activation type services (examples: Three -Way Calling, Verify, Interrupt, Call Return, etc.)
- 4.7.1.1.2 Measured billable Local
- 4.7.1.1.3 Directory Assistance messages
- 4.7.1.1.4 IntraLATA Toll
- 4.7.1.1.5 WATS and 800 Service
- 4.7.1.1.6 N11
- 4.7.1.1.7 Information Service Provider Messages
- 4.7.1.1.8 Operator Services Messages
- 4.7.1.1.9 Operator Services Message Attempted Calls (Network Element only)
- 4.7.1.1.10 Credit/Cancel Records
- 4.7.1.1.11 Usage for Voice Mail Message Service
- 4.7.1.2 Rated Incollects (messages BellSouth receives from other revenue accounting offices) can also be on ODUF. Rated Incollects will be intermingled with BellSouth recorded rated and unrated usage. Rated Incollects will not be packed separately.
- 4.7.1.3 BellSouth will perform duplicate record checks on records processed to ODUF. Any duplicate messages detected will be deleted and not sent to Knology.

- 4.7.1.4 In the event that Knology detects a duplicate on ODUF they receive from BellSouth, Knology will drop the duplicate message and will not return the duplicate to BellSouth.
- 4.7.2 ODUF Physical File Characteristics
- 4.7.2.1 ODUF will be distributed to Knology via CONNECT:Direct or another mutually agreed medium. The ODUF feed will be a variable block format (2476) with a Logical Record Link (LRECL) of 2472. The data on the ODUF feed will be in a non-compacted EMI format (175 byte format plus modules). It will be created on a daily basis Monday through Friday except holidays. Details such as dataset name and delivery schedule will be addressed during negotiations of the distribution medium. There will be a maximum of one dataset per workday per OCN.
- 4.7.2.2 Data circuits (private line or dial-up) will be required between BellSouth and Knology for the purpose of data transmission as set forth in Section 3.10.1 above.
- 4.7.3 ODUF Packing Specifications
- 4.7.3.1 A pack will contain a minimum of one message record or a maximum of 99,999 message records plus a pack header record and a pack trailer record. One transmission can contain a maximum of 99 packs and a minimum of one pack.
- 4.7.3.2 The OCN, From RAO, and Invoice Number will control the invoice sequencing. The From RAO will be used to identify to Knology which BellSouth RAO that is sending the message. BellSouth and Knology will use the invoice sequencing to control data exchange. BellSouth will be notified of sequence failures identified by Knology and resend the data as appropriate.

The data will be packed using ATIS EMI records.

- 4.7.4 ODUF Pack Rejection
- 4.7.4.1 Knology will notify BellSouth within one business day of rejected packs (via the mutually agreed medium). Packs could be rejected because of pack sequencing discrepancies or a critical edit failure on the Pack Header or Pack Trailer records (i.e. out-of-balance condition on grand totals, invalid data populated). Standard ATIS EMI error codes will be used. Knology will not be required to return the actual rejected data to BellSouth. Rejected packs will be corrected and retransmitted to Knology by BellSouth.
- 4.7.5 ODUF Control Data
- 4.7.5.1 Knology will send one confirmation record per pack that is received from BellSouth. This confirmation record will indicate Knology's receipt of the pack and acceptance or rejection of the pack. Pack Status Code(s) will be populated using standard ATIS EMI error codes for packs that were rejected by Knology for reasons stated in the above section.

4.7.6 ODUF Testing

4.7.6.1 Upon request from Knology, BellSouth shall send ODUF test files to Knology. The Parties agree to review and discuss the ODUF content and/or format. For testing of usage results, BellSouth shall request that Knology set up a production (live) file. The live test may consist of Knology's employees making test calls for the types of services Knology requests on ODUF. These test calls are logged by Knology, and the logs are provided to BellSouth. These logs will be used to verify the files. Testing will be completed within 30 calendar days from the date on which the initial test file was sent.

5. ACCESS DAILY USAGE FILE

- 5.1 Upon written request from Knology, BellSouth will provide the Access Daily Usage File (ADUF) service to Knology pursuant to the terms and conditions set forth in this section.
- 5.2 Knology shall furnish all relevant information required by BellSouth for the provision of ADUF.
- 5.3 ADUF will contain access messages associated with a port that Knology has purchased from BellSouth.
- 5.4 Charges for ADUF will appear on Knology's monthly bills. The charges are as set forth in Exhibit A to this Attachment. All messages will be in the standard ATIS EMI record format.
- 5.5 Messages that error in the billing system of Knology will be the responsibility of Knology. If, however, Knology should encounter significant volumes of errored messages that prevent processing by Knology within its systems, BellSouth will work with Knology to determine the source of the errors and the appropriate resolution.
- 5.6 ADUF Messages To Be Transmitted
- 5.6.1 The following messages recorded by BellSouth will be transmitted to Knology:
- 5.6.1.1 Recorded originating and terminating interstate and intrastate access records associated with a port.
- 5.6.1.2 Recorded terminating access records for undetermined jurisdiction access records associated with a port.
- 5.6.2 BellSouth will perform duplicate record checks on records processed to ADUF. Any duplicate messages detected will be dropped and not sent to Knology.

- 5.6.3 In the event that Knology detects a duplicate on ADUF they receive from BellSouth, Knology will drop the duplicate message and will not return the duplicate to BellSouth.
- 5.6.4 ADUF Physical File Characteristics
- 5.6.4.1 ADUF will be distributed to Knology via CONNECT:Direct or another mutually agreed medium. The ADUF feed will be a fixed block format (2476) with an LRECL of 2472. The data on the ADUF feed will be in a non-compacted EMI format (210 byte). It will be created on a daily basis Monday through Friday except holidays. Details such as dataset name and delivery schedule will be addressed during negotiations of the distribution medium. There will be a maximum of one dataset per workday per OCN.
- 5.6.4.2 Data circuits (private line or dial-up) will be required between BellSouth and Knology for the purpose of data transmission as set forth in Section 3.10.1 above.
- 5.6.5 ADUF Packing Specifications
- 5.6.5.1 A pack will contain a minimum of one message record or a maximum of 99,999 message records plus a pack header record and a pack trailer record. One transmission can contain a maximum of 99 packs and a minimum of one pack.
- 5.6.5.2 The OCN, From RAO, and Invoice Number will control the invoice sequencing. The From RAO will be used to identify to Knology which BellSouth RAO is sending the message. BellSouth and Knology will use the invoice sequencing to control data exchange. BellSouth will be notified of sequence failures identified by Knology and resend the data as appropriate.

The data will be packed using ATIS EMI records.

- 5.6.6 ADUF Pack Rejection
- 5.6.6.1 Knology will notify BellSouth within one business day of rejected packs (via the mutually agreed medium). Packs could be rejected because of pack sequencing discrepancies or a critical edit failure on the Pack Header or Pack Trailer records (i.e. out-of-balance condition on grand totals, invalid data populated). Standard ATIS EMI error codes will be used. Knology will not be required to return the actual rejected data to BellSouth. Rejected packs will be corrected and retransmitted to Knology by BellSouth.
- 5.6.7 ADUF Control Data
- 5.6.7.1 Knology will send one confirmation record per pack that is received from BellSouth. This confirmation record will indicate Knology's receipt of the pack and acceptance or rejection of the pack. Pack Status Code(s) will be populated using standard ATIS EMI error codes for packs that were rejected by Knology for reasons stated in the above section.

5.6.8 ADUF Testing

5.6.8.1 Upon request from Knology, BellSouth shall send a test file of generic data to Knology via Connect:Direct or Text File via E-Mail. The Parties agree to review and discuss the test file's content and/or format.

ODUF/ADUF/EODUF/CMDS - Alabama											Attach	ment: 7	Exhibit: A			
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC		RATES(\$)		Submitted Elec	Submitted	Charge -	Charge -	Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l		
						Rec	Nonrecurring Nonrecurring Disconnect				OSS Rates(\$)					
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
ODUF/ADUF/O																
	SS DAILY USAGE FILE (ADUF)															
	ADUF: Message Processing, per message				N/A	0.007037										
	ADUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.000113										
OPTIO	NAL DAILY USAGE FILE (ODUF)															
	ODUF: Recording, per message				N/A	0.000011										
	ODUF: Message Processing, per message				N/A	0.004101										
	ODUF: Message Processing, per Magnetic Tape provisioned				N/A	42.67										
	ODUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.000094										
CENTR	ALIZED MESSAGE DISTRIBUTION SERVICE (CMDS)															
	CMDS: Message Processing, per message				N/A	0.004										
	CMDS: Data Transmission (CONNECT:DIRECT), per message				N/A	0.001										
	ICED OPTIONAL DAILY USAGE FILE (EODUF)				N1/A	0.00								ļ		
	EODUF: Message Processing, per message	L			N/A	0.22					<u> </u>					L
Notes:	If no rate is identified in the contract, the rate for the specific	service	e or fun	ction will be as set	orth in appli	cable BellSout	h tariff or as n	egotiated by t	he Parties upor	n request by e	ther Party.					
ODUF/ADUF	/EODUF/CMDS - Florida												Attach	ment: 7	Exhi	bit: A
-------------	---	-------------	------	----------------------	------	------------	-------	-----------	--------------	--------------	-------------------	-----------	----------	--	-------------------------------------	----------
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Submitted Elec	Submitted	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge - Manual Svc Order vs.	Charge -
						Rec	Nonre	curring	Nonrecurring	g Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
ODUF/ADUF/O																
	SS DAILY USAGE FILE (ADUF)															
	ADUF: Message Processing, per message				N/A	0.014391										
	ADUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00012973										
	NAL DAILY USAGE FILE (ODUF)															
	ODUF: Recording, per message				N/A	0.0000071										
	ODUF: Message Processing, per message				N/A	0.006835										
	ODUF: Message Processing, per Magnetic Tape provisioned				N/A	48.96										
	ODUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.00010811										
CENTR	RALIZED MESSAGE DISTRIBUTION SERVICE (CMDS)															
	CMDS: Message Processing, per message				N/A	0.004										
	CMDS: Data Transmission (CONNECT:DIRECT), per message				N/A	0.001										
					N1/A	0.000400										
	EODUF: Message Processing, per message			ction will be as set	N/A	0.229109					1					

ODUF/ADUF	/EODUF/CMDS - Georgia												Attach	ment: 7	Exhi	bit: A
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Submitted Elec	Submitted	Charge -	Charge -	Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonre	curring	Nonrecurring	g Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
ODUF/ADUF/O																
	SS DAILY USAGE FILE (ADUF)															
	ADUF: Message Processing, per message				N/A	0.0136327										
	ADUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.0000434										
OPTIO	NAL DAILY USAGE FILE (ODUF)															
	ODUF: Recording, per message				N/A	0.0001275										
	ODUF: Message Processing, per message				N/A	0.0082548										
	ODUF: Message Processing, per Magnetic Tape provisioned				N/A	28.85										
	ODUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.0000434										
CENTR	ALIZED MESSAGE DISTRIBUTION SERVICE (CMDS)															
	CMDS: Message Processing, per message				N/A	0.004										
	CMDS: Data Transmission (CONNECT:DIRECT), per message				N/A	0.001										
	ICED OPTIONAL DAILY USAGE FILE (EODUF)		<u> </u>		N1/A	0.0004555										
	EODUF: Message Processing, per message	L	Ļ		N/A	0.0034555		L			<u> </u>					
Notes:	If no rate is identified in the contract, the rate for the specific	service	e or tun	ction will be as set	orth in appli	cable BellSout	n tariff or as n	egotiated by t	ne Parties upor	n request by ei	ther Party.					

Image: Image:	ODUF/ADUF	/EODUF/CMDS - Kentucky												Attach	ment: 7	Exhi	bit: A
N/A N/A 0.0001/36 SOMAN Soman Soman <th< td=""><td>CATEGORY</td><td>RATE ELEMENTS</td><td></td><td>Zone</td><td>BCS</td><td>USOC</td><td></td><td></td><td>RATES(\$)</td><td></td><td></td><td>Submitted Elec</td><td>Submitted Manually</td><td>Charge - Manual Svc Order vs. Electronic-</td><td>Charge - Manual Svc Order vs. Electronic-</td><td>Charge - Manual Svc Order vs. Electronic-</td><td>Charge -</td></th<>	CATEGORY	RATE ELEMENTS		Zone	BCS	USOC			RATES(\$)			Submitted Elec	Submitted Manually	Charge - Manual Svc Order vs. Electronic-	Charge - Manual Svc Order vs. Electronic-	Charge - Manual Svc Order vs. Electronic-	Charge -
Image: state of the state							Bee	Nonre	curring	Nonrecurring	g Disconnect			OSS	Rates(\$)		
ACCESS DAILY USAGE FILE (ADUF) Image: Constraint of the stage processing, per message Image: Co							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
ACCESS DAILY USAGE FILE (ADUF) Image: Constraint of the stage processing, per message Image: Co																	
ADUF: Message Processing, per message N/A 0.001857 Image: Constraint of Constraints of Constrai																	
ADUF: Data Transmission (CONNECT:DIRECT), per message N/A 0.0001245 Image: Constraint of the state																	
OPTIONAL DAILY USAGE FILE (ODUF) Image: Constraint of the state		ADUF: Message Processing, per message				N/A	0.001857										
ODUF: Recording, per message N/A 0.0000136 Image: Constant of the stage						N/A	0.0001245										
ODUF: Message Processing, per message N/A 0.002506 Image: Constraint of the provision of the provisi																	
ODUF: Message Processing, per Magnetic Tape provisioned N/A 35.90 Image: Constraint of the constra																	
ODUF: Data Transmission (CONNECT:DIRECT), per message N/A 0.00010372 CENTRALIZED MESSAGE DISTRIBUTION SERVICE (CMDS) N/A 0.004 CMDS: Message Processing, per message N/A 0.004		ODUF: Message Processing, per message				N/A	0.002506										
CENTRALIZED MESSAGE DISTRIBUTION SERVICE (CMDS) N/A 0.004 CMDS: Message Processing, per message N/A 0.004		ODUF: Message Processing, per Magnetic Tape provisioned				N/A	35.90										
CMDS: Message Processing, per message N/A 0.004						N/A	0.00010372										
	CENTR	RALIZED MESSAGE DISTRIBUTION SERVICE (CMDS)															
		CMDS: Message Processing, per message				N/A	0.004										
		CMDS: Data Transmission (CONNECT:DIRECT), per message				N/A	0.001										
ENHANCED OPTIONAL DAILY USAGE FILE (EODUF)						N1/A	0.005000										ł
EODUF: Message Processing, per message N/A 0.235889 Output Outp			I	L,						<u> </u>		<u> </u>					Ļ

	RATE ELEMENTS	Interi m	Zone	BCS	USOC						Svc Order Submitted			Incremental Charge -	Incremental Charge -	Incremental
ACCESS DAILY					0300			RATES(\$)					Manual Svc Order vs.		Manual Svc Order vs.	Charge - Manual Svc Order vs. Electronic- Disc Add'l
ACCESS DAILY						Rec	Nonre	curring	Nonrecurring	Disconnect			OSS	Rates(\$)		
ACCESS DAILY						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
ACCESS DAILY																
	MDS															
ADUF: N	(USAGE FILE (ADUF)															
	Message Processing, per message				N/A	0.008061										
	Data Transmission (CONNECT:DIRECT), per message				N/A	0.00013036										
	LY USAGE FILE (ODUF)															
	Recording, per message				N/A	0.0000216										
ODUF: N	Message Processing, per message				N/A	0.004704										
ODUF: N	Message Processing, per Magnetic Tape provisioned				N/A	48.87										
	Data Transmission (CONNECT:DIRECT), per message				N/A	0.00010863										
CENTRALIZED M	MESSAGE DISTRIBUTION SERVICE (CMDS)															
CMDS: N	Message Processing, per message				N/A	0.004										
	Data Transmission (CONNECT:DIRECT), per message				N/A	0.001										
	TIONAL DAILY USAGE FILE (EODUF)		1		1											1
Notes: If no rate	: Message Processing, per message		1		N/A	0.258301										

ODUF/ADUF	/EODUF/CMDS - Tennessee												Attach	ment: 7	Exhi	bit: A
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Submitted Elec	Submitted	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrecurring		Nonrecurring	g Disconnect			OSS	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
ODUF/ADUF/O																
	SS DAILY USAGE FILE (ADUF)															
	ADUF: Message Processing, per message				N/A	0.004										
	ADUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.001										
	NAL DAILY USAGE FILE (ODUF)															
	ODUF: Recording, per message				N/A	0.0000044										
	ODUF: Message Processing, per message				N/A	0.0027366										
	ODUF: Message Processing, per Magnetic Tape provisioned				N/A	52.75										
	ODUF: Data Transmission (CONNECT:DIRECT), per message				N/A	0.0000339										
CENTR	ALIZED MESSAGE DISTRIBUTION SERVICE (CMDS)															
	CMDS: Message Processing, per message				N/A	0.004										
	CMDS: Data Transmission (CONNECT:DIRECT), per message				N/A	0.001										
					N1/A	0.004										
	EODUF: Message Processing, per message	L	L		N/A	0.004			L							───
Notes:	If no rate is identified in the contract, the rate for the specific	service	e or fun	ction will be as set	forth in appli	cable BellSout	n tariff or as n	egotiated by t	ne Parties upor	n request by e	ther Party.					

Attachment 8

Rights-of-Way, Conduits and Pole Attachments

Rights-of-Way, Conduits and Pole Attachments

Knology and BellSouth have entered into a stand-alone Agreement pursuant to 47 U.S.C. § 224, for nondiscriminatory access to any Rights-of-Way, Conduits, or Poles owned or controlled by BellSouth.

ATTACHMENT 9

PERFORMANCE MEASUREMENTS

PERFORMANCE MEASUREMENTS

Upon a particular Commission's issuance of an Order pertaining to Performance Measurements in a proceeding expressly applicable to all CLECs generally, BellSouth shall implement in that state such Performance Measurements as of the date specified by the Commission. Performance Measurements that have been Ordered in a particular state can currently be accessed via the internet at https://pmap.bellsouth.com. At the request of the Tennessee Regulatory Authority (TRA), the following Regional Service Quality Measurements (SQM) plan is being included as the performance measurements currently in place for the state of Tennessee. At such time that the TRA issues an Order pertaining to Performance Measurements, such Performance Measurements shall supersede the Regional SQM contained in the Agreement.

BellSouth Service Quality Measurement Plan (SQM)

Region Performance Metrics

Measurement Descriptions Version 0.06

Issue Date: June 4, 2002

Introduction

The BellSouth Service Quality Measurement Plan (SQM) describes in detail the measurements produced to evaluate the quality of service delivered to BellSouth's customers both wholesale and retail. The SQM was developed to respond to the requirements of the Communications Act of 1996 Section 251 (96 Act) which required BellSouth to provide non-discriminatory access to Competitive Local Exchange Carriers (CLEC)¹ and its Retail Customers. The reports produced by the SQM provide regulators, CLECs and BellSouth the information necessary to monitor the delivery of non-discriminatory access.

This plan results from the many divergent forces evolving from the 96 Act. The 96 Act, the Georgia Public Service Commission (GPSC) Order (Docket 7892-U 12/30/97), LCUG 1-7.0, the FCC's NPRM (CC Docket 98-56 RM9101 04/17/98), the Louisiana Public Service Commission (LPSC) Order (Docket U-22252 Subdocket C 04/19/98), numerous arbitration cases, LPSC sponsored collaborative workshops (10/98-02/00), and proceedings in Alabama, Mississippi, and North Carolina have and continue to influence the SQM.

The SQM and the reports flowing from it must change to reflect the dynamic requirements of the industry. New measurements are added as new products, systems, and processes are developed and fielded. New products and services are added as the markets for them develop and the processes stabilize. The measurements are also changed to reflect changes in systems, correct errors, and respond to both 3rd Party audit requirements and Commission requirements.

This document is intended for use by someone with knowledge of telecommunications industry, information technologies and a functional knowledge of the subject areas covered by the BellSouth Performance Measurements and the reports that flow from them.

Once it is approved, the most current copy of this document can be found on the web at URL: <u>https://pmap.bellsouth.com</u> in the Documentation Downloads folder.

Report Publication Dates

Each month, preliminary SQM reports will be posted to BellSouth's SQM web site (https://www.pmap.bellsouth.com) by 8:00 A.M. EST on the 21st day of each month or the first business day after the 21st. Final validated SQM reports will be posted by 8:00 A.M. on the last day of the month. Reports not posted by this time will be considered late for SEEM payment purposes. SEEM reports will posted on the 15th of the following month. Payments due will also be paid on the 15th of the following month. For instance: May data will be posted in preliminary SQM reports on June 21. Final validated SQM reports will be posted on the last day of June. Final validated SEEM reports will be posted and payments mailed on July 15th. In the event the 15th falls on a weekend or holiday, reports and payments will be posted/made the next business day.

1

iv

Alternative Local Exchange Companies (ALEC) and Competing Local Providers (CLP) are referred to as Competitive Local Exchange Carriers (CLEC) in this document.

Report Delivery Methods

CLEC SQM and SEEM reports will be considered delivered when posted to the web site. Commissions will be given access to the web site. In addition, a copy of the Monthly State Summary reports will be filed with the appropriate Commissions as soon as possible after the last day of each month.

Document Number: RGN-V005-122101

v

Contents

Section 1: Operations Support Systems (OSS)	1-1
OSS-1: Average Response Time and Response Interval (Pre-Ordering/ Ordering)	
OSS-2: Interface Availability (Pre-Ordering/Ordering)	
OSS-3: Interface Availability (Maintenance & Repair)	1-7
OSS-4: Response Interval (Maintenance & Repair)	1-9
PO-1: Loop Makeup - Response Time – Manual	
PO-2: Loop Make Up - Response Time - Electronic	1-13
Section 2: Ordering	2-1
O-1: Acknowledgement Message Timeliness	
O-2: Acknowledgement Message Completeness	
O-3: Percent Flow-Through Service Requests (Summary)	
O-4: Percent Flow-Through Service Requests (Detail)	
O-5: Flow-Through Error Analysis	
O-6: CLEC LSR Information	
LSR Flow Through Matrix	
O-7: Percent Rejected Service Requests	
O-8: Reject Interval	
O-9: Firm Order Confirmation Timeliness	
O-10: Service Inquiry with LSR Firm Order Confirmation (FOC) Response Time Manual	
O-11: Firm Order Confirmation and Reject Response Completeness	
O-12: Speed of Answer in Ordering Center	
O-12: Speed of Answer in Ordering Center O-13: LNP-Percent Rejected Service Requests	
O-14: LNP-Reject Interval Distribution & Average Reject Interval	
O-15: LNP-Firm Order Confirmation Timeliness Interval Distribution & Firm Order Confirmat	
Average Interval	
Section 3: Provisioning	
P-1: Mean Held Order Interval & Distribution Intervals	
P-2: Average Jeopardy Notice Interval & Percentage of Orders Given Jeopardy Notices	
P-3: Percent Missed Installation Appointments	
P-4: Average Completion Interval (OCI) & Order Completion Interval Distribution	
P-5: Average Completion Notice Interval	
P-6: % Completions/Attempts without Notice or < 24 hours Notice	
P-7: Coordinated Customer Conversions Interval.	
P-7A: Coordinated Customer Conversions – Hot Cut Timeliness% Within Interval and Average	
Interval.	
P-7B: Coordinated Customer Conversions – Average Recovery Time	
P-7C: Hot Cut Conversions - % Provisioning Troubles Received Within 7 days of a completed	
Service Order	
P-8: Cooperative Acceptance Testing - % of xDSL Loops Tested	
P-9: % Provisioning Troubles within 30 days of Service Order Completion	
P-10: Total Service Order Cycle Time (TSOCT)	
P-11: Service Order Accuracy	
P-12: LNP-Percent Missed Installation Appointments	
P-13: LNP-Average Disconnect Timeliness Interval & Disconnect Timeliness Interval Distribu	tion

vi

P-14: LNP-Total Service Order Cycle Time (TSOCT)	3-36
Section 4: Section 4: Maintenance & Repair	4-1
M&R-1: Missed Repair Appointments	4-1
M&R-2: Customer Trouble Report Rate	4-3
M&R-3: Maintenance Average Duration	4-5
M&R-4: Percent Repeat Troubles within 30 Days	
M&R-5: Out of Service (OOS) > 24 Hours	
M&R-6: Average Answer Time – Repair Centers	
M&R-7: Mean Time To Notify CLEC of Network Outages	
Section 5: Billing	
B-1: Invoice Accuracy	
B2: Mean Time to Deliver Invoices	
B3: Usage Data Delivery Accuracy	
B4: Usage Data Delivery Completeness	
B5: Usage Data Delivery Timeliness	
B6: Mean Time to Deliver Usage	
B0: Weah Time to Deriver Osage B7: Recurring Charge Completeness	
B8: Non-Recurring Charge Completeness	
Section 6: Operator Services And Directory Assistance	
OS-1: Speed to Answer Performance/Average Speed to Answer - Toll	
OS-2: Speed to Answer Performance/Percent Answered with "X" Seconds - Toll	6-2
DA-1: Speed to Answer Performance/Average Speed to Answer - Directory Assistance	· · ·
DA-2: Speed to Answer Performance/Percent Answered within "X" Seconds - Director	ry Assistance
	ry Assistance
DA-2: Speed to Answer Performance/Percent Answered within "X" Seconds - Director (DA)	ry Assistance
DA-2: Speed to Answer Performance/Percent Answered within "X" Seconds - Director (DA)	ry Assistance 6-4 7-1
 DA-2: Speed to Answer Performance/Percent Answered within "X" Seconds - Director (DA) Section 7: Database Update Information. D-1: Average Database Update Interval 	ry Assistance 6-4 7-1 7-1
 DA-2: Speed to Answer Performance/Percent Answered within "X" Seconds - Director (DA) Section 7: Database Update Information. D-1: Average Database Update Interval D-2: Percent Database Update Accuracy. 	ry Assistance 6-4 7-1 7-1 7-3
 DA-2: Speed to Answer Performance/Percent Answered within "X" Seconds - Director (DA) Section 7: Database Update Information. D-1: Average Database Update Interval D-2: Percent Database Update Accuracy. D-3: Percent NXXs and LRNs Loaded by the LERG Effective Date. 	ry Assistance 6-4 7-1 7-1 7-3 7-5
 DA-2: Speed to Answer Performance/Percent Answered within "X" Seconds - Director (DA) Section 7: Database Update Information. D-1: Average Database Update Interval. D-2: Percent Database Update Accuracy. D-3: Percent NXXs and LRNs Loaded by the LERG Effective Date. Section 8: E911 	ry Assistance 6-4 7-1 7-3 7-5 8-1
 DA-2: Speed to Answer Performance/Percent Answered within "X" Seconds - Director (DA) Section 7: Database Update Information. D-1: Average Database Update Interval D-2: Percent Database Update Accuracy. D-3: Percent NXXs and LRNs Loaded by the LERG Effective Date. Section 8: E911 E-1: Timeliness. 	ry Assistance 6-4 7-1 7-3 7-5 8-1
 DA-2: Speed to Answer Performance/Percent Answered within "X" Seconds - Director (DA) Section 7: Database Update Information. D-1: Average Database Update Interval. D-2: Percent Database Update Accuracy. D-3: Percent NXXs and LRNs Loaded by the LERG Effective Date. Section 8: E911. E-1: Timeliness. E-2: Accuracy. 	ry Assistance
 DA-2: Speed to Answer Performance/Percent Answered within "X" Seconds - Director (DA) Section 7: Database Update Information. D-1: Average Database Update Interval D-2: Percent Database Update Accuracy. D-3: Percent NXXs and LRNs Loaded by the LERG Effective Date. Section 8: E911 E-1: Timeliness. 	ry Assistance
 DA-2: Speed to Answer Performance/Percent Answered within "X" Seconds - Director (DA) Section 7: Database Update Information. D-1: Average Database Update Interval. D-2: Percent Database Update Accuracy. D-3: Percent NXXs and LRNs Loaded by the LERG Effective Date. Section 8: E911. E-1: Timeliness. E-2: Accuracy. 	ry Assistance
 DA-2: Speed to Answer Performance/Percent Answered within "X" Seconds - Director (DA) Section 7: Database Update Information. D-1: Average Database Update Interval. D-2: Percent Database Update Accuracy. D-3: Percent NXXs and LRNs Loaded by the LERG Effective Date. Section 8: E911 E-1: Timeliness. E-2: Accuracy. E-3: Mean Interval. Section 9: Trunk Group Performance. 	ry Assistance 6-4 7-1 7-3 7-5 8-1 8-1 8-2 8-3 8-1
 DA-2: Speed to Answer Performance/Percent Answered within "X" Seconds - Director (DA) Section 7: Database Update Information. D-1: Average Database Update Interval. D-2: Percent Database Update Accuracy. D-3: Percent NXXs and LRNs Loaded by the LERG Effective Date. Section 8: E911. E-1: Timeliness. E-2: Accuracy. E-3: Mean Interval. 	ry Assistance 6-4 7-1 7-3 7-5 7-5 8-1 8-1 8-2 8-3 8-3 8-3
 DA-2: Speed to Answer Performance/Percent Answered within "X" Seconds - Director (DA) Section 7: Database Update Information. D-1: Average Database Update Interval D-2: Percent Database Update Accuracy. D-3: Percent NXXs and LRNs Loaded by the LERG Effective Date Section 8: E911 E-1: Timeliness. E-2: Accuracy. E-3: Mean Interval. Section 9: Trunk Group Performance. TGP-1: Trunk Group Performance-Aggregate TGP-2: Trunk Group Performance-CLEC Specific . 	ry Assistance 6-4 7-1 7-3 7-5 7-5 8-1 8-1 8-2 8-3 8-3 9-1 9-3
 DA-2: Speed to Answer Performance/Percent Answered within "X" Seconds - Director (DA) Section 7: Database Update Information. D-1: Average Database Update Interval D-2: Percent Database Update Accuracy. D-3: Percent NXXs and LRNs Loaded by the LERG Effective Date. Section 8: E911 E-1: Timeliness. E-2: Accuracy. E-3: Mean Interval Section 9: Trunk Group Performance TGP-1: Trunk Group Performance-Aggregate TGP-2: Trunk Group Performance-CLEC Specific 	ry Assistance 6-4 7-1 7-3 7-5 7-5 8-1 8-1 8-2 8-3 9-1 9-1 9-3 9-1 9-3
 DA-2: Speed to Answer Performance/Percent Answered within "X" Seconds - Director (DA) Section 7: Database Update Information. D-1: Average Database Update Interval D-2: Percent Database Update Accuracy. D-3: Percent NXXs and LRNs Loaded by the LERG Effective Date. Section 8: E911. E-1: Timeliness. E-2: Accuracy. E-3: Mean Interval. Section 9: Trunk Group Performance TGP-1: Trunk Group Performance-Aggregate	ry Assistance 6-4 7-1 7-1 7-3 7-5 8-1 8-1 8-1 8-2 8-3 9-1 9-1 9-3 9-1 9-3 9-1 9-3
 DA-2: Speed to Answer Performance/Percent Answered within "X" Seconds - Director (DA) Section 7: Database Update Information	ry Assistance 6-4 7-1 7-3 7-5 8-1 8-1 8-2 8-3 9-1 9-1 9-3 9-1 9-3 10-1 10-1
 DA-2: Speed to Answer Performance/Percent Answered within "X" Seconds - Director (DA) Section 7: Database Update Information. D-1: Average Database Update Interval. D-2: Percent Database Update Accuracy. D-3: Percent NXXs and LRNs Loaded by the LERG Effective Date. Section 8: E911. E-1: Timeliness. E-2: Accuracy. E-3: Mean Interval. Section 9: Trunk Group Performance. TGP-1: Trunk Group Performance-Aggregate	ry Assistance 6-4 7-1 7-3 7-5 8-1 8-1 8-1 8-2 8-3 9-1 9-1 9-3 9-3 10-1 10-2 10-3
 DA-2: Speed to Answer Performance/Percent Answered within "X" Seconds - Director (DA) Section 7: Database Update Information. D-1: Average Database Update Interval. D-2: Percent Database Update Accuracy. D-3: Percent NXXs and LRNs Loaded by the LERG Effective Date Section 8: E911 E-1: Timeliness. E-2: Accuracy. E-3: Mean Interval. Section 9: Trunk Group Performance. TGP-1: Trunk Group Performance-Aggregate TGP-2: Trunk Group Performance-CLEC Specific Section 10: Collocation Average Response Time. C-2: Collocation Average Arrangement Time C-3: Collocation Percent of Due Dates Missed. Section 11: Change Management.	ry Assistance 6-4 7-1 7-1 7-3 7-5 8-1 8-1 8-1 8-2 8-3 9-1 9-1 9-3 9-3 10-1 10-2 11-4
 DA-2: Speed to Answer Performance/Percent Answered within "X" Seconds - Director (DA) Section 7: Database Update Information. D-1: Average Database Update Interval. D-2: Percent Database Update Accuracy. D-3: Percent NXXs and LRNs Loaded by the LERG Effective Date. Section 8: E911. E-1: Timeliness. E-2: Accuracy. E-3: Mean Interval. Section 9: Trunk Group Performance. TGP-1: Trunk Group Performance-Aggregate	ry Assistance 6-4 7-1 7-1 7-3 7-5 8-1 8-1 8-1 8-1 8-1 8-1 8-1 8-1 8-1

vii

CM-3: Timeliness of Documents Associated with Change CM-4: Change Management Documentation Average Delay Days CM-5: Notification of CLEC Interface Outages	11-7
Section 12: Bona Fide / New Business Request Process BFR-1: Percentage of BFR/NBR Requests Processed Within 30 Business Days BFR-2: Percentage of Quotes Provided for Authorized BFR/NBR Requests Processed With	12-1 12-1 in X
(10/30/60) Business Days Appendix A: Reporting Scope A-1: Standard Service Groupings A-2: Standard Service Order Activities	1
Appendix B: Glossary of Acronyms and Terms Appendix C: BellSouth Audit Policy	1

Section 1: Operations Support Systems (OSS)

OSS-1: Average Response Time and Response Interval (Pre-Ordering/ Ordering)

Definition

Average response time and response intervals are the average times and number of requests responded to within certain intervals for accessing legacy data associated with appointment scheduling, service & feature availability, address verification, request for Telephone numbers (TNs), and Customer Service Records (CSRs).

Exclusions

None

Business Rules

The average response time for retrieving pre-order/order information from a given legacy system is determined by summing the response times for all requests submitted to the legacy systems during the reporting period and dividing by the total number of legacy system requests for that month.

The response interval starts when the client application (LENS or TAG for CLECs and RNS or ROS for BellSouth) submits a request to the legacy system and ends when the appropriate response is returned to the client application. The number of accesses to the legacy systems during the reporting period which take less than 2.3 seconds, the number of accesses which take more than 6 seconds, and the number which are less than or equal to 6.3 seconds are also captured.

Calculation

Response Time = (a - b)

- a = Date & Time of Legacy Response
- b = Date & Time of Legacy Request

Average Response Time = c / d

- c = Sum of Response Times
- d = Number of Legacy Requests During the Reporting Period

Report Structure

- Not CLEC Specific
- Not Product/Service Specific
- Regional Level

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Report Month
 Legacy Contract (per reporting dimension) 	 Legacy Contract (per reporting dimension)
Response Interval	Response Interval
Regional Scope	Regional Scope

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
• RSAG - Address (Regional Street Address Guide-	
Address) - stores street address information used to	
validate customer addresses. CLECs and BellSouth query	
this legacy system.	
• RSAG – TN (Regional Street Address Guide-Telephone	
number) - contains information about facilities available	
and telephone numbers working at a given address.	

CLECs and BellSouth query this legacy system.	
• ATLAS (Application for Telephone Number Load	
Administration and Selection) – acts as a warehouse for	
storing telephone numbers that are available for	
assignment by the system. It enables CLECs and	
BellSouth service reps to select and reserve telephone	
numbers. CLECs and BellSouth query this legacy system.	
COFFI (Central Office Feature File Interface) – stores	
information about product and service offerings and	
availability. CLECs query this legacy system.	
• DSAP (DOE Support Application) – provides due date	
information. CLECs and BellSouth query this legacy	
system.	
• HAL/CRIS (Hands-Off Assignment Logic/Customer	
Record Information System) – a system used to access the	
Business Office Customer Record Information System	
(BOCRIS). It allows BellSouth servers, including LENS,	
access to legacy systems. CLECs query this legacy	
system.	
• P/SIMS (Product/Services Inventory Management	
system) – provides information on capacity, tariffs,	
inventory and service availability. CLECs query this	
legacy system.	
• OASIS (Obtain Available Services Information Systems)	
– Information on feature and rate availability. BellSouth	
queries this legacy system.	

Table 1: Legacy System Access Times For RNS

System	Contract	Data	< 2.3 sec.	> 6 sec.	<= 6.3 sec.	Avg. Sec.	# of Calls
RSAG	RSAG-TN	Address	х	Х	Х	Х	Х
RSAG	RSAG-ADDR	Address	Х	х	Х	х	Х
ATLAS	ATLAS-TN	TN	Х	х	Х	х	Х
DSAP	DSAP	Schedule	х	Х	Х	Х	Х
CRIS	CRSACCTS	CSR	Х	х	Х	х	Х
OASIS	OASISCAR	Feature/Service	Х	Х	Х	х	Х
OASIS	OASISLPC	Feature/Service	Х	Х	Х	Х	Х
OASIS	OASISMTN	Feature/Service	Х	х	Х	х	Х
OASIS	OASISBIG	Feature/Service	х	Х	Х	Х	Х

Table 2: Legacy System Access Times For R0S

System	Contract	Data	< 2.3 sec.	> 6 sec.	<= 6.3 sec.	Avg. sec.	# of Calls
RSAG	RSAG-TN	Address	х	х	х	х	х
RSAG	RSAG-ADDR	Address	Х	Х	Х	Х	х
ATLAS	ATLAS-TN	TN	х	х	х	х	х
DSAP	DSAP	Schedule	Х	х	Х	Х	Х
CRIS	CRSOCSR	CSR	Х	Х	Х	Х	х
OASIS	OASISBIG	Feature/Service	X	Х	Х	х	X

System	Contract	Data	< 2.3 sec.	> 6 sec.	<6.3 sec.	Avg. sec.	# of Calls
RSAG	RSAG-TN	Address	Х	Х	Х	Х	Х
RSAG	RSAG-ADDR	Address	Х	Х	Х	Х	Х
ATLAS	ATLAS-TN	TN	Х	х	Х	х	Х
DSAP	DSAP	Schedule	Х	Х	Х	Х	Х
HAL	HAL/CRIS	CSR	Х	х	Х	х	х
COFFI	COFFI/USOC	Feature/Service	Х	х	Х	х	Х
P/SIMS	PSIMS/ORB	Feature/Service	Х	Х	Х	Х	Х

Table 3: Legacy System Access Times For LENS

Table 4: Legacy System Access Times For TAG

System	Contract	Data	< 2.3 sec.	> 6 sec.	<6.3 sec.	Avg. sec.	# of Calls
RSAG	RSAG-TN	Address	Х	Х	Х	Х	Х
RSAG	RSAG-ADDR	Address	х	Х	Х	Х	Х
ATLAS	ATLAS-TN	TN	Х	Х	Х	Х	Х
ATLAS	ATLAS-MLH	TN	Х	Х	Х	Х	Х
ATLAS	ATLAS-DID	TN	Х	Х	Х	Х	Х
DSAP	DSAP	Schedule	Х	Х	Х	Х	Х
CRIS	CRSECSRL	CSR	Х	Х	Х	Х	Х
CRIS	CRSECSR	CSR	X	Х	X	Х	X

SEEM Measure

SEEM Measure			
Yes	Tier I		
	Tier II	Х	

Note: CLEC specific data is not available in this measure. Queries of this sort do not have company specific signatures.

SEEM Disaggregation - Analog/Benchmark

SEEM Disaggregation	SEEM Analog/Benchmark
• RSAG – Address (Regional Street Address Guide-	• Percent Response Received within 6.3 seconds: > 95%
Address) - stores street address information used to	• Parity + 2 seconds
validate customer addresses. CLECs and BellSouth query	
this legacy system.	
• RSAG – TN (Regional Street Address Guide-Telephone	
number) – contains information about facilities available	
and telephone numbers working at a given address.	
CLECs and BellSouth query this legacy system.	
• ATLAS (Application for Telephone Number Load	
Administration and Selection) – acts as a warehouse for	
storing telephone numbers that are available for	
assignment by the system. It enables CLECs and	
BellSouth service reps to select and reserve telephone	
numbers. CLECs and BellSouth query this legacy system.	
• COFFI (Central Office Feature File Interface) – stores	
information about product and service offerings and	
availability. CLECs query this legacy system.	
• DSAP (DOE Support Application) – provides due date	
information. CLECs and BellSouth query this legacy	
system.	
• HAL/CRIS (Hands-Off Assignment Logic/Customer	
Record Information System) – a system used to access the	

Business Office Customer Record Information System	
(BOCRIS). It allows BellSouth servers, including LENS,	
access to legacy systems. CLECs query this legacy	
system.	
P/SIMS (Product/Services Inventory Management	
system) – provides information on capacity, tariffs,	
inventory and service availability. CLECs query this	
legacy system.	
• OASIS (Obtain Available Services Information Systems)	
- Information on feature and rate availability. BellSouth	
queries this legacy system.	

SEEM OSS Legacy Systems

System	BellSouth	CLEC		
	Telephone Number/Address			
RSAG-ADDR	RNS, ROS	TAG, LENS		
RSAG-TN	RNS, ROS	TAG, LENS		
ATLAS	RNS,ROS	TAG. LENS		
	Appointment Scheduling			
DSAP	RNS, ROS	TAG, LENS		
CSR Data				
CRSACCTS	RNS			
CRSOCSR	ROS			
HAL/CRIS		LENS		
CRSECSRL		TAG		
CRSECSR		TAG		
	Service/Feature Availability			
OASISBIG	RNS, ROS			
PSIMS/ORB		LENS		

OSS-2: Interface Availability (Pre-Ordering/Ordering)

Definition

Percent of time applications are functionally available as compared to scheduled availability. Calculations are based upon availability of applications and interfacing applications utilized by CLECs for pre-ordering and ordering. "Functional Availability" is defined as the number of hours in the reporting period that the applications/interfaces are available to users. "Scheduled Availability" is defined as the number of hours in the reporting period that the applications/interfaces are scheduled to be available.

Scheduled availability is posted on the Interconnection web site: (www.interconnection.bellsouth.com/oss/oss_hour.html)

Exclusions

- CLEC-impacting troubles caused by factors outside of BellSouth's purview, e.g., troubles in customer equipment, troubles in networks owned by telecommunications companies other than BellSouth, etc.
- Degraded service, e.g., slow response time, loss of non-critical functionality, etc.

Business Rules

This measurement captures the functional availability of applications/interfaces as a percentage of scheduled availability for the same systems. Only full outages are included in the calculations for this measure. Full outages are defined as occurrences of either of the following:

- Application/interfacing application is down or totally inoperative.
- Application is totally inoperative for customers attempting to access or use the application. This includes transport outages when they may be directly associated with a specific application.

Comparison to an internal benchmark provides a vehicle for determining whether or not CLECs and retail BST entities are given comparable opportunities for use of pre-ordering and ordering systems.

Calculation

Interface Availability (Pre-Ordering/Ordering) = (a / b) X 100

- a = Functional Availability
- b = Scheduled Availability

Report Structure

- Not CLEC Specific
- Not Product/Service Specific
- Regional Level

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Report Month
• Legacy Contract Type (per reporting dimension)	 Legacy Contract Type (per reporting dimension)
Regional Scope	Regional Scope
Hours of Downtime	Hours of Downtime

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
Regional Level	• >= 99.5%

OSS Interface Availability

Application	Applicable to	% Availability
EDI	CLEC	X
TAG	CLEC	X
LENS	CLEC	х
LEO	CLEC	х
LESOG	CLEC	х
LNP Gateway	CLEC	х
COG	CLEC	Under Development
SOG	CLEC	Under Development
DOM	CLEC	Under Development
DOE	CLEC/BellSouth	Х
SONGS	CLEC/BellSouth	X
ATLAS/COFFI	CLEC/BellSouth	Х
BOCRIS	CLEC/BellSouth	X
DSAP	CLEC/BellSouth	Х
RSAG	CLEC/BellSouth	Х
SOCS	CLEC/BellSouth	Х
CRIS	CLEC/BellSouth	Х

SEEM Measure

SEEM Measure			
Yes	Tier I		
	Tier II	X	

SEEM Disaggregation - Analog/Benchmark

SEEM Disaggregation	SEEM Analog/Benchmark
Regional Level	• >= 99.5%

SEEM OSS Interface Availability

Application	Applicable to	% Availability
EDI	CLEC	x
HAL	CLEC	x
LENS	CLEC	x
LEO Mainframe	CLEC	x
LESOG	CLEC	x
PSIMS	CLEC	X
TAG	CLEC	x

OSS-3: Interface Availability (Maintenance & Repair)

Definition

Percent of time applications are functionally available as compared to scheduled availability. Calculations are based upon availability of applications and interfacing applications utilized by CLECs for maintenance and repair. "Functional Availability" is defined as the number of hours in the reporting period that the applications/interfaces are available to users. "Scheduled Availability" is defined as the number of hours in the reporting period that the applications/interfaces are scheduled to be available.

Scheduled availability is posted on the Interconnection web site: (www.interconnection.bellsouth.com/oss/oss hour.html)

Exclusions

- CLEC-impacting troubles caused by factors outside of BellSouth's purview, e.g., troubles in customer equipment, troubles in networks owned by telecommunications companies other than BellSouth, etc.
- Degraded service, e.g., slow response time, loss of non-critical functionality, etc.

Business Rules

This measurement captures the functional availability of applications/interfaces as a percentage of scheduled availability for the same systems. Only full outages are included in the calculations for this measure. Full outages are defined as occurrences of either of the following:

- Application/interfacing application is down or totally inoperative.
- Application is totally inoperative for customers attempting to access or use the application. This includes transport outages when they may be directly associated with a specific application.

Comparison to an internal benchmark provides a vehicle for determining whether or not CLECs and retail BST entities are given comparable opportunities for use of maintenance and repair systems.

Calculation

OSS Interface Availability (a / b) X 100

- a = Functional Availability
- b = Scheduled Availability

Report Structure

- Not CLEC Specific
- Not Product/Service Specific
- Regional Level

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Availability of CLEC TAFI	Availability of BellSouth TAFI
• Availability of LMOS HOST, MARCH, SOCS, CRIS,	• Availability of LMOS HOST, MARCH, SOCS, CRIS,
PREDICTOR, LNP and OSPCM	PREDICTOR, LNP and OSPCM
• ECTA	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
Regional Level	• >= 99.5%

OSS Interface Availability (M&R)

OSS Interface	% Availability
BST TAFI	Х
CLEC TAFI	Х
CLEC ECTA	x
BellSouth & CLEC	x
CRIS	x
LMOS HOST	X
LNP	Х
MARCH	X
OSPCM	X
PREDICTOR	X
SOCS	Х

SEEM Measure

SEEM Measure		
Yes	Tier I	
	Tier II	Х

SEEM Disaggregation - Analog/Benchmark

SEEM Disaggregation	SEEM Analog/Benchmark
Regional Level	• >= 99.5%

OSS Interface Availability (M&R)

OSS Interface	% Availability
CLEC TAFI	Х
CLEC ECTA	Х

OSS-4: Response Interval (Maintenance & Repair)

Definition

The response intervals are determined by subtracting the time a request is received on the BellSouth side of the interface from the time the response is received from the legacy system. Percentages of requests falling into each interval category are reported, along with the actual number of requests falling into those categories.

Exclusions

None

Business Rules

This measure is designed to monitor the time required for the CLEC and BellSouth interface system to obtain from BellSouth's legacy systems the information required to handle maintenance and repair functions. The clock starts on the date and time when the request is received on the BellSouth side of the interface and the clock stops when the response has been transmitted through that same point to the requester.

Note: The OSS Response Interval BellSouth Total Report is a combination of BellSouth Residence and Business Total.

Calculation

OSS Response Interval = (a - b)

- a = Query Response Date and Time
- b = Query Request Date and Time

Percent Response Interval (per category) = (c / d) X 100

- c = Number of Response Intervals in category "X"
- d = Number of Queries Submitted in the Reporting Period

where, "X" is <= 4, > 4 <= 10, <= 10, > 10, or > 30 seconds.

Report Structure

- Not CLEC Specific
- Not product/service specific
- Regional Level

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
CLEC Transaction Intervals	BellSouth Business and Residential Transactions
	Intervals

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
Regional Level	• Parity

Legacy System Access Times for M&R

System	BellSouth & CLEC			Count		
		<= 4	> 4 <= 10	<= 10	> 10	> 30
CRIS	Х	Х	Х	Х	Х	Х
DLETH	Х	Х	Х	Х	Х	Х
DLR	Х	Х	Х	Х	Х	Х
LMOS	Х	Х	Х	Х	Х	Х
LMOSupd	Х	х	Х	Х	х	х
LNP	Х	Х	Х	Х	Х	X
MARCH	Х	Х	Х	Х	Х	х
OSPCM	Х	Х	Х	Х	Х	х
Predictor	Х	Х	Х	Х	Х	х
SOCS	Х	Х	Х	Х	Х	х
NIW	Х	Х	Х	Х	х	Х

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation - Analog/Benchmark

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

OSS

PO-1: Loop Makeup - Response Time – Manual

Definition

This report measures the average interval and percent within the interval from the submission of a Manual Loop Makeup Service Inquiry (LMUSI) to the distribution of Loop Makeup information back to the CLEC.

Exclusions

- Inquiries, which are submitted electronically.
- Designated Holidays are excluded from the interval calculation.
- Weekend hours from 5:00PM Friday until 8:00AM Monday are excluded from the interval calculation.
- Canceled Inquiries.

Business Rules

The CLEC Manual Loop Makeup Service Inquiry (LMUSI) process includes inquiries submitted via mail or FAX to BellSouth's Complex Resale Support Group (CRSG).

This measurement combines three intervals:

- 1. From receipt of the Service Inquiry for Loop Makeup to hand off to the Service Advocacy Center (SAC) for "Lookup."
- 2. From SAC start date to SAC complete date.
- 3. From SAC complete date to date the Complex Resale Support Group (CRSG) distributes loop makeup information back to the CLEC.

The "Receive Date" is defined as the date the Manual LMUSI is received by the CRSG. It is counted as day Zero. LMU "Return Date" is defined as the date the LMU information is sent back to the CLEC from BellSouth. The interval calculation is reset to Zero when a CLEC initiated change occurs on the Manual LMU request.

Note: The Loop Make Up Service Inquiry Form does not require the CLEC to furnish the type of Loop. The CLEC determines whether the loop makeup will support the type of service they wish to order or not and qualifies the loop. If the loop makeup will support the service, a firm order LSR is submitted by the CLEC.

Calculation

Response Interval = (a - b)

- a = Date and Time LMUSI returned to CLEC
- b = Date and Time the LMUSI is received

Average Interval = (c / d)

- c = Sum of all Response Intervals
- d = Total Number of LMUSIs received within the reporting period

Percent within interval = (e / f) X 100

- e = Total LMUSIs received within the interval
- f = Total Number of LMUSIs processed within the reporting period

Report Structure

- CLEC Aggregate
- CLEC Specific
- Geographic Scope
 - State
 - Region

• Interval for manual LMUs:

- 0 <= 1 day
- >1 <= 2 days
- >2 <= 3 days
- 0 <= 3 days
- >3 <= 6 days
- >6 <= 10 days
- > 10 days
- Average Interval in days

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Not Applicable
Total Number of Inquiries	
• SI Intervals	
State and Region	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
• Loops	Benchmark
	• 95% <= 3 Business Days

SEEM Measure

SEEM Measure		
Yes	Tier I	Х
	Tier II	Х

SEEM Disaggregation - Analog/Benchmark

SEEM Disaggregation	SEEM Analog/Benchmark
• Loops	Benchmark
	• 95% <= 3 Business Days

OSS

PO-2: Loop Make Up - Response Time - Electronic

Definition

This report measures the average interval and the percent within the interval from the electronic submission of a Loop Makeup Service Inquiry (LMUSI) to the distribution of Loop Makeup information back to the CLEC.

Exclusions

- Manually submitted inquiries.
- · Designated Holidays are excluded from the interval calculation.
- Canceled Requests.
- Scheduled OSS Maintenance.

Business Rules

The response interval starts when the CLEC's Mechanized Loop Makeup Service Inquiry (LMUSI) is submitted electronically through the Operational Support Systems interface, LENS, TAG or RoboTAG. It ends when BellSouth's Loop Facility Assignment and Control System (LFACS) responds electronically to the CLEC with the requested Loop Makeup data via LENS, TAG or RoboTAG Interfaces.

Note: The Loop Make Up Service Inquiry Form does not require the CLEC to furnish the type of Loop. The CLEC determines whether the loop makeup will support the type of service they wish to order or not and qualifies the loop. If the loop makeup will support the service, a firm order LSR is submitted by the CLEC. EDI is not a pre-ordering system, and, therefore, is not applicable in this measure.

Calculation

Response Interval = (a - b)

- a = Date and Time LMUSI returned to CLEC
- b = Date and Time the LMUSI is received

Average Interval = (c / d)

- c = Sum of all response intervals
- d = Total Number of LMUSIs received within the reporting period

Percent within interval = (e / f) X 100

- e = Total LMUSIs received within the interval
- f = Total Number of LMUSIs processed within the reporting period

Report Structure

- CLEC Aggregate
- CLEC Specific
- Geographic Scope
 - State
 - Region
- Interval for electronic LMUs:
 - $0 \ll 1$ minute
 - >1 <= 5 minutes
 - 0 <= 5 minutes
- > 5 <= 8 minutes
- > 8 <= 15 minutes
- > 15 minutes
- Average Interval in minutes

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Not Applicable

- Legacy Contract
- Response IntervalRegional Scope

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	n SQM Analog/Benchmark
• Loops	Benchmark
-	• 90% <= 5 Minutes (05/01/01)
	• 95% <= 1 Minute (08/01/01)

SEEM Measure

SEEM Measure		
Yes	Tier I	Х
	Tier II	Х

SEEM Disaggregation - Analog/Benchmark

SEEM Disaggregation	SEEM Analog/Benchmark
• Loop	• 90% <= 5 Minutes (05/01/01)
	• 95% <= 1 Minute (08/01/01)

Section 2: Ordering

O-1: Acknowledgement Message Timeliness

Definition

This measurement provides the response interval from the time an LSR or transmission (may contain multiple LSRs from one or more CLECs in multiple states) is electronically submitted via EDI or TAG respectively until an acknowledgement notice is sent by the system.

Exclusions

Scheduled OSS Maintenance

Business Rules

The process includes EDI & TAG system functional acknowledgements for all messages/Local Service Requests (LSRs) which are electronically submitted by the CLEC. Users of EDI may package many LSRs into one transmission which will receive the acknowledgement message. EDI users may place multiple LSRs in one "envelope" requesting service in one or more states which will mask the identity of the state and CLEC. The start time is the receipt time of the message at BellSouth's side of the interface (gateway). The end time is when the acknowledgement is transmitted by BellSouth at BellSouth's side of the interface (gateway). If more than one CLEC uses the same ordering center (aggregator), an Acknowledgement Message will be returned to the "Aggregator". However, BellSouth will not be able to determine which specific CLEC or state this message represented.

Calculation

Response Interval = (a - b)

- a = Date and Time Acknowledgement Notices returned to CLEC
- b = Date and Time messages/LSRs electronically submitted by the CLEC via EDI or TAG respectively

Average Response Interval = (c / d)

- c = Sum of all Response Intervals
- d = Total number of electronically submitted messages/LSRs received, from CLECs via EDI or TAG respectively, in the Reporting Period.

Reporting Structure

• CLEC Aggregate

- CLEC Specific/Aggregator
- Geographic Scope
- Region
- Electronically Submitted LSRs
- 0 <= 10 minutes
- >10 <= 20 minutes
- >20 <= 30 minutes
- $0 \le 30$ minutes
- >30 <= 45 minutes
- >45 <= 60 minutes
- >60 <= 120 minutes >120 minutes
- Average interval for electronically submitted messages/LSRs in minutes

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Not Applicable
Record of Functional Acknowledgements	

SQM Disaggregation - Analog/Benchmark

enchmark
01)
01)
tes

SEEM Measure

SEEM Measure		
Yes	Tier I	Х
	Tier II	Х

SEEM Disaggregation - Analog/Benchmark

SEEM Disaggregation	SEEM Analog/Benchmark
• EDI	• EDI
	- 90% <= 30 minutes (05/01/01)
	- 95% <= 30 minutes (08/01/01)
• TAG	• TAG – 95% <= 30 minutes

O-2: Acknowledgement Message Completeness

Definition

This measurement provides the percent of transmissions/LSRs received via EDI or TAG respectively, which are acknowledged electronically.

Exclusions

- Manually submitted LSRs
- Scheduled OSS Maintenance

Business Rules

EDI and TAG send Functional Acknowledgements for all transmissions/LSRs, which are electronically submitted by a CLEC. Users of EDI may package many LSRs from multiple states in one transmission. If more than one CLEC uses the same ordering center, an Acknowledgement Message will be returned to the "Aggregator", however, BellSouth will not be able to determine which specific CLEC this message represented. The Acknowledgement Message is returned prior to the determination of whether the transmission/LSR will be partially mechanized or fully mechanized.

Calculation

Acknowledgement Completeness = (a / b) X 100

- a = Total number of Functional Acknowledgements returned in the reporting period for transmissions/LSRs electronically submitted by EDI or TAG respectively
- b = Total number of electronically submitted transmissions/LSRs received in the reporting period by EDI or TAG respectively

Report Structure

- CLEC Aggregate
- CLEC Specific/Aggregator
- Geographic Scope
- Region

Note: The Order calls for Mechanized, Partially Mechanized, and Totally Mechanized, however, the Acknowledgement message is generated before the system recognizes whether this electronic transmission will be partially or fully mechanized.

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Not Applicable
Record of Functional Acknowledgements	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
• EDI	• Benchmark: 100%
• TAG	

SEEM Measure

SEEM Measure		
Yes	Tier I	Х
	Tier II	Х

SEEM Disaggregation - Analog/Benchmark

SEEM Disaggregation	SEEM Analog/Benchmark
• EDI	• Benchmark: 100%
• TAG	

O-3: Percent Flow-Through Service Requests (Summary)

Definition

The percentage of Local Service Requests (LSR) and LNP Local Service Requests (LNP LSRs) submitted electronically via the CLEC mechanized ordering process that flow through and reach a status for a FOC to be issued, without manual intervention.

Exclusions

- Fatal Rejects
- Auto Clarification
- Manual Fallout
- CLEC System Fallout
- Scheduled OSS Maintenance

Business Rules

The CLEC mechanized ordering process includes all LSRs, including supplements (subsequent versions) which are submitted through one of the three gateway interfaces (TAG, EDI and LENS), that flow through and reach a status for a FOC to be issued, without manual intervention. These LSRs can be divided into two classes of service: Business and Residence, and two types of service: Resale, and Unbundled Network Elements (UNE). The CLEC mechanized ordering process does not include LSRs which are submitted manually (for example, fax and courier) or are not designed to flow through (for example, Manual Fallout.)

Definitions:

Fatal Rejects: Errors that prevent an LSR, submitted electronically by the CLEC, from being processed further. When an LSR is submitted by a CLEC, LEO/LNP Gateway will perform edit checks to ensure the data received is correctly formatted and complete. For example, if the PON field contains an invalid character, LEO/LNP Gateway will reject the LSR and the CLEC will receive a Fatal Reject.

Auto-Clarification: Clarifications that occur due to invalid data within the LSR. LESOG/LAUTO will perform data validity checks to ensure the data within the LSR is correct and valid. For example, if the address on the LSR is not valid according to RSAG, or if the LNP is not available for the NPA NXXX requested, the CLEC will receive an Auto-Clarification.

Manual Fallout: Planned Fallout that occur by design. Certain LSRs are designed to fallout of the Mechanized Order Process due to their complexity. These LSRs are manually processed by the LCSC. When a CLEC submits an LSR, LESOG/LAUTO will determine if the LSR should be forwarded to LCSC for manual handling. Following are the categories for Manual Fallout:

- 1. Complex*
- 2. Special pricing plans
- 3. Some Partial migrations
- 4. New telephone number not yet posted to BOCRIS
- 5. Pending order review required
- 6. CSR inaccuracies such as invalid or missing CSR data in CRIS
- 7. Expedites (requested by the CLEC)

- Denials-restore and conversion, or disconnect and conver sion orders
- 9. Class of service invalid in certain states with some types of service
- 10. Low volume such as activity type "T" (move)
- 11. More than 25 business lines, or more than 15 loops
- 12. Transfer of calls option for the CLEC end users
- 13. Directory Listings (Indentions and Captions)

*See LSR Flow-Through Matrix following O-6 for a list of services, including complex services, and whether LSRs issued for the services are eligible to flow through.

Total System Fallout: Errors that require manual review by the LSCS to determine if the error is caused by the CLEC, or is due to BellSouth system functionality. If it is determined the error is caused by the CLEC, the LSR will be sent back to the CLEC for clarification. If it is determined the error is BellSouth caused, the LCSC representative will correct the error, and the LSR will continue to be processed.

Z Status: LSRs that receive a supplemental LSR submission prior to final disposition of the original LSR.

Calculation

Percent Flow Through = a / [b - (c + d + e + f)] X 100

- a = The total number of LSRs that flow through LESOG/LAUTO and reach a status for a FOC to be issued
- b = the number of LSRs passed from LEO/LNP Gateway to LESOG/LAUTO
- c = the number of LSRs that fall out for manual processing
- d = the number of LSRs that are returned to the CLEC for clarification
- e = the number of LSRs that contain errors made by CLECs
- f = the number of LSRs that receive a Z status

Percent Achieved Flow Through = a / [b-(c+d+e)] X 100

- a = the number of LSRs that flow through LESOG/LAUTO and reach a status for a FOC to be issued
- b = the number of LSRs passed from LEO/LNP Gateway to LESOG/LAUTO
- c = the number of LSRs that are returned to the CLEC for clarification
- d = the number of LSRs that contain errors made by CLECs
- e = the number of LSRs that receive Z status

Report Structure

- CLEC Aggregate
 - Region

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Report Month
• Total Number of LSRs Received, by Interface, by CLEC	 Total Number of Errors By Type
- TAG	- Bellsouth System Error
- EDI	
- LENS	
• Total Number of Errors by Type, by CLEC	
- Fatal Rejects	
- Auto Clarification	
- CLEC Caused System Fallout	
 Total Number of Errors by Error Code 	
Total Fallout for Manual Processing	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark ²
Residence	• Benchmark: 95%
Business	• Benchmark: 90%
• UNE	• Benchmark: 85%
• LNP	• Benchmark: 85%

SEEM Measure

SEEM Measure		
Yes	Tier I	
	Tier II	Х

SEEM Disaggregation - Analog/Benchmark

SEEM Disaggregation	SEEM Analog/Benchmark ³
Residence	• Benchmark: 95%
Business	• Benchmark: 90%
• UNE	Benchmark: 85%
• LNP	• Benchmark: 85%

² Benchmarks do not apply to the "Percent Achieved Flow Through."

³ Benchmarks do not apply to the "Percent Achieved Flow Through."

O-4: Percent Flow-Through Service Requests (Detail)

Definition

A detailed list, by CLEC, of the percentage of Local Service Requests (LSR) and LNP Local Service Requests (LNP LSRs) submitted electronically via the CLEC mechanized ordering process that flow through and reach a status for a FOC to be issued, without manual or human intervention.

Exclusions

- · Fatal Rejects
- Auto Clarification
- Manual Fallout
- · CLEC System Fallout
- Scheduled OSS Maintenance

Business Rules

The CLEC mechanized ordering process includes all LSRs, including supplements (subsequent versions) which are submitted through one of the three gateway interfaces (TAG, EDI, and LENS), that flow through and reach a status for a FOC to be issued, without manual intervention. These LSRs can be divided into two classes of service: Business and Residence, and three types of service: Resale, and Unbundled Network Elements (UNE). The CLEC mechanized ordering process does not include LSRs, which are submitted manually (for example, fax and courier) or are not designed to flow through (for example, Manual Fallout.)

Definitions:

Fatal Rejects: Errors that prevent an LSR, submitted electronically by the CLEC, from being processed further. When an LSR is submitted by a CLEC, LEO/LNP Gateway will perform edit checks to ensure the data received is correctly formatted and complete. For example, if the PON field contains an invalid character, LEO/LNP Gateway will reject the LSR and the CLEC will receive a Fatal Reject.

Auto-Clarification: Clarifications that occur due to invalid data within the LSR. LESOG/LAUTO will perform data validity checks to ensure the data within the LSR is correct and valid. For example, if the address on the LSR is not valid according to RSAG, or if the LNP is not available for the NPA NXXX requested, the CLEC will receive an Auto-Clarification.

Manual Fallout: Planned Fallout that occur by design. Certain LSRs are designed to fallout of the Mechanized Order Process due to their complexity. These LSRs are manually processed by the LCSC. When a CLEC submits an LSR, LESOG/LAUTO will determine if the LSR should be forwarded to LCSC for manual handling. Following are the categories for Manual Fallout:

1. Complex*

6.

- 2. Special pricing plans
- 3. Some Partial migrations
- 4 New telephone number not yet posted to BOCRIS
- 5. Pending order review required CSR inaccuracies such as invalid or missing CSR data in
- 8. Denials-restore and conversion, or disconnect and conver sion orders
- 9 Class of service invalid in certain states with some types of service
- 10. Low volume such as activity type "T" (move)
- 11. More than 25 business lines, or more than 15 loops
- Transfer of calls option for the CLEC end users 12.
- 13. Directory Listings (Indentions and Captions)

Expedites (requested by the CLEC)

*See LSR Flow-Through Matrix following O-6 for a list of services, including complex services, and whether LSRs issued for the services are eligible to flow through.

Total System Fallout: Errors that require manual review by the LSCS to determine if the error is caused by the CLEC, or is due to BellSouth system functionality. If it is determined the error is caused by the CLEC, the LSR will be sent back to the CLEC for clarification. If it is determined the error is BellSouth caused, the LCSC representative will correct the error, and the LSR will continue to be processed.

Z Status: LSRs that receive a supplemental LSR submission prior to final disposition of the original LSR.

Calculation

Percent Flow Through = $a / [b - (c + d + e + f)] \times 100$

- a = The total number of LSRs that flow through LESOG/LAUTO and reach a status for a FOC to be issued
- b = the number of LSRs passed from LEO/LNP Gateway to LESOG/LAUTO
- c = the number of LSRs that fall out for manual processing
- d = the number of LSRs that are returned to the CLEC for clarification
- e = the number of LSRs that contain errors made by CLECs
- f = the number of LSRs that receive a Z status

Percent Achieved Flow Through = a / [b-(c+d+e)] X 100

- a = the number of LSRs that flow through LESOG/LAUTO and reach a status for a FOC to be issued
- b = the number of LSRs passed from LEO/LNP Gateway to LESOG/LAUTO
- c = the number of LSRs that are returned to the CLEC for clarification
- d = the number of LSRs that contain errors made by CLECs
- e = the number of LSRs that receive Z status

Report Structure

Provides the flow through percentage for each CLEC (by alias designation) submitting LSRs through the CLEC mechanized ordering process. The report provides the following:

- CLEC (by alias designation)
- Number of fatal rejects
- Mechanized interface used
- Total mechanized LSRs
- Total manual fallout
- Number of auto clarifications returned to CLEC
- Number of validated LSRs
- Number of BellSouth caused fallout
- Number of CLEC caused fallout
- Number of Service Orders Issued
- Base calculation
- CLEC error excluded calculation

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Report Month
• Total Number of LSRs Received, by Interface, by CLEC	 Total Number of Errors by Type
- TAG	- Bellsouth System Error
- EDI	
- LENS	
• Total Number of Errors by Type, by CLEC	
- Fatal Rejects	
- Auto Clarification	
- CLEC Errors	
Total Number of Errors by Error Code	
Total Fallout for Manual Processing	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark ⁴
Residence	• Benchmark: 95%
Business	• Benchmark: 90%
• UNE	• Benchmark: 85%
• LNP	• Benchmark: 85%

⁴ Benchmarks do not apply to the "Percent Achieved Flow Through."
SEEM Measure

SEEM Measure									
Yes	Tier I	X							
	Tier II								

SEEM Disaggregation	SEEM Analog/Benchmark ⁵
Residence	• Benchmark: 95%
• Business	• Benchmark: 90%
• UNE	• Benchmark: 85%
• LNP	Benchmark: 85%

⁵ Benchmarks do not apply to the "Percent Achieved Flow Through."

O-5: Flow-Through Error Analysis

Definition

An analysis of each error type (by error code) that was experienced by the LSRs that did not flow through or reached a status for a FOC to be issued.

Exclusions

Each Error Analysis is error code specific, therefore exclusions are not applicable.

Business Rules

The CLEC mechanized ordering process includes all LSRs, including supplements (subsequent versions) which are submitted through one of the three gateway interfaces (TAG, EDI, and LENS), that flow through and reach a status for a FOC to be issued. The CLEC mechanized ordering process does not include LSRs which are submitted manually (for example, fax and courier).

Calculation

Total for each error type.

Report Structure

Provides an analysis of each error type (by error code). The report is in descending order by count of each error code and provides the following:

- Error Type (by error code)
- Count of each error type
- Percent of each error type
- Cumulative percent
- Error Description
- CLEC Caused Count of each error code
- Percent of aggregate by CLEC caused count
- Percent of CLEC caused count
- BellSouth Caused Count of each error code
- · Percent of aggregate by BellSouth caused count
- · Percent of BellSouth by BellSouth caused count

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance					
Report Month	Report Month					
 Total Number of LSRs Received 	• Total Number of Errors by Type (by error code)					
• Total Number of Errors by Type (by error code)	- BellSouth System Error					
- CLEC Caused Error						

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark				
Not Applicable	Not Applicable				

SEEM Measure

SEEM Measure							
No	Tier I						
	Tier II						

SEEM Disaggregation	SEEM Analog/Benchmark				
Not Applicable	Not Applicable				

O-6: CLEC LSR Information

Definition

A list with the flow through activity of LSRs by CC, PON and Ver, issued by each CLEC during the report period.

Exclusions

- Fatal Rejects
- LSRs submitted manually

Business Rules

The CLEC mechanized ordering process includes all LSRs, including supplements (subsequent versions) which are submitted through one of the three gateway interfaces (TAG, EDI, and LENS), that flow through and reach a status for a FOC to be issued. The CLEC mechanized ordering process does not include LSRs which are submitted manually (for example, fax and courier).

Calculation

Not Applicable

Report Structure

Provides a list with the flow through activity of LSRs by CC, PON and Ver, issued by each CLEC during the report period with an explanation of the of the columns and content. This report is available on a CLEC specific basis. The report provides the following for each LSR.

- CC
- PON
- Ver
- Timestamp
- Type
- Err #
- Note or Error Description

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Not Applicable
• Record of LSRs Received by CC, PON and Ver	
• Record of Timestamp, Type, Err # and Note or Error	
Description for each LSR by CC, PON and Ver	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark				
Not Applicable	Not Applicable				

SEEM Measure

SEEM Measure							
No	Tier I						
	Tier II						

SEEM Disaggregation	SEEM Analog/Benchmark			
Not Applicable	Not Applicable			

LSR Flow Through Matrix

Product	Product	Reqtype	ACT Type	F/T ³	Comple		Planned		TAG	LEN
	Туре				x		Fallout For		2	S ⁴
					Service	Order				
			NE			3.7	Handling ¹			
2 wire analog DID trunk port	U,C	A	N,T	No	UNE	Yes	NA	N	N	N
2 wire analog port	U	A	N,T	No	UNE	No	Yes	Y	Y	N
2 wire ISDN digital line	U,C	A	N,T	No	UNE	Yes	NA	N	N	N
2 wire ISDN digital loop	U,C	A	N,T	Yes	UNE	Yes	No	Y	Y	N
3 Way Calling	R,B	E,M	N,C,T,V,W	Yes	No	No	No	Y	Y	Y
4 wire analog voice grade loop	U,C	A	N,T	Yes	UNE	Yes	No	Y	Y	N
4 wire DSO & PRI digital loop	U,C	A	N,T	No	UNE	Yes	NA	Ν	Ν	Ν
4 wire DS1 & PRI digital loop	U,C	A	N,T	No	UNE	Yes	NA	N	N	N
4 wire ISDN DSI digital trunk ports	U,C	A	N,T	No	UNE	Yes	NA	N	N	N
Accupulse	С	E	N,C,T,V,W	No	Yes	Yes	NA	Ν	Ν	Ν
ADSL	R,B,C	E	V,W	No	UNE	No	No	Y	Y	Ν
Area Plus	R,B	E,M	N,C,T,V,W	Yes	No	No	No	Y	Y	Y
Basic Rate ISDN	U,C	A	N,T	No	Yes	Yes	Yes	Y	Y	Ν
Basic Rate ISDN 2 Wire	С	E	C, D,T,V,W	No	Yes	Yes	Yes	Y	Y	Ν
Basic Rate ISDN 2 Wire	С	Е	N,T	No	Yes	Yes	N/A	Ν	Ν	Ν
Basic Rate ISDN 2 Wire UNE P	С	М	N,C,D,V	No	YES	Yes	N/A	Ν	Ν	Ν
Analog Data/Private Line	С	E	N, C, T, V, W, D, P,	No	Yes	Yes	N/A	Ν	Ν	Ν
			Q							
Call Block	R,B	E,B,M	N,C,T,V,W	Yes	No	No	No	Y	Y	Y
Call Forwarding	R,B	E,B,M	N,C,T,V,W	Yes	No	No	No	Y	Y	Y
Call Return	R,B	E,B,M	N,C,T,V,W	Yes	No	No	No	Y	Y	Y
Call Selector	R,B	E,B,M	N,C,T,V,W	Yes	No	No	No	Y	Y	Y
Call Tracing	R,B	E,B,M	N,C,T,V,W	Yes	No	No	No	Y	Y	Y
Call Waiting	R,B	E,B,M	N,C,T,V,W	Yes	No	No	No	Y	Y	Y
Call Waiting Deluxe	R,B	E,B,M	N,C,T,V,W	Yes	No	No	No	Y	Y	Y
Caller ID	R,B	E,B,M	N,C,T,V,W	Yes	No	No	No	Y	Y	Y
CENTREX	C	Р	V,P	No	Yes	Yes	NA	Ν	Ν	Ν
DID ACT W	C	Ν	W	No	Yes	Yes	Yes	Y	Y	Y
Digital Data Transport	U	Е	N,C,T,V,W	No	UNE	Yes	NA	Ν	Ν	Ν
Directory Listing Indentions	B,U	B,C,E,F, J,M,N	N,C,T,R,V,W,P,Q	No	No	No	Yes	Y	Y	Y
Directory Listings Captions	R,B,U	B,C,E,F, J,M,N	N,C,T,R,V,W,P,Q	No	No	Yes	Yes	Y	Y	Y
Directory Listings (simple)	R,B,U	B,C,E,F, J,M,N	N,C,T,R,V,W,P,Q	Yes	No	No	No	Y	Y	Y
DS3	U	A,M	N,C,V	No	UNE	Yes	NA	Ν	Ν	Ν
DS1Loop	U	A,M	N,C,V	Yes	UNE	Yes	No	Y	Y	N
DSO Loop	U	A, B	N,C,D,T,V	Yes	UNE	Yes	No	Y	Y	Ν
Enhanced Caller ID	R,B	E,M	C,D,N,T,V,W	Yes	No	No	No	Y	Y	Y
ESSX	C	P	C,D,T,V,S,B,W,L ,P,Q	No	Yes	Yes	NA	N	N	N
Flat Rate/Business	В	E, M	C,D,N,T,V,W	Yes	No	No	No	Y	Y	Y
Flat Rate/Residence	R	E, M	C,D,N,T,V,W	Yes	No	No	No	Y	Y	Y
FLEXSERV	C	E	N,C,D,T,V,W,P,Q	No	Yes	Yes	NA	N	N	N
Frame Relay	C	E	N,C,D,V,W	No	Yes	Yes	NA	N	N	N
FX	C	E	N,C,D,T,V,W,P,Q	No	Yes	Yes	NA	N	N	N
Ga. Community Calling	R,B	E, M	C,D,N,T,V,W	Yes	No	No	No	Y	Y	Y
HDSL	U U	A A	N,C,D	Yes	UNE	No	No	Y	Y	N
Hunting MLH	R,B	E, M	C,D,N,T,V,W	No	C/S4	C/S	Yes	Y	Y	N
Hunting Series Completion	R,B	E, M E, M	C,D,N,T,V,W	Yes	C/S4 C/S	C/S	No	Y	Y	Y
INP to LNP Conversion	U K,D	C C	С, D, N, I, V, W	No	UNE	Yes	Yes	Y	Y	N

Product	Product	Reqtype	ACT Type	F/T ³	Comple	Com	Planned	EDI	TAG	LEN
	Туре		21		x	plex	Fallout For		2	S^4
					Service	Order				
							Handling ¹			<u> </u>
LightGate	С	Е	N,C,D,T,V,W,P,Q	No	Yes	Yes	NA	Ν	Ν	Ν
Line Sharing	U	А	C,D	Yes	UNE	No	No	Y	Y	Y
Local Number Portability	U	C	C,D,P,V,Q	Yes	UNE	Yes	No	Y	Y	Ν
LNP With Complex Listing	C	C	P,V,Q,W	No	UNE	Yes	Yes	Y	Y	Ν
LNP with Partial Migration	U	С	D,P,V,Q	No	UNE	Yes	Yes	Y	Y	Ν
LNP with Complex Services	C	С	P,V,Q,W	No	UNE	Yes	Yes	Y	Y	Ν
Loop+INP	U	В	D,P,V,Q	Yes	UNE	No	No	Y	Y	Ν
Loop+LNP	U	В	C,D,N,V	Yes	UNE	No	No	Y	Y	Ν
Measured Rate/Bus	R,B	E,M	C,D,T,N,V,W	Yes	No	No	No	Y	Y	Y
Measured Rate/Res	R,B	E,M	C,D,T,N,V,W	Yes	No	No	No	Y	Y	Y
Megalink	С	Е	N,V,W,T,D,C,P,Q	No	Yes	Yes	NA	Ν	Ν	Ν
Megalink-T1	С	E,M	N,V,W,T,D,C,P,Q	No	Yes	Yes	NA	Ν	Ν	Ν
Memory Call	R,B	E, M	C,D,N,T,V,W	Yes	No	No	No	Y	Y	Y
Memory Call Ans. Svc.	R,B	E, M	C,D,N,T,V,W	Yes	No	No	No	Y	Y	Y
Multiserv	С	Р	N,C,D,T,V,S,B, W,L,P,Q	No	Yes	Yes	NA	N	N	N
Native Mode LAN Interconnection (NMLI)	С	Е	N,C,D,V,W	No	Yes	Yes	NA	N	N	N
Off-Prem Stations	С	Е	N,C,D,V,W,T,P,Q	No	Yes	Yes	NA	Ν	Ν	Ν
Optional Calling Plan	R,B	E, M	N	Yes	No	No	No	Y	Y	Y
Package/Complete Choice and Area Plus	R,B	E, M	N,T,C,V,W	Yes	No	No	No	Y	Y	Y
Pathlink Primary Rate ISDN	С	Е	N,C,D,T,V,W,P,Q	No	Yes	Yes	NA	Ν	Ν	Ν
Pay Phone Provider	B	E	C,D,T,N,V,W	No	No	No	NA	N	N	N
PBX Standalone Port	C	F	N,C,D	No	Yes	Yes	Yes	Y	Y	N
PBX Trunks	R,B	E	N,C,D,V,W,T,P,Q	No	Yes	Yes	Yes	Y	Y	N
Port/Loop PBX	U	M	A,C,D,V	No	No	No	Yes	Y	Y	N
Port/Loop Simple	U	М	A,C,D,V	Yes	No	No	Yes	Y	Y	Y
Preferred Call Forward	R,B,U	E	C,D,T,N,V,W	Yes	No	No	No	Y	Y	Y
RCF Basic	R,B	E	N,D,W,T,F	Yes	No	No	No	Y	Y	Y
Remote Access to CF	R,B	E,M	C,D,T,N,V,W	Yes	No	No	No	Y	Ŷ	Y
Repeat Dialing	R,B	E,M	C,D,T,N,V,W	Yes	No	No	No	Y	Y	Y
Ringmaster	R,B	E,M	C,D,T,N,V,W	Yes	No	No	No	Y	Y	Y
Smartpath	R,B	E	C,D,T,N,V,W	No	Yes	Yes	NA	N	N	N
SmartRING	C	E	N,D,C,V,W	No	Yes	Yes	NA	N	N	N
Speed Calling	R,B	E	C,D,T,N,V,W	Yes	No	No	No	Y	Y	Y
Synchronet	C	E	<u>N</u>	Yes	Yes	Yes	Yes	Y	Y	N
Tie Lines	C	E	N,C,D,V,W,T,P,Q	No	Yes	Yes	NA	N	N	N
Touchtone	R,B	E	C,D,T,N,V,W	Yes	No	No	No	Y	Y	Y
Unbundled Loop-Analog 2W, SL1, SL2	U	A,B	C,D,T,N,V,W	Yes	UNE	No	No	Y	Y	Y
WATS	R,B	Е	W,D	No	Yes	Yes	NA	N	N	N
XDSL	C,U	A,B	N,T,C,V,D	Yes	UNE	No	No	Y	Y	N
XDSL XDSL Extended LOOP	C,U C,U	A,B A,B	N,T,C,V,D N,T,C,V,D	No	UNE	Yes	NA	N	I N	N
Collect Call Block	R,B	E A,D	N,T,C,V,W,D	Yes	No	No	No	Y	Y	Y
900 Call Block	R,B	E	N,T,C,V,W,D	Yes	No	No	No	Y	Y	Y
3rd Party Call Block	R,B	E	N,T,C,V,W,D N,T,C,V,W,D	Yes	No	No	No	Y Y	Y Y	Y Y
Three Way Call Block	R,B	E	N,T,C,V,W,D N,T,C,V,W,D	Yes	No	No	No	I Y	I Y	Y I
PIC/LPIC Change	R,B	E	T,C,V,	Yes	No	No	No	Y	I Y	I Y
PIC/LPIC Change PIC/LPIC Freeze	R,B	E E	N,T,C,V		No	No		Y Y	Y Y	Y Y
FIC/LFIC FIEEZE	к,в	Ľ	IN, I, C, V	Yes	10	100	No	ľ	ľ	I

Note¹: Planned Fallout for Manual Handling denotes those services that are electronically submitted and are not intended to flow through due to the complexity of the service.

Note²: The TAG column includes those LSRs submitted via Robo TAG.

Note³: For all services that indicate 'No' for flow-through, the following reasons, in addition to errors or complex services, also prompt manual handling: Expedites from CLECs, special pricing plans, denials restore and conversion or disconnect and conversion both required, partial migrations (although conversions-as-is flow through for issue 9), class of service invalid in certain states with some TOS e.g. government, or cannot be changed when changing main TN on C activity, low volume e.g. activity type T=move, pending order review required, more than 25 business lines, CSR inaccuracies such as invalid or missing CSR data in CRIS, Directory listings – Indentions, Directory listings – Captions, transfer of calls option for CLEC end user – new TN not yet posted to BOCRIS. Many are unique to the CLEC environment.

Note⁴: Services with C/S in the Complex Service and/or the Complex Order columns can be either complex or simple.

Note⁵: EELs are manually ordered.

Note⁶: LSRs submitted for Resale Products and Services for which there is a temporary promotion or discount plan will be processed identically to those LSRs ordering the same Products or Services without a promotion or discount plan.

O-7: Percent Rejected Service Requests

Definition

Percent Rejected Service Request is the percent of total Local Service Requests (LSRs) received which are rejected due to error or omission. An LSR is considered valid when it is submitted by the CLEC and passes edit checks to insure the data received is correctly formatted and complete.

Exclusions

- Service Requests canceled by the CLEC prior to being rejected/clarified.
- Scheduled OSS Maintenance

Business Rules

Fully Mechanized: An LSR is considered "rejected" when it is submitted electronically but does not pass LEO edit checks in the ordering systems (EDI, LENS, TAG, LEO, LESOG) and is returned to the CLEC without manual intervention. There are two types of "Rejects" in the Mechanized category:

A **Fatal Reject** occurs when a CLEC attempts to electronically submit an LSR but required fields are either not populated or incorrectly populated and the request is returned to the CLEC before it is considered a valid LSR.

Fatal rejects are reported in a separate column, and for informational purposes ONLY. Fatal rejects are excluded from the calculation of the percent of total LSRs rejected or the total number of rejected LSRs.

An **Auto Clarification** occurs when a valid LSR is electronically submitted but rejected from LESOG because it does not pass further edit checks for order accuracy.

Partially Mechanized: A valid LSR, which is electronically submitted (via EDI, LENS, TAG) but cannot be processed electronically and "falls out" for manual handling. It is then put into "clarification" and sent back (rejected) to the CLEC.

Total Mechanized: Combination of Fully Mechanized and Partially Mechanized LSRs electronically submitted by the CLEC.

Non-Mechanized: LSRs which are faxed or mailed to the LCSC for processing and "clarified" (rejected) back to the CLEC by the BellSouth service representative.

Interconnection Trunks: Interconnection Trunks are ordered on Access Service Requests (ASRs). ASRs are submitted to and processed by the Interconnection Purchasing Center (IPC). Trunk data is reported separately.

Calculation

Percent Rejected Service Requests = (a / b) X 100

- a = Total Number of Rejected Service Requests in the Reporting Period
- b = Total Number of Service Requests Received in the Reporting Period

Report Structure

- Fully Mechanized, Partially Mechanized, Total Mechanized, Non-Mechanized
- CLEC Specific
- CLEC Aggregate
- Geographic Scope
- State
- Region
- Product Specific Percent Rejected
- Total Percent Rejected

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Not Applicable
Total Number of LSRs	
 Total Number of Rejects 	
State and Region	
Total Number of ASRs (Trunks)	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
Mechanized, Partially Mechanized and Non-Mechanized	• Diagnostic
Resale - Residence	
Resale - Business	
• Resale – Design (Special)	
Resale PBX	
Resale Centrex	
Resale ISDN	
• LNP (Standalone)	
• INP (Standalone)	
 2W Analog Loop Design 	
 2W Analog Loop Non-Design 	
2W Analog Loop With INP Design	
 2W Analog Loop With INP Non-Design 	
• 2W Analog Loop With LNP Design	
• 2W Analog Loop With LNP Non-Design	
• UNE Loop + Port Combinations	
Switch Ports	
UNE Combination Other	
• UNE xDSL (ADSL, HDSL, UCL)	
Line Sharing	
UNE ISDN Loop	
• UNE Other Design	
• UNE Other Non-Design	
Local Interoffice Transport	
Local Interconnection Trunks	

SEEM Measure

	SEEM Measure			
No	Tier I			
	Tier II			

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

O-8: Reject Interval

Definition

Reject Interval is the average reject time from receipt of an LSR to the distribution of a Reject. An LSR is considered valid when it is submitted by the CLEC and passes edit checks to insure the data received is correctly formatted and complete.

Exclusions

- Service Requests canceled by CLEC prior to being rejected/clarified
- Designated Holidays are excluded from the interval calculation
- LSRs which are identified and classified as "Projects"
- The following hours for Partially mechanized and Non-mechanized LSRs are excluded from the interval calculation:

Residence Resale Group – Monday through Saturday 7:00PM until 7:00AM From 7:00 PM Saturday until 7:00 AM Monday

Business Resale, Complex, UNE Groups – Monday through Friday 6:00PM until 8:00AM From 6:00 PM Friday until 8:00 AM Monday.

The hours excluded will be altered to reflect changes in the Center operating hours. The LCSC will accept faxed LSRs only during posted hours of operation.

The interval will be the amount of time accrued from receipt of the LSR until normal closing of the center if an LSR is worked using overtime hours.

In the case of a Partially Mechanized LSR received and worked after normal business hours, the interval will be set at one (1) minute.

Scheduled OSS Maintenance

Business Rules

Fully Mechanized: The elapsed time from receipt of a valid electronically submitted LSR (date and time stamp in EDI, LENS or TAG) until the LSR is rejected (date and time stamp or reject in EDI, TAG or LENS). Auto Clarifications are considered in the Fully Mechanized category.

Partially Mechanized: The elapsed time from receipt of a valid electronically submitted LSR (date and time stamp in EDI, LENS or TAG) until it falls out for manual handling. The stop time on partially mechanized LSRs is when the LCSC Service Representative clarifies the LSR back to the CLEC via LENS, EDI, or TAG.

Total Mechanized: Combination of Fully Mechanized and Partially Mechanized LSRs which are electronically submitted by the CLEC.

Non-Mechanized: The elapsed time from receipt of a valid LSR (date and time stamp of FAX or date and time mailed LSR is received in the LCSC) until notice of the reject (clarification) is returned to the CLEC via LON.

Interconnection Trunks: Interconnection Trunks are ordered on Access Service Requests (ASRs). ASRs are submitted to and processed by the Local Interconnection Service Center (LISC). Trunk data is reported separately. All interconnection trunks are counted in the non-mechanized category.

Calculation

Reject Interval = (a - b)

- a = Date and Time of Service Request Rejection
- b = Date and Time of Service Request Receipt

Average Reject Interval = (c / d)

- c = Sum of all Reject Intervals
- d = Number of Service Requests Rejected in Reporting Period

Report Structure

- CLEC Specific
- CLEC Aggregate
- Fully Mechanized, Partially Mechanized, Total Mechanized, Non-Mechanized
- Geographic Scope

- State
- Region
- Mechanized:
 - 0 <= 4 minutes
 - >4 <= 8 minutes
- >8 <= 12 minutes >12 - <= 60 minutes
- 0 <= 1 hour
- >1 <= 4 hours
- >4 <= 8 hours
- >8 <= 12 hours
- >12 <= 16 hours
- >16 <= 20 hours
- >20 <= 24 hours
- >24 hours
- Partially Mechanized:
- 0 <= 1 hour >1 - <= 4 hours
- >1 <= 4 hours >4 - <= 8 hours
- >8 <= 10 hours
- 0 <= 10 hours
- >10 <= 18 hours
- 0 <= 18 hours
- >18 <= 24 hours
- >24 hours
- Non-mechanized:
- 0 <= 1 hour
- >1 <= 4 hours
- >4 <= 8 hours
- >8 <= 12 hours
- >12 <= 16 hours >16 - <= 20 hours
- >10 <= 20 hours >20 - <= 24 hours
- 0 <= 24 hours
- > 24 hours
- Trunks:
 - $\leq 4 \text{ days}$
- >4 <= 8 days
- >8 <= 12 days
- >12 <= 14 days
- >14 <= 20 days
- >20 days

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Not Applicable
Reject Interval	
Total Number of LSRs	
 Total Number of Rejects 	
State and Region	
• Total Number of ASRs (Trunks)	

SQM Level of Disaggregation	SQM Analog/Benchmark
Resale - Residence	• Mechanized:
Resale - Business	- 97% <= I Hour
• Resale - Design (Special)	• Partially Mechanized:
Resale PBX	- 85% <= 24 hours
Resale Centrex	- 85% <= 18 Hours (05/01/01)

Resale ISDN	- 85% <= 10 Hours (08/01/01)
• LNP (Standalone)	• Non-Mechanized: - 85% <= 24 hours
• INP (Standalone)	
2W Analog Loop Design	
2W Analog Loop Non-Design	
• 2W Analog Loop With INP Design	
 2W Analog Loop With INP Non-Design 	
• 2W Analog Loop With LNP Design	
 2W Analog Loop With LNP Non-Design 	
 UNE Loop + Port Combinations 	
Switch Ports	
UNE Combination Other	
• UNE xDSL (ADSL, HDSL, UCL)	
Line Sharing	
UNE ISDN Loops	
UNE Other Non-Design	
Local Interoffice Transport	
UNE Other Design	
Local Interconnection Trunks	• Trunks: - 85% <= 4 Days

SEEM Measure

SEEM Measure		
Yes	Tier I	Х
	Tier II	Х

SEEM Disaggregation	SEEM Analog/Benchmark
Fully Mechanized	• 97% <= 1 Hour
Partially Mechanized	• 85% <= 24 Hours
	• 85% <= 18 Hours (05/01/01)
	• 85% <= 10 Hours (08/01/01)
Non-Mechanized	• 85% <= 24 Hours

O-9: Firm Order Confirmation Timeliness

Definition

Interval for Return of a Firm Order Confirmation (FOC Interval) is the average response time from receipt of valid LSR to distribution of a Firm Order Confirmation.

Exclusions

- Rejected LSRs
- Designated Holidays are excluded from the interval calculation
- LSRs which are identified and classified as "Projects"
- The following hours for Partially Mechanized and Non-mechanized LSRs are excluded from the interval calculation:

Residence Resale Group – Monday through Saturday 7:00PM until 7:00AM From 7:00 PM Saturday until 7:00 AM Monday.

Business Resale, Complex, UNE Groups – Monday through Friday 6:00PM until 8:00AM From 6:00 PM Friday until 8:00 AM Monday.

The hours excluded will be altered to reflect changes in the Center operating hours. The LCSC will accept faxed LSRs only during posted hours of operation.

The interval will be the amount of time accrued from receipt of the LSR until normal closing of the center if an LSR is worked using overtime hours.

In the case of a Partially Mechanized LSR received and worked after normal business hours, the interval will be set at one (1) minute.

Scheduled OSS Maintenance

Business Rules

- Fully Mechanized: The elapsed time from receipt of a valid electronically submitted LSR (date and time stamp in EDI, LENS or TAG) until the LSR is processed, appropriate service orders are generated and a Firm Order Confirmation is returned to the CLEC via EDI, LENS or TAG.
- **Partially Mechanized:** The elapsed time from receipt of a valid electronically submitted LSR (date and time stamp in EDI, LENS, or TAG) which falls out for manual handling until appropriate service orders are issued by a BellSouth service representative via Direct Order Entry (DOE) or Service Order Negotiation Generation System (SONGS) to SOCS and a Firm Order Confirmation is returned to the CLEC via EDI, LENS, or TAG.
- Total Mechanized: Combination of Fully Mechanized and Partially Mechanized LSRs which are electronically submitted by the CLEC.
- Non-Mechanized: The elapsed time from receipt of a valid paper LSR (date and time stamp of FAX or date and time paper LSRs received in LCSC) until appropriate service orders are issued by a BellSouth service representative via Direct Order Entry (DOE) or Service Order Negotiation Generation System (SONGS) to SOCS and a Firm Order Confirmation is sent to the CLEC via LON.
- Interconnection Trunks: Interconnection Trunks are ordered on Access Service Requests (ASRs). ASRs are submitted to and processed by the Local Interconnection Service Center (LISC). Trunk data is reported separately.

Calculation

Firm Order Confirmation Interval = (a - b)

- a = Date & Time of Firm Order Confirmation
- b = Date & Time of Service Request Receipt)

Average FOC Interval = (c / d)

- c = Sum of all FOC Intervals
- d = Total Number of Service Requests Confirmed in Reporting Period

FOC Interval Distribution (for each interval) = (e / f) X 100

- e = Service Requests Confirmed in interval
- f = Total Service Requests Confirmed in the Reporting Period

Report Structure

- Fully Mechanized, Partially Mechanized, Total Mechanized, Non-Mechanized
 - CLEC Specific
- CLEC Aggregate
- Geographic Scope
- State - Region • Fully Mechanized: $0 - \leq 15$ minutes >15 - <= 30 minutes >30 - <= 45 minutes >45 - <= 60 minutes >60 - <= 90 minutes >90 - <= 120 minutes >120 - <= 180 minutes $0 - \le 3$ hours >3 - <= 6 hours >6 - <= 12 hours >12 - <= 24 hours >24 - <= 48 hours >48 hours • Partially Mechanized: $0 - \leq 4$ hours >4 - <= 8 hours >8 - <= 10 hours 0 - <= 10 hours >10 - <= 18 hours 0 - <= 18 hours >18 - <= 24 hours 0 - <= 24 hours >24 - <= 48 hours >48 hours • Non-Mechanized: $0 - \leq 4$ hours >4 - <= 8 hours >8 - <= 12 hours >12 - <= 16 hours >16 - <= 20 hours >20 - <= 24 hours
- >20 <= 24 hours >24 - <= 36 hours 0 - <= 36 hours >36 - <= 48 hours >48 hours

• Trunks: 0 - <= 5 days >5 - <= 10 days 0 - <= 10 days >10 - <= 15 days >15 - <= 20 days >20 days

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Not Applicable
• Interval for FOC	
Total Number of LSRs	
State and Region	
• Total Number of ASRs (Trunks)	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
• Resale – Residence	• Mechanized: - 95% <= 3 Hours
• Resale – Business	Partially Mechanized:
• Resale – Design (Special)	- 85% <= 24 Hours
Resale PBX	- 85% <= 18 Hours (05/01/01)
Resale Centrex	- 85% <= 10 Hours (08/01/01)
Resale ISDN	• Non-mechanized: - 85% <= 36 Hours
• LNP (Standalone)	
• INP(Standalone)	
2W Analog Loop Design	
2W Analog Loop Non-Design	
2W Analog Loop With INP Design	
 2W Analog Loop With INP Non-Design 	
2W Analog Loop With LNP Design	
 2W Analog Loop With LNP Non-Design 	
• UNE Loop + Port Combinations	
Switch Ports	
UNE Combination Other	
• UNE xDSL (ADSL, HDSL, UCL)	
Line Sharing	
UNE ISDN Loops	
UNE Other Design	
UNE Other Non-Design	
Local Interoffice Transport	
Local Interconnection Trunks	• Trunks: - 95% <= 10 Days

SEEM Measure

SEEM Measure		
Yes	Tier I	Х
	Tier II	Х

SEEM Disaggregation	SEEM Analog/Benchmark
Fully Mechanized	• 95% <= 3 Hours
Partially Mechanized	• 85% <= 24 Hours
	• 85% <= 18 Hours (05/01/01)
	• 85% <= 10 Hours (08/01/01)
Non-Mechanized	• 85% <= 36 Hours
IC Trunks	• 95% <= 10 Days

O-10: Service Inquiry with LSR Firm Order Confirmation (FOC) Response Time Manual⁶

Definition

This report measures the interval and the percent within the interval from the submission of a Service Inquiry (SI) with Firm Order LSR to the distribution of a Firm Order Confirmation (FOC).

Exclusions

- Designated Holidays are excluded from the interval calculation
- Weekend hours from 5:00PM Friday until 8:00AM Monday are excluded from the interval calculation of the Service Inquiry
- Canceled Requests
- Electronically Submitted Requests
- Scheduled OSS Maintenance

Business Rules

This measurement combines four intervals:

- 1. From receipt of Service Inquiry with LSR to hand off to the Service Advocacy Center (SAC) for Loop 'Look-up'.
- 2. From SAC start date to SAC complete date.
- 3. From SAC complete date to the Complex Resale Support Group (CRSG) complete date with hand off to LCSC.
- 4. From receipt of SI/LSR in the LCSC to Firm Order Confirmation.

Calculation

FOC Timeliness Interval = (a - b)

- a = Date and Time Firm Order Confirmation (FOC) for SI with LSR returned to CLEC
- b = Date and Time SI with LSR received

Average Interval = (c / d)

- c = Sum of all FOC Timeliness Intervals
- d = Total number of SIs with LSRs received in the reporting period

Percent Within Interval = (e / f) X 100

- e = Total number of Service Inquiries with LSRs received by the CRSG to distribution of FOC by the Local Carrier Service Center (LCSC)
- f = Total number of Service Inquiries with LSRs received in the reporting period

Report Structure

- CLEC Aggregate
- CLEC Specific
- Geographic Scope
 - State
 - Region
- Intervals
- 0 <= 3 days
- >3 <= 5 days
- 0 <= 5 days
- >5 <= 7 days >7 - <= 10 days
- >10 <= 15 days
- >15 days

⁶

See O-9 for FOC Timeliness

• Average Interval measured in days

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Not Applicable
Total Number of Requests	
• SI Intervals	
State and Region	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
• xDSL (includes UNE unbundled ADSL, HDSL and UNE	• 95% Returned <= 5 Business days
Unbundled Copper Loops)	
Unbundled Interoffice Transport	

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

O-11: Firm Order Confirmation and Reject Response Completeness

Definition

A response is expected from BellSouth for every Local Service Request transaction (version). More than one response or differing responses per transaction is not expected. Firm Order Confirmation and Reject Response Completeness is the corresponding number of Local Service Requests received to the combination of Firm Order Confirmation and Reject Responses.

Exclusions

- · Service Requests canceled by the CLEC prior to FOC or Rejected/Clarified
- Non-Mechanized LSRs
- Scheduled OSS Maintenance

Business Rules

Mechanized – The number of FOCs or Auto Clarifications sent to the CLEC from LENS, EDI, TAG in response to electronically submitted LSRs (date and time stamp in LENS, EDI, TAG).

Partially Mechanized – The number of FOCs or Rejects sent to the CLEC from LENS, EDI, TAG in response to electronically submitted LSRs (date and time stamp in LENS, EDI, TAG), which fall out for manual handling by the LCSC personnel.

Total Mechanized - The number of the combination of Fully Mechanized and Partially Mechanized LSRs

Non-Mechanized – The number of FOCs or Rejects sent to the CLEC via FAX Server in response to manually submitted LSRs (date and time stamp in FAX Server).

Note: Manual (Non-Mechanized) LSRs have no version control by the very nature of the manual process, therefore, non-mechanized LSRs are not captured by this report.

For CLEC Results:

Firm Order Confirmation and Reject Response Completeness is determined in two dimensions:

Percent responses is determined by computing the number of Firm Order Confirmations and Rejects transmitted by BellSouth and dividing by the number of Local Service Requests (all versions) received in the reporting period.

Percent of multiple responses is determined by computing the number of Local Service Request unique versions receiving more than one Firm Order Confirmation, Reject or the combination of the two and dividing by the number of Local Service Requests (all versions) received in the reporting period.

Calculation

Single FOC/Reject Response Expected

Firm Order Confirmation / Reject Response Completeness = (a / b) X 100

- a = Total Number of Service Requests for which a Firm Order Confirmation or Reject is Sent
- b = Total Number of Service Requests Received in the Report Period

Multiple or Differing FOC / Reject Responses Not Expected

Response Completeness = $[(a + b) / c] \times 100$

- a = Total Number of Firm Order Confirmations Per LSR Version
- b = Total Number of Reject Responses Per LSR Version
- c = Total Number of Service Requests (All Versions) Received in the Reporting Period

Report Structure

Fully Mechanized, Partially Mechanized, Total Mechanized, Non-Mechanized

- State and Region
- CLEC Specific
- CLEC Aggregate
- BellSouth Specific

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Not Applicable
Reject Interval	
• Total Number of LSRs	
Total Number of Rejects	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
Resale Residence	• 95% Returned
Resale Business	
Resale Design	
Resale PBX	
Resale Centrex	
Resale ISDN	
• LNP (Standalone)	
• INP (Standalone)	
2W Analog Loop Design	
2W Analog Loop Non - Design	
2W Analog Loop With INP Design	
• 2W Analog Loop With INP Non - Design	
2W Analog Loop With LNP Design	
• 2W Analog Loop With LNP Non - Design	
UNE Loop and Port Combinations	
Switch Ports	
UNE Combination Other	
• UNE xDSL (ADSL, HDSL, UCL)	
Line Sharing	
UNE ISDN Loops	
• UNE Other Design	
• UNE Other Non - Design	
Local Interoffice Transport	
Local Interconnection Trunks	

SEEM Measure

SEEM Measure			
Yes	Tier I	Х	
Tier II X			

SEEM Disaggregation	SEEM Analog/Benchmark
Fully Mechanized	• 95% Returned

O-12: Speed of Answer in Ordering Center

Definition

Measures the average time a customer is in queue.

Exclusions

None

Business Rules

The clock starts when the appropriate option is selected (i.e., 1 for Resale Consumer, 2 for Resale Multiline, and 3 for UNE-LNP, etc.) and the call enters the queue for that particular group in the LCSC. The clock stops when a BellSouth service representative in the LCSC answers the call. The speed of answer is determined by measuring and accumulating the elapsed time from the entry of a CLEC call into the BellSouth automatic call distributor (ACD) until a service representative in BellSouth's Local Carrier Service Center (LCSC) answers the CLEC call.

Calculation

Speed of Answer in Ordering Center = (a / b)

- a = Total seconds in queue
- b = Total number of calls answered in the Reporting Period

Report Structure

Aggregate

- CLEC Local Carrier Service Center
- BellSouth
 - Business Service Center
- Residence Service Center

Note: Combination of Residence Service Center and Business Service Center data.

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Mechanized tracking through LCSC Automatic Call	• Mechanized tracking through BellSouth Retail center
Distributor	support system.

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
Aggregate	• Parity with Retail
CLEC – Local Carrier Service Center	
• BellSouth	
- Business Service Center	
- Residence Service Center	

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark	
Not Applicable	Not Applicable	

O-13: LNP-Percent Rejected Service Requests

Definition

Percent Rejected Service Request is the percent of total Local Service Requests (LSRs) which are rejected due to error or omission. An LSR is considered valid when it is electronically submitted by the CLEC and passes LNP Gateway edit checks to insure the data received is correctly formatted and complete, i.e., fatal rejects are never accepted and, therefore, are not included.

Exclusions

- Service Requests canceled by the CLEC
- Scheduled OSS Maintenance

Business Rules

An LSR is considered "rejected" when it is submitted electronically but does not pass edit checks in the ordering systems (EDI, TAG, LNP Gateway, LAUTO) and is returned to the CLEC without manual intervention.

Fully Mechanized: There are two types of "Rejects" in the Fully Mechanized category:

A **Fatal Reject** occurs when a CLEC attempts to electronically submit an LSR (via EDI or TAG) but required fields are not populated correctly and the request is returned to the CLEC.

Fatal rejects are reported in a separate column, and for informational purposes ONLY. They are not considered in the calculation of the percent of total LSRs rejected or the total number of rejected LSRs.

An **Auto Clarification** is a valid LSR which is electronically submitted (via EDI or TAG), but is rejected from LAUTO because it does not pass further edit checks for order accuracy. Auto Clarifications are returned without manual intervention.

Partially Mechanized: A valid LSR which is electronically submitted (via EDI or TAG), but cannot be processed electronically due to a CLEC error and "falls out" for manual handling. It is then put into "clarification", and sent back (rejected) to the CLEC.

Total Mechanized: Combination of Fully Mechanized and Partially Mechanized rejects.

Non-Mechanized: A valid LSR which is faxed or mailed to the BellSouth LCSC.

Calculation

LNP-Percent Rejected Service Requests = (a / b) X 100

- a = Number of Service Requests Rejected in the Reporting Period
- b = Number of Service Requests Received in the Reporting Period

Report Structure

- Fully Mechanized, Partially Mechanized, Total Mechanized, Non-Mechanized
- CLEC Specific
- CLEC Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Not Applicable	Not Applicable

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
• LNP	Diagnostic
• UNE Loop With LNP	

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

O-14: LNP-Reject Interval Distribution & Average Reject Interval

Definition

Reject Interval is the average reject time from receipt of an LSR to the distribution of a Reject. An LSR is considered valid when it is electronically submitted by the CLEC and passes LNP Gateway edit checks to insure the data received is correctly formatted and complete.

Exclusions

- Service Requests canceled by the CLEC
- Designated Holidays are excluded from the interval calculation
- LSRs which are identified and classified as "Projects"
- The following hours for Partially mechanized and Non-mechanized LSRs are excluded from the interval calculation:

Residence Resale Group – Monday through Saturday 7:00PM until 7:00AM From 7:00 PM Saturday until 7:00 AM Monday

Business Resale, Complex, UNE Groups – Monday through Friday 6:00PM until 8:00AM From 6:00 PM Friday until 8:00 AM Monday.

The hours excluded will be altered to reflect changes in the Center operating hours. The LCSC will accept faxed LSRs only during posted hours of operation.

The interval will be the amount of time accrued from receipt of the LSR until normal closing of the center if an LSR is worked using overtime hours.

In the case of a Partially Mechanized LSR received and worked after normal business hours, the interval will be set at one (1) minute.

Scheduled OSS Maintenance

Business Rules

The Reject interval is determined for each rejected LSR processed during the reporting period. The Reject interval is the elapsed time from when BellSouth receives LSR until that LSR is rejected back to the CLEC. Elapsed time for each LSR is accumulated for each reporting dimension. The accumulated time for each reporting dimension is then divided by the associated total number of rejected LSRs to produce the reject interval distribution.

An LSR is considered "rejected" when it is submitted electronically but does not pass edit checks in the ordering systems (EDI, TAG, LNP Gateway, LAUTO) and is returned to the CLEC without manual intervention.

Fully Mechanized: There are two types of "Rejects" in the Fully Mechanized category:

A **Fatal Reject** occurs when a CLEC attempts to electronically submit an LSR but required fields are not populated correctly and the request is returned to the CLEC.

An **Auto Clarification** is a valid LSR which is electronically submitted (via EDI or TAG), but is rejected from LAUTO because it does not pass further edit checks for order accuracy. Auto Clarifications are returned without manual intervention.

Partially Mechanized: A valid LSR which electronically submitted (via EDI or TAG), but cannot be processed electronically due to a CLEC error and "falls out" for manual handling. It is then put into "clarification", and sent back to the CLEC.

Total Mechanized: Combination of Fully Mechanized and Partially Mechanized rejects.

Non-Mechanized: A valid LSR which is faxed or mailed to the BellSouth LCSC.

Calculation

Reject Interval = (a - b)

- a = Date & Time of Service Request Rejection
- b = Date & Time of Service Request Receipt

Average Reject Interval = (c / d)

- c = Sum of all Reject Intervals
- d = Total Number of Service Requests Rejected in Reporting Period

Reject Interval Distribution = (e / f) X 100

- e = Service Requests Rejected in reported interval
- f = Total Number of Service Requests Rejected in Reporting Period

Report Structure

Fully Mechanized, Partially Mechanized, Total Mechanized, Non-Mechanized

- CLEC Specific
- CLEC Aggregate
- State, Region
- Fully Mechanized: 0 - <= 4 minutes >4 - <= 8 minutes >8 - <= 12 minutes >12 - <= 60 minutes 0 - <= 1 hour >1 - <= 4 hours >4 - <= 8 hours >8 - <= 12 hours
- >12 <= 16 hours
- >16 <= 20 hours
- >20 <= 24 hours
- > 24 hours
- Partially Mechanized:
 - 0 <= 1 hour >1 - <= 4 hours
 - >4 <= 8 hours
 - >8 <= 10 hours
- 0 <= 10 hours >10 - <= 18 hours
- 0 <= 18 hours
- >18 <= 24 hours
- > 24 hours
- Non-Mechanized:
 - 0 <= 1 hour
- >1 <= 4 hours
- >4 <= 8 hours
- >8 <= 12 hours
- >12 <= 16 hours >16 - <= 20 hours
- >20 <= 20 hours
- 0 <= 24 hours
- >24 hours
- Average Interval in Days or Hours

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Not Applicable
Reject Interval	
Total Number of LSRs	
Total number of Rejects	
• State and Region	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark	
• LNP	• Mechanized: 97% <= I Hour	
• UNE Loop with LNP	• Partially Mechanized: 85% <= 24 Hours	
-	• Partially Mechanized: 85% <= 18 Hours (05/01/01)	
	• Partially Mechanized: 85% <= 10 Hours (08/01/01)	
	• Non-Mechanized: 85% <= 24 Hours	

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

SEEM Disaggregation	SEEM Analog/Benchmark	
Not Applicable	Not Applicable	

O-15: LNP-Firm Order Confirmation Timeliness Interval Distribution & Firm Order Confirmation Average Interval

Definition

Interval for Return of a Firm Order Confirmation (FOC Interval) is the average response time from receipt of a valid LSR to distribution of a firm order confirmation.

Exclusions

- Rejected LSRs
- Designated Holidays are excluded from the interval calculation
- LSRs which are identified and classified as "Projects"
- The following hours for Partially Mechanized and Non-mechanized LSRs are excluded from the interval calculation:

Residence Resale Group – Monday through Saturday 7:00PM until 7:00AM

From 7:00 PM Saturday until 7:00 AM Monday.

Business Resale, Complex, UNE Groups - Monday through Friday 6:00PM until 8:00AM

From 6:00 PM Friday until 8:00 AM Monday.

The hours excluded will be altered to reflect changes in the Center operating hours. The LCSC will accept faxed LSRs only during posted hours of operation.

The interval will be the amount of time accrued from receipt of the LSR until normal closing of the center if an LSR is worked using overtime hours.

In the case of a Partially Mechanized LSR received and worked after normal business hours, the interval will be set at one (1) minute.

• Scheduled OSS Maintenance

Business Rules

- Fully Mechanized: The elapsed time from receipt of a valid electronically submitted LSR (date and time stamp in EDI, LENS or TAG) until the LSR is processed, appropriate service orders are generated and a Firm Order Confirmation is returned to the CLEC via EDI, LENS or TAG.
- **Partially Mechanized:** The elapsed time from receipt of a valid electronically submitted LSR (date and time stamp in EDI, LENS, or TAG) which falls out for manual handling until appropriate service orders are issued by a BellSouth service representative via Direct Order Entry (DOE) or Service Order Negotiation Generation System (SONGS) to SOCS and a Firm Order Confirmation is returned to the CLEC via EDI, LENS, or TAG.
- Total Mechanized: Combination of Fully Mechanized and Partially Mechanized LSRs which are electronically submitted by the CLEC.
- Non-Mechanized: The elapsed time from receipt of a valid paper LSR (date and time stamp of FAX or date and time paper LSRs received in LCSC) until appropriate service orders are issued by a BellSouth service representative via Direct Order Entry (DOE) or Service Order Negotiation Generation System (SONGS) to SOCS and a Firm Order Confirmation is sent to the CLEC via LON.

Calculation

Firm Order Confirmation Interval = (a - b)

- a = Date & Time of Firm Order Confirmation
- b = Date & Time of Service Request Receipt)

Average FOC Interval = (c / d)

- c = Sum of all FOC Intervals
- d = Total Number of Service Requests Confirmed in Reporting Period

FOC Interval Distribution (for each interval) = (e / f) X 100

- e = Service Requests Confirmed in interval
- f = Total Service Requests Confirmed in the Reporting Period

Report Structure

Fully Mechanized, Partially Mechanized, Total Mechanized, Non-Mechanized

- CLEC Specific
- CLEC Aggregate
- State and Region
- Fully Mechanized:
- 0 <= 15 minutes
- >15 <= 30 minutes >30 - <= 45 minutes
- >45 <= 60 minutes
- >43 <= 60 minutes>60 - <= 90 minutes
- >90 <= 120 minutes
- >120 <= 180 minutes
- 0 <= 3 hours
- >3 <= 6 hours
- >6 <= 12 hours
- >12 <= 24 hours
- >24 <= 48 hours >48 hours
- Partially Mechanized:
- 0 <= 4 hours >4 - <= 8 hours >8 - <= 10 hours 0 - <= 10 hours >10 - <= 18 hours 0 - <= 18 hours >18 - <= 24 hours >24 - <= 48 hours >48 hours • Non-Mechanized: 0 - <= 4 hours >4 - <= 8 hours
- >4 <- 8 hours >8 - <= 12 hours >12 - <= 16 hours >16 - <= 20 hours >20 - <= 24 hours >24 - <= 36 hours >36 - <= 48 hours >48 hours

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Not Applicable
Total Number of LSRs	
 Total Number of FOCs 	
• State and Region	

SQM Level of Disaggregation	SQM Analog/Benchmark	
• LNP	• Mechanized: 95% <= 3 Hours	
• UNE Loop with LNP	• Partially Mechanized: 85% <= 24 Hours	
	• Partially Mechanized: 85% <= 18 Hours (05/01/01)	
	• Partially Mechanized: 85% <= 10 Hours (08/01/01)	
	• Non-Mechanized: 85% <= 36 hours	

SEEM Measure

SEEM Measure				
No	Tier I			
	Tier II			

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

Section 3: Provisioning

P-1: Mean Held Order Interval & Distribution Intervals

Definition

When delays occur in completing CLEC orders, the average period that CLEC orders are held for BellSouth reasons, pending a delayed completion, should be no worse for the CLEC when compared to BellSouth delayed orders. Calculation of the interval is the total days orders are held and pending but not completed that have passed the currently committed due date; divided by the total number of held orders. This report is based on orders still pending, held and past their committed due date at the close of the reporting period. The distribution interval is based on the number of orders held and pending but not completed over 15 and 90 days. (Orders reported in the >90 day interval are also included in the >15 day interval.)

Exclusions

- Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders, Test Orders, etc.)
- Disconnect (D) & From (F) orders
- Orders with appointment code of 'A' for Rural orders

Business Rules

Mean Held Order Interval: This metric is computed at the close of each report period. The held order interval is established by first identifying all orders, at the close of the reporting interval, that both have not been reported as completed in SOCS and have passed the currently committed due date for the order. For each such order, the number of calendar days between the earliest committed due date on which BellSouth had a company missed appointment and the close of the reporting period is established and represents the held order interval for that particular order. The held order interval is accumulated by the standard groupings, unless otherwise noted, and the reason for the order being held. The total number of days accumulated in a category is then divided by the number of held orders within the same category to produce the mean held order interval. The interval is by calendar days with no exclusions for Holidays or Sundays.

CLEC Specific reporting is by type of held order (facilities, equipment, other), total number of orders held, and the total and average days.

Held Order Distribution Interval: This measure provides data to report total days held and identifies these in categories of >15 days and > 90 days. (Orders counted in >90 days are also included in > 15 days).

Calculation

Mean Held Order Interval = a / b

- a = Sum of held-over-days for all Past Due Orders Held for the reporting period
- b = Number of Past Due Orders Held and Pending But Not Completed and past the committed due date

Held Order Distribution Interval (for each interval) = (c / d) X 100

- c = # of Orders Held for >= 15 days or # of Orders Held for >= 90 days
- d = Total # of Past Due Orders Held and Pending But Not Completed)

Report Structure

- CLEC Specific
- CLEC Aggregate
- BellSouth Aggregate
- Circuit Breakout < 10, >= 10 (except trunks)

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
 Report Month CLEC Order Number and PON (PON) Order Submission Date (TICKET_ID) Committed Due Date (DD) Service Type (CLASS_SVC_DESC) Hold Reason Total Line/circuit Count Geographic Scope Note: Code in parentheses is the corresponding header found 	 Report Month BellSouth Order Number Order Submission Date Committed Due Date Service Type Hold Reason Total Line/circuit Count Geographic Scope
in the raw data file.	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
Resale Residence	Retail Residence
Resale Business	Retail Business
Resale Design	Retail Design
Resale PBX	• Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
LNP (Standalone)	Retail Residence and Business (POTS)
• INP (Standalone)	Retail Residence and Business (POTS)
2W Analog Loop Design	Retail Residence and Business Dispatch
2W Analog Loop Non-Design	• Retail Residence and Business - POTS Excluding Switch-
	Based Orders
2W Analog Loop With LNP Design	Retail Residence and Business Dispatch
• 2W Analog Loop With LNP Non-Design	 Retail Residence and Business - POTS Excluding Switch- Based Orders
• 2W Analog Loop With INP-Design	Retail Residence and Business Dispatch
• 2W Analog Loop With INP Non-Design	• Retail Residence and Business - POTS Excluding Switch-
	Based Orders
• UNE Digital Loop < DS1	• Retail Digital Loop < DS1
• UNE Digital Loop >= DS1	• Retail Digital Loop >= DS1
UNE Loop + Port Combinations	Retail Residence and Business
UNE Switch Ports	Retail Residence and Business (POTS)
UNE Combo Other	Retail Residence, Business and Design Dispatch
• UNE xDSL (HDSL, ADSL and UCL)	ADSL Provided to Retail
• UNE ISDN	• Retail ISDN - BRI
UNE Line Sharing	ADSL Provided to Retail
• UNE Other Design	Retail Design
• UNE Other Non-Design	Retail Residence and Business
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice
Local Interconnection Trunks	Parity with Retail

SEEM Measure

SEEM Measure			
No	Tier I		
Tier II			

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

P-2: Average Jeopardy Notice Interval & Percentage of Orders Given Jeopardy Notices

Definition

When BellSouth can determine in advance that a committed due date is in jeopardy for facility delay, it will provide advance notice to the CLEC.

The interval is from the date/time the notice is released to the CLEC/BellSouth systems until 5pm on the commitment date of the order. The Percent of Orders is the percentage of orders given jeopardy notices for facility delay in the count of orders confirmed in the report period.

Exclusions

- · Orders held for CLEC end user reasons
- Disconnect (D) & From (F) orders
- Non-Dispatch Orders

Business Rules

When BellSouth can determine in advance that a committed due date is in jeopardy for facility delay, it will provide advance notice to the CLEC. The number of committed orders in a report period is the number of orders that have a due date in the reporting period. Jeopardy notices for interconnection trunks results are usually zero as these trunks seldom experience facility delays. The Committed due date is considered the Confirmed due date. This report measures dispatched orders only. If an order is originally sent as non-dispatch and it is determined there is a facility delay, the order is converted to a dispatch code so the facility problem can be corrected. It will remain coded dispatched until completion.

Calculation

Jeopardy Interval = a - b

- a = Date and Time of Jeopardy Notice
- b = Date and Time of Scheduled Due Date on Service Order

Average Jeopardy Interval = c / d

- c = Sum of all jeopardy intervals
- d = Number of Orders Notified of Jeopardy in Reporting Period

Percent of Orders Given Jeopardy Notice = (e / f) X 100

- e = Number of Orders Given Jeopardy Notices in Reporting Period
- f = Number of Orders Confirmed (due) in Reporting Period)

Report Structure

- CLEC Specific
- CLEC Aggregate
- BellSouth Aggregate
- Dispatch Orders
- Mechanized Orders
- Non-Mechanized Orders

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
 Report Month CLEC Order Number and PON Date and Time Jeopardy Notice Sent Committed Due Date Service Type Note: Code in parentheses is the corresponding header found in the raw data file. 	 Report Month BellSouth Order Number Date and Time Jeopardy Notice Sent Committed Due Date Service Type

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
% Orders Given Jeopardy Notice	
Resale Residence	Retail Residence
Resale Business	Retail Business
Resale Design	Retail Design
Resale PBX	Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
LNP (Standalone)	Retail Residence and Business (POTS)
• INP (Standalone)	Retail Residence and Business (POTS)
2W Analog Loop Design	Retail Residence and Business Dispatch
2W Analog Loop Non-Design	Retail Residence and Business - (POTS Excluding
	Switch- Based Orders)
2W Analog Loop With LNP Design	Retail Residence and Business Dispatch
2W Analog Loop With LNP Non-Design	Retail Residence and Business - (POTS Excluding
	Switch- Based Orders)
• 2W Analog Loop With INP Design	 Retail Residence and Business Dispatch
• 2W Analog Loop With INP Non-Design	• Retail Residence and Business (POTS Excluding Switch-
	Based Orders)
•UNE Digital Loop < DS1	• Retail Digital Loop < DS1
•UNE Digital Loop >= DS1	• Retail Digital Loop >= DS1
•UNE Loop + Port Combinations	Retail Business and Residence
•UNE Switch Ports	• Retail Residence and Business (POTS)
•UNE Combo Other	Retail Residence, Business and Design Dispatch
•UNE xDSL (HDSL, ADSL and UCL)	ADSL Provided to Retail
•UNE ISDN	• Retail ISDN BRI
•UNE Line Sharing	ADSL Provided to Retail
•UNE Other Design	Retail Design
•UNE Other Non -Design	Retail Residence and Business
•Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice
Local Interconnection Trunks	Parity with Retail
•Average Jeopardy Notice Interval	• 95% >= 48 Hours

SEEM Measure

SEEM Measure			
No	Tier I		
Tier II			

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

P-3: Percent Missed Installation Appointments

Definition

"Percent missed installation appointments" monitors the reliability of BellSouth commitments with respect to committed due dates to assure that the CLEC can reliably quote expected due dates to their retail customer as compared to BellSouth. This measure is the percentage of total orders processed for which BellSouth is unable to complete the service orders on the committed due dates and reported for Total misses and End User Misses.

Exclusions

- Canceled Service Orders
- Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders Test Orders, etc.)
- Disconnect (D) & From (F) orders
- End User Misses on Local Interconnection Trunks

Business Rules

Percent Missed Installation Appointments (PMI) is the percentage of orders with completion dates in the reporting period that are past the original committed due date. Missed Appointments caused by end-user reasons will be included and reported separately. The first commitment date on the service order that is a missed appointment is the missed appointment code used for calculation whether it is a BellSouth missed appointment or an End User missed appointment. The "due date" is any time on the confirmed due date. Which means there cannot be a cutoff time for commitments, as certain types of orders are requested to be worked after standard business hours. Also, during Daylight Savings Time, field technicians are scheduled until 9PM in some areas and the customer is offered a greater range of intervals from which to select.

Calculation

Percent Missed Installation Appointments = (a / b) X 100

- a = Number of Orders with Completion date in Reporting Period past the Original Committed Due Date
- b = Number of Orders Completed in Reporting Period

Report Structure

- CLEC Specific
- CLEC Aggregate
- BellSouth Aggregate
- Report in Categories of <10 lines/circuits >= 10 lines/circuits (except trunks)
- Dispatch/No Dispatch

Report Explanation: The difference between End User MA and Total MA is the result of BellSouth caused misses. Here, Total MA is the total percent of orders missed either by BellSouth or CLEC end user. The End User MA represents the percentage of orders missed by the CLEC or their end user.

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
 Report Month CLEC Order Number and PON (PON) Committed Due Date (DD) Completion Date (CMPLTN DD) Status Type Status Notice Date Standard Order Activity Geographic Scope Note: Code in parentheses is the corresponding header four in the raw data file. 	 Report Month BellSouth Order Number Committed Due Date (DD) Completion Date (CMPLTN DD) Status Type Status Notice Date Standard Order Activity Geographic Scope

SQM LEVEL of Disaggregation	SQM Analog/Benchmark
Resale Residence	Retail Residence
Resale Business	Retail Business
Resale Design	Retail Design
Resale PBX	Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
• LNP (Standalone)	Retail Residence and Business (POTS)
• INP (Standalone)	Retail Residence and Business (POTS)
2W Analog Loop Design	Retail Residence and Business Dispatch
2W Analog Loop Non-Design	Retail Residence and Business - (POTS Excluding
	Switch-Based Orders)
- Dispatch	- Dispatch
- Non-Dispatch (Dispatch In)	- Non-Dispatch (Dispatch In)
2W Analog Loop With LNP Design	Retail Residence and Business Dispatch
2W Analog Loop With LNP Non-Design	Retail Residence and Business - (POTS Excluding
	Switch-Based Orders)
- Dispatch	- Dispatch
- Non-Dispatch (Dispatch In)	- Non-Dispatch (Dispatch In)
2W Analog Loop With INP Design	Retail Residence and Business Dispatch
2W Analog Loop With INP Non-Design	Retail Residence and Business (POTS Excluding Switch
	Based Orders)
- Dispatch	- Dispatch
- Non-Dispatch (Dispatch In)	- Non-Dispatch (Dispatch In)
• UNE Digital Loop < DS1	• Retail Digital Loop < DS1
• UNE Digital Loop >= DS1	• Retail Digital Loop >= DS1
UNE Loop + Port Combinations	Retail Residence and Business
- Dispatch Out	- Dispatch Out
- Non-Dispatch	- Non-Dispatch
- Dispatch In	- Dispatch In
- Switch-Based	- Switch-Based
UNE Switch Ports	• Retail Residence and Business (POTS)
UNE Combo Other	• Retail Residence, Business and Design Dispatch
	(Including Dispatch Out and Dispatch In)
- Dispatch	- Dispatch
- Non-Dispatch (Dispatch In)	- Non-Dispatch (Dispatch In)
• UNE xDSL (HDSL, ADSL and UCL)	ADSL Provided to Retail
• UNE ISDN	• Retail ISDN - BRI
UNE Line Sharing	ADSL Provided to Retail
UNE Other Design	• Retail Design
UNE Other Non - Design	Retail Residence and Business
• Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice
Local Interconnection Trunks	Parity with Retail

SQM Disaggregation - Analog/Benchmark

SEEM Measure

SEEM Measure			
Yes Tier I X			
Tier II X			

SEEM Disaggregation	SEEM Analog/Benchmark
Resale POTS	 Retail Residence and Business (POTS)
Resale Design	Retail Design
UNE Loop + Port Combinations	Retail Residence and Business
UNE Loops	Retail Residence and Business Dispatch
• UNE xDSL	ADSL Provided to Retail
UNE Line Sharing	ADSL Provided to Retail
Local Interconnection Trunks	Parity with Retail

P-4: Average Completion Interval (OCI) & Order Completion Interval Distribution

Definition

The "average completion interval" measure monitors the interval of time it takes BellSouth to provide service for the CLEC or its own customers. The "Order Completion Interval Distribution" provides the percentages of orders completed within certain time periods. This report measures how well BellSouth meets the interval offered to customers on service orders.

Exclusions

- Canceled Service Orders
- Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders, Test Orders, etc.)
- Disconnect (D&F) orders (Except "D" orders associated with LNP Standalone)
- "L" Appointment coded orders (where the customer has requested a later than offered interval)

Business Rules

The actual completion interval is determined for each order processed during the reporting period. The completion interval is the elapsed time from when BellSouth issues a FOC or SOCS date time stamp receipt of an order from the CLEC to BellSouth's actual order completion date. This includes all delays for BellSouth's CLEC/End Users. The clock starts when a valid order number is assigned by SOCS and stops when the technician or system completes the order in SOCS. Elapsed time for each order is accumulated for each reporting dimension. The accumulated time for each reporting dimension is then divided by the associated total number of orders completed. Orders that are worked on zero due dates are calculated with a .33-day interval (8 hours) in order to report a portion of a day interval. These orders are issued and worked/completed on the same day. They can be either flow through orders (no field work-non-dispatched) or field orders (dispatched).

The interval breakout for UNE and Design is: 0.5 = 0.4.99, 5.10 = 5.9.99, 10-15 = 10-14.99, 15-20 = 15-19.99, 20-25 = 20-24.99, 25-30 = 25-29.99, >= 30 = 30 and greater.

Calculation

Completion Interval = (a - b)

- a = Completion Date
- b = Order Issue Date

Average Completion Interval = (c / d)

- c = Sum of all Completion Intervals
- d = Count of Orders Completed in Reporting Period

Order Completion Interval Distribution (for each interval) = (e / f) X 100

- e = Service Orders Completed in "X" days
- f = Total Service Orders Completed in Reporting Period

Report Structure

- CLEC Specific
- CLEC Aggregate
- BellSouth Aggregate
- Dispatch / No Dispatch categories applicable to all levels except trunks
- Residence & Business reported in day intervals = 0, 1, 2, 3, 4, 5, 5+
- UNE and Design reported in day intervals = 0-5, 5-10, 10-15, 15-20, 20-25, 25-30,>= 30
- All Levels are reported <10 line/circuits; >= 10 line/circuits (except trunks)
- ISDN Orders included in Non-Design

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
• CLEC Company Name	 Report Month BellSouth Order Number

 Application Date & Time (TICKET_ID) Completion Date (CMPLTN_DT) Service Type (CLASS_SVC_DESC) Geographic Scope 	 Application Date & Time Order Completion Date & Time Service Type Geographic Scope
Note: Code in parentheses is the corresponding header found in the raw data file.	

SQM Level of Disaggregation	SQM Analog/Benchmark
Resale Residence	Retail Residence
Resale Business	Retail Business
Resale Design	Retail Design
Resale PBX	Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
• LNP (Standalone)	Retail Residence and Business (POTS)
• INP (Standalone)	Retail Residence and Business (POTS)
2W Analog Loop Design	Retail Residence and Business Dispatch
2W Analog Loop Non-Design	• Retail Residence and Business - (POTS Excluding Switch-
	Based Orders)
- Dispatch	- Dispatch
- Non-Dispatch (Dispatch In)	- Non-Dispatch (Dispatch In)
2W Analog Loop With LNP Design	Retail Residence and Business Dispatch
• 2W Analog Loop With LNP Non-Design	• Retail Residence and Business - (POTS Excluding Switch-
	Based Orders)
- Dispatch	- Dispatch
- Non-Dispatch (Dispatch In)	- Non-Dispatch (Dispatch In)
2W Analog Loop With INP Design	Retail Residence and Business Dispatch
 2W Analog Loop With INP Non-Design 	• Retail Residence and Business - (POTS Excluding Switch-
	Based Orders)
- Dispatch	- Dispatch
- Non-Dispatch (Dispatch In)	- Non-Dispatch (Dispatch In)
• UNE Digital Loop < DS1	• Retail Digital Loop < DS1
• UNE Digital Loop >= DS1	• Retail Digital Loop >= DS1
• UNE Loop + Port Combinations	 Retail Residence and Business
- Dispatch Out	- Dispatch Out
- Non-Dispatch	- Non-Dispatch
- Dispatch In	- Dispatch In
- Switch-Based	- Switch-Based
UNE Switch Ports	Retail Residence and Business (POTS)
UNE Combo Other	• Retail Residence, Business and Design Dispatch
	(Including Dispatch Out and Dispatch In)
- Dispatch	- Dispatch
- Non-Dispatch (Dispatch In)	- Non-Dispatch (Dispatch In)
• UNE xDSL (HDSL, ADSL and UCL) without	• 7 Days
conditioning	
• UNE xDSL (HDSL, ADSL and UCL) with conditioning	• 14 Days
• UNE ISDN	Retail ISDN BRI
UNE Line Sharing	ADSL Provided to Retail
UNE Other Design	Retail Design
UNE Other Non-Design	Retail Residence and Business
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice
Local Interconnection Trunks	• Parity with Retail
SEEM Measure

SEEM Measure		
Yes	Tier I	Х
	Tier II	Х

SEEM Disaggregation	SEEM Analog/Benchmark
Resale POTS	• Retail Residence and Business (POTS)
Resale Design	Retail Design
• UNE Loop + Port Combinations	Retail Residence and Business
UNE Loops	Retail Residence and Business Dispatch
UNE xDSL without conditioning	• 7 Days
• UNE xDSL with conditioning	• 14 Days
UNE Line Sharing	ADSL Provided to Retail
Local Interconnection Trunks	Parity with Retail

P-5: Average Completion Notice Interval

Definitions

The Completion Notice Interval is the elapsed time between the BellSouth reported completion of work and the issuance of a valid completion notice to the CLEC.

Exclusions

- Cancelled Service Orders
- Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders, Test Orders, etc.)
- D&F orders (Exception: "D" orders associated with LNP Standalone)

Business Rules

Measurement on interval of completion date and time entered by a field technician on dispatched orders, and 5PM start time on the due date for non-dispatched orders; to the release of a notice to the CLEC/BellSouth of the completion status. The field technician notifies the CLEC the work was complete and then he/she enters the completion time stamp information in his/her computer. This information switches through to the SOCS systems either completing the order or rejecting the order to the Work Management Center (WMC). If the completion is rejected, it is manually corrected and then completed by the WMC. The notice is returned on each individual order.

The start time for all orders is the completion stamp either by the field technician or the 5PM due date stamp; the end time for mechanized orders is the time stamp the notice was transmitted to the CLEC interface (LENS, EDI, OR TAG). For non-mechanized orders the end timestamp will be timestamp of order update to C-SOTS system.

Calculation

Completion Notice Interval = (a - b)

- a = Date and Time of Notice of Completion
- b = Date and Time of Work Completion

Average Completion Notice Interval = c / d

- c = Sum of all Completion Notice Intervals
- d = Number of Orders with Notice of Completion in Reporting Period

Report Structure

- CLEC Specific
- CLEC Aggregate
- BellSouth Aggregate
- Mechanized Orders
- Non-Mechanized Orders
- Reporting intervals in Hours; 0, 1-2, 2-4, 4-8, 8-12, 12-24, >= 24 plus Overall Average Hour Interval (The categories are inclusive of these time intervals: 0-1 = 0.99; 1-2 =1-1.99; 2-4 = 2-3.99, etc.)
- Reported in categories of <10 line/circuits; >= 10 line/circuits (except trunks)

 Report Month CLEC Order Number (so_nbr) Work Completion Date (cmpltn_dt) Work Completion Time Completion Notice Availability Date Service Type Report Month BellSouth Order Number (so_nbr) Work Completion Date (cmpltn_dt) Work Completion Time Completion Notice Availability Time Completion Notice Availability Time Completion Notice Availability Time 	Relating to CLEC Experience	Relating to BellSouth Performance
 Geographic Scope Note: Code in parentheses is the corresponding header found Note: Code in parentheses is the corresponding header found NOTE: Code in parentheses is the corresponding header 	 CLEC Order Number (so_nbr) Work Completion Date (cmpltn_dt) Work Completion Time Completion Notice Availability Date Completion Notice Availability Time Service Type Geographic Scope 	 BellSouth Order Number (so_nbr) Work Completion Date (cmpltn_dt) Work Completion Time Completion Notice Availability Date Completion Notice Availability Time Service Type Geographic Scope

in the raw data file.

found in the raw data file.

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
Resale Residence	Retail Residence
Resale Business	Retail Business
Resale Design	• Retail Design
Resale PBX	• Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	• Retail ISDN
• LNP (Standalone)	• Retail Residence and Business (POTS)
• INP (Standalone)	• Retail Residence and Business (POTS)
2W Analog Loop Design	 Retail Residence and Business Dispatch
 2W Analog Loop Non-Design 	• Retail Residence and Business - (POTS Excluding Switch-
	Based Orders)
- Dispatch	- Dispatch
- Non-Dispatch (Dispatch In)	- Non-Dispatch (Dispatch In)
 2W Analog Loop With LNP Design 	 Retail Residence and Business Dispatch
 2W Analog Loop With LNP Non-Design 	• Retail Residence and Business - (POTS Excluding Switch-
	Based Orders)
- Dispatch	- Dispatch
- Non-Dispatch (Dispatch In)	- Non-Dispatch (Dispatch In)
2W Analog Loop With INP Design	Retail Residence and Business Dispatch
 2W Analog Loop With INP Non-Design 	Retail Residence and Business (POTS Excluding Switch-
	Based Orders)
- Dispatch	- Dispatch
- Non-Dispatch (Dispatch In)	- Non-Dispatch (Dispatch In)
• UNE Digital Loop < DS1	• Retail Digital Loop < DS1
• UNE Digital Loop >= DS1	• Retail Digital Loop >= DS1
 UNE Loop + Port Combinations 	Retail Residence and Business
- Dispatch Out	- Dispatch Out
- Non-Dispatch	- Non-Dispatch
- Dispatch In	- Dispatch In
- Switch-Based	- Switch-Based
UNE Switch Ports	Retail Residence and Business (POTS)
UNE Combo Other	Retail Residence, Business and Design Dispatch (Including
	Dispatch Out and Dispatch In)
- Dispatch	- Dispatch
- Non-Dispatch (Dispatch In)	- Non-Dispatch (Dispatch In)
• UNE xDSL (HDSL, ADSL and UCL)	ADSL Provided to Retail
• UNE ISDN	Retail ISDN BRI
UNE Line Sharing	ADSL Provided to Retail
UNE Other Design	Retail Design
UNE Other Non-Design	Retail Residence and Business
 Local Transport (Unbundled Interoffice Transport 	
Local Interconnection Trunks	• Parity with Retail

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

P-6: % Completions/Attempts without Notice or < 24 hours Notice

Definition

This Report measures the interval from the FOC end timestamp on the LSR until 5:00 P.M. on the original committed due date of a service order. The purpose of this measure is to report if BellSouth is returning a FOC to the CLEC in time for the CLEC to notify their customer of the scheduled date.

Exclusions

"0" dated orders or any request where the subscriber requested an earlier due date of < 24 hours prior to the original commitment date, or any LSR received < 24 hours prior to the original commitment date.

Business Rules

For CLEC Results:

Calculation would exclude any successful or unsuccessful service delivery where the CLEC was informed at least 24 hours in advance. BellSouth may also exclude from calculation any LSRs received from the requesting CLEC with less than 24 hour notice prior to the commitment date.

For BellSouth Results:

BellSouth does not provide a FOC to its retail customers.

Calculation

Percent Completions or Attempts without Notice or with Less Than 24 Hours Notice = $(a / b) \times 100$

- a = Completion Dispatches (Successful and Unsuccessful) With No FOC or FOC Received < 24 Hours of original Committed Due Date
- b = All Completions

Report Structure

- CLEC Specific
- CLEC Aggregate
- Dispatch /Non-Dispatch
- Total Orders FOC < 24 Hours
- Total Completed Service Orders
- % FOC < 24 Hours

Relating to CLEC Experience	Relating to BellSouth Performance
Committed Due Date (DD)	Not Applicable
FOC End Timestamp	
Report Month	
CLEC Order Number and PON	
Geographic Scope	
- State / Region	

SQM Level of Disaggregation	SQM Analog/Benchmark
Resale Residence	Diagnostic
Resale Business	
Resale Design	
Resale PBX	
Resale Centrex	
Resale ISDN	
• LNP (Standalone)	
• INP (Standalone)	
2W Analog Loop Design	
 2W Analog Loop Non-Design 	
2W Analog Loop With LNP-Design	
 2W Analog Loop With LNP Non-Design 	
 2W Analog Loop With INP-Design 	
 2W Analog Loop With INP Non-Design 	
• UNE Digital Loop < DS1	
• UNE Digital Loop >=DS1	
 UNE Loop + Port Combinations 	
UNE Switch ports	
UNE Combo Other	
• UNE xDSL (HDSL, ADSL and UCL)	
• UNE ISDN	
UNE Line Sharing	
UNE Other Design	
• UNE Other Non -Design	
Local Transport (Unbundled Interoffice Transport)	
Local Interconnection Trunks	

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

P-7: Coordinated Customer Conversions Interval

Definition

This report measures the average time it takes BellSouth to disconnect an unbundled loop from the BellSouth switch and cross connect it to CLEC equipment. This measurement applies to service orders with INP and with LNP, and where the CLEC has requested BellSouth to provide a coordinated cut over.

Exclusions

- Any order canceled by the CLEC will be excluded from this measurement
- Delays due to CLEC following disconnection of the unbundled loop
- Unbundled Loops where there is no existing subscriber loop and loops where coordination is not requested

Business Rules

When the service order includes INP, the interval includes the total time for the cut over including the translation time to place the line back in service on the ported line. When the service order includes LNP, the interval only includes the total time for the cut over (the port of the number is controlled by the CLEC). The interval is calculated for the entire cut over time for the service order and then divided by items worked in that time to give the average per-item interval for each service order.

Calculation

Coordinated Customer Conversions Interval = (a - b)

- a = Completion Date and Time for Cross Connection of a Coordinated Unbundled Loop
- b = Disconnection Date and Time of an Coordinated Unbundled Loop

Percent Coordinated Customer Conversions (for each interval) = (c / d) X 100

- c = Total number of Coordinated Customer Conversions for each interval
- d = Total Number of Unbundled Loop with Coordinated Conversions (items) for the reporting period

Report Structure

- CLEC Specific
- CLEC Aggregate
- The interval breakout is 0.5 = 0.4.99, 5.15 = 5.14.99, >=15 = 15 and greater, plus Overall Average Interval.

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	No BellSouth Analog Exists
CLEC Order Number	• No Densouth Analog Exists
• Committed Due Date (DD)	
• Service Type (CLASS_SVC_DESC)	
Cut over Start Time	
Cut over Completion Time	
• Portability Start and Completion Times (INP orders)	
Total Conversions (Items)	
Note: Code in parentheses is the corresponding header found in the raw data file.	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
Unbundled Loops with INP/LNP	• 95% <= 15 minutes
• Unbundled Loops without INP/LNP	

SEEM Measure

SEEM Measure		
Yes	Tier I	X
	Tier II	Х

SEEM Disaggregation	SEEM Analog/Benchmark
Unbundled Loops	• 95% <= 15 minutes

P-7A: Coordinated Customer Conversions – Hot Cut Timeliness% Within Interval and Average Interval

Definition

This category measures whether BellSouth begins the cut over of an unbundled loop on a coordinated and/or a time specific order at the CLEC requested start time. It measures the percentage of orders where the cut begins within 15 minutes of the requested start time of the order and the average interval.

Exclusions

- Any order canceled by the CLEC will be excluded from this measurement
- Delays caused by the CLEC
- Unbundled Loops where there is no existing subscriber loop and loops where coordination is not requested
- All unbundled loops on multiple loop orders after the first loop

Business Rules

This report measures whether BellSouth begins the cut over of an unbundled loop on a coordinated and/or a time specific order at the CLEC requested start time. The cut is considered on time if it starts 15 minutes before or after the requested start time. Using the scheduled time and the actual cut over start time, the measurement will calculate the percent within interval and the average interval. If a cut involves multiple lines, the cut will be considered "on time" if the first line is cut within the interval. <= 15 minutes includes intervals that began 15:00 minutes or less before the scheduled cut time and cuts that began 15 minutes or less after the scheduled cut time; >15 minutes, <= 30 minutes includes cuts within 15:00 – 30:00 minutes either prior to or after the scheduled cut time; >30 minutes includes cuts greater than 30:00 minutes either prior to or after the scheduled cut time.

Calculation

% within Interval = (a / b) X 100

- a = Total Number of Coordinated Unbundled Loop Orders for the interval
- b = Total Number of Coordinated Unbundled Loop Orders for the reporting period

Interval = (c - d)

- c = Scheduled Time for Cross Connection of a Coordinated Unbundled Loop Order
- d = Actual Start Date and Time of a Coordinated Unbundled Loop Order

Average Interval = (e / f)

- Sum of all Intervals
- Total Number of Coordinated Unbundled Loop Orders for the reporting period.

Report Structure

- CLEC Specific
- CLEC Aggregate

Reported in intervals of early, on time and late cuts % <=15 minutes; % >15 minutes, <= 30 minutes; % > 30 minutes, plus Overall Average Interval.

Relating to CLEC Experience	Relating to BellSouth Performance
 Report Month CLEC Order Number (so_nbr) Committed Due Date (DD) Service Type (CLASS_SVC_DESC) Cut over Scheduled Start Time Cut over Actual Start Time Total Conversions Orders 	• No BellSouth Analog exists
Note: Code in parentheses is the corresponding header found in the raw data file.	

SQM Level of Disaggregation	SQM Analog/Benchmark
Product Reporting Level	• 95% Within + or – 15 minutes of Scheduled Start Time
- SL1 Time Specific	
- SL1 Non-Time Specific	
- SL2 Time Specific	
- SL2 Non-Time Specific	

SEEM Measure

SEEM Measure			
Yes	Tier I	Х	
	Tier II	Х	

SEEM Disaggregation	SEEM Analog/Benchmark
• UNE Loops	• 95% Within + or – 15 minutes of Scheduled Start time

P-7B: Coordinated Customer Conversions – Average Recovery Time

Definition

Measures the time between notification and resolution by BellSouth of a service outage found that can be isolated to the BellSouth side of the network. The time between notification and resolution by BellSouth must be measured to ensure that CLEC customers do not experience unjustifiable lengthy service outages during a Coordinated Customer Conversion. This report measures outages associated with Coordinated Customer Conversions prior to service order completion.

Exclusions

- Cut overs where service outages are due to CLEC caused reasons
- Cut overs where service outages are due to end-user caused reasons

Business Rules

Measures the outage duration time related to Coordinated Customer Conversions from the initial trouble notification until the trouble has been restored and the CLEC has been notified. The duration time is defined as the time from the initial trouble notification until the trouble has been restored and the CLEC has been notified. The interval is calculated on the total outage time for the circuits divided by the total number of outages restored during the report period to give the average outage duration.

Calculation

Recovery Time = (a - b)

- a = Date & Time That Trouble is Closed by CLEC
- b = Date & Time Initial Trouble is Opened with BellSouth

Average Recovery Time = (c / d)

- c = Sum of all the Recovery Times
- d = Number of Troubles Referred to the BellSouth

Report Structure

- CLEC Specific
- CLEC Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
• Report Month	• None
CLEC Company Name	· None
• CLEC Order Number (so_nbr)	
• Committed Due Date (DD)	
• Service Type (CLASS_SVC_DESC)	
CLEC Acceptance Conflict (CLEC_CONFLICT)	
• CLEC Conflict Resolved (CLEC_RESOLVE)	
CLEC Conflict MFC (CLEC_CONFLICT_MFC)	
Total Conversion Orders	
Note: Code in parentheses is the corresponding header found	
in the raw data file.	

SQM Level of Disaggregation	SQM Analog/Benchmark
Unbundled Loops with INP/LNP	Diagnostic
 Unbundled Loops without INP/LNP 	

SEEM Measure

SEEM Measure				
No	Tier I			
	Tier II			

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

P-7C: Hot Cut Conversions - % Provisioning Troubles Received Within 7 days of a completed Service Order

Definition

Percent Provisioning Troubles received within 7 days of a completed service order associated with a Coordinated and Non-Coordinated Customer Conversion. Measures the quality and accuracy of Hot Cut Conversion Activities.

Exclusions

- Any order canceled by the CLEC
- Troubles caused by Customer Provided Equipment

Business Rules

Measures the quality and accuracy of completed service orders associated with Coordinated and Non-Coordinated Hot Cut Conversions. The first trouble report received on a circuit ID within 7 days following a service order completion is counted in this measure. Subsequent trouble reports are measured in Repeat Report Rate. Reports are calculated searching in the prior report period for completed Coordinated and Non-Coordinated Hot Cut Conversion service orders and following 7 days after the completion of the service order for a trouble report issue date.

Calculation

% Provisioning Troubles within 7 days of service order completion = (a / b) X 100

- a = The sum of all Hot Cut Circuits with a trouble within 7 days following service order(s) completion
- b = The total number of Hot Cut service order circuits completed in the previous report calendar month

Report Structure

- CLEC Specific
- CLEC Aggregate
- Dispatch/Non-Dispatch

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
 Report Month CLEC Order Number (so_nbr) PON Order Submission Date (TICKET_ID) 	No BellSouth Analog Exists
 Order Submission Time (TICKET_ID) Status Type Status Notice Date 	
 Standard Order Activity Geographic Scope Total Conversion Circuits 	
Note: Code in parentheses is the corresponding header found in the raw data file.	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
UNE Loop Design	• <= 5%
UNE Loop Non-Design	

SEEM Measure

SEEM Measure		
Yes	Tier I	Х
	Tier II	Х

SEEM Disaggregation	SEEM Analog/Benchmark
UNE Loops	• <= 5%

P-8: Cooperative Acceptance Testing - % of xDSL Loops Tested

Definition

The loop will be considered cooperatively tested when the BellSouth technician places a call to the CLEC representative to initiate cooperative testing and jointly performs the tests with the CLEC.

Exclusions

- Testing failures due to CLEC (incorrect contact number, CLEC not ready, etc.)
- xDSL lines with no request for cooperative testing

Business Rules

When a BellSouth technician finishes delivering an order for an xDSL loop where the CLEC order calls for cooperative testing at the customer's premise, the BellSouth technician is to call a toll free number to the CLEC testing center. The BellSouth technician and the CLEC representative at the center then test the line. As an example of the type of testing performed, the testing center may ask the technician to put a short on the line so that the center can run a test to see if it can identify the short.

Calculation

Cooperative Acceptance Testing - % of xDSL Loops Tested = $(a / b) \ge 100$

- a = Total number of successful xDSL cooperative tests for xDSL lines where cooperative testing was requested in the reporting period
- \hat{b} = Total Number of xDSL line tests requested by the CLEC and scheduled in the reporting period

Report Structure

- CLEC Specific
- CLEC Aggregate
- Type of Loop tested

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
 Report Month CLEC Company Name (OCN) CLEC Order Number (so_nbr) and PON (PON) Committed Due Date (DD) Service Type (CLASS_SVC_DESC) Acceptance Testing Completed (ACCEPT_TESTING) Acceptance Testing Declined (ACCEPT_TESTING) Total xDSL Orders 	• No BellSouth Analog Exists
Note : Code in parentheses is the corresponding header found in the raw data file.	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation:	SQM Analog/Benchmark:
• UNE xDSL	• 95% of Lines Tested
- ADSL	
- HDSL	
- UCL	
- OTHER	

SEEM Measure

SEEM Measure		
Yes	Tier I	X
	Tier II	X

SEEM Disaggregation	SEEM Analog/Benchmark
• UNE xDSL	• 95% of Lines Tested

P-9: % Provisioning Troubles within 30 days of Service Order Completion

Definition

Percent Provisioning Troubles within 30 days of Service Order Completion measures the quality and accuracy of Service order activities.

Exclusions

- Canceled Service Orders
- Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders, Test Orders, etc.)
- D & F orders
- Trouble reports caused and closed out to Customer Provided Equipment (CPE)

Business Rules

Measures the quality and accuracy of completed orders. The first trouble report from a service order after completion is counted in this measure. Subsequent trouble reports are measured in Repeat Report Rate. Reports are calculated searching in the prior report period for completed service orders and following 30 days after completion of the service order for a trouble report issue date.

D & F orders are excluded as there is no subsequent activity following a disconnect.

Note: Standalone LNP historical data is not available in the maintenance systems (LMOS or WFA).

Calculation

% Provisioning Troubles within 30 days of Service Order Activity = (a / b) X 100

- a = Trouble reports on all completed orders 30 days following service order(s) completion
- b = All Service Orders completed in the previous report calendar month

Report Structure

- CLEC Specific
- CLEC Aggregate
- BellSouth Aggregate
- Reported in categories of <10 line/circuits; >= 10 line/circuits (except trunks)
- Dispatch / No Dispatch (except trunks)

Relating to CLEC Experience	Relating to BellSouth Performance
 Report Month CLEC Order Number and PON Order Submission Date (TICKET_ID) Order Submission Time (TICKET_ID) Status Type Status Notice Date Standard Order Activity Geographic Scope Note: Code in parentheses is the corresponding header found in the raw data file. 	 Report Month BellSouth Order Number Order Submission Date Order Submission Time Status Type Status Notice Date Standard Order Activity Geographic Scope

SQM Level of Disaggregation	SQM Analog/Benchmark
Resale Residence	Retail Residence
Resale Business	Retail Business
Resale Design	• Retail Design
Resale PBX	• Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
2W Analog Loop Design	Retail Residence and Business Dispatch
2W Analog Loop Non-Design	Retail Residence and Business - (POTS Excluding Switch-
	Based Orders)
- Dispatch	- Dispatch
- Non-Dispatch (Dispatch In)	- Non-Dispatch (Dispatch In)
2W Analog Loop With LNP Design	Retail Residence and Business Dispatch
2W Analog Loop With LNP Non-Design	Retail Residence and Business - (POTS Excluding Switch-
	Based Orders)
- Dispatch	- Dispatch
- Non-Dispatch (Dispatch In)	- Non-Dispatch (Dispatch In)
2W Analog Loop With INP Design	Retail Residence and Business Dispatch
2W Analog Loop With INP Non-Design	• Retail Residence and Business (POTS - Excluding Switch-
	Based Orders)
- Dispatch	- Dispatch
- Non-Dispatch (Dispatch In)	- Non-Dispatch (Dispatch In)
• UNE Digital Loop < DS1	• Retail Digital Loop < DS1
• UNE Digital Loop >= DS1	• Retail Digital Loop >= DS1
• UNE xDSL (HDSL, ADSL and UCL)	ADSL provided to Retail
• UNE ISDN	Retail ISDN BRI
UNE Line Sharing	ADSL Provided to Retail
• INP (Standalone)	• Retail Residence and Business (POTS)
• LNP (Standalone)	Retail Residence and Business (POTS)
• UNE Loop + Port Combinations	 Retail Residence and Business
- Dispatch Out	- Dispatch Out
- Non-Dispatch	- Non-Dispatch
- Dispatch In	- Dispatch In
- Switch-Based	- Switch-Based
UNE Switch Ports	Retail Residence and Business (POTS)
UNE Combo Other	• Retail Residence, Business and Design Dispatch
	(Including Dispatch Out and Dispatch In)
- Dispatch	- Dispatch
- Non-Dispatch (Dispatch In)	- Non-Dispatch (Dispatch In)
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice
UNE Other Non-Design	Retail Residence and Business
UNE Other Design	Retail Design
Local Interconnection Trunks	Parity with Retail

SEEM Measure

SEEM Measure		
Yes	Tier I	Х
	Tier II	Х

SEEM Disaggregation	SEEM Analog/Benchmark
Resale POTS	 Retail Residence and Business (POTS)
Resale Design	Retail Design
UNE Loop + Port Combinations	Retail Residence and Business
UNE Loops	Retail Residence and Business Dispatch
• UNE xDSL	ADSL Provided to Retail
UNE Line Sharing	ADSL Provided to Retail
Local Interconnection Trunks	Parity with Retail

P-10: Total Service Order Cycle Time (TSOCT)

Definition

This report measures the total service order cycle time from receipt of a valid service order request to the return of a completion notice to the CLEC Interface.

Exclusions

- Canceled Service Orders
- Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders, Test Orders, etc.)
- D (Disconnect Except "D" orders associated with LNP Standalone.) and F (From) orders. (From is disconnect side of a move order when the customer moves to a new address)
- "L" Appointment coded orders (where the customer has requested a later than offered interval)
- Orders with CLEC/Subscriber caused delays or CLEC/Subscriber requested due date changes

Business Rules

The interval is determined for each order processed during the reporting period. This measurement combines three reports: FOC Timeliness, Average Order Completion Interval and Average Completion Notice Interval. For UNE XDSL Loop, this measurement combines Service Inquiry Interval (SI), FOC Timeliness, Average Completion Interval, and Average Completion Notice Interval.

This interval starts with the receipt of a valid service order request and stops when a completion notice is sent to the CLEC Interface (LENS, TAG OR EDI) and the BellSouth Legacy Systems. Elapsed time for each order is accumulated for each reporting dimension. The accumulated time for each reporting dimension is then divided by the associated total number of orders completed. Orders that are worked on zero due dates are calculated with a .33 day interval (8 hours) in order to report a portion of a day interval. These orders are issued and worked/completed on same day. They can be either flow through orders (no field work-non-dispatched) or field orders (dispatched).

Reporting is by Fully Mechanized, Partially Mechanized and Non-Mechanized receipt of LSRs.

Calculation

Total Service Order Cycle Time = (a - b)

- a = Service Order Completion Notice Date
- b = Service Request Receipt Date

Average Total Service Order Cycle Time = (c / d)

- c = Sum of all Total Service Order Cycle Times
- d = Total Number Service Orders Completed in Reporting Period

Total Service Order Cycle Time Interval Distribution (for each interval) = (e / f) X 100

- e = Total Number of Service Requests Completed in "X" minutes/hours
- f = Total Number of Service Requests Received in Reporting Period

Report Structure

- CLEC Specific
- CLEC Aggregate
- BellSouth Aggregate
- Fully Mechanized; Partially Mechanized; Non-Mechanized
- Report in categories of <10 line/circuits; >= 10 line/circuits (except trunks)
- Dispatch / No Dispatch categories applicable to all levels except trunks
- Intervals 0-5, 5-10, 10-15, 15-20, 20-25, 25-30, >= 30 Days. The interval breakout is: 0-5 = 0-4.99, 5-10 = 5-9.99, 10-15 = 10-14.99, 15-20 = 15-19.99, 20-25 = 20-24.99, 25-30 = 25-29.99, >= 30 = 30 and greater.

Relating to CLEC Experience	Relating to BellSouth Performance
• Interval for EOC	 Report Month BellSouth Order Number

 CLEC Company Name (OCN) Order Number (PON) Submission Date & Time (TICKET_ID) Completion Date (CMPLTN_DT) Completion Notice Date and Time Service Type (CLASS_SVC_DESC) 	 Order Submission Date & Time Order Completion Date & Time Service Type Geographic Scope
• Geographic Scope Note: Code in parentheses is the corresponding header found in the raw data file	

SQM Level of Disaggregation	SQM Analog/Benchmark
Resale Residence	Diagnostic
Resale Business	
Resale Design	
Resale PBX	
Resale Centrex	
Resale ISDN	
• LNP (Standalone)	
• INP (Standalone)	
2W Analog Loop Design	
2W Analog Loop Non-Design	
2W Analog Loop With LNP Design	
2W Analog Loop With LNP Non-Design	
UNE Switch Ports	
• UNE Loop + Port Combinations	
UNE Combo Other	
• UNE xDSL (HDSL, ADSL and UCL)	
• UNE ISDN	
UNE Line Sharing	
• UNE Other Design	
• UNE Other Non -Design	
• UNE Digital Loops < DS1	
• UNE Digital Loops >= DS1	
• Local Transport (Unbundled Interoffice Transport)	
Local Interconnection Trunks	

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

P-11: Service Order Accuracy

Definition

The "service order accuracy" measurement measures the accuracy and completeness of a sample of BellSouth service orders by comparing what was ordered and what was completed.

Exclusions

- Cancelled Service Orders
- Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders, Test Orders, etc.)
- D & F orders

Business Rules

A statistically valid sample of service orders, completed during a monthly reporting period, is compared to the original account profile and the order that the CLEC sent to BellSouth. An order is "completed without error" if all service attributes and account detail changes (as determined by comparing the original order) completely and accurately reflect the activity specified on the original order and any supplemental CLEC order. For both small and large sample sizes, when a Service Request cannot be matched with a corresponding Service Order, it will not be counted. For small sample sizes an effort will be made to replace the service request.

Calculation

Percent Service Order Accuracy = (a / b) X 100

- a = Orders Completed without Error
- b = Orders Completed in Reporting Period

Report Structure

- CLEC Aggregate
- Reported in categories of <10 line/circuits; >= 10 line/circuits
- Dispatch / No Dispatch

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	 No BellSouth Analog Exist
CLEC Order Number and PON	
 Local Service Request (LSR) 	
Order Submission Date	
Committed Due Date	
Service Type	
Standard Order Activity	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
Resale Residence	• 95% Accurate
Resale Business	
• Resale Design (Specials)	
• UNE Specials (Design)	
• UNE (Non-Design)	
Local Interconnection Trunks	

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

P-12: LNP-Percent Missed Installation Appointments

Definition

"Percent missed installation appointments" monitors the reliability of BellSouth commitments with respect to committed due dates to assure that CLECs can reliably quote expected due dates to their retail customer as compared to BellSouth. This measure is the percentage of total orders processed for which BellSouth is unable to complete the service orders on the committed due dates and reported for total misses and End User Misses.

Exclusions

- Canceled Service Orders
- Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders, Test Orders, etc.) where identifiable

Business Rules

Percent Missed Installation Appointments (PMI) is the percentage of total orders processed for which BellSouth is unable to complete the service orders on the committed due dates. Missed Appointments caused by end-user reasons will be included and reported in a separate category. The first commitment date on the service order that is a missed appointment is the missed appointment code used for calculation whether it is a BellSouth missed appointment or an End User missed appointment. The "due date" is any time on the confirmed due date, which means there cannot be a cutoff time for commitments as certain types of orders are requested to be worked after standard business hours.

Calculation

LNP Percent Missed Installation Appointments = (a / b) X 100

- a = Number of Orders with Completion date in Reporting Period past the Original Committed Due Date
- b = Number of Orders Completed in Reporting Period

Report Structure

- CLEC Specific
- CLEC Aggregate
- Geographic Scope
- State/Region
- Report in Categories of <10 lines/circuits >= 10 lines/circuits (except trunks)

Report explanation: Total Missed Appointments is the total percent of orders missed either by BellSouth or the CLEC end user. End User MA represents the percentage of orders missed by the CLEC end user. The difference between End User Missed Appointments and Total Missed Appointments is the result of BellSouth caused misses.

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Not Applicable
CLEC Order Number and PON (PON)	• Not Applicable
• Committed Due Date (DD)	
Completion Date (CMPLTN DD)	
• Status Type	
Status Notice Date	
Standard Order Activity	
Geographic Scope	
Note: Code in parentheses is the corresponding header found	1
in the raw data file.	

SQM Level of Disaggregation	SQM Analog/Benchmark
• LNP	• Retail Residence and Business (POTS)

SEEM Measure

SEEM Measure		
Yes	Tier I	Х
	Tier II	Х

SEEM Disaggregation - Analog/Benchmark

SEEM Disaggregation	SEEM Analog/Benchmark
• LNP	• 95% Due Dates Met ^a

^aDue to data structure issues, BellSouth is using a benchmark comparison for SEEM rather than the Truncated Z as stated in the Order.

P-13: LNP-Average Disconnect Timeliness Interval & Disconnect Timeliness Interval Distribution

Definition

Disconnect Timeliness is defined as the interval between the time ESI Number Manager receives the valid 'Number Ported' message from NPAC (signifying the CLEC 'Activate') until the time the Disconnect is completed in the Central Office switch. This interval effectively measures BellSouth responsiveness by isolating it from impacts that are caused by CLEC related activities.

Exclusions

- Canceled Service Orders
- Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders, Test Orders, etc.) where identifiable.

Business Rules

The Disconnect Timeliness interval is determined for each telephone number ported associated with a disconnect service order processed on an LSR during the reporting period. The Disconnect Timeliness interval is the elapsed time from when BellSouth receives a valid 'Number Ported' message in ESI Number Manager (signifying the CLEC 'Activate') for each telephone number ported until each telephone number on the service order is disconnected in the Central Office switch. Elapsed time for each ported telephone number is accumulated for each reporting dimension. The accumulated time for each reporting dimension is then divided by the total number of selected telephone numbers disconnected in the reporting period.

Calculation

Disconnect Timeliness Interval = (a - b)

- a = Completion Date and Time in Central Office switch for each number on disconnect order
- b = Valid 'Number Ported' message received date & time

Average Disconnect Timeliness Interval = (c / d)

- c = Sum of all Disconnect Timeliness Intervals
- d = Total Number of disconnected numbers completed in reporting period

Disconnect Timeliness Interval Distribution (for each interval) = (e / f) X 100

- e = Disconnected numbers completed in "X" days
- f = Total disconnect numbers completed in reporting period

Report Structure

- CLEC Specific
- CLEC Aggregate
- Geographic Scope
 - State, Region

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Order Number	Not Applicable
 Telephone Number/Circuit Number 	
Committed Due Date	
• Receipt Date/Time (ESI Number Manager)	
• Date/Time of Recent Change Notice	

SQM Level of Disaggregation	SQM Analog/Benchmark
• LNP	• 95% <= 15 Minutes

SEEM Measure

SEEM Measure		
Yes	Tier I	Х
	Tier II	Х

SEEM Disaggregation	SEEM Analog/Benchmark
LNP Standalone	• 95% <= 15 Minutes

P-14: LNP-Total Service Order Cycle Time (TSOCT)

Definition

Total Service Order Cycle Time measures the interval from receipt of a valid service order request to the completion of the final service order associated with that service request.

Exclusions

- Canceled Service Orders
- Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders, Test Orders, etc.) where identifiable
- "L" appointment coded orders (indicating the customer has requested a later than offered interval)
- "S" missed appointment coded orders (indicating subscriber missed appointments), except for "SP" codes (indicating subscriber prior due date requested). This would include "S" codes assigned to subsequent due date changes.

Business Rules

The interval is determined for each order processed during the reporting period. This measurement combines three reports: FOC Timeliness, Average Order Completion Interval and Average Completion Notice Interval.

This interval starts with the receipt of a valid service order request and stops when a completion notice is sent to the CLEC Interface (LENS, TAG OR EDI). Elapsed time for each order is accumulated for each reporting dimension. The accumulated time for each reporting dimension is then divided by the associated total number of orders completed. Orders that are worked on zero due dates are calculated with a .33 day interval (8 hours) in order to report a portion of a day interval. These orders are issued and worked/completed on the same day.

Reporting is by Fully Mechanized, Partially Mechanized and Non-Mechanized receipt of LSRs.

Calculation

Total Service Order Cycle Time = (a - b)

- a = Service Order Completion Notice Date
- b = Service Request Receipt Date

Average Total Service Order Cycle Time = (c / d)

- c = Sum of all Total Service Order Cycle Times
- d = Total Number Service Orders Completed in Reporting Period

Total Service Order Cycle Time Interval Distribution (for each interval) = (e / f) X 100

- e = Total Number of Service Orders Completed in "X" minutes/hours
- f = Total Number of Service Orders Received in Reporting Period

Report Structure

- CLEC Specific
- CLEC Aggregate
- Fully Mechanized; Partially Mechanized; Non-Mechanized
- Report in categories of < 10 lines/circuits; >= lines/circuits (except trunks)
- Intervals 0-5, 5-10, 10-15, 15-20, 20-25, 25-30, >= 30 Days. The interval breakout is: 0-5 = 0-4.99, 5-10 = 5-9.99, 10-15 = 10-14.99, 15-20 = 15-19.99, 20-25 = 20-24.99, 25-30 = 25-29.99, >= 30 = 30 and greater.

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	• Not Applicable
• Interval for FOC	- Not Applicable
CLEC Company Name (OCN)	
• Order Number (PON)	
• Submission Date & Time (TICKET_ID)	
Completion Date (CMPLTN_DT)	
Completion Notice Date and Time	

Service Type (CLASS_SVC_DESC)Geographic Scope	
Note: Code in parentheses is the corresponding header found in the raw data file	

	SQM Level of Disaggregation	SQM Analog/Benchmark
 LNP 		Diagnostic

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

M&R-1: Missed Repair Appointments

Definition

The percent of trouble reports not cleared by the committed date and time.

Exclusions

- Trouble tickets canceled at the CLEC request
- BellSouth trouble reports associated with internal or administrative service
- Customer Provided Equipment (CPE) troubles or CLEC Equipment Trouble

Business Rules

The negotiated commitment date and time is established when the repair report is received. The cleared time is the date and time that BellSouth personnel clear the trouble and closes the trouble report in his/her Computer Access Terminal (CAT) or workstation. If this is after the Commitment time, the report is flagged as a "Missed Commitment" or a missed repair appointment. When the data for this measure is collected for BellSouth and a CLEC, it can be used to compare the percentage of the time repair appointments are missed due to BellSouth reasons. (No access reports are not part of this measure because they are not a missed appointment.)

Note: Appointment intervals vary with force availability in the POTS environment. Specials and Trunk intervals are standard interval

appointments of no greater than 24 hours. Standalone LNP historical data is not available in the maintenance systems (LMOS or WFA).

Calculation

Percentage of Missed Repair Appointments = (a / b) X 100

- a = Count of Customer Troubles Not Cleared by the Quoted Commitment Date and Time
- b = Total Trouble reports closed in Reporting Period

Report Structure

- Dispatch/Non-Dispatch
- CLEC Specific
- CLEC Aggregate
- BellSouth Aggregate

Relating to CLEC Experience	Relating to BellSouth Performance
 CLEC Company Name Submission Date & Time (TICKET_ID) Completion Date (CMPLTN_DT) Service Type (CLASS_SVC_DESC) Disposition and Cause (CAUSE_CD & CAUSE_DESC) Gaographic Same 	 Report Month BellSouth Company Code Submission Date & Time Completion Date Service Type Disposition and Cause (Non-Design /Non-Special Only) Trouble Code (Design and Trunking Services) Geographic Scope

SQM Level of Disaggregation	SQM Analog/Benchmark
Resale Residence	Retail Residence
Resale Business	Retail business
Resale Design	Retail Design
• Resale PBX	•
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
• LNP (Standalone) (Not Available in Maintenance)	Not Applicable
2W Analog Loop Design	Retail Residence & Business Dispatch
• 2W Analog Loop Non - Design	• Retail Residence & Business (POTS) (Exclusion of
	Switch-Based Feature Troubles)
• UNE Loop + Port Combinations	Retail Residence & Business
UNE Switch Ports	Retail Residence & Business (POTS)
UNE Combo Other	• Retail Residence, Business and Design Dispatch
• UNE xDSL (HDSL, ADSL and UCL)	ADSL Provided to Retail
• UNE ISDN	• Retail ISDN – BRI
UNE Line Sharing	ADSL Provided to Retail
• UNE Other Design	Retail Design
• UNE Other Non - Design	Retail Residence & Business
Local Interconnection Trunks	• Parity with Retail
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice

SEEM Measure

SEEM Measure		
Yes	Tier I	Х
	Tier II	Х

SEEM Disaggregation	SEEM Analog/Benchmark
Resale POTS	• Retail Residence and Business (POTS)
Resale Design	Retail Design
UNE Loop + Port Combinations	Retail Residence and Business
UNE Loops	Retail Residence and Business Dispatch
UNE xDSL	ADSL Provided to Retail
UNE Line Sharing	ADSL Provided to Retail
Local Interconnection Trunks	Parity with Retail

M&R-2: Customer Trouble Report Rate

Definition

Percent of initial and repeated customer direct or referred troubles reported within a calendar month per 100 lines/circuits in service.

Exclusions

- Trouble tickets canceled at the CLEC request
- · BellSouth trouble reports associated with internal or administrative service
- Customer Provided Equipment (CPE) troubles or CLEC Equipment Trouble

Business Rules

Customer Trouble Report Rate is computed by accumulating the number of maintenance initial and repeated trouble reports during the reporting period. The resulting number of trouble reports are divided by the total "number of service" lines, ports or combination that exist for the CLECs and BellSouth respectively at the end of the report month.

Calculation

Customer Trouble Report Rate = (a / b) X 100

- a = Count of Initial and Repeated Trouble Reports closed in the Current Period
- b = Number of Service Access Lines in service at End of the Report Period

Report Structure

- Dispatch/Non-Dispatch
- CLÊC Specific
- CLEC Aggregate
- BellSouth Aggregate

Relating to CLEC Experience	Relating to BellSouth Performance
 Report Month CLEC Company Name Ticket Submission Date & Time (TICKET_ID) Ticket Completion Date (CMPLTN_DT) Service Type (CLASS_SVC_DESC) Disposition and Cause (CAUSE_CD & CAUSE_DESC) # Service Access Lines in Service at the end of period Geographic Scope Note: Code in parentheses is the corresponding header found in the raw data file. 	 Report Month BellSouth Company Code Ticket Submission Date & Time Ticket Completion Date Service Type Disposition and Cause (Non-Design /Non-Special Only) Trouble Code (Design and Trunking Services) # Service Access Lines in Service at the end of period Geographic Scope

SQM Level of Disaggregation	SQM Analog/Benchmark
• Resale Residence	Retail Residence
Resale Business	Retail Business
Resale Design	Retail Design
• Resale PBX	• Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
• LNP (Standalone) (Not Available in Maintenance)	Not Applicable
2W Analog Loop Design	Retail Residence & Business Dispatch
2W Analog Loop Non - Design	• Retail Residence & Business (POTS) (Exclusion of
	Switch-Based Feature Troubles)
• UNE Loop + Port Combinations	Retail Residence & Business
UNE Switch Ports	Retail Residence & Business (POTS)
UNE Combo Other	• Retail Residence, Business and Design Dispatch
• UNE xDSL (HDSL, ADSL and UCL)	ADSL Provided to Retail
• UNE ISDN	• Retail ISDN – BRI
UNE Line Sharing	ADSL Provided to Retail
• UNE Other Design	Retail Design
• UNE Other Non - Design	Retail Residence & Business
Local Interconnection Trunks	Parity with Retail
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice

SEEM Measure

SEEM Measure		
Yes	Tier I	Х
	Tier II	Х

SEEM Disaggregation	SEEM Analog/Benchmark
Resale POTS	• Retail Residence and Business (POTS)
Resale Design	Retail Design
UNE Loop + Port Combinations	Retail Residence and Business
UNE Loops	Retail Residence and Business Dispatch
• UNE xDSL	ADSL Provided to Retail
UNE Line Sharing	ADSL Provided to Retail
Local Interconnection Trunks	Parity with Retail

M&R-3: Maintenance Average Duration

Definition

The Average duration of Customer Trouble Reports from the receipt of the Customer Trouble Report to the time the trouble report is cleared.

Exclusions

- Trouble tickets canceled at the CLEC request
- BellSouth trouble reports associated with internal or administrative service
- Customer Provided Équipment (CPE) troubles or CLEC Equipment Trouble

Business Rules

For Average Duration the clock starts on the date and time of the receipt of a correct repair request. The clock stops on the date and time the service is restored and the BellSouth or CLEC customer is notified (when the technician completes the trouble ticket on his/her CAT or work systems).

Calculation

Maintenance Duration = (a - b)

- a = Date and Time of Service Restoration
- b = Date and Time Trouble Ticket was Opened

Average Maintenance Duration = (c / d)

- c = Total of all maintenance durations in the reporting period
- d = Total Closed Troubles in the reporting period

Report Structure

- Dispatch/Non-Dispatch
- CLEC Specific
- CLEC Aggregate
- BellSouth Aggregate

Relating to CLEC Experience	Relating to BellSouth Performance
 Report Month Total Tickets (LINE_NBR) CLEC Company Name Ticket Submission Date & Time (TICKET_ID) Ticket Completion Date (CMPLTN_DT) Service Type (CLASS_SVC_DESC) Disposition and Cause (CAUSE_CD & CAUSE_DESC) Geographic Scope Note: Code in parentheses is the corresponding header founin the raw data file. 	 Report Month Total Tickets BellSouth Company Code Ticket Submission Date Ticket Submission Time Ticket Completion Date Ticket Completion Time Total Duration Time Service Type Disposition and Cause (Non-Design /Non-Special Only) Trouble Code (Design and Trunking Services) Geographic Scope

SQM Level of Disaggregation	SQM Analog/Benchmark
• Resale Residence	Retail Residence
Resale Business	Retail Business
Resale Design	Retail Design
• Resale PBX	• Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
• LNP (Standalone) (Not Available in Maintenance)	Not Applicable
2W Analog Loop Design	Retail Residence & Business Dispatch
2W Analog Loop Non - Design	• Retail Residence & Business (POTS) (Exclusion of
	Switch-Based Feature Troubles)
• UNE Loop + Port Combinations	Retail Residence & Business
UNE Switch Ports	Retail Residence & Business (POTS)
UNE Combo Other	• Retail Residence, Business and Design Dispatch
• UNE xDSL (HDSL, ADSL and UCL)	ADSL Provided to Retail
• UNE ISDN	• Retail ISDN – BRI
UNE Line Sharing	ADSL Provided to Retail
• UNE Other Design	Retail Design
• UNE Other Non - Design	Retail Residence & Business
Local Interconnection Trunks	Parity with Retail
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice

SEEM Measure

SEEM Measure		
Yes	Tier I	Х
	Tier II	Х

SEEM Disaggregation	SEEM Analog/Benchmark
Resale POTS	• Retail Residence and Business (POTS)
Resale Design	Retail Design
• UNE Loop + Port Combinations	Retail Residence and Business
UNE Loops	Retail Residence and Business Dispatch
• UNE xDSL	ADSL Provided to Retail
UNE Line Sharing	ADSL Provided to Retail
Local Interconnection Trunks	Parity with Retail

M&R-4: Percent Repeat Troubles within 30 Days

Definition

Closed trouble reports on the same line/circuit as a previous trouble report received within 30 calendar days as a percent of total troubles closed reported

Exclusions

- Trouble tickets canceled at the CLEC request
- BellSouth trouble reports associated with internal or administrative service
- Customer Provided Équipment (CPE) troubles or CLEC Equipment Trouble

Business Rules

Includes Customer trouble reports received within 30 days of an original Customer trouble report.

Calculation

Percent Repeat Troubles within 30 Days = (a / b) X 100

- a = Count of closed Customer Troubles where more than one trouble report was logged for the same service line within a continuous 30 days
- b = Total Trouble Reports Closed in Reporting Period

Report Structure

- Dispatch/Non-Dispatch
- CLÉC Specific
- CLEC Aggregate
- BellSouth Aggregate

Relating to CLEC Experience	Relating to BellSouth Performance
 Report Month Total Tickets (LINE_NBR) CLEC Company Name Ticket Submission Date & Time (TICKET_ID) Ticket Completion Date (CMPLTN_DT) Total and Percent Repeat Trouble Reports within 30 Days (TOT_REPEAT) Service Type Disposition and Cause (CAUSE_CD & CAUSE_DESC) Geographic Scope Note: Code in parentheses is the corresponding header found in the raw data file. 	 Report Month Total Tickets BellSouth Company Code Ticket Submission Date Ticket Submission Time Ticket Completion Date Ticket Completion Time Total and Percent Repeat Trouble Reports within 30 Days Service Type

SQM Level of Disaggregation	SQM Analog/Benchmark
Resale Residence	Retail Residence
• Resale Business	Retail Business
Resale Design	Retail Design
• Resale PBX	• Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
• LNP (Standalone) (Not Available in Maintenance)	Not Applicable
• 2W Analog Loop Design	Retail Residence & Business Dispatch
• 2W Analog Loop Non - Design	• Retail Residence & Business (POTS) (Exclusion of
	Switch-Based Feature Troubles)
• UNE Loop + Port Combinations	Retail Residence & Business
UNE Switch Ports	Retail Residence and Business (POTS)
UNE Combo Other	Retail Residence, Business & Design Dispatch
• UNE xDSL (HDSL, ADSL and UCL)	ADSL Provided to Retail
• UNE ISDN	• Retail ISDN – BRI
UNE Line Sharing	ADSL Provided to Retail
• UNE Other Design	Retail Design
• UNE Other Non - Design	Retail Residence & Business
Local Interconnection Trunks	Parity with Retail
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice

SEEM Measure

SEEM Measure		
Yes	Tier I	Х
	Tier II	Х

SEEM Disaggregation	SEEM Analog/Benchmark
Resale POTS	• Retail Residence and Business (POTS)
Resale Design	Retail Design
• UNE Loop + Port Combinations	Retail Residence and Business
UNE Loops	Retail Residence and Business Dispatch
• UNE xDSL	ADSL Provided to Retail
UNE Line Sharing	ADSL Provided to Retail
Local Interconnection Trunks	Parity with Retail
M&R-5: Out of Service (OOS) > 24 Hours

Definition

For Out of Service Troubles (no dial tone, cannot be called or cannot call out) the percentage of Total OOS Troubles cleared in excess of 24 hours. (All design services are considered to be out of service).

Exclusions

- Trouble Reports canceled at the CLEC request
- BellSouth Trouble Reports associated with administrative service
- Customer Provided Equipment (CPE) Troubles or CLEC Equipment Troubles

Business Rules

Customer Trouble reports that are out of service and cleared in excess of 24 hours. The clock begins when the trouble report is created in LMOS/WFA and the trouble is counted if the elapsed time exceeds 24 hours.

Calculation

Out of Service (OOS) > 24 hours = $(a / b) \ge 100$

- a = Total Cleared Troubles OOS > 24 Hours
- b = Total OOS Troubles in Reporting Period

Report Structure

- Dispatch/Non Dispatch
- CLEC Specific
- BellSouth Aggregate
- CLEC Aggregate

Data Retained

SQM Level of Disaggregation	SQM Analog/Benchmark
Resale Residence	Retail Residence
Resale Business	Retail Business
Resale Design	Retail Design
• Resale PBX	• Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
• LNP (Standalone) (Not Available in Maintenance)	Not Applicable
2W Analog Loop Design	Retail Residence & Business Dispatch
2W Analog Loop Non - Design	• Retail Residence & Business (POTS) (Exclusion of
	Switch-Based Feature Troubles)
UNE Loop + Port Combinations	Retail Residence & Business
UNE Switch Ports	Retail Residence & Business (POTS)
UNE Combo Other	Retail Residence, Business and Design Dispatch
• UNE xDSL (HDSL, ADSL and UCL)	ADSL Provided to Retail
• UNE ISDN	• Retail ISDN – BRI
UNE Line Sharing	ADSL Provided to Retail
• UNE Other Design	Retail Design
• UNE Other Non - Design	Retail Residence & Business
Local Interconnection Trunks	Parity with Retail
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice

SQM Disaggregation - Analog/Benchmark

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

M&R-6: Average Answer Time – Repair Centers

Definition

This measures the average time a customer is in queue when calling a BellSouth Repair Center.

Exclusions

None

Business Rules

The clock starts when a CLEC Representative or BellSouth customer makes a choice on the Repair Center's menu and is put in queue for the next repair attendant. The clock stops when the repair attendant answers the call (abandoned calls are not included).

Note: The Total Column is a combined BellSouth Residence and Business number.

Calculation

Answer Time for BellSouth Repair Centers = (a - b)

- a = Time BellSouth Repair Attendant Answers Call
- b = Time of entry into queue after ACD Selection

Average Answer Time for BellSouth Repair Centers = (c / d)

- c = Sum of all Answer Times
- d = Total number of calls by reporting period

Report Structure

- CLEC Aggregate
- BellSouth Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
CLEC Average Answer Time	BellSouth Average Answer Time

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
• Region. CLEC/BellSouth Service Centers and BellSouth	• For CLEC, Average Answer Times in UNE Center and
Repair Centers are regional.	BRMC are comparable to the Average Answer Times in
	the BellSouth Repair Centers.

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

M&R-7: Mean Time To Notify CLEC of Network Outages

Definition

This report measures the time it takes for the BellSouth Network Management Center (NMC) to notify the CLEC of major network outages.

Exclusions

None

Business Rules

BellSouth will inform the CLEC of any major network outages (key customer accounts) via a page or email. When the BellSouth NMC becomes aware of a network incident, the CLEC and BellSouth will be notified electronically. The notification time for each outage will be measured in minutes and divided by the number of outages for the reporting period. These are broadcast messages. It is up to those receiving the message to determine if they have customers affected by the incident.

The CLECs will be notified in accordance with the rules outlined in Appendix D of the CLEC "Customer Guide" which is published on the internet at: www.interconnection.bellsouth.com/guides/other_guides/html/gopue/indexf.htm.

Calculation

Time to Notify CLEC = (a - b)

- a = Date and Time BellSouth Notified CLEC
- b = Date and Time BellSouth Detected Network Incident

Mean Time to Notify CLEC = (c / d)

- c = Sum of all Times to Notify CLEC
- d = Count of Network Incidents

Report Structure

- BellSouth Aggregate
- CLEC Aggregate
- CLEC Specific

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	• Report Month
 Major Network Events 	 Major Network Events
Date/Time of Incident	Date/Time of Incident
Date/Time of Notification	Date/Time of Notification

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
BellSouth Aggregate	Parity by Design
CLEC Aggregate	
CLEC Specific	

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

Section 5: Billing

B-1: Invoice Accuracy

Definition

This measure provides the percentage of accuracy of the billing invoices rendered to CLECs during the current month.

Exclusions

- Adjustments not related to billing errors (e.g., credits for service outage, special promotion credits, adjustments to satisfy the customer)
- Test Accounts

Business Rules

The accuracy of billing invoices delivered by BellSouth to the CLEC must enable them to provide a degree of billing accuracy comparative to BellSouth bills rendered to retail customers of BellSouth. CLECs request adjustments on bills determined to be incorrect. The BellSouth Billing verification process includes manually analyzing a sample of local bills from each bill period. The bill verification process draws from a mix of different customer billing options and types of service. An end-to-end auditing process is performed for new products and services. Internal measurements and controls are maintained on all billing processes.

Calculation

Invoice Accuracy = $[(a - b) / a] \ge 100$

- a = Absolute Value of Total Billed Revenues during current month
- b = Absolute Value of Billing Related Adjustments during current month

Report Structure

- CLEC Specific
- CLEC Aggregate
- BellSouth Aggregate
- Geographic Scope
- Region
- State

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
• Report Month	Report Month
Invoice Type	• Retail Type
- UNE	- CRIS
- Resale	- CABS
- Interconnection	Total Billed Revenue
Total Billed Revenue	Billing Related Adjustments
Billing Related Adjustments	

SQM Level of Disaggregation	SQM Analog/Benchmark
Product/Invoice Type	• CLEC Invoice Accuracy is comparable to BellSouth
- Resale - UNE	Invoice Accuracy
- Interconnection	

SEEM Measure

SEEM Measure		
Yes	Tier I	X
	Tier II	Х

SEEM Disaggregation	SEEM Analog/Benchmark
CLEC State	• Parity With Retail
BellSouth State	-

B2: Mean Time to Deliver Invoices

Definition

Bill Distribution is calculated as follows: CRIS BILLS-The number of workdays is reported for CRIS bills. This is calculated by counting the Bill Period date as the first work day. Weekends and holidays are excluded when counting workdays. J/N Bills are counted in the CRIS work day category for the purposes of the measurement since their billing account number (Q account) is provided from the CRIS system.

CABS BILLS-The number of calendar days is reported for CABS bills. This is calculated by counting the day following the Bill Period date as the first calendar day. Weekends and holidays are included when counting the calendar days.

Exclusions

Any invoices rejected due to formatting or content errors.

Business Rules

This report measures the mean interval for timeliness of billing records delivered to CLECs in an agreed upon format. CRIS-based invoices are measured in business days, and CABS-based invoices in calendar days.

Calculation

Invoice Timeliness = (a - b)

- a = Invoice Transmission Date
- b = Close Date of Scheduled Bill Cycle

Mean Time To Deliver Invoices = (c / d)

- c = Sum of all Invoice Timeliness intervals
- d = Count of Invoices Transmitted in Reporting Period

Report Structure

- CLEC Specific
- CLEC Aggregate
- BellSouth Aggregate
- Geographic Scope
 - Region
 - State

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Report Month
Invoice Type	Invoice Type
- UNE	- CRIS
- Resale	- CABS
- Interconnection	Invoice Transmission Count
 Invoice Transmission Count 	• Date of Scheduled Bill Close
Date of Scheduled Bill Close	

SQM Level of Disaggregation	SQM Analog/Benchmark
Product/Invoice Type	• CRIS-based invoices will be released for delivery within
• Resale	six (6) business days.
• UNE	• CABS-based invoices will be released for delivery within
• Interconnection	eight (8) calendar days.
	 CLEC Average Delivery Intervals for both CRIS and
	CABS Invoices are comparable to BellSouth Average
	delivery for both systems.

SEEM Measure

SEEM Measure		
Yes	Tier I	Х
	Tier II	Х

SEEM Disaggregation	SEEM Analog/Benchmark
CLEC State	• Parity with Retail
- CRIS	
- CABS	
BellSouth Region	

B3: Usage Data Delivery Accuracy

Definition

This measurement captures the percentage of recorded usage that is delivered error free and in an acceptable format to the appropriate Competitive Local Exchange Carrier (CLEC). These percentages will provide the necessary data for use as a comparative measurement for BellSouth performance. This measurement captures Data Delivery Accuracy rather than the accuracy of the individual usage recording.

Exclusions

None

Business Rules

The accuracy of the data delivery of usage records delivered by BellSouth to the CLEC must enable them to provide a degree of accuracy comparative to BellSouth bills rendered to their retail customers. If errors are detected in the delivery process, they are investigated, evaluated and documented. Errors are corrected and the data retransmitted to the CLEC.

Calculation

Usage Data Delivery Accuracy = (a - b) / a X 100

- a = Total number of usage data packs sent during current month
- b = Total number of usage data packs requiring retransmission during current month

Report Structure

- CLEC Specific
- CLEC Aggregate
- BellSouth Aggregate
- Geographic Scope
- Region

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	• Report Month
 Record Type BellSouth Recorded Non-BellSouth Recorded 	Record Type

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
Region	• CLEC Usage Data Delivery Accuracy is comparable to
	BellSouth Usage Data Delivery Accuracy

SEEM Measure

SEEM Measure		
Yes	Tier I	Х
	Tier II	Х

SEEM Disaggregation	SEEM Analog/Benchmark
CLEC State	• Parity With Retail
BellSouth Region	

B4: Usage Data Delivery Completeness

Definition

This measurement provides percentage of complete and accurately recorded usage data (usage recorded by BellSouth and usage recorded by other companies and sent to BellSouth for billing) that is processed and transmitted to the CLEC within thirty (30) days of the message recording date. A parity measure is also provided showing completeness of BellSouth messages processed and transmitted via CMDS. BellSouth delivers its own retail usage from recording location to billing location via CMDS as well as delivering billing data to other companies. Timeliness, Completeness and Mean Time to Deliver Usage measures are reported on the same report.

Exclusions

None

Business Rules

The purpose of these measurements is to demonstrate the level of quality of usage data delivered to the appropriate CLEC. Method of delivery is at the option of the CLEC.

Calculation

Usage Data Delivery Completeness = (a / b) X 100

- a = Total number of Recorded usage records delivered during current month that are within thirty (30) days of the message recording date
- b = Total number of Recorded usage records delivered during the current month

Report Structure

- CLEC Specific
- CLEC Aggregate
- BellSouth Aggregate
- Region

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	• Report Month
Record Type	Record Type
- BellSouth Recorded	
- Non-BellSouth Recorded	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
Region	• CLEC Usage Data Delivery Completeness is comparable
	to BellSouth Usage Data Delivery Completeness

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

B5: Usage Data Delivery Timeliness

Definition

This measurement provides a percentage of recorded usage data (usage recorded by BellSouth and usage recorded by other companies and sent to BellSouth for billing) that is delivered to the appropriate CLEC within six (6) calendar days from the receipt of the initial recording. A parity measure is also provided showing timeliness of BellSouth messages processed and transmitted via CMDS. Timeliness, Completeness and Mean Time to Deliver Usage measures are reported on the same report.

Exclusions

None

Business Rules

The purpose of this measurement is to demonstrate the level of timeliness for processing and transmission of usage data delivered to the appropriate CLEC. The usage data will be mechanically transmitted or mailed to the CLEC data processing center once daily. The Timeliness interval of usage recorded by other companies is measured from the date BellSouth receives the records to the date BellSouth distributes to the CLEC. Method of delivery is at the option of the CLEC.

Calculation

Usage Data Delivery Timeliness Current month = (a / b) X 100

- a = Total number of usage records sent within six (6) calendar days from initial recording/receipt
- b = Total number of usage records sent

Report Structure

- CLEC Aggregate
- CLEC Specific
- BellSouth Aggregate
- Region

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Report Month
• Record Type	• Record Type
- BellSouth Recorded	
- Non-BellSouth Recorded	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
Region	• CLEC Usage Data Delivery Timeliness is comparable to
	BellSouth Usage Data Delivery Timeliness

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

B6: Mean Time to Deliver Usage

Definition

This measurement provides the average time it takes to deliver Usage Records to a CLEC. A parity measure is also provided showing timeliness of BellSouth messages processed and transmitted via CMDS. Timeliness, Completeness and Mean Time to Deliver Usage measures are reported on the same report.

Exclusions

None

Business Rules

The purpose of this measurement is to demonstrate the average number of days it takes BellSouth to deliver Usage data to the appropriate CLEC. Usage data is mechanically transmitted or mailed to the CLEC data processing center once daily. Method of delivery is at the option of the CLEC.

Calculation

Mean Time to Deliver Usage = (a X b) / c

- a = Volume of Records Delivered
- b = Estimated number of days to deliver
- c = Total Record Volume Delivered

Note: Any usage record falling in the 30+ day interval will be added using an average figure of 31.5 days.

Report Structure

- CLEC Aggregate
- CLEC Specific
- BellSouth Aggregate
- Region

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Report Month
• Record Type	• Record Type
- BellSouth Recorded	
- Non-BellSouth Recorded	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
Region	• Mean Time to Deliver Usage to CLEC is comparable to
	Mean Time to Deliver Usage to BellSouth.

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

B7: Recurring Charge Completeness

Definition

This measure captures percentage of fractional recurring charges appearing on the correct bill.

Exclusions

None

Business Rules

The effective date of the recurring charge must be within 30 days of the bill date for the charge to appear on the correct bill.

Calculation

Recurring Charge Completeness = (a / b) X 100

- a = Count of fractional recurring charges that are on the correct bill¹
- b = Total count of fractional recurring charges that are on the correct bill

¹Correct bill = next available bill

Report Structure

- CLEC Specific
- CLEC Aggregate
- BellSouth Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Report Month
Invoice Type	Retail Analog
 Total Recurring Charges Billed 	 Total Recurring Charges Billed
Total Billed on Time	Total Billed on Time

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
Product/Invoice Type	
• Resale	• Parity
• UNE	• Benchmark 90%
Interconnection	• Benchmark 90%

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

B8: Non-Recurring Charge Completeness

Definition

This measure captures percentage of non-recurring charges appearing on the correct bill.

Exclusions

None

Business Rules

The effective date of the non-recurring charge must be within 30 days of the bill date for the charge to appear on the correct bill.

Calculation

Non-Recurring Charge Completeness = $(a / b) \ge 100$

- a = Count of non-recurring charges that are on the correct bill¹
- b = Total count of non-recurring charges that are on the correct bill

¹Correct bill = next available bill

Report Structure

- CLEC Specific
- CLEC Aggregate
- BellSouth Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	• Report Month
Invoice Type	Retail Analog
 Total Non-recurring Charges Billed 	 Total Non-recurring Charges Billed
Total Billed on Time	Total Billed on Time

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
Product/Invoice Type	
• Resale	• Parity
• UNE	• Benchmark 90%
• Interconnection	Benchmark 90%

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

Section 6: Operator Services And Directory Assistance

OS-1: Speed to Answer Performance/Average Speed to Answer - Toll

Definition

Measurement of the average time in seconds calls wait before answered by a toll operator.

Exclusions

None

Business Rules

The clock starts when the customer enters the queue and the clock stops when a BellSouth representative answers the call or the customer abandons the call. The length of each call is determined by measuring, using a scanning technique, and accumulating the elapsed time from the entry of a customer call into the BellSouth call management system queue until the customer call is abandoned or transferred to BellSouth personnel assigned to handle calls for assistance. The system makes no distinction between CLEC customers and BellSouth customers.

Calculation

Speed to Answer Performance/Average Speed to Answer - Toll = a / b

- a = Total queue time
- b = Total calls answered

Note: Total queue time includes time that answered calls wait in queue as well as time abandoned calls wait in queue prior to abandonment.

Report Structure

- Reported for the aggregate of BellSouth and CLECs
- State

Data Retained (on Aggregate Basis)

- For the items below, BellSouth's Performance Measurement Analysis Platform (PMAP) receives a final computation; therefore, no raw data file is available in PMAP
- Month
- Call Type (Toll)
- Average Speed of Answer

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
• None	Parity by Design

SEEM Measure

No Tier I	SEEM Measure		
	No	Tier I	
Tier II		Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

OS-2: Speed to Answer Performance/Percent Answered with "X" Seconds - Toll

Definition

Measurement of the percent of toll calls that are answered in less than ten seconds.

Exclusions

None

Business Rules

The clock starts when the customer enters the queue and the clock stops when a BellSouth representative answers the call or the customer abandons the call. The length of each call is determined by measuring, using a scanning technique, and accumulating the elapsed time from the entry of a customer call into the BellSouth call management system queue until the customer call is abandoned or transferred to BellSouth personnel assigned to handle calls for assistance. The system makes no distinction between CLEC customers and BellSouth customers.

Calculation

The Percent Answered within "X" Seconds measurement for toll is derived by using the BellCore Statistical Answer Conversion Tables, to convert the Average Speed to Answer measure into a percent of calls answered within "X" seconds. The BellCore Conversion Tables are specific to the defined parameters of work time, number of operators, max queue size and call abandonment rates.

Report Structure

- Reported for the aggregate of BellSouth and CLECs
 - State

Data Retained (on Aggregate Basis)

- For the items below, BellSouth's Performance Measurement Analysis Platform (PMAP) receives a final computation; therefore, no raw data file is available in PMAP
- Month
- Call Type (Toll)
- Average Speed of Answer

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
• None	Parity by Design

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

DA-1: Speed to Answer Performance/Average Speed to Answer - Directory Assistance (DA)

Definition

Measurement of the average time in seconds calls wait before answered by a DA operator.

Exclusions

None

Business Rules

The clock starts when the customer enters the queue and the clock stops when a BellSouth representative answers the call or the customer abandons the call. The length of each call is determined by measuring, using a scanning technique, and accumulating the elapsed time from the entry of a customer call into the BellSouth call management system queue until the customer call is abandoned or transferred to BellSouth personnel assigned to handle calls for assistance. The system makes no distinction between CLEC customers and BellSouth customers.

Calculation

Speed to Answer Performance/Average Speed to Answer – Directory Assistance (DA) = a / b

- a = Total queue time
- b = Total calls answered

Note: Total queue time includes time that answered calls wait in queue as well as time abandoned calls wait in queue prior to abandonment.

Report Structure

- Reported for the aggregate of BellSouth and CLECs
- State

Data Retained (on Aggregate Basis)

- For the items below, BellSouth's Performance Measurement Analysis Platform (PMAP) receives a final computation; therefore, no raw data file is available in PMAP
- Month
- Call Type (DA)
- Average Speed of Answer

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
• None	Parity by Design

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

DA-2: Speed to Answer Performance/Percent Answered within "X" Seconds - Directory Assistance (DA)

Definition

Measurement of the percent of DA calls that are answered in less than twelve seconds.

Exclusions

None

Business Rules

The clock starts when the customer enters the queue and the clock stops when a BellSouth representative answers the call or the customer abandons the call. The length of each call is determined by measuring, using a scanning technique, and accumulating the elapsed time from the entry of a customer call into the BellSouth call management system queue until the customer call is abandoned or transferred to BellSouth personnel assigned to handle calls for assistance. The system makes no distinction between CLEC customers and BellSouth customers.

Calculation

The Percent Answered within "X" Seconds measurement for DA is derived by using the BellCore Statistical Answer Conversion Tables, to convert the Average Speed to Answer measure into a percent of calls answered within "X" seconds. The BellCore Conversion Tables are specific to the defined parameters of work time, number of operators, max queue size and call abandonment rates.

Report Structure

- Reported for the aggregate of BellSouth and CLECs
 - State

Data Retained (on Aggregate Basis)

- For the items below, BellSouth's Performance Measurement Analysis Platform (PMAP) receives a final computation; therefore, no raw data file is available in PMAP.
- Month
- Call Type (DA)
- Average Speed of Answer

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
• None	Parity by Design

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

Section 7: Database Update Information

D-1: Average Database Update Interval

Definition

This report measures the interval from receipt of the database change request to the completion of the update to the database for Line Information Database (LIDB), Directory Assistance and Directory Listings. For E-911, see Section 8.

Exclusions

- Updates Canceled by the CLEC
- Initial update when supplemented by CLEC
- · BellSouth updates associated with internal or administrative use of local services

Business Rules

The interval for this measure begins with the date and time stamp when a service order is completed and the completion notice is released to all systems to be updated with the order information including Directory Assistance, Directory Listings, and Line Information Database (LIDB). The end time stamp is the date and time of completion of updates to the system.

For BellSouth Results:

The BellSouth computation is identical to that for the CLEC with the clarifications noted below.

Other Clarifications and Qualification:

- For LIDB, the elapsed time for a BellSouth update is measured from the point in time when the BellSouth file maintenance process makes the LIDB update information available until the date and time reported by BellSouth that database updates are completed.
- Results for the CLECs are captured and reported at the update level by Reporting Dimension (see below).
 The Completion Date is the date upon which BellSouth issues the Update Completion Notice to the CLEC.
- The Completion Date is the date upon which Bensouth issues the Opdate Completion Nonce to the CLEC.
 If the CLEC initiates a supplement to the originally submitted update and the supplement reflects changes in customer requirements (rather than responding to BellSouth initiated changes), then the update submission date and time will be the date and time of BellSouth receipt of a syntactically correct update supplement. Update activities responding to BellSouth initiated changes will not result in changes to the update submission date and time used for the purposes of computing the update completion interval.
- Elapsed time is measured in hours and hundredths of hours rounded to the nearest tenth of an hour.
- Because this should be a highly automated process, the accumulation of elapsed time continues through off-schedule, weekends and holidays; however, scheduled maintenance windows are excluded.

Calculation

Update Interval = (a - b)

- a = Completion Date & Time of Database Update
- b = Submission Date and Time of Database Change

Average Update Interval = (c / d)

- c = Sum of all Update Intervals
- d = Total Number of Updates Completed During Reporting Period

Report Structure

- CLEC Specific (Under development)
- CLEC Aggregate
- BellSouth Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
 Database File Submission Time 	Database File Submission Time
Database File Update Completion Time	 Database File Update Completion Time
CLEC Number of Submissions	BellSouth Number of Submissions
• Total Number of Updates	Total Number of Updates

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation:	SQM Analog/Benchmark:
Database Type	• Parity by Design
• LIDB	
Directory Listings	
Directory Assistance	

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation - Analog/Benchmark

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

DUI

D-2: Percent Database Update Accuracy

Definition

This report measures the accuracy of database updates by BellSouth for Line Information Database (LIDB), Directory Assistance, and Directory Listings using a statistically valid sample of LSRs/Orders in a manual review. This manual review is not conducted on BellSouth Retail Orders.

Exclusions

- Updates canceled by the CLEC
- Initial update when supplemented by CLEC
- CLEC orders that had CLEC errors
- · BellSouth updates associated with internal or administrative use of local services

Business Rules

For each update completed during the reporting period, the original update that the CLEC sent to BellSouth is compared to the database following completion of the update by BellSouth. An update is "completed without error" if the database completely and accurately reflects the activity specified on the original and supplemental update (order) submitted by the CLEC. Each database (LIDB, Directory Assistance, and Directory Listings) should be separately tracked and reported.

A statistically valid sample of CLEC Orders are pulled each month. That sample will be used to test the accuracy of the database update process. This is a manual process.

Calculation

Percent Update Accuracy = (a / b) X 100

- a = Number of Updates Completed Without Error
- b = Number Updates Completed

Report Structure

- CLEC Aggregate
- CLEC Specific (not available in this report)
- BellSouth Aggregate (not available in this report)

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
 Report Month CLEC Order Number (so_nbr) and PON (PON) Local Service Request (LSR) Order Submission Date Number of Orders Reviewed 	• Not Applicable
Note : Code in parentheses is the corresponding header found in the raw data file.	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
Database Type	• 95% Accurate
• LIDB	
Directory Assistance	
Directory Listings	

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

D-3: Percent NXXs and LRNs Loaded by the LERG Effective Date

Definition

Measurement of the percent of NXX(s) and Location Routing Numbers LRN(s) loaded in end office and/or tandem switches by the Local Exchange Routing Guide (LERG) effective date when facilities are in place. BellSouth has a single provisioning process for both NXX(s) and LRN(s). In this measure, BellSouth will identify whether or not a particular NXX has been flagged as LNP capable (set triggers for dips) by the LERG effective date.

An LRN is assigned by the owner of the switch and is placed into the software translations for every switch to be used as an administrative pointer to route NXX(s) in LNP capable switches. The LRN is a result of Local Number Porting and is housed in a national database provided by the Number Portability Administration Center (NPAC). The switch owner is responsible for notifying NPAC and requesting the effective date that will be reflected in the LERG. The national database downloads routing tables into BellSouth Service Control Point (SCP) regional databases, which are queried by switches when routing ported numbers.

The basic NXX routing process includes the addition of all NXX(s) in the response translations. This addition to response translations is what supports LRN routing. Routing instructions for all NXX(s), including LRN(s), are received from the Advance Routing & Trunking System (ARTS) and all routing, including response, is established based on the information contained in the Translation Work Instructions (TWINs) document.

Exclusions

- · Activation requests where the CLEC's interconnection arrangements and facilities are not in place by the LERG effective date
- Expedite requests

Business Rules

Data for the initial NXX(s) and LRN(s) in a local calling area will be based on the LERG effective date or completion of the initial interconnection trunk group(s), whichever is longer. Data for additional NXX(s) in the local calling area will be based on the LERG effective date. The LERG effective date is loaded into the system at the request of the CLEC. It is contingent upon the CLEC to engineer, order, and install interconnection arrangements and facilities prior to that date.

The total Count of NXX(s) and LRN(s) that were scheduled to be loaded and those that were loaded by the LERG effective date in BellSouth switches will be captured in the Work Force Administration -Dispatch In database.

Calculation

Percent NXXs/LRNs Loaded and Tested Prior to the LERG Effective Date = (a / b) X 100

- a = Count of NXXs and LRNs loaded by the LERG effective date
- b = Total NXXs and LRNs scheduled to be loaded by the LERG effective date

Report Structure

- CLEC Specific
- CLEC Aggregate
- BellSouth (Not Applicable)

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Company Name	Not Applicable
Company Code	
NPA/NXX	
LERG Effective Date	
Loaded Date	

SQM Level of Disaggregation	SQM Analog/Benchmark
Geographic Scope	• 100% by LERG Effective Date
- Region	

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

Section 8: E911

E-1: Timeliness

Definition

Measures the percent of batch orders for E911 database updates (to CLEC resale and BellSouth retail records) processed successfully within a 24-hour period.

Exclusions

- Any resale order canceled by a CLEC
- Facilities-based CLEC orders

Business Rules

The 24-hour processing period is calculated based on the date and time processing starts on the batch orders and the date and time processing stops on the batch orders. Mechanical processing starts when SCC (the BellSouth E911 vendor) receives E911 files containing batch orders extracted from the BellSouth Service Order Control System (SOCS). Processing stops when SCC loads the individual records to the E911 database. The E911 database includes updates to the Automatic Location Identification (ALI) database. The system makes no distinction between CLEC resale records and BellSouth retail records.

Calculation

E911 Timeliness = (a / b) X 100

- a = Number of batch orders processed within 24 hours
- b = Total number of batch orders submitted

Report Structure

Reported for the aggregate of CLEC resale updates and BellSouth retail updates

- State
- Region

Data Retained

- Report month
- Aggregate data

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
• None	Parity by Design

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

E-2: Accuracy

Definition

Measures the percent of E911 telephone number (TN) record updates (to CLEC resale and BellSouth retail records) processed successfully for E911 (including the Automatic Location Identification (ALI) database).

Exclusions

- Any resale order canceled by a CLEC
- Facilities-based CLEC orders

Business Rules

Accuracy is based on the number of records processed without error at the conclusion of the processing cycle. Mechanical processing starts when SCC (the BellSouth E911 vendor) receives E911 files containing telephone number (TN) records extracted from BellSouth's Service Order Control System (SOCS). The system makes no distinction between CLEC resale records and BellSouth retail records.

Calculation

E911 Accuracy = (a / b) X 100

- a = Number of record individual updates processed with no errors
- b = Total number of individual record updates

Report Structure

Reported for the aggregate of CLEC resale updates and BellSouth retail updates

- State
- Region

Data Retained

- Report month
- Aggregate data

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
• None	Parity by Design

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark	
Not Applicable	Not Applicable	

E-3: Mean Interval

Definition

Measures the mean interval processing of E911 batch orders (to update CLEC resale and BellSouth retail records) including processing against the Automatic Location Identification (ALI) database.

Exclusions

- Any resale order canceled by a CLEC
- Facilities-based CLEC orders

Business Rules

The processing period is calculated based on the date and time processing starts on the batch orders and the date and time processing stops on the batch orders. Data is posted is 4-hour increments up to and beyond 24 hours. The system makes no distinction between CLEC resale records and BellSouth retail records.

Calculation

E911 Interval = (a - b)

- a = Date and time of batch order completion
- b = Date and time of batch order submission
- **E911 Mean Interval** = (c / d)
- c = Sum of all E911 Intervals
- d = Number of batch orders completed

Report Structure

Reported for the aggregate of CLEC resale updates and BellSouth retail updates

- State
- Region

Data Retained

- Report month
- Aggregate data

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
• None	Parity by Design

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

Section 9: Trunk Group Performance

TGP-1: Trunk Group Performance-Aggregate

Definition

The Trunk Group Performance report displays, over a reporting cycle, aggregate, average trunk group blocking data for each hour of each day of the reporting cycle, for both CLEC affecting and BellSouth affecting trunk groups.

Exclusions

- Trunk groups for which valid data is not available for an entire study period
- Duplicate trunk group information
- Trunk groups blocked due to CLEC network/equipment failure
- Trunk groups blocked due to CLEC delayed or refused orders
- Trunk groups blocked due to unanticipated significant increases in CLEC traffic
- Final groups actually overflowing, not blocked

Business Rules

The purpose of the Trunk Group Performance Report is to provide trunk blocking measurements on CLEC and BellSouth trunk groups for comparison only. It is not the intent of the report that it be used for network management and/or engineering.

Monthly Average Blocking:

- The reporting cycle includes both business and non-business days in a calendar month.
- Monthly average blocking values are calculated for each trunk group for each of the 24 time consistent hours across a reporting cycle.

Aggregate Monthly Blocking:

- Used to compare aggregate blocking across trunk groups which terminate traffic at CLEC points of presence versus BellSouth switches.
- Aggregate monthly blocking data is calculated for each hour of the day across all trunk groups assigned to a category.

Trunk Categorization:

This report displays, over a reporting cycle, aggregate, average blocking data for each hour of a day. Therefore, for each reporting cycle, 24 blocking data points are generated for two aggregate groups of selected trunk groups. These groups are CLEC affecting and BellSouth affecting trunk groups. In order to assign trunk groups to each aggregate group, all trunk groups are first assigned to a category. A trunk group's end points and the type of traffic that is transmitted on it define a category. Selected categories of trunk groups are assigned to the aggregate groups so that trunk reports can be generated. The categories to which trunk groups have been assigned for this report are as follows.

CLEC Affecting Categories:

	Point A	Point B
Category 1:	BellSouth End Office	BellSouth Access Tandem
Category 3:	BellSouth End Office	CLEC Switch
Category 4:	BellSouth Local Tandem	CLEC Switch
Category 5:	BellSouth Access Tandem	CLEC Switch
Category 10:	BellSouth End Office	BellSouth Local Tandem
Category 16:	BellSouth Tandem	BellSouth Tandem
BellSouth Affectin	g Categories:	
	Point A	Point B
Category 9:	BellSouth End Office	BellSouth End Office
Coloulation		

Calculation

Monthly Average Blocking:

• For each hour of the day, each day's raw data are summed across all valid measurements days in a report cycle for blocked and attempted calls.

• The sum of the blocked calls is divided by the total number of calls attempted in a reporting period.

Aggregate Monthly Blocking:

- For each hour of the day, the monthly sums of the blocked and attempted calls from each trunk group are separately aggregated over all trunk groups within each assigned category.
- The total blocked calls is divided by the total call attempts within a group to calculate an aggregate monthly blocking for each assigned group.
- The result is an aggregate monthly average blocking value for each of the 24 hours by group.
- The difference between the CLEC and BellSouth affecting trunk groups are also calculated for each hour.

Report Structure

- CLEC Aggregate
- BellSouth Aggregate
- State

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Report Month
Total Trunk Groups	Total Trunk Groups
Number of Trunk Groups by CLEC	 Aggregate Hourly Blocking Per Trunk Group
Hourly Blocking Per Trunk Group	Hourly Usage Per Trunk Group
Hourly Usage Per Trunk Group	Hourly Call Attempts Per Trunk Group
Hourly Call Attempts Per Trunk Group	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
CLEC aggregate	• Any 2 hour period in 24 hours where CLEC blockage
BellSouth aggregate	exceeds BellSouth blockage by more than 0.5% using
	trunk groups 1, 3, 4, 5, 10, 16 for CLECs and 9 for
	BellSouth

SEEM Measure

SEEM Measure		
Yes	Tier I	
	Tier II	Х

SEEM Disaggregation	SEEM Analog/Benchmark
CLEC Aggregate	• Any 2 hour period in 24 hours where CLEC blockage
BellSouth Aggregate	exceeds BellSouth blockage by more than 0.5% using
	trunk groups 1,3,4,5,10,16 for CLECs and 9 for
	BellSouth

TGP-2: Trunk Group Performance-CLEC Specific

Definition

The Trunk Group Performance report displays, over a reporting cycle, aggregate, average trunk group blocking data for each hour of each day of the reporting cycle, for both CLEC affecting and BellSouth affecting trunk groups.

Exclusions

- Trunk Groups for which valid data is not available for an entire study period
- Duplicate trunk group information
- Trunk groups blocked due to CLEC network/equipment failure
- · Trunk groups blocked due to CLEC delayed or refused orders
- Trunk groups blocked due to unanticipated significant increases in CLEC traffic
- · Final groups actually overflowing, not blocked

Business Rules

The purpose of the Trunk Group Performance Report is to provide trunk blocking measurements on CLEC and BellSouth trunk groups for comparison only. It is not the intent of the report that it be used for network management and/or engineering.

Monthly Average Blocking:

- The reporting cycle includes both business and non-business days in a calendar month.
- Monthly average blocking values are calculated for each trunk group for each of the 24 time consistent hours across a reporting cycle.

Aggregate Monthly Blocking:

- Used to compare aggregate blocking across trunk groups which terminate traffic at CLEC points of presence versus BellSouth switches.
- · Aggregate monthly blocking data is calculated for each hour of the day across all trunk groups assigned to a category.

Trunk Categorization:

• This report displays, over a reporting cycle, aggregate, average blocking data for each hour of a day. Therefore, for each reporting cycle, 24 blocking data points are generated for two aggregate groups of selected trunk groups. These groups are CLEC affecting and BellSouth affecting trunk groups. In order to assign trunk groups to each aggregate group, all trunk groups are first assigned to a category. A trunk group's end points and the type of traffic that is transmitted on it define a category. Selected categories of trunk groups are assigned to the aggregate groups so that trunk reports can be generated. The categories to which trunk groups have been assigned for this report are as follows.

CLEC Affecting Categories:

	Point A	Point B
Category 1: Category 3: Category 4:	BellSouth End Office BellSouth End Office BellSouth Local Tandem	BellSouth Access Tandem CLEC Switch CLEC Switch
Category 5: Category 10:	BellSouth Access Tandem BellSouth End Office	CLEC Switch BellSouth Local Tandem
Category 16: BellSouth Affecting Cate	BellSouth Tandem gories:	BellSouth Tandem
	Point A	Point B

Category 9: BellSouth End Office BellSouth End Office

Calculation

Monthly Average Blocking:

- For each hour of the day, each day's raw data are summed across all valid measurements days in a report cycle for blocked and attempted calls.
- The sum of the blocked calls is divided by the total number of calls attempted in a reporting period.

Aggregate Monthly Blocking:

- For each hour of the day, the monthly sums of the blocked and attempted calls from each trunk group are separately aggregated over all trunk groups within each assigned category.
- The total blocked calls is divided by the total call attempts within a group to calculate an aggregate monthly blocking for each assigned group.
- The result is an aggregate monthly average blocking value for each of the 24 hours by group.
- The difference between the CLEC and BellSouth affecting trunk groups are also calculated for each hour.

Report Structure

- CLEC Specific
- State

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	• Report Month
Total Trunk Groups	Total Trunk Groups
Number of Trunk Groups by CLEC	 Aggregate Hourly Blocking Per Trunk Group
Hourly Blocking Per Trunk Group	Hourly Usage Per Trunk Group
Hourly Usage Per Trunk Group	Hourly Call Attempts Per Trunk Group
Hourly Call Attempts Per Trunk Group	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
CLEC Trunk Group	• Any 2 hour period in 24 hours where CLEC blockage
	exceeds BellSouth blockage by more than 0.5% using
	trunk groups 1, 3, 4, 5, 10, 16 for CLECs and 9 for
	BellSouth

SEEM Measure

SEEM Measure		
Yes	Tier I	Х
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
CLEC Trunk Group	• Any 2 hour period in 24 hours where CLEC blockage
BellSouth Trunk Group	exceeds BellSouth blockage by more than 0.5% using
	trunk groups 1, 3, 4, 5, 10, 16 for CLECs and 9 for
	BellSouth

Section 10: Collocation

C-1: Collocation Average Response Time

Definition

Measures the average time (counted in calendar days) from the receipt of a complete and accurate collocation application (including receipt of application fee if required) to the date BellSouth returns a response electronically or in writing. Within 10 calendar days after having received a bona fide application for physical collocation, BellSouth must respond as to whether space is available or not.

Exclusions

Any application canceled by the CLEC.

Business Rules

The clock starts on the date that BellSouth receives a complete and accurate collocation application accompanied by the appropriate application fee if required. The clock stops on the date that BellSouth returns a response. The clock will restart upon receipt of changes to the original application request.

Calculation

Response Time = (a - b)

- a = Request Response Date
- b = Request Submission Date

Average Response Time = (c / d)

- c = Sum of all Response Times
- d = Count of Responses Returned within Reporting Period

Report Structure

- Individual CLEC (alias) Aggregate
- Aggregate of all CLECs

Data Retained

- Report Period
- Aggregate Data

SQM Disaggregation - Analog/Benchmark

Level of Disaggregation	SQM Analog/Benchmark
• State	Virtual - 20 Calendar Days
Virtual-Initial	 Physical Caged - 30 Calendar Days
Virtual-Augment	 Physical Cageless - 30 Calendar Days
Physical Caged-Initial	
Physical Caged-Augment	
Physical-Cageless-Initial	
Physical Cageless-Augment	

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

C-2: Collocation Average Arrangement Time

Definition

Measures the average time (counted in calendar days) from receipt of a complete and accurate Bona Fide firm order (including receipt of appropriate fee if required) to the date BellSouth completes the collocation arrangement and notifies the CLEC.

Exclusions

- Any Bona Fide firm order canceled by the CLEC
- Any Bona Fide firm order with a CLEC-negotiated interval longer than the benchmark interval

Business Rules

The clock starts on the date that BellSouth receives a complete and accurate Bone Fide firm order accompanied by the appropriate fee. The clock stops on the date that BellSouth completes the collocation arrangement and notifies the CLEC.

Calculation

Arrangement Time = (a - b)

- a = Date Collocation Arrangement is Complete
- b = Date Order for Collocation Arrangement Submitted

Average Arrangement Time = (c / d)

- c = Sum of all Arrangement Times
- d = Total Number of Collocation Arrangements Completed during Reporting Period

Report Structure

- Individual CLEC (alias) Aggregate
- Aggregate of all CLECs

Data Retained

- Report Period
- Aggregate Data

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
• State	• Virtual - 50 Calendar Days (Ordinary)
Virtual-Initial	• Virtual - 75 Calendar Days (Extraordinary)
Virtual-Augment	 Physical Caged - 90 Calendar Days
Physical Caged-Initial	 Physical Cageless - 60 Calendar Days (Ordinary)
Physical Caged-Augment	 Physical Cageless - 90 Calendar Days (Extraordinary)
Physical Cageless-Initial	
Physical Cageless-Augment	

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

C-3: Collocation Percent of Due Dates Missed

Definition

Measures the percent of missed due dates for both virtual and physical collocation arrangements.

Exclusions

Any Bona Fide firm order canceled by the CLEC.

Business Rules

Percent Due Dates Missed is the percent of total collocation arrangements which BellSouth is unable to complete by end of the BellSouth committed due date. The clock starts on the date that BellSouth receives a complete and accurate Bona Fide firm order accompanied by the appropriate fee if required. The arrangement is considered a missed due date if it is not completed on or before the committed due date.

Calculation

% of Due Dates Missed = $(a / b) \times 100$

- a = Number of Completed Orders that were not completed within BellSouth Committed Due Date during Reporting Period
- b = Number of Orders Completed in Reporting Period

Report Structure

- Individual CLEC (alias) Aggregate
- Aggregate of all CLECs

Data Retained

- Report Period
- Aggregate Data

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark	
• State	• >= 95% on time	
• Virtual-Initial		
• Virtual-Augment		
Physical Caged-Initial		
Physical Caged-Augment		
Physical Cageless-Initial		
Physical Cageless-Augment		

SEEM Measure

SEEM Measure		
Yes	Tier I	Х
	Tier II	Х

SEEM Disaggregation	SEEM Analog/Benchmark
All Collocation Arrangements	• >= 95% on time

Section 11: Change Management

CM-1: Timeliness of Change Management Notices

Definition

Measures whether CLECs receive required software release notices on time to prepare for BellSouth interface/system changes so CLEC interfaces are not impaired by change.

Exclusions

- Changes to release dates for reasons outside BellSouth control, such as the system software vendor changes. For example: a patch to fix a software problem.
- Type 6 Change Requests (Defects/Expedites), as defined by the Change Control Process (CCP)

Business Rules

This metric is designed to measure the percent of change management notices sent to the CLECs according to notification standards and time frames set forth in the Change Control Process. The CCP is used by BellSouth and the CLECs to manage requested changes to the BellSouth Local Interfaces.

The clock starts on the notification date. The clock stops on the software release date. When project events occur (scope changes, analysis information, etc.), the software release date may change. A revised notification would be required and the clock would restart. Based on release constraints for defects/expedites, notification may be less than the agreed upon interval in the CCP for new features.

Calculation

Timeliness of Change Management Notices = (a / b) X 100

- a = Total number of Change Management Notifications Sent Within Required Timeframes
- b = Total Number of Change Management Notifications Sent

Report Structure

• BellSouth Aggregate

Data Retained

- Report Period
- Notice Date
- Release Date

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
Region	• 95% >= 30 Days of Release

SEEM Measure

SEEM Measure			
Yes	Tier I		
	Tier II	Х	

SEEM Disaggregation	SEEM Analog/Benchmark
Region	• 95% >= 30 Days of Release

CM-2: Change Management Notice Average Delay Days

Definition

Measures the average delay days for change management system release notices sent outside the time frame set forth in the Change Control Process.

Exclusions

- Changes to release dates for reasons outside BellSouth control, such as the system software vendor changes. For example: a patch to fix a software problem
- Type 6 Change Requests (Defects/Expedites), as defined by the Change Control Process

Business Rules

This metric is designed to measure the percent of change management notices sent to the CLECs according to notification standards and time frames set forth in the Change Control Process. The CCP is used by BellSouth and the CLECs to manage requested changes to the BellSouth Local Interfaces.

The clock starts on the notification due date. The clock stops on the software release date. When project events occur (scope changes, analysis information, etc.), the software release date may change. A revised notification would be required and the clock would restart. Based on release constraints for defects/expedites, notification may be less than the agreed upon interval in the CCP for new features.

Calculation

Change Management Notice Delay Days = (a - b)

- a = Date Notice Sent
- b = Date Notice Due

Change Management Notice Average Delay Days = (c / d)

- c = Sum of all Change Management Notice Delay Days
- d = Total Number of Notices Sent Late

Report Structure

• BellSouth Aggregate

Data Retained

- Report Period
- Notice Date
- Release Date

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
Region	• <= 8 Days

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable
CM-3: Timeliness of Documents Associated with Change

Definition

Measures whether CLECs received requirements or business rule documentation on time to prepare for BellSouth interface/system changes so CLEC interfaces are not impaired by change.

Exclusions

- Documentation for release dates that slip less than 30 days for reasons outside BellSouth control, such as changes due to Regulatory mandate or CLEC request
- Type 6 Change Requests (Defects/Expedites), as defined by the Change Control Process

Business Rules

This metric is designed to measure the percent of requirements or business rule documentation sent to the CLECs according to documentation standards and timeframes set forth in the Change Control Process. The CCP is used by BellSouth and the CLECs to manage requested changes to the BellSouth Local Interfaces.

The clock starts on the business rule documentation release date. The clock stops on the software release date. When project events occur (scope changes, analysis information, etc.), the software release date may change. Revisions to documentation could be required and the clock would restart.

Calculation

Timeliness of Documents Associated with Change = (a / b) X 100

- a = Change Management Documentation Sent Within Required Timeframes after Notices
- b = Total Number of Change Management Documentation Sent

Report Structure

• BellSouth Aggregate

Data Retained

- Report Period
- Notice Date
- Release Date

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
• Region	 95% >= 30 days if new features coding is required 95% >= 5 days for documentation defects, corrections or clarifications

SEEM Measure

SEEM Measure		
Yes	Tier I	
	Tier II	Х

SEEM Disaggregation	SEEM Analog/Benchmark
Region	• 95% >= 30 days of the change

CM-4: Change Management Documentation Average Delay Days

Definition

Measures the average delay days for requirements or business rule documentation sent outside the time frames set forth in the Change Control Process.

Exclusions

- Documentation for release dates that slip less than 30 days for reasons outside BellSouth control, such as changes due to Regulatory mandate or CLEC request
- Type 6 Change Requests (Defects/Expedites), as defined by the Change Control Process

Business Rules

This metric is designed to measure the percent of requirements or business rule documentation sent to the CLECs according to documentation standards and time frames set forth in the Change Control Process. The CCP is used by BellSouth and the CLECs to manage requested changes to the BellSouth Local Interfaces.

The clock starts on the business rule documentation release date. The clock stops on the software release date. When project events occur (scope changes, analysis information, etc.), the software release date may change. Revisions to documentation could be required and the clock would restart.

Calculation

Change Management Documentation Delay Days = (a - b)

- a = Date Documentation Provided
- b = Date Documentation Due

Change Management Documentation Average Delay Days = (c / d)

- c = Sum of all CM Documentation Delay Days
- d = Total Change Management Documents Sent

Report Structure

BellSouth Aggregate

Data Retained

- Report Period
- Notice Date
- Release Date

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
• Region	• <= 8 Days

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

CM-5: Notification of CLEC Interface Outages

Definition

Measures the time it takes BellSouth to notify the CLEC of an outage of an interface.

Exclusions

None

Business Rules

This measure is designed to notify the CLEC of interface outages within 15 minutes of BellSouth's verification that an outage has taken place. This metric will be expressed as a percentage.

Calculation

Notification of CLEC Interface Outages = (a / b) X 100

- a = Number of Interface Outages where CLECS are notified within 15 minutes
- b = Total Number of Interface Outages

Report Structure

CLEC Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Number of Interface Outages	Not Applicable
• Number of Notifications <= 15 minutes	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
• By interface type for all interfaces accessed by CLECs	• 97% in 15 Minutes

Interface	Applicable to
EDI	CLEC
CSOTS	CLEC
LENS	CLEC
TAG	CLEC
ECTA	CLEC
TAFI	CLEC/BellSouth

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

Section 12: Bona Fide / New Business Request Process

BFR-1: Percentage of BFR/NBR Requests Processed Within 30 Business Days

Definition

Percentage of Bona Fide/New Business Requests processed within 30 business days for the development and purchases of network elements not currently offered.

Exclusions

• Any application cancelled by the CLEC

Business Rules

The clock starts when BellSouth receives a complete and accurate application. The clock stops when BellSouth completes application processing for Network Elements that are not operational at the time of the request.

Calculation

Percentage of BFR/NBR Requests Processed Within 30 Business Days = (a / b) X 100

- a = Count of number of requests processed within 30 days
- b = Total number of requests

Report Structure

- Individual CLEC (alias) Aggregate
- Aggregate of all CLECs

Data Retained

- Report Period
- Aggregate Data

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
Region	 90% <= 30 business days

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

BFR-2: Percentage of Quotes Provided for Authorized BFR/NBR Requests Processed Within X (10/30/60) Business Days

Definition

Percentage of quotes provided in response to Bona Fide/New Business Requests within X (10/30/60) business days for network elements not currently offered.

Exclusions

• Requests that are subject to pending arbitration

Business Rules

The clock starts when BellSouth receives a complete and accurate application. The clock stops when BellSouth responds back to the application with a price quote.

Calculation

Percentage of Quotes Provided for Authorized BFR/NBR Requests Processed Within X (10/30/60) Business Days = (a / b) X 100

- a = Count of number of requests processed within "X" days
- b = Total number of requests
 - where "X" = 10, 30, or 60 days

Report Structure

- New Network Elements that are operational at the time of the request
- New Network Elements that are ordered by the FCC
- New Network Elements that are not operational at the time of the request

Data Retained

- Report Period
- Aggregate Data

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
Region	• 90% <= 10/30/60 business days
	- Network Elements that are operational at the time of
	the request -10 days
	- Network Elements that are Ordered by the FCC -30
	days
	- New Network Elements – 90 days

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

Appendix A: Reporting Scope

A-1: Standard Service Groupings

See individual reports in the body of the SQM.

A-2: Standard Service Order Activities

These are the generic BellSouth/CLEC service order activities which are included in the Pre-Ordering, Ordering, and Provisioning sections of this document. It is not meant to indicate specific reporting categories.

Service Order Activity Types

- Service Migrations Without Changes
- Service Migrations With Changes
- Move and Change Activities
- Service Disconnects (Unless noted otherwise)
- New Service Installations

Pre-Ordering Query Types

- Address
- Telephone Number
- Appointment Scheduling
- Customer Service Record
- Feature Availability
- Service Inquiry

Maintenance Query Types:

TAFI - TAFI queries the systems below

- CRIS
- March
- Predictor
- LMOS
 - DLR
 - DLETH
 - LMOSupd
- LNP
- NIW
- OSPCM
- SOCS

Report Levels

- CLEC RESH
- CLEC State
- CLEC Region
- Aggregate CLEC State
- Aggregate CLEC Region
- BellSouth State
- BellSouth Region

Appendix B: Glossary of Acronyms and Terms

Symbols used in calculations

Σ

A mathematical symbol representing the sum of a series of values following the symbol.

A mathematical operator representing subtraction.

+

A mathematical operator representing addition.

/

A mathematical operator representing division.

<

A mathematical symbol that indicates the metric on the left of the symbol is less than the metric on the right.

<=

A mathematical symbol that indicates the metric on the left of the symbol is less than or equal to the metric on the right.

>

A mathematical symbol that indicates the metric on the left of the symbol is greater than the metric on the right.

>=

A mathematical symbol that indicates the metric on the left of the symbol is greater than or equal to the metric on the right.

()

Parentheses, used to group mathematical operations which are completed before operations outside the parentheses.

Α

ACD

Automatic Call Distributor - A service that provides status monitoring of agents in a call center and routes high volume incoming telephone calls to available agents while collecting management information on both callers and attendants.

Aggregate

Sum total of all items in like category, e.g. CLEC aggregate equals the sum total of all CLECs' data for a given reporting level.

ALEC

Alternative Local Exchange Company = FL CLEC

ADSL

Asymmetrical Digital Subscriber Line

ASR

Access Service Request - A request for access service terminating delivery of carrier traffic into a Local Exchange Carrier's network.

ATLAS

Application for Telephone Number Load Administration System - The BellSouth Operations System used to administer the pool of available telephone numbers and to reserve selected numbers from the pool for use on pending service requests/service orders.

ATLASTN

ATLAS software contract for Telephone Number.

Auto Clarification

The number of LSRs that were electronically rejected from LESOG and electronically returned to the CLEC for correction.

В

BFR:

Bona Fide Request

BILLING

The process and functions by which billing data is collected and by which account information is processed in order to render accurate and timely billing.

BOCRIS

Business Office Customer Record Information System (Front-end to the CRIS database.)

BRI

Basic Rate ISDN

BRC

Business Repair Center - The BellSouth Business Systems trouble receipt center which serves business and CLEC customers.

BellSouth

BellSouth Telecommunications, Inc.

С

CABS Carrier Access Billing System

CCC Coordinated Customer Conversions

ССР

Change Control Process

Centrex

A business telephone service, offered by local exchange carriers, which is similar to a Private Branch Exchange (PBX) but the switching equipment is located in the telephone company Central Office (CO).

CKTID

A unique identifier for elements combined in a service configuration

CLEC

Competitive Local Exchange Carrier

CLP

Competitive Local Provider = NC CLEC

СМ

Change Management

CMDS

Centralized Message Distribution System - Telcordia administered national system used to transfer specially formatted messages among companies.

COFFI

Central Office Feature File Interface - Provides information about USOCs and class of service. COFFI is a part of DOE/ SONGS. It indicates all services available to a customer.

COG

Corporate Gateway - Telcordia product designed for the electronic submission of xDSL Local Service Requests.

CRIS

Customer Record Information System - The BellSouth proprietary corporate database and billing system for non-access customers and services.

CRSACCTS

CRIS software contract for CSR information

CRSG

Complex Resale Support Group

C-SOTS

CLEC Service Order Tracking System

CSR

Customer Service Record

CTTG

Common Transport Trunk Group - Final trunk groups between BellSouth & Independent end offices and the BellSouth access tandems.

CWINS Center

Customer Wholesale Interconnection Network Services Center (formerly the UNE Center).

D

DA

Directory Assistance

Design

Design Service is defined as any Special or Plain Old Telephone Service Order which requires BellSouth Design Engineering Activities.

Disposition & Cause

Types of trouble conditions, e.g. No Trouble Found, Central Office Equipment, Customer Premises Equipment, etc.

DLETH

Display Lengthy Trouble History - A history report that gives all activity on a line record for trouble reports in LMOS.

DLR

Detail Line Record - All the basic information maintained on a line record in LMOS, e.g. name, address, facilities, features etc.

DS-0

The worldwide standard speed for one digital voice signal (64000 bps).

DS-1

24 DS-0s (1.544Mb/sec., i.e. carrier systems)

DOE

Direct Order Entry System - An internal BellSouth service order entry system used by BellSouth Service Representatives to input business service orders in BellSouth format.

DOM

Delivery Order Manager - Telcordia product designed for the electronic submission of xDSL Local Service Requests.

DSAP

DOE (Direct Order Entry) Support Application - The BellSouth Operations System which assists a Service Representative or similar carrier agent in negotiating service provisioning commitments for non-designed services and Unbundled Network Elements.

DSAPDDI

DSAP software contract for schedule information.

DSL

Digital Subscriber Line

DUI

Database Update Information

Ε

E911

Provides callers access to the applicable emergency services bureau by dialing a 3-digit universal telephone number.

EDI

Electronic Data Interchange - The computer-to-computer exchange of inter and/or intra-company business documents in a public standard format.

ESSX

BellSouth Centrex Service

F

Fatal Reject

LSRs electronically rejected from LEO, which checks to see of the LSR has all the required fields correctly populated.

Flow-Through

In the context of this document, LSRs submitted electronically via the CLEC mechanized ordering process that flow through to the BellSouth OSS without manual or human intervention.

FOC

Firm Order Confirmation - A notification returned to the CLEC confirming that the LSR has been received and accepted, including the specified commitment date.

FX

Foreign Exchange

GΗ

HAL

"Hands Off" Assignment Logic - Front end access and error resolution logic used in interfacing BellSouth Operations Systems such as ATLAS, BOCRIS, LMOS, PSIMS, RSAG and SOCS.

HALCRIS

HAL software contract for CSR information

HDSL

High Density Subscriber Loop/Line

IJK

ILEC Incumbent Local Exchange Company

INP

Interim Number Portability

ISDN

Integrated Services Digital Network

IPC

Interconnection Purchasing Center

L

LAN Local Area Network

LAUTO

The automatic processor in the LNP Gateway that validates LSRs and issues service orders.

LCSC

Local Carrier Service Center - The BellSouth center which is dedicated to handling CLEC LSRs, ASRs, and Preordering transactions along with associated expedite requests and escalations.

Legacy System

Term used to refer to BellSouth Operations Support Systems (see OSS)

LENS

Local Exchange Negotiation System - The BellSouth LAN/web server/OS application developed to provide both preordering and ordering electronic interface functions for CLECs.

LEO

Local Exchange Ordering - A BellSouth system which accepts the output of EDI, applies edit and formatting checks, and reformats the Local Service Requests in BellSouth Service Order format.

LERG

Local Exchange Routing Guide

LESOG

Local Exchange Service Order Generator - A BellSouth system which accepts the service order output of LEO and enters the Service Order into the Service Order Control System using terminal emulation technology.

LFACS

Loop Facilities Assessment and Control System

LIDB

Line Information Database

LISC

Local Interconnection Service Center - The center that issues trunk orders.

LMOS

Loop Maintenance Operations System - A BellSouth Operations System that stores the assignment and selected account information for use by downstream OSS and BellSouth personnel during provisioning and maintenance activities.

LMOS HOST

LMOS host computer

LMOSupd LMOS updates

LMU Loop Make-up

LMUS

Loop Make-up Service Inquiry

LNP

Local Number Portability - In the context of this document, the capability for a subscriber to retain his current telephone number as he transfers to a different local service provider.

Loops

Transmission paths from the central office to the customer premises.

LRN

Location Routing Number

LSR

Local Service Request - A request for local resale service or unbundled network elements from a CLEC.

Μ

Maintenance & Repair

The process and function by which trouble reports are passed to BellSouth and by which the related service problems are resolved.

MARCH

BellSouth Operations System which accepts service orders, interprets the coding contained in the service order image, and constructs the specific switching system Recent Change command messages for input into end office switches.

Ν

NBR

New Business Request

NC

"No Circuits" - All circuits busy announcement.

NIW

Network Information Warehouse

NMLI

Native Mode LAN Interconnection

NPA

Numbering Plan Area

NXX

The "exchange" portion of a telephone number.

0

OASIS

Obtain Availability Services Information System - A BellSouth front-end processor, which acts as an interface between COFFI and RNS. This system takes the USOCs in COFFI and translates them to English for display in RNS.

OASISBSN

OASIS software contract for feature/service

OASISCAR

OASIS software contract for feature/service

OASISLPC OASIS software contract for feature/service

OASISMTN

OASIS software contract for feature/service

OASISNET

OASIS software contract for feature/service

OASISOCP

OASIS software contract for feature/service

ORDERING

The process and functions by which resale services or unbundled network elements are ordered from BellSouth as well as the process by which an LSR or ASR is placed with BellSouth.

OSPCM

Outside Plant Contract Management System - Provides Scheduling Information.

OSS

Operations Support System - A support system or database which is used to mechanize the flow or performance of work. The term is used to refer to the overall system consisting of hardware complex, computer operating system(s), and application which is used to provide the support functions.

Out Of Service

Customer has no dial tone and cannot call out.

Ρ

PMAP

Performance Measurement Analysis Platform

PMQAP

Performance Measurement Quality Assurance Plan

PON

Purchase Order Number

POTS

Plain Old Telephone Service

PREDICTOR

The BellSouth Operations system which is used to administer proactive maintenance and rehabilitation activities on outside plant facilities, provide access to selected work groups (e.g. RRC & BRC) to Mechanized Loop Testing and switching system I/O ports, and provide certain information regarding the attributes and capabilities of outside plant facilities.

Preordering

The process and functions by which vital information is obtained, verified, or validated prior to placing a service request.

PRI

Primary Rate ISDN

Provisioning

The process and functions by which necessary work is performed to activate a service requested via an LSR or ASR and to initiate the proper billing and accounting functions.

PSIMS

Product/Service Inventory Management System - A BellSouth database Operations System which contains availability information on switching system features and capabilities and on BellSouth service availability. This database is used to verify the availability of a feature or service in an NXX prior to making a commitment to the customer.

PSIMSORB

PSIMS software contract for feature/service.

QR

RNS

Regional Negotiation System - An internal BellSouth service order entry system used by BellSouth Consumer Services to input service orders in BellSouth format.

ROS

Regional Ordering System

RRC

Residence Repair Center - The BellSouth Consumer Services trouble receipt center which serves residential customers.

RSAG

Regional Street Address Guide - The BellSouth database, which contains street addresses validated to be accurate with state and local governments.

RSAGADDR

RSAG software contract for address search.

RSAGTN

RSAG software contract for telephone number search.

S

SAC Service Advocacy Center

SEEM

Self Effectuating Enforcement Mechanism

SOCS

Service Order Control System - The BellSouth Operations System which routes service order images among BellSouth drop points and BellSouth Operations Systems during the service provisioning process.

SOG

Service Order Generator - Telcordia product designed to generate a service order for xDSL.

SOIR

Service Order Interface Record - any change effecting activity to a customer account by service order that impacts 911/E911

SONGS

Service Order Negotiation and Generation System.

Т

TAFI

Trouble Analysis Facilitation Interface - The BellSouth Operations System that supports trouble receipt center personnel in taking and handling customer trouble reports.

TAG

Telecommunications Access Gateway – TAG was designed to provide an electronic interface, or machine-to-machine interface for the bi-directional flow of information between BellSouth's OSSs and participating CLECs.

TN

Telephone Number

Total Manual Fallout

The number of LSRs which are entered electronically but require manual entering into a service order generator.

UV

UNE Unbundled Network Element

UCL Unbundled Copper Link

USOC Universal Service Order Code

WXYZ

WATS Wide Area Telephone Service

WFA Work Force Administration

WMC Work Management Center

WTN Working Telephone Number.

Appendix C: Appendix C: BellSouth Audit Policy

BellSouth currently provides many CLECs with certain audit rights as a part of their individual interconnection agreements. However, it is not reasonable for BellSouth to undergo an audit of the SQM for every CLEC with which it has a contract. BellSouth has developed a proposed Audit Plan for use by the parties to an audit. If requested by a Public Service Commission or by a CLEC exercising contractual audit rights, BellSouth will agree to undergo a comprehensive audit of the aggregate level reports for both BellSouth and the CLEC(s) each of the next five (5) years (2001-2005) to be conducted by an independent third party. The results of that audit will be made available to all the parties subject to proper safeguards to protect proprietary information. This aggregate level audit includes the following specifications:

- 1. The cost shall be borne 50% by BellSouth and 50% by the CLEC or CLECs.
- 2. The independent third party auditor shall be selected with input from BellSouth, the PSC, if applicable, and the CLEC(s).
- 3. BellSouth, the PSC and the CLEC(s) shall jointly determine the scope of the audit.

BellSouth reserves the right to make changes to this audit policy as growth and changes in the industry dictate.

Attachment 10

BellSouth Disaster Recovery Plan

CONTENTS

PAGE

1.0	Purpo	se	2		
2.0	Single Point of Contact				
3.0	Identifying the Problem				
		Site Control	3		
	3.2	Environmental Concerns	4		
4.0	The E	Emergency Control Center (ECC)			
5.0	Recovery Procedures				
	5.1	CLEC Outage	5		
	5.2	BellSouth Outage	5		
		5.2.1 Loss of Central Office	6		
		5.2.2 Loss of a Central Office with Serving Wire Center Functions	6		
		5.2.3 Loss of a Central Office with Tandem Functions	6		
		5.2.4 Loss of a Facility Hub	6		
	5.3	Combined Outage (CLEC and BellSouth Equipment)	7		
6.0	T1 Identification Procedures				
7.0	Acronyms 8				

1.0 PURPOSE

In the unlikely event of a disaster occurring that affects BellSouth's long-term ability to deliver traffic to a Competitive Local Exchange Carrier (CLEC), general procedures have been developed to hasten the recovery process. Since each location is different and could be affected by an assortment of potential problems, a detailed recovery plan is impractical. However, in the process of reviewing recovery activities for specific locations, some basic procedures emerge that appear to be common in most cases.

These general procedures should apply to any disaster that affects the delivery of traffic for an extended time period. Each CLEC will be given the same consideration during an outage, and service will be restored as quickly as possible.

This document will cover the basic recovery procedures that would apply to every CLEC.

2.0 SINGLE POINT OF CONTACT

When a problem is experienced, regardless of the severity, the BellSouth Network Management Center (NMC) will observe traffic anomalies and begin monitoring the situation. Controls will be appropriately applied to insure the sanity of BellSouth's network; and, in the event that a switch or facility node is lost, the NMC will attempt to circumvent the failure using available reroutes.

BellSouth's NMC will remain in control of the restoration efforts until the problem has been identified as being a long-term outage. At that time, the NMC will contact BellSouth's Emergency Control Center (ECC) and relinquish control of the recovery efforts. Even though the ECC may take charge of the situation, the NMC will continue to monitor the circumstances and restore traffic as soon as damaged network elements are revitalized.

The telephone number for the BellSouth Network Management Center in Atlanta, as published in Telcordia's National Network Management Directory, is 404-321-2516.

3.0 IDENTIFYING THE PROBLEM

During the early stages of problem detection, the NMC will be able to tell which CLECs are affected by the catastrophe. Further analysis and/or first hand observation will determine if the disaster has affected CLEC equipment only, BellSouth equipment only or a combination. The initial restoration activity will be largely determined by the equipment that is affected.

Once the nature of the disaster is determined and after verifying the cause of the problem, the NMC will initiate reroutes and/or transfers that are jointly agreed upon by the affected CLECs' Network Management Center and the BellSouth NMC. The type and percentage of controls used will depend upon available network capacity. Controls necessary to stabilize the situation will be invoked and the NMC will attempt to re-establish as much traffic as possible.

For long-term outages, recovery efforts will be coordinated by the Emergency Control Center (ECC). Traffic controls will continue to be applied by the NMC until facilities are re-established. As equipment is made available for service, the ECC will instruct the NMC to begin removing the controls and allow traffic to resume.

3.1 SITE CONTROL

In the total loss of building use scenario, what likely exists will be a smoking pile of rubble. This rubble will contain many components that could be dangerous. It could also contain any personnel on the premises at the time of the disaster. For these reasons, the local fire marshal with the assistance of the police will control the site until the building is no longer a threat to surrounding properties and the companies have secured the site from the general public.

During this time, the majority owner of the building should be arranging for a demolition contractor to mobilize to the site with the primary objective of reaching the cable entrance facility for a damage assessment. The results of this assessment would then dictate immediate plans for restoration, both short term and permanent.

In a less catastrophic event, i.e., the building is still standing and the cable entrance facility is usable, the situation is more complex. The site will initially be controlled by local authorities until the threat to adjacent property has diminished. Once the site is returned to the control of the companies, the following events should occur.

An initial assessment of the main building infrastructure systems (mechanical, electrical, fire and life safety, elevators, and others) will establish building needs. Once these needs are determined, the majority owner should lead the building restoration efforts. There may be situations where the site will not be totally restored within the confines of the building. The companies must individually determine their needs and jointly assess the cost of permanent restoration to determine the overall plan of action.

Multiple restoration trailers from each company will result in the need for designated space and installation order. This layout and control is required to maximize the amount of restoration equipment that can be placed at the site, and the priority of placements.

Care must be taken in this planning to ensure other restoration efforts have logistical access to the building. Major components of telephone and building equipment will need to be removed and replaced. A priority for this equipment must also be jointly established to facilitate overall site restoration. (Example: If the AC switchgear has sustained damage, this would be of the highest priority in order to regain power, lighting, and HVAC throughout the building.)

If the site will not accommodate the required restoration equipment, the companies would then need to quickly arrange with local authorities for street closures, rights of way or other possible options available.

3.2 ENVIRONMENTAL CONCERNS

In the worse case scenario, many environmental concerns must be addressed. Along with the police and fire marshal, the state environmental protection department will be on site to monitor the situation.

Items to be concerned with in a large central office building could include:

1. Emergency engine fuel supply. Damage to the standby equipment and the fuel handling equipment could have created "spill" conditions that have to be handled within state and federal regulations.

2. Asbestos-containing materials that may be spread throughout the wreckage. Asbestos could be in many components of building, electrical, mechanical, outside plant distribution, and telephone systems.

3. Lead and acid. These materials could be present in potentially large quantities depending upon the extent of damage to the power room.

4. Mercury and other regulated compounds resident in telephone equipment.

5. Other compounds produced by the fire or heat.

Once a total loss event occurs at a large site, local authorities will control immediate clean up (water placed on the wreckage by the fire department) and site access.

At some point, the companies will become involved with local authorities in the overall planning associated with site clean up and restoration. Depending on the clean up approach taken, delays in the restoration of several hours to several days may occur.

In a less severe disaster, items listed above are more defined and can be addressed individually depending on the damage.

In each case, the majority owner should coordinate building and environmental restoration as well as maintain proper planning and site control.

4.0 THE EMERGENCY CONTROL CENTER (ECC)

The ECC is located in the Colonnade Building in Birmingham, Alabama. During an emergency, the ECC staff will convene a group of pre-selected experts to inventory the damage and initiate corrective actions. These experts have regional access to BellSouth's personnel and equipment and will assume control of the restoration activity anywhere in the nine-state area.

In the past, the ECC has been involved with restoration activities resulting from hurricanes, ice storms and floods. They have demonstrated their capabilities during these calamities as well as

during outages caused by human error or equipment failures. This group has an excellent record of restoring service as quickly as possible.

During a major disaster, the ECC may move emergency equipment to the affected location, direct recovery efforts of local personnel and coordinate service restoration activities with the CLECs. The ECC will attempt to restore service as quickly as possible using whatever means is available, leaving permanent solutions, such as the replacement of damaged buildings or equipment, for local personnel to administer.

Part of the ECC's responsibility, after temporary equipment is in place, is to support the NMC efforts to return service to the CLECs. Once service has been restored, the ECC will return control of the network to normal operational organizations. Any long-term changes required after service is restored will be made in an orderly fashion and will be conducted as normal activity.

5.0 RECOVERY PROCEDURES

The nature and severity of any disaster will influence the recovery procedures. One crucial factor in determining how BellSouth will proceed with restoration is whether or not BellSouth's equipment is incapacitated. Regardless of whose equipment is out of service, BellSouth will move as quickly as possible to aid with service recovery; however, the approach that will be taken may differ depending upon the location of the problem.

5.1 CLEC OUTAGE

For a problem limited to one CLEC (or a building with multiple CLECs), BellSouth has several options available for restoring service quickly. For those CLECs that have agreements with other CLECs, BellSouth can immediately start directing traffic to a provisional CLEC for completion. This alternative is dependent upon BellSouth having concurrence from the affected CLECs.

Whether or not the affected CLECs have requested a traffic transfer to another CLEC will not impact BellSouth's resolve to re-establish traffic to the original destination as quickly as possible.

5.2 BELLSOUTH OUTAGE

Because BellSouth's equipment has varying degrees of impact on the service provided to the CLECs, restoring service from damaged BellSouth equipment is different. The outage will probably impact a number of Carriers simultaneously. However, the ECC will be able to initiate immediate actions to correct the problem.

A disaster involving any of BellSouth's equipment locations could impact the CLECs, some more than others. A disaster at a Central Office (CO) would only impact the delivery of traffic to and from that one location, but the incident could affect many Carriers. If the Central Office is a Serving Wire Center (SWC), then traffic from the entire area to those Carriers served from that switch would also be impacted. If the switch functions as an Access Tandem, or there is a tandem in the building, traffic from every CO to every CLEC could be interrupted. A disaster that destroys a facility hub could disrupt various traffic flows, even though the switching equipment may be unaffected.

The NMC would be the first group to observe a problem involving BellSouth's equipment. Shortly after a disaster, the NMC will begin applying controls and finding re-routes for the completion of as much traffic as possible. These reroutes may involve delivering traffic to alternate Carriers upon receiving approval from the CLECs involved. In some cases, changes in translations will be required. If the outage is caused by the destruction of equipment, then the ECC will assume control of the restoration.

5.2.1 Loss of a Central Office

When BellSouth loses a Central Office, the ECC will

- a) Place specialists and emergency equipment on notice;
- b) Inventory the damage to determine what equipment and/or functions are lost;

c) Move containerized emergency equipment and facility equipment to the stricken area, if necessary;

d) Begin reconnecting service for Hospitals, Police and other emergency agencies; and

e) Begin restoring service to CLECs and other customers.

5.2.2 Loss of a Central Office with Serving Wire Center Functions

The loss of a Central Office that also serves as a Serving Wire Center (SWC) will be restored as described in Section 5.2.1.

5.2.3 Loss of a Central Office with Tandem Functions

When BellSouth loses a Central Office building that serves as an Access Tandem and as a SWC, the ECC will

a) Place specialists and emergency equipment on notice;

b) Inventory the damage to determine what equipment and/or functions are lost;

c) Move containerized emergency equipment and facility equipment to the stricken area, if necessary;

d) Begin reconnecting service for Hospitals, Police and other emergency agencies;

e) Re-direct as much traffic as possible to the alternate access tandem (if available) for delivery to those CLECs utilizing a different location as a SWC;

f) Begin aggregating traffic to a location near the damaged building. From this location, begin re-establishing trunk groups to the CLECs for the delivery of traffic normally found on the direct trunk groups. (This aggregation point may be the alternate access tandem location or another CO on a primary facility route.)

g) Begin restoring service to CLECs and other customers.

5.2.4 Loss of a Facility Hub

In the event that BellSouth loses a facility hub, the recovery process is much the same as above. Once the NMC has observed the problem and administered the appropriate controls, the ECC will assume authority for the repairs. The recovery effort will include

a) Placing specialists and emergency equipment on notice;

- b) Inventorying the damage to determine what equipment and/or functions are lost;
- c) Moving containerized emergency equipment to the stricken area, if necessary;
- d) Reconnecting service for Hospitals, Police and other emergency agencies; and

e) Restoring service to CLECs and other customers. If necessary, BellSouth will aggregate the traffic at another location and build temporary facilities. This alternative would be viable for a location that is destroyed and building repairs are required.

5.3 COMBINED OUTAGE (CLEC AND BELLSOUTH EQUIPMENT)

In some instances, a disaster may impact BellSouth's equipment as well as the CLECs'. This situation will be handled in much the same way as described in Section 5.2.3. Since BellSouth and the CLECs will be utilizing temporary equipment, close coordination will be required.

6.0 T1 IDENTIFICATION PROCEDURES

During the restoration of service after a disaster, BellSouth may be forced to aggregate traffic for delivery to a CLEC. During this process, T1 traffic may be consolidated onto DS3s and may become unidentifiable to the Carrier. Because resources will be limited, BellSouth may be forced to "package" this traffic entirely differently than normally received by the CLECs. Therefore, a method for identifying the T1 traffic on the DS3s and providing the information to the Carriers is required.

7.0 ACRONYMS

CO	-	Central Office (BellSouth)
DS3	-	Facility that carries 28 T1s (672 circuits)
ECC	-	Emergency Control Center (BellSouth)
CLEC	-	Competitive Local Exchange Carrier
NMC	-	Network Management Center
SWC	-	Serving Wire Center (BellSouth switch)
T1	-	Facility that carries 24 circuits

Hurricane Information

During a hurricane, BellSouth will make every effort to keep CLECs updated on the status of our network. Information centers will be set up throughout BellSouth Telecommunications. These centers are not intended to be used for escalations, but rather to keep the CLEC informed of network related issues, area damages and dispatch conditions, etc.

Hurricane-related information can also be found on line at <u>http://www.interconnection.bellsouth.com/network/disaster/dis_resp.htm</u>. Information concerning Mechanized Disaster Reports can also be found at this website by clicking on CURRENT MDR REPORTS or by going directly to <u>http://www.interconnection.bellsouth.com/network/disaster/mdrs.htm</u>.

BST Disaster Management Plan

BellSouth maintenance centers have geographical and redundant communication capabilities. In the event of a disaster removing any maintenance center from service another geographical center would assume maintenance responsibilities. The contact numbers will not change and the transfer will be transparent to the CLEC.

Attachment 11

Bona Fide Request and New Business Requests Process

BONA FIDE REQUEST AND NEW BUSINESS REQUESTS PROCESS

- 1.0 The Parties agree that Knology is entitled to order any Network Element, Interconnection option, service option or Resale Service required to be made available by the Communications Act of 1934, as modified by the Telecommunications Act of 1996 (the "Act"), FCC requirements or State Commission requirements. Knology also shall be permitted to request the development of new or revised facilities or service options which are not required by the Act. Procedures applicable to requesting the addition of such facilities or service options are specified in this Attachment 11.
- 2.0 Bona Fide Requests ("BFR") are to be used when Knology makes a request of BellSouth to provide a new or modified network element, interconnection option, or other service option pursuant to the Act that was not previously included in the Agreement. New Business Requests ("NBRs") are to be used when Knology makes a request of BellSouth to provide a new or custom capability or function to meet Knology's business needs that was not previously included in the Agreement.
- 3.0 A BFR or a NBR shall be submitted in writing by Knology and shall specifically identify the required service date, technical requirements, space requirements and/or such specifications that clearly define the request such that BellSouth has sufficient information to analyze and prepare a response. Such a request also shall include a Knology's designation of the request as being (i) pursuant to the Telecommunications Act of 1996 (i.e. a "BFR") or (ii) pursuant to the needs of the business (i.e. a "NBR"). The request shall be sent to Knology's Local Contract Manager.
- 4.0Within thirty (30) business days of its receipt of a BFR or NBR from Knology, BellSouth shall respond to Knology by providing a preliminary analysis of such Interconnection, Network Element, or other facility or service option that is the subject of the BFR or NBR. The preliminary analysis shall confirm that BellSouth will either offer access to the Interconnection, Network Element, or other facility or service option, or provide an explanation of why it is not technically feasible and/or why the request does not qualify as an Interconnection or Network Element or is otherwise not required to be provided under the Act. However, if the preliminary analysis is determined to be of such complexity that it causes BellSouth to expend inordinate resources, a fee will be levied upon Knology and collected prior to the beginning of the preliminary analysis and the thirty (30) business days will begin upon receipt of the fee. In addition to the preliminary analysis, an explanation of the fee will be provided.
- 5.0 Knology may cancel a BFR or NBR at any time. If Knology cancels the request more than three (3) business days after submitting it, Knology

shall pay BellSouth's reasonable and demonstrable costs of processing and/or implementing the BFR or NBR up to the date of cancellation. If Knology does not cancel a BFR or NBR, Knology shall pay BellSouth's reasonable and demonstrable costs of processing and implementing the request.

- 6.0 BellSouth shall propose a firm price quote and a detailed implementation plan for BFRs within thirty (30) business days of Knology's acceptance of the preliminary analysis. BellSouth shall propose a firm price and a detailed implementation plan for NBRs within sixty (60) business days of Knology's acceptance of the preliminary analysis.
- 7.0 If Knology accepts the preliminary analysis, BellSouth shall proceed with Knology's BFR or NBR, and Knology agrees to pay the non-refundable amount identified in the preliminary analysis for the initial work required to develop the project plan, create the design parameters, and establish all activities and resources required to complete the BFR or NBR. These costs will be referred to as "development" costs. The development costs identified in the preliminary analysis are fixed. If Knology cancels a BFR or NBR after BellSouth has received Knology's acceptance of the preliminary analysis, Knology agrees to pay BellSouth the reasonable, demonstrable, and actual costs, if any, directly related to complying with Knology's BFR or NBR up to the date of cancellation, to the extent such costs were not included in the non-refundable amount set forth above.
- 8.0 If Knology believes that BellSouth's firm price quote is not consistent with the requirements of the Act, Knology may seek FCC or state Commission arbitration of its request, as appropriate. Any such arbitration applicable to Network Elements and/or Interconnection shall be conducted in accordance with standards prescribed in Section 252 of the Act.
- 9.0 Unless Knology agrees otherwise, all prices shall be consistent with the pricing principles of the Act, FCC and/or the State Commission.
- 10.0 If either Party to a BFR or NBR believes that the other Party is not requesting, negotiating, or processing the Bona Fide Request in good faith, or disputes a determination, or price or cost quote, such Party may seek FCC or state Commission resolution of the dispute, as appropriate.
- 11.0 Upon agreement to the terms of a BFR or NBR, an amendment to the Agreement may be required.